| | AD-A1 | 93 355 551F1E | CHE The Ca D Sap | MICAL DISPO A N I EO-CDI | STOCK DSAL O BARSEL E-IS-8 | PILE D F CHEM L ET A 7006 D | ISPOSI 1. (U) 1. AU(AAA15- | AL PRO GA TEI 3 87 GI -85-D-1 | GRAM R Chnolo 7-C-19 7022 | ISK AN GIES 1 563 | NC SAI | 5 OF 1 DIEG 15/6.3 | 1/ D NL | 13 |
|---|-------|------------------|---------------------------|-----------------------------------|-------------------------------------|--------------------------------------|--------------------------------------|--|------------------------------------|-------------------------|--------|--------------------------|------------|----|
| | | | | e des Sec | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| ļ | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |



523 m 555

12.22.22



GA-C18563

RISK ANALYSIS OF THE DISPOSAL OF CHEMICAL MUNITIONS AT NATIONAL OR REGIONAL SITES

GA TECHNOLOGIES INC.

Prepared under Contract DAAA15-85-D-0022/0007 for the U.S. Army Office of the Program Executive Officer – Program Manager for Chemical Demilitarization Aberdeen Proving Ground, Maryland

> GA PROJECT 3813 AUGUST 1987

SECURITY CLASSIFICATION OF THIS PAGE

AD- A 193355-

REPORT DOCUMENTATION PAGE

Form Approved OMB No. 0704-0188

| | | | | | 15. RESTRICTIVE | MARKINGS | | | | | | |
|--|---|--|---|--|--|--|---|--|---|--|--|--|
| 1a. REPORT S | ECURITY CLASS | SIFICATION | N . | | | | | | | | | |
| Unclassi 2a. Security | JNC1855111ed SECURITY CLASSIFICATION AUTHORITY | | | | 3. DISTRIBUTION / AVAILABILITY OF REPORT | | | | | | | |
| 26. DECLASSIFICATION / DOWNGRADING SCHEDULE 4. PERFORMING ORGANIZATION REPORT NUMBER(S) GA-C18563 | | | | Unlimited Distribution. Approved for Public Release | | | | | | | | |
| | | | | 5. MONITORING | ORGANIZATION | N REPORT | NUMBER(| 5) | | | | |
| | | | | SAPEO-CDE-I | s-87008 | | | | | | | |
| 6a. NAME OF | PERFORMING | ORGANIZ | ATION | 6b. OFFICE SYMBOL | 7a. NAME OF MC | DNITORING OR | GANIZAT | ION | | | | |
| GA Techn | ologies, | Inc. | | | PEO-PM Cml | Demil | | | | | | |
| 6c. ADDRESS | (City, State, an | d ZIP Code | e) | L | 7b. ADDRESS (Cit | y, State, and Z | (IP Code) | | | | | |
| P.O. Box San Dieg | 0, CA 921 | 38 | | | Aberdeen Pr | oving Grou | und, M | D 21010 | -5401 | | | |
| 8a. NAME OF ORGANIZI | FUNDING/SPC | ONSORING | | 8b. OFFICE SYMBOL (If applicable) | 9. PROCUREMENT | T INSTRUMENT | IDENTIFI | CATION NU | IMBER | | | |
| PEO-PM C | ml Demil | 1 ZIP Code |) | l | 10 SOURCE OF F | UNDING NUMP | BERS | | | | | |
| | | | • | | PROGRAM | PROJECT | TASK | < | WORK UNIT | | | |
| Aberdeen | Proving | Ground, | , MD 210 | 010-5401 | ELEMENT NO. | NO. | NO. | | ALCESSION NO. | | | |
| 11. TITLE (Inc | lude Security C | lassificatio |)) | - <u></u> | | | | ` . | | | | |
| | | | | | and the second sec | | | | | | | |
| | | | | | · · · | | | | | | | |
| Risk Ana | lysis of | the Dis | sposal o | of Chemical Muni | itions at Nat | ional or l | Region | al Site | <u>s</u> | | | |
| Risk Ana 12. PERSONAL | lysis of AUTHOR(S) | the Dis | sposal o | of Chemical Muni | itions at Nat | ional or] | Region | al Site | 5 | | | |
| Risk Ana 12. Personai 13a. Type Of | Lysis of AUTHOR(S) REPORT | the Dis | Bb. TIME CO | OVERED | 14. DATE OF REPO | ional or] RT (Year, Mon | Region | al Site | | | | |
| Risk Ane 12. PERSONAL 13. TYPE OF Final | lysis of AUTHOR(S) REPORT | the Dis | sposal o Bb. TIME CO FROM2/3, | of Chemical Muni Overed /86 то <u>8/25/</u> 87 | 14. DATE OF REPO 1987, Augu | ional or] RT (Year, Mons st | Region th, Day) | al_Site | COUNT | | | |
| Risk Ana 12. PERSONAN 13a. TYPE OF Final 16. SUPPLEME | LYSIS OF AUTHOR(S) REPORT | the Dis | sposal o Bb. TIME CO FROMP_/3 | of Chemical Muni DVERED /86 то <u>8/25/</u> 87 | 14 DATE OF REPO 1987, Augu | <u>ional or]</u> RT (Year, Mo n st | Region th, Day) | al Site 15. PAGE 1152 | | | | |
| Risk Ana 12. PERSONAN 13. TYPE OF Final 16. SUPPLEME This rep | lysis of AUTHOR(S) REPORT NTARY NOTA NOTL prepa | the Dis 13 TION red und | sposal (Bb. TIME CO FROM2/3) der cont | DVERED /86 το <u>8/25/</u> 87 tract DAAA15-85- | 14. DATE OF REPO 1987, Augu | <u>ional or]</u> RT (Year, Mon St Order 7. | Region th, Day) | 81 Site 15. PAGE 1152 | COUNT | | | |
| Risk Ana 12. PERSONAL 13. TYPE OF Final 16. SUPPLEME This rep | Lysis of AUTHOR(S) REPORT INTARY NOTA OFT PREPA | the Dis 11 TION red und CODES | Bb. TIME CO FROM 2/3, der cont | DVERED /86 TO <u>8/25/</u> 87 tract DAAA15-85- 18. SUBJECT TERMS (Chaming) Stock | 14 DATE OF REPO 1987, Augu -D-0022 Task Continue on reversa | ional or] RT (Year, Mont st Order 7. e if necessary a | Region th, Day) and ident | al Site 15. PAGE 1152 | S COUNT k number) | | | |
| Risk Ana 12. PERSONAL 13. TYPE OF Final 16. SUPPLEME This rep 17. FIELD | Lysis of AUTHOR(S) REPORT NTARY NOTA OOTT prepa COSATI GROUP | the Dis TION red und CODES SUB-C | BD TIME CO FROM 2/3, der cont | DVERED /86 TO 8/25/87 tract DAAA15-85- 18. SUBJECT TERMS (Chemical Stock Chemical Munit | 14. DATE OF REPO 1987, Augu -D-0022 Task Continue on reverse spile, Chemic tions, Risk A | ional or] RT (Year, Mon st Order 7. e if necessary a sal Munitic nalysis,] | Region th, Day) and ident ons, T Hazard | al Site 15. PAGE 1152 1152 1152 1152 | COUNT k number) tation of is. | | | |
| Risk Ana 12. PERSONA 13. TYPE OF Final 16. SUPPLEM This rep 17. FIELD | Lysis of AUTHOR(S) REPORT NTARY NOTA ORT prepa COSATI GROUP | the Dis 11 TION red und CODES SUB-C | Bb. TIME CO FROM 2/3, der cont | DVERED /86 TO 8/25/87 tract DAA15-85- 18. SUBXECT TERMS (Chemical Stock Chemical Munit Probabilistic | 14 DATE OF REPO 1987, Augu -D-0022 Task Continue on reverse tpile, Chemic tions, Risk A Risk Assessm | ional or] RT (Year, Monast St Order 7. e if necessary a ral Munitic nalysis, 1 ent | Region th, Day) and ident ons, T Hazard | al Site 15. PAGE 1152 1152 1152 1152 1152 1152 1152 1152 | S COUNT k number) tation of is, | | | |
| Risk Ane 12. PERSONAL 13. TYPE OF Final 16. SUPPLEMI This rep 17. FIELD 19. ABSTRACT | Lysis of AUTHOR(S) REPORT NTARY NOTA OOTT prepa COSATI GROUP | the Dis 11 TION red und CODES SUB-C reverse if | Bb. TIME CC FROM2/3, der cont GROUP | DVERED /86 TO 8/25/87 tract DAAA15-85- 18. SUBJECT TERMS (Chemical Stock Chemical Munit Probabilistic and identify by block m | 14. DATE OF REPO 1987, Augu -D-0022 Task Continue on reverse Continue, Chemic Cions, Risk A Risk Assessm | ional or] RT (Year, Mon Ist Order 7. e if necessary a al Munitic nalysis, l ent | Region th, Day) and ident ons, T Hazard | al Site 15. PAGE 1152 1152 trify by bloc transpor Analys | S COUNT k number) tation of is, | | | |
| Risk Ane 12. PERSONAL 13. TYPE OF Final 16. SUPPLEMI This rep 17. FIELD 19. ABSTRACT This doc Environm describe magnitud site tra disposal disposal water tr internal accident | lysis of AUTHOR(S) REPORT NTARY NOTA OOT prepa COSATI GROUP (Continue on ument has ental Imp s the res e of chem nsportati of the c site. H ansportat accident initator | the Dis the Dis reverse if been H act Sta ults of ical ag on, and hemical ail tra ion fro initia s (e.g. | Bb. TIME CC FROM2/3, der cont FROM2/3, der cont FROM2/3, der cont FROM2/3, frecessary preparec atement f a Comp gent rei d chemic atement f a Comp gent rei d chemic atement f a comp gent rei d chemic ators (e , earth | by Chemical Muni- DVERED /86 TO 8/25/87 tract DAAA15-85- 18. SUBJECT TERMS (Chemical Stock Chemical Stock Chemical Munit Probabilistic and identify by block m i for the U.S. A for the Chemicas prehensive proba- lease for the st cal demilitarization pile at two regination from sever site were the off e.g., human error nquakes, airplan | 14 DATE OF REPO 1987, Augu -D-0022 Task Continue on reverse cpile, Chemic cions, Risk A Risk Assessm number) Army to suppo al Stockpile abilistic ass corage, handl ation plant o ional disposa n sites, air ff-site trans pr, equipment he crashes) w | ional or] RT (Year, Mon ast Order 7. e if necessary a cal Munitic nalysis, 1 ent pert the Fin Disposal 1 essment or ing, on-s: perations l sites of transporta portation malfunct: ere include | Region th, Day) and ident ons, T Hazard nal Pr Progra f the ite tr assoc r at a ation modes ion) a ded in | al Site 15. PAGE 1152 1152 1152 1152 1152 ransport Analys ogramma m. Thi frequen ansport iated w single from tw analyz nd exte the an | COUNT k number) tation of is, tic s report cy and ation, off- ith the national o sites and ed. Both rnal alysis. | | | |
| Risk Ana Risk Ana Risk Ana Risk Ana Risk Ana Risk Ana Risk Ana Risk Ana Risk Ana Final Risk Ana Field Risk Ana Field Risko Field Risk Field Risk Ana Field Risk Ana Field F | Iysis of AUTHOR(S) REPORT NTARY NOTATION FOR PREPART COSATI GROUP (Continue on ument has ental Imp s the resident insportation of the construction site. A ansportation ansportation initator | the Dis the Dis TION red und CODES SUB-C SUB- | Bb. TIME Co FROM2/3 der cont BROUP Trecessary prepared atement f a Comp gent re: d chemic t stockj ansports cators (e ., earth | by Chemical Muni- DVERED 786 TO 8/25/87 tract DAAA15-85- 18 SUBJECT TERMS (Chemical Stock Chemical Munit Probabilistic and identify by block of the openensive proba- lease for the U.S. A for the Chemicas prehensive proba- lease for the st cal demilitarization from sever site were the openensite were the openensite site were the openensite were the openensite and identify and the sever site were the openensite were the openensite were the openensite the second several se | 14 DATE OF REPO 1987, Augu -D-0022 Task (Continue on reverse triple, Chemic trions, Risk A Risk Assessm tumber) Army to suppo al Stockpile abilistic ass torage, handl ation plant o tonal disposa h sites, air ff-site trans pr, equipment he crashes) w 21 ABSTRACT SEC | ional or] RT (Year, Mon ist Order 7. e if necessary a cal Munitic nalysis, 1 ent ort the Fin Disposal 1 essment or ing, on-s: perations l sites or transporta portation malfunct: ere includ CURITY CLASSIN | Region th, Day) and ident ons, T Hazard nal Pr Progra f the ite tr assoc r at a ation modes ion) a ded in FICATION | al Site 15. PAGE 1152 1152 1152 1152 1152 ransport Analys ogramma m. Thi frequen ansport iated w single from tw analyz nd exte the an () Cerve () | count tation of is, tic s report cy and ation, off- ith the national o sites and ed. Both rnal alysis. t tre t t t tre t t t tre t t t t t t t t t t t t t t t t t t t | | | |
| Risk Ana Risk Ana Risk Ana Risk Ana Risk Ana Risk Ana Final Risk Ana Field Risk Ana Risk Ana Field Risk Ana Risk Ana Field Risk Ana Risk Ana Field Risk Ana Risk Ana Field Risk Ana Field F | Lysis of AUTHOR(S) REPORT INTARY NOTA OOT Prepa COSATI GROUP (Continue on ument has ental Imp s the res e of chem nsportati of the c site. h ansportati accident initator | the Dis the Dis reverse if been I act Sta ults of ical ag on, and hemical ail tra ion fro initia s (e.g. | Bb. TIME CC FROM 2/3, der cont FROM 2/3, der cont F | DVERED NOVE | 14 DATE OF REPO 1987, Augu -D-0022 Task Continue on reverse cpile, Chemic cions, Risk A Risk Assessm number) Army to suppo al Stockpile abilistic ass corage, handl ation plant o ional disposa n sites, air Eff-site trans pr, equipment he crashes) w 21 ABSTRACT SEC Unclassifie | ional or] RT (Year, Monist Order 7. e if necessary a cal Munitic nalysis, 1 ent Construction ing, on-si operations l sites of transports portation malfunct: ere include CURITY CLASSING d | Region th, Day) and ident ons, T Hazard nal Pr Progra f the ite tr assoc r at a ation modes ion) a ded in FICATION | al Site 15 PAGE 1152 1155 1155 1155 1155 1155 1155 1155 1155 1155 1155 1155 1155 1155 | <pre>count k number) tation of is, tic s report cy and ation, off- ith the national o sites and ed. Both rnal alysis. t tree </pre> | | | |
| Risk Ana 2. PERSONAL 3. TYPE OF Final 6. SUPPLEMI This rep 7. FIELD 9. ABSTRACI This doc Environm describe magnitud site tra disposal disposal disposal water tr internal accident 20. DISTRIBUT DUNCLAS 22. NAME O Thomas S | Ivsis of AUTHOR(S) REPORT NTARY NOTAT FOR Prepa COSATI GROUP (Continue on ument has ental Imp s the res e of chem nsportati of the c site. n ansportat accident initator | the Dis the Dis TION red und CODES SUB-C SUB- | Bb. TIME CC FROM2/3, der cont FROM2/3, der cont FROM2/3, der cont FROM2/3, der cont FROM2/3, frecessory oprepared atement f a Comp gent rei d chemic l chemic l stock ansports om one s ators (e ., earth | DVERED NOVE | 14 DATE OF REPO 1987, Augu -D-0022 Task (Continue on reverse cpile, Chemic cions, Risk A Risk Assessm number) Army to suppo al Stockpile abilistic ass corage, handl ation plant o ional disposa n sites, air ff-site trans or, equipment he crashes) w 21 ABSTRACT SE Unclassifie 22b TELEPHONE((301) 671-3 | ional or RT (Year, Mon st Order 7. e if necessary a al Munitic nalysis, l ent ort the Fin Disposal l essment or ing, on-s: perations l sites or transporta portation malfunct: ere includ CURITY CLASSII d Include Area Co | Region th, Day) and ident ons, T Hazard nal Pr Progra f the ite tr assoc r at a ation modes ion) a ded in FICATION ode) 220 AM | al Site 15. PAGE 1152 1152 1152 1152 1152 ransport analys ogramma m. Thi frequen ansport iated w single from tw analyz nd exte the an () < ft CPEO-CD | COUNT k number) tation of is, tic s report cy and ation, off- ith the national o sites and ed. Both rnal alysis. t tr () () () () () () () () () () | | | |



DISCLAIMERS

The views, opinions and/or findings contained in this report should not be construed as an official Department of the Army position, policy, or decision, unless designated by other documentation.

The use of trade names in this report does not constitute an official endorsement of approval of the use of such commercial products. This report may not cited for purposes of advertisement.

Son Star Car PYES PARAT 1110 2.33 Vacator read Vicition and the 3. Dettering Veile Line Cours Hand I and Jog Dist | Special





ACKNOWLEDGMENT

This report is a team effort of the GA Technologies Inc. (GA) staff, H&R Technical Associates, Inc., JBF Associates, Inc., and Battelle-Columbus Division. Principal technical contributors are:

| GA | H&R |
|--------------------|-------------------|
| A. W. Barsell | D. Crenshaw-Smith |
| E. A. Bellis | G. J. Malek |
| C. A. Bolig | W. R. Rhyne |
| R. K. Deremer | R. J. Robinette |
| C. J. Everline | P. J. Simpkins |
| A. J. Fraissinet | |
| D. M. Ligon | JBF |
| W. S. Robison | D. J. Campbell |
| M. G. Stamatelatos | |
| D. Tow | Battelle |
| | B. J. Bell |
| | S. E. Rose |

The project team thanks many other contributors not the least of which was Jen Lobner and the technical editors and typing staff. GA project management was provided by H. C. Carney and M. G. Stamatelatos. The project team greatly appreciates the review and involvement by the Office of the Program Executive Officer-Program Manager for Chemical Demilitarization, personnel of the Environmental, Safety and Surety, and Engineering and Technology Divisions, as well as by the MITRE Corporation and Oak Ridge National Laboratory.

E. R. Zamejc



PARTICIA VOLUMENTA

ロード・ファー・マ イ

<u>-1</u>-

- 7

LIST OF ABBREVIATIONS

h.

a state to the state

| AAF | Army Air Field |
|--------|--|
| AMC | Army Materiel Command |
| ANAD | Anniston Army Depot |
| APG | Aberdeen Proving Ground |
| | |
| BCS | bulk chemical storage |
| BDS | bulk drain station |
| BRA | brine reduction area |
| BSA | buffer storage area |
| BSR | burster size reduction |
| | |
| CAMDS | Chemical Agent Munition Disposal System |
| CASY | chemical agent storage yard |
| CCDF | complementary cumulative distribution function |
| CHE | cargo handling equipment |
| CONUS | continental United States |
| CSDP | Chemical Stockpile Disposal Program |
| DARCOM | U.S. Army Materiel Development and Readiness Command |
| DATS | drill and transfer system |
| Decon | decontaminate/decontamination |
| DFS | deactivation furnace system |
| DoD | Department of Defense |
| DPE | demilitarization protective ensemble |
| DPG | Dugway Proving Ground |
| DUN | dunnage incinerator |
| ECR | explosive containment room |
| ECV | explosive containment vestibule |

iv

And Salate

| EIS | environmental impact statement |
|--------|---|
| EMP | electromagnetic pulse |
| EPA | expected peak acceleration |
| FAA | Federal Aviation Administration |
| TEIS | Final Environmental Impact Statement |
| FMEA | failure modes and effects analysis |
| | |
| GA | GA Technologies Inc. |
| | |
| HAZOP | hazard and operability analysis |
| HF | handling operation at the facility |
| HC | handling operation related to onsite transportation |
| HP | high pressure |
| H&R | H&R Technical Associates, Inc. |
| HRA | human reliability analysis |
| | |
| IE | initiating event |
| | |
| JACADS | Johnston Atoll Chemical Agent Disposal System |
| | |
| LASH | lighter aboard ship |
| LBAD | Lexington-Blue Grass Army Depot |
| LIC | liquid incinerator |
| LPF | leakers processing facility |
| LPG | liquified propane gas |
| | |
| MDR | munitions demilitarization building |
| MDM | multipurpose demilitarization machine |
| MDE | mine demilitarization equipment |
| MA | munitions holding area |
| MIC | munitions nolding 1g100 |
| MIG | mine glove box |
| MITDE | The MITTE Comments |
| MITE | Ine MITKE Corporation |
| ril U | master logic diagram |





| тс | ton container |
|-------|--|
| TEAD | Tooele Army Depot |
| TECOM | Test and Evaluation Command |
| THERP | Technique for Human Reliability Analysis |
| TOX | toxic cubicle |
| UBC | Uniform Building Code |
| UMDA | Umatilla Depot Activity |
| ΙΤΡΑ | unpack area |





CONTENTS

| ACKN | OWLEDGE | ENT | • • • | • • | •• | • • | • • | ••• | • | • | • | • | ••• | • | iii |
|------|----------|------------------|--------|-------|-----|------|----------|------|-----|-----|----|----|-----|---|------|
| LIST | OF ABB | EVIATIONS | • • • | •• | | • • | • • | ••• | • | • | • | • | ••• | • | ív |
| EXEC | UTIVE SI | MMARY | • • • | | | • • | • | • • | • | • | • | • | ••• | • | S-1 |
| 1. | INTROD | CTION | • • • | | | • • | • | • • | • | • | • | • | ••• | • | 1-1 |
| | 1.1. | Background | • • • | • • | | • • | • | • • | • | • | • | • | ••• | • | 1-1 |
| | 1.2. | Study Objective | es and | Scop | e. | • • | • | • • | • | • | • | • | •• | • | 1-6 |
| | 1.3. | Demilitarizati | on Act | iviti | .es | and | Saf | lety | Co | onc | er | ns | • | • | 1-8 |
| | 1.4. | Study Assumption | ons . | | | | • | • • | • | • | • | • | ••• | • | 1-10 |
| | 1.5. | Report Format | • • • | • • | | • • | • | • • | • | • | • | • | | • | 1-12 |
| | 1.6. | References | • • • | | | • • | • | | • | • | • | • | | • | 1-15 |
| 2. | RISK AS | SESSMENT METHO | OLOGY | •• | | • • | • | • • | • | • | • | • | • • | • | 2-1 |
| | 2.1. | Overview | • • • | | | • • | • | | • | • | • | • | | • | 2-1 |
| | 2.2. | Initiating Even | nts. | • • | | • • | • | | • | • | • | • | | • | 2-4 |
| | 2.3. | Scenario Devel | opment | and | Log | ic M | lode | els | • | • | • | • | ••• | • | 2-5 |
| | 2.4. | Human Factors | • • • | • • | | • • | • | | • | • | • | • | ••• | • | 2-13 |
| | 2.5. | Release Charac | eriza | tion | | • • | • | • • | • | • | • | • | ••• | • | 2-14 |
| | 2.6. | Uncertainty And | alysis | •• | | • • | • | • • | • | • | • | • | ••• | • | 2-16 |
| | 2.7. | References | • • • | | | • • | • | | • | • | • | • | ••• | • | 2-20 |
| 3. | DEMILI | ARIZATION DESC | RIPTIO | N OVE | RVI | EW . | • | | • | • | • | • | •• | • | 3-1 |
| | 3.1. | Collocation Di | sposal | Acti | vit | ies | and | I Ri | sks | 5 | • | • | • • | • | 3-1 |
| | | 3.1.1. Stora | ze | • • | | • • | •• | • • | • | • | • | • | ••• | • | 3-3 |
| | | 3.1.2. Handl | ing . | • • | | • • | • • | • • | • | • | • | • | ••• | • | 3-3 |
| | | 3.1.3. Onsit | e Tran | sport | • | • • | • | • • | • | • | • | • | ••• | • | 3-6 |
| | | 3.1.4. Offsi | e Tra | nspor | t - | Rai | 1 | ••• | • | • | • | • | ••• | • | 3-7 |
| | | 3.1.5. Offsi | te Tra | nspor | t - | Air | . | • • | • | • | • | • | ••• | • | 3-7 |
| | | 3.1.6. Offsi | te Tra | nspor | t - | Shi | Lp | • • | • | • | • | • | | • | 3-7 |
| | | 3.1.7. Plant | Opera | tions | • • | • • | • | • • | • | • | • | • | • • | • | 3-8 |
| | | 3.1.8. Decom | nissio | ning | • • | • • | • | | • | • | • | • | • • | • | 3-9 |



| 3.2. | Munitions Description | 3-10 |
|--------|--|------------------------------|
| | 3.2.1. Rockets | 3-10 |
| | 3.2.2. Land Mines | 3-10 |
| | 3.2.3. Projectiles and Mortars | 3-12 |
| | 3.2.4. Bombs | 3-12 |
| | 3.2.5. Spray Tanks | 3-12 |
| | 3.2.6. Bulk Agent | 3-13 |
| 3.3. | Munition Packaging and Transport | 3-14 |
| 3.4. | References | 3-18 |
| INITIA | TING EVENTS | 4-1 |
| 4.1. | Initiating Event Identification and Selection \ldots . | 4-1 |
| 4.2. | Initiating Event Frequencies | 4-22 |
| | 4.2.1. External Events | 4-22 |
| | 4.2.2. Electromagnetic Radiation | 4-47 |
| | 4.2.3. Internal Events | 4-52 |
| 4.3. | References | 4-56 |
| SCENAR | IO LOGIC MODELS FOR STORAGE | 5-1 |
| 5.1. | Sequence List and Event Trees | 5-1 |
| 5.2. | External Events | 5-21 |
| | 5.2.1. Tornadoes and High Winds | 5-21 |
| | 5.2.2. Meteorite Strikes | 5-36 |
| | 5.2.3. Aircraft Crashes | 5-41 |
| | 5.2.4. Earthquakes | 5-51 |
| | 5.2.5. Lightning | 5-59 |
| | 5.2.6. Floods | 5-62 |
| 5.3. | Special Handling Activities | 5-66 |
| | 5.3.1. Leaking Munitions | 5-66 |
| 5.4. | Scenario Quantification | 5-72 |
| 5.5. | Uncertainty Analysis | 5-102 |
| | 5.5.1. Overview | 5-102 |
| | 5.5.2. Error Factors | 5-102 |
| 5.6. | References | 5-107 |
| | 3.2. 3.3. 3.4. INITIA 4.1. 4.2. 4.3. SCENAR 5.1. 5.2. 5.3. 5.4. 5.5. | 3.2.1. Munitions Description |

x

ALALAU - 1.5

11111

<u>᠃᠃᠃᠃᠃᠃᠃᠃᠃᠃᠃᠃᠃᠃᠃᠃᠃</u>



ዿዸዸኯቔዄቔቔዄ፟ጜዄኯኯዄጞ፝ዹቜፙ፟ቘቘኯኯፙቔኯጞቒኯዄጞኯዄጞጟጞፚኯ፟ጜጟጜጜጜጜጜጞፙኯፙኯፙኯጞኯጞ ፟

| - | 6. SCENARIO LOGIC MODELS FOR HANDLING |
|-----|--|
| | 6.1. General Handling Procedures and Assumptions 6-2 |
| | 6.1.1. Rail Option |
| | 6.1.2. Air Transport Option 6-4 |
| | 6.1.3. Marine Option |
| | 6.2. Chronology of Handling Operations 6-6 |
| | 6.3. Accident Scenarios for Handling Associated with Rail Transport |
| | 6.3.1. Human-Reliability Analysis for Handling |
| | Operations |
| | 6.3.2. Data and Results |
| | 6.4. Analysis for Air Transport 6-66 |
| | 6.5. Analysis for Marine Transport 6-66 |
| | 6.6. References |
| | 7. SCENARIO LOGIC MODELS FOR PLANT OPERATIONS |
| | 7.1. Internal Events |
| ANA | 7.1.1. Explosive Containment Room Vestibule and Munitions Corridor |
| | 7.1.2. Munition Processing Systems |
| | 7.1.3. Buffer Storage Area |
| | 7.1.4. Toxic Cubicle |
| | 7.1.5. Incinerator Systems |
| | 7.1.6. Accident Analysis Summary and Results 7-64 |
| | 7.2. External Events |
| | 7.2.1. Tornadoes and High Winds |
| | 7.2.2. Meteorite Strikes |
| | 7.2.3. Aircraft Crashes |
| | 7.2.4. Earthquakes |
| | 7.2.5. Quantification of Logic Models 7-110 |
| | 7.3. References |
| | 8. SCENARIO LOGIC MODELS FOR TRANSPORT |
| | 8.1. Onsite Transport |
| | 8.1.1. Chronology of Operations 8-4 |
| | 8.1.2. Procedures and Assumptions 8-5 |
| | 8.1.3. Accident Scenario for OFC and Vault Transport |
| | x 1 |
| | |

| | 8.1.4. Accident Scenarios for ONC Transport | 8-31 |
|-------------|---|-----------------|
| | 8.1.5. Analytical Results | 8-32 |
| | 8.2. Offsite Rail Transport | 8-49 |
| | 8.2.1. Accident Scenario Definition | 8-49 |
| | 8.2.2. Rail Transport Procedures and Data | 8-51 |
| | 8.2.3. Event Tree Analysis | 8-58 |
| , 4 • | 8.3. Offsite Air Transport | 8-86 |
| • { | 8.3.1. Procedures and Assumptions | 8-86 |
| | 8.3.2. Accident Scenario Definition | 8-89 |
| | 8.3.3. Accident Sequence Analysis | 8-91 |
| | 8.4. Offsite Marine Transport | 8-109 |
| | 8.4.1. Procedures, Assumptions, and Data | 8-109 |
| | 8.4.2. Accident Sequence Definition and | |
| | Analysis | 8-110 |
| - - | 8.5. Uncertainty Analysis | 8-129 |
| | 8.6. References | 8-141 |
| 9. | QUANTIFICATION BASES | 9- 1 |
| | 9.1. Data Base | 9-1 |
| | 9.1.1. Train Accident Data | 9-1 |
| | 9.1.2. Onsite Truck Accident Data | 9-6 |
| | 9.1.3. Plant Accident Data | 9-11 |
| | 9.1.4. Handling Accident Data | 9-25 |
| | 9.2. Human Factors Data | 9-27 |
| | 9.2.1. Human-Error Probability Estimation - Handling Accidents | 9-27 |
| | 9.2.2. Human-Reliability Analysis for Plant Operations | 9-31 |
| | 9.3. References | 9-54 |
| 10. | AGENT RELEASE CHARACTERIZATION | 10-1 |
| y 4 | 10.1. Release Analysis Approach and Bases | 10-1 |
| (| 10.1.1. Approach | 10-1 |
| | 10.1.2. Mechanical Failure Release | 10-2 |
| | 10.1.3. Detonations | 10-7 |
| Ň | 10.1.4. Fire Release | 10-9 |
| l | 10.1.5. Release Duration | 10-12 |
| | | |
| | xii | |
| | | С., П ъ. |

•



፞ጟዀጟዀጟዀጟዀጟዀዀዀዀዀዀዀዀዀዀዀዀዀ

| 20 | |
|----|--|

| 10.2. | Application to Accident Seque | ences . | ••• | • | • • | • | • | • | 10-17 |
|-------------|-------------------------------|-----------|------|-----|-----|---|---|---|------------|
| | 10.2.1. Handling | • • • • | • • | • | •• | • | • | • | 10-17 |
| | 10.2.2. Warehouse Storage Re | elease Du | irin | B | | | | | |
| | Earthquakes | • • • • | •• | • | • • | ٠ | • | • | 10-20 |
| | 10.2.3. Plant Operation Rele | eases . | ••• | • | •• | • | ٠ | ٠ | 10-24 |
| | 10.2.4. Transport Releases | • • • • | • • | • | •• | • | ٠ | • | 10-35 |
| | 10.2.5. Uncertainties | • • • • | • • | • | • • | • | ٠ | • | 10-40 |
| 10.3. | References | • • • • | • • | • | • • | • | ٠ | • | 10-45 |
| 11. RESULT | 5 | • • • • | • • | • | •• | • | • | • | 11-1 |
| 11.1. | Accident Scenarios During Sto | orage . | ••• | • | •• | • | • | • | 11-2 |
| | 11.1.1. Internal Events | • • • • | • • | • | • • | • | ٠ | ٠ | 11-2 |
| | 11.1.2. External Events | • • • • | • • | • | • • | • | ٠ | ٠ | 11-2 |
| 11.2. | Accident Scenarios During Ham | ndling . | • • | • | •• | • | • | • | 11-4 |
| | 11.2.1. Rail Option | • • • • | ••• | • | • • | • | • | • | 11-4 |
| | 11.2.2. Air Option | • • • • | • • | • | • • | • | • | • | 11-6 |
| | 11.2.3. Marine Option | • • • • | • • | • | • • | • | • | • | 11-6 |
| 11.3. | Accident Scenarios During Ope | eration | • • | • | • • | • | • | • | 11-8 |
| | 11.3.1. Internal Events | | • • | • | | • | • | • | 11-8 |
| | 11.3.2. External Events | • • • • | | • | •• | • | • | • | 11-9 |
| 11.4. | Accident Scenarios During Tra | ansport | • • | • | • • | • | • | • | 11-10 |
| | 11.4.1. Onsite Transportation | on | • • | • | •• | • | • | • | 11-10 |
| | 11.4.2. Offsite Transportat: | ion - Ra | 11. | • | | • | • | • | 11-12 |
| | 11.4.3. Offsite Transportat: | ion - Air | r Op | tio | n. | • | • | • | 11-13 |
| | 11.4.4. Offsite Transportat: | ion - Mar | rine | Opt | tio | n | • | • | 11-14 |
| 11.5. | Uncertainties | | • • | • | | • | • | • | 11-16 |
| | 11.5.1. Sources of Uncertain | nty | • • | • | | • | • | • | 11-16 |
| | 11.5.2. Uncertainties | | • • | • | | • | • | • | 11-17 |
| APPENDIX A: | REFERENCE LIST OF ACCIDENT | SCENARIOS | 5. | ٠ | | • | • | | A-1 |
| APPENDIX B: | SENSITIVITY ANALYSIS | | • • | • | | • | • | • | B-1 |
| APPENDIX C: | STRUCTURAL ANALYSIS | | • • | • | | • | • | • | C-1 |
| APPENDIX D: | SITE INFORMATION | | • • | • | | • | • | • | D-1 |
| APPENDIX E: | (DELETED) | | | | | | | | |
| APPENDIX F: | MUNITION FAILURE THRESHOLDS | | | | | | • | • | F-1 |
| APPENDIX G: | DEMILITARIZATION ACTIVITIES | | • • | • | | • | | • | G-1 |
| APPENDIX H: | (CLASSIFIED INFORMATION) . | | • • | • | | • | • | • | H-1 |
| | - | | | | | | | | |

Ŵ





TUTTURE STATES

NEW CONTRACTOR OF THE OWNER OF THE

FIGURES

| 1-1. | Location of chemical agents and munitions in the U.S | 1-2 |
|------|---|------|
| 1-2. | Logistic phases associated with the munitions storage and disposal options | 1-9 |
| 2-1. | Outline of risk assessment procedure used in this study | 2-2 |
| 2-2. | Accident scenario development using an event tree | 2-6 |
| 2-3. | A fault tree model of a power system failure | 2-8 |
| 2-4. | Definition of fault tree symbols | 2-9 |
| 2-5. | Impact of assumptions on the accident frequency uncertainty assessment | 2-18 |
| 3-1. | Activities associated with munitions handling and transport | 3-2 |
| 4-1. | Master logic diagram - level 1 (public impact) for the collocation options of the chemical demilitarization cycle | 4-5 |
| 4-2. | Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part A - storage release | 4-6 |
| 4-3. | Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part B - handling release | 4-7 |
| 4-4. | Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part C - onsite storage release | 4-8 |
| 4-5. | Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part D - plant operations internal events | 4-9 |
| 4-6. | Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part E - plant operations external events | 4-10 |
| 4-7. | Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part F - offsite rail transport | 4-11 |
| 4-8. | Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part G - offsite air transport | 4-12 |

FIGURES (Continued)

1. 1. 2. 4 A. 1.

| 4-9. | Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part H - offsite sea transport | 4-13 |
|-------|--|------|
| 4-10. | Seismic zone map for the contiguous United States | 4-26 |
| 4-11. | Annual frequency of exceeding various effective peak accelerations for locations on the indicated contours | 4-28 |
| 4-12. | Tornadic winds corresponding to a probability of 1.0 x 10 ⁻⁷ per year | 4-31 |
| 4-13. | Tornadic winds corresponding to a probability of 1.0 x 10 ⁻⁶ per year | 4-32 |
| 4-14. | Tornadic winds corresponding to a probability of 1.0 x 10 ⁻⁵ per year | 4-33 |
| 4-15. | Wind strength versus probability of recurrence, tornado Zone I (TEAD and UMDA sites) | 4-34 |
| 4-16. | Wind strength versus probability of recurrence, tornado Zone II (PUDA and APG sites) | 4-35 |
| 4-17. | Wind strength versus probability of recurrence, tornado Zone III (ANAD, LBAD, PGA, and NAAP sites) | 4-40 |
| 4-18. | Field intensity potentially hazardous to susceptible weapons which require special restriction - communication frequencies | 4-50 |
| 4-19. | Field intensity potentially hazardous to susceptible weapons which require special restrictions - radar | 6 51 |
| 5_1 | Agent release indicated by termede-senerated missiles | 4-51 |
| 5-2 | Metaorite_induced egent release | 5-12 |
| 5_3 | large eireraft areab onto storage isloss containing | J-11 |
| J-J. | burstered munitions | 5-13 |
| 5-4. | Aircraft crash onto storage facilities with nonburstered (NB) munitions | 5-14 |
| 5-5. | Aircraft crash accidents involving munitions in holding area (rail option) | 5-15 |
| 5-6. | Aircraft crash accidents involving munitions in holding area (air option) | 5-16 |
| 5-7. | Aircraft crash accidents involving munitions in holding area (marine option) | 5-17 |
| 5-8. | Aircraft crash accidents involving the lighters at rest (marine option) | 5-18 |

FIGURES (Continued)

| 5-9. | Aircraft crash accidents involving the LASH at rest (marine option) | 5-19 |
|-------|--|------|
| 5-10. | Earthquake-induced agent releases involving munitions in storage igloos | 5-20 |
| 5-11. | Earthquake-induced releases from the warehouses | 5-54 |
| 6-1. | Flow diagram for handling operations, collocation option | 6-8 |
| 6-2. | Handling activities for ship transport option | 6-9 |
| 6-3. | Event tree for drop of munition(s) during handling at facility | 6-14 |
| 6-4. | Event tree for drop of munition(s) during handling operations other than at facility | 6-15 |
| 6-5. | Event tree for forklift tine punctures during handling | 6-16 |
| 6-6. | Event tree for vehicle collisions during handling at facility | 6-17 |
| 6-7. | Event tree for vehicle collisions during handling other than at facility | 6-18 |
| 7-1. | Event tree for spill of one rocket in ECV | 7-6 |
| 7-2. | Event tree for spill of one ton container in ECV | 7-7 |
| 7-3. | Event tree for spill of two ton container in ECV \ldots . | 7-8 |
| 7-4. | Event tree for detonation of burstered munition in ECV $$. | 7-9 |
| 7-5. | Fault tree for agent release through the ventilation | _ |
| | systems | 7-10 |
| 7-6. | Event tree for spill of munition(s) in the ECR | 7-15 |
| 7-7. | Event tree for detonation of burstered munitions in the ECR | 7-16 |
| 7-8. | Event tree for spill of munition in MPB | 7-17 |
| 7-9. | Event tree for detonation of one ton container in | |
| | in the BSA | 7-23 |
| 7-10. | Event tree for a small spill in the TOX | 7-26 |
| 7-11. | Event tree for a large spill in the TOX | 7-27 |
| 7-12. | Fault tree for failure to suppress a fire in the TOX $$ | 7-29 |
| 7-13. | Event tree for LIC-1 initiators | 7-34 |
| 7-14. | Event tree for LIC-1A initiators | 7-35 |
| 7-15. | Event tree for LIC-2 initiators | 7-36 |

Ċ

Į.



41087 STATER KARABARANIA (KARA)

FIGURES (Continued)

| 7-16. | Event tree for LIC-2A initiators | 7-37 |
|-------|--|------|
| 7-17. | Event tree for LIC-3 initiators | 7-38 |
| 7-18. | Event tree for LIC-4 initiators | 7-39 |
| 7-19. | Event tree for LIC-4A initiators | 7-40 |
| 7-20. | LIC room ventilation fault tree | 7-41 |
| 7-21. | Filtered exhaust fault tree | 7-42 |
| 7-22. | Fault tree for LIC fuel flow termination | 7-43 |
| 7-23. | Fault tree for terminating fuel flow to LIC burners | 7-44 |
| 7-24. | Fault tree for LIC agent feed termination | 7-45 |
| 7-25. | Fault tree for LIC PAS shutdown | 7-46 |
| 7-26. | Event tree for LIC PCC fuel flow termination | 7-47 |
| 7-27. | Control and solenoid valve fault trees | 7-48 |
| 7-28. | LIC room explosion fault tree | 7-49 |
| 7-29. | Fault tree for fuel flow forming a flammable mixture in the LIC room | 7-50 |
| 7-30. | Fault tree for draining bulk containers | 7-51 |
| 7-31. | Event tree for MPF shutdown | 7-52 |
| 7-32. | Event tree for unpunched bulk item fed to MPF | 7-53 |
| 7-33. | MPF room ventilation fault tree | 7-54 |
| 7-34. | MPF room filtered exhaust fault tree | 7-55 |
| 7-35. | Fault tree for MPF fuel flow termination | 7-56 |
| 7-36. | Fault tree for failure to stop fuel to MPF burners \ldots | 7-57 |
| 7-37. | MPF room explosion fault tree (loss of offsite power) | 7-58 |
| 7-38. | Fault tree for feeding an unpunched container to the MPF | 7-59 |
| 7-39. | Tornado-induced agent release scenarios | 7-74 |
| 7-40. | Meteorite-induced agent release scenarios | 7-75 |
| 7-41. | Large aircraft crash onto MHI/MDB containing burstered munitions | 7-76 |
| 7-42. | Aircraft crash onto MHI/MDB with nonburstered (NB) munitions | 7-77 |
| 7-43. | Event tree: earthquake-induced releases from the MDB involving bulk containers | 7-78 |





FIGURES (Continued)

25. 30 . Farata

0003030005000

274

SSSS **BEERS**

| 7-44. | Event tree: earthquake-induced releases from the MDB involving burstered munitions | 7-79 |
|-------|--|-------|
| 7-45. | Extrapolated seismic hazard model | 7-99 |
| 7-46. | TOX fragility model | 7-101 |
| 7-47. | Phenomenological fault tree for ignition occurrence | 7-102 |
| 7-48. | Phenomenological success tree for fire suppression | 7-104 |
| 8-1. | Diagram of transportation steps and options (sending sites are APG for marine shipment, AFG or LBAD for air transport, and all nondisposal sites for rail transport | 8-2 |
| 8-2. | Event tree for onsite transportation (truck accident) | 8-8 |
| 8-3. | Event tree for onsite transportation (aircraft crash) | 8-16 |
| 8-4. | Event tree for onsite transportation (earthquake) | 8-19 |
| 8-5. | Event tree for tornado-caused collision/overturn during onsite transportation | 8-26 |
| 8-6. | Event tree for tornado-generated missile affecting onsite transportation | 8-27 |
| 8-7. | Cumulative distribution of fire durations for train- tanker accidents and locomotive fires | 8-56 |
| 8-8. | Impact velocity distribution for large packages in train accidents | 8-57 |
| 8-9. | Cumulative distribution of total crush load for a large package in train accident | 8-59 |
| 8-10. | Event tree for train accident caused by earthquake or human error/equipment failure | 8-60 |
| 8-11. | Event tree for air transportation | 8-90 |
| 8-12. | Approximate duration of fuel fire as a function of aircraft size and on-board fuel inventory | 8-95 |
| 8-13. | Event tree for marine vessel collision initiating event | 8-113 |
| 9-1. | Probability of failure to isolate room by X min | 9-44 |
| 9-2. | HRA event tree of fire suppression model | 9-46 |

TABLES

| 3-1. | Data for onsite transport containers (ONC), vaults, | |
|------|---|------|
| | and offsite transport containers (OFC) | 3-11 |

1. 1. 1. 1.

| 4-1. | Accident sequence coding scheme | 4-4 |
|-------|--|------|
| 4-2. | Initiating event families for storage | 4-15 |
| 4-3. | Initiating event families for handling | 4-16 |
| 4-4. | Initiating event families for onsite truck transport | 4-17 |
| 4-5. | Initiating event families for plant operations | 4-19 |
| 4-6. | Initiating event families for offsite rail, air, or sea transport | 4-21 |
| 4-7. | Site specific frequencies of external initiating events | 4-23 |
| 4-8. | External event frequencies for special cases | 4-24 |
| 4-9. | Maximum modified Mercalli intensities in the vicinity of each site | 4-25 |
| 4-10. | Annual risk of earthquakes | 4-29 |
| 4-11. | Tornado windspeeds and probability of recurrence for chemical storage sites | 4-36 |
| 4-12. | Wind generated missile parameters | 4-37 |
| 4-13. | Windborne missile velocities | 4-38 |
| 4-14. | Aircraft crash probabilities near airports | 4-43 |
| 4-15. | Assumed distribution of air traffic | 4-44 |
| 4-16. | Summary of aircraft crash probabilities | 4-46 |
| 4-17. | Size distribution of meteorites which are 1 lb or larger | 4-48 |
| 4-18. | List of internal initiating events and frequencies | 4-53 |
| 5-1. | Master list of storage accidents | 5-3 |
| 5-2. | Windborne missile velocities | 5-9 |
| 5-3. | Munition penetration through steel igloo doors | 5-24 |
| 5-4. | Munition penetration through concrete igloo doors | 5-26 |
| 5-5. | Probability of a wind hazard sufficient to breach munitions in storage magazines | 5-27 |
| 5-6. | Probabilities for wind-generated missile penetration of ton containers and spray tanks stored in ware- warehouses and open storage | 5-28 |
| 5-7. | Windborne missile velocity (holding/loading area - rail/air option) | 5-30 |



12623404

þ

xix

| 5-8. | Probability of a wind sufficient to generate missiles to beach munition (holding/loading area - rail/air option) | 5-33 |
|-------|--|-------|
| 5-9. | Probability of missile striking munitions in holding/ loading area (rail/air option) | 5-34 |
| 5-10. | Tornado-generation missile analysis of the barge package | 5-35 |
| 5-11. | Meteorite required for penetration of munitions in storage | 5-37 |
| 5-12. | Summary of aircraft crash probabilities | 5-40 |
| 5-13. | Data base for aircraft crash-initiated scenarios for storage | 5-46 |
| 5-14. | Data base for analysis of earthquake-induced agent release in the storage igloos | 5-53 |
| 5-15. | Earthquake-induced accidents in warehouses | 5-55 |
| 5-16. | Site-specific lighting strike information | 5-61 |
| 5-17. | Probable size distribution for steel tanks | 5-64 |
| 5-18. | Data base for analysis of SL1, SL2, and SL9 | 5-70 |
| 5-19. | Frequencies of storage accident sequences | 5-73 |
| 5-20. | Frequency of earthquake storage accidents per year | 5-88 |
| 5-21. | Generic uncertainty models | 5-103 |
| 6-1. | List of accident scenarios (HC and HF) collocation option - rail transport | 6-20 |
| 6-2. | Initiating events frequencies (handling associated with rail transport) | 6-28 |
| 6-3. | Conditional events probabilities (rail transport) | 6-30 |
| 6-4. | Handling accident-collocation processing option (rail transport) | 6-36 |
| 6-5. | Initiating event frequencies (handling associated with air transport) | 6-68 |
| 6-6. | Conditional events probabilities (air transport) | 6-70 |
| 6-7. | Handling accident - collocation processing option (air transport) | 6-73 |
| 6-8. | Accident scenarios for marine transport option han- dling activities | 6-79 |
| 6-9. | Initiating and conditional events frequencies (marine transport) | 6-80 |



14° 15' 181' 144 14

TABLES (Continued)

| 6-10. | Barge transport option | 6-82 |
|-------|---|--------------|
| 7-1. | Events considered for the ECV/munitions corridor | 7-11 |
| 7-2. | Events considered for the munition processing systems . | 7-1 9 |
| 7-3. | Events considered for the BSA | 7-24 |
| 7-4. | Events considered for the TOX | 7-28 |
| 7-5. | LIC initiating event descriptions | 7-33 |
| 7-6. | Internal events accident sequences | 7-65 |
| 7-7. | Plant operations internally-initiated accident sequence frequencies (events/year) | 7-66 |
| 7-8. | Master list of externally-initiated plant accident scenarios | 7-71 |
| 7-9. | Data base for tornado-initiated events for plant operations | 7-83 |
| 7-10. | Event 5 statistical parameters | 7-109 |
| 7-11. | Data base for meteorite initiated plant accident | |
| | sequences | 7-111 |
| 7-12. | Effective target area for aircraft crash analysis | 7-112 |
| 7-13. | Aircraft crash data | 7-113 |
| 7-14. | Plant operations data | 7-117 |
| 8-1. | Onsite transport sequence 1 | 8-11 |
| 8-2. | Onsite transport sequence 2 | 8-12 |
| 8-3. | Onsite transport sequence 3 | 8-12 |
| 8-4. | Onsite transport sequence 4 | 8-13 |
| 8-5. | Onsite transport sequence 5 | 8-14 |
| 8-6. | Onsite transport sequence 6 | 8-17 |
| 8-7. | Onsite transport sequence 7 | 8-18 |
| 8-8. | Onsite transport sequence 9 | 8-21 |
| 8-9. | Onsite transport sequence 10 | 8-22 |
| 8-10. | Onsite transport sequence 11 | 8-23 |
| 8-11. | Onsite transport sequence 12 | . 8-24 |
| 8-12. | Onsite transport sequence 13 | 8-25 |
| 8-13. | Onsite transport sequence 14A | 8-28 |
| 8-14. | Onsite transport sequence 14B | 8-29 |
| 8-15. | Onsite transport sequence 15 | 8-30 |





and the set of the set

Parata', 42'ata'.4

| 8-16. | Onsite transportation - regional and national disposal options | 8-33 |
|-------|--|-------|
| 8-17. | Onsite transportation - onsite package - collocation option | 8-42 |
| 8-18. | Travel miles for rail transport | 8-53 |
| 8-19. | Rail transportation data | 8-54 |
| 8-20. | Rail transport sequence 1 | 8-62 |
| 8-21. | Rail transport sequence 2 | 8-63 |
| 8-22. | Rail transport sequence 3 | 8-64 |
| 8-23. | Rail transport sequence 4 | 8-65 |
| 8-24. | Rail transport sequence 5 | 8-66 |
| 8-25. | Rail transport sequence 6 | 8-67 |
| 8-26. | Rail transport sequence 7 | 8-68 |
| 8-27. | Rail transport sequence 9 | 8-69 |
| 8-28. | Rail transport sequence 10 | 8-70 |
| 8-29. | Rail transport sequence 11 | 8-71 |
| 8-30. | Rail transport sequence 12 | 8-72 |
| 8-31. | Rail transport sequence 13 | 8-73 |
| 8-02. | Rail transport sequence 14 | 8-74 |
| 8-33. | Rail transport sequence 15 | 8-75 |
| 8-34. | Offsite rail transportation - national disposal option | 8-76 |
| 8-35. | Offsite transportation - regional disposal option | 8-81 |
| 8-36. | Accident scenarios for air transport to Tooele Army | |
| | Depot | 8-92 |
| 8-37. | Distribution of aircraft accident impact categories | 8-96 |
| 8-38. | Summary of aircraft accident threat frequency data | 8-100 |
| 8-39. | Air transport sequence 1 | 8-103 |
| 8-40. | Air transport sequence 2 | 8-104 |
| 8-41. | Air transport sequence 3 | 8-105 |
| 8-42. | Air transport sequence 4 | 8-106 |
| 8-43. | Air transport sequence 5 | 8-107 |
| 8-44. | Sequence frequencies for air transportation | 8-108 |
| 8-45. | Summary of offsite marine transport sequences | 8-115 |

122222124 NZZZZZ

بر•۲۹ م م•۲۹ م

| 8-46. | Frequency of agent release for the lighter in the | |
|-------|--|-------|
| | Chesapeake Bay | 8-119 |
| 8-47. | Frequency of agent release for the ship in the | |
| | Chesapeake Bay | 8-120 |
| 8-48. | Frequency of agent release for the ship in the coastal areas | 8-122 |
| 8-49. | Frequency of agent release for the ship on the high seas | 8-123 |
| 8-50. | Frequency of accidents without a release for the ship on the high seas | 8-124 |
| 8-51. | Frequencies of offsite marine transport accident sequences | 8-126 |
| 8-52. | Uncertainty data for the onsite truck transportation (in onsite package) accident sequences | 8-130 |
| 8-53. | Uncertainty data for onsite truck transportation (offsite package) accident sequences | 8-132 |
| 8-54. | Uncertainty data for offsite rail accident transportation sequences | 8-135 |
| 8-55. | Uncertainty data for offsite transportation by air accident sequences | 8-138 |
| 8-56. | Uncertainty data for onsite truck transportation accident sequences (marine option) | 8-139 |
| 9-1. | Comparison of 1972 and 1982 accident data | 9-3 |
| 9-2. | 1982 train accident data by category | 9-4 |
| 9-3. | Train accidents caused by track, roadbed, and structure defects | 9-7 |
| 9-4. | Train accidents caused by mechanical and electrical failures | 9-8 |
| 9-5. | Train accidents caused by human factors | 9-9 |
| 9~6. | Train accidents caused by miscellaneous factors | 9-10 |
| 9-7. | Truck accident rate | 9-12 |
| 9-8. | Human error probabilities per handling operation | 9-28 |
| 9-9. | Screening quantification for human reliability analysis of plant operations | 9-33 |
| 9-10. | Human-ertor events by sequence | 9-35 |
| 9-11. | Human-error events for final quantification | 9-39 |
| 9-12. | THERP quantification of fire-suppression model | 9-47 |

<u>Versioner besternen bereiten besternen besternen besternen besternen besternen besternen besternen</u>

<u>888885</u>

xxiii

| 10-1. | Agent release times for fires and detonations 10-14 |
|--------|---|
| 10-2. | Grouping of handling sequences according to agent release characteristics |
| 10-3. | Inventory data for onsite and offsite transport containers |
| 10-4. | Agent releases (pounds) for handling sequences 10-21 |
| 10-5. | Agent HD releases from ton containers stored in UMDA warehouses during earthquakes |
| 10-6. | Agent VX releases from NAAP warehouse ton containers during earthquakes |
| 10-7. | Agent VX release from spray tanks stored at TEAD warehouses during earthquakes |
| 10-8. | Agent inventories and releases |
| 10-9. | Results of agent release for onsite transport accident sequences (marine option) |
| 10-10. | Results of onsite transport release analysis - air option |
| 10-11. | Release consequences for the air transport mode 10-41 |
| 10-12. | Marine transport agent releases |

a data, -----

xxiv

EXECUTIVE SUMMARY

S.1. INTRODUCTION

S.1.1. Background

Under the direction of the U.S. Army Office of the Program Executive Officer-Program Manager for Chemical Demilitarization (PEO-PM Cml Demil), GA Technologies Inc. (GA) and its subcontractors performed a comprehensive assessment of the frequency and magnitude of accidental agent releases associated with various alternatives under consideration for the Chemical Stockpile Disposal Program (CSDP). This assessment was carried out in support of the environmental impact statement (EIS) for this program and addresses only the stockpile of chemical munitions that is currently stored at eight sites in the continental United States (CONUS). The assessment of potential health consequences to the public resulting from accidental releases calculated in this study will be performed in a separate study. These consequences and the GA-evaluated frequencies of the releases leading to these consequences will form the basis of estimates of the potential public "risks" associated with the CSDP alternatives.

The alternatives investigated in this study are as follows:

- Disposal of the agents and munitions at the eight existing storage sites.
- Collocation (transportation) and disposal of the munitions at two regional sites.

S-1

- 4. Partial collocation of the selected stockpiles from Aberdeen Proving Ground (APG) to Johnston Island by water or to Tooele Army Depot (TEAD) by air and from the Lexington-Blue Grass Army Depot (LBAD) to TEAD by air.
- 5. Continued storage of the munitions at the existing storage sites.

This report addresses the collocation alternatives listed above (i.e., items 2, 3, and 4). The other alternatives are discussed in separate reports.

Anniston Army Depot (ANAD) in northeast Alabama and TEAD in north central Utah have been identified as the regional disposal sites should this collocation alternative be selected. If the national collocation alternative is selected, the disposal facility will be constructed at TEAD.

Demilitarization of the chemical agent and munition stockpiles requires the construction of facilities and planned activities to store, handle, and transport onsite the chemical materiel; to transport the agents and munitions between sites if a collocation alternative is selected; to destroy the munitions; and to decommission the disposal facilities. This report addresses each of these activities, other than facility construction and closure, which do not pose risk to the health and safety of the general public from agent release.

S.1.2. Study Objectives and Deliverables

The primary objectives of the study reported in this document were to:

- 1. Identify events that could initiate the release of agent to the environment (i.e., initiating events).
- 2. Develop the various sequences of events resulting from these initiators and leading to accidental agent release.
- 3. Perform a quantitative analysis of the frequency of occurrence of each relevant accident sequence.
- 4. Characterize the physical state, quantity, and duration of agent released from each accident sequence.

These objectives were accomplished by developing a list of potential accident sequences for each major activity, estimating the frequencies of these sequences, and calculating the magnitudes of released agent associated with these sequences. It should be noted that only accident sequences that survived a conservative screening process, considering both frequency and magnitude of agent release, are included in the deliverables of this project.

S.1.3. Scope of Study

The scope of effort reported in this document, as noted earlier, did not include the evaluation of agent dispersion to the environment and the consequences to the public resulting from such releases. As such, the title of this report is more appropriately that of a probabilistic "release" analysis as opposed to a probabilistic "risk" analysis, since risk is usually defined as the product of frequency and consequence. Therefore, the term "risk," as used in this study, refers to the frequency of accidental agent release and not to the frequency of the agent release consequence to public health.

S.1.4. Plant Description

Demilitarization of the chemical munitions stored at U.S. sites is based on the Johnston Atoll Chemical Agent Disposal System (JACADS) technology. This facility is currently being constructed on the Johnston Atoll in the Pacific Ocean. The demilitarization facility consists of an integrated munitions handling system that can process a variety of munitions types and agents. After disassembly and draining of the munitions, the agent, explosive materials, dunnage, and metal mass are subjected to different combustion trains where the combustibles are consumed by incineration. All materials are subjected to two-stage incineration, and combustion products are released to the environment through a state-of-the-art pollution abatement system.

Two types of demilitarization plants will be constructed: mixedmunition plants and bulk agent plants. Mixed-munition plants are capable of processing all types of chemical materiel. Bulk plants are designed to process ton containers, bombs, and spray tanks. For the national disposal alternative, three mixed-munition plants and two bulk agent plants will be constructed at TEAD. For the regional disposal alternative, two mixed-munition plants and one bulk agent plant will be constructed at TEAD, and one mixed-munition plant and one bulk agent plant will be constructed at ANAD. BUSINESS INTERVENT NEW

To meet the September 1994 deadline for the destruction of the chemical agent stockpile, the plants are projected to begin operation during the period between September 1990 and March 1991. The plants will operate five days per week and twenty-four hours per day.

The analysis of plant operations presented in this assessment was based on a plant design which was approximately 35 percent complete.

S-4

It is recognized that design evolution could have an impact on the results reported herein.

S.1.5. <u>Site Descriptions</u>

ĥ

There are eight sites in the CONUS where chemical munitions are currently being stored. These sites are: Tooele Army Depot (TEAD), Anniston Army Depot (ANAD), Aberdeen Proving Ground (APG), Lexington-Blue Grass Army Depot (LBAD), Newport Army Ammunition Plant (NAAP), Pine Bluff Arsenal (PBA), Pueblo Depot Activity (PUDA), and the Umatilla Depot Activity (UMDA).

TEAD is located in north central Utah. A prototype demilitarization plant, the Chemical Agent Munitions Disposal System (CAMDS) facility, is located at this site. The site currently stores a wide variety of chemical munitions and bulk agent containers of mustard and the nerve agents, GB and VX.

ANAD is located in northeast Alabama. The chemical munitions stockpile at ANAD consists of all chemical munitions types except for bombs, spray tanks, and 8-in. projectiles filled with VX.

APG is located in Maryland near the head of the Chesapeake Bay. APG is comprised of two general areas, the Aberdeen area and the Edgewood area where the chemical munition storage facilities are located. Only mustard-filled ton containers are stored at APG.

LBAD is located south of Richmond, Kentucky. The chemical munition stockpile at LBAD consists of 8-in. projectiles, 155-mm projectiles, and M55 rockets.

NAAP is located west of Indianapolis, Indiana. The chemical munitions stockpile is stored there in a single warehouse and consists of containers of VX. PBA is located southeast of Little Rock, Arkansas. The stockpile at PBA consists of M55 rockets, land mines, ton containers, and some 4.2-in. mortar projectiles.

UMDA is located in northeastern Oregon. The stockpile at UMDA consists of 155-mm and 8-in. projectiles, M55 rockets, M23 land mines, bombs, spray tanks, and ton containers.

S.2. STUDY APPROACH

The risk analysis presented in this report combines the structured safety analysis detailed in MIL-STD-882B (Ref. S-1) and the probabilistic approach outlined in NUREG/CR-2300 (Ref. S-2). The first reference requires that hazards analyses be performed to assess the risk involved during the planned life expectancy of a system. It also provides guidance on the categorization of hazard severity and of probability as a means of identifying which hazards should be eliminated or reduced to an acceptable level. The second reference serves as a guidebook for the risk assessment of nuclear power plants.

Risk assessment can be defined as the quantification of an undesirable effect in probabilistic terms. Relative to the health and safety of the public, the effects of interest are injuries and deaths. Risk assessment has been utilized in various industries for some time. Insurance companies have long used actuarial data for statistical evaluations to justify differences in the insurance premium paid by persons in different "risk" categories. The risk assessments performed for nuclear power plants, on the other hand, are examples of major industry efforts to quantify risks of low-frequency events for which no good actuarial data exist. The nuclear power plant risk assessments have become models for other industrial risk assessments.

S-6

S.2.1. <u>Risk Assessment Methodology</u>

Probabilistic risk assessment (PRA) is a systematic, disciplined approach to quantifying the frequency and consequences of events which can occur at random points in time. In its application to the various chemical munition disposal alternatives currently under consideration, PRA provides a comprehensive framework for estimating and understanding the risks associated with the storage, handling, transportation, and demilitarization activities associated with these alternatives. By applying this methodology to each alternative in a consistent and uniform manner, a statement of the relative risk of these alternatives can be made. Because of the significant uncertainties in the data used to quantify the frequency of occurrence of various accident sequences and the magnitudes of the associated agent releases, extreme caution must be used when addressing the absolute risk associated with each disposal option.

In simplistic terms, the PRA process focuses on answering the following three basic questions:

- 1. What can go wrong?
- 2. How frequently is it expected to happen?
- 3. What would be the associated consequences?

The remainder of this summary describes how these questions are addressed in the risk assessment of the chemical materiel disposal program. In this study, the estimation of consequences is limited to the magnitudes of agent release for each sequence.

S.2.1.1. <u>Identification of Initiating Events</u>. The first step in a probabilistic risk assessment is the identification of initiating events which, by themselves or in combination with additional failures, can lead to the release of agent to the environment. Initiating events are identified for each of the demilitarization activities. Such events generally fall into two broad categories known as "internal" events and "external" events. Internal events originate within the activity and are caused by human error or random equipment failure. Examples of such events are the dropping or puncture of munitions during handling operations, and the random failure of a normally operating piece of equipment in the demilitarization process line. The class of events referred to as external includes aircraft crashes and natural phenomena such as earthquakes and storms. In the context of a risk assessment, events such as internal flooding and fires are also considered to be external events. External events are usually pervasive in nature in that they are assumed to fail redundant equipment that is provided for safe shutdown of the operation and containment of the agent.

S.2.1.2. Accident Sequence Development. Once initiating events are identified, logic models (such as event trees and sequence level fault trees) are developed to display the various paths that the accident can take. For example, an initiating event such as spurious shutdown of an incinerator will not result in a significant release of agent to the environment unless numerous ventilation and automatic shutdown systems fail. In most cases, the probability of failure of multiple systems is so low that the frequencies of such accident sequences are too low to be of any concern. Furthermore, because of inherent system inertia and engineered safety features which are provided, there may be ample time to recover and repair mitigating^{*} systems prior to any release.

AMAMMUT DEPENDENT MED VOLDENT DEPENDENT DE COSECUTION DE COSECUTION DE COSECUTION DE COSECUTION DE CO

As suggested above, operator intervention can influence the course of an accident, and therefore his role must be included in the logic models where appropriate. Of course, operating and emergency personnel also have a significant influence on the potential for and amount of accidental agent release.

*"Mitigation" as used in this report is the act of preventing or limiting the consequence of an accident that has occurred. S.2.1.3. <u>Human Interactions</u>. Human interactions, or interventions, of interest to the chemical munitions disposal risk assessment fall into one of the following six general categories:

- Initiation of an accident by committing an error (e.g., a munitions handler punctures or accidentally drops a munition).
- Test and maintenance actions (e.g., a value is disabled or left in the wrong configuration following a test or maintenance act).
- 3. Termination of an accident by correctly implementing established emergency procedures (e.g., an operator terminates agent feed to the liquid incinerator when automatic termination has failed).
- 4. Aggravation of an accident by taking incorrect action (e.g., a plant operator misdiagnoses the nature of the accident and performs an act which causes the accident to have greater consequences).
- 5. Termination of an accident by actions which are outside the scope of existing procedures (e.g., based on his knowledge of the plant or process, a plant operator performs an act which is not covered by procedures and terminates or mitigates the accident).
- Intentional acts to initiate accidents or render equipment in a failed state (sabotage).

Human interactions that fall in the first three categories are modeled either as a separate event heading in the event tree or as an independent event in the fault tree which is used to model and quantify
the event in the event tree. Human interactions defined by categories 4 and 5 above are difficult to quantify and as such are not given much attention in a risk assessment.

Acts of sabotage (category 6) are outside the scope of this analysis and will be addressed elsewhere.

S.2.1.4. <u>Agent Release Characterization</u>. The consequences of an agentrelease event are dependent on the type of agent, the magnitude of the release, the mode and duration of the release, the dispersion of the agent to the environment, the demographic characteristics of the region impacted by the release, and the toxicity of the dispersed agent at the concentration levels to which members of the public are exposed. The scope of effort reported in this document is limited to the first three characteristics listed above. Agent dispersion to the environment and subsequent effects on humans are addressed elsewhere in a separate report.

BATTELLA ANNARY SOUTH, HAVEN MARKED MURLES SPITTLES SECOND SCOTLES

The characterization of agent release required a systematic review of the potential modes of agent release from its normal confinement. The first result of this review was the separation of the accident scenarios into two categories: (1) scenarios that occur while the agent is contained in the munition; and (2) scenarios that occur after the agent is separated from the munition. For the munition-dependent accident scenarios, the agent release mechanism is dependent on the particular mechanical, thermal, and explosive behavior of the munition, assuming the occurrence of an initiating event such as dropping during handling or aircraft crash, as well as the confinement which is provided, if any. Scenarios included in the second group are limited to those which occur during the actual demilitarization process (i.e., plant operations).

After determining that agent could be released in a particular accident sequence and that the frequency of that sequence exceeded the threshold screening frequency, an analysis was performed to identify the

possible paths by which the agent could be released to the environment and to estimate the quantity of agent released.

S.2.1.5. <u>Sequence Screening</u>. The implementation of PRA methodology in terms of event trees can produce a large number of potential accident sequences. In order to reduce this to a manageable number to focus on the critical scenarios for analysis, the accident sequences are screened for frequency or consequence. By using conservative values for the conditional probabilities of event tree branches, it is possible to show that many of the possible sequences are of sufficiently low frequency (e.g., less than 10^{-10} per year) that they need not be addressed further. In addition, if an accident sequence has a frequency greater than the threshold screening frequency but results in an insignificant release of agent^{*} to the environment, it can also be eliminated from further consideration. The accident sequences contained in this report have been subjected to both types of screening.

S.3. RESULTS

The analysis of the potential for agent release to the atmosphere from accident scenarios related to the collocation disposal option included the following major activities: (1) storage, (2) handling activities associated with the transport of munitions, (3) onsite transportation, (4) offsite transportation, and (5) plant operations associated with the demilitarization of munitions. This section discusses some of the accident probability and agent release results associated with these activities.

^{*}Less than 14 1bm of mustard; less than 0.4 1bm of agent VX; and less than 0.3 1bm of agent GB. These quantities represent the minimum quantities of agent release that would result in a lethal dose of agent at 500 m for the most limiting release modes (Ref. S-3).

The results of the analysis of the various activities encompassing the collocation options cannot be presented in the same units, i.e., annual frequencies, because of the possible divulgence of classified information. This is only possible for some storage and plant operation accident scenarios. For accident scenarios related to the handling activities either at the original site, the regional site, or the national site, the unclassified portion of the probabilistic analysis is given in terms of frequency of accidents per pallet of munitions (or as a container of munitions). For onsite and offsite transportation accidents, the basic results are reported in terms of accident frequency per vehicle mile. These probabilities/unit are then multiplied by the number of handling operations or vehicle miles traveled during the stockpile disposal program.

The evaluation of the actual risk to the public and environment requires agent dispersion calculations which are not in the scope of the study reported here. Despite this limitation, the results discussed herein still provide useful insights on the contributions of the various disposal activities to the risk of an agent release. These insights are discussed below.

S.3.1. Accident Scenarios During Storage

The collocation alternative requires some storage of munitions in their existing location prior to transportation to the disposal site. In addition, it requires storage of munitions in offsite transport containers at the sending and receiving sites and some storage at the disposal site before movement to the demilitarization facility.

S.3.1.1. <u>Internal Events</u>. There were no significant internal event initiators of accidents during storage at the disposal site before movement to the demilitarization facility. Per unit operation, forklift drop accidents occur more frequently than forklift time punctures.

Also, the use of a lifting beam instead of a time leads to an order of magnitude decrease in drop frequency.

S.3.1.2. <u>External Events</u>. These events involve accidents caused by natural phenomena or human activity affecting munitions in storage igloos, open storage areas, holding areas, or warehouses. If these are assumed to be full of munitions, the agent inventories range up to 100, 200, 1000, and 2000 tons, respectively, for storage igloos, holding areas, open areas, and warehouses. The most frequent external accidents having significant release involve mild intensity earthquakes or small airplane crashes (order depending on site). Amounts of available agent inventories released in these events are on the order of fractions of one percent or less (munition punctures, drops, etc.).

The largest releases occur for a large aircraft crash, a meteorite strike, or a severe earthquake, especially when a warehouse (at NAAP, TEAD, or UMDA) is involved. These can result in up to 10 percent of the agent inventory released for scenarios involving a fire which has the potential (duration) for destroying the entire inventory of an igloo or warehouse. The munitions stored in warehouses contain only VX or mustard which have much slower evaporation rates than GB and hence are not easily dispersed into the atmosphere. Thus, warehouse scenarios involving only spills are not significant risk contributors. The warehouse at UMDA has the potential for the largest release. Meteorite strike-initiated sequence median frequencies are one to two orders of magnitude lower than the aircraft crash-induced sequence frequencies. As expected, munitions stored outdoors are generally more susceptible to large aircraft crashes than those stored in warehouses or igloos, but releases are lower. Both APG and PBA have ton containers stored outdoors, and the aircraft crash probabilities at these sites are somewhat higher than at the other sites. Igloos appear to provide only minimal protection from direct crashes of large planes, but releases are an order of magnitude lower. The releases are more severe if burstered munitions are involved.

S.3.2. Accident Scenarios During Handling

Included in the handling analysis are (1) single munition or pallet movements by hand, forklift, or other equipment; (2) packing or unpacking pallets into transportation containers; (3) loading and unloading packages from trucks, railcars, aircraft, or barges; or (4) loading and off-loading barges into the oceanfaring vessel (LASH). There are twice as many handling operations at the receiving sites (RDC or NDC) involving collocated munitions that are not in any transportation container. Furthermore, there are more handling operations involving munitions in onsite transport containers (ONCs) than bare munitions or those in larger offsite transport containers (OFCs).

S.3.2.1. <u>Handling for the Rail Alternative</u>. The results indicate that dropped munitions, whether in palletized form or not, occur more frequently than either forklift time puncture or forklift collision accidents. In fact, the frequency of forklift collision accidents which lead to the munitions falling off the forklift is an order of magnitude lower than the drop accidents. Furthermore, the type of clothing an operator is wearing while handling these munitions influence the drop frequency value. An operator wearing Level A clothing is more likely to commit an error that would cause the munition to be dropped than when he is wearing more comfortable clothing.

The results also indicate that spray tanks (in overpacks) have relatively higher drop frequencies than other munitions. This is largely due to the assumption that spray tanks will be lifted and moved to the truck (for loading or unloading) using forklift with tines. The drop frequency using the tines is an order of magnitude higher than with the use of lifting beams.

For bare munitions, the rockets seem to be the most prone to punctures from drops or forklift time accidents. However, the ONC or OFC itself also affects the puncture probability. Because of its weight and larger surface area, the drop of an OFC increases the munition puncture probability by about a factor of 4 to 5 (depending on the munition type and packing density) when compared to a similar drop of an ONC. However, bare munitions have higher puncture probabilities than munitions in ONCs. This observation is of course not quite evident in the final results presented because there are more handling operations involving possible drops of ONCs than bare munitions.

Children and a characteria and a

Bulk items that are punctured lead to larger releases than other munitions such as projectiles or rockets. Bombs are of concern because they contain GB which evaporates more readily than the other agent types. The agent vapor releases range up to 400 lb (thermal failure of all munitions in an OFC).

Within the types of handling accidents, the events designated as HC, which are related to the packaging of munitions in ONCs or OFCs and their movement from storage (sending sites) to the munitions handling igloo (MHI) (receiving sites), predominate over handling accidents related to the facility (HF). This is largely because (1) there are more handling operations involved in the HC accidents, (2) HF accidents generally involve munitions in ONCs, which provides them with some protection from puncture, and (3) HF accidents involving bare munitions occur inside the munitions demilitarization building (MDB) which is designed for vapor containment; hence, including the probability of a detonation which destroys the vapor containment barrier, both the frequency of a release and the release itself are relatively lower.

The frequency results for the handling accidents could not be compared with the accidents from other activities, such as plant operations, because of differences in units. To get some perspective on how they compare on a yearly basis, we can estimate the number of pallets

that could be handled based on the plant annual processing rates. For illustrative purposes we calculate the number of bomb pallets that are required to meet the annual plant processing rate as:

1 X N)

0.2223

5.4 bombs/h x 24 h/day x 5 day/week

x 52 week/yr /2 bombs/pallet = 16,848 pallets/yr

By multiplying the HCl sequence frequency for TEAD $(1.2 \times 10^{-7}/ \text{pallet})$ with the number of pallets/yr, the annual frequency is 2.0 x $10^{-3}/\text{yr}$. Thus, handling accidents which lead to significant agent releases (in particular, agent GB) are dominant risk contributors because of the relatively higher annual frequency values. Of course depending on the actual munition inventory, the value of annual frequency may either increase or decrease when converted to the more meaningful per stockpile basis.

S.3.2.2. <u>Handling for the Air Option</u>. The accident scenarios discussed for the rail option also apply to the air option. Since the air option involves only the movement of munitions from LBAD and APG to TEAD, agent releases from 155-mm projectiles, 8-in. projectiles, rockets and ton containers are of interest. The general observations noted in the discussion of the accident frequencies for the rail option (Section S.3.2.1) also apply here. The accident release is lower for the handling of these munitions since the amounts of GB agent contained in rockets and projectiles are quite small compared to bombs.

S.3.2.3. <u>Handling for the Marine Option</u>. For this option, the ton containers are placed in a transportation container (vault) that is different from the OFC; hence, the handling steps are somewhat different. There are eight sequences related to handling that were identified. Sequence HW34, which involves the dropping of a lighter by a crane while loading into or unloading from the lighter aboard ship (LASH) vessel, has a relatively high frequency of 6.0 x 10^{-6} per shipment. The structural analysis indicates that dropping of the lighter



S.3.3. Accident Scenarios During Plant Operations

the amount of agent released to the atmosphere is small.

Included in the analysis for this phase are all malfunctions during agent processing/incineration within the MDB or external events affecting drained and undrained agent in the MDB, including those in the unpack area (UPA) (up to 10^4 lb of agent available) and munitions awaiting processing in the MHI, up to 3×10^4 lb of agent available. After unpacking, the munitions are processed by conveyor to the burster removal area, mine punch-and-drain area, projectile mortars disassembly area, rocket and burster shearing machines, mine machine for burster removal, a bulk item drain station, a toxic cubicle (TOX) agent storage tank, furnaces for explosive deactivation, metal parts decontamination, and agent and dunnage incinerators, as appropriate.

S.3.3.1. <u>Internal Events</u>. Because of the engineered safety features provided in the plant design, both the frequency of release and magnitude of release associated with accidents initiated by equipment failure and human error are relatively small. Among the large number of accident scenarios analyzed, the highest frequency scenario (P052) is initiated by an inadvertent feed of an unpunched burstered munition to the dunnage incinerator $(10^{-2}/\text{yr}$ for mines; $5 \times 10^{-3}/\text{yr}$ for other munitions). As a result of detonation, one burstered munition inventory is released to the atmosphere as vapor (only up to 15 1b of agent).

The largest amount of agent vapor release occurs for a metal parts furnace explosion (PO44) with ventilation failure (one bulk item inventory release, up to 1700 lb). However, this scenario was assessed to have a very low frequency, around $10^{-10}/yr$. Another event with up to several hundred pounds of vapor release is PO48, munition detonation in

the explosive containment room vestibule with subsequent fire spreading to unpacked munitions. However, this scenario also has a low frequency, around $10^{-9}/yr$.

S.3.3.2. External Events. Aircraft crashes dominate the external event frequency, and there is little difference between direct and indirect crashes. The small difference is attributed to offsetting effects. Although the indirect crash has smaller conditional probabilities of failures than the direct crash, the risk model utilizes a larger target area for the indirect crash. There is very little distinction in the frequency of aircraft crashes with or without fire, since historical data indicate that there is roughly a 50 percent chance that the crash of an aircraft will involve a fire. The frequency of a crash onto the MDB is considerably larger than that for the MHI because the surface area of the MDB is more than 30 times larger than the MHI.

The frequency of large aircraft crashes is estimated to be higher at ANAD than it is for TEAD. This impacts the regional versus national collocation option. The accident scenario involving the crash of an airplane onto the outdoor agent piping system for the modified CAMDS facility at TEAD has a frequency of about $10^{-8}/yr$ with up to 55 lb of vapor release. This scenario includes both large and small aircraft crashes. The frequency of small aircraft (including helicopters) crashes is at least two orders of magnitude higher than the frequency of large aircraft crashes at TEAD.

The frequencies of earthquake-induced accident scenarios are generally higher for TEAD than for ANAD since TEAD is located in a region more prone to earthquakes. Sequence PO33, which represents an earthquake-initiated munition fall and fire but with the MDB and TOX intact, has the highest frequency (2 x 10^{-6} /yr for ANAD and 5 x 10^{-5} /yr for TEAD). This sequence involves the detonation of all munitions (if burstered) in the UPA since the fire is not suppressed in this sequence.



All accident sequences related to tornadoes or meteorites were estimated to occur at frequencies of less than $10^{-10}/yr$ and thus were screened out.

S.3.4. Accident Scenarios During Transport

S.3.4.1. Onsite Transportation. There are two truck transportation phases considered in the analysis. At the sending sites, munitions in offsite transportation packages are transported by truck to the holding area prior to loading into the train, airplane, or barge. The accidents are identified as the VR, VA, or VW (i.e., for rail, air, and water, respectively) scenarios. At the receiving sites, munitions still in offsite packages are moved to storage locations where they are removed from the offsite package and stored until they are ready for demilitarization. The accidents are also coded VR or VA. Finally, when munitions at their storage locations are ready for demilitarization, they are transferred into onsite containers and then moved by truck to the MHI. The accidents are identified as VO scenarios to distinguish between the transportation risk of using an onsite package versus an offsite package (different failure thresholds). The agent available in a truck carrying an OFC is less than 3400 lb, while up to 7000 lb is available for an ONC truck transport.

As a result of analysis for both internally initiated events (human error or equipment failure) and externally initiated events, the following conclusions were reached:

1. The offsite transportation package provides munitions with more protection from crush forces generated from truck accidents than the onsite package. Hence, sequences with OFC crush have insignificant accident frequency whereas scenarios with ONC crush have frequencies up to 10^{-8} /truck-mile.

2. Both packages provide similar protection from impact forces. The results show that accident frequencies resulting in impact failure are insignificant. This is largely due to the administrative control to be imposed during truck travel which limits truck speed to no more than 20 mph. The impact forces at this velocity are not sufficient to breach the containment.

- 3. The probability of puncture resulting from truck collision/ overturn is at least an order of magnitude higher for offsite containers than onsite containers. This results from the higher likelihood of generating a probe sufficient to puncture the container and the munition when the accident involves a large package such as the OFC.
- 4. Truck accidents which generate fires are more likely to detonate burstered munitions inside onsite packages, since they provide only a 15-min protection from an all engulfing fire (versus 2 h for the OFC). However, all these scenario frequency results are also quite low because of the administrative control for limiting the amount of fuel in the truck so as not to exceed a 10-min fire.
- 5. When rockets are involved in the accidents which generate sufficient impact forces to cause propellant ignition, there is very little distinction in the results for the two packages.
- For tornado-initiated accidents, puncture as a result of truck overturn is the dominant contributor to the sequence frequency.
- 7. Generation of undue forces during truck accidents that could cause burster detonations has a small contribution to the overall truck transportation risk.



BASESSES

PRIMIN SUCCESS ALLENDA ISSOURCE

- 8. The amount of agent spilled or burned during truck accidents resulting in the breach in containment by puncture forces generally involve the agent content of one munition. Up to 10 percent is released as vapor.
- 9. Both containers can fail when an aircraft crashes into the truck (VR6, VR7, V06, V07). The entire truckload is involved, and up to 10 percent is released as a vapor. Hence, aircraft crash-initiated truck accidents have the most severe consequences. It should be noted, however, that none of the accident sequences has a frequency greater than $10^{-7}/yr$.

S.3.4.2. <u>Offsite Transport - Rail</u>. In this option, munitions in OFCs are transported by rail either to two regional destruction centers (RDC-ANAD or RDC-TEAD) or a single national destruction center (NDC-TEAD). The agent inventory available per railcar ranges up to 7000 lb. Results of the accident analysis indicate the following:

- Rail accident crush and impact forces are very unlikely to fail an OFC and munition inside.
- 2. The major risk contribution due to mechanical failure comes from a probe such as a railcar coupler (generated from train accidents) capable of puncturing the OFC and the munition. Munition failure frequency by puncture (RC3) is about an order of magnitude higher than train accidents which lead to fire and cause the thermal detonation or rupture of munitions (RC4 and RC5). However, the consequence (i.e., agent release) from the latter sequence is more severe.
- 3. For tornado-initiated accidents (RC14), puncture as a result of train derailment is the dominant contributor to the agent release frequency.

4. Aircraft crash into a train can damage the munitions (RC6 and RC7). The crash can involve one or two railcars (i.e., up to four OFCs). The largest amounts of agent released are from the bulk items (bombs, ton containers, and spray tanks). A maximum of 10 percent of the inventory is released as vapor (up to 1400 lb). This is the largest release for rail scenarios.

S.3.4.3. Offsite Transport - Air Option. The air transport option applies only to the movement of ton containers from APG to TEAD, and rockets and projectiles from LBAD to TEAD. Five generic sequences related to air transport were identified. These scenarios were evaluated for both the C-141 and C-5 aircrafts. There will be approximately 1500 flights from LBAD and 300 flights from APG for the C-141 aircraft. The C-5 aircraft would decrease the number of required flights by one fourth. The analysis also differentiated among accidents which occur during takeoff, while in flight, and during landing. Each flight would carry up to 3400 lb of agent inside OFCs.

The aircraft accident frequency during landing is about seven times higher than during takeoff and about three times higher than inflight accidents. However, the failure probability of the package due to impact forces is higher inflight than either takeoff or landing. If an aircraft crash occurs, the OFC and the munitions are subjected primarily to impact forces sufficient to fail the package. The accident frequencies from sequences which involve impact only are almost of the same order of magnitude as sequences which involve impact and fire (AAl versus AA20). The accident frequencies involving the C-5 aircraft are an order of magnitude higher than those for C-141 aircraft. A compensating factor is that there will be 75 percent fewer flights if the C-5 is used.



Accident scenarios involving fire of sufficient duration to fail the packages are not credible for the C-141 aircraft because of insufficient fuel available to sustain a fire of duration to fail the package containment.

Accidents which lead to severe impact (AA1 and AA2; AB1 and AB2) without fire have the highest frequency and also lead to the largest amounts of agent released. For severe impact release involving burstered munitions, some of the munitions contained in the aircraft will detonate, and up to just over 400 lb will be released as vapor. For accidents involving moderate impact forces, no agent release occurs from impact alone. The moderate impact accident must be accompanied by fire to fail the package thermally.

S.3.4.4. Offsite Transport - Marine Option. The marine option was analyzed only for the movement of ton containers filled with mustard at APG to the Johnston Atoll. There were five groups of initiating events identified. Impact and puncture are not the dominant failure forces experienced in marine accidents. The cargo will be adequately braced to hold it in place. Furthermore, most of the events are low-velocity, high-momentum events; hence, the dominant failure mode is crush. Fire, immersion, and aircraft crash events were also considered because of the large amount of agent being transported which could be involved in fire or sinking accidents.

The results indicate that:

- For the lighters in the Chesapeake Bay, collision accidents are at least three orders of magnitude more probable than either rammings or groundings.
- 2. For the LASH vessel in the Chesapeake Bay, both grounding and collision accidents are at least one order of magnitude more probable than rammings.

- 3. Grounding of the LASH vessel in the coastal areas is less likely than in shallower inland waters.
- For the LASH vessel in high seas, collision is still the predominant event. However, grounding results in more severe consequences.

The agent release analysis shows that collisions result in the largest number of ton containers (TCs) which fail (8) for barges, but that groundings or heavy weather damage results in the maximum number of TCs failed (68) for the LASH (except for aircraft crash, which is below the frequency screening threshold). The largest amount of agent vapor release to the atmosphere occurs for these worst events, and the amounts are not strongly dependent on whether fire occurs or not. Although a large inventory (up to 4 million 1b on the LASH) is available, no accident leads to a release of more than 0.1 percent.

S.4. UNCERTAINTIES IN THE ANALYSIS

In assessing the risks associated with the CSDP alternatives, every effort was made to perform best-estimate analyses, i.e., "realistic" evaluation and quantification of the accident sequence frequencies and associated agent releases. The use of pessimistic or conservative modeling techniques or data for quantification violates the intent of the probabilistic nature of the study. Realistic modeling and quantification permits a balanced evaluation of risk contributors and comparison of alternatives. However, for realistic or best-estimate calculations, the obvious concern is the accuracy of the results. Uncertainty analysis addresses this concern.

S.4.1. Sources of Uncertainty

Since the event sequences discussed in Section S.3 have not actually occurred, it is difficult to establish the frequency of the sequence and associated consequences with great precision. For this reason, many parameters in a risk assessment are treated as probabilistically distributed parameters, so that the computation of sequence frequencies and resulting consequences can involve the probabilistic combination of distributions.

There are three general types of uncertainty associated with the evaluations reported in this document: (1) modeling, (2) data, and (3) completeness.

There exist basic uncertainties regarding the ability of the various models to represent the actual conditions associated with the sequence of events for the accident scenarios that can occur in the storage and disposal activities. The ability to represent actual phenomena with analytical models is always a potential concern. The use of fundamental models such as fault trees and event trees is sometimes simplistic because most events depicted in these models are treated as leading to one of two binary states: success or failure (i.e., partial successes or failures are ignored). Model uncertainties are difficult to quantify and are addressed in this study by legitimate efforts of the analysts to make the models as realistic as possible. Where such realism could not be achieved, conservative approaches were taken.

No uncertainty from oversights, errors, or omission from the models used (e.g., event trees and fault trees) is included in the uncertainty analysis results. Including these uncertainties is beyond the state-ofthe-art of present day uncertainty analysis.

The uncertainties in the assignment of event probabilities (e.g., component failure rates and initiating event frequencies) are of two types: intrinsic variability and lack of knowledge. An example of intrinsic variability is that where the available experience data is for a population of similar components in similar environments, but not all the components exhibit the same reliability. Intrinsic variations can be caused, for example, by different manufacturers, maintenance practices, or operating conditions. A second example of intrinsic variability is that related to the effects of long-term storage on the condition of the munitions as compared to their original configuration. Lack of knowledge uncertainty is associated with cases where the model parameter is not a random or fluctuating variable, but the analyst simply does not know what the value of the parameter should be. Both of these data uncertainty types are encountered in this study.

S.4.2. Uncertainties

The sequence frequency results discussed in this report are presented in terms of a median value and a range factor of a probability distribution representing the frequency of interest. The range factor represents the ratio of the 95th percentile value of frequency to the 50th percentile (i.e., median) value of frequency. The uncertainty in the sequence frequency is determined using the STADIC-2 program (Ref. S-4) to propagate the uncertainties associated with each of the events in the fault trees or event trees through to the end result. Some scenarios, such as those associated with tornado missiles and lowimpact detonations have rather large uncertainties. The difficulty with tornado-generated missiles lies with the difficulty in accurately modeling the probability that the missile will be in the proper orientation to penetrate the munition and in predicting the number of missiles per square foot of wind. The difficulty with the low-impact detonations lies with the sparse amount of data available and its applicability to the scenarios of interest. In general, uncertainties tend to be large when the amount of applicable data is small and vice versa.

S.5. REFERENCES

- S-1. U.S. Department of Defense, "Military Standard System Safety Program Requirements," MIL-STD-882B, March 30, 1984.
- S-2. U.S. Nuclear Regulatory Commission, "PRA Procedure Guide," NUREG/ CR-2300, 1982.
- S-3. Memorandum to C. A. Bolig (GA) from R. B. Perry (PEO-PM Cml Demil), May 6, 1987.
- S-4. Koch, P., and H. E. St. John, "STADIC-2, A Computer Program for Combining Probability Distribution," GA Technologies Inc., GA-A16277, July 1983.

KUNAN NUNUN NUNUN

2.222.22

1. INTRODUCTION

1.1. BACKGROUND

The U.S. Department of Defense is required by Congress (Public Law 99-145) to destroy the stockpile of lethal chemical agents and munitions stored at eight U.S. Army installations in the continental United States (CONUS) and at the Johnston Atoll Army site in the Pacific Ocean by the end of September 1994. The locations of the CONUS sites are shown in Fig. 1-1. The total Army stockpile at these sites is made up of more than 3,000,000 items consisting of rockets, mines, mortars, projectiles, cartridges, bombs, spray tanks, and bulk containers. These munitions contain the nerve agents GB and VX and the blistering mustard agents H, HD, and HT.

The Army has developed a plan for destruction of the chemical munition stockpile. This plan is set forth in the Chemical Stockpile Disposal Concept Plan submitted to Congress in March 1986 and supplemented in March 1987. In this plan, three disposal alternatives are described:

- Disposal of the agents and munitions at each of the eight existing storage sites.
- Collocation and disposal of the munitions at two regional sites.
- Collocation and disposal of the munitions at a single national site.

These three disposal alternatives were also described in a Draft Progammatic Environmental Impact Statement published by the Army in

Sorra assessed teacada tossessa presentativeressantessessantativera teacada teacada teacada teacada



Fig. 1-1. Location of chemical agents and munitions in the U.S.



July 1986. Additionally, it was required that the status quo, i.e., continued storage, be also evaluated as the fourth alternative. As part of the public commentary on this document, requests were made of the Army to consider also the transport of the inventory from Aberdeen Proving Ground to Johnston Island by water or to Tooele Army Depot by air and from the Lexington-Blue Grass Army Depot to Tooele by air. These alternative options for offsite transport were also investigated during the study reported here. They represent subset options for the collocation option. Received by South Breaker Brown

Under direction from the U.S. Army Office of the Program Executive Officer Program Manager for Chemical Demilitarization (PEO-PM Cml Demil), GA Technologies Inc. (GA) and its subcontractors have performed a comprehensive probabilistic assessment of the frequercy and magnitude of agent release associated with activities involving the three disposal alternatives currently set forth in the Chemical Stockpile Disposal Program (CSDP), as well as the continued storage alternative. This assessment has been carried out in support of the environmental impact statement (EIS) for this program and it addresses only the stockpile of chemical munitions which are currently stored at the eight sites located in the continental United States (CONUS).

When combined with an assessment of the consequences (injuries and/or deaths) to the public resulting from the accident sequences and associated agent releases identified and evaluated in this study, the results form a basis for an assessment of public risk. The dispersion of the agent to the environment and the assessment of consequences related to these releases are outside the scope of this study. A consequence assessment has been performed by MITRE Corporation and Oak Ridge National Laboratory for the EIS, based on the releases identified in this document.

This report addresses the collocation alternatives identified above. The remaining alternatives are discussed in separate reports.

1-3

Anniston Army Depot (ANAD) in northeast Alabama and Tooele Army Depot (TEAD) in north central Utah have been identified as the regional sites, assuming this collocation alternative is selected. Should the single national site collocation alternative be selected, that facility would be at TEAD.

Previous studies have been utilized by GA as reference bases for this assessment. Quantitative hazards analyses were performed by Arthur D. Little, Inc. on the disposal of M55 rockets (Refs. 1-1 to 1-5), and qualitative hazards analyses were performed by the Ralph M. Parsons Company on the Johnston Atoll Chemical Agent Disposal System (JACADS) design (Refs. 1-6 and 1-7). In addition, a probabilistic analysis of chemical agent release during transport of M55 rockets has been performed by H&R Technical Associates (Ref. 1-8), and a probabilistic analysis of selected hazards during the disposal of M55 rockets has been performed by Science Applications International Corporation (Ref. 1-9). These studies provided the set of accident scenarios that was compiled in a systematic order by MITRE Corporation (Refs. 1-10 and 1-11). GA, in turn, used these accident scenarios as a starting point in this study.

The analyses performed by Arthur D. Little, Inc. used a technique known as hazard and operability analysis (HAZOP). HAZOP involves a detailed review of plant design to trace all parts and functions of the demilitarization process. For each piece of equipment or pipe run, deviations from normal operating conditions were examined and possible consequences were discussed. Through this approach, potential failure modes leading to agent release outside of the facility were identified. The expected frequencies of occurrence of all agent release sequences identified in the HAZOP were then evaluated using fault tree analysis.

1-4

The qualitative hazards analysis performed for JACADS used an approach known as failure modes and effects analysis (FMEA). The severity and probability levels of identified hazards were ranked according to the guidelines in Ref. 1-12.

The transportation studies performed by H&R Technical Associates (Ref. 1-8) used a combined fault tree and event tree approach to assess the frequency of agent release from transportation accidents.

The work performed by Science Applications International Corporation (Ref. 1-9) on the disposal of M55 rockets utilized both event tree and fault tree methodology as used in the PRA of nuclear power plants. Demilitarization of the chemical munitions stockpile requires the construction of facilities to destroy the contents of the munitions, the handling, transportation, and storage of munitions at both the "sending" and the "receiving" site(s), the transport of munitions between sites, the destruction of the munitions, and the decommissioning of the constructed facilities. This report addresses each of these activities, except for facility construction and decommissioning.

1.2. STUDY OBJECTIVES AND SCOPE

The primary objectives of the study reported in this document were to:

- 1. Identify events (for each major activity) that could initiate the release of agent to the environment.
- 2. Develop the various sequences of events resulting from these initiators and leading to agent release.
- 3. Perform a quantitative analysis of the frequency of occurrence of each relevant accident sequence.
- 4. Characterize the form, quantity, and duration of agent release from each accident sequence.
- 5. Identify accident sequences which make the most significant contributions to risk.

The major deliverables of this effort are a list of potential accident sequences for each major activity, the estimated frequencies of these sequences, and the magnitudes of released agent associated with these sequences. It should be noted that only accident sequences that survived a conservative screening process, involving both frequency and magnitude of agent release, are included in these deliverables.

This report addresses each of the objectives listed above and presents the analysis of this study. The risk analysis includes an evaluation of potential accidents and natural occurring phenomena such as earthquakes and tornadoes. Acts of war, sabotage, and terrorism, which involve intentionally-initiated events, were not included in the scope of this effort.





The term "chemical munitions" is used here to describe both burstered chemical munitions and chemical bulk items. The 4.2-in. mortars refer to the actual 4.2-in. projectile which is fired from mortar cannons or tubes. The 105-mm cartridge and 4.2-in. mortar projectile can either be configured with propellant (i.e., a cartridge) or without propellant (i.e., a projectile); in this study, it was assumed that the propellant and fuze were removed prior to the onset of the disposal program.



1.3. DEMILITARIZATION ACTIVITIES AND SAFETY CONCERNS

Figure 1-2 shows a comparison of the various logistics phases associated with the various munition disposal and storage alternatives evaluated for the EIS. As indicated in this figure, the demilitarization process associated with the two collocation alternatives can be divided into five general areas of activity: storage, plant operations, handling, onsite transport and offsite transport. Except for the offsite transport activity, the onsite disposal alternative involves the same logistic phases. In contrast, only the storage activity is of concern for the continued storage option.

For each of these activities or phases, the hazards of interest are those involving the evaporative release of agent to the environment resulting from spills, leaks, and mechanical failures, and the release of agent to the environment resulting from fires and explosions. The generation of these potential hazards originates with a number of "internal" and "external" initiating events. The number of hazardinitiating event combinations is rather extensive. However, because of the screening process which was used to remove from further consideration the accident sequences whose frequency was low and/or the associated magnitude of agent release was low, the number of individual sequences which are important to risk is relatively small.

1-8

|--|



1.1.1.1.1.1.1.1.1.1

| ť | | | | | |
|----------------------|---------------------|----------------------|------------------------|------------------|---------------------------------------|
| OFFSITE TRANSPORT | | | | | RAIL AND AIR OR SHIP |
| ONSITE TRANSPORT | |) | A TNO ONC ONC | | ONC AND OFC UNITS |
| HANDLING | | | BARE BARE UNITS | | BARE AND ONC AND OFC UNITS |
| PLANT OPERATIONS | | | 8 CONUS SITES | | 1 OR 2 SITES |
| STORAGE | | LONG TERM | SHORT Short Term | • • • • | SHORT Short Term and Interim |
| L | | CONTINUED Storage | ONSITE DISPOSAL | | REGIONAL/ National Disposal |

Fig. 1-2. Logistic phases associated with the munitions storage and disposal options NAMES OF STREET

EN EREN DEREN EREN EREN

1.4. STUDY ASSUMPTIONS

The risk analysis presented in this report uses an approach that combines the structured safety analysis detailed in MIL-STD-882B (Ref. 1-12) and the probabilistic approach used in the safety analyses of nuclear power plants (Ref. 1-13). Reference 1-12 requires that hazards analyses be performed in order to assess the risk involved during the planned life expectancy of a system. It also provides some guidance on the categorization of hazard severity and probability as a means of identifying which hazards should be eliminated or reduced to a level acceptable to the managing activity.

The risk analysis was performed under the following set of general assumptions:

- Onsite transportation of munitions will be by truck. The baseline offsite transportation mode analyzed was rail. Several specific offsite transport options by air or marine craft for selected stockpiles were also analyzed.
- Munitions will be stored in their current storage locations and will be transferred to the demilitarization site as needed.
- 3. The baseline process design will be used (i.e., JACADS type facility). At TEAD, some existing process equipment will be used. Both of the collocation alternatives include a bulkonly facility, as well as a mixed munition plant design similar to the JACADS design. The design of the CONUS demilitarization facilities is now approximately 35% complete.

and assessmentsseers and a sublimited of the second of the

1-10



4. Munitions are in good condition during the handling, transportation, and disposal activities. 5. Sabotage or terrorism is not considered.

A detailed listing and discussion of assumptions is presented in Appendix E.

1.5. REPORT FORMAT

This report is structured as outlined schematically in Fig. 1-3. The structure follows that typically used in comprehensive probabilistic risk assessment (PRA) studies.

Following the introduction in Section 1 of this report, Section 2 provides a summary of the methodology used in this assessment, including the procedure for accident scenario identification and screening, the approach used for quantifying accident frequencies and characterizing agent release, and the treatment of uncertainties.

Section 3 provides a brief discussion of the various activities involved in the disposal of chemical munitions. This discussion is provided to assist readers in the understanding of the initiating events and accident scenarios that have been identified and are discussed in Sections 4 through 8. This section also discusses site-specific information that is important to a particular site. Appendix D contains additional site information.

The list of accident initiating events which have been analyzed is along with the analysis of their occurrence frequencies are presented in Section 4. These events include accidents from internal causes, such as inadvertent impact during handling, and accidents caused by external events, such as earthquakes or aircraft crashes.

Sections 5 through 8 present the detailed development and analysis of the key accident scenarios resulting from the initiating events.

Section 9 provides the basis for quantification of accident sequence frequencies including munition failure probabilities, the data base used for estimating the probabilities of event-tree top events and fault-tree basic events, and the data used for assessing human error.



N.N.S. 200000 - 2000

1.01.01.01.01

Fig. 1-3. Outline of report structure

12120000000000

 $X \in X \setminus X$

1.1.1.1

۰.

10.78

The characterization of agent released in the various accident sequences is discussed in Section 10.

Section 11 presents the overall results of the analysis. The results presented in Sections 4 through 8 are summarized for both collocation alternatives to highlight the accident sequences which are predicted to have the highest frequencies of occurrence or large agent releases.

Supporting data and calculations for the study are contained in the appendices. References to appropriate appendices are made throughout the body of the report.

1.6. REFERENCES

CONSTRACT

- 1-1. Arthur D. Little, Inc., "Analysis of Risks Associated with Operation of Onsite M55 Rocket Demilitarization Plants," U.S. Army Toxic and Hazardous Materials Agency, M55-OD-3, May 1985.
- 1-2. Arthur D. Little, Inc., "Analysis of Risks Associated with Operation of the JACADS M55 Rocket Demilitarization Plant," U.S. Army Toxic and Hazardous Materials Agency, M55-0D-9, May 1985.
- 1-3. Arthur D. Little, Inc., "Analysis of Converted M55 Rocket Demilitarization Plant at Pine Bluff Arsenal," U.S. Army Toxic and Hazardous Materials Agency, M55-OD-10, May 1985.
- 1-4. Arthur D. Little, Inc., "Analysis of Risks Associated with Operation of the CAMDS M55 Rocket Demilitarization Plant," U.S. Army Toxic and Hazardous Materials Agency, M55-OD-11, May 1985.
- 1-5. Arthur D. Little, Inc., "M55 Rocket Separation Study," U.S. Army Toxic and Hazardous Materials Agency, M55-0D-08, November 1985.
- 1-6. The Ralph M. Parsons Company, "JACADS-Preliminary Hazards Analysis of Final Concept Design," Task G-56 Interim Report, May 1983.
- 1-7. The Ralph M. Parsons Company, "Fault Tree Analysis, Johnson Atoll Chemical Agent Disposal System," Task G-SC, U.S. Army Engineers Division, February 1985.
- 1-8. Rhyne, W. R., et al., "Probabilistic Analysis of Chemical Agent Release During Transport of M55 Rockets," H&R Technical Associates, Inc., Prepared for USATHAMA, M55-CD-4, H&R 255-1, September 1985.
- 1-9. Science Applications International Corporation, "Probabilities of Selected Hazards in Disposition of M55 Rockets," U.S. Army Toxic and Hazardous Materials Agency, M55-CS-2, November 1985.
- 1-10. Fraize, W., et al., "Preliminary Risk Analysis Supporting the Chemical Stockpile Disposal Program (CSDP), Volume 1: Analysis," MTR-86W4-01, the MITRE Corporation, May 1987.
- 1-11. Fraize, W., et al., "Preliminary Risk Analysis Supporting the Chemical Stockpile Disposal Program (CSDP), Volume 2: Appendices," MTR-86W4-02, the MITRE Corporation, May 1987.

- 1-12. U.S. Department of Defense, "Military Standard System Safety Program Requirements," MIL-STD-882B, March 30, 1984.
- 1-13. U.S. Nuclear Regulatory Commission, "PRA Procedure Guide," NUREG/CR-2300, 1982.

2. RISK ASSESSMENT METHODOLOGY

2.1. OVERVIEW

The probabilistic risk assessment (PRA) methodology used in this study is generally consistent with the PRA Procedures Guide (Ref. 2-1) for nuclear power plants. Figure 2-1, adapted from that guide, outlines the risk assessment procedure for this study. Certain specific features of the demilitarization process dictate some different emphasis and treatments from those described in Ref. 2-1. The risk assessment steps corresponding to the procedures in Fig. 2-1 are as follows:

- Identify accident initiators (initiating events) through information collection, hazards analyses, or the use of master logic diagrams. The initiating events are classified as external if they originate from outside the demilitarization process (such as aircraft crash) and as internal otherwise.
- 2. Define accident scenarios, i.e., combination of initiating events and the successes or failures of systems that respond to the initiating event. An "accident sequence" is referred to in this report as a specific end point of an accident scenario, which is usually modeled using event trees. An "event tree" is an inductive logic model which traces the sequence of events that can occur following an initiating event.
- 3. Construct "fault trees" (deductive system logic models) to determine the root causes of individual system failures. The fault tree is reduced to minimal cut sets using Boolean algebra. A "minimal cut set" represents a unique combination of events leading to system failure.



all sea all sea

Outline of risk assessment procedure used in this study Fig. 2-1.

0

U,

97



2-2
4. Assign failure rates or probabilities to events (components or subsystem) modeled in the event trees and fault trees. Quantify the frequencies of occurrence of accident sequences from either the event tree or fault tree by computing the product of the initiating event frequency and the probabilities of the subsequent conditional events in a given accident scenario. 1222223

- 5. Determine the consequences of the accident sequences. In this analysis, the consequence of concern is the amount of agent released to the local free environment. The impact of agent release on the population will be used by others in their CSDP analysis.
- 6. Evaluate the uncertainties in the data base, and predict the uncertainty in each relevant accident sequence frequency by propagating the top event uncertainties through the event trees.
- 7. Present the results (i.e., accident scenario frequency and consequence) in a form that will best show those scenarios that are important to risk and will reflect the uncertainties associated with the accident sequence frequency.

Ľ,

2.2. INITIATING EVENTS

22077777 A. 666 S. 65

An initiating event is a single occurrence or malfunction that has the potential to release one or more agents or to start a sequence of events that could lead to a release. The list of initiating events is developed based on previous demilitarization studies (Section 1.2) and related PRAs such as Waste Repository studies (e.g., Ref. 2-2), in addition to the use of master logic diagrams.

The initiating event list is developed in top-down fashion by structuring a master logic diagram to define a functional set of initiating categories. These categories form a complete set in the sense that any event which leads to agent release must cause at least one of these categories to occur.

Some "common cause initiating events" (e.g., an earthquake) can activate more than one initiating event category and disable controls for release. While there is no way to guarantee that all such events are identified, two areas yield the most significant events. The first includes severe environmental events (such as fire, flood, earthquake, and wind) as well as hazardous activities in the vicinity (such as aircraft patterns). The second area includes malfunctions that can affect multiple controls or barriers for the prevention of release to the atmosphere.

Coincident with the development of the list of initiating events is the assessment of the initiating event frequencies. This is required, first, for subsequent quantification of event trees, since the event initiator is the first event of the tree. Second, it enables screening of the list of initiating events, i.e., events having extremely low frequencies can be eliminated. Where possible, the initiating events are grouped into categories when the subsequent event tree and release analysis development is the same for all initiating events in the category. This grouping is performed by Boolean summation of the occurrence frequencies, accounting for dependencies, if any. TELEVISION CONTRACTOR DESCRIPTION

- 22202/1

2.3. SCENARIO DEVELOPMENT AND LOGIC MODELS

Given the occurrence of an initiating event (IE), accident scenarios are developed, in many cases using logic models of either event trees, fault trees, or both, to arrive at the various outcomes of the scenario progression. Each of these outcomes, termed a sequence, is associated with (or even characterized by) a certain level of agent release. The basic premise of the risk summation process is that release frequencies (initiating event frequency multiplicatively combined with probabilities of subsequent failures necessary to get the release) of entirely different sequences can be additively combined to get the overall frequency of release. The additive and multiplicative combination is performed using Boolean algebra and accounts for dependencies.

Figure 2-2 shows a sample event tree. In this example, the IE is a vehicle collision, having an estimated occurrence frequency which can be a point estimate or be probabilistically distributed. The IE is the first "top event," and potential subsequent failures represent the other top events or branch points. These top events are in the form of questions, and by convention the upper branch represents the positive answer sequence and the lower branch is the negative answer sequence. Branch split fractions or probabilities are assigned at each of these branch points. These split fractions may be point estimates or probabilistic distributions, and may not be the same for all branch points under a specific top event, depending on prior events. That is, the split fractions represent conditional probabilities.

The frequency of an accident sequence is calculated based on the following equation:

$$F_{j} = I_{j} \prod_{i=1}^{n} P_{i,j} ,$$

(2-1)

NEGLIGIBLE AGENT RELEASE HIGH HIGH N/A PACKAGE INTACT YES Q Ŷ NO DETONATION PREVENTED YES **N** Q FIRE PREVENTED **OR CONTAINED** YES **N** INITIATING EVENT VEHICLE COI LISION

Fig. 2-2. Accident scenario development using an event tree

Alta IA Leve



**

where F₁ = frequency of accident sequence j,

- I₁ = initiating event frequency,

Accident frequency and equipment/component failure rate data were derived from various sources, as described in Section 9.

In this study, the event trees are relatively simple in form compared to those developed for nuclear plant PRAs. Most dependencies are modeled explicitly in the event trees by use of conditional branching probabilities which are dependent upon the branch taken for prior events. For example, in an event tree where two consecutive top events represent the availabilities of systems 1 and 2, system 2 might not be called upon unless system 1 fails. This would be shown in the event tree by a dashed line for system 2 in the system 1 success branch, indicating not applicable. Conversely, if system 2 is capable of operating only in conjunction with successful operation of system 1, the dashed line is shown on the system 1 failure (no) branch for system 2 top event. This indicates a guaranteed failure of system 2, given nonoperation of system 1.

For many scenarios, it was found convenient to use fault tree logic for development of the accident progression and quantification of the sequence frequencies. Figure 2-3 depicts a sample fault tree. Logic symbols used in constructing fault trees are defined in Fig. 2-4. The approach taken for treatment of dependencies in the event trees is to identify specific intercomponent and intersystem causes of multiple failures, if any, directly in the fault tree and to make an allowance for those not explicitly identified. A Beta factor method (e.g., Ref. 2-3) is a convenient tool for determining a suitable allowance and was used where appropriate. In this method, multiple failures of redundant components are assumed to occur in a dependent fashion; the





Fig. 2-4. Definition of fault tree symbols

parameter β is defined as the fraction of failures experienced in components that are common cause failures.

Just as there are uncertainties in estimating component failure rates, there are also uncertainties in the β factor. These uncertainties were quantified assuming lognormal distribution for the β factor. The uncertainty distribution accounts for uncertainties due to sparsity of data, as well as those due to classification and the so-called "potential common cause failures." These are events in which one failure actually occurs and additional failures could have occurred under different circumstances, as well as incipient failures and degraded operability states.

In the case where the fault sequence i, given an initiating event, involves a subsystem or equipment failure, the failure probability calculations may involve not only the calculation of the unavailability value (probability of failure per demand) but also the unreliability value (probability of failure while component/equipment is running). In this case, the overall failure probability value for a given equipment or subsystem is calculated using the following equation (Ref. 2-3):

 $P_i = P_{i,d} + (1 - P_{i,d}) P_{i,r}$, (2-2)

where $P_{i,d}$ = failure upon demand (unavailability), $P_{i,r}$ = failure while running (unreliability).

The calculation of component unreliability $(P_{i,r})$ is influenced by several factors: (1) the frequency of periodic maintenance (PM); (2) the use of different failure detection systems; and (3) the various methods used to monitor equipment operation.

For the analysis presented in this report, two options were considered in the calculation of component unreliability. The first option was to consider the periodic maintenance of a component. Thus, when a



component is periodically removed from service for preventative maintenance, the failure probability is dominated by the maintenance interval in addition to the failure rate according to the following equation:

$$P_{i,r} = \frac{1}{\lambda\theta} (1 - e^{-\lambda\theta}) \approx \frac{\lambda\theta}{2} , \qquad (2-3)$$

where λ = failure rate,

 θ = maintenance interval.

The second option was to consider continuous component surveillance which decreases the failure probability by announcing component failure to the operators concurrent with failure initiation. The repair time required to restore the component becomes an important factor as shown in the following equation:

$$P_{i,r} = \frac{\lambda}{\lambda + \nu} \left[1 - e^{-(\lambda + \nu)t} \right] , \qquad (2-4)$$

where $\nu = 1/\tau$ mean repair rate (per h),

- τ = repair time (h),
- t = time interval of interest.

In Eq. 2-5 the failure probability approaches $\lambda \tau$ as the time interval increases and $\lambda \tau$ is small (i.e., $\lambda \tau \langle \langle 1 \rangle$.

In most of the component failures identified in the fault tree models, the first option is used (i.e., calculating reliability as a function of maintenance interval) and a monthly PM interval is assumed (i.e., maintenance interval of 528 h) for the equipment. This is a conservative approach in deriving the failure probability. If a more frequent maintenance policy is adopted or if experience shows that the component restoration time is much less than the maintenance interval, the failure probability will decrease. However, in view of the nature of the fault tree models, this approach seems justified because the







.....

failure contribution of a particular component is not negated by assuming an unnecessarily low failure probability.

2.4. HUMAN FACTORS

The treatment of intersystem and intercomponent equipment dependencies is discussed above, including how dependencies are taken into account by the logic models. This section describes another kind of dependence--that involving human interaction.

To the extent that human beings design, construct, operate, and maintain the plant, it is impossible to fully isolate the role of human interactions from any of the dependencies discussed above in terms of hardware interactions. Hence, all of the common cause analysis methods described above pertain directly or indirectly to human interactions. The discussion is restricted here to human intervention in the operation and maintenance processes.

The procedure used for analysis of intersystem and intercomponent dependencies caused by human interactions was to include human errors of omission and commission explicitly in the event tree/fault tree models and to use the human reliability methods of Swain (Ref. 2-4) to implement quantification. A starting point for the identification of specific errors is the analysis of operation and maintenance procedures if they have been defined for the event sequence being investigated. This is especially important if operator action is required to effect actuation of a system or a collection of systems. Consideration needs to be given to possible incorrect judgments as to the plant state and subsequent implementation of the wrong procedures. Once these acts are identified and modeled, the problem of determining contribution to risk by operator actions is reduced to assigning the correct human error rate values.

2.5. RELEASE CHARACTERIZATION

The risk associated with each accident scenario requires not only the quantification of the frequency of that scenario but a characterization of the agent release as well. This characterization involves the type and amount of agent released, and the mode duration of the release.

At any given time, there is at least one containment barrier separating the agent from the surrounding environment. Thus failure or loss of integrity of this barrier must occur for agent to be released to the environment.

In general, the accident scenarios interest can be divided into two groups: (1) those scenarios in which the agent is inside the munition (e.g., scenarios involving transportation accidents), and (2) those in which the agent has been removed from the munition (plant operations accident scenarios). For both of these groups there are essentially three types of agent release to the environment:

- 1. Evaporation from a liquid spill.
- 2. Releases resulting from detonations.
- 3. Releases resulting from fires.

Various combinations of these releases appear in many of the scenarios. In addition, depending on the location of these events (e.g., indoor versus outdoor spills), the evaporation rates governing these releases may vary somewhat. The approach taken for assessing the amount, type, and duration of agent release is based on deterministic models which stem from previous demilitarization safety studies described in Section 1.1. These models are based largely on data but also engineering judgment. They are described in Section 10.1. Elements of the model include correlations for evaporation release, based on the D2PC computer program. In many cases, the D2PC computer program was used directly to calculate evaporative releases. Other elements include the fraction of burning agent which is released as vapor and the fraction of a detonating munition inventory which is released as vapor. The model relies heavily on data and analysis of munitions failure thresholds, summarized in Appendix F, to determine the extent of munition failures, including the potential for failure propagation of munitions. It is this area where engineering judgment was needed to supplement the data and analysis. Where judgmental factors entered in, they were routinely made in a conservative manner to cover possible uncertainties. BARREN TO LEAVE BOARD TO A REAL PARTY OF

TALENCE STRUCTURE STRUCTURE

RECEIPTION DESIGNATION

65333334

2.6. UNCERTAINTY ANALYSIS

Estimates of failure probabilities derived from various data sources are subject to uncertainties. Data sources do not always specify what failure modes are represented, what environment is applicable, or what is the total statistical population. In some cases, failure data may not be available for a specific event; therefore, data for events that occur under conditions that are similar to the events under consideration are selected as representative. These considerations result in uncertainties that are reflected in the range of possible numerical values for an event.

For events involving equipment failures, a lognormal distribution was assured to define the uncertainty in the failure probability. The lognormal distribution was explicitly used in Ref. 6-18 and other PRA studies of nuclear power plants because of its mathematical behavior. For the analysis covered in this report, equipment failures and accident initiators that are either man-made or arise from natural causes are assumed to be lognormally distributed.

In the analysis of accident scenario probabilities, the STADIC-2 computer program (Ref. 2-5) was used to combine probability distributions of a series of event sequences which make up an accident scenario. STADIC-2 uses a Monte Carlo simulation technique to generate a pseudorandom sample statistical distribution for a user-defined output function. Each input variable exhibits random, statistical variations that are represented by a particular probability distribution (lognormal, normal, etc.). The statistical distribution for the output function (and accident scenario probability in this case) is generated by combining the distributions in accordance with the mathematical operations



specified by that function. This combining of distributions is accomplished as follows:

- Each Monte Carlo sample consists of selecting one pseudorandom sample value for each input variable from its corresponding statistical distribution.
- 2. The set of sample variable values are mathematically combined to find the corresponding value of the function.
- Sampling is continued in this manner until the desired sample size is attained.
- The results consist of the pseudo-randomly generated values of the output function.

Probabilistic data base uncertainties are the only uncertainties explicitly quantified in this analysis. Although data base uncertainties are important, the accident frequency calculations are also sensitive to assumptions incorporated into the probabilistic assessment. Since the uncertainties in these assumptions are extremely difficult to quantify, conservative assumptions are consistently used in this risk analysis.

Figure 2-5 depicts the impact of this methodology (identified as Method 1 in the figure) on the accident frequency assessment results. Essentially, this methodology produces a conservative nominal frequency estimate, and underestimates the size of the confidence bands. However, the error associated with the confidence band estimate primarily results in predicting a much higher value for the lower confidence band than actually exists. (Compare the results of Methods 1 and 3 in Fig. 2-5.) Hence, the uncertainty assessment methodology employed in this analysis overestimates nominal accident frequencies and the confidence in the predicted frequency. 10^{-6} 10^{-7} 10^{-8} 10^{-8} 10^{-8} 10^{-9}

UNCERTAINTY ASSESSMENT METHODOLOGY

U.N.

| and the second s | | | | | |
|--|---|--|--|--|--|
| METHOD | DESCRIPTION CONSERVATIVE ASSUMPTIONS, ONLY DATA BASE UNCERTAINTIES QUANTIFIED | | | | |
| 1 | | | | | |
| 2 | CONSERVATIVE ASSUMPTIONS, ALL UNCERTAINTIES QUANTIFIED | | | | |
| 3 | REALISTIC ASSUMPTIONS, ALL UNCERTAINTIES QUANTIFIED | | | | |

Fig. 2-5. Impact of assumptions on the accident frequency uncertainty assessment



No quantitative uncertainty analysis is performed for the agent release calculations, due to the complexity involved in such an assessment. Instead, conservative releases are calculated. Because of the complex phenomenology that governs agent release, sensitivity studies were conducted to assure that the agent release estimates are, indeed, bounding. These sensitivity analyses are presented in Appendix B.

S MADDE

Presson and

والمراجعة والمحاصر والمحاصر والمحاص وا

2.7. REFERENCES

- 2-1. U.S. Nuclear Regulatory Commission, "PRA Procedures Guide," NUREG/CR-2300, 1982.
- 2-2. GA Technologies Report for Sandia National Laboratories, "High-Level Waste Preclosure Systems Safety Analysis Phase II Progress Report," GA-C18557, August 1986.
- 2-3. Fleming, K. N., "A Reliability Model for Common Mode Failures in Reduidant Safety Systems," Procedures of the Sixth Annual Pittsburgh Conference on Modeling and Simulation, April 1975.
- 2-4. Swain, A. D., and H. E. Guttmann, "Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Applications," NUREG/CR-1278, SAND 80-0200, August 1983.
- 2-5. Koch, P., and H. St. John, "STADIC-2, A Computer Program for Combining Probability Distribution," GA Report GA-A16777, July 1983.



3. DEMILITARIZATION DESCRIPTION OVERVIEW

Chemical munitions are currently stored at eight CONUS sites (Fig. 1-1). A description of the CONUS sites, including local maps, is given in Appendix D. Section 3.2 provides a summary description of the munitions.

The two alternatives for the disposal of the chemical munition stockpile which are discussed in this report are: (1) collocating munitions for disposal at two regional destruction centers (RDCs) located at the Tooele Army Depot (TEAD) in Utah and Anniston Army Depot (ANAD) in Alabama; and (2) collocating munitions for disposal at a single national destruction center (NDC) located at TEAD. A detailed discussion of the storage, handling, operations, transport, and decommissioning activities related to the alternatives is presented in Appendix G. Section 3.1 provides a summary of these activities as they relate to the risk study. Data for the munition transport containers are presented in Section 3.3.

3.1. COLLOCATION DISPOSAL ACTIVITIES AND RISKS

The major activities for the two collocation alternatives are outlined in Fig. 3-1. The activities begin with the munitions at each CONUS site in their existing storage locations in magazine igloos, warehouses or open areas. Long-term risks associated with continued storage, such as earthquakes and munition maintenance, are reduced by shipment to NDC or RDC disposal sites. This risk reduction must be weighed against risks associated with the transfer and disposal of the munitions. Elements of the added risks are: added storage risks created by establishing holding areas and interim storage, handling, plant operations, onsite transport, and offsite transport risks. These are discussed in the following paragraphs.



Fig. 3-1. Activities associated with munitions handling and transport

2000000

S.

and the second secon

NEADOWN DISABAD CONTRACTOR STATES

3.1.1. Storage

During storage, the only planned activities are monitoring for leakage, surveillance, maintenance and repair of munitions in the stockpile. Internal events for storage thus address leakage between inspections and munition drop or forklift time puncture during munition handling. The stored munitions are susceptible to external events, such as fire, tornado, aircraft or meteorite crash, earthquake, flood, and lightning. Storage time is a critical parameter for both the internal and external events. Until the sending site agent inventory is depleted, the holding areas established at both the sending and receiving sites create additional locations where munitions will be present with added interface area with potential external events.

In this study, the munitions are assumed to spend two weeks at each holding area. The munitions are moved from the holding areas to interim storage facilities. They stay considerably longer in interim storage; for purposes of analysis it is conservatively assumed that the interim storage facilities are full.

3.1.2. Handling

The munitions transfer from existing CONUS sites to one or two disposal sites creates a multitude of logistic handling operations. These operations are identified in Fig. 3-1 for handling at the sending and receiving sites and for offsite transport by rail or air. A detailed diagram of the loading/unloading of munitions into/out of the transport packages and loading/unloading the packages off the trucks and onto the trains, etc., is presented in Section 6. The handling operations for the specific option of marine transport from APG to Johnston Atoll differ in that a different offsite package is used, ship crane and dock crane loading/unloading is involved. Basically, the risks associated with these handling operations stem from internal handling accidents, caused by equipment failures or human error. Types of accidents are: vehicle collisions, forklift tine punctures, and drops of munitions. The munitions affected may be single, in pallets or overpacks (bombs and spray tanks), in an onsite container (ONC) or in an offsite container (OFC) or vault. Locations of the agent release may be indoors, or in the open (outdoors). External events causing handling accidents were not considered in this analysis because of the short time involved in actual outdoor handling operations. Also, the analyses for plant operations and storage considers the effect of external events on <u>all</u> munitions within buildings or igloos, regardless of whether or not handling is in progress.

The handling risk depends on the number of handling operations, such as packing, loading, and separating, moving or stacking with a forklift, which in turn depends upon the sites involved, the mode of offsite transport, and the type of munition moved. Section 6 describes how these variables were factored into the analysis.

Packing and unpacking handling operations occur first at the sending site storage area, where the munitions are packed inside an OFC. They remain inside that package until arrival at the interim storage area (igloo or warehouse) of the NDC or RDC. There, they are unpacked and stored in their original palletized configuration until ready for disposal. For disposal processing, they are packed in an ONC for shipment from interim storage until reaching the unpack area of the MDB. This procedure results in the munition always being in an ONC or OFC while outdoors onsite, and in the OFC when enroute offsite. Note: for marine transport, vaults will be used instead of OFCs; for the discussion in this section, the package will be referred to as an OFC.

The procedure assumed here of temporarily storing the munitions arriving at the NDC or RDC in a storage igloo or warehouse and subsequently moving them again (by truck) from storage to the munitions

holding igloo (MHI) has more handling operations than direct delivery to the MHI. Logistics may permit simple direct delivery; nonetheless, the complex logistics scheme is adopted for this risk analysis as a conservative approach. The MHI is a part of the demilitarization facility. The munitions are moved from the MHI to the package unloading area of the facility by forklift.

Loading and unloading handling operations occur at multiple times as follows:

- At the sending site storage area, the OFCs are loaded into trucks for onsite transport to the holding area (e.g., railhead, for rail transport).
- 2. At the sending site holding area, the OFCs are unloaded from the truck and held until reloaded on the railcar for offsite transport by rail. For the air transport option, the OFCs are loaded onto a truck bound for the airport. At the airport the OFCs are off-loaded onto a conveyor which loads them into the aircraft. For marine transport, the vaults are trucked to the loading dock where they are loaded by crane into the barge.
- 3. At the receiving site, the rail shipments arrive directly at the NDC or RDC holding area, where the packages are offloaded. Air shipments are transferred by conveyor from the aircraft to trucks for arrival and off-loading at the NDC or RDC holding area. Marine shipments are off-loaded by crane onto the dock.
- 4. At the holding area, the OFCs are loaded onto a truck for onsite transport to an interim storage area.

5. At the interim storage area, the palletized munitions are unloaded from the OFCs and placed in storage. The munitions

are then placed in ONCs and loaded onto trucks for onsite transport to the MHI.

- 6. At the MHI, the ONCs are unloaded from the trucks and placed in the MHI. For disposal, they are removed from the MHI by electric forklift and loaded onto diesel forklifts for transport to the MDB.
- 7. At the MDB, the forklifts deposit the ONCs in the Package Unloading Area for final processing.

In this risk study forklift transport operations are assumed to belong to the handling phase while truck transport is not.

3.1.3. Onsite Transport

Onsite transport encompasses all truck transfer operations outlined above at the sending and receiving sites. Associated risks consist of truck collision and/or overturn accidents with the munitions configured in ONC or OFC packages (or spray tanks and Weteye bombs in overpacks during transfer from the storage facilities to the demilitarization facility). These risks depend upon the expected distance of truck travel.

At all sending sites, the truck transfer distance from storage to the holding area is assumed to be one mile. For air transport, the departure and the arrival air strips were assumed to be located one mile each from the respective holding areas.

At the disposal site, one mile distances are assumed between the receiving holding area and interim storage and between interim storage and the MHI.

3.1.4. Offsite Transport - Rail

Special munitions trains will be used for rail transportation. Each munition train will be preceded by a pilot train. The munitions train is so configured that cars are divided into groups with buffer cars containing inert material between the groups. Special administrative procedures and controls are used to assure track and equipment reliability, as described in Appendix G. This study assesses the risks due to internally caused train accidents, due to human error or equipment (switching, etc.) failure, as well as externally caused events, such as aircraft crash, earthquake, and tornado, while the train is enroute to the receiving stations. The enroute risks consider the number of rail miles involved for specific site transfers.

Loading or unloading a munitions train is estimated to take approximately one day. During this time, the munitions on the train are susceptible to externally caused accidents.

3.1.5. Offsite Transport - Air

An option of using air transport to move munitions from either APG or LBAD sending sites to the Tooele depot receiving site was evaluated. Actual air flight distances were factored into the risk analysis. These are 1540 and 2066 miles, respectively, from LBAD and APG, pertaining to specific routes which avoid major population centers. The type of military aircraft (affecting the number of flights needed) assumed for this analysis was either C-141 or C-5. The availability of these aircraft during the demilitarization campaign is unknown at this time.

3.1.6. Offsite Transport - Ship

This study examines a specific option of moving mustard-filled ton containers from APG, Maryland, to JACADS. The analysis was based on using the LASH shipping system (lighter aboard ship). In this system, the ton containers are loaded into vaults and trucked to an onsite dock. At the dock, the vaults will be loaded on barges (called lighters) at a loading facility to be constructed on installation property on the Bush River. The barges are towed to the ocean-going LASH vessel anchored nearby in the deeper water of the Chesapeake Bay. The barges with the cargo onboard are lifted onto the LASH vessel, which is designed to carry preloaded barges in the hold. The LASH ship will then sail southward in the Chesapeake Bay, south along the east coast of the United States and Central and South America, around Cape Horn, then across the Pacific Ocean to Johnston Atoll. The ship is then off-loaded in the reverse order at the Johnston Atoll dock.

Risks associated with this ship transport involve internally (mostly human error) caused accidents, such as vessel collisions, bridge or shore rammings or groundings. Also considered are externally caused events, namely on-board fires, heavy weather damage, or aircraft crash.

3.1.7. Plant Operations

The demilitarization activity involves all processes present in a JACADS-type demilitarization facility including removal and deactivation of explosives, draining and incineration of agent, and treatment of all process effluents and ventilation air. For this study, the demilitarization facility is defined to be the MHI, where munitions await processing, and the MDB, where the incineration occurs.

In the MDB the munitions are first unpacked in the UPA. They are then processed by conveyor to the burster removal area, mine punch-anddrain area, projectile mortars disassembly area, rocket and burster shearing machines, mine machine for booster removal, a bulk drain station to punch and drain bulk items, a TOX agent storage tank, furnaces for explosive deactivation, metal parts decontamination, and agent and dunnage incinerators, as appropriate.

| AD-A193 355 UNCLASSIFIE | CHEMICAL THE DISP CA A N D SAPEO-CD | STOCKPILE OSAL OF CHE BARSELL ET E-IS-87006 | DISPOSAL A M. (U) GA AL. AUG 87 DAAA15-85- | ROGRAM RI Technolog GR-C-185 D-0022 | ISK ANALYSI BIES INC SF 563 F/G | S OF IN DIEGO 15/6.3 | 2/13 NL |
|----------------------------|--|--|---|--|--|----------------------------|------------|
| | | | | | | | |
| | | | | | | | |
| | | | | | | | |
| | | | | | | | |
| | | | | | | | |
| | | | | | | | |
| | | | | | | | |



Risks associated with the plant operations (disposal) phase include internally (human error or equipment) caused accidents resulting in munition drops, spills, and fires or explosions in furnace rooms. Externally caused risks involve tornado, meteorite, aircraft crash, or earthquake events. The potential for such events to fail packaged munitions in the MHI or UPA, bare or punched munitions in the MDB, or TOX piping systems was analyzed.

3.1.8. Decommissioning

After the existing stockpile of lethal chemical agent and munitions at each site has been destroyed, the demilitarization facility will be decommissioned. The activities for cleanup and closure of the destruction facilities, as discussed in Chemical Stockpile Disposal Plan (Ref. 3-1), are as follows:

- 1. Decontamination of the MDB and laboratory.
- 2. Disposal of all solid wastes and residues.
- 3. Certification of the plant and site as nontoxic.





3.2. MUNITIONS DESCRIPTION

This section describes the munitions that comprise the CONUS munitions stockpile. The munitions stored at each site are summarized in Table 3-1. As indicated the inventory of munitions and bulk agent in storage differs greatly from site to site. Detailed information on the precise numbers of chemical agent munitions at each site is classified except for the information on M55 rockets. All of the chemical munitions in storage are at least 18 yr old (production of new chemical munitions was stopped in 1968), and some are more than 40 yr old.

The munitions stockpile consists of 11 different munition types. A detailed description of each munition type, including a discussion of their thresholds, is presented in Appendix F. A brief description of the munitions follows.

3.2.1. Rockets

The M55 rockets are filled with either GB or VX. The rockets are equipped with fuzes and bursters which contain explosives. Propellant is also built into the motor of the rocket. The rocket casing is made of aluminum. Some of the rockets have a leakage problem.

The rockets are individually packaged in fiberglass shipping tubes with metal end caps. Fifteen containers with rockets are packed on a wooded pallet.

3.2.2. Land Mines

Mines contain VX and explosive charges. The mines are packaged three to a steel drum. Mine activators and fuzes are packaged separately in the same drum. Twelve drums of mines are contained on a wooden pallet.





TABLE 3-1 DATA FOR ONSITE TRANSPORT CONTAINERS (ONC), VAULTS, AND OFFSITE TRANSPORT CONTAINERS (OFC)

<u>Size</u>:

ONC: 6-ft diameter x 8-ft long cylinder
OFC: 20 ft x 8 ft x 8 ft
Vault: 8.8 ft x 3.7 ft x 4.5 ft

Failure Criteria:

Exposure to engulfing 1850°F fire detonates bursterd munitions ONC: 15 min OFC: 30 min Vault: not used for burstered munitions Exposure to engulfing 1850°F fire thermally fails munitions ONC: 15 min OFC: 2 h Vault: 2 h Impact failure: ONC: 40-ft drop (35 mph) OFC: 40-ft drop (35 mph) Vault: 40-ft drop (35 mph) Puncture: ONC: velocity/radius = 100/s OFC: velocity/radius = 200/s Vault: velocity/radius = 200/s Crush: ONC: 50,000-1b static load OFC: <520,000-1b static load Vault: <520,000-1b static load



3.2.3. Projectiles and Mortars

The munitions stockpile contains 105-mm projectiles with GB or mustard, 155-mm projectiles with GB, VX, or mustard, 8-in. projectiles with GB or VX, and 4.2-in. mortar projectiles with mustard. Some 105-mm projectiles are stored as complete rounds containing fuze, burster with explosive, cartridge case and propellant, while others are stored without bursters, fuzes and propellant. Mortars are stored with fuzes, bursters, and propellants. 155-mm and 8-in. projectiles are also stored with and without bursters. For this study, it was assumed that fuzes and propellants have been removed from the 4.2-in. mortars and 105-mm cartridges.

The 105-mm projectiles are packed 24 projectiles to a pallet; the 4.2-in. mortar projectiles are packed 48 projectiles to a pallet.

155-mm and 8-in. projectiles are packaged eight and six projectiles on a wooden pallet, respectively.

3.2.4. <u>Bombs</u>

There are three types of bombs, all containing GB agent. These are the MC-1, a 750-1b bomb, the MK-94, a 500-1b bomb, and the MK-116 ("weteye"), a 525-1b bomb. The 525-1b bomb is designed to release an aerosol spray of agent on detonation. The bombs are stored without explosives. The MC-1 bombs are packaged two to a wooden pallet and the others in individual metal shipping containers.

3.2.5. Spray Tanks

Spray tanks contain VX agent. They are designed for releasing chemical agent from slow-traveling, low-flying aircraft. The spray tanks are stored in a metal overpack container.





3.2.6. Bulk Agent

All three types of agent are stored in bulk as liquid in standard one-ton steel containers (called ton containers). Ton containers are not palletized.

Ton containers are the only items stored at the Aberdeen Proving Ground (APG) and Newport Army Ammunition Plant (NAAP). The ton containers at APG contain mustard (HD), while NAAP has VX-filled ton containers. The Anniston Army Depot (ANAD) has filled ton containers. Pine Bluff Arsenal (PBA) has mustard-filled ton containers. Tooele Army Depot (TEAD) has all types of bulk agent in storage. Umatilla Depot Activity (UMDA) has mustard-filled ton containers.



3.3. MUNITION PACKAGING AND TRANSPORT

For offsite transport by air or rail, the munitions will be packaged in offsite transport containers (OFCs) at the storage facility of the sending site. They will remain in OFCs until arrival at the disposal site storage. Transport from the disposal site storage to the demil facility is done with munitions in onsite transport containers (OFCs). Offsite transport of ton containers by marine shipment is done with the TCs in vaults. Table 3-3 presents the failure criteria for these munitions packages (Ref. 3-1).

Leakers may be caused by the corrosive nature of the chemical agent on the materials in the munitions agent compartment wall. When leakers are detected in storage, the munitions are packaged in a special leakproof package. No munitions known to be leaking are ever transported unless they are packaged in a special leak-proof package. Realistically, the major impact of corrosion is to degrade the original materials such that, while a leak has not occurred, the material parameters upon which the calculated failure thresholds are based generally do not reflect the actual condition of the munitions. The extent of degradation is unknown and cannot be considered in a meaningful way in the analyses presented in this report. Therefore, a general assumption is that the effect of corrosion or other material degradation is neglected, and a leak is assumed not to be initiated in transport.

If the accident forces are sufficiently severe to cause the OFC to fail, then those munitions with a lower failure threshold are also assumed to fail so that agent release occurs. If the failure threshold of the individual munition is higher than the OFC package, the munition failure threshold is used. In other words, the failure threshold of the package and contained munitions combined is equal to the maximum of either the package threshold or the munition threshold.

It is also assumed that when large fires occur, they engulf the entire transport vehicle. The assumption that the "representative" large fire always engulfs the transport vehicle is very conservative.

The structural calculations are based on the assumption that the munitions impact an unyielding surface, but because such surfaces are seldom encountered in real accidents, the structural failure thresholds are conservative.

The Sandia National Laboratory (SNL) transportation data base (see Sections 8 and 9) is assumed to be applicable to military transport. Where appropriate, modifications are clearly indicated to account for administrative controls. The major benefit of using the SNL transportation data base is that, in addition to providing accident rates for impact, fire, etc., the SNL researchers used sophisticated modeling to produce the accident environments that appear in the figures showing the percentage of accidents that do not exceed a certain force. These curves, or accident force spectra, are based on the best data available to SNL and a number of assumptions. The effect of administrative controls is to change either the data or the assumptions used to generate not only the accident rate but also the accident force spectra. Thus, a major assumption in this report is that when the accident rates are modified to account for factors unique to munitions aircraft, the accident force spectra are essentially unchanged. Use of the SNL curves is conservative, however.

No generally accepted method to quantify the probability of potential sabotage events in a risk analysis has been developed. Thus, any change in sabotage risk which occurs when extra packaging is used is not included in a quantitative way.

The shipping will be accomplished using the LASH (lighter aboard ship) shipping system. In this system, the ton containers will be loaded into vaults at their current storage location, and then trucked





to an onsite dock. At the dock, the packages will be loaded on barges (called lighters) at a loading facility to be constructed on installation property on the Bush River. The lighters will be towed to the ocean-going LASH vessel anchored nearby in the deeper water of the Chesapeake Bay. The lighters with the munitions on board are lifted onto the LASH vessel, which is designed to carry preloaded lighters in the cargo holds. The LASH ship will then sail southward in the Chesapeake Bay, south along the east coast of the United States and Central and South America, around Cape Horn, then across the Pacific Ocean to Johnston Island. The ship will then be off-loaded in the reverse order at Johnston Island.

The mustard-filled ton containers at Aberdeen Proving Ground will be transported to the dock in the vaults and loaded into the lighters. It is assumed that a towboat will then transport the lighters, 10 at a time, to a LASH vessel anchored in deeper water within the Chesapeake Bay. The lighters will be lifted by crane onto the LASH vessel and loaded into the ship's cargo hold.

5

Once the LASH vessel has been loaded, the ton containers will be transported south through the Chesapeake Bay, around South America, and across the Pacific Ocean to Johnston Atoll. This distance, approximately 14,000 nautical miles, is shorter than earlier proposed routes and was selected in order to eliminate the risk from refueling. Due to the significantly increased risk of an accident occurring during the transport if the ship is required to go into port to refuel or is refueled during transit, it is also assumed that the LASH vessel will have adequate fuel for the entire journey.

As the LASH vessel proceeds down the Chesapeake Bay, tugs will be used to assist the LASH under bridges and as otherwise needed to increase the maneuverability. This, in turn, will reduce the risk.

- SPEEDER TO SERVICE


An escort ship will accompany the munitions ship on the voyage. The escort ship will carry support personnel and equipment sufficient to respond to an emergency aboard the LASH ship that cannot be handled by onboard personnel.

It was also assumed that no lighters would be stored above deck for the transit. This was done to decrease the risk of transport.

64020

In Decement



3.4. REFERENCES

637777

3-1. Reed, A., et al., "Analysis of Existing Hazardous Material Containers for Transporting Chemical Munitions," the MITRE Corporation, June 1987.

- KK((())

4. INITIATING EVENTS

This section describes the approach used to identify and select initiating events and to assess or present their occurrence frequencies. As described in Section 2, initiating events are single occurrences or individual malfunctions that either directly cause the release of chemical agents or start a sequence of events that could lead to a release. They are classified as external events when caused by natural phenomena (e.g., earthquakes) or man-made interferences (e.g., aircraft crashes) from outside the demilitarization cycle. They are classified as internal events when caused by human error or equipment failure within the demilitarization process. Section 4.1 describes the logic used for selection of the initiating events. Section 4.2 discusses the generic considerations in specifying the initiating event frequency units (i.e., per unit time or per operation). The application of the generic frequency estimates to specific accident scenarios, locations and demilitarization phases are discussed in the sections dealing with accident logic model development, Sections 5 through 8.

4.1. INITIATING EVENT IDENTIFICATION AND SELECTION

This study used a multifaceted approach for identifying potential initiating events, screening out those which (based on conservative scoping) should not affect the overall risk and selecting those events warranting further analysis. The approach consisted of:

 Developing a master logic diagram (MLD), a logic tool described in the PRA Procedures Guide (Ref. 4-1) for systematically examining potential modes of release, jathways for release, barriers against release, and mitigating safety functions together with root causes (initiators) of release.

4-1

2. Dividing the demilitarization facility (MDB) into spatial zones and examining potential sources of release in each zone to identify internal initiating events for plant operations.

- 3. Cross-referencing results from items 1 and 2 with a list of accident scenarios from safety related studies on the chemical weapons disposal program, compiled by the MITRE Corporation in Ref. 4-2.
- Applying previous munitions risk study experience in Refs. 4-3 through 4-11. (The results of these studies are described in Section 1.1.)
- 5. Peer review by the Army and independent consultants during the early and draft report phases of this study.

Two criteria were used to screen accident scenarios: (1) accidents with extremely low frequency (below 10^{-10} per year), (2) those with low consequences (amount of agent release below 0.3 lb for GB, 14 lb for H or 0.4 lb for VX) were also screened. Events with frequencies below the cutoff have little meaning from a practical standpoint since the expected times between events is measured on a cosmic scale rather than on a scale of human history. The consequence criteria pertain to the minimum release levels that would produce acute human fatalities 0.5 km from the incident, based on environmental impact calculations performed by MITRE (Ref. 4-2).

For bookkeeping purposes, a coding system is used in this report to identify, organize, and refer to accident sequences. Not all accident sequences were encoded; those that could be screened out early because of simple conservative scoping analysis bear no coding. Conversely, many sequences that were screened after detailed analysis retain their coding but may not be in the final lists of results. However, Appendix A contains a record of all encoded sequences.

4-2

 $\mathcal{L}_{\mathcal{L}}$

Table 4-1 shows the coding scheme followed for identification of accident sequences. The coding system is based on that used in Ref. 4-2. The first two letters identify the demilitarization phase (S for storage, H for handling, R for rail transport, V for truck transport, B for barge transport, L for LASH transport, and P for plant operations) and the offsite transport mode option or division of activities for that phase, if any. For example, VR, VA, and VW refer to onsite transport for rail, air, or marine options. The first two letters together with the sequence number at the end uniquely identify an accident sequence of events. The middle letters identify the munition/agent type combinations and the release mode. Throughout this report, either the entire coding is used or sequences are referred to by the first two letters and the sequence number.

The MLD developed for the risk study event identification is shown in Figs. 4-1 through 4-9. Following the PRA Procedures Guide (Ref. 4-1), the top level logic (Fig. 4-1, level 1) pertains to the public impact, in this case, exposure to chemical releases throughout the various phases of the demilitarization process (storage, plant operations, handling, onsite transport and offsite transport).

Figure 4-2 shows MLD level 2 (release mode or pathway) and subsequent levels (barriers to release, safety functions mitigation/failure and, finally event initiators) for storage, including interim storage. It shows three modes for release. One is leakage of agent from corroded munitions, such as leakage of a ton container stored in open areas. Another is inadvertent rupture of a munition during maintenance. The third is a disruptive influence due to an external event. Since handling associated with incoming and outgoing munitions are considered in the handling phase, these three modes logically represent the possible ways a release can occur in the storage phase.

Subsequent levels are developed considering the types of disruptive events that can occur, taking into account information on the potential



1222

~~~~

# TABLE 4-1ACCIDENT SEQUENCE CODING SCHEME

## The Accident Scenario Identification is an 8-Character Code for the Form: XXYZWnnn as Defined Below.

| Activity (XX)                |      |          |                      | Munition/Agent Type                                                                 |
|------------------------------|------|----------|----------------------|-------------------------------------------------------------------------------------|
|                              | Rail | Air      | Ship                 | Combinations (YZ)                                                                   |
| Plant operations             | PO   | PO       | PO                   | BG: bomb containing GB<br>DH: mortar containing H                                   |
| Storage, long term           | SL   | SL       | SL                   | CG: cartridge containing GB<br>CH: cartridge containing H                           |
| Storage, interim             | SR   | SA       | SW                   | KG: ton container with GB<br>KH: ton container with H                               |
| Handling, at facility        | HF   | HF       | HF                   | KV: ton container with VX<br>MV: mine containing VX                                 |
| Handling, onsite             | HC   | HA       | HW                   | PG: projectile (155 mm)<br>containing GB<br>PH: projectile (155 mm)<br>containing H |
| Truck transport, interim     | vo   | VO       | vo                   | PV: projectile (155 mm)<br>containing VX<br>QG: projectile (8-in.)<br>containing GB |
| Truck transport, for offsite | VR   | VA       | VS                   | QV: projectile (8-in.)<br>containing VX<br>RG: rocket containing GB                 |
| Offsite transport(a)         | RC   | AA<br>AB | BI<br>LI<br>LC<br>LS | RV: rocket containing VX<br>SV: spray tank containing VX                            |

|    | Release Mode (W)                     | Sequence No. (nnn) |
|----|--------------------------------------|--------------------|
| S: | Spill or leak                        | 001, 002, 003, 999 |
| C: | Complex (e.g., detonation with fire) |                    |
| F: | Fire only                            |                    |

(a)For air transport, AA is for C-5 and AB is for C-141 aircraft. For ship transport, BI covers barge events; LI, LC, and LS are for LASH events in intercoastal, coastal and high-sea waters, respectively.







<u>᠋</u>





NOTE THERE ARE ADDITIONAL STORAGE SEQUENCES ASSOCIATED WITH HOLDING AREAS



10" . 10" . 10" . 10" . 10"

J. C. C.Y.

والمراقبة المراجب



NAMES OF STREET, STREE

J 22223

Fig. 4-3. Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part B - handling release





4-9







B. COM

Ę

#### Fig. 4-8. Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part G offsite air transport

and the second second



TE LITES IN LITES TO THE T

**LET.E**T.ET.ET.E

5.000

Fig. 4-9. Master logic diagram - levels 2 (release pathway) and lower (barriers, safety functions, and initiators). Part H - offsite sea transport

failure modes of the munitions (puncture, detonation, fire, etc.), given that the event occurs. For illustration, some sequences analyzed in Section 5 are noted under the initiating event boxes. Table 4-2 summarizes the initiating event families for storage selected for analysis.

Figure 4-3 shows the MLD levels 2 and lower for handling operations. There are three modes of release: release due to unpacking of undetected leakers, impact rupture due to handling accidents (drops and forklift collisions), and forklift time puncture. Note that external events are not included here; external events for storage and transport consider the entire munitions inventory available regardless of whether handling operations are in progress. The subsequent level initiating events consider the location where the event occurs (e.g., if the event occurs indoors or in an open area), since different barriers for release are involved. Table 4-3 summarizes the families or handling initiating events selected for analysis.

The MLD for onsite truck transport is developed in Fig. 4-4. A single generic mode of release applies to this phase, involving a vehicle collision or overturn coupled with potential munitions failure modes. In this phase, the munitions are always in offsite or onsite transport containers or overpacks, and failure thresholds may differ from those for bare munitions. Table 4-4 summarizes the initiating event families analyzed for onsite transport.

Figure 4-5 shows the MLD level 2 and subsequent levels for internal events during plant operations. This portion of the MLD was constructed by dividing the MDB into spatial zones and examining the sources for agent release in each zone. The zones are as follows:

1. The explosive containment vestibule (ECV) and munitions corridor.

4-14



# TABLE 4-2 INITIATING EVENT FAMILIES FOR STORAGE

# INTERNAL EVENTS

- 1. Munition drop
  - a. During leaker isolation
  - b. Due to pallet degradation
- 2. Forklift time puncture during leaker isolation
- 3. Leak between inspections

# EXTERNAL EVENTS(a)

- 1. Fires due to
  - a. Spontaneous ignition of a rocket
  - b. Flammable materials in an igloc or warehouse
  - c. LPB ingress into an igloo or warehouse
  - d. Flammable liquids near a warehouse at NAAP
- 2. Meteorite strikes an igloo, warehouse, or interim storage holding area
- 3. Tornado collapses a building or generates a missile
- 4. Aircraft crash due to
  - a. Small aircraft (direct)
  - b. Large aircraft (direct)c. Large aircraft (indirect)
- 5. Earthquake
- 6. Lightning strikes outdoor storage

(a)Note: Floods are shown in Section 5 to be unimportant initiators.



ር እርዲነው የውሃር እርዲ የዲዲ እንዲ የኢትዮጵያ ው የዲዲ የኢትዮጵያ እ



 $\sim \sim$ 

\*,\*\*\*\*\*\*

# TABLE 4-3 INITIATING EVENT FAMILIES FOR HANDLING(@)

a na hana na haran n

| 1. | Drop during operations at the processing facility of a |
|----|--------------------------------------------------------|
|    | a. Pallet or ONC outdoors                              |
|    | b. Pallet or ONC in the MDB                            |
|    | c. Single munition in the MDB                          |
| 2. | Drop during operations outside the facility of a       |
|    | a. Pallet or ONC in a storage igloo                    |
|    | b. Pallet or ONC outdoors                              |
|    | c. ONC in the MHI                                      |
|    | d. Pallet or OFC in the LPF                            |
|    | e. Single munition in the LPF                          |
|    | f. OFC outdoors                                        |
| 3. | Forklift tine puncture of a                            |
|    | a. Bare munition in a storage igloo                    |
|    | b. Bare munition in the LPF                            |
|    | c. Bare munition in the MDB                            |
|    | d. ONC or OFC outside the facility                     |
|    | e. UNC or UFC at the facility                          |
| 4. | Forklift collision at the processing facility for a    |
|    | a. ONC outdoors                                        |
|    | b. ONC in the MDB                                      |
| 5. | Forklift or CHE collision outside the facility for a   |
|    | a. Palletized munition outdoors                        |
|    | b. Palletized munition in a storage igloo              |
|    | c. Bare munition in the LPF                            |
|    | d. ONC outdoors                                        |
|    | e. OFC in the LPF                                      |
|    | I. UNL IN THE MHI                                      |
|    | g. ore outdoors                                        |
| 6. | Failure to detect a leak in an ONC or OFC              |

(a)For the marine transport option, vaults are used instead of OFCs.

and the second secon



#### INTERNAL EVENTS

- 1. Truck collision or overturn due to human error or equipment failure
  - a. With fire
  - b. Without fire

# EXTERNAL EVENTS

- 1. Aircraft crash into a truck
  - a. With fire
  - b. Without fire
- 2. Earthquake causes a truck collision or overturn
  - a. With fire
  - b. Without fire
- 3. Tornado causes a truck collision or overturn
  - a. With fire
  - b. Without fire

- 2. The munitions processing systems within the explosive containment room (ECR) and the munitions processing bay (MPB).
- 3. The buffer storage area (BSA), particularly punched and drained units present there.
- 4. The TOX tanks and associated piping systems.
- 5. The furnaces (MPF and DFS) and incinerators (LIC and DUN) and associated rooms.

For zones 1 and 2, the munitions present are unpunched. Thus, both a fall or other upset and a failure of the munition casing must occur for an agent spill. In zone 3, only the event is needed since the munitions are punched. Zone 4 refers to vessels and piping containing liquid agent; failure or rupture of safety grade metallic barriers are required for spills. Should spills occur in zones 1 through 4, they would drain to the appropriate sump. Evaporation from the floor and sump or a possible burning of the spill could result in a release to the environment if the MDB ventilation system or building structure fails. Zone 5 includes furnace and incinerator rooms where the release pathway is via accidental explosions.

Figure 4-6 shows the corresponding diagram logic diagram for release due to external events during plant operations. Here, the conditional failure of the MDB structure may be more likely or certain, given the catastrophic nature of the external events, such as meteorite strike or aircraft crash. Table 4-5 summarizes the initiating event families for plant operations.

Figures 4-7, 4-8, and 4-9 show the logic for agent release during offsite transport by rail, air or sea, respectively. For the first two transport options, release is contingent on the accident occurrence and munition failure. In the sea transport case, a liquid agent release to

L. L. L. L. L.

᠘ᠰᡓᡕᡘᡓᠺᡡᡓᡳᡸᡟᡘᡡ᠘ᢣᡘᡄᠧ᠋ᠴᠧᠧᠧᠧᠧ᠘ᡬᡭᡚᠺᠯᡳᠺᡳᡬᠧᡘᠽᢌᡬᢣᡬᢌᡬᠶᡬᢐᡘᡘᢌᡗᢌᡘᢌᡗᢌᡗᢌᡬᢌᡬᢌᡬᢌᡬᢌᡬᢌᡬᢌᡬᢌᡬᢌᡬ

ate at 6. 21 feat feat feat feat feat feat feat

# INTERNAL EVENTS

NCOCOCOCIN

\$\*#.\$\*4.\$`#.\$`\$!\$\*\$#\$\*\$#\$`#.\$`#.\$`#.\$`#.\$`#.\$`#.\$`#.

- 1. Accident in the ECV fails a munition
- 2. Accident in the ECR or MPB fails a munition
- 3. Accident in the BSA causes a punched munition spill
- 4. Failure of TOX tank or piping causes a spill
- 5. Accident associated with a furnace or incinerator which releases agent vapor

#### EXTERNAL EVENTS

- 1. A tornado generated missile fails
  - a. MHI munitions
  - b. UPA munitions
  - c. TOX/BDS piping (outdoor for CAMDS)
- 2. A meteorite fails
  - a. MHI munitions
  - b. UPA munitions
  - c. TOX/BDS piping
  - d. Agent collection tanks in TOX
- 3. A direct large aircraft crash fails
  - a. MHI munitions
  - b. UPA munitions
  - c. TOX/BDS piping (outdoor for CAMDS)
  - d. Agent collection tanks in TOX
- 4. An indirect large aircraft crash fails
  - a. MHI munitions
  - b. UPA munitions
  - c. Agent collection tanks in TOX
- A direct small aircraft crash fails TOX/BDS piping (outdoor for CAMDS)
- 6. An earthquake fails
  - a. MHI munitions
  - b. UPA munitions
  - c. Agent collection tanks in TOX
- 7. A truck accident fails
  - a. TOX/BDS piping (outdoor for CAMDS)

4-19



the surface of the water before sinking is needed in addition, in order for an evaporative release to occur. Table 4-6 summarizes the initiating event families for offsite transport. TABLE 4-6 INITIATING EVENT FAMILIES FOR OFFSITE RAIL, AIR, OR SEA TRANSPORT

## RAIL TRANSPORT

- 1. Train accident due to human error or equipment failure
- 2. Aircraft crash onto a train
- 3. Earthquake causes a train derailment
- 4. Tornado winds or missiles cause a train derailment

## AIR TRANSPORT

- 1. Aircraft crash into ground
  - a. Severe collision fails munitions by impact or fire
  - b. Moderate collision causes fire
- 2. On-board fire causes thermal failure of munitions

#### SEA TRANSPORT (barge or LASH vessels)

#### INTERNAL

- 1. Vessel collision fails munitions
- 2. Shore or bridge ramming fails munitions
- 3. Grounding fails munitions

#### EXTERNAL

- 1. Heavy weather fails munitions
- 2. On-board fire fails munitions
- 3. Aircraft crash into barge or LASH vessel fails munitions

#### 4.2. INITIATING EVENT FREQUENCIES

#### 4.2.1. External Events

This section presents the site-specific frequencies of external initiating events considered in this study. Table 4-7 summarizes the results for occurrences at each of the eight CONUS sites. Table 4-8 presents the nonsite specific occurrence frequencies. The bases for these results are discussed in the following subsections.

4.2.1.1. <u>Earthquakes</u>. The frequency at which a major earthquake occurs at a specific site varies significantly throughout the U.S. (Table 4-9). In an attempt to quantify the seismic risk associated with a particular site, the Seismology Committee of the Structural Engineers Association of California (SEAOC) has divided the U.S. into five seismic zones. Maps of these seismic zones are presented in the Uniform Building Code (Ref. 4-11) and in Army TM 5-809-10 (Ref. 4-12). Figure 4-10 presents the seismic zone map from TM 5-809-10, and Table 4-9 presents the seismic zones indicated for each of the storage sites. The probability of seismic damage in each of the zones is defined in Ref. 4-11 as follows:

Zone 0 - None Zone 1 - Minor Zone 2 - Moderate Zone 3 - Major Zone 4 - Great

The determination of a seismic zone of a site is based on the history of past earthquakes and the proximity of known faults. Appendix D presents listings of the earthquakes that have occurred in the vicinity of each of the storage sites. The magnitudes of the earthquakes are expressed as Modified Mercalli Intensities (MMI). Table 4-9 presents a summary of the maximum earthquake occurring in the vicinity of each of





TRANSAL SAL

# TABLE 4-7 SITE SPECIFIC FREQUENCIES OF EXTERNAL INITIATING EVENTS

|                                                                                                                                                                                        | APG                                                      | ANAD                                                     | LBAD                                                     | NAAP                                         | PBA                                                      | PUDA                                                     | TEAD                                                 | ADM                                                                                          |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------|----------------------------------------------------------|----------------------------------------------------------|----------------------------------------------|----------------------------------------------------------|----------------------------------------------------------|------------------------------------------------------|----------------------------------------------------------------------------------------------|
| Large aircraft crash<br>(events/yr-m1 <sup>2</sup> )                                                                                                                                   | 5.3x10-7                                                 | 7.9×10-6                                                 | 4.5x10-6                                                 | 4.6x10 <sup>-6</sup>                         | 1.5×10 <sup>-6</sup>                                     | 5.9x10 <sup>-5</sup>                                     | 3.6×10 <sup>-7</sup>                                 | 1.5×10 <sup>-5</sup>                                                                         |
| Small aircraft crash<br>(events/yr-mi <sup>2</sup> )                                                                                                                                   | 7.8x10 <sup>-3</sup>                                     | 1.2×10 <sup>-5</sup>                                     | 1.8×10 <sup>-7</sup>                                     | 2.3x10 <sup>-5</sup>                         | 1.1x10 <sup>-4</sup>                                     | 1.0×10 <sup>-4</sup>                                     | 1.5x10 <sup>-5</sup>                                 | 1.2x10 <sup>-5</sup>                                                                         |
| Meteorite (>1.0 lb) strikes<br>(events/yr-ft <sup>2</sup> )                                                                                                                            | 6.4x10-13                                                | 6.4×10-13                                                | 6.4x10-13                                                | 6.4x10-13                                    | 6.4x10-13                                                | 6.4x10-13                                                | 6.4x10-13                                            | 6.4x10-13                                                                                    |
| Lightning (events/yr-mi <sup>2</sup> )                                                                                                                                                 | 7.8                                                      | 23.2                                                     | 23.3                                                     | 12.9                                         | 28.5                                                     | 10.4                                                     | 7.8                                                  | 5.2                                                                                          |
| Earthquakes (events/yr)<br>- 0.15 g<br>- 0.2 g                                                                                                                                         | 1.5×10-4<br>7.0×10-5                                     | 1.5×10-4<br>7.0×10-5                                     | 1.5×10-4<br>7.0×10-5                                     | 7.5×10-4<br>3.6×10-4                         | 1.5×10-4<br>7.0×10-5                                     | 1.5x10-4<br>7.0x10-5<br>4.0x10-5                         | 4.0x10-3<br>2.0x10-3                                 | 1.5×10-4<br>7.0×10-5<br>4.0×10-5                                                             |
| - 0.25 g<br>- 0.3 g<br>- 0.5 g<br>- 0.5 g<br>- 0.5 g                                                                                                                                   | 4.0×10-5<br>2.5×10-5<br>1.2×10-5<br>6.0×10-6<br>3.5×10-6 | 4.0x10-5<br>2.5x10-5<br>1.2x10-5<br>6.0x10-6<br>3.5x10-6 | 4.0×10-5<br>2.5×10-5<br>1.2×10-5<br>6.0×10-6<br>3.5×10-6 | 2.0×10-5<br>5.0×10-5<br>2.0×10-5<br>1.0×10-5 | 2.5×10-5<br>2.5×10-5<br>1.2×10-5<br>6.0×10-6<br>3.5×10-6 | 2.5×10-5<br>2.5×10-5<br>1.2×10-5<br>6.0×10-6<br>3.5×10-6 | 7.0×10-4<br>2.6×10-4<br>1.0×10-4<br>4.5×10-5         | 2.5×10 <sup>-5</sup><br>1.2×10 <sup>-5</sup><br>6.0×10 <sup>-6</sup><br>3.5×10 <sup>-6</sup> |
| - 0.7 8                                                                                                                                                                                | 2.5×10-6                                                 | 2.5×10 <sup>-6</sup>                                     | 2.5×10 <sup>-6</sup>                                     | 7.0×10 <sup>-6</sup>                         | 2.5x10-6                                                 | 2.5×10 <sup>-6</sup>                                     | 2.0x10 <sup>-5</sup>                                 | 2.5x10 <sup>-6</sup>                                                                         |
| Tornadoes (events/yr)<br>- 100 mph windspeed<br>- 140 mph windspeed<br>- 150 mph windspeed<br>- 200 mph windspeed<br>- 250 mph windspeed<br>- 260 mph windspeed<br>- 260 mph windspeed | <br>1.0×10-5<br>1.0×10-6<br>1.0×10-6                     | <br><br><br>1.0×10-5<br>1.0×10-6<br>1.0×10-7             | <br><br>1.0×10-5<br>1.0×10-6<br>1.0×10-6                 | <br><br>1.0×10-5<br>1.0×10-6<br>1.0×10-6     | <br><br>1.0×10-5<br>1.0×10-6<br>1.0×10-6                 | <br>1.0×10-5<br><br>1.0×10-6<br>1.0×10-7<br>1.0×10-7     | 1.0×10-5<br>1.0×10-6<br><br>1.0×10-7<br>1.0×10-7<br> | 1.0x10-5<br>1.0x10-6<br>1.0x10-7<br>1.0x10-7                                                 |

4-23

|    | Event                                                                                                                                                    | Frequency                                                                      |
|----|----------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------|
| 1. | Fires                                                                                                                                                    |                                                                                |
|    | <ul> <li>a. Spontaneous rocket ignition</li> <li>b. Flammable material (inside)</li> <li>c. LNPG ingress</li> <li>d. Flammable liquids nearby</li> </ul> | (a)<br>(b)<br>(c)<br>(d)                                                       |
| 2. | Marine transport events                                                                                                                                  |                                                                                |
|    | a. Heavy weather damage to lighters                                                                                                                      | $3 \times 10^{-9} / trip$                                                      |
|    | b. Heavy weather damage to LASH                                                                                                                          | $3 \times 10^{-9} / trip$                                                      |
|    | c. On-board fire (LASH)                                                                                                                                  | $3 \times 10^{-9} / trip$                                                      |
| 3. | Aircraft events                                                                                                                                          |                                                                                |
|    | a. On-board fire, C-141<br>b. On-board fire, C-5                                                                                                         | 7.6 x $10^{-9}$ accidents/flight-mile<br>3.2 x $10^{-8}$ accidents/flight-mile |

# TABLE 4-8EXTERNAL EVENT FREQUENCIES FOR SPECIAL CASES

(a)Negligibly low probability based on AMSAA report.

(b)Insufficient flammable material in storage areas; analyzed by plant area for the demil facility.

(c)Negligibly low rate of ingress relative to that needed for flammability.

(d)Applies only to NAAP; quantity of flammable material determined to be insufficnet to threaten munitions.



| TABLE 4-9                             |       |
|---------------------------------------|-------|
| MAXIMUM MODIFIED MERCALLI INTENSITIES | (MMI) |
| IN THE VICINITY OF EACH SITE          |       |
|                                       |       |

 $\left\{ \mathbf{v}_{i}\right\}$ 

| Site                                   | Seismic<br>Zone | MMI  | No. of<br>Occurrences |
|----------------------------------------|-----------------|------|-----------------------|
| Aberdeen Proving Ground (APG)          | 1               | VII  | 1                     |
| Pine Bluff Arsenal (PBA)               | 1               | VI   | 3                     |
| Pueblo Depot Activity (PUDA)           | 1               | VI   | 1                     |
| Umatilla Depot Activity (UMDA)         | 1               | VII  | 1                     |
| Anniston Army Depot (ANAD)             | 2               | VII  | 1                     |
| Newport Army Ammunition Plant (NAAP)   | 2               | VII  | 1                     |
| Lexington-Blue Grass Army Depot (LBAD) | 2               | VII  | 1                     |
| Tooele Army Depot (TEAD)               | 3               | VIII | 2                     |





1 e.



20072226 N. 19722202



Seismic zone map for the contiguous United States Fig. 4-10.

AR's 1 - AR's 80 - 512 - 551 - 601



د ر ز

. A A S

the storage sites. The maximum earthquake recorded at any of the eight storage sites is an MMI VIII.

Currently, the Applied Technology Council, which is associated with the SEAOC, is developing a new seismic regulations for buildings (Ref. 4-13). When this work is completed, it is expected to be the basis for future federal, state, and local building codes. Part of this work was the development of a seismic risk map which divides the U.S. into seven seismic map areas similar to the five seismic zones used in Refs. 4-11 and 4-12. The seismic risk is approximately constant throughout a seismic map area.

Figure 4-11 (from Ref. 4-13) presents a set of curves that can be used to estimate the probabilities of earthquakes of various g-levels occurring within a particular seismic map area. The dashed portions of the curves indicate possible extrapolations to larger and smaller annual probabilities. Table 4-10 identifies the seismic map areas for each of the CONUS sites and tabulates the annual frequencies of earthquakes of various g-levels being exceeded at the storage sites. The data in Table 4-10 were obtained from Fig. 4-11. Straight line, logarithmic extrapolation was used to extrapolate to accelerations beyond the curves shown in Fig. 4-11. This method of extrapolation is believed to produce conservative estimates of the probabilities.

4.2.1.2. <u>Wind Hazards</u>. Methods for estimating the frequency and intensity of extreme winds can be found in ANSI/ANS-2.3-1983 (Ref. 4-14). The discussion which follows is largely based on the referenced national standard.

4.2.1.2.1. <u>Tornadoes</u>. A tornado is a violently rotating column of air whose circulation reaches the ground. The velocity of tornadic winds can exceed 300 miles per hour. The path of a tornado can be more than a mile in width, but generally ranges from 1/8 to 3/4 mile wide. The path width is defined as the tornado diameter corresponding to a 75 mph wind



Fig. 4-11. Annual frequency of exceeding various effective peak accelerations for selected map areas defined by the Applied Technology Council (Ref. 4-13)

4-28

20090610209061060610610610606606666



:

10.00

K

TABLE 4-10 ANNUAL RISK OF EARTHQUAKES

|                          |             |                        |                        |                        | Acceleration           | (g-level)              |                        |                        |                        |
|--------------------------|-------------|------------------------|------------------------|------------------------|------------------------|------------------------|------------------------|------------------------|------------------------|
| Site                     | Map<br>Area | 0.15                   | 0.20                   | 0.25                   | 0.3                    | 0.4                    | 0.5                    | 0.6                    | 0.7                    |
| TEAD                     | 2           | 4.0 × 10 <sup>-3</sup> | 2.0 × 10 <sup>-3</sup> | $1.0 \times 10^{-3}$   | 7.0 × 10 <sup>-4</sup> | 2.6 × 10 <sup>-4</sup> | $1.0 \times 10^{-4}$   | 4.5 x 10 <sup>-5</sup> | 2.0 × 10 <sup>-5</sup> |
| NAAP                     | e           | 7.5 × 10 <sup>-4</sup> | 3.6 x 10 <sup>-4</sup> | 2.3 × 10 <sup>-4</sup> | 1.3 × 10 <sup>-4</sup> | 5.0 × 10 <sup>-5</sup> | $2.0 \times 10^{-5}$   | 1.0 × 10 <sup>-5</sup> | 7.0 × 10 <sup>-6</sup> |
| APG. ANAD.               | 7           | 1.5 × 10 <sup>-4</sup> | 7.0 × 10 <sup>-5</sup> | 4.0 × 10 <sup>-5</sup> | $2.5 \times 10^{-5}$   | 1.2 x 10 <sup>-5</sup> | 6.0 × 10 <sup>-6</sup> | 3.5 × 10 <sup>-6</sup> | 2.5 × 10 <sup>-6</sup> |
| LBAD, PBA,<br>UMDA, PUDA |             |                        |                        |                        |                        |                        |                        |                        |                        |

5

4-29

velocity. The path of a tornado is seldom more than 10 miles long, although extreme cases are on record where the storm path extended more than 200 miles.

┙<u>╴╸┛╴╙┍┖╴┶┛╴┶┖╶┶┚╶┰┚╴┶┚</u>╌┰┚╌⋩┚╌⋩┚╌⋩┚╌⋩┚╌┟╨╌┟╨╌┟╨╌┟╨┟╔╘╼╻╉┶╔╉⋎╔┛┱╝┷╔┦╖╝┸┉╝┶┉╝╌┉╝╶┉╝╴┱┙┶╓╝╲╓┙╘╓╛╔╓╝╶╓╝

Meteorological and topographic conditions, which vary significantly from site to site, influence the frequency of occurrence and intensity of tornadoes. Reference 4-14 presents three regionalized maps of tornadic windspeeds corresponding to return frequencies of  $1.0 \times 10^{-7}$ ,  $1.0 \times 10^{-6}$ , and  $1.0 \times 10^{-5}$  per year. These maps (Figs. 4-12 through 4-14) are expected to bound the intensities and return probabilities at the various sites (Ref. 4-17). A tabulation of maximum tornado windspeed and return frequency for each of the storage sites based on these figures is presented in Table 4-11.

4.2.1.2.2. <u>Tornado-Generated Missiles</u>. One of the characteristics of a tornado is its capability to generate missiles from objects lying within the strike area and from nearby structural debris. The selection of tornado-generated missiles is dependent on the intensity of the tornado, the number of potential missiles present, their position relative to the tornado path, and the physical properties of the missiles. Reference 4-18 presents a spectrum of actual wind-generated missiles. Characteristics of these missiles are listed in Table 4-12, and expected windborne missile velocities are listed in Table 4-13.

4.2.1.2.3. Other Extreme Winds. The approach used for the determination of extreme windspeed (other than tornado) including hurricane winds is the method suggested by Science Applications International Corporation (SAIC). SAIC (Ref. 4-15) suggested the use of a basic wind speed as defined in Ref. 4-19. A frequency of occurrence of 2.0 x  $10^{-2}$  per year is associated with a basic wind speed of 70 mph. SAIC concluded that the basic wind speed was applicable to all of the sites that store M55 rockets. Lacking site-specific meteorological data, it is assumed that the basic wind speed is applicable to the other sites as well.





200013.5

Ç.

NUMBER NUMBER

19:55 Parts

22.22 C

Fig. 4-12. Tornadic winds corresponding to a probability of 1.0 x  $10^{-7}$  per year



10.00.00

NOR ON ON OUT

ŝ

Fig. 4-13. Tornadic winds corresponding to a probability of  $1.0 \times 10^{-6}$  per year



Fig. 4-14. Tornadic winds corresponding to a probability of 1.0 x  $10^{-5}$  per year

Marian



Fig. 4-15. Wind strength versus probability of recurrence, tornado Zone I (TEAD and UMDA sites)




|                      | Probabili            | ty of Occurrence<br>[Windspeed (mph) | Per Year<br>]        |
|----------------------|----------------------|--------------------------------------|----------------------|
| Size                 | $1.0 \times 10^{-5}$ | 1.0 x 10 <sup>-6</sup>               | $1.0 \times 10^{-7}$ |
| ANAD (Anniston, AL)  | 200                  | 260                                  | 320                  |
| LBAD (Lexington, KY) | 200                  | 260                                  | 320                  |
| UMDA (Umatilla, OR)  | 100                  | 140                                  | 180                  |
| PBA (Pine Bluff, AR) | 200                  | 260                                  | 320                  |
| TEAD (Tooele, UT)    | 100                  | 140                                  | 180                  |
| PUDA (Pueblo, CO)    | 150                  | 200                                  | 250                  |
| NAAP (Newport, IN)   | 200                  | 260                                  | 320                  |
| APG (Aberdeen, MD)   | 150                  | 200                                  | 250                  |

. 111 . • •

# TABLE 4-11 TORNADO WINDSPEEDS AND PROBABILITY OF RECURRENCE FOR CHEMICAL STORAGE SITES



| Missile                                            | Weight<br>(1b) | Projected<br>Area<br>(ft <sup>2</sup> ) | Cross Sectional<br>Area<br>(ft <sup>2</sup> ) |
|----------------------------------------------------|----------------|-----------------------------------------|-----------------------------------------------|
| Timber plank<br>4 in. x 12 in. x<br>12 ft          | 139            | 11.50                                   | 0.29                                          |
| Three-indiameter<br>standard steel<br>pipe x 10 ft | 75.8           | 2.29                                    | 0.0155(b)                                     |
| Utility pole<br>13.5-indiameter x<br>35 ft         | 1490           | 39.4                                    | 0.99                                          |
| Automobile                                         | 4000           | 100.0                                   | 20.0                                          |

# TABLE 4-12 WIND GENERATED MISSILE PARAMETERS(a)

(a) Source: Ref. 4-18.

(b)Value given is metal area. In penetration calculations the gross cross sectional area may be used.

|                                   | Hori | zontal | Missile | Veloci | ty <sup>(b)</sup> (m | ph) | Maximum        |
|-----------------------------------|------|--------|---------|--------|----------------------|-----|----------------|
| Design Wind Speed                 | 100  | 150    | 200     | 250    | 300                  | 350 | Height<br>(ft) |
| Timber plank                      | 60   | 72     | 90      | 100    | 125                  | 175 | 200            |
| Three-indiameter<br>standard pipe | 40   | 50     | 65      | 85     | 110                  | 140 | 100            |
| Utility pole                      | (c)  | (c)    | (c)     | 80     | 100                  | 130 | 30             |
| Automobile                        | (c)  | (c)    | (c)     | 25     | 45                   | 70  | 30             |

# TABLE 4-13 WINDBORNE MISSILE VELOCITIES(a)

(a)Source: Ref. 4-18.

(b)Vertical velocities are taken as two-thirds the horizontal missile velocity. Horizontal and vertical velocities should not be combined vectorially.

(c)Missile will not be picked up or sustained by the wind; however, for this analysis, any initial missile velocity of 80 mph or less was assigned a wind velocity of 250 mph.



4.2.1.3. <u>Aircraft Operations</u>. Much of the data in this section were taken from the SAIC report (Ref. 1-9).

There are three major concerns in assessing potential hazards due to aircraft operations:

- 1. Proximity of aircraft operations to munitions areas.
- 2. The frequency of aircraft flights.

3. The characteristics of the aircraft traffic.

The proximity of aircraft operations to munitions activities is an important consideration in that approximately 50% of aircraft accidents which result in fatalities or destroy aircraft occur within 5 miles of airports (Ref. 1-9). Also, the close proximity of flight paths to munitions activities increases the likelihood of these areas receiving falling debris from aircraft accidents. The frequency of flight activity increases the possibility of damage to munitions by increasing the overall likelihood of an aircraft accident.



23-2 • •

h

Fig. 4-17. Wind strength versus probability of recurrence, tornado Zone III (ANAD, LBAD, PGA, and NAAP sites)



Per the recommendations of NUREG-0800 (Ref. 4-16), the probability of an aircraft crash can be considered small if the distance to the site meets the following requirements:

- 1. The plant-to-airport distance (D) is between 5 and 10 statute miles, and the projected annual number of operations is less than 500 D<sup>2</sup>, or the plant-to-airport distance is greater than 10 statute miles, and the projected annual number of operations is less than 1000 D<sup>2</sup>.
- 2. The plant is at least 5 statute miles from the edge of military training routes, including low-level training routes, except those associated with a usage greater than 1000 flights per year, or where activities may create an unusual stress situation.
- 3. The plant is at least 2 statute miles beyond the nearest edge of a federal airway, holding pattern, or approach pattern.

The characteristics of an aircraft, such as its weight, number of engines, etc., are important in determining the energy of potential missiles generated in an aircraft accident, and depending on the structure they hit, the magnitude of the damage they may cause.

The frequency of an aircraft crashing while in an airway can be computed as follows (Ref. 4-16):

$$P_{FA} = C \times N \times A/W , \qquad (4-1)$$

where C = inflight crash rate per mile for aircraft using airway,

W = width of airway (plus twice the distance from the airway edge to the site when the site is outside the airway) in miles, A = effective area of facility in square miles,

N = number of flights per year along the airway.

For commercial aircraft, a value for C of 1.0 x  $10^{-10}$  has been used (Ref. 4-16). For military aircraft, C is estimated to be five times the value for commercial flights (Ref. 4-13). For general aviation, C was estimated to be the same as for military aircraft.

The frequency of an aircraft crashing in the vicinity of an airport or heliport can be computed as follows (Ref. 4-16):

$$P_{A} = \sum_{j=1}^{L} \sum_{j=1}^{M} C_{j} N_{ij} A_{j} , \qquad (4-2)$$

where L = number of flight trajectories affecting the target,

- M = number of different flights using the airport,
- C<sub>j</sub> = probability per square mile of a crash per aircraft movement for j<sup>th</sup> aircraft,

 $N_{11}$  = number per year of movements by the j<sup>th</sup> aircraft,

 $A_i$  = effective target area in square miles for the j<sup>th</sup> aircraft.

The values for  $C_j$  which were used in the analysis are listed in Table 4-14. The total crash probability is the sum of  $P_{FA}$  and  $P_A$ . The methodology for selecting these values is discussed in Appendix C.

The Federal Aviation Administration (FAA) does not monitor the number of certain types of aircraft which fly the high and low altitude airways. Consequently, the air traffic was estimated. Since air traffic is not the same on all airways, the airways are divided into five categories with regard to air traffic: very low, low, medium, high, and very high. Table 4-15 presents estimates of the air traffic on each of these airways. Each airway was assigned to one of these categories based on the traffic expected between the cities that the airway



\* 16\* 26\* 16\* 16\* 16\* 1

| Distance From | Probability (x 10 <sup>8</sup> ) of a Fatal Crash per Square<br>Mile per Aircraft Movement |                  |          |            |  |
|---------------|--------------------------------------------------------------------------------------------|------------------|----------|------------|--|
| End of Runway | Commercial                                                                                 | General Aviation | Military | Helicopter |  |
| 0-1           | 16.7                                                                                       | 84               | 7.0      | 168        |  |
| 1-2           | 4.0                                                                                        | 15               | 1.7      | 30         |  |
| 2-3           | 0.96                                                                                       | 6.2              | 0.72     | 12         |  |
| 3-4           | 0.68                                                                                       | 3.8              | 0.37     | 7.6        |  |
| 4-5           | 0.27                                                                                       | 1.2              | 0.30     | 2.4        |  |
| 5-6           | 0.14                                                                                       | 0.70             | 0.14     | 1.4        |  |
| 6-7           | 0.14                                                                                       | 0.70             | 0.14     | 1.4        |  |
| 7-8           | 0.14                                                                                       | 0.70             | 0.14     | 1.4        |  |
| 8-9           | 0.14                                                                                       | 0.70             | 0.14     | 1.4        |  |
| 9-10          | 0.12                                                                                       | 0.60             | 0.12     | 1.2        |  |

# TABLE 4-14AIRCRAFT CRASH PROBABILITIES NEAR AIRPORTS

|       | Aircraft         | Very<br>Low      | Low        | Medium | High   | Very<br>High |
|-------|------------------|------------------|------------|--------|--------|--------------|
|       |                  | <u>High Alti</u> | tude Jet R | loutes |        |              |
| Large | commercial       | 1,000            | 2,000      | 5,000  | 10,000 | 20,000       |
| Large | military         | 500              | 1,000      | 2,500  | 5,000  | 10,000       |
| Large | general aviation | 500              | 1,000      | 2,500  | 5,000  | 10,000       |
| Tot   | tal              | 2,000            | 4,000      | 10,000 | 20,000 | 40,000       |
|       |                  | Low Alt          | titude Air | ways   |        |              |
| Large | commercial       | 400              | 800        | 2,000  | 4,000  | 8,000        |
| Large | military         | 240              | 480        | 1,200  | 2,400  | 4,800        |
| Large | general aviation | 400              | 800        | 2,000  | 4,000  | 8,000        |
| Small | general aviation | 6,960            | 13,920     | 34,800 | 69,600 | 139,200      |
| Tot   | tal              | 8,000            | 16,000     | 40,000 | 80,000 | 160,000      |

# TABLE 4-15 ASSUMED DISTRIBUTION OF AIR TRAFFIC(a)

(a)Flights per year.

(b) The number of small commercial and small military flights is assumed to be small compared to other types of flights.

connects. If there are no low altitude airways near a site, it is assumed that the air traffic over the site is at least equal to that for a very low air traffic airway.

Appendix C presents tables which summarize the input data that were used to calculate the annual frequencies of both small and large aircraft crashes at each of the eight sites. The frequencies were computed using the equations given above. The annual frequencies for all the sites and for large and small aircraft and helicopters are summarized in Table 4-16. Note that for the air collocation option the annual frequencies for large aircraft crashes at APG, LBAD, and TEAD have to be adjusted by the additional flights expected into and out of these locations when munitions are moved by air from LBAD and APG to TEAD. It is expected that there will be an additiona. 1500 flights/yr at LBAD, 300 flights/yr at APG, and 1800 flights (1500 from LBAD and 300 from APG) at TEAD.

A major source of air crashes is the proximity of airports and heliports. This is of particular concern at APG, PBA, and PUDA. The air traffic for the APG analysis was supplied by POE-PM Cml Demil (Ref. 4-15). The helicopter air traffic at PBA was estimated by SAI (Ref. 4-15). The air traffic at PUDA was based on data collected at Pueblo Memorial Airport and communicated to GA by telephone. The helicopter traffic at TEAD is light and was assumed to be 15 flights per month.

The annual frequency of a crash into a specific facility is computed by multiplying the appropriate frequency taken from Table 4-16 by the effective target area of the facility (see Appendix C).

4.2.1.4. <u>Meteorites</u>. The frequency of meteorite strikes for meteorites 1.0 lb or greater is 4.3 x  $10^{-13}/ft^2$  (Ref. 4-20). For small meteorites (a ton or less), stone meteorites are approximately ten times more common than iron meteorites (Ref. 4-21). However, iron meteorites are more

n dia. Rana



# TABLE 4-16 SUMMARY OF AIRCRAFT CRASH PROBABILITIES (Crashes/Square-Mile/Year)

 $(a)_N/A = not applicable.$ 



TARANA SA CARACTER

dense and tend to have higher impact velocities, and consequently, represent a significant portion of the total meteorites that can rupture munitions. Table 4-17 shows the size distribution of striking meteorites for both iron and stone meteorites. The table was compiled from the data presented in Refs. 4-20 and 4-21.

### 4.2.2. <u>Electromagnetic Radiation</u>

でしていたい

14

Electromagnetic (E-M) radiation, either as a continuous source of energy or a short duration but higher energy pulse (EMP), has been considered as a potential hazard for control systems, sensitive explosive materials, and various munition components. The EMP field is a short pulse which might contain higher energies due to some uncontrollable phenomenon. Solid-state electrical circuits associated with systems which are national security sensitive are designed for protection from EMP produced electrical energies which could result from atmospheric nuclear blasts. These protection systems generally are designed as a Faraday's cage or have been designed to include "sacrificial" (i.e., expendable) electrical components. However, since nuclear warfare is out of this study's scope, the potential for these levels of energies to exist have been qualitatively screened out as not being credible as potential hazards to control systems. All munitions with the exception of M55 rockets are inherently enclosed in metal that acts as a Faraday's cage for protecting the munition's internals for normal and stray E-M fields. A Faraday's cage would provide a conducting shield for induced electrical energy which results from E-M fields passing through it. This E-M phenomenon is the basic physics principle, represented by the well-known Maxwell's equations, which enables an electrical generator to change mechanical energy to electrical energy by rotating a conducting system through a magnetic field. Therefore, with the exception of further examination of the possible effects of E-M on M55 rockets, normal or stray E-M fields have been eliminated as a potential initiating event in this hazard analysis.

PERSONAL DESCRIPTION

|                                            | ~                                           |                                            |                      |
|--------------------------------------------|---------------------------------------------|--------------------------------------------|----------------------|
| SIZE DI                                    | TE<br>STRIBUTION OF METEO                   | NGLE 4-17<br>RIVES WHICH ARE 1-11          | b OR LARGER(a)       |
|                                            |                                             |                                            |                      |
|                                            |                                             |                                            |                      |
| Greater                                    |                                             |                                            |                      |
| Than                                       | Stone                                       | Iron                                       | A11                  |
| (1b)                                       | Meteorites(b)                               | Meteorites <sup>(b)</sup>                  | Meteorites(b)        |
| 1                                          |                                             | 0.1                                        | 1.0                  |
| -                                          | 0.3                                         | 3 - 10-2                                   | 0.2                  |
| 2                                          | 0.5                                         | 5 x 10 -                                   | 0.5                  |
| 20                                         | 0.1                                         | 1 x 10 <sup>-2</sup>                       | 0.1                  |
| 200                                        | $3 \times 10^{-2}$                          | 3 x 10 <sup>-3</sup>                       | 3 x 10 <sup>-2</sup> |
| 2,000                                      | $2 \times 10^{-3}$                          | $2 \times 10^{-4}$                         | $2 \times 10^{-3}$   |
| 20,000                                     | $3 \times 10^{-4}$                          | 3 x 10 <sup>-5</sup>                       | $3 \times 10^{-4}$   |
| (a) <sub>Data</sub><br>(b) <sub>Frac</sub> | a compiled from Refs<br>tion of total numbe | a. 4-20 and 4-21.<br>ar of meteorites 1.0  | ) lb or greater.     |
| (a) <sub>Data</sub><br>(b) <sub>Frac</sub> | a compiled from Refs<br>tion of total numbe | a. 4-20 and 4-21.<br>For of meteorites 1.0 | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.<br>For of meteorites 1.0 | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.                          | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.                          | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.                          | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.                          | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.                          | ) lb or greater.     |
| (a)Data<br>`(b)Frac                        | a compiled from Refs                        | a. 4-20 and 4-21.                          | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.<br>For of meteorites 1.0 | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.<br>For of meteorites 1.0 | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.<br>For of meteorites 1.0 | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.<br>For of meteorites 1.0 | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.<br>For of meteorites 1.0 | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.<br>For of meteorites 1.0 | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | a. 4-20 and 4-21.<br>For of meteorites 1.0 | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | 4-48                                       | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | 4-48                                       | ) lb or greater.     |
| (a)Data<br>(b)Frac                         | a compiled from Refs                        | 4-48                                       | ) lb or greater.     |

M55 rockets, and in particular the rocket motors and ignition systems, have been evaluated for their susceptibility to E-M energies or short duration pulses (EMPs in an earlier study (Ref. 4-22). M55 rockets warranted special investigation because they contain their own motors and firing systems (igniters), and because of propellant instability which could be increasing as the rockets age. The SAI M55 study (Ref. 4-22) further investigated the rocket's internals and concluded that all the critical components were contained within metallic Faraday's cage type of shields. This study screened out the "rare" event of a simultaneous failure of the igniter's shunt, which prevents electrical energies from reaching the motor, and the existence of an incident delivering sufficient electrical energy to this M55 rocket. However, if any M55 rockets have a nonworking igniter shunt, then it is not really a case of two simultaneous occurring events. There are guidelines for naval vessels (Ref. 4-23) for maximum radar and communication energies for ensuring that E-M hazards are controlled. Figures 4-18 and 4-19 are from NAVSEA HERO document (Ref. 4-23) and represent the safe field strength and power densities for fully assembled ordnance. These curves are based on experimental results of HERO tests. The boundaries were established by the most susceptible ordnance items. We recommend that further effort be expended in determining whether or not the most sensitive ordnance onboard the naval vessels include items similar to the M55 rockets and in determining what the field strength and power density boundaries mean terms of radio or radar transmission energies which can be more easily understood and enforced.

In summary, E-M and EMP have been screened out as potential sources for plant operations' initiating events; however, further analysis and study are recommended to administratively control the safe demilitarization of munitions well within the safe E-M boundaries.



20

E = VERTICAL ELECTRIC FIELO STRENGTH--V/M (PEAK ENVELOPE)



The state in the second

1. 1. 1. 1. M. 1. 1. 1. 1. 1.

Table 4-18 summarizes the internal initiating events for all demilitarization phases of the collocation disposal option. Also summarized in the table are the event occurrence frequencies. The bases for these frequencies are discussed in the individual phase sections dealing with the event tree analysis, Sections 5 through 8, and are not repeated here.



|                                         |                                                             |                      | Frequency              |                        |  |  |  |
|-----------------------------------------|-------------------------------------------------------------|----------------------|------------------------|------------------------|--|--|--|
|                                         | Event                                                       | A C                  | lothing Leve<br>C      | F                      |  |  |  |
| STORAGE/HANDLING EVENTS (per operation) |                                                             |                      |                        |                        |  |  |  |
| 1.                                      | Munition drop from CHE (bulk containers)                    | 3 x 10 <sup>-5</sup> | 1.5 x 10 <sup>-6</sup> | 3 x 10 <sup>-6</sup>   |  |  |  |
| 2.                                      | Munition drop from forklift<br>(pallets or ST in overpacks) | 3 x 10-4             | 1.5 x 10 <sup>-5</sup> | 3 x 10 <sup>-5</sup>   |  |  |  |
| 3.                                      | Munition drop from hand (single units)                      | 6 x 10 <sup>-4</sup> | 3 x 10 <sup>-4</sup>   | 6 x 10 <sup>-5</sup>   |  |  |  |
| 4.                                      | Forklift tine accident                                      | $1 \times 10^{-4}$   | 5 x 10 <sup>-5</sup>   | $1 \times 10^{-5}$     |  |  |  |
| 5.                                      | Forklift or CHE collision                                   | $4.3 \times 10^{-6}$ | 4.3 x 10 <sup>-6</sup> | 4.3 x 10 <sup>-6</sup> |  |  |  |
| 6.                                      | Leak between inspections (stored pallets)                   | Munition de          | pendent                |                        |  |  |  |
| 7.                                      | Leak in ONC or OFC; failure to<br>detect                    | Munition de          | pendent                |                        |  |  |  |

# TABLE 4-18LIST OF INTERNAL INITIATING EVENTS AND FREQUENCIES

|      | Events                                                 | Frequency                  |
|------|--------------------------------------------------------|----------------------------|
| TRAN | ISPORT EVENTS                                          |                            |
| 1.   | Truck collision or overturn in convoy                  | 1.4 x $10^{-7}$ /road mile |
| 2.   | Truck fire in convoy                                   | 2.8 x $10^{-8}$ /road mile |
| 3.   | Train derailment (human error<br>or equipment failure) | 5.5 x $10^{-6}$ /road mile |
| 4.   | Aircraft crash at APG                                  | $4.2 \times 10^{-7}/yr$    |
| 5.   | Aircraft crash at LBAD                                 | $1.6 \times 10^{-9}/yr$    |
| 6.   | Aircraft crash at TEAD                                 | 9.1 x $10^{-10}/yr$        |

TABLE 4-18 (Continued)

|     | Events                                         | Frequency                                                                                                            |
|-----|------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|
| 7.  | Barge collision                                | 5.0 x $10^{-5}$ /shipment                                                                                            |
| 8.  | Barge ramming                                  | 4.1 x $10^{-5}$ /shipment                                                                                            |
| 9.  | Barge grounding                                | 8.6 x $10^{-5}$ /shipment                                                                                            |
| 10. | LASH collision, inland                         | $1.8 \times 10^{-4}/\text{shipment}$                                                                                 |
|     | LASH collision, coastal<br>LASH collision, sea | 8.1 x 10 <sup>-5</sup> /shipment<br>1.8 x 10 <sup>-5</sup> /shipment                                                 |
| 11. | LASH ramming, inland                           | $2.5 \times 10^{-5}$ /shipment                                                                                       |
|     | LASH ramming, coastal                          | $1.7 \times 10^{-5}$ /shipment                                                                                       |
|     | LASH ramming, sea                              | 1.3 x $10^{-5}$ /shipment                                                                                            |
| 12. | LASH grounding, inland                         | 2.3 x $10^{-4}$ /shipment                                                                                            |
|     | LASH grounding, coastal                        | 6.6 x 10 <sup>-</sup> /shipment                                                                                      |
| 1.  | Munition spill in ECV                          | K: $4 \times 10^{-5}/yr$<br>R: $3 \times 10^{-7}/yr$<br>M: $4 \times 10^{-7}/yr$                                     |
|     |                                                | M: $4 \times 10^{-7}/yr$<br>Q: $3 \times 10^{-7}/yr$<br>C: $1 \times 10^{-8}/yr$<br>P: $6 \times 10^{-7}/yr$         |
| 2.  | Munition(s) spill in ECR                       | 1M: $10^{-1}/yr$<br>2M: $10^{-2}/yr$<br>1Q: $10^{-1}/yr$<br>2Q: $10^{-2}/yr$<br>1R: $10^{-2}/yr$<br>2R: $10^{-3}/yr$ |
| 3.  | Munition detonates in ECR                      | M: 4 x 10 <sup>-4</sup> /yr<br>R: 1 x 10 <sup>-2</sup> /yr<br>others: 2 x 10 <sup>-3</sup> /yr                       |
| 4.  | Munition(s) spill in MPB                       | K: $4 \times 10^{-5}/yr$<br>Q: $3 \times 10^{-3}/yr$<br>2Q: $3 \times 10^{-4}/yr$                                    |
| 5.  | Ton container spill in BSA                     | $4 \times 10^{-5}/yr$                                                                                                |
| 6.  | Small TOX spill                                | $1 \times 10^{-3}/yr$                                                                                                |

4-54

Ś

|    |           | Events               | Frequency                                                                                                                                           |
|----|-----------|----------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------|
| 7. | Large TOX | spill                | $1 \times 10^{-3}/yr$                                                                                                                               |
| 8. | Unpunched | bulk item fed to MPF | KH: $1 \times 10^{-9}/yr$<br>KV: $6 \times 10^{-1}/yr$<br>KG: $9.2 \times 10^{-10}/yr$<br>B: $6.4 \times 10^{-9}/yr$<br>S: $7.2 \times 10^{-10}/yr$ |





S. California and

A D



0000000

n

N)

# 4.3. REFERENCES

00122254

- 4-1. "PRA Procedures Guide," Nuclear Regulatory Commission, NUREG/ CR-2300, 1982.
- 4-2. Fraize, W., et al., "Preliminary Risk Analysis Supporting the Chemical Weapons Stockpile Disposal Program (CSDP)," Volume 1: Analysis (MTR-86W4-01) and Volume 2: Appendices (MTR-86W4-02), the MITRE Corporation, May 1987.
- 4-3. Arthur D. Little, Inc., "Analysis of Risks Associated with Operation on Onsite M55 Rocket Demilitarization Plants," U.S. Army Toxic and Hazardous Materials Agency, M55-OD-3, May 1985.
- 4-4. Arthur D. Little, Inc., "Analysis of Risks Associated with Operation of the JACADS M55 Rocket Demilitarization Plant," U.S. Army Toxic and Hazardous Materials Agency, M55-OD-9, May 1985.
- 4-5. Arthur D. Little, Inc., "Analysis of Converted M55 Rocket Demilitarization Plant at Pine Bluff Arsenal," U.S. Army Toxic and Hazardous Materials Agency, M55-0D-10, May 1985.
- 4-6. Arthur D. Little, Inc., "Analysis of Risks Associated with Operation of the CAMDS M55 Rocket Demilitarization Plant," U.S. Army Toxic and Hazardous Materials Agency, M55-OD-11, May 1985.
- 4-7. Arthur D. Little, Inc., "M55 Rocket Separation Study," U.S. Army Toxic and Hazardous Materials Agency, M55-OD-08, November 1985.
- 4-8. The Ralph M. Parsons Company, "JACADS-Preliminary Hazards Analysis of Final Concept Design," Task G-56 Interim Report, May 1983.
- 4-9. The Ralph M. Parsons Company, "Fault Tree Analysis, Johnston Atoll Chemical Agent Disposal System," Task G-SC, U.S. Army Engineers Division, February 1985.
- 4-10. Rhyne, W. R., et al., "Probabilistic Analysis of Chemical Agent Release During Transport of M55 Rockets," H&R Technical Associates, Inc., Prepared for USATHAMA, M55-OD-4, H&R 255-1, September 1985.
- 4-11. Uniform Building Code, International Conference of Building Officials, 1985.

4-12. "Technical Manual, Seismic Design for Buildings," TM 5-809-10, Department of the Army, the Navy, and the Air Force, February 1982.

- 4-13. "Tentative Provisions for the Development of the Seismic Regulations for Buildings," Applied Technology Council, ATC 3-06, NBS SP-510, NSF 78-8, June 1978.
- 4-14. "Standard for Estimating Tornado and Extreme Wind Characteristics at Nuclear Power Sites," ANSI/ANS-2.3-1983, 1983.
- 4-15. Science Applications International Corporation, "Probabilities of Selected Hazards in Disposition of M55 Rockets," U.S. Army Toxic and Hazardous Materials Agency, M55-CS-2, November 1985.
- 4-16. "USNRC Standard Review Plan," Section 3.5.1.6, "Aircraft Hazards," U.S. Nuclear Regulatory Commission, NUREG-0800, July 1981.
- 4-17. Memo from OPMCM to GA Technologies Inc., November 20, 1986.
- 4-18. Coats, D. W., "Natural Phenomena Hazards Modeling Project: Extreme Wind/Tornado Hazards Model for Department of Energy Sites," UCRL-53526, Lawrence Livermore National Laboratory, February 1984.
- 4-19. "Minimum Design Loads for Buildings and Structures," American National Standards Institute, ANSI A58.1-1982.
- 4-20. Solomon, K. A., K. C. Erdmann, and D. Okrent, "Estimate of the Hazards to a Nuclear Reactor from the Random Impact of Meteorites," <u>Nuclear Technology</u>, <u>Vol. 25</u>, January 1975.
- 4-21. Blake, V. E., "A Prediction of the Hazards from the Random Impact of Meteorites on the Earth's Surface," Sandia Laboratories SC-RR-68-838, December 1968.
- 4-22. Science Applications International Corporation, "Probabilities of Selected Hazards in Disposition of M55 Rockets," U.S. Army Toxic and Hazardous Materials Agency, M55-CS-2, November 1985.
- 4-23. NAVSEA OD 30393, Design Principles and Practices for Controlling Hazards of Electromagnetic Rariation to Ordnance (Hero Design Guide), First Edition.

## 5. SCENARIO LOGIC MODELS FOR STORAGE

### 5.1. SEQUENCE LIST AND EVENT TREES

The accident scenarios involving the interim storage of chemical munitions were categorized as follows:

- External event-induced agent releases (e.g., earthquakes, aircraft crashes, etc.).
- 2. Releases due to leakage of munitions while in storage.
- 3. Releases from accidents that could occur during the isolation of leaking munitions while in storage.

For the collocation option, interim storage encompasses several phases: (1) storage of munitions at their original location (in igloos, warehouses, or open yards) before transfer to a destruction center; (2) storage of munitions in offsite transportation containers at the holding area of a sending site while awaiting loading onto a train car, aircraft, or barge; (3) storage of munitions in offsite transportation containers in the holding area of a receiving site upon arrival and while awaiting movement to a storage location; and (4) storage of the unpackaged munitions at this storage location before movement to a demil facility.

As discussed in Section 3, there are three transport modes discussed in this report. The rail transport mode applies to all sending sites; the air option applies only to movement of munitions from APG and LBAD to TEAD; and the marine option applies only to movement of

munitions from APG to Johnston Atoll. The storage analysis considers not only the existing storage locations but also interim storage at the disposal site and storage at the holding areas of the sending and disposal sites.

There are three types of transportation packages addressed here. For the rail and air options, the offsite transportation container consists of 90-in. diameter by 18-ft long inner container, and 8 x 8 x 20 ft steel outer container. For the marine option, the offsite container is a 106 x 44 x 54 in. steel vault designed to provide protection from impact, crush, puncture, and fire. To distinguish between the two containers, the offsite transport container for rail and air is referred to as an OFC, and the one for marine as the vault. Details on the OFC and vault are provided in Section 3.3 and Appendix F. For onsite transport from interim storage to the demil facility at the disposal site, an onsite transport container (ONC) is used. Table 3-3 gives dimensions and failure criteria for all three packages.

Table 5-1 presents the list of accident sequences identified and evaluated for the continued storage option. Accident sequences involving munitions in offsite transportation containers are designated SR, SA, SW for rail, air, and marine options, respectively. The event tree models are shown in Figs. 5-1 through 5-10. They will be discussed in the following sections by initiating event category. In these event trees, the following notations are used:

- NR = no release of agent
  - F = sequence screened based on low frequency criterion
  - C = sequence screened based on low release criterion.

TABLE 5-1 MASTER LIST OF STORAGE ACCIDENTS

| Event        | ID | Description                                                                                                                                                                                                 |
|--------------|----|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| SL1          |    | Munition develops a leak between inspections.                                                                                                                                                               |
| SL2          |    | Munition punctured by forklift tine during leaker-handling activities.                                                                                                                                      |
| SL3          |    | Spontaneous ignition of rocket during storage(a)                                                                                                                                                            |
| SL4          |    | Large aircraft direct crash onto storage area; fire not<br>contained in 30 min. (Note: Assume detonation occurs if<br>burstered munitions hit; fire involving burstered munitions<br>not contained at all.) |
| SL5          |    | Large aircraft indirect crash onto storage area; fire not contained in 30 min. (See note in SL4.)                                                                                                           |
| SL6          |    | Tornado-generated missiles strike the storage magazine,<br>warehouse, or open storage area; munitions breached (no<br>detonation).                                                                          |
| SL7          |    | Severe earthquake breaches the munitions in storage igloos; no detonations.                                                                                                                                 |
| SL8          |    | Meteorite strikes the storage area; fire occurs; munitions breached (if burstered, detonation also occurs).                                                                                                 |
| SL9          |    | Munition dropped during leaker isolation operation; munition punctured.                                                                                                                                     |
| SL10         |    | Storage igloo or warehouse fire from internal sources.(a)                                                                                                                                                   |
| <b>SL</b> 11 |    | Munitions are dropped due to pallet degradation.(a)                                                                                                                                                         |
| SL12         |    | Liquid propane gas (LPG) infiltrates igloo/building.(a)                                                                                                                                                     |
| SL13         |    | Flammable liquids stored in nearby facilities explode; fire propagates to munition warehouse (applies to NAAP).(a)                                                                                          |
| SL14         |    | Tornado-induced building collapse leads to breaching/<br>detonation of munitions.(a)                                                                                                                        |
| <b>SL</b> 15 |    | Small aircraft direct crash onto warehouse or open storage yard; fire occurs; not contained in 30 min.                                                                                                      |

5-3

(a)Screened out for the reasons stated in Table 5-2.

TABLE 5-1 (Continued)

| Event ID | Description                                                                                         |
|----------|-----------------------------------------------------------------------------------------------------|
| SL16     | Large aircraft direct crash; no fire; detonation (if burstered).                                    |
| SL17     | Large aircraft direct crash; fire contained within 30 min (applies to nonburstered munitions only). |
| SL18     | Small aircraft direct crash onto warehouse or open storage yard; no fire.                           |
| SL19     | Small aircraft direct crash onto warehouse or open storage yard; fire contained in 30 min.          |
| SL20     | Large aircraft indirect crash onto storage area; no fire.                                           |
| SL21     | Large aircraft indirect crash onto storage area; fire con-<br>tained in 30 min.                     |
| SL22     | Severe earthquake leads to munition detonation.                                                     |
| SL23     | Tornado-generated missiles strike the storage igloo and leads to munition detonation.               |
| SL24     | Lightning strikes ton containers stored outdoors.                                                   |
| SL25     | Munition dropped during leaker isolation; munition detonates.                                       |
| SL261    | Earthquake occurs; NAAP warehouse is intact; no ton containers damaged; fire occurs.                |
| SL262    | Earthquake occurs; NAAP warehouse is intact; ton container damaged; no fire.                        |
| \$1263   | Earthquake occurs; NAAP warehouse is intact; ton container damaged; fire occurs.                    |
| SL264    | Earthquake occurs; NAAP warehouse is damaged; ton containers damaged; no fire.                      |
| SL265    | Earthquake occurs; NAAP warehouse is damaged; ton containers damaged; fire occurs.                  |
| SL271    | Earthquake occurs; TEAD warehouses intact; munitions intact; fire occurs at one warehouse.          |
| SL272    | Earthquake occurs; TEAD warehouses intact; munitions intact; fire occurs at two warehouses.         |

<u>. 146 k</u>

TABLE 5-1 (Continued)

| Event ID | Description                                                                                                                             |
|----------|-----------------------------------------------------------------------------------------------------------------------------------------|
| SL273    | Earthquake occurs; one TEAD warehouse is damaged; munitions intact; fire occurs at one warehouse.                                       |
| SL274    | Earthquake occurs; one TEAD warehouse is damaged; munitions intact; fire occurs at two warehouses.                                      |
| SL275    | Earthquake occurs; two TEAD warehouses damaged; munitions intact; fire occurs at one warehouse.                                         |
| SL276    | Earthquake occurs; two TEAD warehouses damaged; munitions intact; fire occurs at two warehouses.                                        |
| SL281    | Earthquake occurs; UMDA warehouses intact; munitions intact; fire occurs at one warehouse.                                              |
| SL282    | Earthquake occurs; UMDA warehouses intact; munitions intact; fire occurs at two warehouses.                                             |
| SL283    | Earthquake occurs; UMDA warehouses intact; munitions in one warehouse damaged; no fire occurs.                                          |
| SL284    | Earthquake occurs; UMDA warehouses intact; munitions in one warehouse damaged; fire occurs at warehouse with damaged munitions.         |
| SL285    | Earthquake occurs; UMDA warehouses intact; munitions in one<br>warehouse damaged; fire occurs at warehouse with undamaged<br>munitions. |
| SL286    | Earthquake occurs; UMDA warehouses intact; munitions in one warehouse damaged; fire occurs at two warehouses.                           |
| SL287    | Earthquake occurs; UMDA warehouses intact; munitions in two<br>warehouses damaged; no fire occurs.                                      |
| SL288    | Earthquake occurs; UMDA warehouses intact; munitions in two<br>warehouses damaged; fire occurs at warehouse with damaged<br>munitions.  |
| SL289    | Earthquake occurs; UMDA warehouses intact; munitions in two<br>warehouses damaged; fire occurs at two warehouses.                       |
| SL2810   | Earthquake occurs; one UMDA warehouse damaged; munitions in one warehouse damaged; no fire occurs.                                      |
|          |                                                                                                                                         |

<u>مېرىم مەرىپەر بەرىپەر بەرىپەر</u>



| Event ID  | Description                                                                                                                               |
|-----------|-------------------------------------------------------------------------------------------------------------------------------------------|
| SL2811    | Earthquake occurs; one UMDA warehouse damaged; munitions in<br>one warehouse damaged; fire occurs at warehouse with damaged<br>munitions. |
| SL2812    | Earthquake occurs; one UMDA warehouse damaged; munitions in one warehouse damaged; fire occurs at two warehouses.                         |
| SL2813    | Earthquake occurs; one UMDA warehouse damaged; munitions in two warehouses damaged; no fire occurs.                                       |
| SL2814    | Earthquake occurs; one UMDA warehouse damaged; munitions in<br>two warehouses damaged; fire occurs warehouse with damaged<br>munitions.   |
| SL2815    | Earthquake occurs; one UMDA warehouse damaged; munitions in two warehouses damaged; fire occurs at two warehouses.                        |
| SL2816    | Earthquake occurs; two UMDA warehouses damaged; munitions in two warehouses damaged; no fire occurs.                                      |
| SL2817    | Earthquake occurs; two UMDA warehouses damaged; munitions in two warehouses damaged; fire occurs at both warehouses.                      |
| Rail Opti | on                                                                                                                                        |
| SR1       | Large aircraft direct crash onto transportation containers in holding area; no fire.                                                      |
| SR2       | Large aircraft direct crash onto transportation containers in holding area; fire occurs but not contained.                                |
| SR3       | Large aircraft direct crash onto transportation containers in holding area; fire contained but agent spill is burned.                     |
| SR4       | Small aircraft direct crash onto transportation containers in holding area; no fire.                                                      |
| SR5       | Small aircraft direct crash onto transportation containers in holding area; fire not contained.                                           |
| SR6       | Small aircraft direct crash onto transportation containers in holding area; fire contained.                                               |
| SR7       | Tornado-generated missiles strike munitions in transportation containers in holding area; no detonation.                                  |



# TABLE 5-1 (Continued)

ù

| Event  | ID   | Description                                                                                                       |
|--------|------|-------------------------------------------------------------------------------------------------------------------|
| SR8    |      | Tornado-generated missiles strike munitions in holding area;<br>detonation occurs.                                |
| SR9    |      | Meteorite strikes munitions in transportation containers in holding area; fire occurs; detonation (if burstered). |
| Air O  | otio | <u>n</u>                                                                                                          |
| SA1    |      | Large aircraft direct crash onto transportation containers in holding area; no fire.                              |
| SA2    |      | Large aircraft direct crash onto transportation containers in holding area; fire not contained.                   |
| SA3    |      | Large aircraft direct crash onto transportation containers in holding area; fire contained.                       |
| SA4    |      | Small aircraft direct crash onto transportation containers in holding area; no fire.                              |
| SA5    |      | Small aircraft direct crash onto transportation containers in holding area; fire not contained.                   |
| SA6    |      | Small aircraft direct crash onto transportation containers in holding area; fire contained.                       |
| SA7    |      | Tornado-generated missiles strike munitions in transportation containers in holding area; no detonation.          |
| SA8    |      | Tornado-generated missiles strike munitions in holding area;<br>detonation occurs.                                |
| SA9    |      | Meteorite strikes munitions in transportation containers in holding area; fire occurs; detonation (if burstered). |
| Marine | e Op | tion                                                                                                              |
| SW1    |      | Large aircraft direct crash onto transportation containers in holding area; no fire.                              |
| SW2    |      | Large aircraft direct crash onto transportation containers in holding area; fire not contained.                   |
| SW3    |      | Large aircraft direct crash onto transportation containers in holding area; fire contained.                       |

5-7



# TABLE 5-1 (Continued)

24124720422412462042241224824824824 19 19 19 19

| Event I | D Description                                                                                            |
|---------|----------------------------------------------------------------------------------------------------------|
| SW4     | Small aircraft direct crash onto transportation containers in holding area; no fire.                     |
| SW5     | Small aircraft direct crash onto transportation containers in holding area; fire not contained.          |
| SW6     | Small aircraft direct crash onto transportation containers in holding area; fire contained.              |
| SW7     | Tornado-generated missiles strike munitions in transportation containers in holding area; no detonation. |
| SW9     | Meteorite strikes munitions in transportation containers in holding area; fire occurs.                   |





| $\langle \cdot \rangle$ |
|-------------------------|





# TABLE 5-2 Accident sequences eliminated from detailed analysis

| Basis for Elimination  | Recent study indicates that the propellant is<br>stable and will continue to be so for some<br>time (Ref. 5-18). There is an enhanced mon-<br>itoring program to sample propellant in stor-<br>age. There are also accelerated tests being<br>performed to provide advanced warning of the<br>onset of propellant destabilization. | The pallets in storage are in very good con-<br>dition and are expected to remain so for many<br>more years. The munitions are periodically<br>inspected, and if deterioration is observed,<br>the causes are identified and corrected, and<br>the degraded pallet replaced. | There is no source of fire in the storage<br>igloos or storage buildings (Ref. 5-1). | LPG cloud due to release of largest conceiv-<br>able inventory (35,000 gal) cannot deposit<br>flammable concentration inside the igloo or<br>building (Ref. 5-1). | Several empty storage vessels are located<br>approximately 350 ft from the nearest ton<br>containers outside the exclusion area at<br>NAAP. These tanks were used in conjunction<br>with the former VX production facility. It |
|------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Description            | Spontaneous ignition of rocket during storage.                                                                                                                                                                                                                                                                                     | Munitions are dropped and damaged due to<br>pallet degradation.                                                                                                                                                                                                              | Storage igloo or building fire from internal sources.                                | Liquid propane gas (LPG) infiltrates igloo/<br>building.                                                                                                          | Flammable liquids stored in nearby facilities<br>explode; fire propagates to munition warehouse<br>(applies to NAAP only).                                                                                                     |
| <u>i</u> dent<br>uence | sr3                                                                                                                                                                                                                                                                                                                                | SLII                                                                                                                                                                                                                                                                         | SL10                                                                                 | SL12                                                                                                                                                              | SL13                                                                                                                                                                                                                           |
| Acc<br>Seq             | :                                                                                                                                                                                                                                                                                                                                  | 2.                                                                                                                                                                                                                                                                           | з.                                                                                   | 4.                                                                                                                                                                | 5.                                                                                                                                                                                                                             |
|                        |                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                              |                                                                                      |                                                                                                                                                                   |                                                                                                                                                                                                                                |

\*.64\*.04\*.b4\*.t

Ŋ,

TABLE 5-2 (Continued)

its to take

| Acc      | ldent<br>luence | Description                                                                           | Basis for Elimination                                                                                                                                                                      |
|----------|-----------------|---------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|          |                 |                                                                                       | is the Army's position to ensure that these<br>tanks will remain empty while munitions are<br>being stored at the NAAP warehouse.                                                          |
| <b>.</b> | (a)             | Tornado-induced munition drop leads to munition<br>ignition and detonation.           | Calculations indicate that tornado winds at 200 mph will not lift munitions (Ref. 5-2).                                                                                                    |
| ۲.       | (B)             | Tornado winds lift ton containers and drop them<br>to the ground; container ruptures. | Same as item 6.                                                                                                                                                                            |
| <b>.</b> | (8)             | A vehicle fire spreads to the storage area/<br>igloo and sets off munitions           | Previous analysis indicated that even if the<br>fire was just outside the igloo and the igloo<br>door was open, the munition thermal failure<br>threshold will not be exceeded (Ref. 5-1). |
| .6       | (a)             | Electrostatic ignition of rocket motor leads<br>to detonation and fire.               | Previous study indicated there was no source<br>of spark capable of igniting a rocket motor<br>accidentally (Ref. 5-1).                                                                    |
| 10.      | (a)             | Electromagnetic pulse (EMP) effects cause<br>detonation.                              | Previous study concluded there were no<br>sources of EMP of sufficient strength to<br>cause damage to the munitions (Ref. 5-1).                                                            |
|          | a)Sequ          | ence number not identified in GA's list.                                              |                                                                                                                                                                                            |

V

166683054

ANKA ANKA

20

5-10

.



 $\omega \omega$ 





-SR9 (NONBURSTERED) SA9 (NONBURSTERED) **SLB (NONBURSTERED)** AGENT RELEASE SEQUENCE - F (SCREENED OUT) •F (SCREENED OUT) -SR9 (BURSTERED) = SA9 (BURSTERED) -SL8 (BURSTERED) 6MS I ž ۳ ٤I 51 DETONATION Avoided No YES YES YES YES No. Si Si YES ŝ QNO 2NO I FIRE AVOIDED Ñ NO NO YES NO YES 0 N YES YES CONTAINMENT INTACT YES YES Ñ Q YES No 0N N YES LOCATION HOLDING AREA (BARGE) IGLOO/WAREHOUSE/ OPEN YARD HOLDING AREA (RAIL) HOLDING AREA (AIR) METEORITE STRIKE

6666666



5-12

alle te te the tetre to the tetre tetre to the tetre tetre







Large aircraft crash onto storage igloos containing burstered munitions F1g. 5-3.

5-13
AGENT Release Sequence - SL16 (NB) SL17 (NB) - SL20 (NB) SL21 (NB) SL 18 (NB) SL19 (NB) SL15 (NB) SL5 (NB) SL4 (NB) ЯN М EN ЯN RN Ë .... 1 1 FIRE CONTAINED\* i YES YES ۶ ۲ YES NON ON i 1 MUNITION YES YES YES YES NO No YES YES NO N NO 2 1 FIRE Avoided YES YES 2 YES ŝ R STRUCTURE INTACT YES YES ŝ NO YES 2 IGLOD/ WAREHOUSE/ OPEN YARD IGLOO/ WAREHOUSE/ WAREHOUSE/ OPEN YARD OPEN YARD LOCATION \*SUCCESS DEFINED AS ABILITY TO CONTAIN FIRE IN 30 MIN OR LESS DIRECT OR INDIRECT DIRECT INDIRECT DIRECT AIRCRAFT Crash Occurs LARGE LARGE SMALL



5-14

•

Ch Ci



Aircraft crash accidents involving munitions in holding area (marine option) F1g. 5-5.

<u> 1998 - Entrol Franker, success sourcester bretter bretter bretter bretter bretter bretter bretter</u>

5-15

Contraction of the second second

Aircraft crash accidents involving munitions in holding area (air option) Fig. 5-6.







Aircraft crash accidents involving munitions in holding area (rail option) F1g. 5-7.

للدنديديكا



803083-592

223B

21

: 15 SX 97 A



and the second second

Ż



Fig. 5-9. Aircraft crash accidents involving the LASH at rest (marine option)

5-19



<u>~~~</u>

Fig. 5-10. Earthquake-induced agent releases involving munitions in

The external events that were evaluated include:

- Tornadoes and high winds.
- Meteorite strikes.
- Aircraft crashes.
- Earthquakes.
- Lightnings.
- Floods.

In general, the amount of agent released to the atmosphere from accidents induced by such events depends on the extent of damage incurred to the building structure and the munition itself. The munitions are currently stored in igloos, warehouses, or open storage yards. Appendix D discusses the types of storage structures present at each CONUS site, as well as the kinds of munitions stored. Munitions in OFCs and barge packages temporarily stored in open holding areas are also vulnerable to these natural and man-caused events.

### 5.2.1. Tornadoes and High Winds

The accident scenarios identified involve the breaching of the munitions in the storage facilities (i.e., igloos, warehouses, or open yards) by tornado- or high-wind-generated missiles. This failure mode was determined to be more credible than that identified in sequence SL14, which is a tornado/high-wind-induced building collapse that could lead to the crushing of munitions by the falling structure. For UBC designed structures such as a warehouse, the wind loads will fail the walls of the structure before the structure will collapse. Storage igloos have been designed to resist the direct effects of tornadoes with winds up to 320 mph except for the possibility of missiles breaching the igloo doors (Ref. 5-1). For the above reasons, sequence SL14 has been screened out from further analysis.

The event tree developed to define relevant accident sequences is shown in Fig. 5-1. None of the accident sequences could be screened out initially as more detailed quantitative analysis is required to determine the necessary wind velocity to generate missiles which could penetrate the munitions. Hence, all the accident sequences shown in the event trees were quantified. They are SL6, SL23, SR7, SR8, SA7, SA8, and SW7.

Essentially, the missile penetration of the munition occurs if (1) a tornado or extremely high wind occurs with a velocity sufficient to generate a missile that could penetrate the igloo door, warehouse wall, or transportation container wall, and the munition itself; and (2) the missile actually hits the target munition.

The probability of a missile hitting and rupturing a munition is the product of four variables: (1) the probability that the velocity vector of the missile is nearly perpendicular to the target; (2) the probability that the missile is oriented properly to penetrate the target; (3) the number of missiles per square foot of wind; and (4) the target area. More details on the derivation of these variables are provided in Appendix C and Ref. 5-2.

If the missile hits a burstered munition, two failure modes are possible: (1) the munition is opened up due to puncture or crush, or (2) the missile impact causes munition detonation due to the application of a force greater than the "undue force." The undue force is defined as "a force greater than that generally required to assemble the munition" or as "any force which could cause deformation to the munition (other than minor surface deformation) or damage to the explosive train" (Ref. 5-3).



5.2.1.1. <u>Storage Magazines</u>. The analysis of the vulnerability of the igloo door to the tornado-generated missile considered the two types of igloo doors present at the CONUS sites, i.e., steel and concrete. PBA and TEAD have igloos with either steel or concrete doors, while the igloos at ANAD, LBAD, PUDA, and UMDA have steel doors only. For conservatism, all igloos at PBA and TEAD were assumed to have concrete igloo doors.

The steel doors require a missile velocity of 94 mph for penetration by a 3-in. steel pipe or 66 mph for penetration by a utility pole. For the concrete doors, the penetration velocity for a 3-in. steel pipe is 66 mph and for the utility pole, 54 mph. After penetrating the door, the remaining missile velocity must be large enough to rupture the munition. The formula for the required initial missile velocity is as follows:

$$\sqrt{v_{\rm I}} = v_{\rm d}^2 + v_{\rm m}^2$$
, (5-1)

where  $V_{I}$  = required initial velocity,

 $V_d$  = required velocity to penetrate the door,  $V_m$  = required velocity to rupture the munition.

In order for a missile to reach the velocity required to penetrate the igloo door and the munitions inside, a wind with a significantly higher velocity is required. Table 5-3 presents the relationship between wind velocity and missile velocity.

The frequency of a wind-generated missile penetrating an igloo and a munition inside the igloo, is the product of the following:

 The frequency of a tornado or wind which has sufficient velocity to generate a missile that can penetrate the igloo and munition.

|                                      | Hor | izonta | l Miss<br>(mj | ile Ve<br>ph) | locity | (b) | Maximum Height |
|--------------------------------------|-----|--------|---------------|---------------|--------|-----|----------------|
| Design Wind Speed                    | 100 | 150    | 200           | 250           | 300    | 350 | (ft)           |
| Timber plank                         | 60  | 72     | 90            | 100           | 125    | 175 | 200            |
| Three-inch-diameter<br>standard pipe | 40  | 50     | 65            | 85            | 110    | 140 | 100            |
| Utility pole                         | (c) | (c)    | (c)           | 80            | 100    | 130 | 30             |
| Automobile                           | (c) | (c)    | (c)           | 25            | 45     | 70  | 30             |

### TABLE 5-3 WINDBORNE MISSILE VELOCITIES(a)

(a)Source: Ref. 5-4.

(b)Vertical velocities are taken as 2/3 the horizontal missile velocity. Horizontal and vertical velocities should not be combined vectorially.

(c)Missile will not be picked up or sustained by the wind, however, for this analysis any initial missile velocity of 80 mph or less was assigned a wind velocity of 250 mph.

Aresteed Villing

10000C

2. The probability of a missile penetrating the igloo and hitting the munition in such a way as to cause damage and is calculated as follows:

$$P_{p} = P_{d} \times P_{o} \times D_{e} \times A_{t} , \qquad (5-2)$$

where P<sub>d</sub> = probability that the velocity of the missile is nearly perpendicular to the target plane,

- P<sub>o</sub> = probability that the missile is oriented to penetrate the target (i.e., missile not tumbling or going sideways),
- De = density of number of missiles per square foot of wind,

 $A_t = target area.$ 

Details on the calculation of these variables are given in Ref. 5-2.

The site-specific tornado frequency versus velocity curves has been presented in Section 4. Two types of missiles were initially considered: (1) a 3-in. pipe and (2) a utility pole. For all munition types, it was found that the utility pole had a higher probability of penetrating munitions.

Tables 5-4 and 5-5 present the wind velocities required to generate missiles which have sufficient velocity to penetrate the igloo door and the various munitions stored inside. Table 5-6 presents the annual frequencies of these winds occurring at each of the sites that have igloos. The frequencies were read from the curves presented in Figs. 4-9 through 4-11. The conditional probability of a missile hitting the igloo door and the munitions stored inside is  $3.2 \times 10^{-6}$  (see Appendix C). TABLE 5-4 MISSILE PENETRATION THROUGH STEEL IGLOO DOORS AND MUNITIONS

| Munition         | Missile          | Munition<br>Rupture Velocity<br>(mph) | Door Penetration<br>Velocity<br>(mph) | Required Initial<br>Missile Velocity<br>(mph) | Required<br>Wind Velocity<br>(mph) |
|------------------|------------------|---------------------------------------|---------------------------------------|-----------------------------------------------|------------------------------------|
| Ton container    | 3-in. pipe       | 108                                   | 94                                    | 143                                           | >350                               |
|                  | Utility pole     | 67                                    | 66                                    | 94                                            | 285(a)                             |
| 4.2-in. mortar   | 3-in. pipe       | 60                                    | 94                                    | 112                                           | 303                                |
|                  | Utility pole     | 8                                     | 66                                    | 67                                            | 250(a)                             |
| 750-1b bomb      | 3-in. pipe       | 101                                   | 94                                    | 138                                           | 347                                |
|                  | Utility pole     | 63                                    | 66                                    | 91                                            | 278(a)                             |
| 8-in. projectile | 3-in. pipe       | 162                                   | 94                                    | 187                                           | >350                               |
|                  | Utility pole     | 25                                    | 66                                    | 71                                            | 250(8)                             |
| M23 land mine    | 3-in. pipe       | 43                                    | 94                                    | 103                                           | 286                                |
|                  | Utility pole     | 6                                     | 66                                    | 66                                            | 250(a)                             |
| M55 rocket       | 3-in. pipe       | 22                                    | 94                                    | 97                                            | 274                                |
|                  | Utility pole     | 8                                     | 66                                    | 67                                            | 250(a)                             |
| (a)Critical mi   | ssile for munit: | ion.                                  |                                       |                                               |                                    |

5-26

ᢓᠼᡬᡷᠧᠼᠧᠼᢉᠼᡗᡄᢧᡗᠵ᠓ᡔᡘᡄᡸᠴ᠓ᠼᡸᠴᠺᠱ᠋ᡘᢣᡭᢣᡭᢣᡯᢣᡯᢣᡯᢐᠯᡐᡘᡈᡱᡐᠣ᠕ᡆᡢᡅᡟᠣᡭᠣ







### TABLE 5-5 MISSILE PENETRATION THROUGH CONCRETE IGLOO DOORS AND MUNITONS

| Munition         | Missile          | Munition<br>Rupture Velocity<br>(mph) | Door Penetration<br>Velocity<br>(mph) | Required Initial<br>Missile Velocity<br>(mph) | Reguired<br>Wind Velocity<br>(mph) |
|------------------|------------------|---------------------------------------|---------------------------------------|-----------------------------------------------|------------------------------------|
| Ton container    | 3-in. pipe       | 108                                   | 66                                    | 127                                           | 329                                |
|                  | Utility pole     | 67                                    | 54                                    | 86                                            | 285(a)                             |
| 4.2-in. mortar   | 3-in. pipe       | 60                                    | 66                                    | 89                                            | 258                                |
|                  | Utility pole     | 8                                     | 54                                    | 55                                            | 250(a)                             |
| 750-1b bomb      | 3-in. pipe       | 101                                   | 66                                    | 121                                           | 318                                |
|                  | Utility pole     | 63                                    | 54                                    | 83                                            | 258(a)                             |
| 8-in. projectile | 3-in. pipe       | 162                                   | 66                                    | 175                                           | >350                               |
|                  | Utility pole     | 25                                    | 54                                    | 60                                            | 250(a)                             |
| M23 land mine    | 3-in. pipe       | 6 4                                   | 66                                    | 79                                            | 235(a)                             |
|                  | Utility pole     | 9                                     | 54                                    | 54                                            | 250                                |
| M55 rocket       | 3-in. pipe       | 22                                    | 66                                    | 70                                            | 213(a)                             |
|                  | Utility pole     | 8                                     | 54                                    | 55                                            | 250                                |
| (a)Critical miv  | seila for muniti | g                                     |                                       |                                               |                                    |

|                        | ANAD   | LBAD   | PBA(b) | PUDA   | TEAD(b) | UMDA    |
|------------------------|--------|--------|--------|--------|---------|---------|
| Cartridges and mortars | 1.5E-6 |        |        | 1.0E-7 | 1.8E-9  |         |
| Projectiles            | 1.5E-6 | 1.5E-6 |        | 1.0E-7 | 1.8E-9  | 1.8E-9  |
| Mines                  | 1.5E-6 |        | 2.6E-6 |        | 4.2E-9  | 1.8E-9  |
| Rockets                | 1.5E-6 | 1.5E-6 | 6.1E-6 |        | 1.5E-8  | 1.8E-8  |
| Ton containers         | 3.8E-7 |        |        |        | 7.5E-10 | 2.4E-10 |
| Bombs                  |        |        | ~-     |        | 1.1E-9  | 3.6E-10 |
| Spray tanks            |        |        |        |        |         | 1.1E-9  |

TABLE 5-6FREQUENCY OF A WIND HAZARD SUFFICIENT TO BREACH<br/>MUNITIONS IN STORAGE MAGAZINES(a) (PER YEAR)

(a)Frequencies obtained from the curves presented in Figs. 4-9 through 4-11.

ver

(b)Concrete doors.

5.2.1.2. <u>Warehouses</u>. The warehouses at TEAD are designed for 100-mph wind loads (Ref. 5-1). Assuming that the warehouses at NAAP and UMDA are designed to the UBC requirements, they should be designed for at least 70 mph winds. An analysis of the UBC requirements shows that winds will fail the walls of UBC designed structures before the frame of the structure will fail. Based on the margins of safety required by the UBC, the concrete walls of the warehouses at TEAD are not expected to be breached by winds less than 160 mph. Breaching of the concrete walls is expected to involve cracking and spalling of the concrete and the possibility of the wall partially separating from the frame. The sheet metal walls of the warehouses at NAAP and UMDA are expected to be blown away by 115-mph winds. Neither of these failures are expected to damage the bulk containers.

In order for a wind blown missile to penetrate a spray tank in a warehouse at TEAD, it must pass through the 6-in. concrete wall, the spray tank overpack, and finally the spray tank itself. This would require a 283-mph wind.

A 250-mph wind can generate a missile that will penetrate an unprotected ton container. Since a 115-mph wind is expected to blow away the walls of the warehouses at NAAP and UMDA, the walls will offer no protection. Therefore, a 250-mph wind has the potential to generate missiles that will penetrate the ton containers stored in these warehouses. Table 5-6 presents the frequency of occurrence of such winds at these sites. The conditional probability of a missile hitting a ton container in an orientation which could breach the container is  $2.2 \times 10^{-4}$  at NAAP and  $2.7 \times 10^{-4}$  at UMDA (see Appendix C).

5.2.1.3. Open Storage. Ton containers are stored in open storage at APG, PBA, and TEAD. A wind velocity of 250 mph is required to generate a missile that can penetrate these ton containers. The frequences of generating the 250-mph wind are presented in Table 5-7. The probability of a missile hitting a ton container in an orde could breach the container is  $6.6 \times 10^{-4}$  (see Appendix CC).

| AD-A1 | 93 355<br>551F1E | Che<br>The<br>Ca<br>D Sap | MICAL<br>DISPO<br>A W I<br>PEO-CDI | STOCK<br>DSAL O<br>BARSEL<br>E-IS-8 | PILE D<br>F Chen<br>L et A<br>7 <b>000</b> d | ISPOSA<br>(U)<br>L. AUG<br>AAA15- | AL PRO<br>GA TE<br>87 Gi<br>85-D-1 | GRAM R<br>Chnolo<br>R-C-18<br>D022 | ISK AN<br>GIES 1<br>563 | IALYSI<br>NC SAI<br>F/G : | 5 OF<br>N DIEG(<br>15/6.3 | ) 3/<br>NL | 13 |
|-------|------------------|---------------------------|------------------------------------|-------------------------------------|----------------------------------------------|-----------------------------------|------------------------------------|------------------------------------|-------------------------|---------------------------|---------------------------|------------|----|
|       |                  |                           |                                    |                                     |                                              |                                   |                                    |                                    |                         |                           |                           |            |    |
|       |                  |                           |                                    |                                     |                                              |                                   |                                    |                                    |                         |                           |                           |            |    |
|       |                  |                           |                                    |                                     |                                              |                                   |                                    |                                    |                         |                           |                           |            |    |
|       |                  |                           |                                    |                                     |                                              |                                   |                                    |                                    |                         |                           |                           |            |    |
|       |                  |                           |                                    |                                     |                                              |                                   |                                    |                                    |                         |                           |                           |            |    |
|       |                  |                           |                                    |                                     |                                              |                                   |                                    |                                    |                         |                           |                           |            |    |
|       |                  |                           |                                    |                                     |                                              |                                   |                                    |                                    |                         |                           |                           |            |    |
|       |                  |                           | L                                  |                                     |                                              |                                   |                                    |                                    |                         |                           |                           | <u></u>    |    |



reverse exclanated fractional environmal fractionate

### TABLE 5-7 FREQUENCIES FOR WIND-GENERATED MISSILE PENETRATION OF TON CONTAINERS AND SPRAY TANKS STORED IN WAREHOUSES AND OPEN STORAGE

| Site | Storage         | Required<br>Wind | Frequency<br>of Wind                   | Probability<br>of Hitting and<br>Rupturing TC |
|------|-----------------|------------------|----------------------------------------|-----------------------------------------------|
| APG  | Open            | 250              | 1.0E-7                                 | 6.6E-4                                        |
| PBA  | Open            | 250              | 1.5E-6                                 | 6.6E-4                                        |
| NAAP | Warehouse(a)    | 250              | 1.5E-6                                 | 2.2E-4                                        |
| UMDA | Warehouse(a)    | 250              | 1.8E-9                                 | 2.7E-4                                        |
| TEAD | Warehouse(b)    | 283              | 2.7E-10                                | 4.4E-4                                        |
| (a)] | Metal walls.    |                  | ······································ |                                               |
| (Þ)  | Concrete walls. |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 | 5 20             |                                        |                                               |
|      |                 | 5-30             |                                        |                                               |
|      |                 |                  |                                        |                                               |
|      |                 |                  |                                        |                                               |

きんてい ちちちち





5.2.1.4. Tornado-Generated Missiles Cause Munition Detonation. The analysis of scenario SL23 included the estimation of the probability that a missile impacting a munition would cause it to detonate or in the case of rockets, cause the rocket motor to ignite and subsequently detonate the burster. The data presented in Ref. 5-5 indicated that a projectile with Comp B explosive could ignite when subjected to a minimum impact velocity of 123 mph. Because the conditions of the tests described in Ref. 5-5 do not fully apply to the conditions being considered here (i.e., the shell casing provides protection for the bursters), it is assumed that there is a 50% chance that a munition will detonate at 123 mph. Furthermore, Army data indicate that dropping of thousands of burstered munitions from 40 ft did not lead to any detonations (Ref. 5-6). However, these are newer munitions and do not fully represent the chemical munitions in the stockpile. Therefore, based on the consensus of risk experts (Ref. 5-19), an estimated probability of  $10^{-6}$ /munition was assigned to all drops of 6 ft or lower (equivalent to a free fall drop of 13.5 mph). To determine the probability of detonating a munition at an impact velocity equivalent to that of a missile required to penetrate the igloo and the munition, we assumed a lognormal distribution and derived the necessary parameters (e.g., standard deviation and standard normal deviate) from these two data points. The calculation details are given in the calculation sheets (Ref. 5-2).

The overall frequency for this scenario is the product of the following:

- The frequency of a tornado or wind which has sufficient velocity to generate a missile that can penetrate the igloo and munition.
- 2. The probability of a missile penetrating the igloo and hitting the munition in such a way as to cause damage.
- 3. The probability of burster detonation from impact.

The values for the first two variables have already been presented in Section 5.2.1.1. The probability of a detonation given penetration of burstered munitions stored inside the igloos with steel doors is 0.07 and for concrete doors, 0.055. See Ref. 5-2 for calculations.

5.2.1.5. <u>Holding Areas</u>. The holding area is a concrete pad constructed to support equipment for loading containers onto a train, aircraft, or barge. The analysis was based on the following assumptions:

- The maximum number of containers stored at the holding area at any given time is 140 for the rail and barge options and 15 for the air option.
- 2. Since no design information or data is given regarding the holding area nor the arrangement of the containers in the holding area, the largest possible target area for potential missile penetration is used. The target area is munition specific and is a function of the arrangement of the munitions inside the containers and the packing density.
- 3. The puncture resistance for the OFC and the barge package is assumed to be equivalent to 0.75-in thick steel. This is in accordance with the package design criteria provided by MITRE.

Table 5-8 gives the windborne missile velocities for munitions in OFC. The critical missile is the utility pole. Table 5-9 gives the wind frequency sufficient to breach the munitions at the various sites. These frequencies were read from the curves presented in Figs. 4-9 through 4-11. The conditional probability of a missile hitting the munition as calculated using Eq. 5-2 is given in Table 5-10. The target area is munition specific and is calculated as follows based on the package configuration given in Ref. 5-7:

 $\mathbf{A} = \mathbf{N} \times \mathbf{H} \times \mathbf{L}/144$ 

(5-3)

| Munition                    | Container<br>Penetration<br>Velocity,<br>V <sub>c</sub><br>(mph) | Munition<br>Rupture<br>Velocity,<br>V <sub>m</sub><br>(mph) | Required Initial<br>Missile Velocity,<br>V<br>(mph) | Required Wind<br>Velocity<br>(mph) |
|-----------------------------|------------------------------------------------------------------|-------------------------------------------------------------|-----------------------------------------------------|------------------------------------|
| Ton container               | 106                                                              | 67                                                          | 125                                                 | 342                                |
| 4.2-in. mortar              | 106                                                              | 8                                                           | 106                                                 | 310                                |
| 750-1b bomb                 | 106                                                              | 63                                                          | 123                                                 | 338                                |
| 8-in. projectile            | 106                                                              | 25                                                          | 109                                                 | 315                                |
| M23 land mine               | 106                                                              | 6                                                           | 106                                                 | 310                                |
| 105-mm pro-<br>jectile      | 106                                                              | 17                                                          | 107                                                 | 312                                |
| M55 rocket                  | 106                                                              | 8                                                           | 106                                                 | 310                                |
| Spray tank with<br>overpack | 106                                                              | 51                                                          | 118                                                 | 330                                |

TABLE 5-8 WINDBORNE MISSILE VELOCITY (HOLDING/LOADING AREA - RAIL/AIR OPTION) ond see that the second second second

Notes: 1. Critical missile is the utility pole.

2. See Ref. 5-2 for details.



### TABLE 5-9PROBABILITY OF A WIND SUFFICIENT TO GENERATE MISSILES TO BEACH<br/>MUNITION (HOLDING/LOADING AREA - RAIL/AIR OPTION)

|                             |                                              | Probabi                     | lity of Occu<br>(P <sub>w</sub> )            | irrence/Year                          |
|-----------------------------|----------------------------------------------|-----------------------------|----------------------------------------------|---------------------------------------|
|                             | Required<br>Windborne Missile<br>Velocity(&) | Transporta<br>Hol<br>(Sendi | tion Contain<br>ding/Loading<br>ng or Receiv | ner (TRC) in<br>3 Area<br>7 Ing Site) |
| Munition                    | (mph)                                        | Zone I(b)                   | Zone II(c)                                   | Zone III(d)                           |
| Ton container               | 342                                          | 8.9E-12                     | 1.4E-9                                       | 4.3E-8                                |
| 4.2-in. mortar              | 310                                          | 5.6E-11                     | 6.3E-9                                       | 1.5E-7                                |
| 750-1b bomb                 | 338                                          | 1.1E-11                     | 1.7E-9                                       | 5.0E-8                                |
| 8-in. projectile            | 315                                          | 4.2E-11                     | 5.0E-9                                       | 1.2E-7                                |
| M23 land mine               | 310                                          | 5.6E-11                     | 6.3E-9                                       | 1.5E-7                                |
| 105-mm projectile           | 312                                          | 5.0E-11                     | 5.8E-9                                       | 1.4E-7                                |
| M55 rocket                  | 310                                          | 5.6E-11                     | 6.3E-9                                       | 1.5E-7                                |
| Spray tank with<br>overpack | 330                                          | 1.8E-11                     | 2.5E-9                                       | 6.8E-8                                |

(a)From Table 5-8.

(b)Zone I - TEAD and UMDA sites (probability values obtained from Fig. 4-9).

(c)Zone II - PUDA and APG sites (probability values obtained from Fig. 4-10).

(d)Zone III - ANAD, PBA, LBAD, and NAAP sites (probability values obtained from Fig. 4-11).





## PROBABILLITY OF MISSILE STRIKING MUNITIONS IN HOLDING/LOADING AREA (RAIL/AIR OPTION) TABLE 5-10

|                             | Pď                              | Po                          | DE                                      | AT                                | P <sub>P</sub> (a)                                |
|-----------------------------|---------------------------------|-----------------------------|-----------------------------------------|-----------------------------------|---------------------------------------------------|
| Munition                    | Missile Velocity<br>Probability | Missile Axis<br>Probability | Missile Density<br>(1/ft <sup>2</sup> ) | Target Area<br>(ft <sup>2</sup> ) | Conditional<br>Probability of<br>Munition Rupture |
| Ton container               | 0.17                            | 0.015                       | 1.9 x 10 <sup>-5</sup>                  | 1155                              | 5.6 x 10 <sup>-5</sup>                            |
| 4.2-in. mortar              | 0.17                            | 0.015                       | 1.9 x 10 <sup>-5</sup>                  | 2036                              | 9.9 x 10 <sup>-5</sup>                            |
| 750-1b bomb                 | 0.17                            | 0.015                       | 1.9 x 10 <sup>-5</sup>                  | 1792                              | 8.7 x 10 <sup>-5</sup>                            |
| 8-in. projectile            | 0.17                            | 0.015                       | 1.9 x 10 <sup>-5</sup>                  | 1329                              | 6.4 x 10 <sup>-5</sup>                            |
| M23 land mine               | 0.17                            | 0.015                       | 1.9 x 10 <sup>-5</sup>                  | 1805                              | 8.7 × 10 <sup>-5</sup>                            |
| 105-mm projectile           | 0.17                            | 0.015                       | 1.9 x 10 <sup>-5</sup>                  | 1341                              | 6.5 x 10 <sup>-5</sup>                            |
| M55 rocket                  | 0.17                            | 0.015                       | 1.9 x 10 <sup>-5</sup>                  | 2195                              | 1.1 × 10 <sup>-4</sup>                            |
| Spray tank with<br>overpack | 0.17                            | 0.015                       | 1.9 x 10 <sup>-5</sup>                  | 1966                              | 9.5 x 10 <sup>-5</sup>                            |
| (a) For sir out             | ion multinlu D_ hu              | 10717517                    |                                         |                                   |                                                   |

.(041/CI) yd gy yrgun muitipin y gy (12/140).

5-35

11.1



where N = number of munition pallets that a missile could hit,

- H = effective height of the pallet (in.),
- L = effective length or width of the pallet, whichever gives the most critical target area.

Table 5-11 gives the windborne missile velocity for the vault, the wind frequency sufficient to generate missiles, and the probability of the missile hitting and rupturing the munitions, based on Eqs. 5-1 through 5-3. Calculations are provided in Ref. 5-2.

For the marine option, the tornado scenario addressed only the packages temporarily stored in the holding area. A tornado-generated missile strike of the lighter or ship at rest was not considered credible.

### 5.2.2. <u>Meteorite Strikes</u>

Like tornado-generated missiles, meteorites striking the igloos, warehouses, and the outdoor yards can lead to a significant amount of agent release. The consequence of such an accident is more severe than that from a tornado-generated missile because meteorite strikes generally involve fires. Hence, if burstered munitions are involved, explosive detonations could occur from the fire or from direct impact, leading to instantaneous agent releases.

The event tree developed for meteorite-initiated accidents is shown in Fig. 5-2. The scenarios could not be subjected to any preliminary screening without doing a more detailed analysis of the what type (stone or iron) and size of meteorite is capable of penetrating munitions stored igloos, warehouses, or outdoors. The accident sequences identified are: SL8, SR9, SA9, and SW9. - ARCENT BURNEL NAME







LINGSCONDUCTION CONTINUES OF A STAT

# TABLE 5-11 TORNADO-GENERATION MISSILE ANALYSIS OF THE VAULT (MARINE OPTION)<sup>(&)</sup>

NAKA MANJUTA ANGRA MATUMATIN N

|                                       | windborne<br>Veloc<br>(mph | missile<br>ity<br>)     |                              |                           |                         | Probability                |
|---------------------------------------|----------------------------|-------------------------|------------------------------|---------------------------|-------------------------|----------------------------|
|                                       | Package<br>Penetration     | Munition<br>Penetration | Required<br>Initial Velocity | Required<br>Wind Velocity | Tornado<br>Frequency/yr | of Munition<br>Penetration |
| n container at APG<br>ornado Zone II) | 106                        | 67                      | 125                          | 342                       | 1.45-9                  | 4.6E-4                     |

(a) See Ref. 5-2 for details.

· · ·

ζ.,

### Storage Magazines

In this scenario (SL8), the meteorite penetrates the storage magazine and ruptures some of the munitions stored inside. The meteorite is expected to be sufficiently hot to cause ignition of the exposed burster, propellant, and/or agent. The fire is expected to spread, resulting in the destruction of the entire inventory of the storage magazine. WANNA TEANAN TEAN TO TAKE A TAKA TA

### Warehouses

This scenario is similar to the storage magazines. The meteorite penetrates the warehouse and ruptures some of the bulk munitions stored inside. The meteorite causes the ignition of the exposed agent. Fire spreads and results in the destruction of the entire warehouse inventory.

### **Open Storage**

In this scenario, the meteorite directly impacts and ruptures some ton containers. The heat from the meteorite is expected to ignite the exposed agent, but is not expected to cause the rupture of additional munitions.

### Holding Area

This scenario (sequences SR9, SA9, SW9) applies to munitions in OFCs and barge packages temporarily stored in the holding area. As with the tornado-generated missile scenario, the OFC or barge package provide the first structural barrier for missile penetration. The same assumptions used in the tornado analysis apply here. For the marine option, the meteorite strike scenario addressed only the packages temporarily stored in the holding areas. A meteorite strike while munitions are in the lighter or ship at rest was not considered credible. 5.2.2.1. Meteorite Strike Accident Analysis. About 3500 meteorites, each weighing over 1 lb, strike the earth each year; the majority of them are of small sizes (Ref. 5-8). Given the earth's surface area of 5.48 x  $10^{-15}$  ft<sup>2</sup>, the frequency of meteorite strikes for meteorites weighing 1.0 lb or greater is 6.4 x  $10^{-13}/\text{ft}^2$  (Ref. 5-8). For meteorites one ton or less, stone meteorites are approximately 10 times more common than iron. However, iron meteorites are more dense and tend to have higher impact velocities and therefore represent a significant portion of the total meteorites that can rupture the munitions. Table 4-18 shows the size distribution of both iron and stone meteorites. The table was compiled from data presented in Refs. 5-8 and 5-9.

For agent to be released, the meteorite has to penetrate the storage structure and the munition wall. In the case of an igloo, this would require initial penetration of a 6-in. concrete roof. The minimum meteorite impact velocity that would collapse the earth cover and the 6-in. concrete roof is 1500 fps for stone meteorite and 3800 fps for iron meteorite. The overall frequency of a meteorite capable of penetrating and rupturing the munitions in the igloo is:

$$P = F (F_s + F_j) A \times S , \qquad (5-4)$$

- where F = the frequency of a meteorite weighing one pound or more striking the earth, 6.4 x  $10^{-13}/\text{ft}^2$ ,
  - $F_8$  = fraction of stone meteorites which can penetrate the target,  $F_1$  = fraction of iron meteorites which can penetrate the target, A = target area (igloo, warehouse, or open storage yard,

S = spacing factor.

Table 5-12 presents the frequencies for meteorite penetration of munitions stored in the various storage configurations along with the size of the meteorites required to penetrate the munitions and the data

|                        |                                      | Stone M        | eteorite         | Iron M         | eteorite         | Tarvet Area               |                       |
|------------------------|--------------------------------------|----------------|------------------|----------------|------------------|---------------------------|-----------------------|
| Storage Area           | Munition                             | Weight<br>(1b) | Fraction<br>(fs) | Weight<br>(1b) | Fraction<br>(fi) | (ft <sup>2</sup> )<br>(A) | Spacing Factor<br>(s) |
| Igloo                  | All                                  | 1,000          | 0.02             | 200            | 0.003            | 960                       | 0.5                   |
| Warehouse-NAAP         | Ton container                        | 20             | 0.11             | 2              | 0.03             | 22,000                    | 0.5                   |
| Warehouse-UMDA         | Ton container                        | 20             | 0.11             | 7              | 0.03             | 46,000                    | 0.4                   |
| Warehouse-TEAD         | Spray tank                           | 100            | 0.08             | 10             | 0.02             | 67,000                    | 0.4                   |
| Open                   | Ton container                        | 20             | 0.11             | 7              | 0.03             | 139(8)                    | 1.0                   |
| Rail holding area      | All                                  | 100            | 0.05             | 20             | 0.013            | 154,000(b)                | 0.030 to 0.053(c)     |
| Air holding area       | Ton container,<br>projectile, rocket | 100            | 0.05             | 20             | 0.013            | 16,500(d)                 | 0.030 to 0.035(c)     |
| Marine holding<br>area | Ton container                        | 100            | 0.05             | 20             | 0.013            | 50,400(e)                 | 0.19                  |
| (a)Area of one         | pallet (15 ton contai                | ners sta       | cked two h       | igh).          |                  |                           |                       |
| (b)700 x 220 ft        | (Ref. 5-17).                         |                |                  |                |                  |                           |                       |
|                        |                                      |                |                  |                |                  |                           |                       |

METEORITE REQUIRED FOR PENETRATION OF MUNITIONS IN STORAGE TABLE 5-12

See Ref. 5-2 for details. (c)Spacing factor is munition specific.

(d)(15/140)(154,000).

(e)Based on 14 x 10 array of 106 x 44 x 54 in. vaults with 1-ft clearance.

122222222

CODUCATE CONSECCE - SERVICE - SAVERED AND VERTING REPORT REPORT OF THE SAVER THE SAVERED REPORT

1.1.1.1.1.1.1



- SSEE NEWS

required to evaluate Eq. 5-4. Supporting calculations are presented in Ref. 5-2, and the methodology is discussed in Appendix C.

### 5.2.3. Aircraft Crashes

The sequences describing the effects of an aircraft crash on munitions in storage are SL4, SL5, SL15, SL16, SL17, SL18, SL19, SL20, SL21, SR1 through SR6, SA1 through SA6, SW1 through SW6, and SW10 through SW21.

The effects of large (>12,500 lb) and small (12,500 lb or less, including helicopters) aircraft crashes on the munitions in storage igloos, warehouses, and open yards were evaluated. Because of the potential for large quantities of fuel to be carried by large aircraft and the potential for large, high-velocity missiles (e.g., engines), the large aircraft crash scenarios were further divided into direct and indirect crashes. For direct and indirect large aircraft crashes onto the storage area that do not result in fire, it is assumed that the impact of the crash is strong enough to cause the detonation of burstered munitions. For munitions in OFCs or vaults, only direct aircraft crashes were considered, since the target area considered was large enough to include indirect hits and the effects on the munitions will be the same.

For a small aircraft crash adjacent to the storage site to produce a credible event, the crash would have to be so close that it would virtually be a direct hit. Therefore, the small aircraft crash scenarios address only direct hits into the storage areas including holding areas.

The event trees developed to identify the agent release scenarios from aircraft crashes are shown in Figs. 5-3 through 5-9.

5.2.3.1. <u>Aircraft Crash Accident Analysis</u>. In summary, the following general assumptions were made in deriving the large/small aircraft accident scenarios:

£184.882.884.8°0.834.8°0.8°0.828.8

- 1. For large aircraft crashes onto burstered munitions, it is assumed that detonations will occur for both indirect and direct hits, and, if a fire occurs, it is uncontained.
- 2. No small aircraft crashes were assumed to be able to sufficiently damage the igloo to cause agent releases.

### Direct Crash of Large Aircraft (Sequences SL4, SL16, SL17, SR1 Through SR3, SA1 Through SA3, SW1 Through SW3, SW10 Through SW12, and SW16 Through SW18)

For a direct aircraft crash, the target area is the surface area of the building or open yard.

<u>Storage Magazines</u>. The direct crash of the main body of a heavy military or commercial aircraft into the shell or front face of a storage magazine (igloo) can breach the igloo and allow crash-generated missiles and/or aviation fuel to enter into the igloo. There is a high probability that one or more munitions will be crushed or punctured by the missiles. Burstered munitions could also detonate from impact. If the crash produces a fire, the fire is expected to spread through the igloo, resulting in the destruction of the entire igloo inventory.

<u>Warehouses</u>. Since a warehouse is not expected to offer any substantial resistance to the crash of a large aicraft, the direct impact of any part of a large airacraft is expected to breach the warehouse and subject the stored munitions to crash-generated missiles. Bulk containers will be crushed or punctured. If the crash produces a fire, the fire is expected to spread, resulting in the destruction of the entire inventory. Open Storage. The crash of a large aircraft into an open area is expected to breach a large number of ton containers. If the crash produces a fire, and it is not contained, it is expected to breach additional containers in the immediate vicinity of the initial container that is on fire.

<u>Holding Area</u>. The crash of a large aircraft onto the OFCs temporarily stored in the holding area is expected to breach a large number of munitions. For the rail and marine options analysis, the 140 OFCs are assumed to be arranged in a 14 by 10 array in the holding area. This configuration was used to determine the target area for a plane crash. Since there will only be 15 OFCs in the holding area for the air option, the target area was adjusted proportionately.

Lighter/LASH at Rest. For the marine option, the direct crash of a large aircraft onto (1) a flotilla of lighters while awaiting loading onto a LASH and (2) LASH vessel were also considered. Ten lighters are assumed to be in the area at any given time. The size of a lighter is  $6.88 \times 10^{-5}$  mi<sup>2</sup>. The size of a LASH is  $2.94 \times 10^{-3}$  mi<sup>2</sup>. The entire time for operations of loading the lighters and the LASH is expected to be 14 days. Hence, the lighters and the LASH will be exposed for only a fraction of the year (i.e., 14 days/365 days).

### Indirect Crash of a Large Aircraft (Sequences SL5, SL20, SL21)

For an indirect crash, the target area is determined by increasing all perimeters for the direct crash by 200 ft.

Storage Magazines. Should a large aircraft crash adjacent to an igloo, the area that is most vulnerable is the igloo door. The crashgenerated missiles can breach the igloo door which essentially provides a pathway to the breaching of munitions in the line of site of the missile. Alternatively, the igloo door may already be open at the time of the crash and the missile could directly penetrate the munitions. If

and the second state of the se

fire is involved, the missile could already be on fire or the fire could propagate into the igloo opening. Thus, if fire is not contained, the amount of agent release is the same as for the direct crash of a large aircraft into an igloo.

<u>Warehouses</u>. The designs of the warehouses are such that the crash of a large aircraft into an area adjacent to a warehouse may also breach the warehouse if the aircraft is flying towards the warehouse at the time of the crash. The amount of munitions that are initially impacted would be less than the direct crash scenario. However, if fire is involved and uncontained, the amount of agent release is the same as for the diect crash of large aircraft into a warehouse.

<u>Open Storage</u>. The accident scenario for the crash of a large aircraft into an area adjacent to the open storage area considers that there is a 50% chance that some ton containers would be breached by the crash-generated missile. If fire is involved and not contained, additional containers would rupture due to excessive heating.

Holding Area. This scenario was not considered for the munitions in OFCs or barge package temporarily stored in the holding area since the effects on the munitions will be the same as the direct crash.

Lighter/LASH at Rest. This scenario was not considered for the marine option since the target area considered for the direct crash was sufficient to include indirect hits, and the effects on the munitions will be the same as the direct crash.

Direct Crash of a Small Aircraft ( Sequences SL15, SL18, SL19, SR4 Through SR6, SA4 Through SA6, SW4 Through SW6, SW13 Through SW15, and SW19 Through SW21)

<u>Storage Magazines</u>. Due to the high strength of the storage magazine, the crash of a small aircraft is not expected to breach an igloo or affect the structural integrity of an igloo. <u>Warehouses</u>. The crash of a small aircraft into a warehouse would very likely breach the warehouse. The resulting crash-generated missiles are expected to crush or puncture some munitions. If the crash produces a fire and it is not contained, the fire would involve the entire inventory.

<u>Open Storage</u>. The crash of a small aircraft into an open storage area is similar to the large aircraft crash into an open storage area except a smaller number of ton containers is breached.

Lighter/LASH at Rest. The crash of a small aircraft onto a flotilla of lighters or onto the LASH vessel is similar to the large aircraft crash except that the extent of damage could be less severe.

5.2.3.2. <u>Aircraft Crash Frequency</u>. The frequency of an aircraft crashing while in an airway or the vicinity of an airport can be computed as shown in Section 4.2.1.3.

The annual frequency of a crash into a specific facility was computed by multiplying the appropriate frequency taken from Table 4-16 by the effective target area of the facility (see Appendix C). Table 5-13 summarizes these annual frequencies. The calculations of the effective areas are contained in Ref. 5-2 and take into account such factors as aircraft wing span, facility height, and facility vulnerability.

5.2.3.3. <u>Probability of Fire Resulting From An Aircraft Crash</u>. The probability of a fire resulting from the crash has been estimated to be 0.45 (Ref. 5-12). The successful containment of the fire is defined here to be 0.5 h for unpackaged nonburstered munitions. This time was selected based on the thermal failure threshold data presented in Appendix F, which indicate that direct heating of ton containers for 36 min leads to hydraulic rupture. For unpackaged burstered munitions, the thermal failure threshold range from 4 min for rockets to 23 min for mines. Since the Army policy is not to fight a fire involving direct KCCCCC ASSOCIATE RESSERVE PRODUCES RECORDER REPORTE REPORT 

|            | STORAGE         |
|------------|-----------------|
|            | FOR             |
|            | SCENARIOS       |
| TABLE 5-13 | CRASH-INITIATED |
|            | AIRCRAFT        |
|            | FOR             |
|            | BASE            |
|            | DATA            |

| Event                              | Variable<br>ID | Frequency or<br>Probability | Unit         | Error<br>Factor | Reference |
|------------------------------------|----------------|-----------------------------|--------------|-----------------|-----------|
| Large aircraft direct crash storag | 80             |                             |              |                 |           |
| area:                              |                |                             |              |                 |           |
| ANAD - 60 ft igloo                 | LDAN160        | 4.5E-10                     | Per facility | 10              | Ref. 5-2  |
| ANAD - 80 ft igloo                 | LDAN180        | 6.0E-10                     | year         | 10              |           |
| APG - open                         | LDAPOP         | 2.4E-09                     |              | 10              |           |
| LBAD - 89 ft igloo                 | LDLBI89        | 3.7E-10                     |              | 10              |           |
| NAAP – wh                          | LDNAWH         | 3.6E-09                     |              | 10              |           |
| PBA - 80 ft igloo                  | LPDBI80        | 1.1E-10                     |              | 10              |           |
| - open                             | LDPBOP         | 1.7E-08                     |              | 10              |           |
| PUDA - 80 ft igloo                 | LDPUI80        | 4.5E-09                     |              | 10              |           |
| TEAD - 80 ft igloo                 | LDTEI80        | 2,7E-11                     |              | 10              |           |
| - 89 ft igloo                      | LDTE189        | 3.0E-11                     |              | 10              |           |
| - wh                               | LDTEWH         | 8.7E-10                     |              | 10              |           |
| - open                             | LDTEOP         | 7.9E-09                     |              | 10              |           |
| UMDA - 80 ft igloo                 | LDUM180        | 1.1E-09                     |              | 10              |           |
| - wh                               | LDUMWH         | <b>2.5E-08</b>              |              |                 |           |
| Large aircraft indirect crash:     |                |                             |              |                 |           |
| ANAD - 60 ft igloo                 | <b>LAANI60</b> | 5.5E-08                     | Per facility | 10              | Ref. 5-2  |
| ANAD - 80 ft igloo                 | LAAN180        | 5.7E-08                     | year         | 10              |           |
| APG - open                         | LAAPOP         | 9.4E-09                     |              | 10              |           |
| LBAD - 89 ft igloo                 | LALB189        | 3.3E-08                     |              | 10              |           |
| NAAP - wh                          | LANAWH         | 5.0E-08                     |              | 10              |           |
| PBA - 80 ft igloo                  | LAPB180        | <b>1.1E-08</b>              |              | 10              |           |
| - open                             | LAPBOP         | <b>3.5E-08</b>              |              | 10              |           |
| PUDA - 80 ft igloo                 | LAPUI80        | 4.3E-07                     |              | 10              |           |
| TEAD - 80 ft igloo                 | LATE180        | 2.6E-09                     |              | 10              |           |
| - 89 ft igloo                      | LATE189        | 2.7E-09                     |              | 10              |           |
| - wh                               | LATEWH         | 7.9E-09                     |              | 10              |           |
| - open                             | LATEOP         | 1.3E-08                     |              | 10              |           |

ad de la desta de la seconda de la desta de la dest

1028582838387 F238889393

init.



hat a fait a

TABLE 5-13 (Continued)

DADADADADADADADADADADADADA

| Event                                                  | Variable<br>ID    | Frequency or<br>Probability | Unit                 | Error<br>Factor |      | Reference          |
|--------------------------------------------------------|-------------------|-----------------------------|----------------------|-----------------|------|--------------------|
| UMDA - 80 ft igloo<br>- wh                             | LAUMI80<br>LAUMWH | 1.1E-07<br>1.3E-07          | Per facility<br>year | 10              |      |                    |
| Igloo breached given direct crash                      | 11<br>D           | 8.0E-01                     | None                 | 1.4             | ЕJ   |                    |
| Igloo breached given indirect crash                    | IA                | 2.3E-03                     | None                 | e               | Ref. | 5-2                |
| Warehouse/outdoor container<br>breached (direct crash) | DHM               | 1.0E+00                     | None                 | None            | EJ   |                    |
| Warehouse breached given indirect<br>crash             | MHA               | 1.7E-01                     | None                 | 2               | Ref. | 5-2                |
| Outdoor container breached<br>(indirect crash)         | <b>V</b> O        | 5.0E-01                     | None                 | 1.4             | Ref. | 5-2                |
| Crash does not involve fire                            | NF                | 5.5E-01                     | None                 | None            | Ref. | 5-10               |
| Crash results in fire                                  | ΥF                | 4.5E-01                     | None                 | None            | Ref. | 5-10               |
| Fire not contained in 1/2 h<br>(burstered)             | FNCD              | 1.0E+00                     | None                 | None            | Ref. | 5-2 and Appendix J |
| Fire contained in 1/2 h<br>(nonburstered)              | FCNB              | 3.4E-04                     | None                 | Ś               | Ref. | 5-2 and Appendix J |
| Fire not contained in 1/2 h<br>(nonburstered)          | FNCNB             | 1.0E+00                     | None                 | None            | Ref. | 5-2 and Appendix J |
| Fire contained (wh or op) small                        | SFNB              | 1.9E-02                     | None                 | ę               | Ref. | 5-2                |
| Small aircraft crash warehouse NAAP                    | SANAAP            | <b>1.8E-08</b>              | Per facility<br>vear | 10              | Ref. | 5-2                |
**MAADOON** 00000014 Ņ

TABLE 5-13 (Continued)

|                                  | Event                            | Variable<br>ID | Frequency or<br>Probability | Unit                 | Error<br>Factor | Referen  | lce |
|----------------------------------|----------------------------------|----------------|-----------------------------|----------------------|-----------------|----------|-----|
| Small aircraft                   | crash warehouse UMDA             | SAUMDA         | 2.0E-08                     | Per facility<br>year | 10              | Ref. 5-2 |     |
| Small aircraft                   | crash warehouse TEAD             | SATEAD         | 3.5E-08                     |                      | 10              | Ref. 5-2 |     |
| Small aircraft                   | crash open APG                   | SAOAPG         | 3.6E-05                     |                      | 10              | Ref. 5-2 |     |
| Small aircraft                   | crash open PBA                   | SAOPBA         | 1.3E-06                     |                      | 10              | Ref. 5-2 |     |
| Small aircraft                   | crash open TEAD                  | SAOTEAD        | 3.2E-07                     | -                    | 10              | Ref. 5-2 |     |
| Large aircraft<br>area (rail and | direct crash holding<br>marine): |                |                             | -                    |                 |          |     |
| ANAD                             |                                  | LDHAN          | 1.4E-08                     | Per year             | 10              | Ref. 5-2 |     |
| APG                              |                                  | LDHAP          | 9.6E-10                     | Per year             | 10              |          |     |
| LBAD                             |                                  | LDHLB          | 8.1E-09                     | Per year             | 10              |          |     |
| NAAP                             |                                  | LDHNA          | 8.3E-09                     | Per year             | 10              |          |     |
| PBA                              |                                  | LDHPB          | 2.7E-09                     | Per year             | 10              |          |     |
| PUDA                             |                                  | гднри          | 1.1E-07                     | Per year             | 10              |          |     |
| TEAD                             |                                  | LDHTE          | 6.5E-10                     | Per year             | 10              |          |     |
| UMDA                             |                                  | LDHUM          | 2.7E-08                     | Per year             | 10              |          |     |
| Small aircraft<br>area (rail and | direct crash holding<br>marine): |                |                             |                      |                 |          |     |
| ANAD                             |                                  | SDHAN          | 2.2E-08                     | Per year             | 10              | Ref. 5-2 |     |
| APG                              |                                  | SDHAP          | 1.4E-05                     | Per year             | 10              |          |     |
| LBAD                             |                                  | SDHLB          | 3.3E-10                     | Per year             | 10              |          |     |
| NAAP                             |                                  | <b>NHUS</b>    | 4.2E-08                     | Per year             | 10              |          |     |
| PBA                              |                                  | SDHPB          | 2.0E-07                     | Per year             | 10              |          |     |
| PUDA                             |                                  | SDHPU          | 1.8E-07                     | Per year             | 10              |          |     |
| TEAD                             |                                  | SDHTE          | 2.6E-08                     | Per year             | 10              |          |     |
| NDA                              |                                  | SDHUM          | 2.2E-08                     | Per year             | 10              |          |     |
|                                  |                                  |                |                             |                      |                 |          |     |

14.25.23 N. 19.20

\*\*\*

- 222 • •







## TABLE 5-13 (Continued)

| Event                                              | Variable<br>ID | Frequency or<br>Probability | Unit            | Error<br>Factor |      | Reference |
|----------------------------------------------------|----------------|-----------------------------|-----------------|-----------------|------|-----------|
| Fire contained (offsite) large<br>aircraft         | LFC2           | 3.5E-01                     | None            | 2               | Ref. | 5-2       |
| Fire contained (offsite) small<br>aircraft         | SFC2           | 8.7E-01                     | None            |                 | Ref. | 5-2       |
| Large aircraft direct crash holding<br>area (air): |                |                             |                 |                 |      |           |
| APG                                                | ALAP           | 1.1E-09                     | Per year        | 10              | Ref. | 5-2       |
| LBAD                                               | ALLB           | 5.8E-09                     | Per year        | 10              |      |           |
| TEAD                                               | ALTE           | 6.0E-09                     | Per year        | 10              |      |           |
| Small aircraft direct crash holding<br>area (air): |                |                             |                 |                 |      |           |
| APG                                                | ASAP           | 1.5E-06                     | Per year        | 10              | Ref. | 5-2       |
| LBAD                                               | ASLB           | 3.5E-11                     | Per year        | 10              |      |           |
| TEAD                                               | ASTE           | 2.8E-09                     | Per year        | 10              |      |           |
| Large aircraft crash (marine only)<br>crashes/year |                |                             | ſ               | 1               |      |           |
| Lighter                                            | BLHDAP         | 1.4E-11                     | <b>Per year</b> | 10              | Ref. | 5-2       |
| Ship                                               | SLHDAP         | 6.0E-11                     | Per year        | 10              | Ref. | 5-2       |
| Small aircraft crash (marine only)                 |                |                             | I               |                 | 1    |           |
| Lighter                                            | DSHDAP         | 2.1E-07                     | Per year        | 10              | Ref. | 5-2       |
| Ship                                               | SSHDAP         | 8.8E-07                     | Per year        | 10              | Ref. | 5-2       |

heating of burstered munitions, the probability of the "failure to contain fire" event is essentially 1.0.

Thus, the amount of agent released from bulk containers subjected to aircraft crash fires depends on the ability to contain the fire. If fire is allowed to progress for more than 30 min, more containers will rupture.

The ability of the fire-fighting team to extinguish an aircraft crash fire depends on many variables such as the precise crash site, the burn time of the resulting fire, the availability of resources necessary to contain the fire, etc. If fire fighters arrive at the crash site in a relatively short period of time, the fire will be easier to extinguish since it is not likely to have spread very far. Because the fire will involve chemical agent, additional precautions will have be taken before the fire-fighting team can start extinguishing the fire. Their arrival at the perimeter of the MDB or MHI is assumed to occur about 5 min after the crash. The crew will have to put on agent protective clothing in addition to their normal, fire-fighting suits of thermal protective clothing. Donning these clothes and checking for proper mask fit would take several more minutes, if it is assumed that the crew was partially dressed; i.e., in a standby readiness mode. Because of all the detection, observation, communication, preparation, and travel tasks involved, it is estimated that it would take the fire-fighting team 15 min to get to the scene of the fire.

Once at the scene, the time it takes to actually extinguish the fire is difficult to estimate. GA interviewed local fire fighting personnel to get their opinion on how long it takes to extinguish a fire from a small aircraft crash versus large aircraft crash. No definite time can be given because of the many variables involved. But based on local experience, it would take 1 to 3 h to extinguish a fire from a small aircraft; while it would take 3 to 10 h for a large aircraft fire. Using the lognormal distribution, GA then derived the probability of

containing the fire in 0.5 h or less and took no credit for the first 15 min of the fire. More details are provided in the calculation sheets (Ref. 5-2).

For munitions in OFCs or barge packages, it is assumed that the intact containers can withstand a 2-h all engulfing fire. Therefore, the successful fire containment for the SR, SA, and SW aircraft fire scenarios is defined as the ability to put out the fire in 2.5 h since thermal rupture of munitions take additional minutes.

### 5.2.4. Earthquakes

5.2.4.1. <u>Storage Magazines</u>. The earthquake-initiated accident affecting the storage igloos assumes that the earthquake causes the munitions in the igloo to fall and be punctured given the presence of a probe on the igloo floor or the fall could cause a burstered munition to detonate (Sequence SL7). This sequence is modeled using the event tree illustrated in Fig. 5-10.

The storage magazines are expected to survive the largest credible earthquake with little or no damage. Some cracking or spalling of the concrete is possible, but this should not produce a threat to the munitions or significantly change the containment capability of the magazine. Igloos have been tested by very large external explosions and have survived without damage (Ref. 5-11). The data from these tests indicate that the igloo experienced accelerations which were in excess of 20 g. Though an explosion is not as potentially damaging to an igloo as an earthquake of equal acceleration, the similarities are sufficient to conclude that a very large earthquake, in the range of 1.0 g, is not likely to damage an igloo.

Sequence SL7 postulates that the earthquake causes the stacked munitions to fall and may be punctured upon impact. Based on the coefficient of friction between pallets of munitions, a 0.3-g earthquake will likely cause some stacked munitions to fall and a 0.5-g earthquake will cause a large number to fall. The highest stacked munitions in an igloo can potentially fall 6 ft. The munition failure threshold data indicate that all palletized munitions and bulk containers can survive the impact of a drop from this height but could be punctured if they were to land on a probe which was sufficiently sharp and rigid. For this analysis a 0.3-g earthquake was assumed to cause 25% of the stacked pallets to fall while a 0.5-g earthquake will cause 100% of the stacked pallets to fall. The number of pallets which have the potential of impacting a probe was estimated for each munition type based on (1) how the pallets are stacked and (2) the floor area available for the pallets to fall. The calculation details are provided in Ref. 5-2.

The analysis of the presence of a probe in the igloo has indicated that it is unlikely that there is a probe inside the igloo that is sufficiently rigid and sharp to damage a munition. Table 5-14 provides the earthquake frequency data for each of the eight sites and the puncture probability of a munition type given a 6-ft drop.

Sequence SL22 involves the detonation of burstered munitions resulting from an earthquake-induced fall. The probability of a munition detonating from a 6-ft drop is estimated using the same approach discussed for detonations due to impact by wind-generated missiles.

5.2.4.2. <u>Warehouses</u>. The event tree describing release scenarios resulting from earthquake-induced accidents in warehouses is shown in Fig. 5-11. The event tree applies to the long-term storage warehouses at TEAD, NAAP, and UMDA. Spray tanks are stored at the two warehouses at TEAD. Ton containers are stored at NAAP in one warehouse and at UMDA in two adjacent warehouses.

Accident sequences describing releases from long-term storage warehouses are given in Table 5-15. Sequence designations are SLxxx26x for the NAAP warehouse, SLxxx27x for the TEAD warehouses, and SLxxx28x

|                                          | Map Area 5<br>Site: TEAD | Map Area 2<br>Site: ANAD, LBAD, PBA,<br>UMDA, and PUDA |
|------------------------------------------|--------------------------|--------------------------------------------------------|
| Earthquake frequency (/yr) at            |                          |                                                        |
| 0.3 to 0.5 g (F <sub>1</sub> )           | 6.0E-4                   | 1.9E-5                                                 |
| >0.5 g (F <sub>2</sub> )                 | 1.0E-4                   | 6.0E-6                                                 |
| Probability stacked pallets will fall at |                          |                                                        |
| 0.3 to 0.5 g (P <sub>1</sub> )           | 0.25                     | 0.25                                                   |
| >0.5 g (P <sub>2</sub> )                 | 1.0                      | 1.0                                                    |

|      |       |     | T    | ARL | E 5- | 14      |          |     |
|------|-------|-----|------|-----|------|---------|----------|-----|
| DATA | BASE  | FOR | ANAL | YSI | S OF | EARTHQU | AKE-INDU | CED |
|      | AGENT | REL | EASE | IN  | THE  | STORAGE | IGLOOS   |     |

|                       | Number of Mun<br>Falling A                               | itions<br>t                              |
|-----------------------|----------------------------------------------------------|------------------------------------------|
| Munition Type         | $(N_1)$<br>0.3 to 0.5 g                                  | (N <sub>2</sub> )<br>>0.5 g              |
| Bomb                  | 3                                                        | 11                                       |
| 105-mm cartridge      | 5                                                        | 20                                       |
| 4.2-in. mortar        | 5                                                        | 18                                       |
| Ton container         | 6                                                        | 22                                       |
| Mine                  | 4                                                        | 14                                       |
| Projectile            | 11                                                       | 46                                       |
| Rocket                | 5                                                        | 20                                       |
| Spray tank            | N/A                                                      | N/A                                      |
| SL7 (accident frequen | cy) = $(F_1 * P_1 * + (F_2 * P_2 * + (F_2 * P_2 * P_2))$ | r N <sub>1</sub> )<br>r N <sub>2</sub> ) |



• • • Ě

Ř

RANGE AND AND A MERINA SERVICES

15

10.03

0.007.0



さん さんえい さいさいがく





21-Aug-87 PAGE 1

STORAGE EARTHQUANE - WAREHOUSES

CLORAGE - EARTHQUAKE-INDUCED ACCIDENTS IN THE WAREHDUSES (FER YEAR)

| SCEMARIC         NJ         PAME         AFE         AFA         AF                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                          |               |                 |              |                 |               |                 |              |                 |              |                 |              |                |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------|---------------|-----------------|--------------|-----------------|---------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|----------------|
| State     25     N:A     N:A     N:A     N:A       State     27     N:A     N:A     N:A     N:A       State     281     N:A     N:A     N:A     N:A       State     281                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | APG RAVGE<br>FREG FACTOR | L BAD<br>Freq | RANGE<br>FACTOR | NAAP<br>Fred | RANGE<br>Factor | F BA<br>F REQ | RANGE<br>FACTOR | PUDA<br>FREQ | RANGE<br>Factor | TEAO<br>Freg | RANGE<br>Factor | UNDA<br>Freg | FACTCP         |
| State         261         K/A         K/A </th <th></th>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                          |               |                 |              |                 |               |                 |              |                 |              |                 |              |                |
| SUPTC     252     N-6     N/6     N/6     N/6     N/6       SUPUC     255     N/6     N/6     N/6     N/6     N/6     N/6       SUSUC     255     N/6     N/6     N/6     N/6     N/6     N/6       SUSUC     277     N/6     N/6     N/6     N/6     N/6       SUSUC     273     N/6     N/6     N/6     N/6     N/6       SUPHC     281     N/6     N/6     N/6     N/6     N/6       SUPHC     281     N/6<                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A N/A                  | M/ A          | N/A             | 1.16-06      | 1.05+01         | N/A           | N/A             | N/A          | N/A             | N/A          | 3/6             | N/A          | Ă              |
| Streff         2bs         N.A         N.A<                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A N/A                  | N/A           | N/A             | 9.56-07      | 2.06+01         | N/A           | A/A             | M/A          | N/A             | N/A          | N:A             | N/A          | N/A            |
| SLIVE     264     M.A     M.A     M.A     M.A     M.A       SLAVC     265     M.A     M.A     M.A     M.A     M.A       SLSF     271     N.A     M.A     M.A     M.A     M.A       SLSF     271     N.A     M.A     M.A     M.A     M.A       SLSF     271     N.A     M.A     M.A     M.A     M.A       SLSF     273     N.A     M.A     M.A     M.A     M.A       SLSF     274     N.A     M.A     M.A     M.A       SLPHE     281     N.A     M.A     M.A </td <td>N/A N/A</td> <td>N/A</td> <td>M/A</td> <td>1.16-09</td> <td>2.95+01</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>A/A</td> <td>N/A</td> <td>N/A</td> <td>A, A</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | N/A N/A                  | N/A           | M/A             | 1.16-09      | 2.95+01         | N/A           | N/A             | N/A          | N/A             | A/A          | N/A             | N/A          | A, A           |
| SLUC         265         N/A         N/A <td>N/A N/A</td> <td>N/A</td> <td>N/A</td> <td>5. 36-04</td> <td>5. 5E + 00</td> <td>N/A</td> <td>N/A</td> <td>A/A</td> <td>8/N</td> <td>A. N</td> <td>A, A</td> <td>e.N</td> <td>4/W</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | N/A N/A                  | N/A           | N/A             | 5. 36-04     | 5. 5E + 00      | N/A           | N/A             | A/A          | 8/N             | A. N         | A, A            | e.N          | 4/W            |
| SLEAF     271     N/A     N/A     N/A     N/A     N/A       SLEAF     277     N/A     N/A     N/A     N/A     N/A       SLEAF     275     N/A     N/A     N/A     N/A       SLEAF     275     N/A     N/A     N/A     N/A       SLEAF     291     N/A<                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | N/A N/A                  | N/A           | N/A             | 1.4E-04      | 8.66+00         | N/A           | A/A             | A/A          | N/A             | N/A          | N/A             | N/A          | N/A            |
| 5154     272     N/A     N/A     N/A     N/A       5154     273     N/A     N/A     N/A     N/A     N/A       5154     275     N/A     N/A     N/A     N/A     N/A       5154     278     N/A     N/A     N/A     N/A       5154     281     N/A     N/A     N/A       5154     281     N/A     N/A     N/A       5154     2813     N/A     N/A     N/A       5154     2813     N/A     N/                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | N/A N/A                  | N/A           | N/A             | R/A          | N/A             | ¥/¥           | N/A             | N/A          | N/N             | 2.76-04      | 9.6E •00        | N/A          | A/A            |
| 51.54     273     N/A     N/A     N/A     N/A       51.54     275     N/A     N/A     N/A     N/A       51.14     276     N/A     N/A     N/A     N/A       51.14     281     N/A     N/A     N/A     N/A       51.14     281     N/A     N/A     N/A     N/A       51.14     283     N/A     N/A     N/A     N/A       51.14     284     N/A     N/A     N/A     N/A       51.14     284     N/A     N/A     N/A     N/A       51.14     284     N/A     N/A     N/A     N/A       51.14     2813     N/A     N/A     N/A     N/A       51.14     2814     N/A     N/A     N/A       51.14     2813     N/A     N/A     N/A       51.14     2814     N/A     N/A     N/A       51.14     2815     N/A <td< td=""><td>N/A N/A</td><td>A/A</td><td>A/A</td><td>N/A</td><td>A/A</td><td>N/A</td><td>N/A</td><td>N/A</td><td>N/A</td><td>8. 3E - 06</td><td>7. IE+00</td><td>N/A</td><td>NA</td></td<>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | N/A N/A                  | A/A           | A/A             | N/A          | A/A             | N/A           | N/A             | N/A          | N/A             | 8. 3E - 06   | 7. IE+00        | N/A          | NA             |
| 21.5     1.5     N.5     N.6     N.6     N.6     N.6       21.5     N.5     N.6     N.6     N.6     N.6     N.6       21.5     N.6     N.6     N.6     N.6     N.6     N.6       21.5     N.7     N.6     N.6     N.6     N.6     N.6       21.6     N.6     N.6     N.6     N.6     N.6     N.6       21.6     N.7     N.7     N.7     N.7     N.7     N.7       21.6     N.7     N.7     N.7     N.7     N.7     N.7       21.6     N.7     N.7     N.7     N.7     N.7       21.6     21.6     N.7     N.7     N.7 <td>N/A N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>3.16-05</td> <td>9.35+0)</td> <td>N/A</td> <td>N/A</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | N/A N/A                  | N/A           | N/A             | N/A          | N/A             | N/A           | N/A             | N/A          | N/A             | 3.16-05      | 9.35+0)         | N/A          | N/A            |
| Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave         Ave         Ave         Ave           Ave         Ave         Ave         Ave         Ave                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | ANN ANN                  | N/A           | N/A             | K/A          | N/A             | N/A           | N/A             | A/4          | N'A             | 1.96-06      | 1.16+01         | 1/F          | N/ 4           |
| SL-5F     276     N/A     N/A     N/A     N/A     N/A       SL-1FF     281     N/A     N/A     N/A     N/A     N/A       SL-1FF     281     N/A     N/A     N/A     N/A     N/A       SL-1FF     283     N/A     N/A     N/A     N/A     N/A       SL-1FF     281     N/A     N/A     N/A     N/A     N/A       SL-1FF     289     N/A     N/A     N/A     N/A     N/A       SL-1FF     289     N/A     N/A     N/A     N/A     N/A       SL-1FF     289     N/A     N/A     N/A     N/A       SL-1FF     2811     N/A     N/A     N/A     N/A       SL-1FF     2811     N/A     N/A     N/A     N/A       SL+1FF     2814     N/A     N/A     N/A     N/A       SL+1FF     2814     N/A     N/A     N/A     N/A       SL+1FF     2814     N/A     N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | N/A N'S                  | H/A           | N/A             | N/A          | N/A             | N/G           | A/A             | A/A          | M/A             | 7.46-07      | 3.45+01         | 878          | A'N            |
| 5:1+H     281     N/A     N/A     N/A     N/A       5:1+H     282     N/A     N/A     N/A     N/A       5:1+H     293     N/A     N/A     N/A     N/A       5:1+H     293     N/A     N/A     N/A     N/A       5:1+H     293     N/A     N/A     N/A     N/A       5:1+H     295     N/A     N/A     N/A     N/A       5:1+H     291     N/A     N/A     N/A       5:1+H     2813     N/A     N/A     N/A       5:1+H     2814     N/A     N/A     N/A       5:1+H     2815     N/A     N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | N/A N/A                  | N/A           | N/A             | N/A          | N/A             | A/A           | N/A             | A/N          | ¥/N             | 4.86-08      | 2.86+01         | N/A          | N/6            |
| SLIVI         282         N.A         N.A         N.A         N.A         N.A           SLIVIF         293         N/A         Y/A         N/A         N/A         N/A           SLIVIF         293         N/A         Y/A         N/A         N/A         N/A           SLIVIF         293         N/A         N/A         N/A         N/A         N/A           SLIVIF         284         N/A         N/A         N/A         N/A         N/A           SLIVIF         289         N/A         N/A         N/A         N/A         N/A           SLIVIF         289         N/A         N/A         N/A         N/A         N/A           SLIVIF         289         N/A         N/A         N/A         N/A         N/A           SLIVIF         2891         N/A         N/A         N/A         N/A         N/A           SLIVIF         2813         N/A         N/A         N/A         N/A         N/A           SLIVIF         2814         N/A         N/A         N/A         N/A         N/A           SLIVIF         2814         N/A         N/A         N/A         N/A         N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N/A N/A                  | A/A           | N/A             | N/A          | N/A             | #/A           | N.A             | M/A          | N/A             | A.A          | N/A             | 1, 36 - 97   | 1.25+0         |
| CLUME         293         N/A         M/A         N/A         N/A </td <td>N/A N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N.A</td> <td>N/A</td> <td>N/A</td> <td>M/A</td> <td>N/A</td> <td>N/A</td> <td>6. JE - CS</td> <td>8, 8E + 0</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N/A N/A                  | N/A           | N/A             | N/A          | N/A             | N.A           | N/A             | N/A          | M/A             | N/A          | N/A             | 6. JE - CS   | 8, 8E + 0      |
| SLIPHC         284         N/F         Y/A         N/A         N/A<                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N:A N/A                  | N/A           | N/A             | N/A          | N/A             | N/A           | A/A             | R/A          | N.A             | M/A          | N/A             | 1.95-07      | - 36           |
| STPHE         285         N         N/A         N/A <td>N/A N/A</td> <td>N/A</td> <td>R/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>A.A</td> <td>N/A</td> <td>3.16-10</td> <td>3.15-0</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A N/A                  | N/A           | R/A             | N/A          | N/A             | N/A           | N/A             | N/A          | N/A             | A.A          | N/A             | 3.16-10      | 3.15-0         |
| Quantum         Quantum <t< td=""><td>N/A N/A</td><td>N/A</td><td>N/A</td><td>N/A</td><td>M/A</td><td>N/A</td><td>N/A</td><td>N/A</td><td>N/A</td><td>N/A</td><td>N/A</td><td>3.16-10</td><td>3.16+0</td></t<>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | N/A N/A                  | N/A           | N/A             | N/A          | M/A             | N/A           | N/A             | N/A          | N/A             | N/A          | N/A             | 3.16-10      | 3.16+0         |
| SLRH         297         N/A         N/A <td>N/A N/A</td> <td>A/A</td> <td>A/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>A/A</td> <td>N, A</td> <td>N/A</td> <td>N/A</td> <td>NEGL</td> <td></td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A N/A                  | A/A           | A/A             | N/A          | N/A             | N/A           | N/A             | A/A          | N, A            | N/A          | N/A             | NEGL         |                |
| SLIPIC         288         N/A         N/A<                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A N/A                  | N/A           | NIA             | N/A          | N/A             | A/K           | N/A             | N/A          | N/A             | A/A          | N/A             | 8.5E-10      | 5.66.0         |
| SLENF         281         N.A         N.A </td <td>N/A N/A</td> <td>M/A</td> <td>N/A</td> <td>A/A</td> <td>A/A</td> <td>N/6</td> <td>A'A</td> <td>A/A</td> <td>H/A</td> <td>N/A</td> <td>R/A</td> <td>NEGL</td> <td></td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | N/A N/A                  | M/A           | N/A             | A/A          | A/A             | N/6           | A'A             | A/A          | H/A             | N/A          | R/A             | NEGL         |                |
| SUPPE         2010         N/A         N/A<                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A N/A                  | N/A '         | N/A             | N/A          | A/A             | N/A           | N/A             | N/4          | N/A             | N/A          | d / N           | 1934         |                |
| SLPH         2811         N/A         N/A </td <td>N/A N/A</td> <td>N/A</td> <td>N/A</td> <td>A/A</td> <td>A.A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/A</td> <td>N/4</td> <td>N/A</td> <td>1.46-05</td> <td>1.25</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A N/A                  | N/A           | N/A             | A/A          | A.A             | N/A           | N/A             | N/A          | N/A             | N/4          | N/A             | 1.46-05      | 1.25           |
| SLEME 2012 N/A 1/A N/A N/A N/A SLEME 2013 N/A N/A N/A N/A 3/A 2014 N/A N/A N/A 3/A 2014 N/A N/A N/A N/A 2014 N/A N/A N/A N/A N/A 2014 1/A | N/A N/A                  | N/8           | NrA             | N'A          | N/A             | A/A           | 8/N             | 4/H          | AIA             | N/A          | A N             | 2.96-05      | 7.56+0         |
| SLRHF 2813 N/A N/A N/A N/A N/A SLPHF 2814 N/A N/A N/A SLPHF 2814 N/A N/A N/A SLPHF 2815 N/A N/A N/A N/A N/A SLPHF 2010 V/A N/A N/A N/A N/A N/A N/A SLPHF 2010 V/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A N/A                  | N/A           | N/A             | A'A          | N/A             | N/A           | N/A             | N/A          | N/A             | N/A          | 4/4             | 1.25-97      | 9.25.6         |
| Stiff 2014 N/A N/A N/A N/A N/A Stiff 2015 N/A N/A N/A N/A Stiff 2015 N/A N/A N/A N/A N/A N/A Stiff 2010 N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | N/A N/A                  | NZA           | N.A             | N/A          | A/A             | 4/H           | N/A             | N/A          | N / A           | A/A          | N/A             | 7.6E-03      |                |
| Striff 2015 N.A. N.A. N.A. N.A. N/A<br>Striff 2016 V/A N/A N/A N/A<br>Striff 2018 V/A N/A N/A N/A<br>Striff 2017 V/A V/A N/A V/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A N/A                  | N/A           | N/A             | H/A          | N/A             | N/A           | N/A             | N/A          | A,A             | A/A          | N/A             | 6.95-06      | 7.75+6         |
| Steve 2016 4/A N/A N/A N/A N/A Steve 2017 5.00                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | NA NA                    | 8/N           | N : A           | k, A         | N.A             | 4<br>4        | N.A             | H/A          | A, N            | 11 - Ĥ       | 4.7             | 3.65 1"      | 1.<br>1.<br>1. |
| C. 101 1.0 1.0 1.0 1.0 1.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | N/A N/A                  | 8/8           | N/A             | R/A          | 4 : N           | ¥. ħ          | 47 N            | M/A          | 4 · X           | N / P        | 4:4             | 5.45-05      | <u>.</u><br>۲  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | N.A NIA                  | N/A           | N.'A            | 47N          | A N             | R/H           | N. 4            | M.A          | 4               | N/A          | 4.4             | 1.15-75      | 5. Ff          |

for the warehouses at UMDA. The accident sequence designations are also shown on the Fig. 5-11 event tree. For those accident sequences where no agent release occurs, the release scenario is labeled "NR." Those release scenarios whose frequency is below  $1.0 \times 10^{-10}$  for all sites have been screened using the frequency criterion labeled with an "F" in the event tree. The events modeled in Fig. 5-11 are discussed below:

 Earthquake Occurs. The initiating event (Event 1) in Fig. 5-11 is earthquake occurrence. To simplify the event tree evaluation, Event 1 further restricts the earthquake intensity to an acceleration range from gl (0.15 to 0.2 g) to gu (>0.7 g). Seven ranges are considered:

a. 0.15 to 0.2 g.
b. 0.2 to 0.3 g.
c. 0.3 to 0.4 g.
d. 0.4 to 0.5 g.
e. 0.5 to 0.6 g.
f. 0.6 to 0.7 g.

g. Greater than 0.7 g.

Earthquakes below 0.15 g are not considered in the analysis because the damage probabilities associated with such tremors are negligibly small. Detailed examination of seismic ranges above 0.7 g is unnecessary because earthquakes above 0.7 g have a probability of almost 1.0 of causing damage.

The initiating event frequency at each site is the sitespecific frequency at which earthquakes in the range gl to gu occur.

 <u>"K" Warehouses Damaged by Earthquake</u>. Warehouse damage is defined as structural collapse. This is the only failure mode of interest because it will crush stored ton containers.



Although less severe damage can result from an earthquake, it was screened in quantifying the Event 2 probability because it does not induce ton container failure.

Three damage combinations are considered in Event 2:

- a. No warehouses are damaged (K = 0).
- b. Only one warehouse is damaged (K = 1).
- c. Both warehouses are damaged (K = 2).

Tracking these three probabilities is necessary in order to estimate the agent release source term. Note that since there is only one warehouse at NAAP, the probability that K = 2 is zero for that site.

Event 2 damage probabilities are based upon a generic study of damage to structures designed to the Uniform Building Code.

3. <u>Munitions Damaged in "L" Warehouses</u>. Event 3 addresses whether the earthquake causes an agent release from the stored munitions. Two failure modes are analyzed: puncture and crushing.

Only ton containers are subject to these failures. Spray tanks are in overpacks which protect them from crush forces. Furthermore, they are not stacked while in storage, hence can't be punctured.

Three damage combinations are considered in Event 3:

- a. No agent releases result from the earthquake (L = 0).
- b. The earthquake causes an agent release in one warehouse (L = 1).



The puncture probability is the probability that at least one ton container falls and strikes a probe of sufficient size and density to penetrate it. The probability that ton containers are crushed is correlated to warehouse damage. If K is 0, 1, or 2 in Event 2, then ton containers in none, 1, or 2 warehouses are crushed, respectively. Since the NAAP site has only one warehouse, the probability that L = 2 is zero for that site. In addition, since only spray tanks are stored in the TEAD warehouses, L can only be zero at that site.

4. <u>Ignition at "M" Warehouses</u>. Seismically initiated fires are an important consideration because they influence agent dispersion and can thermally fail agent containers. This second aspect is particularly important at TEAD because fire damage is the only spray tank container failure mode.

Electrical fires are the only concern in warehouses. The three conditions necessary for an electrical fire are:

- a. An electrical fault capable of causing arcing.
- b. A supply of electric power to sustain the arc.
- c. Contact with an ignition source.

Including this second condition in the fire ignition probability calculation is important because available data indicate that offsite power can be lost at a relatively low seismic intensity.

Condition three considers both the agent and wood dunnage assemblies as possible ignition sources in the warehouses. If ton containers have been damaged by either crush or puncture, the probability of igniting spilled agent given an electrical arc has occured is essentially unity. If no munition damage has occurred, the probability of ignition is represented as the ratio of exposed wood surface area to the total area of the warehouse.

Similar to previous events, Event 4 addresses how many warehouses experience ignition.

5. Ignition at Warehouse With Damaged Munitions. If the earthquake only damages the containers stored in one warehouse and ignition occurs at only one warehouse, it is necessary to discern whether the fire is in the warehouse with the damaged containers. If the fire is in the same warehouse as the damaged containers, thermal failure and the subsequent release of agent from the second warehouse is averted. However, if the damaged containers and fire are in different warehouses, then the agent release source term will be increased.

Suppression of fires has a negligible probability since the warehouses have no fire alarms nor automatic fire suppression systems. For this reason it is not considered in the warehouse analysis.

### 5.2.5. Lightning

Munitions stored in igloos and warehouses are protected from lightning. Hence, only ton containers stored outdoors at APG, PBA, and TEAD may be susceptible to lightning strikes. No event tree model has been developed for this scenario. Basically, if a lightning strikes a ton container, the container will be breached and agent will spill to the ground.

A lightning strike density for the contiguous United States was previously determined (Ref. 5-12) based on the correlation developed from the duration of thunderstorms. Based on this empirical correlation, the frequency (events/yr-km<sup>2</sup>) for the different storage locations has been determined, as shown in Table 4-7.

Using conservative assumptions, a threshold lightning energy required to burn through the ton container wall was found to be proportional to the fourth power of the wall thickness as described in the calculation sheets (Ref. 5-2). Neglecting corrosion thinning of the container wall, the maximum value of failure frequency for each cluster of 15 ton containers at PBA is  $5.1 \times 10^{-10}$ , as shown in Table 5-16.

The results indicate that the threshold lightning energy required to burn through the container wall is a strong function of wall thickness. In order to assess the sensitivity of the failure frequency to corrosion, a probability density function for wall thickness was derived by conservatively assuming that one ton container stored outdoors has a leak through its wall. This is a conservative assumption since no wall leak has been reported. This probability density function for wall thickness is used in conjunction with the lightning energy requirements to calculate the failure frequency of a cluster of 21 containers at the different sites. As expected for the PBA site, the failure probability is increased by approximately 55 from the previous value of  $5.1 \times 10^{-10}$ .

If all other agent release scenarios have frequencies that are below this bounding value, then the extent of container corrosion must be investigated. However, if other scenarios involving comparable or larger amounts of agent release also have frequencies much higher than the bounding value for the lightning initiated release, then lightning release scenarios can be ignored. This is true for aircraft crash accidents which lead to much larger releases and also higher frequencies for some sites.







# TABLE 5-16 SITE-SPECIFIC LIGHTING STRIKE INFORMATION

| Projecte<br>Area for<br>Each Clus<br>21 Contai<br>(km)<br>2.5 x 10<br>2.5 x 10                                                                                                                                                                                                                                                |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ITE-SPECIFIC LIGHTI<br>Ground Density<br>[1]<br>Event/Yr/km <sup>2</sup> (<br>N <sub>1</sub><br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>3<br>3<br>2<br>2                                                                                                                                                                                                                                                                      |
| S]<br>Name of Site<br>Aberdeen Proving Ground (APG)<br>Anniston Army Deport (ANAD)<br>Laxington - Blue Grass Army Depot<br>(LBAD)<br>Laxington - Blue Grass Army Depot<br>(LBAD)<br>Newport Army Ammunition Depot<br>(LBAD)<br>Newport Army Ammunition Depot<br>(NAAP)<br>Pine Bluff Arsenal (PBA)<br>Pine Bluff Arsenal (PBA)<br>Pueblo Depot Activity (PUDA)<br>Tooele Army Deport (TEAD)<br>Umatilla Depot Activity (UMDA) |

1111

### 5.2.6. <u>Floods</u>

During a flood, materials such as lumber, crates, storage tanks, and other lightweight containers may be carried away by flood flows and cause damage to downstream structures. Water velocities during floods depend largely on the size and shape of the cross sections, conditions of the stream, and the slope bed, all of which vary on different streams and at different locations. In the upper reaches of a flood basin, main channel flows could be as high as 14 ft/s, but typical overbank flow is less than 2 ft/s (Ref. 5-13).

Munitions stored in igloos and warehouses are considered protected against flood-generated projectiles. The only munition stored outdoors are mustard-filled ton containers (APG, PBA, and TEAD).

The puncture equation is as follows:

$$V_{\rm m}^2 = \{64 \ (672 \ DT)^{3/2}\}/W$$
, (5-5)

where D = probe diameter (in.),

- T = wall thickness to be punctured (in.),
- W = weight of projectile (i.e., moving object) (1b),
- $V_m$  = velocity of projectile (ft/s).

The wall thickness of the ton container is 0.41 in. Assuming the smallest probe size is 0.8-in. in diameter,

 $V_m^2$  (W) = (64)(672 DT)<sup>3/2</sup> = 217,335

For puncture, the following conditions must be met:

| v <sub>m</sub> | W           |
|----------------|-------------|
| (ft/s)         | <u>(1b)</u> |
| 1              | 217,335     |
| 2              | 53,334      |
| 6              | 6,037       |
| 10             | 2,173       |
| 14             | 1,108       |

A credible flood-generated projectile is assumed to be a light, steel tank with a rigidly attached 0.8-in. diameter probe. This could be a water storage tank or a gasoline tank, using a tank height to diameter ratio of 1.2 and a wall thickness of 0.25 in. Table 5-17 presents the data developed for steel tanks. Tanks larger than 10 ft in diameter would not be credible except in main channel flows. Thus, typical overbank flows, i.e., 2 ft/s, would not produce puncture.

Puncture could be initiated by using an extreme overbank velocity of 6.13 ft/s combined with a 10-ft diameter floating tank with a rigidly attached 0.8-in. probe. The probability of a 6.13 ft/s overbank velocity is estimated to be less than 10%. This condition will be designated as the reference flood-generated projectile.

The probability of puncture of a single ton container from the reference single floating tank condition is as follows:

$$P_{\rm F} = L_{\rm p} \times T_{\rm p} \times P_{\rm p} \quad , \tag{5-6}$$

where  $L_p$  = location probability, i.e., the probability that the probe attached to the floating tank is pointing towards the ton container wall at the moment of collision,

PREMAMENTAL AND PRESSOR

うちゅうかいて ましていたいかん

| D<br>Diameter<br>(ft) | 1.2D<br>Height<br>(ft)                | 57.67D <sup>2</sup><br>Weight<br>(1b) | 5.3407D <sup>2</sup><br>Surface Area<br>(ft <sup>2</sup> ) |
|-----------------------|---------------------------------------|---------------------------------------|------------------------------------------------------------|
| 2                     | 2.4                                   | 231                                   | 21.36                                                      |
| 4                     | 4.8                                   | 923                                   | 84.45                                                      |
| 6                     | 7.2                                   | 2076                                  | 192.0                                                      |
| 8                     | 9.6                                   | 3690                                  | 342.0                                                      |
| 10                    | 12.0                                  | 5767                                  | 534.0                                                      |
| <del></del>           | · · · · · · · · · · · · · · · · · · · | <u></u>                               | ······                                                     |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       |                                       |                                                            |
|                       |                                       | 5-64                                  |                                                            |
|                       | -                                     |                                       |                                                            |
|                       |                                       |                                       |                                                            |

### TABLE 5-17 PROBABLE SIZE DISTRIBUTION FOR STEEL TANKS

22222222

 $T_p$  = target probability, i.e., the probability that the tank collides with the ton container,

CARACTER AND REACT A

 $P_p$  = probability of probe being present.

 $L_p$  can be approximated by the ratio of total surface area to the effective surface position. Assuming that the probe must be within a 1 ft<sup>2</sup> location, then:

 $L_p = 1/(7.06)^2 (5.3407) = 0.0038$ 

 $T_p$  can be approximated by assuming a flood channel width at the point of collision and comparing that to the length of a ton container (82 in.). Using a three-mile wide channel, which is conservative for a typical flood, then:

 $T_p = 82/\{(5280) (12) (3)\} = 0.00043 \text{ or } 0.0043$ 

for the total width of 10 containers.

 $P_p$  is estimated to be 1 x 10<sup>-3</sup>. Thus the probability of a reference tank hitting and rupturing a ton container is

 $P_F = (0.0038) (0.0043) (0.001) = 1.6 \times 10^{-8}$ 

It would seem reasonable from the flood basin size to assume no more than one reference floating projectile per flood and the flood reoccurrence to be greater than 100 years. In addition, the probability of a 6 ft/s overbank velocity is estimated as 10%. Thus, the probability of rupture is approximately 1.63 x  $10^{-11}$ /yr.

Thus, based on the above calculations this scenario can be screened out on the basis that its frequency is below the criterion.

N/X/X/0-R

### 5.3. SPECIAL HANDLING ACTIVITIES

### 5.3.1. Leaking Munitions

22359

Several scenarios were identified that specifically address the leakage of stored munitions and the accidents that could occur in the process of isolating leaking munitions which could aggravate the existing situation. The event trees are shown in Figs. 5-12 and 5-13.

LY COXA

Sequence SL1 addresses the possibility that a munition could leak from the time the periodic inspection has been performed until the next periodic inspection. It is assumed that the leaking munition will be detected at the time the next inspection is made. For all sites, except at APG, the inspections are assumed to be performed quarterly (90 days). At APG, the ton containers are inspected daily. No event tree was developed for this scenario since it is represented by a single event failure.

Sequences SL2 and SL9 address accidents related to the movements of munitions for inspection or isolation of leakers. The forklift time puncture or drop of munition was determined to be largely due to human error. The quantification of these events required a detailed human reliability study (Ref. 5-14). Essentially a task analysis was performed to identify those errors that could potentially impact agent release probabilities. Available data was used to quantify the probabilities of some of these errors and extrapolations were made from these fixed data to quantify the remainder.

Isolation of leaking rockets require special tasks. The leaking rockets are isolated in the storage igloo at the original location, where the pallet containing the leaking rocket is unpacked. Only those rockets blocking access to the leaking rocket are removed and are placed in a holding fixture. This rocket is hand-carried by a two-man team wearing Level A protective clothing to the PIG (which has been placed

| FORKLIFT TINE<br>ACCIDENT<br>(INSIDE IGLOO) | MUNITION<br>Integrity | AGENT<br>RELEASE<br>SEQUENCE |
|---------------------------------------------|-----------------------|------------------------------|
|                                             | INTACT                |                              |
|                                             | PUNCTURED             | SL2                          |

DCOC AA

i

Fig. 5-12. Munition punctured by forklift time during leaker - handling activities

and the first of the state of the

| MUNITION<br>DROPPED INSIDE<br>IGLOO | MUNITION<br>Integrity | AGENT<br>RELEASE<br>SEQUENCE |
|-------------------------------------|-----------------------|------------------------------|
|                                     | INTACT                | NB                           |
|                                     | DETONATED             | SL25                         |
|                                     | PUNCTURED             | SL9                          |
|                                     |                       |                              |

Ŀ

22

DESERVED DESERVE DERENER REALERS

Š.



on a plastic sheet) and secured in it. The handlers lift the decontaminated PIG by its handles, carry it outside, and place it on the truck that will carry it to an igloo reserved for leaking munitions (Ref. 5-1). The analysis assumes that the same procedure is followed for isolating other leaking munitions, except that overpacks (other than PIGs) are used.

Three types of operator errors related to leaker isolation were identified in the task analysis: (1) puncturing a munition with a forklift tine, (2) dropping a munition or pallet from a forklift, and (3) dropping a single munition while hand-carrying it. Details on these handling errors are discussed in Section 6 (Handling Activities).

A previously identified scenario involving the improper replacement of a corroded valve or plug in a ton container (Sequence SL16, Ref. 5-15), has been deleted in the present evaluation. It is expected that few ton containers with GB will require that their valves be replaced before collocation and disposal are initiated. The human reliability analysis (see Appendix J) concluded that this event has a low frequency of occurrence. Furthermore, the amount of mustard or VX that could be dispersed to the atmosphere from a valve or plug replacement operation is insignificant.

Table 5-18 presents the data used to evaluate the accident frequencies for the scenarios addressed above. The frequency of scenario SL1 was derived by determining the leakage rate for each munition type based on the leaker data at each site and the total munition inventory at each site. Since the two parameters are classified information, they will be presented and discussed further in a classified appendix.



NAME SERVICE DEPOSITION

TABLE 5-18 DATA BASE FOR ANALYSIS OF SEQUENCES SL1, SL2, AND SL9

CULCULA C

1224-2020

11111

3-3-1-5-3

لز الأحداث

| Event                                                     | Frequency<br>of Probability                                                                 | Reference |
|-----------------------------------------------------------|---------------------------------------------------------------------------------------------|-----------|
| Munition develops a leak during storage (Scenario SL1):   |                                                                                             |           |
| Bomb (TEAD)<br>(UMDA)                                     | 7.5E-5 per year Ref<br>4.5E-4 per year                                                      | . 5-16    |
| 4.2-in. mortar (ANAD)<br>(PUDA)<br>(TEAD)                 | 2.8E-7 per year<br>1.0E-6 per year<br>7.0E-6 per year                                       |           |
| 105-mm cartridge (ANAD)<br>(PUDA)<br>(TEAD)               | 2.8E-7 per year<br>1.0E-6 per year<br>7.0E-6 per year                                       |           |
| Ton container                                             |                                                                                             |           |
| Mine (ANAD)<br>(PBA)<br>(TEAD)<br>(UMDA)                  | 9.0E-6 per year<br>1.1E-6 per year<br>2.5E-4 per year<br>3.1E-4 per year                    |           |
| Projectile (ANAD)<br>(LBAD)<br>(PUDA)<br>(TEAD)<br>(UMDA) | 4.9E-6 per year<br>9.3E-6 per year<br>5.0E-6 per year<br>8.1E-5 per year<br>6.2E-5 per year |           |
| Rocket (ANAD)<br>(LBAD)<br>(PBA)<br>(TEAD)<br>(UMDA)      | 6.1E-5 per year<br>4.3E-5 per year<br>9.1E-7 per year<br>1.3E-3 per year<br>1.8E-4 per year |           |
| Spray tank                                                | 9.8E-5 per year                                                                             |           |
| Forklift tine accident (SL2)                              | 1.0E-4 per operator Ref                                                                     | . 5-15    |
| Munition puncture given tine accident:                    |                                                                                             |           |
| Bomb                                                      | 1.29E-2 Ref                                                                                 | . 5-2     |
| 4.2-in. mortar                                            | 3.68E-2                                                                                     |           |
| 105-mm cartridge                                          | 8.90E-3                                                                                     |           |
| Mine                                                      | 7.07E 2                                                                                     |           |



### TABLE 5-18 (Continued)

Ş

| Event                                              | Frequency<br>of Probability | Reference            |
|----------------------------------------------------|-----------------------------|----------------------|
| Projectile                                         | 5.00E-2                     |                      |
| Rocket                                             | 2.63E-1                     |                      |
| Spray tank                                         | 1.53E-2                     |                      |
| Munition dropped during leaker<br>isolation (SL9): |                             |                      |
| Pallet and bulk (B, S)                             | 3.0E-4                      | Human Reliability    |
| Single (C, D, M, P, Q, R)                          | 6.0E-4                      | Analysis (Ref. 5-15) |
| Ton container (K)                                  | 3.0E-5                      |                      |
| Munition punctured given drop:                     |                             |                      |
| Bomb (pallet)<br>(single)                          | 4.72E-4<br>1.62E-4          | Ref. 5-2             |
| 4.2-in. mortar (pallet)<br>(single)                | 1.24E-4<br>0.0              |                      |
| 105-mm cartridge (pallet)<br>(single)              | 2.71E-5<br>0.0              |                      |
| Ton continer                                       | 1.55E-3                     |                      |
| Mine (pallet)<br>(single)                          | 9.27E-5<br>4.08E-5          |                      |
| <b>Projectile (pallet</b> or single)               | 0.0                         |                      |
| Munition detonates given 6 ft<br>drop              | 1.6E-8/munition             | Ref. 5-2             |





### 5.4. SCENARIO QUANTIFICATION

Tables 5-19 and 5-20 present the results of the accident scenario frequency analysis for all the storage sequences discussed previously except those which were initially screened (i.e., SL10, SL11, SL12, SL13, and SL14). From the results it is evident that the following sequences could be screened out further based on the 1.0 x  $10^{-10}/yr$  criterion:

NEW CONTRACTOR OF THE AND A CONTRACTOR OF THE ACCOUNT OF

- SL17 Large aircraft direct crash; fire contained in 30 min.
- SL21 Large aircraft indirect crash; fire contained in 30 min.
- SL23 Tornado-generated missiles cause munition detonation upon impact.
- SR7, SA7, SW7 Tornado-generated missile penetrate munitions in OFCs (SR, SA) or barge packages (SW); no detonations occur.
- SR8, SA8 Tornado-generated missile penetrates munitions in OFCs and cause munition detonations.

Since handling-related accidents are given in terms of events per munition operation, no screening can be performed without divulging classified information.

AND CLOUDED IN

File: CONSTOR.WK1 date21-Aug-87 page 1

a alata a second

TABLE 5-19 FREQUENCIES OF STORAGE ACCIDENT SEQUENCES (SL)

STORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

Accident Frequencies

RANGE FACTOR

UMDA Fred

RANGE Factor

TEAD Freq

RANGE Factor

PUDA Freq

RANGE Factor

PRA Freg

RANGE Factor

NAAP Freq

RANGE Factor

LBAD Fred

FACTOR

AFG FREQ

RANGE FACTOR

AKAD Freq

ġ.

SCENARIO

| N/A<br>55-07 1.01<br>55-07 1.01 | •         |           |         |            |           |            |         |          |         |            |         |          |         |             |        |
|---------------------------------|-----------|-----------|---------|------------|-----------|------------|---------|----------|---------|------------|---------|----------|---------|-------------|--------|
| ю-1<br>1                        | ۱         | N/A       | •       | N/A        | ,         | N/A        | '       | N/A      | '       | A/A        | •       | 7.56-05  | 1.0E+01 | 4.56-04 1.0 | 9      |
| 4                               | 10+3      | N/A       | ,       | N/A        | •         | N/A        | •       | N/A      | '       | 1.0E-06    | 1.0E+01 | 7.0E-06  | 1.06+01 | N/A         |        |
|                                 | E+01      | N/A       | ۰       | N/A        | •         | <b>N/A</b> | ı       | M/M      | '       | A/N        | 1       | 7.0E-06  | 1.0€+01 | • N/A       |        |
| 1.0                             | E+01      | N/A       | •       | N/A        | ,         | N/A        | •       | N/A      | •       | 1.06-06    | 1.0E+01 | N/A      | ۱       | N/A         |        |
|                                 | •         | N/A       | •       | N/A        | •         | N/A        | •       | N/N      | •       | A/A        | •       | 1.96-04  | 1.0E+01 | N/A         |        |
| 1.01                            | E+01      | N/A       | •       | A/A        | •         | N/A        | •       | N/A      | •       | N/A        | '       | A/N      | ٠       | N/A         |        |
|                                 | t         | H/A       | ,       | N/A        | ,         | A/A        | 1       | 5.96-06  | 1.0E+01 | N/N        | •       | 5.95-06  | 1.05401 | N/A         |        |
|                                 | - 5.9     | E-06 1.0  | E+01    | A/N        | •         | N/A        | '       | N/A      | •       | N/N        | '       | N/A      | •       | N/A         |        |
| _                               | •         | N/A       | ,       | N/A        | '         | N/A        | ı       | N/A      | '       | N/A        | ١       | N/A      | ,       | 5.96-06 1.0 | 0530   |
| æ                               | •         | N/A       | •       | N/A        | •         | N/A        | •       | N/A      | '       | N/A        | •       | 5.9E-06  | 1.05+01 | 6/N         |        |
| A                               | •         | N/A       | •       | N/A        | •         | 5.9E-(\b   | 1.0E+01 | N/A      | •       | N/A        | •       | N/A      | '       | N/A         |        |
| 0.1 40                          | E+01      | N/A       | •       | N/A        | •         | N/A        | •       | 1. IE-06 | 1.0E+01 | N/A        | •       | 2.5E-04  | 1.05+01 | 3.1E-04 1.0 | 0F 30  |
| -09 1.01                        | E+01      | N/A       | •       | N/A        | ,         | N/A        | '       | N/A      | •       | N/A        | •       | 8.1E-05  | 1.0E+01 | 6.2E-05 1.0 | Ú÷ B   |
| -06 1.01                        | E+01      | N/A       | 1       | 9. JE-06 1 | 1.0E+01   | N/A        | •       | N/A      | •       | 5.0E-06    | 1.0E+01 | 8.1E-05  | 1.0E+01 | N/A         |        |
| -06 1.01                        | 10+3      | N/A       | •       | 9.3E-06 1  | 1.0E+01   | N/A        | I       | N/A      | •       | N/A        | ı       | 8. 1E-05 | 1.0E+01 | 6.2E-05 1.0 | ÷      |
| -06 1.01                        | E+01      | N/A       | 1       | 9. JE-06 1 | 1.05+01   | N/A        | I       | N/A      | •       | N/N        | '       | B. 1E-05 | 1.0£+01 | 6.2E-05 1.0 | (if gi |
| /A                              | •         | N/A       | ,       | N/A        | •         | N/A        | •       | N/A      | '       | N/A        | '       | 8.1E-(/5 | 1.05+01 | 6.2E-05 1.0 | j.     |
| -05 1.01                        | E+01      | N/A       | •       | 4. 3E-05 1 | 1.0E+01   | N/A        | 1       | 9.1E-07  | 1.0€+01 | N/A        | '       | 1.75-03  | 1.05+01 | 1.85-04 1.0 | ц.     |
| -05 1.01                        | E+01      | N/A       | ,       | 4. JE-05 1 | 1.0E+01   | N/A        | ·       | 9.1E-07  | 10+30.1 | N/A        | '       | 1.36-03  | 1.05+01 | 1.86-04 1.0 | ÷      |
| /A                              | ,         | N/A       | ,       | N/A        | 1         | A/A        | '       | N/A      | •       | N/A        | ı       | N/A      | ,       | 9.86-05 1.0 | Ξ      |
| ۲ <b>۹</b>                      | •         | N/A       | ı       | N/A        | r         | N/A        | ,       | N/A      | •       | N/A        | '       | 9.85-05  | 1.05+01 | N/A         |        |
| by fork!                        | lift tine | during le | aker ha | ndling ac  | ctivities |            |         |          |         |            |         |          |         |             |        |
| /A                              | ۰         | N/A       | ı       | N/A        | ,         | N/A        | '       | N/A      | 1       | A/N        | •       | 5.2E-06  | 1.35+01 | 5.26-06 1.3 | Ĩ.     |
| -05 1.31                        | E+01      | N/A       | ł       | N/A        | •         | N/A        | 1       | N/A      | ſ       | 4.4E-05    | 1.35+01 | 4.4E-05  | 1.35401 | N/A         |        |
| -05 1.31                        | E+01      | N/A       | ,       | N/A        | ,         | N/A        | •       | N/A      | •       | N/A        | •       | 1.16-05  | 1.3£+01 | N/A         |        |
| -05 1.31                        | E+01      | N/A       | ,       | N/A        | •         | N/A        | '       | N/A      | '       | 1. IE -(15 | 1.36+01 | N/A      | •       | N/A         |        |
| æ                               | ,         | N/A       | ,       | N/A        | ,         | N/A        | ,       | N/N      | ı       | N/A        | •       | 0.0E+00  | '       | N/A         |        |

See notes at end of table.

TABLE 5-19 (Continued)

File: CONSTOR.WK1 date21-Aug-87 page 2

STORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EXISING SITES

Accident Frequencies

|   | FAC TOR         |         | •          | '       | 1. 35+01 | 1.35+01 | '       | 1.36+01    | 1.35+01    | 1.35+01    | 1.36+01   | 1.35+01   | 1. 3E+01        |               | 1. (E+01       | •            | ,            | •            | •            | '            | '            |              | '            | '            | ,            |              |              | r          | 1.05+01 | '            |
|---|-----------------|---------|------------|---------|----------|---------|---------|------------|------------|------------|-----------|-----------|-----------------|---------------|----------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|------------|---------|--------------|
|   | UMDA<br>FREQ    | A 05.00 | 0. UE + UU | N/A     | 8.56-05  | 6.0E-05 | N/A     | 6. OE -1)5 | 6.0E-US    | 6.0E-05    | 3.26-04   | 3. 25-04  | 4°-39' <b>1</b> |               | 4.1E-10        | N/A          | N/A          | N/A          | N/A          | A/A          | e/N          | N/A          | N/A          | A:A          | N/A          | N/N          | N.A          | K/A        | 1.16-08 | 6/N          |
|   | FACTOR          |         | •          | •       | 1.35+01  | 1.36+01 | 10+31.1 | 1. JE +01  | 1. XE +01  | 1.36+01    | 1.36+01   | 1.36+01   | 1.3€+01         |               | 1.0€+01        | 1.0E+01      | ,            | 1.0€+01      | 10+30.1      | ,            | •            | •            | •            | 1.05+01      | 1,05+01      | 1.05+01      | •            | 1.05401    | •       | 1.05+01      |
|   | TEAD<br>Fred    | 0.00    | 0. VE 10U  | 0.0E+00 | 8.56-05  | 6.05-05 | 6.0E-05 | 6.0E-05    | 6.0E-05    | 6. 0E - 05 | 3.26-04   | 3. 2E-04  | 4.6E-06         |               | 9.8E-12        | 1. IE-11     | N/A          | 9.8E-12      | 1.1E-11      | A/A          | 9.85-12      | 1. IE - 11   | A/N          | A/N          | A/A          | 9.86-12      | N/A          | 3. 5E - US | N/A     | 9.8E-12      |
|   | FANGE<br>FACTOR |         | r          | 4       | 1        | •       | 1.36+01 | ٠          | •          | •          | •         | •         | •               | nt).          | '              | •            | ,            | 1.0E+01      | ı            | 1            | •            | •            | •            | 1.05+01      | •            | ł            | ,            | '          | •       | •            |
|   | PUDA<br>FREQ    |         | H/H        | N/N     | N/A      | N/A     | 6.0E-05 | N/A        | N/A        | N/A        | N/A       | N/A       | N/A             | onate if l    | N/A            | N/A          | N/A          | 1.6E-09      | A/A          | N/A          | N/A          | N/A          | N/A          | 1.6E-09      | N/A          | N/A          | N/A          | A/A        | N/A     | N/A          |
|   | RANGE<br>FACTOR |         | ۱          | •       | 1.36+01  | ٠       | •       | '          | •          | •          | 1. 3E +01 | 1.36+01   | ,               | tions det     | ı              | ı            | ,            | ı            | 1            | 1            | •            | •            | '            | •            | •            | '            | '            | 1.05+01    | 1       | •            |
|   | PBA<br>Freq     | 0 00    | 0. VE +UU  | N/A     | 8.56-05  | N/A     | N/A     | N/A        | N/A        | N/A        | 3.2E-04   | 3.2E-04   | N/A             | ered auni     | N/A            | N/A          | N/A          | N/A          | N/A          | N/A          | N/A          | N/A          | N/A          | N/A          | N/A          | N/A          | N/A          | 7.6E-09    | N/A     | N/A          |
| 2 | FACTOR          |         | ı          | •       | ı        | •       | •       | ı          | '          | •          | •         | '         | '               | es (burst     | '              | ı            | ,            | •            | •            | •            | '            | ١            | 1            | '            | ı            | •            | ,            | •          | 1       | '            |
|   | NAAP<br>Fred    |         |            | 0.0E+00 | N/A      | A/N     | N/A     | N/A        | N/A        | N/A        | N/A       | N/A       | N/A             | 30 minute     | N/A            | A/A          | N/A          | A/A          | N/A          | N/A        | N/A     | N/A          |
|   | RANGE<br>Factor |         | •          | •       | '        | •       | 1.36+01 | 1.35+01    | 1.36+01    | •          | 1.36+01   | 1.3E+01   | '               | tained in     | •              | •            | 1            | '            | '            | ı            | •            | 4            | ť            | '            | •            | '            | ı            | 1          | ł       | ۰            |
|   | L BAD<br>Freq   |         |            | N/A     | N/A      | N/A     | 6.06-05 | 6.0E-05    | 9° 0E - U2 | N/A        | 3.26-04   | 3. 2E -04 | N/A             | e not con     | N/A            | N/A          | N/A          | N/A          | N/A          | N/A          | A/A          | N/A          | N/A          | N/A          | N/A          | N/A          | A/A          | N/A        | N/A     | N/A          |
|   | RANGE<br>Factor |         | •          | •       | •        | •       | •       | •          | •          | '          | •         | •         | •               | area; fire    | •              | '            | •            | ı            | ı            | ,            | •            | 1            | ſ            | ı            | 1            | •            | '            | 1.0€+01    | •       | •            |
|   | AP6<br>Fred     | 0.00    | 0.01-00    | N/A     | N/A      | N/A     | N/A     | N/A        | N/A        | N/A        | N/A       | N/A       | N/A             | o storage     | N/A            | A/A          | N/A          | 1.1E-09    | N/A     | N/A          |
|   | RANGE<br>FACTOR |         | ,          | •       | 1.36+01  | 1.3E+01 | 1.36+01 | 1.36+01    | 10+35.1    | •          | 1.3E+01   | 1.35+01   | '               | crash onti    | ,              | ,            | 1.0E+01      | 10+30.1      | •            | 1.01.01      | 1.05+01      | 1            | 1.0E+01      | i.0E+01      | ł            | 1            | 10+30-1      |            | ı       | •            |
|   | ANAD<br>FKEQ    | 0 M 100 | 0. UE 100  | N/A     | 8.5E-05  | 6.0E-05 | 6.0E-05 | 6.0E-05    | 6.0E-05    | A/A        | 3. 2E-04  | 3.2E-04   | N/A             | t direct      | N/A            | N/A          | 1.6E-10      | 2.2E-10      | N/A          | 1.6E-10      | 2.2E-10      | N/A          | 1.6E-10      | 2.2E-10      | N/A          | N/A          | 1.66-10      | N/A        | A/N     | N/A          |
|   | 9               | ſ       | 4          | ~       | ~        | ~       | ۲٩      | 2          | 2          | 2          | 64        | ~         | ~               | rcraf         | -              | -            | -            | *            | -            | *            | -            | +            | +            | *            | -            | -            | -            | -          | -       | 4            |
|   | SCENARIO        | -       | Ĩ.         | VC      | VC       | 6C      | 분       | VC         | 60         | ٨C         | 6C        | VC        | VC              | it - Large ai | 191, 180, 191) | GF (89° EGL) | HC (90. 16L) | +C (80. IGL) | HC (86. 16L) | 2C (90. [81) | EC (80. IGL) | 6C (89° IGL) | HC (90. 16L) | HC (80. [6L) | HC (86. 16L) | GF (80° IGL) | HE (90, 16L) | HF (OFEN)  | HF (MH) | AF (80' IGL) |
|   |                 | 2       | 2          | Ľ,      | 5        | 5       | 5       | 5          | ដី         | 2          | 5         | Ľ         | S               | <i>a</i>      | SL             | 25           | SLD          | 25           | SLDI         | 2            | 20           | SC           | SLC          | SC           | 35           | Ĕ            | Ľ,           | 5          | Ē       | Ľ            |

See notes at end of table.

Marca - Marca

REG THE RESIDE THE REAL PROPERTY AND

<u>.</u>





11111



File: CONSIOR.MK1 date21-Aug-87 page 3

TABLE 5-19 (Continued)

## STORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EXISTING STIES

### Accident Frequencies

| SCENARIO         | NO.   | ANAD<br>Fefo | FANGE<br>FAFTOR | AF6<br>Fred | RANGE      | LBAD       | RANGE<br>Fartne | NAAF     | RANGE<br>Factor | F8A<br>FRF0 | RANGE<br>Factor | PUDA<br>Ered | RANGE   | TEAD        | RANGE<br>FACTOR | UMDA<br>Fred | RANGE<br>FACTO |
|------------------|-------|--------------|-----------------|-------------|------------|------------|-----------------|----------|-----------------|-------------|-----------------|--------------|---------|-------------|-----------------|--------------|----------------|
|                  | ţ     |              |                 |             |            |            |                 |          |                 |             |                 |              |         |             |                 |              |                |
| SLEVE (NH)       | -     | N/A          | ı               | N/A         | ı          | N/A        | ,               | 1.66-09  | 1.0E+01         | N/A         | ۱               | N/A          | •       | N/A         | •               | N/A          |                |
| SLAVC (60' 16L)  | +     | 1.65-10      | 1.05+01         | N/A         | •          | N/A        | •               | N/A      | '               | N/A         | 1               | N/A          | •       | N/A         | •               | N/A          |                |
| SLAVE (BO IGL)   | -     | 2.2E-10      | 1.úE+01         | N/A         | '          | N/A        | •               | N/A      | ı               | 4. IE-11    | 1.05+01         | N/A          | ı       | 9,86-12     | 1.05+01         | 4.1E-10      | 1.05+0         |
| SLFEC (60° 16L)  | +     | 1.66-10      | 1.0€+01         | N/A         | •          | N/A        | •               | N/A      | '               | N/A         | •               | N/A          | •       | N/A         | '               | N/A          |                |
| SLFEC (80' 16L)  | -     | 2.2E-10      | 1.0€+01         | N/A         | ,          | A/N        | ı               | N/A      | •               | N/A         | •               | N/A          | •       | 9.86-12     | 1. NE + (* 1    | 4.16-10      | 1.06+0         |
| SLFEC (89' 16L)  | -     | N/A          | •               | N/A         | •          | N/A        | 1               | N/A      | •               | N/A         | •               | A/A          | ı       | 1.16-11     | 1.0E+01         | N/A          |                |
| SLFHC (60 16L)   | +     | 1.65-10      | 1.0€+01         | N/A         | •          | N/A        | •               | N/A      | ı               | N/A         | •               | N/A          | ı       | N/A         | ı               | N/A          |                |
| SLFHC (80 IGL)   | +     | 2.2E-10      | 1.05+01         | N/A         | •          | N/A        | ,               | N/A      | ,               | N/A         | '               | 1.65-09      | 1.0E+01 | 9.8E-12     | 1.05+01         | N/A          |                |
| SLFHC (89' 16U)  | +     | N/A          | ۱               | N/A         | •          | 1. JE-10   | 1.0E+01         | N/A      | ı               | N/A         | •               | A/A          | ,       | 1.16-11     | 1.0E+01         | N/A          |                |
| SLPYC (60: 161)  | -     | 1.65-10      | 10+30.1         | A/A         | ,          | N/A        | •               | N/A      | ı               | N/A         | ı               | N/A          | •       | N/A         | '               | A/N          |                |
| CLFVC 180 16L)   | -     | 2.2E-10      | 1.05+01         | N/A         | •          | N/A        | ٠               | N/A      | '               | N/A         | '               | N/A          | •       | 9.86-12     | 1.0€+01         | 4. IE-19     | 1.6E+0         |
| SLFVC (89' 16L)  | -     | N/A          | •               | N/A         | •          | 1. 35 - 10 | 1.0E+01         | N/A      | 1               | N/A         | •               | N/A          | •       | 1.16-11     | 1.0E+(-1        | N/A          |                |
| SLDEC (60° 16L)  | -     | 1.65-10      | 1.0€+01         | N/A         | ı          | N/A        | '               | N/A      | ۱               | N/A         | '               | N/A          | ı       | N/A         | •               | N/A          |                |
| CT01 080 290 15  | -     | 2.2E-10      | 1.05+01         | N/A         | ı          | A/N        | ,               | N/A      | •               | N/A         | ı               | N/A          | '       | 9.86-12     | 1.0€+01         | 01-31 I      | 1.05+0         |
| SLOGC (89° 16L)  | -     | A/A          | ,               | N/A         | •          | 1.3E-10    | 1.0E+01         | N/A      | •               | N/A         | ۱               | N/A          | ı       | 1.16-11     | 1.0E+01         | N/A          |                |
| SL PVC (60 16L)  | •     | A/A          | •               | N/A         | ł          | A/N        | ı               | N/A      | •               | N/A         | ı               | N/A          | ł       | N/A         | '               | A / A        |                |
| CT91 .081 JAD 15 | -     | N/A          | •               | N/A         | •          | N/A        | '               | N/A      | ,               | N/A         | ı               | N/A          | ,       | 9.86-12     | 1.05+01         | 4.1E-10      | 1.65.0         |
| SL QVC (89 16L)  | -     | N/A          | ı               | N/A         | •          | N/A        | •               | A/N      | •               | N/A         | •               | N/A          | ı       | 1.16-11     | 1.0€+01         | N/A          |                |
| SLREC (60° 161)  | -     | 1.66-10      | 10+3v.1         | N/A         |            | N/A        | ,               | N/A      | •               | N/A         | ł               | N/A          | ı       | N/A         | ۱               | A.A          |                |
| St KGC (80 1GL)  | -     | 2.26-10      | 1.05+01         | N/A         | ,          | N/A        | •               | A/N      | '               | 4. IE-H     | 1.05+01         | N/A          | ı       | 9.86-12     | 1.05+01         | 4. IE - 10   |                |
| SLRGC (89° 16L)  | -     | N/A          | ,               | N/A         | •          | 1. 36 - 10 | 1.0E+01         | N/A      | •               | N/A         | •               | N/A          | 1       | 1. JE - I I | 1.0€+01         | N/A          |                |
| SLRVC (60° 16L)  | •     | 1.66-10      | 1.05+01         | N/A         | •          | N/A        | •               | N/A      | •               | N/A         | •               | N/A          | ı       | N/A         | •               | 47N          |                |
| SLEVC (80' 16L)  | +     | 2.2E-10      | 1.05+01         | N/A         | ı          | A/N        | •               | N/A      | •               | 4. IE-11    | 1.0£+01         | N/A          | •       | 9.86-12     | 1.06+01         | 4.1E-10      | 1.0E+i         |
| SLEVC (89° 16L)  | -     | N/A          | •               | N/A         | 1          | 1. JE - 10 | 1.05+01         | N/A      | •               | N/A         | ٠               | N/A          | •       | 1.15-11     | 1.05-01         | A . A        |                |
| SLEVF (80° 16L)  | -     | N/A          | •               | N/A         | •          | N/A        | •               | N/A      | '               | N/A         | ،               | N/A          | '       | N/A         | •               | 4.16-10      | 1.1540         |
| SLSVF (NH)       | +     | N/A          | ,               | N'N         | •          | N/A        | ,               | N/A      | •               | N/A         | ,               | N/A          | 1       | 3, 9E - 10  | 1,05401         | N/A          |                |
| SL5 - Large an   | rcraf | t indirec    | t crash on      | to storage  | e area; fi | re nnt co  | it parret i     | n 30 mmu | ites (burs      | tered oun   | itions de       | tonate 1f    | hit).   |             |                 |              |                |
| SLEGF (80 (GL)   | רש    | 11. J        | r               | N/A         | ·          | N/A        | •               | 4.7      | ,               | N/A         | ,               | N/A          | ,       | 2.76-17     | 1. 3E +01       | 1.1E-10      | 1. 35 00       |

See notes at end of table.

TABLE 5-19 (Continued)

File: CONSTOR.WX1 date21-Aug-87 page 4

K

## STORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR NUNITIONS AT EXISTING SITES

|           | RANGE<br>FACTOF |   | ·<br>_  |             | -         | '             | •          |                |                 |         | •           | _           | _       | •       |         | 10+31.1E+01 | '<br>-       |         | •            | 6-1.36+01  | _                 | 0 1. XE +++1 |                | -       |            |              |            | 0 1. 35 +01  |                     | •        |
|-----------|-----------------|---|---------|-------------|-----------|---------------|------------|----------------|-----------------|---------|-------------|-------------|---------|---------|---------|-------------|--------------|---------|--------------|------------|-------------------|--------------|----------------|---------|------------|--------------|------------|--------------|---------------------|----------|
|           | UNDA<br>Freq    |   | N/A     | A/N         | N/N       | N/A           | N/A        | N/A            | N/A             | N/A     | N/A         | A/N         | N/A     | A/N     | N/A     | 1.4E-0      | N/A          | A/N     | e/N          | 1. IE - I  | H/H               | 1. IE - I    | A / N          | N.'N    | N.A        | R/A          | A/N        | 1.16-1       | €/N                 | A/A      |
|           | RANGE<br>Factor |   | 1.35+01 | •           | 1. 35 +01 | 1. 3E+01      | ı          | '              | '               | '       | 1.35+01     | 1.35+01     | 1.36+01 | ı       | 1.05+01 | '           | 1. 3E+01     | •       | •            | 1.35401    | •                 | 1.75+01      | 1. 3E +01      | '       | 1. JE +01  | 1.36+01      | 1          | 1.36491      | 1.35+01             |          |
|           | TEAD<br>Freq    |   | 2.7E-12 | N/A         | 2.7E-12   | 2.7E-12       | N/A        | 2.7E-12        | 2.7E-12         | 8/N     | A/M         | N/A         | 2.7E-12 | A/A     | 2.7E-12 | A/N         | 2, 7E-12     | A/N     | N/A          | 2.7E-12    | A./ N             | 2.7E-12      | 2.7E-12        | N/A     | 2. 7E - 12 | 2.7E-12      | N/A        | 2.7E-12      | 2.7E-12             | N/A      |
|           | RANGE<br>Factor |   | ı       | ı           | 1.3E+01   | ı             | ı          | •              | •               | •       | 1.35+01     | •           | •       | •       | •       | •           | '            | 1       | '            | ı          | •                 | •            | •              | •       | 1.35+01    | •            | •          | •            | '                   | ı        |
|           | PUDA<br>Fred    |   | N/A     | N/A         | 4.4E-10   | N/A           | N/A        | N/A            | N/A             | A/A     | 4.46-10     | N/A         | N/A     | N/A     | N/A     | N/A         | N/N          | N/A     | N/A          | N/A        | A/A               | N/A          | N/A            | N/A     | 4.4E-10    | A/N          | N/A        | N/A          | N/A                 | N/A      |
|           | RANGE<br>Factor |   | •       | •           | •         | '             | 1          | •              | •               | •       | •           | •           | •       | •       | 1.0E+01 | •           | •            | ı       | ı            | 1. 35 +01  | 5                 | ,            | •              | •       | •          | •            | 1          | ,            | 1                   | ı        |
|           | PBA<br>Fred     |   | N/A     | N/A         | N/A       | N/A           | N/A        | N/A            | A/4             | N/A     | N/A         | N/A         | N/A     | N/A     | 7.96-09 | N/A         | N/A          | N/A     | N/A          | 1.16-11    | N/A               | N/A          | N/A            | N/A     | N/A        | N/A          | N/A        | A/A          | N/A                 | N/A      |
| es        | RANGE<br>Factor | 1 | '       | '           | •         | ı             | ı          | •              | •               | ٠       | •           | 1           | '       | ł       | '       | •           | '            | I.1E+01 | '            | ,          | '                 | •            | •              | •       | •          | '            | 1          | •            | •                   | 1        |
| Frequenci | NAAP<br>Freg    |   | N/A     | N/A         | N/A       | N/A           | N/A        | N/A            | N/A             | N/A     | N/A         | R/A         | N/A     | N/A     | N/A     | N/A         | N/A          | 3.8E-09 | N/A          | N/A        | N/A               | N/A          | A/M            | A/M     | N/A        | N/N          | N/A        | N/A          | N/A                 | R/A      |
| Accident  | RANGE<br>Factor |   | •       | 1           | ,         | •             | 1          | ,              | 1               | ١       | •           | •           | 1       | •       | ۱       | 1           | 1            | 1       |              | •          | •                 | '            | •              | •       | '          | 1.36+01      | •          | •            | 1.35+01             | 1        |
|           | LEAD<br>Freq    |   | A/A     | N/A         | M/A       | N/A           | M/A        | N/A            | N/A             | N/A     | N/A         | N/A         | N/A     | N/A     | N/A     | N/A         | <b>M/M</b>   | M/A     | A/A          | N/A        | A/A               | A/A          | M/A            | N/A     | N/A        | 3.46-11      | N/A        | N/A          | 3, 45 - 11          | M/M      |
|           | RANGE<br>Factor |   | ı       | '           | '         | •             | •          | ,              | •               | •       | '           | ſ           | 1       | '       | 1.05+01 | •           | •            | •       | ,            | ,          | '                 | ,            | '              | •       | ı          | •            | •          | •            | •                   |          |
|           | AFG<br>Freq     |   | N/A     | N/A         | N/A       | N/A           | N/A        | N/A            | A/N             | A / N   | N/A         | N           | 4/N     | N/A     | 2.1E-09 | N A         | N/A          | N/A     | N/A          | A/A        | N / A             | A × N        | A/N            | N/A     | 4 · N      | 4<br>2       | 4 × X      | A.N          | <b>A</b> . <b>N</b> | A / A    |
|           | RANGE<br>Factor |   | •       | 1.35+01     | 1.36+01   | '             | 1.36+01    | 1.35+01        | •               | 10+35.1 | 1.35+01     | •           | •       | 1.36+01 | T       | •           | 1            |         | 10+ 3x - 1   | 1. 36 +61  | 1.7.5 +01         | 1.35+01      | •              | 10+3(   | 10+3: 1    | •            | 10+3.1     | 10-3-1       |                     | 10+32.12 |
|           | ANAD<br>FREQ    |   | A/A     | 5. 7E - L1  | 5.96-11   | N/A           | 5. 7E - 11 | 5.9E-11        | N/A             | 5.76-11 | 5.96-11     | 8/8         | A/A     | 5.76-11 | N/A     | N/A         | M/A          | M/A     | 5.7E H       | 5.95 11    | 5. E-H            | 11 30.5      | 4<br>7         | 5.76 11 | 11 36 .    | 4            | 11-37      | 11 3e-5      | A N                 | 5.76 11  |
|           | NO.             | ł | ŝ       | ŝ           | v٦        | v٦            | ŝ          | ŝ              | ru              | n       | 'n          | <b>e</b> 7) | WD.     | 'n      | v٦      | ŝ           | רש           | ŝ       | wh           | <b>u</b> n | <b>u</b> *•       | 'n           | Ś              | ŝ       | Ŷ          | •            | <b>u</b> . | •            | •                   | ŗ        |
|           | 2               |   | 16L)    | <u>1</u> 61 | 16.)      | 1 <b>6</b> 1) | 16L)       | <b>.</b><br>18 | 181.)           | ( PI )  | 181         | 191         | 16L)    | 161)    | ī       |             | โย           |         | ાશ           | .0         | $\overline{\tau}$ | - 19         | 161.)          | ಕ್ಷ     | 161        | E            | Ē          | H            | 5                   | ଞ        |
|           | ENAR            |   | 6       | (9)         | 68)       | . 68)         | . 09)      | 08             | 68)             | 99      | <b>9</b> 8) | . 68)       | (8)     | 097     | 13101   | (H)         | 0 <b>8</b> I | Ĥ       | ψ <b>9</b> ι | 80         | ję,               | u P          | 68)            | 99      | Ū,         | 6 <b>6</b> ) | 092        | 5 <b>8</b> . | ŝ                   | 96.      |
|           | ŝ               |   | 51 AGF  | SLCHC       | SL OHC    | SL DHC        | SL ( 6C    | 31 CGC         | 51 ( <b>6</b> C | SLCHC   | SLCHC       | SLCHC       | SLEGF   | SLEHF   | SLUHF   | SLEHF       | SLEVE        | 34.475  | 31 #15       | C^¥ ij     | 31 F GC           | 393 S        | 3 <b>9</b> : 5 | SLFHC   | 5 ا ا ال   | JH t 15      | 5,         | 31 8 15      | 51975               | រម្មតិ 🖯 |

See notes at end of table.

<u>.</u>

•

5-76





17.D2-32-02-15-15-05

ч (C



TABLE 5-19 (Continued)

## STORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

### Accident Frequencies

|                                                                                                  |                    |            |                 |             | 1               |                |                 |              |                  |                       |                 |               |                 |              |           |              |                  |
|--------------------------------------------------------------------------------------------------|--------------------|------------|-----------------|-------------|-----------------|----------------|-----------------|--------------|------------------|-----------------------|-----------------|---------------|-----------------|--------------|-----------|--------------|------------------|
| SCENARIO                                                                                         | Û.                 | FRED       | RANGE<br>FACTOR | AFG<br>Freq | RANGE<br>Factor | L BAD<br>F RED | RANGE<br>FACTOR | NAAF<br>Freq | RANGE<br>Factor  | PRA<br>Fred           | RANGE<br>Factor | PUDA<br>Fikeo | RANGE<br>Factor | TEAD<br>Freq | FAC TOR   | UMDA<br>Freq | RANGE<br>Frictor |
| 4<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1 | <b>t</b><br>5<br>1 |            |                 |             |                 |                |                 |              | ;<br>;<br>;<br>; | 1<br>1<br>1<br>1<br>1 |                 |               |                 |              |           |              |                  |
| 190. 180. 181                                                                                    | . 5                | 5. 9E - 11 | 1.3E+01         | N/A         | ,               | N/A            | •               | N/A          | •                | N/A                   | ı               | N/A           | •               | 2.7E-12      | 1. 36+01  | 1. IE-10     | 1.38 001         |
| 06C (89' 16                                                                                      | 5                  | N/A        | ,               | N/A         | ,               | 3.4E-11        | 1.3E+01         | N/A          | •                | N/A                   | •               | N/A           | ı               | 2.7E-12      | 1. 3E +01 | N/A          | ,                |
| OVC (60' 161                                                                                     | <b>.</b>           | N/A        | ,               | N/A         | ,               | N/A            | ı               | N/A          | ,                | M/A                   | ,               | N/A           | •               | N/A          | ,         | N/A          | ,                |
| .0vC (80' 161                                                                                    | <u>.</u>           | N/A        | •               | N/A         | '               | N/A            | ,               | N/A          | ,                | N/A                   | •               | N/A           | '               | 2.76-12      | 1. XE+01  | 1.16-10      | 1. 3E ±01        |
| QVC (89° 161                                                                                     | <u>.</u>           | N/A        | ,               | N/A         | ,               | N/A            | ı               | N/A          | ,                | A/A                   | ,               | N/A           | •               | 2.76-12      | 1. 3E+01  | N/A          |                  |
| FGC (60' 161                                                                                     | 0.5                | 5.7E-11    | 1.3E+01         | N/A         | ,               | N/A            | ı               | N/A          | ł                | N/A                   | •               | N/A           | •               | N/A          | •         | N/A          | •                |
| RGC (80' 16L                                                                                     | . 5                | 5. fE-11   | 1.3E+01         | N/A         | •               | N/A            | •               | N/A          | •                | 1.16-11               | 1.36+01         | N/A           | ,               | 2. TE - 12   | 1. JE+01  | 1.16-10      | 1. 3( +01        |
| RGC (89° 1GL                                                                                     | 5                  | N/A        | ,               | N/A         | ı               | 3.4E-11        | 1.35+01         | N/A          | ,                | N/A                   | •               | N/A           | •               | 2.76-12      | 1.36401   | N/A          | '                |
| PVC 160' 161                                                                                     | 5                  | 5.7E-11    | 1.36+01         | N/A         | •               | N/A            | •               | N/A          | •                | N/A                   | •               | N/A           | •               | N/A          | ,         | H/N          | •                |
| RVC 180 161                                                                                      | 5                  | 5.96-11    | L. 3E+01        | N/A         | '               | N/A            | ,               | N/A          | •                | 1.16-11               | 1.35+01         | N/A           | •               | 2.7E-12      | 1.35+01   | 1. IE-10     | 1.36+01          |
| FVC 189' 161                                                                                     | с.                 | N/A        | ,               | N/A         | '               | 3.4E-11        | 1.3E+01         | N/A          | •                | N/A                   | •               | N/A           | ,               | 2.7E-12      | 1. 3E+01  | N/A          | •                |
| SVF (80 16L                                                                                      | 5                  | N/A        | ı               | N/A         | •               | N/A            | •               | N/A          | '                | N/A                   | •               | N/A           | •               | N/A          | •         | 1.16-10      | 1, 36+01         |
| SVF (NH)                                                                                         | ŝ                  | N/A        | •               | N/A         | ,               | N/A            | ı               | N/A          | •                | A/N                   | •               | N/A           | •               | 6.0E-10      | 1. IE +01 | N/A          | •                |
| SL6 - Tornac                                                                                     | lo gene            | rated miss | illes strik     | e the sto   | rage sagaz      | ine, warel     | house, or       | open stor    | age area         | ; sunitio             | ns breacht      | ed ino de     | tonation)       |              |           |              |                  |
| FIGC E                                                                                           | ÷                  | N/A        | 1               | N/A         |                 | N/A            | •               | N/A          | •                | N/A                   | ı               | N/A           | '               | 3.5E-15      | 9.4E+01   | 1.2E-15      | 9.4E+ú]          |
| DHC                                                                                              | \$                 | 4.86-12    | 9.4E+01         | N/A         | ı               | N/A            | t               | N/A          | ١                | N/A                   | '               | 3. 2E-13      | 9.45+01         | 5.BE-15      | 9.4E+01   | N/A          | '                |
| C6C                                                                                              | -0                 | 4.8E-12    | 9.4E+01         | N/8         | •               | N/A            | ı               | N/A          | •                | N/A                   | ,               | N/A           | •               | 5.86-15      | 9.4E+01   | A/N          | •                |
| CHC                                                                                              | <b>\$</b>          | 4.8E-12    | 9.4E+01         | N/A         | ı               | N/A            | ,               | N/A          | ı                | N/A                   | ,               | 3.2E-13       | 9.45+01         | N/A          | 1         | N/A          | ı                |
| FEC (80. 161                                                                                     | <b>9</b> (:        | N/A        | ł               | N/A         | '               | N/A            | ı               | A/A          | •                | N/A                   | ,               | N/A           | '               | 2.46-15      | 9.45+01   | N/A          | 1                |
| KHC (60' 16L                                                                                     | 9 (                | 1.26-12    | 9.4E+01         | N/A         | •,              | H/H            | ,               | N/A          | •                | N/A                   | •               | N/A           | •               | N/A          | •         | N/A          |                  |
| KHS (OFEN)                                                                                       | 4                  | N/A        | ı               | 6.65-11     | 9.4E+01         | N/A            | ı               | N/A          | •                | 9.96-10               | 9.4E+01         | N/A           | •               | 1.25-12      | 9.4E+01   | N/H          | ,                |
| KHC (NH)                                                                                         | 9                  | N/A        | •               | N/A         | •               | N/A            | ,               | N/A          | ı                | N/A                   | 1               | N/A           | ,               | N/6          |           | 1-36.4       | 5.4E+111         |
| KVC (8) 161                                                                                      | <b>9</b> (         | N/A        | •               | N/A         | '               | N/A            | ,               | N/A          | •                | N/A                   | ı               | N/A           | ,               | 2.46-15      | 9.4E+01   | N/A          |                  |
| KVC (NH)                                                                                         | 9                  | N/A        | ,               | N/A         | '               | N/A            | •               | 3. 35-10 9   | 1.4E+01          | N/A                   | t               | N/A           | •               | N/4          | ı         | D/N          |                  |
| AVC                                                                                              | <b>,</b> 9         | 4,85-12    | 9.46+01         | N/A         | '               | N/A            | ı               | N/A          | ı                | 8. XE-12              | 9.4E+01         | N/A           | ŀ               | 1.35-11      | 9. 4E+ů1  | 51-39.5      | 5.4[+11]         |
| F 6C                                                                                             | \$                 | 4.86-12    | 9.45+01         | N.'A        | •               | N/A            | i               | N/A          | '                | N/A                   | ı               | N/A           | •               | 5.86-15      | 9.4E+31   | 5.66-15      | 9.4E+u1          |
| FHC                                                                                              | ¢.                 | 4.86-12    | 9.4E+01         | N/A         | ı               | 4.86-12        | 9.4E+ú)         | N/A          | •                | N/A                   | '               | 3. 25-13      | 9.4E+ú1         | 5.86 15      | 9, 4E +01 | N/A          |                  |
| FVC                                                                                              | 9                  | 4.BE-12    | 9.46+01         | N/A         | •               | 4.85-12        | 9.46+01         | N/A          | 1                | N/A                   | ,               | N/A           | 1               | 5.86-15      | 9.4E 401  | 5.86-15      | 9.4E+01          |

See notes at end of table.

LEARING STORY

5-77

11.

File: CONSTOR. Whi date21-Aug-87 page b

TABLE 5-19 (Continued)

STORAGE ACCIPENIS - (Frequency units given at bottom of table) FOR MUMITIONS AT EXISTING STIES

Accident Frequencies

| SCENAR10         | NO.        | ANAD<br>FREQ    | FANGE<br>FACTOR | 486<br>FREQ | HANGE<br>Factor | L RAD<br>F REQ | RANGE<br>FACTOR | FRED  | RANGE<br>FACTOR | PRA<br>Fred | RANGE<br>Factor | PUDA<br>Freg | FANGE<br>Factor | TEAD<br>Freq | RANGE<br>FACTOR | UMDA<br>FREO | FACTOR    |
|------------------|------------|-----------------|-----------------|-------------|-----------------|----------------|-----------------|-------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|-----------|
| 1 DEC            | 4          | <b>4</b> .86-12 | 9.46+01         | N/A         | ,               | 4. RE - 12     | 9.46+01         | N/A   |                 | N/A         | ,               | NIA          | '               | 5.86-15      | 9.4E+01         | 5.86-15      | 9.4E +11  |
| 1 PVC            | o          | A/A             | , '<br>!        | N/A         | •               | N/A            | . '             | N/A   | •               | N/A         | ı               | N/A          | 1               | 5.0E-15      | 9.4E+01         | 5.86-15      | 9, 4E +01 |
| 31 F.GC          | م          | 1.86-12         | 9.45+01         | N/A         | •               | 4.86-12        | 9.45+01         | N/A   | '               | 1.96-11     | 9.46+01         | N/A          | ,               | 41-36.1      | 9.46+01         | 5.6E-15      | 9.4E+01   |
| LRVC             | <b>.</b> 0 | 4.86-12         | 9.46+01         | N/A         | •               | 4.86-12        | 9.4E+01         | N/A   | ,               | 1.96-11     | 9.4E+01         | N/A          | •               | 4.8E-14      | 9.4E+01         | 5.8E-15      | 9.4E+01   |
| (191 .08) JOIN   | -0         | N/A             |                 | N/A         | •               | N/A            | •               | N/A   | •               | N/A         | •               | N/A          | ı               | N/A          | ,               | 31-35-12     | 9.4E+01   |
| (HIH) STEAD      | •          | N/A             |                 | A/A         | '               | N/A            |                 | A/N   | •               | N/A         | •               | N/A          |                 | 1.2E-13      | 9.4E+(-1        | N/A          | •         |
| SL7 - Severe     | earthg     | iuale brea      | iches the c     | euritions   | in storage      | igles; n       | o detonat:      | 1605. |                 |             |                 |              |                 |              |                 |              |           |
| 1. PGC           | 1          | A/A             | ,               | N/A         | ۰ ،             | N/A            | ,               | N/A   | •               | N/A         | ı               | A/A          | •               | 1.66-06      | 10+32.1         | 7. OE - n8   | 1. 35 461 |
| 3H0 1:           | 1          | 3.0E (iB        | 1. 3E +01       | N/A         | •               | N/A            | •               | N/A   | ı               | N/A         | ١               | 3.0E-08      | 1. 35 +01       | 7, 05-07     | 1. XE 401       | N/A          | '         |
| 1000             |            | 7.06-49         | 1.36+01         | N/A         | ٠               | N/A            | ,               | N/A   | ı               | N/A         | •               | A/A          | •               | 1.6E-(i7     | 1.36+01         | Η/Ĥ          |           |
| ALCHC            | •          | 7.05-07         | 10+32.1         | N/A         | •               | N.'N           | •               | N/A   | ,               | A/A         | •               | 7.05-09      | 1. JE +01       | £/X          | '               | A/A          |           |
| (191 .08) 19(1)  | 1          | A/A             |                 | N/A         | ,               | A/A            | •               | N/A   | •               | N/A         | ۱               | A/A          | ł               | 1. IE -(15   | 10+3-1          | N/A          | ,         |
| (191 . UR) 3HATS | 1          | 4.6E-07         | 10+32.1         | N/A         |                 | N/A            | '               | N/A   | •               | N/A         | ſ               | N/A          | ١               | N/A          | •               | N/A          |           |
| ILLHS (OPEN)     | ~          | N/A             | '               | 0°0E+00     | ,               | N/A            | ,               | N/A   | ı               | 0.0E+00     | ,               | N/A          | •               | 0.05+(i))    | •               | N/A          | ,         |
| SLEHC (MH)       | 5          | A.'. N          | ı               | N/A         | ,               | N/A            | 1               | N/A   | ,               | N/A         | •               | A/N          | ı               | N/A          | •               | 4/N          | ,         |
| 118VC (80' 16L)  | ~          | N/A             |                 | N/A         | ,               | A/A            | ١               | N/A   | •               | N/A         | •               | N/A          | ۰               | 1. IE - (15  | 1.36+01         | N/A          | '         |
| SLEVE (NH)       | 1          | N/A             | •               | N/A         | ·               | N/A            | •               | N/A   | •               | N/A         | ,               | N/A          | ,               | N/A          | •               | N/A          | ,         |
| 11 MVC           | 1          | 1.86-08         | 1.3E+01         | N/A         | ·               | N/A            | 1               | N/A   | ı               | 1.85-08     | 1.3E+01         | N/A          | ,               | 4. IE -07    | 1.36+01         | 1.86-08      | 1.36+     |
| 31 FGC           | -          | 0.0E+()()       | •               | N/A         | •               | N/A            | 1               | M/A   | 1               | N/A         | '               | N/A          | ı               | 0.0E+00      | ٠               | 0.0E+110     | ·         |
| IL FHC           | 1          | 0.0E+00         |                 | N/A         | ,               | 0.0E+00        | •               | N/A   | •               | N/N         | '               | 0.0E+00      | ٠               | 0.05400      | ,               | N/A          |           |
| SLFVC            | ^          | 0.0E+110        | •               | N/A         | '               | 0.05+()()      | ,               | N/A   | '               | N/A         | •               | N/A          | '               | 0.0E+00      |                 | 0.01400      |           |
| 31 DGC           | ~          | 0.0E+00         | •               | A/A         | ı               | 0° 0E +00      | ,               | N/A   | '               | N/A         | •               | N/A          | •               | 0.0E+00      | •               | 0.0E+00      |           |
| 31.070           | 1          | N/A             | ٠               | N/A         | •               | N/A            | •               | N/A   | ۱               | N/A         | •               | N/A          | 4               | 0.0E+00      | 1               | 0, 0E +00    |           |
| 1. REC           | ~          | 9.75-08         | 1.36+01         | N/A         | ,               | 9.75-08        | 1. 3E+01        | N/A   | •               | 9.75-08     | 1.3E+01         | N/A          | ı               | 2.1E-06      | 10+32-1         | 9.75-08      | 1. XE+01  |
| JLAVE            | 1          | 9.75-08         | 1.35+01         | N/A         | •               | 9.76-08        | 1.3E+01         | N/A   | 1               | 9.7E-08     | 1. 35+01        | N/A          | •               | 2.1E-06      | 1.35+01         | 9. JE - 08   | 1. 7E+01  |
| (191 08) JAS 1   | ٢          | N./A            | •               | N/A         | ,               | N/A            | 1               | N/A   | •               | N/A         | 1               | N/A          | '               | 0.0E+00      | ,               | 0. 0E + 0    |           |
| SUC (HH)         | 1          | N.A             | •               | N/A         | ,               | N/A            | ı               | N/A   | •               | N/A         | •               | N/N          | '               | N/A          | I               | N/A          | ,         |

See notes at end of table.

2,22,21

Ē







File: CONSTOR.MK1 date21-Aug-87 page 7

## TABLE 5-19 (Continued)

## STORAGE ACCIDEMTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

### Accident Frequencies

| SCENARIO     | NO.         | ANAD<br>Freq      | RANGE<br>Factor | AFG<br>Freq   | RANGE<br>FACTOR | L FAD<br>F KEQ | RANGE<br>Factor | NAAF<br>FREU | RANGE<br>Factor | PRA<br>Freq | RANGE<br>FACTOR | PUDA<br>Freq | RANGE<br>Factor | TEAD<br>FRED | RANGE<br>FACTOR   | UMDA<br>Freg      | RANGE<br>FACTU |
|--------------|-------------|-------------------|-----------------|---------------|-----------------|----------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-------------------|-------------------|----------------|
|              |             | ******            |                 |               |                 |                |                 |              |                 |             |                 |              |                 |              |                   |                   |                |
| 51 HGF       | 89          | A/A               | 1               | N/A           | I               | N/A            | •               | N/A          | •               | N/A         | ı               | N/A          | 1               | 6. 7E-12     | 2.6E+01           | 6.7E-12           | 2.66+13        |
| SLDHC        | 80          | 6.7E-12           | 2.6E+01         | NZA           |                 | N/A            | •               | A/A          | '               | N/A         | •               | 6.7E-12      | 2.6E+01         | 6. 7E-12     | 2.6E+(i)          | N/A               |                |
| 51 CGC       | 8           | 6. 7E-12          | 2.65+01         | N/A           | •               | N/A            | •               | N/A          | •               | N/A         | •               | N/A          | •               | 6. 7E-12     | 2.6E+01           | R/A               |                |
| SLCHC        | 80          | 6.7E-12           | 2.46+01         | N/A           | •               | N/A            | '               | N/A          | ,               | d/N         | •               | 6.7E-12      | 2.65+01         | N/A          | •                 | N/A               |                |
| SUBER (16L)  | 8           | N/A               | •               | N/A           | ,               | N/A            | '               | N/A          | Ŧ               | N/A         | '               | N/A          | •               | 6. 7E-12     | 2.66+01           | N/A               |                |
| SLINF (IGL)  | 8           | 6.7E-12           | 2.6E+01         | N/A           | •               | N/A            | •               | N/A          | •               | N/A         | 1               | N/A          | •               | N/A          | •                 | N/A               |                |
| SLEHF (DFEN) | 8           | N/A               | •               | 1.26-11       | 1.7E+01         | N/A            | •               | N/A          | •               | 1.25-11     | 1.7E+01         | N/A          | •               | 1.75-11      | 1.75+01           | N/A               |                |
| (HM) JHAIS   | B           | N/A               | •               | N N           | ,               | N/A            | •               | A/A          | •               | N/A         | ľ               | N/A          | '               | N/A          | '                 | 1.45-10           | 2.6E+0         |
| (191) JAIIS  | 8           | A/A               | •               | N/A           | •               | N/A            | •               | N/A          | •               | N/A         | •               | N/A          | '               | 6.7E-12      | 2.46+01           | N/A               |                |
| SLEVE THH)   | 8           | N/A               | ŀ               | N/A           | •               | N/A            | '               | 1.06-09      | 2.6E+01         | N/A         | ,               | N/A          | •               | N/A          | •                 | N/A               |                |
| SLAVC        | 8           | 6. JE-12          | 2.66+01         | NA            | •               | N.A            | •               | N/A          | ,               | 6.7E-12     | 2.6E+01         | N/A          | •               | 6. 71-12     | 2.6E+01           | 6.7E-12           | 2.6f+ñ         |
| 51 PGC       | 8           | 6.7E-12           | 10+39.5         | N/A           | •               | N/A            | •               | N/A          | •               | N/A         | •               | N/A          |                 | 6.7E-12      | 2.6E+01           | 6.7E-12           | 2.6E+n         |
| 3L FHC       | 8           | 6.7E-12           | 2.6E+01         | N/A           | ı               | 6. 75 - 12     | 2.6E+01         | N/A          | ۰               | A/A         | ·               | 6.75-12      | 2.65+01         | 6.76-12      | 2.66+"1           | N/A               |                |
| SLFVE        | ę           | 6.7E-12           | 2.6E+01         | N/A           | •               | 6.7E-12        | 2.6E+01         | N/A          | '               | N/A         | ,               | M/A          | '               | 6. /E -12    | 2.6E+01           | 6. <i>J</i> E -12 | 1. kE +11      |
| 51 26C       | 8           | 6. <i>1</i> E-12  | 2.66+01         | N/A           |                 | 6.7E-12        | 2.6E+01         | N/A          | ı               | N/A         | •               | N.'N         | •               | 6. 7E-12     | 2.65+0]           | 6.7E-12           | 2. 6E +11      |
| 346 IS       | ത           | N/A               | •               | N/A           | '               | N/A            | 1               | N/A          | '               | N/A         | •               | M/A          | •               | 6. 7E-12     | 2.66+01           | 6. JE - L         | 2.6E+ii        |
| SLFGC        | <b>m</b>    | 6.75-12           | 2.6E+01         | N/A           | t               | 6.7E-12        | 2.66+01         | N/A          | ,               | 6.75-12     | 2.65+01         | N/A          | ı               | 6.7E-12      | 2. <u>6</u> E +01 | 6.7E-12           | 2.66 +11       |
| SI FVC       | an)         | 6. <i>J</i> E -12 | 2.6E+01         | N/A           |                 | 6.7E-12        | 2.6E+01         | N/A          | '               | 6.7E-12     | 2.6E+01         | N/A          | '               | 6. JE - 12   | 2.66+01           | 6. <i>i</i> E -12 | <b>6€</b> ₩    |
| 1911 34515   | æ           | N/A               | 1               | N/A           | •               | N/A            | ,               | N/A          | •               | N/A         | ,               | N/A          | ,               | N/A          | •                 | 6.7E-12           | 2.6E+11        |
| SLSVE (NH)   | 8           | N/A               |                 | N/A           | •               | N/A            | •               | N/A          | •               | N/A         | ,               | N/A          | •               | 1.75 69      | 2.65+61           | N/A               |                |
| SLP - Munits | on drop     | ped durit         | ig leater       | i sal et i on | activities      |                |                 |              |                 |             |                 |              |                 |              |                   |                   |                |
| 51.630       | ٠           | N.A               | •               | N/A           | r               | N/A            | •               | N/A          | •               | A/N         | •               | N/A          | 1               | 6.6E-01      | 1.35401           | 6.6E-ü7           | 1. 16 +11      |
| SL DHC       | 5           | (i-35°)           | 1.35+01         | N/A           | ,               | N/A            |                 | N/A          |                 | N/A         | ı               | 10-35.1      | 1. 3E +01       | 1.55-07      | 1 ° 3E + 0 I      | N/A               |                |
| 51 CGC       | ¢           | 9, BE -08         | 1. 35 +01       | 8/N           | •               | N/A            | '               | N/A          | •               | N/A         | 1               | N/A          | •               | 9.86-08      | 1. 76 461         | A/A               |                |
| SLCHC        | ۶           | 9. RE -118        | 1.35+01         | N/A           | ı               | N/A            | ,               | A/A          | •               | A/N         | '               | 9. 96. 08    | 1.35+01         | N/A          | •                 | N/4               |                |
| 511.51       | ٥           | A/A               | •               | N , A         | '               | N/A            | •               | A/A          | •               | N/A         | ı               | N/A          | 1               | 1.95-07      | 1. 75 +01         | N/A               |                |
| 5L n HC      | <b>C</b> +- | 1, 96 - 47        | 1.36+01         | 1.96-07       | 1.35+01         | N/A            | ı               | N/A          | ,               | 1.96-07     | 1.35+01         | e/N          | •               | 1.36-07      | 1. 35 401         | 1.55 07           | 1. 35 🗤        |
| 16415        | ه           | N/A               | •               | н, А          | 1               | N/A            | ı               | 1.96-07      | 1 VE +01        | N/A         |                 | N/A          | ,               | 1.96 11/     | 1.7.5 + 0.1       | N/A               |                |

See notes at end of table.

File: CONSTUR.WH 1 date21-kug-87 page 8

TABLE 5-19 (Continued)

STORAGE ACCIDENTS - IFrequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

Accident Frequencies

| RANGE<br>Factor | . 36+01 | '         | ١         | ,        | ١          | t          | 1.36+01   | . 3E+01   | 10+35.1  |              | •        | •        | ı        | . 06 +01 | ı        | •       | ľ     | •       |              | . OE +U1   | ı          | •           | •         | •          | •         | ı         | ı           |     |
|-----------------|---------|-----------|-----------|----------|------------|------------|-----------|-----------|----------|--------------|----------|----------|----------|----------|----------|---------|-------|---------|--------------|------------|------------|-------------|-----------|------------|-----------|-----------|-------------|-----|
| UNDA<br>FRED    | 3.6E-07 | 0.0E+00   | N/A       | 0.0€+00  | 0.0€+00    | 0.0E+00    | 1.56-06   | 1.56-06   | 3.6E-06  |              | NYA      | N/A      | N/A      | 8.96-09  | N/A      | N/A     | N/A   | N/A     |              | 5.06-10    | N/A        | N/A         | N/A       | N/A        | N/A       | N/A       | N/A         |     |
| RANGE<br>Factor | 1.3E+01 | •         | ı         | ı        | ı          | •          | 1.35+01   | 1. 3E +01 | 1.36+01  |              | 1        | •        | 10+30.1  | ۱        | ı        | •       | 1     | 1.0E+01 |              | 1.0E+01    | 1.0E+01    | •           | 1.0E+Ú1   | 1. (E +ù I | ,         | ı         | •           |     |
| TEAD<br>FRED    | 3.6E-07 | 0.0E+00   | 0.0E+00   | 0.0E+00  | 0. 0E +(i0 | 0.0E+00    | 1.5E-06   | 1.5E-06   | 3, 66-06 |              | 0.0E+00  | N/A      | 1.4E-07  | N/A      | 0.0E+00  | N/A     | N/A   | 1.6E-0B |              | 1.26-11    | 1.36-11    | N/A         | 1.2E-11   | 1.35-11    | N/A       | 1.26-11   | 1.35-11     |     |
| RANGE<br>FACTOR | •       | '         | •         | •        | •          | •          | •         | •         | ı        |              | •        | •        | •        | •        | ı        | ı       | •     | '       |              | •          | •          | •           | 1.0E+01   | '          | '         | 1         | 1           |     |
| PUDA<br>FRED    | N/A     | N/A       | 0.0E+00   | N/A      | N/A        | N/A        | N/A       | N/A       | N/A      |              | N/A      | N/A      | N/A      | N/A      | N/A      | N/A     | N/A   | N/A     |              | N/A        | N/A        | N/A         | 2.0E-09   | N/A        | N/A       | N/A       | N/A         | : : |
| RANGE<br>Factor | 1.3£+01 | •         | ı         | ,        | 1          | ı          | 1. 3E +01 | 1.36+01   | '        | es.          | •        | '        | 1.05+01  | •        | ٠        | ı       | •     | •       |              | ٠          | •          | •           | ۰         | ,          |           | ı         | •           |     |
| PBA<br>FKED     | 3.6E-07 | N/A       | N/A       | N/A      | N/A        | N/A        | 1.5E-06   | 1.5E-06   | N/A      | 30 ainut     | N/A      | N/A      | 5.6E-07  | N/A      | N/A      | N/A     | N/A   | N/A     |              | N/A        | N/A        | N/A         | N/A       | N/A        | N/A       | N/A       | N/A         |     |
| RANGE<br>Factor | ,       | '         | •         | •        | •          | '          | '         | •         | ,        | tained in    | 1        | '        | •        | r        | ١        | 1.0E+01 | ,     | '       |              | •          | ١          | '           | ı         | •          | ı         | ١         | 1           |     |
| NAAP<br>FRED    | N/A     | N/A       | N/A       | N/A      | N/A        | N/A        | N/A       | N/A       | N/A      | e hot con    | A/A      | N/A      | A/N .    | N/A      | A/A      | B.1E-09 | N/A   | N/A     | ~            | N/A        | N/A        | N/A         | N/A       | A/A        | N/A       | N/A       | N/A         |     |
| RANGE<br>FACTOR | ,       | ,         | ,         | ,        | ı          | '          | 1. 3E +01 | 1. 3E+01  | ı        | yard; fir    | ı        | ۱        | ١        | 1        | ١        | •       | ı     | ı       | detonate     | ٠          | •          | I           | ,         | ı          | ı         | ı         | •           |     |
| L BAD<br>FREQ   | N/A     | N/A       | 0.0E+00   | 0.0E+00  | 0.0E+00    | N/A        | 1.5E-06   | 1.56-06   | N/A      | storage      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A     | N/A   | N/A     | unitions     | N/A        | e/N        | N/A         | A/A       | N/A        | N/A       | N/A       | N/A         |     |
| RANGE<br>FACTOR | ,       | •         | •         | •        | ,          | ,          | •         | •         | •        | se or open   |          | '        | 1.05+01  | ı        | •        | ١       | •     | ,       | irstered 🖷   | ı          | •          | ı           | •         | ,          | ١         | ۱         | •           |     |
| APG<br>Freu     | N/A     | N/A       | N/A       | N/A      | N/A        | N/N        | N/A       | N/A       | N,A      | o warehou:   | A'X      | N/A      | 1.65-05  | N/A      | N/A      | N/A     | N/A   | N/A     | fire. (bi    | N/A        | N/A        | N/A         | A/N       | N/A        | N/6       | N/A       | N/A         | : - |
| RANGE<br>FACTUR | 1.36+01 | •         | •         | ,        | ,          | •          | 1.35+01   | 1.36+01   | ,        | crash ont    | ,        | ı        |          | •        | •        | '       | ,     | ı       | crash; no    | '          | T          | 10+30.1     | 1.6+01    | •          | 1. u£ +01 | 1.0E+01   | •           |     |
| ANAD<br>FKEQ    | 3.6E-07 | 0, JE +0U | 0. (E +0) | 0, 6E+00 | (i, VE+0)) | N/A        | 1.56-06   | 1.56-06   | N/A      | ft direct    | N/A      | 0.0E+++0 | N/A      | A/A      | N/A      | N/A     | N/A   | N/A     | ft direct    | A/A        | E/N        | 2.06-10     | 2.66-10   | N/A        | 2.0E-10   | 2.6E-10   | N/A         |     |
| NO.             | œ       | •         | •         | ъ        | <b>.</b>   | •          | o-        | ۰         | ۰        | arcra        | <u>5</u> | <u>.</u> | 5        | 5        | <u>.</u> | £       | 5     | 5       | arcrà        | 16         | 16         | 16          | 16        | 16         | 16        | 91        | 16          | ! . |
| CENAÑ IO        |         |           |           |          |            | <i>,</i> , | L         |           | <u>ب</u> | 15 - Small é | F (16L)  | 1161)    | F (DFEN) | (NH)     | : (191)  | (Hill)  | (161) | (In)    | ió - Large a | (191.08) ; | C (86 16L) | C (60° 16L) | (191 .08) | (191 48)   | (10] (19) | (191191.) | C 189 101.) |     |

See notes at end of table.

3. • E S

23 • 1999 • 1999 • 1999 • 1999 • 1999 • 1999 • 1999 • 1999 • 1999 • 1999 • 1999 • 1999 • 1999 • 1999 • 1999 • 1

 $\hat{\mathbf{z}}$ 

1.1.1.1.1.1.1.1.1

والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمحاج والمحا







File: CONSIOR.HK1 date21-Aug-87 page 9

## TABLE 5-19 (Continued)

## STORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

### Accident Frequencies

| SC              | ENAR    | 01                | NC. | ANAD<br>Fred | RANGE<br>Factor | AP6<br>Freq | RANGE<br>FACTOR | L BAD<br>F REQ | RANGE<br>FACTOR | NAAP<br>Freq | RANGE<br>FACTOR | PBA<br>Fred | RANGE<br>FACTOR | PUDA<br>Fred | RANGE<br>Factor | TEAD<br>Freq | RANGE<br>FACTOR | UNDA<br>FREQ | FANGE    |
|-----------------|---------|-------------------|-----|--------------|-----------------|-------------|-----------------|----------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|----------|
|                 |         |                   |     |              |                 |             |                 |                |                 |              |                 |             |                 |              |                 |              |                 |              |          |
| SLCHC           | . 081   | 181)              | 16  | 2.6E-10      | 1.0E+01         | N/A         | •               | N/A            | ı               | N/A          | ı               | A/A         |                 | 2.0E-09      | 1.0E+01         | N/A          | 1.0E+01         | N/A          |          |
| SLCHC           | .68)    | (19)              | 16  | N/A          | •               | N/A         | •               | N/A            | •               | N/A          | ٠               | N/A         | •               | N/A          | ı               | N/A          | 1.0€+01         | N/A          |          |
| SLKGC           | .081    | 191<br>1          | 16  | N/A          | ٠               | N/A         | '               | M/A            | '               | N/A          | ı               | N/A         | •               | A/N          | 1               | 1.26-11      | 1.05+01         | N/A          |          |
| SLKHC           | .091    | (J)               | 16  | 2.06-10      | 1.0E+01         | N/A         | 1               | N/A            | •               | N/A          | ı               | N/A         | '               | N/N          | •               | N/A          | ۱               | N/A          |          |
| SLKHS           | 101     | Ĥ                 | 16  | N/A          | •               | 1.3E-09     | 1.0E+01         | N/A            | •               | N/A          | '               | 9.4E-09     | 1.06+01         | N/A          | '               | 4. 35 -09    | 1.0E+01         | N/A          |          |
| SLXHS           | (HN)    |                   | 91  | N/A          | •               | N/A         | ٠               | N/A            | '               | N/A          | •               | N/A         | •               | N/A          | •               | N/A          | •               | 1.46-08      | 1.0E+0   |
| SLKVC           | .08)    | 16L)              | 16  | N/A          | •               | N/A         | •               | N/A            | ı               | N/A          | ı               | N/A         | '               | N/A          | •               | 1.25-11      | 1.0E+01         | A/N          |          |
| SLILVS          | Ĥ       |                   | 16  | N/A          | •               | N/A         | ı               | N/A            | ,               | 2.0E-09      | 1.0E+01         | N/A         | •               | N/A          | •               | N/A          | ı               | N/A          |          |
| SLAVC           | . 09)   | 161)              | 16  | 2.0E-10      | 10+30.1         | N/A         | ł               | N/A            | '               | N/A          | ı               | N/A         | '               | N/A          | •               | N/A          | 1               | N/A          |          |
| 3/ M-15         | . (18)  | 16L)              | 16  | 2.6E-10      | 10+30°1         | N/A         | ,               | N/A            | ,               | N/A          | •               | 5.0E-11     | 1.0E+01         | N/A          | ٠               | 1.25-11      | 1.0E+C1         | 5.0E-10      | 1.0E+()  |
| SLFEC           | . 091   | 1GL)              | 16  | 2.0E-10      | 10+30°1         | N/A         | •               | N/A            | •               | N/A          | •               | N/A         | '               | N/A          | •               | A/A          | I               | e/N          |          |
| SLFGC           | . 08)   | [EL)              | 16  | 2.6E-10      | 1.0E+01         | N/A         | •               | 4/4            | ۰               | N/A          | 1               | N/A         | •               | N/A          | •               | 1.25-11      | 1.0E+01         | 5.0E-10      | 1.05+0   |
| SLFGC           | . 68)   | 161)              | 16  | N/A          | 1               | N/A         | 1               | N/A            | '               | N/A          | ,               | N./A        | '               | N/A          | •               | 1. J H       | 1.05+01         | N/A          |          |
| 3LFHC<br>SL FHC | .091    | 161 )             | 16  | 2.0E-10      | 10+30.1         | N/A         | •               | N/A            | •               | N/A          | •               | N/A         | ,               | A/N          | ı               | N/A          | •               | N/A          |          |
| SLFHC           | . 08)   | 19T )             | 16  | 2.6E-10      | 1.0E+01         | N/A         | ı               | N/A            | '               | N/A          | 1               | N/A         | ١               | 2.0E-09      | 1.0E+01         | 1.25-11      | 1, 0E+01        | N/A          |          |
| 3H PHC          | . 68)   | 16U)              | 16  | N/A          | ı               | N/A         | ۱               | N/A            | 1.0E+01         | N/A          | '               | A / N       | ,               | N/A          | ,               | 1            | 1.0E+01         | N/A          |          |
| SLFVC           | . 091   | 161)              | 16  | 2.06-10      | 10+30.1         | N/A         | •               | N/A            | •               | N/A          | •               | N/A         | •               | N/A          | •               | N/A          | 1               | N/A          |          |
| SLPVC           | .081    | ( <del>1</del> 91 | 91  | 2.66-10      | 1.úE+01         | N/A         | •               | N/A            | '               | N/A          | ,               | N/A         | •               | N/A          | •               | 1.25-11      | 1.0E+01         | 5.0E-10      | 1.0E+0   |
| SLFUC           | . 68)   | 16L)              | 16  | N/A          | ŀ               | N/A         | ,               | 1.6E-10        | 10+30.1         | N/A          | '               | N/A         | •               | N/A          | •               | 1.35-11      | 10+30.1         | N/A          |          |
| 2196 JS         | .091    | 16L)              | 16  | 2.0E-10      | 10+30°1         | N/A         | •               | A/N            | ı               | N/A          | '               | N/A         |                 | N/A          | ,               | N/A          |                 | N/A          |          |
| 31 <b>0</b> 60  | .081    | 101)              | 16  | 2.66-10      | 1.06+01         | N/A         | ,               | N/A            | •               | N/A          | '               | N/A         | •               | A/A          | ı               | 1.25-11      | 1. QE +01       | 5.06-10      | 1.05+0   |
| 25050           | 68)     | 191               | 16  | N/A          | ١               | N/A         | •               | 1.65-10        | 1.0E+01         | A/A          | ı               | N/A         | ١               | N/A          | ۲               | 1. 35 -11    | 1.06+01         | A/A          |          |
| SLOVC           | . (191) | (J)]              | 16  | 2.0E-10      | 1.0E+01         | N/A         | '               | N/A            | ı               | N/A          | ł               | N/A         | •               | N/A          | ·               | R/N          | '               | N/A          |          |
| SLOVC           | .080    | 18L)              | 16  | 2.66-10      | 1.0E+01         | N/A         | ,               | N/A            | ı               | N/A          | •               | N/A         | •               | N/A          | •               | 1.26-11      | 1.05+01         | 5.0E-10      | 1.0E+n   |
| SLOVC           | . 68)   | 16L)              | 16  | N/A          | ,               | N/A         | •               | N/A            | ı               | N/A          | •               | N/A         | ı               | N/A          | •               | 1. 36 - 11   | 1.0E+ù1         | N/A          |          |
| SL RGC          | . (191  | 161.)             | 91  | 2.06-10      | 1.0E+01         | N/A         | ,               | N/A            | '               | N/A          | '               | N/A         | ı               | N/A          | •               | N/A          | •               | N/A          |          |
| SLRGC           | .081    | 16L)              | 16  | 2.6E-10      | 1.0E+01         | N/A         | ı               | N/A            | ı               | N/A          | ı               | 5.06-11     | 1.0E+01         | N/A          | •               | 1.26-11      | 1 o£ +01        | 5.0E-10      | 1.05 +++ |
| SLRGC           | .68)    | 16L)              | 16  | N/A          | T               | N/A         | ı               | 1. 6E - 10     | 1,0E+01         | N/A          | '               | N/A         | •               | R/A          | ı               | 1.35 11      | 1.05+01         | N/A          |          |

See notes at end of table.

File: CONSTOR.MK1 date21-Aug-87 page 10

TABLE 5-19 (Continued)

## SIORAGE ACCIDENTS - IFrequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

### Accident Frequencies

See notes at end of table.









## TABLE 5-19 (Continued)

# STORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

### Accident Frequencies

| Г FREG                                                                                           | N.A   | N:A<br>N/A<br>1.76-06 1.16<br>N/A<br>N/A  | N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>1.46-10<br>1.36-                    | К:А<br>N/A<br>I.76-96 I.16-<br>N/A<br>N/A<br>I.46-10 I.36-<br>N/A<br>I.46-10 I.36-<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A | К/А<br>К/А<br>1.76-06 1.16-<br>К/А<br>К/А<br>К/А<br>К/А<br>К/А<br>К/А<br>К/А<br>К/А                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N.A<br>N.A<br>N.A<br>N.A<br>N.A<br>N.A<br>N.A<br>N.A<br>N.A<br>N.A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|--------------------------------------------------------------------------------------------------|-------|-------------------------------------------|---------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| 9 FACTOR<br>                                                                                     |       | /A 1.0E+01<br>/A<br>-12 1.3E+01<br>/A - / |                                                                                             |                                                                                                                              | 78<br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 월 1 월 호호로 ()국<br>4 1 년 년 년<br>1 년 년 년                                                            |       | 979777<br>                                | ,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>, | స్. స్. స్. సి.<br>సి. సి. సి. సి. సి. సి. సి. సి. సి. సి.                                                                   | ;)::::::::::::::::::::::::::::::::::::                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | , "<br>· · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| FACIOR<br>                                                                                       |       | . , , , ;                                 |                                                                                             |                                                                                                                              | 1, 5, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1,                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| FREQ<br>                                                                                         |       | М/М<br>М/М<br>М/М<br>М/М<br>М/М           |                                                                                             | К/А<br>К/А<br>К/А<br>К/А<br>К/А<br>К/А<br>К/А<br>К/А<br>К/А<br>С/С<br>К/А<br>К/А                                             | N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| FACTOR                                                                                           |       | 1,0E+01<br>-<br>-                         | 1.0E+01<br>-<br>-<br>1.3E+01<br>-                                                           | 1.0E+01<br>                                                                                                                  | 1.0E+01<br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1.0E+01<br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| К.К.<br>К.А.А.<br>К.А.А.<br>К.А.А.<br>К.А.А.<br>К.А.А.                                           |       | 9.75-09<br>N/A<br>N/A<br>N/A<br>N/A       | 9.75-09<br>N/A<br>N/A<br>N/A<br>1.45-11<br>N/A<br>N/A                                       | 9.7E-09<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                                | 9.75-99<br>1.46<br>1.46<br>1.46<br>1.46<br>1.4<br>1.46<br>1.4<br>1.4<br>1.4<br>1.4<br>1.4<br>1.4<br>1.4<br>1.4<br>1.4<br>1.4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 9.75-99<br>1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                                                                                                  |       | -<br>-<br>16+01                           |                                                                                             | · · · 10• · · · · · · · · · · · · · · · · · · ·                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|                                                                                                  |       | N/A<br>N/A<br>N/A<br>N/A<br>N/A           | N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                        | м.м<br>м.м.<br>. 76 - 05 г.<br>м.м.<br>м.м.<br>м.м.<br>м.м.<br>м.м.<br>м.м.<br>м.м.<br>м                                     | мла<br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | N.A<br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|                                                                                                  |       | ي.<br>د د د د د                           | ₩<br>1 4 1 4 4 4 4 4 4                                                                      |                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | بة في المحمد بالمالياتين.<br>القالية المحمد بالمالياتين                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|                                                                                                  |       | <b>दददद</b> द                             |                                                                                             | аааааааааа<br>                                                                                                               | анарарарарарарарарарарарарарарарарарара                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | анарарарарарарарарарарарарарарарарарара                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|                                                                                                  |       | * 2 * 2 X                                 | * 2 * 2 * 2 * 2 2 2                                                                         | * * * * * * * * * * * * * * * * * * *                                                                                        | * * * * * * * * * * * * * * * * * * *                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N N N N N N N N N N N N N N N N N N N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| FACTOR                                                                                           |       | 1.0E+01<br>-<br>-<br>-                    | 1.06.401                                                                                    | 1. 0E+01<br>                                                                                                                 | 1.0E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 10:5401<br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 4<br>4<br>4<br>4<br>7<br>7<br>7<br>7<br>7<br>7<br>7<br>7<br>7<br>7<br>7<br>7<br>7<br>7<br>7<br>7 |       | 2.6E-09<br>N/A<br>N/A<br>N/A<br>N/A       | 2.66-09<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                             | 2.65-07<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                                | 2.55-20<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 2.<br>67-70<br>7 7 7<br>7 7 7<br>7 7<br>7 7<br>7 7<br>7 7<br>7 7<br>7 7<br>7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| FACTOR<br><br>1.35+01<br>1.35+01<br>                                                             |       | -<br>-<br>1. 35 +01                       |                                                                                             | 1. 36 +01<br>1. 36 +01<br>1. 36 +01<br>1. 36 +01<br>1. 36 +01<br>1. 36 +01<br>1. 36 +01                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 1, 36-01<br>1, |
| FFEQ<br>N/A<br>7.0E-11<br>7.5E-11<br>N/A<br>N/A                                                  | N . N | N/A<br>N/A<br>N/A<br>N/A<br>N/A           | N/A<br>N/A<br>N/A<br>N/A<br>2.36-11<br>7.36-11<br>7.36-11                                   | N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                                           | N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | NA<br>NA<br>NA<br>NA<br>NA<br>NA<br>NA<br>NA<br>NA<br>NA<br>NA<br>NA<br>NA<br>N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5                                                            | •     | 2 2 2 2 2 2                               | 2 3 3 3 5 3 5 8 5 8 5                                                                       | 2 2 3 2 3 3 3 3 5 3 5 3 5 5 5 5 5 5 5 5                                                                                      | 223232323232323232                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| (10)<br>(10)<br>(10)<br>(10)<br>(10)<br>(10)<br>(10)<br>(10)                                     |       | ()<br>[[6()                               | 1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1 | 0<br>180<br>180<br>180<br>180<br>180<br>180<br>180<br>180<br>180<br>18                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| (9)<br>(9)<br>(9)<br>(9)<br>(9)<br>(9)                                                           |       | (0FEN<br>(HH)<br>(AH)<br>(AH)             | (05EN<br>(MH)<br>(80<br>(80<br>(80<br>(80<br>(80<br>(80<br>(80<br>(80<br>(80                | (10FEN<br>(MH)<br>(MH)<br>(HM)<br>(HM)<br>(HM)<br>(HM)<br>(HM)<br>(HM)                                                       | (0FEN<br>(MH)<br>(80<br>(80<br>(80<br>(80<br>(80<br>(80<br>(80<br>(80<br>(80<br>(80                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | (8)<br>(8)<br>(8)<br>(8)<br>(8)<br>(8)<br>(8)<br>(8)<br>(8)<br>(8)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 영동문동영북                                                                                           |       | SHAC<br>SHAC<br>SHAC                      | SUPHS<br>SUPHC<br>SUPVC<br>SUPVC<br>SUPPC<br>SUPPC<br>SUPPC                                 | SUPPS<br>SUPPS<br>SUPPC<br>SUPPC<br>SUPPC<br>SUPPC<br>SUPPC<br>SUPPC<br>SUPPC<br>SUPPC<br>SUPPC                              | SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE<br>SUPPE |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |

See notes at end of table.
File: CONSTOR.WK1 date21-Aug-87 page 12

.

TABLE 5-19 (Continued)

STORAGE AFCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EXISTIMG SITES

|            | RANGE<br>FACTOR | 1.1.1            |                 | •               | '               | 1.35+01         | 1                    | 1.3E+01         | ,           |                | 1                | ,               | '               | '               | '            | 1.16+01    | ,               | ,          | ľ               | '          |               |          | •       | •       | ı       | '           | '           | •            | ١          | •           |
|------------|-----------------|------------------|-----------------|-----------------|-----------------|-----------------|----------------------|-----------------|-------------|----------------|------------------|-----------------|-----------------|-----------------|--------------|------------|-----------------|------------|-----------------|------------|---------------|----------|---------|---------|---------|-------------|-------------|--------------|------------|-------------|
|            | UMDA<br>Fkeq    |                  | 1-1-1           | N/7             | N/A             | 1.4E-10         | N/A                  | 1.46-10         | N/A         |                | 0° 0E +00        | N/A             | 4/N             | N/A             | N/A          | 4.86-12    | N/A             | N/A        | (i • 0E +i)0    | N/A        |               | 0. (E+00 | N/A     | E/N     | N/G     | N/A         | N/A         | N'A          | 0.0E+00    | W/A         |
|            | RANGE<br>Factor |                  | 1. 36 11        | 1. 36+01        | '               | 1.36+01         | 1.36+01              | •               | 1.1E+01     |                | ŀ                | ١               | ١               | ,               | 1.0E+01      | ı          | 1. 3E+01        | ı          | ,               | 1.1E+01    |               | •        | 2.6E+01 | 2.65+01 | •       | •           | •           | •            | '          | •           |
|            | TEAD<br>FREQ    | 11 I             | 21-30-0         | 3.46-12         | N/A             | 3. 36 - 12      | 3. <del>IE</del> -12 | N/A             | 7.45-10     |                | 9.26-16          | 9. 3E-16        | 9.25-16         | N/A             | 9.7E-13      | N/A        | 9.25-16         | N/A        | N/A             | 2. IE-13   |               | 0.0E+00  | 2.76-07 | 1.46-07 | N/A     | 0.0E+00     | R . N       | 0.05+00      | N/A        | 0.05+00     |
|            | RANGE<br>FACTOR | r<br>1<br>1<br>1 | •               | '               | ۱               | •               | •                    | •               | •           |                | •                | •               | 1               | 1               | •            | ſ          | ,               | ٠          | •               | ۲          |               | •        | 2.6E+01 | •       | 2.6E+01 | '           | •           | •            | ,          | •           |
|            | PUDA<br>Fred    |                  | H               | N/A             | N/A             | N/A             | N/A                  | N/A             | N/A         |                | N/R              | N/A             | N/A             | N/A             | N/A          | N/A        | N/A             | N/A        | N/A             | N/A        |               | N/A      | 1.25 08 | M/A     | 6.2E-07 | N/A         | N/A         | N/A          | N/A        | N/A         |
|            | RANGE<br>Factor |                  | 1.36401         | ı               | ·               | 1.3E+01         | '                    | ı               | '           |                | 1                | •               | ,               | •               | 10+30.1      | •          | '               | 1          | •               | ٠          |               | ı        | •       | ı       | 1       | •           | •           | •            | •          | ı           |
|            | PBA<br>Freq     |                  | 1.44-11         | N/A             | N/A             | 1.4E-11         | N/A                  | N/A             | N/A         |                | N/A              | N/A             | N/A             | N/A             | 2.7E-12      | N/A        | N/A             | N/A        | N/A             | N/A        |               | N/A      | N/A     | N/A     | N/A     | N/A         | N/A         | 0.0E+00      | N/A        | N/A         |
| 53         | RANGE<br>Factor | 1                | •               | ,               | ,               | ,               | •                    | ۰               | •           |                | •                | ı               | ·               | •               | 1            | •          | 1               | 1.15+01    | ı               | '          |               | '        | ,       | •       | ,       | •           | •           | ı            | •          | •           |
| r requenci | NAAP<br>FREQ    |                  | H/N             | N/A             | N/A             | N/A             | N/A                  | N/A             | N/A         | 0 minutes      | N/A              | N/A             | N/A             | A/N             | N/A          | N/A        | N/A             | 1.3E-12    | N/A             | N/A        |               | N/A      | M/A     | N/A     | N/N     | N/A         | N/A         | N/A          | N/A        | ¥/¥         |
| Accident   | RANGE<br>FACTOR |                  | ,               | 1. JE +01       | ı               | •               | 10+36-1              | •               | ı           | Ined in J      | ,                | 1               | I               | •               | ı            | •          | ı               | •          | •               | '          |               | ı        | '       | '       | •       | ı           | ,           | ι            | •          | •           |
|            | LBAD<br>Freq    |                  | A/A             | 4.2E-11         | N/A             | N/A             | 4.2E-11              | N/A             | N/A         | ire conta      | N/A              | A/A             | N/A             | N/A             | N/N          | N/A        | N/A             | N/A        | N/A             | N/A        |               | N/A      | N/A     | N.A     | N/A     | N/A         | N/A         | N/A          | N/A        | N/A         |
|            | RANGE<br>FACTOR | 9<br>9<br>9<br>9 | •               | '               | '               | '               | ,                    | ·               | •           | ge area; f     | •                | ł               | •               | ,               | 1.05+01      | 1          | ,               | ,          | ,               | ı          | nation        | •        | •       | ·       | ,       | •           | •           | •            | •          | ,           |
|            | AFG<br>Freq     |                  | N/A             | N/A             | N.A             | N/A             | N/A                  | N/A             | N/A         | into stora     | N/A              | N/A             | N/A             | N/A             | 7.26-13      | N/A        | N/A             | N/A        | N/A             | N/N        | tion deto     | N/A      | N/A     | N/A     | N/A     | N/A         | N/A         | 0°.0E+00     | N/A        | 8/N         |
|            | RANGE<br>Facior |                  | 1.34+01         | •               | 1.36+01         | 1.36+01         | •                    | ,               | I           | ct crash c     | ·                | •               | ı               | •               | ۰            | ı          | •               | '          | •               | '          | ls to auni    | •        | 2.6E+01 | 2.6E+01 | 2.4E+01 | •           | ,           | ,            | •          | •           |
|            | ANAD<br>Freq    |                  | / 11            | N/A             | 7.06-11         | 7. 36 - 11      | N/A                  | A/W             | N/A         | ft indire      | N/A              | N/A             | N/A             | 11-36-11        | A/A          | N/A        | N/A             | N/A        | N/A             | N/A        | quake lea     | N/A      | 1.26-08 | 6.25-07 | 6.25-07 | N/A         | 0.0E+00     | N/A          | N/A        | R/N         |
|            | ND.             | ; ;              | 0.7             | 2               | 02              | 20              | 20                   | 20              | 20          | ircra          | Ы                | 5               | 21              | 21              | 21           | 21         | 21              | 21         | 21              | 21         | earth         | 72       | 13      | 22      | 22      | 3           | 22          | 22           | 22         | 22          |
|            | SCENARIO        |                  | SUREC (80, 16L) | SLKGC (89' 16L) | SLRVC (60' 16L) | STRVC (80. 16L) | SLRVC 189' 161 )     | SLSVE (80' 16L) | SLSVC (INH) | SL21 - large a | Streff (80' 16L) | SLRGF (89° 16L) | SLK6F (80° 16L) | SLKHF (60° 16L) | SLEHF (OFEN) | SLEHF (WH) | SLKVF (80° 16L) | SLKVF (NH) | SLSVF (80° 16L) | SLSVF (NH) | SL22 - Severe | SI 86C   | SLDHC   | SI CGC  | SL CHC  | SLEGS (16L) | SLAHS (16L) | SLEHS (OFEN) | SLKHS (NH) | SLKVS (16L) |

See notes at end of table.

 $\sim$ 



POTENSION DAMAGE DAMANCE DAMANCE







File: CONSTOR.MA1 date21-Aug-87 page 13

0.0.000000

Charles Carl

TABLE 5-19 (Continued)

## SIORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUWFILDWS A? EXISTING SITES

### Accident Frequencies

يعت

| <u>.</u>   | ANAD<br>Fred | FACTOR     | AF6<br>Freq | FACTOR     | L RAD<br>F RED | FACTOR     | NAAP<br>Fred | RANGE<br>Factor | FBA<br>FREQ | RANGE<br>Factor | PUDA<br>FRED | FACTOR  | TEAD<br>Fred | FACTOR       | UNDA<br>Fred | FACTO    |
|------------|--------------|------------|-------------|------------|----------------|------------|--------------|-----------------|-------------|-----------------|--------------|---------|--------------|--------------|--------------|----------|
|            | 0 / M        |            | 0 / N       |            | 47 M           | ,          | 0.06+00      |                 | 6 / N       | I               | 0/2          | ,       | 0/N          | ,            | N/A          |          |
|            | 1.0F-09      | 2. AF +01  |             | ,          | N/A            | ,          | N/A          |                 | 7.05-09     | 2 46 401        |              | ,       | 1.AF-07      | 2. 4F 401    | 7.0F-09      | S. AF et |
|            | 4. 7F -09    | 10+34.5    | N/A         | ,          | N/A            | •          | N/A          | ,               | N/A         |                 | M/M          | ,       | t. 0F-07     | 2.66+01      | 4. 7F-69     | 2.65     |
|            | 4.75-09      | 2.46+01    | N/A         | ,          | 4.75-09        | 2.66+01    | A/N          | ı               | N/A         | '               | 4.7E-09      | 2.6E+01 | 1.06-07      | 2,66+01      | N/A          |          |
|            | 4.7E-05      | 2.6E+01    | N/A         | ,          | 4.76-09        | 2.6E+01    | N/A          | •               | N/A         | ,               | N/A          | •       | 1.06-07      | 2.6E+01      | 4, 76 -119   | 2.6E     |
|            | 3.4E-09      | 2.6E+01    | N/A         | '          | 3.4E-09        | 2.6E+01    | N/A          | ı               | N/A         | ,               | N/A          | ,       | 7.65-08      | 2.6E+01      | 3° 4E - Úð   | 2.6E+(   |
| ~~         | N/A          | •          | N/A         | '          | N/A            | '          | M/A          | ı               | N/A         | 1               | N/A          | 1       | 7.65-118     | 2.66+01      | 2. 4E -09    | 2.66+1   |
| ~          | 3.9E-09      | 2.6E+01    | N/A         | •          | 3.9E-09        | 2.6E+01    | N/A          | ı               | 3.96-09     | 2.6E+01         | N/A          | ,       | 8°-36-08     | 2.65+01      | 3. 96 -119   | 2.6E+i   |
| ٢v         | 3°-36-08     | 2.66+01    | N/A         | ,          | 3. 9E -(19     | 2.6E+01    | N/A          | ı               | 3.9E-09     | 2.6E+01         | A/N          | •       | 8.96-08      | 2.65+01      | 30-36 °2     | 2.6E+    |
| 2          | N/A          | •          | N/A         | ,          | N/A            | •          | N/A          | 1               | N/A         | •               | N/A          | ،       | N/A          | •            | 0,05+00      |          |
| <b>e</b> 4 | N/A          | •          | N/A         | •          | N/A            | 1          | N/A          | •               | N/A         | '               | N/A          | •       | 0,06+00      | ,            | N/A          |          |
| ia uel     | rated aus    | siles stri | ike the sto | orage iglo | io and cal     | ise munit  | ton detona   | t10n.           |             |                 |              |         |              |              |              |          |
|            | N/A          | ,          | N/A         |            | N/A            | •          | N/A          | •               | N/A         | '               | N/A          | •       | 0.01+000     | ,            | 0.0E+00      |          |
| ŗ          | 3.4E-13      | 10+36.6    | N/A         | •          | N/A            | •          | N/A          | •               | N/A         | •               | 2.2E-14      | 9.9E+01 | 3.2E-16      | 9.96+01      | N/A          |          |
| -          | 3.4E-13      | 10+36.9    | N/A         | ,          | N/A            | •          | N/A          | •               | N/A         | •               | N/A          | '       | 3. 26-16     | 9. 95 +01    | N/A          |          |
| m          | 3.46-13      | 9.95+01    | N,A         | •          | N/A            | •          | A/A          | •               | N/A         | •               | 2.26-14      | 9.95+01 | N/A          | ,            | N/A          |          |
| ~          | N/A          | '          | N/A         | •          | N/A            | '          | N/A          | •               | N/A         | '               | N/A          | •       | 0.01 10.0    | ,            | N/A          |          |
| ~          | 0.05+00      | •          | N/A         | ,          | N/A            | •          | N/A          | •               | N/A         | •               | N/A          | ı       | N/A          |              | N/A          |          |
| ٠.         | N/A          | •          | 0°*0E+00    | •          | M/A            | •          | N/A          | •               | 0, 0E +00   | •               | N/A          | '       | 0.05+00      | ,            | A N          |          |
| p.,        | N/A          | ,          | N/A         | '          | N/A            | ,          | A/A          | ۱               | N/A         | •               | N/A          | ,       | N.A          | •            | 0.05 +110    |          |
| <b>.</b>   | N/A          | ,          | N/A         | '          | N/A            | '          | M/A          | •               | N/A         | •               | M/A          | •       | 0.0E+00      |              | N/A          |          |
| -          | N/A          | ,          | N/A         | •          | N/A            | '          | 0.0E+00      | •               | M/A         | '               | N/A          | •       | N/A          | •            | N/A          |          |
| •          | 3.4E-13      | 9.9E+01    | N/A         | ı          | N/A            | •          | A/A          | •               | 1-39.4      | 9.95+01         | N/A          | '       | 7.45-16      | 10+34-9      | 4.(E-16      | 3.56     |
| •          | 3.4E-13      | 9.95+01    | N/A         | ,          | N/A            | •          | N/A          | •               | N/A         | •               | N/A          | 1       | 3. 26 - 16   | 9. 9E +111   | 4.06-16      | 9.96     |
| м,         | 3.4E-13      | 10+36.9    | N/A         | •          | 3.46-13        | 9.9E + N   | N/A          | •               | N/A         | ,               | 2.26-14      | 9.96+01 | 91-32 · ;    | 9.95 +111    | N/A          |          |
| -          | 3.4E-13      | 9.9E+01    | N/A         | ,          | 3. 46 - 13     | 9. 9E +01  | N/A          | ,               | A/A         | '               | N/A          | •       | 3. 2E - 16   | 9. 9E +01    | 4.01-16      | ς. 9€ •  |
|            | 3.4E-13      | 10+36°6    | N/A         | '          | 3.4E-13        | 9. 9E tú 1 | A/A          | ,               | N/A         | •               | N/A          | ,       | 3. 21 - 16   | t (in 35 ° 5 | 4. nE - 1 é  | 9. 3E+   |
| •••        | N/A          | •          | N/A         | '          | A/A            | '          | N/A          | ,               | N/A         | '               | N/A          | •       | 3. 25 -16    | 10+36.9      | 4.(6-16      | 3, 96 +1 |

See notes at end of table.

<u>Developed and the second between the second managed and the second presents and the second managed and the s</u>

5-85

File: CONSTOR.WK1 date21-Aug-87 page 14

TABLE 5-19 (Continued)

### STORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

### Accident Frequencies

| SCENARIO       | NO.       | ANAD<br>F REQ | RANGE<br>FACTOR | APG<br>Fred      | RANGE<br>FACTOR | L RAD<br>F KE D | RAtige<br>Factor | NAAP<br>Fred | RANGE<br>Factor | FRA<br>Fred | RANGE<br>Factor | FUDA<br>Freq | RANGE<br>Facior | TEAD<br>Fred | PANGE<br>Factor | UMDA<br>Fred | RANGE<br>Factor |
|----------------|-----------|---------------|-----------------|------------------|-----------------|-----------------|------------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|-----------------|
|                | !         |               |                 |                  | •               |                 |                  |              |                 |             |                 |              |                 |              | 1               |              |                 |
| SLRGC          | 23        | 3, 46 - 13    | 9.9E+01         | N/A              | '               | 3.4E-13         | 6.9E+01          | N/A          | ı               | 1.16-12     | 9.9E+ú1         | N/A          | ,               | 2.6E-15      | 9.9E+01         | 4. ()E - ] b | 9.9[+11]        |
| SLAVE          | 52        | 3.46-13       | 9.9E+01         | N/A              | •               | 3.4E-13         | 9.9E+01          | N/A          | •               | 1.16-12     | 9.96+01         | N/A          | •               | 2.6E-15      | 9.9E+01         | 4.06-16      | 9.96401         |
| SLSVS (TEL)    | 23        | N/A           | •               | N/A              | •               | A/N             | •                | N/A          | •               | N/A         | ·               | N/A          | '               | N/A          | ı               | 0.05+00      | ı               |
| SLEVS (NH)     | 23        | N/A           | •               | N/A              | ,               | N/A             | •                | N/A          | •               | N/A         | •               | N/A          | '               | 0.0E+00      | •               | N/A          |                 |
| SL24 - Lightn  | ing st    | crikes ton    | n container     | 's stored        | outdoor s.      |                 |                  |              |                 |             |                 |              |                 |              |                 |              |                 |
| SLKHS (CPEN)   | 24        | N/A           | ı               | 1.4E-10          | 1.0E+01         | N/A             | ,                | N/A          | ,               | 5. IE-10    | 1.05+01         | A/N          | ,               | 1.4E-10      | 1.0£+/1]        | N/A          | ı               |
| 5L25 - Muniti  | ons dr    | up paddo.     | 'ing leaker     | isolatio         | n; aunitio      | in detonati     | PS.              |              |                 |             |                 |              |                 |              |                 |              |                 |
| St 66C         | ង         | N/A           | 1               | N/A              | •               | N/A             | '                | N/A          | •               | N/A         | ı               | N/A          | •               | 0.0E+00      | •               | 0. 0E 400    | •               |
| SLDHC          | 52        | 1.75-07       | 2.6E+01         | N/A              | '               | N.'A            | •                | N/A          | ,               | N/A         | •               | 1.7E-07      | 2.6E+01         | 1.76-07      | 2.6E+01         | N/A          | `               |
| SLCGC          | 25        | 8.9E-08       | 2.6E+01         | N/A              | '               | N/A             | 1                | N/A          | •               | A/A         | 1               | N/A          | 1               | 8°-36-08     | 2.6E+01         | N/A          | •               |
| SLCHC          | 55        | 8°-36-08      | 2.4E+01         | N/A              | ı               | N/A             | ,                | A/A          | ,               | N/A         | •               | 8°-36-08     | 2.6E+01         | N/A          | ٠               | N/A          | '               |
| SLIFEC         | 23        | N/A           | ı               | N/A              | •               | N/A             | ı                | N/A          | ı               | N/A         | ,               | N/A          | 1               | 0.0E+00      | ,               | N/A          | `               |
| 21 MC          | 22        | 0.05+00       | '               | 0 <b>°</b> 0E+00 | 1               | N/A             | •                | N/A          | •               | N/A         | ı               | N/A          | ľ               | 0, 0E +00    | ,               | 0 0E +00     | ,               |
| JAATS          | <u>بر</u> | N/A           | •               | N/A              | •               | N/A             | •                | 0,0E+00      | •               | N/A         | •               | A.'N         | •               | 0.CE+00      | •               | N/A          | ١               |
| SLAVC          | 25        | 1. XE -07     | 2.6E+01         | N/A              | ٠               | N/A             | •                | A/N          | •               | 1. 35-07    | 2.6E+01         | N/A          | •               | 1.3E-07      | 2.6E+01         | 1.35-07      | 2.66401         |
| SLFGC          | 53        | 80-37-1       | 2.6E+01         | N/A              | '               | A/A             | ۰                | N/A          | r               | N/A         | ı               | N/A          | •               | 3.2E-08      | 2.6E+01         | 3, 26-08     | 2.66+01         |
| SLFHC          | ۲3<br>ا   | 3, 26 - 08    | 2.6E+01         | N/A              | ,               | 3.2E-(iB        | 2.6E+01          | A/N          | •               | N/A         | •               | 3.2E-08      | 2.6E+01         | 3, 2E -08    | 2.6E+01         | N/A          | 3               |
| SLFVC          | 33        | 3.2E-08       | 2.6E+01         | N/A              | ,               | 3. 25 -08       | 2.6E+01          | N/A          | ı               | A/N         | '               | R/A          | •               | 3.2E-0E      | 2.66+01         | 3, 2E-08     | 2.66+01         |
| SL <b>D</b> 5C | 53        | 3.2E-MB       | 2.6E+01         | N/A              | •               | 3. 26-08        | 2.6E+()1         | 67N          | •               | N/A         | ,               | N/A          | •               | 3.2E-08      | 2.6E+01         | 3.2E -08     | 2.6E+01         |
| SLOVC          | 53        | N.A           |                 | N/A              | •               | N/A             | •                | N/A          | '               | N/A         | ,               | N/A          | ۱               | 3, 26 - 98   | 2,65+01         | 3. 25 -08    | 2.66 401        |
| 51.F.GC        | 2         | 5.7E-118      | 2.6E+01         | N/A              | 4               | 5.76-08         | 2.6E+01          | <b>U.</b> N  | '               | 5.7E-08     | 2.6E+01         | N/A          | •               | 5. 76 -09    | 2.65401         | 5. )E -08    | 2.65 401        |
| JAN IS         | 5         | 5.76-08       | 2.6E+01         | N/A              | 1               | 5. 7E-08        | 2.6E+01          | N/A          | ,               | 5.7E-08     | 2.6E+01         | N/A          | •               | 5.7E-08      | 2.6E+01         | 5.7E-08      | 2.6E+U1         |
| SLSVC          | ដ         | N/A           | '               | N/A              | I               | N/A             | '                | M/A          | •               | N/A         | ł               | A/N          | •               | 0.0E+00      |                 | 0, 0E+00     | •               |

NOTES:

See notes at end of table.



132A







File: CONSTOR.WK1 date21-Aug-87 page 15

## TABLE 5-19 (Continued)

### STORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

### Accident Frequencies

| RANGE        | FACTOR   |   |
|--------------|----------|---|
| UFDA         | F KEC    |   |
| RANGE        | F AC TOR |   |
| 1E 4D        | FKEQ     |   |
| RANGE        | FACIOR   |   |
| PUDA         | FRED     |   |
| RANGE        | FACTOR   |   |
| PBA          | FREQ     |   |
| RANGE        | FACTOR   |   |
| NAAF         | FREQ     |   |
| RANGE        | FAC 10R  |   |
| LEAD         | FREQ     |   |
| RANGE        | FACTOR   |   |
| AFG          | FREQ     |   |
| <b>FANGE</b> | F AC TOR |   |
| ANAD         | FREQ     |   |
| ж0.          |          | i |
| SCENAR10     |          |   |

- 1. Frequency units for scenario 1 are events per munition year.
- 2. Frequency units for scenarios 2, 9, and 25 are events per leaker.
- Frequency units for scenarios 4, 5, 8, 15 through 21, and 23 are events per storage unit-year ligloo or warehouse). For ton containers stored outdoors, frequency units for scenarios B and 24 are events per cluster-year of ton containers (15 TC/cluster).
- Agent release for SLMHS 1 (open) assumes outdoor spill onto a porous surface.
- 5. Frequency units for scenarios 7 and 22 are events per year.

file: STORHOLD.Mkl Fage 1 Date 21-Aug-87

٠.

Ю 27 . FREQUENCY OF INTERIM STORAGE SEQUENCES (SA, SR, SW) INTERIM STORAGE AIR OFIION (EVENTS/TR)

| HARE      FANGE      FBA      RANGE      FBA      FBA      RANG      FBA      RANG      FBA      RANGE      FBA      RANGE      FBA      RANGE      FBA      RANGE      FBA      RANGE      FFACTOR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | N/A N/A I.5E-09 (0 N/A<br>N/A N/A I.5E-09 (0 N/A<br>N/A U/A 1.5E-09 (0 N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A N/A 1.5E-09 1º N/A N/A N/A 1.6E-10 11 P/A                     |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------|
| HARE      FAUGE      FBA      RAVEE      FUDA      EANGE      FBA      RAVEE        FACTOR      FACTOR      FACTOR      FACTOR      FACTOR      FACTOR      FACTOR        N/A       N/A       N/A       S.3E-09      10        N/A       N/A       N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A N/A 1.55-19 30<br>N/A N/A 1.56-19 10<br>N/A 1.56-19 10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A N/A 1.5E-09 10<br>N/A N/A 1.6E-10 11                          |
| NARE      FAUGE      FUDA      EANGE      TEAD        FACTOR      FACTOR      FACTOR      FACTOR      FACTOR      FACTOR        N/A       N/A       N/A       3.3E-09        N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N/A N/A 1.5E-19<br>N/A N/A 1.5E-09<br>N/A 1.5E-09                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Nyê NyA 1,5E-09<br>NyA NyA 1,6E-09                                |
| NAAF      RANGE      FBA      RANGE      FBA      RANGE      FACTOR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | N/A N/A<br>N/A N/A<br>N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A N/A N/A                                                       |
| NAAF  RANGE  FBA  FACTOR    N/A   N/A    N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A N/A<br>N/A N/A<br>N/A N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A 9/A<br>N/A N/A                                                |
| NAAF      RANGE      FBA      RANGE      FACTOR        N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ¥/X                                                               |
| NAAF      RANCE      FBA        N/A       H/A        N/A       H/A        N/A       N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | d in a second se |                                                                   |
| ИААР<br>ИААР<br>Кансе<br>Кила<br>Кила<br>Кила<br>Кила<br>Кила<br>Кила<br>Кила<br>Кила                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | : :                                                               |
| NNN NNN NNNN NNNN<br>AAAAAAAAAAAAAAAAAAAAA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | : : :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                   |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | N/N<br>N/A<br>N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/N<br>A/N                                                        |
| FANSE<br>FACTOR<br>FACTOR<br>FOC                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 29                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 9                                                                 |
| LBAD<br>N/A<br>3.2E-09<br>3.2E-09<br>3.2E-09<br>3.2E-09<br>1.7E-09<br>1.7E-09<br>1.7E-09<br>1.7E-09<br>1.7E-09<br>1.7E-09<br>9.1E-10<br>9.1E-10<br>9.1E-10<br>9.1E-10<br>9.1E-10<br>9.1E-10<br>9.1E-10<br>9.1E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11<br>1.7E-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 1.96-11<br>1.96-11<br>.ontained                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 1.75-11<br>Ontained<br>N/A                                        |
| RANSE<br>RANSE<br>In fire<br>10<br>10<br>10<br>10<br>10<br>10<br>11<br>10<br>10<br>11<br>10<br>11<br>10<br>11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | ų                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | <br>                                                              |
| AFG<br>allers: r<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>3.25-10<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | <br><br>1re not                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 2                                                                 |
| SANGE<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIOR<br>SCIO<br>SCIO<br>SCIO<br>SCIO<br>SCIO<br>SCIO<br>SCIO<br>SCIO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A N/A<br>N/A N/A<br>alferst fire not<br>a ac.no                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | К∕А<br>атбег5; ftr<br>8,36-08                                     |
| inter and interest in the second seco | N/A<br>N/A<br>Dig containers; fire mot                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | KrA<br>Mia containers: fir<br>8.36-08                             |
| AHAB        ANAB        N.Y.A.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 47.6 N/A<br>N/6 N/A<br>Crash onto contansers; fire not<br>A.A. Darugo 10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/G N/A<br>. crash onto containers: fir<br>N/A 8.3E-08            |
| 10 No.      ANAD      F        aircraft crash on      i      N/A        i      N/A      i                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 4 1/2 N/A<br>4 N/6 N/A<br>arrotations; fire not<br>5 A.A 0.354.00 10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 4 N/G N/A<br>Bircraft crash onto containers; fir<br>5 N/A 8,36-08 |

2226 • 122222

COLOR RESIDENT



# TABLE 5-20 (Continued)

STATEMENTA STORAGE AIR DETION (EVENTS/YR)

| RANGE<br>Factor | ł       | ł       | !        | ł       | ;       |             | ł       | 1       | :       | !           | :       | :        |             | ł          | !       | 1       | ł               | ;       | ;       |             | :        | :        | ;        | ;               | ;       |             | ł          | :       | ;              | ;       |
|-----------------|---------|---------|----------|---------|---------|-------------|---------|---------|---------|-------------|---------|----------|-------------|------------|---------|---------|-----------------|---------|---------|-------------|----------|----------|----------|-----------------|---------|-------------|------------|---------|----------------|---------|
| KUKU            | N/A     | N/A     | M/A      | N/A     | N/A     |             | N/A     | N/A     | N/A     | N/A         | N/A     | N/A      |             | K/A        | N/A     | H/A     | N/A             | N/A     | N/A     |             | K/A      | N/A      | N/A      | 4/H             | N/A     |             | N/A        | N/A     | N/A            | н/д     |
| RANGE<br>actor  | 11      | 11      | Ξ        | 11      | П       |             | 10      | lů      | 9       | 01          | 10      | 9        |             | 94         | 5       | 94      | 5               | 5       | 94      |             | 66       | 66       | 66       | 54              | 94      |             | 26         | 26      | 5.6            | 26      |
| TEAD            | 1.6E-10 | 1.66-10 | 1.66-10  | 1.65-10 | 1.6E-10 |             | 1.1E-09 | 1.1E-09 | 1.1E-09 | 1.15-09     | 1.16-09 | 1. IE-09 |             | 5.0E-16    | 2.7E-15 | 2.7E-15 | 2.7E-15         | 6.2E-15 | 6.2E-15 |             | 1.0E-15  | 1.0E-15  | 1.0E-15  | 6.6E-15         | 6.6E-15 |             | 2.0E-11    | 2.2E-11 | 2. ZE-11       | 2.26-11 |
| KANGE<br>Factor | ;       | ;       | :        | :       | :       |             | ł       | :       | :       | ;           | ;       | ;        |             | ;          | 1       | ;       | !               | ł       | ;       |             | ;        | ;        | ł        | :               | ;       |             | ;          | ;       | :              | 1       |
| FUEA            | N/A     | N/A     | N/A      | N/A     | N/A     |             | N/A     | N/A     | N/A     | N/A         | N/A     | N/A      |             | N/A        | N/A     | N/A     | N/A             | N/A     | N/A     |             | N/A      | N/A      | N/A      | N/A             | N/A     |             | N/A        | N/A     | N/A            | N/A     |
| KÅNGE<br>Factor | ;       | •<br>1  | ł        | ł       | ł       |             | 1       | :       | 1       | ł           | 1       | 1        |             | :          | ł       | 1       | :               | 1       | ;       |             | !        | ł        | ;        | ;               | ;       |             | ;          | 1       | :              | 1       |
| FBA             | N/A     | N/A     | N/A      | N/A     | N/A     |             | N/A     | R/A     | N/A     | N/A         | N/A     | N/A      |             | N/A        | N/A     | N/A     | N/A             | N/A     | N/A     | ts)         | N/A      | N/A      | N/A      | A/A             | N/A     |             | N/A        | N/A     | N/A            | N/A     |
| RANGE<br>Factor | ł       | ;       | :        | 1       | ;       |             | ł       | 1       | !       | ;           | ;       | :        |             | ;          | :       | ;       | :               | ;       | 1       | n for rocke | ł        | ł        | :        | ł               | ;       |             | 1          | ;       | ;              | :       |
| NAAP            | N/A     | N/A     | N/A      | N/A     | 8/N     |             | N/A     | N/A     | N/A     | H/A         | N/A     | N/A      |             | N/A        | N/A     | N/A     | N/A             | N/Å     | N/A     | r ignitio   | N/A      | N/A      | N/A      | A/A             | N/A     |             | N/A        | N/A     | N/A            | N/A     |
| RANGE<br>FACTOR | =       | Ξ       | 11       | П       | Π       |             | 1       | 10      | 1Ú      | 61          | 10      | 01       | ition       | ;          | 44      | 94      | <b>5</b>        | 94      | 94      | tor acto    | 69       | 66       | 66       | <b>\$</b> 5     | 54      |             | :          | 26      | 26             | 26      |
| LEAD            | 2.0E-12 | 2.0E-12 | 2.0E-12  | 2.0E-12 | 2.0E-12 | bined       | N/A     | 1.46-11 | 1.4E-11 | 11-31-1     | 1.46-11 | 1.4E-11  | ; no detona | N/A        | 8.4E-13 | 8.4E-13 | B.4E-13         | 1.8E-12 | I.BE-12 | detonation  | 3. IE-13 | 3. IE-13 | 3. IE-13 | 1.96-12         | 1.96-12 |             | N/A        | 2.26-11 | 2.26-11        | 2.2E-11 |
| RANGE<br>Factor | ;       | 1       | ;        | ;       | ł       | ire conti   | 10      | с<br>8  | ;       | ;           | !       | ;        | atairers    | <b>1</b> 5 | :       | :       | ;               | ł       | ł       | tainers;    | 1        | ;        | ;        | !               | ł       |             | 56         | ł       | !              | :       |
| μFG             | N/A     | Υ.À     | N/H      | N/A     | N/A     | it (2.1901) | 5.98-07 | N/A     | A/A     | ₽/N         | A/A     | N/A      | etrate co   | 8.46-14    | N.A     | e/N     | ħ/Ĥ             | N/6     | ĥ/ĥ     | rate con    | A/A      | N/A      | 87N      | N/H             | h/ĥ     | lfêà        | 2.0E-11    | A/A     | R.P            | ₽/N     |
| RANGE<br>FACTOR | :       | ł       | !        | 1       | ;       | anta conta  | !       | 1       | ł       | ł           | 1       | 1        | siles peri  | ;          | ;       | ł       | ;               | :       | :       | eile penel  | :        | 1        | ł        | •               | :       | helding a   | ;          | !       | ;              | •       |
| ÂNÂÛ            | 4/N     | ы/А     | N Å      | ñ/ñ     | ۲/۳     | Cresh (     | 14/H    | 1.A     | N/A     | <b>N</b> /A | N/A     | R/A      | ited ais:   | 1.1N       | N.A     | A.A     | N/A             | NA      | NZA     | ited ats:   | ۸/ ĥ     | ħ/Å      | щ, н     | йÂ              | N. F    | ies the     | й/н        | 47N     | Nin            | ц.н     |
| NG.             | υJ      | רני     | רי       | τu      | רט      | rirart      | 4       | -13     | 4       | 9           | ÷       | c,       | gener a     | <b>r</b>   | 1       | 1       | 1               | r.,     | 7       | genera      | na       | Ð        | 8        | <del>ري</del> . | 80      | e stri      | <b>r</b> - |         | ( <b>r</b>     | *       |
| Stenar10        | shirtu  | S-F1.   | كالمنابغ | Crit J. | Бил . L | ite lleid   | 54 HF   | SHPHC   | SåFvC   | ວິຊີບູລິດ   | SAROL   | 34445    | י יר יופלט- | an hô      | ákř n3  | ort 10  | <u> ĉ</u> u lec | juĥ.ŭS  | SARVS   | Tor feado-  | SHPHC    | ShFVû    | 3-43.    | Sanol.          | Ĵhĥ J   | is teurn tu | 1<br>T     |         | <u>анг</u> . Г | CHIC    |

5-89

1.1.1.1.1.1

م و من

1.1.

1.1

TABLE 5-20 (Continued)INTERIM STORAGE AIR OFTION (EVENTS/VE)

| FANGE<br>FACTOR |                                |                    |
|-----------------|--------------------------------|--------------------|
| ปรายจ           |                                | N/A<br>N/A         |
| RAHSE<br>FACTOR |                                | 2 2                |
| TEAD            |                                | 2.0E-11<br>2.0E-11 |
| FANGE<br>Factor |                                | : :                |
| 6'UDA           |                                | N/A<br>N/A         |
| RANGE<br>Factor |                                | : :                |
| PBA             | 4<br> <br> <br> <br> <br> <br> | N/A<br>N/A         |
| PANGE<br>Factor |                                | : :                |
| NAAP            |                                | N/A<br>N/A         |
| RANGE<br>Factor |                                | 26<br>26           |
| LBAD            |                                | 2.0E-11<br>2.6E-11 |
| RANGE<br>Factor |                                | ;;                 |
| AF G            |                                | N/A<br>N/A         |
| RANGE<br>Factor |                                | 1 :                |
| ANAD            |                                | N/A<br>N/A         |
| No.             | ł                              | • •                |
| Scenario        |                                | SARGC<br>SARVC     |

SUSSER TRACKS TRAVELY TRACKS AND SUSSER TRACKS AND SUSSER

utete ● Type

5-90



100 xxxx

64666666

いるないでした

NUMBER OF STREET, STRE

File: STORHOLD.NK1 Fage 1 Date 21-Aug-B7

TABLE 5-20 (Continued)

INTERIM STORAGE SCENARIOS - NATIONAL/REGIONAL DISPOSAL OPTIONS (FER YEAR)

| RANGE<br>Factor                                                     |                                                                                                                                                                           | 01       | :                        | ł                        | ł                        | ;                 | 01                              | ł                     | 01                       | 01                       | ;                               | 10<br>1                         | 1¢                              | 61                | 01                              | 01                              | 10                |                                                   | 61                | ;                        | ;                        | ;                        | {                 | 61                              | ł                     | 10                       | ú.                       | ł                               | 10                              |
|---------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------|--------------------------|--------------------------|--------------------------|-------------------|---------------------------------|-----------------------|--------------------------|--------------------------|---------------------------------|---------------------------------|---------------------------------|-------------------|---------------------------------|---------------------------------|-------------------|---------------------------------------------------|-------------------|--------------------------|--------------------------|--------------------------|-------------------|---------------------------------|-----------------------|--------------------------|--------------------------|---------------------------------|---------------------------------|
| UNDA                                                                |                                                                                                                                                                           | 1.5F-09  | N/A                      | N/A                      | N/A                      | A/A               | 1.56-08                         | N/A                   | 1.56-08                  | 1.5E-08                  | N/A                             | 1.55-08                         | 1.5E-0B                         | 1.5E-08           | 1.5E-08                         | 1,5E-08                         | 1.5E-08           |                                                   | 7. 9E-07          | N/A                      | A/P                      | N/A                      | N/A               | 7.96-09                         | A/A                   | 5ú-36°L                  | 7.96-09                  | N/A                             | 1.98-69                         |
| RANGE<br>FACTOR                                                     |                                                                                                                                                                           | 10       | 2                        | ł                        | 10                       | ;                 | 10                              | 10                    | 10                       | 10                       | 01                              | 10                              | 9                               | 10                | 9                               | i)                              | 10                |                                                   | ΰI                | 10                       | ;                        | 9                        | ł                 | Ē                               | 10                    | 10                       | 91                       | 9                               | 61                              |
| TEAD                                                                |                                                                                                                                                                           | 1. 4F-10 | 3.66-10                  | N/A                      | 3.6E-10                  | N/A               | 3.6E-10                         | 3.6E-10               | 3.6E-10                  | 3.6E-10                  | 3.6E-10                         | 3.6E-10                         | 3.6E-10                         | 3.66-10           | 3.6E-10                         | 3.6E-10                         | 3. bE-10          |                                                   | 1.9E-10           | 1.9E-10                  | N/A                      | 1.96-10                  | A/A               | 1.96-10                         | 1.96-10               | 1.96-19                  | 1.35-10                  | 1.96-10                         | 1.92-10                         |
| RANGE<br>Factor                                                     |                                                                                                                                                                           | ;        | 9                        | ;                        | 2                        | !                 | 1                               | ;                     | ;                        | ;                        | 91                              | :                               | ł                               | ;                 | ł                               | ł                               | ł                 |                                                   | ;                 | 10                       | ł                        | 10                       | ;                 | 1                               | 1                     | ;                        | 1                        | 10                              | ;                               |
| PUDA                                                                | 1<br>2<br>1<br>1<br>1<br>1                                                                                                                                                | N/0      | 5.9E-08                  | N/A                      | 5.9E-08                  | N/A               | N/A                             | N/A                   | N/A                      | N/A                      | 5.9E-08                         | N/A                             | N/A                             | N/A               | N/A                             | N/A                             | N/A               |                                                   | N/A               | 3. IE-08                 | N/A                      | 3. IE - 08               | N/A               | N/A                             | N/A                   | N/A                      | N/A                      | 3. IE - 98                      | N/A                             |
| RANGE<br>Factor                                                     |                                                                                                                                                                           | :        | ł                        | 1                        | ;                        | !                 | 10                              | 1                     | 10                       | ;                        | ;                               | :                               | :                               | 1                 | 10                              | 10                              | 1                 |                                                   | ł                 | :                        | ł                        | 1                        | ł                 | 10                              | :                     | 01                       | :                        | :                               | ł                               |
| 68 <b>4</b>                                                         | )<br> <br> <br> <br> <br> <br> <br> <br>                                                                                                                                  | N/A      | N/A                      | N/A                      | N/A                      | N/A               | 1.5E-09                         | N/A                   | 1.5E-09                  | N/A                      | N/A                             | N/A                             | N/A                             | A/A               | 1.5E-09                         | 1.56-09                         | N/A               |                                                   | N/A               | N/A                      | N/A                      | N/A                      | N / A             | 7.95-10                         | N/A                   | 7.96-10                  | N/A                      | N/A                             | N/A                             |
| RANGE<br>Factor                                                     | )<br>)<br>)<br> <br>                                                                                                                                                      | :        | ł                        | ł                        | :                        | :                 | ł                               | 10                    | ;                        | ;                        | ;                               | ;                               | ł                               | ;                 | :                               | ł                               | ;                 |                                                   | :                 | :                        | ;                        | :                        | :                 | !                               | 9                     | }                        | :                        | ;                               | ;                               |
|                                                                     |                                                                                                                                                                           |          |                          |                          |                          |                   |                                 |                       |                          |                          |                                 |                                 |                                 |                   |                                 |                                 |                   |                                                   |                   |                          |                          |                          |                   |                                 |                       |                          |                          |                                 |                                 |
| NAAF                                                                |                                                                                                                                                                           | M/A      | N/A                      | N/A                      | A/A                      | N/A               | N/A                             | 4.65-09               | N/A                      | N/A                      | N/A                             | N/A                             | N/A                             | R/A               | N/A                             | N/A                             | N/A               |                                                   | N/A               | N/A                      | N/A                      | N/A                      | N/A               | N/A                             | 2.4E-09               | N/A                      | A.A                      | N/A                             | N/A                             |
| RANGE NAAF<br>Factor                                                | L<br>L<br>L<br>L<br>L<br>L<br>L                                                                                                                                           | N/A      | N/A                      | N/A                      | N/A                      | N/A               | N/A                             | \$.bE-09              | N/A                      | N/A                      | 10 N/A                          | 10 N/A                          | 10 N/A                          | N/A               | 10 N/A                          | 10 N/A                          | N/A               |                                                   | N/A               | N/A                      | N/A                      | N/A                      | N/A               | N/A                             | 2.4E-09               | N/A                      |                          | 10 N/A                          | 10 N/A                          |
| LBAD RANGE NAAF<br>Factor                                           | 1 ( )<br>1 )<br>1 )<br>1 )<br>1 )<br>1 )<br>1 ( )<br>1 ( )<br>1 ( )<br>1 )<br>1 ( )<br>1 )<br>1 ( )<br>1 )<br>1 ( )<br>1 )<br>1 )<br>1 )<br>1 )<br>1 )<br>1 )<br>1 )<br>1 | H/A N/A  | N/A N/A                  | N/A N/A                  |                          | N/A N/A           | N/A N/A                         | N/A 4.6E-09           | N/A N/A                  | N/A N/A                  | 4.5E-09 10 N/A                  | 4.5E-09 10 N/A                  | 4.5E-09 10 N/A                  | N/A N/A           | 4.5E-09 10 N/A                  | 4.5E-09 10 N/A                  | N/A N/A           | contained                                         | N/A N/A           | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A           | N/A N/A                         | N/A 2.4E-09           | N/A N/A                  | N/A N/A                  | 2.4E-09 10 N/A                  | 2.4E-09 10 N/A                  |
| RANGE LBAD RANGE WAAF<br>Factor Factor                              |                                                                                                                                                                           |          | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A           | 10 N/A N/A                      | N/A 4.6E-09           | N/A N/A                  | N/A N/A                  | 4.5E-09 10 N/A                  | 4.5E-09 10 N/A                  | 4,5E-09 10 N/A                  | N/A N/A           | 4.5E-09 10 N/A                  | 4.5E-09 10 N/A                  | N/A N/A           | ire not contained                                 | N/A N/A           | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A           | 10 N/A N/A                      | N/A 2.4E-09           | N/A N/A                  | N/A N/A                  | 2.4E-09 10 N/A                  | 2.4E-09 10 N/A                  |
| AFG RANGE LBAD RANGE NAAF<br>Factor factor                          |                                                                                                                                                                           |          | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A       | 5.3E-10 10 N/A N/A              | N/A N/A 4.6E-09       | N/A N/A N/A              | N/A N/A N/A              | N/A 4.5E-09 10 N/A              | N/A 4.5E-09 10 N/A              | N/A 4.5E-09 10 N/A              | N/A N/A N/A       | N/A 4.5E-09 10 N/A              | N/A 4.5E-09 10 N/A              | N/A N/A N/A       | ainers; fire not contained                        | N/A N/A N/A       | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A       | 2.8E-10 10 N/A N/A              | N/A N/A 2.4E-09       | N/A N/A N/A              | N/A N/A N/A              | N/A 2.4E-09 10 N/A              | N/A 2.4E-09 10 N/A              |
| KAWGE AFG RANGE LBAD RAMGE WAAP<br>Factor factor factor             |                                                                                                                                                                           |          | 10 N/A N/A N/A           | 10 N/A N/A N/A           | 10 N/A N/A N/A           | N/A N/A N/A       | 10 5.3E-10 10 N/A N/A           | N/A N/A 4.6E-09       | 10 N/A N/A N/A           | 10 N/A N/A N/A           | 10 N/A 4.5E-09 10 N/A           | 10 N/A 4.5E-09 10 N/A           | 10 N/A 4.5E-09 10 N/A           | N/A N/A N/A       | 10 N/A 4.5E-09 10 N/A           | 10 N/A 4.5E-09 10 N/A           | N/A N/A N/A       | nto containers; fire not contained                | N/A N/A N/A       | 10 N/A N/A N/A           | 10 N/A N/A N/A           | 10 N/A N/A N/A           | N/A N/A N/A       | 10 2.8E-10 10 N/A N/A           | N/A N/A 2.4E-09       | 10 N/A N/A N/A           | IC N/A N/A N/A           | 10 N/A 2.4E-09 10 N/A           | 10 N/A 2.4E-09 10 N/A           |
| ANAD KANGE APG RANGE LBAD RANGE NAAP<br>Factor factor factor        |                                                                                                                                                                           |          | 7.8E-09 10 N/A N/A N/A   | 7.8E-09 10 N/A N/A N/A   | 7.8E-09 10 N/A N/A N/A   | N/A N/A N/A N/A   | 7.8E-09 10 5.3E-10 10 N/A N/A   | NIA NIA NIA 4.6E-09   | 7.8E-09 10 N/A N/A N/A   | 7.8E-09 10 N/A N/A N/A   | 7.8E-09 10 N/A 4.5E-09 10 N/A   | 7.8E-09 10 N/A 4.5E-09 10 N/A   | 7.8E-09 10 N/A 4.5E-09 10 N/A   | N/A N/A N/A N/A   | 7.8E-09 10 N/A 4.5E-09 10 N/A   | 7.8E-09 10 N/A 4.5E-09 10 N/A   | N/A N/A N/A N/A   | t crash ento containers; fire not contained       | N/A N/A N/A N/A   | 4.2E-09 10 N/A N/A N/A   | 4.2E-09 10 N/A N/A N/A   | 4.2E-09 10 N/A N/A N/A   | N/A N/A N/A N/A   | 4.2E-09 10 2.8E-10 10 N/A N/A   | N/A N/A N/A 2,4E-09   | 4.2E-0? 10 N/A N/A N/A   | 4.2E-09 IO N/A N/A N/A   | 4.2E-09 10 N/A 2.4E-09 10 N/A   | 4.2E-09 10 N/A 2.4E-09 10 N/A   |
| IO MO. ANAD RANGE AFG RANGE LBAD RANGE NAAP<br>Factor Factor Factor |                                                                                                                                                                           |          | I 7.8E-09 IO N/A N/A N/A | 1 7.8E-09 10 N/A N/A N/A | 1 7.8E-09 10 N/A N/A N/A | I N/A N/A N/A N/A | I 7.8E-09 10 5.3E-10 10 N/A N/A | 1 N/A N/A N/A 4.6E-09 | I 7.8E-09 IO N/A N/A N/A | I 7.8E-09 IO N/A N/A N/A | I 7.8E-09 10 N/A 4.5E-09 10 N/A | I 7.8E-09 IO N/A 4.5E-09 IO N/A | 1 7.8E-05 10 N/A 4.5E-09 10 N/A | 1 N/A N/A N/A N/A | I 7.8E-09 10 N/A 4.5E-09 10 N/A | 1 7.8E-09 10 N/A 4.5E-09 10 N/A | I N/A N/A N/A N/A | nrcraft crash ento containers; fire not contained | 2 N/A N/A N/A N/A | 2 4.2E-09 10 N/A N/A N/A | Z 4.2E-09 10 N/A N/A N/A | 2 4.2E-09 10 N/A N/A N/A | Z N/A N/A N/A N/A | 2 4.2E-09 10 2.8E-10 10 N/A N/A | 2 N/A N/A N/A 2.4E-09 | 2 4.2E-0? 10 N/A N/A N/A | 2 4.2E-09 IO N/A N/A N/A | 2 4.2E-09 10 N/A 2.4E-09 10 N/A | 2 4.2E-09 10 N/A 2.4E-09 10 N/A |

o. Dava - Marka Tamana Rarawi reserve davarati kasara kasara kasara bereka berrara berrara tamana berra

File: STORHOLD.WHI Fage 2 Date 21-Aug-97

# TABLE 5-20 (Continued)

INTERIM STORAGE SCENARIOS - NATIONAL/REGIOMAL DISPOSAL OFTIONS (SCENARIOS - NATIONAL/REGIOMAL DISPOSAL OFTIONS

| ~         |
|-----------|
| œ:        |
|           |
| -         |
| -         |
| -         |
| _         |
|           |
| ~         |
|           |
| -         |
| <u>a.</u> |
| _         |
| _         |
|           |
|           |
|           |
|           |
|           |
|           |
|           |
|           |

| RANGE<br>FACTOR         | 01              | 16      | 61       | 10         | 01      |               | 11        | ł       | ;       | 1       | ;        | Ξ        | ;       | 11       | Ξ       | ;        | 11       | Ξ        | Ξ        | Ξ        | Ξ        | Ξ       |             | u l     | ł        | :       | ;       | {        | 01      |
|-------------------------|-----------------|---------|----------|------------|---------|---------------|-----------|---------|---------|---------|----------|----------|---------|----------|---------|----------|----------|----------|----------|----------|----------|---------|-------------|---------|----------|---------|---------|----------|---------|
| UMDA                    | 7.9E-09         | 7.95-03 | 7.96-09  | 30-36°.2   | 7.95-09 |               | 4. 3E-09  | A/A     | 3.4     | N/A     | N/A      | 4. 3E-09 | A/A     | 4. 3E-09 | 4.35-09 | A/A      | 4. 3E-09 | 4. 3E-09 | 4. 3E-09 | 4.35-09  | 4. 3E-09 | 4.36-03 |             | 1.25-09 | N/A      | ₩/₩     | N/A     | A/A      | i.2E-08 |
| RANGE<br>actor          | 16              | 9       | 9        | 10         | 2       |               | Π         | 11      | :       | 11      | ;        | Ξ        | 11      | Ξ        | =       | Π        | Ξ        | Ξ        | Ξ        | Ξ        | 11       | Η       |             | Û.      | 91       | ł       | 10      | ł        | 61      |
| TEAD                    | <b>1.9€-1</b> 0 | 1.96-10 | 1.96-10  | 1.96-10    | 1.96-10 |               | 1. (,E-10 | 1.06-10 | N/A     | 1.96-10 | N/A      | 1.06-10  | 1.06-10 | 1.0E-10  | 1.05-10 | 1.0E-10  | 1.06-10  | 1.06-10  | 1.0E-10  | 1.0E-10  | 1.05-10  | 1.0E-10 |             | 1.4E-08 | 1.4E-08  | R/A     | 1.4E-08 | ¥74      | 1.46-08 |
| RANGE<br>Factor         | ;               | ;       | ;        | ;          | ;       |               | :         | =       | ł       | Ξ       | :        | ;        | ;       | 1        | ł       | Ξ        | 1        | ł        | !        | ;        | ;        | :       |             | ;       | ýl       | ;       | 10      | ;        | !       |
| FUDA                    | N/A             | N/A     | N/A      | N/A        | N/A     |               | N/A       | 1.7E-09 | N/A     | 1.75-08 | N/A      | N/A      | N/A     | N/A      | N/A     | 1.7E-08  | N/A      | N/A      | N/A      | N/A      | N/A      | N/A     |             | N/A     | 9. 9E-09 | N/A     | 9.9E-08 | N/A      | N/A     |
| RANGE<br>Factor         | ł               | :       | 10       | 10         | i       |               | :         | :       | 1       | :       | ;        | ;        | ;       | Π        | ;       | :        | ;        | 1        | ł        | Ξ        | =        | 1       |             | ;       | ;        | ;       | ;       | 1        | 10      |
| FBA                     | N/A             | N/A     | 7.96-10  | 7.96-10    | N/A     |               | N/A       | N/A     | N/A     | N/A     | N/A      | N/A      | N/A     | 4.3E-10  | H/A     | N/A      | N/A      | N/A      | N/A      | 4.35-10  | 4. 3E-10 | N/A     |             | N/A     | N/A      | N/A     | NZA     | A/A      | 1.1E-07 |
| RANGE<br>Factor         | ;               | :       | ;        | ;          | ł       |               | ;         | ;       | :       | ł       | 1        | !        | 11      | :        | ;       | !        | !        | ;        | ;        | 1        | !        | 1       |             | !       | ł        | 1       | :       | ;        | !       |
| HAAP                    | N/A             | N/A     | N/A      | N/A        | N/A     |               | N/A       | N/R     | R/A     | R/A     | N/A      | N/A      | 1.3E-09 | N/A      | N/A     | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A     |             | N/A     | A/R      | N/A     | N/A     | N/A      | N/A     |
| RANGE<br>Factor         | 10              | ł       | 10       | 10         | ;       |               | :         | ł       | ;       | :       | !        | ł        | ;       | 1        | ;       | Η        | 11       | 11       | 1        | 11       | Ξ        | ;       |             | :       | ;        | 1       | ;       | :        | ;       |
| LBAD                    | 2.4E-03         | N/A     | 2.4E-09  | 2.4E-03    | N/A     | ained         | A/A       | N/A     | N/A     | 8/N     | N/A      | N/A      | N/A     | N/A      | N/A     | 1. 35-09 | 1.3E-09  | 1.35-09  | N/A      | 1.35-09  | 1.3E-09  | N/A     |             | N/A     | N/A      | A/A     | A/A     | N/A      | N/A     |
| RANGE<br>Factor         | :               | !       | !        | ;          | :       | ire cont      | ;         | ;       | ;       | ł       | ł        | Ξ        | ł       | :        | ;       | ;        | :        | ;        | ;        | ł        | ;        | ł       | o fire      | ;       | ł        | ;       | ;       | ;        | 01      |
| AF6                     | A'N             | N/A     | N/A      | N/A        | N/A     | uners; f      | N/A       | N/A     | N/A     | R/A     | N/A      | 1.56-10  | N/A     | N/A      | R/A     | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A     | n:srs; n    | N/A     | N/A      | N/A     | N/A     | A/A      | 7.7E-06 |
| FANGE<br>Actor          | 9               | :       | 10       | 10         | ł       | ito conta     | ;         | 11      | Π       | Ξ       | ;        | 11       | ł       | 11       | 11      | Π        | Ξ        | Ξ        | :        | Ξ        | =        | :       | ito conta   | ;       | 01       | 10      | 10      | ;        | 61      |
|                         | 10-30           | N/A     | 4.25-(19 | 4.25-09    | N/A     | crash on      | N/A       | 2.2E-09 | 2.2E-09 | 2.2E-09 | N/A      | 2.25-09  | N/A     | 2.26-09  | 2.2E-09 | 2.2E-09  | 2.2E-09  | 2.2E-09  | N/A      | 2.2E-09  | 2.2E-07  | 8/A     | crash on    | N/A     | 1.2E-09  | 1.25-09 | 1.25-08 | N/A      | 1.2E-09 |
| ANAD                    | -               |         |          |            |         | ~             |           |         |         |         |          |          |         |          |         |          |          |          |          |          |          |         | aft         |         |          |         |         |          |         |
| ) NO. ANAD              | 2 4.            | 7       | ¢.1      | <b>C</b> 4 | ~       | rcraf         | ~         | M       | **      | 57      | ~        | *7       | ••••    | •7       | ~.      | ~        | ~        | ~        | ~        | **       | ~        | m       | 5           | -       | 4        | •       | •       | -        | -       |
| SCENARIO NO. ANAD<br>Id | 5406C 2 4.      | SPOVC 2 | SFRGL 2  | SKRUC 2    | SESVE 2 | Large aircraf | Skeef 3   | SRDHC 3 | SFC6C 3 | SECHC 3 | SKI 6F 3 | SRE4F 3  | SRFVF 3 | SPRVC 3  | SFPEC 3 | SRPHC 3  | SKFVC 3  | SPDEC 3  | SROVC 3  | 58.96C 3 | SGRUC 3  | SGSVF 3 | Seall aircr | SEB55   | SF0HC 4  | SFCBC 4 | SFCHC 4 | 5FA.65 4 | SKEHS 4 |

с., т

3354 • 5377.

5

) )

.





file: STORHOLD.WK1 Fage 3 Date 21-Aug-B7

# TABLE 5-20 (Continued)

INTERIM STORAGE SCENARIOS - NATIONAL/REGIONAL DISPOSAL OFTIONS (FER YEAR)

| FRNGE<br>FRCTOR                               | t             | 10                   | 61                   | ;                        | 10                       | 10                       | Ξ             | 0<br>I                   | <u>01</u>                | ýl            |                                                   | 2             | ł                     | 1                    | i                    | 1             | 01                          | :             | 9                    | Ξ                    | ł                        | 11                       | £:                       | S.              | -                        | 1                        | 3.            |                                              | <u>.</u>      |
|-----------------------------------------------|---------------|----------------------|----------------------|--------------------------|--------------------------|--------------------------|---------------|--------------------------|--------------------------|---------------|---------------------------------------------------|---------------|-----------------------|----------------------|----------------------|---------------|-----------------------------|---------------|----------------------|----------------------|--------------------------|--------------------------|--------------------------|-----------------|--------------------------|--------------------------|---------------|----------------------------------------------|---------------|
| httph                                         | 4/6           | 1.26-(8              | 1.26-09              | N/A                      | 1.2E-08                  | 1.2E-08                  | 1.25-08       | 1.26-98                  | 1.2E-03                  | 1.26-98       |                                                   | 1.35-05       | N/A                   | N.A                  | 4. H                 | A/A           | 1.36-03                     | н .<br>Н      | 1.35-03              | 1.35-09              | 4/4                      | 1. 35-45                 | 1.36-69                  | $1. \times .09$ | 1. 35-03                 | 1.16-09                  | 1.35-09       |                                              | 51-35-8       |
| RANGE<br>FACTOR                               | 01            | 10                   | 61                   | 10                       | 01                       | 91                       | 10            | 01                       | 10                       | 0;            |                                                   | 0             | 10                    | :                    | 2                    | ;             | 9                           | 01            | 19                   | 10                   | <u>e</u>                 | 1                        | 91                       | ol              | 10                       | 10                       | 61            |                                              | Ξ             |
| IEAD                                          | 1.46-08       | 1.45-08              | 1.45-08              | 1.46-08                  | 1.4E-08                  | 1.4E-08                  | 1.45-08       | 1.4E-08                  | 1.4E-08                  | 1.4E-08       |                                                   | 1.55-09       | 1.56-09               | N/A                  | 1.5E-09              | N/A           | 1.56-03                     | 1.5E-09       | 1.55-09              | 1.56-99              | 1.5E-09                  | 50-35°I                  | 1.5E-09                  | 1.56-49         | 1.55-09                  | 1.56-19                  | 1.56-09       |                                              | 1.01-08       |
| FACTOR                                        | ł             | 1                    | ;                    | 10                       | ;                        | ł                        | ;             | ;                        | ;                        | 1             |                                                   | ;             | <u>01</u>             | ;                    | 10                   | ł             | ;                           | ;             | ;                    | :                    | ŝ                        | :                        | 1                        | ;               | 1                        | ;                        | 1             |                                              | ł             |
| FUDA                                          | N/A           | N/A                  | N/A                  | 9.96-08                  | N/G                      | N/A                      | N/A           | N/A                      | N/A                      | N/A           |                                                   | N/A           | 1. IE-08              | N/A                  | 1.1E-08              | N/A           | N/A                         | N/A           | N/A                  | N/A                  | 1.15-08                  | N/A                      | N/A                      | N/A             | N/A                      | A N                      | NZA           |                                              | N/A           |
| RANGE<br>Factor                               | :             | 10                   | ;                    | ;                        | :                        | :                        | ;             | 10                       | 10                       | ł             |                                                   | ;             | ;                     | ;                    | ;                    | :             | 10                          | ;             | 2                    | :                    | ;                        | ;                        | :                        | 1               | 91                       | 10                       | ł             |                                              | ;             |
| FBA                                           | N/A           | 1.15-07              | N/A                  | N/A                      | N/A                      | N/A                      | N/A           | 1.1E-97                  | 1.1E-07                  | N/A           |                                                   | N/A           | N/A                   | N/A                  | N/A                  | N/A           | 1.26-08                     | N/A           | 1.25-08              | N/A                  | N/A                      | N/A                      | N/A                      | A/A             | 1.2E-19                  | 80-31.1                  | N/A           |                                              | N/A           |
| RANGE<br>Factor                               | 01            | ;                    | ;                    | :                        | !                        | :                        | :             | ;                        | :                        | ;             |                                                   | ;             | 1                     | :                    | 1                    | ;             | ;                           | 10            | :                    | :                    | ;                        | ł                        | :                        | ;               | :                        | ;                        | :             |                                              | ;             |
| NAAF                                          | 2. 3E-08      | A/M                  | N/A                  | N/A                      | N/A                      | A/A                      | A/A           | N/A                      | N/A                      | N/A           |                                                   | N/A           | N/A                   | N/A                  | N/A                  | N/A           | N/A                         | 2.4E-03       | 4/H                  | N/A                  | N/A                      | N/A                      | N/A                      | A, N            | 4/H                      | N/A                      | N/A           |                                              | N             |
| RANGE<br>Factor                               | ;             | :                    | 1                    | 10                       | 10                       | 10                       | ;             | Ξ                        | 10                       | :             |                                                   | :             | :                     | :                    | ;                    | :             | ;                           | :             | ;                    | ;                    | 10                       | 10                       | 01                       | ;               | 10                       | ů.                       | ;             |                                              | ł             |
| BAD                                           |               |                      |                      |                          |                          |                          |               |                          |                          |               |                                                   |               |                       |                      |                      |               |                             |               |                      |                      |                          |                          |                          |                 |                          |                          |               |                                              |               |
|                                               | N/A           | A/A                  | N/A                  | 1.86-10                  | 1.86-10                  | 1.6E-10                  | N/A           | 1.86-10                  | 1.86-10                  | N/A           | contained                                         | N/A           | N/A                   | H/H                  | N/A                  | N/A           | N/A                         | 6/N           | N/A                  | N/A                  | 1.96-11                  | 11-35-11                 | 1.96-11                  | N/A             | 11-36-11                 | 1.9£-11                  | N/A           | paure                                        | N/A           |
| RANSE L<br>Factor                             | N/A           | N/A                  | N/A                  | 1.86-10                  | 1.86-10                  | 1.6E-10                  | N/A           | 1.8E-10                  | 1.86-10                  | N/A           | fire not contained                                | N/A           | N/A                   | N/A                  | N/A                  | N/A           | N/N ()                      | N/A           | N/A                  | N/A                  | 1.9E-11                  | 1.56-11                  | 1.96-11                  | N/A             | 1.95-11                  | 1.96-11                  | N/A           | fire contained                               | N/A           |
| AF5 RANGE L<br>Factor                         | N/A N/A       | N/A N/A              | N/A N/A              | N/A 1.8E-10              | N/A 1.8E-10              | N/A 1.6E-10              | N/A N/A       | N/A 1.8E-10              | N/A 1.8E-10              | N/A N/A       | ainers; fire not contained                        | N/A N/A       | N/A N/A               | N/A N/A              | M/A 14/A             | NZA NZA       | 8.26-07 10 N/A              | R/A R/A       | N/A N/A              | N/A N/A              | N/A 1.9E-11              | N/A 1.5E-11              | N/A 1.9E-11              | N/A N/A         | N/A 1.9E-11              | N/A 1.9E-11              | N/A N/A       | ainers; fire contained                       | N/A N/A       |
| RANGE AFG RANGE L<br>Factor Factor            | N/A N/A       | 10 N/A N/A           | 10 N/A N/A           | 10 N/A 1.8E-10           | 10 N/A 1.8E-10           | 10 N/A 1.6E-10           | N/A N/A       | 10 N/A 1.8E-10           | 10 N/A 1.8E-10           | N/A N/A       | nto containers; fire not contained                | N/A N/A       | 10 N/A N/A            | 10 N/A N/A           | 10 M/A 34/A          | N/A N/A       | 10 8.2E-07 10 N/A           | N/A N/A       | 10 N/A N/A           | 10 N/A N/A           | 10 N/A 1.9E-11           | 10 N/A 1.5E-11           | 10 N/A 1.9E-11           | N/A N/A         | 10 N/A 1.9E-11           | 10 N/A 1.9E-11           | N/A N/A       | nto containers; fire contained               | N/A N/A       |
| ANAD RANGE AFS RANGE L<br>Freidr              | N/A N/A N/A   | 1.2E-09 10 N/A N/A   | 1.2E-05 10 N/A N/A   | 1.2E-0B 10 N/A 1.BE-10   | 1.2E-03 10 N/A 1.8E-10   | 1.2E-08 10 N/A 1.6E-10   | 4/A N/A N/A   | 1.2E-08 10 N/A 1.8E-10   | 1.2E-08 10 N/A 1.8E-10   | N/A N/A N/A   | t crash onto containers; fire not contained       | N/A N/A N/A   | L. 3E-1)? 10 N/A N/A  | 1.3E-09 10 N/A N/A   | 1.3E-07 10 N/A N/A   | N/A N/A N/A   | 1.3E-09 10 8.2E-07 10 N/A   | R/A R/A R/A   | 1. 3E -09 10 N/A N/A | 1.3E-09 10 N/A N/A   | 1.3E-09 10 N/A 1.9E-11   | 1.7E-07 10 N/A 1.5E-11   | 1.3E-09 10 N/A 1.9E-11   | N/A N/A N/A     | 1.3E-09 10 N/A 1.9E-11   | L.SE-09 10 N/A 1.9E-11   | N/A N/A N/A   | : crash onto containers; fire contained      | N/A N/A N/A   |
| D ND. ANAD RANGE AFS RANGE L<br>Factor Factor | 4 N/A N/A N/A | 4 1.2E-09 10 N/A N/A | 4 1.2E-05 10 N/A N/A | 4 1.2E-08 10 N/A 1.8E-10 | 4 1.2E-03 10 N/A 1.8E-10 | 4 1.2E-08 10 N/A 1.8E-10 | 4 V/A N/A N/A | 4 1.2E-08 10 N/A 1.8E-10 | 4 1.2E-08 10 N/A 1.8E-10 | 4 N/A N/A N/A | ircraft crash onto containers; fire not contained | 5 N/A N/A N/A | 5 1.3E-1)? 10 N/A N/A | 5 1.3E-09 10 N/A N/A | 5 1.3E-07 10 N/A N/A | 5 N/A N/A N/A | 5 1.3E-09 10 8.2E-07 10 N/A | 5 N/A N/A N/A | 5 1.3E-09 10 N/A N/A | 5 1.3E-09 10 N/A N/A | 5 1.3E-09 10 N/A 1.9E-11 | 5 1.3E-07 10 N/A 1.5E-11 | 5 1.3E-09 10 N/A 1.9E-11 | 5 N/A N/A N/A   | 5 1.3E-09 10 N/A 1.9E-11 | 5 1.56-09 10 8/4 1.96-11 | 5 N/A N/A N/A | scraft crash onto containers; fire contained | 6 N/A N/A N/A |

File: STORHOLD. Mk1 Fage 4 Date 21-Aug-87

TABLE 5-20 (Continued)

INTERIM STORAGE SCEMARIDS - MATIOMAL/REGIDNAL DISPOSAL OPTIONS (PER YEAR)

| 照음!                                                                        |                          |                          |                          |                   | ŝ                               |                       | 0                        | 0                        |                                 | 2                               | 9                               | Ξ                 | ŝ                               | ŝ                               | ů,                |                                                        |                   |                          |                          |                          |                   |                                 |                       | 56                       | 2                        |                                 |                                 |                                  | -                 |
|----------------------------------------------------------------------------|--------------------------|--------------------------|--------------------------|-------------------|---------------------------------|-----------------------|--------------------------|--------------------------|---------------------------------|---------------------------------|---------------------------------|-------------------|---------------------------------|---------------------------------|-------------------|--------------------------------------------------------|-------------------|--------------------------|--------------------------|--------------------------|-------------------|---------------------------------|-----------------------|--------------------------|--------------------------|---------------------------------|---------------------------------|----------------------------------|-------------------|
| FACTO                                                                      | ;                        | !                        | ;                        | !                 | -                               | ;                     | _                        | _                        | ;                               | _                               |                                 | _                 | _                               | -                               |                   |                                                        |                   | !                        | 1                        | ;                        | 1                 |                                 | :                     |                          |                          | ;                               |                                 |                                  |                   |
| UNDA                                                                       | A/N                      | N/A                      | H/A                      | A/A               | 8.56-09                         | H.A                   | <b>8.</b> 5E-05          | 8.5E-99                  | N/A                             | 8.55-09                         | 8.55-05                         | 8.5E-09           | 8.56-09                         | 3. SE-09                        | <b>3.5</b> E-09   |                                                        | 9.46-16           | N/A                      | A/A                      | A/A                      | 4/H               | 5.0E-18                         | A/A                   | 4.96-15                  | 2.7E-15                  | N/A                             | 2.7E-1                          | 2. 7E-15                         | 2.76-15           |
| RANGE<br>FACTOR                                                            | 9                        | ł                        | 10                       | ł                 | i)                              | 9                     | Ē                        | 9                        | 61                              | 10                              | 01                              | 9                 | 01                              | 2                               | 10                |                                                        | 45                | 16                       | 94                       | 1                        | ;                 | ₽£                              | 94                    | 16                       | 96                       | 94                              | 55                              | 54                               | 46                |
| TEAD                                                                       | 1.0E-08                  | N/A                      | 1.95-08                  | N/A               | 1.0E-0B                         | 1.06-08               | 1.0E-08                  | 1,0E-08                  | 1.0E-09                         | 1.06-08                         | 1.0E-08                         | 1.05-08           | 1.05-08                         | 1.0E-08                         | 1.06-08           |                                                        | 9.6E-16           | 5.5E-15                  | 3.2E-15                  | N/A                      | A/A               | 5.0E-16                         | 5. (E-16              | 4.96-15                  | 2.7E-15                  | 2. 7E-15                        | 2.7E-15                         | 2.7E-15                          | 2.7E-15           |
| RANGE<br>Factor                                                            | 10                       | ;                        | 61                       | 1                 | í                               | 1                     | ſ                        | ł                        | 10                              | ł                               | 1                               | {                 | 1                               | {                               | ł                 |                                                        | ł                 | 54                       | ;                        | <b>1</b> 5               | ł                 | ł                               | ł                     | !                        | 1                        | 94                              | ł                               | ł                                | ł                 |
| FUDA                                                                       | 7.16-09                  | N/A                      | 7.1E-08                  | N/A               | N/A                             | A/A                   | N/A                      | N/A                      | 7. IE-08                        | N/N                             | N/A                             | N/A               | N/A                             | N/A                             | N/A               |                                                        | A/A               | 6.2E-13                  | N/A                      | 3.8E-13                  | N/A               | N/A                             | N/A                   | N/A                      | N/A                      | 3.2E-13                         | N/A                             | N/A                              | N/A               |
| RANGE<br>Factor                                                            | ;                        | ;                        | 1                        | :                 | 8                               | :                     | 10                       | !                        | 1                               | 1                               | :                               | :                 | 10                              | 10                              | :                 |                                                        | 1                 | 1                        | !                        | ł                        | 1                 | 86                              | 1                     | 64                       | ł                        | ł                               | 1                               | 1                                | ł                 |
| FBA                                                                        | N/A                      | N/A                      | N/A                      | N/A               | 7.85-08                         | N/A                   | 7.8E~(8                  | N/A                      | NIA                             | N/A                             | N/A                             | N/A               | 7.85-03                         | 7.85-08                         | N/A               |                                                        | N/A               | N/A                      | N/A                      | N/A                      | N/A               | 2.4E-12                         | N/A                   | 1.3E-11                  | N/A                      | N/A                             | N/A                             | 8/N                              | N/A               |
| NGE<br>CTOR                                                                | :                        | :                        | ;                        | !                 | 1                               | 10                    | :                        | ł                        | 1                               | :                               | {                               | 1                 | 1                               | ſ                               | {                 |                                                        | ſ                 | ;                        | {                        | ;                        | {                 | ł                               | 94                    | ;                        | 1                        | {                               | ł                               | ł                                | 1                 |
| RA -                                                                       |                          |                          |                          | ·                 | -                               |                       |                          |                          |                                 |                                 |                                 |                   |                                 |                                 |                   |                                                        |                   |                          |                          |                          |                   |                                 |                       |                          |                          |                                 |                                 |                                  |                   |
| NAAF RA                                                                    | N/A                      | N/A                      | N/A                      | N/A               | N/A                             | 1.65-08               | N/A                      | N/A                      | N/A                             | N/A                             | N/A                             | N/A               | N/A                             | N/A                             | N/A               |                                                        | N/A               | N/A                      | N/A                      | N/A                      | N/A               | N/A                             | 2.4E-12               | N/A                      | N/A                      | N/A                             | N/A                             | N/A                              | N/A               |
| RANGE NAAF RA<br>Factor Fa                                                 | N/A                      | N/A                      | N/A                      | N/A               | N/A                             | 1.6E-i)B              | N/A                      | N/A                      | 10 N/A                          | 10 N/A                          | 10 N/A                          | N/A               | 10 N/A                          | 10 N/A                          | N/A               | ation                                                  | N/A               | N/A                      | N/A                      | N/A                      | N/A               | 4/H                             | 2.4E-12               | N/A                      | N/A                      | 94 K/A                          | 94 N/A                          | 94 N/A                           | N/A               |
| LBAD RANGE NAAF RA<br>Factor Fa                                            | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A           | N/A N/A                         | N/A 1.6E-08           | N/A N/A                  | K/A N/A                  | 1.3E-10 10 N/A                  | 1.3E-10 10 N/A                  | 1.3E-10 10 N/A                  | N/A N/A           | 1.3E-10 10 N/A                  | 1.3E-10 10 N/A                  | N/A N/A           | i no detonation                                        | N/A N/A           | N/A N/A                  | N/A N/A                  | N/A N/A                  | H/A N/A           | N/A N/A                         | N/A 2.4E-12           | N/A N/A                  | N/A N/A                  | 7.8E-12 94 K/A                  | 7.8E-12 94 N/A                  | 7.8E-12 94 N/A                   | N/A N/A           |
| RANGE LBAD RANGE NAAF RA<br>Factor Factor                                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A           | 10 N/A N/A                      | N/A 1.6E-08           | N/A N/A                  | N/A N/A                  | 1.3E-10 10 N/A                  | 1.3E-10 10 N/A                  | 1.3E-10 10 N/A                  | N/A N/A           | 1.3E-10 10 N/A                  | 1.3E-10 10 N/A                  | N/A N/A           | ntainers; no detonation                                | N/A N/A           | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A           | 94 N/A H/A                      | N/A 2.4E-12           | N/A N/A                  | N/A N/A                  | 7.8E-12 94 K/A                  | 7.8E-12 94 N/A                  | 7.8E-12 94 N/A                   | N/A N/A           |
| AFG RANGE LBAD RANGE NAAF RA<br>Factor Factor Fa                           | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A       | 5.5E-06 10 N/A N/A              | N/A N/A 1.6E-08       | N/A N/A N/A              | N/A N/A N/A              | N/A 1.3E-10 10 N/A              | N/A 1.3E-10 10 N/A              | N/A 1.3E-10 10 N/A              | N/A N/A N/A       | N/A 1.3E-10 10 N/A              | N/A 1.3E-10 10 N/A              | N/A N/A N/A       | etrate containers; no detonation                       | N/A N/A N/A       | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A       | 7.8E-14 94 N/A N/A              | N/A N/A 2.4E-12       | K/A N/A N/A              | N/A N/A N/A              | N/A 7.8E-12 94 K/A              | N/A 7.8E-12 94 N/A              | N/A 7.8E-12 94 N/A               | N/A N/A N/A       |
| FANGE AFG RANGE LBAD RANGE NAAF RA<br>-Actor Factor Factor Fa              | 10 N/A N/A N/A           | 10 N/A N/A N/A           | 10 N/A N/A N/A           | N/A N/A N/A       | 10 5.5E-06 10 N/A N/A           | N/A N/A 1.6E-08       | 10 N/A N/A N/A           | 10 N/A N/A N/A           | 10 N/A 1.3E-10 10 N/A           | 10 N/A 1.3E-10 10 N/A           | 10 N/A 1.3E-10 10 N/A           | N/A N/A N/A       | 10 N/À 1.3E-10 19 N/A           | 10 N/A 1.3E-10 10 N/A           | N/A N/A N/A       | iles penetrate containers; no detonation               | N/A N/A N/A       | 94 N/A N/A N/A           | 54 H/A N/A N/A           | 94 N/A N/A N/A           | N/A N/A N/A       | 94 7.8E-14 94 N/A N/A           | N/A N/A 2.4E-12       | 94 K/A K/A N/A           | 94 N/A N/A N/A           | 54 N/A 7.8E-12 94 N/A           | 94 N/A 7.8E-12 94 N/A           | , 94 M/A 7.8E-12 94 N/A          | N/A N/A N/A       |
| ANGO FANSE AFG RANGE LBGO RANGE NAAF RA<br>Factor Factor Factor Factor     | 5.6E-09 10 N/A N/A N/A   | 5.6E-09 10 N/A N/A N/A   | 5.6E-09 10 N/A N/A N/A   | N/A N/A N/A N/A   | 5.6E-07 10 5.5E-06 10 N/A N/A   | N/A N/A N/A 1.6E-08   | 5.6E-09 10 N/A N/A N/A   | 5.6E-09 10 N/A N/A N/A   | 5.6E-09 10 N/A 1.3E-10 10 N/A   | 5.6E-09 10 N/A 1.3E-10 10 N/A   | 5.6E-03 10 N/A 1.3E-10 10 N/A   | N/A N/A N/A N/A   | 5.6E-09 10 N/A 1.3E-10 19 N/A   | 5.6E-09 10 N/A 1.3E-10 10 N/A   | N/A N/A N/A N/A   | ated missiles penetrate containers; no detonation      | N/A N/A N/A N/A   | 1.5E-11 94 N/A N/A N/A   | 9.1E-12 94 N/A N/A N/A   | 9.1E-12 94 N/A N/A N/A   | N/A N/A N/A N/A   | 2.4E-12 94 7.8E-14 94 N/A N/A   | N/A N/A N/A 2.4E-12   | 1.3E-11 94 N/A N/A N/A   | 7.8E-12 94 N/A N/A N/A   | 7.86-12 54 N/A 7.86-12 94 N/A   | 7.8E-12 94 N/A 7.8E-12 94 N/A   | 7.8E-12 , 94 M/A 7.8E-12 94 N/A  | N/A N/A N/A N/A   |
| ND. AMAD FANSE AFG RANGE LBAD RANGE NAAF RA<br>Factor Factor Factor Factor | 6 5.6E-09 10 N/A N/A N/A | 6 5.6E-09 10 N/A N/A N/A | 6 5.6E-09 10 N/A N/A N/A | 6 N/A N/A N/A N/A | 6 5.6E-0? 10 5.5E-06 10 N/A N/A | 6 N/A N/A N/A 1.6E-08 | 6 5.6E-09 10 N/A N/A N/A | 6 5.6E-09 10 N/A N/A N/A | 6 5.6E-09 10 N/A 1.3E-10 10 N/A | 6 5.6E-09 10 N/A 1.3E-10 10 N/A | 6 5.6E-0? 10 N/A 1.3E-10 10 N/A | 6 N/A N/A N/A N/A | 6 5.6E-09 10 N/A 1.3E-10 19 N/A | 6 5.6E-09 10 N/A 1.3E-10 10 N/A | 6 N/A N/A N/A N/A | generated aissiles penetrate containers; no detonation | 7 N/A N/A N/A N/A | 7 1.5E-11 94 N/A N/A N/A | 7 9.1E-12 54 N/A N/A N/A | 7 9.1E-12 94 N/A N/A N/A | 7 N/A N/A N/A N/A | 7 2.4E-12 94 7.8E-14 94 N/A H/A | 7 N/A N/A N/A 2.4E-12 | 7 1.3E-11 94 N/A N/A N/A | 7 7.6E-12 94 N/A N/A N/A | 7 7.8E-12 54 N/A 7.8E-12 94 K/A | 7 7.8E-12 94 N/A 7.8E-12 94 N/A | 7 7.8E-12 \$4 M/A 7.8E-12 94 N/A | 7 N/A N/A N/A N/A |

5



File: STORHULD.WK1 Fage 5 Date 21-Aug 87

6.

STORE COLO

TABLE 5-20 (Continued)

INTERIM STORAGE SCENPPIOS - NATIONAL/REGIONAL DISPOSAL OPTIONS (FER YEAR)

| P SRUGE<br>FACTOR       | -:-<br>-:-       | 5           | -15 34      |                  | ;                | ,       | I.       | H 31-   | -15 96   |             | 15      | 5e -,1 -   | 56 51-   | -15 - 64 | 51-         |              | і.<br>П.    | ;          |          |          |          | т.<br>Т. | :        | 19<br>12   | 21<br>91- | ;          | -1-<br>-1- |  |
|-------------------------|------------------|-------------|-------------|------------------|------------------|---------|----------|---------|----------|-------------|---------|------------|----------|----------|-------------|--------------|-------------|------------|----------|----------|----------|----------|----------|------------|-----------|------------|------------|--|
| 1940                    | 5. E             | Å. 75       | 1.7E        |                  | H. H             | н.<br>Н | 4        | 1.85    | 1.05     | 4 2         | 1.1     | <u>.</u>   | <u>.</u> | ¢.¢Ε     | 6.45        |              | 2.1         | ч <b>.</b> | 11/1     | N.A      | Ч N      | ц;<br>1  | 4-14     | 3.5        | 7. H      | 12         | . IE       |  |
| FANGE<br>FACTOR         | 75               | 1<br>1<br>1 | <b>T</b> (* |                  | 6                | έs      |          | 53      | 55       | 63          | :6      | 55         | 66       | 94       | 34          |              | 1           | ę.         | ł        | 26       | :        | 55       | ;        | 92         | Ë         | ۲,         | £1         |  |
| TEAD                    | 6.2E 15          | 6.2E-15     | 1.7E-15     |                  | 5.96-15          | 3.5E-15 | N/A      | 1.35.1  | 1.0E-15  | 1.06-15     | 1.06-15 | 1.15.15    | 1.0E-15  | 6.6E-15  | 6. 6E 15    |              | 01 - 36 · 3 | 1.85-11    | 5. K     | 2.26-10  | 67H      | 1 SE-10  | N.4      | 01-30.0    | 2. IE-10  | 2. IE - 10 | 2. IE-10   |  |
| FANGE<br>FACTOR         | ;                | :           | !           |                  | 65               |         | 56       | ;       | ŧ<br>;   | 66          | ;       | 1          | ;        | ;        | 1           |              | ł           | 4          | ;        | ÷        | ;        | :        | !        |            | ;         | 52         | ł          |  |
| FUDA                    | N/A              | N/6         | N/A         |                  | 2.36-13          | 4/N     | 1.46-13  | N/4     | N/A      | 3.56-17     | NrA     | N, A       | a/N      | 11, G    | N/A         |              | N/4         | íl-36•1    | N/A      | 2.76-10  | ۲/H      | N/4      | 47.4     | N/A        | N/A       | 2.1E-10    | 4/6        |  |
| RANGE<br>Factor         | <b>7</b>         | 94          | 1           |                  | ;                | ;       | 1        | 65      | ;        | 1           | ł       | 1          | :<br>4   | 94       | <b>7</b> 5  |              | ;           | ;          | ;        | :        | ,        | 36       | ł        | 26         | 1         | 1          | ;          |  |
| FBA                     | 11-3 <u>(</u> ,) | 1.7E-11     | N/A         | t :)             | N/A              | A/A     | N/A      | 4.9E-12 | N/A      | N/A         | N/A     | N/A        | N/G      | 1.96-11  | 1.8E-11     |              | N.A         | N/E        | 4 1      | N/P      | 1/1      | 01-32-10 | N/A      | 2.46-40    | N/A       | A / P      | ⊽/H        |  |
| RANSE<br>FACTOR         | ;                | ;           | ;           | i for rocke      | :                | ł       | ;        | :       | ;        | ;           | 1       | ;          | 1        | :        | ;           |              | :           | ·          | ;        | r<br>R   | !        | 1        | 36       | i          | ;         | ļ          | ;          |  |
| ИААР                    | N/A              | N/A         | N/A         | r ignitio        | N/A              | N/A     | N/A      | N/A     | N/F      | N/A         | N/A     | A/A        | N/6      | e, N     | A/A         |              | N/9         | ₹.N        | N/A      | 4, A     | 21<br>12 | N A      | 1.75-10  | N/9        | 67 N      | A/A        | N/A        |  |
| FACTOF                  | ₹5               | 20          | ;           | n (or moto       | :                | ;       | :        | :       | ;        | 1<br>1<br>1 | 54      | <b>₽</b> 6 | 1        | ÞŚ       | <b>\$</b> 6 |              | :           | !          | ;        | :        | !        | ł        | ;        | ;          | ł         | 26         | 26         |  |
| L 840                   | 1.75-11          | 11-301      | A.A         | detonatio        | N/A              | N/A     | N/A      | 8/N     | 212      | 2.95-12     | 2.96-12 | 2.9E-12    | N/A      | 1.9E-11  | 1.86-11     |              | N/A         | 4/A        | A/A      | N/A      | с 'н     | 4/F      | 6/N      | N/A        | N/A       | 2.1E-10    | ž. 15-10   |  |
| RANGE<br>FACTOR         | ł                | !           | 1           | nta:rers;        | ;                | ;       | <b>)</b> | ;       | :        | !           | !       | 1          | ł        | 1        | ;           |              | :           | ;          | !        | ;        | ;        | 92       | !        | 1          | ļ         | ł          | 1          |  |
| AFG                     | A/A              | н. А        | N:A         | trate co         | N.A              | A/A     | A.N      | N/A     | N/A      | <b>N</b> 7A | A.A     | N/A        | N:A      | A, A     | 4. N        | area         | N B         | NA         | N/A      | A/A      | N/6      | 1.9E-10  | 11/A     | A/A        | A/A       | 4/H        | A/A        |  |
| EANGE<br>FACTOR         | 94               | 94          | ;           | ile pere         | έċ               | 55      | 49       | 66      | 66       | 59          | 56      | 65         | 1        | 94       | 4£          | - polting -  | :           | -n<br>Ci   | -9<br>С  | 26       | :        | 92       | !        | 26         | 25        | 26         | 55         |  |
| - ,                     | =                | 11-7;       | 4           | 64 NI 59         | 3 <b>.6E-1</b> 2 | 5.46-12 | 3.45-12  | 4.75-12 | 31-36-12 | 2.95-12     | 2.9€ 12 | 2.76-12    | N/F      | 1.96-11  | 1.36-11     | lies the t   | ų v         | (:1-38-1:) | 2.26-10  | 2.7E-1i) | N/A      | (1-36-1) | 5,4      | 2. 4E - 10 | 2.16-10   | 2.16-19    | 2.15-16    |  |
| ANAD                    | ىپ<br>1          |             | -           | Ľ,               | • •              |         |          |         |          |             |         |            |          |          |             | ÷            |             |            |          |          |          |          |          |            |           |            |            |  |
| .0 NO. ANAD             | 1 1.1            | 7 1.        | а.<br>г.    | -generat         | 60               | œ       | 8        | 8       | æ        | 8           | c,      | B          | æ        | æ        | æ           | te st        | 0           | "          | 6        | 9        | C.P-     | ¢        | Ċ        | 6          | 6         | ć          | 6          |  |
| SCENASIO NO. ANAD<br>ID | 3.11 / SEB32     | See. 7 1.   | 1 1 Et 515  | -Ioroado-generat | Skuht 8          | SFCEC 8 | SECHC B  | 8 3%24S | 8 JS435  | 8 Shash     | SRFVC G | SF33C 8    | SFRVC 8  | SREGC B  | SFRUG B     | Retearlie st | 5-861 7     | 3F0HC 7    | 5erc80 9 | SPEHC 9  | SFILEF 7 | 5Fi HF 3 | Sct 77 3 | SEMVC 9    | 5FF 5C 9  | E DHd-JS   | 6 J.t.J.5  |  |

5-95

131.441.434.434.434.43

File: STORHOLD.WK1 Page 6 Date 21-Aug-87

# TABLE 5-20 (Continued)

INTERIM STORAGE SCEMARIOS - MATTONAL/REGIONAL DISPOSAL OPTIONS (PER YEAR)

| FANGE<br>Pactor |   | 26<br>26<br>26                |
|-----------------|---|-------------------------------|
| MBA             |   | 1.9E-10<br>1.9E-10<br>3.2E-10 |
| RANGE<br>Factor |   | 3 3 3                         |
| TEAD            |   | 1.9E-10<br>1.9E-10<br>3.2E-10 |
| RANGE<br>Factor |   | :::                           |
| FUDA            |   | N/A<br>N/A<br>N/A             |
| RANGE<br>Factor |   | 26<br>26<br>                  |
| PBA             |   | 1.9E-10<br>1.9E-10<br>N/A     |
| RANGE<br>Factor |   | :::                           |
| NAAP            |   | N/N<br>N/N<br>N/N             |
| RANGE<br>Factor |   | 26<br>26<br>                  |
| LBAD            |   | 1.9E-10<br>1.9E-10<br>N/A     |
| RANGE<br>Factor |   | 111                           |
| AF6             |   | N/A<br>N/A<br>N/A             |
| RANGE<br>Factor |   | 26<br>26<br>                  |
| ANAD            |   | 1.9E-10<br>1.9E-10<br>N/A     |
| NO.             | ł | ~ ~ ~                         |
| SCENARID<br>Id  |   | SRRGC<br>SARVC<br>SASVC       |

TOOOL I

andreis N Serri am



### TABLE 5-20 (Continued)

Interia Storage Barge Only (events/yr)

|          |     | <br> |     |        |
|----------|-----|------|-----|--------|
| ID       |     |      |     | FACTOR |
| SCENARIO | NO. |      | APG | RANGE  |

Large aircraft crash onto holding area; no fire SWKHS 1 2.0E-11 10 Large aircraft crash onto holding area; fire not contained SWKHF 2 1.1E-11 10 Large aircraft crash onto holding area; fire contained SWKHE 3 5.8E-12 11 Small aircraft crash onto holding area; no fire SWKHS 4 3.0E-07 10 Small aircraft crash onto holding area; fire not contained SWKHF 5 3.2E-08 11 Small aircraft crash onto holding area; fire contained SWIKHF 6 2.1E-07 10 Tornado-generated eissile penetrates vault SWKHC 7 2.5E-14 94 Heteorite strikes vault SWKHF 9 1.0E-10 26 Large aircraft crash onto lighter; no fire SWKHS 10 7.7E-12 10 Large aircraft crash onto lighter; fire not contained SWKHE 11 4.1E-12 10 Large aircraft crash onto lighter: fire contained SWKHE 12 2.2E-12 11 Small aircraft crash onto lighter; no fire SWKHS 13 1.2E-07 10 Small aircraft crash onto lighter: fire not contained SWKHE 14 1.25-08 11 Small aircraft crash onto lighter: +ire contained SWKHE 15 6.2E-98 16 Large aircraft crash onto ship: no fire SWEHS 16 3.0E-11 :0 Large aircraft crash onto ship; "ire not contained

14 LAN 47 LAN 147 LA

W. A. W. A. K

### TABLE 5-20 (Continued) Interie Storage Barge Only (events/yr)

|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1.1    |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------|
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| No. and the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| NS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| NG                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| 55 (C)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| <u> </u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| R                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | TABLE 5-20 (Continued)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | THESE J-20 (Conclined)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Interie Storage Barge Only (events/yr)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | _      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | SCENARID NG. APS RANG                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | E      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ID FACTO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | R      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | -      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | SWKHF 17 1.8E-11 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 0      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Large aircraft crash onto ship; fire contain                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | ed     |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | SWKHF 18 9.5E-12 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 1      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Small aircraft crash onto ship; no fire                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | SWKHS 19 4.8E-07 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Ú      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Small aircraft crash noto shint fire not con                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | tained |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | SWKNE 20 5.1E-08 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 1      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Small sircraft crash onto chin; fire contain                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | eri    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Δ.<br> |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 3WKRC 21 5.42-07 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | •      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| MA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| K                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| 25                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 5-98                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |        |
| 23                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | J~30                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| r<br>tra                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| C/-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| Ref.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |        |
| and the second sec | المراجع العرابية المراجع فالمراجعة بالأزار الراجع والمراجع                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |        |
| MEANING CARDEN DATA CARACATA CARACATA CARACATA CARACATA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | and a tall of the solution of the second of | and    |

A CONTRACTOR OF A REAL OF A REAL OF A DESCRIPTION OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A 

1.1.1



The trends indicated by the frequency results are as follows:

### Externally-Induced Events

- 1. Tornado and high wind
  - a. Munitions stored outdoors or in warehouses are generally more susceptible to tornado strikes. APG, PBA, NAAP, TEAD, and UMDA have warehouses. PBA and NAAP are in Tornado Zone I while APG is in Tornado Zone II (Zone I has the highest tornado frequency). TEAD and UMDA are in Tornado Zone III.
  - b. The transportation containers provide some protection to the munitions which are temporarily stored in open holding areas.

### 2. Meteorite strike

- a. Munitions stored in warehouses are more susceptible to meteorite strikes. Since fire is generally present, a meteorite strike may involve the entire warehouse inventory.
- b. The frequency of breaching ton containers in OFCs or barge packages in the holding area is of the same order of magnitude as the unpackaged ton containers stored in warehouses (at NAAP and UMDA). However, the OFC provide the spray tanks an additional layer of protection than spray tanks in warehouses (at TEAD) which are normally stored in their overpacks.

### 3. Aircraft crashes

- a. Munitions stored outdoors are generally more susceptible to these events. APG, PBA, and TEAD have ton containers stored outdoors. However, the aircraft crash probabilities at APG and PBA are relatively higher than the other sites.
- b. Igloos provide minimal protection from direct crashes of large aircraft. The accident becomes more serious when burstered munitions are involved.
- c. Large aircraft crash frequencies at APG, LBAD, and TEAD greatly increase for the air option because of the additional landings and takeoffs at these sites.
- d. The OFCs and barge packages do not provide additional protection to the munitions from direct aircraft crashes.

### 4. Earthquakes

- Earthquakes, particularly in high seismic locations such as TEAD, could cause stacked munitions to be punctured. However, the probability of having a probe present inside an igloo is quite low.
- b. Detonations due to earthquake-induced drops are at least two orders of magnitude less likely than punctures.
- c. There is a significantly high frequency earthquakeinduced agent releases to munitions stored in warehouses at NAAP, TEAD, and UMDA.



### Leaker-Related Events

- Forklift drop accidents can occur more frequently than forklift time puncture accidents.
- Use of a lifting beam instead of a time leads to an order of magnitude decrease in drop frequency.





### 5.5. UNCERTAINTY ANALYSIS

### 5.5.1. Overview

The frequency results presented in Tables 5-16, 5-17, and 5-18 are median values. The values shown in the range factor column represent the ratios of the 95<sup>th</sup> percentile values to the median values. The range factors vary from 10 to almost 100. The tornado frequency results have the highest uncertainties, largely because of the difficulty to accurately model the probability that the missile will be in the proper orientation to penetrate the munition and how many missiles per square foot of wind will actually be present. The ability to model low-impact detonations also leads to large uncertainties in the final results. The data available are scarce and sometimes not directly applicable to the scenario being analyzed.

### 5.5.2. Error Factors

In those cases where sufficient information exists to determine the upper and lower bound values, the error factor was derived by assuming that the upper bound value is equivalent to the 95<sup>th</sup> percentile. The engineers' best estimate is taken as the median value based on the properties of the lognormal distribution. This choice is rather conservative, since the mean value of the resulting distribution becomes larger than the best estimate or recommended value.

In many cases, however, the data sources were limited. Therefore, the assignment of error factors was entirely based on engineering judgment, taking into consideration the important parameters which may influence a particular variable. The generic guidelines for the uncertainty assessment is shown in Table 5-21.

5.5.2.1. <u>Tornado Sequence Uncertainties</u>. The frequency of the initiating event itself (i.e., tornado wind of sufficient intensity to

### TABLE 5-21 GENERIC UNCERTAINTY MODELS

- External events (both from natural causes and human-caused events external to the operation, e.g., aircraft crash): EF = 10.
- Component or equipment failure rates were generally assigned an error factor of 3. An exception to this rule is when the analyst does not feel confident with the applicability of the data to a particular demil equipment, component, or operation. In such case, a larger error factor was used, ranging from 5 to 10.
- In cases where the event probability range from 0.1 to 0.9, and was derived largely from engineering judgment, the error factor used is:

| Probability: | 0.1 to 0.3 | EF = 2.0 |
|--------------|------------|----------|
| Probability: | 0.4 to 0.6 | EF = 1.5 |
| Probability: | 0.7 to 0.8 | EF = 1.4 |
| Probability: | 0.9        | EF = 1.0 |

Munition failure probability due to puncture that was calculated using standard mathematical models was assigned an error factor of 5. generate missiles occurs) is assigned an error factor of 10, per Table 5-19. The conditional probability of a missile's hitting the structure and penetrating the munition is assigned an error factor of 50. As explained in Section 5.2.1.1 (Eq. 5-2), this event is the product of four variables. The uncertainty is largely due to the variable  $D_e$  which is the number of missiles per square foot of wind. The conditional probability of a burstered munition's detonating when hit by a missile is assigned an error factor of 2.

5.5.2.2. <u>Meteorite Strike Sequence Uncertainty</u>. The frequency of a meteorite strike is assigned an error factor of 10. The conditional probability of a meteorite's penetrating and rupturing the munition is the product of (1) fraction of stone and iron meteorites capable of penetrating the target; (2) target area; and (3) spacing factor. This event is assigned an error factor of 10. The uncertainty is largely due to the fraction of stone and iron meteorites capable of penetrating the structure.

5.5.2.3. <u>Aircraft Crash Sequence Uncertainties</u>. The aircraft crash frequency is assigned an error factor of 10. Aircraft crash accident sequences with or without fires (from impact) have been considered. For this reason no uncertainties were assigned to either the probability of having a fire (0.45) or no fire (0.55). The uncertainties associated with the structural damage (i.e., igloo or warehouse) given an aircraft crash are given in Table 5-8. For events with probabilities greater than 0.1, the uncertainties assigned followed the guidelines given in Table 5-19.

### 5.5.2.4. Earthquake Sequence Uncertainties

### Storage Igloos

The initiating event, earthquake occurs, is assigned an error factor of 10. The conditional event, munition punctured given a



drop, is assigned an error factor of 5. The puncture probability is a function of drop height, weight and pressure of a probe of sufficient length and density. The uncertainty is largely due to the last variable. Note also that no uncertainty from errors with the models has been considered, since this is beyond the state-of-the-art of presentday uncertainty analysis.

### Warehouse Storage

Event 1: Earthquake Occurs

The initiating event frequency is assigned an error factor of 10. Event 2: "K" Warehouses Damaged by Earthquake

Uncertainty factors for values above 0.1 are taken from Table 5-21. For probabilities between 0.01 and 0.1, an uncertainty factor of 3 is recommended. Probabilities below  $10^{-2}$  are assigned an uncertainty factor of 3. The uncertainty distribution in each case is lognormal with a median equal to P<sub>2</sub>. Recall that P<sub>2</sub> is the independent warehouse damage probability, given an earthquake.

Event 3: Munitions Damaged in "L" Warehouses

If munition damage results from building collapse, the uncertainty in Event 3 is negligible because the analysts are very confident (i.e., essentially certain) that munition damage occurs. If the warehouse remains intact, the uncertainty in Event 3 is dominated by the uncertainty in  $P_p$  - the conditional probability that a fallen container is punctured. From Table 5-21 the uncertainty distribution is lognormal with an uncertainty factor of 5 and a median equal to the point estimate for  $P_p$ . Event 4: Ignition at "M" Warehouses

The ignition probability is a function of  $P_{\rm OSP}$  and  $P_{\rm EL}$ , that is, the probability that offsite power is available following the quake, and that an electrical fault occurs. The uncertainty in these probabilities was quantified using the methodology reported in the Zion PRA. Moreover, the data used to quantify the uncertainty in  $P_{\rm OSP}$  also comes from the Zion study.

The major uncertainty in  $P_{EL}$  is due to the application of a generic Modified Mercalli fragility model to the warehouses. Depending upon the actual, as-built design features, the median failure threshold can vary by a factor of 2 about the nominal value. Thus, an uncertainty factor of 2 was applied to the uncertainty in the failure threshold.

Event 5: Ignition at Warehouse with Damaged Munitions

All parameters and distributions required to quantify the uncertainty in Event 5 are presented in the Event 4 analysis.

5.5.2.5. <u>Handling Accident Sequence Uncertainties</u>. All initiating events associated with munitions handling (i.e., drops, collisions, forklift time punctures) were assigned an error factor of 10. The conditional probability of puncturing the munitions given any one of the initiating events is assigned an error factor of 3. The probability of causing a low-impact detonation (i.e., drop from 6 ft or lower) is assigned an error factor of 10.

### 5.6. REFERENCES

- 5-1. Science Applications International Corporation, "Probabilities of Selected Hazards in Disposition of M55 Rockets," U.S. Army Toxic and Hazardous Materials Agency, M55-CS-2, November 1985.
- 5-2. "Supporting Calculations for the Risk Analysis of the Continued Storage of Chemical Munitions," GA Technologies Inc., (Issue 2), Volumes 1, 2, and 3, August 1987.
- 5-3. Department of the Army, U.S. Army Materiel Development and Readiness Command, "Safety Manual," DARCOM-R-385-100, August 1981.
- 5-4. Coats, D. W., "Natural Phenomena Hazards Modeling Project: Extreme Wind/Tornado Hazards Models For Department of Energy Sites," UCRL-53526, Lawrence Livermore National Laboratory, February 1984.
- 5-5. Lawrence Livermore National Laboratory, "Explosives Handbook, Properties of Chemical Explosives and Explosive Simulants," March 16, 1981.
- 5-6. Telephone conversation: R. Perry (OPMCM) and C. Everline (GA), June 15, 1987.
- 5-7. "Status Report on Packaging Concepts for the Transportation of Chemical Munitions," presented by MITRE Corporation to OPMCM, June 17, 1987 (modified on June 26, 1987).
- 5-8. Blake, V. E., "A Prediction of the Hazards from Random Impact of Meteorites on the Earth's Surface," Sandia Laboratories SC-RR-68-838, December 1968.
- 5-9. Solomon, K. A., K. C. Erdmann, and D. Okrent, "Estimate of the Hazards to a Nuclear Reactor from the Random Impact of Meteorites," <u>Nuclear Technology</u>, Vol. 25, January 1975.
- 5-10. Clarke, R. K., et al., "Severities of Transportation Accidents," SCA-74-0001, Sandia National Laboratories, July 1976.
- 5-11. Weals, F. H., and B. Finder, "ESKIMO V Magazine Separation Test," NUC-TP-6076, Naval Weapons Center, China Lake, February 1979.



- 5-12. MacGorman, D. R., et al., "Lightning Strike Density fot the Contiguous United States From Thunderstorm Duration Records," U.S. NRC Report NUREG/CR-3759, April 1984.
- 5-13. Flood Plain Information Part 11 Arkansas River, Caney Bayou, and Tributaries; Department of the Army, Corps of Engineers, Vicksburg, Miss., November 1973.
- 5-14. Bell, B. J., et al., "Human Reliability and Sensitivity Analyses Supporting the Chemical Munitions Demilitarization Risk Assessment," Battelle Columbus Division, July 10,1987.
- 5-15. GA Technologies Inc., "Risk Analysis of the Continued Storage Option," GA Draft Report GA-C18564, Rev. 1, November 1986.
- 5-16. U.S. Army Armament, Munition, and Chemical Command, "Quarterly Chemical Munitions Leaker Report (RCS-RCQA-123, Classified)," October 15, 1985, January 1986.
- 5-17. "Cost Estimates Prepared in Support of the Transportation Concept Plan," Draft Report prepared by MITRE Corporation, June 1987.
- 5-18. "Independent Evaluation/Assessment of Rocket, 115-mm: Chemical Agent (GB or VX), M55," U.S. Army Materiel Systems Analysis Activity (AMSAA), October 1985.
- 5-19. Meeting at OPMCM, Aberdeen, Maryland on August 13-14, 1987 (Attendees: OPMCM, ORNL, JBF Associates, H&R Technical Associates, Battelle, and GA staff).

### 6. SCENARIO LOGIC MODELS FOR HANDLING

The objectives of this section are to: (1) define those activities considered as "handling" in the analysis; (2) address the assumptions and data that have been used to evaluate the handling accident scenarios; (3) present the analytical structure of the evaluation; and (4) discuss the quantification of the accident scenarios.

Section 3 provides an overview of how munitions are handled at the sending site for transfer to the collocation sites (NDC or RDC) and at the NDC or RDC itself prior to the demilitarization operations. The activities associated with the handling of munitions at the original storage site and at the receiving site (NDC or RDC) are diagramed in overview form in Fig. 3-1. In brief the Army's plan is to package the munitions that are to be transported outside an installation's boundaries in offsite transportation containers to protect them against impact, crush, puncture, fire, and immersion while being transported. The munitions will be subjected to many handling operations during their movement to the railhead, airstrip or loading dock for offsite transport by rail, air or sea. Upon arrival at the receiving site, handling operations include unloading, movement to an interim storage facility, and unpackaging from offsite containers. The collocated munitions and the items originally stored at the receiving site would ultimately be subject to packaging in onsite containers, and finally movement by truck to the demilitarization facility. At the disposal facility, handling operations include unpacking the transport containers and transfer of the munitions to the materials handling equipment within the plant. For this study, movement by forklifts is considered to be a handling operation rather than transportation. However, onsite truck transportation is considered a transportation operation.

6-1

### 6.1. GENERAL HANDLING PROCEDURES AND ASSUMPTIONS

### 6.1.1. Rail Option

Although there may be some slight differences in the munition handling procedures at each site, for this analysis the following general assumptions were made and are intended to apply to all the sites, as appropriate:

- Forklifts are used to move munition pallets for short distances. Electric forklifts are used inside storage igloos, warehouses, maintenance facilities, storage facilities, MHIs or MDBs. Fossil fueled forklifts are used outside these facilities.
- 2. A forklift will handle one pallet or container at a time.
- 3. A forklift equipped with a lifting beam is used to move and carry the ton containers.
- 4. Ton containers will have been tested ultrasonically to determine susceptibility to leak development in the plug and value area during transportation. The ton containers indicating potential leak development will have both their values and plugs replaced with plugs. The handling activities associated with these operations are considered "preparatory" procedures and are not part of this risk analysis. Further, it is assumed that the ton containers will not leak thereafter and this analysis does not address handling of leaking ton containers.
- 5. The mines will be transported with their fuzes still in the drums.

6. The spray tanks and weteye bombs will not be removed from their overpack for placement inside the offsite transportation container. These items are handled with forklift with tines.

- 7. Munition pallets or single items will be placed directly inside the offsite transportation container using the storage area electric forklift, except for the ton containers stored in outdoor storage yards (at APG and PBA) where a diesel forklift is used. The transportation container will be already loaded and secured on a 40-ft flatbed truck parked just outside the storage facility (igloo apron, warehouse or storage yard's entrance).
- 8. The offsite transportation package inner container is 72-in. inner diameter, 90-in. outer diameter and 18-ft long. The outer container is an 8 x 8 x 20 ft steel iso-container. The package is designed to provide the munitions with maximum protection from impact, crush, puncture, and fire. It is capable of providing a seal.
- 9. The 4.2-in. mortars and 105-mm cartridges will have their propellant removed prior to the start of the demil campaign and these munitions will be handled as palletized projectiles. The handling activities associated with the propellant removal operations are considered "preparatory" procedures and are not part of this risk analysis.
- 10. Munitions found to be leaking prior to transport during igloo monitoring and inspection will be treated before further movement. The activities associated with the treatment of leakers which have developed during storage were considered as part of the interim storage activities and are not part of the handling accidents presented in this section.

10.55 The 642 (1999) In the second of the form of the

-36°-36

- 11. At the holding area, the containers are unloaded one at a time from the truck using standard Cargo Handling Equipment (CHE) similar to a piggypacker. This is a four-wheeled munition handling equipment that lifts the containers from the top. Loading procedures are similar.
- 12. The containers will not be stacked while in the holding area.
- 13. This analysis addresses leaking munitions which may have developed during onsite transportation, at the storage igloo, or at the holding area. At the holding area, daily low-level monitoring will be performed on the transportation containers for munitions leakage. The containers found to have leaking munitions will be brought by truck to a leakers processing facility.
- 14. At the leakers processing facility, the transportation container will be brought to an inner area where the leaker is identified, removed from the pallet, decontaminated and repacked in an overpack. For the purpose of this analysis, once overpacked the leakers are like all other munitions and no further specific handling activities need be considered.

- 15. A maximum of 16 ONC containers will be stored inside the MHI. The containers will not be stacked.
- 16. In the UPA there could be as many as six onsite transportation containers at any given time. The onsite containers will not be stacked.

There is a possibility of transporting arriving munitions directly from the rail car to the MDB. If this was the case, there will be less handling operations involved than what is assumed in this analysis. However, the analysis conservatively assumes movement from the rail to the interim storage areas.

### 6.1.2. Air Transport Option

The procedures and assumptions listed in Section 6.1.1, except for those dealing with munition types other than ton containers, projectiles, and rockets, apply to the air transport option handling. Items 5, 6, and 9 do not apply because only these three munition types will be airlifted.

### 6.1.3. Marine Option

For offsite marine shipment, the only munition type is the ton container. Thus, the procedure items number 5, 6, and 9 in Section 6.1.1 do not apply here. Another difference is that the ton container will be shipped in a vault instead of an offsite transport container (OFC). This affects items 7 and 8 which are amended to read as follows:

- 7a. Ton containers for marine shipment will be placed in a vault positioned just outside the storage facility (storage yard entrance) using forklifts.
- 8a. The vault is designed to provide the munitions with maximum protection from impact, crush, puncture, and fire. Vaults are handled with equipment which lifts the vaults from the top, such as a forklift with lifting beams (not with times).

Another difference is that the receiving site risk in terms of agent release only (not in public effects) can be assumed to be the same as disposal of the APG stocks at TEAD.



### 6.2. CHRONOLOGY OF HANDLING OPERATIONS

The handling operations were categorized primarily into two groups: (1) handling operations (HC, HA, and HW for rail, air, and water transport, respectively) between the storage facilities and the offsite transporter at the sending site, and between the offsite transporter and the MHI at the receiving site and (2) handling operations at the facility (HF), including movement from the MHI to the MDB entrance and then to the UPA. A third category of handling operations was also considered: handling at the igloo prior to onsite transport at the MHI and unloading at the MHI. These operations served as the basis for the identification of relevant handling accident initiating events presented in Section 4.

Handling operations specific to the sending sites are the placement of munitions inside offsite transportation containers and the handling of leakers found to have developed during onsite transportation from their storage area to the holding area. Handling operations specific to the receiving site for collocated munitions are the leakers processing facility-related handling operations for leakers which have developed during offsite transportation, the unloading of munitions from the offsite containers and the loading of munitions (both those that were collocated and those that were originally stored at the site) in the onsite containers at the storage area.

The accident scenario analysis also addressed both leaking and nonleaking munitions. Army experience on the movement of various munitions suggests that rockets, MC-1 bombs, and ton containers will more likely leak during transport. However, the valves and plugs of ton containers will be tested and replaced, if necessary, prior to offsite movement. Thus, only additional handling operations necessary to isolate and overpack leaking rockets and bombs are evaluated both at the sending and receiving sites.



A flow diagram of the handling operations at both the sending and receiving sites is shown in Figs. 6-1 and 6-2. The handling steps at the sending sites are as follows:

- For rail and air transport options, an electric forklift picks up a pallet of munitions inside the storage area, carries it to a truck with an OFC already secured on it, and places the pallet inside the container. Ton containers stored in open yards are normally moved to the truck by a diesel forklift.
- 2. For the marine transport option, the ton containers will be placed in the transportation package positioned at the storage yard's entrance, using a diesel forklift with lifting beams.
- The 40-ft flatbed truck transports the container to the holding area about one mile away.
- 4. A cargo handling equipment (CHE) picks up the OFC from the truck and places it in the holding area.
- 5. When ready for shipment, a CHE picks up the OFC at the holding area and places it on a train car for rail transport, or on a truck for air or marine transport.
- 6. At the air strip, the OFC is loaded into the aircraft by conveyor. At the marine loading dock, the OFC is loaded by crane into the barge, and the barge is loaded by crane later into the LASH.
- 7. At the sending site, the leaker isolation operations are as follows: after evidence of a leak in the offsite transportation container at the holding area, the container is loaded on a truck for transport to the leaker processing facility (LPF). At the LPF, the container is unloaded from the truck

6-7



見たいためでい



using a CHE and is brought to an inner area of the facility. After identification of the pallet containing the leaker, an electric forklift picks up the pallet from the container and unloads it on the floor of the facility. The leaker is removed from the pallet, carried to a PIG or overpack ready for reception of the leaker, and placed inside it. These operations are done manually for rockets and using an electric forklift for bombs.

The handling steps at the receiving sites are as follows:

- 1. For rail transport, after arrival of the train at the destruction site (NDC or RDC), a CHE immediately unloads the containers from the train car and places it in a holding area, where the container is kept and periodically monitored for agent for up to 14 days.
- 2. For air transport, the OFC is offloaded from the plane by conveyor onto a truck and transported to the holding area.
- 3. For sea transport, this analysis assumes risk is the same as at TEAD for unloading rail transport vehicles.
- 4. A CHE picks up the offsite transportation container at the holding area and loads it on a flatbed truck.
- 5. The truck transports the container to the storage igloo about one mile away.
- 6. Upon arrival at the storage igloo, the transportation container stays on the truck and is monitored prior to being opened. An electric forklift rolls up to the truck, picks up the pallet of munitions from the container and moves it directly into the storage igloo. The spray tanks arriving at



the storage igloo of the receiving site are in their overpack inside the offsite transportation container. They are removed from the offsite container like all the other munitions at the storage facility and remain in their overpack.

- 7. When ready for demilitarization, an electric forklift picks up the pallet inside the storage igloo and loads it into an onsite transportation container (single cylindrical container) located immediately outside the facility (igloo apron). Two exceptions are the spray tank and weteye bombs, which are not placed in an ONC but which are handled in their overpacks using forklifts.
- A diesel forklift with a lifting beam picks up the onsite transportation container at the storage facility apron and loads it on a flatbed truck.
- 9. The truck transports the onsite container to the munitions holding igloo (MHI) located one mile away.
- 10. At the MHI, a diesel forklift with lifting beams picks up the container from the truck and places it at the igloo apron.
- 11. An electric forklift picks up the container at the MHI apron and places it inside the MHI to await demilitarization.
- 12. When ready for further processing, an electric forklift picks up the container inside the MHI and brings it to the MHI apron.
- 13. A diesel forklift picks up the container at the MHI apron and carries it to the MDB elevator where it is taken to the second level.

6-11
14. An electric forklift takes the container out of the elevator and moves it to the UPA.

15. Whenever a leaker is suspected (during periodic monitoring activities at the holding area or at the storage facility), the truck transports the offsite transportation container to a leakers processing facility for leaker isolation. The leakers processing facility at the receiving site is capable of handling contaminated offsite transportation containers.

Based on these handling procedures, the number of operations for each scenario is calculated.

6.3. ACCIDENT SCENARIOS FOR HANDLING ASSOCIATED WITH RAIL TRANSPORT

According to the Master Logic Diagram (Section 4), there were four types of initiating events which could lead to agent release: munition drop, forktine puncture, forklift collision, and leakage. The list was further expanded to specific accident sequences to address conditions such as (1) where the accident occurs (i.e., storage area, leakers processing facility, etc.); (2) munition configuration (i.e., handled as pallets or singularly); (3) the presence of any packaging (i.e., bare or in transportation container); and (4) whether it applies only to leaking or nonleaking munitions. This resulted in the identification of six families of initiating events for handling, as given in Table 4-3.

Event tree logic models were developed for the first five of these six families of initiating events, as shown in Figs. 6-3 through 6-7. Leakage scenarios were analyzed without using logic models. For each tree, the scenario begins with the disruptive occurrence at a specified location and munition configuration; the subsequent events, which affect whether or not agent is released or how much is released, were then developed.

The initiating events for the accident scenarios evaluated are largely due to operator error. Except for forklift collision accidents in which the frequency data used was derived from industry data which already incorporated human error contribution to the overall event frequency, a human reliability task analysis was performed as described below to determine the occurrence of such events as dropping of munitions, forklift time accidents, etc. The forklift collision frequency is 4.3 x 10<sup>-6</sup> per operation.

The event tree sequences for the onsite handling operations (HC), related to the movement of munitions to various locations, are coded

6-13



0.00000



Event tree for drop of munitions(s) during handling at facility Fig. 6-3.



6-14

| INITIATIN<br>CONFIGURATION | LOCATION                                   | DETONATION<br>AVOIDED | MUNITION<br>INTACT | SEQUENCE | SEQUENCE<br>FREQUENCY | TYPE OF<br>AGENT RELEASE    |
|----------------------------|--------------------------------------------|-----------------------|--------------------|----------|-----------------------|-----------------------------|
| PALLET OR ONC              | STORAGE IGLOO                              | I                     |                    | · · ·    | <u> </u>              | NONE                        |
|                            | 3 x 10 <sup>-5</sup> /3 x 10 <sup>-6</sup> |                       |                    | нсі      |                       | EVAPORATION                 |
|                            |                                            |                       |                    | •        |                       | DETONATION +<br>EVAPORATION |
|                            | OUTDOORS                                   | ~                     |                    |          |                       | NONE                        |
|                            | 3 x 10 <sup>5</sup> /3 x 10 <sup>-6</sup>  |                       |                    | HC5      |                       | LIQUID SPILL                |
|                            |                                            |                       |                    | HC22     |                       | DETONATION +<br>SPILL       |
|                            | мні                                        | ~                     | <u></u>            |          |                       | NONE                        |
|                            | 3 x 10 <sup>-5</sup> /3 x 10 <sup>-6</sup> |                       |                    |          |                       | EVAPORATION                 |
|                            |                                            |                       |                    | HC11     |                       | DETONATION +<br>EVAPORATION |
| INGLE MUNITION             | LPF                                        | ~                     |                    |          |                       | NONE                        |
|                            | 6 x 10 <sup>-4</sup>                       |                       |                    | HC18     |                       | EVAPORATION                 |
|                            |                                            |                       |                    | НСЗО     |                       | DETONATION                  |
| OFC                        | OUTDOORS                                   | ~                     |                    |          |                       | NONE                        |
|                            | 3 x 10 <sup>-5</sup> /3 x 10 <sup>-6</sup> |                       |                    | HC8      |                       | LIQUID SPILL                |
|                            |                                            |                       |                    | HC23     |                       | DETONATION +<br>SPILL       |
|                            | LPF                                        | ~                     | ~                  |          |                       | NONE                        |
|                            |                                            |                       |                    | HC17     |                       | EVAPORATION                 |
|                            |                                            |                       |                    | НС29     |                       | DETONATION +<br>EVAPORAION  |
|                            |                                            |                       |                    |          |                       |                             |
|                            |                                            |                       |                    |          |                       |                             |





Fig. 6-4. Event tree for drop of munition(s) during handling operations other than at facility

Fig. 6-5. Event tree for forklift tine punctures during handling

والأحراق والأحراقي والأرقاق

•

1.1.1. BUEN

| INITIATIN                  | G EVENT                 | MULTION | VENTILATION           | SEQUENCE | SEQUENCE  | TYPE OF       |
|----------------------------|-------------------------|---------|-----------------------|----------|-----------|---------------|
| CONFIGURATION<br>PUNCTURED | LOCATION<br>OF PUNCTURE | INTACT  | INTACT                | 0        | FREQUENCY | AGENT RELEASE |
| BABE MINITION              | STORAGE IGLOO           |         |                       |          |           | NONE          |
|                            |                         |         | 1<br>1<br>1<br>1<br>1 | HC3      |           | EVAPORATION   |
|                            |                         |         |                       |          |           | NONE          |
|                            |                         |         |                       |          |           | NEGLIGIBLE    |
|                            | 1 × 10                  |         |                       | HC19     |           | EVAPORATION   |
|                            | MUR                     |         |                       |          |           | NONE          |
|                            |                         |         |                       |          |           | NEGLIGIBLE    |
|                            | 5×10 ~                  |         |                       | HF9      |           | EVAPORATION   |
|                            | REFORE FACILITY         |         |                       |          |           | NONE          |
|                            |                         |         |                       | HC14     |           | EVAPORATION   |
| <u></u>                    |                         |         |                       |          |           | NONE          |
|                            | 1 × 10 <sup>-5</sup>    |         |                       | HF4      |           | EVAPORATION   |
|                            | -                       |         | <br> <br> <br> <br>   |          |           |               |
|                            |                         |         |                       |          |           |               |

6-16

------

| -         |
|-----------|
| 1 Section |
|           |
|           |



Event tree for vehicle collisions during handling at facility Fig. 6-6. 6-17

| 1.1    | ÁD-A1:<br>UNCLA | 93 355<br>55IFIE | Che<br>The<br>Ca<br>D Saf | MICAL<br>DISP<br>A N I<br>Peo-Cdi | STOCK<br>DSAL O<br>BARSEL<br>E-IS-8 | PILE C<br>F Chem<br>L et r<br>7006 d | ISPOSI<br>I. (U)<br>IL. AU(<br>AAA15- | AL PRO<br>GA TE<br>3 87 G<br>-85-d- | GRAM R<br>Chnolo<br>R-C-18<br>9022 | ISK AN<br>GIES 1<br>563 | IALYSI<br>INC SAI<br>F/G : | 5 OF<br>N DIEG<br>15/6.3 | i0 4/ | '13 |
|--------|-----------------|------------------|---------------------------|-----------------------------------|-------------------------------------|--------------------------------------|---------------------------------------|-------------------------------------|------------------------------------|-------------------------|----------------------------|--------------------------|-------|-----|
|        |                 |                  |                           |                                   |                                     |                                      |                                       |                                     |                                    |                         |                            |                          |       |     |
|        |                 |                  |                           |                                   |                                     |                                      |                                       |                                     |                                    |                         |                            |                          |       |     |
|        |                 |                  |                           |                                   |                                     |                                      |                                       |                                     |                                    |                         |                            |                          |       |     |
|        |                 |                  |                           |                                   |                                     |                                      |                                       |                                     |                                    |                         |                            |                          |       |     |
|        |                 |                  |                           |                                   |                                     |                                      |                                       |                                     |                                    |                         |                            |                          |       |     |
|        |                 |                  |                           |                                   |                                     |                                      |                                       |                                     |                                    |                         |                            |                          |       |     |
|        |                 |                  |                           |                                   |                                     |                                      |                                       |                                     |                                    |                         |                            |                          |       |     |
| $\sim$ |                 |                  |                           |                                   |                                     |                                      |                                       |                                     |                                    |                         |                            |                          |       |     |



BSS STATE RELEASES

1. 2. 1 . 1. 1.



Y

なななないです。と言語であるというな

3

### Fig. 6-7. Event tree for vehicle collisions during handling other than at facility

Keererood, Possoorer

differently from those at the demilitarization facility (HF). The complete list of sequences is shown in Table 6-1. Due to recent programmatic changes, a few sequences have now become obsolete and are no longer applicable. For example, onsite container is not handled with a forklift with times as had been previously assumed (delete HCl4). An electric forklift is always used at the leaker processing facility, which deletes sequence HC20. For facility-related handling operations (HF), 14 sequences were identified and are shown in Table 6-1. The applicability of these sequences to the specific munitions stored at each site is also shown in Table 6-1. For clarity, the onsite handling accident sequences (HC) are listed for both the sending site and the receiving site, since differences in agent release dispersion will apply depending on which site the accident occurs.

### 6.3.1. Human-Reliability Analysis for Handling Operations

Ś

0.0000014

TELEVICE SCRUTTERS DECEMBER 1000001444

A human-reliability analysis (HRA) was performed in support of the handling operations analysis. This section discusses the objective of the HRA, the methodology used, the task analysis performed, the errors described, and the quantification of those errors.

6.3.1.1. <u>Objective</u>. The objective of the human-reliability analysis of the munitions handling operations is to identify, define, and quantify operator errors that could lead to agent release to the environment. The handling operations examined consist of all handling activities at the sending and receiving sites that take place before the demilitarization operations. These include all activities involving loading and unloading munitions, moving munitions with forklifts<sup>\*</sup> and by hand, and packing and unpacking munition pallets. The equipment and personnel involved and the order in which the events occur are based on site visit



<sup>\*</sup>For this study, forklifts and other rubber-tire vehicles performing the same functions as forklifts are referred to as forklifts, and no difference in the error probabilities assigned to these various vehicles is assumed.

200000000 NVEM "FYSSYNYNYNYNYNYNYN "ECCECECE" - ECCECECE

TABLE 6-1 LIST OF ACCIDENT SCENARIOS (HC AND HF) COLLOCATION OPTION - RAIL TRANSPORTATION

|      |                                                                                         | Applicable<br>Munition | Munition Con                           | figuration                                |
|------|-----------------------------------------------------------------------------------------|------------------------|----------------------------------------|-------------------------------------------|
|      | Scenarios Description                                                                   | Types                  | At Sending Site                        | At Receiving Site                         |
| HCI  | Drop of bare pallet or single item at storage<br>area                                   | All                    | Spray Tank (ST) in<br>overpack         | ST in overpack                            |
| HC2  | Forklift collision with short duration fire<br>at storage area involving bare munitions | TC only                | Ton container (TC)<br>only at APG, PBA | Not applicable                            |
| нсз  | Forklift tine accident involving bare muni-<br>tions at storage area                    | All but TC             | ST+overpack                            | ST+overpack at<br>storage area and<br>MHI |
| HC4  | Forklift collision accident without fire at<br>storage area involving bare munitions    | AII                    | ST+overpack                            | ST+overpack                               |
| HC5  | Drop of onsite container                                                                | All                    | Not applicable                         | ST+overpack                               |
| HC6  | Forklift collision with short duration fire<br>during handling of onsite container      | <b>A11</b>             | Not applicable                         | ST+overpack                               |
| HC 7 | Forklift collision without fire during han-<br>dling of onsite container                | All                    | Not applicable                         | ST+overpack                               |
| HC8  | Drop of offsite container                                                               | All                    | ST+overpack+0FC                        | ST+overpack+OFC                           |
| HC9  | Collision accident with short duration fire<br>during handling of offsite container     | AII                    | ST+overpack+0FC                        | ST+overpack+0FC                           |
| HC10 | Collision accident without fire during han-<br>dling of offsite container               | A11                    | ST+overpack+0FC                        | ST+overpack+0FC                           |
| HCII | Drop of bare palletized munition leads to<br>detonation                                 | Burstered              | Pallet                                 | Pallet                                    |

いたいたい

うちたいとうない 「たちちちち」 かいていたい 「「「ちちちちち」」 こうちょうの しょう ほうちょう しょう しょうしょう



MALONG CALCULATE STREET STREET





### TABLE 6-1 (Continued)

|         |                                                                                                            | Applicable<br>Munition                    | Munition Con    | figuration        |
|---------|------------------------------------------------------------------------------------------------------------|-------------------------------------------|-----------------|-------------------|
|         | Scenarios Description                                                                                      | Types                                     | At Sending Site | At Receiving Site |
| HC12    | Forklift collision accident at storage area<br>leads to detonation of burstered munition                   | Burstered                                 | Bare            | Bare              |
| HC13(a) | Forklift collision accident without fire at<br>maintenance facility                                        | 4.2-in.<br>mortar,<br>105-mm<br>cartridge | Ваге            | Bare              |
| HC14    | Forklift tine accident involving munitions in<br>onsite container                                          | All but TC                                | Not applicable  | ONC               |
| HC17    | Drop of pallet containing a leaking munition<br>during leaker isolation operations at LPF                  | Rockets,<br>bombs(a)                      | Pallet          | Pallet            |
| HC18    | Drop of single leaking munition in vapor con-<br>tainment room of leakers processing facility              | Rockets,<br>bombs(a)                      | Bare            | Bare              |
| HC19    | Forklift tine puncture during leaker isola-<br>tion operations                                             | Rockets,<br>bombs(a)                      | Bare            | Bare              |
| HC20    | Collision accident with short duration fire<br>during handling of leaking munition<br>(munition in pallet) | Rockets,<br>bombs(a)                      | Pallet          | Pallet            |
| HC21    | Collision accident without fire during han-<br>dling of leaker                                             | Rockets,<br>bombs <sup>(a)</sup>          | Bare            | Bare              |

6-21

ሃል እና እና

6643623686863645

8. 8 a. 8. Mate. 9.

*222* 2260 2

SSSS TOSS

Ň

| -        |
|----------|
|          |
| .0       |
| e        |
| 2        |
| <b>C</b> |
| •••      |
| L.       |
| <u> </u> |
| 5        |
| ~~       |
| <u> </u> |
| $\sim$   |
|          |
| -        |
|          |
| Ý        |
|          |
| 61       |
|          |
| Ξ.       |
| -        |
| ≤.       |
|          |

|      |                                                                                                          | Applicable<br>Munition | Munition Con    | figuration        |
|------|----------------------------------------------------------------------------------------------------------|------------------------|-----------------|-------------------|
|      | Scenarios Description                                                                                    | Types                  | At Sending Site | At Receiving Site |
| HC22 | Drop of munition in onsite container leads to<br>detonation                                              | Burstered              | Not applicable  | ONC               |
| HC23 | Drop of munition in offsite container leads<br>to detonation                                             | Burstered              | OFC             | OFC               |
| HC24 | Collision accident during munition handling<br>in onsite container leads to detonation due<br>to impact  | Burstered              | Not applicable  | ONC               |
| HC25 | Collision accident during munition handling<br>in offsite container leads to detonation due<br>to impact | Burstered              | OFC             | OFC               |
| HC26 | Collision accident in onsite container with prolonged fire leads to thermal detonation                   | All                    | Not applicable  | ST+overpack       |
| HC27 | Collision accident in offsite container with prolonged fire leads to thermal detonation                  | AII                    | ST+overpack+0FC | ST+overpack+0FC   |
| HC29 | Drop of pallet containing leaker leads to<br>detonation                                                  | Rockets(a)             | Pallet          | Pallet            |
| HC30 | Drop of single leaking munition leads to<br>detonation                                                   | Rockets(a)             | Bare            | Bare              |

6-22

1.1



TABLE 6-1 (Continued)

|      |                                                                                                           | Applicable<br>Munition | Munition Con    | nfiguration            |
|------|-----------------------------------------------------------------------------------------------------------|------------------------|-----------------|------------------------|
|      | Scenarios Description                                                                                     | Types                  | At Sending Site | At Receiving Site      |
| HC31 | Collision accident involving a leaker leads<br>to detonation due to impact                                | Rockets <sup>(a)</sup> | Bare            | Bare                   |
| HC32 | Failure to detect a leak in the offsite con-<br>tainer                                                    | Rockets,<br>bombs(a)   | OFC             | OFC                    |
| HF1  | Munition dropped during movement from the MHI<br>to the MDB                                               | <b>A</b> 11            | Not applicable  | ST+overpack            |
| HF2  | Bare single munition dropped during handling inside the MDB                                               | <b>111</b>             | Not applicable  | ST without<br>overpack |
| HF3  | Forklift collision accident with short dura-<br>tion fire during handling from MHI to MDB                 | A11                    | Not applicable  | ST+overpack            |
| HF4  | Forklift tine accident during handling from<br>MHI to MDB                                                 | ST only                | Not applicable  | ST+overpack            |
| HFS  | Forklift collision accident with prolonged<br>fire during handling from MHI to MDB leads to<br>detonation | All                    | Not applicable  | ONC                    |
| HF7  | Collision accident without fire during move-<br>ment from the MHI to the MDB                              | A11                    | Not applicable  | ST+overpack            |
| HF8  | Munition dropped inside the MDB (in onsite container)                                                     | All                    | Not applicable  | ST+overpack            |
| HF9  | Forklift tine accident occurs inside the MDB                                                              | ST only                | Not applicable  | ST+overpack            |
| HFIO | Forklift collision accident without fire inside the MDB                                                   | A11                    | Not applicable  | ST+overpack            |

6-23

TABLE 6-1 (Continued)

f

|      |                                                                                                          | Applicable<br>Munition | Munition        | Configuration     |
|------|----------------------------------------------------------------------------------------------------------|------------------------|-----------------|-------------------|
|      | Scenarios Description                                                                                    | Types                  | At Sending Site | At Receiving Site |
| HF11 | Munition pallet dropped during movement from<br>the MHI to the MDB leads to detonation                   | Burstered              | Not applicable  | ONC               |
| HF12 | Bare single munition dropped during handling<br>inside the MDB leads to detonation                       | Burstered              | Not applicable  | Bare              |
| HF13 | Palletized munition in onsite container<br>dropped during handling inside the MDB leads<br>to detonation | Burstered              | Not applicable  | Pallet            |
| HF14 | Collision accident from MHI to MDB leads to<br>detonation due to impact                                  | Burstered              | Not applicable  | ONC               |

(a)Leakers developing during transportution from storage igloo to holding area.

a de la compañía de la compañía de la decaración de casa de la casa de la compañía de compañía de la compañía d

ELLER PRESS NUMBER





observations, telephone conversations, and reviews of documents including "Transportation of Chemical Agents and Munitions: A Concept Plan" (Ref. 6-1) and the list of GA's handling assumptions (Ref. 6-2).

6.3.1.2. <u>Methodology</u>. The approach used for the human-reliability analysis is similar to the one used for plant operations (described in Plant Operations, Chapter 9). First, a task analysis was performed to identify those errors that could potentially impact agent release probabilities. Those errors were categorized according to the human operations involved; usually, no munition-specific differences were cited. Available data were used to quantify the probabilities of some of these errors, and extrapolations were made from these fixed data to quantify the remainder. Conservative error factors were selected to account for the uncertainty associated with the data, the models, the extrapolations, and site-specifics.

6.3.1.3. <u>Task Analysis</u>. A task analysis was performed to identify credible human errors associated with the handling operations. The sequence of handling events related to rail transport on which this task analysis was based is described in Sections 6.3.2 and 6.3.3. Figures 6-2 and 6-3 schematically represent the various handling steps. Section 9.2 contains the task-analysis table that shows precisely which human errors were identified as applicable to each operation.

All of the handling operations analyzed are performed with forklifts or by hand. Electric forklifts are used inside storage igloos, warehouses, leakers processing facilities, storage facilities, MHIs and MDBs to move single munitions and pallets between the inside of the building and its apron or loading dock. Diesel forklifts are used for moving single munitions, pallets, and transportation containers between the apron or loading docks and trucks and for movement elsewhere outside. Larger forklifts, referred to as container handling equipment (for example, a "piggypacker"), are used to move transport containers (onsite and offsite types). Forklift tines are used to lift pallets and



6-25

spray tanks inside their overpacks. Forklift lifting beams are used to lift ton containers and transportation containers.

6.3.1.4. <u>Human-Error Description</u>. Six types of operator errors were identified in the task analysis: (1) puncturing a munition with a forklift tine, (2) dropping a munition or pallet from a forklift, (3) dropping a single munition while hand-carrying it, (4) damaging a munition or munitions in a forklift collision, (5) failing to detect a leaking munition which has developed during transportation, and (6) replacing a ton container valve or plug improperly. These errors are described in the following paragraphs:

- Puncturing a munition with a forklift time might occur any time a munition or pallet is approached with a forklift time. Puncture probability is a function of the human error that results in impact of the time with the munition and of the vulnerability of the munition to such an impact.
- 2. Dropping a munition or pallet from a forklift could occur any time a forklift is carrying a load (single munitions, pallets, TCs, spray tanks, package containers, etc.). This action could be caused by operating the forklift in a way that causes the load to fall or by loading the forklift such that the load is misaligned or the weight distribution within the pallet or the package container is unbalanced. It could also result from the pallet's getting caught on and pulled off by something it has run into. Sudden acceleration or deceleration, sharp turns, high-speed operation, or operation over uneven ground could all be contributors to munition drops.
- 3. Dropping a munition while hand-carrying it may occur any time the munition is picked up, put down, or carried without using a forklift or other lifting device. It could be caused by the

6-26



operator's falling as he carries the munition or by the munition's slipping from his grasp.

- 4. A forklift colliding with another vehicle or with a fixed structure is a credible human-error event, since a human is at the controls at the time of the collision. However, the data available does not distinguish between collisions caused by human error and those caused by mechanical failure. Since the two are accounted for in the collision probability estimate, the human-error factor will not be counted again by quantifying it separately in the human-reliability analysis.
- 5. Failing to detect that there is a leaking munition in a transportation container is probable every time the operator fails to check the monitor on the container before opening it (since there may or may not be a leaker inside).
- 6. Improperly replacing a value or a plug on a TC involves operations that will be conducted before TCs are transported. The TCs frequently have been found to be severely corroded around the brass fill and drain values and on the threaded plugs installed in the container ends. This replacement is outside the scope of normal plant operation but will be discussed as a separate case in Appendix J.

6.3.1.5. <u>Human Error Probability Estimation</u>. Section 9.2 discusses the human error probability estimation for the handling accidents. Much of the data is based on Ref. 6-4.

### 6.3.2. Data and Results

Tables 6-2 and 6-3 present the input data used for the accident frequency analysis. The basis for the initiating events frequencies has been discussed in the Human Reliability Analysis Section. Given the

### TABLE 6-2 INITIATING EVENTS FREQUENCIES (HANDLING ASSOCIATED WITH RAIL TRANSPORT)

### INITIATING EVENTS FREQUENCIES

|        | INITIATING EVENT                                                                                | FREQUENCY<br>Events/op | ERROR<br>Factor | REFERENCE<br>(NOTES)                            | APPLICABLE<br>SCENARIOS                                                                       |
|--------|-------------------------------------------------------------------------------------------------|------------------------|-----------------|-------------------------------------------------|-----------------------------------------------------------------------------------------------|
| *****  |                                                                                                 | (Col.H)                | (Col.I)         |                                                 | #===##################################                                                        |
| HE10   | Pallet or single item dropped during<br>handling of non-leaking munition<br>outside the NDB (1) |                        |                 |                                                 | HC1,HC5,HC8,HC11,HC22,HC23<br>HF1,HF11                                                        |
| HEIQA  | Iteas lifted with times                                                                         | 3.0E-05                | 10.0            | 6.7                                             |                                                                                               |
| HE10B  | Items lifted with lifting beams                                                                 | 3.0E-06                | 10.0            | 8                                               |                                                                                               |
| HE15   | Pallet or container dropped during<br>handling of non-leaking munition<br>inside the MDB (2)    |                        |                 |                                                 | NF8, HF 13                                                                                    |
| HE15A  | Items lifted with tines                                                                         | 1.5E-04                | 10.0            | 7                                               |                                                                                               |
| HE 158 | Items lifted with lifting beams                                                                 | 1.5E-05                | 10.0            | 8                                               |                                                                                               |
| HE 20  | Pallet or container dropped during handling of leaking munition (3)                             |                        |                 |                                                 | HC17,HC29                                                                                     |
| HE20A  | Items lifted with tines                                                                         | 3.0E-04                | 10.0            | 7                                               |                                                                                               |
| HE20B  | Items lifted with lifting beams                                                                 | 3.0E-05                | 10.0            | 8                                               |                                                                                               |
| HE25   | Single munition dropped inside the MDB (4)                                                      | 3.0E-04                | 10.0            |                                                 | HF2,HF12                                                                                      |
| HE35   | Single leaking munition dropped (3)                                                             | 6.0E-04                | 10.0            |                                                 | HC18,HC30                                                                                     |
| HE40   | Forklift tine accident involving<br>munition handling outside the MDB (1)                       | 1.0E-05                | 10.0            |                                                 | HC3,HF4                                                                                       |
| HE 45  | Forklift tine accident involving<br>munition handling inside the NDB (2)                        | 5.0E-05                | 10.0            |                                                 | HF9                                                                                           |
| HE50   | Forklift tine accident involving handling of leaking munition (3)                               | 1.0E-04                | 10.0            |                                                 | NC19                                                                                          |
| HE55   | Vehicle collision accident                                                                      | 4.3E-06                | 10.0            | GA derive<br>data, see<br>details i<br>Appendix | HC2,HC4,HC6,HC7,HC9,HC10,<br>HC12,HC21,HC24,HC25,HC26,<br>HC27,HC31,HF3,HF5,HF7,<br>KF10,HF14 |
| HE65   | Failure to detect a leak in the                                                                 | 1.0E-03                | 10.0            |                                                 | HC32                                                                                          |

~1`~ ● ~~~ TABLE 6-2 (Continued)

### transportation container

NOTES:

(1) Handled by forklift or other handling equipment; operators wearing street clothes with mask

TATATAR TANANG SALAMAT NAVANA TAN

- (2) Handled by forklift; operators wearing mask, gloves, and boots; excluding ton container
- (3) Operators in level A clothing
- (4) Handled singly by hand; operators wearing mask, gloves and boots.
- (6) 3.0e-5 = 3\*10-5
- (7) For all items lifted with times (spray tanks in overpacks and bare munitions)
- (8) Items lifted by a lifting beam or by a cargo handling equipment

Service Service

Decessus

**REALESS** 

TABLE 6-3CONDITIONAL EVENTS PROBABILITIES (RAIL TRANSPORT)

33787

### CONDITIONAL EVENTS PROBABILITIES

|         | EVENT SEQUENCE                                                                              | EVENT<br>ROBABILIT | ERROR<br>Factor | REFEREI | NCEAPPLICABLE<br>SCENARIO |
|---------|---------------------------------------------------------------------------------------------|--------------------|-----------------|---------|---------------------------|
| HE100   | Palletized or single munition<br>punctured given a drop outside<br>the MDB (Drop ht = 6ft.) |                    |                 |         | HC1,HF1(for SW)           |
| HE1008  | Boab                                                                                        | 1.02E-03           | 3.0             | See Ga  | calc                      |
| HE100D  | 4.2-in Nortar                                                                               | 2.67E-04           | 3.0             | sheets  |                           |
| HE100C  | 105-se Cartridge                                                                            | 4.73E-05           | 3.0             | lRef.   | )                         |
| HE100K  | Ton Container                                                                               | 3.34E-03           | 3.0             |         |                           |
| HE100M  | Nine (in drums)                                                                             | 2.00E-04           | 3.0             |         |                           |
| HEIDOP  | 155-mm Projectile                                                                           | 0.00E+00           |                 |         |                           |
| HE1000  | 8-in Projectile                                                                             | 0.002+00           |                 |         |                           |
| HE100R  | Rocket                                                                                      | 7.95E-04           | 3.0             |         |                           |
| HE100SW | Spray Tank (with overpack)                                                                  | 8.63E-03           | 3.0             |         |                           |
| HE110   | Offsite container and munition pun                                                          | ctured             |                 |         | HC8,HF1                   |
|         | (4ft drop)                                                                                  | ner                |                 |         |                           |
| HE1108  | Boeb                                                                                        | 1.93E-03           | 3.0             |         |                           |
| HE1100  | 4.2-in Mortar                                                                               | 1.02E-03           | 3.0             |         |                           |
| HEIIOC  | 105-se Cartridge                                                                            | 7.30E-04           | 3.0             |         |                           |
| HEIIOK  | Ton Container                                                                               | 1.83E-03           | 3.0             |         |                           |
| HEILON  | Mine (in drums)                                                                             | 1.00E-03           | 3.0             |         |                           |
| HE110P  | 155-mm Projectile                                                                           | 8.40E-04           | 3.0             |         |                           |
| HE1100  | 8-in Projectile                                                                             | 8.40E-04           | 3.0             |         |                           |
| HEILOR  | Rocket                                                                                      | 1.06E-03           | 3.0             |         |                           |
| HE110SW | Spray Tank (with overpack)                                                                  | 7.10E-04           | 3.0             |         |                           |
| HE120   | Container and munition punctured g                                                          | i ven              |                 |         | HC9,HC10                  |
|         | drop of the offsite container (2ft                                                          | drop)              |                 |         |                           |
| HE120B  | Baab                                                                                        | 1.38E-03           | 3.0             |         |                           |
| HE120D  | 4.2-in Nortar                                                                               | 8.60E-04           | 3.0             |         |                           |
| HE120C  | 105-mm Cartridge                                                                            | 4.20E-04           | 3.0             |         |                           |
| HE120K  | Ton Container                                                                               | 1.10E-03           | 3.0             |         |                           |
| HE120H  | Nine (in drums)                                                                             | 8.50E-04           | 3.0             |         |                           |
| HE120P  | 155-am Projectile                                                                           | 2.00E-04           | 3.0             |         |                           |
| HE1200  | 8-in Prajectile                                                                             | 2.00E-04           | 3.0             |         |                           |
|         |                                                                                             |                    |                 |         |                           |



TABLE 6-3 (Continued)

| HE120R<br>HE120SH | Rocket<br>Spray Tank (with overpack)                                                                 | 9.00E-04<br>4.40E-04 | 3.0<br>3.0 |                                        |
|-------------------|------------------------------------------------------------------------------------------------------|----------------------|------------|----------------------------------------|
| HE140             | Palletized or single munition<br>punctured given a drop resulting<br>from collision (Drop ht = 2ft.) |                      |            | HC2,HC4,HC21<br>HF3,HF7,HF10 (SW only) |
| HE 1408           | Roeb                                                                                                 | 3.94E-04             | 3.0        |                                        |
| HE1400            | 4.2-in Hortar                                                                                        | 1.87E-04             | 3.0        |                                        |
| HE 140C           | 105-an Cartridge                                                                                     | 4.57E-06             | 3.0        |                                        |
| HE140K            | Ton Container                                                                                        | 1.662-03             | 3.0        |                                        |
| HE140M            | Nine (in druas)                                                                                      | 1.60E-04             | 3.0        |                                        |
| HE140P            | 155-an Projectile                                                                                    | 0.00E+00             | •          |                                        |
| HE1400            | 8-in Projectile                                                                                      | 0.00E+00             |            |                                        |
| HE140R            | Rocket                                                                                               | 7.16E-04             | 3.0        |                                        |
| HE140SH           | Sorav Tank (with overpack)                                                                           | 6.31E-03             | 3.0        |                                        |
|                   |                                                                                                      |                      | • • •      |                                        |
| HE150             | Palletized munition in onsite contai                                                                 | ner                  |            | HFB_HC5                                |
|                   | puncture given a drop of container                                                                   |                      |            | )                                      |
|                   | (Drop ht = 4ft., also applies to han                                                                 | dling in UPA         | )          |                                        |
|                   | •                                                                                                    |                      |            |                                        |
| HE1508            | Baeb                                                                                                 | 3.508-04             | 3.0        |                                        |
| HE1500            | 4.2-in Mortar                                                                                        | 3.50E-04             | 3.0        |                                        |
| HE150C            | 105-an Cartridge                                                                                     | 4.00E-05             | 3.0        |                                        |
| HE150K            | Ton Container                                                                                        | 7.20E-04             | 3.0        |                                        |
| HE150H            | Nine (in drums)                                                                                      | 4.80E-04             | 3.0        |                                        |
| HE 150P           | 155-se Projectile                                                                                    | 6.00E-05             | 3.0        |                                        |
| HE150Q            | 8-in Projectile                                                                                      | 6.00E-05             | 3.0        |                                        |
| HE150R            | Rocket                                                                                               | 2.70E-04             | 3.0        |                                        |
|                   |                                                                                                      | :                    |            | NCE 1103                               |
| HE190             | Pailetized or single munition in ons                                                                 | lte                  |            | nco,nc/                                |
|                   | container punctured given drop resul                                                                 | ting                 |            |                                        |
|                   | from collision (24t drop)                                                                            |                      |            |                                        |
| HE 1608           | Bosh                                                                                                 | 1.00E-04             | 3.0        |                                        |
| HEILOO            | 4.2-in Nortar                                                                                        | 3.00E-04             | 3.0        |                                        |
| HEILOC            | 105-em Cartridge                                                                                     | 0.00E+00             | •••        |                                        |
| HEILOK            | Ton Container                                                                                        | 3.30E-04             | 3.0        |                                        |
| HE 160M           | Nine (in drues)                                                                                      | 4.40E-04             | 3.0        |                                        |
| HEILOP            | 155-ae Projectile                                                                                    | 0.00E+00             | 2          |                                        |
| HE 160D           | 8-in Projectile                                                                                      | 0.00E+00             |            |                                        |
| HEILOR            | Rocket                                                                                               | 2.60E-04             | 3.0        |                                        |
| ********          |                                                                                                      |                      |            |                                        |

### TABLE 6-3 (Continued)

SALAR ISTITUTE SALAR

HF2, HFB (HE2505W) KE 250 Single bare munition punctured given drop in UPA (Drop ht = 4ft.)3,0 3.50E-04 HE250B Boab 0.00E+00 HE250D 4.2-in Mortar HE250C 105-em Cartridge 0.00E+00 2.80E-03 3.0 HE250k Ton Container 3.0 HE250H Hine (in drums) 8.82E-05 0.00E+00 HE250P 155-mm Projectile 0.00E+00 HE2500 B-in Projectile 3.0 5.93E-04 HE250R Rocket 1.512-02 3.0 HE250SD Spray Tank (no overpack) 7.87E-03 3.Ú HE2505W Spray Tank (with overpack) HE3, HE19, HE4 (5W) HE400 Munition punctured by HE9 (SH) forklift times 1.29E-02 3.0 HE4008 BORD 3.68E-02 3.0 HE400D 4.2-in Mortar 8.90E-03 3.0 HE400C 105-mm Cartridge N/A 3.0 HE400K Ton Container 7.07E-02 3.0 HE400H Mine (in drums) 5.00E-03 3.0 HE400P 155-mm Projectile 5.008-03 3.6 HE4000 8-in Projectile 2.63E-01 3.0 HE400R Rocket HE4005W Spray Tank (with overpack) 1.53E-02 3.0 7.25E-02 10.0 See ADD F HC2. HCb. HC9. HE550 Fire results from vehicle HC26, HC27, HF3, HF5 collision See App F HC7.HC10. 9.27E-01 none HE555 Collision does not cause HF7 fire HC2.HC6.HC9.HF3 HE560 Fire contained HE560A 4 min - Burstered munitions 5.00E-01 none HE5608 30 min - Non burstered munitions 1.00E+00 none HE560C >15 min - On/Offsite container 1.00E+00 none HE26.HE27.HF3 HE570 Fire not contained 5.00E-01 none HE570A 4 min - Burstered munitions 0.00E+00 HE570B 30 min - Non burstered munitions none

6<sub>3</sub>9



NATALAN



| HE570C | >15 min - On/Offsite container                                                              | 0.00E+00 | none |                                             |
|--------|---------------------------------------------------------------------------------------------|----------|------|---------------------------------------------|
| HE590  | Munition in on/offsite container<br>detonates or ruptures given prolonged<br>fire (>15 min) | 1.00E+00 | NONE | HC26,HC27,HF5                               |
| HE 500 | Munition detonates given drop (6ft)<br>or collision (per munition)                          | 9.50E-09 |      | HC11,HC12<br>HC29,HC30,HC31 (R)             |
| HE600D | 4.2-in Mortar (48)                                                                          | 4.56E-07 | 10.0 |                                             |
| HE600C | 105-am Cartridge (24)                                                                       | 2.28E-07 | 10.0 |                                             |
| HE600N | Mine (in drums) (36)                                                                        | 3.42E-07 | 10.0 |                                             |
| HE600P | 155-mm Projectile (8)                                                                       | 7.60E-08 | 10.0 |                                             |
| HE6000 | 8-in Projectile (6)                                                                         | 5.70E-08 | 10.0 |                                             |
| HE600R | Rocket (15)                                                                                 | 1.43E-07 | 10.0 |                                             |
| HE620  | Single bare munition detonates<br>given 4 ft drop (in UPA)                                  | 3.20E-10 | 10.0 | HF12                                        |
| HE700  | Munition in onsite container<br>detonates given drop (per munition)                         | 3.20E-11 |      | HC22,HC24,<br>HF11,HF13,HF14                |
| HE7000 | 4.2-in Mortar (48)                                                                          | 1.54E-09 | 10.0 |                                             |
| HE700C | 105-mm Cartridge (24)                                                                       | 7.68E-10 | 10.0 |                                             |
| HE700H | Mine (in drums) (36)                                                                        | 1.15E-09 | 10.0 |                                             |
| HE700P | 155-ma Projectile (8)                                                                       | 2.56E-10 | 10.0 |                                             |
| HE7000 | 8-in Projectile (6)                                                                         | 1.92E-10 | 10.0 |                                             |
| HE700R | Rocket (15)                                                                                 | 4.80E-10 | 10.0 |                                             |
| HE710  | Munition in offsite container<br>detonates given drop (per munition)                        | 3.20E-12 |      | HC23,HC25                                   |
| HE710D | 4.2-in Mortar (48)                                                                          | 6.14E-10 | 10.0 |                                             |
| HE710C | 105-mm Cartridge (24)                                                                       | 9.22E-10 | 10.0 |                                             |
| HE710H | Mine (in drums) (36)                                                                        | 3.46E-10 | 10.0 |                                             |
| HE710P | 155-se Projectile (8)                                                                       | 3.84E-10 | 10.0 |                                             |
| HE7100 | 8-in Projectile (6)                                                                         | 1.92E-10 | 10.0 |                                             |
| HE710R | Rocket (15)                                                                                 | 1.92E-10 | 10.0 |                                             |
| HEBOO  | NDB Ventilation System Failure                                                              | 1.00E-09 | 10.0 | HC17,HC18,HC19,HC21<br>HF2,<br>HF8,HF9,HF10 |



6-33

initiating event, additional events have to occur to cause an agent release to the environment. The mechanisms for release could be the breaching of munitions by puncture, impact, or detonation because of some undue force. If the accident involves a fire (e.g., collisions), thermal detonation of burstered munitions or hydraulic rupture of nonburstered munitions is possible if the fire is not suppressed. For accidents which occur in the leakers processing facility or in the UPA (some HF scenarios), failure of the ventilation system is critical to the amount of agent released to the environment.

<u>Puncture Probability</u>. The probability of puncturing a munition whether it is inside or outside a transportation container has been evaluated based on a puncture model that is a function of the probe density and length, the possible number of such probes in the area, the munition size and configuration, and drop height. Details of this model are discussed in Appendix C.

<u>Munition Detonation</u>. The probability of a bare munition detonating when dropped from a height of 6 ft (equivalent to a collision at 13.5 mph) is assumed to be  $9.5 \times 10^{-9}$ /munition. For a 4-ft drop, the corresponding probability is  $3.2 \times 10^{-10}$ . The probability of a munition inside a transportation container detonating when dropped is judged to be lower. Here we take credit for the cushioning effect provided by the dunnage and packaging material inside the container. We assume that this will essentially reduce the impact velocity experienced by the munition itself by 30%, thus reducing the impact velocity to 9.5 mph. Using the approach outlined in Appendix C of Ref. 6-3, this results in a probability of  $3.2 \times 10^{-11}$ /munition for the onsite container and  $3.2 \times 10^{-12}$  for the offsite container.

<u>Collision Leads to Fire</u>. The probability value of 0.0725 was derived from Ref. 6-3, which presents data indicating that 25% of collision accidents lead to fire and 29% of collision accidents occur at

6-34

20 mph or less. This is the assumed maximum speed of the forklift during a collision.

<u>Fire Contained</u>. The amount of available fuel in any transportation vehicle will be limited such that it cannot sustain a prolonged fire (greater than a few minutes). For nonburstered munitions that are not in transportation containers, it takes 30 min of direct heating before hydraulic rupture occurs (36 min for ton containers). Since the available fuel will be insufficient to support this fire duration, the probability of fire containment is 1.0. When munitions are in transportation containers, it takes at least 15 min of direct heating of an intact container to cause a thermal explosion. Again the available fuel will not be sufficient to support this fire. Hence, the probability of fire containment is also 1.0.

The results of the handling analysis are presented in Section 11 of this report. Table 6-4 summarizes the results of the frequency and uncertainty calculations for rail transport handling. Frequency results are median values.

TABLE 6-4 HANDLING ACCIDENT-COLLOCATION PROCESSING OPTION (RAIL TRANSPORT)

File: HDLREEST.Mkl 17-Aug-87 Fage 1

HANDLING ACCIDENTS - REGIONAL PROCESSING OPTION - PER PALLET

Accident Frequencies and Range Factors

|                   |                                           |         |          |         |           |         |          |          | •        |         |         |           |         |         |           | -            |          |          |          |         |          |             | _         |          |           |          |           |         | -        |        |
|-------------------|-------------------------------------------|---------|----------|---------|-----------|---------|----------|----------|----------|---------|---------|-----------|---------|---------|-----------|--------------|----------|----------|----------|---------|----------|-------------|-----------|----------|-----------|----------|-----------|---------|----------|--------|
| RANGE<br>Factor   | )<br> <br> <br> <br> <br> <br>            | 1.3E+0) | ļ        | !       | ;         | ;       | 1. 3E+0  | 1        | 1. 36+0  | 1       | ļ       | 1         | 1       | 1       | 1.3E+0    | 1.3E+0       | 1. JE +0 | ı        | 1. 3E+0  | ł       | ſ        | ł           | 1. 36+0   | 1. 75+0  | ł         | 1.35+0   | 1. XE+0   | 1.36+0  | 1.35+0   | 11 · · |
| UMDA<br>Fred      |                                           | 6.1E-08 | N/A      | N/A     | N/A       | N/A     | 2.0E-08  | N/A      | 1.2E-08  | 0°0E+00 | N/A     | 0° 0E +00 | 0.0E+00 | 0.0E+00 | 4.8E-08   | 4.86-08      | 5.2E-07  | N/A      | 1. 3E-07 | N/A     | N/A      | N/A         | 7.16-07   | 5.06-08  | N/A       | 5.05-08  | 5.0E-08   | 5.0E-08 | 2.65-06  | 11 11  |
| IANGE<br>.actor   |                                           | 1.36+01 | 1.36+01  | 1.3E+01 | 1.3E+01   | 1.3E+01 | 1.3E+01  | 1.36+01  | 1.3E+01  | 1       | ;       | ł         | ł       | ł       | 1.35+01   | 1.JE+01      | 1.35+01  | 3. IE+01 | 1.35+01  | 1.3E+01 | 1.35+01  | 1.36+01     | 1.3E+01   | 1.35+01  | 1. JE +01 | 1. 3E+01 | 1.35+01   | 1.3£+01 | 1.36+01  |        |
| TEAD F            |                                           | 1.25-07 | 3.2E-08  | 5.7E-09 | 5.7E-09   | 4.0E-08 | 4.0E-08  | 4.(iE-08 | 2.4E-08  | 0.0E+00 | 0.0E+00 | 0.0E+00   | 0.0E+00 | 0,0E+00 | 9.5E-08   | 9.5E-08      | 1.0E-06  | 5.2E-10  | 2.6E-07  | 7.4E-07 | 1.8E-07  | 1.8E-07     | 1.4E-06   | 1.05-07  | 1.05-07   | 1.05-07  | 1.0E-07   | 1.0E-07 | 5.3E-06  |        |
| iange<br>actor    |                                           | ;       | 1. 3E+01 | !       | 1.3E+01   | ł       | ł        | ł        | 1        | ł       | ł       | ;         | ł       | !       | ł         | ł            | ł        | ;        | ;        | 1.3E+01 | ;        | 1.3E+01     | ;         | ;        | 1.3E+01   | ;        | ;         | ;       | ;        |        |
| FUDA F            |                                           | N/A     | 1.65-08  | N/A     | 2.8E-09   | N/A     | N/A      | N/A      | N/A      | N/A     | 0.0E+00 | N/A       | N/A     | N/A     | N/A       | A/A          | N/A      | N/A      | N/A      | 3.7E-07 | N/A      | 80-36-8     | N/A       | N/A      | 5.0E-08   | N/A      | N/A       | N/A     | N/A      |        |
| RANGE<br>Factor   |                                           | ł       | ;        | ł       | ł         | ł       | 1.3E+01  | ;        | 1. 3E+01 | ;       | ł       | ;         | ł       | ł       | 1.3E+01   | 1.3E+01      | :        | 3. IE+01 | ;        | ;       | ;        | :           | 1. 3E+01  | 1        | ;         | ł        | ł         | 1       | 1.36+01  | :      |
| PBA 1<br>FREG F   |                                           | N/A     | N/A      | N/A     | N/A       | N/A     | 2.0E-08  | N/A      | 1.25-08  | N/A     | N/A     | N/A       | N/A     | N/A     | 4.86-08   | 4.8E-08      | N/A      | 5.2E-10  | N/A      | N/A     | N/A      | N/A         | 7.1E-07   | N/A      | N/A       | N/A      | W/A       | N/A     | 2.6E-06  |        |
| RANGE<br>Factor   |                                           | 1       | ł        | :       | ł         | ;       | ł        | 1.3E+01  | :        | :       | ;       | :         | ;       | :       | ;         | 1            | :        | ł        | ;        | :       | :        | :           | !         | ł        | 1         | ł        | :         | !       | :        |        |
| NAAP<br>FREG      |                                           | N/A     | N/A      | N/A     | N/A       | N/A     | N/A      | 2.0E-08  | N/A      | N/A     | N/A     | N/A       | 5/N     | N/A     | N/A       | N/A          | N/A      | N/A      | N/A      | N/A     | N/A      | N/A         | N/A       | N/A      | N/A       | N/A      | N/A       | N/A     | N/A      |        |
| IANGE<br>Actor    |                                           | 1       | :        | ;       | :         | ;       | ł        | :        | ;        | ł       | 1       | :         | ł       | ;       | 1.35+01   | 1. JE +01    | ł        | ł        | ł        | ;       | :        | {           | ;         | 1        | 1. XE 401 | 1.35+01  | 1. 3E +01 | ł       | 1. 3E+01 |        |
| LBAD F            |                                           | N/A     | N/A      | N/A     | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | 0.0E+00 | 0.0E+00   | 0.01400 | R/A     | 4.8E-09   | 4.85-08      | N/A      | N/A      | N/A      | N/A     | N/A      | N/A         | N/A       | N/A      | 5.06-08   | 5. 0E-08 | 5.0E-08   | N/A     | 2.6E-06  |        |
| RANGE<br>Factor   |                                           | ;       | ;        | :       | ł         | ;       | 1.35+01  | !        | ;        | :       | ł       | :         | :       | :       | ł         | ;            | ;        | 3. 1E+01 | ł        | ł       | :        | 1           | :         | ;        | ţ         | 1        | ł         | ł       | ;        |        |
| AFG I<br>FRED F   |                                           | N/A     | N/A      | N/A     | N/A       | A / A   | 2.0E-08  | N/A      | N/A      | N/A     | N/A     | N/A       | N/A     | N/A     | N/A       | N/A          | N/A      | 5.26-10  | A/A      | N/A     | N/A      | N/A         | N/A       | N/A      | N/A       | N/A      | N / A     | N/A     | N/A      | •      |
| RANGE<br>Factor   |                                           | ;       | 1. 3E+01 | 1.36+01 | 1. XE +01 | ;       | 1. 3E+01 | 1. 35 41 | 1.3E+01  | :       | ł       | ;         | ł       | ł       | 1. 3E +0: | 1.3E+01      | ;        | ;        | ;        | 1.36+01 | 1. 3E+01 | 1.35+01     | 1. 36 401 | 1. 3E+01 | 1.36+01   | 1.36+01  | 1.36+01   | ;       | 1.35+01  |        |
| ANAD I<br>Freg F  |                                           | N/A     | 3.26-08  | 5.76-09 | 5.76-09   | A.A     | 4.0E-08  | 4.0E-08  | 2.4E-08  | 0,0E+(0 | 0.06+00 | 0.0E+00   | 0.6E+00 | N/A     | 9.56-08   | 9.5E-08      | N/A      | N/A      | N/A      | 7.46-07 | 1.86-07  | 1.8E-07     | 1.45-06   | 1.0E-07  | 1.0E-07   | 1.05-07  | 1.0E-07   | N/A     | 5.36-06  |        |
| 9.<br>1           |                                           |         | 1        | -       | 1         | -       | -        |          |          | -       |         | -         | -       | -       | -         | -            | -        | 7        | ٣,       | М       | ۰۰       | r)          | ٣,        | ~        | m         | Ŷ        | •••       | r       | ⊷,       | ļ      |
| SCEN- DF.<br>Ario | 1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1 | HCRGC   | HCDHC    | HCCGC   | HCCHC     | HCIr BC | HCKHC    | HCEVC    | HERVE    | HCFEC   | HCFHC   | HCPVC     | HCQAC   | HCQVC   | HCR6C     | <b>ICRVC</b> | JUSJE    | IC HF    | HC RGC   | 1CDHC   | 10,060   | <b>LCHC</b> | JANC      | ICPGC    | ICFHC     | CFVC     | 1000      | HCDVC   | 1CFGC    |        |

6-36

<mark>⋛⋬⋻⋛⋶⋌⋌⋨⋶⋨⋛⋨⋛⋺⋚⋺⋚⋺⋚⋺⋚⋺⋚⋺⋚⋺⋚⋨⋚⋛⋺⋚⋺⋚⋨⋚⋨⋛⋛⋛⋧⋛⋧⋛⋧⋚⋧⋺⋧⋺⋩⋺⋩⋺⋩⋺⋩⋳⋛⋧⋛⋧⋛⋛</mark>⋳⋛⋧⋛⋧⋛⋧⋚⋧⋺⋧⋻⋶⋧⋺⋧⋻⋶⋺⋧⋺⋧⋺⋧⋺⋧⋺⋧⋺⋧⋺⋧⋺⋧⋺⋧⋺⋧⋺⋧⋺⋧⋺⋧⋺⋧







File: HDLREGST.MK1 17-Aug-87 Page 2

TABLE 6-4 (Continued)

# HANDLING ACCIDENTS - REGIONAL PROCESSING OPTION - PER PALLET

## Accident Frequencies and Range Factors

| SCFN- D | G. ND. | ANAD       | RANGE     | AFG     | RANGE            | L BAD    | RANGE   | NAAP    | RANGE    | PBA      | RANGE    | FUDA    | RANGE    | TEAD     | RANGE    | MDA              | RANGE             |
|---------|--------|------------|-----------|---------|------------------|----------|---------|---------|----------|----------|----------|---------|----------|----------|----------|------------------|-------------------|
| ARIO    |        | FREQ       | FACTOR    | FREQ    | FACTOR           | FREQ     | FACTOR  | FREQ    | FACTOR   | FRED     | FACTOR   | FRED    | FACTOR   | FREQ     | FACTOR   | FREQ             | FACTOR            |
|         |        |            |           |         | 9<br>8<br>9<br>9 |          |         |         |          |          |          |         |          |          |          |                  |                   |
| HCSVC   | •      | N/A        | :         | N/A     | 1                | N/A      | ł       | N/A     | ;        | N/A      | ł        | N/A     | ;        | 6.1E-07  | 1. 3E+01 | 1.5E-07          | 1. 3E+01          |
| HCRGC   | +      | N/A        | !         | N/A     | ;                | N/A      | ;       | N/A     | ;        | N/A      | ;        | N/A     | ;        | 3.4E-09  | 1. 3E+01 | 1.7E-09          | 1, <u>3E</u> +01  |
| HCDHC   | -      | 1.65-09    | 10.35.01  | N/A     | ;                | N/A      | ſ       | N/A     | ;        | N/A      | 1        | 8.0E-10 | 1.3E+01  | 1.6E-09  | 1. 3E+01 | N/A              | ;                 |
| HCCGC   | -      | 3.96-11    | 1.3E+01   | N/A     | ł                | N/A      | 1       | N/A     | ;        | N/A      | 1        | N/A     | ;        | 3.9E-11  | 1.36+01  | N/A              | ł                 |
| HCCHC   | +      | 11-36.5    | 1.36+01   | N/A     | ł                | N/A      | :       | N/A     | ;        | N/A      | ;        | 2.0E-11 | 1. 3E+01 | 3.96-11  | 1, 3E+01 | A/A              | ł                 |
| HCK6C   | -      | N/A        | ;         | N/A     | 1                | N/A      | 1       | N/A     | ;        | N/A      | ł        | N/A     | 1        | 1.4E-08  | 1.35+01  | N/A              | ł                 |
| JH 4 JH | -      | 1.4E-CB    | 1.36+01   | 6.7E-09 | 1.3E+01          | N/A      | ł       | N/A     | ;        | 6.7E-09  | 1. 3E+01 | N/A     | ł        | 1.4E-08  | 1.36401  | 7.26-09          | 1.35+01           |
| 37. 43H | 4      | 1.46-08    | 1.3£+01   | N/A     | ţ                | N/A      | ł       | 7.26-09 | 1. JE+01 | N/A      | 1        | N/A     | ł        | 1.4E-08  | 1. 3E+01 | N/A              | ł                 |
| HCINC   | -      | 1.4E-09    | 1.35+01   | N/A     | 1                | N/A      | ;       | N/A     | ;        | 6.9E-10  | 1.3E+01  | N/A     | ;        | 1.4E-09  | 1.3E+01  | 01-36 <b>.</b> 9 | 1, 35 +01         |
| HCF6C   | -      | 0.01400    | !         | N/A     | ł                | N/A      | 1       | N/A     | ;        | N/A      | ł        | N/A     | ;        | 0.0E+00  | ł        | 0.0€+00          | ;                 |
| HCFHC   | 4      | 0°*00      | 1         | N/A     | ;                | 0°0E+00  | ł       | N/A     | ;        | N/A      | ł        | 0.0E+00 | ł        | 0.0130.0 | ł        | N.A              | !                 |
| HCFVC   | -      | 0.01400    | •         | N, A    | 1                | 0.0E+00  | ;       | N/A     | !        | N/A      | 1        | N/A     | :        | 0.0E+00  | ;        | 0.0E+00          | ł                 |
| HCDEC   | -      | 0.0E+00    | ;         | N/A     | ;                | 0.0E+00  | ſ       | N/A     | ł        | N/A      | 1        | N/A     | ł        | 0.0E+00  | ;        | 0.0E+00          | ļ                 |
| HCOVC   | 4      | N/A        |           | N/A     | ;                | N/A      | {       | N/A     | :        | N/A      | ł        | N/A     | !        | 0.0E+00  | !        | 0.01400          | ł                 |
| HCKGC   | -      | 6.2E-09    | 1. 3E+01  | N/A     | 1                | 3.1E-09  | 1.3E+01 | N/A     | ÷        | 3. IE-09 | 1.35+01  | N/A     | !        | 6.2E-09  | 1.3E+01  | 3. IE-09         | 1.35+01           |
| JVAJH   | -      | 6.2E-09    | 1.36+01   | N/A     | ;                | 3. IE-09 | 1.3E+01 | N/A     | :        | 3.1E-09  | 1.3E+01  | N/A     | ;        | 6.2E-09  | 1.3E+01  | 3.1E-09          | 1. <u>35</u> + 01 |
| HCSVC   | -      | N/A        | :         | N/G     | ;                | N/A      | ł       | N/A     | ;        | N/A      | ł        | N/A     | ł        | 5.46-08  | 1, 3E+01 | 2.7E-08          | 1. 35+01          |
| HCRES   | ŝ      | N/A        | ;         | N/A     | 1                | N/A      | 1       | N/A     | ;        | N/A      | ł        | N/A     | ł        | 60-35-09 | 1. 3E+01 | N/A              | ı                 |
| HCDHS   | ŝ      | 6. 3E - 09 | 1. 35 +01 | N/A     | 1                | N/A      | ;       | N/A     | 1        | N/A      | ł        | N/A     | 1        | 6.3E-09  | 1.3E+01  | N/A              | •                 |
| HCCGS   | ı,     | 7.2E-10    | 1.35+01   | N/A     | ł                | N/A      | ł       | N/A     | ł        | N/A      | 1        | N/A     | 1        | 7.26-10  | 1.35+01  | N/A              | ;                 |
| HCCHS   | ŝ      | 7.2E-10    | 1.36+01   | N/A     | 1                | N/A      | ;       | N/A     | ł        | N/A      | 1        | N/A     | 1        | 7.2E-10  | 1.36+01  | N/A              | ;                 |
| HCF6S   | רט     | N/A        | ł         | N/A     | 1                | N/A      | ł       | N/A     | 1        | N/A      | ł        | N/A     | ;        | 1.36-08  | 1.3E+01  | N/A              | ł                 |
| HCKHS   | רש     | 1. 3E - 08 | 1.35+01   | N/A     | ;                | N/A      | ł       | N/A     | :        | N/A      | 1        | N/A     | ł        | 1.36-08  | 1.36+01  | N/A              | ;                 |
| HCk VS  | רש     | 1.3E-08    | 1.36+01   | N/A     | !                | N/A      | ;       | N/A     | 1        | N/A      | ł        | N/A     | :        | 1.36-08  | 1. 3E+01 | N/A              | :                 |
| HCHVS   | 'n     | 8.6E-09    | 1.3E+01   | N/A     | :                | N/A      | ;       | N/A     | 1        | N/A      | ;        | N/A     | ;        | 8.6E-09  | 1.36+01  | N/A              | ;                 |
| HCF6S   | ליו    | 1.1E-09    | 1. 3E+01  | N/A     | ;                | N/A      | ł       | N/A     | ł        | N/A      | ł        | N/A     | :        | 1.1E-09  | 1.36+01  | A/A              | :                 |
| HCFHS   | Ś      | 1.1E-09    | 1.3E+01   | N/A     | ;                | N/A      | ł       | N/A     | :        | N/A      | ł        | N/A     | ł        | 1.1E-09  | 1.35+01  | N/A              | ł                 |
| HCFVS   | רט     | 1.16-09    | 1.35+01   | N/A     | 1                | N/A      | ł       | N/A     | ł        | N/A      | 1        | N/A     | ł        | 1.16-09  | 1.36+01  | N/A              | ł                 |
| HCB6S   | ŝ      | 1. IE -09  | 1.3E+01   | N/A     | ł                | N/A      | ;       | N/A     | ;        | N/A      | 1        | N/A     | ł        | 1.1E-09  | 1, 36+01 | N/A              | ļ                 |

22222

TABLE 6-4 (Continued)

Fage 3

File: HDLREGST.Mk1 17-Aug-87

HANDLING ACCIDENTS - REBIONAL PROCESSING OPTION - PER PALLET

Accident Frequencies and Range Factors

| 2               | 1      |          |          |           |         |          |          |          |         |         |          |          |          |         |         |         |         |         |          |          |           |         |          |          |         |         |         |          |         |          |
|-----------------|--------|----------|----------|-----------|---------|----------|----------|----------|---------|---------|----------|----------|----------|---------|---------|---------|---------|---------|----------|----------|-----------|---------|----------|----------|---------|---------|---------|----------|---------|----------|
| RANGE<br>Facto  | 1      | 1        | 1        | 1         | ;       | ;        | 1        | ;        | ;       | !       | ì        | ;        | ;        | ;       | ;       | 1       | ;       | ;       | ;        | •        | 1         | ł       | ł        | 1        | 1       | !       | ;       | ;        | ;       | 1        |
| UMDA<br>FREQ    |        | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | 8/N     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/N     | N/A     | N/A     | N/A      | N/A     | N/A      |
| RANGE<br>Factor |        | 1. JE+01 | 1. 35+01 | 1.35+01   | 1.35+01 | 3. IE+01 | 3. IE+01 | ļ        | ł       | 3.16+01 | 3. IE+01 | 3. IE+01 | 3. IE+01 | :       | ł       | ;       | 1       | ;       | 3, IE+01 | 3, 16+01 | 3. IE +01 | 1.3E+01 | 1.3E+01  | ł        | 1       | 10+31.1 | 1.36+01 | 1.36+01  | 1.36+01 | ;        |
| TEAD<br>Freq    |        | 1.1E-09  | 4.96-09  | 4.9E-09   | 1.4E-07 | 6.25-11  | 1.96-10  | 0°.0E+00 | 0.0€+00 | 2.1E-10 | 2.1E-10  | 2.1E-10  | 2.7E-10  | 0.01+00 | 0.0E+00 | 0.0E+00 | 0.0E+00 | 0.0E+00 | 1.66-10  | 1.66-10  | 3.9E-09   | 3.7E-09 | 1.16-08  | 0°.(E+0) | 0.0E+00 | 1.26-08 | 1.2E-08 | 1.26-09  | 1.6E-08 | 0.0E+00  |
| RANGE<br>Factor |        | 1        | 1        | ł         | :       | ł        | ;        | ł        | 1       | ſ       | 1        | 1        | ł        | ;       | 1       | ł       | 1       | 1       | ł        | ;        | ł         | ł       | ł        | ł        | :       | 1       | ł       | ;        | ł       | ł        |
| PUDA<br>FREQ    |        | N/A      | N/A      | N/A       | N/A     | N/A      | A/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A      |
| RANGE<br>Factor |        | ;        | ł        | :         | ł       | ;        | ł        | :        | ;       | ł       | ł        | 1        | ;        | !       | ł       | 1       | 1       | ł       | ł        | ١        | {         | 1       | ł        | ſ        | ł       | 1       | 1       | 1        | ł       | 1        |
| PBA<br>Freq     |        | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A      |
| RANGE<br>Factor |        | ł        | :        | ;         | 1       | ł        | :        | 1        | ł       | 1       | 1        | ł        | :        | ;       | ł       | 1       | :       | ł       | ł        | 1        | 1         | ł       | ł        | 1        | ;       | :       | ;       | ł        | :       | ł        |
| NAAP<br>Freq    |        | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | A/A       | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A      |
| RANGE<br>FACTOR |        | !        | ;        | ł         | 1       | ţ        | 1        | :        | ł       | :       | 1        | ł        | 1        | 1       | ;       | ł       | ł       | 1       | ;        | ;        | ł         | :       | ł        | 1        | 1       | 1       | ł       | ;        | ;       | :        |
| L BAD<br>FRED   |        | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | A/A     | N/A     | N/A     | R/A     | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A      |
| RANGE<br>FACTOR |        | :        | 1        | ł         | ١       | ;        | 1        | 1        | 1       | ;       | ;        | :        | ł        | 1       | 1       | ł       | 1       | ł       | 1        | ł        | ł         | 1       | ł        | ł        | 1       | 1       | ł       | :        | ł       | ;        |
| AF6<br>Freg     |        | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A      |
| RANGE<br>Factor |        | ;        | 1.3E+01  | 1. JE +01 | ;       | ;        | 3. IE+01 | !        | ;       | ł       | 3. JE+01 | 3. IE+01 | 3. 1E+01 | ;       | ł       | :       | 1       | ;       | 3. IE+01 | 3. IE+01 | ;         | !       | 1. 3E+01 | 1        | ł       | ł       | 1.35+01 | 1, 35+01 | 1.3€+01 | :        |
| ANAD<br>Freq    |        | N/A      | 4.96-09  | 4.95-09   | N/A     | N/A      | 1.95-10  | 0.0E+00  | 0.0E+00 | N/A     | 2.16-10  | 2.1E-10  | 2.7E-10  | 0.0E+00 | 0.0E+00 | 0.0E+00 | 0.0E+00 | N/A     | 1.65-10  | 1.6E-10  | N/A       | N/A     | 1.16-08  | 0(+30*0  | 0.0E+00 | N/A     | 1.2E-08 | 1.25-08  | 1.6E-08 | 0, 0E+00 |
| )F. NO.         | ;<br>; | ŝ        | 5        | ŝ         | r»      | 9        | 9        | 9        | ę       | \$      | -9       | -0       | 9        | 9       | s.      | -0      | 9       | ÷       | 9        | 9        | 9         | 7       | ſ        | 1        | 1       | 1       | 7       | 7        | 7       | 1        |
| SCEN- C<br>Ario |        | HCBVS    | HCF65    | HCKVS     | HCSVS   | HCBGF    | HCDHF    | HCCGF    | HCCHF   | HCKBF   | HCKHF    | HCKVF    | HCHVF    | HCF6F   | HCPHF   | HCFVF   | HCDGF   | HCOVF   | HCKGF    | HCRVF    | HCSVF     | HCRES   | HCDHS    | HCCGS    | HCCHS   | HCKES   | HCKHS   | HCKVS    | HCMVS   | HCF6S    |

(;;;)







File: HDLREGSL.MK1 17-Aug-87 Page 4 TABLE 6-4 (Continued)

HANDLING ACCIDENTS - REGIONAL PROCESSING OPTION - PER PALLET

Accident Frequencies and Range Factors

| RANGE<br>Factor                                                                         |   | :                       | ł                     | :                     | ;                 | ţ                             | 1                             | !                 | 1.36+01           | 1                             | ;                             | !                             | ł                 | 1. 3E+01                                  | ;                                    | 1. 3E +01                     | 1.36+01                       | ;                                         | 1. 36 •••1                                | 1. XE +01                                 | 1.36+01           | 1.35+01                                   | 1. 3E+01                                  | 1.76401           | 10+31             | !                             | ;                             | 1                             | :                 | 3. 1E+01                                  |
|-----------------------------------------------------------------------------------------|---|-------------------------|-----------------------|-----------------------|-------------------|-------------------------------|-------------------------------|-------------------|-------------------|-------------------------------|-------------------------------|-------------------------------|-------------------|-------------------------------------------|--------------------------------------|-------------------------------|-------------------------------|-------------------------------------------|-------------------------------------------|-------------------------------------------|-------------------|-------------------------------------------|-------------------------------------------|-------------------|-------------------|-------------------------------|-------------------------------|-------------------------------|-------------------|-------------------------------------------|
| UNDA<br>Freq                                                                            |   | H/H                     | N/A                   | N/A                   | N/A               | N/A                           | N/A                           | N/A               | 2.36-08           | N/A                           | N/A                           | N/A                           | N/A               | 2.2E-08                                   | N/A                                  | 1.2E-08                       | 1.06-08                       | N/A                                       | 1.05-08                                   | 1.0E-09                                   | 1. ñE-08          | 1.3E-08                                   | 1.35-08                                   | <b>B.5E</b> -09   | 8.6E-10           | N/A                           | A/A                           | N/A                           | N/A               | 6.96-10                                   |
| RANGE<br>FACTOR                                                                         |   | ł                       | ł                     | ;                     | :                 | 1. 3E+01                      | 1.3E+01                       | 1.3E+01           | 1. 3E+01          | 1.35+01                       | 1.35+01                       | 1.36+01                       | 1. 3E+01          | 1.3E+01                                   | 1° 3E +01                            | 1.3E+01                       | 1°+38-1                       | 1. 3E +01                                 | 1. 3E+01                                  | 1. 3E+01                                  | 1. 3E +01         | 1.36+01                                   | 1.3E+01                                   | 1.36+01           | 3. IE +01         | 3. IE+01                      | 3. IE +01                     | 3.1E+01                       | 3. IE +01         | 3. IE+01                                  |
| IEAD<br>Freq                                                                            |   | 0. UE + UU              | 0.0E+00               | 0.0E<00               | 0.0E+00           | 9.6E-09                       | 9.6E-09                       | 2.3E-07           | 2.36-08           | 1. 2E - AB                    | <b>B.</b> BE - 09             | 8.85-09                       | 2.2E-08           | 2.2E-08                                   | 2.2E-08                              | 1.2E-08                       | 1.0E-09                       | 1.05-08                                   | 1.0E-08                                   | 1.0E-09                                   | 1.0E-08           | 1.3E-08                                   | 1.3E-08                                   | 8.5E-09           | 8.65-10           | 5.4E-10                       | 2.6E-10                       | 2.6E-10                       | 6.9E-10           | 6.9E-10                                   |
| RANGE<br>Factor                                                                         |   | 1                       | :                     | ;                     | ł                 | ł                             | ł                             | 1                 | ;                 | 1.JE+01                       | ł                             | 1.3E+01                       | 1                 | :                                         | 1                                    | ł                             | 1                             | 1.3E+01                                   | :                                         | ł                                         | ł                 | ;                                         | :                                         | ł                 | !                 | 3.1E+01                       | ł                             | 3. JE +01                     | ł                 | ł                                         |
| PUDA<br>Fred                                                                            |   | H/H                     | N/A                   | N/A                   | N/A               | N/A                           | N/A                           | N/A               | N/A               | 1.25-08                       | N/A                           | <b>B.</b> 8E-09               | N/A               | N/A                                       | N/A                                  | N/A                           | N/A                           | 1.05-08                                   | N/A                                       | N/A                                       | N/A               | N/A                                       | N/A                                       | N/A               | N/A               | 5.4E-10                       | N/A                           | 2.6E-10                       | N/A               | N/A                                       |
| RANGE<br>Factor                                                                         |   | ł                       | ł                     | :                     | :                 | 1                             | ;                             | 1                 | 1                 | ł                             | ;                             | !                             | 1                 | 1.3E+01                                   | ł                                    | 1.3E+01                       | ł                             | í                                         | ļ                                         | ł                                         | ł                 | 1.3E+01                                   | 1.36+01                                   | ł                 | 1                 | ł                             | ł                             | 1                             | ł                 | 3.1E+01                                   |
| PBA                                                                                     |   | H/H                     | N/A                   | N/A                   | N/A               | N/A                           | N/A                           | N/A               | N/A               | N/A                           | N/A                           | N/A                           | N/A               | 2.2E-08                                   | N/A                                  | 1.2E-09                       | N/A                           | N/A                                       | N/A                                       | N/A                                       | N/A               | 1.3E-08                                   | 1.35-08                                   | NZA               | N/A               | N/A                           | N/A                           | N/A                           | N/A               | 6.9E-10                                   |
| щ <sup>85</sup>                                                                         | ! | :                       | !                     | ;                     | :                 | ;                             | ;                             | ł                 | 1                 | ;                             | ;                             | :                             | ł                 | 1                                         | 3E+i)1                               | ;                             | ;                             | ;                                         | :                                         | :                                         | ;                 | ;                                         | ;                                         | -                 | ;                 | :                             | ;                             | ;                             | ;                 | 1                                         |
| RANG<br>FAC1                                                                            | 1 |                         |                       |                       |                   |                               |                               |                   |                   |                               |                               |                               |                   |                                           |                                      |                               |                               |                                           |                                           |                                           |                   |                                           |                                           |                   |                   |                               |                               |                               |                   |                                           |
| NAAP RANG<br>Freq Fact                                                                  |   | N/A                     | N/A                   | N/A                   | N/A               | N/A                           | N/A                           | N/A               | N/A               | N/A                           | N/A                           | N/A                           | N/A               | N/A                                       | 2.2E-08 1.                           | N/A                           | N/A                           | N/A                                       | N/A                                       | N/A                                       | N/A               | N/A                                       | N/A                                       | N/A               | N/A               | N/A                           | N/A                           | N/A                           | N/A               | N/A                                       |
| RANGE NAAP RANG<br>Factor freq fact                                                     |   | N/A                     | N/A                   | N/A                   | N/A               | N/A                           | N/A                           | N/A               | N/A               | N/A                           | N/A                           | N/A                           | N/A               | N/A                                       | 2.2E-08 1.                           | N/A                           | N/A                           | 1.3E+01 N/A                               | 1.3E+01 N/A                               | 1.3E+01 N/A                               | N/A               | 1.3E+01 N/A                               | 1. 3E+01 N/A                              | N/A               | N/A               | N/A                           | N/A                           | N/A                           | N/A               | N/A                                       |
| LBAD RANGE NAAP RANG<br>Freq Factor Freq Fact                                           |   | N/A N/A                 | N/A N/A               | N/A N/A               | N/A N/A           | N/A N/A                       | N/A N/A                       | N/A N/A           | N/A N/A           | N/A N/A                       | N/A N/A                       | N/A N/A                       | N/A N/A           | N/A N/A                                   | N/A 2.2E-08 1.                       | N/A N/A                       | N/A N/A                       | 1.0E-08 1.3E+01 N/A                       | 1.0E-08 1.3E+01 N/A                       | 1.0E-08 1.3E+01 N/A                       | N/A N/A           | 1.3E-08 1.3E+01 N/A                       | 1.3E-08 1.3E+01 N/A                       | N/A N/A           | N/A N/A           | N/A N/A                       | N/A N/A                       | N/A N/A                       | N/A N/A           | N/A N/A                                   |
| RANGE LBAD RANGE NAAP RANG<br>Factor Freq Factor                                        |   |                         | N/A N/A               | N/A N/A               | N/A N/A           | N/A N/A                       | N/A N/A                       | N/A N/A           | N/A N/A           | N/A N/A                       | N/A N/A                       | N/A N/A                       | N/A N/A           | 1.3E+01 N/A N/A                           | N/A 2.2E-08 1.                       | N/A N/A                       | N/A N/A                       | 1.0E-08 1.3E+01 N/A                       | 1.0E-08 1.3E+01 N/A                       | 1.0E-08 1.3E+01 N/A                       | N/A N/A           | 1.3E-08 1.3E+01 N/A                       | 1.3E-08 1.3E+01 N/A                       | N/A N/A           | N/A N/A           | N/A N/A                       | N/A N/A                       | N/A N/A                       | N/A N/A           | 3.1E+01 N/A N/A                           |
| APG RANGE LBAD RANGE NAAP RANG<br>Freq Factor Freq Factor Freq Fact                     |   |                         | N/A N/A N/A           | N/A N/A N/A           | N/A N/A N/A       | N/A N/A N/A                   | N/A N/A N/A                   | N/A N/A N/A       | N/A N/A N/A       | N/A N/A N/A                   | N/A N/A N/A                   | N/A N/A N/A                   | N/A N/A N/A       | 2.2E-08 1.3E+01 N/A N/A                   | N/A N/A 2.2E-08 1.                   | N/A N/A N/A                   | N/A N/A N/A                   | N/A 1.0E-08 1.3E+01 N/A                   | N/A 1.0E-08 1.3E+01 N/A                   | N/A 1.0E-08 1.3E+01 N/A                   | N/A N/A N/A       | N/A 1.3E-08 1.3E+01 N/A                   | N/A 1.3E-08 1.3E+01 N/A                   | N/A N/A N/A       | N/A N/A N/A       | N/A N/A N/A                   | N/A N/A N/A                   | N/A N/A N/A                   | N/A N/A N/A       | 6.9E-10 3.1E+01 N/A N/A                   |
| RAMGE APG RANGE LBAD RANGE NAAP RANG<br>Factor Freq Factor Freq Factor Freq Fact        |   | N/A N/A                 | N/A N/A N/A           | N/A N/A N/A           | N/A N/A N/A       | 1.3E+01 N/A N/A N/A           | 1.3E+01 N/A N/A N/A           | N/A N/A N/A       | N/A N/A N/A       | 1.3E+01 N/A N/A N/A           | 1.3E+01 N/A N/A N/A           | 1.3E+01 N/A N/A N/A           | N/A N/A N/A       | 1.3E+01 2.2E-08 1.3E+01 N/A N/A           | 1.3E+01 N/A N/A 2.2E-08 1.           | 1.3E+01 N/A N/A N/A           | 1.3E+01 N/A N/A N/A           | 1.3E+01 N/A 1.0E-08 1.3E+01 N/A           | 1.3E+01 N/A 1.0E-08 1.3E+01 N/A           | 1.3E+01 N/A 1.0E-08 1.3E+01 N/A           | N/A N/A N/A       | 1.3E+01 N/A 1.3E-08 1.3E+01 N/A           | 1.3E+01 N/A 1.3E-08 1.3E+01 N/A           | N/A N/A N/A       | N/A N/A N/A       | 3.1E+01 N/A N/A N/A           | 3.1E+01 N/A N/A N/A           | 3.1E+01 N/A N/A N/A           | N/A N/A N/A       | 3.1E+01 6.9E-10 3.1E+01 N/A N/A           |
| ANAD RANGE APG RANGE LBAD RANGE NAAP RANG<br>Freg factor freg factor freg factor        |   | U., !!E+UV] N/A N/A N/A | 0.0E+00 N/A N/A N/A   | 0.0E+00 N/A N/A N/A   | N/A N/A N/A N/A   | 9.6E-09 1.3E+01 N/A N/A N/A   | 9.6E-09 1.3E+01 N/A N/A N/A   | N/A N/A N/A N/A   | N/A N/A N/A N/A   | 1.2E-08 1.3E+01 N/A N/A N/A   | 8.8E-09 1.3E+01 N/A N/A N/A   | 8.86-09 1.35+01 N/A N/A N/A   | N/A N/A N/A N/A   | 2.2E-08 1.3E+01 2.2E-08 1.3E+01 N/A N/A   | 2.2E-08 1.3E+01 N/A 2.2E-08 1.       | 1.2E-09 1.3E+01 N/A N/A N/A   | 1.0E-08 1.3E401 N/A N/A N/A   | 1.0E-08 1.3E+01 N/A 1.0E-08 1.3E+01 N/A   | 1.0E-08 1.3E+01 N/A 1.0E-08 1.3E+01 N/A   | 1.0E-08 1.3E+01 N/A 1.0E-08 1.3E+01 N/A   | N/A N/A N/A N/A   | 1.3E-08 1.3E+01 N/A 1.3E-08 1.3E+01 N/A   | 1.3E-08 1.3E+01 N/A 1.3E-08 1.3E+01 N/A   | N/A N/A N/A N/A   | N/A N/A N/A N/A   | 5.4E-10 3.1E+01 N/A N/A N/A   | 2.6E-10 3.1E+01 N/A N/A N/A   | 2.6E-10 3.1E+01 N/A N/A N/A   | N/A N/A N/A N/A   | 6.9E-10 3.1E+01 6.9E-10 3.1E+01 N/A N/A   |
| F. NO. AMAD RANGE APG RANGE LBAD RANGE MAAP RANG<br>Freg factor freg factor freg factor |   | / 0.vie+00 N/A N/A N/A  | 7 0.0E+00 N/A N/A N/A | 7 0.0E+00 N/A N/A N/A | 7 N/A N/A N/A N/A | 7 9.6E-09 1.3E+01 N/A N/A N/A | 7 9.6E-09 1.3E+01 N/A N/A N/A | 7 N/A N/A N/A N/A | 8 N/A N/A N/A N/A | B 1.2E-08 1.3E+01 N/A N/A N/A | 8 8.8E-09 1.3E+01 N/A N/A N/A | B B.BE-09 1.3E+01 N/A N/A N/A | B N/A N/A N/A N/A | B 2.2E-0B 1.3E+01 2.2E-0B 1.3E+01 N/A N/A | 8 2.2E-08 1.3E+01 N/A N/A 2.2E-08 1. | B 1.2E-09 1.3E+01 N/A N/A N/A | 8 1.0E-08 1.3E401 N/A N/A N/A | 8 1.0E-08 1.3E+01 N/A 1.0E-08 1.3E+01 N/A | 8 1.0E-08 1.3E+01 N/A 1.0E-08 1.3E+01 N/A | B 1.0E-08 1.3E+01 N/A 1.0E-08 1.3E+01 N/A | d N/A N/A N/A N/A | B 1.3E-08 1.3E+01 N/A 1.3E-08 1.3E+01 N/A | 8 1.3E-08 1.3E+01 N/A 1.3E-08 1.3E+01 N/A | 8 N/A N/A N/A N/A | P N/A N/A N/A N/A | 9 5.4E-10 3.1E+01 N/A N/A N/A | 9 2.6E-10 3.1E+01 N/A N/A N/A | 9 2.6E-10 3.1E+01 N/A N/A N/A | 9 N/A N/A N/A N/A | 9 6.9E-10 3.1E+01 6.9E-10 3.1E+01 N/A N/A |

TAX BANK

6-39

File: HDLRE65T.uki 17-Aug-87 Page 5

TABLE 6-4 (Continued)

6.0335553

المحمد ومرا

LP22222000

HANDLING ACCIDENTS - REGIONAL PROCESSING OPTION - PER PALLET

Accident Frequencies and Range Factors

|                                                                              |                                        |                                 |                                 | _                               |                                     |                                     | -                                   |                     |                                     |                                     | -                   | _                    |                                  |                                  |                                  |                        |                                              |                                  |                                  | _                                |                                      | _                                    |                                      | _                    | -                                    | _                                    |                      |                                  |                                  |                                  |
|------------------------------------------------------------------------------|----------------------------------------|---------------------------------|---------------------------------|---------------------------------|-------------------------------------|-------------------------------------|-------------------------------------|---------------------|-------------------------------------|-------------------------------------|---------------------|----------------------|----------------------------------|----------------------------------|----------------------------------|------------------------|----------------------------------------------|----------------------------------|----------------------------------|----------------------------------|--------------------------------------|--------------------------------------|--------------------------------------|----------------------|--------------------------------------|--------------------------------------|----------------------|----------------------------------|----------------------------------|----------------------------------|
| RANGE<br>Factor                                                              |                                        | :                               | 3. JE+0                         | 3. IE+01                        | ;                                   | 3. IE+01                            | 3. 1E+0                             | 3.1E+01             | 3.1E+0                              | 3. IE+01                            | 3. IE+0             | 1. 3E+01             | :                                | ;                                | ł                                | ;                      | 1. 3E+01                                     | ;                                | 1. JE +01                        | 1.3E+0]                          | ;                                    | 1.36+01                              | 1.36+0                               | 1. 3E +01            | 1.3E+0                               | 1. 3E+01                             | 1.36+0               | 1                                | ;                                | 1                                |
| UNDA<br>Fred                                                                 | *<br>*<br>*                            | N/A                             | 5.36-10                         | 1.2E-10                         | N/A                                 | 1.2E-10                             | 1.2E-10                             | 1.2E-10             | 5.6E-10                             | 5.6E-10                             | 2.7E-10             | 1.15-09              | N/A                              | N/A                              | M/6                              | N/A                    | <b>B.</b> 8E - 09                            | N/A                              | 6. BE -09                        | 1.65-03                          | N/A                                  | 1.6E-09                              | 1.65-09                              | 1.6E-09              | 7.2E-09                              | 7.26-09                              | 3.5E-09              | N/A                              | N/A                              | N/A                              |
| RANGE<br>Factor                                                              |                                        | 3. IE+01                        | 3.1E+01                         | 3. IE+01                        | 3.1E+01                             | 3. IE+01                            | 3. IE+01                            | 3. IE+01            | 3.1E+01                             | 3. IE+01                            | 3. IE+01            | 1.JE+01              | 1, 35+01                         | 1.3E+01                          | 1.3E+01                          | 1.3E+01                | 1.36+01                                      | 1.36+01                          | 1.35+01                          | 1.3E+01                          | 1.36+01                              | 1.3E+01                              | 1.3E+01                              | 1.36+01              | 1. 3E+01                             | 1.36+01                              | 1.36+01              | 2.6E+01                          | 2.6E+01                          | 2.6F+01                          |
| TEAD<br>Freq                                                                 |                                        | 6.9E-10                         | 5.3E-10                         | 1.26-10                         | 1.2E-10                             | 1.26-10                             | 1.26-10                             | 1.2E-10             | 5.6E-10                             | 5.6E-10                             | 2.7E-10             | 1.16-08              | 6.9E-09                          | 3.4E-09                          | 3.4E-09                          | 8.8E-09                | 8.86-09                                      | 8.86-09                          | 6.8E-09                          | 1.65-09                          | 1.6E-09                              | 1.6E-09                              | 1.6E-09                              | 1.6E-09              | 7.2E-09                              | 7.2E-09                              | 3.5E-09              | 5.85-09                          | 2.95-09                          | 2.9E-09                          |
| RANGE<br>Factor                                                              |                                        | ;                               | :                               | :                               | 3. IE+01                            | ł                                   | :                                   | ł                   | 1                                   | :                                   | :                   | 1                    | 1.3E+01                          | !                                | 1.3E+01                          | ;                      | ł                                            | ;                                | ł                                | !                                | 1.36+01                              | :                                    | ł                                    | ;                    | :                                    | :                                    | ł                    | 2.6E+01                          | :                                | 2. AF +01                        |
| FUDA                                                                         |                                        | N/A                             | N/A                             | N/A                             | 1.25-10                             | N/A                                 | N/A                                 | N/A                 | N/A                                 | N/A                                 | N/A                 | N/A                  | 60-36-09                         | N/A                              | 3.4E-09                          | N/A                    | N/A                                          | N/A                              | N/A                              | N/A                              | 1.65-09                              | N/A                                  | N/A                                  | N/A                  | N/A                                  | N/A                                  | N/A                  | 2.9E-09                          | N/A                              | 1.46-09                          |
| lange<br>.actor                                                              |                                        | ł                               | 3. IE+01                        | ł                               | :                                   | 1                                   | 1                                   | ł                   | 3. IE+01                            | 3. 1E+01                            | ł                   | ł                    | 1                                | ;                                | :                                | :                      | 1. 3E+01                                     | ł                                | 1.36+01                          | :                                | :                                    | ł                                    | ł                                    | :                    | 1. 3E+01                             | 1. JE +01                            | ;                    | 1                                | ł                                | 1                                |
| PBA A<br>Freg F                                                              |                                        | N/A                             | 5. 3E-10                        | N/A                             | N/9                                 | N/A                                 | N/A                                 | N/A                 | 5.6E-10                             | 5.6E-10                             | N/A                 | N/A                  | N/A                              | N/A                              | N/A                              | N/A                    | <b>9.8E-09</b>                               | N/A                              | 6.8E-09                          | <b>N/A</b>                       | N/A                                  | N/A                                  | N/A                                  | N/A                  | 7.2E-09                              | 7.2E-09                              | N/A                  | N/A                              | N/A                              | N/A                              |
| RANGE<br>Pactor                                                              |                                        | 3. IE+01                        | ;                               | :                               | !                                   | ł                                   | ł                                   | ł                   | ;                                   | !                                   | :                   | ;                    | ;                                | ł                                | !                                | :                      | :                                            | 1. 3E+01                         | :                                | :                                | :                                    | ;                                    | :                                    | ;                    | :                                    | :                                    | 1                    | :                                | ł                                | {                                |
| REQ F                                                                        |                                        | 6.9E-1()                        | N/A                             | N/A                             | N/A                                 | N/A                                 | N/A                                 | N/A                 | N/A                                 | N/A                                 | N/A                 | N/A                  | N/A                              | N/A                              | N/A                              | N/A                    | N/A                                          | 8.8E-09                          | N/A                              | N/A                              | N/A                                  | N/A                                  | N/A                                  | N/A                  | N/A                                  | N/A                                  | N/A                  | N/A                              | N/A                              | N/A                              |
| range<br>Pictor                                                              |                                        | ;                               | ł                               | ;                               | 3. IE+01                            | 3. IE +01                           | 3. IE+01                            | ;                   | 3. 1E+01                            | 3. IE+01                            | ;                   | ;                    | ;                                | :                                | ł                                | :                      | :                                            | :                                | :                                | 1                                | 1. JE +01                            | 1.3E+01                              | 1. 3E+01                             | :                    | 1. JE +01                            | 1.3E+01                              | 1                    | ;                                | ;                                | ;                                |
|                                                                              |                                        |                                 |                                 |                                 |                                     |                                     |                                     |                     |                                     |                                     |                     |                      |                                  |                                  |                                  |                        |                                              |                                  |                                  |                                  |                                      |                                      |                                      |                      |                                      |                                      |                      |                                  |                                  |                                  |
| L PAD<br>Fred                                                                |                                        | N/A                             | N/A                             | N/A                             | 1.26-10                             | 1.2E-10                             | 1.2E-10                             | N/A                 | 5.6E-10                             | 5.66-10                             | N/A                 | N/A                  | N/A                              | N/A                              | N/A                              | N/A                    | N/A                                          | N/A                              | N/A                              | N/A                              | 1.6E-09                              | 1.6E-09                              | 1.6E-09                              | N/A                  | 7.2E-09                              | 7.2E-09                              | N/A                  | N/A                              | N/A                              | N/N                              |
| RANGE L'PAD<br>PACTOR FRED                                                   |                                        | N/A                             | N/A                             | N/A                             | 1.2E-10                             | 1.2E-10                             | 1.2E-10                             | N/A                 | 5.6E-10                             | 5.6E-10                             | N/A                 | A/A                  | N/A                              | A/N                              | N/A                              | N/A                    | 1.3E+01 N/A                                  | N/A                              | N/A                              | N/A                              | 1.6E-09                              | 1.6E-09                              | 1.6E-09                              | N/A                  | 7.2E-09                              | 7.2E-09                              | N/A                  | N/A                              | N/A                              | N/A                              |
| AFG RANGE LPAD<br>Freq Factor Freq                                           |                                        | N/A N/A                         | N/A N/A                         | N/A N/A                         | N/A 1.2E-10                         | N/A 1.2E-10                         | N/A 1.2E-10                         | N/A N/A             | N/A 5.6E-10                         | N/A 5.6E-10                         | N/A N/A             | N/A N/A              | N/A N/A                          | N/A N/A                          | N/A N/A                          | N/A N/A                | 8.8E-09 1.3E+01 N/A                          | N/A N/A                          | N/A N/A                          | N/A N/A                          | N/A 1.6E-09                          | N/A 1.6E-09                          | W/A 1.6E-09                          | N/A N/A              | N/A 7.2E-09                          | N/A 7.2E-09                          | N/A N/A              | N/A N/A                          | N/A N/A                          | N/A N/A                          |
| ANGE AFG RANGE LAAD<br>Actor Fred Factor Fred                                |                                        | 3.1E+01 N/A N/A                 | 3.1E+01 N/A N/A                 | 3.1E+01 N/A N/A                 | 3.1E+01 N/A 1.2E-10                 | 3.1E+01 N/A 1.2E-10                 | 3.1E+01 N/A 1.2E-10                 | N/A N/A             | 3.1E+01 N/A 5.6E-10                 | 3.1E+01 N/A 5.6E-10                 | N/A N/A             | N/A N/A              | 1.3E+01 N/A N/A                  | 1.3E+01 N/A N/A                  | 1.3E+01 N/A N/A                  | N/A N/A                | 1.3E+01 8.8E-09 1.3E+01 N/A                  | 1.3E+01 N/A N/A                  | 1.3E+01 N/A N/A                  | 1.3E+01 N/A N/A                  | 1.3E+01 N/A 1.6E-09                  | 1.3E+01 N/A 1.6E-09                  | 1.3E+01 W/A 1.6E-09                  | N/A N/A              | 1.3E+01 N/A 7.2E-09                  | 1.3E+01 N/A 7.2E-09                  | N/A N/A              | 2.6E+01 N/A N/A                  | 2.6E+01 N/A N/A                  | 2.6E+01 N/A N/A                  |
| ANAD RANGE AFG RANGE LEAD<br>Freq Factor Freq Factor Freq                    | ······································ | 6.9E-10 3.1E+01 N/A N/A         | 5.3E-10 3.1E+01 N/A N/A         | 1.2E-10 3.1E+01 N/A N/A         | 1.2E-10 3.1E+01 N/A 1.2E-10         | 1.2E-10 3.1E+01 N/A 1.2E-10         | 1.2E-10 3.1E+01 N/A 1.2E-10         | N/A N/A N/A         | 5.6E-10 3.1E+01 N/A 5.6E-10         | 5.6E-10 3.1E+01 N/A 5.6E-10         | N/A N/A N/A         | N/A N/A N/A          | 6.9E-09 1.3E+01 N/A N/A          | 3.4E-09 1.3E+01 N/A N/A          | 3.4E-09 1.3E+01 N/A N/A          | N/A N/A N/A            | 8.8E-09 1.3E+01 8.8E-09 1.3E+01 N/A          | 8.8E-09 1.3E+01 N/A N/A          | 6.8E-09 1.3E+01 N/A N/A          | 1.6E-09 1.3E+01 N/A N/A          | 1.6E-09 1.3E+01 N/A 1.6E-09          | 1.6E-09 1.3E+01 N/A 1.6E-09          | 1.6E-09 1.3E+01 M/A 1.6E-09          | N/A N/A N/A          | 7.2E-09 1.3E+01 N/A 7.2E-09          | 7.2E-09 1.3E+01 N/A 7.2E-09          | N/A N/A N/A          | 5.8E-09 2.6E+01 N/A N/A          | 2.9E-09 2.6E+01 N/A N/A          | 7.96-09 7.66+01 N/A N/A          |
| OF. NO. ANAD RANGE AFG RANGE LEAD<br>Freq factor freq factor freq            |                                        | 9 6.9E-10 3.1E+01 N/A N/A       | 9 5.3E-10 3.1E+01 N/A N/A       | 9 1.2E-10 3.1E+01 N/A N/A       | 9 1.2E-10 3.1E+01 N/A 1.2E-10       | 9 1.2E-10 3.1E+01 N/A 1.2E-10       | 9 1.2E-10 3.1E+01 N/A 1.2E-10       | 9 N/A N/A N/A       | 9 5.6E-10 3.1E+01 N/A 5.6E-10       | 9 5.6E-10 3.1E+01 N/A 5.6E-10       | 9 N/A N/A N/A       | 10 N/A N/A N/A       | 10 6.9E-09 1.3E+01 N/A N/A       | 10 3.4E-09 1.3E+01 N/A N/A       | 10 3.4E-09 1.3E+01 N/A N/A       | 10 - N/A N/A N/A       | 10 8.8E-09 1.3E+01 8.8E-09 1.3E+01 N/A       | 10 8.8E-09 1.3E+01 N/A N/A       | 10 6.8E-09 1.3E+01 N/A N/A       | 10 1.6E-09 1.3E+01 N/A N/A       | 10 1.6E-09 1.3E+01 N/A 1.6E-09       | 10 1.6E-09 1.3E+01 N/A 1.6E-09       | 10 1.6E-09 1.3E+01 M/A 1.6E-09       | 10 N/A N/A N/A       | 10 7.2E-09 1.3E+01 N/A 7.2E-09       | to 7.2E-09 1.3E+01 N/A 7.2E-09       | 10 N/A N/A N/A       | 11 5.8E-09 2.6E+01 N/A N/A       | 11 2.9E-09 2.6E+01 N/A N/A       | 11 2.9F-09 2.6F+01 N/A N/A       |
| SCEM- OF. NO. ANAD RANGE AFG RANGE LEAD<br>Ario Freq Factor Freq Factor Fred |                                        | HCKVF 9 6.9E-10 3.1E+01 N/A N/A | HCRVF 9 5.3E-10 3.1E+01 N/A N/A | HCFGF 9 1.2E-10 3.1E+01 N/A N/A | HCPHF 9 1.2E-10 3.1E+01 N/A 1.2E-10 | HCFVF 9 1.2E-10 3.1E+01 N/A 1.2E-10 | HCQ6F 9 1.2E-10 3.1E+01 N/A 1.2E-10 | HCQVF 9 N/A N/A N/A | HCK6F 9 5.6E-10 3.1E+01 N/A 5.6E-10 | HCKVF 9 5.6E-10 3.1E+01 N/A 5.6E-10 | HCSVF 9 N/A N/A N/A | HCEGS 10 N/A N/A N/A | HEDHS 10 6.9E-09 1.3E+01 N/A N/A | HCCGS 10 3.4E-09 1.3E+01 N/A N/A | HCCHS 10 3.4E-09 1.3E+01 N/A N/A | HCK6S 10 · N/A N/A N/A | HCkHS 10 8.8E-09 1.3E+01 8.8E-09 1.3E+01 N/A | MCKVS 10 8.8E-09 1.3E+01 N/A N/A | HERVS 10 6.8E-09 1.3E+01 N/A N/A | HCFGS 10 1.6E-09 1.3E+01 N/A N/A | HCPHS 10 1.6E-09 1.3E+01 N/A 1.6E-09 | HCFVS 10 1.6E-09 1.3E+01 N/A 1.6E-09 | HC96S 10 1.6E-09 1.3E+01 H/A 1.6E-09 | HCOVS 10 N/A N/A N/A | HCR6S 10 7.2E-09 1.3E+01 N/A 7.2E-09 | MCRVS 10 7.2E-09 1.3E+01 N/A 7.2E-09 | HCSVS 10 N/A N/A N/A | HCDHC 11 5.8E-09 2.6E+01 N/A N/A | HCCGC 11 2.9E-09 2.6E+01 N/A N/A | HCCHC 11 2.9F-09 2.6F+01 N/A N/A |

20000







File: HDLREGST.Nki · 17-Aug-B7 Fage b

TABLE 6-4 (Continued)

# HANDLING ACCIDENTS - REGIONAL FROCESSING OPTION - PER PALLET

### Accident Frequencies and Range Factors

| I, ANAD RANGE AFG RANGE LBAD RANGE NAAP RANGE P&A RANGE<br>Fred Faciok Freg Factor Freg Factor Fred Factor<br> | RANGE AFG RANGE LBAD RANGE NAAP RANGE P&A RANGE<br>Facior freq facior freq facior freq facior<br> | AFG RANGE LBAD RANGE WAAP RANGE PBA RANGE<br>Freg Factor Freg Factor Freg Factor<br> | RANGE LBAD RANGE WAAP RANGE PRA RANGE<br>Factor Freg Factor Freg Factor Freg Factor<br> | LBAD RANGE NAAP RANGE PBA RANGE<br>Freg Factor Freg Factor Freg Factor<br> | RANGE NAAP RANGE PRA RANGE<br>Factor Freq Factor Freq Factor<br> | NAAP RANGE PRA RANGE<br>Freq Factor Freq Factor | RANGE PRA RANGE<br>Factor Freg Factor | PRA RANGE<br>Freg Factor | RANGE<br>Factor<br> | •        | FUDA<br>Freq | RANGE<br>Factor | TEAD<br>Fred | RANGE<br>Factor | UMDA<br>Fred | RANGE<br>Factor |
|----------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------|----------------------------------------------------------------------------|------------------------------------------------------------------|-------------------------------------------------|---------------------------------------|--------------------------|---------------------|----------|--------------|-----------------|--------------|-----------------|--------------|-----------------|
| 4 3C-00 3 45401 M/A W/A 7 2E-00 2 4                                                                            | 3 15401 W/A W/A 7 25-00 2 6                                                                       | N/A 7.75-00 2.6                                                                      | NIA 275-00 2.6                                                                          | NYA 7 2F-00 2 F                                                            | NA 7 3E-00 2 6                                                   | MA 7 2E-00 2 6                                  | 7 2E-00 2 4                           | 7 7E-00 7 6              | ر<br>م              | 5401     | 4/A          | :               | 15-00        | 2 464ÚT         | 7 7E-00      |                 |
| 9.6E-10 2.6E+01 N/A N/A N/A                                                                                    | 1 2.6E+01 N/A N/A N/A N/A                                                                         | N/A N/A N/A N/A                                                                      | N/A N/A N/A                                                                             | N/A N/A N/A                                                                | N/A N/A                                                          | N/A N/A                                         | N/A                                   | N/A                      |                     | :        | N/A          | ł               | 9.66-10      | 2.6E+01         | 4.86-10      | 9               |
| 9.6E-10 2.6E+01 N/A 4.5E-10 2.6E+01 N/A N/A                                                                    | ) 2.6E+01 N/A 4.5E-10 2.6E+01 N/A N/A                                                             | N/A 4.5E-10 2.6E+01 N/A N/A                                                          | 4.5£-10 2.6E+01 N/A N/A                                                                 | 4.5E-10 2.6E+01 N/A N/A                                                    | 2.6E+01 N/A N/A                                                  | N/A N/A                                         | N/A                                   | N/A                      |                     | 1        | 4.85-10      | 2.6E+01         | 9.65-10      | 2.6E+01         | N/A          | 1               |
| 9.6E-10 2.6E+01 N/A 4.8E-10 2.6E+01 N/A N/A                                                                    | ) 2.6E+01 N/A 4.8E-10 2.6E+01 N/A N/A                                                             | N/A 4.8E-10 2.6E+01 N/A N/A                                                          | 4.8E-10 2.6E+01 N/A N/A                                                                 | 4.8E-10 2.6E+01 N/A N/A                                                    | 2.6E+01 N/A N/A                                                  | N/A N/A                                         | N/A                                   | N/A                      |                     | ł        | N/A          | :               | 9.6E-10      | 2.6E+01         | 4.85-10      | 2 6             |
| 7.2E-10 2.6E+01 N/A 3.6E-10 2.6E+01 N/A N/A                                                                    | * 2.6E+01 N/A 3.6E-10 2.6E+01 N/A N/A                                                             | N/A 3.6E-10 2.6E+01 N/A N/A                                                          | 3.6E-10 2.6E+01 N/A N/A                                                                 | 3.6E-10 2.6E+01 N/A N/A                                                    | 2.6E+01 N/A N/A                                                  | N/A N/A                                         | N/A                                   | N/A                      |                     | ;        | N/A          | ł               | 7.2E-10      | 2.6E+01         | 3.6E-10      | 2.6             |
| N/A N/A N/A N/A N/A                                                                                            | N/A N/A N/A N/A                                                                                   | N/A N/A N/A N/A                                                                      | N/A N/A N/A                                                                             | N/A N/A N/A                                                                | N/A N/A                                                          | N/A N/A                                         | N/A                                   | N/A                      |                     | ;        | N/A          | :               | 7.2E-10      | 2.6E+01         | 3.6E-10      | 2.6             |
| 1.8E-09 2.6E+01 N/A 9.0E-10 2.6E+01 N/A 9.0E-10                                                                | * 2.6E+01 N/A 9.0E-10 2.6E+01 N/A 9.0E-10                                                         | N/A 9.0E-10 2.6E+01 N/A 9.0E-10                                                      | 9.0E-10 2.6E+01 N/A 9.0E-10                                                             | 9.0E-10 2.6E+01 N/A 9.0E-10                                                | 2.6E+01 N/A 9.0E-10                                              | N/A 9.0E-10                                     | 9.0E-10                               | 9.06-10                  |                     | 2.6E+01  | N/A          | ł               | 1.85-09      | 2. bE +01       | 9.0E-10      | 2.6             |
| 1.8E-09 2.6E+01 N/A 9.0E-10 2.6E+01 N/A 9.0E-10                                                                | 2.6E+01 N/A 9.0E-10 2.6E+01 N/A 9.0E-10                                                           | N/A 9.0E-10 2.6E+01 N/A 9.0E-10                                                      | 9.0E-10 2.6E+01 N/A 9.0E-10                                                             | 9.0E-10 2.6E+01 N/A 9.0E-10                                                | 2.6E+01 N/A 9.0E-10                                              | N/A 9.0E-10                                     | 9,0E-10                               | 9.0E-10                  |                     | 2.6E+01  | N/A          | :               | 1.85-09      | 2.6E+01         | 9.06-10      | 7.6E            |
| 4.1E-10 2.6E+01 N/A N/A N/A N/A                                                                                | n 2.6E+01 N/A N/A N/A N/A                                                                         | N/A N/A N/A N/A                                                                      | N/A N/A N/A                                                                             | N/A N/A N/A                                                                | N/A N/A                                                          | N/A N/A                                         | N/A                                   | N/A                      |                     | 1        | 2. IE-10     | 2.6E+01         | 4.1E-10      | 2.6E+01         | N/A          | ;               |
| · 2.16-10 2.66+01 N/A N/A N/A N/A                                                                              | 1 2.6E+01 N/A N/A N/A N/A                                                                         | N/A N/A N/A N/A                                                                      | N/A N/A N/A                                                                             | N/A N/A N/A                                                                | N/A N/A                                                          | N/A N/A                                         | N/A                                   | N/A                      |                     | 1        | N/A          | 1               | 2.1E-10      | 2.6E+01         | N/A          | 1               |
| · 2.1E-10 2.6E+01 N/A N/A N/A N/A                                                                              | ) 2.6E+i)] N/A N/A N/A N/A                                                                        | N/A N/A N/A N/A                                                                      | N/A N/A N/A                                                                             | N/A N/A N/A                                                                | N/A N/A                                                          | N/A N/A                                         | N/A                                   | N/A                      |                     | ł        | 1.05-10      | 2.6E+(11        | 2.1E-10      | 2.6E+01         | N/A          | 1               |
| · 3.1E-10 2.6E+01 N/A N/A 1.5E-                                                                                | ) 2.6E+01 N/A N/A 1.5E-                                                                           | N/A N/A N/A 1.5E-                                                                    | N/A N/A 1.5E-                                                                           | N/A N/A 1.5E-                                                              | N/A 1.5E-                                                        | N/A 1.5E-                                       | 1.5E-                                 | 1.5E-                    | 91                  | 2.6E+01  | N/A          | ł               | 3. IE-10     | 2.6E+01         | 1.51-10      | 2.6E            |
| · 6.9E-11 2.6E+01 N/A N/A N/A N/A                                                                              | 2.6E+01 N/A N/A N/A N/A                                                                           | N/A N/A N/A N/A                                                                      | N/A N/A N/A                                                                             | N/A N/A N/A                                                                | N/A N/A                                                          | N/A N/A                                         | N/A                                   | N/A                      |                     | ł        | N/A          | ļ               | 6.9E-11      | 2.6E+01         | 3.4E-11      | 2.6E            |
| · 6.9E-11 2.6E+01 N/A 3.4E-11 2.6E+01 N/A N/A                                                                  | : 2.6E+01 N/A 3.4E-11 2.6E+01 N/A N/A                                                             | N/A 3.4E-11 2.6E+01 N/A N/A                                                          | 3.4E-11 2.6E+01 N/A N/A                                                                 | 3.4E-11 2.6E+01 N/A N/A                                                    | 2.6E+01 N/A N/A                                                  | N/A N/A                                         | N/A                                   | N/A                      |                     | ;        | 3.4E-11      | 2.6E+01         | 6.9E-11      | 2.6E+01         | N/A          | ł               |
| 6.9E-11 2.6E+01 N/A 3.4E-11 2.5E+01 N/A N/                                                                     | 2.6E+01 N/A 3.4E-11 2.6E+01 N/A N/                                                                | N/A 3.4E-11 2.6E+01 N/A N//                                                          | 3.4E-11 2.6E+01 N/A N/6                                                                 | 3.4E-11 2.6E+01 N/A N/A                                                    | 2.6E+01 N/A N/F                                                  | N/A N/A                                         | 1/N                                   | 1/N                      | -                   | ł        | N/A          | 1               | 6.9E-11      | 2.6E+ń          | 3.4E-11      | 2.6E            |
| · 5.2E-11 2.6E+01 N/A 2.6E-11 2.6E+01 N/A N/                                                                   | 2.6E+01 N/A 2.6E-11 2.6E+01 N/A N/                                                                | N/A 2.6E-11 2.6E+01 N/A N/                                                           | 2.6E-11 2.6E+01 N/A N/                                                                  | 2.6E-11 2.6E+01 N/A N/                                                     | 2.6E+01 N/A N/                                                   | N/A N/                                          | W                                     | ¥                        | æ                   | 1        | N/A          | ł               | 5.2E-11      | 2.6E+01         | 2.6E-11      | 2.6E            |
| N/A N/A N/A N/A N/                                                                                             | N/A N/A N/A N/                                                                                    | N/A N/A N/A N/                                                                       | N/A N/A N/                                                                              | N/A N/A N/                                                                 | N/A N/                                                           | N/A N/                                          | / <b>X</b>                            | ż                        | 4                   | ;        | N/A          | -               | 5.2E-11      | 2.6E+01         | 2.65-11      | 2.6E            |
| · 1.36-10 2.6E401 N/A 6.5E-11 2.6E401 N/A 6.5                                                                  | 1 2.6E+01 N/A 6.5E-11 2.6E+01 N/A 6.5                                                             | N/A 6.5E-11 2.6E+01 N/A 6.5                                                          | 6.5E-11 2.6E+01 N/A 6.5                                                                 | 6.5E-11 2.6E+01 N/A 6.5                                                    | 2.6E+01 N/A 6.5                                                  | N/A 6.5                                         | 6.5                                   | 6.5                      | E-11                | 2.6E+01  | N/A          | ;               | 1.36-10      | 2.6E+01         | 6.5E-11      | 2.6E            |
| 1.3E-10 2.6E+01 N/A 6.5E-11 2.6E+01 N/A 6.51                                                                   | 0 2.6E+01 N/A 6.5E−11 2.6E+01 N/A 6.51                                                            | N/A 6.5E-11 2.6E+01 N/A 6.5                                                          | 6.5E-11 2.6E+01 N/A 6.51                                                                | 6.5E-11 2.6E+01 N/A 6.5                                                    | 2.6E+01 N/A 6.5                                                  | N/A 6.5                                         | 6.51                                  | 6.5                      | = <u>-</u>          | 2.6E+01  | N/A          | ;               | 1.35-10      | 2.6E+01         | 6.5E-1       | 2.6E            |
| N/A N/A N/A N/A N/A                                                                                            | N/A N/A N/A N/A                                                                                   | N/A N/A N/A N/I                                                                      | N/A N/A N/I                                                                             | N/A N/A N/I                                                                | N/A N/I                                                          | N/A N/I                                         | N/I                                   | N/I                      | -                   | 1        | N/A          | ł               | 6.0E-13      | 2.6E+0)         | 6.nE-13      | 2.6F            |
| 6.0E-13 2.6E+01 N/A 6.0E-13 2.6E+01 N/A 6.0                                                                    | : 2.6E+01 N/A 6.0E-13 2.6E+01 N/A 6.0                                                             | N/A 6.0E-13 2.6E+01 N/A 6.0                                                          | 6.0E-13 2.6E+01 N/A 6.0                                                                 | 6.0E-13 2.6E+01 N/A 6.0                                                    | 2.6E+01 N/A 6.0                                                  | N/A 6.0                                         | 9.0                                   | 9°0                      | E-13                | 2.6E+01  | N/A          | ;               | 6.0E-13      | 2.6E+01         | 6.0E-13      | 2, 6E           |
| 6.0E-13 2.6E+01 N/A 6.0E-13 2.6E+01 N/A 6.0E                                                                   | : 2.6E+01 N/A 6.0E-13 2.6E+01 N/A 6.0E                                                            | N/A 6.0E-13 2.6E+01 N/A 6.0E                                                         | 6.0E-13 2.6E+01 N/A 6.0E                                                                | 6.0E-13 2.6E+01 N/A 6.0E                                                   | 2.6E+01 N/A 6.0E                                                 | N/A 6.0E                                        | 6.05                                  | <b>6.</b> 0E             | -13                 | 2.6E+01  | N/A          | ł               | 6.0E-13      | 2,66+01         | 6.0E-13      | 2.6E            |
| N/A N/A N/A N/A N/A                                                                                            | N/A N/A N/A N/A                                                                                   | N/A N/A N/A N/A                                                                      | N/A N/A N/A                                                                             | N/A N/A N/A                                                                | N/A N/A                                                          | N/A N/A                                         | N/A                                   | N/A                      |                     | :        | N/A          | ;               | 1.2E-12      | 2.6E+01         | 1.2E-12      | 2. 6E           |
| 1.2E-12 2.6E+01 N/A 1.2E-12 2.6E+01 N/A 1.2E                                                                   | 2.66+01 N/A 1.26-12 2.66+01 N/A 1.26                                                              | N/A 1.2E-12 2.6E+01 N/A 1.2E                                                         | 1.2E-12 2.6E+01 N/A 1.2E                                                                | 1.2E-12 2.6E+01 N/A 1.2E                                                   | 2.6E+01 N/A 1.2E                                                 | N/A 1.2E                                        | 1.25                                  | Ч.<br>Ч                  | -12                 | 2.6E+01  | N/A          | 1               | 1.2E-12      | 2.6E+01         | 1.26-12      | 2.6E            |
| 1.2E-12 2.6E+01 N/A 1.2E-12 2.6E+01 N/A 1.2E-                                                                  | 1 2.6E+01 N/A 1.2E-12 2.6E+01 N/A 1.2E-                                                           | N/A 1.2E-12 2.6E+01 N/A 1.2E-                                                        | 1.2E-12 2.6E+01 N/A 1.2E-                                                               | 1.2E-12 2.6E+01 N/A 1.2E-                                                  | 2.6E+01 N/A 1.2E-                                                | N/A 1.2E-                                       | 1.2E-                                 | 1.25-                    | 12                  | 2.6E+01  | N/A          | ł               | 1.2E-12      | 2.6E+01         | 1.2E-12      | 2. KE +         |
| N/A N/A N/A N/A N/A                                                                                            | N/A N/A N/A N/A                                                                                   | N/A N/A N/A N/A                                                                      | N/A N/A N/A                                                                             | N/A N/A N/A                                                                | N/A N/A                                                          | N/A N/A                                         | N/A                                   | N/A                      |                     | !        | N/A          | ł               | 2.66-15      | 3. IE+01        | 2.6E-15      |                 |
| 2.46-14 3.16+01 N/A 2.46-14 3.16+01 N/A 2.46-1                                                                 | 1 3.1E+01 N/A 2.5E-14 3.1E+01 N/A 2.6E-1                                                          | N/A 2.5E-14 3.1E+01 N/A 2.6E-1                                                       | 2.6E-14 3.1E+01 N/A 2.6E-1                                                              | 2.6E-14 3.1E+01 N/A 2.6E-1                                                 | 3.1E+01 N/A 2.6E-1                                               | N/A 2.6E-1                                      | 2.6E-1                                | 2.6E-1                   | -                   | 3. IE+01 | N/A          | 1               | 2.6E-14      | 3. IE+01        | 2.65-14      | <br>1E          |
| 2.6E-14 J.1E+11 N/A 2.6E-14 J.1E+01 N/A 2.6E-                                                                  | U 3.1E+U1 N/A 2.6E-14 3.1E+U1 N/A 2.6E-                                                           | N/A 2.6E-14 3.1E+01 N/A 2.6E-                                                        | 2.6E-14 3.1E+01 N/A 2.6E-                                                               | 2.6E-14 3.1E+01 N/A 2.6E-                                                  | 3.1E+01 N/A 2.6E-                                                | N/A 2.6E-                                       | 2.6E-                                 | 2.6E-                    | 1                   | 3.16+01  | N/A          | :               | 2.6E-14      | 3. IE+01        | 2.66-14      | 3. IE           |
| N/A N/A N/A N/A N/A                                                                                            | N/A N/A N/A N/A                                                                                   | N/A N/A N/A N/A                                                                      | N/A N/A N/A                                                                             | N/A N/A N/A                                                                | N/A N/A                                                          | N/A N/A                                         | N/A                                   | N/A                      |                     | !        | N/A          | ;               | 3.4E-18      | 3. IE+01        | 3.46-18      | 3. IE           |

6-41

TABLE 6-4 (Continued)

Page 7

17-Aug-87

File: HDLREGST.Wk1

4. 8. 8 ( And 10. 0.

# HANDLING ACCIDENTS - REGIONAL PROCESSING OPTION - PER PALLET

### Accident Frequencies and Range Factors

| SCEN-<br>ARID | 0F. NO | . ANAD<br>Freq | RANGE<br>Factor  | AFG<br>Freq | RANGE<br>Factor | L BAD<br>Freq         | RANGE<br>Factor | NAAP<br>Freg | RANGE<br>Factor | FEA<br>Freg | RANGE<br>Factor | PUDA<br>FREG | RANGE<br>Factor | TEAD<br>Freq | RANGE<br>Factor | UMDA<br>Freg | RANGE<br>Factor |
|---------------|--------|----------------|------------------|-------------|-----------------|-----------------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|-----------------|
|               | •      |                | ,<br>,<br>,<br>, |             | <br> <br>       | 1<br>4<br>4<br>1<br>1 |                 |              |                 |             |                 |              |                 |              |                 |              |                 |
| HCRGC         | 21     | 3.1E-18        | 3.1E+01          | N/A         | 1               | 3.16-18               | 3. JE+01        | N/A          | ;               | 3. 1E - 18  | 3. IE+01        | N/A          | ;               | 3, 1E-18     | 3.1E+01         | 3.1E-18      | 3. 1E+01        |
| HCRVC         | 21     | 7.16-18        | 3. IE+01         | N/A         | ;               | 3.16-18               | 3.1E+01         | N/A          | ł               | 3.16-18     | 3. IE+01        | N/A          | 1               | 3.1E-18      | 3.16+01         | 3.1E-18      | 7, 1E+01        |
| HCDHC         | 22     | 8.75-11        | 2.66+01          | N/A         | :               | N/A                   | :               | N/A          | ;               | N/A         | ;               | N/A          | ł               | B. 7E-11     | 2.6E+01         | N/A          | ;               |
| HCCEC         | 22     | 4.3E-11        | 2.6E+01          | NZA         | 1               | N/A                   | 1               | N/A          | ;               | N/A         | ;               | N/A          | 1               | 4.3E-11      | 2.6E+01         | N/A          | 1               |
| HCCHC         | 72     | 4.36-11        | 2.6E+01          | A'N         |                 | N/A                   | ł               | N/A          | ;               | A/A         | ł               | N/A          | ;               | 4.35-11      | 2.6E+01         | N/A          | ;               |
| HCMITC        | 22     | 6.5E-11        | 2.6E+01          | N/A         | ł               | N/A                   | :               | N/A          | ;               | N/A         | 1               | N/A          | ;               | 6.5E-11      | 2.6E+01         | N/A          | :               |
| HCFGC         | 22     | 1.4E-11        | 2.6E+01          | 4.'N        | ł               | N/A                   | :               | N/A          | ;               | N/A         | 1               | N/A          | :               | 1.46-11      | 2.6f+01         | A/A          | ;               |
| Энд Эн        | 22     | 1.46-11        | 2.6E+ü1          | A.A         | ł               | N/A                   | ;               | N,A          | :               | N/A         | ;               | N/A          | ;               | 1.46-11      | 2.66+01         | N/A          | ;               |
| RCFVC         | 22     | 1.4E-11        | 2.6E+01          | N/A         | ;               | N/A                   | 1               | N/A          | ;               | N , A       | :               | N/A          | 1               | 1.46-11      | 2.6E+U]         | A'A          | ;               |
| JJO JH        | 11     | 1.16-11        | 2.6E+01          | N/A         | ;               | N/A                   | 8<br>1          | N/A          | ł               | N/A         | ł               | N/A          | 1               | 1.16-11      | 2.6E+01         | N/A          | ţ               |
| HCOVC         | 22     | N/A            | :                | N/A         | :               | A/X                   | :               | N/A          | ;               | A/A         | :               | N/A          | :               | 1.16-11      | 2.6E+01         | N/A          | ;               |
| нсрес         | 22     | 2.7E-11        | 2.6E+01          | NA          | ;               | N/A                   | :               | A/K          | ţ               | N/A         | ;               | N/A          | 1               | 2.76-11      | 2.6E+01         | N/4          | 1               |
| HCRVC         | 22     | 2.76-11        | 2.6E+01          | N/A         | 1               | N/A                   | !               | N/A          | 1               | N/A         | ;               | N/A          | +               | 2.7E-11      | 2.6E+01         | N/A          | ;               |
| HC DHC        | 23     | 2.3E-11        | 2.6E+01          | N/A         | ţ               | N/A                   | 1               | N/A          | !               | A/A         | 1               | 2.3E-11      | 2.6E+:)1        | 2.36-11      | 2.6E+01         | N/A          | ;               |
| ::093         | 23     | 3.5E-11        | 2.6€+01          | N/A         | 1               | N/A                   | ;               | N/A          | ;               | N/A         | :               | N/A          | ;               | 3.56-11      | 2. 6E+01        | N/A          | 1               |
| HCCHC         | 23     | 3.56-11        | 2.6E+U1          | N/A         | ;               | N/A                   | ł               | N/A          | ;               | N, A        | ;               | 1.56-11      | 2. 5E +Ù]       | 3.5E-11      | 2.6E+01         | N/A          | ;               |
| HCMVC         | 23     | 1.35-11        | 2.66+01          | N/A         | ;               | N/A                   | ;               | N/A          | ţ               | 1.3E-11     | 2.6E+01         | N/A          | ;               | 1.35-11      | 2.6E+01         | 1.35-11      | 2.6E+11         |
| HCFGC         | 23     | 1.45-11        | 2.6E+01          | A/N         | !               | N/A                   | ł               | N/A          | ł               | 4/X         | ;               | N/A          | ;               | 1.45-11      | 2.6E+01         | 1.46-11      | 2. 6E +01       |
| HCFHC         | 23     | 1.46-11        | 2.6E+01          | N/A         | :               | 1.46-11               | 2.6E+01         | N/A          | ì               | N/A         | 1               | 1.46-11      | 2.6E+01         | 1.46-11      | 2.6E+01         | N/A          | ;               |
| JV4JH         | 23     | 1.4E-11        | 2.6E+01          | N/A         | 1               | 1.4E-11               | 2.6E+01         | N/A          | 1               | N / A       | ;               | N/A          | 1               | 1.4E-11      | 2.6E+01         | 1.46-11      | 2. 6E 401       |
| HCGGC         | 23     | 7.2E-12        | 2.6E+01          | A/A         | ł               | 7.25-12               | 2.6E+01         | N/A          | i               | N/A         | ţ               | N/A          | :               | 7.2E-12      | 2.6E+01         | 7.2E-12      | 2.6[+11]        |
| HCQVC         | 21     | N/A            | 1                | N/A         | ;               | N/A                   | ;               | N/A          | ;               | N/A         | 1               | N/A          | ;               | 7.2E-12      | 2.6E+U1         | 7.2E-12      | 2.6E+01         |
| HCRGC         | 23     | 7.2E-12        | 2.6E+01          | N/A         | ł               | 7.2E-12               | 2.6E+(i)        | N/A          | ;               | 1.2E 12     | 2.6E+01         | N/A          | ;               | 7.2E-12      | 2,6E+01         | 7.26-12      | 2.6E+01         |
| HCAVC         | 23     | 7.2E-12        | 2.6E+01          | N/A         | ł               | 7.2E-12               | 2.6E+01         | N/A          | 1               | 7.26-12     | 2.6E+01         | N/A          | :               | 7.2E-12      | 2.6E+01         | 7.2E-12      | 2.66+01         |
| HCDHC         | 24     | 6.2E-11        | 2.6€+01          | N/A         | ;               | N/A                   | :               | A/A          | ſ               | N/A         | 1               | N/A          | :               | 6.2E-11      | 2.6E+01         | N/A          | ;               |
| HCC6C         | 24     | 3. 16-11       | 2.6E+01          | N/A         | ;               | N/A                   | ł               | N/A          | ł               | N/A         | :               | N/A          | :               | 3.1E-11      | 2.6E+01         | N/A          | ;               |
| HCCHC         | 24     | 3.1E-11        | 2.6E+01          | A/A         | ;               | N/A                   | ;               | N/A          | ;               | A'N         | ;               | N/A          | ;               | 3. IE-11     | 2.6E+Ù1         | N/A          | ;               |
| JANJH         | 24     | 4.65-11        | 2.6E+01          | A/A         | ;               | N/A                   | ł               | N/A          | ł               | 8/N         | :               | N/A          | ;               | 4.6E-11      | 2.6E+01         | N/A          | :               |
| HCPGC         | 24     | 1.05-11        | 2.6E+01          | N/A         | ł               | N/A                   | ;               | N/A          | :               | N/A         | •               | N/A          | :               | 1.0E-11      | 2.6E+01         | N/A          | ;               |

1.2222

- AARAA TARRAA

6-42

C.

いたいとうとうという。こことであるので、たちからないので

TABLE 6-4 (Continued)

Fage B

17-Aug-87

File: HDLREGST.WK1

HANDLING ACCIDENTS - REGIONAL PROCESSING OFTION - PER PALLET

Accident Frequencies and Range Factors

| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | •       | •       | !        | ł       | ļ       | ,       | ţ       | ŗ         | !       | €+a]     | E +i)]  |         | E 401              | f+0]          | E+i)]   | E+u)        | 10+3     | !       | ,        | 1       |             |         | ł       |         |         | 1       |         | I       |              |  |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------|---------|----------|---------|---------|---------|---------|-----------|---------|----------|---------|---------|--------------------|---------------|---------|-------------|----------|---------|----------|---------|-------------|---------|---------|---------|---------|---------|---------|---------|--------------|--|
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | '       | '       | I        | '       | 1       | ł       | I       |           | 1       | -12 2.6  | 11 2.6  | •       | 11 2.6             | -12 2.6       | 12 2.6  | -12 2.6     | 12 2.6   | 1       | ſ        | 1       | ,           |         | '       | '       | •       | ,       | ,       |         |              |  |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | I N/A   | 1 N/A   | 1 N/A    | I N/A   | I N/A   | 1 N/A   | I N/A   | I N/A     | 1 N/A   | 1 9.3E-  | 1 1.0E  | 1 N/A   | - <b>J</b> 0-1 - i | 1 5. JE-      | 1 5.2E- | 1 5.2E-     | l 5.2E-  | N/A     | N/A      | N/A     | N/A         | N/A     | N/A     | N, A    | R/A     | N/6     | N/A     | A/A     | N/A          |  |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 2.6E+0  | 2.6E+0  | 2.6E+()  | 2.6E+0  | 2.6E+U  | 2.6E+0  | 2.6E+ú  | 2.6E+0    | 2.6E+0  | 2.6E+0   | 2.6E+0  | 2.6E+() | 2.6E+0             | 2.6E+0        | 2.6E+ú  | 2.6E+U      | 2.6E+0   | !       | ;        | ;       | 1           | ;       | ;       | :       | :       | ;       | ł       | ł       | :            |  |
| 21       1.0F-11       2.6F-01       WA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 1.0E-11 | 1.0E-11 | 7. JE-12 | 7.7E-12 | 1.9E-11 | 1.96-11 | 1.7E-11 | 2.5E-11   | 1.5E-11 | 9.3E-12  | 1.0E-11 | 1.0E-11 | 11-30-1            | 5.2E-12       | 5.2E-12 | 5.2E-12     | 5.2E-12  | 0.0E+00 | 0°*0E+00 | 0.0E+00 | 0.05+00     | 0.01400 | 0.0E+00 | 0.01400 | 0,0E+0n | 0.05+00 | 0.0E+00 | 0.0E+00 | 0•0E+00      |  |
| 21       1.0E-11       2.6E+01       WA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | :       | 1       | ;        | ł       | ł       | 1       | 2.6E+01 | 1         | 2.6E+01 | ł        | ;       | 2.6E+01 | ;                  | ;             | 1       | ł           | ;        | ;       | :        | ;       | ł           | :       | •       | !       | 1       | :       | ;       | 1       | t<br>i       |  |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | .7E-11  | N/A       | 2.5E-11 | N/A      | N/A     | 1.0E-11 | N/A                | N/A           | N/A     | N/A         | N/A      | N/A     | N/A      | N/A     | N/A         | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | A.A          |  |
| 24       1.06-11       2.66+01       WA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | ţ       | ;       | :        | 1       | ł       | ;       | -       | ;         | ; 4     | ., 6E+01 | ;       | -       | :                  | :             | !       | .,6E+01     | 6E+01    | 1       | ;        | ;       | ;           | :       | ;       | :       | ;       | ;       | ;       | !       | :            |  |
| 21       1.0E-11       2.6E+01       W/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A       | N/A     | .3E-12 2 | N/A     | N/A     | N/A                | N/A           | N/A     | . 2E - 12 2 | .2E-12 2 | N/A     | N/A      | N/A     | N/A         | N/A     | N/A     | N/A     | N/A     | N/A     | NA      | N/A     | N/A          |  |
| 24       1.0E-11       2.6E+01       W/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ;       | ;       | ł        | ;       | 1       | 1       | ;       | ;         | ļ       | 6        | ;       | ł       | :                  | ł             | ł       | :           | -<br>-   | ;       | :        | :       | ;           | ;       | :       | ;       | ;       | ;       | !       | 1       | ;            |  |
| 71       1.06-11       2.66+01       N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Α,      | A/      | A /      | A /     | A /     | A /     | A /     | /A        | A /     | /A       | Å,      | /A      | / A                | (A            | A /     | ۲۹          | A.       | A)      | đ,       | A /     | (A          | A /     | A,      | /A      | A /     | đ       | đ,      | ۲A      | ۲.           |  |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Ż       | Ń       | Ŷ        | Ŷ       | Ż       | Ì       | Ì       | Ŷ         | Ż       | Ŷ        | 2       | 101 N/  | /N 10+             | 'N 10+        | Ì       | /N 10+      | /N 10+   | Ŷ       | Ń        | Â.      | Ŷ           | N,      | Ì       | Ì       | Ń       | Ŷ       | /H/     | Ŷ       | Z            |  |
| 21       1.0E-11       2.6E+01       N/A        N/A         21       2.7E-12       2.6E+01       N/A        N/A         22       2.4       1.9E-11       2.6E+01       N/A        N/A         22       2.4       N/A        N/A        N/A         23       1.9E-11       2.6E+01       N/A        N/A         25       1.7E-11       2.6E+01       N/A        N/A         26       25       1.6E-11       2.6E+01       N/A        N/A         26       25       2.5E-11       2.6E+01       N/A        N/A         27       1.0E-11       2.6E+01       N/A        N/A         28       1.0E-11       2.6E+01       N/A        N/A         28       1.0E-01       2.6E+01       N/A        N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 1       | ;       | ;        | •       | ;       | 1       | 1       | •         | :       | 1        | :       | 1 2.6E  | 1 2.6E             | 2 2.6E        | 1       | 2 2.6E      | 2 2.6E   | ;       | 1        | 1       | ;           | 1       | ;       | ;       | 1       | :       | :       | ;       | {            |  |
| 6       21       1.0E-11       2.6E+01       N/A          7       7.7E-12       2.6E+01       N/A        N/A          7       7.7E-12       2.6E+01       N/A        N/A          7       7.7E-12       2.6E+01       N/A        N/A          7       7.7E-11       2.6E+01       N/A        N/A          8       7.7E-11       2.6E+01       N/A        N/A          8       25       1.7E-11       2.6E+01       N/A            8       25       2.5E-11       2.6E+01       N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A       | N/A     | N/A      | A/A     | 1.0E-1  | 1-90-1             | 5.26-1        | N/A     | 5.2E-1      | 5.2E-1   | N/A     | N/A      | N/A     | NJA         | N/A     | N/F     | N/A     | N/A     | ¥,¤     | 4 / N   | N/A     | ₹/N          |  |
| C       24       1.0E-11       2.6E+01       N/A         C       24       1.0E-11       2.6E+01       N/A         C       24       1.0E-11       2.6E+01       N/A         C       24       1.9E-11       2.6E+01       N/A         C       24       1.9E-11       2.6E+01       N/A         C       24       1.9E-11       2.6E+01       N/A         C       25       1.7E-11       2.6E+01       N/A         C       25       1.7E-11       2.6E+01       N/A         C       25       1.0E-11       2.6E+01       N/A         C       25       1.0E-01       2.6E+01       N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | ł       | ;       | ;        | ł       | t       | ;       | 1       | ł         | ł       | ;        | ;       | ł       | ł                  | 1             | ;       | 1           | ;        | ;       | 1        | ł       | ł           | t<br>L  | ł       | ţ       | ļ       | ;       | ł       | :       | ł            |  |
| C       24       1.06-11       2.66401         C       24       1.06-11       2.66401         C       24       1.06-11       2.66401         C       24       1.96-11       2.66401         C       25       1.76-11       2.66401         C       25       1.06-11       2.66401         C       25       1.06-10       1.         C       25       0.06-00       1.         C       25       0.06-00       1.         C       26       0.06-00       1.         Z       26                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A       | N/A     | N/A      | N 2 A   | N A     | N/À                | N/A           | N/A     | NZA         | N/A      | N/A     | N / G    | N/C     | 4/A         | N/A     | N/A     | N/A     | N/A     | N, A    | N/A     | U.S.    | -<br>        |  |
| 1     24     1.06-11       1     24     1.06-11       1     24     1.06-11       24     1.96-11       24     1.96-11       24     1.96-11       25     1.96-11       26     24       27     1.96-11       26     26       27     1.96-11       28     1.96-11       29     1.96-11       25     2.56-12       26     1.06-11       27     1.06-11       28     1.06-11       29     1.06-11       20     25       21     26       26     0.06-00       27     0.06-00       28     0.06-00       29     0.06-00       20     0.06-00       21     26       28     0.06-00       28     0.06-00       29     0.06-00       20     0.06-00       21     26       28     0.06-00       29     0.06-00       20     0.06-00       20     0.06-00       20     0.06-00       21     0.06-00       28     0.06-00       29     0.06-00                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 2.6E+01 | 2.6E+01 | 2.6E+01  | :       | 2.6E+01 | 2.6E+01 | 2.6E+01 | 2. bE +m1 | 2.6E+01 | 2.6F+01  | 2.6E+01 | 2.6E+01 | 2.6E+ii1           | 2.6E+01       | ;       | 2.6E+01     | 2.6E+U1  | ł       | 1        | !       | ;           | ;       | ;       | ;       | 2       | !       | 1       | ł       | :            |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 11-30.  | .06-11  | .76-12   | M/A     | 11-36.  | .96-11  | .76-11  | 11-35.    | .5E-11  | .36-12   | N-30.   | 11-30.  | 11-30.             | .25-12        | N/A     | .21-12      | , 2E-12  | N/A     | .0E+00   | úu+30.  | .0E+0û      | N/A     | 0€+00   | ùQ+3ù*  | 00+30°  | 00+3ú°  | 00+30°  | 0E+00   | · 10 • 311 · |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 24 1.   | 24 1    | 24 7.    | 24      | 24 1.   | 24 1.   | 25 1.   | 25 2      | 25 2.   | 25 9.    | 25 1    | 25 1.   | 25 1.              | 25. 5.        | 25      | 25 5.       | 25 5.    | 26      | 26 0.    | 26 11   | 25 0.       | 26 1    | 26 11,  | 26 11.  | 26 (i.  | 26 0.   | 26 0.   | .0 9.7  | 26 9.        |  |
| <ul> <li>HCFA</li> <li>HCFA</li></ul> | HCFHC   | HCFVC   | HCDEC    | HCOVC   | HCK6C   | HCRVC   | HC DHC  | HCCBC     | HCCHC   | HCHVC    | HCF6C   | HCFHC   | HCEVC              | <u> Э</u> ӨСН | HCGVC   | HCRGC       | HLRVE    | HC PGF  | HUDHC    | HCCGC   | <b>JHDH</b> | HCKEF   |         | HCFVF   | 36438   | нСББС   | JH JH   | JV 13H  | JODON        |  |

File: HDLREGSJ.WK1 17-Aug-87 Page 9

TABLE 6-4 (Continued)

HANDLING ACCIDENTS - REGIONAL PROCESSING OPTION - PER PALLET

Accident Frequencies and Range Factors

| IANGE<br>Actor   |   | ;       | !          | ;        | ł        | ;         | i       | !       | ;       | !       | ;       | ;       | ;       | ;         | ;          | ł        | :        | !       | ;                  | ł                    | ;        | 2.6E+1)1   | 2.66401      | 2. 6E 401  | 2.66+01    | 2. éE turl       | 2. 6E HUT | 1. QE +01 | 1, (E tụ 1 | 1.06401   |
|------------------|---|---------|------------|----------|----------|-----------|---------|---------|---------|---------|---------|---------|---------|-----------|------------|----------|----------|---------|--------------------|----------------------|----------|------------|--------------|------------|------------|------------------|-----------|-----------|------------|-----------|
| UNDA F<br>FRED F |   | N/N     | N/A        | N/A      | N/A      | 0° ÚE +00 | N/A     | N/A     | N/A     | A/A     | 0.0E+00 | N/A     | 0.0E+00 | 0°.0E+00  | M/A        | 0,0130.0 | 0°+30°0  | 0.0E+00 | 0°0£400            | 0° 0E+00             | n, 0E+0u | 9.05-09    | 9. (iE - (i9 | 1.8E-0.8   | 1.85-08    | 1.36-10          | 1.36-10   | 1.6E-(5)  | 1.0E-U;    | 1.0E-0.   |
| RANGE<br>Factor  |   | !       | !          | 1        | ł        | 1         | ł       | ;       | ł       | ;       | ł       | ;       | ł       | ł         | 1          | ł        | ;        | :       | i                  | ;                    | !        | 2.6E+01    | 2.66+01      | 2.6E+01    | 2.66401    | 2.6E+(i)         | 2.6E+01   | 1. nE +ú1 | 1.((E+1))  | 1.05+01   |
| TEAD<br>Freq     |   | 9.05+00 | 0.0E+00    | 0.0E+00  | 0°.0E+00 | 0.015+00  | 0.01400 | 0.0E+00 | 0.0£+00 | 0.0E+00 | 0.0E+00 | 0.01400 | 0.0E+00 | 0.0+30.00 | 0.0E+00    | 0°.0E+00 | 0.0430.0 | 0.0E+00 | 0.0€+00            | 0°.0E+00             | 0. nE+00 | 9. ñE - 09 | 9. nE -09    | 1.86-09    | 1.86-08    | 6.55-11          | 6.5E-11   | 1.ņE-03   | 1. UE -03  | 1.01-03   |
| RANGE<br>Factor  |   | ł       | ;          | ł        | ł        | 1         | 1       | ł       | 1       | ł       | ;       | ;       | ļ       | ł         | ł          | ;        | ł        | ;       | ;                  | {                    | ;        | ;          | ;            | ;          | ;          | ;                | ;         | ł         | ł          | ;         |
| FUDA<br>Freq     |   | N/A     | N/A        | N/A      | N/A      | N/A       | 0.0E+00 | N/A     | 0.0E+00 | N/A     | N/A     | N/A     | N/A     | N/A       | 0° ÚE + 0Ú | N/A      | N/A      | N/A     | N/A                | N/A                  | N/A      | N/A        | N/A          | N/A        | N/A        | N/A              | N/A       | N/A       | R/A        | N/A       |
| RANGE<br>Factor  |   | 1       | 1          | 1        | 1        | !         | ł       | 1       | 1       | ł       | 1       | :       | ;       | ;         | 1          | 1        | ;        | ł       | ;                  | ł                    | 1        | 2.6E+01    | 2.6E+01      | 2.6E+01    | 2.6E +11   | 2.6E+/1          | 2.6E+01   | ;         | 1.0E+01    | 1.05+01   |
| PRA<br>FRED      |   | N/A     | N/A        | N/A      | N/A      | N/A       | N/A     | N/A     | A/A     | N/A     | 0.0E+00 | N/A     | 0.0E+00 | N/A       | N/A        | N/A      | N/A      | N/A     | 0.0E+00            | 00+30.0              | N/A      | 9.0E-09    | 9.0E-09      | 1.85-08    | 1.85-08    | 1.35-10          | 1.35-10   | N/A       | 1.05-03    | 1.05-03   |
| RANGE<br>Factor  |   | 1       | !          | !        | ł        | ;         | ;       | ł       | 1       | 1       | ł       | ;       | ł       | 1         | ;          | ;        | ł        | ł       | ł                  | ;                    | 1        | :          | ł            | ł          | ;          | ;                | ł         | 1         | ł          | 1         |
| NAAP<br>Fred     |   | N/A     | N/A        | N/A      | N/A      | N/A       | N/A     | N/A     | N/A     | N/A     | N/A     | 0.0E+00 | N/A     | N/A       | N/A        | N/A      | N/A      | N/A     | N/A                | N/A                  | N/A      | N/A        | N/A          | N/A        | N/A        | N/A              | N/A       | N/A       | N/A        | A/A       |
| RANGE<br>Factor  |   | ł       | 1          | 1        | ;        | ł         | ł       | :       | ł       | !       | ł       | ł       | ł       | 1         | ł          | ;        | 1        | ;       | ;                  | ł                    | 1        | 2.6E+01    | 2.6E+01      | 2.6E+01    | 2.6E+01    | 2 <b>.6E</b> +01 | 2.6E+01   | 1         | 1.05+01    | 1.6E+01   |
| LBAD<br>Fred     |   | N/N     | N/A        | N/A      | N/A      | N/A       | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A       | 0.0E+00    | 0.0E+00  | 0.0E+00  | N/A     | 0.01430.0          | 00+30°0              | N/A      | 9.0E-03    | 9. nE - 09   | 1.86-08    | 1.85-08    | 1.36-10          | 1.3E-10   | N/A       | 1.0E-03    | 1.05-03   |
| RANGE<br>Factor  |   | ;       | 1          | ;        | ł        | 1         | 1       | ł       | ;       | ł       | ;       | !       | 1       | ł         | 1          | ;        | ;        | 1       | 1                  | 1                    | ł        | 1          | ;            | ;          | ļ          | 4<br>2           | ;         | ;         | ;          | ;         |
| APG<br>Fred      |   | N/A     | N/A        | N/A      | N/A      | N/A       | N/A     | N/A     | N/A     | N/A     | 0.05+00 | N/A     | N/A     | N/A       | N/A        | N/A      | N/A      | N/A     | N/A                | N/A                  | N/A      | N/A        | N/A          | N/A        | N/A        | N/A              | N/A       | N/A       | N/A        | N.A       |
| RANGE<br>Factor  |   | ł       | :          | ł        | :        | ;         | ł       | ł       | ;       | ;       | 1       | 1       | ;       | :         | :          | :        | ł        | ł       | :                  | 1                    | 1        | 2.6E+01    | 2.6E+01      | 2.6E+01    | 2. 6E +1-1 | 2.6E+01          | 2.66401   | :         | 1. (iE +01 | 1.^E+01   |
| ANAD<br>FREQ     |   | N/N     | 0°*9E+00   | 0°.0E+00 | N/A      | N/A       | 00+30-0 | 0.0E+00 | 0.05+00 | N/A     | 0.0E+00 | 0°0E+00 | 0.0E+00 | 0.0E+00   | 0.0130.0   | 0.0130.0 | 0.0140.0 | N.'A    | 0° 0 <b>€ +</b> 00 | 0 <b>0 • ∃</b> 0 • 0 | N/A      | 9. ñE -03  | 9.05-09      | 1, 86 - 69 | 1.85-08    | 6.5E-11          | 6.56-11   | N/A       | 1. nF - 03 | 1. u£ -03 |
| P. NO.           | i | 97      | <b>3</b> 6 | 26       | 36       | 27        | 27      | 27      | 27      | 17      | 27      | 27      | 27      | 27        | 27         | 27       | 21       | 11      | 12                 | 17                   | 11       | 52         | 52           | ġ:         | 02         | 5                |           | 22        | 72         | 3         |
| SCEN- 0<br>AR10  |   | HUUVU   | HCRGC      | HCKVC    | HCSVF    | HCEGF     | HCDHC   | HCCGC   | HCCHC   | HENGF   | HCFHF   | HCEVE   | HEMVE   | HCF GC    | HCFHC      | HCF VC   | HCBGC    | HCOVC   | HCFGC              | HCKVC                | HCSVF    | HCRGC      | HCRVC        | HCKGF      | HURVE      | HCKSC            | )%.sJH    | ACF65     | 11C FGS    | HCRVS     |

- 27422









File: HFRE6.Whl Fage 1 Date 20-Aug-87

CARACTER STORE

TABLE 6-4 (Continued)

REGIDNAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Handling Operations (HF) (Events per Fallet or Container)

|         | SCENAR10 | ANAD              | RANGE    | AF6   | RANGE | LRAD | RANGE  | NAAP  | RANGE  | PBA<br>Ford | RANGE | FUDA | KANGE | TEAD     | RANGE      | UNDA | RANGE   |
|---------|----------|-------------------|----------|-------|-------|------|--------|-------|--------|-------------|-------|------|-------|----------|------------|------|---------|
|         | NUNBEK   | - HER             |          | + KEV |       | THE  | raLIUK | THE W | LHLIUK |             |       |      |       | A HEA    |            |      | LHC LOK |
|         |          |                   |          |       |       |      |        |       |        |             |       |      |       |          |            |      |         |
| HF PGS  | -        | 0.0E+00           | ;        | N/A   | ;     | N/A  | í      | N/A   | ;      | N/A         | 1     | N/A  | 1     | 4.25-09  | 1.3E+01    | N/A  | :       |
| HF DHS  |          | 4, 4E - 09        | 1.3E+01  | N/A   | ;     | N/A  | ſ      | N/A   | 1      | N/A         | 1     | N/A  | ;     | 4.2E-09  | 1.36+01    | N/A  | ;       |
| HFC65   |          | 4.86-10           | 1. 3E+01 | N/A   | ł     | N/A  | :      | N/A   | ;      | N/A         | !     | A/N  | ł     | 4.8E-10  | 01.3E+01   | N/A  | 1       |
| HFCHS   | -        | 4.BE-LÚ           | 1.3E+01  | N/A   | :     | N/A  | ;      | N/A   | ;      | A/A         | ł     | N/A  | ł     | 4.8E-10  | 0.1.35+01  | N/A  | 1       |
| HFAGS   | -        | 0.0E+00           | :        | N/A   | :     | N/A  | ł      | N/A   | ļ      | N/A         | :     | N/A  | ł     | 8.65-09  | 1.36+01    | N/A  | 1       |
| HEAHS   | -        | 8, 8E- <i>0</i> 9 | 1.3E+01  | N/A   | ;     | N/A  | 1      | N/A   | ł      | N/A         | 1     | A/N  | ł     | 8.65-09  | 10+31.1    | N/A  | 1       |
| HFKVS   | -        | 8.85-09           | 1.3E+01  | N/A   | 1     | N/A  | 1      | N/A   | ł      | N/A         | 1     | N/A  | 1     | 8.65-09  | ) 1.3E+01  | N/Å  | 1       |
| HFMVS   | -        | 5.86-09           | 1.35+01  | N/A   | ;     | N/A  | ;      | A/N   | ł      | N/A         | !     | N/A  | ł     | 5.86-09  | 1.36+01    | N/Å  | 1       |
| HFF65   |          | 7.26-10           | 1.36+01  | N/A   | :     | N/A  | 1      | N/A   | :      | N/A         | :     | N/A  | ł     | 7.26-10  | 10+35-11   | N/A  | !       |
| SHJJH   |          | 7.2E-10           | 1.3E+01  | N/A   | ł     | N/A  | ;      | N/A   | 1      | N/A         | ;     | N/A  | ł     | 7.2E-10  | 0+36-11    | N/A  | ;       |
| SV73H   | 1        | 7.25-10           | 1.3E+01  | N/A   | 1     | N/A  | 1      | N/A   | 1      | N/A         | 1     | N/A  | !     | 7.2E-10  | 0.1.35+01  | N/A  | :       |
| HFQES   | -        | 7.2E-10           | 1.3E+01  | N/A   | ;     | N/A  | ł      | N/A   | ;      | N/A         | ;     | N/A  | :     | 7.2E-10  | 10+31.1    | N/A  | ;       |
| HFEVS   | -        | 0.0E+00           | ł        | N/A   | ;     | A/N  | ;      | N/A   | ł      | N/A         | 1     | N/A  | ł     | 7.2E-10  | 1.36+01    | N/A  | ł       |
| HF RES  | -        | 3. 2E -09         | 1.3E+01  | N/A   | ;     | N/A  | ł      | N/A   | ł      | N/A         | 1     | N/A  | ł     | 3. 2E-05 | 10+35-1    | N/A  | 1       |
| HFRVS   | -        | 3.26-09           | 10+35.1  | N/A   | 1     | N/A  | :      | N/A   | ;      | N/A         | :     | N/A  | :     | 3.2E-09  | 10+35-11   | N/A  | ;       |
| HF SVS  | -        | 0.0E+00           | :        | N/A   | ;     | N/A  | 1      | N/A   | 1      | N/A         | 1     | N/A  | :     | 1.06-06  | 5 1. JE+01 | A/A  | :       |
| HF BGC  | 2        | 0.0E+00           | 1        | N/A   | ;     | N/A  | {      | N/A   | ł      | N/A         | ;     | N/A  | 1     | 1.0E-16  | 5 3. JE+01 | N/A  | ł       |
| HEDHC   | 2        | 0.0E+00           |          | N/A   | ł     | N/A  | {      | N/A   | 1      | N/A         | ;     | N/A  | 1     | 0°0E+00  | -          | A/A  | ;       |
| HFCGC   | 2        | 0.0E+00           |          | N/A   | ;     | N/A  | 1      | N/A   | ł      | N/A         | 1     | N/A  | ł     | 0.0E+00  |            | N/A  | ;       |
| NFCHC   | 2        | 0.0E+00           |          | N/A   | 1     | N/A  | 1      | A/A   | 1      | N/A         | ;     | N/A  | !     | 0.0E+00  |            | NA   | ;       |
| HFI GC  | 2        | 0.0E+00           | 1        | N/A   | :     | N/A  | ;      | N/A   | ;      | N/A         | ł     | N/A  | ł     | 8.46-16  | 5 3.16+01  | N/A  | :       |
| HF HC.  | 2        | 8.4E-16           | 3. 1E+01 | N/A   | ;     | N/A  | 1      | N/A   | 1      | N/A         | :     | A/A  | ł     | 9.4E-16  | 5 3. IE+01 | N/A  | ;       |
| HFAUC   | 2        | 8.46-16           | 3. 1E+01 | N/A   | ;     | N/A  | ł      | N/A   | ł      | N/A         | ;     | N/A  | 1     | 8.4E-16  | 5.1E+01    | N/A  | ;       |
| HFMVC   | 2        | 2.6E-17           | 3. IE+01 | N/A   | ţ     | N/A  | ;      | N/A   | ;      | N/A         | ł     | N/A  | 1     | 2.6E-17  | 7 3. IE+01 | N/A  | [       |
| HFFEC   | C4       | 0,0E+00           |          | N/A   | ;     | N/A  | ł      | N/A   | ł      | N/A         | 1     | N/A  | ł     | 0.014    | _          | N/A  | ł       |
| HFFHC   | 2        | 0.0E+00           |          | N/A   | {     | N/A  | {      | N/A   | ł      | N/A         | 1     | N/A  | 1     | 0+30.0   | _          | N. A | ;       |
| HFFVC   | 2        | 0°.0E+UÙ          |          | N/A   | :     | A/A  | ł      | N/A   | 1      | N/A         | ;     | N/A  | :     | 0,01400  | -          | N.A  | :       |
| HF QG5  | 2        | 0.0E+(i)          |          | N/A   | ł     | N/A  | ł      | A/A   | 1      | N/A         | ł     | N/A  | :     | 0.0E+00  | ~          | N.A  | 1       |
| 5.Vg HF | 2        | 0, ((E+())        | :        | N/A   | ;     | N/A  | ł      | N/A   | ;      | N/A         | !     | N/A  | ;     | 0.0E+00  | •          | N/A  | 1       |

Destruct

~

6-45

ፚኯፚኯፚኯፚኯዸኯዸኯፚኯፚኯፚኯዀዀዀዀዀዀዀዀዀዀዀዀዀዀ
File: HFRE6.MM1 Fage 2 Date 20-Aug-87

 $\mathcal{A}$ 

Ŷ

TABLE 6-4 (Continued)

## REGIONAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Handling Operations (HF) (Events per Fallet or Container)

|          | SCENARIU<br>NUMBER | ANAD<br>FRED     | RANGE<br>FACTOR | AF6<br>FREQ | KANGL<br>Factor | LEAD<br>FRED | KANGE<br>FACTOR | NAAF<br>Fred | KANGE<br>Factor | FREQ | KANGE<br>FACTOR | FUUA<br>FREQ | KANDC<br>FACTOR | ILAU<br>FREG | KANDE<br>FACIUR | UNUH<br>FREQ | FACT OF |
|----------|--------------------|------------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|------|-----------------|--------------|-----------------|--------------|-----------------|--------------|---------|
|          |                    |                  |                 |             |                 |              |                 |              |                 |      |                 |              |                 |              |                 |              |         |
| HFREC    | 2                  | 1.86-16          | 3, 1E+01        | N/A         | ł               | N/A          | ł               | N/A          | 1               | N/A  | :               | N/A          | 1               | 1.86-16      | 3. IE+01        | N/A          | ł       |
| HFRVC    | 7                  | 1.86-16          | 3. IE+01        | N/A         | ;               | N/A          | ;               | N/A          | ł               | N/A  | :               | N/A          | :               | 1.8E-16      | 3. IE+01        | N/A          | :       |
| HESVC    | 2                  | 0.0E+00          | :               | N/A         | ł               | N/A          | ł               | N/A          | 1               | N/A  | ł               | N/A          | 1               | 4.56-15      | 3. l£+0l        | N/A          | ł       |
| 4F 86F   |                    | 0.0E+00          | :               | N/A         | ł               | N/A          | ł               | N/A          | 1               | N/A  | ł               | N/A          | ł               | 3. IE-11     | 3. IE+01        | N/A          | :       |
| 4F DHF   | ς.                 | 9.46-11          | 3. IE+01        | N/A         | !               | N/A          | 1               | N/A          | :               | N/A  | 1               | NIA          | :               | 9.4E-11      | 3. IE+01        | N/A          | ;       |
| HF L GF  | 2                  | 0.0E+00          |                 | N/A         | ;               | N/A          | ;               | N/N          | ł               | N/A  | ;               | N/A          | ;               | 0.0E+00      |                 | N/A          | 1       |
| #C.HF    | P~1                | 0.0E+00          |                 | N/A         | ł               | N/A          | ;               | N/A          | 1               | N/A  | ;               | N/A          | :               | 0.05+00      |                 | N/A          | ;       |
| HFIL GF  | ~                  | 0.05+00          | ļ               | N/A         | ł               | N/A          | ;               | N/A          | :               | N/A  | 1               | N/A          | ;               | 1.06-10      | 3. IE+01        | N/A          | 1       |
| 4FA.HF   | r                  | 1:05-10          | 3.16+01         | A/A         | ;               | N/A          | ;               | N/A          | ;               | A/A  | ;               | A/A          | ł               | 1,05-10      | 3.16+01         | N/A          | ;       |
| HERVE    | r                  | 1.0E-10          | 3, IE+01        | N/A         | :               | N/A          | ;               | N/A          | :               | A/N  | ł               | N/A          | ł               | 1.06-10      | 3. IE+01        | N/A          | ;       |
| HENVE    | **                 | 1.4E-10          | 3. IE+01        | N/A         | :               | N/A          | !               | N/A          | ł               | N/A  | 1               | N/A          | ;               | 1.45-10      | 3. 1E+01        | N/N          | 1       |
| 45 F.GF  | 2                  | 0.0E+00          |                 | N/N         | :               | N/A          | :               | N/A          | :               | N/A  | :               | N/A          | 1               | 0°.0E+00     |                 | A/A          | ;       |
| 4F P HF  | r                  | 0.05+00          |                 | N/A         | !               | N/A          | ł               | N/A          | :               | N/A  | ;               | N/A          | 1               | 0,05+00      |                 | A/A          | ł       |
| 46 P.VF  | m                  | 0.0E+00          |                 | N/A         | ;               | N/A          | ;               | N/A          | ;               | A/N  | :               | N/A          | ;               | 0°.0E+00     |                 | N/A          | ;       |
| 4F 06F   | 2                  | 0.0E+00          |                 | A/A         | ;               | N/A          | ;               | N/A          | 1               | N/A  | ţ               | N/A          | ł               | 0.0E+00      |                 | N/A          | ;       |
| HFBVF    | 3                  | 0, 0E+00         | ;               | N/A         | ł               | N/A          | ;               | A/A          | !               | N/A  | ł               | N/A          | 1               | 0.0E+00      |                 | N/A          | ;       |
| HF R.GF  | m                  | 8.15-11          | 3. IE+01        | N/A         | ł               | N/A          | ł               | N/A          | :               | N/A  | ;               | N/A          | ;               | 8. IE-11     | 3.1E+01         | N/A          | ;       |
| 4F RVF   | •••                | 8. IE-11         | 3, IE+01        | N/A         | ł               | N/A          | :               | 4/N          | ;               | N/A  | !               | N/A          | ;               | 8. IE-11     | 3. IE+01        | N/A          | ;       |
| HF SVF   | 1                  | 0.0E+00          | 1               | N/A         | ł               | N/A          | ;               | N/A          | 1               | N/A  | ł               | N/Ĥ          | ;               | 2.0E-09      | 3. IE+0I        | N/A          | ;       |
| 4F SVS   | -                  | 0.05+00          | ;               | N/A         | ł               | N/A          | ;               | N/A          | ;               | N/A  | 1               | N/A          | ł               | 3. IE-07     | 1.3E+01         | N/A          | ;       |
| 46 EIGF  | ŝ                  | 0.0E+00          | :               | N/À         | ł               | N/A          | 1               | N/A          | 1               | N/A  | 1               | N/A          | !               | 00+30°0      |                 | N/A          | ;       |
| 4F DHC   | רע                 | 0.0E+00          |                 | N/A         | ł               | N/A          | ł               | N/A          | !               | N/A  | :               | A.N          | :               | 0.0E+00      |                 | N/A          | ;       |
| HF C.E.C | 5                  | 0.0E+00          |                 | N/A         | ł               | N/A          | ł               | N/A          | ł               | N/A  | ł               | N/H          | ł               | 0°.0E+00     |                 | N/A          | ;       |
| HE CHC   | 5                  | 0.0E+00          |                 | N/A         | ł               | N/A          | 1               | N/A          | ł               | N/A  | ;               | N/A          | ;               | 0.0E+00      |                 | N.A          | ;       |
| IF).GF   | ŝ                  | 0.0E+00          | ;               | N/A         | ;               | N/A          | ł               | N/A          | i               | N/A  | i               | N/A          | !               | 0.0E+00      |                 | N/A          | ;       |
| HEALF    | ŝ                  | 00+30 <b>*</b> 0 |                 | N/A         | ł               | N/A          | 1               | N/A          | 1               | A/A  | 1               | N/A          | :               | 0.0E+00      |                 | N/A          | ;       |
| 4Fb VF   | 5                  | 0.0E+00          |                 | N/A         | ;               | N/A          | 1               | N/A          | 1               | N/A  | 1               | N/A          | 1               | 0°.0E+00     |                 | N/A          | 1       |
| HF MUC   | ŝ                  | 0, 0E+00         |                 | N/A         | ł               | N/A          | ;               | N/A          | 1               | A/A  | 1               | N/A          | ;               | 0.0E+00      |                 | N/A          | ;       |
| HFP6C    | ŝ                  | 0.0E+00          |                 | N/A         | ;               | N/A          | ł               | A/A          | :               | A/A  | 1               | N/A          | ł               | 0.0E+00      |                 | N/A          | ;       |

BEERED CONTRACTOR BEERE

PSSSSSS \_

V. V. V. V. A. A. A.



and an an



File: HFKE6.Wk1 Fage 3 Date 20-Aug-87

TABLE 6-4 (Continued)

REGIONAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Mandling Operations (HF) (Events per Pallet or Container)

|        | SCENAR10 | ANAD        | <b>RANGE</b> | AP6  | KANGE  | LBAD | RANGE  | NAAP | RANGE  | FEA  | RANGE  | PUDA | KANGE  | TEAD       | RANGE    | UNDA | KANGE  |
|--------|----------|-------------|--------------|------|--------|------|--------|------|--------|------|--------|------|--------|------------|----------|------|--------|
|        | NUMBER   | FREQ        | FAC10R       | FRED | FACTOR | FREQ | FACTOR | FRED | FACIOR | FRED | FACTOR | FREQ | FACTOR | FREQ       | FACTOR   | FREQ | FACTOR |
|        |          |             |              |      |        |      |        |      |        |      |        |      |        |            |          |      |        |
| HFPHC  | ניע      | 0.0E+00     |              | N/A  | ł      | N/A  | !      | N/A  | ;      | N/A  | ł      | N/A  | ł      | 0.0E+00    | _        | N/A  | !      |
| HFPVC  | ŝ        | 0.0E+00     |              | N/A  | ł      | N/A  | ;      | N/A  | Í      | N/A  | 1      | N/A  | ł      | 0.01400    |          | N/A  | 1      |
| HFQGC  | 5        | 0.0E+00     |              | N/A  | ;      | N/A  | ;      | N/A  | ł      | N/A  | ;      | N/A  | ;      | 0.0E+00    |          | N/A  | ;      |
| HFQVC  | 5        | 0.0E+00     | 1            | N/A  | ł      | N/A  | ;      | N/A  | ł      | N/A  | :      | N/A  | :      | 0.0E+00    | _        | N/A  | :      |
| HFRGC  | S        | 0.0E+00     |              | N/A  | 1      | N/A  | :      | N/A  | ł      | N/A  | ;      | N/A  | :      | 0.0E+00    |          | N/A  | 1      |
| HFRVC  | ŝ        | 0.0E+00     |              | N/A  | :      | N/A  | ł      | N/A  | ł      | N/A  | ł      | N/A  | ł      | 0.0E+00    |          | N/A  | :      |
| HESVE  | ŝ        | 0.0E+00     | :            | N/A  | :      | N/A  | ;      | N/A  | ;      | N/A  | :      | N/A  | 1      | 0.0E+00    |          | N/A  | ł      |
| HF BGS | 1        | 0.0E+00     | ;            | N/A  | ;      | N/A  | ;      | N/A  | ;      | N/A  | !      | N/A  | :      | 8. 3E-10   | 1.3E+01  | N/A  | ł      |
| HEDHS  | 2        | 2.5E-09     | 1.36+01      | N/A  | 1      | 8/N  | ;      | A/N  | :      | N/A  | :      | N/A  | ļ      | 2.5E-09    | 1.3E+01  | N/A  | ł      |
| HFC65  | 2        | 0.0E+00     |              | N/A  | ł      | N/A  | :      | N/A  | ł      | N/A  | 1      | N/A  | ;      | 0.0£+00    |          | N/A  | ł      |
| HFCHS  | 7        | 0,0E+00     |              | N/A  | 1      | N/A  | :      | N/N  | ł      | N/A  | ł      | N/A  | ł      | 0.0E+00    | _        | N/A  | :      |
| HFLES  | 7        | 0.0E+00     | ł            | N/A  | ;      | N/A  | ;      | N/A  | ł      | N/A  | :      | N/A  | 1      | 2.7E-09    | 1.3E+01  | N/A  | :      |
| HFAHS  | 7        | 2.7E-09     | 1.36+01      | N/A  | ;      | N/A  | 1      | N/A  | ł      | N/A  | ł      | N/A  | ł      | 2.7E-09    | 1.3E+01  | N/A  | :      |
| SVIJH  | 1        | 2.7E-09     | 1.3E+01      | N/A  | 1      | N/A  | ł      | N/A  | ;      | N/A  | ł      | N/A  | !      | 2.7E-09    | 1.3E+01  | N/A  | 1      |
| HFAVS  | 1        | 3.6E-09     | 1.3E+01      | N/A  | 1      | N/A  | ł      | N/A  | ;      | N/A  | 1      | N/A  | 1      | 3.66-09    | 1.36+01  | N/A  | 1      |
| HFF65  | 7        | 0.0E+00     |              | N/A  | 1      | N/A  | ł      | N/A  | :      | N/A  | ł      | N/A  | ł      | 0.0E+00    |          | N/A  | ;      |
| HFFHS  | 1        | 0, ((E+()() |              | N/A  | :      | N/A  | ł      | N/A  | 1      | N/A  | ł      | N/A  | ł      | 0.0E+00    | _        | N/A  | :      |
| HFFVS  | 1        | 0.0E+00     |              | N/A  | 1      | N/A  | :      | N/A  | ł      | N/A  | :      | N/A  | ł      | 0.0E+00    |          | N/A  | :      |
| HFD65  | 1        | 0,0E+00     |              | N/A  | ţ      | N/A  | ;      | N/A  | :      | N/A  | 1      | N/A  | 1      | 0.0E+00    |          | N/A  | ;      |
| HFBVS  | 1        | 0.0E+00     | 1            | N/A  | 1      | N/A  | 1      | N/A  | 1      | N/A  | ;      | N/A  | ł      | 0.0E+00    |          | N/A  | :      |
| HFRGS  | 7        | 2.2E-09     | 1.3E+01      | N/A  | ł      | N/A  | :      | N/A  | ;      | N/A  | ł      | N/A  | 1      | 2.2E-09    | 1.3E+01  | N/A  | ;      |
| HFRVS  | 2        | 2.26-09     | I.3E+01      | N/A  | :      | N/A  | !      | N/A  | ł      | N/A  | ł      | N/A  | ł      | 2.2E-09    | 1. 3E+01 | N/A  | ;      |
| HFSVS  | 1        | 0.0E+00     | ł            | N'A  | 8      | N/A  | 1      | N/A  | !      | N/A  | :      | N/A  | ;      | 5.2E-08    | 1.35+01  | N/N  | ;      |
| HFREC  | 80       | 0.0E+00     | ;            | N/A  | ł      | N/A  | ;      | N/A  | ;      | N/A  | !      | N/A  | 1      | 5. JE-18   | 3. IE+01 | N/A  | ;      |
| HFDHC  | 8        | 5.3E-18     | 3. IE+01     | N/A  | :      | N/A  | !      | N/A  | 1      | N/A  | !      | N/A  | ;      | 5.3E-18    | 3. IE+01 | N/A  | !      |
| HFCGC  | 80       | 6.0E-19     | 3.16+01      | N/A  | 1      | N/A  | 1      | N/A  | 1      | A/A  | ;      | N/A  | 1      | 6.0E-19    | 3. IE+01 | N/A  | ;      |
| HFCHC  | 80       | 61-30.0     | 3. IE+01     | N/A  | 1      | N/A  | 1      | N/A  | 1      | N/A  | ł      | N/A  | ł      | 6. ûE - 19 | 3. 1E+01 | N/A  | !      |
| HF> GC | 80       | 0, 0E+00    | ;            | N/A  | ;      | N/A  | 1      | N/A  | 1      | N/A  | :      | N/A  | ł      | 1.16-17    | 3. IE+01 | N/A  | ;      |
| HEAHC  | œ        | 1.16-17     | 3.1E+01      | N/A  | ;      | N/A  | 1      | N/A  | :      | N/A  | 1      | N/A  | :      | 1.16-17    | 3.15+01  | N/A  | :      |

RECORD DEPENDENCE DEPENDENCE

SA G

File: HFKEG.Mk1 Page 4 Date 20-Aug-87

**14046474 (**4.040)

TABLE 6-4 (Continued)

## REBIONAL COLLOCATION OFTION - FACILITY HANDLING

Accident Frequencies for Facility Mandling Operations (MF) (Events per Pallet or Container)

|        | SCENARIO<br>NIMBER | ANAD     | RANGE<br>Factor | AF6<br>Fred | RANSE<br>FALTOR | L RAD<br>F RF D | RANGE<br>Factor | NAAP<br>Fred | RANGE<br>Factor | P.B.A<br>FRFQ | RANGE<br>Factor | PUDA | RANGE<br>Factor | 1EAD<br>Fred | RANGE<br>Factor | UNDA<br>Fred | RANGE<br>Factor |
|--------|--------------------|----------|-----------------|-------------|-----------------|-----------------|-----------------|--------------|-----------------|---------------|-----------------|------|-----------------|--------------|-----------------|--------------|-----------------|
|        |                    |          |                 |             |                 |                 |                 |              |                 |               |                 |      |                 |              |                 |              |                 |
|        |                    |          |                 |             |                 |                 |                 |              |                 |               |                 |      |                 |              |                 |              |                 |
| HFKVC  | 6                  | 1.15-17  | 3. IE+01        | N/A         | ;               | N/A             | ł               | N/A          | 1               | N/A           | ;               | N/A  | ł               | 1. IE-17     | 3.16+01         | N/A          | ł               |
| HFMVC  | 8                  | 7. 2E-18 | 3. IE+01        | N/A         | :               | A/N             | ;               | N/A          | ļ               | N/A           | ;               | N/A  | 1               | 7.2E-18      | 3. IE+01        | N/A          | !               |
| HFPGC  | 6                  | 9.0E-19  | 3. IE+01        | N/A         | ł               | N/A             | ;               | N/A          | 1               | N/A           | ;               | N/A  | ł               | 9.0E-19      | 3. IE+01        | N/A          | :               |
| HEPHC  | 60                 | 9.0E-19  | 3. IE+01        | N/A         | !               | N/A             | ţ               | N/A          | ;               | A/A           | ;               | N/A  | ł               | 9.0E-19      | 3. IE+01        | N/A          | :               |
| HFFVC  | 80                 | 9.0E-19  | 3. IE+01        | N/A         | ;               | A/A             | ;               | N/A          | ł               | N/A           | 1               | N/A  | ł               | 9.0E-19      | 3.1E+01         | N/A          | :               |
| HFBGC  | 80                 | 9.06-19  | 3. IE+01        | N/A         | ł               | N/A             | ;               | N/A          | ł               | N/A           | ;               | N/A  | :               | 9.0E-19      | 3. IE+01        | N/A          | ;               |
| HFBVC  | 80                 | 0.0E+00  | ł               | N/A         | ł               | N/A             | ;               | N/A          | ;               | N/A           | ł               | N/A  | ;               | 9.0E-19      | 3. IE+01        | N/A          | ł               |
| HFRGC  | 8                  | 4.16-18  | 3.16+01         | N/A         | ł               | N/A             | ;               | N/A          | ;               | N/A           | ;               | N/A  | ;               | 4. IE-18     | 3.1E+01         | N/A          | 1               |
| HFRVC  | 80                 | 4. JE-18 | 3. IE+01        | N/A         | ;               | N/A             | ;               | A/N          | :               | N/N           | ł               | N/A  | ł               | 4.1E-18      | 3. IE+01        | N/A.         | !               |
| HF SVC | 80                 | 0.0E+00  | ;               | N/A         | ;               | A/A             | ł               | N/A          | ł               | N/A           | ;               | N/A  | ;               | 1.2E-15      | 3. IE+01        | N/A          | 1               |
| HESVC  | 6                  | 0.0E+00  | ;               | N/A         | ł               | N/A             | !               | N/N          | ;               | N/A           | ;               | N/A  | ł               | 7.7E-16      | 3.16+01         | N/A          | ;               |
| HF BGC | 10                 | 0.0E+00  | :               | N/A         | ł               | N/A             | ;               | N/A          | ;               | N/A           | ł               | N/A  | ;               | 4.3E-19      | 3.1E+01         | N/A          | :               |
| HFDHC  | 01                 | 1. 3E-18 | 3. IE+01        | R/A         | ;               | N/A             | ł               | N/A          | !               | N/A           | ţ               | A/A  | 1               | 1.36-18      | 3.1E+01         | N/A          | ł               |
| HFCGC  | 10                 | 0.05+00  |                 | A/A         | ;               | N/A             | ł               | N/A          | 1               | N/A           | ;               | N/A  | ł               | 0.0E+00      |                 | N/A          | :               |
| HECHC  | 10                 | 0.0E+00  |                 | N/A         | ;               | N/A             | ł               | N/N          | ;               | N/A           | ;               | N/N  | ł               | 0.0E+00      |                 | N/A          | ;               |
| HFLEC  | 10                 | 0.0E+00  | ;               | N/A         | ;               | N/A             | ł               | N/A          | ;               | N/A           | ļ               | N/A  | ;               | 1.4E-18      | 3.16+01         | N/A          | :               |
| HFKHC  | 10                 | 1.4E-18  | 3.1E+01         | A/A         | ţ               | N/A             | 1               | N/A          | ;               | A/A           | ł               | N/A  | ;               | 1.46-18      | 3.1E+01         | N/A          | ł               |
| HFLVC  | 10                 | 1.4E-18  | 3. IE+01        | N/A         | 1               | N/A             | ł               | N/A          | :               | N/A           | ł               | N/A  | ł               | 1.46-18      | 3. IE+01        | N/A          | ł               |
| JUNT   | 10                 | 1.96-18  | 3.1E+01         | A/A         | ;               | N/A             | ł               | A/N          | 1               | N/A           | ł               | N/A  | ł               | 1.96-18      | 3. IE+01        | N/A          | :               |
| HFF6C  | 10                 | 0.0E+00  |                 | N/A         | ;               | N/A             | 1               | N/A          | 1               | N/A           | 1               | N/A  | 1               | 0.0E+00      |                 | N/A          | :               |
| HEPHC  | 10                 | 0.0+30.0 |                 | N/A         | }               | N/A             | ;               | N/A          | ;               | N/A           | 1               | N/A  | ł               | (). ()E+00   |                 | N/A          | ł               |
| HFPVC  | 10                 | 0, 0E+00 |                 | N/A         | ;               | N/A             | :               | N/A          | 1               | N/A           | ;               | N/A  | ;               | 0.0E+00      |                 | N/A          | ť               |
| HFGGC  | 10                 | 0.05+00  |                 | N/A         | ;               | N/A             | ł               | A/A          | ł               | N/A           | 1               | N/A  | ;               | 0.0130.0     |                 | N/A          | ;               |
| HFGVC  | 01<br>10           | 0,0E+00  | ł               | N/A         | ;               | N/A             | 1               | N/A          | ;               | N/A           | 1               | N/A  | ;               | 0.0E+00      |                 | N/A          | {               |
| HFRGC  | 10                 | 1.1E-18  | 3. IE+01        | N/A         | ;               | N/A             | ł               | N/A          | ł               | N/A           | :               | N/A  | ł               | 1. IE-18     | 3. IE+ÙI        | N/A          | 1               |
| HFRVC  | <b>1</b> 0         | 1.16-18  | 3. IE+01        | N/A         | }               | N/A             | ł               | N/A          | 1               | N/A           | ;               | N/A  | 1               | 1.1E-18      | 3. IE+ůl        | N/A          | ſ               |
| HESVC  | 10                 | 0.0E+00  | ł               | N/A         | 1               | N/A             | 1               | N/A          | ;               | N/A           | ł               | N/A  | ;               | 2.76-17      | 3.1E+01         | N/A          | 1               |
| HEDHC  | 11                 | 5.8E-11  | ;               | N/A         | ;               | N/A             | 1               | N/A          | :               | N/A           | 1               | N/A  | 1               | 5.86-11      | 2.66+01         | N/A          | ł               |
| HFCGC  | Ξ                  | 2.9E-11  | 2.6E+01         | N/A         | ;               | N/A             | ł               | N/A          | :               | N/A           | ;               | N/A  | ;               | 2.96-11      | 2.6E+01         | N/A          | ł               |



Ċ.

ኯፚ፝ኯ*፝ኯፚኯ*ቘ፝፝ኯቘኯቘኯፚኯፚኯፚኯቘኯቘኯቘኯጜኯጜኯጜኯፚኯፚኯፚኯፚኯ





a.''.a.'.a.'.a.'.'.a.J.'.a



TABLE 6-4 (Continued)

## REGIONAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Mandling Operations (MF) (Events per Pallet or Container)

|           | SCENARIO<br>NUMBER | ANAD<br>FREQ | RANGE<br>FACTOR | APG<br>FREQ | RANGE<br>FACTOR | LFAD<br>FKED | RANGE<br>FACTOR | NAAP<br>Fred | RANGE<br>Factor | PEA<br>Fred | RANGE<br>Factor | PUDA<br>Fred | KANGE<br>Factor | TEAD<br>Freq        | RANGE<br>Factor | UMDA<br>Freq | KANGE<br>FACTOR |
|-----------|--------------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|---------------------|-----------------|--------------|-----------------|
|           |                    |              |                 |             |                 |              |                 |              |                 |             |                 |              |                 |                     | :               |              |                 |
| HF CHC    | Ξ                  | 2.96-11      | 2.6E+01         | A/A         | ;               | N/A          | ł               | N/A          | ;               | A/A         | 1               | N/A          | :               | 2.96-11             | 2.6E+01         | A N          | ;               |
| HENC      | Ξ                  | 4. 35-11     | 2.6E+01         | N/A         | ł               | N/A          | ;               | N/A          | 1               | N/A         | !               | N/A          | 1               | 4. 36-11            | 2.6E+01         | A'A          | ;               |
| HF P.GC   | 11                 | 9.6E-12      | 2.6E+01         | N/A         | 1               | N/A          | ;               | N/A          | ł               | N/A         | ł               | N/A          | :               | 9.6E-12             | 2.6E+01         | N/A          | :               |
| HFFHC     | Ξ                  | 9.6E-12      | 2.65+01         | N/A         | ţ               | N/A          | ;               | A/N          | ł               | N/A         | 1               | N/A          | :               | 9.6E-12             | 2.6E+01         | N/A          | :               |
| HEPVC     | Π                  | 9.6E-12      | 2.6E+01         | N'A         | ;               | N/A          | 1               | N/A          | ;               | N/A         | :               | N/A          | ;               | 9.66-12             | 2.6E+01         | N/A          | :               |
| HFGGC     | 11                 | 7.26-12      | 2.6E+01         | N/A         | 4<br>1          | N/A          | :               | N/A          | ;               | N/A         | :               | N/A          | ;               | 7.26-12             | 2.6E+01         | N/A          | :               |
| HFQVC     | Ξ                  | 0, ()E +0()  | ;               | N/A         | ;               | N/A          | !               | N/A          | ;               | N/A         | ;               | N/A          | :               | 7.2E-12             | 2.6E+()1        | N/A          | :               |
| RFREC     | 11                 | 1.8E-11      | 2.6E+(i)        | \$/\$       | ;               | 8/8          | ;               | 4/4          | ;               | 4/4         | 1               | A/A          | :               | 1.96-11             | 2. èE +01       | H/N          | :               |
| HFRVC     | 11                 | 1.86-11      | 2.6E+01         | N/A         | ;               | N/A          | ł               | N/A          | ;               | N/A         | ;               | N/A          | ;               | 11-38.1             | 2.6E+01         | N/A          | :               |
| HFDHC     | 12                 | 3, 0E-10     | 2.6E+Ù1         | N/A         | ł               | N/A          | ;               | N/A          | ł               | N'A         | :               | N/A          | ;               | 3. 0E-10            | 2.66+01         | 4/H          | ;               |
| HFCGC     | 12                 | 3.0E-10      | 2.6E+01         | N/A         | ł               | N/A          | !               | A/N          | 1               | N/A         | 1               | N/A          | ł               | 3.0E-10             | 2.6E+01         | N/A          | ł               |
| HECHE     | 12                 | 3.0E-10      | 2.6E+01         | N/Å         | ;               | N/A          | :               | N/A          | ;               | N/A         | :               | N/A          | ;               | 3.0E-10             | 2.6E+01         | N/A          | :               |
| HFAVC     | 11                 | 3.0E-10      | 2.6E+01         | N/A         | 1               | N/A          | ;               | A/N          | !               | N/A         | ł               | N/A          | ł               | 3.06-10             | 2.6E+UI         | N/A          | :               |
| HFFGC     | 12                 | 3.0E-10      | 2.6E+01         | N/A         | ;               | N/A          | ;               | N/A          | :               | N/A         | ł               | A/A          | ;               | 3. oE - 10          | 2.5E+(i1        | ₹.N          | :               |
| JH-F HC   | 12                 | 3,05-10      | 2.6E+01         | N/A         | ;               | N/A          | !               | N/A          | ;               | N/A         | :               | N/A          | ;               | 3.06-10             | 2.4E+01         | N/A          | ;               |
| HFFVC     | 12                 | 3.0E-10      | 2.66+01         | A'N         | 1               | N/A          | 1               | N/A          | ;               | N/A         | ł               | A/A          | ;               | 3.0E-10             | 2.6E+01         | 47k          | :               |
| нг ըն(    | 12                 | 3.06-10      | 2.66+01         | A'N         | 1               | N/A          | ł               | N/A          | ;               | N/A         | :               | N/A          | ;               | 3.06-10             | 2.6E+01         | N/A          | ł               |
| J'VQ H    | 12                 | 0.0E+U0      | :               | NA          | :               | N/A          | F<br>I          | A / N        | 8<br>7          | N/A         | ;               | N/A          | ;               | 3,06-10             | 2.6E+01         | N / A        | ;               |
| HFREC     | 12                 | 3.0E-10      | 2.6E+01         | N/A         | 1               | N/A          | ;               | N/A          | ;               | N/A         | ł               | N/A          | :               | 3. (i <b>f</b> - 10 | 2.6E+i)         | N/Ĥ          | ł               |
| HFRVC     | 12                 | 3.0E-10      | 2.6E+U1         | A'N         | ;               | N/A          | :               | N/A          | ;               | N/A         | 1               | A/A          | ;               | 3.0E-10             | 2.5E+01         | N/G          | ;               |
| HF DHC    | 13                 | 7.26-11      | 2.6E+01         | N A         | ;               | A/N          | :               | N/A          | ;               | N/A         | :               | N/A          | :               | 7.2E-11             | 2.6E+01         | 4/H          | ;               |
| HFCEC     | 13                 | 3. 6E - 11   | 2.6E+v1         | N/A         | 1               | N/A          | ł               | N/A          | ;               | N/A         | :               | N:A          | ;               | J. 6E-11            | 2.6E+01         | N/A          | ł               |
| HECHC     | 1                  | 3.6E-11      | 2.6E+()}        | N/A         | i               | 4/N          | 1               | N/A          | :               | N/A         | ł               | N/A          | :               | 3.6E-11             | 2.66+01         | N/A          | ;               |
| Ditter an | <u> </u>           | 5,4E-11      | 2.6E+01         | N.A         | :               | N/A          | :               | N/A          | :               | N/A         | :               | N/A          | ;               | 5.46-11             | 2.6E+v1         | N, G         | ı               |
| HFFGC     | 11                 | 1.26-11      | 2.66+01         | N/A         | ;               | 4/N          | 1               | N/N          | ł               | N/A         | ł               | N/A          | 1               | 1.25-11             | 2.66+01         | N/A          | !               |
| HEFHC     | 13                 | 1.2E-11      | 2.6E+01         | N/A         | 1               | N/A          | ł               | N/A          | :               | N/A         | 1               | N/A          | ;               | I. 2E-11            | 2.6E+U1         | 878          | 8               |
| HFFVC     | :                  | 1.26-11      | 2.6E+01         | N/A         | ;               | N/A          | :               | N/A          | :               | N/A         | ł               | N/A          | :               | 1.26-11             | 2.6E+01         | N/A          | !               |
| HF 06C    | 1                  | 9, ()E - 12  | 2.6€+01         | A'N         | ł               | N/A          | :               | N/A          | ;               | N/A         | :               | N A          | :               | 9.uE-12             | 2.5E+01         | N/A          | ;               |
| HFQVC     | 13                 | 0°.0E+00     | ;               | 4/N         | ł               | N/A          | ;               | N/A          | :               | N/A         | :               | N/A          | ;               | 9.(iE-12            | 2.6E+01         | N/A          | :               |

33335

E STATES 

tree of

TABLE 6-4 (Continued) File: MFREG.MA1 Page 6 Date 20-Aug-87

REGIONAL COLLOCATION OPTION - FACILITY HANDLING

for Facility Mandling Operations (MF) (Events per Pallet or Container)

| Container)  |  |
|-------------|--|
| 2           |  |
| Pallet      |  |
| per         |  |
| lEvents     |  |
| Ē           |  |
| Dperations  |  |
| Handling    |  |
| Facility    |  |
| f or        |  |
| Frequencies |  |
| Accident    |  |
|             |  |

9

Ł

|         | SCENAR 10<br>NUMBER | ANAD<br>Freg | RANGE<br>FACTOR | AF6<br>Fked | RANGE<br>FACTOR | L BAD<br>F RED        | RANGE<br>Factor | NAAP<br>Fred | KANGE<br>FACTOR | PRA<br>Fred | RANGE<br>Factor | PUDA<br>Freq | KANGE<br>FACTOR | TEAD<br>Freq | RANGE<br>Factor | UMDA<br>FREQ | RANGE<br>Factor |
|---------|---------------------|--------------|-----------------|-------------|-----------------|-----------------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|-----------------|
|         |                     |              |                 |             |                 | 1<br>1<br>1<br>1<br>1 |                 |              |                 |             |                 |              |                 |              |                 |              |                 |
| IF R.C. |                     | 2.2E-11      | 2.6E+01         | A/A         | ł               | N/A                   | ;               | N/A          | ;               | M/A         | ł               | N/A          | ;               | 2.26-11      | 2.45+01         | N/A          | 1               |
| IF KVC  | : ==                | 2.26-11      | 2.66+01         | N/A         | 1               | N/A                   | ;               | N/A          | :               | N/A         | :               | N/A          | ł               | 2.2E-11      | 2.6E+01         | N/A          | ţ               |
| IF DHC  |                     | 1. 35-14     | 2.6E+01         | N/A         | 1               | N.'A                  | ł               | N/A          | ;               | N/A         | ł               | N/A          | ;               | 1.36-14      | 2.6E+01         | NÌA          | !               |
| IF CGC  | Ξ                   | 6.66-15      | 2.6E+01         | N'A         | ;               | N/A                   | ;               | N/A          | :               | N/A         | 1               | N/A          | ;               | 6.65-15      | 2.6E+01         | N/A          | ;               |
| IF CHC  | -                   | 6.6E-15      | 2.6E+01         | 4           | ł               | N/A                   | 1               | A/A          | 1               | M/A         | 1               | N/A          | ;               | 6.66-15      | 2.6E+01         | N/A          | 1               |
| 4F MVC  |                     | 9.96-15      | 2.6E+01         | A/A         | ;               | N A                   | ;               | N/A          | ł               | N/A         | ł               | N/A          | ;               | 9.96-15      | 2.6E+01         | N/A          | :               |
| IFP6C   | 11                  | 2.2E-15      | 2.66+01         | A/A         | ł               | A/A                   | ł               | A/A          | ł               | N/A         | !               | N/A          | :               | 2.25-15      | 2.6E+01         | 8/N          | ;               |
| JH 1    | -                   | 2.26-15      | 2.6E+01         | N/A         | ;               | N/A                   | 1               | N/A          | :               | A/N         | 1               | N/A          | ł               | 2.2E-15      | 2.6E+01         | N/A          | :               |
| IF PUC  | Ξ                   | 2.26-15      | 2.66+01         | N/A         | ;               | A/A                   | :               | A/A          | ł               | N/A         | ;               | N/A          | 1               | 2.2E-15      | 2.6E+01         | N/A          | !               |
| IF BGC  | *                   | 1.76-15      | 2.46+01         | N/A         | :               | M/A                   | :               | A/A          | 1               | N/A         | ł               | N/A          | :               | 1.7E-15      | 2.6E+01         | N/A          | ;               |
| #F PVC  | =                   | 0,0E+00      | ;               | N/A         | ;               | A/A                   | :               | A/A          | :               | N/A         | 1               | N/A          | 1               | 1.7E-15      | 2.6E+01         | N/A          | ;               |
| IF RGC  | 1                   | 4.16-15      | 2.66+01         | N/A         | :               | A/A                   | :               | N/A          | :               | N/A         | 1               | N/A          | 1               | 4, IE-15     | 2.66+01         | N/A          | ł               |
| IFRVC   |                     | 4. IE-15     | 2.6E+01         | N/A         | ;               | A/A                   | ;               | N/A          | 1               | N/A         | 1               | A/N          | ;               | 4.16-15      | 2.6E+01         | N/A          | :               |







file: HDLMAI.Wki, Page 1 Date 18-Aug-87

TABLE 6-4 (Continued)

# HANDLING ACCIDENTS - NATIONAL PROCESSING OPTION - PER PALLET OR CONTAINER

### Accident Frequencies and Range Factors

| SCEN-  | OF. NO.    | ANAD                       | RANGE      | AFG       | RANGE    | LPAD             | RANGE     | NAAP    | RANGE    | PBA     | RANGE    | FUDA           | RANGE   | TEAD     | RANGE       | UMDA       | RANGE     |
|--------|------------|----------------------------|------------|-----------|----------|------------------|-----------|---------|----------|---------|----------|----------------|---------|----------|-------------|------------|-----------|
| ARID   |            | FREQ                       | FACTOR     | FRED      | FACTOR   | FREQ             | FACTOR    | FREQ    | FACTOR   | FREQ    | FACTOR   | FREQ           | FACTOR  | FREQ     | FACTOR      | FREQ       | FAC TOR   |
|        | 1          | 8<br>1<br>8<br>8<br>1<br>1 |            |           |          |                  |           |         |          |         |          |                |         |          | *<br>*<br>1 |            |           |
| HCEGC  |            | N/A                        |            | N/A       | ;        | N/A              | ;         | N/A     | ;        | N/A     | !        | N/A            | 1       | 1.2E-07  | 1.3E+01     | 6.1E-08    | 1.3E+01   |
| JHCDHC | -          | 1.66-08                    | 1 . 3E +01 | N/A       | ;        | NZA              | :         | N/A     | ł        | N/A     | ł        | 1.6E-08        | 1.3E+01 | 3.2E-08  | 1.35+01     | N/A        | :         |
| MCCGC  | -          | 2.85 -(19                  | 1.36 + 01  | A/N       | ;        | 4/N              | :         | N/A     | ;        | N/A     | !        | N/A            | ł       | 5.7E-09  | 1.3E+01     | N/A        | ;         |
| HCCHC  | -          | 2.86-09                    | 1. 3E+01   | N/A       | :        | N/A              | 1         | N/A     | ;        | N/A     | ł        | 2.8E-09        | 1.3E+01 | 5.7E-09  | 1. 3E+01    | N/A        | :         |
| HCINEC | -          | N/A                        | ;          | N/A       | ;        | N/A              | ;         | N/A     | :        | N/A     | :        | N/A            | ;       | 4°.0E-08 | 1°.3E+01    | N/A        | :         |
| HCKHC  | -          | 2.05-08                    | 1. 3E+01   | 2. nE -08 | 1.3E+01  | N/A              | 1         | N/A     | ŀ        | 2.0E-08 | 1.3E+01  | N/A            | ł       | 4.0E-08  | 1. 3E+01    | 2.0E-08    | 1.35+01   |
| HCKVC  | -          | N/A                        | ;          | N/A       | :        | N/A              | ł         | 2.0E-08 | 1, 3E+01 | N/A     | {        | N/A            | ;       | 4.0E-08  | 1.3E+01     | N/A        | ;         |
| HENVE  | -          | 1.25-08                    | 1, 3E+01   | N, A      | ;        | N/A              | ;         | N/A     | ł        | 1.2E-08 | 1. 3E+01 | N/A            | :       | 2.4E-08  | 1.3E+01     | 1.2E-08    | 1. 36+01  |
| HCF6C  |            | 00+ 30° 11                 | t<br>1     | N/A       | ;        | N/A              | ;         | N'A     | ;        | N/A     | ;        | N/A            | ;       | 0°,0E+00 | ;           | 0.05+00    | 1         |
| Эндэн  | -          | 0.05+00                    | ;          | N/A       | ;        | 0°.0E+00         | ł         | N/A     | !        | N/A     | ;        | 0°*0E+00       | ł       | 0.0E+00  | ł           | N/A        | ;         |
| HEFVC  | -          | 0° 0E+00                   | :          | N/A       | ;        | 0 <b>.</b> (E+00 | ;         | N/A     | :        | N/A     | ł        | N/A            | ţ       | 0.0E+00  | ;           | 0.0E+00    | :         |
| HC06C  | -          | 0.05+00                    | ;          | N/A       | ł        | 00+30.0          | ;         | N/A     | 1        | NA      | 1        | N/A            | :       | 0.05+00  | :           | 0.0E+00    | ;         |
| HCOVC  | -          | N/A                        | !          | NA        | ;        | N/A              | :         | N/A     | ;        | N/A     | ;        | N/A            | ;       | 0.0E+00  | ł           | 0.0E+00    | :         |
| HCFGC  | ` <b>-</b> | 4.85-08                    | 1. 3E +01  | N/A       | ŗ        | 4.86-08          | 1. 3E+01  | N/A     | ;        | 4.85-08 | 1.36+01  | N/A            | ;       | 9.5E-08  | 1.3E+01     | 4.85-08    | 1.36+01   |
| HCRVC  | -          | 4.86-08                    | 1. 3E+01   | N/A       | ;        | 4.85-08          | 1.36+01   | N/A     | 1        | 4.8E-08 | 1.3E+01  | N/A            | ł       | 9.5E-08  | 1.3E+01     | 4.8E-08    | 1.35+01   |
| HCSVC  | -          | N/A                        | ţ          | N/A       | ;        | N/A              | ;         | N/A     | ł        | N/A     | ١        | N/A            | :       | 1.0E-06  | 1.35401     | 5.2E-07    | 1, 35+01  |
| HCHE   | ~          | N/A                        | :          | 5.2E-10   | 3. IE+01 | N/A              | ł         | N/A     | 1        | 5.2E-10 | 3. 16+01 | N/A            | ł       | 5.2E-10  | 3. IE+01    | N/A        | ;         |
| HCFIEC | •          | N/A                        | ;          | N/A       | ;        | N/A              | ł         | N/A     | :        | N/A     | ;        | N/A            | ;       | 2.6E-07  | 10+32.1     | 1.36-07    | 1, 3E+01  |
| HCDHC  | 2          | 3.7E-07                    | 1. 3E +01  | N/A       | :        | N/A              | 1         | A/A     | 1        | A/A     | ١        | 3.7E-07        | 1.3E+01 | 7.4E-07  | 1.35+01     | N/A        | ł         |
| 393JH  | ñ          | <b>8.9E</b> -0 <b>B</b>    | 1.3E+01    | N/A       | ł        | N/A              | ł         | N/A     | 1        | N/A     | ;        | N/A            | ł       | 1.86-07  | 1. 3E+01    | N/A        | ł         |
| HCCHC  | r          | 80-36-08                   | 1.3£+01    | N/A       | ł        | N/A              | ł         | N/A     | ł        | N/A     | 1        | <b>8.9E-08</b> | 1.3E+01 | 1.8E-07  | 1.35+01     | N/A        | ł         |
| HCRVC  | 5          | 7.16-07                    | 1.35+01    | N/A       | ł        | N/A              | ł         | N/A     | ł        | 7.IE-07 | 1.35+01  | N/A            | 1       | 1.4E-06  | 1.3E+01     | 7.1E-07    | 1, 3E+01  |
| HCFGC  | ~          | 5.05-08                    | 1.36+01    | N/A       | ł        | N/A              | ł         | N/A     | 1        | N/A     | ;        | N/A            | ;       | 1.05-07  | 1. 3E+01    | 5.05-08    | 1. 3E +01 |
| HCFHC  | r          | 5.06-08                    | 1.36+01    | N/A       | ł        | 5.0E-08          | 1. XE+01  | N/A     | ;        | N/A     | ١        | 5.0E-08        | 1.JE+01 | 1. 0E-07 | 1.36+01     | K/A        | ;         |
| HCFVC  | ~          | 5.0E-08                    | 10+31.1    | N/A       | ;        | 5.01-08          | 1. 3E +01 | N/A     | ;        | N/A     | ;        | N/A            | :       | 1.05-07  | 1.3E+01     | 5.0E-08    | 1. 3E+01  |
| 100C   | •*•        | 5.05-08                    | 10+35-1    | N/A       | ;        | 5.06-08          | 1.36+01   | N/A     | ;        | N/A     | 1        | N/A            | :       | 1.0E-07  | 1.3E+01     | 5.0E-08    | 1. 3E +01 |
| HCOVC  | ~          | N/A                        | ;          | N/A       | :        | N/A              | ł         | N/A     | ţ        | N/A     | ;        | N/A            | :       | 1.06-07  | 1. JE +01   | 5. QE - 08 | 1.35+01   |
| HCFEC  | ~          | 2.6E-06                    | 1.36+01    | N/A       | ;        | 2.6E-06          | 1.3E+01   | N/A     | ţ        | 2.6E-06 | 1.36+01  | N/A            | :       | 5. 36-06 | 1.35+01     | 2.65-06    | 1.35+01   |
| HCRVC  | ~          | 2.6E-06                    | 1.36+01    | N/A       | 1        | 2.66-06          | 1.3E+01   | N/A     | ł        | 2.6E-06 | 1.36+01  | N/A            | ;       | 5. 3E-06 | 1.3E+01     | 2.6E-06    | 1.3E+01   |

80001 - 3000001 - **200001 - 200001 - 2000** 

6-51

۰.

File: HDLMAT.Mk1, Page 2 Date 18-Aug-87

TABLE 6-4 (Continued)

# HANDLING ACCIDENTS - NATIONAL PROCESSING OPTION - PER PALLET OR CONTAINER

### Accident Frequencies and Range Factors

|                 | -       | Ξ        |           |           |           |          | Ξ         |             | Ξ        |          |         |         |          |         | -        | Ξ        | Ξ       |          |          |         |         |         |          |             |         |          |           |             |         |
|-----------------|---------|----------|-----------|-----------|-----------|----------|-----------|-------------|----------|----------|---------|---------|----------|---------|----------|----------|---------|----------|----------|---------|---------|---------|----------|-------------|---------|----------|-----------|-------------|---------|
| RANGE<br>FACTOR | 1.3E+0  | 1. 3E+0  | 1         | ;         | ;         | ;        | 1. 35+0   | ;           | 1. 3E +0 | ;        | ł       | ;       | ;        | ;       | 1. 3E +0 | 1. 3E+0  | 1. 3E+0 | ł        | ;        | ;       | ţ       | ł       | ł        | ;           | ;       | :        | ł         | ;           | ;       |
| UMDA<br>FREQ    | 1.5E-07 | 1.7E-09  | N/A       | N/A       | N/A       | N/A      | 7.2E-09   | N/A         | 6.9E-10  | 0.0E+00  | N/A     | 0.05+00 | 0.01400  | 0.05+00 | 3.1E-09  | 3. IE-09 | 2.7E-08 | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A         | N/A     | N/A      | N/A       | N/A         | N/A     |
| RANGE<br>Factor | 1.3E+01 | 1, 35+01 | 1. JE +01 | 1. 3E+01  | 1. JE +01 | 1. 3E+01 | 1.3E+01   | 1. 36+01    | 1.3E+01  |          |         |         |          |         | 1.3E+01  | 1.36+01  | 1.3E+01 | 1. 76+01 | 1. XE+01 | 1.36+01 | 1.35+01 | 1.36+01 | 1. 3E+01 | 1.3E+01     | 1.35+01 | 1.36+01  | 1. JE +01 | 1, 35 +01   | 1.35+01 |
| TEAD<br>Freq    | 6.1E-07 | 3.4E-09  | 1.6E-09   | 3.9E-11   | 3.9E-11   | 1.4E-08  | 1.4E-08   | 1.4E-08     | 1.4E-09  | 0.0E+00  | 0.0E+00 | 0.0E+00 | 0.0E+00  | 0.0E+00 | 6.2E-09  | 6.2E-09  | 5.4E-08 | 6.3E-09  | 6.3E-09  | 7.2E-10 | 7.26-10 | 1.3E-08 | 1.35-09  | 1. JE-08    | 8.6E-09 | 1.1E-(19 | 1.1E-09   | 1.15-09     | 1.15-09 |
| RANGE<br>Factor | :       | 1        | 1.36+01   | ł         | 1.3E+01   | 1        | 1         | 1           | ;        | :        | ł       | ł       | ł        | 1       | 1        | ł        | ;       | ł        | 1        | ł       | ;       | ł       | ł        | ł           | ;       | ;        | ;         | 1           | ;       |
| FUDA<br>FREQ    | N/A     | N/A      | 8.0E-10   | N/A       | 2.0E-11   | N/A      | N/A       | <b>K</b> /A | N/A      | N/A      | 0.0E+00 | N/A     | N/A      | N/A     | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A         | N/A     | N/A      | N/A       | N/A         | N/A     |
| RANGE<br>FACTOR | 1       | :        | ;         | ſ         | 1         | ſ        | 1. JE +01 | ł           | 1.3E+01  | ł        | ł       | ł       | ł        | ;       | 1.3E+01  | 1.3E+01  | ł       | 1        | ł        | ł       | 5       | ł       | ł        | ł           | ł       | ;        | ł         | 1           | :       |
| FBA<br>FREQ     | N/A     | N/A      | N/A       | N/A       | N/A       | N/A      | 6.7E-09   | N/A         | 6.9E-10  | N/A      | N/A     | N/A     | N/A      | N/A     | 3.1E-09  | 3. IE-09 | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A         | N/A     | N/A      | N/A       | N/A         | N/A     |
| RANGE<br>Factor | ;       | ;        | ;         | ;         | ;         | ;        | ;         | 1. JE+01    | ļ        | ;        | ļ       | ;       | ;        | ;       | ;        | ł        | !       | ;        | ;        | ;       | :       | ł       | ;        | ;           | ł       | 1        | :         | ł           | :       |
| NAAP<br>Freg    | N/A     | N/A      | N/A       | N/A       | N/A       | N/A      | N/A       | 7.2E-09     | N/A      | N/A      | A/A     | N/A     | N/A      | N/A     | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A         | N/A     | N/A      | N/A       | N/A         | N/A     |
| RANGE<br>Factor | :       | ;        | :         | :         | ł         | :        | ł         | t           | ;        | 1        | !       | ;       | ł        | ;       | 1.3E+01  | 1.3E+01  | :       | ١        | 1        | 1       | ;       | ;       | ;        | ;           | 1       | 1        | ł         | ł           | :       |
| LBAD<br>FREQ    | N/A     | N/A      | N/A       | N/A       | N/A       | N/A      | N/A       | N/A         | N/A      | N/A      | 0.0E+00 | 0.0E+00 | 0.0E+00  | N/A     | 3.1E-09  | 3.1E-09  | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A         | N/A     | N/A      | N/A       | N/A         | N/A     |
| RANGE<br>Factor | !       | :        | 1         | ;         | ł         | ;        | 1.3E+01   | :           | !        | 1        | ł       | ł       | 1        | ł       | ł        | ł        | 1       | ;        | ł        | ł       | 1       | 1       | ;        | ;           | ł       | 1        | ł         | 1           | :       |
| AF6<br>FREQ     | N/A     | N/A      | N/A       | N/A       | N/A       | N/A      | 6.7E-09   | N/A         | N/A      | N/A      | N/A     | N/A     | N/A      | N/A     | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A         | N/A     | N/A      | N/A       | N/A         | N/A     |
| RANGE<br>Factor | :       | ;        | 1. JE +01 | 1. JE +01 | 1. 3E +01 | ſ        | 1.36+01   | ſ           | 1. 35+01 | {        | ł       | 1       | ;        | {       | 1.35+01  | 1.35+01  | :       | ;        | {        | ł       | ł       | !       | :        | ;           | ;       | ;        | :         | ;           | ;       |
| ANAD I<br>FRED  | N/A     | N/A      | 8.05-10   | 2.0E-11   | 2.0E-11   | N/A      | 7.2E-09   | N/A         | 6.9E-10  | 0°.0E+00 | 0°0E+00 | 0.06+00 | 0° 0E+00 | N/A     | 3.1E-09  | 3.1E-09  | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A         | N/A     | N/A      | N/A       | N/A         | N/A     |
| <b>10</b>       | μ.      | -        | -         | -         | +         | +        | -         | -           | -        | -        | -       | -       | -        | -       | -        | *        | •       | υ        | Ś        | ŝ       | ŝ       | 67      | ŝ        | <b>u</b> 7) | ŝ       | 67       | ç         | <b>u</b> 7) | 67      |
| 66              |         |          |           |           |           |          |           |             |          |          |         |         |          |         |          |          |         |          |          |         |         |         |          |             |         |          |           |             |         |
| SCEN-<br>AF10   | HCSVC   | HCRGC    | HCDHC     | HCCBC     | HCCHC     | HCKGC    | Энасн     | HCKVC       | HCHVC    | HCP6C    | HCFHC   | HCFVC   | HCOCC    | HCQVC   | HCRGC    | HCKVC    | HCSVC   | HC B6S   | HCDHS    | HCC6S   | HCCHS   | HCK6S   | HCKHS    | SVAJH       | HCMVS   | HCF65    | HCFHS     | HCFVS       | HCOES   |





\_520). ● ●

<u>ند</u>







1.1.1.1.1.1.1.1

File: HDLNAT.NK1, Page 3 Date 18-Aug-87

TABLE 6-4 (Continued)

# HANDLING ACCIDENTS - NATIONAL PROCESSING OPTION - PER PALLET OR CONTAINER

.

### Accident Frequencies and Range Factors

| RANGE<br>Factor | 1       | ł        | !        | ł       | 1        | ł        | 1       | ;       | ł        | !        | ł        | :        | 1       | i       | !       | ;       | ł       | ;        | 1        | 1        | ł       | ;        | :        | :       | ;       | :       | ;        | ;       | 1       |
|-----------------|---------|----------|----------|---------|----------|----------|---------|---------|----------|----------|----------|----------|---------|---------|---------|---------|---------|----------|----------|----------|---------|----------|----------|---------|---------|---------|----------|---------|---------|
| UMDA<br>Freg    | N/A     | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     |
| RANGE<br>Factor | 1.3E+01 | 1. 3E+01 | 1. 3E+01 | 1.36+01 | 3. IE+01 | 3. IE+01 | :       | ł       | 3. IE+01 | 3. JE+01 | 3. IE+01 | 3. IE+01 | :       | ;       | :       | :       | ;       | 3. IE+01 | 3. IE+01 | 3. IE+01 | 1.36+01 | 1, 3E+01 | 1        | :       | 1.3E+01 | 1.35+01 | 1. 3E+01 | 1.36+01 | ł       |
| TEAD<br>Freq    | 1.16-09 | 4.95-09  | 4.96-09  | 1.46-07 | 6.2E-11  | 1.96-10  | 0.0E+00 | 0.0E+00 | 2.16-10  | 2.1E-10  | 2.16-10  | 2.7E-10  | 0.0E+00 | 0.0E+00 | 0.0E+00 | 0.0E+00 | 0.0E+00 | 1.65-10  | 1.66-10  | 3.96-09  | 3.76-09 | 1.16-08  | 0°.0E+00 | 0.0E+00 | 1.2E-08 | 1.2E-08 | 1.26-08  | 1.65-08 | 0.0E+00 |
| RANGE<br>Factor | ;       | ł        | 1        | ł       | ;        | :        | 1       | ;       | ł        | ł        | ;        | ł        | ;       | :       | ł       | !       | ;       | 1        | ;        | !        | ł       | ;        | 5        | ł       | ;       | ł       | :        | ł       | 1       |
| FRED            | A/A     | N/A      | N/A      | N/A     | N/N      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     |
| RANGE<br>Factor | ;       | ;        | ł        | :       | ;        | 1        | :       | :       | ł        | :        | ł        | ;        | :       | 1       | ;       | :       | :       | :        | ;        | ł        | :       | ł        | 1        | 1       | :       | ł       | ł        | ł       | ;       |
| PBA<br>FREQ     | N/A     | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     |
| RANGE<br>Factor | ;       | ł        | ;        | ł       | ;        | :        | :       | ł       | ;        | 1        | ł        | 1        | ł       | ł       | ł       | ľ       | ۰<br>۱  | ;        | ł        | ;        | ;       | ;        | 1        | !       | 1       | ł       | }        | }       | ;       |
| NAAP<br>Freg    | N/A     | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     |
| RANGE<br>Factor | :       | ;        | ł        | ;       | ;        | ł        | 1       | ;       | ;        | ł        | :        | ;        | ;       | ;       | 1       | :       | ł       | :        | 1        | ;        | 1       | 1        | ł        | ł       | ł       | ;       | :        | ;       | ;       |
| LBAD<br>Freq    | N/A     | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     |
| RANGE<br>Factor | ;       | ;        | 1        | ł       | ł        | ;        | 1       | ;       | ;        | ł        | :        | :        | ł       | ;       | 1       | ł       | :       | ł        | :        | :        | :       | !        | ;        | 1       | ;       | ;       | ł        | 1       | :       |
| AP6<br>Fred     | N/A     | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     |
| RANGE<br>Factor | ;       | ;        | ;        | ;       | ;        | :        | :       | ;       | :        | 1        | :        | ł        | !       | :       | ł       | ł       | ł       | 1        | :        | :        | ł       | !        | ł        | 1       | !       | ł       | ł        | ;       | ;       |
| ANAD<br>FREQ    | N/A     | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | A/N      | N/A     | N/A     |
| DF. NO.         | 5       | ŝ        | ŝ        | רי      | 9        | 9        | ¢       | 9       | 4        | 9        | 9        | 9        | 9       | 9       | -0      | 9       | œ-      | 9        | 9        | 9        | 1       | 1        | 1        | 1       | 1       | 1       | ٢        | 1       | 1       |
| SCEN- C<br>AKIO | HCBVS   | HCR6S    | HCRVS    | HCSVS   | HC F6F   | HCDHF    | HCC6F   | HCCHF   | HCKEF    | HCKHF    | HCKVF    | HCMVF    | HCFGF   | HCPHF   | HCPVF   | HCQGF   | HCQVF   | HCRGF    | HCRVF    | HCSVF    | HCR65   | HCDHS    | HCCGS    | HCCHS   | HCK6S   | HCKHS   | HCKVS    | HCHUS   | HCF6S   |
|                 |         |          |          |         |          |          |         |         |          |          |          |          |         |         |         |         |         |          |          |          |         |          |          |         |         |         |          |         |         |

6-53

ر بر

File: HDLNAI.Mkl, Page 4 Date 18-Aug-87

TABLE 6-4 (Continued)

HANDLING ACCIDENTS - NATIONAL PROCESSING OPTION - FER PALLET OR CONTAINER

ŝ

Accident Frequencies and Range Factors

|       | 19460      |             | DAUCE    |         | ANCT.    |         | DANC     | 400     | DANFE    | CUDA            | DANCE    | TEAD            | DANCC     | 1 MOV   | DANCE     |
|-------|------------|-------------|----------|---------|----------|---------|----------|---------|----------|-----------------|----------|-----------------|-----------|---------|-----------|
| . I . | ALLER CTOR | Arb<br>Freg | FACTOR   | FREG    | FACTOR   | FRED    | FACTOR   | FRED    | FACTOR   | FRED            | FACTOR   | FRED            | FACTOR    | FREQ    | FACTOR    |
|       |            |             |          |         |          |         |          |         |          |                 |          |                 |           |         |           |
|       | :          | N/A         | ł        | N/A     | ł        | N/A     | :        | N/A     | 1        | N/A             | :        | 0.0E+00         | ;         | N/A     | :         |
|       | ;          | N/A         | :        | N/A     | ;        | N/A     | ;        | N/A     | ;        | N/A             | 1        | 0.0E+00         | :         | N/A     | ł         |
|       | :          | N/A         | :        | N/A     | ;        | N/A     | ;        | N/A     | :        | N/A             | 1        | 0°0E+00         | ļ         | N/A     | ;         |
|       | ;          | N/A         | ;        | N/A     | :        | N/A     | ł        | N/A     | 1        | N/A             | 1        | 0.0E+00         | :         | N/A     | !         |
|       | :          | N/A         | :        | N/A     | :        | N/A     | :        | N/A     | ;        | N/A             | 1        | 9.6E-09         | 1. 3E+01  | N/A     | ł         |
|       | :          | N/A         | :        | N/A     | ;        | N/A     | ł        | N/A     | !        | N/A             | ł        | 9.6E-09         | 1.3E+01   | N/A     | ;         |
|       | :          | N/A         | ;        | N/A     | :        | N/A     | :        | N/A     | :        | N/A             | ;        | 2.3E-07         | 1. JE +01 | N/A     | ł         |
|       | :          | N/A         | 1        | N/A     | :        | N/A     | 1        | N/A     | !        | N/A             | ł        | 2.3E-08         | 1. 3E +01 | 2.3E-08 | 1.35+01   |
| -     | 1.36+01    | N/A         | ł        | N/A     | :        | N/A     | :        | N/A     | ł        | 1.25-08         | 1. 3E+01 | 1.2E-08         | 1.3E+01   | N/A     | ;         |
| ٥.    | 1.36+01    | N/A         | :        | N/A     | :        | N/A     | 1        | N/A     | ;        | N/A             | :        | 8.8E-09         | 1. 3E+01  | N/A     | ł         |
| œ     | 1.3E+01    | N/A         | :        | N/A     | :        | N/A     | ;        | N/A     | ;        | <b>8.8</b> E-09 | 1.35+01  | <b>8.8E-09</b>  | 1.3E+01   | N/A     | :         |
|       | ;          | N/A         | ł        | N/A     | :        | N/A     | :        | N/A     | ł        | N/A             | 1        | 2.2E-08         | 1. 3E+01  | N/A     | ;         |
| 8     | 1.36+01    | 2.2E-08     | 1.3E+01  | N/A     | :        | N/A     | :        | 2.2E-08 | 1.3E+01  | N/A             | ;        | 2.25-08         | 1.3E+01   | 2.2E-08 | 1.36+01   |
|       | ;          | N/A         | !        | N/A     | ł        | 2.25-08 | 1. JE+01 | N/A     | ;        | N/A             | ł        | 2.2E-08         | 1.3E+01   | N/A     | :         |
| 8     | 1.3€+01    | N/A         | ;        | N/A     | :        | N/A     | :        | 1.2E-08 | 1.3E+01  | N/A             | 1        | 1.2E-08         | 1.35+01   | 1.2E-09 | 1.35+01   |
| В     | 1. 3E+01   | N/A         | ł        | N/A     | ;        | M/A     | :        | N/A     | ;        | N/A             | ł        | 1.05-08         | 1.3E+01   | 1.0E-08 | 1. 3E+01  |
| 8     | 1.3E+01    | N/A         | :        | 1.0E-08 | 1.3E+01  | N/A     | ;        | N/A     | !        | 1.0E-0B         | 1.3E+01  | 1.(E-08         | 1. 3E+01  | N, A    | ł         |
| 8     | 1.36+01    | N/A         | ;        | 1.05-08 | 1.36+01  | N/A     | ł        | N/A     | ł        | N/A             | 1        | 1.05-08         | 1.36+01   | 1.0E-08 | 1. JE +01 |
| 8     | 1. XE+01   | N/A         | ;        | 1.0E-09 | 1.3E+01  | N/A     | ;        | N/A     | ;        | N/A             | 1        | 1.0E-08         | 1. 3E+01  | 1.0E-08 | 1,35+01   |
|       | ł          | N/A         | !        | N/A     | ;        | N/A     | ;        | N/A     | ł        | N/A             | ł        | 1.05-08         | 1. JE+01  | 1.0E-08 | 1.3E+01   |
| 8     | 1.3E+01    | N/A         | :        | 1.36-08 | 1. 3E+01 | N/A     | 1        | 1.3E-08 | 1. 3E+01 | N/A             | 1        | 1.36-08         | 1.35+01   | 1.3E-09 | 1. 3E 401 |
| 8     | 1.36+01    | N/A         | !        | 1.3E-08 | 1.3E+01  | N/A     | ;        | 1.36-08 | 1.3E+01  | N/A             | ;        | 1. 3E -08       | 1. 3E+01  | 1.3E-08 | 1. 3E+01  |
|       | :          | N/A         | :        | N/A     | :        | N/A     | ł        | N/A     | ;        | N/A             | ;        | <b>B.5</b> E-09 | 1. 3E+01  | 8.5E-03 | 1. 3E +01 |
|       | ;          | N/A         | ;        | N/A     | ;        | N/A     | ١        | N/A     | 1        | N/A             | ;        | 8.6E-10         | 3. IE+01  | 8.6E-10 | 3. IE+01  |
| 0     | 3. 1E+01   | N/A         | :        | N/A     | :        | N/A     | . 1      | N/A     | :        | 5.46-10         | 3. IE+01 | 5.4E-10         | 3. IE+01  | N/A     | ł         |
| ÷     | 3. IE+01   | N/A         | :        | N/A     | ;        | N/A     | 1        | N/A     | :        | N/A             | ł        | 2.6E-10         | 3.1E+01   | N/A     | ł         |
| 0     | 3.1E+01    | N/A         | :        | A/A     | ;        | N/A     | ;        | N/A     | :        | 2.65-10         | 3, 1E+ù1 | 2.6E-10         | 3. IE+01  | N/A     | ;         |
|       | !          | N/A         | :        | N/A     | ł        | N/A     | ;        | N/A     | 1        | N/A             | ;        | 6.9E-10         | 3. IE+01  | N/A     | !         |
| 0     | 3, IE+01   | 6.9E-10     | 3. JE+01 | N/A     | :        | N/A     | ł        | 6.95-10 | 3. 1E+01 | N/A             | ;        | 6.9E-10         | 3.1E+01   | 6.9E-10 | 3.1E+01   |

REALES DEPEND

277.2

SSENS CARAC

<u>ن</u>

6-54



0.566.5.5



File: HDLNAT.Mk1, Fage 5 Date 18-Aug-87

TABLE 6-4 (Continued)

HANDLING ACCIDENTS - NATIONAL PROCESSING OFTION - PER PALLET OR CONTAINER

Accident Frequencies and Range Factors

| -NEUS   | 0F. NO. | . ANAD           | RANGE      | AFG     | RANGE     | LBAD    | RANGE     | NAAP    | RANGE    | FBA     | RANGE    | FUDA    | RANGE     | TEAD     | RANGE     | UMDA              | RANGE     |
|---------|---------|------------------|------------|---------|-----------|---------|-----------|---------|----------|---------|----------|---------|-----------|----------|-----------|-------------------|-----------|
| AK10    |         |                  | FACTOR     | PHE9    | FACTUR    | P HER   |           | + + + F |          | HER     |          | FRE     |           | HER      |           | + KEA             | F.AL. IUK |
| 30 A JH | o       | N7/A             | 1          | N/0     | ;         | N/0     | :         | 6.9F-10 | 3, 16401 | N/A     | :        | N/A     | :         | A. 9E-10 | 3. IE+01  | N/A               | ;         |
| HCMVF   | • •     | 5.3E-10          | ) 3.1E+01  | N/A     | ;         | N/A     | ;         | N/A     |          | 5.3E-10 | 3.1E+01  | N/A     | ;         | 5. 3E-10 | 3, 1E+01  | 5. 3E-10          | 3. 1E+0   |
| HCF6F   | • •     | 1.25-10          | ) 3.1E+01  | N/A     | ł         | N/A     | ;         | N/A     | ł        | N/A     | 1        | N/A     | {         | 1.2E-10  | 3. IE+UI  | 1.2E-10           | 3. IE+0   |
| HEFHF   | 6       | 1.2E-10          | ) 3. JE+01 | N/A     | ;         | 1.2E-10 | 3. 1E+01  | N/A     | 1        | N/A     | ;        | 1.2E-10 | 3. JE+01  | 1.2E-10  | 3. IE+01  | N/A               | ;         |
| HCFVF   | 6       | 1.2E-10          | ) 3.1E+01  | N/A     | :         | 1.2E-10 | 3. IE+01  | N/A     | ł        | N/A     | !        | N/A     | :         | 1.2E-10  | 3. 1E+01  | 1.25-10           | 3.1E+0    |
| HCOGF   | 0-      | 1.2E-10          | 0 3° 1E+01 | N/A     | ł         | 1.2E-10 | 3. IE+01  | N/A     | 1        | N/A     | :        | N/A     | ł         | 1.25-10  | 3. JE+01  | 1.26-10           | 3. 1E+0   |
| HCOVF   | 6       | N/A              | !          | N/A     | ;         | N/A     | :         | N/A     | 1        | N/A     | :        | N/A     | 1         | 1.2E-10  | 3. 1E+01  | I.2E-10           | 3.1E+0    |
| HCRGF   | 6       | 5.6E-10          | ) 3. IE+01 | N/A     | ;         | 5.6E-10 | 3. IE+01  | N/A     | ;        | 5.6E-10 | 3. IE+01 | N/A     | ;         | 5.66-10  | 3. JE+01  | 5.6E-10           | 3. IE+0   |
| HCKVF   | 6       | 5.6E-10          | 0 3.1E+01  | N/A     | 1         | 5.6E-10 | 3. 1E+01  | N/A     | 1        | 5.6E-10 | 3. IE+01 | N/A     | :         | 5.6E-10  | 3. IE+01  | 5.6E-10           | 7. 1E+0   |
| HCSVF   | •       | N/A              | 1          | N/A     | :         | N/A     | :         | N/A     | 1        | N/A     | ł        | N/A     | ;         | 2.7E-10  | 3. 1E+01  | 2.7E-10           | 3. IE+0   |
| HCRGS   | 10      | N/A              | :          | N/A     | :         | N/A     | ;         | N/A     | ;        | N/A     | ł        | N/A     | 1         | 1.16-08  | 1.36+01   | 1.1E-08           | 1.3E+û    |
| HCDHS   | 10      | 6.9E-09          | P 1.3E+01  | N/A     | ł         | N/A     | ;         | N/A     | ;        | N/A     | ;        | 60-36-9 | 1.36+01   | 6.9E-()9 | 1.36+01   | N/A               | ;         |
| HCCGS   | 10      | 3.4E-09          | 0 1.3E+01  | N/A     | ţ         | N/A     | :         | N/A     | 1        | N/A     | :        | N/A     | ;         | 3.4E-09  | 1.35+01   | N/A               | ;         |
| HCCHS   | 10      | 3.4E-09          | 7 1. 3E+01 | N/A     | !         | N/A     | 1         | N/A     | ;        | N/A     | ;        | 3.4E-09 | 1. JE +01 | 3.4E-09  | 1.3E+01   | N/A               | ;         |
| HCK6S   | 10      | N/A              | ;          | N/A     | !         | N/A     | 1         | N/A     | ;        | N/A     | !        | N/A     | !         | 8.85-03  | 1. 3E+01  | N/A               | ;         |
| HCKHS   | 10      | 8.BE-05          | 0 1.3E+01  | 8.BE-09 | 1. JE +01 | N/A     | 1         | N/A     | ł        | 8.85-09 | 1. 3E+01 | N/A     | :         | 8.85-09  | 1.35+01   | <b>B.</b> 8E - 09 | 1.3E40    |
| HCKVS   | 10      | N/A              | ;          | N/A     | ;         | N/A     | :         | 8.86-09 | 1. 3E+01 | N/A     | ;        | N/A     | ţ         | 8.8E-09  | 1.36+01   | N/A               | ;         |
| HCHVS   | 91      | 6.BE-09          | 1.35+01    | N/A     | !         | N/A     | :         | N/A     | ł        | 6.BE-09 | 1.36+01  | N/A     | ;         | 6.8E-09  | 1.36+01   | 6.8E-09           | 1. 3E+0   |
| HCF6S   | 10      | 1.6E-05          | 10+32-1    | N/A     | ł         | N/A     | 1         | N/A     | ł        | N/A     | ł        | N/A     | ł         | 1.6E-09  | 1.38+01   | 1.65-09           | 1. JE +0  |
| HCFHS   | 10      | 1. <u>4</u> E-05 | 0 1.3E+01  | N/A     | :         | 1.6E-09 | 1. 3E+01  | N/A     | ł        | N/A     | 1        | 1.65-09 | 1.3E+01   | 1.6E-09  | 1.3E+01   | N/A               | ;         |
| HCFVS   | 10      | 1.6E-09          | 1.3E+01    | N/A     | !         | 1.65-09 | 1. JE +01 | N/A     | ;        | N/A     | 1        | N/A     | ;         | 1.6E-07  | 1, 3E +01 | 1.6E-09           | 1.3E+0    |
| HCDGS   | 10      | 1.6E-05          | 1.36+01    | N/A     | ;         | 1.65-09 | 1.36+01   | N/A     | 1        | N/A     | ;        | N/A     | ł         | 1.65-09  | 1.3E+01   | 1.6E-09           | 1.35+0    |
| HCQVS   | 10      | N/A              | 1          | N/A     | ;         | N/A     | ;         | N/A     | ł        | N/A     | ł        | N/A     | ł         | 1.6E-09  | 1, JE +01 | 1.66-09           | 1.3E+0    |
| HCK65   | Ú1      | 7.2E-09          | 1.36+01    | N/A     | ł         | 7.2E-09 | 1. 3E+01  | N/A     | ł        | 7.2E-09 | 1.3E+01  | N/A     | 1         | 7.2E-09  | 1. 3E+01  | 7.2E-09           | 1. 3E +ú  |
| HCRVS   | 10      | 7.2E-09          | 1.3E+01    | N/A     | 1         | 7.2E-09 | 1.3E+01   | N/A     | 1        | 7.2E-09 | 1.3E+01  | N/A     | ł         | 7.2E-09  | 1. JE +01 | 7.2E-09           | 1. YE +0  |
| HCSVS   | 10      | N/A              | :          | N/A     | 1         | N/A     | 1         | N/A     | 1        | N/A     | ł        | N/A     | ;         | 3.5E-09  | 1.3E+01   | 3.56-09           | 1.36+0    |
| нсрнс   | Ξ       | 2.9E-09          | 1 2.6E+01  | N/A     | ł         | N/A     | ł         | N/A     | ł        | N/A     | ł        | 2.9E-09 | 2.6E+01   | 5.8E-09  | 2.6E+01   | N/A               | !         |
| HCCGC   | Ξ       | 1.4E-09          | 1 2.6E+01  | N/A     | 1         | N/A     | 1         | N/A     | ;        | N/A     | ļ        | N/A     | !         | 2.9E-09  | 2.6E+01   | N/A               | 1         |
| HCCHC   | Ξ       | 1.4E-03          | 1 2.6E+01  | N/A     | ;         | N/A     | ;         | N/A     | ł        | N/A     | ł        | 1.4E-09 | 2.6E+01   | 2.9E-03  | 2.6E+01   | N/A               | :         |

<u>arana nakata nakakata nanana akakata kakata nakata nananan nanana nanana nanana natata na</u>

TABLE 6-4 (Continued)

File: HDLNAT.Wk1, Page 6 Date 18-Aug-87

HANDLING ACCIDENTS - NATIONAL PROCESSING OPTION - PER PALLET OR CONTAINER

Accident Frequencies and Range Factors

| SCEN- OF<br>Arid | , NO. | ANAD<br>FREG | RANGE<br>Factor | AP6<br>Freq | RANGE<br>Factor | L BAD<br>Fred | RANGE<br>Factor | NAAP<br>Freq | RANGE<br>Factor | PBA<br>Freq | RANGE<br>Factor | FUDA<br>Fred | RANGE<br>Factor | TEAD<br>Freq | RANGE<br>Factor | UNDA<br>Fred | range<br>Factor |
|------------------|-------|--------------|-----------------|-------------|-----------------|---------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|-----------------|
|                  | 1     |              |                 |             |                 |               |                 |              |                 |             |                 |              |                 |              | 5               |              |                 |
| HCMVC            | 11    | 2.2E-09      | 2.6E+01         | N/A         | 1               | N/A           | ł               | N/A          | ł               | 2.2E-09     | 2.6E+01         | N/A          | ł               | 4.35-09      | 2.6E+01         | 2.2E-09      | 2.6E+(1)        |
| HCPGC            | Ξ     | 4.86-10      | 2.6E+01         | N/A         | ;               | N/A           | ł               | N/A          | ;               | N/A         | ł               | N/A          | ł               | 9.6E-10      | 2.6E+01         | 4.86-10      | 2, 6E+i)1       |
| HCPHC            | Ξ     | 4.85-10      | 2.6E+01         | N/A         | ;               | 4.8E-10       | 2.6E+01         | N/A          | ;               | N/A         | !               | 4.8E-10      | 2.6E+01         | 9.6E-10      | 2.6E+01         | N/A          | ł               |
| HCFVC            | Ξ     | 4.86-10      | 2.6E+01         | N/A         | 1               | 4.86-10       | 2.6E+01         | N/A          | ;               | N/A         | ł               | N/A          | ;               | 9.6E-10      | 2.6E+01         | 4.86-10      | 2.6E+01         |
| HCDEC            | Ξ     | 3.6E-10      | 2.6E+01         | N/A         | ;               | 3.6E-10       | 2.6E+01         | N/A          | ;               | N/A         | ł               | N/A          | 1               | 7.2E-10      | 2.6E+01         | 3.66-10      | 2.6E+01         |
| HCQVC            | Ξ     | N/A          | 1               | N/A         | 1               | N/A           | ł               | N/A          | 1               | N/A         | ;               | N/A          | 1               | 7.2E-10      | 2.6E+01         | 3.6E-10      | 2.6E+01         |
| HCRGC            | Ξ     | 9.0E-10      | 2.6E+01         | N/A         | ;               | 9.05-10       | 2.6E+01         | N/A          | :               | 9.0E-10     | 2.6E+01         | N/A          | 1               | 1.8E-09      | 2.6E+()1        | 9.0E-10      | 2.6E+01         |
| HCRVC            | Ξ     | 9.0E-10      | 2.6E+01         | N/A         | 1               | 9.0E-10       | 2.6E+01         | N/A          | ł               | 9.0E-10     | 2.6E+01         | N/A          | 1               | 1.86-09      | 2.6E+01         | 9.05-10      | 2.6E+01         |
| HCDHC            | 12    | 2.15-10      | 2.6E+01         | N/A         | 1               | N/A           | ł               | N/A          | 1               | N/A         | ł               | 2. IE-10     | 2.6E+01         | 4. iE-10     | 2,6€+01         | N/A          | ł               |
| HCCGC            | 12    | 1.0E-10      | 2.6E+01         | N/A         | ł               | N/A           | ł               | N/A          | ł               | N/A         | ;               | N/A          | 1               | 2. IE-10     | 2.6E+01         | N/A          | ;               |
| HCCHC            | 12    | 1.0E-10      | 2.6E+01         | N/A         | 1               | N/A           | :               | N/A          | 1               | N/A         | ł               | 1.0E-10      | 2.6E+01         | 2.1E-10      | 2.6E+01         | N/A          | 1               |
| HCHVC            | 12    | 1.56-10      | 2.6E+01         | N/A         | ;               | N/A           | ł               | N/A          | ;               | 1.56-10     | 2.6E+01         | N/A          | ;               | 3.1E-10      | 2.6E+01         | 1.SE-10      | 2.6E+01         |
| HCF6C            | 12    | 3.4E-11      | 2.6E+01         | N/A         | ;               | N/A           | 1               | N/A          | !               | N/A         | ł               | N/A          | ;               | 6.9E-11      | 2.6E+01         | 3.4E-11      | 2.6E+01         |
| HCFHC            | 12    | 3.4E-11      | 2.6E+01         | N/A         | 1               | 3.4E-11       | 2.6E+01         | N/A          | ;               | N/A         | ł               | 3.4E-11      | 2.6E+01         | 6.9E-11      | 2.6E+01         | N/A          | ;               |
| HCPVC            | 12    | 3.4E-11      | 2.6E+01         | N/A         | :               | 3.46-11       | 2.6E+01         | N/A          | ;               | N/A         | ł               | N/A          | !               | 6.9E-11      | 2.6E+01         | 3.4E-11      | 2.6E+(i)        |
| HCBGC            | 12    | 2.6E-11      | 2.6E+01         | N/A         | 1               | 2.6E-11       | 2.6E+01         | N/A          | ł               | N/A         | ł               | N/A          | 1               | 5.2E-11      | 2.6E+01         | 2.6E-11      | 2.6E+01         |
| HCQVC            | 12    | N/A          | ł               | N/A         | ł               | N/A           | ł               | N/A          | ;               | N/A         | ;               | N/A          | ;               | 5. ZE-11     | 2.6E+01         | 2.6E-11      | 2.6E+01         |
| HCRGC            | 12    | 6.5E-11      | 2.6E+01         | N/A         | 1               | 6.5E-11       | 2.6E+01         | N/A          | 1               | 6.5E-11     | 2.6E+01         | N/A          | ł               | 1.3E-10      | 2.6E+01         | 6.5E-11      | 2.6E+01         |
| HCRVC            | 12    | 6.5E-11      | 2.6E+01         | N/A         | 1               | 6.5E-11       | 2.6E+01         | N/A          | ;               | 6.5E-11     | 2.6E+01         | N/A          | :               | 1.3E-10      | 2.6E+01         | 6.5E-11      | 2.6E+01         |
| HCRGC            | 17    | N/A          | ;               | A/A         | :               | N/A           | ;               | N/A          | !               | N/A         | !               | N/A          | :               | 6.0E-13      | 2.6E+01         | 6.0E-13      | 2.6E+01         |
| HCRGC            | 1     | 6.0E-13      | 2.6E+01         | N/A         | !               | 6.0E-13       | 2.6E+01         | N/A          | ł               | 6.0E-13     | 2.6E+01         | N/A          | :               | 6.0E-13      | 2.6E+01         | 6.0113       | 2.6E+01         |
| HCRVC            | 11    | 6.0E-13      | 2.6E+01         | N/A         | 1               | 6.0E-13       | 2.6E+01         | N/A          | ;               | 6.0E-13     | 2.6E+01         | N/A          | ł               | 6.0E-13      | 2.6€+01         | 6.((E-13     | 2.6E+01         |
| HCBGC            | 81    | N/A          | ł               | N/A         | 1               | M/A           | ŗ               | N/A          | 1               | N/A         | 1               | N/A          | 1               | 1.25-12      | 2.6E+01         | 1.2E-12      | 2.66+01         |
| HLRGC            | 8     | 1.25-12      | 2.6E+01         | N/A         | ;               | 1.2E-12       | 2.6E+01         | N/A          | !               | 1.2E-12     | 2.6E+01         | N/A          | ;               | 1.2E-12      | 2.6E+01         | 1.2E-12      | 2.6E+(1         |
| HCRVC            | 8     | 1.2E-12      | 2.6E+01         | N/A         | ł               | 1.2E-12       | 2.6E+01         | N/A          | ;               | 1.26-12     | 2.6E+01         | R/A          | :               | 1.2E-12      | 2.6E+ú}         | 1.2E-12      | 2.6E+01         |
| HCBGC            | 19    | N/A          | 1               | N/A         | 1               | N/A           | :               | N/A          | ;               | N/A         | ;               | N/A          | ł               | 2.6E-15      | 3. 1E+01        | 2.6E-15      | 3. JE+01        |
| HCRGC            | 61    | 2.66-14      | J. 1E+01        | N/A         | !               | 2.65-14       | 3. JE+01        | N/A          | 1               | 2.6E-14     | 3. IE+01        | N/A          | ł               | 2.6E-14      | 3. IE+01        | 2.6E-14      | 3. IE+01        |
| HCRVC            | 16    | 2.6E-14      | 3, 1E+01        | N/A         | :               | 2.6E-14       | 3. IE+01        | N/A          | ;               | 2.6E-14     | 3. IE+01        | N/A          | :               | 2.6E-14      | 3.1E+01         | 2.6E-14      | 3. IE+01        |
| HCBEC            | 21    | A/A          | ł               | N/A         | 1               | N/A           | 1               | N/A          | ł               | N/A         | ;               | N/A          | 1               | 3.46-18      | 3. IE+01        | 3.4E-18      | 3.16+01         |

Υ.

**77** 

Ŋ

7

5







TABLE 6-4 (Continued)

File: HDLNAT.Wk1, Fage 7 Date 18-Aug-87

ኯዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀ

HANDLING ACCIDENTS - NATIONAL PROCESSING OPTION - PER PALLET OR CONTAINER

Accident Frequencies and Range Factors

| RANGE                                              | FAC TOR                                       | 1<br>1<br>1<br>1                                                                                 | J. 1E+01                                         | 3, 1E+01                                         | :                        | ł                        | :                        | ;                        | !                        | ł                        | ţ                        | ;                        | ;                        | ;                        | ;                        | !                                    | ;                                    | !                                    | 2.6E+111                             | 2.6E+01                              |                                                  | 2. bE +01                                        | 2.6E+01                                          | 2.66+01                  | 2.6E+m1                                          | 2.6E401                                          | ;                        | ;                        | !                        | ;                        | ;                        |
|----------------------------------------------------|-----------------------------------------------|--------------------------------------------------------------------------------------------------|--------------------------------------------------|--------------------------------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------------------|--------------------------------------|--------------------------------------|--------------------------------------|--------------------------------------|--------------------------------------------------|--------------------------------------------------|--------------------------------------------------|--------------------------|--------------------------------------------------|--------------------------------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|
| NDA                                                | FRED                                          |                                                                                                  | 3.1E-18                                          | 3.1E-18                                          | N/N                      | N,'A                     | N/A                                  | N/A                                  | N/A                                  | 1.3E-11                              | 11-34.1                              | N/A                                              | 1.4E-11                                          | 7.25-12                                          | 7.25-12                  | 7.2E-12                                          | 7.26-12                                          | N/A                      | N/A                      | N/A                      | N, A                     | N/A                      |
| RANGE                                              | : AC TOK                                      |                                                                                                  | 3. IE+01                                         | 3.16+01                                          | 2.6E+01                  | 2.6E+U1                  | 2.6E+01                  | 2.6E+01                  | 2.6E+01                  | 2.6E+01                              | 2.6E+01                              | 2.6E+01                              | 2.6E+01                              | 2.66+01                              | 2.6E+01                                          | 2.6E+01                                          | 2.6E+01                                          | 2.66+01                  | 2.6E+01                                          | 2.6E+#1                                          | 2.6E+01                  | 2.66+01                  | 2.6E+01                  | 2.6E+01                  | 2.6E+ůl                  |
| TEAD                                               | FREQ                                          |                                                                                                  | 3. IE-18                                         | 3.1E-18                                          | 8. 7E - 11               | 4. 36-11                 | 4.36-11                  | 6.5E-11                  | 1.4E-11                  | 1.46-11                  | 1.4E-11                  | 1.16-11                  | 1.16-11                  | 2.7E-11                  | 2.76-11                  | 2.36-11                              | 3.5E-11                              | 7.56-11                              | 1.36-11                              | 1.4E-11                              | 1.46-11                                          | 1.46-11                                          | 7.26-12                                          | 7.2E-12                  | 7.2E-12                                          | 7.2E-12                                          | 6.2E-11                  | 3. IE-11                 | 3.16-11                  | 4.65-11                  | 1.06-11                  |
| RANGE                                              | FACTOR                                        |                                                                                                  | :                                                | ł                                                | ł                        | ł                        | ł                        | :                        | ;                        | 1                        | ļ                        | ;                        | ł                        | ł                        | 1                        | 2.6E+01                              | ;                                    | 2.6E+01                              | !                                    | 1                                    | 2.66+01                                          | 1                                                | 1                                                | ;                        | ;                                                | ;                                                | ;<br>1                   | 1                        | :                        | ;                        | ;                        |
| FUDA                                               | FREQ                                          |                                                                                                  | N/A                                              | N/A                                              | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | 2.3E-11                              | N/A                                  | 3.56-11                              | N/A                                  | N/A                                  | 1.45-11                                          | N/A                                              | N/A                                              | N/A                      | N/A                                              | NA                                               | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      |
| ANGE                                               | ACTOR                                         |                                                                                                  | 3.1E+01                                          | 3. IE+01                                         | ł                        | 1                        | ł                        | ł                        | ł                        | ł                        | ł                        | 1                        | ł                        | ł                        | ł                        | ţ                                    | ;                                    | ł                                    | 2.6E+01                              | ł                                    | 1                                                | ł                                                | ł                                                | ł                        | 2.6E+01                                          | 2.6E+01                                          | ł                        | 1                        | ł                        | ;                        | ł                        |
| P.R.A                                              | FREG                                          | 1<br>2<br>2<br>1<br>1                                                                            | 3. IE-18                                         | 3. IE-18                                         | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | K/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                                  | N/A                                  | N/A                                  | 1.35-11                              | N/A                                  | N/A                                              | N/A                                              | N/A                                              | N/A                      | 7.25-12                                          | 7.26-12                                          | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      |
| RANGE                                              | FACTOR                                        | 1                                                                                                | ļ                                                | ;                                                | ;                        | ł                        | 1                        | :                        | ł                        | 1                        | ;                        | 1                        | :                        | !                        | ł                        | 1                                    | 1                                    | :                                    | ł                                    | ł                                    | ;                                                | ł                                                | 1                                                | 1                        | 1                                                | i                                                | ł                        | 1                        | ł                        | ł                        | ł                        |
|                                                    |                                               |                                                                                                  |                                                  |                                                  |                          |                          |                          |                          |                          |                          |                          |                          |                          |                          |                          |                                      |                                      |                                      |                                      |                                      |                                                  |                                                  |                                                  |                          |                                                  |                                                  |                          |                          |                          |                          |                          |
| NAAP                                               | FREQ                                          |                                                                                                  | N/A                                              | N/A                                              | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                                  | N/A                                  | N/A                                  | N/A                                  | N/A                                  | N/A                                              | N/A                                              | N/A                                              | N/A                      | N/A                                              | N/A                                              | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      |
| RANGE NAAP                                         | FACTOR FREQ                                   | 4<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9 | 3.1E+01 N/A                                      | 3.1E+01 N/A                                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | A/A                                  | N/A                                  | N/A                                  | N/A                                  | N/A                                  | 2.6E+01 N/A                                      | 2.6E+01 N/A                                      | 2.6E+01 N/A                                      | N/A                      | 2.6E+01 N/A                                      | 2.6E+U1 N/A                                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      |
| LBAD RANGE NAAP                                    | FREQ FACTOR FREQ                              |                                                                                                  | 3.1E-18 3.1E+01 N/A                              | 3.1E-18 3.1E+01 N/A                              | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                              | N/A N/A                              | N/A N/A                              | N/A N/A                              | N/A N/A                              | 1.4E-11 2.6E+01 N/A                              | 1.4E-11 2.6E+01 N/A                              | 7.7E-12 2.6E+01 N/A                              | N/A N/A                  | 7.2E-12 2.6E+01 N/A                              | 7.2E-12 2.6E+01 N/A                              | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  |
| RANGE LBAD RANGE NAAP                              | FACTOR FREQ FACTOR FREQ                       |                                                                                                  | 3.1E-18 3.1E+01 N/A                              | 3.1E-18 3.1E+01 N/A                              | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                              | N/A N/A                              | N/A N/A                              | N/A N/A                              | N/A N/A                              | 1.4E-11 2.6E+01 N/A                              | 1.4E-11 2.6E+01 N/A                              | 7.7E-12 2.6E+01 N/A                              | N/A N/A                  | 7.2E-12 2.6E+01 N/A                              | 7.2E-12 2.6E+U1 N/A                              | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A                  |
| AFG RANGE LBAD RANGE NAAP                          | FREQ FACTOR FREQ FACTOR FREQ                  |                                                                                                  | N/A 3.1E-18 3.1E+01 N/A                          | N/A 3.1E-18 3.1E+01 N/A                          | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A                          | N/A N/A N/A                          | N/A N/A N/A                          | N/A N/A N/A                          | N/A N/A N/A                          | N/A 1.4E-11 2.6E+01 N/A                          | N/A 1.4E-11 2.6E+01 N/A                          | N/A 7.7E-12 2.6E+01 N/A                          | N/A N/A N/A              | N/A 7.2E-12 2.6E+01 N/A                          | N/A 7.2E-12 2.6E+01 N/A                          | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              |
| RANGE AFG RANGE LBAD RANGE NAAP                    | ACTOR FREQ FACTOR FREQ FACTOR FREQ            |                                                                                                  | 3.1E+01 N/A 3.1E-18 3.1E+01 N/A                  | 3.1E+01 N/A 3.1E-18 3.1E+01 N/A                  | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | 2.6E+01 N/A N/A N/A                  | 2.6E+01 N/A N/A N/A                  | 2.6E+01 N/A N/A N/A                  | 2.6E+01 N/A N/A N/A                  | 2.6E+n1 N/A N/A N/A                  | 2.6E+01 N/A 1.4E-11 2.6E+01 N/A                  | 2.6E+01 N/A 1.4E-11 2.6E+01 N/A                  | 2.6E+01 N/A 7.7E-12 2.6E+01 N/A                  | N/A N/A N/A              | 2.6E+01 N/A 7.2E-12 2.6E+01 N/A                  | 2.6E+01 N/A 7.2E-12 2.6E+01 N/A                  | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              |
| ANAD RANGE AFG RANGE LBAD RANGE NAAP               | FREQ FACTOR FREQ FACTOR FREQ FACTOR FREQ      |                                                                                                  | 3.1E-18 3.1E+01 N/A 3.1E-18 3.1E+01 N/A          | 3.1E-18 3.1E+01 N/A 3.1E-18 3.1E+01 N/A          | N/A N/A N/A N/A          | N/A :- N/A N/A N/A       | N/A N/A N/A N/A          | N/A N/A N/A N/A          | N/A N/A N/A N/A          | N/A N/A N/A N/A          | N/A N/A N/A N/A          | N/A N/A N/A N/A          | 2.3E-11 2.6E+01 N/A N/A N/A          | 3.5E-11 2.6E+01 N/A N/A N/A          | 3.5E-11 2.6E+01 N/A N/A N/A          | 1.3E-11 2.6E+01 N/A N/A N/A          | 1.4E-11 2.6E+01 N/A N/A N/A          | 1.4E-11 2.6E+01 N/A 1.4E-11 2.6E+01 N/A          | 1.4E-11 2.6E+01 N/A 1.4E-11 2.6E+01 N/A          | 7.2E-12 2.6E+01 N/A 7.7E-12 2.6E+01 N/A          | N/A N/A N/A N/A          | 7.2E-12 2.6E+01 N/A 7.2E-12 2.6E+01 N/A          | 7.2E-12 2.6E+01 N/A 7.2E-12 2.6E+01 N/A          | N/A N/A N/A N/A          |
| F. ND. ANAD RANGE AFG RANGE LBAD RANGE NAAP        | FREQ FACTOR FREQ FACTOR FREQ FACTOR FREQ      |                                                                                                  | 21 3.1E-18 3.1E+01 N/A 3.1E-18 3.1E+01 N/A       | 21 3.1E-18 3.1E+01 N/A 3.1E-18 3.1E+01 N/A       | 22 N/A N/A N/A N/A       | 23 2.3E-11 2.6E+01 N/A N/A N/A       | 23 3.5E-11 2.6E+01 N/A N/A N/A       | 23 3.5E-11 2.6E+01 N/A N/A N/A       | 23 1.3E-11 2.6E+01 N/A N/A N/A       | 23 1.4E-11 2.6E+n1 N/A N/A N/A       | 23 1.4E-11 2.6E+01 N/A 1.4E-11 2.6E+01 N/A       | 23 1.4E-11 2.6E+01 N/A 1.4E-11 2.6E+01 N/A       | 23 7.2E-12 2.6E+01 N/A 7.7E-12 2.6E+01 N/A       | 23 N/A N/A N/A N/A       | 23 7.2E-12 2.6E+01 N/A 7.2E-12 2.6E+01 N/A       | 23 7.2E-12 2.6E+01 N/A 7.2E-12 2.6E+01 N/A       | 24 N/A N/A N/A N/A       |
| SCEN- OF, ND, ANAD RANGE AFG RANGE LOAD RANGE MAAP | AKIO FREG FACTOR FREG FACTOR FREG FACTOR FREG |                                                                                                  | HCKGC 21 3.1E-18 3.1E+01 N/A 3.1E-18 3.1E+01 N/A | HCRVC 21 3.1E-18 3.1E+01 N/A 3.1E-18 3.1E+01 N/A | HCCHC 22 N/A N/A N/A N/A | HCEGE 22 N/A N/A N/A N/A | HCCHC 22 N/A N/A N/A N/A | HCNVC 22 N/A N/A N/A N/A | HCFGC 22 N/A N/A N/A N/A | HCFHC 22 N/A N/A N/A N/A | HGEVC 22 N/A N/A N/A N/A | HCBGC 22 N/A N/A N/A N/A | HEQVC 22 N/A N/A N/A N/A | HERGE 22 N/A N/A N/A N/A | HCKVC 22 N/A N/A N/A N/A | HCDHC 23 2.3E-11 2.6E+01 N/A N/A N/A | HCCGC 23 3.5E-11 2.6E+01 N/A N/A N/A | HCCHC 23 3.5E-11 2.6E+01 N/A N/A N/A | HCHVC 23 1.3E-11 2.6E+01 N/A N/A N/A | HCF6C 23 1.4E-11 2.6E+01 N/A N/A N/A | HCFHC 23 1.4E-11 2.6E+01 N/A 1.4E-11 2.6E+01 N/A | HCFVC 23 1.4E-11 2.6E+01 N/A 1.4E-11 2.6E+01 N/A | HCCGC 23 7.2E-12 2.6E+01 N/A 7.2E-12 2.6E+01 N/A | HCOVC 23 N/A N/A N/A N/A | HGRGC 23 7.2E-12 2.6E+01 N/A 7.2E-12 2.6E+01 N/A | HCRVC 23 7.2E-12 2.6E+01 N/A 7.2E-12 2.6E+01 N/A | HCDHC 24 N/A N/A N/A N/A | HCCGC 24 N/A N/A N/A N/A | HECHC 24 N/A N/A N/A N/A | HCMVC 24 N/A N/A N/A N/A | HCF6C 24 N/A N/A N/A N/A |

2.55.55

File: HDLNAT.NK1, Page 8 Date 18-Aug-87

TABLE 6-4 (Continued)

HANDLING ACCIDENTS - MATIONAL PROCESSING OPTION - PER PALLET OR CONTAINER

Service of

Accident Frequencies and Range Factors

| SCEN-  | OF. NO. | ANAD<br>Freq | RANGE<br>FACTOR | AF6<br>Freq | RANGE<br>Factor | L BAD<br>Freq | RANGE<br>Factor | NAAP<br>Freq | RANGE<br>Factor | PBA<br>Freq | RANGE<br>Factor | FUDA<br>Freq | RANGE<br>Factor | TEAD<br>Fred     | RANGE<br>Factor | UMDA<br>Freq | RANGE<br>Factor |
|--------|---------|--------------|-----------------|-------------|-----------------|---------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|------------------|-----------------|--------------|-----------------|
|        | ;       |              |                 |             | 1               |               |                 |              |                 |             |                 |              |                 |                  |                 | 5            |                 |
| HCFHC  | 24      | N/A .        | ;               | N/A         | ł               | N/A           | 1               | N/A          | :               | N/A         | ł               | N/A          | 1               | 1.0E-11          | 2.6E+01         | N/A          | ;               |
| HCPVC  | ŧ.      | N/A          | :               | N/A         | ;               | N/A           | ;               | N/A          | ł               | N/A         | :               | N/A          | ;               | 1.0E-11          | 2.6E+01         | N/A          | ;               |
| HCQGC  | 24      | N/A          | ł               | N/A         | ł               | N/A           | ł               | N/A          | ł               | N/A         | 1               | N/A          | ;               | 7.7E-12          | 2.6E+01         | N/A          | !               |
| HCQVC  | 24      | N/A          | 1               | N/A         | ł               | N/A           | ;               | N/A          | ;               | N/A         | 1               | N/A          | 1               | 7.7E-12          | 2.6E+01         | N/A          | ;               |
| нскес  | 24      | N/A          | ;               | N/A         | ;               | N/A           | 1               | N/A          | ł               | N/A         | 1               | A/A          | ł               | 11-36-11         | 2.6E+01         | N/A          | ;               |
| HCRVC  | 24      | N/A          | 1               | N/A         | ł               | N/A           | ;               | N/A          | ;               | N/A         | 1               | N/A          | ł               | 1.96-11          | 2.6E+01         | N/A          | ł               |
| HCDHC  | 25      | 1.7E-11      | 2.6E+01         | A/A         | :               | N/A           | ;               | N/A          | ;               | N/A         | 1               | 1.7E-11      | 2.6E+01         | 1.7E-11          | 2.6E+01         | N/A          | ;               |
| HCCGC  | 25      | 2.5E-11      | 2.6E+01         | N/A         | 1               | N/A           | ;               | N/A          | :               | N/A         | ł               | N/A          | 1               | 2.56-11          | 2.6E+01         | N/A          | ;               |
| HCCHC  | 25      | 2.5E-11      | 2. 6E+01        | N/A         | ;               | N/A           | 1               | N/A          | į               | N/A         | ł               | 2.5E-11      | 2.6E+01         | 2.5E-11          | 2.6E+01         | N/A          | ;               |
| HCMVC  | 25      | 9.3E-12      | 2.6E+01         | N/A         | ;               | N/A           | ł               | N/A          | ł               | 9.36-12     | 2.6E+01         | N/A          | 1               | 9.3E-12          | 2.6E+01         | 9.36-12      | 2.6E+01         |
| HCF6C  | 5       | 1.06-11      | 2.6E+01         | N/A         | I<br>T          | N/A           | ł               | N/A          | ;               | N/A         | ł               | N/A          | ł               | 11-3ÿ <b>*</b> 1 | 2.6E+01         | 1. (E-11     | 2.6E+01         |
| HCFHC  | 25      | 1.0E-11      | 2.6E+01         | N/A         | ;               | 1.06-11       | 2.6E+01         | N/A          | ;               | N/A         | ľ               | 1.0E-11      | 2.6E+01         | 1.0E-11          | 2.6E+01         | N/A          | ł               |
| HCPVC  | ដ       | 1.0E-11      | 2.6E+01         | N/A         | ł               | 1.0E-11       | 2.6E+01         | N/A          | 1               | N/A         | 1               | N/A          | 1               | 11-30-11         | 2.6E+01         | 1.05-11      | 2.6E+01         |
| HCQGC  | 5       | 5.2E-12      | 2.6E+01         | N/A         | ł               | 5.2E-12       | 2.6E+01         | N/A          | !               | N/A         | ;               | N/A          | ł               | 5.2E-12          | 2.6E+Ù1         | 5.2E-12      | 2.6E+01         |
| HCGVC  | 25      | N/A          | ;               | N/A         | ł               | N/A           | ;               | N/A          | 1               | N/A         | ł               | N/A          | ;               | 5.26-12          | 2.6E+01         | 5.2E-12      | 2.6E+01         |
| HCRGC  | 5       | 5.2E-12      | 2.6E+01         | N/A         | ł               | 5.2E-12       | 2.6E+01         | N/A          | 1               | 5.2E-12     | 2.6E+01         | N/A          | ł               | 5.2E-12          | 2.6E+01         | 5.2E-12      | 2.6E+01         |
| HCRVC  | 22      | 5.26-12      | 2.6E+01         | N/A         | ;               | 5.2E-12       | 2.6E+01         | N/A          | ;               | 5.2E-12     | 2.6E+01         | N/A          | 1               | 5.2E-12          | 2.6E+01         | 5.2E-12      | 2.6E+01         |
| HCRGF  | 26      | N/A          | ;               | N/A         | 1               | N/A           | !               | N/A          | ł               | N/A         | 1               | N/A          | ł               | 0.0E+00          | ;               | N/A          | 1               |
| HCDHC  | 56      | N/A          | 1               | N/A         | ;               | N/A           | ł               | N/A          | ;               | N/A         | ł               | N/A          | ł               | ()° ()E + ()()   | :               | M/A          | 1               |
| JUCCEC | 56      | N/A          | :               | N/A         | ;               | N/A           | ;               | N/A          | 1               | N/A         | ł               | N/A          | ;               | 0.0E+00          | ł               | N / A        | ł               |
| HCCHC  | 56      | N/A          | ;               | N/A         | :               | N/A           | ;               | N/A          | :               | N/A         | ł               | N/A          | :               | 0.01400          | :               | N/A          | ,               |
| HCKGF  | 26      | N/A          | ;               | N/A         | ;               | N/A           | 1               | N/A          | 1               | N/A         | ;               | N/A          | ;               | 0.0E+00          | ;               | N/A          | 1               |
| HC, HF | 26      | N/A          | :               | N/A         | :               | N/A           | :               | N/A          | ;               | N/A         | :               | N/A          | :               | 0.0E+00          | :               | N/A          | :               |
| HCKVF  | 26      | N/A          | 1               | N/A         | :               | N/A           | 1               | N/A          | ł               | N/A         | ;               | N/A          | :               | 0.0E+00          | ;               | N/A          | ł               |
| HCHVC  | 26      | N/A          | :               | N/A         | ;               | N/A           | ;               | N/A          | :               | N/A         | ;               | N/A          | :               | 0.0E+00          | ;               | N/A          | :               |
| HCFGC  | 26      | N/A          | 1               | N/A         | ;               | N/A           | ;               | N/A          | 1               | N/A         | ł               | N/A          | ł               | 0.0E+00          | ł               | N/A          | ;               |
| HCFHC  | 26      | N/A          | 1               | N/A         | ;               | N/N           | ;               | N/A          | !               | N/A         | ł               | N/A          | 1               | 0.05+00          | 1               | N/A          | !               |
| HCFVC  | 26      | N/A          | ł               | N/A         | !               | N/A           | ł               | N/A          | 1               | N/A         | ł               | A/A          | 1               | 0.0E+00          | 1               | N/A          | ;               |
| HCQGC  | 26      | N/A          | ;               | N/A         | ;               | R/A           | ;               | N/A          | :               | N/A         | 1               | N/A          | :               | 0°.0E+00         | ;               | N/A          | ;               |

くいい

25.25







File: HDLNAI.Nkl, Page 9 Date 21-Aug-87

TABLE 6-4 (Continued)

HAMDLING ACCIDENTS - NATIONAL PROCESSING OPTION - PER FALLET OR CONTAINER

Accident Frequencies and Range Factors

| F.A                                                | FACTO                                           | l                                                                                           | ;                  | ł                  | ł                  | 1                  | i                  | 1                      | {                      | ;                      | 1                    | {                          | ł                        | ;                        | ſ                        | ;                            | ;                            | 1                            | ;                    | 1                          | :                            | :                    | 2.65+0                                     | 2.6E+0                                       | 2.6[+3                                       | 2.4E+0                                       | 2.6E+0                                       | 2.6E+0                                       | 1.0040               | 0+3ú*1                                       | 1.0E+0                                       |
|----------------------------------------------------|-------------------------------------------------|---------------------------------------------------------------------------------------------|--------------------|--------------------|--------------------|--------------------|--------------------|------------------------|------------------------|------------------------|----------------------|----------------------------|--------------------------|--------------------------|--------------------------|------------------------------|------------------------------|------------------------------|----------------------|----------------------------|------------------------------|----------------------|--------------------------------------------|----------------------------------------------|----------------------------------------------|----------------------------------------------|----------------------------------------------|----------------------------------------------|----------------------|----------------------------------------------|----------------------------------------------|
| UMDA                                               | FREQ                                            |                                                                                             | N/A                | N/A                | N/A                | N/A                | 0.0E+00            | N/A                    | N/A                    | N/A                    | A/A                  | 0.0E+90                    | R/A                      | 0.06+00                  | 0.0E+00                  | R/A                          | 0.0E+00                      | 0.0140.0                     | 0.0E+00              | 0.0E+00                    | 0.0E+00                      | 0.0E+00              | 9.05-09                                    | 9.0E-U9                                      | 1.85-09                                      | 1.85-08                                      | 1.36-10                                      | 1.3E-10                                      | 1.0E-03              | 1.0E-03                                      | 1.05-03                                      |
| ANGE                                               | ACTOR                                           |                                                                                             | )<br>1             | ł                  | ;                  | ł                  | ł                  | ;                      | ;                      | :                      | ;                    | ł                          | ł                        | ł                        | ł                        | ;                            | :                            | ;                            | ;                    | ;                          | !                            | ł                    | 2.6E+01                                    | 2.6E+01                                      | 2.6E+01                                      | 2.6E+01                                      | 2.6E+01                                      | 2.6E+01                                      | 1.0E+Ù1              | 1.0E+01                                      | 1.0E+01                                      |
| TEAD                                               | FREG                                            |                                                                                             | 0.0E+00            | 0.0E+00            | 0.0E+00            | 0.05+00            | 0.0E+90.0          | 0.0€+00                | 0.0E+00                | 0.0E+00                | 0.0E+0)              | 0.0E+00                    | 0.0E+00                  | 0.0E+00                  | 0.0E+00                  | 0.0E+00                      | 0.0E+00                      | 00+30.0                      | 0.01+30.0            | 0.0E+00                    | 0.0E+00                      | 0.0E+00              | 9.05-03                                    | 9.0E-09                                      | 1.85-08                                      | 1.36-09                                      | 1.35-10                                      | 1.JE-10                                      | 1.0E-03              | 1.CE-03                                      | 1.0E-03                                      |
| RANGE                                              | FACTOR                                          |                                                                                             | ;                  | ;                  | ;                  | :                  | :                  | 1                      | ł                      | ;                      | ;                    | ł                          | ;                        | ł                        | ;                        | ;                            | ;                            | ;                            | ;                    | ;                          | ;                            | ł                    | ;                                          | ;                                            | ;                                            | ;                                            | ;                                            | ;                                            | ;                    | ;                                            | 1                                            |
| FUDA                                               | FREG                                            |                                                                                             | N/A                | N/A                | N/A                | N/A                | N/A                | 0.0E+00                | N/A                    | 0.0£+00                | N/A                  | N/A                        | A/A                      | N/A                      | N/A                      | 0.0E+00                      | N/A                          | N/A                          | N/A                  | N/A                        | N/A                          | N/A                  | N/A                                        | N/A                                          | N/A                                          | N/A                                          | N/A                                          | N/A                                          | N/A                  | N/A                                          | A/A                                          |
| RANGE                                              | FACTOR                                          |                                                                                             | ;                  | ;                  | ļ                  | ;                  | ;                  | ;                      | 1                      | ;                      | ;                    | }                          | ;                        | ţ                        | ;                        | ;                            | 1                            | ł                            | ;                    | :                          | ł                            | ł                    | 2.6E+01                                    | 2.6E+01                                      | 2.6E+01                                      | 2.6E+01                                      | 2.6E+01                                      | 2.6E+(i]                                     | :                    | 1.0E+01                                      | 1.0E+01                                      |
| PBA                                                | FREG                                            |                                                                                             | N/A                | N/A                | N/A                | N/A                | N/A                | N/A                    | N/A                    | RYA                    | N/A                  | 0.0E+00                    | N/A                      | 0.0E+00                  | N/A                      | N/A                          | N/A                          | N/A                          | N/A                  | 0.0E+00                    | 0.0E+00                      | N/A                  | 9.0E-09                                    | 9.0E-09                                      | 1.86-08                                      | 1.86-08                                      | 1.3E-10                                      | 1.3E-10                                      | N/A                  | 1.0E-03                                      | 1.0E-03                                      |
|                                                    |                                                 | •                                                                                           |                    |                    |                    |                    |                    |                        |                        |                        |                      |                            | ,                        |                          |                          |                              | ,                            | 1                            | :                    | :                          | ł                            | !                    | :                                          |                                              |                                              | ,                                            | ,                                            |                                              | •                    | ŗ                                            | •                                            |
| RANGE                                              | FACTOR                                          |                                                                                             | 1                  | 1                  | i                  | 1                  | i                  | ;                      | i                      | i                      | ł                    | i                          | '                        | •                        | •                        | •                            | •                            | •                            | ·                    | •                          | •                            | •                    | •                                          | •                                            | •                                            | 1                                            | •                                            | 1                                            | •                    | •                                            | '                                            |
| NAAP RANGE                                         | FREQ FACTOR                                     |                                                                                             | A/N                | N/A                | N/A                | N/A                | N/A                | N/A                    | A/M                    | N/A -                  | - N/A                | N/A -                      | 0.0E+00 -                | N/A -                    | - N/A -                  | N/A -                        | N/A -                        | N/A -                        | N/A                  | N/A                        | N/A                          | - W/A                | N/A                                        | N/A -                                        | N/A -                                        | - 4/N                                        | N/A -                                        | - A/N                                        | N/A -                | N/A -                                        | N/A -                                        |
| RANGE NAAP RANGE                                   | FACTOR FREQ FACTOR                              | 9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9 | N/A                | N/A                | N/A                | N/A                | N/A                | N/A                    | W/A                    | N/A                    | N/A                  | N/A                        | - 0.0E+00                | N/A                      | N/A                      | N/A                          | N/A                          | N/A                          | N/A                  | N/A                        | N/A                          | - N/N                | 2.6E+01 N/A -                              | 2.6E+01 N/A -                                | 2.6E+01 N/A -                                | 2.6E+01 N/A -                                | 2.6E+01 N/A -                                | 2.6E+01 N/A -                                | N/A                  | 1.0E+01 N/A -                                | 1.0E+01 N/A -                                |
| LBAD RANGE NAAP RANGE                              | FREQ FACTOR FREQ FACTOR                         | 9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9<br>9 | N/A N/A            | N/A N/A            | N/A N/A            | N/A A/A            | N/A N/A            | N/A N/A                | N/A N/A                | N/A N/A                | - N/A N/A            | N/A A/A                    | N/A 0.0E+00 -            | N/A N/A -                | N/A N/A -                | 0.0E+00 N/A -                | 0.0E+00 N/A -                | 0.0E+00 N/A -                | N/A N/A -            | 0.0E+00 N/A                | 0.0E+00 N/A -                | M/A M/A -            | 9.0E-09 2.6E+01 N/A                        | 7.0E-09 2.6E+01 N/A -                        | 1.8E-08 2.6E+01 N/A -                        | 1.8E-08 2.6E+01 N/A -                        | 1.3E-10 2.6E+01 N/A -                        | 1.3E-10 2.6E+01 N/A -                        | N/A N/A              | 1.0E-03 1.0E+01 N/A -                        | 1.0E-03 1.0E+01 N/A -                        |
| RANGE LBAD RANGE NAAP RANGE                        | FACTOR FREQ FACTOR FREQ FACTOR                  |                                                                                             | N/A N/A            | N/A N/G            | N/A N/A            | N/A N/A            | N/A N/A            | N/A N/A                | N/A N/A                | N/A N/A                | N/A N/A              | N/A N/A                    | N/A 0.0E+00 -            | N/A N/A -                | N/A N/A                  | 0.0E+00 N/A -                | 0.0E+00 N/A -                | 0.0E+00 N/A -                | N/A N/A              | 0,0E+00 N/A                | 0.0E+00 N/A -                | N/A N/A              | 9.0E-09 2.6E+01 N/A                        | 7.0E-09 2.6E+01 N/A -                        | 1.8E-08 2.6E+01 N/A                          | 1.8E-08 2.6E+01 N/A -                        | 1.3E-10 2.6E+01 N/A -                        | 1.3E-10 2.6E+01 N/A -                        | N/A N/A              | 1.0E-03 1.0E+01 M/A -                        | 1.0E-03 1.0E+01 N/A -                        |
| AFG RANGE LBAD RANGE NAAP RANGE                    | FREQ FACTOR FREQ FACTOR FREQ FACTOR             |                                                                                             | N/A N/A N/A        | N/A N/A N/A        | N/A N/A N/A        | W/A W/A W/A        | N/A N/A N/A        | N/A N/A N/A            | 11/A N/A N/A           | N/A N/A N/A            | N/A N/A N/A          | 0.0E+00 N/A N/A            | N/A N/A 0.0E+00 -        | N/A N/A N/A -            | - N/A N/A N/A            | N/A 0.0E+00 N/A -            | N/A 0.0E+00 N/A -            | N/A 0.0E+00 N/A -            | N/A N/A N/A -        | N/A 0.0E+00 N/A            | N/A 0.0E+00 N/A -            | N/A N/A N/A -        | N/A 9.0E-09 2.6E+01 N/A                    | N/A 7.0E-09 2.6E+01 N/A -                    | N/A 1.8E-08 2.6E+01 N/A -                    | N/A 1.8E-08 2.6E+01 N/A -                    | N/A 1.3E-10 2.6E+01 N/A -                    | N/A 1.3E-10 2.6E+01 N/A -                    | N/A N/A N/A          | N/A 1.0E-03 1.0E+01 N/A -                    | N/A 1.0E-03 1.0E+01 N/A -                    |
| RANGE AFG RANGE LBAD RANGE NAAP RANGE              | FACTOR FREQ FACTOR FREQ FACTOR FREQ FACTOR      |                                                                                             | N/A N/A N/A            | N/A N/A N/A            | N/A N/A N/A            | N/A N/A N/A          | 0,0E+00 N/A N/A            | N/A N/A 0.0E+00 -        | N/A N/A N/A              | N/A N/A                  | N/A 0.0E+00 N/A -            | N/A 0.0E+00 N/A -            | N/A 0.0E+00 N/A -            | N/A N/A N/A -        | N/A 0.0E+00 N/A            | N/A 0.0E+00 N/A -            | N/A N/A N/A          | 2.6E+01 N/A 9.0E-09 2.6E+01 N/A            | 2.6E+01 N/A 7.0E-09 2.6E+01 N/A -            | 2.6E+01 N/A 1.8E-08 2.6E+01 N/A -            | 2.6E+01 N/A 1.8E-08 2.6E+01 N/A -            | 2.66+01 N/A 1.3E-10 2.6E+01 N/A -            | 2.6E+01 N/A 1.3E-10 2.6E+01 N/A -            | N/A N/A N/A          | 1.0E+01 N/A 1.0E-03 1.0E+01 N/A -            | 1.0E+01 N/A 1.0E-03 1.0E+01 N/A -            |
| ANAD RANGE AFG RANGE LBAD RANGE NAAP RANGE         | FREO FACTOR FREQ FACTOR FREQ FACTOR FREQ FACTOR |                                                                                             | N/A N/A N/A N/A    | 0.0E+00 N/A N/A N/A    | 0.0E+0i) N/A N/A N/A   | 0.0E+00 N/A N/A N/A    | N/A N/A N/A N/A      | 0.0E+00 0.0E+00 N/A N/A    | N/A N/A N/A 0.0E+00 -    | 0.0E+00 N/A N/A N/A -    | 0.0E+00 N/A N/A N/A -    | 0.9E+00 N/A 0.0E+00 N/A -    | 0.0E+00 N/A 0.0E+00 N/A -    | 0.0E+00 N/A 0.0E+00 N/A -    | N/A N/A N/A N/A -    | 0.0E+00 N/A 0.0E+00 N/A    | 0.0E+00 N/A 0.0E+00 N/A -    | N/A N/A N/A N/A -    | 9.0E-07 2.6E+01 N/A 9.0E-09 2.6E+01 N/A    | 7.0E-09 2.6E+01 N/A 7.0E-09 2.6E+01 N/A -    | 1.86-08 2.66+01 N/A 1.86-08 2.66+01 N/A -    | 1.8E-08 2.6E+01 N/A 1.8E-08 2.6E+01 N/A -    | 1.3E-10 2.6E+01 N/A 1.3E-10 2.6E+01 N/A -    | 1.3E-10 2.6E+01 N/A 1.3E-10 2.6E+01 N/A -    | N/A N/A N/A N/A -    | 1.0E-03 1.0E+01 N/A 1.0E-03 1.0E+01 N/A -    | 1.0E-03 1.0E+01 N/A 1.0E-03 1.0E+01 N/A -    |
| JP. NO. ANAD RANGE AFG RANGE LBAD RANGE NAAP RANGE | FREO FACTOR FREQ FACTOR FREO FACTOR             |                                                                                             | 26 N/A N/A N/A N/A | 27 N/A N/A N/A N/A | 27 0.0E+00 N/A N/A N/A | 27 0.0E+00 N/A N/A N/A | 27 0.0E+00 N/A N/A N/A | 2? N/A N/A N/A N/A - | 27 0.0E+00 0.0E+00 N/A N/A | 27 N/A N/A N/A 0.0E+00 - | 27 0.0E+00 N/A N/A N/A - | 27 0.0E+00 N/A N/A N/A - | 27 0.9E+00 N/A 0.0E+00 N/A - | 27 0.0E+00 N/A 0.0E+00 N/A - | 27 0.0E+00 N/A 0.0E+00 N/A - | 27 N/A N/A N/A N/A - | 27 0.0E+00 N/A 0.0E+00 N/A | 27 0.0E+00 N/A 0.0E+00 N/A - | 27 N/A N/A N/A N/A - | 29 9.0E-07 2.6E+01 N/A 9.0E-09 2.6E+01 N/A | 29 7.0E-09 2.6E+01 N/A 7.0E-09 2.6E+01 N/A - | 30 1.8E-08 2.6E+01 N/A 1.8E-08 2.6E+01 N/A - | 30 1.8E-08 2.6E+01 N/A 1.8E-08 2.6E+01 N/A - | 31 1.3E-10 2.6E+01 N/A 1.3E-10 2.6E+01 N/A - | 31 1.3E-10 2.6E+01 N/A 1.3E-10 2.6E+01 N/A - | 32 N/A N/A N/A N/A - | 32 1.0E-03 1.0E+01 N/A 1.0E-03 1.0E+01 N/A - | 32 1.0E-03 1.0E+01 N/A 1.0E-03 1.0E+01 N/A - |

፝ፚኇፚ፟ኯፚዹዄዀዀዀዀ ፟

File: HFNAT.WFL Fage 1 Date 20 Aug-87

TABLE 6-4 (Continued)

NATIONAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Handling Operations (HF) (Events per Pallet or Container)

|             | SCENAR IO<br>NUMBER | UENH<br>FFEQ | FANGE<br>FALTOR | AF6<br>FREQ | FANGE<br>FACTOR | LRAD<br>Freq | KARGE<br>FACTOR | FREQ | FACTOR | FREQ | RANGE<br>Factor | FREQ | KANGE<br>Factor | TEAD<br>Freq            | RANGE<br>Factor | FRED | FACTOR |
|-------------|---------------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|------|--------|------|-----------------|------|-----------------|-------------------------|-----------------|------|--------|
|             |                     |              |                 |             |                 |              |                 |      |        |      |                 |      |                 |                         |                 |      |        |
| 204         |                     | N/A          | 1               | A . N       | 1               | N/A          | 1               | N/A  | 1      | N/A  | ;               | N/A  | 1               | 4.2E-09                 | 1.36+01         | N/A  | ;      |
| 2H3         | -                   | N, A         | ;               | N/A         | :               | N/A          | ;               | N/A  | ł      | N/A  | !               | A/A  | 1               | 4.2E-09                 | 1.36+01         | N/A  | !      |
| 540         |                     | N. A         | ļ               | NrA         | ;               | N/A          | ;               | N/A  | 1      | N/A  | ;               | N/A  | :               | 4.85-10                 | 1.35+01         | N/A  | -      |
| (HS         | -                   | N/A          | :               | A/A         | :               | N/A          | :               | N/A  | :      | N/Å  | ł               | N/A  | :               | 4.85-10                 | 1.3E+01         | N/A  | !      |
| <b>1.65</b> |                     | N/A          | ;               | N/A         | :               | N/A          | ł               | N/A  | :      | N/A  | :               | N/A  | :               | 8.4E-()9                | 1.36+01         | N/A  | :      |
| SH I        |                     | N/A          | :               | A/A         | :               | N/A          | :               | N/A  | ł      | N/A  | 1               | N/A  | ;               | 8.6E-19                 | 1.35+01         | N/A  | ł      |
| 511         | -                   | N/A          | ;               | N/A         | ;               | N/A          | ł               | N/A  | :      | N/A  | ;               | N/A  | :               | 8.6E-09                 | 1.3E+01         | N/A  | ł      |
| H'S         | -                   | N.A          | :               | N/N         | 1               | N/A          | ;               | N/A  | 1      | N/A  | ;               | N/A  | ł               | 5.8E-09                 | 1.35+01         | N'A  | 1      |
| FGS         | -                   | N/A          | ;               | N/A         | :               | N/A          | :               | N/A  | :      | N/A  | ł               | N/A  | 1               | 7.26-10                 | 1.3E+01         | N/À  | ;      |
| FHS         |                     | N/A          | :               | N/A         | :               | N/A          | :               | N/A  | ;      | N/A  | 1               | N/A  | ł               | 7.2E-10                 | 1.3E+01         | N/A  | !      |
| FV5         |                     | N/A          | ;               | N/A         | ;               | N/A          | ;               | N/A  | 1      | N/A  | 1               | N/A  | 1               | 7.2E-10                 | 1.35+01         | N/A  | !      |
| 598.        |                     | N / A        | :               | N/A         | ;               | NA           | 1               | N/A  | 1      | N/A  | ł               | N/A  | ł               | 7.26-10                 | 1.35+01         | N/A  | ł      |
| 518         | 1                   | N, A         | ;               | N/A         | ;               | N/A          | ļ               | N/A  | 1      | N/A  | ł               | N/A  | 1               | 7.26-10                 | 1.36+01         | N/A  | 1      |
| <b>Fi5S</b> | -                   | A.N          | ;               | N/A         | 1               | N/A          | ł               | N/A  | ł      | N/A  | 1               | N/A  | ł               | 3.2E-09                 | 1. JE+01        | N, A | :      |
| FIVS        | -                   | N/A          | ;               | N/A         | ;               | N/A          | :               | N/A  | ;      | N/A  | ł               | N/A  | ;               | 3.2E-09                 | 1.35+01         | N/A  | 1      |
| 545         |                     | N/A          | ;               | N/A         | ł               | N/A          | ł               | N/A  | :      | N/A  | 1               | N/A  | ł               | 1.0E-06                 | 1.36+01         | N/A  | !      |
| PGC         | 2                   | N/A          | ;               | N/A         | ;               | N/A          | 1               | N/A  | ;      | N/A  | ł               | N/A  | :               | 1.06-16                 | 3.16+01         | N/A  | :      |
| DHC         | 7                   | N/A          | 1               | N/A         | :               | N/A          | ł               | A/N  | 1      | N/A  | !               | N/A  | 1               | 0.01400                 |                 | N/A  | :      |
| C6C         | 2                   | N/A          | ;               | N/A         | ;               | N/A          | 1               | N/A  | ł      | N/A  | 1               | N/A  | :               | 0.0E+00                 |                 | N/A  | 1      |
| CHC         | 2                   | N/A          | ;               | N/A         | 1               | N/A          | ł               | N/A  | ł      | N /  | 1               | N/A  | 1               | 0.0E+00                 |                 | N/A  | 1      |
| r 6C        | 2                   | N/A          | ;               | N/A         | !               | N/A          | ł               | N/A  | 1      | N/4  | ł               | N/A  | ł               | B.4E-16                 | 3. 16+01        | N/A  | 1      |
| ) HC        | 2                   | N/A          | ;               | N/A         | ;               | N/A          | 1               | N/A  | :      | N/A  | ;               | N/A  | ł               | 8.4E-16                 | 3.16+01         | N/A  | 1      |
| ) N I       | 2                   | NIA          | 1               | N/A         | ;               | N/A          | ł               | N/A  | ł      | A/N  | 1               | N/A  | !               | 8.4E-16                 | 3.16+01         | N/A  | ;      |
| AVC         | 7                   | N/A          | ;               | N/A         | ;               | N/A          | !               | N/A  | ł      | N/A  | ł               | N/A  | !               | 2.6E-17                 | 3. 16+01        | N/A  | ł      |
| PGC         | 2                   | A/A          | ;               | N/A         | ;               | N/A          | ;               | N/A  | 1      | N/A  | 1               | N/A  | :               | 0°.0E+00                |                 | N/A  | 1      |
| FHC         | ~                   | N/A          | ;               | N/A         | :               | N/A          | ł               | N/A  | ł      | N/A  | ;               | N/A  | ł               | 0,0E+00                 |                 | N A  | ;      |
| PVC         | 2                   | N/A          | ;               | N/A         | ł               | N/A          | 1               | N/A  | ;      | N/A  | 1               | N/A  | !               | () <b>• ()E + (</b> )() |                 | N/A  | ł      |
| 965         | 7                   | N/A          | ;               | N/A         | ;               | N/A          | :               | N/A  | 1      | N/A  | :               | N/A  | ł               | 0.05+00                 |                 | N.A  | ;      |
| 200         | ſ                   |              |                 | AL M        |                 |              |                 |      |        |      |                 |      |                 |                         |                 |      |        |

يستعطينا والمستعملين المشرقين فستعط

200005555554 (MILLOUGHORDS) 18850 2520-271 (MILLOUGHORDS)

للالماليات

ፚ፝ኇፚ፟ኯዄ፟ዄዄ፝ኯፚ፝ኯፚ፟ኯዄጜዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀ







File: HFNAT.WK1 Page 2 Date 20-Aug-87

TABLE 6-4 (Continued)

## NATIONAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Handling Operations (HF) (Events per Fallet or Container)

|        | SCENAR 10 | <b>ANAD</b> | RANGE  | AP6  | RANGE                      | LEAD | RANGE  | NAAP | RANGE  | FBA  | RANGE  | PUDA | RANGE                      | TEAD      | RANGE    | MDA  | <b>KANGE</b> |
|--------|-----------|-------------|--------|------|----------------------------|------|--------|------|--------|------|--------|------|----------------------------|-----------|----------|------|--------------|
|        | NUMBER    | FREQ        | FAC10R | FREQ | FACTOR                     | FRED | FACTOR | FREQ | FACTOR | FREQ | FACTOR | FREQ | FACTOR                     | FRED      | FAC TOR  | FREQ | FACTOR       |
|        |           |             | 8      |      | •<br>•<br>•<br>•<br>•<br>• |      |        |      |        |      |        |      | 5<br>1<br>1<br>5<br>6<br>6 |           |          |      |              |
| HFRGC  | 2         | N/A         | ;      | N/A  | ;                          | N/A  | ١      | N/A  | ;      | N/A  | :      | N/A  | ;                          | 1.86-16   | 3.16+01  | N/A  | 1            |
| HFRVC  | 2         | N/A         | ;      | N/A  | 1                          | N/A  | ;      | N/A  | ;      | N/A  | ł      | N/A  | ł                          | 1.86-16   | 3. IE+01 | N/A  | ł            |
| HFSVC  | 2         | N/A         | ł      | N/A  | ł                          | N/A  | ł      | A/A  | ;      | N/A  | 1      | N/A  | 1                          | 4.56-15   | 3.1E+01  | N/A  | ţ            |
| HF B6F | £         | N/A         | 1      | N/A  | ;                          | N/A  | 1      | N/A  | ;      | N/A  | ;      | N/A  | :                          | 3. IE-11  | 3. IE+01 | N/A  | 1            |
| HE DHF | £         | N/A         | ;      | N/A  | 1                          | N/A  | ł      | A/A  | ;      | N/A  | ł      | N/A  | ł                          | 9.4E-11   | 3.16+01  | N/A  | !            |
| HFC6F  | m         | N/A         | ;      | N/A  | :                          | N/A  | :      | N/A  | ł      | N/A  | ;      | N/A  | 1                          | 0.01100   |          | N/A  | !            |
| HFCHF  | •         | N/A         | ;      | N/A  | 1                          | N/A  | ļ      | N/A  | 1      | N/A  | ł      | N/A  | ł                          | 0.0E+00   |          | A'N  | 1            |
| HFK6F  | 2         | N/A         | ł      | N/A  | :                          | N/A  | 1      | A/A  | ł      | N/A  | ł      | N/A  | ł                          | 1.0E-10   | 3. IE+01 | N/A  | 1            |
| HFKHF  | m         | N/A         | ;      | N/A  | 1                          | N/A  | ł      | N/A  | ł      | N/A  | ł      | N/A  | ł                          | 1.05-10   | 3.1E+01  | N/A  | 1            |
| HEKVE  | •         | N/A         | !      | N/A  | ;                          | N/A  | ;      | N/A  | ł      | N/A  | !      | N/A  | {                          | 1.0E-10   | 3. IE+01 | N/A  | ;            |
| HEAVE  |           | N/A         | 1      | N/A  | ł                          | N/A  | ł      | A/A  | ;      | N/A  | ł      | N/A  | ł                          | 1.4E-10   | 3.1E+01  | N/A  | 1            |
| HFPGF  | •••       | N/A         | 1      | N/A  | 1                          | N/A  | 1      | N/A  | ł      | N/A  | ;      | N/A  | 1                          | 0.0E+00   |          | N/A  | ł            |
| HFFHF  | ~         | N/A         | ł      | N/Å  | 1                          | N/A  | 1      | N/A  | ł      | N/A  | 1      | N/A  | 1                          | 0.0E+00   |          | N/A  | ł            |
| HEPVE  | •         | N/A         | 1      | N/A  | 1                          | N/A  | ;      | N/A  | ł      | N/A  | 1      | N/A  | 1                          | 0.0E+00   |          | N/A  | ł            |
| HFQ6F  | 2         | N/A         | 1      | N/A  | 1                          | N/A  | ł      | N/A  | ;      | N/A  | ł      | N/A  | ł                          | 0.0E+00   |          | NA   | 1            |
| HFQVF  | r         | N/A         | 1      | N/A  | 1                          | N/A  | ;      | N/A  | ł      | N/A  | 1      | N/A  | ł                          | 0.0E+00   |          | N/A  | 1            |
| HFRGF  | £         | N/A         | ł      | N/A  | ł                          | N/A  | 1      | N/A  | ;      | N/A  | ;      | N/A  | 1                          | 8. IE-11  | 3. IE+01 | N/A  | :            |
| HFRVF  | m         | N/A         | ł      | N/A  | 1                          | N/A  | ;      | N/A  | ;      | N/A  | :      | N/A  | !                          | 8.1E-11   | 3. IE+01 | N/A  | :            |
| HESVE  | 6.4       | N/A         | :      | N/A  | ł                          | N/A  | ł      | N/A  | 1      | N/A  | 1      | N/A  | ł                          | 2.0E-09   | 3. IE+01 | NIA  | ł            |
| HF SVS | 4         | N/A         | 1      | N/A  | 1                          | N/A  | ;      | N/A  | ;      | N/A  | ;      | N/A  | !                          | 3. IE-07  | 1.3E+01  | N/A  | ;            |
| HF BGF | ŝ         | N/A         | ;      | N/A  | 1                          | N/A  | ł      | N/A  | ;      | N/A  | :      | N/A  | ł                          | 0.0E+00   | _        | N/A  | :            |
| HF DHC | 0         | N/A         | ſ      | A'A  | 1                          | N/A  | 1      | N/A  | ;      | N/A  | ł      | N/A  | ;                          | 0°.0E+00  |          | N/A  | 1            |
| HFCGC  | ŝ         | N/A         | ł      | N/A  | l                          | N/A  | ;      | N/N  | !      | N/A  | 1      | N/A  | ł                          | 0,0E+00   |          | N/A  | :            |
| HFCHC  | ŝ         | N/A         | 1      | N/A  | ł                          | N/A  | ;      | N/A  | ;      | N/A  | ł      | N/A  | ł                          | 0.05+00   |          | N/A  | 1            |
| HFEGF  | 5         | A/A         | ;      | A/A  | ł                          | N/A  | ţ      | N/A  | ;      | N/A  | 1      | N/A  | ł                          | 0.0E+00   |          | N/A  | ł            |
| HF NHF | לש        | N/A         | ;      | N/A  | 1                          | N/A  | ł      | N/A  | ;      | N/A  | 1      | N/A  | ;                          | 0.05+00   |          | N/A  | ;            |
| HFLVF  | 5         | N/A         | 1      | N/A  | 1                          | N/A  | ł      | N/A  | ;      | N/A  | ł      | N/A  | !                          | 0.0E+00   |          | N/A  | !            |
| HFINC  | ŝ         | N/A         | ł      | N/A  | 1                          | N/A  | ;      | N/A  | ;      | N/A  | :      | N/A  | ;                          | 0.0€+00   |          | H/A  | 1            |
| HFPGC  | ŝ         | N/A         | ł      | N/A  | ł                          | N/A  | ;      | N/A  | ;      | N/A  | ł      | N/A  | ;                          | 0° 0E +00 |          | N/A  | :            |

File: HFWAI.Wk1 Fage 3 Date 20-Aug-B7

TABLE 6-4 (Continued)

NATIONAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Mandling Operations (MF) (Events per Fallet or Container)

|         | SCENAR10<br>NUMBER | ANAD<br>Freq | RANGE<br>Factor | AF6<br>Fred | RANGE<br>Factor | L&AD<br>Freq | RANGE<br>FACTOR | NAAP<br>Freq | RANGE<br>Factor | PBA<br>Fred | RANGE<br>FACTOR | FUDA<br>Freq | KANGE<br>FACTOR | TEAD<br>Freq | RANGE<br>FACTOR | UMDA<br>FREQ | RANGE<br>FACTOR |
|---------|--------------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|-----------------|
|         | 1                  |              |                 |             |                 |              |                 |              |                 |             |                 |              |                 |              |                 |              |                 |
| HEPHC   | 5                  | N/A          | ;               | N/A         | ١               | N/A          | !               | N/A          | ;               | N/A         | ;               | N/A          | ;               | 0.0E+00      |                 | N/A          | !               |
| HF FVC  | ŝ                  | N/A          | ;               | N/A         | :               | N/A          | 1               | N/A          | :               | N/A         | 1               | N/A          | ł               | 0.0E+00      |                 | N, A         | :               |
| HF QEC  | ŝ                  | N/A          | ;               | N/A         | ł               | N/A          | ł               | N/A          | 1               | N/A         | 1               | N/A          | :               | 0.0E+00      |                 | A/A          | ;               |
| HFOVC   | <b>i</b>           | N/A          | ;               | N/A         | ;               | N/A          | ;               | N/A          | !               | N/A         | ;               | N/A          | ł               | 0.0E+00      |                 | N/A          | :               |
| HFREC   | 5                  | N/A          | ł               | N/A         | 1               | N/A          | 1               | N/A          | 1               | N/A         | ł               | N/A          | ;               | 0.0E+00      |                 | N/A          | !               |
| HFEVC   | ŝ                  | N/A          | !               | N/A         | ł               | N/A          | ;               | N/A          | 1               | N/A         | ;               | N/A          | :               | 0.0E+00      |                 | N/A          | ł               |
| HF SVF  | S                  | N/A          | ł               | N/A         | 1               | N/A          | ł               | N/A          | ;               | N/A         | ł               | N/A          | :               | 0.0E+00      |                 | N/A          | ;               |
| HF R6S  | 1                  | A N          | ł               | N/A         | ;               | N/A          | :               | N/A          | ł               | N/A         | 1               | N/A          | :               | 8.3E-10      | 1.3E+01         | N/A          | 1               |
| SHQ 3H  | (                  | N/A          | 1               | N/A         | 1               | N/A          | ł               | N/A          | ;               | N/A         | ł               | N/A          | ;               | 2.5E-09      | 1.3E+01         | N/A          | ;               |
| HFCES   | 1                  | N/A          | ;               | N/A         | ł               | A/N          | ;               | N/A          | 1               | N/A         | ł               | N/A          | :               | 0.0E+00      |                 | N/A          | :               |
| SFCHS   | 1                  | N/A          | 1               | N/A         | ;               | N/A          | ł               | N/A          | :               | N/A         | 1               | N/A          | ;               | 0.0E+00      |                 | N/A          | :               |
| HFI.65  | 1                  | N/A          | ;               | N/A         | 1               | N/A          | :               | N/N          | ł               | N/A         | ł               | N/A          | 1               | 2.7E-09      | 1.3E+01         | N/A          | :               |
| SH 4 JH | 7                  | N/A          | :               | N/A         | 1               | N/A          | 1               | N/A          | 1               | N/A         | ł               | N/A          | ł               | 2.7E-09      | 1.3E+01         | N/A          | :               |
| SAAR    | r.                 | N/A          | ;               | N/A         | 1               | N/A          | ł               | N/A          | 1               | N/A         | ł               | N/A          | ;               | 2.7E-09      | 1.3E+01         | N/A          | :               |
| SAMAH   | 7                  | N/A          | :               | N/A         | 1               | N/A          | ł               | N/A          | ;               | A/A         | 1               | N/A          | ł               | 3.6E-()9     | 1.3E+01         | N/A          | :               |
| HFF6S   | 1                  | N/A          | :               | N/A         | ł               | N/A          | ł               | N/A          | 1               | N/A         | ł               | N/A          | ł               | 0.0E+00      |                 | N/A          | ;               |
| HFFHS   | 1                  | N/A          | ;               | N/A         | :               | N/A          | ;               | N/A          | 1               | N/A         | ł               | N/A          | :               | 0°*0E+00     |                 | N/A          | :               |
| HFFVS   | 1                  | N/A          | !               | N/N         | ;               | N/A          | ;               | N/A          | 1               | N/A         | ;               | N/A          | ł               | 0.0E+00      |                 | N/A          | :               |
| HF 865  | 7                  | N/A          | !               | N/A         | 1               | N/A          | ;               | N/A          | 1               | N/A         | 1               | N/A          | :               | 0.0E+00      |                 | N/A          | :               |
| HFBVS   | 1                  | N/A          | ł               | N/A         | !               | N/A          | 1               | N/A          | :               | N/A         | :               | N/A          | ;               | 0.0E+00      |                 | N/A          | :               |
| HFR6S   | 7                  | N/A          | ;               | N/A         | ł               | N/A          | ;               | N/A          | ł               | N/A         | :               | N/A          | !               | 2.2E-09      | 1.36+01         | N/A          | ;               |
| HFRVS   | 1                  | N/A          | :               | N/A         | 1               | N/A          | ł               | N/N          | ł               | N/A         | 1               | N/A          | ł               | 2.2E-09      | 1.36+01         | A'N          | :               |
| HFSVS   | 1                  | N/A          | ;               | N/A         | ;               | N/A          | 1               | A/A          | ;               | N/A         | ;               | N/A          | :               | 5.2E-08      | 1.36+01         | N/A          | :               |
| HF RGC  | 8                  | N/A          | :               | N/A         | 1               | N/A          | ;               | N/A          | :               | N/A         | 1               | N/A          | !               | 5.3E-18      | 3.1E+01         | N/A          | :               |
| HF DHC  | 80                 | N/A          | ;               | N/A         | ł               | N/A          | 1               | N/A          | :               | N/A         | 1               | N/A          | ;               | 5.36-18      | J. 1E+01        | N/A          | :               |
| HFCGC   | 80                 | N/A          | 1               | N/A         | ;               | N/A          | ;               | N/A          | :               | N/A         | :               | N/A          | :               | 61-30.6      | 3. IE+01        | N/A          | :               |
| HFCHC   | 8                  | N/A          | :               | N/N         | ;               | N/A          | 1               | N/N          | 1               | N/A         | ;               | N/A          | ;               | 6.0E-19      | 3.16+01         | N/A          | ł               |
| HF # 6C | 80                 | N/A          | :               | N/A         | ł               | N/A          | ·               | -            | :               | N/A         | 1               | N/A          | :               | 1.1E-17      | 3. IE+01        | e/N          | ;               |
| HEAHC   | 80                 | N/A          | 1               | N/A         | 1               | N/A          | ;               | N/A          | ;               | N/A         | ;               | N/A          | ;               | 1. IE-17     | 3. IE+01        | N/A          | ł               |

•

Contraction of

File: HFNAT.WK1 Fage 4 Date 20-Aug-87

TABLE 6-4 (Continued)

NATIONAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Handling Operations (HF) (Events per Pallet or Container)

| range<br>Factor    |      | ;          | 1       | :          | !       | :       | ;          | ł          | ł        | :        | 1       | ł          | !        | ;       | ť       | 1          | :       | 1       | :       | í          | ſ       | í      | ł        | ;       | 1       | ;          | :         | ;          | í       |  |
|--------------------|------|------------|---------|------------|---------|---------|------------|------------|----------|----------|---------|------------|----------|---------|---------|------------|---------|---------|---------|------------|---------|--------|----------|---------|---------|------------|-----------|------------|---------|--|
| UMDA<br>Freq       |      | N/A        | N/A     | N/A        | N/A     | N/A     | N/A        | N/A        | N/N      | N/A      | N/A     | A/A        | N/A      | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A        | N/A     | A/A    | A/N      | A/A     | Ν,Α     | N/H        | N/4       | N/A        | N/A     |  |
| RANSE<br>FACTOR    |      | 7 3. IE+01 | 3.1E+01 | ) 3. IE+01 | 3.16+01 | 3.16+01 | ) J. IE+01 | 9 3. IE+01 | 3. IE+01 | 3.1E+01  | 3.1E+01 | 5 3. IE+01 | 3.16+01  | 3.1E+01 | _       |            | 3.1E+01 | 3.JE+0J | 3.1E+01 | 10+31.5 (  |         |        |          |         | _       | 1.1E+01    | 10+31.5 i | 7. JE + 01 | 2.6E+01 |  |
| 7EAD<br>Freq       |      | 1.16-1     | 7.2E-16 | 9.0E-19    | 9.0E-19 | 9.0E-19 | 9.0E-19    | 9.05-19    | 4.15-16  | 4. IE-16 | 1.2E-1  | 7.76-16    | 4. 3E-19 | 1.3E-18 | 0•0E+00 | 0.0E+0(    | 1.4E-18 | 1.4E-1  | 1.46-16 | 1. 36 - 16 | 0.05+00 | 0.0110 | 0, 0E+9( | 0,0E+00 | 0°0E+0( | 1. IE - JE | 1. IE-16  | 2.76-11    | 5.86-11 |  |
| FANGE<br>FACTOR    |      | ł          | ;       | 1          | 1       | ł       | 1          | ;          | ł        | 1        | ł       | 1          | ;        | ;       | !       | 1          | !       | :       | 1       | !          | 1       | ;      | ł        | 1       | :       | :          | ł         | ;          | 1       |  |
| PUDA<br>Fred       |      | N/A        | N/A     | N/A        | N/A     | A/A     | N/A        | A/N        | N/A      | N/A      | N/A     | N/A        | N/A      | N/A     | A/A     | N/A        | N/A     | A/A     | N/A     | N/A        | N/A     | N/A    | N/A      | N/A     | N/A     | N/A        | N/A       | N/A        | A/A     |  |
| RANGE<br>Factor    |      | ;          | 1       | 1          | ł       | ł       | ;          | ł          | ł        | f        | ſ       | 1          | !        | ł       | ł       | ł          | ;       | ł       | {       | 1          | ł       | ;      | !        | ł       | ;       | ł          | ł         | 1          | ;       |  |
| F8A<br>Freq        |      | N/A        | N/A     | N/A        | N/A     | N/A     | N/A        | N/A        | N/A      | N/A      | N/A     | N/A        | A/N      | N/A     | N/A     | N/A        | N/A     | A/N     | N/A     | N/A        | N/A     | N/A    | N/A      | A/A     | N/A     | N/A        | A/A       | N/A        | N/A     |  |
| RANGE<br>FACTOR    |      | ;          | ;       | ļ          | ;       | 1       | 1          | ;          | :        | :        | ł       | 1          | 1        | ;       | :       | !          | ł       | :       | ł       | ł          | ;       | ł      | ł        | ł       | ;       | :          | 1         | ł          | !       |  |
| NAAF<br>Fred       |      | N/A        | NIA     | N/A        | A/A     | N/A     | N/A        | N/A        | N/A      | N/A      | N/N     | N/A        | N/A      | N/A     | N/A     | N/A        | A/A     | N/A     | N/A     | A/N        | N/A     | N/A    | N/A      | N/A     | A/A     | N/A        | N/A       | N/H        | N/A     |  |
| RANGE<br>FACTOR    |      | 1          | ł       | ł          | ;       | ł       | ;          | 1          | ;        | 1        | ;       | ł          | :        | ł       | ł       | 1          | ł       | ł       | ;       | ;          | !       | ;      | 1        | ۱       | ł       | ;          | :         | ;          | 1       |  |
| L RAD<br>Freq      |      | N/A        | N/A     | N/A        | N/A     | A/N     | N/A        | NIA        | N/A      | N/A      | N/A     | N/A        | N/A      | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A        | N/A     | N/A    | Ň/À      | N/A     | N/A     | N/A        | N/A       | N/A        | N/A     |  |
| FACTOR             | 8688 | ł          | ł       | ;          | 1       | 1       | ł          | ł          | ;        | ł        | 1       | 1          | ;        | ;       | ;       | ł          | ;       | ł       | 1       | ł          | ;       | 1      | 1        | ţ       | ł       | 1          | 1         | ;          | ;       |  |
| AF6<br>Freq        |      | N/A        | N/A     | N/A        | N/A     | N/A     | N/A        | N/A        | N/A      | N/A      | N/A     | N/A        | N/A      | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A        | N/A     | A/A    | N/A      | N/A     | N/A     | N/A        | A.A       | A/A        | N/A     |  |
| KANGE<br>FACTOR    |      | 1          | 1       | ;          | ł       | 1       | 1          | ;          | 1        | :        | 1       | ł          | 1        | :       | !       | ;          | ł       | :       | ł       | 1          | ;       | !      | !        | ł       | 1       | ;          | ;         | ;          | ł       |  |
| ANAD<br>FREQ       |      | N/A        | N/A     | N/A        | A/N     | N/A     | N/A        | N/A        | N/A      | A/A      | e'n     | N/A        | N/A      | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A        | NA      | N/A    | N.A      | N/A     | N/A     | N/A        | A N       | N/A        | NA      |  |
| SCENAR10<br>NUMBER |      | 8          | 80      | 30         | 80      | 8       | 8          | œ          | œ        | 8        | 80      | 0-         | 10       | 10      | 10      | <u>1</u> 0 | 10      | 01      | 10      | 10         | 01      | Ú      | 10       | ļŷ      | 01      | ė          | 10        | 10         | 11      |  |
|                    |      | HFKVC      | HFRVC   | HFFGC      | HFFHC   | JAJH    | HF QGC     | HF QVC     | HFRGC    | HFRVC    | HESVC   | HFSVC      | HF BGC   | HF DHC  | HFEGE   | HFCHC      | HFI.GC  | HFP HC  | HFAVE   | HEAVE      | HFFGC   | HFFHC  | JV7 3H   | HF 05C  | HFBVC   | HFFGC      | HFRVC     | 342 JH     | HF DHC  |  |

File: HFNAT.Wk1 Fage 5 Date 20-Aug-B7

TABLE 6-4 (Continued)

NATIONAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Mandling Operations (MF) (Events per Fallet or Container)

|         | SCENAR 10<br>Minged | ANAD | KANGE<br>CALTOD | AP6<br>CCCO | KANGE<br>East tro | LEAD | RANGE  | NAAP | RANGE | P BA<br>C D C D | RANGE<br>CALTOD | PUDA  | RANGE<br>CACTOD | 1EAD<br>EPECO | RANGE    | UMDA | RANGE<br>Eacting |
|---------|---------------------|------|-----------------|-------------|-------------------|------|--------|------|-------|-----------------|-----------------|-------|-----------------|---------------|----------|------|------------------|
|         |                     |      |                 |             |                   |      |        |      |       |                 |                 |       |                 |               |          |      |                  |
| HFCHC   | Ξ                   | N/A  | ;               | N/A         | :                 | N/A  | ;      | N/A  | :     | N/A             | ;               | N/A   | ţ               | 2.96-11       | 2.6E+01  | N/A  | 8                |
| HFMVC   | Ш                   | N/A  | :               | N/A         | :                 | N/A  | ;      | N/A  | :     | N/A             | ;               | N/A   | ;               | 4.35-11       | 2.6E+01  | N/A  | ;                |
| HFFGC   | =                   | N/A  | :               | N/A         | :                 | N/A  | :      | N/A  | 1     | N/A             | :               | N/A   | !               | 9.66-12       | 2.6E+01  | N/A  | ;                |
| HFFHC   | -                   | N/A  | !               | N/A         | :                 | N/A  | ł      | N/A  | :     | N/A             | :               | N/A   | 1               | 9.6E-12       | 2.6E+01  | N/A  | ;                |
| HFFVC   | =                   | N/A  | :               | N/A         | ţ                 | N/A  | ł      | N/A  | ;     | N/A             | ;               | A/A   | :               | 9.6E-12       | 2.6E+01  | N/A  | ;                |
| HFRGC   | Π                   | N/A  | ;               | N/A         | :                 | N/A  | :      | N/A  | :     | N/A             | ;               | N/A   | :               | 7.2E-12       | 2.6E+01  | N/A  | ł                |
| HFBVC   | П                   | N/A  | {               | A/A         | :                 | N/A  | :      | N/A  | ł     | N/A             | ł               | N/A   | :               | 7.2E-12       | 2.66+01  | A/N  | :                |
| HFRGC   | 11                  | N/A  | !               | N/A         | :                 | N/A  | :      | N/A  | ;     | N/A             | :               | N/A   | :               | 1.86-11       | 2.6E+01  | N/A  | :                |
| HFRVC   | 11                  | N/A  | ;               | N/A         | 1                 | N/A  | !      | N/A  | :     | N/A             | ;               | N/A   | :               | 11-38.1       | 2.6E+01  | N/A  | :                |
| HF [HC  | 12                  | N/A  | ;               | NIA         | ł                 | N/A  | ł      | N/A  | ;     | N/A             | :               | A/A   | ;               | 3.0E-10       | 2.6E+01  | N.A  | :                |
| HF CEC  | 12                  | N/A  | ł               | N/A         | ł                 | N/A  | ł      | N/A  | ;     | N/A             | :               | A/N   | 1               | 3.06-10       | 2.6E+01  | N/A  | ;                |
| HF CHC  | 12                  | N/A  | !               | N/A         | :                 | N/A  | ;      | N/A  | 1     | N/A             | ł               | A/N   | ;               | 3.0E-10       | 2.6E+01  | N/A  | :                |
| HF AVC  | 12                  | N/A  | 4<br>1          | N/A         | ł                 | N/A  | !      | N/A  | :     | N/A             | :               | N/A   | :               | 3.06-10       | 2.46+01  | N/A  | ł                |
| HFFGC   | 12                  | N/A  | 1               | N/A         | ł                 | N/A  | ;      | N/A  | ;     | N/A             | 1               | N/A   | :               | 3, 06-10      | 2.6E+01  | N/À  | ;                |
| HFFHC   | 12                  | N/A  | 4<br>1          | N/A         | ;                 | N/A  | 1      | N/A  | :     | N/A             | :               | N/A   | !               | 3.0E-10       | 2.66+01  | N/A  | ţ                |
| HF FVC  | 12                  | N/A  | 1               | N/A         | ł                 | N/A  | !      | N/A  | ;     | N/A             | :               | N/A   | ţ               | 3, 0E-10      | 2.6E+01  | N/A  | ;                |
| HFQGC   | 11                  | N/A  | ;               | N/A         | ;                 | N/A  | ;      | N/A  | ;     | N/A             | ;               | N/A   | ł               | 3.06-10       | 2.66+01  | N/A  | ł                |
| HF 9VC  | 12                  | N/A  | ;               | N/A         | ł                 | N/A  | 1      | N/A  | :     | N/A             | 1               | N/A   | ;               | 3,05-10       | 2.66+01  | N/A  | :                |
| HFRGC   | 12                  | N/A  | 1               | N/A         | ł                 | N/A  | 1      | N/A  | ł     | N/A             | ł               | N/A   | 1               | 3.0E-10       | 2.6E+01  | N/A  | ;                |
| JURIN   | 12                  | N/A  | :               | N/A         | ;                 | N/A  | 1      | N/A  | 1     | N/A             | 1               | N/A   | ;               | 3.0E-10       | 2.6E+0)  | N/A  | ;                |
| HF DHC  | 13                  | N/A  | :               | N/A         | ł                 | N/A  | 1      | N/A  | ;     | N/A             | 1               | N/A   | :               | 7.26-11       | 2.4E+01  | A/A  | ;                |
| HFCGC   | 11                  | N/A  | ;               | N/A         | 1                 | N/A  | 1      | N/A  | 1     | N/A             | ł               | N/A   | ;               | 3. 6E - 11    | 2.6E+01  | N/A  | ;                |
| HFCHC   | 1                   | A/A  | :               | N/A         | 1                 | N/A  | !      | N/A  | ;     | N/A             | :               | A/A   | :               | 3.66-11       | 2.66401  | N/A  | :                |
| HFRVC   | 11                  | A/A  | !               | N/A         | ł                 | N/A  | ;      | A/A  | !     | N/A             | 1               | N/A   | ;               | 5.46-11       | 2.6E+01  | N/A  | :                |
| HFFGC   | 13                  | N/A  | 1               | N/A         | ;                 | N/A  | ł      | N/A  | ;     | N/A             | ł               | 4 / N | :               | 1.26-11       | 2.66+01  | N/A  | ;                |
| JH 5 JH | 13                  | N/A  | !               | N/A         | :                 | N/A  | l<br>F | M/A  | ;     | N/A             | :               | A'N   | ;               | 1.25-11       | 2. 6E+01 | N/A  | :                |
| HFPVC   | 5                   | 4 N  | ;               | N/A         | ;                 | N/A  | ŧ      | N/A  | ;     | N/A             | ;               | M / Å | ;               | 11-32-11      | 2.6E+()] | N A  | :                |
| HFQEC   | 13                  | N/A  | ;               | N/A         | ł                 | N/A  | ;      | N/A  | 1     | N/A             | :               | A/N   | 1               | 9.01-12       | 2.6E+01  | N/A  | :                |
| HFQVC   |                     | A/A  | ;               | N/A         | ;                 | N/A  | ;      | N/A  | :     | N/A             | :               | N/A   | :               | 9.0E-12       | 2.66+01  | N/A  | 1                |

33

•







Fage b Date 20-Aug-87 File: HFNAT.W.1

TABLE 6-4 (Continued)

NATIONAL COLLOCATION OPTION - FACILITY HANDLING

ŝ

|        |          |      |        |               |        | Accident | Frequenci | tes tor t | acility H | and I ha | Uper at 1 on!    |       | tvents pe | i rallel   | t or Lonta | 1 1 3 0 |                 |
|--------|----------|------|--------|---------------|--------|----------|-----------|-----------|-----------|----------|------------------|-------|-----------|------------|------------|---------|-----------------|
|        | SCENARIO | ANAD | RANGE  | APG           | RANGE  | LBAD     | KANGE     | NAAP      | RANGE     | PEA      | RANGE<br>Lacade. | FUDA  | RANGE     | 1EAD       | RANGE      | UNDA    | KANGE<br>FALLEE |
|        | NUMBER   | FKED | FACIUK | • • • • • • • | FALTUR | FKEU     | FACTOR    | FKEQ      | FACTOR    | + KE 4   | F RC (UK         | r KEN | CHC FUN   |            |            |         |                 |
| HFRGC  | 13       | N/A  | 1      | N/A           | ;      | N/A      | ;         | N/A       | ;         | N/A      | ł                | N/A   | 1         | 2.2E-11    | 2.6E+01    | N/A     | !               |
| HFRVC  | 13       | N/A  | ł      | N/A           | 1      | N/A      | ;         | N/A       | ł         | N/A      | ;                | N/A   | :         | 2.2E-11    | 1 2.6E+01  | A/A     | ļ               |
| HF DHC | 14       | N/A  | ;      | N/A           | !      | N/A      | ;         | N/A       | ł         | N/A      | 1                | N/A   | ;         | 1.3E-14    | 1 2.6E+01  | N/A     | !               |
| HFC6C  | 14       | N.A  | ł      | N/A           | ſ      | N/A      | ł         | N/A       | 1         | N/A      | 1                | N/A   | :         | 6.6E-15    | 5 2.66+01  | A.A     | ţ               |
| HFCHC  | =        | N/A  | 1      | N/A           | ł      | A/A      | ł         | A/A       | 1         | N/A      | ;                | N/A   | :         | 6.6E-15    | 5 2.65+01  | N/A     | ;               |
| HENVC  | 14       | N/A  | ;      | A/A           | 1      | N/A      | 1         | A/A       | ;         | N/A      | :                | N/A   | :         | 31-36.2    | 5 2.6E+01  | A'N     | ;               |
| HFFEC  | 14       | N/A  | :      | N/A           | ł      | N/A      | :         | N/A       | ;         | N/A      | ;                | N/A   | ;         | 2.2E-15    | 5 2.6E+01  | N/A     | ł               |
| HEFHC  | 14       | N/A  | ł      | N/A           | :      | N/A      | ł         | N/A       | ;         | N/A      | :                | N/A   | ł         | 2. ZE-15   | 5 2.6E+01  | N/A     | ;               |
| HFFVC  | 11       | N/A  | ;      | A/A           | ł      | N/A      | !         | A/N       | ł         | N/A      | 1                | N/A   | ł         | 2.2E-15    | 5 2.66+01  | N/A     | ł               |
| HFGEC  | 14       | N/A  | :      | N A           | ł      | N/A      | :         | M/A       | ;         | N/A      | !                | N/A   | ;         | 1.7E-15    | 5 2.6E+01  | N/A     | ;               |
| JYBJH  | 14       | N/A  | ł      | N/A           | 1      | N/A      | 1         | N/A       | 1         | N/A      | ł                | N     | ł         | 1.7E-15    | 5 2.6E+01  | N/A     | ;               |
| HF RGC | 14       | N/A  | 1      | N/A           | ł      | N/A      | ;         | N/A       | 1         | N/A      | :                | N/A   | ;         | 4. IE - 15 | 5 2.6E+Ú]  | N/A     | ;               |
| HFRVC  | 14       | N/A  | ;      | N/A           | ;      | N/A      | 1         | N/A       | ł         | N/A      | ;                | N/A   | ł         | 4.16-15    | 5 2.6E+ûl  | N/A     | ;               |
|        |          |      |        |               |        |          |           |           |           |          |                  |       |           |            |            |         |                 |

<u> 2250 - 122222222 - 12523</u>

### 6.4. UNCERTAINTY ANALYSIS FOR HANDLING ASSOCIATED WITH RAIL TRANSPORT

The values shown in the range factor column in Table 6-4 represent the ratios of the 95th percentile values to the median values. The range factors vary from 13 to 31. The accident sequence frequencies with the largest uncertainty involve (1) forklift collision accidents with fire (H06, HF3) and (2) munition drop accidents inside the MDB. For the latter, the additional failure of the ventilation system for an agent release to the atmosphere to occur is a contributor to the overall uncertainty in the results.

The assignment of error factors to the accident frequency or event probability data was based entirely on engineering judgment. For the handling accidents, the initiating event itself (drop, collision, forklift time puncture) is assigned an error factor of 10. The puncture probability given a drop or collision is assigned an error factor of 3. An error factor of 10 is assigned to the following events: (1) probability of fire given a collision; (2) ventilation system failure; and (3) low-impact detonation of burstered munitions. The error factors for specific events identified in the accident analysis are shown in Tables 6-2 and 6-3.

### ¥

### 6.5. ANALYSIS FOR AIR TRANSPORT

The generic handling operations at both the sending and receiving sites are identical to the operations described above for the rail transportation option for the ton containers, the projectiles, and the rockets except for the operations occurring between the holding area and the air strip which are specific to the air transportation option. The list of accident sequences identified for the rail transportation option was used and applied to the air transportation option. Onsite handling operations (HA) and facility-related handling operations (HF) are identified. For the air transportation option only the results which pertain to ton containers, projectiles, and rockets apply.

Table 6-5 presents the initiating event frequencies for the accident scenarios. These frequencies correspond, in terms of events per operation, closely to those used for the rail option handling. Table 6-6 presents the conditional event probabilities for the subsequent events along the sequence pathways.

The sequence frequencies and uncertainty analysis results are presented in Table 6-7 for the air transport option.



# TABLE 6-5 INITIATING EVENTS FREQUENCIES (HANDLING ASSOCIATED WITH AIR TRANSPORT) INITIATING EVENTS FREQUENCIES

|                     | INITIATING EVENT                                                                                | FREQUENCY<br>Events/op | ERROR<br>Factor | REFERENCE<br>(NOTES)                            | APPLICABLE<br>SCENARIOS                                                                                   |
|---------------------|-------------------------------------------------------------------------------------------------|------------------------|-----------------|-------------------------------------------------|-----------------------------------------------------------------------------------------------------------|
| 8469 <del>4</del> 9 | ##************************************                                                          | (Col.X)                | (Col.I)         |                                                 |                                                                                                           |
| HE10                | Pallet or single item dropped during<br>handling of non-leaking munition<br>outside the MDB (1) |                        |                 |                                                 | HA1,HA5,HA8,HA11,HA22,H<br>HF1,HF11                                                                       |
| HEIÚA               | Items lifted with times                                                                         | 3.0E-05                | 10.0            | 6.7                                             |                                                                                                           |
| HE10B               | Items lifted with lifting beams                                                                 | 3.02-06                | 10.0            | 8                                               |                                                                                                           |
| HE15                | Pallet or container dropped during<br>handling of non-leaking munition<br>inside the MDB (2)    |                        |                 |                                                 | HF8,HF13                                                                                                  |
| HE15A               | Items lifted with tines                                                                         | 1.5E-04                | 10.0            | 7                                               |                                                                                                           |
| HE15D               | Items lifted with lifting beams                                                                 | 1.5E-05                | 10.0            | 8                                               |                                                                                                           |
| HE20                | Pallet or container dropped during handling of leaking eunition (3)                             |                        |                 |                                                 | HA17,HA29                                                                                                 |
| HE20A               | Items lifted with tines                                                                         | 3.0E-04                | 10.0            | 7                                               |                                                                                                           |
| HE20B               | Items lifted with lifting beams                                                                 | 3.0E-05                | 10.0            | 8                                               |                                                                                                           |
| HE25                | Single munition dropped inside the NDB (4)                                                      | 3.0E-04                | 10.0            |                                                 | HF2,HF12                                                                                                  |
| HE35                | Single leaking munition dropped (3)                                                             | 6.0E-04                | 10.0            |                                                 | HA18,HA30                                                                                                 |
| HE40                | Forklift tine accident involving<br>munition handling outside the NDB (1)                       | 1.0E-05                | 10.0            |                                                 | HA3,HF4                                                                                                   |
| HE45                | Forklift tine accident involving<br>munition handling inside the MDB (2)                        | 5.0E-05                | 10.0            |                                                 | HF9                                                                                                       |
| HE50                | Forklift time accident involving handling of leaking eunition (3)                               | 1.0E-04                | 10.0            |                                                 | HA19                                                                                                      |
| HE55                | Vehicle collision accident                                                                      | 4.3E-06                | 10.0            | GA derive<br>data, see<br>details i<br>Appendix | HA2, HA4, HA6, HA7, HA9, HA10<br>HA12, HA21, HA24, HA25, HA26<br>HA27, HA31, HF3, HF5, HF7,<br>HF10, HF14 |
| HE6S                | Failure to detect a leak in the                                                                 | 1.0E-03                | 10.0            | ł                                               | HA32                                                                                                      |

TABLE 6-5 (Continued)

12222

### transportation container

### NOTES:

- (1) Handled by forklift or other handling equipment; operators wearing street clothes with mask slung
- (2) Handled by forklift; operators wearing mask, gloves, and boots; excluding ton container
- (3) Operators in level A clothing

ፚዹዄዿዄዿዀዿዀጟዄዀዀዀዀዀዀዀዀዀዀዀ

- (4) Handled singly by hand; operators wearing mask, gloves and boots.
- (6)  $3.0e^{-5} = 3*10^{-5}$
- (7) For all items lifted with times (spray tanks in overpacks and bare munitions)
- (8) Items lifted by a lifting beam or by a cargo handling equipment

l.

R

TABLE 6-6CONDITIONAL EVENTS PROBABILITIES (AIR TRANSPORT)

1. J. J. J. J.

े के कि जन्म जन्म

0m

|        | EVENT SEQUENCE                                                                                                             | EVENT<br>PROBABILITY | ERROR<br>Factor | REFERENCEAPPLICABLE<br>SCENARID |
|--------|----------------------------------------------------------------------------------------------------------------------------|----------------------|-----------------|---------------------------------|
|        |                                                                                                                            |                      |                 |                                 |
| HE100  | Palletized or single munition<br>punctured given a drop outside<br>the MDB (Drop ht = 6ft.)                                |                      |                 | HAI                             |
| HE100K | Ton Container                                                                                                              | 3.34E-03             | 3.0             |                                 |
| HE100P | 155-sa Projectile                                                                                                          | 0.00E+00             |                 |                                 |
| HE1000 | 8-in Projectile                                                                                                            | 0.00E+00             |                 |                                 |
| HE100R | Rocket                                                                                                                     | 7.95E-04             | 3.0             |                                 |
| HE110  | Offsite container and munition punctured given a drop of the offsite container (4ft drop)                                  |                      |                 | HA8,HF1                         |
| HE110K | Ton Container                                                                                                              | 1.83E-03             | 3.0             |                                 |
| HEIIOP | 155-am Projectile                                                                                                          | 8.40E-04             | 3.0             |                                 |
| HEILOQ | 8-in Projectile                                                                                                            | 8.40E-04             | 3.0             |                                 |
| HEIIOR | Rocket                                                                                                                     | 1.06E-03             | 3.0             |                                 |
| HE120  | Container and munition punctured given<br>drop of the offsite container (2ft drop)                                         |                      |                 | HA9,HA10                        |
| HE120K | Ton Container                                                                                                              | 1.10E-03             | 3.0             |                                 |
| HE120P | 155-mm Projectile                                                                                                          | 2.00E-04             | 3.0             |                                 |
| NE1200 | 8-in Projectile                                                                                                            | 2.00E-04             | 3.0             |                                 |
| HE120R | Rocket                                                                                                                     | 9.00E-04             | 3.0             |                                 |
| HE140  | Palletized or single munition<br>punctured given a drop resulting<br>from collision (Drop ht = 2ft.)                       |                      |                 | Ha2, Ha4, Ha21                  |
| HE140K | Ton Container                                                                                                              | 1.68E-03             | 3.0             |                                 |
| HE140P | 155-se Projectile                                                                                                          | 0.00E+00             | •               |                                 |
| HE1400 | 8-in Projectile                                                                                                            | 0.00E+00             |                 |                                 |
| HE140R | Rocket                                                                                                                     | 7.16E-04             | 3.0             |                                 |
| HE150  | Palletized munition in onsite container<br>puncture given a drop of container<br>(Drop ht = 4ft., also applies to handling | in UPA)              |                 | HF8 HAS                         |

6-70

istrain.

TABLE 6-6 (Continued)

| HE150K | Ton Container                            | 7.20E-04 | 3.0  |                         |
|--------|------------------------------------------|----------|------|-------------------------|
| HE150P | 155-mm Projectile                        | 6.00E-05 | 3.0  |                         |
| HE1509 | 8-in Projectile                          | o.00E-05 | 3.Ú  |                         |
| HE150R | Rocket                                   | 2.70E-04 | 3.0  |                         |
| HEIOÙ  | Palletized or single munitich in onsite  |          |      | HHO, HHT                |
|        | container punctured given drop resulting |          |      |                         |
|        | from collision (2ft drop)                |          |      |                         |
| HE160k | Ton Container                            | 3.30E-04 | 3.Ú  |                         |
| HE160P | 155-mm Projectile                        | 0.V0E+00 |      |                         |
| HE1600 | 8-in Frojectile                          | 0.00E+09 |      |                         |
| HE160R | Rocket                                   | 2.60E-04 | 3.9  |                         |
| HE250  | Sinole bace eucition                     |          |      | HF∠                     |
|        | ounctured given dron in UPA              |          |      | _                       |
|        | (Drop ht = 4ft.)                         |          |      |                         |
| HE250K | Ton Container                            | 2.80E-03 | 3.0  |                         |
| HE250P | 155-mm Projectile                        | 0.00E+00 |      |                         |
| HE2500 | 8-in Projectile                          | Ú.VVE+ŬŬ |      |                         |
| HE250R | facket                                   | 5.93E-04 | 3.0  |                         |
| HE400  | Munition punctured by                    |          |      | HH2, HH19               |
|        | torklift tines                           |          |      |                         |
| HE400P | 155-mm Projectile                        | 5.00E-03 | 3.0  |                         |
| HE4009 | 8-in Projectile                          | 5.00E-03 | 3.0  |                         |
| HE400R | Rocket                                   | 2.63E-v1 | 3.0  |                         |
| HE55V  | Fire results from vehicle                | 7.25E-V2 | 10.0 | See нрр F НА2.Ннс. Нн5. |
|        | collision                                |          |      | nH∠c,dH27,8F3,8F5       |
| HE555  | Collision does not cause                 | 9.27E-01 | none | See Hop F HHT, HHIV.    |
|        | +1re                                     |          |      | Hr ·                    |
| HE560  | Fire contained                           |          |      | HHL, HHS, HH?, HF]      |
| HE56VA | 4 min - Burstered munitions              | 5.00E-01 | none |                         |
| HESOUD | 30 min - Non burstered munitions         | 1.00E+00 | none |                         |
| HESHOE | √15 min - Ún/Úffsite container           | 1.00F+00 | none |                         |



W.Z.L. LYLLEN

TABLE 6-6 (Continued)

Ē

| HE570  | Fire not contained                                                                          |          |      | HA26,HA27,HF5                               |
|--------|---------------------------------------------------------------------------------------------|----------|------|---------------------------------------------|
| HES70A | 4 min - Burstered munitions                                                                 | 5.00E-01 | none |                                             |
| HE5709 | 30 min - Non burstered munitions                                                            | 0.00E+00 | none |                                             |
| HE570C | >15 min - On/Offsite container                                                              | 0.00E+00 | none |                                             |
| HE590  | Nunition in on/offsite container<br>detonates or ruptures given prolonged<br>fire (>13 min) | 1.00E+00 | none | HA26,HA27,HF5                               |
| HE600  | Munition detonates given drop (6ft)<br>or collision (per munition)                          | 9.50E-09 |      | HA11,HA12<br>HA29,HA30,HA31 (R)             |
| HE600P | 155-se Projectile (8)                                                                       | 7.60E-08 | 10.0 |                                             |
| HE6000 | 8-in Projectile (6)                                                                         | 5.70E-08 | 10.0 |                                             |
| HE600R | Rocket (15)                                                                                 | 1.43E-07 | 10.0 |                                             |
| HE620  | Single bare munition detonates<br>given 4 ft drop (in UPA)                                  | 3.20E-10 | 10.0 | HF12                                        |
| HE700  | Munition in onsite container<br>detonates given drog (ger munition)                         | 3.20E-11 |      | HA22,HA24,<br>HF11.HF13.HF14                |
|        |                                                                                             |          |      |                                             |
| HE700P | 155-me Projectile (8)                                                                       | 2.56E-10 | 10.0 |                                             |
| HE7000 | 8-in Projectile (6)                                                                         | 1.92E-10 | 10.0 |                                             |
| HE700R | Rocket (15)                                                                                 | 4.80E-10 | 10.0 |                                             |
| HE710  | Munition in offsite container                                                               | 3.20E-12 |      | HA23,HA25                                   |
|        | detonates given drop (per munition)                                                         |          |      |                                             |
| HE710P | 155-se Projectile (8)                                                                       | 3.84E-10 | 10.0 |                                             |
| HE7100 | 8-in Projectile (6)                                                                         | 1.92E-10 | 10.0 |                                             |
| HE710R | Rocket (15)                                                                                 | 1.92E-10 | 10.0 |                                             |
| HEB00  | MDB Ventilation System Failure                                                              | 1.00E-09 | 10.0 | HA17,HA18,HA19,HA21<br>HF2,<br>HF8,HF9,HF10 |

.....

6-72







TABLE 6-7 HANDLING ACCIDENT FREQUENCIES - COLLOCATION PROCESSING OPTION (AIR TRANSPORT)

DATE18-Aug-87 PAGE1

HAMDLING ACCIDENTS - COLLOCATION PROCESSING OFTION (AIR TRANSPORT) (PER CONTAINER OR FALLET)

Accident Frequencies and Range Factors

| RANGE<br>Factor | 8                             | !        | !       | 1       | ;        | !        | ł       | :       | ;       | ;       | :        | ;       | ;       | !       | ;       | :         | ;       | ;       | :          | :        | ;       | :       | ;       | :         | ;        | •        | ;       | :        | :        | !        |
|-----------------|-------------------------------|----------|---------|---------|----------|----------|---------|---------|---------|---------|----------|---------|---------|---------|---------|-----------|---------|---------|------------|----------|---------|---------|---------|-----------|----------|----------|---------|----------|----------|----------|
| UMDA<br>Fred    |                               | N/A      | N/A     | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | A/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A       | N/A     | A/A     | N/A        | N/A      | N/A     | A/A     | N/A     | N/A       | N,À      | N/A      | N/A     | N/A      | N/A      | N/A      |
| RANGE<br>Factor |                               | 1. 3E+01 | ł       | :       | 1        | 1. 3E+01 | 1.3E+01 | ł       | 1.3E+01 | 1.36+01 | 1. 3E+01 | 1.36+01 | 1.35+01 | 1.3E+01 | :       | ;         | ;       | 1.36+01 | 1. 3E H) 1 | 1. 3E+01 | ;       | 1       | ;       | 1. JE +01 | 1. 3E+01 | 3. IE+01 | !       | ;        | ł        | J. 16+01 |
| TEAD<br>F9EQ    |                               | 4.0E-08  | 0.0E+00 | 0.0E+00 | 0.0E+00  | 9.56-08  | 9.5E-08 | N/A     | 1.0E-07 | 1.05-07 | 1.0E-07  | 5.3E-06 | 5.36-06 | 1.46-08 | 0.0E+00 | ().0E+0() | 0.0E+00 | 6.2E-09 | 6.25-09    | 1.36-08  | 1.1E-09 | 1.16-09 | 1.1E-09 | 4.9E-09   | 4.95-09  | 2.16-10  | 0.06+00 | 0.0130.0 | 0. 0E+00 | 1.65-10  |
| RANGE<br>FACTOR | 1                             | ;        | :       | :       | ł        | !        | ;       | ;       | :       | :       | ł        | ł       | :       | !       | ;       | ;         | ;       | :       | ;          | ;        | 1       | ;       | ;       | :         | 1        | :        | :       | i        | !        | ;        |
| PUDA<br>Fred    | 1<br>                         | N/A      | N/A     | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A       | N/A     | N/A     | N/A        | N/A      | N/A     | N/A     | N/A     | N/A       | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      |
| RANGE<br>Factor |                               | :        | :       | 1       | ;        | ł        | :       | ;       | :       | ł       | ;        | :       | ł       | :       | ;       | ł         | ;       | 1       | ł          | :        | {       | ł       | 1       | ;         | ;        | :        | :       | ;        | ;        | ł        |
| PBA<br>FREQ     |                               | N/A      | N/A     | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A       | N/A     | N/A     | N/A        | N/A      | N/A     | N/A     | N/A     | N/A       | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      |
| RANGE<br>FACTOR |                               | 1        | :       | :       | ł        | ;        | :       | ł       | 1       | ;       | 1        | ;       | 1       | :       | :       | :         | 1       | ;       | ;          | 1        | 1       | {       | L<br>1  | ļ         | ł        | :        | :       | ł        | ;        | -        |
| NAAP<br>Freq    |                               | N/A      | N/A     | N/A     | N/A      | N/N      | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A       | N/A     | N/A     | N/A        | N/A      | N/A     | N/A     | N/A     | N/A       | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      |
| RANGE<br>Factor |                               | ;        | 1       | :       | ;        | 1.3E+01  | 1.3E+01 | ;       | 1.36+01 | 1.36+01 | 1.3E+01  | 1.3E+01 | 1.3E+01 | 1       | ;       | :         | ;       | 1.3E+01 | 1. 3E+01   | :        | ł       | 1       | ł       | 1         | ł        | ;        | ;       | 1        | ł        | :        |
| L BAD<br>F REQ  |                               | N/A      | 0.0E+00 | 0.0E+00 | 0°.0E+00 | 4.86-08  | 4.86-08 | N/A     | 5.0E-08 | 5.05-08 | 5.0E-08  | 2.65-06 | 2.65-06 | N/A     | 0.0E+00 | 0.0E+00   | 0.01400 | 3.15-09 | 3. 1E-09   | N/A      | N/A     | N/A     | N/A     | N/A       | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      |
| RANGE<br>Factor |                               | 1.35+01  | :       | ł       | ł        | ;        | ł       | 3.1E+01 | :       | :       | ł        | :       | ł       | 1.35+01 | :       | ł         | ;       | {       | ;          | 1        | 1       | ſ       | í       | ;<br>1    | {        | ;        | ;<br>1  | ;        | ł        | ;        |
| AFG<br>Freq     |                               | 2.0E-08  | N/A     | N/A     | N/A      | N/A      | N/A     | 5.2E-10 | N/A     | N/A     | N/A      | N/A     | N/A     | 6.7E-09 | N/A     | N/A       | N/A     | N/A     | N/A        | N/A      | N/A     | N/A     | N/A     | N/A       | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      |
| RANGE<br>FACTOR |                               | 1        | !       | 1       | ł        | ł        | 1       | :       | ;       | ł       | 1        | :       | ł       | ;       | ;       | :         | . ¦     | :       | ;          | ;        | ;       | ;       | ł       | ł         | ł        | :        | ;       |          | ł        | 1        |
| ANAD<br>Freq    | <br> <br> <br> <br> <br> <br> | N/A      | N/A     | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A       | N/A     | N/A     | N/A        | N/A      | N/A     | N/A     | N/A     | N/A       | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      |
| Q               |                               | -        | -       | -       |          |          | -       | ~       | 2       | ~       | 2        | M       | 2       | -       | 4       | *         | -       | 4       | 4          | ŝ        | רי      | ŝ       | לע      | רש        | רש       | 9        | 9       | ę        | Ģ        | 9        |
| SCEN-<br>AR10   |                               | SH ARH   | HAFHC   | HAFVC   | HADEC    | HARGC    | HARVC   | HAKHF   | HAFHC   | HAFVC   | HAGGC    | HAFGC   | HARVC   | HAKHS   | HAFHC   | HAFVC     | HADGC   | HARGC   | HARVC      | HAPHS    | HAFHS   | HAFVS   | HADES   | HARGS     | HARVS    | HAKHF    | HAFHF   | HAFVF    | HADGF    | HARGF    |

DATE18-Aug-87 FAGE2

TABLE 6-7 (Continued)

HAMDLING ACCIDENTS - COLLOCATION PROCESSING OFTION (AIR TRANSPORT) (PER CONTAINER OR PALLET)

Accident Frequencies and Range Factors

| or i                |          |          |       |         |         |         |          |          |          |          |         |           |          |          |          |          |          |          |          |         |          |           |         |          |          |          |         |         |         |
|---------------------|----------|----------|-------|---------|---------|---------|----------|----------|----------|----------|---------|-----------|----------|----------|----------|----------|----------|----------|----------|---------|----------|-----------|---------|----------|----------|----------|---------|---------|---------|
| FARGE               | :        | •        | 1     | ł       | ;       | ;       | :        | ;        | :        | 1        | !       | 1         | •        | !        | ;        | ;        | :        | ł        | :        | ł       |          | ł         | 1       | :        | :        | 1        | :       | ;       | J<br>L  |
| UMDA<br>FREG        | N/A      | N/A      | N/A   | N/A     | N/A     | N/A     | N/A      | N, A     | N/A      | N/A      | N/A     | N/A       | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A     | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     |
| RANGE<br>Factor     | 3. IE+01 | 1. 3E+01 | 1     | ł       | ;       | 1.3E+01 | 1, 35+01 | 1.35401  | 1, 3E+01 | 1.3E+01  | 1.3E+01 | 1, 35+01  | 1.3€+01  | 3. IE+01 | 3.1E+01  | 3. JE+01 | 3. 1E+01 | 3. 1E+01 | 3. 1E+01 | 1.35+01 | 1.35+01  | 1.35+01   | 1.JE+01 | 1, 36+01 | 1.35+01  | 2.6E+i)1 | 2.6E+01 | 2.6E+01 | 2.6E+01 |
| TEAD<br>FREQ        | 1.66-10  | 4.1E-09  | 0°*00 | 0.0E+00 | 0°0E+00 | 3.2E-09 | 3.2E-09  | 3. 3E-08 | 1.5E-08  | 1.56-08  | 1.55-08 | 1.95-08   | 1.9E-08  | 1.05-09  | 1.9E-10  | 1.96-10  | 1.96-10  | 8.4E-10  | 8.4E-10  | 1.3E-08 | 2.4E-09  | 2.4E-09   | 2.4E-09 | 1.1E-08  | 1.16-08  | 9.46-10  | 9.65-10 | 7.2E-10 | 1.86-09 |
| RANGE<br>Factor     | ł        | ;        |       | ł       | ł       | 1       | ;        | 1        | 1        | ł        | 1       | 1         | 1        | ł        | ł        | ;        | ;        | ł        | !        | ;       | :        | !         | ;       | 1        | ;        | :        | ł       | ł       | ;       |
| FUDA<br>FREQ        | N/A      | N/A      | N/A   | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A       | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A     | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     |
| RANGE<br>Factor<br> | 1        | ł        | ;     | ł       | !       | ł       | ł        | :        | ;        | ł        | 1       | 1         | ł        | ł        | ł        | !        | ;        | ł        | ł        | ł       | ;        | ;         | :       | ł        | ;        | 1        | 1       | 1       | ł       |
| PBA<br>FREQ         | N/A      | N/A      | N/A   | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A       | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A     | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     |
| FACTOR              | ł        | 1        | 1     | ł       | ł       | 1       | ;        | ł        | ł        | ł        | ł       | ł         | ł        | 1        | ł        | :        | 1        | I        | ;        | i       | ł        | ł         | ł       | 1        | 1        | ł        | 1       | 1       | 1       |
| NAAP<br>FREQ        | N/A      | N/A      | N/A   | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A       | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A     | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     |
| RANGE<br>Factor     | :        | ;        | 1     | ţ       | ;       | 1       | ;        | !        | 1.3E+(i1 | 1°.3E+01 | 1.3E+01 | 1. 3E+01  | 1. 3E+01 | 1        | 3. IE+01 | 3. IE+01 | 3. IE+01 | 3. 1E+01 | 3. IE+01 | 1       | 1. 3E+01 | 1. JE +01 | 1.3E+01 | 1, 35+01 | 1, 3E+01 | 2.6E+01  | 2.6E+01 | 2.6E+01 | 2.6E+01 |
| LBAD<br>FREQ        | N/A      | N/A      | N/A   | N/A     | N/A     | N/A     | N/A      | N/A      | 1.5E-08  | 1.5E-08  | 1.5E-08 | 1.9E-08   | 1.9E-08  | N/A      | 1.9E-10  | 1.9E-10  | 01-36-10 | 01-31-8  | 8.4E-10  | N/A     | 2.4E-09  | 2.4E-09   | 2.4E-09 | 1.1E-08  | 1.1E-08  | 4.86-10  | 4.86-10 | 3.6E-10 | 9.0E-10 |
| RANGE<br>Factor     | ł        | :        | 1     | !       | 1       | ;       | :        | 1.3E+01  | 1        | ;        | ;       | ł         | ;        | 3. 1E+01 | 1        | ;        | ;        | ;        | ł        | 1.36+01 | 1        | ł         | ;       | ;        | ;        | ;        | ł       | ł       | !       |
| APG<br>FREQ         | N/A      | N/A      | N/A   | N/A     | N/A     | N/A     | N/A      | 3. 3E08  | N/A      | N/A      | N/A     | N/A       | N/A      | 1.0E-09  | N/A      | N        | N/A      | N/A      | N/A      | 1.3E-08 | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     |
| RANGE<br>Factor     | ;        | ;        | ł     | t<br>#  | ;       | ÷       | :        | ;        | 1        | ;        | ł       | ;         | ł        | :        | :        | 1        | :        | 1        | 1        | ł       | ļ        | ;         | 1       | ł        | {        | ł        | ł       | ;       | :       |
| ANAD<br>Freq        | N/A      | N/A      | N/A   | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A       | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A      | N/A     | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     |
|                     | <b>.</b> | 1        | 7     | 7       | 7       | 1       | 1        | æ        | 8        | 8        | 8       | 80        | 80       | 6        | 0-       | 6        | ð        | ъ        | 6        | 10      | 10       | 10        | 10      | 10       | 10       | Ξ        | Ξ       | Ξ       | Ξ       |
| CEN-<br>RID         | ۲۶       | SH       | HS    | ۲S      | 65      | 65      | SA       | SH       | ¥S       | 5A       | 65      | <b>55</b> | ٨S       | H        | 놔        | 5        | 5        | BF       | 5        | HS      | St       | ۲s        | 35      | 35       | ٧S       | ť        | ٧C      | 6C      | 90      |

5





DATE18-Aug-87 PAGE3

Marken Marken Charles Contract

TABLE 6-7 (Continued)

# HANDLING ACCIDENTS - COLLOCATION PROCESSING OFTION (AIR TRANSFORT) (PER CONTAINER OR FALLET)

### Accident Frequencies and Range Factors

|   | !                 | ;                                                       | ł                                                                                                           | ł                                                                                                                                                                                                                            | !                                                                                                                                                                                                                                                                                                                                                                                                                                             | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                 | !                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | !                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ;                                                                                                                                                                                                                                                                                                                                                                                                                                               | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1          | !       | ł       | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ;       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | i       |                                                                                                      | !       | •                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|---|-------------------|---------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------|---------|---------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------|------------------------------------------------------------------------------------------------------|---------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|   | N/A               | N/A                                                     | N/A                                                                                                         | N/A                                                                                                                                                                                                                          | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                           | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                               | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A        | N/A     | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/G                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | NA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A     | N/A                                                                                                  | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|   | 2.6E+01           | 2.6E+01                                                 | 2.6E+01                                                                                                     | 2.6E+01                                                                                                                                                                                                                      | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                       | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                           | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 2.6E+(i)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 3. IE+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 3. 1E+01                                                                                                                                                                                                                                                                                                                                                                                                                                        | 3. IE+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 3. IE+01   | 2.6E+01 | 2.6E+01 | 2.65+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 2.6E+ů1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 2.6E+01 | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 2.65+01 | 2.6E+01                                                                                              | 2.6E+01 | 2.6E+ii1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|   | 1.8E-09           | 6.9E-11                                                 | 6.9E-11                                                                                                     | 5.2E-11                                                                                                                                                                                                                      | 1.3E-10                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1.3E-10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 6.0E-13                                                                                                                                                                                                                                                                                                                                                                                                                                           | 6.0E-13                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 1.2E-12                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1.2E-12                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 2.6E-14                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 2.6E-14                                                                                                                                                                                                                                                                                                                                                                                                                                         | 3. 1E-19                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 3. IE-18   | 1.45-11 | 1.4E-11 | 1.1E-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 2.7E-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 2.7E-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 2.2E-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 2.2E-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1.16-11 | 1.16-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 1.1E-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 1.06-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 1.0E-11 | 7.7E-12                                                                                              | 1.9E-11 | 1.9E-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|   | :                 | ;                                                       | ł                                                                                                           | ł                                                                                                                                                                                                                            | {                                                                                                                                                                                                                                                                                                                                                                                                                                             | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | í                                                                                                                                                                                                                                                                                                                                                                                                                                                 | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | !                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ł                                                                                                                                                                                                                                                                                                                                                                                                                                               | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ł          | 1       | 1       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ł       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ;       | !                                                                                                    | ł       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|   | N/A               | N/A                                                     | N/A                                                                                                         | N/A                                                                                                                                                                                                                          | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                           | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                               | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A        | N/A     | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A     | N/A                                                                                                  | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|   | ;                 | ;                                                       | ;                                                                                                           | ł                                                                                                                                                                                                                            | 1                                                                                                                                                                                                                                                                                                                                                                                                                                             | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                 | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ;                                                                                                                                                                                                                                                                                                                                                                                                                                               | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ;          | ;       | ;       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ;       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ;       | :                                                                                                    | ;       | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|   | N/A               | N/A                                                     | N/A                                                                                                         | N/A                                                                                                                                                                                                                          | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                           | A/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                               | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A        | N/A     | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A     | N/A                                                                                                  | NZA     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|   | 1                 | :                                                       | ;                                                                                                           | 1                                                                                                                                                                                                                            | ;                                                                                                                                                                                                                                                                                                                                                                                                                                             | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                                                                                                                                                                                                                                                                                                                                                                                                                                               | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ۱          | 1       | ł       | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | :       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | :       | •                                                                                                    | 1       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|   | N/A               | N/A                                                     | N/A                                                                                                         | N/A                                                                                                                                                                                                                          | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                           | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                               | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A        | N/A     | N/A     | N.A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A     | N/A                                                                                                  | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|   | 2 <b>.</b> 6E+(i] | 2.6E+01                                                 | 2.6E+01                                                                                                     | 2.6E+01                                                                                                                                                                                                                      | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                       | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                           | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 3. IE+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 3. IE+01                                                                                                                                                                                                                                                                                                                                                                                                                                        | 3.1E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 3.1E+01    | i<br>i  | ł       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 2.6E+01 | 2.6E+01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 2.6E+i)1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 1       | ;                                                                                                    | •       | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| 1 | 9.0E-10           | 3.4E-11                                                 | 3.4E-11                                                                                                     | 2.6E-11                                                                                                                                                                                                                      | 6.5E-11                                                                                                                                                                                                                                                                                                                                                                                                                                       | 6.5E-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 6.0E-13                                                                                                                                                                                                                                                                                                                                                                                                                                           | 6.0E-13                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 1.2E-12                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1.2E-12                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 2.6E-14                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 2.6E-14                                                                                                                                                                                                                                                                                                                                                                                                                                         | 3.1E-18                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 3. IE-18   | R/A     | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 2.25-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 2.2E-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1.16-11 | 1.16-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 1.16-11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N A     | N,A                                                                                                  | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|   | 1                 | ;                                                       | ;                                                                                                           | 1<br>1                                                                                                                                                                                                                       | ł                                                                                                                                                                                                                                                                                                                                                                                                                                             | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 1<br>1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                                                                                                                                                                                                                                                                                                                                                                                                                                               | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ;          | !       | ;       | !                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | :       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ;       | :                                                                                                    | !       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|   | N/A               | N/A                                                     | N/A                                                                                                         | N/A                                                                                                                                                                                                                          | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                           | N / A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                               | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N.A        | N/A     | N / A   | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | A/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A     | 97N                                                                                                  | 2       | 41 - <b>X</b>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|   | ł                 | 1                                                       | ;                                                                                                           | ł                                                                                                                                                                                                                            | 1                                                                                                                                                                                                                                                                                                                                                                                                                                             | :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | ł                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                                                                                                                                                                                                                                                                                                                                                                                                                                               | !                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | !          | ł       | ł       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | :       | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | ;                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | :       | i                                                                                                    | ;       | 4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|   | A/A               | N/A                                                     | N/A                                                                                                         | N/A                                                                                                                                                                                                                          | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                           | A/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                               | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                             | N / A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | N/A        | A/A     | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N, A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | N/A     | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | H/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N/A     | N/6                                                                                                  | N/A     | <b>h</b> /f                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|   | 2                 | 12                                                      | 2                                                                                                           | 2                                                                                                                                                                                                                            | 12                                                                                                                                                                                                                                                                                                                                                                                                                                            | 12                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 17                                                                                                                                                                                                                                                                                                                                                                                                                                                | 11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 18                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 8                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 13                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 61                                                                                                                                                                                                                                                                                                                                                                                                                                              | 21                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 21         | 22      | đ       | 22                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 22                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 23                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | Ľ.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 23      | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 23                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 24                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 24      | 7.                                                                                                   | 74      | t.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|   |                   |                                                         |                                                                                                             |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |            |         |         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |         |                                                                                                      |         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|   |                   | N/A N/A 9.0E-19 2.6E+01 N/A N/A N/A 1.8E-09 2.6E+01 N/A | N/A N/A 9.0E-19 2.6E+01 N/A N/A N/A 1.8E-09 2.6E+01 N/A N/A 1.8E-09 2.6E+01 N/A N/A N/A 0.9E-11 2.6E+01 N/A | N/A   9.0E-10  2.6E+01  N/A   N/A   1.8E-09  2.6E+01  N/A     N/A   N/A   1.4E-11  2.6E+01  N/A     N/A   N/A   N/A   1.8E-09  2.6E+01  N/A     N/A   N/A   N/A   1.2.6E+01  N/A     N/A   N/A   N/A   6.9E-11  2.6E+01  N/A | N/A     9,0E-10    2,6E+01    N/A     N/A     1,8E-09    2,6E+01    N/A       N/A     N/A     9,0E-10    2,6E+01    N/A     1,8E-09    2,6E+01    N/A       N/A     N/A     N/A     1,8E-09    2,6E+01    N/A       N/A     N/A     N/A     6,9E-11    2,6E+01    N/A       N/A     N/A     N/A     6,9E-11    2,6E+01    N/A       N/A     N/A     N/A     6,9E-11    2,6E+01    N/A       N/A     N/A     N/A     5,2E-11    2,6E+01    N/A | N/A     9,0E-10    2,6E+01    N/A     N/A     1,8E-09    2,6E+01    N/A       N/A     9,0E-10    2,6E+01    N/A     N/A     1,8E-09    2,6E+01    N/A       N/A     N/A     N/A     N/A     6,9E-11    2,6E+01    N/A       N/A     N/A     N/A     6,9E-11    2,6E+01    N/A       N/A     N/A     N/A     6,9E-11    2,6E+01    N/A       N/A     N/A     N/A     6,9E-11    2,6E+01    N/A       N/A     N/A     N/A     5,2E-11    2,6E+01    N/A       N/A     N/A     N/A     5,2E-11    2,6E+01    N/A       N/A     N/A     N/A     5,2E-11    2,6E+01    N/A <td>N/A   9,0E-10  2,6E+01  N/A   N/A   1,8E-09  2,6E+01  N/A     N/A   9,0E-10  2,6E+01  N/A   N/A   1,8E-09  2,6E+01  N/A     N/A   N/A   N/A   N/A   6,9E-11  2,6E+01  N/A     N/A   N/A   N/A   6,9E-11  2,6E+01  N/A     N/A   N/A   N/A   6,9E-11  2,6E+01  N/A     N/A   N/A   N/A   1,3E-10  2,6E+01  N/A</td> <td>N/A     N/A     N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     9.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     N/A     N/A     6.9E-11    2.6E+01    N/A       N/A     N/A     N/A     6.9E-11    2.6E+01    N/A       N/A     N/A     N/A     6.9E-11    2.6E+01    N/A     6.9E-11    2.6E+01    N/A     6.9E-11    2.6E+01    N/A     5.2E-11    2.6E+01    N/A     1.3E-10    2.6E+01    N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A   </td> <td>N/A     N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     9.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     N/A     N/A     6.9E-11    2.6E+01    N/A       N/A     N/A     N/A     6.9E-11    2.6E+01    N/A     6.9E-11    2.6E+01    N/A     6.9E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     5.2E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A   <td>N/A     N/A     N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     9.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     N/A     N/A     6.9E-11    2.6E+01    N/A     1.8E-09    2.6E+01    N/A     1.8E-09    2.6E+01    N/A     1.8E-09    2.6E+01    N/A     1.7     6.9E-11    2.6E+01    N/A     1.3E-12    2.6E+01    N/A     N/A     &lt;</td><td>N/A     N/A     N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     9.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     N/A     N/A     6.9E-11    2.6E+01    N/A     N/A     5.2E-11    2.6E+01    N/A     N/A     5.2E-11    2.6E+01    N/A     N/A     5.2E-11    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     N/A     N/A     N/A     N/A<!--</td--><td>W/A     W/A     W/A    W/A    -</td><td>N/A     0.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     0.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E-11    2.6E+01    N/A     1.8E-09    2.6E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E+11    2.6E+01    N/A     N/A     6.9E+01    N/A     N/A     1.3E+01    N/A     N/A     1.3E+01    N/A   </td><td>N/A    N/A   </td><td>N/A   </td><td>N/A   </td><td>W/A    W/A    W</td><td>WA    WA    <td< td=""><td>WA    WA    WA    WA    WA    WA    HA    <td< td=""><td>WA     WA     WA    <td< td=""><td>WA     WA     WA    <td< td=""><td>WA   </td><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA   </td><td>WA     WA     WA   </td><td>WA   </td><td>WA    WA    <td< td=""></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td></td> | N/A   9,0E-10  2,6E+01  N/A   N/A   1,8E-09  2,6E+01  N/A     N/A   9,0E-10  2,6E+01  N/A   N/A   1,8E-09  2,6E+01  N/A     N/A   N/A   N/A   N/A   6,9E-11  2,6E+01  N/A     N/A   N/A   N/A   6,9E-11  2,6E+01  N/A     N/A   N/A   N/A   6,9E-11  2,6E+01  N/A     N/A   N/A   N/A   1,3E-10  2,6E+01  N/A | N/A     N/A     N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     9.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     N/A     N/A     6.9E-11    2.6E+01    N/A       N/A     N/A     N/A     6.9E-11    2.6E+01    N/A       N/A     N/A     N/A     6.9E-11    2.6E+01    N/A     6.9E-11    2.6E+01    N/A     6.9E-11    2.6E+01    N/A     5.2E-11    2.6E+01    N/A     1.3E-10    2.6E+01    N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A | N/A     N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     9.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     N/A     N/A     6.9E-11    2.6E+01    N/A       N/A     N/A     N/A     6.9E-11    2.6E+01    N/A     6.9E-11    2.6E+01    N/A     6.9E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     5.2E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     8.9E-11    2.6E+01    N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A <td>N/A     N/A     N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     9.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     N/A     N/A     6.9E-11    2.6E+01    N/A     1.8E-09    2.6E+01    N/A     1.8E-09    2.6E+01    N/A     1.8E-09    2.6E+01    N/A     1.7     6.9E-11    2.6E+01    N/A     1.3E-12    2.6E+01    N/A     N/A     &lt;</td> <td>N/A     N/A     N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     9.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     N/A     N/A     6.9E-11    2.6E+01    N/A     N/A     5.2E-11    2.6E+01    N/A     N/A     5.2E-11    2.6E+01    N/A     N/A     5.2E-11    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     N/A     N/A     N/A     N/A<!--</td--><td>W/A     W/A     W/A    W/A    -</td><td>N/A     0.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     0.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E-11    2.6E+01    N/A     1.8E-09    2.6E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E+11    2.6E+01    N/A     N/A     6.9E+01    N/A     N/A     1.3E+01    N/A     N/A     1.3E+01    N/A   </td><td>N/A    N/A   </td><td>N/A   </td><td>N/A   </td><td>W/A    W/A    W</td><td>WA    WA    <td< td=""><td>WA    WA    WA    WA    WA    WA    HA    <td< td=""><td>WA     WA     WA    <td< td=""><td>WA     WA     WA    <td< td=""><td>WA   </td><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA   </td><td>WA     WA     WA   </td><td>WA   </td><td>WA    WA    <td< td=""></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td> | N/A     N/A     N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     9.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     N/A     N/A     6.9E-11    2.6E+01    N/A     1.8E-09    2.6E+01    N/A     1.8E-09    2.6E+01    N/A     1.8E-09    2.6E+01    N/A     1.7     6.9E-11    2.6E+01    N/A     1.3E-12    2.6E+01    N/A     N/A     < | N/A     N/A     N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     9.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     N/A     N/A     N/A     6.9E-11    2.6E+01    N/A     N/A     5.2E-11    2.6E+01    N/A     N/A     5.2E-11    2.6E+01    N/A     N/A     5.2E-11    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     1.3E-10    2.6E+01    N/A     N/A     N/A     N/A     N/A     N/A </td <td>W/A     W/A     W/A    W/A    -</td> <td>N/A     0.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     0.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E-11    2.6E+01    N/A     1.8E-09    2.6E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E+11    2.6E+01    N/A     N/A     6.9E+01    N/A     N/A     1.3E+01    N/A     N/A     1.3E+01    N/A   </td> <td>N/A    N/A   </td> <td>N/A   </td> <td>N/A   </td> <td>W/A    W/A    W</td> <td>WA    WA    <td< td=""><td>WA    WA    WA    WA    WA    WA    HA    <td< td=""><td>WA     WA     WA    <td< td=""><td>WA     WA     WA    <td< td=""><td>WA   </td><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA   </td><td>WA     WA     WA   </td><td>WA   </td><td>WA    WA    <td< td=""></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td> | W/A     W/A    W/A    - | N/A     0.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     0.0E-10    2.6E+01    N/A     N/A     1.8E-09    2.6E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E-11    2.6E+01    N/A     1.8E-09    2.6E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E+01    N/A       N/A     0.4E-11    2.6E+01    N/A     N/A     6.9E+11    2.6E+01    N/A     N/A     6.9E+01    N/A     N/A     1.3E+01    N/A     N/A     1.3E+01    N/A | N/A    N/A | N/A     | N/A     | W/A    W | WA    WA <td< td=""><td>WA    WA    WA    WA    WA    WA    HA    <td< td=""><td>WA     WA     WA    <td< td=""><td>WA     WA     WA    <td< td=""><td>WA   </td><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA   </td><td>WA     WA     WA   </td><td>WA   </td><td>WA    WA    <td< td=""></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td<> | WA    WA    WA    WA    WA    WA    HA    HA <td< td=""><td>WA     WA     WA    <td< td=""><td>WA     WA     WA    <td< td=""><td>WA   </td><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA   </td><td>WA     WA     WA   </td><td>WA   </td><td>WA    WA    <td< td=""></td<></td></td<></td></td<></td></td<></td></td<></td></td<></td></td<> | WA     WA <td< td=""><td>WA     WA     WA    <td< td=""><td>WA   </td><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA   </td><td>WA     WA     WA   </td><td>WA   </td><td>WA    WA    <td< td=""></td<></td></td<></td></td<></td></td<></td></td<></td></td<> | WA     WA <td< td=""><td>WA   </td><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA   </td><td>WA     WA     WA   </td><td>WA   </td><td>WA    WA    <td< td=""></td<></td></td<></td></td<></td></td<></td></td<> | WA      | WA    WA <td< td=""><td>WA    WA    <td< td=""><td>WA    WA    <td< td=""><td>WA   </td><td>WA     WA     WA   </td><td>WA   </td><td>WA    WA    <td< td=""></td<></td></td<></td></td<></td></td<> | WA    WA <td< td=""><td>WA    WA    <td< td=""><td>WA   </td><td>WA     WA     WA   </td><td>WA   </td><td>WA    WA    <td< td=""></td<></td></td<></td></td<> | WA    WA <td< td=""><td>WA   </td><td>WA     WA     WA   </td><td>WA   </td><td>WA    WA    <td< td=""></td<></td></td<> | WA      | WA     WA | WA      | WA    WA <td< td=""></td<> |

) N

DATE18-Aug-87 PAGE4

TABLE 6-7 (Continued)

10000000

HANDLING ACCIDENTS - COLLOCATION PROCESSING OFTION (AIR TRANSFORT) (PER CONTAINER DR PALLET)

### Accident Frequencies and Range Factors

| RANGE<br>Factor |         |         | ł        | : 1                | ;        | !            | !       | 1       | ;       | ł       | ;       | ł       | ;          | {       | 1         | !       | :       | ;       | ł       | :          | ;       | ;       | !       | ;       | !       |
|-----------------|---------|---------|----------|--------------------|----------|--------------|---------|---------|---------|---------|---------|---------|------------|---------|-----------|---------|---------|---------|---------|------------|---------|---------|---------|---------|---------|
| UMDA<br>FREQ    | 47 M    |         |          | 4/N                |          | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A       | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A     | N:A     | N.'A    | N/A     |
| RANGE<br>Factor |         | 10120.1 | 10120.2  | 2.0E+UI<br>7 4E+01 | 2,65+01  |              | 1       | :       | :       | :       | ł       | ł       | ;          | ;       | ;         | ;       | ţ       | 2.6E+01 | 2.6E+01 | 2.6E+01    | 2.6E+01 | 2.65+01 | 2.6E+01 | 1.0E+01 | 1 05401 |
| IEAD<br>Freq    | 11 22 1 |         |          | 7 76-12            | 7.7E-12  | 0.0E+00      | 0.0E+00 | 0.0E+00 | 0.0E+00 | 0.0E+00 | 0.0E+00 | 00+30°0 | 0.0E+00    | 0.01400 | 0,01430.0 | 0.0E+00 | 0.0E+00 | 9.0E-09 | 60-30°6 | 1.8E-08    | 1.86-08 | 6.5E-11 | 6.5E-11 | 1.01-03 | 1 05-03 |
| RANGE<br>Factor |         | :       | :        | 1                  | 1        | :            | ł       | :       | :       | ł       | :       | :       | ;          | :       | ;         | :       | ;       | :       | :       | ;          | ;       | !       | :       | :       | :       |
| FUDA<br>FREQ    |         |         |          |                    | A/A      | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A       | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A     | N / D   |
| RANGE<br>Factor |         |         | ł        | : 1                | :        | :            | 1       | ł       | ł       | ł       | ;       | ;       | :          | ;       | ł         | ;       | ;       | ;       | ł       | ł          | 1       | ł       | !       | !       | ;       |
| PBA<br>Freq     |         | H / H   |          | 8/8<br>8/8         | A /N     | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A       | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A     | N/D     |
| PANGE<br>FACTOR |         | ;       | 1        | 1                  | 1        | ;            | 1       | 1       | ł       | ł       | :       | 1       | ;          | ł       | ł         | :       | ;       | ;       | ;       | ţ          | ;       | ţ       | ;       | 1       | ;       |
| NAAP<br>Freg    |         | H / H   |          |                    | A/N      | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A       | N/A     | N/N     | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A     | N/A     |
| RANGE<br>FACTOR | 2 15.01 | 1012012 | 111120-7 | 2.0E 101           | 2.6E+01  | :<br>  }<br> | :       | 1       | ;       | ;       | ;       | :       | :          | ;       | ;         | :       | ;       | 2.6E+01 | 2.6E+01 | 2.6E+01    | 2.6E+01 | 2.6E+01 | 2.6E+01 | 1.0E+01 | 1.05+01 |
| LBAD<br>FRED    |         |         |          | 7 75-12            | 7. 7E-12 | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | 0° ()E +00 | 0.0E+00 | 0°.0E+00  | 0.0E+00 | 0.0E+00 | 9.0E-09 | 9.0E-09 | 1.85-08    | 1.95-08 | 6.5E-11 | 6.5E-11 | 1.0E-03 | 1_0F-03 |
| RANGE<br>Factor |         |         | ł        | : :                | :        | ;            | 1       | 1       | ;       | :       | ;       | !       | ;          | 1       | :         | ;       | :       | ł       | :       | 1          | ;       | !       | ;       | !       | ;       |
| AP6<br>Freq     | ¢, h    | 4/2     | H /H     | 4/H                | A/N      | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | 0.0E+00 | N/A        | N/A     | N/A       | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A     | N,A     | N/A     | N/A     |
| RANGE<br>Factor |         |         | :        | : :                |          | :            | ł       | ţ       | 1       | ;       | :       | 1       | :          | :       | ;         | •       | !       | ;       | 1       | 1          | ;       | 4       | ł       | ł       | ;       |
| ANAD<br>FREQ    |         |         |          |                    | N/A      | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A       | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A     | N / A   | N/A     | N/A     |
| ÛN              | Ļ       | 3 2     | 3 2      | G K                | 3 23     | 26           | 26      | 26      | 26      | 26      | 26      | 27      | 11         | 27      | 27        | 11      | 27      | 65      | 29      | 0 <u>:</u> | 30      | 15      | 15      | 12      | ;       |
| SCEN-<br>Ario   | JUARU   | UNEUC   |          | HREFC              | HARVE    | HANHE        | HAFHC   | HAFVC   | HAGGC   | HAREC   | HARVC   | HAR HF  | HAFHC      | HAFVC   | HAQGC     | HAFGC   | HARVE   | HARGC   | JUREN   | HARGC      | HARVC   | HARGC   | HARVE   | HARGS   | MARVS   |

La contrata de



The activities associated with the handling of the ton containers at the original storage site (Aberdeen Proving Ground, APG) and at the receiving site (Johnston Atoll, JA) differ from those for other transport options. In brief, the Army's plan is to package the ton containers in vaults to protect them against impact, crush, puncture, fire, and immersion while being transported. At the sending site the munitions will be subjected to a few handling operations during their movement to the loading dock for offsite ship transportation. Regarding the receiving site (JA), this analysis addresses only the handling activities associated with the unloading of the vessel and of the lighters since information is lacking on the remaining process activities at the atoll.

The handling operation differences between the offsite transport modes are described in Section 6.2. Further details on the marine transport activities are as follows:

- 1. The vaults will be loaded onto a truck by forklift for transfer to the loading dock on the Bush River.
- 2. At the loading dock, a crane is used to lift the vault from the truck and to load it on the lighter.
- 3. Ten lighters will be towed to the ocean-going LASH vessel anchored in the deeper water of the Chesapeake Bay.
- The lighters will be loaded on the LASH vessel using a shipboard crane. The LASH vessel will then sail southward in the Chesapeake Bay.
- 5. At the receiving site (JA) the lighters and the vaults will be unloaded from the LASH vessel using the same handling equipment as the one used at the sending site.

Eight accident sequences were identified for the handling activities taking place at the sending site (APG). Only four accident sequences are identified for the receiving site, since this analysis stops after the lighter and the vault have been unloaded from the LASH vessel. Table 6-8 shows the list of accident sequences for both the sending and the receiving sites, since differences in agent release dispersion will apply depending on which site the accident occurs. The event tree models for the generic accident scenarios presented in Section 6.3 still apply. The additional event for the lighter drop accident consists only of the initiating event and the conditional probability that the vault will be crushed. The release sequence is designated as HW34.

Table 6-9 presents the input data used for the accident frequency analysis. The basis for the initiating events frequencies and for the conditional events probability have been discussed in Section 6.3. The probability of crush of the ton container is new and is discussed in the following. The supporting analysis is found in Appendix C.

The probability of crushing the ton container as a result of dropping the lighter containing the vault has been evaluated based on a simple one dimensional impact model. The results of this analysis indicates that if a lighter is dropped 70 ft to the bottom of the vessel during loading operations, the 56 vaults and ton containers at the bottom of the lighter will fail. In the case where a lighter is accidentally dropped on another lighter already on the vessel, no munition is expected to fail.

The resulting sequence frequencies and uncertainties are summarized in Table 6-10 for the marine transport option.

### TABLE 6-8

### ACCIDENT SCENARIOS FOR MARINE TRANSPORT OPTION HANDLING ACTIVITIES

storage area (no vault)

area

### Sending Site

| 11401 | prop of concerner at storage area (no value)                      |
|-------|-------------------------------------------------------------------|
| HW02  | Forklift collision with short duration fire at storage (no vault) |

- HW04 Forklift collision without fire at storage area (no vault)
- HW05 Drop of vault

Dean

- HW06 Forklift (or crane) collision with fire involving vault
- HW07 Forklift (or crane) collision without fire involving vault
- HW26 Collision accident (no vault) with prolonged fire leads to thermal detonation
- HW34 Drop of lighter while handled with shipboard crane (vault crush)

### Receiving Site

| HW05 | Drop | of | vault |  |
|------|------|----|-------|--|
|------|------|----|-------|--|

- HW06 Forklift (or crane) collision with fire involving vault
- HW07 Forklift (or crane) collision without fire involving vault
- HW34 Drop of lighter while handled with shipboard crane (vault crush)

|            |     |             | TABLE  | 6-9         |         |            |
|------------|-----|-------------|--------|-------------|---------|------------|
| INITIATING | AND | CONDITIONAL | EVENTS | FREQUENCIES | (MARINE | TRANSPORT) |

-

|             | INITIATING                                    | EVENT                                        |                            |                        |     | FREQUENCY<br>EVENTS/OP | ERROR<br>Factor | REFEREN                                         | 52  | APPLICABLE<br>SCENARIO |
|-------------|-----------------------------------------------|----------------------------------------------|----------------------------|------------------------|-----|------------------------|-----------------|-------------------------------------------------|-----|------------------------|
| (B)<br>HE10 | (C) (D)<br>Pallet or<br>handling<br>outside t | (E)<br>container<br>of non-lea<br>he MD8 (1) | (F)<br>dropped<br>king aun | (6)<br>during<br>ition | (H) | ([)<br>3.0E-06         | (J)<br>10.0     | (K)                                             | (L) | (M)<br>HOl,HOS,        |
| HESS        | Vehicle c                                     | ollision a                                   | ccident                    |                        |     | <b>4.3E~)6</b>         | 10.0            | GA derived<br>data, see<br>details in<br>Aoo. F |     | HO2,HO4,<br>HO26       |

### CONDITIONAL EVENTS PROBABILITIES

| 4 <b></b>        | EVENT SEQUENCE                                                                           | EVENT<br>PROBABILI   | ERROR<br>FAC. | REFERENCE | APPLICABLE<br>SCENARIO |
|------------------|------------------------------------------------------------------------------------------|----------------------|---------------|-----------|------------------------|
| HE100K           | Ton Container punctured given<br>a drop (drop height = 6ft.)                             | 3.34E-)3             | 2.0           |           | HOL                    |
| HEIIOK           | Vault and ton container punctured given a drop of vault                                  | 4.90E-04             | 3.0           |           | HO5,HO6,               |
| HE140K           | Ton container punctured given<br>a drop resulting from<br>collision (drop height = 2ft.) | 1.68E-03             | 3.0           |           | HO2,HO4                |
| HE 150K          | Ton container crushed given a drop of the lighter                                        | 1.00E+00             |               |           | H034                   |
| HE550            | Fire results from vehicle collision                                                      | 7.25E-02             | 10.0          |           | HO2,HO6,               |
| HE555            | Collision does not cause<br>fire                                                         | 9.27E-01             | none          |           | HQ4,HQ7                |
| HE360            | Fire contained within                                                                    |                      |               |           |                        |
| HE5608<br>HE560C | 30 ain - non-burstered aunition<br>>4 hrs - vault                                        | 1.00E+00<br>1.00E+00 | nane          |           | H02<br>H06             |
| 45570            | Fire at retained with a                                                                  |                      |               |           |                        |

HE570 Fire not contained within



TABLE 6-9 (Continued)

| HE5701 | 8 30 min - Non-burstered munition         | 0.00E+00 none | H026 |
|--------|-------------------------------------------|---------------|------|
| HE580  | Nunition ruptures given<br>prolonged fire | 1.00E+00      | H026 |





ዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀ
TABLE 6-10 Frequencies of Handling Accidents for Marine Transport Option Fage 1

21-Aug-87

|           |          |           | (1)          | KANGE              | FACTOR |                       | ;       | 1       | i        | ł       | ł        | ;       | 1        | ł        |
|-----------|----------|-----------|--------------|--------------------|--------|-----------------------|---------|---------|----------|---------|----------|---------|----------|----------|
|           |          |           | (5)          | ADAD               | FRED   |                       | N/A     | N/A     | N/A      | N/A     | N/A      | N/A     | N/A      | N/A      |
| )Li       |          |           | (R)          | RANGE              | FACTOR |                       | t       | ;       | ł        | 1       | ł        | ł       | ł        | ;        |
| STON ATC  |          |           | ( <b>0</b> ) | TEAD               | FRE®   |                       | N/A     | K/A     | N/A      | N/A     | N/A      | A/N     | N/A      | H/A      |
| TO JOHN   |          |           | (d)          | KANGE              | FACTOR |                       | 1       | ļ       | 1        | ł       | ł        | ł       | 1        | ł        |
| ROM AP6   |          |           | (j)          | FUDA               | FRED   |                       | N/A     | N/A     | N/A      | N/A     | N/A      | N/A     | N/A      | M/A      |
| AINER F   |          | (APG)     | (N)          | RANGE              | FACTOR | <br> <br> <br> <br>   | !       | ł       | ł        | 1       | ł        | ł       | ;        | 1        |
| TON CONT  | iner     | ng Site   | Û            | FBA                | FREQ   | 6<br>3<br>5<br>8<br>8 | N/A     | N/A     | N/A      | N/A     | N/A      | N/A     | N/A      | N/A      |
| ION FOR   | on conta | - Sendi   | (j           | RANSE              | FACTOR | 8<br>8<br>8<br>8      | 1       | ł       | :        | ;       | ł        | !       | ł        | ļ        |
| TION OPT  | t per t  | luencies  | ())          | NAAF               | FREQ   |                       | N/A     | N/A     | A/A      | N/A     | N/A      | N/A     | N/A      | 6/N      |
| ANSPORTAT | accident | lent Freu | (ľ)          | RANGE              | FACTOR |                       | ;       | ł       | ł        | 1       | 1        | 1       | ;        | ł        |
| ARGE TRI  | gnilbns  | hccu      | (1)          | LBAD               | FRED   |                       | N/À     | A/A     | N/A      | N/A     | N/A      | N/Ä     | N/A      | N/A      |
| ш         | -        |           | ŧ            | RANGE              | FACTOR |                       | 12.8    | 31.1    | 12.8     | 12.8    | 31.1     | 12.8    | ł        | 10.0     |
|           |          |           | (9)          | APG                | FRED   |                       | 2.0E-08 | 5.26-16 | 6.7E-09  | 5.9E-09 | 3. IE-10 | 3.95-09 | 0.0E+00  | 6. 0E-06 |
|           |          |           | (F)          | RANGE              | FACTOR |                       | ł       | ł       | !        | ;       | :        | ł       | ł        | ł        |
|           |          |           | (E)          | ANAD               | FREQ   |                       | N/A     | N/A     | N/A      | K/A     | N/R      | N/A     | N/A      | R/A      |
|           |          |           | 9            | NG.                |        | ł                     |         | e4      | Ŧ        | 67      | <b>.</b> | 1       | 25       | 5        |
|           |          |           | ទ្ធ          | - <mark>-</mark> - |        | -                     | 50      | сл      | <b>س</b> |         | сл       | ري.     | <b>س</b> | m        |
|           |          |           | (8)          | SCEK               | ARIO   |                       |         | Ξ ŵΞ    | Hitt     | H 19    | HAR      | Hakh    | HARH     | 1 41     |

Accdent Freq. Receiving Site

| FANSE<br>Factor |   | ( <u>-</u> ) | 12.8       | 1.1     | а<br>С  | t<br>f | 10.0            |
|-----------------|---|--------------|------------|---------|---------|--------|-----------------|
| JA<br>FRE       |   | (E)          | 2.95-09    | 3.1E-10 | 3.56-09 | 4/4    | <b>6.</b> 0E-05 |
|                 |   |              |            |         |         |        |                 |
| .ON             | : | 9            | <b>L</b> J | iJ      | r       | 26     | ×.              |
| 0P. ND.         |   | (C) (D)      | ני         | ن.      | 1       | 26     | <b>1</b>        |

÷

6-82

6.7. REFERENCES

- 6-1. "Transportation of Chemical Agent and Munitions: A Concept Plan," Final Report prepared for PEO-PM Cml Demil by MITRE Corp., 30 June 1987.
- 6-2. List of Assumptions in letter from GA to OPMCM dated April 29, 1987.
- 6-3. "Probabilities of Selected Hazards in Disposition of M55 Rockets,"
  M55-CS-02, USATHAMA, September 22, 1985.
- 6-4. "Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Applications," NUREG/CR-1278, U.S. Nuclear Regulatory Commission, August 1983.

## 7. SCENARIO LOGIC MODELS FOR PLANT OPERATIONS

#### 7.1. INTERNAL EVENTS

a\*\$\_a\*\$\_a\_1\_a\*\$\_a\*\$\_a\*b\_a\*b\_a\*b\_a\*

The discussion presented in the following paragraphs and the discussion and figures presented in Sections 7.1.1 through 7.1.4 have been taken from documentation provided by JBF Associates, Inc. with only minor editing. The material presented in Section 7.1.5 is based on material supplied by JBF Associates, Inc. but has been augmented by GA to address explosions occurring in the incineration systems.

The development of plant operations accident scenarios involved systematically evaluating each functional area of the plant to identify initiating events which, if unchecked, could lead to agent releases above the screening thresholds set by MITRE. Then, for each initiating event, possible successes and failures of the plant systems that have the potential to check the release of agent were considered. Event trees were used to identify the possible modes of accident progression.

All of the initiating events considered for the analysis of internal events are in the following categories:

- 1. Agent spills.
- 2. Detonations.
- 3. Fires.
- 4. Process upsets.

Accidents initiated by external events are discussed in Section 7.2.

Event trees show the possible modes of accident progression. The events included in the event trees are successes and failures of functions (plant systems and/or operator actions) designed to prevent agent releases. The plant systems considered include the ventilation/ filtration systems, the fire suppression systems, the explosion containment system, and the process control systems.

Each event tree contains a statement of the initiating event at the top, on the left-hand side. The functions that can limit agent releases are listed across the top of the event tree. The event tree branches at each function. The upward path at each branch is success (yes, the stated function worked) and the downward path is failure (no, the stated function did not work).

The order in which the functions are considered is specified by the analyst according to the order in which functions are challenged unless logical considerations of the analysis dictate otherwise. An example of a case where logical considerations dictate the listed order of a function involves the ventilation system. The ventilation system is challenged immediately whenever agent is released within the plant. However, the ventilation system is considered last on most of the event trees because (1) its function may be irrelevant (e.g., if the building integrity is lost because of a fire or explosion, agent will be released regardless of whether the ventilation system works) and (2) its failure probability is a function of other conditions that may develop (e.g., a large fire may saturate the ventilation system's filters, thus increasing the probability the ventilation system fails to prevent an agent release).

The last consideration stated above applies generally to each branch in the event tree; the failure probability of each function depends on the specific conditions implied by the path that leads to a challenge of the function. In other words, the probabilities of success and failure at each branch point in the event tree are conditioned on

7-2

the occurrence of the initiating event and the successes or failures of the preceding functions along the path that leads to the challenge of the function being considered. That is why some of the event tree functions are assigned different failure probabilities within the same tree; they are challenged on different paths of the tree.

Some scenarios were screened from the analysis based on frequency considerations. If the product of the initiating event frequency and conservative estimates of the failure probabilities of plant safety systems for a scenario is less than  $10^{-10}$ /year, that scenario was screened from further consideration (Ref. 7-1). (The initiating event frequencies and system failure probabilities used for screening are shown on the event trees.) Other scenarios were screened based on successful operation of plant safety systems preventing significant agent releases.

Each accident scenario on each event tree is labeled with a "C" if it has been screened based on low consequence, an "F" if it has been screened based on low frequency, or a scenario identified if it is being analyzed.

A discussion of the data, and its basis, used in quantifying the fault trees and event trees is provided in Section 9.1.

#### 7.1.1. Explosive Containment Room Vestibule and Munitions Corridor

The analysis reported in this section examined potential release scenarios that could occur in the Explosive Containment Room Vestibule (ECV) or Munitions Corridor. These scenarios all involve damage to one or more munitions or containers of agent with subsequent catastrophic failure of the building structure or ventilation system. This analysis considered the following types of initiating events:

- Simple spills of munitions that would create an evaporative source of agent greater than the screening thresholds discussed earlier.
- 2. Detonations of munitions that would result in a source of agent vapor greater than the screening thresholds.
- Fires that cause rupture or damage of munitions, thereby creating a source of agent greater than the screening thresholds.

For Type 1 initiators, spills of one or two of each munition or container type were analyzed. For all munitions, it was assumed that spills of more than two at a time will not occur. It was also assumed that all processing operations will make use of two identical conveyor lines. Upsets that cause munition damage in both lines will most likely be detected immediately by some of the many sensors that monitor the system status on a continuous basis. Early detection should result in shutdown of the conveyor lines before additional munitions are damaged.

The principal mechanisms considered for munition spills in the ECV/ Munitions Corridor include falls of munitions from the conveyors, resulting in puncture damage to the casings, and equipment failures (e.g., failures of conveyor stops or control system logic) that cause the munitions to fall from the conveyors.

For Type 2 initiators, detonations of one of each munition type that contains explosive components were analyzed. The principal mechanism considered for detonations in the ECV/Munitions Corridor includes falls of munitions from the conveyor with detonation on impact. A detonation of

A detonation of an 8-in. projectile in the ECV will cause failure of the building and direct release of agent vapors to the environment.

Type 3 initiators were not analyzed. A fire of sufficient intensity and duration to rupture or detonate a munition or agent container is not credible for the ECV/Munitions Corridor due to the low inventory of combustibles in these areas. However, there are ignition sources in these areas (e.g., motors and cables). Therefore, scenarios involving fire subsequent to an agent spill or munition detonation were considered.

The event trees developed for initiating events in the ECV with estimated frequencies above the screening threshold are shown in Figs. 7-1 through 7-4. The ventilation system event was quantified using the fault tree presented in Fig. 7-5. Table 7-1 defines the event tree functions.

The following is a summary of the assumptions used in developing these event trees:

- All processing operations will use two identical conveyor lines.
- Upsets that cause munition damage in two conveyor lines at once will be detected immediately.
- 3. Detonations of 8-in. projectiles in the ECV will cause structural failure of the MDB, resulting in direct agent release to the environment, based on performed analysis.
- 4. A fire (as an initiating event) of sufficient intensity and duration to rupture or detonate a munition or agent container is not credible for the ECV/Munitions Corridor due to the low combustible inventory in these areas.

7-5

.



5

N. 6. 6' 1. 4'

110.00

1.01

Fig. 7-1. Event tree for spill of munition(s) in the ECV

• •

| _     |
|-------|
| dian. |
|       |
| 10 C  |

| SEQUENCE                                |           | C    |      | F    | 3                                     |     | F                                | P045                    |   |
|-----------------------------------------|-----------|------|------|------|---------------------------------------|-----|----------------------------------|-------------------------|---|
| VENTILATION<br>System                   | VENT      | ~1.0 |      | 10-9 | ~1.0                                  |     | 10-9                             |                         |   |
| STRUCTURE<br>Contains<br>Fire           | ECV-SFIRE |      |      |      |                                       | 0.9 |                                  | 10-1                    |   |
| AVOID                                   | ECV-A1    |      | ~1.0 |      |                                       |     | ▼ FAILURE<br>10 <sup>-3</sup> GB | 10 <sup>-4</sup> VX, HD | - |
| SPILL OF ONE<br>TON CONTAINER<br>IN ECV | ECV-ST1   |      |      |      | 4 × 10 <sup>-5</sup> YR <sup>-1</sup> |     |                                  | -                       |   |

Fig. 7-2. Event tree for spill of one ton container in ECV

B282-38-3

2.52.000

1357-775 •

| SPILL OF TWO<br>TON CONTAINERS<br>IN ECV<br>IN ECV<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV-ST2<br>ECV- |                                          |           | 0    |      | ۳.   | <u>ц</u>                   |         | u<br>               |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------|-----------|------|------|------|----------------------------|---------|---------------------|
| SPILL OF TWO<br>TON CONTAINERS<br>IN ECV<br>IN ECV<br>ECV-ST2<br>ECV-A1<br>ECV-SFIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>CONTAINS<br>FIRE<br>FIRE<br>FIRE<br>FIRE<br>FIRE<br>FIRE<br>FIRE<br>FIRE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | VENTILATION                              | VENT      | ~1.0 |      | 10-9 | ~1.0                       |         |                     |
| SPILL OF TWO<br>TON CONTAINERS<br>IN ECV<br>ECV-A1<br>ECV-A1<br>ECV-A1<br>-1.0<br>-1.0<br>B x 10 <sup>-8</sup> YR -1<br>falLURE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | STRUCTURE<br>Contains<br>Fire            | ECV-SFIRE |      |      |      |                            | 0.9     |                     |
| SPILL OF TWO<br>TON CONTAINERS<br>IN ECV<br>ECV-ST2<br>B x 10 <sup>-8</sup> YR <sup>-1</sup>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | AVOID<br>IGNITION                        | ECV-A1    |      | ~1.0 |      |                            | FAILURE | 10 <sup>-3</sup> GB |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | SPILL OF TWO<br>TON CONTAINERS<br>IN ECV | ECV-ST2   |      |      |      | 8 × 10 <sup>-6</sup> YR -1 |         |                     |

3

Event tree for spill of two ton container in ECV Fig. 7-3.

Sector and the sector of the s

June 1

| DETONATION OF<br>ONE BURSTERED<br>MUNITION<br>IN ECV                                             | FIRE<br>AVOIDED                                   | PROPAGATION<br>AVOIDED                              | SEQUENCE<br>ID |
|--------------------------------------------------------------------------------------------------|---------------------------------------------------|-----------------------------------------------------|----------------|
| ECV-D81                                                                                          | ECV-A1B                                           | ECV-PR                                              |                |
|                                                                                                  | 0.5 ROCKETS<br>0.99 OTHERS                        |                                                     | <b>P04</b> 6   |
| FREQUENCY, YR <sup>-1</sup>                                                                      | SUCCESS                                           | 0.99 ROCKETS<br>0.9 OTHERS                          | P047           |
| R: $3 \times 10^{-7}$<br>M: $4 \times 10^{-7}$<br>Q: $3 \times 10^{-7}$<br>P: $6 \times 10^{-7}$ | FAILURE<br>0.5 ROCKETS<br>10 <sup>-2</sup> OTHERS | 10 <sup>-2</sup> ROCKETS<br>10 <sup>-1</sup> Others | P048           |
| C: 1 x 10 <sup>-0</sup>                                                                          |                                                   |                                                     |                |

Fig. 7-4. Event tree for detonation of burstered munition in ECV

<u>ĸĔĸĔŀŔĸĔŀĔĸĔĸĔĸĔĸĔŧŔĊŀĹĸĔŀĊĸĔŀĊĔŀĊĔĸĔĸŎŀĊŎ</u>ĿĊŎĿŎĿŎĿŎĿŎĿŎĿŎĿŎĿŎĿŎĿ





|                 | EVENTS CONSIDERE                                                 | TABLE 7-1<br>D FOR THE ECV/MUNITIONS CORRIDOR                                                                                                                                                                                                                             |
|-----------------|------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                 | Event                                                            | Description                                                                                                                                                                                                                                                               |
|                 | Spill of one rocket in ECV<br>(ECV-SR1)                          | One rocket falls off the conveyor due to a process upset or improper loading and is punctured. The spill is not cleaned up in 1 h.                                                                                                                                        |
|                 | Spill of one mine in ECV<br>(ECV-SM1)                            | One mine falls off the input conveyor due<br>to a process upset or improper loading and<br>is punctured. The spill is not cleaned up<br>in 1 h.                                                                                                                           |
|                 | Spill of two bombs in ECV<br>(ECV-SB2)                           | One tray of bombs falls off a bypass con-<br>veyor due to improper loading or switch<br>failures that prevent the conveyor stop<br>from being raised until the charge can<br>arrives. The bombs are punctured, and<br>the spill is not cleaned up in 1 h.                 |
| íants<br>●<br>C | Spill of one ton container<br>in ECV (ECV-ST1)                   | One ton container falls off a bypass con-<br>veyor due to improper loading or switch<br>failures that prevent the conveyor stop<br>from being raised until the charge car<br>arrives. The container is punctured,<br>and the spill is not cleaned up in 1 h.              |
|                 | Spill of two ton containers<br>in the ECV/COR (ECV-ST2)          | One ton container on each line is damaged<br>when a control system failure prevents the<br>stops at the ends of the bypass conveyors<br>from being raised when the charge car is<br>unavailable. The containers are punctures,<br>and the spill is not cleaned up in 1 h. |
|                 | Detonation of one rocket in<br>ECV (ECV-DR1)                     | A rocket falls off the input conveyor and detonates.                                                                                                                                                                                                                      |
|                 | Detonation of one mine in<br>ECV (ECV-DM1)                       | A mine falls off the input conveyor and detonates.                                                                                                                                                                                                                        |
|                 | Detonation of one 8-in.<br>projectile in ECV (ECV-D81)           | A projectile falls off a conveyor and detonates.                                                                                                                                                                                                                          |
|                 | Detonation of one 105-mm<br>projectile in ECV/COR<br>(ECV-D1051) | A projectile falls off a conveyor and detonates.                                                                                                                                                                                                                          |
| Ŷ               |                                                                  |                                                                                                                                                                                                                                                                           |
|                 |                                                                  | 7-11                                                                                                                                                                                                                                                                      |

. De se de la de

TABLE 7-1 (Continued)

| Event                                                            | Description                                                                                                                                                                                               |
|------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Detonation of one 155-mm<br>projectile in ECV/COR<br>(ECV-D1551) | A projectile falls off a conveyor and detonates.                                                                                                                                                          |
| Avoid ignition (ECV-AI)                                          | Failure on this event tree branch implies<br>ignition of an agent spill. Motors and<br>cables are potential ignition sources in<br>the ECV.                                                               |
| Avoid ignition (ECV-AIB)                                         | Failure on this event tree branch implies<br>ignition of agent vapors and/or liquid<br>agent spills following a munition detona-<br>tion. Motors and cables are potential<br>ignition sources in the ECV. |
| Ventilation system (VENT)                                        | Failure on this event tree branch implies<br>a release of agent through the ventilation<br>system due to (1) duct failure or (2) fil-<br>ter failures. (See fault tree in<br>Fig. 7-5.)                   |
| Propagation Avoided<br>(ECV-PROP)                                | Failure on this event tree branch implies<br>that fragments from a detonated munition<br>hit other munitions in the ECV or the<br>unpack area causing additional agent<br>spillage.                       |

55

P



5. Fires in the ECV/Munitions Corridor will not be suppressed since there are no fire suppression systems and personnel will not be sent in to fight a fire.

### 7.1.2. Munition Processing Systems

The analysis reported in this section examined potential failures involving all seven of the munitions processing systems. They include:

- Mine machine (MIN).
- Rocket shear machine (RSM).
- Rocket punch and drain station (RDS).
- Projectile/mortar disassembly machine (PMD).
- Burster size reduction (BSR) machine.
- Bulk drain station (BDS).

Multipurpose demilitarization machine (MDM).

This evaluation assumed that the machines are capable of processing munitions at designed rates by completely draining agent and disassembling munitions. Also, any situation that prevents the machines from attaining those design parameters requires that the machine be shut down.

Based on these assumptions the following types of events were evaluated for each machine:

- Simple spills of munitions that would create an evaporative source of agent greater than the screening thresholds discussed earlier.
- Detonations of munitions that would result in a source of agent vapor greater than the screening thresholds.

3. Fires that cause rupture or damage of munitions, thereby creating a source of agent greater than the screening thresholds.

For Type 1 initiators, spills of one or more of each munition or container type were analyzed. The mechanisms considered for munition spills in the ECR or MPB include (1) random falls of munitions from the conveyors, resulting in puncture damage to the casings, (2) equipment failures (e.g., failures of conveyor stops or control system logic) that cause the munitions to fall from the conveyors, and (3) equipment failures (e.g., shearing of a partially drained rocket) that cause munitions to be processed improperly.

For Type 2 initiators, detonations of one of each munition type that contains explosive components were analyzed. The mechanisms considered for detonations in the ECR or MPB include (1) falls of munitions from the conveyor with detonation on impact and (2) process upsets or equipment failures (e.g., loss of water spray during rocket shearing) that cause munitions to be processed improperly.

It was assumed that the ECR is likely to contain a blast within its confines since it is designed and constructed to do so.

Type 3 initiators were not analyzed. A fire of sufficient intensity and duration to rupture or detonate a munition or agent container is not credible for the ECR or MPB due to the low inventory of combustibles in these areas. However, there are ignition sources in these areas (e.g., motors and cables). Therefore, scenarios involving fire subsequent to an agent spill or munition detonation were considered.

Figures 7-6 through 7-8 show the event trees for the munitions processing systems. The ventilation system event was quantified using

554 STORES DESCRIPTION

Fig. 7-6. Event tree for spill of munition(s) in the ECR

| SEQUENCE<br>1D                             |          | U    |      | u.                          | ť                                            |                                                                      | u                   | U    |      | Ľ    |  |
|--------------------------------------------|----------|------|------|-----------------------------|----------------------------------------------|----------------------------------------------------------------------|---------------------|------|------|------|--|
| VENTILATION<br>System                      | ECR-VS   | ~1.0 |      | 10 <sup>-9</sup>            | ~1.0                                         |                                                                      | 10-9                | ~1.0 |      | 10-9 |  |
| NOISSEUR                                   | ECR-SUPP |      |      |                             |                                              | ~1.0                                                                 |                     |      | 10-2 |      |  |
| AVOID                                      | ECR-A1   |      | ~1.0 |                             |                                              | FAILURE                                                              | 10-3                |      |      |      |  |
| SPILL OF ONE<br>OR TWO MUNITIONS<br>IN ECR | ECR-PROJ |      |      | FREQUENCY, YR <sup>-1</sup> | 1M: 10 <sup>-1</sup><br>2M: 10 <sup>-2</sup> | 10: 10 <sup>-1</sup><br>20: 10 <sup>-2</sup><br>1R: 10 <sup>-2</sup> | 2R: 10 <sup>3</sup> |      |      |      |  |

\*\*\*\*

•

7-15

5.5

Event tree for detonation of burstered munitions in the ECR F1g. 7-7.

DAN ZEREN KEREN SERVER SERVER



G

Contraction of the second s





Event tree for spill of munition in MPB F1g. 7-8.

 $\sim \kappa$ 

sa kaada baaada baadad sanada kaadada kaadaa kaadada kaada kaada haasaa haadaa haada haada haada haada

7-17

 $\mathbf{x}$ 

Ŵ

the fault tree presented in Fig. 7-5. Table 7-2 defines the event tree functions.

The following is a summary of the assumptions we made in developing these event trees:

- 1. All processing operations will use two identical conveyor lines.
- Upsets that cause munition damage in two conveyor lines at once will be detected immediately.
- Agent reservoirs within munitions are at or near atmospheric pressure.

## 7.1.3. Buffer Storage Area

The analysis reported in this section examined potential release scenarios that could occur in the Buffer Storage Area (BSA) on the first floor of the MDB. The BSA contains only conveyors that hold drained munitions and containers (projectiles, cartridges, bombs, and ton containers) awaiting decontamination in the Metal Parts Furnace. The only items that will contain a significant amount of residual agent after being drained are ton containers. These containers could contain 75 to 85 lb of residual agent. Therefore, the spill of one drained ton container in the BSA was analyzed. To account for the chance that an undrained munition or container could be in the BSA (due to failures in the Bulk Drain Station or the Multipurpose Demilitarization Machine), the spill of one full ton container was also analyzed. Other undrained munitions could also spill their contents in the BSA; the full ton container was selected as a representative worst-case spill for this area.



. \$`\$. \$`\$.\$"\$. \$"\$. \$"\$(\$"\$)`\$"\$`\$"\$`\$"\$."\$"\$"

5./10.445.446.360.87.616.616.61

| Event                                          | Description                                                                                                                                       |
|------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------|
| Spill of mine in ECR<br>(ECR-MIN)              | Any process upset resulting in the release<br>of the agent inventory of a mine in the<br>ECR.                                                     |
| Avoid ignition (ECR-AI)                        | Failure on this event tree branch implies<br>ignition of an agent spill. Motors and<br>cables are potential ignition sources in<br>the ECR.       |
| Suppression (ECR-SUPP)                         | Failure on this event tree branch implies<br>that the dampers for inlet ventilation to<br>the ECR do not close.                                   |
| Ventilation system (ECR-VS)                    | Failure on this event tree branch implies<br>a release of agent through the ventilation<br>system.                                                |
| Spill of two mines in ECR<br>(ECR-MINES)       | Any process upset resulting in the release<br>of the agent inventory of two or more mines<br>in the ECR.                                          |
| Spill of 8-in. projectile<br>in ECR (ECR-PROJ) | Any process upset resulting in the release<br>of the agent inventory of 8-in. projectiles<br>in the ECR.                                          |
| Spill of projectiles in ECR<br>(ECR-PROJS)     | Any process upset resulting in the release<br>of the agent inventory of 8-in. projectile<br>in the ECR.                                           |
| Spill of rocket in ECR<br>(ECR-ROC)            | Any process upset resulting in the release<br>of the agent inventory of a rocket in the<br>ECR.                                                   |
| Spill of rockets in ECR<br>(ECR-ROCS)          | Any process upset resulting in the release<br>of the agent inventory of two or more<br>rockets in the ECR.                                        |
| Detonation of mine(s) in<br>ECR (ECR-DM1)      | Any process upset resulting in the detona-<br>tion of one or more mines in the ECR.                                                               |
| Structure contains blast<br>(ECR-BLAST)        | Failure on this branching operator implies<br>that the walls, ceilings, blast dampers, or<br>blast gates of the ECR are breached by the<br>blast. |

7-19

TABLE 7-2 (Continued)

| Event                                           | Description                                                                                                                                                                                |
|-------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Avoid ignition (ECR-AIB)                        | Failure on this event tree branch implies<br>ignition of agent vapors and/or liquid<br>agent spills following a munition detona-<br>tion.                                                  |
| Detonation of projectile(s)<br>in ECR (ECR-DP1) | Any process upset resulting in the detona-<br>tion of one or more projectiles in the ECR.                                                                                                  |
| Detonation of rocket(s) in<br>ECR (ECR-DR1)     | Any process upset resulting in the detona-<br>tion of one or more rockets in the ECR.                                                                                                      |
| Spill of bulk item in MPB<br>(MPB-BULK)         | Any process upset resulting in the release<br>of the agent inventory of a bulk item in<br>the MPB.                                                                                         |
| Avoid ignition (MPB-AI)                         | Failure on this event tree branch implies<br>ignition of an agent spill. Motors and<br>cables are potential ignition sources in<br>the MPB.                                                |
| Suppression (MPB-SUPP)                          | Failure on this event tree branch implies<br>that the fire brigade does not successfully<br>extinguish a fire in the MPB by spraying it<br>with either decon solution or CO <sub>2</sub> . |
| Ventilation system (MPB-VS)                     | Failure on this event tree branch implies<br>a release of agent through the ventilation<br>system.                                                                                         |
| Spill of bulk items in MPB<br>(MPB-BULKS)       | Any process upset resulting in the release<br>of the agent inventory of two or more bulk<br>items in the MPB.                                                                              |
| Spill of 8-in. projectile<br>in MPB (MPB-PROJ)  | Any process upset resulting in the release<br>of the agent inventory of an 8-in. projec-<br>tile in the MPB.                                                                               |
| Spill of projectiles in MPB<br>(MPB-PROJS)      | Any process upset resulting in the release<br>of the agent inventory of two or more pro-                                                                                                   |

7-20



110.00

TABLE 7-2 (Continued)

| Event                    | Description                                                                                                                               |  |  |  |
|--------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| Avoid ignition (MPB-AIB) | Failure on this event tree branch implies<br>ignition of agent vapors and/or liquid<br>agent spills following a munition detona-<br>tion. |  |  |  |





Ř. M Spills in the BSA can result from a ton container falling off the conveyor. For this analysis, it was assumed that the full container is punched at the Bulk Drain Station but not drained. Therefore, no puncture is required to release its contents.

Scenarios in which fires cause a release from the BSA were not analyzed since there are no combustibles in this area for sustaining a fire. However, there are ignition sources (motors and cables), so the possibility of an agent spill igniting was considered.

The event tree developed for the BSA 15 shown in Fig. 7-9. Descriptions of the events included in this tree are in Table 7-3. The ventilation event tree branch was quantified using the fault tree presented in Fig. 7-5.

The following is a summary of the assumptions used in developing the event tree shown in Fig. 7-9.

- The Bulk Drain Station removes 95% of the agent in a munition or container under normal conditions.
- 2. All ton containers that reach the BSA have been punched at the Bulk Drain Station.

#### 7.1.4. Toxic Cubicle

The analysis presented in this section examined potential release scenarios that could occur in the toxic cubicle (TOX). The only sources of agent in this area are the agent collection tanks and the agent collection and transfer lines. The scenarios which were analyzed involve spills of agent from these sources with subsequent failure of either the building structure or the ventilation system, resulting in a release of agent to the environment.

ᢉᠵᢆᢉ᠊ᡘᡯᠴᠯᡄ᠋ᠮᡅᢉᡅᡃᡗᠣᡗᡯ᠅ᡗᢑᡃᡗᢑ᠊ᢉᢑ᠂ᡘᢑᢉᢑᠽ



\*\*\*\*\*

KEERS PRESEST WANNED BASE

2773223

E222223



STATISTICS AND A STATIST

CARLENDARY STATE

eritiks •

7-23

na na manana manana mangka manana Ang manana ma ):

۰.

۰.

1. March March March

# TABLE 7-3 EVENTS CONSIDERED FOR THE BSA

0.00000

| Spill of one ton container<br>in BSA      | A punched ton container falls off the buffer storage conveyor in the BSA.                                                                                                                      |
|-------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Container drained at BDS (BDS-FDT)        | Failure on this event tree branch implies<br>that the Bulk Drain Station (BDS) did not<br>drain a ton container before sending it on<br>to the BSA.                                            |
| Orientation precludes spill<br>(DTC1-ORI) | Failure on this event tree branch implies<br>that a drained ton container that is<br>dropped lands in the proper orientation for<br>drainage of its residual agent contents.                   |
| Avoid ignition (BSA-AI)                   | Failure on this event tree branch implies<br>ignition of the agent spill. Motors and<br>cables are potential ignition sources in<br>the BSA.                                                   |
| Suppression (BSA-SUPP)                    | Failure on this event tree branch implies<br>that the dampers do not successfully<br>extinguish a fire.                                                                                        |
| Ventilation system (VENT)                 | Failure on this event tree branch implies<br>a release of agent through the ventilation<br>system due to (1) duct failure or (2) fil-<br>ter failures. (See fault tree shown in<br>Fig. 7-12.) |



KUCCUCUTTU

272

7-24

Spills in the TOX can result from equipment ruptures (tanks, piping, or valves) or from overfilling of an agent collection tank. Rupture of a tank or of the tank outlet valves or piping would result in the spill of the entire contents of one agent collection tanks. (A 500-gal spill was assumed for this case.) On the other hand, rupture of the tank inlet valves or piping or overfilling a tank would result in a substantially smaller spill. Therefore, these two classes of spills were analyzed separately.

Scenarios in which a fire in the TOX causes a release were not analyzed since there are no combustibles in the TOX for sustaining a fire. However, there are ignition sources (motors and cables), so scenarios in which agent spills are ignited were analyzed.

The accident event trees developed for the TOX are shown in Figs. 7-10 and 7-11. Table 7-4 provides descriptions of the events used to construct these event trees, and Fig. 7-12 shows the fault tree which was constructed to quantify the fire suppression event. The ventilation system event was quantified using the fault tree presented in Fig. 7-5.

#### 7.1.5. Incinerator Systems

7.1.5.1. Furnace Explosions. Four furnaces are used in the MDB:

- 1. The Liquid Incinerator (LIC).
- 2. The Metal Parts Furnace (MPF).
- 3. The Deactivation Furnace System (DFS).
- 4. The Dunnage Incinerator (DUN).

Analyses of explosions resulting from operating these furnaces focused upon two generic explosion scenarios:

 Furnace explosions - in which the combustible material is initially confined to the furnace interior.



Fig. 7-10. Event tree for a small spill in the TOX

٢

7-26

د ے ц. VENTILATION SYSTEM VENT 10-9 10-9 ~1:0 ~1.0 STRUCTURE Contains Fire TOX-SFIRE SUPPRESSION TOX-SUPP TOX-SUMPP SUMP PUMP 10-1 0.9

SEQUENCE 10

AV0ID IGNITION

LARGE SPILL IN TOX TOX-AI

T0X-SL



U

-1:0 2

¶0

~1: -

6 x 10<sup>-7</sup>

GB: 10<sup>-4</sup> VX, HD: 10<sup>-5</sup>

~1.0

6.0

FAILURE

10<sup>-9</sup>

-1:0 /

1-01 I

10\_<u>1</u>

-1:-

6 × 10<sup>-7</sup>

10-3

~1.0

u.

<u>ب</u>

Fig. 7-11. Event tree for a large spill in the TOX

10<sup>-3</sup>/YR

SUCCESS

~ 1:0



# TABLE 7-4EVENTS CONSIDERED FOR THE TOX

1.1.0.0

| Event                       | Description                                                                                                                                                                                                                                                                        |  |  |  |  |  |  |
|-----------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|--|
| Large spill in TOX (TOX-SL) | The contents of one agent collection tank (500 gal) are spilled onto the floor of the TOX due to rupture of the tank itself or rupture of outlet valves or piping. The frequency is dominated by pipe failure with a rate of $10^{-3}/yr$ .                                        |  |  |  |  |  |  |
| Small spill in TOX (TOX-SS) | An amount of agent less than the volume of<br>one agent collection tank is spilled onto<br>the floor of the TOX (typically less than<br>50 gal) due to tank overfill or rupture of<br>the tank inlet piping or valves.                                                             |  |  |  |  |  |  |
| Avoid ignition (TOX-AI)     | Failure on this event tree branch implies<br>ignition of the agent spill. Motors and<br>cables are potential ignition sources in<br>the TOX. This probability was subjectively<br>estimated.                                                                                       |  |  |  |  |  |  |
| Suppression (TOX-SUPP)      | Failure on this event tree branch implies<br>that the fire suppression system does<br>not start and that the operator fails to<br>either (1) close the room inlet dampers or<br>(2) turn on the dry chemical fire suppres-<br>sion system. (See fault tree shown in<br>Fig. 7-12.) |  |  |  |  |  |  |
| Ventilation system (VENT)   | Failure on this event tree branch implies<br>a release of agent through the ventila-<br>tion system due to (1) duct failure or<br>(2) filter failures. (See fault tree<br>shown in Fig. 7-5.)                                                                                      |  |  |  |  |  |  |







2. Room explosions - in which a flammable mixture forms outside of the furnace.

Room explosions do not preclude accompanying deflagration inside of the furnace.

Structural evaluations show that the LIC can contain a furnace explosion. Since there is no resultant agent release to the environment, LIC furnace explosions can be screened due to their low consequence.

A LIC room explosion can occur if, following a LIC shutdown, continued agent or fuel flow into the LIC results in a flammable mixture forming in the LIC room. However, the LIC room ventilation flow rate precludes flammable mixture formation, even if 100% agent or fuel flow continues. Because of the high ventilation system reliability, the frequency of independent failures resulting in an LIC shutdown, continued fuel or agent flow, and ventilation system failure is below the  $10^{-10}/yr$  screening criteria.

Loss of offsite power was also investigated as an LIC room explosion initiating event, because both LIC shutdown and loss of ventilation flow occur without any electric power. Thus, at frequencies on the order of 0.1 per year, a single initiating event can cause an LIC shutdown and ventilation system failure. However, the loss of offsite power terminates agent flow since, without the pressure developed by the agent feed pump, the agent cannot physically flow through the LIC atomizer. Moreover, the valves on the LIC fuel lines are designed to fail closed upon a loss of power. These design features, in conjunction with procedures requiring that the operators close the manual fuel block valves, result in the frequency of loss of offsite power initiated explosions also being below  $10^{-10}/yr$ .

7-30

| ľ | AD-A1 | 93 355<br>SSIFIE | Che<br>The<br>Ca<br>D Sap | CHEMICAL STOCKPILE DISPOSAL PROGRAM RISK ANALYSIS OF<br>THE DISPOSAL OF CHEM (U) GA TECHNOLOGIES INC SAN DIEGO<br>CA A W BARSELL ET AL. AUG 87 GA-C-18563<br>SAPEO-CDE-IS-87000 DARA13-85-D-8022 F/G 15/6.3 |  |  |  |  |  | 5/<br>0 NL | 5/13 |  |  |
|---|-------|------------------|---------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|------------|------|--|--|
|   |       |                  |                           |                                                                                                                                                                                                             |  |  |  |  |  |            |      |  |  |
|   |       |                  |                           |                                                                                                                                                                                                             |  |  |  |  |  |            |      |  |  |
|   |       |                  |                           |                                                                                                                                                                                                             |  |  |  |  |  |            |      |  |  |
|   |       |                  |                           |                                                                                                                                                                                                             |  |  |  |  |  |            |      |  |  |
|   |       |                  |                           |                                                                                                                                                                                                             |  |  |  |  |  |            |      |  |  |
|   |       |                  |                           |                                                                                                                                                                                                             |  |  |  |  |  |            |      |  |  |
|   |       |                  |                           |                                                                                                                                                                                                             |  |  |  |  |  |            |      |  |  |



An MPF explosion can result in an agent release to the environment if it involves an undrained or unpunched bulk item (i.e., a ton container, spray tank, or bomb). If an undrained bulk item is inadvertently fed to the MPF, the explosion involves agent deflagration. However, this type of explosion can only occur if the MPF is shut down while an undraind bulk item is being processed. Although MPF shutdowns are rather common (~7 per year), the probability of failing to drain a bulk item is so low that the frequency of an MPF explosion occurring while an undrained bulk item is being processed is below  $10^{-10}/yr$ .

An MPF explosion will occur if an unpunched bulk item is fed to the MPF as a result of the bulk item experiencing hydraulic rupture. Hydraulic ruptures are capable of damaging the MDB and releasing virtually all of the bulk item inventory to the environment. Hydraulic ruptures have frequencies about  $10^{-10}/yr$ .

A natural gas deflagration can also cause an MPF explosion. Since the MPF is subjected to structural failure during natural gas deflagrations, these explosions contribute to the plant risk. However, MPF room explosions are screened from the risk assessment because their frequency is below  $10^{-10}/yr$ . This is due to the high room ventilation system reliability, a fail-safe fuel valve design, and instituted procedural requirements. Both DFS and DUN room explosions have frequencies below  $10^{-10}/yr$  for the same reason.

Structural evaluation of DFS furnace explosions conclude that the blast is insufficient to fail the DFS room walls. Hence, any agent present when the explosion occurs will remain in the DFS room, and there will be no damage to any munitions, containers, or equipment outside of the DFS room.

The DUN furnace can contain a natural gas deflagration. Consequently, no agent release results from this scenario. However, the DUN furnace cannot survive a munition detonation. Although the probability of inadvertently feeding a munition to the DUN is low (on the order of  $10^{-7}$  per munition pallet or mine drum), the high munition processing rates result in DUN explosion frequencies ranging from  $\sim 10^{-2}$  to  $\sim 10^{-3}$  per year, depending upon the munition type. If a munition detonates in the DUN, its entire inventory is released to the environment by the detonation.

Table 7-5 describes the initiating events for LIC shutdowns. Figures 7-13 through 7-38 present the corresponding incinerator system logic models.

## 7.1.5.2. Dunnage Incinerator Accidents Analysis

# <u>Mines</u>

Inadvertently feeding a mine to the Dunnage Incinerator (DUN) requires that the following three faults occur:

- 1. The operators mistakenly leave a mine in the dunnage box.
- 2. The mine counter fails.
- 3. The operator responsible for inspecting the dunnage box prior to charging it to the DUN fails to detect the mine.

Because of all the packing in the dunnage box, the ability of an operator to detect a mine by visual inspection is severely limited. Hence, the probability that the operator responsible for inspecting the dunnage box fails to detect the mine is essentially unity.

The mine counter has two failure modes: mechanical and human error. The dominant failure modes involves an operator failing to properly initialize the mine counter prior to unloading a drum of mines. This human error is estimated to have a 0.01 probability (Ref. 7-2).
| Initiator          | Description                                                                                                                                                                        |  |  |  |  |  |  |
|--------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|--|
| LIC-1              | These initiators are all spurious shutdown signals and<br>process upsets which are not expected to cause agent<br>release if no action is taken to stop the furnace<br>operations. |  |  |  |  |  |  |
|                    | These initiators cause the loss of CA to the LIC-AB.(a)                                                                                                                            |  |  |  |  |  |  |
|                    | These initiators cause the loss of all CA to the LIC. <sup>(b)</sup>                                                                                                               |  |  |  |  |  |  |
|                    | These initiators cause a temporary loss of fuel or CA to<br>the LIC-AB.(C)                                                                                                         |  |  |  |  |  |  |
|                    | These initiators cause excess feed agent to the LIC.(d)                                                                                                                            |  |  |  |  |  |  |
| LIC-2              | These initiators cause the loss of air flow through the LIC PAS.                                                                                                                   |  |  |  |  |  |  |
| LIC-3              | These initiators cause the loss of natural gas to all furnaces.                                                                                                                    |  |  |  |  |  |  |
| LIC-4              | . These initiators cause the loss of fuel to the lIC-AB.                                                                                                                           |  |  |  |  |  |  |
| (a) <sub>Thi</sub> | s initiator was previously designated LIC-5.                                                                                                                                       |  |  |  |  |  |  |
| (b) <sub>Thi</sub> | s initiator was previously designated LIC-6.                                                                                                                                       |  |  |  |  |  |  |
| (c) <sub>Thi</sub> | s initiator was previously designated LIC-7.                                                                                                                                       |  |  |  |  |  |  |
| (d) <sub>Thi</sub> | s initiator was previously designated LIC-8.                                                                                                                                       |  |  |  |  |  |  |

# TABLE 7-5LIC INITIATING EVENT DESCRIPTIONS







これ、 というとう しょう しょう たい ひょう しょう かんかん からた



Fig. 7-14. Event tree for LIC-1A initiators

SCENARIO GO TO LIC-2A POAACO41 ပ STOP AGENT FEED LIC-SAF 4.6×10<sup>-11</sup> ~1.0 VENTILATION SYSTEM LIC-VENT 3.1×10 Success Falture ~1.0 LIC PAS SHUTDOWN 2.3×10 <sup>-1</sup>/ m LIC-2

Fig. 7-15. Event tree for LIC-2 initiators

20

7-36

6 V.

A CARLES AND A CARLES AND A

.

30.30 A.A.A.





Fig. 7-16. Event tree for LIC-2A initiators

aller Aller

and the second second second

and to be

Fig. 7-17. Event tree for LIC-3 initiators

\*\*\*\*\*\*

**U** (1)



| 44.2 |
|------|
|      |



Fig. 7-18. Event tree for LIC-4 initiators

ROM EXEMPTOR VOLVAL STRAND SURVEY SUSSA RECEVED SURVEY DURAN DURAN DURAN DURAN DURAN DURAN DURAN DURAN DURAN D

7-39

2,2,2,2,2,2,2,2,2,2,2,2,

1 A . A .



22225

Fig. 7-19. Event tree for LIC-4A initiators



6275 233





W. B. W. J. W.

4" B.S. La" . 84" . 84" . 8

المراز ال

· . 64° . 64° . 174° . 64° . 64° . 75° . 17° . 64° . 68° . 69

Intaka" datak

Fig. 7-21. Filtered exhaust fault tree

and the factor of the factor o

I TERET TEREST AND A SOUTH THE TEREST TEREST TO A SOUTH THE SOUTH THE SOUTH THE SOUTH THE SOUTH THE SOUTH THE S

2163 • •

c.

7-42

עי



Fig. 7-23. Fault tree for terminating fuel flow to LIC burners

\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*

instation of the second se

 $\sim \sim \sim$ 

٩Ţ.





E CAR SAL VALUE

. . .



n Vha

بربرية. م

-----

C D

Fig. 7-25. Fault tree for LIC PAS shutdown



22222

24.542

Fig. 7-26. Fault tree for LIC PCC fuel flow termination







N.

<u>ᠵᡔᠴᠿᡄ᠋ᡱᢓᡵ᠋ᡀᠼᢂᡷᡵ᠘ᡀᡵᡛᠼᡳᠿᡄ</u>ᡏᢓᡵᡄᡗᢓᡵᡬᢢᡵ᠕᠋ᠴᢤᡘᡵᡛᡘᡵᡌᡵᡌᡵᡌᡵᢂᡘᢛᡚᡵ᠖᠋ᡵᢣᡟ᠋ᡵ᠘ᡱᢣᡛᡵ᠘ᡱ᠕ᢣ᠕ᡀᢣ᠕ᡆᢣ᠕ᠥ᠕ᢣ᠕ᡧ᠕ᠳᡐ᠕ᠳᡐ</u>

⋒⋬⋽⋒⋻⋝⋌⋒⋏⋜⋳⋳⋬⋎⋳⋺⋬⋎⋳⋵⋪⋜⋳⋳⋬⋎⋒⋳⋬⋝⋒⋳⋬⋎⋒⋳⋬⋎⋒⋶⋬⋎⋒⋳⋬⋎⋒⋎⋬⋐⋺⋶⋬⋐⋺⋶⋬⋐⋺⋶⋬⋐∊⋧⋬⋳∊⋧⋬⋼⋎⋨⋛⋼⋎⋒⋛⋼⋎⋳⋛⋼⋎⋳⋛⋼⋎⋳⋛⋼⋎⋳⋛⋼⋎⋳⋛⋼⋎⋳



7-50



Fig. 7-30. Fault tree for draining bulk containers

DATA 2020, SANGA

CAL A

**W**. W



Fig. 7-31. Event tree for MPF shutdown





Fig. 7-32. Event tree for unpunched bulk item fed to MPF

**A** 5 0

ΥX.

100000

Кn

irinir.







Fig. 7-35. Fault tree for MPF fuel flow termination



ر مرجم مرجم

Fig. 7-36. Fault tree for failure to stop fuel to MPF burners

.



 $\sim$ 



7-59



Fig. 7-38. Fault tree for feeding an unpunched container to the MPF (sheet 2 of 4)

• •



الاستان المراجع المراجع



7-62

STAND SECOND

The probability that the operators mistakenly leave a mine in the dunnage box involves two operator errors, one of which is recoverable. First, an operator must inadvertently begin to place a mine in the dunnage box. This human error is estimated to have the probability of 0.01. However, the mine weighs 23 lb, and is of a different shape than the drum packing material. Since the extra weight and shape difference are sensory cues to alert the operator of the initial error, a 0.1 recovery factor (Ref. 7-3) is applied to the initial human error probability. Moreover, the second operator assisting with the unloading operation can prevent a mine from being left in the dunnage box if he sees the first operator placing it in the box, or if he sees the mine in the box while loading it. A human error probability of 0.01 was assigned to the second operator. Therefore, the overall failure probability per drum was calculated as follows:

 $10^{-2} \times 10^{-2} \times 0.1 \times 10^{-2} = 10^{-7}/\text{drum}$ 

The frequency of inadvertently feeding a mine to the DUN is the product of the failure probability per drum multiplied by the number of mine drums processed per year.

#### Rockets, Mortars, and 105s

Inadvertently feeding a rocket, mortar, or 105 to the DUN requires that the following two faults occur:

- 1. The operators mistakenly leave a munition in the dunnage box.
- 2. The operator responsible for inspecting the dunnage box prior to charging it to the DUN fails to detect the munition.

From the analysis for the mines, the probability that the operators mistakenly leave a munition in the dunnage box is  $10^{-5}$ . Since rockets, mortars, and 105s are sent to the UPA without all of the packing used for mines, the operator responsible for inspecting the dunnage box has

an excellent chance of detecting a munition mistakenly left in the dunnage box. Assigning an error probability of 0.01 to this inspection results in an overall failure probability per pallet of: 1110011

 $10^{-5} \times 10^{-2} = 10^{-7}/\text{pallet}$ 

The frequency of inadvertently feeding a rocket, mortar, or 105 to the DUN is the product of the failure probability per pallet multiplied by the number of pallets processed per year.

#### Other Munitions

Sec. Carlos

55757773m1/2/2/2/3m1/2/2/2/2/an1035/57523

Mines, mortars, 105 mm projectiles, and rockets weigh 23, 25, 32, and approximately 56 lb, respectively. Because of their weight, these munitions can be handled by a single operator. All other munitions weigh in excess of 100 lb, except 155 mm projectiles which have a 95-lb minimum weight. Because of their weight, these munitions cannot be easily handled by a single operator. Although these other munitions can be fed to the DUN, the likelihood of this occurring is dominated by the probability that at least one operator commits an act of sabotage. The probability of this event cannot be quoted in an unclassified document.

#### 7.1.6. Accident Analysis Summary and Results

Table 7-6 lists the internally-initated plant accident sequences which survived the preliminary screening. A complete list of all the accident sequences identified is provided in Appendix A.

Table 7-7 presents the accident frequency results for these sequences. The values shown are median values. The range factor column represents the rates of the 95th percentile value to the median value. More details on the uncertainty analysis are discussed in Section 7.3.



AN AL

|          | TAB    | LE 7-6   |           |
|----------|--------|----------|-----------|
| INTERNAL | EVENTS | ACCIDENT | SEQUENCES |

1222-222

| Scenario ID                                        | Description                                                                                                      |   |
|----------------------------------------------------|------------------------------------------------------------------------------------------------------------------|---|
| P041                                               | Failure to stop agent feed to the LIC, overloads the ventilation system.                                         |   |
| P042                                               | MPF explosion due to failure to stop fuel flow after a shutdown.                                                 |   |
| P043                                               | MPF explosion due to hydraulic rupture of an unpunched bulk item. MPF room and ventilation integrity maintained. |   |
| P044                                               | MPF explosion due to hydraulic rupture of an unpunched bulk item. MPF room or ventilation integrity lost.        |   |
| P045                                               | Ton container is spilled in the ECV, MDB structure fails due to subsequent agent fire.                           |   |
| P046                                               | Munition detonation in the ECV, no fire.                                                                         |   |
| P047                                               | Munition detonation in the ECV, fire results but does not propagate.                                             |   |
| P048                                               | Munition detonation in ECV, fire results and propagates.                                                         |   |
| P049                                               | Munition detonation in ECR causes structural and ventilation system failure.                                     |   |
| P050                                               | Munition detonation in ECR causes structural failure, a fire, and ventilation failure.                           |   |
| P051                                               | Ton container spile in the MPB results in fire and structural failure.                                           |   |
| P052                                               | A burstered munition is fed to the DUN.                                                                          |   |
|                                                    |                                                                                                                  |   |
|                                                    |                                                                                                                  |   |
|                                                    |                                                                                                                  | • |
|                                                    |                                                                                                                  |   |
|                                                    |                                                                                                                  |   |
|                                                    |                                                                                                                  |   |
|                                                    |                                                                                                                  |   |
|                                                    | 7-65                                                                                                             |   |
|                                                    |                                                                                                                  |   |
|                                                    |                                                                                                                  |   |
| ም ብር ምር የስም እም | ĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸĸ                                                                           |   |









1.12

## TABLE 7-7 PLANT OPERATIONS INTERNALLY-INITIATED ACCIDENT SEQUENCE FREQUENCIES (EVENTS/YEAR)

### DATE21-Aug-97 PAGE1 PLTOPSOS

UNU

PLANT OPERATIONS INTERNAL INITIATING EVENTS RDC OPTION MEDIAN ACCIDENT FREQUENCY ( PER FACILITY-YEAR)

ĸŴĸĔŧŇĸĔŧŴĸĔŧŴĸŨĸĊĸĊŧſĬĸĔĔĬĸĔĔĬĬĸĔŧŇĸĔŧĬŎŧĬĬĸĔŧĬĸĔŧĬĸĔŧĬĸĔŧĬĸĔ

PLANT OPERATIONS INTERNAL INITIATING EVENTS NDC OPTION MEDIAN ACCIDENT FREQUENCY ( PEP FACILITY-(EAR)

|               | SCENA | ANAD RDC | RANGE   | TEAD RDC | RANGE   | TEAD NOC         | RANGE   |
|---------------|-------|----------|---------|----------|---------|------------------|---------|
|               | NUMBE | FREQ     | FACTOR  | FRED     | FACTOR  | FRED             | FACTOR  |
| ••••          |       |          |         | ******   |         |                  |         |
| POAGC         | 41    | 3.3E-10  | 4.8E+01 | 3.3E-10  | 4.8E+01 | 3.3E-10          | 4.8E+01 |
| POAHC         | 41    | 3.3E-10  | 4.8E+01 | 3.3E-10  | 4.8E+01 | 3.3E-10          | 4.8E+01 |
| POAVE         | 41    | 3.3E-10  | 4.8E+01 | 3.3E-10  | 4.8E+01 | 3.3E-10          | 4.8E+01 |
| P096C         | 42    | N/A      |         | 9.9E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| PODHC         | 42    | 9.9E-09  | 3.7E+01 | 9.9E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| POCSC         | 42    | 9.9E-09  | 3.7E+01 | 9.9E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| POCHC         | 42    | 7.9E-09  | 3.7E+01 | 9.9E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| POKSC         | 42    | N/A      |         | 9.9E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| POKHC         | 42    | 9.9E-09  | 3.7E+01 | 9.9E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| POKVC         | 42    | 9.9E-09  | 3.7E+01 | 9.9E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| PONVC         | 42    | N/A      |         | N/A      |         | N/A              |         |
| POPSC         | 42    | 9.9E-09  | 3.7E+01 | 9.9E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| POPHC         | 42    | 9.9E-09  | 3.7E+01 | 9.9E-09  | 3.7E+01 | 7.9E-09          | 3.7E+01 |
| POPVC         | 42    | 9.9E-09  | 3.7E+01 | 9.7E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| P096C         | 42    | 9.9E-09  | 3.7E+01 | 9.9E-09  | 3.7E+01 | 9.7E-09          | 3.7E+01 |
| PORVC         | 42    | N/A      |         | 9.9E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| PORGC         | 42    | N/A      |         | N/A      |         | N/A              |         |
| PORVC         | 42    | N/A      |         | N/A      |         | N/A              |         |
| POSVC         | 42    | N/A      |         | 9.9E-09  | 3.7E+01 | 9.9E-09          | 3.7E+01 |
| F0 <b>86C</b> | 43    | N/A      |         | 1.6E-09  | 4.1E+01 | 1.68-09          | 4.1E+01 |
| POK 6C        | 43    | N/A      |         | 2.3E-10  | 4.1E+01 | 2.3 <b>E-1</b> 0 | 4.1E+01 |
| POKHC         | 43    | 2.7E-10  | 4.1E+01 | 2.7E-10  | 4.1E+01 | 2.7E-10          | 4.1E+01 |
| POKVC         | 43    | 1.5E-10  | 4.1E+01 | 1.5E-10  | 4.1E+01 | 1.58-10          | 4.1E+01 |
| POSVC         | 43    | N/A      |         | 1.8E-10  | 4.1E+01 | 1.3E-10          | 4.16+01 |
| 20860         | 44    | N/A      |         | 1.6E-10  | 4.1E+01 | 1.0E-10          | 4.1E+01 |
| POKEC         | 44    | N/A      |         | 2.3E-10  | 4.iE+01 | 2.3E-10          | 4.1E+01 |
| POKHC         | 44    | 2.7E-10  | 4.1E+01 | 2.7E-10  | 4.1E+01 | 2.7E-10          | 4.1E+01 |
| POKVC         | 44    | 1.5E-10  | 4.1E+01 | 1.5E-10  | 4.1E+01 | 1.5E-10          | 4.1E+01 |
| POSVC         | 44    | N/A      |         | 1.8E-10  | 4.1E+01 | 1.8E-10          | 4.1E+01 |

TABLE 7-7 (Continued)

DATE21-Aug-87 PAGE2 PLTOPSOS

I VILVILL

MARA CONTRACTOR

100000

r F

PLANT OPERATIONS INTERNAL INITIATING EVENTS RDC OPTION MEDIAN ACCIDENT FREQUENCY ( PER FACILITY-(EAR)

|               | SCENA | ANAD RDC | RANGE   | TEAD RDC | RANGE   |
|---------------|-------|----------|---------|----------|---------|
|               | NUMBE | FRED     | FACTOR  | FRED     | FACTOR  |
|               |       |          |         |          |         |
| POKGF         | 45    | N/A      |         | 4.0E-09  | 1.4E+01 |
| POKHF         | 45    | 4.0E-10  | 1.4E+01 | 4.0E-10  | 1.4E+01 |
| POKVE         | 45    | 4.0E-10  | 1.4E+01 | 4.0E-10  | 1.4E+01 |
| PODHC         | 46    | 9.0E-09  | 2.6E+01 | 9.0E-09  | 2.6E+01 |
| POCSC         | 46    | 1.0E-08  | 2.6E+01 | 1.0E-08  | 2.6E+01 |
| POCHC         | 46    | 1.0E-08  | 2.6E+01 | 1.0E-08  | 2.6E+01 |
| PONVC         | 46    | 4.0E-07  | 2.6E+01 | 4.0E-07  | 2.6E+01 |
| POPGC         | 46    | 6.0E-07  | 2.6E+01 | 6.0E-07  | 2.6E+01 |
| POPHC         | 46    | 6.0E-07  | 2.6E+01 | 6.0E-07  | 2.6E+01 |
| POPVC         | 46    | 6.0E-07  | 2.6E+01 | 6.0E-07  | 2.6E+01 |
| P006C         | 46    | 3.0E-07  | 2.6E+01 | 3.0E-07  | 2,6E+01 |
| POOVC         | 46    | N/A      |         | 3.0E-07  | 2.6E+01 |
| PORGC         | 46    | 1.5E-07  | 2.7E+01 | 1.5E-07  | 2.7E+01 |
| PORVC         | 46    | 1.5E-07  | 2.7E+01 | 1.5E-07  | 2.7E+01 |
| PODHC         | 47    | 8.1E-11  | 3.1E+01 | 8.1E-11  | 3.1E+01 |
| P0C6C         | 47    | 9.0E-11  | 3.1E+01 | 9.0E-11  | 3.4E+01 |
| POCHC         | 47    | 9.0E-11  | 3.1E+01 | 9.0E-11  | 3.1E+01 |
| POHVC         | 47    | 3.68-09  | 3.1E+01 | J.6E-09  | 3.1E+01 |
| POPSC         | 47    | 5.4E-09  | 3.1E+01 | 5.4E-09  | 3.1E+01 |
| POPHC         | 47    | 5.4E-09  | 3.1E+01 | 5.4E-09  | 3.1E+01 |
| POPVC         | 47    | 5.48-09  | 3.1E+01 | 5.4E-09  | 7.1E+01 |
| P0 <b>96C</b> | 47    | 2.7E-09  | 3.1E+01 | 2.7E-09  | 3.1E+01 |
| POQVC         | 47    | N/A      |         | 2.7E-09  | 3.1E+01 |
| PORSC         | 47    | 1.58-07  | 2.7E+01 | 1.5E-07  | 2.7E+01 |
| PORVC         | 47    | 1.5E-07  | 2.7E+01 | 1.5E-07  | 2.7E+01 |
| ₽ <b>00HC</b> | 48    | 9.0E-12  | 3.3E+01 | 9.0E-12  | 3.3E+01 |
| P0C6C         | 48    | 1.6E-11  | 3.3E+01 | 1.0E-11  | 3.3E+01 |
| POCHE         | 48    | 1.0E-11  | 3.3E+01 | 1.JE-11  | 3.3E+01 |
| FORVE         | 48    | 4. 0E-10 | 3.3E+01 | 4.0E-10  | 3.3E+01 |

| LAI | IT OPER | ATIONS | INTERNAL | INITIATING | EVENTS | 5              |
|-----|---------|--------|----------|------------|--------|----------------|
| 00  | OPTION  | MEDIAN | ACCIDENT | FREQUENCY  | ( PER  | FACILITY-YEAR) |

| TEAD NDC | RANGE    |
|----------|----------|
| FREC     | FACTOR   |
|          |          |
|          |          |
| 4 OF-09  | 1 45+01  |
| 4 05-10  | 1 45401  |
| A 05-10  | 1 45+01  |
| 0 05-00  | 2 45401  |
| 1 05-00  | 2.00.001 |
| 1 05-09  | 2.001    |
| 1.VE-VO  | 2.001    |
| 4.VE-07  | 2.05101  |
| 0.05-07  | 2.05+01  |
| 6.VE-07  | 2.00+01  |
| 0.VE-0/  | 2.06+01  |
| 3.0E-07  | 2.62+91  |
| 3.08-07  | 2.62+01  |
| 1.5E-0/  | 2.78+01  |
| 1.58-07  | 2.7E+01  |
| 8.1E-11  | 3.1E+01  |
| 9.0E-11  | 3.1E+01  |
| 9.0E-11  | 3.1E+01  |
| 3.6E-09  | 3.1E+01  |
| 5.4E-09  | 3.1E+01  |
| 5.4E-09  | 3.1E+01  |
| 5.48-09  | 3.1E+01  |
| 2.7E-09  | 3.1E+01  |
| 2.7E-09  | 3.1E+01  |
| 1.5E-07  | 2.7E+01  |
| 1.5E-07  | 2.7E+01  |
| 9.0E-12  | 3.3E+01  |
| 1.0E-11  | 3.3E+01  |
| i.0E-11  | 3.3E+01  |
| 4.0E-10  | 3.3E+01  |
|          |          |

## TABLE 7-7 (Continued)

DATE21-Aug-87 PAGE3 PLTOPSOS

A's.1's.8's.0"s. 0"

PLANT OPERATIONS INTERNAL INITIATING EVENTS RDC OPTION MEDIAN ACCIDENT FREQUENCY ( PER FACILITY-YEAR)

|               | SCENA | ANAD RDC | RANGE   | TEAD RDC | RANGE   |  |
|---------------|-------|----------|---------|----------|---------|--|
|               | NUMBE | FRED     | FACTOR  | FREQ     | FACTOR  |  |
|               |       |          |         |          |         |  |
| P <b>OP6C</b> | 48    | 6.0E-10  | 3.3E+01 | 6.0E-10  | 3.3E+01 |  |
| POPHC         | 48    | 6.0E-10  | 3.3E+01 | 6.0E-10  | 3.3E+01 |  |
| DV909         | 49    | 6.0E-10  | 3.3E+01 | 6.0E-10  | 3.3E+01 |  |
| P006C         | 48    | 3.0E-10  | 3.3E+01 | 3.0E-10  | 3.3E+01 |  |
| POQVC         | 49    | N/A      |         | 3.0E-10  | 3.3E+01 |  |
| PORGC         | 48    | 1.5E-09  | 3.2E+01 | 1.5E-09  | 3.2E+01 |  |
| PORVC         | 48    | 1.5E-09  | 3.2E+01 | 1.5E-09  | 3.2E+01 |  |
| PODHC         | 49    | 3.0E-06  | 3.1E+01 | 3.0E-06  | 3.1E+01 |  |
| POCGC         | 49    | 4.0E-06  | 3.1E+01 | 4.0E-06  | 3.1E+01 |  |
| POCHC         | 49    | 4.0E-06  | 3.1E+01 | 4.0E-06  | 3.1E+01 |  |
| POHVC         | 49    | 4.0E-09  | 3.7E+01 | 4.0E-09  | 3.7E+01 |  |
| POPSC         | 49    | 2.0E-06  | 3.1E+01 | 2.0E-06  | 3.1E+01 |  |
| POPHC         | 49    | 2.0E-06  | 3.1E+01 | 2.0E-06  | 3.1E+01 |  |
| POPVC         | 49    | 2.0E-06  | 3.1E+01 | 2.0E-06  | 3.1E+01 |  |
| P0 <b>96C</b> | 49    | 8.0E-07  | 3.1E+01 | 8.0E-07  | 3.1E+01 |  |
| POQVC         | 49    | N/A      |         | 8.0E-07  | 3.1E+01 |  |
| PORGC         | 49    | 5.0E-07  | 3.4E+01 | 5.0E-07  | 3.4E+01 |  |
| PORVC         | 49    | 5.0E-07  | 3.4E+01 | 5.0E-07  | 3.4E+01 |  |
| PODHC         | 50    | 3.0E-08  | 3.7E+01 | 3.0E-08  | 3.7E+01 |  |
| P0C6C         | 50    | 4.0E-08  | 3.7E+01 | 4.0E-08  | 3.7E+01 |  |
| POCHC         | 50    | 4.0E-08  | 3.7E+01 | 4.0E-08  | 5.7E+01 |  |
| PONVC         | 50    | 4.0E-11  | 3.7E+01 | 4.0E-11  | 3.7E+01 |  |
| POPGC         | 50    | 2.0E-08  | 3.7E+01 | 2.0E-08  | 3.7E+01 |  |
| POPHC         | 50    | 2.0E-08  | 3.7E+01 | 2.0E-08  | 3.7E+01 |  |
| POPVC         | 50    | 2.0E-08  | 3.7E+01 | 2.0E-08  | 3.7E+01 |  |
| 20900         | 50    | 8.0E-09  | 3.7E+01 | 8.0E-09  | 3.7E+01 |  |
| POQVC         | 50    | N/A      | ••      | 8.0E-09  | 3.7E+01 |  |
| PORGC         | 50    | 5.0E-07  | 3.4E+01 | 5.0E-07  | 3.4E+01 |  |
| PORVC         | 50    | 5.0E-07  | 3.4E+01 | 5.0E-07  | 3.4E+01 |  |

NDC OPTION MEDIAN ACCIDENT FREQUENCY ( PER FACILITY-YEAR)

PLANT OPERATIONS INTERNAL INITIATING EVENTS

TEAD NOC RANGE FRED

6.0E-10 3.3E+01 6.0E-10 3.3E+01 6.0E-10 3.3E+01 3.0E-10 3.3E+01 3.0E-10 3.3E+01 1.5E-09 3.2E+01 1.5E-09 3.2E+01 3.0E-06 3.1E+01 4.0E-06 3.1E+01 4.0E-06 3.1E+01 4.0E-09 3.7E+01 2.0E-06 3.1E+01 2.0E-06 3.1E+01 2.0E-06 3.1E+01 8.0E-07 3.1E+01 8.0E-07 3.1E+01 5.0E-07 3.4E+01 5.0E-07 3.4E+01 3.0E-08 3.7E+01 4.0E-08 3.7E+01 4.0E-08 3.7E+01 4.0E-11 3.7E+01 2.0E-08 3.7E+01 2.0E-08 3.7E+01 2.0E-08 3.7E+01 8.0E-09 3.7E+01 8.0E-09 3.7E+01 5.0E-07 3.4E+01 5.0E-07 3.4E+01

-----

FACTOR

-----

· · · 

7-68
TABLE 7-7 (Continued)

## DATE21-Aug-87 PAGE4 PLTOPSOS

#### PLANT OPERATIONS INTERNAL INITIATING EVENTS RDC OPTION MEDIAN ACCIDENT FREDUENCY ( PER FACILITY-YEAR)

|       | SCENA | ANAD RDC | RANGE   | TEAD RDC | RANGE   |
|-------|-------|----------|---------|----------|---------|
|       |       |          |         |          |         |
| POKGF | 51    | N/A      |         | 4.0E-09  | 1.4E+01 |
| POKHF | 51    | 4.0E-09  | 1.4E+01 | 4.0E-09  | 1.4E+01 |
| POKVF | 51    | 4.0E-09  | 1.4E+01 | 4.0E-09  | 1.4E+01 |
| POOHC | 52    | 4.4E-03  | 5.7E+01 | 4.4E-03  | 5.7E+01 |
| POCGC | 52    | 5.0E-03  | 5.7E+01 | 5.0E-03  | 5.7E+01 |
| POCHC | 52    | 5.0E-03  | 5.7E+01 | 5.0E-03  | 5.7E+01 |
| PONVC | 52    | 1.1E-02  | 5.7E+01 | 1.1E-02  | 5.7E+01 |
| POPGC | 52    | NEGL     |         | NEGL     |         |
| POPHC | 52    | NEGL     |         | NEGL     |         |
| POPVC | 52    | NEGL     |         | NEGL     |         |
| P096C | 52    | NEGL     |         | NEGL     |         |
| POOVC | 52    | N/A      |         | NEGL     | *-      |
| PORSC | 52    | 1.6E-03  | 5.7E+01 | 1.6E-03  | 5.7E+01 |
| PORVC | 52    | 1.6E-03  | 5.7E+01 | 1.6E-03  | 5.7E+01 |

PLANT OPERATIONS INTERNAL INITIATING EVENTS NDC OPTION MEDIAN ACCIDENT FREQUENCY ( PER FACILITY-YEAR)

| FRED    | FACTOR  |
|---------|---------|
|         |         |
|         |         |
| 4.0E-09 | 1.4E+01 |
| 4.0E-09 | 1.4E+01 |
| 4.0E-09 | 1.4E+01 |
| 4.4E-03 | 5.7E+01 |
| 5.0E-03 | 5.7E+01 |
| 5.0E-03 | 5.7E+01 |
| 1.1E-02 | 5.7E+01 |
| NEGL    |         |
| 1.6E-03 | 5.7E+01 |
| 1.6E-03 | 5.7E+01 |

TEAD NDC RANGE



### 7.2. EXTERNAL EVENTS

The following external event initiators were considered in the development of plant-related accident scenarios which could lead to the release of a significant amount of chemical agent:

- 1. Tornadoes and high winds.
- 2. Meteorite strikes.
- 3. Aircraft crashes.
- 4. Earthquakes.
- 5. Lightning.

For this study, the demil facility is defined to include (1) the MHI where munitions awaiting demilitarization are temporarily stored and (2) the MDB which houses the systems and equipments to destroy the explosives and agent contained in the various munitions. The accident sequences identified were subjected to a preliminary screening process by assigning very conservative failure probability values. The screening criteria for frequency and agent release are described in Section 4.

The initiating event families for plan operations were identified in Section 4.1. Table 7-8 lists the accident sequences related to plant operations initiated by external events. The event tree models are presented in Figs. 7-39 through 7-44. 22222

2222223

#### 7.2.1. Tornadoes and High Winds

The accident scenarios identified involve the breaching of the munitions in the MHI and the UPA by tornado- or high wind-generated missiles. This failure mode was determined to be more credible than a tornado/high wind-induced building collapse which could lead to the crushing of munitions by the falling structure. For UBC designed structures such as the MDB, the wind loads will fail the walls of the



B

and the second second

# TABLE 7-8 MASTER LIST OF EXTERNALLY-INITIATED PLANT ACCIDENT SCENARIOS

| Scenario ID | Description                                                                                                     |
|-------------|-----------------------------------------------------------------------------------------------------------------|
| P01         | Tornado-generated missile puncture/crush munitions in the MHI.                                                  |
| PO2         | Tornado-generated missile detonate munitions in the MHI.                                                        |
| P03         | Tornado-generated missile puncture/crush munitions in the UPA.                                                  |
| P04         | Tornado-generated missile detonate munitions in the UPA.                                                        |
| P05         | Tornado-generated missile damages the agent piping system between the BDS and TOX at TEAD (bulk-only facility). |
| P06         | Meteorite strikes the MHI.                                                                                      |
| P07         | Meteorite strikes the UPA.                                                                                      |
| P07A        | Meteorite strikes the TOX.                                                                                      |
| P08         | Meteorite strikes the agent piping system between the BDS and TOX at TEAD (bulk-only facility).                 |
| P09         | Direct large aircraft crash onto the MHI; no fire.                                                              |
| P010        | Direct large aircraft crash onto the MHI; fire not contained in 0.5 h.                                          |
| P011        | Direct large aircraft crash onto the MHI; fire contained in 0.5 h.                                              |
| P012        | Direct large aircraft crash damages the MDB; no fire.                                                           |
| P013        | Direct large aircraft crash damages the MDB; fire not contained in 0.5 h.                                       |
| P014        | Direct large aircraft crash damages the MDB; fire contained in 0.5 h.                                           |
| P015        | Indirect large aircraft crash damages the MHI; no fire.                                                         |
| P016        | Indirect large aircraft crash damages the MHI; fire not contained in 0.5 h.                                     |
| P017        | Indirect large aircraft crash damages the MHI; fire contained in 0.5 h.                                         |

7-71

TABLE 7-8 (Continued)

Press and the second

Š

ないと言うでです。

1212222

LANCOSCI L

. .

| Scenario ID | Description                                                                                                                            |
|-------------|----------------------------------------------------------------------------------------------------------------------------------------|
| P018        | Indirect large aircraft crash damages the MDB; no fire.                                                                                |
| P019        | Indirect large aircraft crash damages the MDB; fire not contained in 0.5 h.                                                            |
| P020        | Indirect large aircraft crash damages the MDB; fire contained in 0.5 h.                                                                |
| P021        | Direct crash of a large or small aircraft damages the outdoor agent piping system at TEAD; no fire.                                    |
| P022        | Direct crash of a large or small aircraft damages the<br>outdoor agent piping system at TEAD; fire occurs and not<br>contained.        |
| P023        | Earthquake causes the munitions in the $MH^{\intercal}$ to fall and be punctured.(a)                                                   |
| P024        | Earthquake causes munitions in the MHI to fall and detonate. <sup>(a)</sup>                                                            |
| P025        | Earthquake damages the MDB structure, munitions fall and are punctured; fire suppressed.                                               |
| P026        | Earthquake damages the MDB structure, munitions fall and are punctured; earthquake also initiates fire; fire suppression system fails. |
| PO28A(b)    | Earthquake damages the MDB structure, munitions fall and are punctured; TOX damaged; fire occurs; fire suppressed.                     |
| P028        | Earthquake damages the MDB structure, munitions fall and are punctured; TOX damaged; fire occurs; fire suppression system fails.       |
| P029        | Earthquake damages the MDB; munitions are intact; fire occurs; fire suppression system fails.                                          |
| P030        | Earthquake damages the MDB; munitions are intact; TOX damaged; no fire occurs. <sup>(c)</sup>                                          |
| P031A       | Earthquake damages the MDB; munitions are intact; TOX damaged; fire occurs; fire suppressed.                                           |
| P031        | Earthquake damages the MDB; munitions are intact; TOX damaged; fire occurs; fire not suppressed.                                       |

TABLE 7-8 (Continued)

| Scenario ID              | Description                                                                                                                                                              |
|--------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| P032                     | Earthquake causes munitions to fall and detonate; MDB breached by detonation; the TOX is intact; no fire.(c)                                                             |
| P033                     | Earthquake causes munitions to fall but no detonation<br>occurs; the MDB is intact; the TOX is intact; earthquake<br>also initiates fire; fire suppression system fails. |
| P034                     | Earthquake causes munitions to fall but no detonation occurs; the MDB is intact; the TOX is damaged; fire occurs; fire suppression system fails.                         |
| (a)Screend<br>(b)Sequend | ed out due to design changes.<br>ce 27 not used.                                                                                                                         |

(c)Screened out on the basis of frequency.

X X

[ |



~\.·\.

STATES C

555555





Fig. 7-40. Meteorite-induced agent release scenarios



Fig. 7-41. Large aircraft crash onto MHI/MDB containing burstered munitions

5.7

ABD STREET DOOD DOOD STREET TURNED AND STREET DOOD AND STREET DOOD STREET STREET.



7-76

£1413 ●

T-D-D-C-C-





Fig. 7-42. Aircraft crash onto MHI/MDB with nonburstered (NB) munitions

SAME ZARANA TAAAAN MANATI KANANA TAARAATIAN MANANI KASAMATIKASAN TAANAN TAANAATI KASAMATIKAA MANANI TAANAA MA

7-77

Fig. 7-43. Event tree: earthquake-induced releases from the MDB involving bulk containers

ないという

2222

AND THE PARTY IN THE

| AGENT RELEASE<br>SEQUENCE                                             | NONE         | NONE   | <u>81</u> *  | B2*   | <u>83</u> (P025) | <u>84</u> (P026) | <u>85</u> (P027) | <u>B</u> 6 (P028A) | <u>87</u> (P028) | NONE  | NONE     | <u>88</u> (P029) | <u>6</u> 9* | <u>B10</u> (P031A) | <u>811</u> (P031)               |
|-----------------------------------------------------------------------|--------------|--------|--------------|-------|------------------|------------------|------------------|--------------------|------------------|-------|----------|------------------|-------------|--------------------|---------------------------------|
| FIRE<br>SUPPRESSION<br>SUCCESSFUL                                     | (N/N)        | (N/A)  | (UD)         | (N/N) |                  |                  | (N/N)            |                    |                  | (N/N) |          |                  | (N/N)       |                    |                                 |
| IGNITION<br>Avoided                                                   |              | (FIRE) | (DETONATION) |       | DN               | (FIRE)           |                  | NO                 | (FIRE)           |       | NO       | (FIRE)           |             | ND                 | (FIRE)                          |
| TOX<br>INTEGRITY<br>MAINTAINED                                        |              |        |              | YES   |                  |                  | NO               |                    |                  | YES   | -        |                  | ON          |                    |                                 |
| EARTHQUAKE<br>IMPACT<br>ON MUNITION<br>INTEGRITY                      | N/R          |        |              |       | (PUNCTURED)      |                  |                  |                    |                  |       | (INTACT) |                  |             |                    |                                 |
| MD8 NDT<br>Damaged<br>BY The<br>Earthquake                            | VES          |        |              | NO    |                  |                  |                  |                    |                  |       |          |                  |             | -10,v.5            | <br>IVE HORIZONTAL<br>ELERATION |
| EARTHQUAKE<br>OCCURS WHILE<br>PROCESSING<br>NONBURSTERED<br>MUNITIONS | (√ eq ≈ a**) |        |              | -     |                  |                  |                  |                    |                  |       |          |                  |             |                    | ** a = PEAK EFFECT              |

7-78

N037

972K02KK49



15 - Co.

\*\*\*\*\*\*\*\*\*

|                                                                    |             |  | <br>_   |      |      | _ | _    |      |      | <br>        | <br>        |                            |                     |             |       |      |          |      |      |       |      |  |
|--------------------------------------------------------------------|-------------|--|---------|------|------|---|------|------|------|-------------|-------------|----------------------------|---------------------|-------------|-------|------|----------|------|------|-------|------|--|
| AGENT RELEASE<br>SEQUENCE                                          | P032        |  | NONE    | NONE | P033 |   | NONE | NONE | P034 |             |             | P025                       | P026                | P027        | P028A | P028 | NONE     | NONE | P029 | P031A | P031 |  |
| FIRE<br>SUPPRESSION<br>SUCCESSFUL                                  |             |  |         |      |      |   |      |      |      |             |             |                            |                     |             |       |      |          |      |      |       |      |  |
| IGNITION<br>AVOIDED                                                |             |  |         |      |      |   |      |      |      |             |             |                            |                     |             |       |      |          |      |      |       |      |  |
| TOX<br>INTEGRITY<br>Maintained                                     |             |  |         |      |      |   |      |      |      |             |             |                            | -                   |             |       | -    |          |      | _    |       | -    |  |
| EARTHQUAKE<br>IMPACT<br>ON MUNITION<br>INTEGRITY                   | (DETONATED) |  | (OTHER) |      |      |   |      |      |      | (DETONATED) | (PUNCTURED) |                            |                     |             |       |      | (INTACT) |      |      |       |      |  |
| MDB NOT<br>Damaged<br>By<br>Earthquake                             |             |  |         |      |      |   |      |      |      |             |             | 10 <sub>.</sub> YR         | VE HORIZONTAL       | ELERATION   |       |      |          |      |      |       |      |  |
| EARTHQUAKE<br>OCCURS WHILE<br>PROCESSING<br>BURSTERED<br>MUNITIONS | (< eq ⇔a**) |  |         |      |      |   |      |      |      |             |             | *FREQUENCY 10 <sup>-</sup> | •• a = PEAK EFFECTI | GROUND ACCI |       |      |          |      |      |       |      |  |



7-79

WAYN NATAT

structure before the structure will collapse. Storage igloos like the MHI have been designed to resist the direct effects of tornadoes with winds up to 320 mph except for the possibility of missiles breaching the igloo doors (Ref. 7-4). In the MDB, only the UPA has been determined to be vulnerable to tornado/high wind-generated missiles that could result in significant agent releases.

An additional scenario that applies to the modified CAMDS facility at TEAD has been identified. This involves the susceptibility of the outdoor agent piping system that links the bulk drain station and the TOX which will be located in a separate building.

The event tree developed to define relevant accident sequences is shown in Fig. 7-39. No sequences could be screened out initially as more detailed quantitative analysis is required to determine the necessary wind velocity to generate missiles which could penetrate the munitions. Hence, all the accident sequences numbered in the event tree were quantified. They are:

- PO1 Tornado-generated missiles puncture/crush the munitions in the MHI.
- PO2 Tornado-generated missiles detonate the burstered munitions in the MHI.
- PO3 Tornado-generated missiles breach the munitions in the UPA.
- PO4 Tornado-generated missiles detonate the burstered munitions in the UPA.
- PO5 Tornado-generated missiles breach the agent piping system between the BDS and TOX at TEAD (CAMDS-modified bulk only facility).



7.2.1.1. Tornado and High Wind Accident Analysis. Essentially, the missile penetration of the munition inside the MHI or UPA occurs if (1) a tornado or extremely high wind occurs with a velocity sufficient to generate a missile that could penetrate the MHI door or UPA wall and a munition, and (2) the missile actually hits the target munition. The probability of a missile hitting and rupturing a munition is the product of four variables: (1) the probability that the velocity vector of the missile is nearly perpendicular to the target; (2) the probability that the missile is oriented properly to penetrate the target; (3) the number of missiles per square foot of wind; and (4) the target area. More details on the derivation of these variables are provided in the calculation sheets (Ref. 7-5). If the missile hits a burstered munition, two failure modes are possible, (1) the munition is opened up due to puncture or crush, or (2) the missile impact causes munition detonation due to the application of a force greater than the "undue force."

## <u>Scenario PO1 - Tornado-Generated Missile Penetrates Munitions in</u> the MHI (No Detonations Occur)

The MHI is assumed to be an 80-ft long by 27-ft wide igloo with a concrete door for all sites. The munitions are stored in onsite transportation containers, except the spray tanks and wet eye bombs which are in their existing overpacks and not in onsite transportation containers. There will be a maximum of 16 containers in the MHI.

For an agent release to occur, the missile must penetrate the igloo door, the onsite container, and the munition itself. The required initial velocity (V) to puncture the munition is given by:

$$\mathbf{v} = \mathbf{v}_{\rm D}^2 + \mathbf{v}_{\rm C}^2 + \mathbf{v}_{\rm m}^2 , \qquad (7-1)$$

where  $V_D$  = door penetration velocity,

 $V_{C}$  = onsite container penetration velocity,

 $V_m$  = munition penetration velocity (munition specific).



The puncture velocity for a concrete igloo door has been analyzed previously (Ref. 7-5) and was calculated to be 54 mph assuming the missile is a utility pole. The puncture velocity for the onsite container was calculated to be 63 mph. The penetration velocity for the munition itself is munition specific and is largely a function of the thickness of the munition. Details are provided in the calculation sheets. Having calculated the required initial missile velocity, the required wind velocity to generate the missile is determined in Section 4.2. The frequency of occurrence of a given wind speed is determined from the set of curves given in Section 4.

## <u>Scenario PO2 - Tornado-Generated Missile Penetrate Munitions in the</u> <u>MHI; Detonation Results from Impact</u>

The analysis of scenarios PO2 included the estimation of the probability that a missile impacting a munition would cause it to detonate. The data presented in Ref. 7-6 indicated that a projectile with Comp B explosive could ignite when subjected to a minimum impact velocity of 123 mph. Because the conditions of the tests described in Ref. 7-6 do not fully apply to the conditions being considered here (i.e., the shell casing provides protection for the bursters), it is assumed that there is a 50% chance that a munition will detonate at 123 mph. Furthermore, we also estimate the probability of a detonation resulting from a drop of the munition from a height of 40 ft to be  $10^{-3}$  (Ref. 7-9). The 40 ft drop height corresponds to a free fall velocity (in a vacuum) of about 34.6 mph. To determine the probability of detonating a munition at an impact velocity equivalent to that of a missile required to penetrate the igloo and the munition, we assumed a lognormal distribution and derived the necessary parameters (e.g., standard deviation and standard normal deviate) from these two data points. The results are shown in the data base table (Table 7-9). The calculation details are given in the calculation sheets (Ref. 7-5).

5

200000



STATE STATE

# TABLE 7-9 DATA BASE FOR TORNADO-INITIATED EVENTS FOR PLANT OPERATIONS

PERSON

LA-Jul-or Fage L

#### Data Base For Tornado-Initiated Events For Plant Operations

115.57

| Event                        | Site       | Munition<br>Type | Variable<br>Name | 148     | Variable<br>Name | M08     | Variable<br>Name | Open Arza | Error<br>Factor | Reierence |
|------------------------------|------------|------------------|------------------|---------|------------------|---------|------------------|-----------|-----------------|-----------|
| L. Frequency wind sufficient | ANAD       | 105-ee rartre    | ANNH 1."         | # 9E-07 | ANNORE           | 1.5E-0A |                  |           |                 | SEE CALL  |
| to concrate aissile          |            | 4.2-in mortar    | ARMHID           | 1.12-06 | ANNOBD           | 1.52-06 |                  |           | <br>Iú          | SHEETS    |
| ,                            |            | ton contar       | ANNHIK           | 1.4E-07 | ANNOBK           | 7.3E-07 |                  |           | 10              | )         |
|                              |            | 0146             | ANHH1R           | 1.18-06 | ANNOBH           | 1.5E-06 |                  |           | 10              | 1         |
|                              |            | 155-ee proj      | ANNHIP           | 7.4E-07 | ANNOSP           | 1.5E-06 |                  |           | 10              | r         |
|                              |            | 8-in proj        | ANNH10           | 7.4E-07 | ANNGBQ           | 1.5E-06 |                  |           | 10              | r.        |
|                              |            | rackets          | ANNHIR           | 1.1E-96 | ANNOBR           | 1.5E-96 |                  |           | 10              |           |
|                              | AP6        | ton costar       |                  |         | APHOSK           | 4.5E-00 |                  |           |                 |           |
|                              | LIM        | 155-an proj      | LONHIP           | 1.38-06 | Landab           | 1.5E-06 |                  |           | 10              | i.        |
|                              |            | 8-in proj        | LBMRIQ           | 1.5E-06 | LBHOBQ           | 1.5E-06 |                  |           | 10              |           |
|                              |            | rocket           | LBHHIR           | 1.5E-06 | LBHDBR           | 1.5E-06 |                  |           | 10              |           |
|                              | NAAP       | ton conter       |                  |         | NAHOBK           | 7.3E-07 |                  |           | 10              |           |
|                              | <b>P3A</b> | ton conter       | PBNHIK           | 3.8E-07 | PBMOBK           | 7.3E-07 |                  |           |                 |           |
|                              |            | sine             | PBNHIN           | 1.5E-06 | PBMCBM           | 1.5E-06 |                  |           | 10              |           |
|                              |            | rocket           | POMHIR           | 1.5E-06 | PBMOBR           | 1.5E-06 |                  |           | 10              |           |
|                              | PUBA       | 105-mm cartre    | PUNH1C           | L.0E-07 | PUNDEC           | 1.08-07 |                  |           | 10              |           |
|                              |            | 4.2-in mortar    | PUMHID           | 1.0E-07 | PUMOBO           | 1.0E-07 |                  |           | 10              |           |
|                              |            | 155-en proj      | PUNHIP           | 1.0E-07 | PUNDBP           | 1.0E-07 |                  |           | 19              |           |
|                              | TEAD       | boeb             | TEMIIB           | 3.6E-10 | TEMOBO           | 8.12-10 |                  |           | 10              |           |
|                              |            | 105-ee cartry    | TERHIC           | 1.8E-09 | TENDBC           | 1.82-09 |                  |           | 19              |           |
|                              |            | 4.2-in mortar    | TENHID           | 1.82-09 | TEMOBO           | 1.8E-09 |                  |           | Įú              |           |
|                              |            | ton conter       | TENHIK           | 2.48-10 | TENDBK           | 7.JE-10 |                  |           | 10              |           |
|                              |            | nine             | TEAHIN           | 1.8E-09 | TENDON           | 1.8E-09 |                  |           | 10              |           |
|                              |            | 155-an proj      | TEXHIP           | 1.5E-09 | TENOBP           | 1.8E-09 |                  |           | 10              |           |
|                              |            | 8-in proj        | TENHIQ           | 1.8E-09 | TENOFO           | 1.8E-09 |                  |           | 10              |           |
|                              |            | rocket           | TENHIR           | 1.8E-09 | TENOBR           | 1.8E-09 |                  |           | 10              |           |
|                              |            | spray tank       | TEMHIS           | 1.12-09 | TEMOBS           | 1.88-09 |                  |           | 10              |           |
|                              | ACHE       | bonb             | URMH 1 B         | 3.6E-10 | UHMOBB           | 8.1E-10 |                  |           | 10              |           |
|                              |            | ton contar       | UNNHIK           | 2.4E-10 | UMMOBK           | 7.3E-10 |                  |           | 10              |           |
|                              |            | øine             | UMMHEN           | 1.8E-09 | UMNOSH           | 1.8E-09 |                  |           | 10              |           |
|                              |            | 155-ee proj      | UMMH I P         | 1.5E-09 | UNHOSP           | 1.8E-09 |                  |           | 10              |           |
|                              |            | D-in proj        | UNINH [ Q        | 1.8E-09 | UMHORO           | 1.8E-09 |                  |           | 10              |           |
|                              |            | racket           | UMMHIR           | 1.82-09 | UMMOBR           | 1.8E-09 |                  |           | 10              |           |
|                              |            | spray Lank       | URMH I S         | 1.1E-09 | UARD85           | 1.8E-09 |                  |           | 10              |           |
|                              | TEAD       | 8.K.S(P(PE)      |                  |         |                  |         | COOPA            | 1.8E-09   | 10              |           |







# TABLE 7-9 (Continued)

| 2. | Probability sumition  | A11  | boeb          | MHIPTB | 4.JE-07   | NDBPTB  | 2.0E-06 |       |         | 5ú |
|----|-----------------------|------|---------------|--------|-----------|---------|---------|-------|---------|----|
|    | penetrated            |      | 105-mm cartrg | MHIPTC | 2.4E-07   | MOBPTC  | 1.58-06 |       |         | 50 |
|    |                       |      | 4.2-in mortar | MHIPTO | 7.3E-07   | MDBPTD  | 4.48-06 |       |         | 50 |
|    |                       |      | tan contnr    | NHIPTK | 8.2E-07   | HDBP TK | 4.98-06 |       |         | Sú |
|    |                       |      | 61.ne         | NHIPTN | 8.65-07   | MDBPTM  | 5.1E-06 |       |         | 50 |
|    |                       |      | 155-ee proj   | MHIPTP | 3.8E-07   | MOSPTP  | 2.38-06 |       |         | 50 |
|    |                       |      | 8-in proj     | NHIPTO | 3.8E-07   | MOBPIC  | 2.JE-06 |       |         | 50 |
|    |                       |      | racket        | HHIPTR | 7.8E-07   | MOSPIR  | 4.7E-06 |       |         | 50 |
|    |                       |      | spray tank    | MHIPTS | 3.4E-)7   | MOBPTS  | 8.4E-06 |       |         | 50 |
|    |                       | A11  | A11           | 16LPT  | 3.28-06   |         |         |       |         | 50 |
| 3. | Prob. pipe penetrated | TEAD | B,K,S(PIPE)   |        |           |         |         | COPTP | 1.JE-02 | 50 |
| 4. | Munition decomates    | ALL  | A11           | DEMHI  | 1.7E-01 D | EMDÐ    | 7.0E-02 |       |         | 2  |



1.2

•

5

े-√ |

## <u>Sequence PO3 - Tornado-Generated Missile Penetrate Munitions in the</u> <u>UPA</u>

Except for accounting for the difference in the structure of the UPA, the same analytical approach described in scenario PO1 was used. The UPA is located on the second floor of the MDB and will contain as many as six onsite containers at any given time. The wall of the MDB is constructed of two layers of thin steel sheets (thickness is approximately 0.047 in.), separated by an insulation material for a total thickness of approximately 2 in. Details of the analysis are given in the calculation sheets (Ref. 7-5).

## <u>Sequence PO4 - Tornado-Generated Missile Penetrate Munitions in the</u> <u>UPA; Burstered Munitions Detonate Upon Impact</u>

The analysis of sequence PO4 follows the same approach as sequence PO2. The probability of munition detonation is calculated from the missile impact velocity upon penetration. Details are given in the calculation sheets (Ref. 7-5).

## <u>Sequence PO5 - Tornado-Generated Missile Breach the Outdoor Agent</u> <u>Piping System at the Modified CAMDS Bulk-Only Facility</u>

Analysis of sequence PO5 also followed the same approach described above except that only a double-walled pipe had to be breached in order to result in an agent release.

#### 7.2.2. <u>Meteorite Strikes</u>

Like tornado-generated missiles, meteorites striking the MHI, MDB, and the outdoor agent piping system at TEAD can lead to a significant amount of agent release. The consequence of such an accident is more severe than that from a tornado-generated missile because meteorite strikes generally involve fires. Hence, if burstered munitions are involved, explosive detonations could occur from the fire or from direct impact, leading to instantaneous agent releases.

The event tree developed for meteorite-initiated accidents is shown in Fig. 7-30. The sequences could not be subjected to any preliminary screening without doing a more detailed analysis of the what type (stone or iron) and size of meteorite is capable of penetrating munitions in the MHI or damaging the MDB which contain not only intact munitions (primarily in the UPA) but a large agent holding tank (in the TOX). 22222534 State 2222222

The accident sequences identified are:

デンロンシングは、「たいいいい」

- PO6 Meteorite strikes the MHI and if burstered munitions are involved, detonations are assumed to occur.
- PO7 Meteorite strikes the UPA and if burstered munitions are involved, detonations are assumed to occur.
- PO7A Meteorite strikes the TOX.
- PO8 Meteorite strikes the outdoor agent piping system at TEAD (CAMDS-modified bulk only facility).

7.2.2.1. <u>Meteorite Strike Accident Analysis</u>. The frequency of meteorite strikes for meteorites weighing 1.0 lb or greater is  $(6.4 \times 10^{-13})/\text{ft}^2$  (Ref. 7-7). For small meteorites (one ton or less), stone meteorites are approximately ten time more common than iron. However, iron meteorites are more dense and tend to have higher impact velocities and therefore represent a significant portion of the total meteorites that can rupture the munitions. The meteorite size distribution data has been presented in Section 4.2.

## Sequence PO6 - Meteorite Strikes the MHI

The munitions in the MHI are stored in their onsite transportation containers. For agent to be released given a meteorite strike, the meteorite has to penetrate 2 ft of soil and 6 in. of concrete roof, the onsite container, and the munition wall. Hence, there are essentially four layers of structural barrier. The minimum meteorite impact velocity that would collapse the 6-in. thick concrete roof is 1500 fps for a stone meteorite and 3800 fps for an iron meteorite. The overall frequency of a meteorite capable of penetrating and rupturing the munitions in the MHI is:

$$\mathbf{F}_{t} = \mathbf{F}(\mathbf{f}_{s} + \mathbf{f}_{i}) \mathbf{A} \mathbf{x} \mathbf{S} , \qquad (7-2)$$

- where F = the frequency of a meteorite weighing one pound or more striking the earth, 6.4 x  $10^{-13}/\text{ft}^2$ ,
  - $f_s$  = fraction of stone meteorites which can penetrate the target,
  - $f_1$  = fraction of iron meteorites which can penetrate the target,
  - $A = target area (80 \times 12 ft),$
  - S = spacing factor.

It is assumed that burstered munitions will detonate when struck by a meteorite. Fire is also expected to occur. Details of the calculations are given in Ref. 7-5.

## Sequence PO7 - Meteorite Strikes the UPA

In this sequence the meteorite has to penetrate the 6-in. thick concrete roof of the MDB, the onsite container, and the munition itself. The same approach described in PO6 is used here. Quantification details are provided in Ref. 7-5.

### Sequence PO7A - Meteorite Strikes the TOX

The TOX is located in the first floor of the MDB. The ceiling of the TOX is a minimum 12-in. thick. This is the most likely area vulnerable to a meteorite strike. Detailed calculations presented in Ref. 7-5 indicate that either a 200-lb stone meteorite or 20-lb iron meteorite can penetrate the TOX ceiling.

## 7.2.3. Aircraft Crashes

The aircraft crash-initiated accidents affecting the MHI and the MDB are similar to those affecting the storage igloos and warehouses. Both direct and indirect (i.e., adjacent to the building) crashes were considered. The aircraft crash may or not result in a fire. Furthermore, the ability to contain the fire in the shortest time possible influences the severity of the accident.

The event trees developed are shown in Figs. 7-41 and 7-42. No preliminary screening could be performed until the actual aircraft crash frequencies at each site had been analyzed. However, once the accident frequencies were quantified, those which have frequencies of  $10^{-10}/yr$  or less were not analyzed for the agent release quantities. The accident sequences that have been defined from the event trees are as follows:

PO9 - Direct large aircraft crash onto the MHI; no fire.

- PO10 Direct large aircraft crash onto the MHI; fire not contained in 0.5 h.
- PO11 Direct large aircraft crash onto the MHI; fire contained in 0.5 h.



P012 - Direct large aircraft crash damages the MDB; no fire.\*

- PO13 Direct large aircraft crash damages the MDB; fire not contained in 0.5 h.\*
- PO14 Direct large aircraft crash damages the MDB; fire contained in 0.5 h.\*
- P015 Indirect large aircraft crash damages the MHI; no fire.
- P016 Indirect large aircraft crash damages the MHI; fire not contained in 0.5 h.
- PO17 Indirect large aircraft crash damages the MHI; fire contained in 0.5 h.
- PO18 Indirect large aircraft crash damages the MDB; no fire.\*
- P019 Indirect large aircraft crash damages the MDB; fire not contained in 0.5 h.\*
- PO20 Indirect large aircraft crash damages the MDB; fire contained in 0.5 h.\*
- PO21 Large and small aircraft direct crash damages the outdoor agent piping system at TEAD; no fire.
- PO22 Large and small aircraft direct crash damages the outdoor agent piping system at TEAD; fire occurs and not contained.

\*Does not include effects of crash on outdoor piping sytem of the modified CAMDS facility at TEAD, which is considered separately.



7.2.3.1. <u>Aircraft Crash Accident Analysis</u>. In summary, the following general assumptions were made in deriving the large/small aircraft accident sequences:

 For a large aircraft crash onto burstered munitions, it is assumed that detonations will occur for direct hits; only rockets and mines detonate from indirect hits; and, if a fire occurs, it is uncontained.

- 2. No small aircraft crashes were assumed to be able to sufficiently damage the MHI or the MDB to cause agent releases.
- 3. The vulnerability of the outdoor agent piping system at the modified CAMDS bulk facility (TEAD) was analyzed separately.

#### Direct Large Aircraft Crash Onto the MHI/MDB; No Fire (PO9, PO12)

Only large aircraft crashes have been found to significantly damage the MDB or the MHI. For a direct aircraft crash, the target area is the surface area of the building. Even if the crash does not lead to a fire, the impact of the crash is strong enough to cause the detonation of burstered munitions. The transportation data presented in Ref. 7-8 indicate that 55% of all air crashes do not involve fires. Quantification details are provided in Ref. 7-5.

# Direct Large Aircraft Crash onto the MHI/MDB; Fire Not Contained in 0.5 h (PO10, PO13)

The analysis of these sequences follows the same approach as PO9 and PO12. The transportation data indicate that 45% of all aircraft crashes result in fires.

The successful containment of the fire is defined here to be 0.5 h for nonburstered munitions. This time was selected based on the thermal failure threshold data presented in Appendix F, which indicate that direct heating of ton containers for 36 min leads to hydraulic rupture. For burstered munitions in onsite containers, the thermal failure threshold is conservatively defined as 15 min, which is the package design criteria for an all engulfing fire. Since the Army policy is not to fight a fire involving direct heating of burstered munitions, the probability of the "not containing the fire in 0.5 h" event is essentially unity.

The amount of agent released from bulk containers subjected to aircraft crash fires depends on the ability to contain the fire. If fire is allowed to progress for more than 30 min, more containers will rupture. The approach for quantifying the probability of successful containment of an aircraft crash fire has been discussed in Section 5.

## Direct Large Aircraft Crash onto the MHI; Fire Contained in 0.5 h (PO11, PO14)

These scenarios essentially apply to nonburstered munitions only. If an airplane crashes directly onto the MHI or MDB containing nonburstered munitions, it is expected that every means available will be employed to terminate the fire immediately. The sooner the fire is extinguished the fewer munitions will be subjected to thermal rupture. Although the munitions are stored in onsite containers they are only provided 15-min protection from an all engulfing fire. The approach for calculating the probability of containing the fire in 0.5 h or less has been discussed in PO10. The quantification details are provided in Ref. 7-5.

## Indirect Large Aircraft Crash onto MHI/MDB; No Fire (P015, P018)

For an indirect crash, the target area is determined by increasing all building perimeters by 200 ft. To determine the probability that the building will be damaged by flying debris from an aircraft crash in the vicinity of the buildinz, the following assumptions were made:

0.0000

- 1. The airplane can skid 100 ft and still damage the MHI.
- 2. The airplane can skid 150 ft and still damage the MDB.
- 3. 10% of all crashes are directed towards the igloo door.
- 4. 25% of all crashes are directed towards the MDB (i.e, either the TOX or the UPA may be hit).

Sector Sector

For the MHI, the total probability of an aircraft part damaging the munition in containers is the sum of the probability that the missile will rupture the structure (including the munition at its line of sight) and the probability that the door is open at the time of the crash and the missile enters the open door and hits the munitions.

The probability that the missile will rupture the structure and the munitions is calculated as follows:

$$P_i = 0.10 \times A_i / A_{LA}$$
, (7-3)

where  $A_i =$  the area of the crash that could damage the igloo door if closed,

 $A_{LA}$  = the target area for an indirect large aircraft crash.

The SAI study (Ref. 7-4) indicates that the igloo door may be open 1% of the time. Since only 10% of all crashes are directed towards the door, the probability that the door is open and missile hits the munition through the open door is 0.001.

For the MDB, it is assumed that the either the TOX or the UPA may be the most vulnerable to a missile strike. Assuming that there was a



25% chance of the airplane crashing towards the TOX or the UPA, the probability of damaging the TOX or UPA is:

$$P_t = 0.25 \times A_t / A_{LA}$$
, (7-4)

where  $A_t$  = the area of crash capable of damaging the TOX or UPA,  $A_{LA}$  = the target area for an indirect crash of a large aircraft.

Quantification details are provided in Ref. 7-5.

## Indirect Large Aircraft Crash Damages the MHI/MDB; Fire Not Contained in 0.5 h (PO16, PO19)

The same approach discussed above is applied to the analysis of these scenarios.

Indirect Large Aircraft Crash Damages the MHI/MDB; Fire Contained in 0.5 h (P017, P020)

The same approach discussed above is applied to the accident frequency analysis of these scenarios. This scenario applies to nonburstered munitions only based on the discussion of scenario PO11.

# Aircraft Direct Crash Damages the Outdoor Agent Piping System at TEAD; No Fire (PO21, PO22)

The present CAMDS facility at TEAD which will be modified to process bulk items only will have a separate building housing the TOX and the LIC. The two buildings will be connected by a 330 ft agent piping system to allow transfer of agent from the bulk drain station to the TOX. This pipe may be damaged by a both a large and small aircraft. The consequence is the same for both large and small aircraft crashes, hence the total aircraft crash frequency is the sum of the large and small aircraft crashes.

### 7.2.4. Earthquakes

The earthquake-initiated accident affecting the MHI is not a credible event since the current plan is to store unstacked munitions in onsite transportation containers in the MHI. The igloo is known to withstand very high intensity earthquakes and the only possibility for an agent release is if the munitions were to fall on a probe and be punctured. Since munitions will be stored in cylindrical containers and will not be stacked, puncture is not possible.

Several areas within the MDB are sensitive to earthquakes in the sense that damage to any of these areas could lead to a significant agent release. The areas of concern are: (1) the UPA where up to six onsite containers may be present; (2) the toxic cubicle (TOX) which houses two agent collection tanks, one of which may be completely full at the time of an earthquake; (3) the ventilation duct; (4) the agent piping system from the bulk drain station (BDS) to the TOX and from the TOX to the liquid incinerator (LIC); and (5) the fuel lines which could break and be ignited by earthquake-initiated electrical sparks.

Figures 7-43 and 7-44 show the event trees developed to identify relevant accident sequences in the MDB involving nonburstered and burstered munitions, respectively. Many event sequences have been screened out from further analysis based on the screening criteria described previously.

The accident sequences which survived the initial screening and have been analyzed further are listed in Table 7-6. Several more sequences were finally screened out after some analysis were performed on the basis of the frequency screening criterion of  $10^{-10}/yr$ .

7.2.4.1. <u>Earthquake Accident Analysis</u>. The earthquake intensity is usually given in terms of maximum acceleration (i.e., g-level). There

is an approximate relationship between the Modified Mercalli Intensity (MMI) scale and the g-level. For example, MMI of VIII is approximately equivalent to 0.15 to 0.30 g.

7.1.4.2. <u>Releases from Earthquake-Induced Accidents in the MDB</u>. Sequences P025 to P034 involve the earthquake-initiated events inside the MDB. Lower intensity earthquakes may keep the munitions in the UPA as well as the agent collection tanks in the TOX intact but could initiate a fire that could subsequently cause the thermal detonation or hydraulic rupture of munitions in the UPA. Otherwise, high intensity earthquakes could cause munitions in the UPA to fall and be punctured, damage the agent collection tanks and the piping system, and also cause fire/explosion due to fuel line breaks. The events modeled are discussed below.

### Releases Involving Bulk Containers

1. Earthquake Occurs. The initiating event (Event 1) in Fig. 7-43 is earthquake occurrence while bulk containers are being processed. To simplify the event tree evaluation, Event 1 further restricts the earthquake intensity to an acceleration range from  $g_1$  to  $g_u$ . Seven ranges are considered:

a. 0.15 g to 0.2 g.
b. 0.2 g to 0.3 g.
c. 0.3 g to 0.4 g.
d. 0.4 g to 0.5 g.
e. 0.5 g to 0.6 g.
f. 0.6 g to 0.7 g.
g. Greater than 0.7 g.

Earthquakes below 0.15 g are not considered in the analysis because the damage probabilities associated with such tremors

is negligibly small. Detailed examinations of seismic ranges above 0.7 g are unnecessary for the MDB because earthquakes above 0.7 g have a probability of almost 1.0 of damaging the MDB. With respect to the TOX, its high seismic design criterion precludes earthquake damage at frequencies above  $10^{-10}/yr$ (see Section 4). Since release scenarios with frequencies below  $10^{-10}/yr$  require no detailed examination, a detailed event tree analysis of seismic ranges above 0.7 g is also unnecessary relative to releases from the TOX.

The initiating event frequency at each site is the sitespecific frequency at which earthquakes in the range,  $g_1$  to  $g_u$ , occur multiplied by the fraction of all bulk containers processed at the site. (Note: since this is classified information, the final frequency results will be adjusted accordingly in the classified appendix.). For an annual risk above ~3 x  $10^{-5}/yr$ , the initiating event frequencies were taken from Fig. 4-11.

2. <u>MDB Not Damaged by the Earthquake</u>. MDB damage is defined as any loss of the MDB's agent containment capability. This includes damage to the MDB confinement walls or the ventilation system. As long as the MDB containment capability is maintained, any agent release inside the MDB (e.g., a release from a punctured munition) results in no appreciable release to the environment. Event 2 damage probabilities are based upon a generic study of damage to structures designed to the UBC.

The MDB (including the pipes and ducts) is designed to meet UBC seismic standards which means that the building is designed with a factor of safety and should not fail given an earthquake of a certain magnitude, depending on the site's seismic zone location. The CONUS facilities are being

designed for a minimum of seismic zone 2 design earthquakes, even though some of the sites may be in seismic zone 1 (i.e., APG, PBA, PUDA, and UMDA). ANAD, LBAD, and NAAP are in seismic zone 2 while TEAD is in seismic zone 3. Thus, the MDB at TEAD is designed to meet seismic zone 3 earthquake standards while the rest of the sites are designed to meet seismic zone 2 standards. The design level for a UBC structure with concrete walls (such as the MDB) is 0.14 g for seismic zone 3 and 0.07 g for seismic zone 2. The design safety factor is generally equal to 2. More details on the failure probabilities are presented in Appendix C. Earthquake Impact on Munition Integrity. The munitions in the UPA represent a significant agent inventory. Event 3 addresses whether the earthquake causes a release from any of these munitions. Puncture is the dominant munition failure

mode. The puncture probability is the probability that the earthquake causes an unpacked munition to fall from the conveyor or while it is being placed on the conveyor (this probability is conditionally dependent on seismic intensity) and that the fallen munition strikes a probe of sufficient size and density to penetrate it (the probe penetration probability is a function of munition type, see Ref. 7-5).

3.

Packed munitions are not stacked in the UPA. Ancillary studies indicate that the probability that a packed (or palletized) munition falls or is knocked over and strikes a probe of sufficient size to penetrate it (including penetrating any intervening packing material) is negligibly small relative to the  $10^{-10}/yr$  screening criterion. Thus, only single munition punctures are addressed in Fig. 7-44.

TOX Integrity Maintained. The TOX, which may contain up to 500 gal of agent, also represents a potentially significant

7-97

release source. To minimize the potential of a release, the TOX room, tanks, and piping are being designed to meet the more stringent NRC standards and can survive earthquakes that engender MDB damage. The design g-level has not yet been determined but the intent is to ensure that the TOX will withstand relatively high g-forces. The same criteria will be applied to all sites regardless of the seismic zone location. For this analysis, it is assumed that the TOX will be designed for a 1-g safe shutdown earthquake (SSE) at all sites.

SUCCESSION STATES

The high TOX design criterion virtually assures that TOX integrity will be maintained after all but the strongest (i.e., greater than 1 g) earthquakes. In order to quantify this contention, it is necessary to extrapolate the seismic hazard model in Fig. 4-11 to higher acceleration levels. This extrapolation is depicted in Fig. 7-45. The extrapolation is conservative for two reasons:

- a. Linear logarithmic extrapolation results in the seismicity models for contour levels 0.05 through 0.20 intersecting the contour level 0.40 curve. Since the seismic hazard of a geological region is directly related to the associated contour level value, it is unlikely that the seismicity model for a region with a low hazard (e.g., a 0.10 contour level) will intersect the seismicity model for a region with a larger seismic hazard (e.g., a 0.20 contour level).
- b. Most seismologists now believe that there is a physical upper limit to the amount of seismic energy that the earth can transmit. Although this upper limit depends upon site specific geological characteristics, for the MDB sites being considered it is estimated that this upper limit restricts ground acceleration to a maximum



Fig. 7-45. Extrapolated seismic hazard model

7-99

value of around 0.6 to 0.8 g. Therefore, Fig. 7-45 is probably conservative by including effective peak accelerations above 0.8 g.

Figure 7-46 is the TOX fragility model corresponding to a 1 g SSE design (Appendix B includes the TOX fragility model derivation). By combining Figs. 7-45 and 7-46, it was determined that no event sequences involving TOX damage have a frequency of  $10^{-10}/yr$  or greater.

5. <u>Ignition Avoided</u>. Available data indicate a high likelihood of earthquake-induced fires in both residential and commercial structures. Fig. 7-47 is the fault tree used to quantify the probability that an earthquake-initiated fire (or detonation) originates in the MDB.

Three mechanisms for ignition are identified. The first involves combustible material ignition by hot process equipment (e.g., a kiln or burner). Because of the high operating temperatures of this equipment, the ignition probability for Event X1 is essentially the probability that combustible material remains in contact with a hot surface long enough to ignite. If the MDB is not damaged by the earthquake the Event X1 probability is small relative to the probability of ignition from other mechanisms identified in Fig. 7-47. However, if the MDB is damaged by the earthquake, the Event X1 probability is essentially unity.

Natural gas ignition can result in either a fire or a detonation, depending upon the MDB integrity. If the MDB is intact, it is expected that detonation will result from a natural gas ignition. However, if the MDB is damaged by the earthquake, the buoyant natural gas cannot readily form a large detonable mass. Therefore, Fig. 7-47 models fire as the consequence of





34 **336334 "**262204" <u>2626</u>20

Fig. 7-46. TOX fragility model

ŊĸŊĸŎĸŎĊŎĔŎĸŎĔŎĊŎŎŎŎŎŎŎŎŎŎŎŎŎŎŎŎ



CALCULATION OF THE OWNER

1.1.1.1.1.1.1.1.1.1.1.

. ) . . X



7-102



natural gas ignition when the MDB is damaged by the earthquake, and detonation as the consequence of natural gas ignition when the MDB is intact.

Three criteria must be satisfied for natural gas to ignite inside the MDB:

- a. A natural gas line leak must occur inside the MDB.
- A supply of natural gas must be available from the external distribution system.
- c. An ignition source is required.

The third ignition mechanism addressed in Fig. 7-47 is an electrical fire. The conditions necessary for an electrical fire are:

a. An electrical fault (i.e., arcing) inside the MDB.b. A supply of electric power to the faulted equipment.

Event X3 is an important factor in evaluation Fig. 7-47 because available data indicate that offsite power can be lost at a relatively low seismic intensity.

6. <u>Fire Suppression Successful</u>. Successful fire suppression is defined as extinguishing a fire before it increases the amount of agent available for release to the environment. The UPA and TOX are the major areas of concern. Since the TOX tank is vented, over pressurization is not a problem. Moreover, the temperatures produced by a fire are insufficient to directly fail the tank or agent piping. Hence, the principal concern is thermal failure of munitions in the UPA. Fig. 7-48 is the fire suppression success tree. If the fire originates in the UPA (Event X1), 30 min are available to



7-103



1000

ALL PRESS DURING SAMA DURING DURING

 $\tilde{\mathcal{A}}$ 

21



<u>}</u>

<u>ᠵ᠆ᢣ᠋᠘ᢉᠼᡳᡭ᠕ᡀᡀᡀᡀᡀᡀᡀᡀᡀᡀ</u>

323-X-X


suppress it before the bulk containers fail. If the MDB is intact (i.e., it has not been damaged by either the earthquake or a natural gas detonation), applicable data indicate a 76% chance of successfully suppressing the fire. If the MDB is damaged, the likelihood of suppressing a UPA fire within 30 min is effectively zero.

Fires that originate outside the UPA must propagate to the UPA and burn for 30 min before any bulk containers fail (Gate G3 in Fig. 7-48). If the MDB is intact, the fire walls preclude the propagation to the UPA. If the MDB is damaged by the earthquake, the probability of Event X4 is predicated upon extrapolating a fire propagation model developed for nuclear power plants, and is a function of the distance from the fire to the UPA. Finally, if the MDB is damaged by a natural gas detonation, successful fire suppression is conservatively ignored.

Event 6 is quantified with respect to whether the fire damages any containers in the UPA. However, if agent is released from the TOX, the dispersion mechanism is dependent upon agent combustion. Agent dispersion with combustion occurs only if any one of the following conditions is satisfied:

- a. Natural gas detonation occurs.
- b. The TOX and MDB are both damaged and a fire occurs.

#### Releases Involving Burstered Munitions

The salient differences between Figs. 7-43 and 7-44 relate to Events 1, 3, 5, and 6. The initiating event frequency (Event 1) in Fig. 7-34 is the site-specific frequency at which earthquakes in the range,  $g_1$  to  $g_u$ , occur multiplied by the fraction of all munitions that



will be processed at the site that are burstered (this will be given in the classified appendix).

In addition to puncture, detonation is an important failure mode when burstered munitions are being processed (Event 3, Fig. 7-44). If the earthquake causes a munition detonation in the UPA, the probabilities of ignition and successful fire suppression (Events 5 and 6) are altered. Specifically, the conditional ignition probability is unity, subsequent to a munition detonation in the UPA. Moreover, a munition detonation in the UPA essentially precludes successful fire suppression. If the earthquake causes a fire but does not directly detonate any munitions, the fire suppression probability is quantified with the Fig. 7-12 success tree. However, the time available to suppress the fire is only 10 min for burstered munitions and there is no intervention from plant personnel or site fire fighters.

Uncertainties for the MDB earthquake events were evaluated as follows:

## Event 1: Earthquake Occurs

The uncertainty in the initiating event frequency is represented by a lognormal distribution with an uncertainty factor of 10 and a median value equal to the point frequency estimate. This is predicated upon the generic guidelines issued for the uncertainty assessment (see Table 5-21).

# Event 2: MDB Not Damaged by the Earthquake

Uncertainty factors for MDB damage probabilities above 0.1 will also be taken from Table 5-21. For failure probabilities below 0.1 an uncertainty factor of 3 is assigned. The uncertainty distribution in each case is lognormal with a median equal to the MDB failure probability. Event 3: Earthquake Impact on Munition Integrity

Table 5-21 recommendations for probabilities of 0.1 or greater are applicable to the uncertainty in the probability that a munition falls from the conveyor. An uncertainty factor of 5 is applied to  $P_p$  - the conditional probability that a munition is punctured subsequent to a fall. Since all event sequences involving a munition detonation have frequencies below 10<sup>-10</sup>/yr, they require no uncertainty analysis. The uncertainty distributions for the Event 3 parameters are lognormal with medians equal to the point probability estimates.

# Event 4: TOX Integrity Maintained

Uncertainty factors for TOX damage probabilities above 0.1 will also be taken from Table 5-21. For failure probabilities below 0.1 an uncertainty factor of 3 is assigned. The uncertainty distribution in each case is lognormal with a median equal to the TOX failure probability.

Event 5: Ignition Avoided

The Event 5 uncertainty results from the uncertainties in the following functions and parameter.

- 1.  $f_{X2}(x) \rightarrow$  probability density function for inside pipe failure.
- 2. f<sub>X3</sub> (x) + probability density function for underground pipe
  failure.
- 3. Pr  $(X_4) \rightarrow$  natural gas ignition probability.
- f<sub>X5L</sub> (x) → probability density function for light fixture failure.

7-107

5. f<sub>X5I</sub> (x) → probability density function for industrial circuit failure.

Y Y MANAKANA KANA TATATA YA MAYANA YA MANAKANA

- Y. M. N. N.

6.  $f_{X6}(x)$  + probability density function for offsite power loss.

In general:

$$f_{j}(x) = \frac{1}{\beta_{R,J} \times \sqrt{2\pi}} \exp \frac{-[\ln(x) - \ln(\alpha_{J} \epsilon_{U,J})]^{2}}{2\beta_{R,J}^{2}}$$

Moreover, the uncertainty in each Event 5 fragility is a function of the uncertainty on  $\epsilon_{U,J}$ , as was described previously for warehouse fires. From Table 5-20, the uncertainty factors for  $\epsilon_{U,X5L}$  and  $\epsilon_{U,X6}$  are 2 and 1.5, respectively. Uncertainty factors for  $\epsilon_{U,X2}$  and  $\epsilon_{U,X5I}$  are from the Zion and Seabrook PRAs. The value of  $\epsilon_{U,X2}$  is directly applicable to the MDB, but the uncertainty factor for  $\epsilon_{U,X51}$  is obtained from the Seabrook data plus an additional factor of 2 that arises from concerns about the applicability of a nuclear data base on the MDB design.

The major uncertainty in  $\epsilon_{U,X3}$  is due to applying a generic Modified Mercalli fragility model to the MDB. Depending upon the actual soil conditions and pipeline characteristics, the median failure threshold can vary about the nominal value by a factor of 2. Thus, an uncertainty factor of 2 is adopted for  $\epsilon_{U,X3}$ .

Approximately a binominal distribution with a normal distribution, the uncertainty factor for Pr (X4) is 1.5. A lognormal distribution is modeled. These results are tabulated in Table 7-10.

Event 6: Fire Suppression Successful

The uncertainty in most fire suppression model functions and parameters (e.g., the probability of a pipe failure or loss of offsite



Service Services

201

| Parameter | Median | Uncertainty<br>Factor |
|-----------|--------|-----------------------|
| €U,X2     | 1      | 2.2                   |
| €U,XE     | 1      | 2.0                   |
| Pr (X4)   | 0.0067 | 1.5                   |
| €U,X5L    | 1      | 2.0                   |
| €U,X5I    | 1      | 2.8                   |
| €U,X6     | 1      | 1.5                   |

# TABLE 7-10EVENT 5 STATISTICAL PARAMETERS

3







power) was previously addressed for Event 5. Only three additional parameters require uncertainty models:

- 1. Operator error probability.
- 2. Damper failure probability.
- 3. Fire suppression failure probability.

According to information from Battelle-Columbus, the uncertainty in the operator error probability is lognormally distributed with an uncertainty factor of 10 and a median equal to the error probability. Data in EGG-EA5887 support a similar model for the damper failure probability. The fire propagation probability has a lognormal distribution with an uncertainty factor of 3 for fires originating outside of the UPA. For fire suppression inside the UPA the Table 5-21 guidelines are recommended. In both cases the nominal probabilities represent distribution medians.

7.2.4.3. <u>Earthquake-Induced Releases Involving the Outdoor Agent Piping</u> <u>System at TEAD</u>. The analysis of the earthquake scenarios involving the MDB for the modified CAMDS facility at TEAD includes the rupture of the agent piping system between the BDS and the TOX at TEAD. The agent pipe line is assumed to be double walled and approximately 330 ft long. The analysis also assumes that this pipe will be designed to NRC standards which means that the pipe should not fail at 1.0-g earthquake.

# 7.2.5. Quantification of Logic Models

The data base used for the quantification of the external event sequences are presented in Table 7-9 and in Tables 7-11 through 7-13.

7.2.5.1. <u>Tornado Accident Frequencies</u>. The data base used for the accident scenario analysis is listed in Table 7-9. The site-specific tornado frequency versus velocity curves have been presented in Section 4. Two types of missiles were initially considered: a (1) 3-in.



# TABLE 7-11 DATA BASE FOR METEORITE INITIATED PLANT ACCIDENT SEQUENCES

24-Jul-87 Page I

# Data Base For Meteorite Initiated Plant Accident Scenarios

| Event                                                 | Site       | Munition /<br>structure                                                                                     | Variadie                                                             | Input Data                                                                                      | Error                        | Fererence                     |
|-------------------------------------------------------|------------|-------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------|-------------------------------------------------------------------------------------------------|------------------------------|-------------------------------|
| I. Frequency of meteorite<br>strike (events/sq-ft-yr) | ə11        | All                                                                                                         | METEOR                                                               | 6.4E-13                                                                                         | 1.0E+01                      | kei. 7 <b>-8</b>              |
| 2. Probability munition in igloo breached             | 411        | Bomb<br>4.2-in mort<br>105-mm catrg<br>ton contnr<br>mine<br>155-mm proj<br>B-in proj<br>rocket<br>sp. tank | NEIB<br>Meid<br>Meic<br>Neik<br>Meik<br>Meip<br>Meip<br>Meir<br>Meis | 2.4E-06<br>1.6E-06<br>9.7E-07<br>3.3E-06<br>2.4E-06<br>7.4E-07<br>7.4E-07<br>3.3E-06<br>5.6E-06 | 1.0E+01                      | See cair sheets<br>(Ref. 7-3) |
| 3. Probability munition in<br>UFA breached            | Ali        | Bomb<br>4.2-in mort<br>105-sm catrg<br>ton contmr<br>mine<br>155-sm proj<br>B-in proj<br>rocket<br>sp. tank | NEUPB<br>Meupd<br>Meupc<br>Meupk<br>Meupp<br>Meupp<br>Meupg<br>Meups | 7.9E-04<br>5.5E-04<br>3.0E-04<br>1.1E-03<br>7.9E-04<br>2.4E-04<br>2.4E-04<br>1.1E-03<br>1.8E-03 | 1.0E+01                      | See calc sheets<br>(Raf.7-3)  |
| 4. Probability TOX is breached                        | <u>411</u> | H) 1                                                                                                        | METDX                                                                | 1.1E-05                                                                                         | 1.0E+01                      | Rej. 7-3                      |
| 5. Probability outside agent<br>pipe breached         | TEHD       | Pipe                                                                                                        | MEPIPE                                                               | 7.2E-02                                                                                         | 5.vE+00                      | Rey 7-3                       |
| o. Target area (sq-it)                                | +i!        | lgica<br>uPu<br>Tūt<br>Pipe                                                                                 | IGL<br>UP4<br>MD8<br>FIFE                                            | 9.68+++2<br>5.78++13<br>4.48+94<br>6.68+42                                                      | אסמפ<br>המחפ<br>המחפ<br>המחפ | Rey 7-3                       |







| Εł | fecti | ve Tarc | jet Area | (Sq-Mi) | Direct | Crash |
|----|-------|---------|----------|---------|--------|-------|
|----|-------|---------|----------|---------|--------|-------|

| SITE | VARIABLE NAME | AREA (SQ-MI) |  |
|------|---------------|--------------|--|
| (B)  | (C))          | (D)          |  |

| SO-FT IG  | 100 IGL8 | BODR 7. | . 6 | 0E-05 |
|-----------|----------|---------|-----|-------|
| MDB       | MDBI     | DR 1.   | . 7 | 7E-03 |
| CAMDS_FIF | PE CDSP  | PI 1.   | . 1 | 8E-03 |

Effective Target Area (Sq-Mi) Indirect Crash

| SITE | VARIABLE NAME | AREA (SQ-MI) |
|------|---------------|--------------|
| (B)  | (0)           | (D)          |

| 80-FT | IGLOO | IGLSOIR | 7. | 2   | 68-03  |
|-------|-------|---------|----|-----|--------|
| MDB   |       | MDBIR   | ٤. | . 2 | 12E-02 |



22.22

in the second second

# TABLE 7-13 AIRCRAFT CRASH DATA

| (U)<br>DATA BASE | (P)<br>. FOR ALKCRAFT       | (0)<br>CRASH- INTTIATEI  | (R)<br>) SCENARIC | (S)<br>DS FOR PLANT OP | (T)<br>ERATIONS |
|------------------|-----------------------------|--------------------------|-------------------|------------------------|-----------------|
| IN AS J          | Vaki) and E. Fr.<br>To j P. | REDENCY OR<br>RODABILITY | LINN              | ERROR FACTOR           | REFERENCE       |
| (U)              | (F)                         | (0)                      | (R)               | (2)                    | (1)             |
| 1. Latrop-       | aircratt cra÷               | A (direct) onto          | MDB               |                        |                 |
| AHAD             | LDAN                        | 1.4E-08 per              | facility          | 10                     | Ref. 7-3        |
| 計し               | 1. Dral -                   | 9.4E 10 Yr               |                   | 10                     |                 |
| 1 HAD            | LOUP                        | 8.0107                   |                   | 10                     |                 |
| Net of the       | t DNA                       | 8.1E 05                  |                   | 10                     |                 |
| PBA              | 8. IO'I                     | 2.76-03                  |                   | 10                     |                 |
| E UDA            | 0.40.)                      | 1.05-07                  |                   | 10                     |                 |
| TEAD             | 1.01C                       | 6.4E-10                  |                   | 10                     |                 |
| H-II-H-I         | 1411/1                      | 2. 6E~08                 |                   | 10                     |                 |
| 2. Large         | aircraft cras               | sh (Indirect) ont        | to MDB            |                        |                 |
| ÂNHD             | L.A.A.N                     | 0 6E-08 per              | facility          | 10                     | Ref. 7-3        |
| AL 10            | Leven                       | 6.3E+02 yr               |                   | 10                     |                 |
| 1 HAD            | LALF                        | 5.5E -08                 |                   | 10                     |                 |
| the no           | L milter                    | S. 4F 00                 |                   | 10                     |                 |
| 111.1            | LEALE                       | 1.80-06                  |                   | 10                     |                 |
| Fit Joés         | 1 H H                       | 7.2E 402                 |                   | 10                     |                 |
| TLAD             | LATE                        | 4.40.09                  |                   | 10                     |                 |
| 111.001          | 1 2012                      | 1. 8E-07                 |                   | 10                     |                 |
| á. Large         | direrati erde               | h (direct) onto          | THM               |                        |                 |
| 신태산 /            | DIAN                        | 6.0E 10 per              | facility          | 10                     | Ref. 73         |
| eat a l          | 1)1-4                       | 4.0E 11 yr               |                   | 10                     |                 |
| LEAD             | 1111                        | 3.4E - 10                |                   | 10                     |                 |
| NGAL             | DING                        | 5. SE -10                |                   | 10                     |                 |
| 1.116            | D.11-0                      | 1.16-10                  |                   | 10                     |                 |
| 1104             | 0.414                       | 4 <b>.</b> 5E 05         |                   | 10                     |                 |
| 11-1412          | D110                        | 2.76 11                  |                   | 10                     |                 |

South and the second second stress and second second second second second measured because the second second s

# TABLE 7-13 (Continued)

(...) 

| REFERENCE                | (1)            | Kef. 7 3                                                                                                                         | LJ<br>Pert, 2005<br>EJ                                                       | isef. ≻<br>An internationalise                 | Ref. 2 5<br>Ref. 2 5                                                  |
|--------------------------|----------------|----------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------|------------------------------------------------|-----------------------------------------------------------------------|
| ERROR FACTOR             | (5) 10         | 000000000                                                                                                                        | none<br>1.4                                                                  |                                                | none<br>crash<br>none<br>and indirect)                                |
| LING                     | (R.)           | onto MIN<br>er facility<br>-                                                                                                     | л соле<br>1100е<br>1100е<br>1100е                                            |                                                | none<br>none<br>rd) direct<br>none<br>d -direct a<br>none             |
| COUENCY OR<br>COBABILITY | (0)<br>1.1E-00 | (indirect) c<br>5.7E-08 pe<br>3.3E 09 yr<br>3.3E-08<br>3.3E-08<br>1.1E-08<br>1.1E-08<br>1.1E-08<br>1.1E-07<br>1.1E-07<br>1.1E-07 | irect crash<br>1.0E400<br>ndirect crash<br>1.7F-01<br>hrect crash<br>8.0E-01 | ndirect crast<br>2.05-03<br>ve fire<br>5 55-01 | re<br>4.5F 01<br>1/2 hr (burst<br>1.0F+00<br>hr (nonburstr<br>3.4E-04 |
| VARIARLE FRE<br>ID FF    | (F)<br>D1UN    | aircraft crast<br>AIAF<br>AIAF<br>AINA<br>AIPB<br>AIPU<br>AIPU<br>AITE<br>AIUM                                                   | eached given c<br>BD<br>eached given i<br>BA<br>BA<br>eached given c         | eached given i<br>IA<br>does not invol<br>NE   | results in fi<br>YF<br>not contud in<br>FNCH<br>contud in 1/2<br>FCNE |
| EVENI                    | (D)<br>UMDA    | 4. Large<br>ANAD<br>APG<br>LEAD<br>NAAF<br>FFA<br>FFA<br>FUDA<br>TEAD<br>UMDA                                                    | 5. HDB bri<br>6. MDB bri<br>7. MHI bri                                       | 9. Orash (                                     | 13. Crash<br>14. Fire (<br>15. Fire (                                 |



TOOOSE PYTON TOUR PORTAGE STATES







TABLE 7-13 (Continued)

|       | SN              |
|-------|-----------------|
|       | OFERATIO        |
| (B)   | FLANT           |
|       | FOR             |
| Э     | SCENAR 105      |
| ((1)) | CRASH-INITIATED |
| (P)   | AIRCRAFT        |
| Ū     | f:0R            |
| (0)   | DAIA BASE       |

|                             |                         | . 7 -3     | ю<br>              |
|-----------------------------|-------------------------|------------|--------------------|
| REFERENCE                   | (1)                     | Ref        | Ref                |
| ERROR FACTOR                | (3)                     | anon       | 10                 |
| UNIT                        | (R)<br>urstrd)          |            | e ac crima<br>none |
| FREQUENCY OR<br>FROBABILITY | (g)<br>in 1/2 hr (nent) | 1.0E+00    | 01140 00000 P1P    |
| ARIABLE<br>ID               | (P)<br>contrid          | FNONB      |                    |
| E VENT VI                   | (0)<br>16. Fire not     | 17 Airreat |                    |

pipe and a (2) utility pole. For all munition types, it was found that the utility pole had a higher probability of penetrating munitions in the UPA and the igloo (with a steel door). Hence the data shown in Table 7-9 apply only to the cases where a utility pole was the missile. Also shown in the table are the error factors assigned to each variable. In many cases there was insufficient statistical information to adequately assess the data uncertainty and, therefore, the assignment of error factors was by engineering judgment. The results of the accident frequency analysis are presented in Table 7-14. All the accident scenarios were screened out on the basis of 1 x  $10^{-10}$ )/yr frequency criterion.

7.2.5.2. <u>Meteorite Strike Frequencies</u>. The data base used for the accident scenario analysis is presented in Table 7-11. More details on the derivation of these values are given in the calculation sheets (Ref. 7-5). The results of the accident frequency analysis are presented in Table 7-14. As indicated in the results, the frequencies of meteorite-initiated accidents for all scenarios are below  $10^{-10}/yr$  and hence these scenarios have been screened out from further analysis.

7.2.5.3. <u>Aircraft Crash Frequencies</u>. The data used in the analysis of the aircraft crash accidents are presented in Tables 7-12 and 7-13. The derivation of the aircraft crash frequency values at each site has been discussed in Section 4. The results of the analysis are shown in Table 7-14. The following scenarios can be screened out on the basis of the 1.0 x  $10^{-10}/yr$ :

PO11 - Direct large aircraft crash onto the MHI; fire contained in 0.5 h.

P014 - Direct large aircraft crash onto the MDB; fire contained in 0.5 h.

P015 - Indirect large aircraft crash onto the MHI; no fire.





1.6.1.8.6°6.6°2.6°2.8\*1

TABLE 7-14 PLANT OPERATIONS DATA File: FLTCOL.Wk1, 20-Aug-87 FAGE1

N MANANA MANANA NA KAKAMAN

\*\*\*\*

| PL:<br>Media    | ANT G<br>N ACC | PERATIONS  | COLLOCAT:<br>UENCY ( I | ION<br>Per year :    |                           | PLAN<br>Median 4 | FOPERATIONS COLLOCATION<br>ACCIDENT FREQUENCY + FER | YEAF |
|-----------------|----------------|------------|------------------------|----------------------|---------------------------|------------------|-----------------------------------------------------|------|
| SCENARIO        | NC.            | ANAD RDC   | RANSE                  | TEAD RDC             | FANGE                     | TEAD NDC         | AANGE                                               |      |
| 1.6.            |                | FREQ       | FACTOP                 | FREQ                 | FACTOR                    | FRED             | FACTOR                                              |      |
|                 |                |            |                        |                      | •••••                     |                  | •••••                                               |      |
| Púl - Ta        | rnado          | -generateo | #1551ie                | puncture/cr          | ush munitions in the MHL. |                  |                                                     |      |
| FOBSC           | 1              | N. Á       | •-                     | 1.5E-16              | <b>94</b>                 | 1.5E-16          | 92                                                  |      |
| ÷00⊭C           | 1              | 52-17      | 94                     | 1.3E-15              | 94                        | 1.3E-15          | C.4                                                 |      |
| P0030           | -              | 2.15-17    | 94                     | 4.3E-16              | 94                        | 4.3E-16          | 34                                                  |      |
| POCHC           | :              | 1.12-13    | ÷4                     | 4.3E-16              | 94                        | 4.3E-16          | S4                                                  |      |
| FOLGO           | 1              | N A        |                        | 2.0E-1o              | 94                        | 2.0E-16          | 94                                                  |      |
| F 💷 H i         | :              | 1.12-13    | 94                     | 2.0E-16              | 94                        | 2.0E-16          | ;;                                                  |      |
| PGr V I         | 1              | 1.12-13    | 94                     | 2.0E-1o              | <u>94</u>                 | 2.0E-16          | 94                                                  |      |
| P1 5 4          | 1              | 5.5E-13    | 94                     | 1.5E-15              | 94                        | 1.5E-15          | 94                                                  |      |
| POP3C           | :              | 2.8E-13    | <b>\$</b> 4            | 5.7E-18              | C4                        | 5.7E-16          | 94                                                  |      |
| FIF             | :              | 2.8E-13    | <del>94</del>          | 5.78-10              | 94                        | 5.7E-16          | 94                                                  |      |
| FCPU <u>C</u>   | 1              | 1.62-13    | 94                     | 5.7E-10              | 94                        | 5.7E-16          | 94                                                  |      |
| P1130           | 1              | 2.85-17    | 94                     | 6.35-16              | 94                        | 6.8E-16          | <del>7</del> 4                                      |      |
| F02VC           | 1              | 11 A       |                        | 0.85-16              | 94                        | 6.8E-16          | 94                                                  |      |
| - 265 [         | :              | 9. sE-10   | 94                     | 1.4E-15              | <b>9</b> 4                | 1.4E-15          | <u>94</u>                                           |      |
| PGRVC           | i              | 8.65-13    | 94                     | 1.4E-15              | 94                        | 1.4E-15          | 94                                                  |      |
| FOSVE           | :              | N/4        |                        | 3.7E-16              | 94                        | 3.7E-1o          | 92                                                  |      |
| 201 - Co        | rnado          | -generated | Alssile                | detonate mu          | nitions in the MHI.       |                  |                                                     |      |
| -05KC           | 2              | 1.72-13    | 99                     | 2. <sup>-</sup> E-10 | 99                        | 2.7E-16          | 99                                                  |      |
| 20190           | 2              | 4.5E-14    | <del>9</del> 9         | 9.1E-17              | 99                        | 9.1E-17          | 65                                                  |      |
| F1146           | :              | 4.5E-14    | 99                     | 9.1E-17              | <b>99</b>                 | 9.1E-17          | <b>?</b> \$                                         |      |
| PCHVC           | 2              | 2.0E-13    | 99                     | 3.2E-16              | 99                        | 3.2E-16          | ę=                                                  |      |
| P0P61           | 1              | 5.0E-14    | 99                     | 1.28-16              | 95                        | 1.2E-10          | 00                                                  |      |
| \${\$4 <u>}</u> | 2              | 6.0E-14    | 99                     | 1.25-16              | ବହ                        | 1.2E-16          | 0 <b>9</b>                                          |      |
| 20800           | -              | 5.0E-14    | 63                     | 1.2E-16              | 99                        | 1.2E-1o          | 95                                                  |      |
| C 1081          | 2              | :.0E-14    | 20                     | 1.5E-16              | 99                        | 1.5E-1c          | ĉċ                                                  |      |
| 400vC           | 2              | R/A        |                        | 1.58-16              | 99                        | 1.5E-16          | 63                                                  |      |
| F 3F 6C         | 2              | 1.9E-13    | 00                     | 3.0E-16              | 99                        | 3.08-16          | 99                                                  |      |
| P.3P. 2         | 1              | :.9E-13    | <del>7</del> 9         | 3.0E-16              | ¢Ş                        | 3.0E-16          | <b>0</b> 0                                          |      |

POT - Toroado-generated esselle puncture/crush eunitions in the UPA.





File: PLTCOL.WK1, 20-Aug-87 FASE2

Ŕ

8463335

Ţ,

| PL)                  | NT OPERATIONS COLLOCATION<br>NACCIDENT FREQUENCY & FER NYEAR / |                     |                 |                  |                    |                 | PLANT OPERATIONS COLLOCATION |                  |                 |           |       |
|----------------------|----------------------------------------------------------------|---------------------|-----------------|------------------|--------------------|-----------------|------------------------------|------------------|-----------------|-----------|-------|
| ~edia                | 450                                                            | ISENT FREE          | NGENCY († 1     | FER YEAR /       |                    |                 |                              | MEDIAN           | ACCIDENT FRESL  | ENCY PE   | r Enf |
| SCEN4719<br>1.0.     | NC.                                                            | ANAD RDC<br>Freq    | RANGE<br>Factor | TEAD POC<br>Freq | RANGE<br>Factor    |                 |                              | TEAC NDC<br>Fred | RANGE<br>FACTOR |           | ,     |
|                      |                                                                | •                   |                 |                  | <b>.</b>           |                 |                              |                  |                 |           |       |
| PC350                | :                                                              | N/A                 |                 | 2.18-15          | Ç3                 |                 |                              | 2.1E-15          | 54              |           |       |
| PC 0HC               | 3                                                              | a.83-12             | 94              | 7.92-15          | <b>\$</b> 4        |                 |                              | 7.9E-15          | 94              |           |       |
| P0080                | 2                                                              | 2.02-12             | 94              | 2.7E-15          | 74                 |                 |                              | 1.7E-15          | \$4             |           |       |
| F00H0                | 5                                                              | 2.3E-13             | 74              | 2.7E-15          | 94                 |                 |                              | 1.7E-15          | 74              |           |       |
| P0x50                | 7                                                              | N/4                 |                 | 3.5E-15          | 74                 |                 |                              | 2.6E-15          | 94              |           |       |
| POTHE                | ;                                                              | J. 68-12            | 94              | 3. oE-15         | <b>94</b>          |                 |                              | 3.6E-15          | 94              |           |       |
|                      | -                                                              | 7.6E-17             | 94              | J.6E-15          | 94                 |                 |                              | 3.08-15          | 54              |           |       |
| F(*7)                | ÷                                                              | a.96-13             | 54              | 9.2E-15          | 94                 |                 |                              | 9.28-15          | 4               |           |       |
| - 195                | -                                                              | 1.58-10             | <b>94</b>       | 4,15-15          | 94                 |                 |                              | 4.15-15          | 54              |           |       |
| FOFHE                | -                                                              | 1.57-12             | 04              | 4,18-15          | 94                 |                 |                              | 4.15-15          | 94              |           |       |
| ÷                    | -                                                              | 1.55-11             | 19              | 4, 12-15         | 94                 |                 |                              | 4 15-15          | 14              |           |       |
| F0160                | -                                                              | 7,57-12             | 94              | 4.16-15          | 94                 |                 |                              | 4 15-15          | 94              |           |       |
| 10001                | ÷                                                              | N. 1                |                 | 4 15-15          | 94                 |                 |                              | 1 16-15          | 54              |           |       |
|                      | :                                                              |                     | 94              | 8 55-15          | C4                 |                 |                              | 6 55-15          | 64              |           |       |
| 215.25               | •                                                              | 7 15-15             | 94              | 8 55-15          | Q.1                |                 |                              | 5 55-15          | 40              |           |       |
| 572 K                | -                                                              |                     |                 | : 5E-14          | 54                 |                 |                              | 1 52-12          | 24              |           |       |
|                      |                                                                | n.r.<br>Tateraneter |                 | 11.25 1-         | ,-                 | the URA         |                              | 1.JL-14          | ••              |           |       |
|                      | 1000                                                           | - 4646, 2000        | 04<br>80        | 2 SELIS          | 01 292 1910.<br>04 | the pro.        |                              | 0 45             |                 |           |       |
|                      | -                                                              | 40                  | 6.4             | 0.4E-12          | 74                 |                 |                              | 0.4E-10          |                 |           |       |
| 5000                 | -                                                              | 3 40-12             | 04              | 14<br>14         | C 1                |                 |                              | 5 05.10          | - 4             |           |       |
| 5 4 4 7 4<br>5 5 6 1 |                                                                |                     | 2.4<br>2.4      | 2.72-10          | 04<br>D4           |                 |                              | 1.72-10          | 74              |           |       |
| - 1                  | 7                                                              |                     | · · •           | 7.02-10          | 74                 |                 |                              | 05-10            |                 |           |       |
| toret                |                                                                | 2472742             | 54              | 4.4E-16          | 74                 |                 |                              | 4.41-10          |                 |           |       |
| 5.5 M.               | ;                                                              | 3.703               | 14              | 7.4E-10          | 7-                 |                 |                              | 4.4E-10          | **              |           |       |
| F.F.Y.               | 1                                                              | /1-12               | 74              | 4.42-10          | 44                 |                 |                              | 4.4L-15          | 44              |           |       |
| 1.0.35               | -                                                              | 2. (L=1.            | 44              | 4.42-10          | 44                 |                 |                              | 4.4L-10          | 74              |           |       |
|                      | *                                                              | 11 H                |                 | 4.42-10          | 44                 |                 |                              | 4.41-16          | 44              |           |       |
| **.•*.               | -                                                              | •3E-11              | 94              | 4.4E-15          | 44                 |                 |                              | 4.4E-15          | 54              |           |       |
| PC VI                | ÷                                                              | v. ( <b>E+6</b> 0   | ₹4              | 4.4E-16          | 74                 |                 |                              | 4.4E-16          | 34              |           |       |
|                      |                                                                | -generated          | l ≋issiie       | damages the      | agent pipi         | ng syster betwe | en the BDS and T             | iO¥ at T€        | 140 (bull-only  | facility. |       |
| P0426                | 5                                                              | N/A                 |                 | 2.JE-11          | ÷4                 |                 |                              | 2.3E-11          | 94              |           |       |



File: PLTCOL.WK1, 20-Aug-87 PAGED

| PLA<br>Median    | PLAKT OPERATIONS COLLOCATION<br>MEDIAN ACCIDENT FREQUENC: ( PER YEAR )<br>(HAFIO NO. ANAD ADC RANGE TEAD RDC RANGE<br>1.D. FREQ FACTOR FREQ FACTOR |                  |                 |                  |                 | PLANT OPERATIONS COLLOCHTIGN<br>Nedian accident frequency frequency |  |  |  |  |
|------------------|----------------------------------------------------------------------------------------------------------------------------------------------------|------------------|-----------------|------------------|-----------------|---------------------------------------------------------------------|--|--|--|--|
| SCEHAFID<br>I.J. | NC.                                                                                                                                                | ANAD RDC<br>Freg | RANGE<br>Factor | TEAD RDC<br>Freq | FANGE<br>Factor | TEAD NDC RANGE<br>Freq Freitof                                      |  |  |  |  |
|                  |                                                                                                                                                    |                  |                 |                  |                 |                                                                     |  |  |  |  |
| PCAHE            | 5                                                                                                                                                  | N/A              |                 | (.0 <b>E+</b> 00 | 94              | 0.0E+6) 54                                                          |  |  |  |  |
| 10110            | c                                                                                                                                                  | 6.2              |                 | 0.0F+0L          | 40              | 6.0F+6.                                                             |  |  |  |  |

12195:

1. Frequency unit = events/operating year

I. Scenario 5 applies only to the TEAD bulk-only facility



File: PLTCOL.WK1, 20-Aug-87 PAGE1

| FLI<br>Median    | FLANT OPERATIONS COLLOCATION<br>MEDIAN ACCIDENT FREQUENCY ( PER YEAR )<br>CENARIO NO. ANAD RDC RANGE TEAD RDC RANGE<br>1.0. FREQ FACTOR FREQ FACTOR |                  |                 |                  |                 | PLANT OPERATIONS COLLOCATION<br>Median Accident frequenci - feb (fab |
|------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------|------------------|-----------------|------------------|-----------------|----------------------------------------------------------------------|
| SCENARIO<br>1.9. | ND.                                                                                                                                                 | ANAD RDC<br>Freq | RANGE<br>Factor | TEAD RDC<br>Freq | RANGE<br>Factor | TEAC NDC RANGE<br>Freq Factor                                        |
|                  |                                                                                                                                                     |                  |                 |                  |                 |                                                                      |
| Cis - Met        | eor i                                                                                                                                               | te strikes       | the MHI.        |                  |                 |                                                                      |
| FOREF            | 5                                                                                                                                                   | N/H              |                 | 1.4E-15          | 26              | 1.4E-15 26                                                           |
| : <u>]</u> [-4]  | c                                                                                                                                                   | 9.8E-10          | 26              | 9.8E-1o          | 26              | 9 <b>.8E-16</b> 20                                                   |
| F0060            | ċ                                                                                                                                                   | o.0E-16          | 26              | 6.0E-16          | 26              | 6.0E~16 26                                                           |
| FUCHE            | 5                                                                                                                                                   | o.∂E-le          | 26              | 6.0E-16          | 26              | 6.0E-16 26                                                           |
| €0+9F            | c                                                                                                                                                   | N/A              |                 | 2.0E-15          | 26              | 2.0E-15 26                                                           |
| Filket           | c                                                                                                                                                   | 2.0E-15          | 26              | 2.0E-15          | 26              | 2.0E-15 26                                                           |
| FQ) (F           | 5                                                                                                                                                   | 2.0E-15          | 2è              | 2.0E-15          | 26              | 2.0E-15 26                                                           |
| P3805            | o                                                                                                                                                   | 1.5E-15          | 26              | i.5E-15          | 26              | :.SE-15 26                                                           |
| P0250            | ÷                                                                                                                                                   | 4.62-16          | 26              | 4.oE-16          | 26              | 4.6E-16 20                                                           |
| FOFHC            | 5                                                                                                                                                   | 4.oE-16          | 26              | 4.6E-1ċ          | 26              | 4.6E-10 20                                                           |
| FORVE            | ٥                                                                                                                                                   | 4.oE-10          | 26              | 4.6E-16          | 26              | 4.6E-10 20                                                           |
| F0090            | c                                                                                                                                                   | 4.02-10          | 26              | 4.6E-16          | 26              | 4 <b>.</b> 5E-16 26                                                  |
| PGGVC            | 5                                                                                                                                                   | N/A              |                 | 4.6E-16          | 26              | 4.68-16 26                                                           |
| ÷3930            | e                                                                                                                                                   | 2.1E-15          | 26              | 2.1E-15          | 26              | 2.1E-15 26                                                           |
| 20240            | ÷                                                                                                                                                   | J.4E-15          | 26              | 2.1E-15          | 26              | 2.1E-15 26                                                           |
| FOSVF            | 5                                                                                                                                                   | N/4              |                 | 3.48-15          | 25              | 3.4E-15 26                                                           |
| 207 - Me         | teori                                                                                                                                               | te stri-e        | s the UPA       |                  |                 |                                                                      |
| POSSE            | 7                                                                                                                                                   | N, A             |                 | 2.92-12          | 26              | 2.9E-12 26                                                           |
| FODHC            | -                                                                                                                                                   | 1.0E-12          | 26              | 2.0E-12          | 26              | 2.0E-12 26                                                           |
| 16003            | -                                                                                                                                                   | 1.18-12          | 26              | 1.1E-12          | 26              | 1.1E-12 26                                                           |
| FOCHC            | 7                                                                                                                                                   | 1.1E-12          | 26              | 1.12-12          | 26              | 1.1E-12 20                                                           |
| P0195            | -                                                                                                                                                   | N/6              |                 | 4.0E-12          | 26              | 4.0E-12 26                                                           |
| POrnE            | 7                                                                                                                                                   | 4.0E-12          | 20              | 4,0E-12          | 26              | 4.0E-12 26                                                           |
| FOR F            | 7                                                                                                                                                   | 4E-12            | 26              | 4.0E-12          | 26              | 4.0E-12 2o                                                           |
| ÷0#,5            | ٦                                                                                                                                                   | 2.9E-12          | 26              | 2.9E-12          | 26              | 2.9E-12 26                                                           |
| 2020             | 7                                                                                                                                                   | 9.8E-13          | 26              | 8.8E-13          | 26              | 8.8E-13 26                                                           |
| 202-40           | -                                                                                                                                                   | 8.8E-13          | 26              | 8.8E-13          | 26              | 6.8E-13 26                                                           |
| FCP              | 7                                                                                                                                                   | 8.8E-13          | 26              | 8.8E-13          | 26              | 8.8E-13 26                                                           |
| PC 380           | 7                                                                                                                                                   | 8.9E-17          | 26              | 8.8E-13          | 26              | <b>8.6E-1</b> 3 25                                                   |

7-120

5

KKKSAN

# File: PLTCOL.WK1, 20-Aug-87 PAGE2

| PLI<br>Mediai    | ANT D<br>N ACC | PERATIONS<br>IDENT FREE | COLLOCATI<br>NENCY ( P | ION<br>Per Tear ( |                         | PLANT<br>Median A                        | OPERATIONS CO<br>COLDENT FREQUE | )LLOCATION<br>ENCY - FER YEAF - |
|------------------|----------------|-------------------------|------------------------|-------------------|-------------------------|------------------------------------------|---------------------------------|---------------------------------|
| SCENARIO<br>1.D. | NO.            | ANAD ROC<br>Fieq        | RANGE<br>Factor        | TEAD RUC<br>Fred  | R <b>HNGE</b><br>Factor | TEAD NDC<br>Freq                         | FANGE<br>Factor                 |                                 |
| P00v0            |                | N/4                     |                        | 9.9E-13           |                         | <br>8.8F-13                              | 26                              |                                 |
| FCFGC            | 7              | 4.0E-12                 | 26                     | 4.0E-12           | 20                      | 4.0F-17                                  | 26                              |                                 |
| P09V2            | 7              | 4.0E-12                 | 26                     | 4.0E-12           | 26                      | 4.0E-12                                  | 25                              |                                 |
| P05.F            | 7              | N/4                     |                        | 6.7E-12           | 26                      | 6.7E-12                                  | 20                              |                                 |
| 1074 - Me        | eteor          | ite strike              | s the TGG              | <.                |                         |                                          |                                 |                                 |
| FCABE            | 7A             | 3.4E-13                 | 26                     | 3.4E-13           | 26                      | 3.4E-13                                  | 26                              |                                 |
| <b>FORHE</b>     | -4             | 3.4E-13                 | 26                     | 3.4E-13           | 26                      | 3.4E-13                                  | 26                              |                                 |
| 504 F            | 7ê             | 0.4E-13                 | 20                     | 3.4E-13           | 26                      | <b>J.4E-1</b> 3                          | 26                              |                                 |
| 903 - Mes        | ceors.         | te strikes              | the agen               | nt piping sy      | sten betw               | n the BOS and TO4 at TEAD (bulk-only fac | ality).                         |                                 |
| PENGF            | Ξ              | N-H                     |                        | 3.0E-11           | 17                      | 3.0E-11                                  | 17                              |                                 |
| ₽. an₽           | 2              | <b>N</b> /4             |                        | 5.0E-11           | 17                      | 3.0E-11                                  | 17                              |                                 |
| FOAVE            | 8              | N/A                     |                        | 3.0E-11           | 17                      | 3.0E-11                                  | 17                              |                                 |

## Notes:

i. Frequency unit = events-operating year
I. Sceneric B applies only to the TEAD bulk-only facility

File: PLTCOL.Wk1, 20-Aug-87 PAGE1

| PLI<br>Mediai    | ANT O<br>N ACE | PERATIONS<br>1DENT FREG | COLLOCATE<br>NUENCY ( P | ON<br>Er year (  |                 | PL<br>MEDIA                | ANT DPERATIONS COLLO<br>AN ACCIDENT FREQUENCI | CATION<br>( { PEP YEAF ) |
|------------------|----------------|-------------------------|-------------------------|------------------|-----------------|----------------------------|-----------------------------------------------|--------------------------|
| SCENAFIC<br>1.D. | NO.            | ANAD RDC<br>FREQ        | RANGE<br>Factor         | TEAD RDC<br>Freq | RANGE<br>Factor | TEAD NG<br>Freq            | DC RANJE<br>Factor                            |                          |
|                  |                |                         |                         |                  |                 | *****                      |                                               |                          |
| 20100            | C C C          | iarye airo<br>N/A       |                         | 1 75-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| 20545<br>20545   | 9              | 5 AF-10                 | 10                      | 1 25-11          | 10              | 1.75-1                     | 10                                            |                          |
| FLOOSE           | 3              | 2.02 10<br>2.AE-10      | tù                      | 1.7E-11          | 10              | 1.2E-1                     | 1 10                                          |                          |
| 20190            | Q              | 2.56-10                 | 10                      | 1.2E-11          | 10              | 1.2E-1                     | 1 10                                          |                          |
| 20,65            | 4              | N:A                     |                         | 1.25-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| 5 h = 5          | 2              | 2.65-16                 | 10                      | 1.7E-11          | 10              | 1.2E-1                     | 11 16                                         |                          |
| FOLVS            | ç              | 2.6E-10                 | 10                      | 1.2E-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| FORCE            | :              | 2.6E-10                 | 10                      | 1.2E-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| F 1960           | 9              | 1.6E-10                 | 10                      | 1.2E-11          | 10              | 1.2E-1                     | 1 10                                          |                          |
| £522]            | 3              | 2.6E-10                 | 10                      | 1.2E-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| FORVE            | Ģ              | 2.6E-10                 | 10                      | 1.2E-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| P0060            | :              | 2.oE-16                 | 10                      | 1.2E-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| P09.0            | ş              | N/A                     |                         | 1.2E-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| F 3F 3C          | Q              | 2.oE-10                 | 10                      | 1.2E-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| POFVC            | Ģ              | 2.65-10                 | 10                      | 1.2E-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| F06V3            | :              | £i i é                  |                         | 1.2E-11          | 10              | 1.2E-1                     | 11 10                                         |                          |
| 4019 <b>- 5</b>  | irect          | liarge air              | craft cra               | sh onto th       | e MHI; f:       | not contained in 0.5 hours |                                               |                          |
| FOBGF            | 11             | N/A                     |                         | 9.8E-12          | 10              | 9.8E-1                     | 12 10                                         |                          |
| FOSHC            | 10             | 2.2E-10                 | 46                      | 9.8E-12          | 10              | 9.8E-1                     | 12 10                                         |                          |
| F0060            | 12             | 2.2E-10                 | 10                      | 9.8E-12          | 10              | 9.8E-1                     | 12 10                                         |                          |
| P0080            | 19             | 1.2E-10                 | 1ů                      | 9.8E-12          | 10              | 9.8E-1                     | 12 19                                         |                          |
| ÷ G⊭ GF          | :              | N/A                     |                         | 9.8E-12          | 10              | 9.8E-1                     | 12 10                                         |                          |
| £ÇkH£            | 10             | 2.2E-10                 | 10                      | 9.8E-12          | 10              | 9.8E-1                     | 12 10                                         |                          |
|                  | 1              | 2.2E-16                 | 10                      | 9.8E-12          | 10              | 9.9E-1                     | 12 10                                         |                          |
|                  | К              | 2.2E-10                 | $10^{-10}$              | 9.8E-12          | 10              | <b>₹.8</b> E-1             | 12 10                                         |                          |
| PGP80            |                | 2.2E-10                 | 10                      | 9.8E-12          | 10              | 9.8E-1                     | 12 10                                         |                          |
| FOFHO            | 10             | 2 <b>.2E</b> -10        | 10                      | 9.6E-12          | 10              | 9.8E+1                     | 12 10                                         |                          |
| 20°\C            | ų.             | 2.28-10                 | 10                      | 9.8E-12          | 10              | 9.0E-1                     | 12 10                                         |                          |
| P9660            | 10             | 2.25-10                 | 10                      | 9.8E-12          | 10              | 9.8E-1                     | 12 10                                         |                          |

žina <u>strona pinekuj kunkoji kunkoji kunkoji konkontroli kosto kunkona pinekoji kontrol</u> kontroli kunku kun

ALCONT ON A

File: PLTCOL.WK1, 20-Aug-87 PAGE2

| PL#<br>Median    | ANT D | PERATIONS<br>IDENT FREG | COLLOCAT<br>RUENCY ( ) | ION<br>PER YEAR ) |                 | PLANT OPERATIONS COLLOCATION<br>Median accident frequency ( PER ( 76+P) |  |
|------------------|-------|-------------------------|------------------------|-------------------|-----------------|-------------------------------------------------------------------------|--|
| BCENARIO<br>I.D. | ٩0.   | ANAD RDC<br>Freq        | RANGE<br>Factor        | TEAD RDC<br>Freq  | RANGE<br>FACTOR | TEAD NDC RANGE<br>FREQ FACTOR                                           |  |
|                  |       |                         |                        |                   |                 |                                                                         |  |
| 24504            | 10    | N/A                     |                        | 9.8E-12           | 10              | 9.88-12 10                                                              |  |
| FOREC            | 10    | 2.2E-10                 | 10                     | 9.8E-12           | 10              | 9.8E-12 10                                                              |  |
| PGPVC            | :6    | 2.2E-10                 | 10                     | 9.8E-12           | 10              | 9.8E-12 10                                                              |  |
|                  |       |                         |                        |                   |                 |                                                                         |  |

|           |       |             |            |          |           |                     |              | • • |
|-----------|-------|-------------|------------|----------|-----------|---------------------|--------------|-----|
| PGPVC     | 10    | 2.2E-10     | 10         | 9.8E-12  | 10        |                     | 9.8E-12      | 10  |
| F05.F     | 16    | N/A         |            | 9.8E-12  | 10        |                     | 9.8E-12      | 10  |
| FC11 - 5: | irect | large airci | raft crash | onto the | e MH1; fi | re contained in 0.5 | hours        |     |
| POBGE     | 11    | N/4         |            | 3.3E-15  | 13        |                     | 3.3E-15      | 13  |
| PQ SF     | 11    | N/A         |            | 3.32-15  | 13        |                     | 3.3E-15      | 13  |
| FORME     | 11    | 7.3E-14     | 13         | 3.3E-15  | 13        |                     | 3.3E-15      | 13  |
| POKVF     | 11    | 7.32-14     | 13         | 3.3E-15  | 13        |                     | 3.3E-15      | 13  |
| POSVE     | 11    | NZA         |            | 3.3E-15  | 13        |                     | 3.3E-15      | 13  |
| PC12 - 5: | :rect | large airci | raft crash | damages  | the MDB;  | no fire             |              |     |
| POB65     | 12    | N/A         |            | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| PODHC     | 12    | 7.7E-09     | 10         | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| PGC6C     | 12    | 7.78-09     | 10         | 3.5E-10  | 10        |                     | 3.5E-10      | 19  |
| POCHC     | 12    | 7.7E-09     | 10         | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| PGK.GS    | 12    | K;A         |            | 3.5E-10  | 16        |                     | 3.5E-10      | 1¢  |
| FCHHS     | 12    | 7.7E-09     | 10         | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| POKIS     | 12    | 7.7E-07     | 10         | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| P0r.v(    | 12    | 7.7E-09     | 10         | 3.58-10  | 10        |                     | 3.5E-10      | 10  |
| P0P30     | 12    | 7.7E-07     | 10         | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| FOFH2     | :2    | 7.72-09     | 10         | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| POPVE     | 12    | 7.72-09     | 10         | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| P0000     | .2    | 7.7E-09     | 10         | 3.5E-10  | 16        |                     | 3.5E-10      | 10  |
| Fûê+C     | 12    | N/m         |            | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| PORAS     | 11    | 7.7E-09     | 10         | 3.5E-10  | 10        |                     | 3.5E-10      | 16  |
| FORVE     | 12    | 7.7E-)9     | 10         | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| P0505     | 12    | 3/A         |            | 3.5E-10  | 10        |                     | 3.5E-10      | 10  |
| F013 - 0: | irect | large airc  | raft crash | damages  | the MDB;  | fire not contained  | in 0.5 hours |     |
| P0867     | 17    | N/4         |            | 2.9E-10  | 10        |                     | 2.9E-10      | 10  |
| FODHC     | 13    | 6.JE-09     | 10         | 2.9E-10  | 10        |                     | 2.9E-10      | 10  |





File: FLTCDL. KK1, 20-Aug-87 PAGE3

F

C

| FLA<br>MEDIAN         | int o<br>I acc   | PERATIONS<br>IDENT FRED | COLLOCATI<br>UENCY ( P | ON<br>Er year (  |                 | PLAN<br>Median           | PLANT OPERATIONS COLLOCATION<br>MEDIAN ACCIDENT FREQUENCY ( PER ) |   |  |  |  |
|-----------------------|------------------|-------------------------|------------------------|------------------|-----------------|--------------------------|-------------------------------------------------------------------|---|--|--|--|
| SCENAPIG<br>I.D.      | ĸJ.              | AHAD RDC<br>Freq        | RANGE<br>FACTOR        | TEAD RDC<br>Fred | RANGE<br>Factor | TEAD NOC<br>Fred         | RANGE<br>Factor                                                   |   |  |  |  |
|                       | <br>, •          | . 75-09                 |                        | 7 95-16          |                 |                          | 10                                                                | - |  |  |  |
| 50000                 | 12               | 4 35-39                 | 10                     | 2.9E-10          | 10              | 2.9E-10                  | 10                                                                |   |  |  |  |
| 51101<br>55155        |                  | 6. JE=V7<br>E/A         |                        | 2 95-10          | 10              | 2.9E-10                  | 10                                                                |   |  |  |  |
| ruxur<br>Dovue        | د <u>د</u><br>جه | N/R<br>10-25-10         | 10                     | 2.95-10          | 10              | 2.95-10                  | 10                                                                |   |  |  |  |
| 501.05                | 17               | 0.JE-07                 | 10                     | 2.9E-10          | 10              | 2.9E-16                  | 10                                                                |   |  |  |  |
| 50697<br>53895        |                  | 5.30-04                 | 10                     | 2.96-10          | 10              | 2.9E-10                  | 10                                                                |   |  |  |  |
| 50052                 |                  | 5.35-09                 | 10                     | 2.9E-10          | 10              | 2.9E-10                  | 10                                                                |   |  |  |  |
| 00510                 |                  | - 3E-09                 | 10                     | 2.9F-10          | 10              | 2.9E-10                  | 10                                                                |   |  |  |  |
| E OPOLI<br>E OPOLI    | 11               | 6.3E V7                 | 16                     | 2.9E-10          | 10              | 2.9E-10                  | 10                                                                |   |  |  |  |
| ECCERCIC<br>ECCCERCIC | 17               | A 35-05                 | 10                     | 2.9E-10          | 10              | 2.9E-10                  | 10                                                                |   |  |  |  |
| P0000                 |                  | 610E V7                 |                        | 7.9E-10          | 19              | 2.9E-10                  | 10                                                                |   |  |  |  |
| 20220                 | 17               | a. 32-69                | 10                     | 2.5E-10          | 10              | 2.9E-10                  | 10                                                                |   |  |  |  |
| 20202                 | 17               | 6.72-09                 | 10                     | 2.9E-10          | 10              | 2.9E-10                  | 10                                                                |   |  |  |  |
| COCUE                 | 11               | N/A                     |                        | 1.9E-10          | 10              | 2.98-10                  | ) 10                                                              |   |  |  |  |
| PC:1 - T              | 178-1            | large al                | rcrait cr              | ash damages      | the MD9:        | e contained in 0.5 hours |                                                                   |   |  |  |  |
| DORGE                 | :4               |                         |                        | ?.7E-14          | 13              | 9.78-14                  | 13                                                                |   |  |  |  |
| 101.85                |                  | N/A                     |                        | 5.7E-14          | 13              | 9.7E-14                  | 13                                                                |   |  |  |  |
| PCLAR                 | :4               | 2.18-12                 | 10                     | 9.7E-14          | 13              | 9.7E-14                  | i 13                                                              |   |  |  |  |
| POYVE                 | 14               | 2.18-12                 | 13                     | 9.7E-14          | 13              | 9.7E-14                  | i 13                                                              |   |  |  |  |
| PESVE                 | 14               | N/A                     |                        | 9.7E-14          | 13              | 9,7E-14                  | 17 17                                                             |   |  |  |  |
| FC15 - 1              | กล่าก            | ect large               | aircraft               | crash damag      | es the MH       | o fire                   |                                                                   |   |  |  |  |
| PCBBS                 | 15               | 11/A                    |                        | 2.9E-12          | 13              | 2.9E-12                  | 2 13                                                              |   |  |  |  |
| PODEC                 | :5               | 6.3E-11                 | 13                     | 2.9E-12          | 13              | 2.9E-12                  | 2 13                                                              |   |  |  |  |
| 23004                 | 15               | 6.3E-11                 | 13                     | 2.9E-12          | 13              | 2.9E-12                  | 2 13                                                              |   |  |  |  |
| FOCHE                 | 15               | 5.3E-11                 | 13                     | 2.9E-12          | 13              | 2.9E-12                  | 2 13                                                              |   |  |  |  |
| FDEBB                 | 15               | N: A                    |                        | 2.9E-12          | 13              | 2.98-12                  | 2 13                                                              |   |  |  |  |
| POR13                 | 15               | 6.3E-11                 | 17                     | 2.9E-12          | 13              | 2.9E-12                  | 2 13                                                              |   |  |  |  |
| PORVS                 | 15               | 6.3E-11                 | 13                     | 2.9E-12          | 13              | 2.9E-12                  | 2 13                                                              |   |  |  |  |
| PORVO                 | 15               | ь.3E-11                 | 13                     | 2.9E-12          | 13              | 2.9E-12                  | 13                                                                |   |  |  |  |
| FORAC                 | 15               | 6.37-11                 | 13                     | 2.9E-12          | 13              | 2.9E-12                  | 2 13                                                              |   |  |  |  |

1202003

|           |                                                               |                           | TABLE 7-14 (Continue                         | d)                 |                                                       |
|-----------|---------------------------------------------------------------|---------------------------|----------------------------------------------|--------------------|-------------------------------------------------------|
|           |                                                               |                           |                                              | -,                 |                                                       |
|           | File: PLTCCL.Wk1, 20-Aug-87 PAG                               | GE4                       |                                              |                    |                                                       |
|           | PLANT OPERATIONS COLLOCATIO<br>MEDIAN ACCIDENT FREQUENCY ( PR | DN<br>ER YEAR→            |                                              | PLANT<br>Median Ad | OPERATIONS COLLOCATION<br>CIDENT FRECVENCY ( PER YEAR |
|           | SCENAFIO NO. ANNO RDC RANGE<br>I.D. FRED FACTOR               | TEAD RDC<br>Freq Fi       | RANGE<br>Ictor                               | TEAD NDC<br>Freq   | FANGE<br>Factor                                       |
|           | PDFmC 15 0.3E-11 13<br>PCFVC 15 0.3E-11 13                    | 2.9E-12<br>2.9E-12        |                                              | 2.7E-12<br>2.9E-12 | :3<br>13                                              |
|           | FGEBC 15 5.3E-11 13<br>PCOVC 15 N/A                           | 2.9E-12<br>2.9E-12        | 13<br>13                                     | 2.9E-12<br>2.9E-12 | 13<br>13                                              |
|           | PORGE 15 0.3E-11 13<br>PORVI 15 6.3E-11 13                    | 2.9E-12<br>2.9E-12        | 13<br>13                                     | 2.9E-12<br>2.9E-12 | 13<br>13                                              |
|           | PCSV5 15 N/A<br>PC16 - Indirect large aircraft cr             | 2.9E-12<br>ash damages t  | 13<br>he MHI; fire not contained in 0.5 hour | 2.9E-12<br>s       | 13                                                    |
|           | POBOF 1a N/A<br>PODHC 16 5.2E-11 13                           | 2.JE-12<br>1.4E-12        | 13<br>13                                     | 2.3E-12<br>2.4E-12 | 13<br>17                                              |
|           | FCCSC 16 5.2E-11 13<br>FCCHC 16 5.2E-11 13                    | 2.4E-12<br>2.4E-12        | 13<br>13                                     | 2.4E-12<br>2.4E-12 | 13                                                    |
|           | POFUE IC N/A<br>PC-4F 16 5.15-11 13                           | 2.3E-12<br>2.3E-12        | 13<br>13                                     | 2.3E-12            | 13                                                    |
| Fer.<br>D | FD:15 15 5.1E-11 13<br>F0:54 15 5.1E-11 13                    | 2.3E-12<br>2.4E-12        | 13<br>13                                     | 2.3E-12            | 15                                                    |
|           | FORAC 16 5.2E-11 17<br>FORAC 16 5.2E-11 17                    | 2.4E-12<br>2.4E-12        | 13                                           | 2.48-12            |                                                       |
|           | FARIC 10 5.22-11 13<br>FARIC 10 5.22-11 13                    | 2.48-12                   | 13                                           | 2.4E-12<br>2.4E-12 | 13                                                    |
|           | PDD:0 10 0.2011 10<br>PDD:0 10 N/A                            | 2.4E-12<br>2.4E-12        | 15                                           | 2.4E-12<br>2.4E-12 | 15<br>13                                              |
|           | PSF/C is 5.2E-11 13                                           | 2.4E-12<br>2.4E-12        | 13                                           | 1.4E-12<br>2.4E-12 | 13<br>13                                              |
|           | FUDYr 16 N/H<br>Flil - Indirect large eircraft cra            | 2.3E-12<br>ash damages ti | 13<br>ne MHI; fire contained in 0.5 hours    | 2.3E-12            | :3                                                    |
|           | - Uper 17 N/4<br>FD(3F 17 N/4<br>FF 3F 17 N/4                 | 8.0E-16<br>8.0E-16        | 16                                           | 6.0E-16<br>E.OE-1a | io<br>16                                              |
|           | FLAME 17 1.8E-14 10<br>FLAVE 17 1.8E-14 10                    | 8.0E-16<br>8.0E-16        | 16<br>16                                     | 8.0E-16<br>8.0E-16 | le<br>lo                                              |
|           | Hubah 17 N/A                                                  | 8.0E-16                   | 16                                           | 8.0E-16            | 10                                                    |
|           |                                                               |                           |                                              |                    |                                                       |
|           |                                                               |                           |                                              |                    |                                                       |
|           |                                                               |                           |                                              |                    |                                                       |
|           |                                                               |                           |                                              |                    |                                                       |
|           |                                                               |                           |                                              |                    |                                                       |
|           |                                                               |                           |                                              |                    |                                                       |
|           |                                                               |                           |                                              |                    |                                                       |
|           |                                                               |                           |                                              |                    |                                                       |
|           |                                                               |                           |                                              |                    |                                                       |
|           |                                                               |                           |                                              |                    |                                                       |



File: PLTCOL.Wk1, 20-Aug-87 FAGE5

Ģ

| PL:<br>Hegia:    | ANT D<br>N ACC | PERATIONS<br>IDENT FREG | COLLOCAT<br>PUENCY ( | ION<br>Per year ) |                 | PLAI<br>MEDIAN                | IT OPERATIONS<br>ACCIDENT FREQ | COLLOCATION<br>JENCY (PER YEAR | ۲. |
|------------------|----------------|-------------------------|----------------------|-------------------|-----------------|-------------------------------|--------------------------------|--------------------------------|----|
| SCENARIO<br>I.D. | NŪ.            | ANAD RDC<br>Fred        | RANGE<br>Factor      | TEAD RDC<br>Freq  | RANGE<br>Factor | TEAD NDC<br>Freq              | RANGE<br>Factor                |                                |    |
| P018 - 10        | ndire          | ct large a              | urcraft              | crash damaqe      | s the MDB:      | ) fire                        |                                |                                |    |
| POBGS            | 19             | N/A                     |                      | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| PODHC            | 19             | S.8E-07                 | 11                   | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| F0060            | 18             | B.8E-09                 | 11                   | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| POCHC            | 18             | 8.8E-09                 | 11                   | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| P01 63           | 18             | N/A                     |                      | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| POKAS            | 18             | 6.9E-09                 | 11                   | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| POKVS            | 18             | 8.8E-09                 | 11                   | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| POMVE            | 18             | 6.8E-09                 | 11                   | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| PJP60            | 18             | 6.9E-09                 | 11                   | 4.02-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| FOFHC            | ta             | 8.8E-09                 | 11                   | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| FORVE            | 18             | <b>3.8⊾</b> -09         | 11                   | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| 20260            | 13             | 3.6E-09                 | 11                   | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| POQVC            | 18             | N/A                     |                      | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| POPSC            | 18             | 8.8E-(%                 | 11                   | 4.0E-10           | 11              | 4.0E-10                       | 11                             |                                |    |
| FURVC            | 18             | 8.8E-09                 | 11                   | 4.0E-10           | 11              | 4.0E-16                       | 11                             |                                |    |
| POSVS            | 18             | N/A                     |                      | 4.0E-10           | 11              | 4. )E-10                      | 11                             |                                |    |
| P019 - 1         | ndire          | ct large a              | aircraft             | crash damage      | s the MDB       | re not contained in 0.5 hours |                                |                                |    |
| f GBƏF           | 19             | N/A                     |                      | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |
| FOD-C            | 19             | 7.2E-09                 | 11                   | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |
| FOOST            | 15             | 7.2E-09                 | 11                   | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |
| FOCHU            | ١ć             | 7.2E-09                 | 11                   | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |
| POKSF            | 13             | N/A                     |                      | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |
| FOKHF            | 19             | 7.1E-03                 | 11                   | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |
| PC VF            | 19             | 7.1E-09                 | 11                   | 3.3E-10           | 11              | <b>J. 3E-1</b> 0              | 11                             |                                |    |
| P0470            | 19             | 7.2E+09                 | 11                   | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |
| P0660            | 19             | 7.2E-09                 | 11                   | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |
| FORHC            | 19             | 7.2E-09                 | 11                   | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |
| ະງະ              | 1.2            | 7.2E-09                 | 11                   | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |
| - 6666           | 19             | 7.2E-09                 | 11                   | 3.3E-10           | 11              | 3.3E-10                       | 11                             |                                |    |

TALEND TO A THE TANK NO SAME TANK THE SAME TANK THE SAME TANK THE SAME TANK THE TANK THE TANK THE TANK THE TANK

. Y 📢



| AD-A1  | 93 355 | CHE<br>THE<br>CA | MICAL<br>DISPO<br>A N I | STOCK<br>DSAL O<br>DARSEL | PILE D<br>F CHEM<br>L ET A<br>7006 D | ISPOSI<br>(U)<br>L. RU(<br>AAA15- | AL PRO<br>GA TE<br>3 87 GI<br>85-D-1 | GRAM R<br>Chnolo<br>R-C-19<br>Ra22 | ISK AN<br>GIES I<br>563 | ALYSI<br>NC SA | S OF<br>N DIEG | 0 <b>6</b> / | 213 |
|--------|--------|------------------|-------------------------|---------------------------|--------------------------------------|-----------------------------------|--------------------------------------|------------------------------------|-------------------------|----------------|----------------|--------------|-----|
| UNCEN. |        |                  |                         |                           |                                      |                                   |                                      |                                    |                         |                |                |              |     |
|        |        |                  |                         |                           |                                      |                                   |                                      |                                    |                         |                |                |              |     |
|        |        |                  |                         |                           |                                      |                                   |                                      |                                    |                         |                |                |              |     |
|        |        |                  |                         |                           |                                      |                                   |                                      |                                    |                         |                |                |              |     |
|        |        |                  |                         |                           |                                      |                                   |                                      |                                    |                         |                |                |              |     |
|        |        |                  |                         |                           |                                      |                                   |                                      |                                    |                         |                |                |              |     |
|        |        |                  |                         |                           |                                      |                                   |                                      |                                    |                         |                |                |              |     |
|        |        |                  |                         |                           |                                      |                                   |                                      |                                    |                         |                |                |              |     |



122222222 miles/22223

File: PLTCDL.WK1, 20-Aug-87 FAGE6

PLANT OPERATIONS COLLOCATION MEDIAN ACCIDENT FREQUENCY ( PEP YEAR ) PLANT OPERATIONS COLLOCATION MEDIAN ACCIDENT FREQUENCY ( PER YEAR ) 155.23

56333766

227727

2000

22.22

| SCENARI<br>I.D | 0 NO.  | ANAD RDC<br>Freq | RANGE<br>Factor | TEAD RDC<br>FRED | RANGE<br>Factor | TEAD NDC<br>Freq                           | RANGE<br>Factor |
|----------------|--------|------------------|-----------------|------------------|-----------------|--------------------------------------------|-----------------|
|                |        |                  |                 |                  |                 |                                            |                 |
| P00V1          | ::     | N/4              | •-              | J.JE-10          | 11              | 3.3E-10                                    | 1:              |
| FORGE          | 19     | 7.2E-09          | 11              | 5.3E-10          | 11              | 3.3E-10                                    | 11              |
| F05V5          | 10     | 7.2E-09          | 11              | 3.3E-10          | 11              | 3.3E-10                                    | 11              |
| FOSVF          | : :    | N/A              |                 | 3.3E-10          | 11              | 3.3E-10                                    | 11              |
| P020 -         | Indire | ct large a       | urcraft         | crash damage     | s the MDB;      | fire contained in 0.5 hours                |                 |
| POBGE          | 20     | N/A              |                 | 1.1E-13          | 14              | 1.1E-13                                    | 14              |
| FOISE          | 10     | N/A              |                 | 1.18-17          | 14              | 1.1E-13                                    | 14              |
| POK-F          | 26     | 2.42-12          | 14              | 1.1E-13          | 14              | 1.1E-13                                    | 14              |
| POKVE          | 20     | 2.48-12          | 14              | 1.1E-13          | 14              | 1.1E-13                                    | 14              |
| ¢03,¢          | X      | N. A             |                 | 1.1E-13          | 14              | 1.1E-13                                    | 14              |
| P02: -         | Large  | or small a       | urcraft         | crash damage     | s the outd      | oor acent piping system at TEAD; no fire   |                 |
| POAGE          | 21     | N . A            |                 | 1.0E-08          | 10              | 1.0E-08                                    | 10              |
| POARS          | 21     | N - 4            |                 | 1.0E-08          | 10              | 1.0E-08                                    | 10              |
| PCAVE          | ::     | 14.1 A           |                 | 1.02-03          | 10              | 1.0E-08                                    | 10              |
| FC21 - :       | Large  | or small a       | secraft         | crash damage     | s the outd      | oor agent piping system at TEAD: fire occu | Irs             |
| P0469          | 22     | N/A              |                 | 8.2E-09          | 10              | 8.2E-09                                    | 16              |
| PCAHS          | 22     | 87A              |                 | E.2E-09          | 10              | 8.2E-09                                    | 10              |
| POANS          | 22     | N/A              |                 | 8.2E-09          | 10              | 8.2E-09                                    | 10              |

## Notes:

1. Frequency unit = events/operating year

1. Scenarios 21 and 22 apply only to the TEAD bulk-only facility



File: PLTCOL.WEI, 20-Aug-67 PAGE1

| ECENARI           | D NG.    | ANAD RDC        | RANGE     | TEAD RDC           | RANGE       | TEAD NDC                            | RANGE      |                           |                |
|-------------------|----------|-----------------|-----------|--------------------|-------------|-------------------------------------|------------|---------------------------|----------------|
| I.C               |          | FRED            | FACTOR    | FREQ               | FACTOR      | FREQ                                | FACTOR     |                           |                |
| ÷025 -            | Earth    | juake damag     | es the N  | /8 structure       | , munitions | 1 & puncture; fire suppressed       |            |                           |                |
| POSEC             | 13       | N/A             |           | 1.9E-07            | 7           | 1.92-07                             | 7          |                           |                |
| PCDHC             | - 25     | NESL            |           | NEGL               |             | NE 5L                               |            |                           |                |
| F3661<br>57787    | 40<br>25 | NEGL            |           | NEGL               |             | NEGL NEGI                           |            |                           |                |
| Potro             | 23       | NEGL<br>N/A     |           | 1. AF-OA           | 7           | 1.6E-96                             | 7          |                           |                |
| PORHC             | 25       | 7.12-08         | 7         | 1.65-06            | 7           | 1.65-06                             | 7          |                           |                |
| FORNO             | 25       | 7.1E-08         | 7         | 1.6E-06            | 7           | 1.65-06                             | 7          |                           |                |
| PERVI             | 25       | 2.3E-09         | 6         | 5.0E-03            | 7           | 5.0E-08                             | 7          |                           |                |
| POPSC             | 25       | HEEL            |           | NEGL               |             | NEGL                                |            |                           |                |
| POPHC             | - 25     | NEGL            |           | NEGL               |             | NEGL                                |            |                           |                |
| PSPV2             | 25       | NESL            |           | NEG.               |             | NESL NCCI                           |            |                           |                |
| PUEBE             | 20<br>ne | NE JL           |           | NEGL               |             | REDL MC 21                          |            |                           | -4 =           |
| F SUVU<br>F DE 10 | 22<br>75 | 67.8<br>1.52-09 |           | HECL<br>3.35-07    | 7           | NEJL<br>7.35-07                     | 7          |                           | £ <sup>Q</sup> |
| 2 3 B 4 2         | 20<br>23 | 1.5E-V8         | 6         | 3.32-07<br>3.35-07 | ;           | 3.38-07                             | 7          |                           |                |
| -38.2             | 25       | 3.9E-07         | -         | 3.4E-C4            | , 7         | 8.46-06                             | 7          |                           | 4 L            |
| - 1025 -          | Earth    | 1.3-8 38#20     | jes tha " | E structure        | . aunitions | 11 & puncture: earthquake initiates | fire; fire | suppression system fails. |                |
| PC160             | 2:       | N'A             |           | 6.1E-09            | 13          | 6.1E-09                             | 13         |                           |                |
| -3E-C             | _¢       | NEGL            |           | HES:               |             | NEGL                                |            |                           |                |
| FICEC             | Is.      | NEG_            | ••        | NEGL               |             | NEGL                                |            |                           |                |
| P35+3             |          | 4E51            |           | NEGL               |             | NEGL A DE OF                        |            |                           |                |
| 1946.<br>Source   | 16       | N 75-00         |           | 4.95-08            | 12          | 4.7E-00<br>A CC_00                  | 15         |                           |                |
| POK: 1            | 40<br>24 | 1.05-19         | • •       | 4.76-00            | 13          | 4.75-08                             | 13         |                           |                |
| FUR 0             | 24<br>24 | 1101-77<br>FA   |           | 4.72-05            |             | 4:72-08<br>NEGL                     |            |                           |                |
| FOPSI             | 25       | NEGL            |           | NESL               |             | NEGL                                |            |                           |                |
| FORM              | 10       | HE SL           |           | NEGL               |             | NEGL                                |            |                           |                |
| ، من <i>ج</i>     | 2:       | NEGL            |           | NEGL               |             | NEGL                                |            |                           |                |
| P1030             | 20       | NEGL            |           | NEGL               |             | HE6L                                |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           | ~              |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           | - Q.           |
|                   |          |                 |           |                    |             |                                     |            |                           | $\mathcal{N}$  |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             | -128                                |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |
|                   |          |                 |           |                    |             |                                     |            |                           |                |



File: PLTCOL.WKI, 20-Aug-87 PAGE2

| PLANT OPERATIONS COLLOCATION<br>MEDIAN ACCIDENT FREQUENCY ( PER YEAR ) |          |                  |                 |                  |                 | PLANT OPERATIONS COLLOCATION<br>MEDIAN ACCIDENT FREQUENCY ( PER YEAR )          |  |  |  |
|------------------------------------------------------------------------|----------|------------------|-----------------|------------------|-----------------|---------------------------------------------------------------------------------|--|--|--|
| SCENARIC<br>I.D.                                                       | NC.      | ANAD ROC<br>Freq | RANGE<br>Factor | TEAD RDC<br>Freq | RANGE<br>Factor | TEAD NDC RANGE<br>FREQ FACTOR                                                   |  |  |  |
| FORVC                                                                  | 25       | N/A              |                 | NEGL             | *-              | NFE:                                                                            |  |  |  |
| PORGO                                                                  | 26       | 4.0E-10          | 11              | 1.0E-08          | 14              | 1.05-06 14                                                                      |  |  |  |
| PORVO                                                                  | 26       | 4.0E-10          | 11              | 1.02-08          | 14              | 1.05-08 14                                                                      |  |  |  |
| FOSVE                                                                  | 26       | N/A              |                 | 2.7E-07          | 13              | 2.75-07 13                                                                      |  |  |  |
| 9029 - Ea                                                              | irthg    | uake damag       | es the Mi       | 5: eucition      | s are intac     | t: fire occurs: fire suppression system fails.                                  |  |  |  |
| 26860                                                                  | 29       | N/A              |                 | 2.2E-05          | 10              | 2.2E-05 10                                                                      |  |  |  |
| FOEHE                                                                  | 29       | 7.9E-07          | 9               | 2.2E-05          | 10              | 2.25~05 10                                                                      |  |  |  |
| P0060                                                                  | 29       | 7.85-97          | 9               | 2.22-05          | 10              | 2.25-05 10                                                                      |  |  |  |
| PECHC                                                                  | 25       | 7.8E-07          | 9               | 2.28-05          | 10              | 2.22-05 10                                                                      |  |  |  |
| PC+SC                                                                  | 29       | N/A              |                 | 2.28-05          | 10              | 2.22-05 10                                                                      |  |  |  |
| POKHC                                                                  | 29       | 7.82-07          | ų               | 2.2E-05          | 10              | 2.2E-05 10                                                                      |  |  |  |
| Fêr vê                                                                 | 29       | .3E-07           | 9               | 2.2E-05          | 10              | 2.22-05 10                                                                      |  |  |  |
| FORMS                                                                  | 25       | 7.9E-07          | 9               | 2.28-05          | 10              | 2.2E-05 10                                                                      |  |  |  |
| POPSO                                                                  | 20       | 7.8E-07          | 9               | 2.25-05          | 10              | 2.2E-05 10                                                                      |  |  |  |
|                                                                        | 25       | 7.8E-07          | 9               | 2.28-05          | 10              | 2.2E-05 10                                                                      |  |  |  |
| - (F-1 <b>C</b>                                                        | 19       | 7.65-07          | 9               | 2E-05            | 10              | 2.25-05 10                                                                      |  |  |  |
| P0260                                                                  | 29       | <b>.8E-</b> 07   | 9               | 2.2E-05          | 10              | 2.25-05 10                                                                      |  |  |  |
| P36VC                                                                  | 23       | N/A              |                 | 2.2E-05          | 10              | 2.2E-05 10                                                                      |  |  |  |
| 28405                                                                  | 29       | .5E-∋⊺           | 9               | 2.2E-05          | 10              | 2.2E-05 10                                                                      |  |  |  |
| PORVO                                                                  | <u>(</u> | 7.8E-07          | c               | 2.25-05          | :0              | 2.2E~05 10                                                                      |  |  |  |
| PCSVD                                                                  | 27       | N/A              |                 | 2.2E-05          | 10              | 2.2E~05 10                                                                      |  |  |  |
| P223 - Ea                                                              | ertha.   | iske causel      | s munitia       | ns to fall !     | but no deto     | nation occurs; the MDB is intact; the TOX is intact; earthquake initiates fire: |  |  |  |
| fi                                                                     | re 51    | ( <b>b</b> b     |                 |                  |                 | · · · · · · · · · · · · · · · · · · ·                                           |  |  |  |
| PLASC                                                                  | 33       | 17.2             |                 | N/A              |                 | N/A                                                                             |  |  |  |
| P02-0                                                                  | 33       | 1.7E-0e          | 2ů              | 4.8E-05          | 20              | A/R                                                                             |  |  |  |
| F0030                                                                  |          | 1.7E-06          | 20              | 4.8E-05          | 25              | 4 BF-05 70                                                                      |  |  |  |
| FGCHC                                                                  | 25       | 1.75-00          | 20              | 4.88-05          | 20              | 4.62.03 20                                                                      |  |  |  |
| POKEC                                                                  | 33       | '/A              |                 | N/A              |                 | 100 JO 20                                                                       |  |  |  |
| PGKHC                                                                  | 33       | ÷₊/∆             |                 | N/A              |                 | N/A                                                                             |  |  |  |
| POr VC                                                                 | 32       | N/#              |                 | N/A              | ••              | N/A                                                                             |  |  |  |
| FORVI                                                                  | 37       | 1.7E-06          | 2)              | 4.8E-05          | 20              | 4.8E-05 20                                                                      |  |  |  |

PERSONAL REPORT DESCRIPTION

6 MAR

She Marak

Stor & te. 6" 4. 6"4





File: P\_TCOL.WK1, 20-Aug-87 PAGE3

| FLANT DEFRATIONS COLLBEATION    | PLANT  |          |
|---------------------------------|--------|----------|
| MEDIAN ACCIDENT FREQUENCY ( PER | YEAR ) | MEDIAN ( |

| SCENARIO | 40. | ANAD ROC | RANSE  | TEAD RDC | RANGE  |  |
|----------|-----|----------|--------|----------|--------|--|
| 1.3.     |     | FKEN     | THUIGH | FR28     | CHUIUN |  |
|          |     |          |        |          | 50     |  |
| 5063C    | 33  | 1.78-96  | 20     | 4.35-03  | 20     |  |
| POPHC    | 23  | 1.7E-va  | 20     | 4.8E-05  | 20     |  |
| FORVO    | 33  | 1.75-06  | 20     | 4.8E-05  | 20     |  |
| F6460    | 13  | 1.72-04  | 20     | 4.82-05  | 20     |  |
| FCQVC    | 33  | N/A      |        | 4.85-05  | 20     |  |
| FORGE    | 37  | 1.7E-06  | 20     | 4.8E-05  | 20     |  |
| ₽G₽vj    | 33  | 1.7E-06  | 20     | 4.3E-05  | 20     |  |
| PC340    | 33  | N/A      |        | N/A      |        |  |

Notes:

Kx53233.

103300

.. Frequency unit # events/operating year

PLANT OPERATIONS COLLOCATION MEDIAN ACCIDENT FREQUENCY ( PER YEAR )

| TEAD NDC<br>Freq                                                                            | RANGE<br>Factor                        |  |
|---------------------------------------------------------------------------------------------|----------------------------------------|--|
| 4.8E-05<br>4.8E-05<br>4.8E-05<br>4.8E-05<br>4.8E-05<br>4.8E-05<br>4.8E-05<br>4.8E-05<br>N/A | 20<br>20<br>20<br>20<br>20<br>20<br>20 |  |

5.5

- P016 Indirect large aircraft crash onto the MHI; fire not contained in 0.5 h.
- P017 Indirect large aircraft crash onto the MHI; fire contained in 0.5 h.
- P020 Indirect large aircraft crash onto the MDB; fire contained in 0.5 h.

There is very little distinction in the frequency of aircraft crashes with or without fire since the historical data indicate that there is only a 45% probability that an aircraft crash will involve a fire. The frequency of a crash onto the MDB is greater than the MHI because the surface area of the MDB is more than 100 times larger than the MHI.

For the regional collocation option, it is evident that large aircraft crashes occur more frequently at ANAD than TEAD. The frequency of an aircraft crash onto the outdoor agent piping system for the modified CAMDS facility is a dominant risk contributor. This scenario includes both large and small aircraft crashes and the frequency of small aircraft crashes (including helicopters) is at least two orders of magnitude higher than the frequency of large aircraft crashes at TEAD.

7.2.5.4. Earthquake-Induced Accident Frequencies. Reference 7-5 contains the frequency and failure probability data for each event modeled in the event trees that served as input data for the analysis of the accident scenario frequencies. The results of the frequency analysis are presented in Table 7-14. The earthquake accident frequencies for the scenarios analyzed are generally higher at TEAD since it is located in a more earthquake-prone region. Sequence P033, which postulates an earthquake-initiated munition fall and fire but with the MDB and TOX intact, has the highest frequency value (1.7 x  $10^{-6}$ /yr for ANAD and 4.8 x  $10^{-5}$ /yr for TEAD). This scenario involves the

detonation of all munitions (if burstered) in the UPA since fire is not suppressed. The agent release results are discussed in Section 10.

7.2.5.5. <u>Uncertainty Analysis</u>. The results of the uncertainty analysis indicate that the 95% percentile values may be 7 to 20 times higher than the reported median values. The uncertainties arise mainly from the general applicability of the raw data used to the perceived conditions of environment of a demilitarization program.



# 7.3. REFERENCES

- 7-1. Memorandum to C. A. Bolig, from R. Perry (OPMCM), May 6, 1987.
- 7-2. Swain, A. D., and H. E. Guttmann, "Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Application," Sandia National Laboratory, NUREG/CR-1278, August 1983.
- 7-3. Swain, A. D., "Accident Sequence Evaluation Program Human Reliability Procedure," Sandia National Laboratory, NUREG/CR-4772, February 1987.
- 7-4. Science Applications International Corporation, "Probabilities of Selected Hazards in Disposition of M55 Rockets," U.S. Army Toxic and Hazardous Materials Agency, M55-CS-2, November 1985.
- 7-5. "Supporting Calculations for the Risk Analysis of Chemical Munitions," GA Technologies, Inc., EPS-GA/DP-0092 (Issue 2), Volums 1, 2, and 3, August 1987.
- 7-6. Lawrence Livermore National Laboratory, "Explosives Handbook, Properties of Chemical Explosives and Explosive Simulants," March 16, 1981.
- 7-7. Solomon, K. A., K. C. Erdmann, and D. Okrent, "Estimate of the Hazards to a Nuclear Reactor from the Random Impact of Meteorites," <u>Nuclear Technology, Vol. 25</u>, January 1975.
- 7-8. Clarke, R. K., et al., "Severities of Transportation Accidents," SLA-74-0001, Sandia National Laboratories, July 1976.
- 7-9. Telephone conversation C. Everline (GA Technologies) and R. Perry (OPMCM), June 15, 1987.

7-133

#### 8. SCENARIO LOGIC MODELS FOR TRANSPORT

This section describes the development of accident scenarios for onsite (truck) transport at the originating and receiving sites, as well as offsite rail, air or marine transport from the storage site to the NDC or RDC disposal site. The work was performed by H&R Technical Associates, Inc. under subcontract to GA. The analysis covers only the transport in the hauling vehicle; risks associated with loading, unloading, and other handling activities are considered as part of the handling phase in Section 6. Risks while the munitions are in the holding area awaiting transport are treated in the storage phase analysis in Section 5.

Figure 8-1 shows a diagram of the transportation steps and options for the collocation disposal plan. As shown in the figure, there are three offsite carrier options: air, rail or marine transport. There are attendant differences in the sending/receiving site combinations. For example, marine shipment pertains only to transfer from APG to JACADS and air transfer is only for flights from the easternmost sites (APG and LBAD) to TEAD. Rail transport pertains to all combinations of nondisposal and disposal sites. There are additional attendant differences in the munitions package. Marine shipment pertains only to ton containers packed inside of vaults. For rail and air transport, the munitions are shipped in OFCs. For the last leg of truck transport from interim storage to the demil facility, the munitions are packed in ONCs.

Section 8.1 discusses logic models for accident scenarios during on-installation transport by truck to the offsite transporter (train, aircraft, or barge) or to the disposal plant. This phase is called "onsite transport," although it does not preclude the possibility that the railhead, air strip, or marine loading dock may be located outside

8-1



rung buruny buruny hurung burung b B

<u></u>ᡘᡧᢧᡶᢣᡁᢄᢤᡕᢄᡬ᠈ᢓ᠋ᡬᢞᠧᢄᢞᡄ᠔ᢞᡄ᠘ᡄᢧ᠐᠘ᡩᡄᡬᡶ᠋᠋᠕᠘᠅ᡩᢘᡬᡎᢓᡬᠲ᠌᠘ᡫᢖᢞᡬᠼᠣᡘᢢᢄᢜᡘᡆᡭᠱᢘᡭᡬᢁᡭᢤᠼᡭᠼᢘᡬᢢᡵᡬᢤᡵᡭᢤᢑᡭᢤᢑᡭᢤ

the sending or receiving site. Note that the onsite scenarios are developed in terms of risk per mile per site and they apply for all three offsite transport options. Differences lie in the number of operations involved in the offsite transfer options and because of site-specific data. Also, a different transport package is involved in the marine transport option, which affects the quantification.

Offsite transport is treated separately for the three modes of transport in Sections 8.2 to 8.4, since the respective accident scenarios differ fundamentally. Uncertainties for the transportation analyses in Sections 8.1 through 8.4 are discussed in Section 8.5.

Much of the basic data for rail and truck transport are given in Refs. 8-1 through 8-5. Details on the event tree branching probabilities are documented in the H&R Associates Inc. worksheets in Volume 3 of Ref. 8-6 for all onsite and offsite transportation. References 8-7 through 8-26 provide added sources of data.

The transportation accident analyses are consistent with the Army's plan for packaging and transport in Ref. 8-8. Administrative controls and procedures for transport are discussed in Section 3 and in Appendix G.

#### 8.1. ONSITE TRANSPORT

# 8.1.1. Chronology of Operations

Figures 6-1 and 6-2 show flow diagrams of the handling and transport operations associated with the collocation options, including rail, air, or marine modes of offsite transport. This section analyzes onsite transport by truck for the following steps:

- The munitions in OFCs (rail or air transport options) are taken by truck from their storage locations to the sending site holding area. For the marine transport case, the munitions are transported in vaults.
- 2. For the air transport mode, the OFCs are taken by truck from the holding area to an aircraft loading conveyor at the air strip. For rail transport, the holding area is at the railhead, so no truck transport is needed. For marine transport, the truck takes the vaults from the holding area to the barge loading dock.
- 3. At the destination, the munitions in their packages are placed in the receiving site holding area. For air transport, this is done by truck, from the air strip to the holding area. For rail, the holding area adjoins the railhead and no truck is needed.
- 4. For rail and air transport, the munitions, still in OFCs, are transported by truck from the receiving site holding area to the site's storage area (igloos, warehouse, or open area).
- 5. When ready for disposal, the munitions, now in ONCs, are trucked from storage to the MHI.

8-4

A generic set of accident sequences was developed (Section 8.1.3) which apply to each of these five transport steps. The set is designated as VR, VA, and VW for the rail, air, and marine offsite transport options, respectively, where they apply to the first four steps (first three steps for VW). In these steps, the munitions are always within the offsite packages (OFCs or vaults). For step 5, the munitions are trucked packaged within the ONC, so that quantification of the accident sequences is affected by the different package failure thresholds. For this step, the sequences are designated as VO, applicable for rail and air transport options only. Note that the accident sequence list and description is the same for VR, VA, VW, and VO. Only the resulting frequencies and/or agent releases differ.

# 8.1.2. Procedures and Assumptions

For this analysis it was assumed that all munitions for rail or air transport will be placed inside large OFC packages with outer dimensions of 20 x 8 x 8 ft (with failure thresholds as discussed in Section 3.3) prior to any offsite movement. For marine transport, the ton containers are in vaults which have similar failure thresholds as OFCs. It was also assumed that a 40-ft flatbed truck will be used as the transport vehicle. Each truck will carry one OFC or four vaults. Munition inventories for the OFC packages are shown in Table 10-3. A vault contains only one ton container. The munitions remain in these packages until arrival at the interim storage area of the disposal site. During the last transport leg, truck transport from interim storage to the MHI, the munitions are packed in ONCs, with four ONCs per truck. Each ONC contains one munition pallet (Table 10-3).

A standard distance of one mile was used for the distance between the storage area and the holding area at sending or receiving sites. This same distance was assumed for other truck transport legs, including to and from the air strip, to the marine dock, and from storage to the MHI.

8-5
The data base for truck accidents is discussed in Section 9.1.2, based largely on information in Refs. 8-1 through 8-5. A five-vehicle convoy will be used to transport munitions onsite. There is a lead security vehicle, one munition vehicle, a decontamination vehicle, an emergency vehicle, and a following security vehicle (Ref. 8-8). The small distance that the convoy travels, and the small number of trucks per convoy, make traffic control feasible to provide front, rear, and side collision protection. The major controls that affect the truck accident rate are:

- No other movement activities or other activities which might pose a hazard to the munitions will be allowed to be carried out within 500 ft of the convoy route during munitions transport.
- 2. No fires external to the cargo and capable of challenging the package containment (i.e., an engulfing fire of 1850°F) will last longer than 10 min due to limits placed on the amount of truck fuel available.
- Truck/train collisions are not credible because of the escort and the absence of train traffic during convoy movement.
- 4. No munitions movement will take place during periods of extreme weather conditions such as storms, tornado advisories, and blizzards, although a fully loaded truck may have to remain at rest during the bad weather.

Using this convoy model, several general assumptions can be made about the types of accidents that are possible:

1. Head-on collisions with a munitions vehicle are not credible.



- 2. Collisions in which a munitions vehicle rear-ends another vehicle are low-speed events limited by convoy speed.
- Collisions in which a munitions vehicle hits a stationary object or overturns are low-speed events limited by convoy speed.
- Collisions in which a munitions vehicle is rear-ended by another vehicle are low-speed events limited by convoy speed.
- Collisions in which a munitions vehicle is struck from the side are not credible because of restrictions on other movement activities during convoy movement.

These assumptions limit the type of accident scenarios envisioned for local munition transport to truck collisions and overturns, spontaneous fires, and nonpreventable external events such as aircraft crashes, earthquakes, and tornadoes during transport.

#### 8.1.3. Accident Scenarios for OFC and Vault Transport

Section 4.1 describes the logic for initiating event selection of onsite transport accidents. Table 4-4 shows four families of initiating events: (1) truck collision or overturn accident due to human error or equipment failure, (2) aircraft crash into the truck, (3) earthquakeinduced collision or overturn, and (4) tornado-caused collision/overturn or missile impact. These four initiating event families were used to develop the scenario event trees as described in the subsections below.

8.1.3.1. <u>Truck Collision/Overturn</u>. Figure 8-2 shows the event tree for truck collision or overturn due to human error or mechanical failure. There are five important sequences resulting from this scenario,

BURN OR DETON. BURN OR DETON. DETONATION TYPE OF AGENT RELEASE NONE SPILL SPILL SPILL NONE NONE 2.2 × 10<sup>-10</sup> (c) SEQUENCE FREQUENCY PER MILE 2.8 × 10<sup>-10</sup> SEQUENCE 10 VR4, VR5 VR3 VR2 VR4 VRI MONITOR Failure Mode PUNCTURE THERMAL INTACT INSUL. LOSS PACKAGE Failure Mode PUNCTURE INTACT IMPACT INTACT CRUSH 0.0 0.002 . INSUFFICIENT FIRE DURATION TO FAIL CONTAINMENT 7 FIRE Avoided 0.07 (a) 0.17 (b) (a) COLLISION/OVERTURN (b) ALL TRUCK ACCIDENTS (c) ROCKETS DNLY, ¢ FOR OTHER MUNITIONS DETONATION DUE TO UNDUE FORCE AVOIDED 0.0016 ROCKETS 2.2 × 10<sup>-5</sup> others 1.4 × 10<sup>-7</sup> (a) 1.7 × 10<sup>-7</sup> (b) PER MILE TRUCK Accident Occurs

. 84". 86". 86". 86". 84". 86". 86". 86".

Fig. 8-2. Event tree for onsite transportation (truck accident)



differentiated by the types of force that could cause agent release (crush, impact, puncture, and fire). These are:

| VR1<br>VA1<br>VW1                                                   | - A munitions vehicle collision/overturn occurs<br>and crush forces fail the agent containment.                                                                                                                                                                    |
|---------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| VR2<br>VA2<br>VW2                                                   | - A munitions vehicle collision/overturn occurs<br>and impact forces fail the agent containment.                                                                                                                                                                   |
| VR3<br>VA3<br>VW3                                                   | - A munitions vehicle collision/overturn occurs<br>and puncture forces fail the agent containment.                                                                                                                                                                 |
| VR4<br>VA4<br>(not applicable<br>to the marine<br>transport option) | - Detonation of burstered munitions occurs by<br>either (1) fire only accident, (2) mechanical<br>force and fire, (3) truck collision/overturn<br>impact-induced rocket propellant ignition, or<br>(4) truck collision/overturn induced undue<br>force detonation. |
| VR5<br>VA5<br>VW5                                                   | - A munitions vehicle accident with fire occurs, causing nonburstered munitions to fail.                                                                                                                                                                           |

Note that the sequence coding beginning with VR denotes onsite transport for the rail shipment option, VA denotes onsite transport event for the air transport option and VW pertains to the marine shipment option (nonburstered munitions only). VO refers to truck transport to the disposal facility, which is analyzed separately in Section 8.1.4.

Data base information (Section 9.1.2) indicates that for generic highway accidents the rate is  $2.5 \times 10^{-6}$  collisions/overturns per mile. However, this rate is modified for the use of convoy and administrative controls (Table 9-7). The convoy speed will be selected so that the maximum velocity at which a collision or rollover involving a munitions vehicle can occur in convoy conditions is estimated to be no greater than 30 mph, even assuming gross driver error or mechanical failure (e.g., brakes) on a hill. Because the convoy is moving at low speed relative to highway traffic and under closely controlled conditions, the time allowed for driver response to threatening conditions is much greater at the lower speed, and collision-type accidents and overturn-

type accidents are more avoidable. Convoy accident frequencies have been decreased by a factor of 10 from highway accident frequencies because of greater driver awareness and control during convoy conditions. The probability of accidental collisions and overturns involving mechanical forces thus becomes  $1.4 \times 10^{-7}$  per mile. Mechanical force accident scenarios represent 83% of the total accidents expected.

Fires can break out in the cargo and in the vehicle without the occurrence of a mechanical force accident. The SNL standard highway frequency for this type of accident is  $2.8 \times 10^{-8}$  per mile. The use of convoy controls does not change the probability of a fire occurring, so the accident rate used for convoy traffic is unchanged. Fire-only scenarios represent 17% of the total accidents expected.

The probabilities of mechanical forces (crush, impact, and puncture) being generated in a truck accident were taken from Ref. 8-2, consistent with the failure criteria in Table 3-1. These values are consistent with the data in Ref. 8-3. The probability of an undue mechanical force causing burster detonation was derived from the truck velocity data in Ref. 8-1, assuming a log normal distribution with a 50% probability of detonation at 123 mph and a  $10^{-6}$  probability at 13.5 mph (Ref. 8-6).

The probabilities of the top events of the event tree in Fig. 8-2 are discussed in Tables 8-1 through 8-5 for the munitions in OFCs or vaults for sequences 1 through 5 above. Sequence 1, 2, and 3 involve mechanical failure of the package by crush, impact force, and puncture, respectively. Fire, in conjunction with the crush, will cause different consequences (agent release) than mechanical failure without fire. The truck accident data base was examined by H&R Associates, Inc. to derive the fraction of truck collision/overturn accidents which are accompanied by fire. It was determined that 7% of the time a fire is also present. Thus, 93% of the time the consequence is an agent spill only; the remainder results in some unburned vapor release to the atmosphere.

# TABLE 8-1ONSITE TRANSPORT SEQUENCE 1

.....

VR1 - A truck collision/overturn occurs in which the munitions are subjected primarily to crush forces with other forces being
 VW1 negligible. The agent release frequency is the product of three basic events: BE31, BE68, and BE73.

| No.  | Name                                      | Probability                        | Reference/Remarks                                                                                                                                                                                                                                                                                                                                        |
|------|-------------------------------------------|------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BE31 | Truck collision/<br>overturn              | 1.4 x 10 <sup>-7</sup><br>per mile | Table 9-7.                                                                                                                                                                                                                                                                                                                                               |
| BE68 | Crush force<br>generated                  | 0.05 VR, VA<br>1 VW                | Reference 8-2.                                                                                                                                                                                                                                                                                                                                           |
| BE73 | Crush force<br>fails agent<br>containment | ε                                  | The offsite package is designed<br>to withstand an evenly dis-<br>tributed static crush load of<br>520,000 lb. The nonuniform<br>static crush failure threshold<br>is not known exactly, but is<br>well above the maximum expected<br>crush force of 15,000 lb (maxi-<br>mum crush load for a large<br>package being transported by<br>truck = Ref. 8-2) |



VR2 - A truck collision/overturn occurs in which munitions are subjected primarily to impact forces with other forces being VA2 VW2 negligible. The agent release frequency is the product of three basic events: BE31, BE60, and BE71.

| No.  | Name                                                    | Probability                        | Reference/Remarks                                                                                                                                                                                                                                                                 |
|------|---------------------------------------------------------|------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BE31 | Truck collision/<br>overturn                            | 1.4 x 10 <sup>-7</sup><br>per mile | Table 9-7.                                                                                                                                                                                                                                                                        |
| BE60 | Impact force<br>generated                               | 0.80 VR, VA<br>1 VW                | Reference 8-2.                                                                                                                                                                                                                                                                    |
| BE71 | Impact force<br>fails agent<br>containment<br>(>35 mph) | £                                  | The impact failure threshold<br>for the package is 35 mph.<br>The maximum postulated impact<br>velocity in any accident is<br>30 mph, thereafter, the proba-<br>bility of agent release due to<br>impact to zero, or very close<br>to it, signified by epsilon<br>( $\epsilon$ ). |

#### TABLE 8-3 **ONSITE TRANSPORT SEQUENCE 3**

to

VR3 - A truck collision/overturn occurs in which the munitions VA3 are subjected primarily to puncture forces with other forces VW3 being negligible. The agent release frequency is the product of three basic events: BE31, BE64, and BE67.

| No.  | Name                                                                                           | Probability                        | Reference/Remarks |
|------|------------------------------------------------------------------------------------------------|------------------------------------|-------------------|
| BE31 | Truck collision/<br>overturn                                                                   | l.4 x 10 <sup>-7</sup><br>per mile | Table 9-7.        |
| BE64 | Puncture envi-<br>ronment occurs                                                               | 0.2 VR, VA<br>0.02 VW              | Reference 8-2.    |
| BE67 | Probe fails<br>agent contain-<br>ment (0.75 in.<br>mild steel wall<br>equivalent<br>thickness) | 0.01                               | Reference 8-2.    |

**F** 

## TABLE 8-4ONSITE TRANSPORT SEQUENCE 4

VR4 - Detonation of burstered munitions by (1) fire-only accident, VA4 (2) mechanical force and fire, (3) truck collision/overturn impact-induced rocket propellant ignition, or (4) truck collision/overturn induced undue force detonation. The release frequency is calculated by: (BE31) (BE62) (BE63) + (BE31A) (BE52') (BE62A) (BE63A) + (BE31A) (BE60) (BE61R) + (BE31A) (BE61). The third term is for rockets<sup>(a)</sup> only.

| Event<br>No. | Name                                                                                      | Probability                        | Reference/Remarks                                                     |
|--------------|-------------------------------------------------------------------------------------------|------------------------------------|-----------------------------------------------------------------------|
| BE31         | Truck accident<br>occurs                                                                  | 1.7 x 10 <sup>-7</sup><br>per mile | Table 9-7.                                                            |
| BE62         | Fire generated                                                                            | 0.17                               | Table 9-7.                                                            |
| BE63         | Fire has heat<br>and duration to<br>detonate burster<br>(>2 h)                            | E                                  | Trucks limited to only enough<br>fuel for the fire to last<br>10 min. |
| BE31A        | Truck collision/<br>overturn occurs                                                       | 1.4 x 10 <sup>-7</sup><br>per mile | Table 9-7.                                                            |
| BE52'        | Mechanical forces<br>destroy package<br>insulation                                        | 0.01                               | Reference 8-6.                                                        |
| BE62A        | Fire occurs,<br>given a collision<br>or overturn                                          | 0.07                               | Reference 8-6.                                                        |
| BE63A        | Fire has heat<br>and duration to<br>detonate burster<br>(>30 min for<br>degraded package) | E                                  | Trucks limited to only enough<br>fuel for the fire to last<br>10 min. |
| BE60         | Impact force<br>generated                                                                 | 0.80                               | Reference 8-1.                                                        |
| BE61(R)      | Impact force suf-<br>ficient to deto-<br>nate burster<br>through propel-<br>lant ignition | 0.002                              | Reference 8-5.                                                        |
| BE61         | Undue force deto-<br>nation occurs                                                        | 2.2 x 10 <sup>-5</sup>             | Reference 8-6.                                                        |

(a)Puncture-induced rocket propellant ignition has not been included because there is no evidence that a probe exists or could occur at the velocities necessary to cause puncture-induced propellant ignition. A 30-caliber bullet traveling about 1500 mph is required.



# TABLE 8-5ONSITE TRANSPORT SEQUENCE 5

VR5 - A truck accident occurs and a resulting fire fails non-VA5 burstered munitions. The agent release frequency is the

VW5 product of three basic events: BE31, BE62, and BE75, added to the product of BE31A, BE52', BE62A, and BE75A.

| No.   | Name                                                                                  | Probability                        | Reference/Remarks                                                        |
|-------|---------------------------------------------------------------------------------------|------------------------------------|--------------------------------------------------------------------------|
| BE31  | Truck accident<br>occurs                                                              | 1.7 x 10 <sup>-7</sup><br>per mile | Table 9-7.                                                               |
| BE62  | Fire occurs                                                                           | 0.17 VR, VA<br>0.02 VW             | Table 9-7.                                                               |
| BE75  | Thermal force<br>fails agent con-<br>tainment (>2 h)                                  | E                                  | Trucks are limited to carrying<br>only enough fuel for a 10-min<br>fire. |
| BE31A | Truck collision/<br>overturn occurs                                                   | 1.4 x 10 <sup>-7</sup><br>per mile | Table 9-7.                                                               |
| BE52' | Mechanical<br>forces destroy<br>package insula-<br>tion                               | 0.01                               | Reference 8-6.                                                           |
| BE62A | Fire, given a collision                                                               | 0.07                               | Reference 8-6.                                                           |
| BE75A | Thermal force<br>fails agent con-<br>tainment inside<br>degraded package<br>(>30 min) | E                                  | Trucks are limited to carrying<br>only enough fuel for a 10-min<br>fire. |

8-14

- 2522233

Ľ,

a state of the state

The results in Tables 8-1 through 8-5 show that sequences 1 and 2 can be screened out on the basis of low frequency due to the extremely low probabilities of mechanical failure by impact or crush. Sequence 3, involving mechanical failure by puncture, has a frequency of  $3.8 \times 10^{-10}$ /mile.

8.1.3.2. <u>Aircraft Crash</u>. Figure 8-3 shows the event tree for aircraft crash into a truck. The initiating event frequency is discussed and quantified in Section 4.2 in terms of number of crashes of small and large aircraft per year per unit area at each site (Table 4-7). Aircraft crash (large and small) values from Table 4-7 were modified to account for uncontrolled crashes and then multiplied by the truck cross-sectional area. An uncontrolled crash is defined as one where the impact angle is greater than 10 deg. It was assumed that for an aircraft to actually hit a truck, the crash would have to be uncontrolled. Modification consisted of multiplying the accident frequency contributions during takeoff, inflight and landing phases by the fraction of time that the crash has an impact angle greater than 10 deg for that phase (see Table 8-6).

An inherent assumption is that an accident involving an aircraft crashing onto a munitions vehicle more closely resembles the Sandia National Laboratory (SNL) model of a typical aircraft crash rather than the SNL model of a typical truck crash. In a typical SNL aircraft crash, the crush and puncture forces are negligible compared to the impact forces. Further details are available in Ref. 8-1.

There are two important accident sequences resulting from the aircraft crash event tree, sequences 6 and 7. These are described and quantified in Tables 8-6 and 8-7:

8.1.3.3. <u>Earthquake</u>. Figure 8-4 shows the event tree for the earthquake occurrence impact on onsite transport. Section 4.2 presents earthquake frequencies as a function of earthquake intensity and site. In this study, an earthquake intensity of 0.5 g is assumed to be needed

SPILL OR Detonation SPILL OR Detonation NONE NONE NONE NONE 1.3 × 10<sup>-10</sup> TEAD 1.5 × 10<sup>-10</sup> TEAD VR 6 **VR 7** NOTE: INITIATING EVENT AND SEQUENCE FREQUENCIES ARE IN UNITS OF PER EXPOSURE YEAR. I 7 7 I I FIRST IMPACT 0.55 0.45 FEAD 2.8 × 10<sup>-10</sup> NAAP 9.1 × 10<sup>--10</sup> UMDA 1.8 x 10<sup>-9</sup> ANAD 1.1 × 10<sup>--9</sup> PUDA 8.4 x 10<sup>--9</sup>  $10^{-10}$ APG 1.3 × 10<sup>-7</sup> PBA 2 × 10<sup>-9</sup>

TYPE OF AGENT RELEASE

SEQUENCE Frequency Pervear

SE QUENCE ID

MUNITION

PACKAGE INTACT

IMPACT ONLY (NO FIRE)

AIRCRAFT CRASHES

INTO TRUCK

Event tree for onsite transportation (aircraft crash) F1g. 8-3. POPOSANA AND PARA

C BUANNASS

<u>ᡵ᠆ᡵ᠆ᡵᠮᡵ᠆ᡵ᠆ᡵ᠆ᡵ᠆ᡵ᠆ᡵ᠘ᡵ᠘ᡵ᠘ᡵ᠆ᡵ᠘</u>ᢌ

: • • • دخد

## TABLE 8-6ONSITE TRANSPORT SEQUENCE 6

200

VR6 - An aircraft crashes into a munitions truck; no fire results.
VA6 In aircraft accidents, the SNL data indicate that the preVW6 dominant mechanical force employed against cargo packages is impact, with crush and puncture having negligible effect. The agent release frequency is the product of three impact basic events: BE31, BE60, and BE71.

| Event<br>No. | Name                                                                                             | Probability                                                                                                 | Reference/Remarks                                                                                                                                                                                                                                         |
|--------------|--------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BE31         | Aircraft crash<br>(per truck year)<br>APG<br>ANAD<br>LBAD<br>NAAP<br>PBA<br>PUDA<br>TEAD<br>UMDA | 1.3 x 10-7<br>1.1 x 10-9<br>5 x 10-10<br>9.1 x 10-10<br>2 x 10-9<br>8.4 x 10-9<br>2.8 x 10-10<br>1.8 x 10-9 | Nine percent of all crashes are<br>on takeoff, 32% inflight, and<br>58% on landing. Fifteen per-<br>cent have impact angles greater<br>than 10 deg in takeoff crashes,<br>70% in midflight crashes, and<br>13% in landing crashes<br>(Refs. 8-6 and 8-7). |
| BE60         | Impact force<br>only generated<br>(no fire)                                                      | 0.55                                                                                                        | Derived from data in Ref. 8-1;<br>49% of all aircraft crashes<br>involve impact with or without<br>other forces; $27\%$ of them are<br>impact only; $0.27/0.49 = 0.55$ .                                                                                  |
| BE71         | Impact force<br>fails agent<br>containment                                                       | 1                                                                                                           | Conservative value.                                                                                                                                                                                                                                       |

# TABLE 8-7ONSITE TRANSPORT SEQUENCE 7

VR7 - An aircraft crashes onto a munition truck, fire occurs but

VA7 impact forces fail agent containment. The agent release fre-VW7 quency is the product of three basic events: BE31, BE60/62, and BE71.

| No.     | Name                                                       | Probability                                                                                                                                                            | Reference/Remarks                                                                                                                                                          |
|---------|------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BE31    | Aircraft crash<br>(per truck year)                         |                                                                                                                                                                        |                                                                                                                                                                            |
|         | APG<br>ANAD<br>LBAD<br>NAAP<br>PBA<br>PUDA<br>TEAD<br>UMDA | $1.3 \times 10^{-7}$ $1.1 \times 10^{-9}$ $5 \times 10^{-10}$ $9.1 \times 10^{-10}$ $2 \times 10^{-9}$ $8.4 \times 10^{-9}$ $2.8 \times 10^{-10}$ $1.8 \times 10^{-9}$ | See remarks, sequence 6.                                                                                                                                                   |
| BE60/62 | Impact and fire<br>generated                               | 0.45                                                                                                                                                                   | 0.22/0.49 = 0.45 (fire and impact/all impact).                                                                                                                             |
| BE71    | Impact force<br>fails agent<br>containment                 | 1                                                                                                                                                                      | Conservatively assumes that at<br>least one package fails every<br>time. Burstered munitions<br>detonate. Nonburstered muni-<br>tions release agent by spill<br>and vapor. |



•



Fig. 8-4. Event tree for onsite transportation (earthquake)

ELECTRONIC CONTRACT

to cause a truck collision or overturn. Thus, the initiating event frequency is taken to that for a 0.5 g earthquake or greater (called a "severe earthquake") at the specific site.

The following sequences resulted from the earthquake event tree analysis:

VR9 - A severe earthquake occurs, causing a munitions vehicle VA9 accident, and crash forces fail the agent containment. VW9 VR10 - A severe earthquake occurs, causing a munitions vehicle VA10 accident, and impact forces fail the agent containment. **VW10** VR11 - A severe earthquake occurs, causing a munitions vehicle VA11 accident, and puncture forces fail the agent containment. VW11 VR12 - A severe earthquake occurs, causing a munitions vehicle VA12 accident, and fire detonates burstered munitions. VR13 - A severe earthquake occurs, causing a munitions vehicle **VA13** accident, and fire fails nonburstered munitions. **VW13** VR15 - An earthquake or tornado occurs, generating undue mechanical

Note that sequence 15 has a dual initiator, either a severe earthquake or a tornado (analyzed in the next subsection). Quantification of the earthquake event tree is described in Tables 8-8 through 8-12.

forces which cause detonation of burstered munitions.

**VA15** 

8.1.3.4. Tornado. Figures 8-5 and 8-6 show the event trees for a tornado or high winds causing a truck collision overturn or generating an impacting missile. The tornado frequency is presented in Section 4.2 for the specific sites. Quantification of the event trees is summarized in Tables 8-13 through 8-15.

It is assumed that, given the high winds present during a tornado, and the high probability of accompanying rain, that a significant fire will not be initiated by the tornado, or sustained during the tornado. Therefore, fire sequences were not included in the tornado scenarios.

# TABLE 8-8ONSITE TRANSPORT SEQUENCE 9

VR9 - An earthquake occurs in which the munitions are subjected
VA9 primarily to crush forces with other forces being negligible.
VW9 The agent release frequency is the product of three basic events: BE31, BE68, and BE73.

RECEIPTION DESCRIPTION DESCRIPTION

| Event<br>No. | Name                                                         | Probability                      | Reference/Remarks                                              |
|--------------|--------------------------------------------------------------|----------------------------------|----------------------------------------------------------------|
| BE31         | Earthquake occurs<br>(per year)<br>TEAD<br>NAAP<br>Elsewhere | 1 x 10-4<br>2 x 10-5<br>6 x 10-6 | Table 4-7; a ≥0.5-g earthquake<br>is assumed. See Section 4.2. |
| BE68         | Crush force<br>generated                                     | 0.05 VR, VA<br>1 VW              | Reference 8-2.                                                 |
| BE73         | Crush force fails agent containment                          | E                                | Same as sequence 1.                                            |





# TABLE 8-9ONSITE TRANSPORT SEQUENCE 10

VR10 - An earthquake occurs in which the munitions are subjected
VA10 primarily to impact forces with other forces being negliVW10 gible. The accident release frequency is the product of three basic events: BE31, BE60, and BE71.

| Event<br>No. | Name                                                         | Probability                      | Reference/Remarks                        |
|--------------|--------------------------------------------------------------|----------------------------------|------------------------------------------|
| BE31         | Earthquake occurs<br>(per year)<br>TEAD<br>NAAP<br>Elsewhere | 1 x 10-4<br>2 x 10-5<br>6 x 10-6 | Table 4-7; assumes ≥0.5-g<br>earthquake. |
| BE60         | Impact force<br>generated                                    | 0.80 VR, VA<br>1 VW              | Reference 8-2.                           |
| BE71         | Impact force<br>fails agent<br>containment                   | E                                | Same as sequence 2.                      |

<u>-00</u> • • •

# TABLE 8-10ONSITE TRANSPORT SEQUENCE 11

VR11 - An earthquake occurs in which the munitions are subjected
 VA11 primarily to puncture forces with other forces being negli VW11 gible. The agent release frequency is the product of three basic events: BE31, BE64, and BE67.

| Event<br>No. | Name                             | Probability                                                    | Reference/Remarks |
|--------------|----------------------------------|----------------------------------------------------------------|-------------------|
| BE31         | Earthquake occurs<br>(per year)  |                                                                |                   |
|              | TEAD<br>NAAP<br>Elsewhere        | $1 \times 10^{-4}$<br>$2 \times 10^{-5}$<br>$6 \times 10^{-6}$ | Table 4-7.        |
| BE64         | Puncture environ-<br>ment occurs | 0.2 VR, VA<br>0.02 VW                                          | Reference 8-2.    |
| BE67         | Probe fails agent containment    | 0.01                                                           | Reference 8-2.    |





a na sana na s

### TABLE 8-11ONSITE TRANSPORT SEQUENCE 12

VR12 - An earthquake occurs and accidental forces cause detona-VA12 tion of burstered munitions. This sequence is similar to sequence VR4 and VR15. The agent release frequency is the product of three basic events: BE31, BE62, and BE63 added to the product of BE31, BE52', BE62, and BE63A. The product of three propellant-ignition events: BE31, BE60, and BE61R is added to the result for rockets.(a)

| No.     | Name                                                                                        | Probability                                                    | Reference/Remarks                                                        |
|---------|---------------------------------------------------------------------------------------------|----------------------------------------------------------------|--------------------------------------------------------------------------|
| BE31    | Earthquake occurs<br>(per year)<br>TEAD<br>NAAP<br>Elsewhere                                | $1 \times 10^{-4}$<br>$2 \times 10^{-5}$<br>$6 \times 10^{-6}$ | Table 4-7.                                                               |
| BE62    | Fire generated                                                                              | 0.07                                                           | Reference 8-6.                                                           |
| BE63    | Fire has heat and<br>duration to deto-<br>nate burster<br>(>2 h)                            | E                                                              | Trucks are limited to only<br>enough fuel for a 10-min fire.             |
| BE52'   | Mechanical forces<br>destroy package<br>insulation                                          | 0.01                                                           | Reference 8-6.                                                           |
| BE63A   | Fire has heat and<br>duration to deto-<br>nate burster for<br>degraded package<br>(>30 min) | E                                                              | Trucks are limited to carrying<br>only enough fuel for a 10-min<br>fire. |
| BE60    | Impact force<br>generated                                                                   | 0.80                                                           | Reference 8-2.                                                           |
| BE61(R) | Impact force suf-<br>ficient to deto-<br>nate burster                                       | 0.002                                                          | Reference 8-2.                                                           |

(a)Puncture-induced rocket propellant ignition has not been included because there is no evidence that a probe exists or could occur at velocities necessary to cause propellant ignition (30-caliber bullets traveling about 1500 mph are required).

### TABLE 8-12ONSITE TRANSPORT SEQUENCE 13

VR13 - An earthquake occurs and fire fails nonburstered munitions.
VA13 The agent release frequency is the product of three external
VW13 fire basic events: BE31, BE62, and BE75 added to the product of BE31, BE52', BE62, BE75A.

| No.   | Name                                                                                  | Probability                      | Reference/Remarks                                                        |
|-------|---------------------------------------------------------------------------------------|----------------------------------|--------------------------------------------------------------------------|
| BE31  | Earthquake occurs<br>(per year)<br>TEAD<br>NAAP<br>Elsewhere                          | 1 x 10-4<br>2 x 10-5<br>6 x 10-6 | Table 4-7.                                                               |
| BE62  | Fire generated                                                                        | 0.07                             | Reference 8-6.                                                           |
| BE75  | Thermal force<br>fails agent con-<br>tainment (>2 h)                                  | ε                                | Trucks are limited to only<br>enough fuel for a 10-min fire.             |
| BE52' | Mechanical forces<br>destroy package<br>insulation                                    | 0.01                             | Reference 8-6.                                                           |
| BE75A | Thermal force<br>fails agent con-<br>tainment inside<br>degraded package<br>(>30 min) | ε                                | Trucks are limited to carrying<br>only enough fuel for a 10-min<br>fire. |

DETONATION

SPILL

Event tree for tornado-caused truck collision/overturn during onsite transportation F1g. 8-5.

|                                             |        |                                                                       | · · · · · · · · · · · · · · · · · · · |             | _                |                              |                        | 4                    |
|---------------------------------------------|--------|-----------------------------------------------------------------------|---------------------------------------|-------------|------------------|------------------------------|------------------------|----------------------|
| SEQUENCE<br>Frequency<br>Per year           |        |                                                                       | ç                                     | ij          | 6.6 × 10-11 TEAD | 1.6 × 10 <sup>-17</sup> TEAD |                        |                      |
| SEQUENCE                                    |        |                                                                       |                                       |             | VR 14B           | VR 15                        |                        |                      |
| AGENT<br>CONTAINMENT<br>FAILURE MODE        | INTACT | INTACT                                                                | CRUSH                                 | €<br>IMPACT | ¢<br>PUNCTURE    | 0.002                        |                        | (POSURE YEAR.        |
| DETONATION<br>DUE TO UNDUE<br>FORCE AVOIDED |        | <br> <br> <br> <br> <br>                                              | 7                                     |             |                  | -                            | 2.2 × 10 <sup>-5</sup> | EQUENCIES ARE PER EX |
| TRUCK NOT<br>CAUGHT IN<br>BAD WEATHER       |        | 6.0                                                                   | 1.0                                   |             |                  |                              | -                      | ENT AND SEQUENCE FR  |
| TORNADO<br>Occurs<br>WINDS 160 MPH          |        | 3.3 × 10 <sup>-7</sup> TEAD, UMDA<br>5.6 × 10 <sup>-6</sup> PHINA APG | 1.0 × 10 -4 ELSEWHERE                 |             |                  |                              |                        | NOTE: INITIATING EVE |

TYPE OF AGENT RELEASE

NONE

NONE

SPILL

SPILL

8-26



| ন্য |
|-----|

| SEQUENCE<br>1D                                                             |                 | VB14A  |                      |     |                                         |                                        |                                                                               |                                            |
|----------------------------------------------------------------------------|-----------------|--------|----------------------|-----|-----------------------------------------|----------------------------------------|-------------------------------------------------------------------------------|--------------------------------------------|
| AGENT RELEASE                                                              |                 | I IIdo |                      |     |                                         | NU HELEASE                             | NO DEL FAST                                                                   | NU NELEASE                                 |
| MISSILE HAS<br>Orientation to<br>Puncture<br>Package                       |                 | YES    | 3 × 10 <sup>-5</sup> |     | NO                                      |                                        |                                                                               | S PER EXPOSURE YEAR                        |
| A MISSILE<br>CAPABLE OF<br>PUNCTURING THE<br>PACKAGE IS<br>GENERATED       | WINDS > 310 MPH |        |                      | YES | TEAD, UMDA<br>(1.7 × 10 <sup>-4</sup> ) | PUUA, APG<br>(1.1 × 10 <sup>-3</sup> ) | ELSEWHERE<br>(1.4 × 10 <sup>-3</sup> )                                        | NO<br>(a) <sub>ACCIDENT</sub>              |
| TORNADO<br>OCCURS<br>WINDS > 160 MPH<br>AND TRUCK CAUGHT<br>IN BAD WEATHER |                 |        |                      |     | TORNADO <sup>(a)</sup>                  | TEAD, UMDA                             | (3.3 × 10 ) (10 )<br>PUDA, APG<br>(5 × 10 <sup>-6</sup> ) (10 <sup>-1</sup> ) | $(1.0 \times 10^{-4})$ (10 <sup>-1</sup> ) |

Event tree for tornado-generated missile affecting onsite transportation F1g. 8-6.

8-27

.....

ا م م ب

and the second secon

# TABLE 8-13ONSITE TRANSPORT SEQUENCE 14A

Ì

VR14A - Tornado-generated missile causes agent containment to fail. VA14A The release frequency is the product of events BE31, BE31', VW14A BE64, and BE51.

| No.   | Name                                                                                                                                               | Probability                                                              | Reference/Remarks                                                                                                                           |
|-------|----------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|
| BE31  | Tornado occurs<br>(per year, winds<br>>160 mph)<br>TEAD, UMDA<br>PUDA, APG<br>Elsewhere                                                            | $3.3 \times 10^{-7}$<br>5.6 x 10 <sup>-6</sup><br>1.0 x 10 <sup>-4</sup> | Site specific; see Section 4.2.                                                                                                             |
| BE31' | Truck traveling<br>in bad weather                                                                                                                  | 0.1                                                                      | Assumes a 10% chance that the<br>administrative control pro-<br>hibiting travel in bad weather<br>will be violated.                         |
| BE64  | Tornado-generated<br>missile capable<br>of puncturing and<br>failing agent<br>containment<br>occurs (winds<br>)310 mph)<br>TEAD, UMDA<br>PUDA, APG | $1.7 \times 10^{-4}$<br>$1.1 \times 10^{-3}$                             | Fraction of winds >160 mph that<br>are also >310 mph. See Sec-<br>tion 4.2 and Appendix C. Con-<br>servative for heavy-walled<br>munitions. |
|       | Elsewhere                                                                                                                                          | $1.4 \times 10^{-3}$                                                     |                                                                                                                                             |
| BE51  | Missile fails<br>agent containment                                                                                                                 | 3.0 x 10 <sup>-5</sup>                                                   | Methodology in Appendix C.                                                                                                                  |
|       |                                                                                                                                                    |                                                                          |                                                                                                                                             |

Recent

#### TABLE 8-14 ONSITE TRANSPORT SEQUENCE 14B

Calendar de la de la de la des

VR14B - Tornado causes a truck collision/overturn, generating VA14B mechanical forces that fail agent containment. The release VW14B frequency is calculated by: (BE31) (BE31') (BE68) (BE53) + (BE31) (BE31') (BE60) (BE52) + (BE31) (BE31') (BE64A) (BE51A).

Event No. Name Probability Reference/Remarks **BE31** Tornado occurs See external events section. (per year, winds >160 mph)  $3.3 \times 10^{-7}$ TEAD, UMDA 5.6 x  $10^{-6}$ PUDA, APG  $1.0 \times 10^{-4}$ Elsewhere BE31' Truck traveling 0.1 Assumes a 10% chance that the in bad weather administrative control prohibiting travel during bad weather is violated. **BE68** Crush force 0.05 VR, VA Reference 8-2. generated 1 VW **BE60** Impact force 0.80 VR, VA Reference 8-2. generated 1 VW BE64A Puncture 0.2 VR, VA Reference 8-2. environment 0.02 VW BE53 Crush fails € Maximum expected crush force agent containequals 15,000 lb; Ref. 8-8. ment BE52 Impact fails Maximum postulated velocity € containment change (30 mph) does not (>35 mph) exceed package failure threshold of 35 mph. BE51A Probe fails 0.01 Reference 8-2. agent containment (0.75 in. mild steel wall equivalent thickness)

# TABLE 8-15ONSITE TRANSPORT SEQUENCE 15(a)

VR15 - An earthquake or tornado occurs, generating undue mechanical
 VA15 forces which cause detonation of burstered munitions (types
 C, P, M, R). The release frequency is the product of events
 BE31 and BE61 added to the product of BE31A, BE31A', and
 BE61.

| No.    | Name              | Probability          | Reference/Remarks                                                                     |
|--------|-------------------|----------------------|---------------------------------------------------------------------------------------|
| BE31   | Earthquake occurs |                      |                                                                                       |
|        | TEAD              | $1 \times 10^{-4}$   | See Section 4.2.                                                                      |
|        | NAAP              | 2 x 10 <sup>-5</sup> |                                                                                       |
|        | Elsewhere         | 6 x 10 <sup>-6</sup> |                                                                                       |
| BE31A  | Tornado occurs    |                      |                                                                                       |
|        | (per year)        |                      |                                                                                       |
|        | TEAD, UMDA        | $3.3 \times 10^{-7}$ | See Section 4.2.                                                                      |
|        | PUDA, APG         | 5.6 x $10^{-6}$      |                                                                                       |
|        | Elsewhere         | $1.0 \times 10^{-4}$ |                                                                                       |
| BE31A' | Trucks traveling  | 0.1                  | Assumes a 107 probability that                                                        |
|        | in bad weather    |                      | the administrative control pro-<br>hibiting travel during bad<br>weather is violated. |
| BE61   | Undue mechanical  | $2.2 \times 10^{-5}$ | Reference 8-6.                                                                        |
|        | force sufficient  |                      |                                                                                       |
|        | to detonate       |                      |                                                                                       |
|        | burster occurs    |                      |                                                                                       |

The following sequences resulted from the event tree analysis (Figs. 8-5 and 8-6): 14A, 14B, and 15. In 14A, a penetrating missile is generated. In 14B, the tornado causes a truck collision/overturn with mechanical failure of a package. For 14B, only the puncture failure mode was found to be significant (impact and crush failure are negligible). Thus, the frequencies of 14A and 14B were summed to form sequence 14, since the consequences were the same.

#### 8.1.4. Accident Scenarios for ONC Transport

Transport of munitions in ONCs occurs during the last leg, from interim storage to the MHI at the disposal site (TEAD or ANAD). The structure of the accident event trees and the list of accident sequences is the same for this leg as for other onsite transport legs. However, quantification of the accident sequences differs because the ONC package has different failure criteria and thresholds compared to the OFC package (see Section 3.3). To distinguish the ONC transport, the sequences are denoted by V1 through V15: For the NDC option, they apply only to TEAD; for the RDC option, they apply to both TEAD and ANAD. Corresponding initiating event frequencies for external events at these specific sites are used.

There are three areas where conditional failure probabilities change from those in Section 8.1.3. These are mechanical failure (impact, crush or puncture), detonation of burstered munitions due to a fire, and tornado-generated missile puncture probability. All other branch probabilities are as described in Section 8.1.3.

For crush, the ONCs and munitions themselves can withstand a crush load to 50,000 lb (Table 3-1). Thus, compared to the negligible probability of crush for an OFC, the ONC package has nonnegligible crush failure probability estimated to be 0.1 based on data in Ref. 8-1 on the load environment for a small package. This is the combined probability that the crush force is generated and that the force fails the containment.

For impact, the ONC failure threshold is 35 mph (Table 3-1), compared to the maximum impact velocity in any accident of 30 mph. Thus, the probability of agent release due to impact is close to zero, signified by  $\epsilon$ , as it was for an OFC.

For puncture, ONC failure occurs at a velocity/radius threshold of 100/s, which is half that for the OFC package (Table 3-1). Based on information in Ref. 8-1, the probability that a puncture environment occurs during an accident is 0.016, while the probability that the probe fails the agent containment is 0.024. Thus, the ONC puncture failure probability is about five times higher than for an OFC.

For burstered munitions in an ONC, the time duration to detonate in an engulfing fire is 15 min (Table 3-1). Since the trucks are limited to only enough fuel for a 10 min fire, the detonation probability is quite low, estimated at  $10^{-6}$  per accident.

A tornado-generated missile capable of puncturing the munition inside an ONC is estimated to require winds greater than 250 mph (compared to 310 mph for OFCs). The conditional probability of occurrence for these high winds, given a tornado with winds greater than 160 mph, is 1.7 x  $10^{-4}$  for TEAD and 1.4 x  $10^{-3}$  for ANAD. Given winds >250 mph, the conditional probability of agent containment failure is estimated at 3.0 x  $10^{-5}$  (Appendix C).

#### 8.1.5. Analytical Results

The results of the probabilistic analysis of the accident sequences (median frequency values) are shown in Table 8-16. Results for the onsite transportation from interim storage to the dwnil facility for the collocation options are shown separately in Table 8-17, including the



CONTRACTOR OF CONTRACT,





6.0

# TABLE 8-16 ONSITE TRANSPORTATION - REGIONAL AND NATIONAL DISPOSAL OPTIONS (MOVEMENT TO AND FROM RAIL, AIR, AND SEA IN OFFSITE PACKAGE)

Scenario Frequencies and Range Factors

| SCEN-<br>Ario | ¥o. | ANAD<br>Freq | RANGE<br>FACTOR | APG<br>Freq | RANGE<br>FACTOR  | L BAD<br>FREQ    | RANGE<br>FACTOR | NAAP<br>FREQ | RANGE<br>Factor | PBA<br>Fred | RANGE<br>Factor | PUDA<br>Freq | range<br>Factor | TEAD<br>Freq | KANGE<br>FACTOR | LMDA<br>FREQ | RANGE<br>FACTOF |
|---------------|-----|--------------|-----------------|-------------|------------------|------------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|-----------------|
|               | •   | 1            | 1               |             | J<br>1<br>1<br>1 | ,<br>,<br>,<br>, | 4               |              |                 |             |                 |              |                 |              |                 |              |                 |
| VRBGS         | *7  | N/A          | :               | N/A         | ł                | N/A              | ;               | N/A          | }               | N/A         | ł               | N/A          | :               | 2.86-10      | 2.2E+01         | 2.86-10      | 2.2E+0          |
| VRDHS         | ~   | 2.8E-10      | 2.2E+01         | N/A         | ;                | A/M              | ;               | N/A          | ;               | N/N         | !               | 2.86-10      | 2.2E+01         | 2.8E-10      | 2.2E+01         | N/A          | !               |
| VRCES         | m   | 2.8E-10      | 2.26+01         | N/A         | 1                | N/A              | ł               | N/A          | ;               | N/A         | :               | N/A          | ;               | 2.86-10      | 2.2E+01         | N/A          | 1               |
| VRCHS         | ~   | 2.8E-10      | 2.2E+01         | N/A         | ;                | N/A              | 1               | N/A          | ;               | N/A         | ;               | 2.BE-10      | 2.2E+01         | 2.8E-10      | 2.2E+01         | N/A          | 1               |
| VRK6S         | ы   | N/A          | 1               | N/A         | !                | N/A              | ł               | N/A          | ;               | N/A         | ł               | N/A          | ł               | 2.86-10      | 2.2E+01         | N/A          | ł               |
| VRKHS         | ~   | 2.8E-10      | 2.2E+01         | 2.8E-10     | 10-32.26         | N/A              | ł               | N/A          | ;               | 2.86-10     | 2.2E+01         | N/A          | ł               | 2.86-10      | 2.2E+01         | 2.8E-10      | 2.2E+(          |
| VRKVS         | ~   | 2.86-10      | 2.2E+01         | N/A         | !                | N/A              | :               | 2.8E-10      | 2.2E+01         | N/A         | ;               | N/A          | ;               | 2.86-10      | 2.2E+01         | N/A          | :               |
| VRRVS         | ~   | 2.8E-10      | 2.2E+01         | N/A         | ;                | N/A              | ł               | N/A          | ;               | 2.8E-10     | 2.2E+01         | N/A          | 1               | 2.86-10      | 2.2E+01         | 2.86-10      | 2.2E+(          |
| VRPGS         | 3   | 2.86-10      | 2.2E+01         | N/A         | :                | N/A              | ł               | N/A          | ł               | N/A         | ł               | N/A          | ;               | 2.86-10      | 2.2E+01         | 2.86-10      | 2.2E+(          |
| VRFHS         | m   | 2.86-10      | 2.26+01         | N/A         | 1                | 2.8E-10          | 2.25+01         | A/A          | ļ               | N/A         | ł               | 2.86-10      | 2.2E+01         | 2.0E-10      | 2.2E+01         | N.'A         | 1               |
| VRPVS         | m   | 2.8E-10      | 2.2E+01         | N/R         | ;                | 2.8E-10          | 2.2E+01         | N/A          | ł               | N/A         | ł               | N/A          | ł               | 2.86-10      | 2.2E+01         | 2.86-10      | 2.26+(          |
| VRQ6S         | m   | 2.8E-10      | 2.26+01         | N/A         | ł                | 2.6E-10          | 2.2E+01         | N/A          | }               | N/A         | ł               | N/A -        | 1               | 2.86-10      | 2.2E+01         | 2.86-10      | 2.2E+(          |
| VROVS         | ~   | N/A          | 1               | N/A         | :                | N/A              | ł               | N/A          | ł               | N/A         | ;               | N/A          | ł               | 2.86-10      | 2.2E+01         | 2.86-10      | 2.25+0          |
| VRAGS         | ~   | 2.66-10      | 2.2E+ÙI         | N/A         | :                | 2.86-10          | 2.2E+01         | N/A          | ;               | 2.86-10     | 2.2E+01         | N/A          | 1               | 2.8E-10      | 2.2E+01         | 2.8E-10      | 2.2E+(          |
| VRRVS         | ~   | 2.86-10      | 2.2E+01         | N/A         | ;                | 2.8E-10          | 2. ZE+01        | N/A          | 1               | 2.8E-10     | 2.2E+01         | N/A          | ł               | 2.BE-10      | 2.2E+01         | 2.8E-10      | 2.2E+(          |
| VRSVS         | m   | N/A          | :               | N/A         | :                | N/A              | ł               | N/A          | ;               | N/A         | 1               | N/A          | ł               | 2.BE-10      | 2.2E+01         | 2.86-10      | 2.2E+(          |
| VRDHC         | 4   | 3.0E-12      | 2.65+01         | N/A         | :                | N/A              | ł               | N/A          | ţ               | N/A         | ;               | 3.0E-12      | 2.6E+01         | 3.06-12      | 2.6E+01         | N/A          | ;               |
| VRCGC         | -   | 3.0E-12      | 2.6E+01         | N/A         | ;                | N/A              | 1               | N/A          | 1               | N/A         | 1               | N/A          | ł               | 3.0E-12      | 2.6E+01         | N/A          | :               |
| VRCHC         | -   | 3.0E-12      | 2.6E+01         | N/A         | ł                | N/A              | ;               | N/A          | ;               | N/A         | :               | 3.06-12      | 2.6E+01         | 3.06-12      | 2.6E+01         | N/A          | !               |
| VRAVC         | -   | 3.0E-12      | 2.6E+01         | N/A         | :                | N/A              | :               | N/A          | ;               | 3.0E-12     | 2.6E+01         | N/A          | ;               | 3.0E-12      | 2.6E+01         | 3.0E-12      | 2.6E+(          |
| VRFGC         | -   | 3.0E-12      | 2.6E+01         | A/A         | :                | N/A              | ;               | N/A          | ;               | N/A         | :               | N/A          | :               | 3.0E-12      | 2.6E+01         | 3.0E-12      | 2.6E+(          |
| VRPHC         | *   | 3.0E-12      | 2.6E+01         | N/A         | 1                | 3.0E-12          | 2.6E+01         | N/A          | 1               | N/A         | ;               | 3.06-12      | 2.6E+01         | 3.0E-12      | 2.6€+01         | N/A          | :               |
| VRPVC         | 4   | 3. 0E-12     | 2.6E+01         | N/A         | ;                | 3.06-12          | 2.6E+01         | N/A          | :               | N/A         | 1               | N/A          | ł               | 3.0E-12      | 2.66+01         | 3.0E-12      | 2.6E+(          |
| VRDGC         | 4   | 3.0E-12      | 2.6E+01         | N/A         | 1                | 3.0E-12          | 2.6E+01         | N/A          | ;               | N/A         | :               | N/A          | :               | 3.0E-12      | 2.6E+01         | 3. UE-12     | 2.bE+(          |
| VRBVC         | -   | N/A          | 1               | N/A         | ł                | N/A              | ;               | N/A          | :               | N/A         | ;               | N/A          | :               | 3.0E-12      | 2.6E+01         | 3.06-12      | 2.6E+(          |
| VRRGC         | 4   | 2.2E-10      | 2.6€+01         | N/A         | ł                | 2.2E-10          | 2.6E+01         | N/A          | :               | 2.2E-10     | 2.6E+01         | N/A          | !               | 2.2E-10      | 2.6E+01         | 2.2E-10      | 2.6E+(          |
| VRKVC         | *   | 2.26-10      | 2.6E+01         | N/A         | ļ                | 2.2E-10          | 2.6E+Ù1         | N/A          | !               | 2.2E-10     | 2.6€+01         | N/A          | ł               | 2.2E-10      | 2.6E+01         | 2.26-10      | 2.6E+(          |
| VKBGS         | Ŷ   | N/A          | :               | N/A         | ;                | N/A              | ł               | N/A          | 1               | N/A         | ł               | N/A          | ;               | 1.66-10      | 2.0E+01         | 1.06-09      | 2.0E+C          |
| VRDHC         | þ   | 6.0E-1ù      | 2.ÚE+01         | N/A         | !                | N/A              | :               | N/A          | 1               | N/A         | ;               | 4.7E-09      | 2.0E+01         | 1.èE-10      | 2.0E+01         | N/A          | ł               |

See notes at end of table.

File: ONSITREG.Mk1 Page 2 Date 19-Aug-87

TABLE 8-16 (Continued)

\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*

LORGER L

action o best of the appropriate interaction interaction and an action of the action o

ONSITE TRANSPORTATION - REGIONAL AND NATIONAL DISPOSAL OPTIONS (movement to and from rail in offsite package)

Scenario Frequencies and Kange Factors

| 3. AMAD KANGE APG RANGE LBAD RANGE MAAP RANGE PBA RANGE<br>Freg Factor Freg Factor Freg Factor Freg Factor | KANGE APG RANGE LØAD RANGE NAAP RANGE PBA RANGE<br>Factor freg factor freg factor | APG RANGE LBAD RANGE NAAP RANGE PBA RANGE<br>Freq Factor Freg Factor Freg Factor | RANGE LBAD RANGE NAAP RANGE PBA RANGE<br>Factor Freg Factor Freg Factor | LBAD RANGE WAAP RANGE PBA RANGE<br>Freg Factor Freg Factor<br> | RANGE WAAP RANGE PBA RANGE<br>Factor Freg Factor | MAAP RANGE PBA RANGE<br>Freg Factor<br> | RANGE PBA RANGE<br>Factor Freq Factor | PBA RANGE<br>Fred Factor | RANGE<br>Factor |          | PUDA R    | NUGE<br>Actor      | TEAD<br>Freq | RANGE<br>Factor | LINDA<br>FREG   | RANGE<br>Factor |
|------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------|----------------------------------------------------------------------------------|-------------------------------------------------------------------------|----------------------------------------------------------------|--------------------------------------------------|-----------------------------------------|---------------------------------------|--------------------------|-----------------|----------|-----------|--------------------|--------------|-----------------|-----------------|-----------------|
| , 6.0E-10 2.0E+01 N/A N/A N/A N/A                                                                          | 0 2.0E+01 N/A N/A N/A N/A                                                         | - N/A N/A N/A                                                                    | N/A N/A                                                                 | - N/A N/A N/A                                                  | N/A                                              |                                         | <br>N/A                               | A/N                      | •               | !        | A/A       | ł                  | 1.66-10      | 2.0E+01         | N/A             | •               |
| 5 6.0E-10 2.0E+01 N/A N/A N/A N/A                                                                          | 0.2.0E+01 N/A N/A N/A N/A                                                         | N/A N/A N/A N/A                                                                  | N/A N/A N/A                                                             | N/A N/A N/A                                                    | N/A N/A                                          | N/A N/A                                 | N/A                                   | N/A                      |                 | ł        | 4.7E-09 2 | 0E+01              | 1.6E-10      | 2.0€+01         | N/A             | 1               |
| G N/A N/A N/A N/A N/A                                                                                      | N/A N/A N/A N/A                                                                   | N/A N/A N/A N/A                                                                  | N/A N/A N/A                                                             | N/A N/A N/A                                                    | N/A N/A                                          | N/A N/A                                 | N/A                                   | N/A                      |                 | ;        | N/A       | :                  | 1.6E-10      | 2.0€+01         | N/A             | :               |
| 3 6.0E-10 2.0E+01 7.2E-08 2.0E+01 N/A N/A 1.1E-09 2.                                                       | 0 2.0E+U1 7.2E-U8 2.0E+U1 N/A N/A 1.1E-U9 2.                                      | 7.2E-08 2.0E+01 N/A N/A 1.1E-09 2.                                               | 2.0E+01 N/A N/A 1.1E-09 2.                                              | N/A N/A 1.1E-09 2.                                             | N/A 1.1E-09 2.                                   | N/A 1.1E-09 2.                          | 1.1E-09 2.                            | 1. IE-09 2.              | цч.             | 0E +01   | N/A       | ļ                  | 1.6E-10      | 2.0E+01         | 1.0E-09         | 2.0E+0          |
| 5 6.0E-10 2.0E+01 N/A N/A 5.0E-10 2.0E+01 N/A                                                              | 0 2.0E+01 N/A N/A 5.0E-10 2.0E+01 N/A                                             | N/A N/A 5.0E-10 2.0E+01 N/A                                                      | N/A 5.0E-10 2.0E+01 N/A                                                 | N/A 5.0E-10 2.0E+01 N/A                                        | 5.0E-10 2.0E+01 N/A                              | 5.0E-10 2.0E+01 N/A                     | 2.0E+01 N/A                           | N/A                      |                 | ł        | N/A       | ł                  | 1.6E-10      | 2.0E+01         | N/A             | ł               |
| 5 6.0E-10 2.0E+01 N/A N/A N/A 1.1E-09 2                                                                    | .0 2.0E+01 N/A N/A 1.1E-09 2                                                      | N/A N/A N/A 1.1E-09 2                                                            | N/A N/A 1.1E-09 2                                                       | N/A N/A 1.1E-09 2                                              | N/A 1.1E-09 2                                    | N/A 1.1E-09 2                           | 1.1E-09 2                             | 1.1E-09                  |                 | 2. 0E+01 | N/A       | ł                  | 1.6E-10      | 2.0E+01         | 1.0E-09         | 2.0E+0          |
| 5 6.0E-10 2.0E+01 N/A N/A N/A N/A                                                                          | .0.2.0E+01 N/A N/A N/A N/A                                                        | N/A N/A N/A N/A                                                                  | N/A N/A N/A                                                             | N/A N/A N/A                                                    | N/A N/A                                          | N/A N/A                                 | N/A                                   | N/A                      |                 | 1        | N/A       | ;                  | 1.6E-10      | 2.0E+01         | 1.0E-09         | 2.0€+0          |
| ) 6.0E-10 2.0E+01 N/A 2.7E-10 2.0E+01 N/A N/A                                                              | 0 2.0E+01 N/A 2.7E-10 2.0E+01 N/A N/A                                             | N/A 2.7E-10 2.0E+01 N/A N/A                                                      | 2.7E-10 2.0E+01 N/A N/A                                                 | 2.7E-10 2.0E+01 N/A N/A                                        | 2.0E+01 N/A N/A                                  | N/A N/A                                 | N/A                                   | N/A                      |                 | ;        | 4.76-09 2 | 10+30              | 1.6E-10      | 2.0E+01         | N/A             | :               |
| ) 6.0E-10 2.0E+01 N/A 2.7E-10 2.0E+01 N/A N/A                                                              | 0 2.0E+01 N/A 2.7E-10 2.0E+01 N/A N/A                                             | M/A 2.7E-10 2.0E+01 M/A N/A                                                      | 2.7E-10 2.0E+01 N/A N/A                                                 | 2.7E-10 2.0E+01 N/A N/A                                        | 2.0E+01 N/A N/A                                  | N/A N/A                                 | N/A                                   | N/A                      |                 | 1        | N/A       | ł                  | 1.66-10      | 2.06+01         | 1.0E-09         | 2.0E+0          |
| 5 6.0E-10 2.0E+01 N/A 2.7E-10 2.0E+01 N/A N/A                                                              | 0 2.0E+01 N/A 2.7E-10 2.0E+01 N/A N/A                                             | N/A 2.7E-10 2.0E+01 N/A N/A                                                      | 2.7E-10 2.0E+01 N/A N/A                                                 | 2.7E-10 2.0E+01 N/A N/A                                        | 2.0E+01 N/A N/A                                  | NIA NIA                                 | N/A                                   | <b>N/A</b>               |                 | ł        | N/A       | ;                  | 1.66-10      | 2.0€+01         | 1.0E-09         | 2.0E+0          |
| N. N.A N/A N/A N/A N/A                                                                                     | N/A N/A N/A N/A                                                                   | N/A N/A N/A N/A                                                                  | N/A N/A N/A                                                             | M/A N/A N/A                                                    | N/A N/A                                          | N/A N/A                                 | N/A                                   | N/N                      |                 | ł        | N/A       | ł                  | 1.6E-10      | 2.06+01         | 1.06-09         | 2.0E+01         |
| 5 6.0E-10 2.0E+01 N/A 2.7E-10 2.0E+01 N/A 1.1E-                                                            | 0 2.0E+01 N/A 2.7E-10 2.0E+01 N/A 1.1E-                                           | N/A 2.7E-10 2.0E+01 N/A 1.1E-                                                    | 2.7E-10 2.0E+01 M/A 1.1E-                                               | 2.7E-10 2.0E+01 N/A 1.1E-                                      | 2.0E+01 N/A 1.1E-                                | N/A 1.1E-                               | 1.IE-                                 | 1.16-                    | 5               | 2.0E+01  | N/A -     | ł                  | 1.6E-10      | 2.0E+01         | 1.05-09         | 2.0E+01         |
| ) 6.0E-10 2.0E+01 N/A 2.7E-10 2.0E+01 N/A 1.1E-                                                            | 0.0.00401 N/A 2.76-10 2.06401 N/A 1.16-                                           | N/A 2.7E-10 2.0E+01 N/A 1.1E-                                                    | 2.7E-10 2.0E+01 N/A 1.1E-                                               | 2.7E-10 2.0E+01 N/A 1.1E-                                      | 2.0E+01 N/A 1.1E-                                | N/A 1.1E-                               | 1.IF                                  | 1. IF                    | 6               | 2.0E+01  | N/A       | ;                  | 1.6E-10      | 2.0E+01         | 1.0E-09         | 2.0E+Ù1         |
| N/A N/A N/A N/A N/A                                                                                        | N/A N/A N/A N/A                                                                   | N/A N/A N/A N/                                                                   | N/A N/A N/                                                              | N/A N/A N/                                                     | N/A N/                                           | N/A N/                                  |                                       | N                        | æ               | •        | N/A       | ł                  | 1.6E-10      | 2.0E+01         | 1.0E-09         | 2.0E+01         |
| 7 N/A N/A N/A N/A N/                                                                                       | N/A N/A N/A N/                                                                    | N/A N/A N/A N/                                                                   | N/A N/A N/                                                              | N/A N/A N/                                                     | N/A N/                                           | N/A N/                                  | N                                     | Ì                        | æ               | 1        | N/A       | :                  | 1. JE-10     | 2.0E+01         | <b>B.</b> 5E-10 | 2.0E+01         |
| r 4.9E-10 2.0E+01 N/A N/A N/A N/                                                                           | 0.2.0E+01 N/A N/A N/A N/A                                                         | N/A N/A N/A N/A                                                                  | N/A N/A N/                                                              | N/A N/A N/                                                     | N/A N/                                           | N/A N/                                  | N                                     | N                        | æ               | 1        | 3.96-09 2 | 10+30.             | 1.3E-10      | 2.0E+01         | N/A             | ;               |
| r 4.9E-10 2.0E+01 N/A N/A N/A N/                                                                           | .0 2.0E+01 N/A N/A N/A N/                                                         | N/A N/A N/A N/A                                                                  | N/A N/A N/                                                              | N/A N/A N/                                                     | N/A N/                                           | N/A N/                                  | /N                                    | ¥                        | æ               | !        | N/A       | ł                  | 1.3E-10      | 2.0E+01         | N/A             | 1               |
| 7 4.9E-10 2.0E+01 N/A N/A N/A N/                                                                           | (0.2.0E+U1 N/A N/A N/A N/A                                                        | N/A N/A N/A N/                                                                   | N/A N/A N/                                                              | N/A N/A N/                                                     | N/A N/                                           | N/A N/                                  | <b>N</b>                              | Ì                        | æ               | ł        | 3.9E-09 2 | 10+30 <sup>.</sup> | 1.3E-10      | 2.0E+01         | N/A             | ł               |
| NIA NIA NIA - NIA - NIA NI                                                                                 | N/A N/A - N/A N/A                                                                 | N/A N/A - N/A N/                                                                 | N/A - N/A N/                                                            | N/A N/A N/                                                     | - N/A N/                                         | N/A N/                                  | N                                     | X                        | æ               | 1        | N/A       | :                  | 1.3E-10      | 2.0€+01         | N/A             | ł               |
| 7 4.9E-10 2.0E+01 5.9E-08 2.0E+01 M/A 9.1E                                                                 | 0 2.0E+01 5.9E-08 2.0E+01 N/A 9.1E                                                | 5.9E-08 2.0E+01 M/A N/A 9.1E                                                     | 2.0E+01 M/A N/A 9.1E                                                    | N/A N/A 9.1E                                                   | N/A 9.1E                                         | N/A 9.1E                                | 9.1E                                  | 9. JE                    | -10             | 2.0E+01  | N/A       | 1                  | 1.36-10      | 2.0E+01         | 8.5E-10         | 2.0E+01         |
| 7 4.9E-10 2.0E+01 N/A N/A 4.1E-10 2.0E+01 N/                                                               | 0 2.0E+01 N/A N/A 4.1E-10 2.0E+01 N/                                              | N/A N/A 4.1E-10 2.0E+01 N/                                                       | N/A 4.1E-10 2.0E+01 N/                                                  | N/A 4.1E-10 2.0E+01 N/                                         | 4.1E-10 2.0E+01 N/                               | 4.1E-10 2.0E+01 N/                      | 2.0E+01 N/                            | ¥                        | æ               | :        | N/A       | ;                  | 1.3E-10      | 2.0E+01         | N/A             | 1               |
| * 4.9E-10 2.0E+01 M/A N/A 9.1E                                                                             | 0.2.0E+01 N/A N/A N/A 9.1E                                                        | N/A N/A N/A 9.1E                                                                 | N/A N/A 9.1E                                                            | N/A N/A 9.1E                                                   | N/A 9.1E                                         | N/A 9.1E                                | 9.1E                                  | 9.1E                     | 9-              | 2.06.01  | N/A       | 1                  | 1. JE-10     | 2.0E+01         | 8.5E-10         | 2.0E+01         |
| 7 4.9E-10 2.0E+01 M/A N/A N/A                                                                              | .0.2.0E+01 N/A N/A N/A N/                                                         | N/A N/A N/A N/                                                                   | N/A N/A N/                                                              | N/A N/A N/                                                     | N/A N/                                           | N/A N/                                  | NV                                    | N                        | æ               | 1        | N/A       | !                  | 1.3E-10      | 2.06+01         | 8.5E-10         | 2.0E+01         |
| 7 4.9E-10 2.0E+01 N/A 2.3E-10 2.0E+01 N/A N/                                                               | .0 2.0E+01 N/A 2.3E-10 2.0E+01 N/A N/                                             | N/A 2.3E-10 2.0E+01 N/A N/                                                       | 2.3E-10 2.0E+01 N/A N/                                                  | 2.3E-10 2.0E+01 N/A N/                                         | 2.0E+01 N/A N/                                   | N/A N/                                  | N/                                    | N                        | æ               | ł        | 3.9E-09 2 | 0E+01              | 1.3E-10      | 2.0E+01         | N/A             | !               |
| 7 4.9E-10 2.0E+01 N/A 2.3E-10 2.0E+01 N/A N/                                                               | 10 2.0E+01 N/A 2.3E-10 2.0E+01 N/A N/                                             | N/A 2.3E-10 2.0E+01 N/A N/                                                       | 2.3E-10 2.0E+01 N/A N/                                                  | 2.3E-10 2.0E+01 N/A N/                                         | 2.0E+01 N/A N/                                   | N/A N/                                  | N                                     | Ň                        | æ               | ;        | N/A       | ;                  | 1.3E-10      | 2.0E+01         | <b>B.5E-1</b> 0 | 2.0E+01         |
| 7 4.9E-10 2.0E+01 N/A 2.3E-10 2.0E+01 N/A N/                                                               | .0 2.0E+01 N/A 2.3E-10 2.0E+01 N/A N/                                             | N/A 2.3E-10 2.0E+01 N/A N/                                                       | 2.3E-10 2.0E+01 N/A N/                                                  | 2.3E-10 2.0E+01 N/A N/                                         | 2.0E+01 N/A N/                                   | N/A N/                                  | /N                                    | N                        | ¥               | ;        | N/A       | ł                  | 1. JE-10     | 2.0E+01         | 8.5E-10         | 2.0E+01         |
| N/A N/A N/A N/A N/A                                                                                        | N/A N/A N/A N/A                                                                   | N/A N/A N/A N/A                                                                  | N/A N/A N/                                                              | N/A N/A N/                                                     | VI N/A N/                                        | N/A N/                                  | N/                                    | N                        | æ               | ł        | N/A       | ł                  | 1.3E-10      | 2.0E+01         | 8.56-10         | 2.0E+01         |
| r 4.9E-10 2.9E+01 N/A 2.3E-10 2.0E+01 N/A 9.1E                                                             | 0 2.0E+01 N/A 2.3E-10 2.0E+01 N/A 9.1E                                            | N/A 2.3E-10 2.0E+01 N/A 9.1E                                                     | 2.3E-10 2.0E+01 N/Å 9.1E                                                | 2.3E-10 2.0E+01 N/A 9.1E                                       | 2.0E+01 N/Å 9.1E                                 | N/A 9.1E                                | 9.16                                  | 9.16                     | 9-              | 2.06+01  | N/A       | 1                  | 1.3E-10      | 2.0E+01         | 8.56-10         | 2.0E+01         |
| r 4.9E-10 2.0E+01 N/A 2.3E-10 2.0E+01 N/A 9.1                                                              | 0.2.0E+01 N/A 2.3E-10 2.0E+01 N/A 9.1                                             | N/A 2.3E-10 2.0E+01 N/A 9.1                                                      | 2.3E-10 2.0E+01 N/A 9.1                                                 | 2.3E-10 2.0E+01 N/A 9.1                                        | 2.0E+01 N/A 9.1                                  | N/A 9.1                                 | 9.1                                   | 9.1                      | E-10            | 2.0E+01  | N/A       | ;                  | 1.3E-10      | 2.0E+01         | 8.56-10         | 2.0E+01         |

See notes at end of table.

Ο,



a "h\_\_ Cayser







File: ONSITREG.WK1 Page 3 Date 19-Aug-B7

TABLE 8-16 (Continued)

# ONSITE TRANSPORTATION - REGIONAL AND NATIONAL DISPOSAL OPTIONS

Scenario Frequencies and Range Factors

|   | PACKAGE   |
|---|-----------|
|   | OFFSITE   |
|   | =         |
|   | RAIL      |
|   | FROM      |
| 5 | AND       |
| 5 | 2         |
|   | (HOVENENT |
|   |           |

|        |          |              |                 |             |                 |              | •                     |              |                 |             |                 |              |                 |              | ;               |              |                 |
|--------|----------|--------------|-----------------|-------------|-----------------|--------------|-----------------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|-----------------|
| SCEN-  | <b>%</b> | ANAD<br>Freg | RANGE<br>Factor | APG<br>Free | RANGE<br>FACTOR | LBAD<br>FREQ | RANGE<br>Factor       | INAP<br>FREQ | RANGE<br>Factor | PBA<br>Freq | RANGE<br>Factor | PUDA<br>Freq | RANGE<br>Factor | TEAD<br>Free | RANGE<br>Factor | UNDA<br>Fred | RANGE<br>Factor |
|        | :        |              |                 |             |                 |              | ł<br>1<br>8<br>8<br>1 |              |                 | -           |                 |              |                 |              |                 |              |                 |
| WISUF  | 1        | N/A          | 1               | N/A         | ł               | N/A          | ;                     | N/A          | ;               | N/A         | ;               | N/A          | 1               | 1.36-10      | 2.0€+01         | 8.5E-10      | 2.0E+0          |
| VRIBES | Π        | N/A          | :               | N/A         | 1               | N/A          | ;                     | N/A          | 1               | N/A         | ;               | N/A          | ł               | 2.06-07      | 1.1E+01         | 1.25-08      | 1. IE+0         |
| VRDHS  | 11       | 1.26-08      | 1.1E+01         | N/A         | !               | N/A          | ł                     | N/A          | 1               | N/A         | !               | 1.26-08      | 1.1E+01         | 2.06-07      | 1.1£+01         | N/A          | 1               |
| VRCGS  | Ξ        | 1.26-08      | 1.1E+01         | A/N         | :               | A/A          | :                     | N/A          | :               | N/A         | ;               | N/A          | :               | 2.06-07      | 1.1E+01         | N/A          | :               |
| VACHS  | 11       | 1.25-08      | 1.11.101        | N/A         | ł               | N/A          | ł                     | N/A          | 1               | N/N         | ;               | 1.26-08      | 10+31.1         | 2.0E-07      | 1.1E+01         | M/A          | :               |
| VRKGS  | 11       | N/A          | :               | N/A         | ;               | N/A          | ł                     | N/A          | 1               | N/A         | ;               | N/A          | ł               | 2.06-07      | 1.16+01         | N/A          | :               |
| VRKHS  | Π        | 1.2E-08      | 1.1E+01         | 1.26-08     | 1.1E+01         | N/A          | ;                     | N/A          | {               | 1.25-08     | 1.1E+01         | M/A          | :               | 2.06-07      | 1.16+01         | 1.25-08      | 1. IE+6         |
| VRKVS  | Ξ        | 1.26-08      | 1.1E+01         | N/A         | :               | N/A          | :                     | 4.1E-08      | 1.16+01         | N/A         | 1               | N/A          | :               | 2.0E-07      | 1.1E+01         | N/A          | ł               |
| VRINVS | Π        | 1.26-08      | l. IE+01        | N/N         | :               | N/A          | :                     | N/A          | ł               | 1. 25-08    | 1.1E+01         | N/A          | ;               | 2.0E-07      | 1.16+01         | 1.2E-08      | 1.16+6          |
| VRP6S  | II       | 9.96-98      | 1.1E+01         | N/A         | :               | N/A          | !                     | N/A          | 1               | N/A         | :               | N/A          | ł               | 2.0E-07      | 1.1E+01         | 1.25-08      | 1.1E+0          |
| VRPHS  | Η        | 9.06-08      | 1. IE+01        | N/A         | ;               | 9.96-08      | 1.1E+01               | N/A          | ;               | N/A         | 1               | 1.2E-08      | 1.16+01         | 2.06-07      | 1.16+01         | N/A          | ;               |
| VAPVS  | Ξ        | 9.96-38.9    | 1.15+01         | N/A         | !               | 9.86-08      | 1.1€+01               | N/A          | 1               | N/A         | ;               | K/A -        | :               | 2.0E-07      | 1. IE+01        | 1.26-08      | 1.16+0          |
| VRDES  | Ξ        | 9.86-08      | 1.1E+01         | N/N         | ;               | 9.86-08      | 1.1E+01               | N/A          | ł               | N/N         | ;               | N/A          | 1               | 2.05-07      | 1.1E+01         | 1.26-08      | 1.15+0          |
| VREVS  | Ξ        | N/A          | ł               | N/N         | ł               | N/A          | ł                     | A/A          | ;               | N/N         | 1               | N/A          | ;               | 2.06-07      | 1.1E+01         | 1.26-08      | 1.16+0          |
| VRRGS  | 11       | 1.25-08      | 1.1E+01         | N/A         | 1               | 1.2E-08      | 1.1E+01               | N/A          | ł               | 1.25-08     | 1.1E+01         | N/A          | ;               | 2.0E-07      | 1.1E+01         | 1.25-09      | I. IE+0         |
| VRRVS  | Ξ        | 1.25-08      | 1.1E+01         | N/N         | ł               | 1.2E-08      | 1.1E+01               | N/A          | ł               | 1.26-08     | 1.16+01         | N/A          | :               | 2.0E-07      | 1.16+01         | 1.2E-08      | 1.1E+0          |
| VRSVS  | 11       | N/A          | :               | N/A         | ;               | N/A          | ł                     | N/A          | ;               | N/N         | ;               | N/A          | 1               | 2.0E-07      | 1.1E+01         | 1.26-08      | 1. IE+0         |
| VRDHC  | 12       | 0.0E+00      | ł               | N/A         | 1               | N/A          | ł                     | N/A          | 1               | N/A         | ł               | 0.0E+00      | ł               | 0.0E+00      | ł               | N/A          | ł               |
| VRCGC  | 12       | 0.0E+00      | 1               | N/A         | 1               | N/A          | 1                     | N/A          | ;               | N/N         | ;               | N/A          | 1               | 0.0E+00      | :               | N/A          | ł               |
| VRCHC  | 2        | 0.0E+00      | 1               | N/A         | :               | N/A          | 1                     | A/A          | ;               | N/A         | ;               | 0.0E+00      | 1               | 0.0€+00      | ;               | N/A          | 1               |
| VRINC  | 12       | 0.05+00      | 1               | N/A         | ;               | N/A          | ;                     | N/A          | ;               | 0.0E+00     | 1               | N/A          | 1               | 0.0E+00      | ł               | 0.0E+00      | ł               |
| VRPGC  | 12       | 0.0E+00      | ;               | N/A         | ;               | N/A          | 1                     | N/A          | !               | N/A         | 1               | A/A          | ;               | 0.0E+00      | :               | 0.0E+00      | ł               |
| VRPHC  | 21       | 0.06+00      | 1               | N/A         | !               | 0. 0E+00     | ;                     | N/A          | !               | N/A         | 1               | 0.0E+00      | ł               | 0.0E+00      | :               | N/A          | :               |
| VRPVC  | 12       | 0.05+00      | ł               | N/A         | ł               | 0.0E+0i)     | 1                     | N/A          | 1               | N/A         | ţ               | N/A          | ł               | 0.0E+00      | ;               | 0.0E+00      | 1               |
| VROGC  | 12       | 0.0E+00      | ł               | N/A         | ;               | 0.0E+00      | :                     | N/A          | ;               | N/A         | ;               | N/A          | ł               | 0.0E+00      | ;               | 0.0E+00      | 1               |
| VROVC  | 2        | N/A          | ;               | N/A         | !               | N/A          | :                     | N/A          | !               | N/A         | ł               | N/A          | 1               | 0.0E+00      | :               | 0.0€+00      | :               |
| VRRGC  | 12       | 9.6E-09      | 2.0E+01         | N/A         | ;               | 9.6E-09      | 2.0E+01               | N/A          | ł               | 9.65-09     | 2.0E+01         | N/A          | 1               | 1.65-07      | 1.66+01         | 9.46-09      | 2.06+0          |
| VRRVC  | 2        | 9.6E-09      | 2. uE+ù1        | N/A         | ;               | 9.6E-09      | 2.0E+01               | N/A          | ł               | 9.6E-09     | 2.0E+01         | N/A          | ;               | 1.66-07      | 1.6E+01         | 9.65-09      | 2.0E+0          |
| VRBGÜ  | []       | N/A          | ;               | N/A         | !               | N/A          | :                     | N/A          | ;               | N/A         | :               | N/A          | 1               | 0.0E+00      | :               | 0.0€+00      | :               |

1

See notes at end or table.

File: ONSITRE6. UKI Fage 4 Date 19-Aug-87 TABLE

62.25

TABLE 8-16 (Continued)

# QNSITE TRANSPORTATION - REGIONAL AND WATIONAL DISPOSAL OPTIONS (novement to and from rail in offsite package)

| Factors     |
|-------------|
| Range       |
| and         |
| Frequencies |
| Scenario    |

| SCEN-<br>Ario | No. | ANAD<br>Freq | RANGE<br>FACTOR | AP6<br>FREQ | RANGE<br>Factor | LBAD<br>Freq | RANGE<br>Factor | NAAP<br>FREQ | RANGE<br>Factor | PBA<br>Freg | RANGE<br>Factor | PUDA F    | RANGE<br>Factor | TEAD<br>FREQ | RANGE<br>Factor | UMDA<br>Freq | RANGE<br>Factor |
|---------------|-----|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|-------------|-----------------|-----------|-----------------|--------------|-----------------|--------------|-----------------|
|               | •   |              |                 |             |                 |              |                 |              |                 |             |                 |           |                 |              |                 |              |                 |
| VRK6C         | 13  | N/A          | 1               | N/A         | !               | N/A          | ;               | N/A          | 1               | N/A         | ł               | N/A       | ;               | 0.06+00      | ;               | M/A          | 1               |
| VRXHC         | 1   | 0.0E+00      | :               | 0.0E+00     | :               | N/A          | :               | N/A          | ł               | 0.0E+00     | :               | N/A       | :               | 0.0E+00      | ;               | 0.0€+00      | ł               |
| VRKVC         | 13  | 0.0E+00      | :               | N/A         | ł               | N/A          | !               | 0.0E+00      | 1               | N/A         | ł               | N/A       | ł               | 0.0E+00      | ;               | N/A          | 1               |
| VRSVC         | 13  | N/A          | ł               | N/A         | ;               | N/A          | ;               | N/A          | :               | N/A         | :               | N/A       | ł               | 0.0E+00      | :               | 0.0E+00      | 1               |
| VRBGC         | =   | N/A          | :               | N/A         | ł               | N/A          | :               | N/A          | :               | N/A         | ;               | N/A       | !               | 6.85-11      | 2.56+01         | 6.85-11      | 2.5E+0          |
| VRDHC         | 1   | 0.0E+00      | ł               | N/A         | :               | M/A          | ;               | N/A          | ł               | N/A         | :               | 1.1E-09   | 2.4E+01         | 6.86-11      | 2.5E+01         | N/A          | 1               |
| VRCGC         | 1   | 0.0E+00      | :               | N/A         | !               | N/A          | ł               | N/A          | :               | N/A         | ł               | N/A       | ;               | 6.86-11      | 2.5E+01         | N/A          | ١.              |
| VRCHC         | 1   | 2.16-08      | 2.56+01         | N/A         | :               | N/A          | :               | N/A          | :               | N/A         | :               | 1.1E-09   | 2.4E+01         | 6.8E-11      | 2.56+01         | M/A          | ł               |
| VRKEC         | 1   | N/A          | ;               | N/A         | ł               | N/A          | :               | N/A          | !               | N/A         | :               | N/A       | 1               | 6.85-11      | 2.5E+01         | N/A          | ;               |
| VRKHC         | 2   | 2.1E-CB      | 2.56+01         | 1.1E-Ù9     | 2.4E+01         | N/A          | ;               | N/A          | ;               | M/A         | :               | N/A       | ;               | 6.8E-11      | 2.5E+01         | 6.85-11      | 2.5E+0          |
| VRKVC         | 1   | 2.1E-08      | 2.5E+01         | N/A         | ;               | N/A          | :               | 2.1E-08      | 2.5E+01         | N/A         | !               | N/A       | :               | 6.85-11      | 2.5E+01         | A/A          | :               |
| VRRVC         | =   | 2.1E-08      | 2.56+01         | N/A         | 1               | N/A          | :               | N/A          | 1               | 2.1E-08     | 2.5E+01         | N/A -     | ł               | 6.85-11      | 2.5E+01         | 6.8E-11      | 2.5E+0          |
| VRPGC         | =   | 2.1E-08      | 2.5E+01         | N/A         | :               | N/A          | ;               | N/A          | :               | N/A         | :               | N/A       | :               | 6.8E-11      | 2.5E+01         | 6.8E-11      | 2.5E+0          |
| VRFHC         | =   | 2.1E-08      | 2.56+01         | N/A         | :               | 2.16-08      | 2.5E+01         | N/A          | ;               | N/A         | :               | 1.1E-09   | 2.4E+01         | 6.9E-11      | 2.5E+01         | N/A          | :               |
| VRPVC         | 1   | 2.1E-08      | 2.5E+01         | N/A         | ł               | 2.1E-08      | 2.5E+01         | N/A          | ;               | N/A         | ł               | N/A       | !               | 6.85-11      | 2.5E+01         | 6.8E-11      | 2.5E+0          |
| VRPEC         | 2   | 2.1E-08      | 2.5E+01         | A/N         | ţ               | 2.1E-08      | 2.5E+01         | N/A          | ١.              | N/A         | !               | N/A       | :               | 6.BE-11      | 2. SE+01        | 6.85-11      | 2.5E+0          |
| VRQVC         | 2   | N/A          | ;               | N/A         | ;               | N/A          | :               | N/A          | :               | N/A         | ;               | N/A       | ł               | 6.86-11      | 2.5E+01         | 6.8E-11      | 2.5E+0          |
| VRRGC         | 2   | 2.1E-08      | 2.5E+01         | N/A         | ł               | 2.1E-08      | 2.5E+01         | N/A          | 1               | 2.1E-08     | 2.5E+01         | N/A       | ł               | 6.86-11      | 2.5E+01         | 6.BE-11      | 2.5E+0          |
| VRRVC         | 1   | 2.1E-08      | 2.5E+01         | N/A         | ;               | 2.1E-08      | 2.5E+01         | N/A          | 1               | 2. IE-08    | 2.5E+01         | N/A       | ;               | 6.85-11      | 2.5E+01         | 6.8E-11      | 2.56+0          |
| VRSVC         | 2   | N/A          | 1               | N/A         | ;               | N/A          | 1               | N/A          | ŀ               | N/A         | ;               | N/A       | 1               | 6.8E-11      | 2.56+01         | 6.85-11      | 2.5E+0          |
| VRDHC         | 5   | 3.7E-10      | 5.5E+01         | N/A         | 1               | N/A          | ł               | N/A          | 1               | N/A         | :               | 2.6E-10   | 10+36*1         | 2.2E-09      | 5.8£+01         | N/A          | 1               |
| VRCGC         | 5   | 3.7E-10      | 5.5E+01         | N/A         | ;               | N/A          | :               | N/A          | ł               | N/A         | ;               | N/A       | ;               | 2.2E-09      | 5.86+01         | N/A          | ł               |
| VRCHC         | 5   | 3.7E-10      | 5.56+01         | N/A         | ;               | N/A          | 1               | N/A          | ł               | N/A         | ;               | 2.66-10 4 | 1.96+01         | 2.2E-09      | 5.86+01         | N/A          | ;               |
| VRMVC         | 5   | 3.7E-10      | 5.56+01         | N/A         | ł               | N/A          | ł               | N/A          | ł               | 3.7E-10     | 5.5E+01         | N/A       | ;               | 2.2E-09      | 5.8E+01         | 1.4E-10      | 5.8E+0          |
| VRPGC         | 5   | 3.7E-10      | 5.56+01         | N/A         | l<br>I          | N/A          | ł               | N/A          | ł               | N/A         | ł               | N/A       | 1               | 2.2E-09      | 5.86+01         | 1.4E-10      | 5.8E+0          |
| VRPHC         | 5   | 3.7E-10      | 5.5E+01         | N/A         | ł               | 3.7E-10      | 5.5E+01         | N/A          | 1               | N/A         | 1               | 2.66-10 4 | 10+36.1         | 2.2E-09      | 5.8E+01         | N/A          | 1               |
| VRPVC         | 5   | 3.7E-10      | 5.56+01         | N/A         | ł               | 3.7E-10      | 5.56+01         | N/A          | 1               | N/A         | ł               | N/A       | ;               | 2.2E-09      | 5.8E+01         | 1.4E-10      | 5.86+0          |
| VRGGC         | 5   | 3.7E-10      | 5.5E+01         | N/A         | ł               | 3.7E-10      | 5.5E+01         | N/A          | 1               | N/A         | 1               | N/A       | 1               | 2.2E-09      | 5. BE+01        | 1.46-10      | 5. BE+0         |
| VRQVC         | 51  | N/A          | !               | N/A         | :               | N/A          | ł               | N/A          | 1               | N/A         | ;               | N/A       | ;               | 2.2E-09      | 5.8E+01         | 1.46-10      | 5.8E+0          |

See notes at end of table.

19.2.2.S

REAL DESCRIPTION DESCRIPTION

yrky € €







с.,

ուտ

TABLE 8-16 (Continued) File: ONSITRE6.NK1 Fage 5 Date 19-Aug-87

# DUSITE TRANSPORTATION - REGIONAL AND MATIONAL DISPOSAL OPTIONS (MOVEMENT TO AND FROM RAIL IN OFFSITE PACKAGE)

Scenario Frequ

| actor  |  |
|--------|--|
| y abu  |  |
| nd Rai |  |
| les al |  |
| enc 1  |  |

| RANGE<br>Factor |  | 0 5.BE+01 | 0 5.8E+01 |
|-----------------|--|-----------|-----------|
| UNDA<br>Freg    |  | 1-16-1    | 1.46-1    |
| RANGE<br>Factor |  | 5.8E+01   | 5.8E+01   |
| TEAD<br>FREQ    |  | 2.2E-09   | 2. 2E-09  |
| RANGE<br>Factor |  | ;         | ł         |
| PUDA<br>Freq    |  | N/A       | N/A       |
| rânge<br>Factor |  | ) 5.5E+01 | ) 5.5E+0I |
| PBA<br>Freq     |  | 3.7E-11   | 3.76-11   |
| RANGE<br>Factor |  | 1         | ł         |
| NAAP<br>FREQ    |  | N/A       | N/A       |
| RANGE<br>Factor |  | 5.5E+01   | 5.56+01   |
| LBAD<br>Freg    |  | 3.7E-10   | 3.7E-10   |
| RAMGE<br>Factor |  | ;         | ł         |
| AP6<br>Freq     |  | N/A       | N/A       |
| RANGE<br>Factor |  | 10+35.5   | j 5.5E+Ut |
| ANAD<br>FREQ    |  | 3.7E-16   | 3.76-10   |
| No.             |  | 15        | 5         |
| SCEN-<br>Ario   |  | VRRBC     | VRRVC     |

NUTES: 1. Scenarios 1-5 are per truck mile; scenarios 6-15 are per exposure year. 2. Duration time shown for scenarios with agent releases due to both detonations and spills is for spills only. Duration time for detonation is instantaneous. 3. National Disposal Option VRXVS ANAD is N/A.



File: ONSITBRG.WK1 Page 1 Date 19-Aug-87

#### ONSITE TRANSPORTATION - AIR

A DADAMY

**n**.:

ና ፈንባ እንዲዝ ብለ የ

#### Scenario Frequencies and Range Factors

| CCENADIO | - | AP6     | RANGE   |         | RANGE   | TEAD    | RANGE   |
|----------|---|---------|---------|---------|---------|---------|---------|
|          |   |         |         |         |         |         |         |
|          |   |         |         |         |         |         |         |
| VAKHS    | 1 | 0.0E+00 |         | N/A     |         | 0.0E+00 |         |
| VAPHS    | 1 | N/A     |         | 0.0E+00 |         | 0.0E+00 |         |
| VAPVS    | 1 | N/A     |         | 0.0E+00 |         | 0.0E+00 |         |
| VARGS    | 1 | N/A     |         | 0.0E+00 |         | 0.0E+00 |         |
| VARGS    | 1 | N/A     |         | 0.0E+00 |         | 0.0E+00 |         |
| VARVS    | 1 | N/A     |         | 0.0E+00 |         | 0.0E+00 |         |
| VAKHS    | 2 | 0.0E+00 |         | N/A     |         | 0.0E+00 |         |
| VAPHS    | 2 | N/A     |         | 0.0E+00 |         | 0.0E+00 |         |
| VAPVS    | 2 | N/A     |         | 0.0E+00 |         | 0.0E+00 |         |
| VARGS    | 2 | N/A     |         | 0.0E+00 |         | 0.0E+00 |         |
| VARGS    | 2 | N/A     |         | 0.0E+00 |         | 0.0E+00 |         |
| VARVS    | 2 | N/A     |         | 0.0E+00 |         | 0.0E+00 |         |
| VAKHS    | 2 | 2.8E-10 | 2.2E+01 | N/A     |         | 2.8E-10 | 2.2E+01 |
| VAPHS    | 3 | N/A     |         | 2.8E-10 | 2.2E+01 | 2.8E-10 | 2.2E+01 |
| VAPVS    | 3 | N/A     |         | 2.8E-10 | 2.2E+01 | 2.8E-10 | 2.2E+01 |
| VAQ6S    | 3 | N/A     |         | 2.8E-10 | 2.2E+01 | 2.9E-10 | 2.2E+01 |
| VARGS    | 3 | N/A     |         | 2.8E-10 | 2.2E+01 | 2.8E-10 | 2.2E+01 |
| VARVS    | 3 | N/A     |         | 2.BE-10 | 2.2E+01 | 2.8E-10 | 2.2E+01 |
| VAPHC    | 4 | N/A     |         | 3.0E-12 | 2.6E+01 | 3.0E-12 | 2.5E+01 |
| VAPVC    | 4 | N/A     |         | 3.0E-12 | 2.6E+01 | 3.0E-12 | 2.6E+01 |
| VAREC    | 4 | N/A     |         | 3.0E-12 | 2.6E+01 | 3.0E-12 | 2.6E+01 |
| VARGC    | 4 | N/A     |         | 2.2E-10 | 2.6E+01 | 2.2E-10 | 2.6E+01 |
| VARVC    | 4 | N/A     |         | 2.2E-10 | 2.6E+01 | 2.2E-10 | 2.6E+01 |
| VAKHF    | 5 | 0.0E+00 |         | N/A     |         | 0.0E+00 |         |
| VAKHS    | 6 | 7.2E-08 | 2.0E+01 | N/A     | •-      | 4.8E-09 | 2.0E+01 |
| VAPHC    | 6 | N/A     |         | 2.7E-10 | 2.0E+01 | 4.8E-09 | 2.0E+01 |
| VAPVC    | 6 | N/A     |         | 2.7E-10 | 2.0E+01 | 4.8E-09 | 2.0E+01 |
| VAQ6C    | 6 | N/A     |         | 2.7E-10 | 2.0E+01 | 4.8E-09 | 2.0E+01 |
| VARGC    | 6 | N/A     |         | 2.7E-10 | 2.0E+01 | 4.8E-09 | 2.0E+01 |





File: ONSITBR6.WK1 Page 2 Date 19-Aug-87

#### ONSITE TRANSPORTATION - AIR

Scenario Frequencies and Range Factors

|          |     | AP6     | RANGE   | LBAD     | RANGE   | TEAD    | RANGE    |
|----------|-----|---------|---------|----------|---------|---------|----------|
| SCENARIO | NO. | FRED    | FACTOR  | FRED     | FACTOR  | FRE     | FACTOR   |
|          |     |         |         |          |         |         |          |
| 1140110  |     |         |         | a 75 ia  |         |         | -        |
| VARVL    | 0   | N/A     |         | 2./E-10  | 2.02+01 | 4.85-09 | 2. VE+UI |
| VAKH     | 1   | 2.4F-08 | 2.0E+01 | N/8      |         | 4.0E-09 | 2.0E+01  |
| VAFHC    | 1   | N/A     |         | 2.3E-10  | 2.0E+01 | 4.0E-09 | 2.0E+01  |
| VAPVC    | 1   | N/A     |         | 2.3E-10  | 2.0E+01 | 4.0E-09 | 2.0E+01  |
| VAQGC    | 7   | N/A     |         | 2.3E-10  | 2.0E+01 | 4.0E-09 | 2.0E+01  |
| VARSC    | 7   | N/A     |         | 2.3E-10  | 2.0E+01 | 4.0E-09 | 2.0E+01  |
| VARVC    | 7   | N/A     |         | 2.3E-10  | 2.0E+01 | 4.0E-09 | 2.0E+01  |
| VAKHS    | 9   | 0.0E+00 |         | N/A      |         | 0.0E+00 |          |
| VAPHS    | 9   | N/A     |         | 0.0E+00  |         | 0.0E+00 |          |
| VAPVS    | 9   | N/A     |         | 0.0E+00  |         | 0.0E+00 |          |
| VAQGS    | 9   | N/A     |         | 0.0E+00  |         | 0.0E+00 |          |
| VARGS    | 9   | N/A     |         | 0.0E+00  |         | 0.0E+00 |          |
| VARVS    | 9   | N/A     |         | 0.0E+00  |         | 0.0E+00 | -+       |
| VAKHS    | 10  | 0.0E+00 |         | N/A      |         | 0.0E+00 |          |
| VAPHS    | 10  | N/A     |         | 0.0E+00  |         | 0.0E+00 |          |
| VAPVS    | 10  | N/A     |         | 0.0E+00  |         | 0.0E+00 |          |
| VAQGS    | 10  | N/A     |         | 0.0E+00  |         | 0.0E+00 |          |
| VARSS    | 10  | N/A     |         | 0.0E+00  |         | 0.0E+00 |          |
| VARVS    | 10  | N/A     |         | 0.0E+00  |         | 0.0E+00 |          |
| VAKHS    | 11  | 1.2E-0B | 2.2E+01 | N/A      |         | 2.0E-07 | 2.2E+01  |
| VAPHS    | 11  | N/A     |         | 1.2E-08  | 2.2E+01 | 2.0E-07 | 2.2E+01  |
| VAPVS    | 11  | N/A     |         | 1.2E-08  | 2.2E+01 | 2.0E-07 | 2.2E+01  |
| VARGS    | 11  | N/A.    |         | 1.2E-08  | 2.2E+01 | 2.0E-07 | 2.2E+01  |
| VARGS    | 11  | N/A     |         | 1.25-08  | 2.2E+01 | 2.0E-07 | 2.2E+01  |
| VARVS    | 11  | N/A     |         | 1.75-08  | 2.7E+01 | 2.0F-07 | 2.2E+01  |
| VAPHC    | 12  | N/A     |         | 0.0E+00  |         | 0.05+00 |          |
| VAPUC    | 12  | N/A     |         | 0.05+00  |         | 0 06+00 |          |
| VAGED    | 17  | N/A     |         | 0.05+00  |         | 0.02.00 |          |
| THEUL    | 44  | 21.12   | - *     | A*AF 400 |         | A106-00 |          |





File: GNSITBRG.Wk1 Page 1 Date 19-Aug-87

655544

ayyya addarada araayaya araasaya araacaada addaraa addaraa acceeda araacaada ad

ONSITE TRANSPORTATION - AIR

Scenario Frequencies and Range Factors

|          |     | APS     | RANGE   | LBAD    | RANGE   | TEAD    | RANGE   |
|----------|-----|---------|---------|---------|---------|---------|---------|
| SCENARIO | NC. | FREQ    | FACTOR  | FRED    | FACTOR  | FREQ    | FACTOR  |
|          |     |         |         |         |         |         |         |
| VARVC    | 12  | N/A     |         | 9.6E-09 | 2.0E+01 | 1.6E-07 | 2.0E+01 |
| VAKHF    | 13  | 0.0E+00 |         | N/A     |         | 0.0E+00 |         |
| VAKHC    | 14  | 1.1E-09 | 2.4E+01 | N/A     | ••      | 6.8E-11 | 2.5E+01 |
| VAPHC    | 14  | N/A     |         | 2.1E-08 | 2.5E+01 | 6.8E-11 | 2.5E+01 |
| VAFVC    | 14  | N/A     |         | 2.18-08 | 2.5E+01 | 6.8E-11 | 2.5E+01 |
| VAREC    | 14  | N/A     |         | 2.1E-08 | 2.5E+01 | 6.8E-11 | 2.5E+01 |
| VAR6C    | 14  | N/A     |         | 2.1E-08 | 2.5E+01 | 6.8E-11 | 2.5E+01 |
| VARVC    | 14  | N/A     |         | 2.1E-08 | 2.5E+01 | 6.8E-11 | 2.5E+01 |
| VAPHC    | 15  | N/A     |         | 3.7E-10 | 5.5E+01 | 2.2E-09 | 5.8E+01 |
| VAPVC    | 15  | N/A     |         | 3.7E-10 | 5.5E+01 | 2.2E-09 | 5.8E+01 |
| VADEC    | 15  | N/A     |         | 3.7E-10 | 5.5E+01 | 2.2E-09 | 5.8E+01 |
| VARGE    | 15  | N/A     |         | 3.7E-10 | 5.5E+01 | 2.2E-09 | 5.8E+01 |
| VARVC    | 15  | N/A     |         | 3.7E-10 | 5.5E+01 | 2.2E-09 | 5.8E+01 |



Page 1 Date 21-Aug-87

ONSITE TRANSPORTATION - BARGE

Scenario Frequencies and Range Factors

|           |            | 195     | PANEE  |
|-----------|------------|---------|--------|
| CCENADIO  | мЛ         | 5250    | EALTID |
| SUCKMENIO | <b>au.</b> | LUC R   | PHLIJR |
|           |            |         |        |
| VWKHS     | 1          | 0.0E+00 | ~-     |
| VWKHS     | 2          | 0.0E+00 |        |
| VWKHS     | 3          | 2.7E-11 | 26     |
| VWKHF     | 5          | 0.0E+00 |        |
| VWKHS     | 6          | 2.3E-07 | 20     |
| VWKHF     | 7          | 1.9E-07 | 20     |
| VWKHS     | 9          | 0.0E+00 |        |
| VWKHS     | 10         | 0.0E+00 |        |
| VWKHS     | 11         | 1.2E-09 | 14     |
| VWKHF     | 13         | 0.0E+00 |        |
| VWKHE     | 14         | 1.12-09 | 24     |


TABLE 8-17 ONSITE TRANSPORTATION - ONSITE PACKAGE - COLLOCATION OPTION (MOVEMENT FROM STORAGE TO DEMIL FACILITY IN ONSITE PACKAGE)

Scenario Frequencies and Range Factors

| SCEN- | No. | ANAD    | RANGE<br>FACTOR | AP6<br>Fred | RANGE<br>Factor | LBAD | RANGE<br>Factor | KAAP | RANGE<br>FACTOR | PBA<br>Fred | RANGE<br>Factor | PUDA | RANGE<br>Factor | TEAD<br>Fred | RANGE<br>Factor | UNDA<br>Freq | RANGE<br>Factor |
|-------|-----|---------|-----------------|-------------|-----------------|------|-----------------|------|-----------------|-------------|-----------------|------|-----------------|--------------|-----------------|--------------|-----------------|
|       | 1   |         |                 |             |                 |      |                 |      |                 |             |                 |      |                 |              |                 |              |                 |
| VOB65 | -   | N/A     |                 | N/A         | ;               | N/A  | ;               | N/A  | ł               | N/A         | ;               | N/A  | ł               | 1.46-08      | 2.2E+01         | N/A          | ł               |
| SHODA |     | 1.46-08 | 2.2E+01         | N/A         | :               | N/A  | :               | N/A  | 1               | N/A         | ;               | N/A  | 1               | 1.4E-08      | 2.2£+01         | N/A          | :               |
| VOCGS | -   | 1.46-08 | 2.2E+01         | N/A         | !               | N/A  | ;               | N/A  | 1               | N/A         | :               | N/A  | 1               | 1.46-08      | 2.2E+01         | A/A          | ;               |
| VOCHS |     | 1.4E-08 | 2.2E+01         | N/A         | :               | N/A  | ;               | N/A  | 1               | N/A         | :               | N/A  | ł               | 1.4E-08      | 2.2€+01         | N/A          | 1               |
| VOKES | -   | N/A     | ;               | N/A         | ł               | N/A  | :               | N/A  | ;               | N/A         | ł               | N/A  | 1               | 1.46-08      | 2.2E+01         | N/A          | :               |
| VQKHS | -   | 1.46-08 | 2.2E+01         | N/A         | 1               | N/A  | ;               | N/A  | ;               | N/A         | 1               | N/A  | ł               | 1.4E-08      | 2.2E+01         | N/A          | !               |
| VOKVS | -   | 1.4E-08 | 2.2E+01         | N/A         | 1               | N/A  | ;               | N/A  | ţ               | N/A         | ;               | N/A  | ;               | 1.46-08      | 2.2E+01         | N/A          | 1               |
| VUNVS | -   | 1.46-08 | 2.2E+01         | N/A         | 1               | N/A  | ;               | N/A  | 1               | N/A         | ;               | N/A  | ;               | 1.46-08      | 2.2E+01         | N/A          | ł               |
| VOP6S | -   | 1.46-08 | 2.2E+01         | N/A         | ł               | N/A  | !               | N/A  | 1               | N/A         | :               | N/A  | ;               | 1.4E-08      | 10-32.2         | N/A          | ł               |
| VOPHS | -   | 1.4E-08 | 2.2E+01         | N/A         | ;               | A/A  | 1               | N/A  | :               | N/A         | 1               | N/A  | 1               | 1.4E-08      | 2.2E+01         | N/A          | ł               |
| VOPVS | -   | 1.4E-08 | 2.2E+01         | N/A         | ;               | N/A  | :               | N/A  | ;               | N/A         | ţ               | N/A  | ł               | 1.46-08      | 2.2E+01         | N/A          | 1               |
| V006S | -   | 1.46-08 | 2.2E+01         | N/A         | ł               | N/A  | ;               | N/A  | !               | N/A         | :               | N/A  | ţ               | 1.46-08      | 2.26+01         | N/A          | 1               |
| 2000  |     | N/A     | ł               | N/A         | :               | N/A  | ł               | N/A  | :               | N/A         | :               | N/A  | :               | 1.46-08      | 2.2E+01         | N/A          | ł               |
| VORGS |     | 1.46-08 | 2.2E+01         | N/A         | 1               | N/A  | :               | N/A  | !               | N/A         | :               | N/A  | !               | 1.45-08      | 2.2E+01         | N/A          | 1               |
| VORVS |     | 1.4E-08 | 2.2E+01         | N/A         | :               | N/A  | :               | N/A  | :               | N/A         | :               | N/A  | :               | 1.46-08      | 2.2E+01         | N/A          | :               |
| SVSDV |     | N/A     | ;               | N/A         | ;               | N/A  | :               | N/A  | ;               | N/A         | :               | N/A  | 1               | 1.46-08      | 2.2E+01         | N/A          | ;               |
| VONGS | -   | N/A     | ł               | N/A         | :               | N/A  | :               | N/A  | 1               | N/A         | ł               | N/A  | :               | 1.46-08      | 2.26+01         | N/A          | ł               |
| VOB65 | ~   | N/A     | ;               | N/A         | :               | N/A  | 1               | N/A  | :               | N/A         | :               | N/A  | :               | 5.46-11      | 2.6E+01         | A/N          | :               |
| SHODA | M   | 5.4E-11 | 2.6E+01         | N/A         | ;               | N/A  | 1               | N/A  | :               | N/A         | :               | A/A  | 1               | 5.4E-11      | 2.6E+01         | A/N          | ;               |
| VOCES | m   | 5.4E-11 | 2.6E+01         | N/A         | ;               | N/N  | :               | N/A  | :               | N/A         | ;               | N/A  | :               | 5.46-11      | 2.6E+01         | N/A          | :               |
| VOCHS | m   | 5.4E-11 | 2.6E+01         | N/A         | :               | N/A  | 1               | N/A  | ł               | N/A         | ;               | A/N  | 1               | 5.4E-11      | 2.66+01         | N/A          | :               |
| VOK65 | 3   | N/A     | ł               | N/A         | :               | N/A  | :               | N/A  | ;               | N/A         | :               | N/A  | :               | 5.46-11      | 2.6E+01         | A/A          | 1               |
| VOKHS | ы   | 5.4E-11 | 2.6E+01         | N/A         | :               | N/A  | ł               | N/A  |                 | N/A         | :               | N/A  | :               | 5.46-11      | 2.6E+01         | A/A          | ł               |
| VDKVS | m   | 5.4E-11 | 2.6E+01         | N/A         | ł               | N/A  | 1               | N/A  | :               | N/A         | :               | N/A  | 1               | 5.4E-11      | 2.6E+01         | N/A          | ł               |
| SVNDV | ~   | 5.4E-11 | 2.6E+01         | N/A         | ;               | N/A  | :               | N/A  | ł               | N/A         | :               | N/A  | :               | 5.4E-11      | 2.6E+01         | A/A          | ;               |
| VOF65 | ~   | 5.4E-11 | 2.6£+Ùl         | N/A         | ;               | NIA  | ;               | N/A  | ;               | N/A         | ;               | A/M  | ;               | 5.46-11      | 2.6€+01         | N/A          | 1               |
| VOPHS | n   | 5.4E-11 | 2.6E+01         | N/A         | 1               | N/A  | :               | N/A  | ;               | N/A         | ;               | M/A  | :               | 5.4E-11      | 2.6E+01         | N/A          | :               |
| VOFVS | m   | 5.4E-11 | 2.6E+01         | N/A         | ;               | N/A  | ;               | N/A  | :               | N/A         | !               | N/A  | :               | 5.4E-11      | 2.6E+01         | N/A          | 1               |
| V006S | ~   | 5.4E-11 | 2.6E+01         | N/A         | ;               | N/A  | ł               | N/A  | ł               | N/A         | ;               | N/A  | 1               | 5.4E-11      | 2.6E+Ül         | N/A          | ;               |

See notes at end of table.

SSS CONTRACTOR







File: OMS/IRE6.4KKi Page 2 Date 19-Aug-87

## TABLE 8-17 (Continued)

# Scenario Frequencies and Range Factors

| ŝ       |  |
|---------|--|
|         |  |
| ۔<br>پر |  |
|         |  |
|         |  |
| C       |  |
|         |  |
| Ż       |  |
|         |  |
|         |  |
| ž       |  |
|         |  |

| RANGE       | F RC LOK |   | ;       | ;       | 1       | ;          | ;       | :       | ;       | ;       | ;       | ;       | ł       | :        | :       | :       | ;        | :        | 1       | ;       | 1       | !       | ;          | :       | ł       | :       | :       | :       | :       | :          | ł        |
|-------------|----------|---|---------|---------|---------|------------|---------|---------|---------|---------|---------|---------|---------|----------|---------|---------|----------|----------|---------|---------|---------|---------|------------|---------|---------|---------|---------|---------|---------|------------|----------|
| CHDA<br>CCC | F K F G  |   | N/A     | N/A     | A/N     | N/A        | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | M/A     | N/A     | N/A     | N/A     | N/A     | N/A        | N/A      |
| RANGE       | FACTOR   |   | 2.6E+01 | 2.6E+01 | 2.6E+01 | 2.6€+01    | 2.66+01 | 1.4E+02 | 1.4€+02 | 1.4E+02 | 1.46+02 | 1.46+02 | 1.4€+02 | 1.4E+02  | 1.45+02 | 1.46+02 | 3. 36+01 | 3. 3E+01 | 1.6€+02 | 1.6E+U2 | 1.6E+02 | 1.6E+02 | 1.6E+02    | 1.6E+02 | 2.0E+u1 | 2.0E+01 | 2.0E+01 | 2.0£+01 | 2.0E+01 | 2.0E+U1    | 2.0E+01  |
| TEAD        | FREG     |   | 5.46-11 | 5.46-11 | 5.46-11 | 5.46-11    | 5.46-11 | 3.0E-12 | 3.06-12 | 3.0E-12 | 3.06-12 | 3.06-12 | 3.06-12 | 3.06-12  | 3.0E-12 | 3.0E-12 | 2.2E-10  | 2.2E-10  | 2.86-14 | 2.06-14 | 2.86-14 | 2.86-14 | 2.86-14    | 2.86-14 | 1.66-10 | 1.66-10 | 1.66-10 | 1.66-10 | 1.6E-10 | 1. 66 - 10 | 1.65-10  |
| RANGE       | FACTOR   |   | :       | ;       | ł       | :          | ł       | ;       | :       | ;       | ;       | :       | :       | ľ        | :       | !       | ł        | :        | ł       | :       | ;       | :       | ;          | ;       | ;       | ;       | ;       | ł       | :       | ;          | ;        |
| PUDA        | FRE      |   | N/A     | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A     | N/A     | M/A     | N/A     | N/A      | N/A     | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A        | N/A      |
| RANGE       | FACTOR   |   | :       | !       | :       | :          | :       | :       | ;       | ł       | ł       | ;       | ;       | :        | :       | 1       | :        | :        | :       | :       | ;       | :       | ;          | :       | :       | 1       | ;       | :       | ;       | ;          | :        |
| PBA         | FRE      |   | N/A     | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A     | N/A     | M/A     | N/A     | N/A      | N/A     | N/A     | N/A      | N/A      | N/A     | N/A     | M/A     | N/A     | R'N        | N/A     | N/A     | N/A     | N/A     | M/A     | M/A     | N/A        | N/A      |
| RANGE       | FACTOR   |   | ;       | ļ       | ;       | ;          | ;       | ;       | ;       | ;       | 1       | ;       | :       | :        | ;       | ;       | ţ        | :        | 1       | ł       | 1       | ł       | :          | :       | :       | ;       | :       | !       | ;       | !          | :        |
| NAAP        | FRE      |   | N/A     | N/N     | N/N     | <b>M/A</b> | N/A     | N/A     | N/A     | N/A     | M/A     | N/A     | N/A     | R/N      | N/N     | N/A     | N/A      | M/A      | N/A     | N/A     | M/A     | N/A     | N/A        | N/A     | M/A     | N/A     | N/A     | A/A     | N/A     | N/A        | N/A      |
| RANGE       | F AC FUR |   | 1       | :       | ;       | :          | ;       | ;       | ł       | :       | ;       | ;       | :       | !        | :       | ;       | ;        | :        | ļ       | 1       | {       | :       | :          | :       | :       | :       | ł       | !       | 1       | ;          | ;        |
| LBAD        | FREQ     |   | N/A     | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A        | N/A     | A/A     | N/A     | N/A     | N/A     | N/A     | N/A        | N/A      |
| RANGE       | FACTUR   |   | ł       | ł       | ļ       | ;          | !       | :       | ;       | ł       | :       | :       | ł       | :        | :       | :       | :        | :        | 1       | :       | :       | ;       | ł          | 1       | 1       | ;       | :       | :       | ;       | ;          | :        |
| AP6         | F KES    |   | N/A     | N/A     | N/A     | N/A        | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | M/A     | N/A      | N/A      | N/A     | A/A     | N/A     | N/N     | N/A        | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | ĥ/Å        | N/A      |
| RANGE       | FACTOR   |   | ł       | 2.6E+01 | 2.46+01 | ;          | ;       | 1.4E+02 | 1.4E+02 | 1.4E+02 | 1.4E+02 | 1.4€+02 | 1.46+02 | 1.45+02  | 1.4€+02 | :       | 3. 36+01 | 3. 36+01 | ;       | ;       | 1.6E+02 | 1.66+02 | ;          | :       | 1       | 2.0E+01 | 2.0E+01 | 2.0E+01 | :       | 2.0E+01    | 2.úE+01  |
| ANAD        | FREG     |   | N/A     | 5.4E-11 | 5.46-11 | N/A        | N/A     | 3.0E-12 | 3.06-12 | 3.0E-12 | 3.06-12 | 3.06-12 | 3.06-12 | 3. úE-12 | 3.0E-12 | N/A     | 2.86-10  | 2.86-10  | N/A     | N/A     | 2.8E-14 | 2.86-14 | N/A        | N/A     | N/A     | 6.0E-10 | 6.0E-10 | 6.0E-10 | N/A     | 6.0E-1ù    | 6. ÚE-1Ù |
| No.         |          |   | ~       | m       | м       | n          | ~       | -       | -       | -       | -       | +       | +       | •        | -       | *       | +        | -        | 5       | רי      | רט      | רש      | <b>د</b> ا | ŝ       | 9       | 9       | 9       | 9       | 9       | 4          | 9        |
| SCEN-       | ARIO     | ; | 2VD0V   | VOKES   | VORVS   | SV2DV      | VONGS   | VODHC   | VOCGC   | VOCHC   | VONVC   | VDPGC   | VOPHC   | VOPVC    | V006C   | VOGVC   | VORGC    | VORVC    | VOBGF   | VOK 6F  | VOKHF   | VOK VF  | VOSVF      | VONGF   | VOBES   | VDDHC   | VOCGC   | VOCHC   | VOK6S   | VOKHS      | VOLVS    |

See notes at end of table.

File: OMSITRE6.MKI Page 3 Date 19-Aug-87

TABLE 8-17 (Continued)

Scenario Frequencies and Range Factors

| RANGE<br>Factor |                     | 1       | 1       | !          | ł        | 1       | ł       | 1       | 1       | 1       | 1       | ł        | :        | :        | ł       | 1       | ł       | !       | ł        | :       | ł       | ł       | 1       | ;       | 1       | ł       | 1        | 1       | ł       | ;        |
|-----------------|---------------------|---------|---------|------------|----------|---------|---------|---------|---------|---------|---------|----------|----------|----------|---------|---------|---------|---------|----------|---------|---------|---------|---------|---------|---------|---------|----------|---------|---------|----------|
| LINDA<br>FREQ   | <br> <br> <br> <br> | N/A     | M/A     | N/A        | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/N     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A      |
| RAMGE<br>Factor |                     | 2.0E+01 | 2.0E+01 | 2.0E+01    | 2.0E+01  | 2.0E+01 | 2.0E+01 | 2.0E+01 | 2.0£+01 | 2.0E+01 | 2.0E+01 | 2.0E+01  | 2.0E+01  | 2.0E+01  | 2.0E+01 | 2.0E+01 | 2.0E+01 | 2.0E+01 | 2.0E+01  | 2.0E+01 | 2.0E+01 | 2.0E+01 | 2.0E+01 | 2.0E+01 | 2.0E+01 | 2.0E+01 | 2.0E+01  | 2.0E+01 | 1.1E+01 | 1.1E+01  |
| TEAD<br>Freq    |                     | 1.6E-10 | 1.6E-10 | 1.66-10    | 1.66-10  | 1.66-10 | 1.6E-10 | 1.66-10 | 1.66-10 | 1.6E-10 | 1.66-10 | 1. 3E-10 | 1. 3E-10 | 1. 3E-10 | 1.3E-10 | 1.3£-10 | 1.3E-10 | 1.36-10 | 1. 3E-10 | 1.36-10 | 1.36-10 | 1.3E-10 | 1.3E-10 | 1.3E-10 | 1.3E-10 | 1.36-10 | 1. 3£-10 | 1.36-10 | 1.06-05 | 1.0E-05  |
| RANGE<br>Factor |                     | 1       | ;       | ł          | 1        | 1       | !       | ;       | ;       | 1       | ;       | !        | ;        | ;        | ;       | :       | ;       | 1       | :        | :       | ;       | :       | 1       | !       | :       | :       | ł        | 1       | ł       | ł        |
| FREQ            |                     | N/A     | N/A     | N/A        | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/N      | A/A      | N/A     | N/A     | N/A     | N/A     | A/A      | N/A     | N/N     | N/A     | N/A     | NJA     | N/A     | N/A     | N/A      | N/A     | A/A     | N/A      |
| RANGE<br>Factor |                     | :       | :       | :          | ;        | :       | :       | :       | ;       | 1       | ł       | ł        | :        | ł        | ł       | ł       | :       | 1       | {        | ł       | 1       | :       | ł       | 1       | 1       | ł       | ł        | ſ       | :       | 1        |
| PBA<br>Fred     |                     | N/A     | N/A     | N/A        | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | A/A      | N/A      | NVA     | N/A     | N/A     | N/N     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A      |
| RANGE<br>Factor |                     | ł       | 1       | ł          | 1        | ;       | ;       | ;       | ł       | ł       | ł       | ;        | :        | ł        | ł       | :       | 1       | ;       | 1        | 1       | ł       | 1       | ł       | :       | ł       | :       | ;        | :       | ;       | ł        |
| KRED            |                     | N/A     | N/A     | N/A        | N/A      | M/A     | N/A     | N/A     | N/A     | N/A     | M/A     | A/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A      |
| RANGE<br>Factor |                     | ;       | •       | :          | :        | ł       | ;       | :       | 1       | 1       | 1       | :        | :        | :        | :       | :       | 1       | 1       | :        | ;       | :       | :       | ;       | :       | ł       | ł       | ł        | ;       | :       | ;        |
| LBAD<br>FREQ    |                     | N/N     | N/A     | N/A        | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A      |
| RANGE<br>FACTOR |                     | :       | ;       | :          | ł        | 1       | ;       | 1       | ;       | ;       | 1       | 1        | ;        | :        | :       | :       | ;       | ;       | ł        | :       | 1       | :       | :       | ;       | :       | :       | :        | :       | :       | 1        |
| AP6<br>Free     |                     | N/A     | N/A     | N/A        | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A      | N/A      | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A      | N/A     | N/A     | N/A      |
| RANGE<br>Factor |                     | 2.0E+01 | 2.06+01 | 2.0E+01    | 2.0E+01  | 2.0E+01 | 1       | 2.0E+01 | 2.0E+01 | 1       | 1       | :        | 2.0E+01  | 2.0E+01  | 2.0E+01 | :       | 2.0E+01 | 2.0E+01 | 2.0E+01  | 2.0E+01 | 2.06+01 | 2.0E+01 | 2.0E+01 | ;       | 2.0E+01 | 2.0E+01 | ;        | ;       | ;       | 1.1E+01  |
| ANAD<br>FREQ    |                     | 6.0E-10 | 6.06-10 | 6. 0E - 10 | 6. ÚE-1Ú | 6.0E-10 | N/A     | 6.0E-10 | 6.0E-10 | N/A     | N/A     | N/A      | 4.96-10  | 4.96-10  | 4.96-10 | N/A     | 4.96-10 | 4.95-10 | 4.96-10  | 4.9E-10 | 4.96-10 | 4.96-10 | 4.9E-10 | N/A     | 4.96-10 | 4.96-10 | N/A      | N/A     | N/A     | 6. UE-07 |
| No.             | •                   | -0      | -9      | 4          | •        | -9      | •       | 9       | -0      | \$      | •       | 2        | 1        | 1        | -       | 1       | 2       | -       | 1        | 1       | ~       | ~       | 1       | ~       | 1       | 1       | -        | -       | •       | •        |
| SCEN-<br>Ario   |                     | VONVC   | VOPEC   | VOPHC      | VOPVC    | V006C   | VOOVC   | VOKGC   | VORVC   | SV2DV   | VONGS   | VOBGF    | VODHC    | VOCGC    | VUCHC   | VOK6F   | VOKHF   | VOKVF   | VONUC    | VOP6C   | VOPHC   | VOPVC   | VDQGC   | 700VC   | VOREC   | VORVC   | VOSVF    | VONGF   | VOB65   | SHODA    |

See notes at end of table.

 $\sim$  N

22.22.22

1.1.1.1.1



File: OMSITREG.MKI Page 4 Date 19-Aug-87

TABLE 8-17 (Continued)

Scenario Frequencies and Range Factors

| RANGE<br>Factor         |           | ł       | :        | :        | ł        | :       | ;        | 1        | ;        | ł         | !       | ;       | ł       | ;       | 1       | ;            | ;       | ţ       | :       | 1       | 1       | ł       | !       | ;       | ;       | !       | !       | ;       | ł       | :       |
|-------------------------|-----------|---------|----------|----------|----------|---------|----------|----------|----------|-----------|---------|---------|---------|---------|---------|--------------|---------|---------|---------|---------|---------|---------|---------|---------|---------|---------|---------|---------|---------|---------|
| UMDA<br>FREQ            |           |         | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      | N/A       | N/A     | N/A     | N/A     | N/A     | N/A     | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | M/A     |
| RANGE<br>FACTOR         |           | ł       | 1.1E+01  | 1. IE+01 | 1.1E+01  | 1.16+01 | 1. IE+01 | 1.1E+01  | 1.1E+Ù1  | 1.16+01   | 1.1E+01 | 1.1E+01 | 1.16+01 | 1.16+01 | 1.1E+01 | 1. iE+01     | 1.4E+01 | 1.4E+01 | 1.4E+01 | 1.4E+01 | 1.4E+ù1 | 1.4E+01 | 1.4E+01 | 10+34.1 | 1.46+01 | 1.4E+01 | 1.4E+01 | 1.4£+01 | 1.4E+01 | 1.4E+01 |
| TEAD<br>FREQ            | - VC - VC | rn_30*1 | 1.0E-05  | 1.06-05  | 1.06-05  | 1.06-05 | 1.06-05  | 1.06-05  | 1.0E-05  | 1.06-05   | 1.06-05 | 1.06-05 | 1.06-05 | 1.05-05 | 1.0E-05 | 1.06-05      | 3.9E-08 | 3.96-08 | 3.96-08 | 3.96-08 | 3.96-08 | 3.96~08 | 3.9E-0B | 3.9E-08 | 3.96-08 | 3.96-08 | 3.96-08 | 3.96-08 | 3.96-08 | 3.96-08 |
| RANGE<br>FACTOR         |           |         | :        | 1        | 1        | ;       | :        | 1        | :        | :         | :       | :       | ;       | ł       | :       | :            | ł       | :       | 1       | ł       | ł       | :       | :       | ł       | :       | :       | :       | 1       | :       | :       |
| PUDA<br>FREQ            |           |         | R/N      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      | N/A       | N/A     | N/A     | N/A     | N/A     | N/A     | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | M/A     | N/A     | N/A     |
| RANGE<br>Factor         |           | 1       | ł        | 1        | ;        | ;       | 1        | 1        | ł        | 1         | ;       | !       | ł       | ;       | ;       | :            | ł       | :       | ;       | ł       | 1       | :       | 1       | !       | ;       | :       | :       | ;       | 1       | 1       |
| PBA<br>FREQ             |           |         | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      | N/A       | N/A     | N/A     | N/A     | N/A     | N/A     | N/A          | N/A     | N/A     | N/A     | N/A     | N/N     | N/A     | A/N     |
| RANGE<br>FACTOR         |           | ł       | ;        | ;        | :        | :       | ;        | 1        | :        | :         | ;       | :       | :       | ;       | !       | ;            | ł       | ł       | 1       | :       | ;       | :       | ļ       | :       | ;       | :       | :       | ;       | :       | ł       |
| NAAP<br>FREQ            |           |         | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      | N/A       | N/A     | N/A     | N/A     | N/A     | N/A     | N/A          | N/A     | A/A     | N/A     |
| frange<br>F ac tok      |           | ſ       | 1        | ;        | ;        | :       | !        | ł        | 1        | ;         | :       | ł       | !       | ;       | :       | ;            | 1       | :       | 1       | 1       | 1       | !       | ł       | :       | :       | ł       | :       | ł       | ;       | {       |
| L BAD<br>Freq           |           |         | N/A      | N/A      | N/A      | N/A     | N/A      | A/A      | N/A      | N/A       | N/A     | N/A     | N/A     | N/A     | M/A     | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     |
| RAMGE<br>Factor         |           | ł       | ;        | :        | 1        | 1       | ;        | !        | 1        | 1         | ;       | :       | ;       | ł       | :       | :            | ;       | ;       | :       | ;       | ł       | :       | :       | ł       | :       | :       | :       | ł       | 1       | !       |
| APG<br>FRED             |           |         | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      | N/A      | N/A       | N/A     | N/A     | N/A     | A/A     | N/A     | N/A          | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     | N/A     |
| R <b>ange</b><br>Factor |           | 1.16701 | 1.1E+01  | :        | 1.16+01  | 1.1E+01 | 1. IE+01 | 1.16+01  | 1. 1E+01 | 1.16+31.1 | 1.1E+01 | ;       | 1.1E+01 | 1.1E+01 | ;       | :            | ;       | 1.4E+01 | 1.4E+01 | 1.4E+01 | ;       | 1.4£+01 | 1.4E+01 | 1.4E+01 | 1.4E+01 | 1.46+01 | 1.4E+01 | 1.4E+01 | !       | 1.4E+ù1 |
| ANAD<br>FREQ            |           | 0.00-0/ | 6. UE-07 | A/A      | 6. 0E-07 | 6.05-07 | 6.0E-07  | 6. ÜE-07 | 6.05-07  | 6.0E-07   | 6.05-07 | N/A     | 6.0E-07 | 6.0E-07 | N/A     | N/A          | N/A     | 2.4E-09 | 2.4E-09 | 2.4E-09 | N/A     | 2.4E-09 | 2.4E-09 | 2.4E-09 | 2.4E-09 | 2.4E-09 | 2.4E-09 | 2.46-09 | N/A     | 2.4E-09 |
| 2                       | ; •       | -       | •        | •        | 6        | •       | 6        | •        | 6        | 6         | •       | 6       | 6       | œ       | •       | <del>۰</del> | Π       | Π       | =       | Ξ       | Ξ       | Ξ       | 11      | Π       | Ξ       | Ξ       | П       | Ξ       | Ξ       | Π       |
| SCEN-<br>AR10           | 10101     | COUL    | VOCHS    | VOK 65   | VOKHS    | VOKVS   | SVNDV    | VOP6S    | VOPHS    | VOPVS     | V0065   | SVGDV   | VORES   | VORVS   | VDSVS   | VONGS        | VOB6S   | SHODA   | VOC65   | VOCHS   | VOK6S   | VORHS   | VUKVS   | SVNDV   | VOP6S   | VOPHS   | VOPVS   | V006S   | SVGOV   | VOK6S   |

See notes at end of table.

'*r* 

File: OMS[TRE6.44ki Page 5 Date 19-Aug-87

TABLE 8-17 (Continued)

Scenaric Frequencies and Range Factors

| <br>KANGE<br>Factor |                       | 1       | 1       | ł       | !        | ;        | ł       | ;        | ;       | ;         | ;         | ł        | !        | ;       | ;       | ł        | ł       | 1        | :        | ;       | ł       | ;          | 1          | 1        | ;        | ;       | ;        | ;         | ;        | ;         |
|---------------------|-----------------------|---------|---------|---------|----------|----------|---------|----------|---------|-----------|-----------|----------|----------|---------|---------|----------|---------|----------|----------|---------|---------|------------|------------|----------|----------|---------|----------|-----------|----------|-----------|
| UNUA<br>FREQ        |                       | N/A     | N/A     | N/A     | N/A      | N/A      | M/A     | N/A      | N/A     | N/A       | N/A       | N/A      | N/A      | N/A     | N/A     | N/A      | N/A     | N/A      | N/A      | A/A     | N/A     | N/A        | N/A        | M/A      | N/A      | N/A     | N/A      | N/A       | N/A      | N/A       |
| <br>KANGE<br>FACTOR |                       | 10+34.1 | 1.4E+01 | 10+3+.1 | B. BE+01 | 8.8E+01  | B.8E+01 | 8.8E+01  | 8.8E+01 | B. BE +01 | B. BE +01 | 8.8E+01  | 8.8E+01  | 2.0E+01 | 2.0E+01 | 1.0€+02  | 1.0E+02 | 1.0€+02  | 1.0E+02  | 1.0E+02 | 1.0E+02 | 1. JE +01  | 1.3E+01    | 10+3£-1  | 1.36+01  | 10+32.1 | 1. 3E+01 | 1. 3E +01 | 10+3E.1  | 1.JE+01   |
| I LAD<br>Free       |                       | 3.96-08 | 3.9E-08 | 3.96-08 | 7.1E-12  | 7. IE-12 | 7.1E-12 | 7. IE-12 | 7.16-12 | 7.16-12   | 7.1E-12   | 7. IE-12 | 7. IE-12 | 2.06-07 | 2.0E-07 | 7. IE-12 | 7.1E-12 | 7. IE-12 | 7. 1E-12 | 7.16-12 | 7.1E-12 | 3. 3E - 09 | 3. 36 - 09 | 3.36-09  | 3. 35-09 | 3.36-09 | 3. 3E-09 | 3. 36-09  | 3. 36-09 | 3. JE -09 |
| RANGE<br>Factor     | ,<br>,<br>,<br>,<br>, | ;       | ł       | :       | ł        | 1        | !       | ł        | !       | !         | ł         | :        | ł        | ;       | ł       | :        | :       | ł        | :        | !       | ;       | ;          | :          | ł        | ;        | ł       | :        | :         | :        | ;         |
| <br>PUDA<br>Free    |                       | N/A     | N/A     | N/A     | N/A      | N/A      | N/A     | N/A      | N/A     | N/A       | N/A       | N/A      | N/A      | N/A     | N/A     | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A        | N/A        | N/A      | N/A      | N/A     | N/A      | N/A       | N/A      | N/A       |
| RANGE<br>Factor     | ;                     | ł       | ;       | {       | ;        | ;        | {       | !        | {       | {         | ł         | !        | :        | ;       | :       | ;        | ;       | ;        | ;        | ł       | ł       | :          | :          | ;        | ł        | ł       | ;        | !         | ł        | ;         |
| FREQ                |                       | N/A     | N/A     | N/A     | N/A      | N/A      | N/A     | N/A      | N/A     | N/A       | N/A       | N/A      | N/A      | N/A     | N/A     | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A        | N/A        | N/A      | N/A      | N/A     | N/A      | N/N       | N/A      | A/A       |
| RANGE<br>Factor     |                       | 1       | ;       | ;       | ł        | ł        | :       | ;        | ;       | ļ         | ;         | :        | :        | :       | ;       | ;        | ;       | !        | 1        | :       | 1       | ;          | ;          | :        | :        | 1       | ;        | ;         | :        | ١         |
| KEQ FREQ            |                       | N/A     | N/A     | N/A     | N/A      | N/A      | N/A     | M/A      | N/A     | N/A       | N/A       | N/A      | N/A      | N/A     | N/A     | N/A      | N/A     | N/A      | N/A      | A/N     | N/A     | N/A        | N/A        | N/A      | N/A      | N/A     | A/A      | N/A       | N/A      | N/A       |
| RANGE<br>Factor     |                       | ;       | ;       | ł       | ;        | ł        | ;       | ł        | ;       | 4<br>5    | ł         | ł        | !        | ;       | ł       | ł        | 1       | ;        | ł        | !       | ł       | ;          | ł          | ł        | ;        | :       | :        | :         | :        | :         |
| LBAD<br>Freq        |                       | N/A     | N/A     | N/A     | N/A      | N/A      | N/A     | N/A      | N/N     | N/A       | N/A       | N/A      | N/A      | N/A     | N/A     | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A        | N/A        | N/A      | N/A      | A/A     | N/A      | N/A       | N/A      | N/A       |
| KANGE<br>Factor     |                       | :       | !       | ;       | :        | 1        | ;       | 1        | 1       | ;         | :         | ł        | ;        | ;       | ;       | 1        | ł       | :        | ;        | 1       | :       | :          | 1          | 1        | :        | :       | :        | 1         | :        | ł         |
| AP6<br>Freq         |                       | N/A     | N/A     | N/A     | N/A      | N/A      | M/A     | N/A      | N/A     | N/A       | A/N       | N.A      | N/A      | N/A     | N/A     | N/A      | N/A     | N/A      | N/A      | N/A     | N/A     | N/A        | N/A        | N/A      | N/A      | N/A     | N/A      | N/A       | N/A      | N/A       |
| RANGE<br>Factor     |                       | 1.4E+01 | :       | 1       | 1.0E+02  | L. ÜE+02 | 1.0E+02 | 1.0E+02  | 1.0E+02 | 1. üE+02  | 1.05+02   | 1.0€+02  | :        | 1.76+01 | 1.75+01 | ł        | 1       | 1.1E+02  | 1.1E+02  | ;       | :       |            | 1.0E+01    | 1.0E+01  | 1.0E+01  | ;       | 1.05+01  | 1.0E+01   | 1.0€+01  | 1.0E+01   |
| <b>FREQ</b>         |                       | 2.46-09 | N/A     | N/A     | 4.26-13  | 4.2E-13  | 4.2E-13 | 4.25-13  | 4.2E-13 | 4.25-13   | 4.2E-13   | 4.26-13  | M/A      | 1.26-08 | 1.26-08 | N/A      | N/A     | 4.25-13  | 4.26-13  | N/A     | N/A     | N/A        | 1.16-06    | 1.16-06  | 1.1E-06  | N/A     | 1.16-06  | 1.1E-06   | 1.1E-06  | 1.1E-06   |
| No.                 | •                     | Ξ       | 11      | Π       | 12       | 12       | 12      | 1        | 12      | 51        | 12        | 12       | 12       | 12      | 13      | 13       | 13      | 13       | 13       | 11      | 13      | 1          | 1          | <b>1</b> | =        | =       | 2        | <b>1</b>  | =        | =         |
| SCEN-<br>Ario       |                       | VORVS   | SV2DV   | VOMES   | VODHC    | VOCEC    | VOCHC   | VDRVC    | VDP6C   | VOFHC     | VOPVC     | VODEC    | VOQVC    | VOKEC   | VORVC   | 2980A    | VOKEC   | VOKHC    | VOKVC    | VOSVC   | V0#6C   | VOBEC      | VODHC      | V0C6C    | VOCHC    | VOKES   | VOKHS    | VOKVS     | VONVC    | VOPGC     |

See notes at end of table.

•



ou ou ou

b



File: OMSITAE6.Mkl Page b Date 19-Aug-87

## TABLE 8-17 (Continued)

# Scenario Frequencies and Kange Factors

| KANGE        | FACTOR  |                  | :         | !        | 1       | ;        | 4         | ;       | !            | 1       | ;        | 1        | ;         | !       | 1<br>4   | ;        | ł        | ł        | I<br>T  | 1        | ;        |   |
|--------------|---------|------------------|-----------|----------|---------|----------|-----------|---------|--------------|---------|----------|----------|-----------|---------|----------|----------|----------|----------|---------|----------|----------|---|
| UNDA         | FREQ    |                  | N/A       | N/A      | N/A     | N/A      | N/A       | N/A     | N/A          | N/A     | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      |   |
| KANGE        | FACT OR |                  | 1.3E+01   | 1.36+01  | 1.36+01 | 1.3E+01  | 1. JE +u1 | 1.36+01 | 1. JE+01     | 1.3E+01 | 5. IE+01 | 5. IE+01 | 5. IE +01 | 5.1E+01 | 5. IE+01 | 5. 1E+úl | 5. IE+UI | 5. 1E+ùl | 5.16+01 | 5. IE+VI | 5. 1E+01 |   |
| TEAD         | FRED    |                  | 3. 3E -09 | 3. 36-09 | 3.3E-09 | 3. 3E-09 | 3.36-09   | 3.36-09 | 3.36-09      | 3.36-09 | 2.2E-09  | 2.2E-09  | 2.2E-U9   | 2.2E-09 | 2.2E-09  | 2.2E-09  | 2.2E-09  | 2.2E-09  | 2.2E-09 | 2.2E-09  | 2.2E-09  |   |
| RANGE        | FACTOR  | 1                | 1         | ł        | ;       | !        | ;         | :       | ;            | !       | :        | ;        | 1         | ł       | :        | ;        | :        | :        | ;       | ł        | ;        |   |
| PUDA         | FREQ    |                  | N/A       | N/A      | N/A     | A/A      | A/M       | N/A     | N/A          | N/A     | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      |   |
| RANGE        | FACTOR  |                  | ł         | ;        | ţ       | ;        | :         | :       | 1            | ł       | ;        | ;        | 1         | ł       | :        | :        | ;        | :        | ;       | !        | 1        |   |
| FBA          | FREQ    |                  | N/A       | N/A      | N/A     | N/A      | NIA       | A/A     | N/A          | N/A     | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | N/A      | N/A     | N/A      | N/A      |   |
| RANGE        | FACTOR  |                  | ;         | :        | :       | 1        | :         | :       | ł            | ł       | :        | ;        | ;         | ł       | :        | ;        | ;        | :        | 1       | ;        | ;        |   |
| NAAP         | FREQ    |                  | N/A       | N/A      | N/A     | N/A      | N/A       | N/A     | N/A          | N/A     | N/A      | N/A      | N/A       | N/A     | N/A      | N/A      | N/A      | H/N      | N/A     | N/A      | 8/N      |   |
| RANGE        | FACTOR  | 1<br>5<br>1<br>1 | ;         | :        | ;       | ;        | ;         | ł       | 1            | 1       | ;        | ;        | ł         | ;       | 1        | !        | ;        | ł        | !       | ;        | ;        |   |
| LBAD         | FREQ    |                  | N/A       | N/A      | N/A     | N/A      | A/A       | N/A     | N/A          | K/A     | A/A      | R/A      | N/A       | N/A     | A/A      | N/A      | N/A      | R/N      | A/A     | N/A      | N/A      |   |
| KANGE        | FACTOR  |                  | ;         | :        | ;       | ł        | ;         | ł       | :            | ;       | ł        | ;        | :         | :       | ;        | ţ        | 1        | 1        | 1       | i        | ;        | _ |
| AP6          | FREQ    |                  | N/A       | N/A      | N/A     | N/A      | N/A       | N/A     | NZA          | N/A     | A/A      | N/A      | N/A       | N, A    | N/A      | N/A      | N/A      | N/A      | N/A     | N/A      | H/A      | - |
| <b>RANGE</b> | FACTOR  |                  | 1.0E+01   | 1.0€+01  | 1.0E+01 | 1        | 1.0€+01   | 1.06+01 | ;            |         | 10+32.4  | 4.2E+01  | 4.26+01   | 4.25+01 | 4.2E+01  | 4.2E+01  | 4.26+01  | 4.26+01  | !       | 4.2E+U1  | 4.2E+01  |   |
| ANAD         | FREQ    |                  | 1.1E-06   | 1.1E-06  | 1.15-06 | N/A      | 1.1E-05   | 1.1E-06 | N/A          | N/A     | 2.4E-09  | 2.4E-09  | 2.4E-09   | 2.4E-09 | 2.46-09  | 2.4E-09  | 2.4E-09  | 2.4E-09  | N/A     | 2.4E-09  | 2.46-09  |   |
| No.          |         | }                | =         | =        | =       | =        | =         | =       | Ξ            | =       | 5        | 5        | 5         | 5       | 5        | 5        | 5        | <u>5</u> | 5       | 2        | 5        |   |
| SCEN-        | AR 10   | •                | VOPHC     | VOPVC    | V00GC   | 24907    | VORGC     | VORVC   | <b>VOSVS</b> | VONGC   | VODHC    | VOCGC    | VOCHC     | VUNVC   | VOPEC    | VOPHC    | VOFVC    | V006C    | VOOVC   | VOREC    | VORVC    |   |

NUTES: 1. Scenarios 1–5 are per truck mile; scenarios 6–15 are per exposure year. 2. Duration time shuwn for scenarios with agent releases due to both detonations and spills is for spills only. Duration time for detonation is instantaneous.

results of the uncertainty analysis of the sequence frequency values. The range factor is the ratio of the 95th percentile value to the 50th percentile value of a log normal distribution. The accident frequencies for sequences 1 to 5 are reported per truck mile. The accident frequencies for sequences 6 to 15 are reported per exposure year. No quantitative screening of the sequences was done at this point in the analysis because the accidents per mile need to be multiplied by the number of miles (a classified number) prior to a meaningful screening analysis. 22.000000000

THE REPORT NOTION

-

The number of munitions truckloads is computed from the classified stockpile values divided by the number of munitions per truck load from Table 10-3. The accident frequency is determined by first multiplying the values in Table 8-16 by the number of truckloads. This product is multiplied either by the number of onsite truck miles or by the number of onsite truck exposure years. It is assumed that the trucks move individually to and from the railhead at an effective speed of 10 mph. The total exposure time is the onsite distance divided by 10 mph.

The calculation models described in Section 10 were used to determine the agent released for the onsite transportation accident sequences. The agent release results for these accident sequences are also given in Section 10.

The final results of the accident sequence analysis (per munition inventory) are contained in a classified appendix to this report.

#### 8.2. OFFSITE RAIL TRANSPORT

#### 8.2.1. Accident Scenario Definition

The transport accident scenarios involve train accidents (derailment, collision, highway grade crossings) with and without fires, and nonpreventable external events. The four types of force which could fail the munition casing or its package and cause an agent release (crush, impact, puncture, and fire) were also considered when the accident scenarios were developed. Section 4-1 describes the selection of the initiating events. As shown in Table 4-6, there are four families of initiating events for rail transport: (1) train accident (e.g., derailment) due to human error or equipment failure, (2) an aircraft crash into a railcar, (3) an earthquake-caused train accident, and (4) a tornado-caused train accident or generated missile penetration. Fourteen sequences were analyzed, resulting from the logic model development (Section 8.2.3) of these four initiating event families. These are as follows:

1. Train Accident Due to Human Error or Equipment Failure.

- RC1 A train accident involving a munitions railcar occurs and crush forces fail the agent containment.
- RC2 A train accident involving a munitions railcar occurs and impact forces fail the agent containment.
- RC3 A train accident involving a munitions railcar occurs and puncture forces fail the agent containment.
- RC4 A train accident with fire occurs. Either the OFC insulation is torn away due to mechanical forces and the fire is able to heat the munitions inside the OFC,

or the fire lasts long enough to cause burstered munitions to detonate. Undue force created by the accident may also detonate burstered munitions.

- RC5 A train accident with fire occurs. Either the OFC insulation is torn away due to mechanical forces and the fire is able to heat the munitions inside the OFC, or the fire lasts long enough to cause thermal rupture of the munitions.
- RC15 A train accident occurs due to an earthquake or tornado, generating undue mechanical forces which cause detonation of burstered munitions.
- 2. Aircraft Crash Into Railcars.

₩₽₩₽₩₽₩₽₩₽₩₽₩₽₩₽₩₽₩₽₩₽₩₽

- RC6 An aircraft crashes on a munitions railcar. No fire occurs, but impact forces lead to detonations and/or failure of agent containment.
- RC7 An aircraft crashes on a munitions railcar. Fire occurs, but impact forces lead to detonation and/or failure of agent containment.
- RC8 Open due to scenario revisions.
- 3. Earthquake-caused Train Accident.

- RC9 A severe earthquake occurs involving a munitions railcar and crush forces fail the agent containment.
- RC10 A severe earthquake occurs involving a munitions railcar and impact forces fail the agent containment.



- RC11 A severe earthquake occurs involving a munitions railcar and puncture forces fail the agent containment.
- RC12 A severe earthquake occurs involving a munitions railcar and subsequent fire detonates burstered munitions.
- RC13 A severe earthquake occurs involving a munitions railcar and subsequent fire fails nonburstered munitions.
- 4. Tornado Event.

RC14 - A tornado-generated missile leads to failure of the agent containment, or a tornado occurs causing overturn or derailment of a munitions railcar.

An inherent assumption in this list of sequences is that the accidents involving an aircraft crashing on a munitions railcar more closely resemble the SNL model of a typical aircraft crash rather than the SNL model of a typical train crash. Unlike the train accident scenarios where the impact failure probability is  $\epsilon$ , in a typical SNL aircraft crash the crush and puncture forces are negligible compared to the impact forces. It is also assumed that all aircraft crashes onto a railcar are totally uncontrolled crashes and therefore always have impact forces and that the crash is a severe one ( $\Delta V$  > 300 mph). Reference 8-1 contains further details.

#### 8.2.2. Rail Transport Procedures and Data

Prior to rail transportation, all chemical munitions will be secured in 20 x 8 x 8 ft offsite transportation containers. It is assumed that the munitions within the transportation container will not experience any impact as a result of a train accident. The transportation container provides additional protection from crush, impact, puncture, or fire. Two transportation containers are securely mounted one-high on each railcar. Inventories of agent for each munition and agent type are shown in Table 10-3.

For the regional option, munitions from NAAP, APG, LBAD, and PBA will be transported to ANAD and from all other sites to TEAD. In the national option, all munitions are sent to a single destruction center at TEAD. Mileages for these routes are given in Table 8-18.

Data on rates of rail accidents are presented in Section 9.1.1 and are summarized in Table 8-19. A rate of  $5.5 \times 10^{-6}$  accidents per rail mile was derived. The train fire accident rate was derived from data from Refs. 8-1 and 8-13. Train fires, without any special mitigation, involve only a single car in 90% of all cases (Ref. 8-13).

The fires of interest, then, are (1) derailment fires involving locomotive fuel, (2) collision fires involving locomotive fuel, and (3) grade-crossing collisions with a tanker on the track.

In accidents involving derailment of munitions trains, the most likely source of fire is the locomotive fuel; therefore, the locomotive itself must be severely damaged. For general cargo trains, about 1% of the derailments results in fire (Ref. 8-1) and it is assumed that the probability that the locomotive is one of the derailed cars is 0.5 (Ref. 8-13). Furthermore, since five buffer cars containing inert material are always placed between the locomotives and the first munitions car, the probability that a munitions car is exposed to the fire is assumed to be 0.01. (Reference 8-13 assumes that with only one buffer car there is a 0.1 probability of munitions car exposure to the fire.) Thus, the probability of a munitions car fire given a derailment equals  $(5.5 \times 10^{-6} \text{ accidents/train mile}) \times (0.83 \text{ derailments/accident}) \times (0.51 \text{ locomotive derailments}) \times (0.01 \text{ locomotive fires}) = 2.3 \times 10^{-10} \text{ fires/train mile}.$ 

|                         | P         | lail        |
|-------------------------|-----------|-------------|
| Storage Depot           | To Tooele | To Anniston |
| Aberdeen Proving Ground | 3035      | 1805        |
| Anniston                | 2834      |             |
| Lexington-Blue Grass    | 2546      | 1106        |
| Newport                 | 2201      | 980         |
| Pine Bluff              | 2624      | 1243        |
| Pueblo                  | 732       |             |
| Umatilla                | 1250      |             |

#### TABLE 8-18 TRAVEL MILES FOR RAIL TRANSPORT(a)

(a)Reference 8-11.





#### TABLE 8-19 RAIL TRANSPORTATION DATA

| Train accident rate                          | 5.5 x $10^{-6}$ accidents/mile |
|----------------------------------------------|--------------------------------|
| Fire accident probability, given an accident | $3.9 \times 10^{-4}$           |
| Impact environment probability               | 1.0                            |
| Crush environment probability                | 0.002                          |
| Puncture environment probability             | 5.9 x $10^{-4}$                |
| Aircraft crash onto a railcar probability    | $3.1 \times 10^{-11}$          |
| Undue mechanical force probability           | $5.8 \times 10^{-7}$           |

| Route                  | 0.35-g Earthquake<br><u>Probabilities</u> | Tornado<br><u>Probabilities (yr<sup>-1</sup>)</u> |
|------------------------|-------------------------------------------|---------------------------------------------------|
| Umatilla to Tooele     | 8.2 x $10^{-4}$                           | $3.3 \times 10^{-7}$                              |
| Pueblo to Tooele       | $2.5 \times 10^{-4}$                      | $3.7 \times 10^{-7}$                              |
| Lexington to Tooele    | $5.1 \times 10^{-4}$                      | $7.3 \times 10^{-5}$                              |
| Aberdeen to Tooele     | $1.9 \times 10^{-4}$                      | $6.8 \times 10^{-5}$                              |
| Anniston to Tooele     | $2.4 \times 10^{-4}$                      | 7.6 x $10^{-5}$                                   |
| Newport to Tooele      | $2.2 \times 10^{-4}$                      | 6.6 x $10^{-5}$                                   |
| Pine Bluff to Tooele   | $2.7 \times 10^{-4}$                      | 7.3 x 10 <sup>-5</sup>                            |
| Lexington to Anniston  | $2.8 \times 10^{-4}$                      | $1.0 \times 10^{-4}$                              |
| Pine Bluff to Anniston | $5.3 \times 10^{-4}$                      | $1.0 \times 10^{-4}$                              |
| Aberdeen to Anniston   | $2.0 \times 10^{-4}$                      | 9.3 x 10 <sup>-5</sup>                            |
| Newport to Anniston    | $3.0 \times 10^{-4}$                      | $1.0 \times 10^{-4}$                              |

For munitions train collision accidents, the most likely source of fire is again the locomotives; however, the probability that a munitions car will be affected by the fire is assumed to be 0.05 since the five buffer cars are expected to be less effective in a collision than in a fire. (Reference 8-13 assumes a value of 0.3 for only one buffer car.) Thus, using data presented earlier in this section:  $(5.5 \times 10^{-6}$ accidents/train mile) (0.066 collisions/accident) (0.01 fires/collision) (0.26 locomotive fires/collision) (0.05 munitions car exposures/ locomotive fires) = 4.7 x 10<sup>-11</sup> fires/train mile.

The third fire component, fires given a grade crossing collision is:  $(5.5 \times 10^{-6} \text{ accidents/train mile}) (0.038 \text{ grade crossing accidents/} \text{ accident}) (0.02 \text{ tanker accidents/grade crossing accident}) (0.5) (0.9) = 1.9 \times 10^{-9} \text{ fires/train mile}.$  Here, 0.5 is the probability that the tanker is full of fuel and 0.9 is the probability that when the train comes to a stop, a munitions car is in the fire.

The total munition car fires/train mile is the sum of the three contributors, 2.3 x  $10^{-10}$  + 4.7 x  $10^{-11}$  + 1.9 x  $10^{-9}$  = 2.2 x  $10^{-9}$ . For use in the scenario calculations, the train accident rate was divided out (2.2 X  $10^{-9}/5.5 \times 10^{-6}$ ) for 3.9 x  $10^{-4}$  fires, given a train accident. The train fire duration curves are shown in Fig. 8-7 for locomotive fires and train-tanker accident fires. These curves are used to obtain the probabilities of a fire having the duration to fail the agent containment. The shape of the train tanker fire duration curve is due to a number of factors including: (1) variation in tanker truck size, (2) distribution of spilled fuel, and (3) type and effectiveness of fire fighting response.

The worst-case impact environment for small packages (munitions inside OFCs) is no more severe than normal rough handling; therefore, impact is not a threat to small cargo in a railcar (Ref. 8-1). The severity distribution for large package impacts is shown in Fig. 8-8.



200000



8-56

Ŀ



Fig. 8-8. Impact velocity distribution for large packages in train accidents

The probability of a crush environment for large packages is 0.002, and the severity distribution is given in Fig. 8-9 (Ref. 8-2).

The probability of a puncture environment for large packages with a 0.75-in. steel wall thickness is  $5.9 \times 10^{-4}$  given a cargo damage accident. The probe considered capable of puncture was a railcar coupler (Ref. 8-2).

The probability of undue mechanical crush force causing detonation of burstered munitions was derived from the dynamic crush probability data in Ref. 8-1, assuming a lognormal distribution with a 50% probability of detonation at 3 x  $10^6$  lb of crush force and a  $10^{-6}$  probability for 36,000 lb of force.

#### 8.2.3. Event Tree Analysis

The event tree used in the analysis of a train accident due to earthquake or human error or equipment failure is shown in Fig. 8-10. Top events identified in the event tree and the analysis elements needed to evaluate the probabilities of these top events are as follows:

- BE33 Initiating event (train accident occurs, e.g., derailment, collision, grade crossing collision, aircraft crash, earthquake, etc.). The appropriate value for BE33 is given in the discussion for each sequence.
- BE51 Probe fails munition inside OFC.
- BE52 Impact force sufficient to fail munition inside OFC.
- BE52' Mechanical forces remove insulation from OFC.
- BE53 Crush force sufficient to fail munition inside OFC.



Fig. 8-9. Cumulative distribution of total crush load for a large package in train accident

DETONATION DETONATION DETONATION DETONATION TYPE OF AGENT Release BURN BURN BURN BURN NONE BURN BURN BURN NONE NONE SPILL BURN SPILL SPILL SPILL NON SPILL SPILL 3.2 × 10<sup>-1</sup>2 2.8 × 10<sup>-10</sup> 3.6 × 10<sup>-10</sup> FREQUENCY PER MILE 3.2 × 10<sup>-9</sup> SEQUENCE 10 RC13 RC4A RC4B RC12 RC10 RC11 RC2 RC3 RC9 RCI RC5 PUNCTURE 5.9 × 10<sup>-4</sup> PUNCTURE 5.9 x 10<sup>-4</sup> I MECHANICAL Failure Avdided I IMPACT € PUNCTURE IMPACT ¢ PUNCTURE CRUSH € CRUSH € IMPACT € CRUSH € CRUSH € IMPACT I I I 1 ļ 1 THERMAL FAILURE AVOIDED 1 1 **•FREQUENCIES ARE ROUTE DEPENDENT, RANGE COVERS ALL SITE COMBINATIONS** 0.16 0.16 0.84 0.84 ļ I 1 I DETONATION AVOIDED 5.8 × 10<sup>-7</sup> 0.17 8.0 0.17 0.83 0.83 -3.9 × 10<sup>-4</sup> FIRE Avoided 3.9 × 10<sup>-4</sup> 7 ī HUMAN/EQUIP 5.5 × 10<sup>-6</sup>/MI 2 × 10<sup>-4</sup> TO 8 × 10<sup>-4</sup> YR<sup>-1</sup> TRAIN ACCIDENT CAUSE EARTHQUAKE

Event tree for train accident caused by earthquake or human error/equipment failure F1g. 8-10.

aada baaada haadada baadada baadada baadaa haadaan kacada baadaa haadada baadaa baadaa baadaa baadaa baada baad

NOTE: < INDICATES SEQUENCES SCREEMED ON LOW FREQUENCY

8-60

**BE60** - Impact force generated.

BE61 - Impact force sufficient fails munition.

BE62 - Fire generated.

BE60/62 - Impact and fire generated.

BE63 - Fire has heat and duration to detonate burster in OFC.

BE64 - Probe generated.

BE68 - Crush force generated.

BE75 - Fire has heat and duration to fail nonburstered munitions in OFC.

The probability data for the accident sequences are summarized in Tables 8-20 through 8-33.

The results of the rail transport probabilistic analysis as well as the agent release calculations are summarized in Appendix I for the national and regional disposal options. The probabilities are given on a per train mile basis for sequences RC1 through RC5 and per train exposure year for sequences RC6 through RC14. The final probability values per munition inventory are contained in a classified appendix to this report.

Tables 8-34 and 8-35 present the results of the rail transport frequency and uncertainty assessment for the national and regional collocation options, respectively.

#### TABLE 8-20 RAIL TRANSPORT SEQUENCE 1

RC1 - A train accident occurs and crush forces fail OFC and munition: the sequence frequency is the product of the events: BE33, BE68, and BE53.

| Event<br>No. | Name                                     | Probability                        | Reference/Remarks                                                                                                                                                           |
|--------------|------------------------------------------|------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BE33         | Train accident<br>occurs                 | 5.5 x 10 <sup>-6</sup><br>per mile | Derived by H&R from Refs. 8-1<br>and 8-14.                                                                                                                                  |
| BE68         | Crush force                              | 0.002                              | Reference 8-1, for small                                                                                                                                                    |
|              | generated                                |                                    | packages.                                                                                                                                                                   |
| BE53         | Crush force<br>fails OFC and<br>munition | £                                  | From SNL rail crush curve<br>Fig. 8-8; at 160,000 lb,<br>probability approaches zero.<br>OFC is designed to withstand<br>520,000 lb of evenly dis-<br>tributed static load. |

PARAMETER CONTRACTOR

فكححط

8-62

### TABLE 8-21RAIL TRANSPORT SEQUENCE 2

RC2 - A train accident occurs and impact forces fail OFC and munition: the sequence frequency is the product of the events: BE33, BE60, BE52, and BE61.

| No.  | Name                           | Probability                        | Reference/Remarks                                                                                                                                                                                                                                                                |
|------|--------------------------------|------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BE33 | Train accident<br>occurs       | 5.5 x 10 <sup>-6</sup><br>per mile | Derived by H&R from Refs. 8-1<br>and 8-14.                                                                                                                                                                                                                                       |
| BE60 | Impact force<br>generated      | 1                                  | Reference 8-2.                                                                                                                                                                                                                                                                   |
| BE52 | Impact force<br>fails OFC      | 0.003                              | Figure 8-8.                                                                                                                                                                                                                                                                      |
| BE61 | Impact force<br>fails munition | E                                  | All munitions fail at thresh-<br>olds greater than 35 mph.<br>Impact energy is absorbed dur-<br>ing package failure and insuf-<br>ficient force is left to fail<br>the munition. Per Ref. 8-1,<br>small packages experience no<br>impact greater than during<br>normal handling. |

7

ഹരം

## TABLE 8-22RAIL TRANSPORT SEQUENCE 3

RC3 - A train accident occurs and puncture forces fail the container and munition: the sequence frequency is the product of the events: BE33, BE64, and BE51.

| Event<br>No. | Name                                                            | Probability                        | Reference/Remarks                                |
|--------------|-----------------------------------------------------------------|------------------------------------|--------------------------------------------------|
| BE33         | Train accident<br>occurs                                        | 5.5 x 10 <sup>-6</sup><br>per mile | Derived by H&R from Refs. 8-1<br>and 8-14.       |
| BE64         | Probe generated<br>capable of punc-<br>turing 0.75-in.<br>steel | 5.9 x $10^{-4}$                    | The probe is defined as a railcar coupler.       |
| BE51         | Probe fails OFC<br>and munition<br>(>200 s <sup>-1</sup> )      | 1                                  | This is conservative for heavy walled munitions. |

Note: The probability of puncture and fine is insignificant in comparison with puncture alone or fire-only events.

#### TABLE 8-23RAIL TRANSPORT SEQUENCE 4

RC4 - A train accident occurs with subsequent fire and accident forces cause detonation of burstered munitions (types C, P, M, R). Detonation may be due to fire only, impact plus fire, or undue force. The sequence frequency is given by the equation (BE33) (BE62) (BE63) + (BE33) (BE52') (BE62) (BE63A) + (BE33) (BE53).

Event No. Name Probability Reference/Remarks **BE33**  $5.5 \times 10^{-6}$ Train accident Derived by H&R from Refs. 8-1 occurs per mile and 8-14. **BE62**  $3.9 \times 10^{-4}$ Fire generated Derived by H&R from Refs. 8-1 and 8-13. BE63 Fire has heat 0.16 Figure 8-7. (1850°F) and duration to detonate burstered munition in OFC (>2 h) BE52' Mechanical 0.01 Ref. 8-6. forces destroy OFC insulation BE63A Fire has heat SNL train fire duration curve, (1475°F) and Fig. 8-7. duration to detonate burstered munition in degraded package ()30 min) C 49 min 0.55 P 89 min 0.3 M 68 min 0.4 R 10.5 min 0.75 BE53 Undue force  $8.3 \times 10^{-6}$ Reference 8-6. detonates burstered munitions

#### TABLE 8-24RAIL TRANSPORT SEQUENCE 5

RC5 - A train accident occurs with subsequent fire and accident forces fail nonburstered munitions, types B, S, K. The sequence frequency is the product of events: BE33, BE62, and BE75, added to the product of events: BE33, BE62, BE52', and BE75A.

Event Reference/Remarks No. Name Probability  $5.5 \times 10^{-6}$ BE33 Derived by H&R from Refs. 8-1 Train accident and 8-14. occurs per mile  $3.9 \times 10^{-4}$ BE62 Fire generated Derived by H&R from Refs. 8-1 and 8-13. **BE75** Fire has heat 0.16 Figure 8-7. (1850°F) and duration to fail nonburstered munition in OFC (>2 h) BE52 Mechanical 0.01 Ref. 8-6. forces destroy OFC insulation BE75A Fire has heat 0.16 SNL train fire duration curve, (1475°F) and Fig. 8-7. duration to fail nonburstered munitions in degraded package (>30 min)

## TABLE 8-25RAIL TRANSPORT SEQUENCE 6

RC6 - Aircraft crashes onto railcar, no fire; impact force fails OFC and munition: the sequence frequency is the product of events: BE33, BE60, BE52, and BE61. SUMMER RECEIPT BARRED

| No.  | Name                                   | Probability                         | Reference/Remarks                                             |
|------|----------------------------------------|-------------------------------------|---------------------------------------------------------------|
| BE33 | Aircraft crash                         | 3.1 x 10 <sup>-11</sup><br>per year | See Section 4.2.                                              |
| BE60 | Impact-only<br>force generated         | 0.55                                | <b>Ref. 8-1; 0.27</b> impact only/0.49<br>all impacts = 0.55. |
| BE52 | Impact force<br>fails OFC<br>(>35 mph) | 1                                   | Assumes a severe crash<br>(>300 mph)                          |
| BE61 | Impact force<br>fails munition         | 1                                   | Conservative for heavy-walled munitions.                      |



## TABLE 8-26RAIL TRANSPORT SEQUENCE 7

RC7 - Aircraft crashes onto railcar, fire occurs, impact-induced failure of OFC and munitions: the sequence frequency is the product of the events: BE33, BE60/62, BE52, and BE61.

| No.         | Name                                   | Probability                         | Reference/Remarks                                      |
|-------------|----------------------------------------|-------------------------------------|--------------------------------------------------------|
| BE33        | Aircraft crash                         | 3.1 x 10 <sup>-11</sup><br>per year | See Section 4.2.                                       |
| BE60/<br>62 | Impact and fire generated              | 0.45                                | Ref. 8-1; 0.22/0.49 fire and impact/all impact = 0.45. |
| BE52        | Impact force<br>fails OFC<br>(>35 mph) | 1                                   | Assumes a severe crash<br>(>300 mph).                  |
| BE61        | Impact force<br>fails munition         | 1                                   | Conservative for heavy-walled munitions.               |

#### TABLE 8-27RAIL TRANSPORT SEQUENCE 9

RC9 - An earthquake occurs and crush forces fail the OFC and the munition: the sequence frequency is the products of events: BE33, BE68, and BE53. 222222

ASSESS REAR

**\*\*\***\*\*\*\*

| No.  | Name                                                                                                                                                                                       | Probability                                                                                                                                                                                | Reference/Remarks                                                                                                                                           |
|------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BE33 | Earthquake occurs<br>(per year)<br>UDA to TEAD<br>PUDA to TEAD<br>LBAD to TEAD<br>APG to TEAD<br>ANAD to TEAD<br>NAAP to TEAD<br>PBA to TEAD<br>PBA to ANAD<br>APG to ANAD<br>NAAP to ANAD | 8.2 x $10^{-4}$<br>2.5 x $10^{-4}$<br>5.1 x $10^{-4}$<br>1.9 x $10^{-4}$<br>2.4 x $10^{-4}$<br>2.2 x $10^{-4}$<br>2.8 x $10^{-4}$<br>5.3 x $10^{-4}$<br>2.0 x $10^{-4}$<br>3.0 x $10^{-4}$ | Route- and site-specific;<br>derived by H&R from basic<br>earthquake probability value<br>(GA) and summing length of time<br>train is in each seismic zone. |
| BE68 | Crush force<br>generated                                                                                                                                                                   | 0.002                                                                                                                                                                                      | SNL, for small packages.                                                                                                                                    |
| BE53 | Crush force<br>fails OFC                                                                                                                                                                   | ε                                                                                                                                                                                          | See sequence 1.                                                                                                                                             |

 $\hat{X}$ 

•

## TABLE 8-28RAIL TRANSPORT SEQUENCE 10

RC10 - An earthquake occurs and impact forces fail the OFC and munition: the sequence frequency is the product of the events: BE33, BE60, BE52, and BE61.

| No.  | Name                                                     | Probability       | Reference/Remarks                                                                |
|------|----------------------------------------------------------|-------------------|----------------------------------------------------------------------------------|
| BE33 | Earthquake occurs                                        | See<br>sequence 9 | Route- and site-specific<br>derived by H&R from basic<br>earthquake probability. |
| BE60 | Impact force<br>generated                                | 1                 | Reference 8-2.                                                                   |
| BE52 | Impact force suf-<br>ficient to fail<br>munition and OFC | 0.003             | Figure 8-8.                                                                      |
| BE61 | Impact force<br>fails agent con-<br>tainment.            | ε                 | See sequence 2.                                                                  |

**P**-----



\$~\$\$\*\_\$\$\*\_\$<u>\$</u>\*\$\*\_\$

\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*

10

## TABLE 8-29RAIL TRANSPORT SEQUENCE 11

Level V

S.S.S.S.

RC11 - An earthquake occurs and puncture forces fail the munition and OFC: the sequence frequency is the product of events: BE33, BE64, and BE51.

| Event<br>No. | Name                                                            | Probability            | Reference/Remarks                                                                                   |
|--------------|-----------------------------------------------------------------|------------------------|-----------------------------------------------------------------------------------------------------|
| BE33         | Earthquake occurs                                               | See<br>sequence 9      | Route- and site-specific;<br>derived by H&R from basic<br>earthquake probability in<br>Section 4.2. |
| BE64         | Probe generated<br>capable of punc-<br>turing 0.75-in.<br>steel | 5.9 x 10 <sup>-4</sup> | Reference 8-2.                                                                                      |
| BE51         | Probe fails OFC<br>(>200 s <sup>-1</sup> )                      | 1                      | Conservative for heavy-walled munitions.                                                            |





#### TABLE 8-30RAIL TRANSPORT SEQUENCE 12

0000

RC12 - An earthquake occurs causing a train accident with fire, resulting in detonation of burstered munitions (types C, P, M, R). The sequence frequency is given by the equation (BE33) (BE62) (BE63) + (BE33) (BE62) (BE52') (BE63A)

| Event<br>No. | Name                                                                                                      | Probability       | Reference/Remarks                                                                 |
|--------------|-----------------------------------------------------------------------------------------------------------|-------------------|-----------------------------------------------------------------------------------|
| BE33         | Earthquake occurs                                                                                         | See<br>sequence 9 | Route- and site-specific;<br>derived by H&R from basic<br>earthquake probability. |
| BE62         | Fire generated<br>(given a train<br>accident)                                                             | 3.9 x 10-4        | Derived by H&R from Refs. 8-1<br>and 8-13.                                        |
| BE63         | Fire has heat<br>(1850°F) and<br>duration to<br>detonate<br>burstered<br>munition<br>in OFC<br>(>2 h)     | 0.16              | Figure 8-7.                                                                       |
| BE52'        | Mechanical<br>forces destroy<br>OFC insulation                                                            | 0.01              | Ref. 8-6.                                                                         |
| BE63A        | Fire has heat<br>(1475°F) and<br>duration to fail<br>burstered muni-<br>tion in degraded<br>OFC (>30 min) | See<br>sequence 4 | SNL train fire duration curve;<br>Fig. 8-7.                                       |

## TABLE 8-31RAIL TRANSPORT SEQUENCE 13

RC13 - An earthquake occurs causing a train accident with fire, resulting in failure of nonburstered munitions (types B, S, K): the sequence frequency is given by the equation: (BE33) (BE62) (BE75) + (BE33) (BE62) (BE52') (BE75A)

| No.   | Name                                                                                             | Probability          | Reference/Remarks                                                                                             |
|-------|--------------------------------------------------------------------------------------------------|----------------------|---------------------------------------------------------------------------------------------------------------|
| BE33  | Earthquake occurs                                                                                | See<br>sequence 9    | Route- and site-specific<br>derived by H&R from basic<br>earthquake probability; site-<br>and route-specific. |
| BE62  | Fire generated                                                                                   | $3.9 \times 10^{-4}$ | Derived by H&R from Refs. 8-1<br>and 8-13.                                                                    |
| BE75  | Fire has heat<br>(1850°F) and<br>duration to fail<br>nonburstered<br>munition in OFC<br>(>2 h)   | 0.16                 | Figure 8-7.                                                                                                   |
| BE52' | Mechanical<br>forces destroy<br>OFC insulation                                                   | 0.01                 | Ref. 8-6.                                                                                                     |
| BE75A | Fire has heat<br>(1475°F) and<br>duration to fail<br>nonburstered<br>munition in<br>degraded OFC | See<br>sequence 4    | SNL train fire duration curve,<br>Fig. 8-7.                                                                   |



L'ALLANDO

### TABLE 8-32RAIL TRANSPORT SEQUENCE 14

RC14 - (A) A tornado-generated missile penetrates munition: the sequence frequency is the product of events: BE33, BE64, and BE51, or

 (B) Tornado-generated derailment with mechanical forces occurs causing OFC/munition failure: the sequence frequency is given by the equation (BE33) (BE68, 60, or 64A) (BE53, 52, or 51A)

| No.   | Name                                                              | Probability                                                                                    | Reference/Remarks                                                      |
|-------|-------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------|
| BE33  | Tornado occurs<br>(winds<br>>160 mph)                             | Table 8-19                                                                                     | Route- and site-specific;<br>derived by H&R.                           |
| BE64  | Probe generated<br>capable of fail-<br>ing agent con-<br>tainment | UMDA<br>$1.7 \times 10^{-4}$<br>PUDA<br>$3.0 \times 10^{-3}$<br>Others<br>$1.4 \times 10^{-3}$ | Fraction of winds >160 mph that are also >310 mph.                     |
| BE51  | Missile has<br>orientation to<br>fail agent con-<br>tainment      | 5.3 x 10 <sup>-5</sup>                                                                         | Appendix C.                                                            |
| BE68  | Crush force<br>generated                                          | 0.002                                                                                          | Reference 8-2.                                                         |
| BE60  | Impact force<br>generated                                         | 1                                                                                              | Reference 8-2.                                                         |
| BE64A | Probe generated                                                   | 5.9 x $10^{-4}$                                                                                | Probe is defined as a railcar coupler.                                 |
| BE53  | Crush fails OFC                                                   | ε                                                                                              | Same as sequence 1.                                                    |
| BE52  | Impact fails<br>OFC and munition                                  | £                                                                                              | Same as sequence 2.                                                    |
| BE51A | Probe fails<br>agent contain-<br>ment                             | 1                                                                                              | Same as sequence 3, conserva-<br>tive for heavy-walled muni-<br>tions. |

#### TABLE 8-33RAIL TRANSPORT SEQUENCE 15

200 D 0 0 0 0 0 0

RC15 - A train accident occurs as a result of an earthquake or tornado, generating undue mechanical forces which cause detonation of burstered munitions: the sequence frequency is the product of events BE33 and BE53 added to the product of event BE33A and BE53.

| No.   | Name                                                       | Probability            | Reference/Remarks        |
|-------|------------------------------------------------------------|------------------------|--------------------------|
| BE33  | Earthquake occurs                                          | Table 8-19             | Derived by H&R Ref. 8-4. |
| BE33A | Tornado occurs                                             | Table 8-19             | Based on Section 4.2.    |
| BE53  | Crush force<br>sufficient to<br>detonate burster<br>in OFC | 8.3 x 10 <sup>-6</sup> | Ref. 8-6.                |



Ľ,

TABLE 8-34 OFFSITE RAIL TRANSPORTATION - NATIONAL DISPOSAL OPTION

Accident Frequencies and Range Factors

| ANGE<br>ACTOR                                                                                 | 10            | :                        | ١.                       | ł                        | ;                 | 9                               | 1                     | 01                       | 9                        | ł                               | 10                              | 10                              | 2                 | 9                               | 2                               | 2                 | ſ                        | ł                        | :                        | 21                       | 27                       | 1                               | 27                              | 27                              | 27                | 21                              | 21                              | 11                | 1                 | 11                              |
|-----------------------------------------------------------------------------------------------|---------------|--------------------------|--------------------------|--------------------------|-------------------|---------------------------------|-----------------------|--------------------------|--------------------------|---------------------------------|---------------------------------|---------------------------------|-------------------|---------------------------------|---------------------------------|-------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|---------------------------------|---------------------------------|---------------------------------|-------------------|---------------------------------|---------------------------------|-------------------|-------------------|---------------------------------|
| UNDA F                                                                                        | .2E-09        | N/A                      | N/A                      | N/A                      | N/A               | 5. 2E -09                       | N/A                   | (.2E-09                  | 5. 2E-09                 | N/A                             | 5.2E-09                         | (. 2E-09                        | i. 2E-09          | 1.25-09                         | 2E-09                           | i. 2E-09          | N/A                      | N/A                      | N/A                      | I.4E-10                  | I.4E-10                  | N/A                             | I.4E-10                         | 1.46-10                         | 1.46-10           | 1.56-10                         | 1.5E-10                         | 1.56-10           | N/A               | 6.56-10                         |
| RANGE<br>Factor                                                                               |               | ł                        | ł                        | ł                        | ł                 | ۳<br>۱                          | 1                     | -                        | -                        | ;                               | +                               | ;                               |                   | *                               |                                 | ;                 | 1                        | !                        | 1                        | -                        | -                        | ;                               | +                               | -                               | -                 | +                               | +                               | -                 | !                 | 1                               |
| TEAD<br>Freq.                                                                                 | N/A           | N/A                      | N/A                      | N/A                      | N/A               | N/A                             | N/A                   | N/A                      | N/A                      | W/A                             | N/A                             | N/A                             | N/A               | N/A                             | N/A                             | N/A               | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                             | N/A                             | N/A                             | N/A               | N/A                             | N/A                             | N/A               | N/A               | N/A                             |
| RANGE<br>Factor                                                                               | :             | 10                       | 1                        | 10                       | ;                 | !                               | 1                     | 1                        | !                        | 9                               | ł                               | ļ                               | ł                 | 1                               | 1                               | 1                 | 53                       | !                        | 53                       | ;                        | 1                        | 21                              | ;                               | 1                               | :                 | 1                               | :                               | ł                 | 1                 | ł                               |
| PUDA<br>Freg.                                                                                 | N/A           | 3.2E-09                  | N/A                      | 3. 26-09                 | N/A               | N/A                             | N/A                   | M/A                      | N/A                      | 3.2E-09                         | N/A                             | N/A                             | N/A               | N/A                             | N/A                             | N/A               | 4.4E-10                  | N/A                      | 4.4E-10                  | N/A                      | N/A                      | 4.4E-10                         | A/A                             | 8/N                             | N/A               | N/A                             | A/A                             | N/A               | N/A               | N/A                             |
| range<br>Factor                                                                               | 1             | 1                        | ł                        | ;                        | ł                 | 01                              | 1                     | 10                       | ;                        | 1                               | ;                               | 1                               | !                 | 01                              | 01                              | 1                 | ;                        | ł                        | ;                        | 21                       |                          | 1                               | ;                               | :                               | :                 | 21                              | 21                              | ł                 | ;                 | 41                              |
| FBA<br>Fred.                                                                                  | N/A           | N/A                      | N/A                      | N/A                      | N/A               | 3.2E-09                         | N/A                   | 3.2E-ù9                  | N/A                      | N/A                             | N/A                             | N/A                             | N/A               | 3.2E-09                         | 3.2E-09                         | N/A               | N/A                      | N/A                      | N/A                      | 4.46-10                  | N/A                      | N/A                             | N/A                             | N/A                             | N/A               | 4.5E-10                         | 4.56-10                         | N/A               | N/A               | 3.5E-10                         |
| ANGE<br>ACTOR                                                                                 | :             | ;                        | :                        | :                        | ;                 | ;                               | 10                    | 1                        | ł                        | ł                               | !                               | ł                               | ł                 | 1                               | 1                               | ;                 | ţ                        | 1                        | ſ                        | 1                        | ;                        | :                               | :                               | ſ                               | {                 | {                               | {                               | ?                 | í                 | {                               |
| i                                                                                             |               |                          |                          |                          |                   |                                 |                       |                          |                          |                                 |                                 |                                 |                   |                                 |                                 |                   |                          |                          |                          |                          |                          |                                 |                                 |                                 |                   |                                 |                                 |                   |                   |                                 |
| NAAP R                                                                                        | N/A           | N/A                      | N/A                      | N/A                      | N/A               | N/A                             | 5.2E-09               | N/A                      | N/A                      | N/A                             | N/A                             | N/A                             | N/A               | N/A                             | N/A                             | N/A               | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                             | N/A                             | N/A                             | N/A               | N/A                             | N/A                             | N/A               | N/A               | N/A                             |
| RANGE NAAP RI<br>Factor Fred. Fi                                                              | N/A           | N/A                      | N/A                      | N/A                      | N/A               | N/A                             | 3.2E-09               | N/A                      | N/A                      | 10 N/A                          | 10 N/A                          | 10 N/A                          | N/A               | 10 M/A                          | 10 N/A                          | N/A               | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | 27 N/A                          | 27 N/A                          | 27 N/A                          | N/A               | 21 N/A                          | 21 N/A                          | N/A               | A.A.              | N/A                             |
| LBAD RANGE NAAP RI<br>Faco. Factor Fred. Fi                                                   | N/A N/A       | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A           | N/A N/A                         | N/A 3.2E-09           | N/A N/A                  | N/A N/A                  | 5.2E-09 10 N/A                  | 3.2E-09 10 N/A                  | 5.2E-09 10 N/A                  | N/A N/A           | 5.2E-09 10 M/A                  | 5.2E-09 10 N/A                  | N/A N/À           | N/A N/A                  | 1.4E-10 27 N/A                  | 4.4E-10 27 N/A                  | J.4E-10 27 N/A                  | N/A N/A           | 1.5E-10 21 N/A                  | 4.5E-10 21 N/A                  | N/A N/A           | N/A N/A           | N/A N/A                         |
| RAMGE LBAD RANGE NAAP R<br>Factor Freg. Factor Freg. F                                        | N/A N/A       | N/A N/A                  | N/A N/A                  | N/A N/A                  | N/A N/A           | 10 N/A N/A                      | N/A 3.2E-09           | N/A N/A                  | N/A N/A                  | 3.2E-09 10 N/A                  | 3.2E-09 10 N/A                  | 3.2E-09 10 N/A                  | N/A N/A           | 3.2E-09 10 M/A                  | 3.2E-09 10 N/A                  | N/A N/À           | N/A N/A                  | 4.4E-10 27 M/A                  | 4.4E-10 27 N/A                  | 4.4E-10 27 N/A                  | N/A N/A           | 4.5E-10 21 N/A                  | 4.5E-10 21 N/A                  | N/A N/A           | N/A N/A           | 47 N/A N/A                      |
| AFG RÅNGE LBAD RANGE NAAP R<br>Fred. Factor Fred. Factor Fred. Fi                             | N/A N/A N/A   | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A              | N/A N/A N/A       | 5.2E-09 10 N/A N/A              | N/A N/A 3.2E-09       | N/A N/A N/A              | N/A N/A N/A              | N/A 3.2E-09 10 N/A              | N/A 3.2E-09 10 N/A              | N/A 3.2E-09 10 N/A              | N/A N/A N/A       | N/A 3.2E-09 10 N/A              | N/A 3.2E-09 10 N/A              | N/A N/A N/À       | N/A N/A N/A              | N/A 4.4E-10 27 N/A              | N/A 4.4E-10 27 N/A              | N/A 4.4E-10 27 N/A              | N/A N/A N/A       | N/A 4.5E-10 21 N/A              | N/A 4.5E-10 21 N/A              | N/A N/A N/A       | N/A N/A N/A       | 3.5E-10 47 N/A N/A              |
| KANGE AFG RÅNGE LBAD RÅNGE NÅAP R<br>Factor fred. factor fred. factor fred. f                 | N/A N/A       | 10 N/A N/A N/A           | 10 N/A N/A N/A           | 10 N/A N/A N/A           | N/A N/A N/A       | 10 3.2E-09 10 N/A N/A           | N/A N/A 3.2E-09       | 10 N/A N/A N/A           | 10 N/A N/A N/A           | 10 N/A 3.2E-09 10 N/A           | 10 N/A 3.2E-09 10 N/A           | 10 N/A 3.2E-09 10 N/A           | N/A N/A N/A       | 10 N/A 3.2E-09 10 N/A           | 10 N/A 3.2E-09 10 N/A           | N/A N/A N/A       | 23 N/A N/A N/A           | 23 N/A N/A N/A           | 23 N/A N/A N/A           | 21 N/A N/A N/A           | 27 N/A N/A N/A           | 27 N/A 4.4E-10 27 N/A           | 27 N/A 4.4E-10 27 N/A           | 27 N/A 4.4E-10 27 N/A           | N/A N/A N/A       | 21 N/A 4.5E-10 21 N/A           | 21 N/A 4.5E-10 21 N/A           | N/A N/A N/A       | N/A N/A           | 47 3.5E-10 47 N/A N/A           |
| ANAO KANGE AFG RANGE LBAD RANGE NAAP R<br>Freq. Factor Fred. Factor Freq. Factor Free. F      | N/A N/A N/A   | 3.2E-09 10 N/A N/A N/A   | 3.2E-U9 10 N/A N/A N/A   | 3.2E-09 10 N/A N/A N/A   | N/A N/A N/A N/A   | 3.2E-09 10 3.2E-09 10 N/A N/A   | N/A N/A N/A 3.2E-09   | 3.2E-09 10 N/A N/A N/A   | 3.2E-09 10 N/A N/A N/A   | 3.2E-09 10 N/A 3.2E-09 10 N/A   | 3.2E-09 10 N/A 3.2E-09 10 N/A   | 3.2E-09 10 N/A 3.2E-09 10 N/A   | N/A N/A N/A N/A   | 3.2E-09 10 N/A 3.2E-09 10 N/A   | 3.2E-U9 10 N/A 3.2E-U9 10 N/A   | N/A N/A N/A N/A   | 4.4E-10 23 N/A N/A N/A   | 4.4E-10 23 N/A N/A N/A   | 4.4E-10 23 N/A N/A N/A   | 4.4E-10 21 N/A N/A N/A   | 4.4E-10 27 N/A N/A N/A   | 4.4E-10 27 N/A 4.4E-10 27 N/A   | 4.4E-10 27 N/A 4.4E-10 27 N/A   | 4.4E-10 27 N/A 4.4E-10 27 N/A   | N/A N/A N/A N/A   | 4.5E-10 21 N/A 4.5E-10 21 N/A   | 4.5E-10 21 N/A 4.5E-10 21 N/A   | N/A N/A N/A N/A   | N/A N/A N/A N/A   | 3.5E-10 47 3.5E-10 47 N/A N/A   |
| 40, ANAJ RANGE AFG RÀNGE LBAD RANGE NAAP R<br>Freq. Factor Freq. Factor Freq. Factor Free. Fi | 3 N/A N/A N/A | 3 3,2E-09 10 N/A N/A N/A | 3 3.2E-09 10 N/A N/A N/A | 3 3.2E-09 10 N/A N/A N/A | 3 N/A N/A N/A N/A | 3 3.2E-09 10 3.2E-09 10 N/A N/A | 3 N/A N/A N/A 3.2E-09 | 3 3.2E-09 10 N/A N/A N/A | 3 3.2E-09 10 N/A N/A N/A | 3 3.2E-09 10 N/A 3.2E-09 10 N/A | 3 J.2E-09 10 N/A J.2E-09 10 N/A | 3 3.2E-09 10 N/A 3.2E-09 10 N/A | 3 N/A N/A N/A N/A | 3 3.2E-09 10 N/A 3.2E-09 10 N/A | 3 3.2E-U9 10 N/A 3.2E-09 10 N/A | 3 N/A N/A N/A N/A | 4 4.4E-10 23 N/A N/A N/A | 4 4.4E-10 23 N/A N/A N/A | 4 4.4E-10 23 N/A N/A N/A | 4 4.4E-10 21 N/A N/A N/A | 4 4.4E-10 27 N/A N/A N/A | 4 4.4E-10 27 N/A 4.4E-10 27 N/A | 4 4.4E-10 27 N/A 4.4E-10 27 N/A | 4 4.4E-10 27 N/A 4.4E-10 27 N/A | 4 N/A N/A N/A N/A | 4 4.5E-10 21 N/A 4.5E-10 21 N/A | 4 4.5E-10 21 N/A 4.5E-10 21 N/A | 5 N/A N/À N/A N/A | 5 N/A N/A N/A N/A | 5 3.5E-10 47 3.5E-10 47 N/A N/A |

See notes at end of table.

「「「「「」」のためので、







TABLE 8-34 (Continued)

## ccident Frequencies and Range Factors

| RANGE<br>FACTOR | ;       | 10 47 | 11 20 | :       | 1       | ;       | ł   | 11 20    | ł       | 11 20   | 11 20      | ł       | 11 20   | 11 20   | 11 20  | 11 20   | 11 20            | 11 20 | 11 20 | ł       | :       | 1       | 1   | 11 20   | :       | 11 20   | 11 20          | ;       | 11 20   | 11 20   |
|-----------------|---------|-------|-------|---------|---------|---------|-----|----------|---------|---------|------------|---------|---------|---------|--------|---------|------------------|-------|-------|---------|---------|---------|-----|---------|---------|---------|----------------|---------|---------|---------|
| LINDA<br>FRED   | N/A     | 3.56- | 1.76- | N/A     | N/A     | N/A     | N/A | 1.75-    | N/A     | 1. 7E-  | ו. אָי     | N/A     | 1.7E-   | 1.75-   | I. 7E- | 1.7E-   | 1. <i>T</i> E-   | 1.76- | ÷.    | N/A     | N/A     | N/A     | N/A | 1.45-   | N/A     | 1.4-    | - <del>1</del> | N/A     | 1.45-   | 1.45-   |
| FACTOR          | ł       | 1     | ;     | :       | ł       | ł       | ÷   | ;        | ;       | ł       | :          | ;       | 1       | ۲       | ł      | ;       | !                | :     | 1     | ;       | 1       | ł       | 1   | :       | :       | !       | ţ              | ;       | ;       | :       |
| TEAD<br>FRED.   | N/A     | N/A   | N/A   | N/A     | N/A     | N/A     | N/A | N/A      | N/A     | N/A     | N/A        | N/A     | N/A     | A/A     | N/A    | N/A     | N/A              | N/A   | N/A   | N/A     | A/A     | N/A     | N/A | N/A     | N/A     | N/A     | N/A            | N/A     | N/A     | A/A     |
| FACTOR          | ł       | :     | ;     | 20      | :       | 20      | ;   | ;        | :       | !       | ;          | 20      | ;       | :       | ;      | :       | ;                | 1     | :     | 20      | :       | 20      | ;   | ł       | ł       | ;       | ;              | 20      | ;       | :       |
| PUDA<br>FREQ.   | N/A     | N/A   | N/A   | 1.7E-11 | N/A     | 1.7E-11 | N/A | ۲/N      | N/A     | N/A     | N/A        | 1.76-11 | N/A     | N/A     | N/A    | N/A     | N/A              | N/A   | N/A   | 1.4E-11 | N/A     | 1.46-11 | N/A | N/A     | N/A     | N/A     | N/A            | 1.4E-11 | N/A     | M/A     |
| FACTOR          | !       | ;     | ł     | ;       | ł       | :       | :   | 2        | ;       | 20      | :          | !       | 1       | ł       | ł      | 20      | 20               | 1     | ł     | :       | ł       | :       | 1   | 20      | :       | 20      | 1              | :       | ł       | :       |
| PBA<br>Freq.    | N/A     | N/A   | N/A   | N/A     | N/A     | N/A     | N/A | 1.7E-11  | N/A     | 1.7E-11 | N/A        | N/A     | A/A     | N/A     | N/A    | 1.7E-11 | 1.7E-11          | M/A   | N/A   | N/A     | N/A     | N/A     | N/A | 1.4E-11 | N/A     | 1.4E-11 | N/A            | N/A     | N/A     | N/A     |
| RANGE<br>Factor |         | ł     | ł     | ;       | ł       | ł       | ;   | ;        | 20      | :       | ł          | ;       | 1       | ł       | ł      | ł       | ţ                | ł     | :     | ;       | ł       | ł       | ł   | ł       | 20      | ł       | ł              | ł       | 1       | ;       |
| NAAP<br>FREQ.   | 3.5E-10 | M/A   | N/A   | N/A     | A/A     | N/A     | A/A | N/A      | 11-37.1 | N/A     | M/A        | N/A     | N/A     | N/A     | N/A    | N/A     | N/A              | N/A   | N/A   | N/A     | N/A     | N/A     | N/A | N/A     | 1.4E-11 | N/A     | N/A            | N/A     | N/A     | N/A     |
| RANGE<br>FACTOR | ;       | :     | ł     | ł       | :       | :       | ;   | ;        | ;       | !       | ;          | 20      | 20      | 20      | 1      | 20      | 20               | :     | 1     | 1       | :       | 1       | ł   | ;       | !       | ;       | ;              | 20      | 20      | 50      |
| LBAD<br>Freg.   | N/A     | N/A   | N/A   | N/A     | N/A     | N/A     | N/A | N/A      | N/A     | A/A     | N/A        | 1.7E-11 | 1.7E-11 | 1.76-11 | N/A    | 1.7E-11 | 1.7E-11          | N/A   | N/A   | N/A     | N/A     | N/A     | N/A | N/A     | N/A     | N/A     | N/A            | 1.4E-11 | 1.4E-11 | 1.4E-11 |
| RANGE<br>FACTOR |         | ł     | 1     | 1       | !       | 1       | 1   | 20       | 1       | ł       | ;          | 1       | !       | 1       | ;      | ł       | ;                | 1     | ł     | ł       | !       | ł       | ;   | 20      | ł       | 1       | 1              | 1       | ;       | ;       |
| APG.<br>Freg.   | N/A     | N/A   | N/A   | N/A     | N/A     | N/A     | N/A | 1.7E-11  | N/A     | M/A     | N/A        | N/A     | N/A     | Ň/Å     | N/A    | N/A     | N/A              | N/A   | N/A   | N/A     | N/A     | N/A     | N/A | 1.4E-11 | N/A     | N/A     | A/A            | N/A     | N/A     | N/A     |
| RANGE<br>Factor | :       | :     | 1     | 20      | 2ú      | 20      | :   | 20       | ;       | 20      | 20         | 26      | 30      | 20      | ;      | 20      | 20               | 1     | 1     | 20      | 20      | 20      | ;   | 20      | :       | 20      | 20             | 20      | 50      | 20      |
| ANAD<br>Fred.   | N/A     | N/A   | N/A   | 1.7E-11 | 1.7E-11 | 1.7E-11 | N/A | 1.7E-11  | N/A     | 1.7E-11 | 1. 7E - 11 | 1.76-11 | 1.7E-11 | 1.7E-11 | N/A    | 1.76-11 | 1. <i>T</i> E-11 | N/A   | M/A   | 1.46-11 | 11-34.1 | 1.46-11 | N/A | 1.46-11 | N/A     | 1.4E-11 | 1.4E-11        | 1.4E-11 | 1.4E-11 | 1.46-11 |
| NO.             | 67      | 'n    | 4     | 9       | 4       | -9      | -9  | <b>9</b> | 4       | 9       | 4          | -0      | 4       | •       | Ð      | -a      | -0               | -0    | ~     | ٢       | 1       | 1       | 1   | 1       | 1       | 1       | 1              | 1       | -       | '       |
|                 |         |       |       |         |         |         |     |          |         |         |            |         |         |         |        |         |                  |       |       |         |         |         |     | • •     |         |         |                |         |         |         |

See notes at end of table.

, Texes

122222

1937 **- 22** (23)
Page 3 Date 20-Aug-87

TABLE 8-34 (Continued)

Accident Frequencies and Range Factors

| UMDA RANGE<br>Freq. Factor | 1.46-11 20 | 1.4E-11 20 | 1.4E-11 20 | 1.46-11 20 | 4.8E-07 37 | N/A     | M/A     | N/A     | W/A   | 4.8E-07 37 | W/A   | 4.86-07 37 | 4.86-07 37 | N/A     | 4.86-07 37 | 4.86-07 37   | 4.8E-07 37 | 4.86-07 37   | 4.BE-07 37   | 4.8E-07 37 | N/A     | N/A     | M/A     | 5.3E-08 71 | 5.26-08 73 | M/A     | 5.2E-08 73 | 5.2E-08 73 |  |
|----------------------------|------------|------------|------------|------------|------------|---------|---------|---------|-------|------------|-------|------------|------------|---------|------------|--------------|------------|--------------|--------------|------------|---------|---------|---------|------------|------------|---------|------------|------------|--|
| RANGE<br>Factor            | :          | ł          | ;          | ;          | ;          | ł       | 1       | ł       | ;     | ł          | ł     | ;          | ł          | ł       | ł          | ;            | ;          | ;            | ;            | ł          | ;       | ;       | ;       | ;          | ;          | ;       | ţ          | ;          |  |
| TEAD<br>Fred.              | N/A        | N/A        | N/A        | N/A        | N/A        | N/A     | N/A     | N/A     | N/A   | N/A        | N/A   | N/A        | N/A        | N/A     | N/A        | N/A          | N/A        | N/A          | N/A          | N/A        | N/A     | N/A     | N/A     | N/A        | N/A        | N/A     | N/A        | N/A        |  |
| RANGE<br>Factor            |            | ;          | ;          | !          | !          | 37      | ;       | 37      | !     | ;          | ;     | :          | :          | 37      | ł          | ;            | ł          | ;            | ł            | ;          | 10      | 1       | 01      | :          | {          | 8       | ;          | !          |  |
| PUDA<br>Freq.              | N/A        | N/A        | N/A        | N/A        | N/A        | 1.5E-07 | N/A     | 1.56-07 | N/R   | N/A        | N/A   | N/A        | N/A        | 1.5E-07 | N/A        | N/A          | N/A        | N/A          | N/A          | N/A        | 1.66-08 | N/A     | 1.66-08 | N/A        | N/A        | 1.6E-08 | N/A        | N/A        |  |
| KANGE<br>Factor            | :          | 20         | 20         | ł          | ;          | ţ       | ł       | 1       | !     | 37         | ł     | 37         | 1          | 1       | ţ          | ;            | ł          | 37           | 37           | ł          | ł       | ł       | 1       | 61         | ł          | ;       | 1          | !          |  |
| PBA<br>Freq.               | N/A        | 1.46-11    | 1.4E-11    | N/A        | N/A        | N/A     | N/A     | A/A     | N/A   | 1.65-07    | N/A   | 1.6E-07    | N/A        | N/A     | N/A        | N/A          | N/A        | 1.65-07      | 1.6E-07      | N/A        | N/A     | N/A     | N/A     | 1.7E-08    | N/A        | N/A     | N/A        | N/A        |  |
| RANGE<br>Factor            |            | ļ          | :          | ţ          | 1          | ;       | :       | ł       | 1     | ;          | 37    | :          | ;          | ļ       | ł          | ł            | :          | ł            | 1            | ;          | ;       | 1       | 1       | ;          | ł          | ł       | 1          | :          |  |
| NAAP<br>Freq.              | N/A        | N/A        | N/A        | M/A        | N/A        | N/A     | N/A     | N/A     | N/A   | N/A        | 3E-07 | N/A        | N/A        | N/A     | N/A        | N/A          | N/A        | N/A          | N/A          | N/A        | N/A     | N/A     | N/A     | N/A        | N/A        | N/A     | N/A        | N/A        |  |
| RANGE<br>Factor            | 1          | 20         | 20         | ł          | ;          | :       | ;       | :       | ł     | ţ          | 1     | !          | ł          | 37      | 37         | 37           | ;          | 37           | 37           | ;          | !       | ;       | ł       | ł          | ł          | 105     | 105        | 105        |  |
| LBAD<br>Freq.              | N/A        | 11-34-11   | 1.46-11    | N/A        | N/A        | N/A     | N/A     | N/A     | N/A   | N/A        | N/A   | N/A        | N/A        | 3.0E-07 | 3.0E-07    | 3.06-07      | N/A        | 3.0E-07      | S. 0E-07     | N/A        | N/A     | N/A     | N/A     | N/A        | N/A        | 5.2E-08 | 3.2E-ÙB    | 5.26-08    |  |
| RANGE<br>FACTOR            |            | 1          | ;          | ;          | :          | :       | :       | :       | ;     | 37         | ;     | ;          | ł          | 1       | 1          | 1            | ł          | 1            | 1            | 1          | !       | ;       | 1       | !          | ;          | 1       | 1          | 1          |  |
| APG<br>Freq.               | N/A        | N/A        | N/A        | N/A        | N/A        | N/A     | N/A     | N/A     | N/A   | I.1E-07    | N/A   | N/A        | N/A        | N/A     | N/A        | N/A          | N/A        | N/A          | N/A          | N/A        | A/A     | N/A     | N/A     | N/A        | N/A        | N/A     | N/A        | N/A        |  |
| RANGE<br>Factor            |            | 20         | 20         | ł          | ;          | 37      | 37      | 37      | ł     | 37         | ł     | 37         | 37         | 37      | 37         | 37           | ł          | 37           | 37           | ł          | 96      | 96      | 96      | 105        | 100        | 100     | 100        | 106        |  |
| ANAD<br>Freg.              | N/A        | 1.46-11    | 1.46-11    | N/A        | N/A        | 1.4E-07 | 1.4E-07 | 1.4E-07 | N/A   | 1.4E-07    | N/A   | 1.4E-07    | 1.4E-07    | 1.4E-07 | 1.4E-07    | 1.4E-07      | N/A        | 1.4E-07      | 1.46-07      | N/A        | 1.65-08 | 1.6E-08 | 1.66-08 | 1.6E-08    | 1.5E-08    | 1.56-08 | 1.55-08    | 1.5E-08    |  |
|                            |            | 1          | 1          | 1          | Π          | Π       | 11      | Π       | 11    | Ξ          | Ξ     | Ξ          | 11         | 11      | Ξ          | 11           | 11         | 11           | 11           | 11         | 12      | 12      | 12      | 12         | 12         | 12      | 12         | 12         |  |
| SCEN-<br>Ario              | RCEVC      | ACREC      | RCRVC      | ACSVF      | RCB65      | RCDHS   | RCCGS   | RCCHS   | RCKGS | RUKHS      | RCKVS | RCMVS      | RCP6S      | RCPHS   | RCPVS      | <b>KCQGS</b> | RCEVS      | <b>KCR6S</b> | <b>RCRVS</b> | RCSVS      | RCDHC   | RCCGC   | RCCHC   | RCMVC      | RCFGC      | RCPHC   | RCPVC      | RC GGC     |  |



See notes at end of table.

12050





DOCT-MARCO

Page 4 Date 20-Aug-87

TABLE 8-34 (Continued)

### Accident Frequencies and Range Factors

|                                              | RANGE<br>Factor                                                                       | - 98                            | 69                | 1                 | 88                               | ţ                   | 68                   | 76                | ł                        | :                        | ł                        | {                 | 76                              | ł                   | 76                      | 76                      | ;                              | 76                             | 76                             | 76               | 76                             | 76                             | 76                | ł                        | ł                        | ;                        | 117                      | 117                      | :                              | 117                             | 117                             |
|----------------------------------------------|---------------------------------------------------------------------------------------|---------------------------------|-------------------|-------------------|----------------------------------|---------------------|----------------------|-------------------|--------------------------|--------------------------|--------------------------|-------------------|---------------------------------|---------------------|-------------------------|-------------------------|--------------------------------|--------------------------------|--------------------------------|------------------|--------------------------------|--------------------------------|-------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------------|---------------------------------|---------------------------------|
|                                              | UNDA<br>Freq.                                                                         | 5.4E-08                         | 5.2E-08           | N/A               | 5. IE-12                         | N/A                 | 5.2E-08              | 1.9E-10           | N/A                      | N/A                      | N/A                      | N/A               | 1.9E-10                         | N/A                 | 1.9E-10                 | 1.96-10                 | A/A                            | 1.9E-10                        | 1.9E-10                        | 1.96-10          | 1.9E-10                        | 1.9E-10                        | 1.96-10           | N/A                      | N/A                      | N/A                      | 1.36-08                  | 1.3E-08                  | N/A                            | 1.3E-08                         | 1.3E-08                         |
|                                              | RANGE<br>Factor                                                                       | ;                               | 1                 | ł                 | !                                | ;                   | ;                    | !                 | ł                        | ł                        | ł                        | 1                 | ;                               | :                   | ۲                       | ł                       | ł                              | ;                              | 1                              | ł                | ł                              | :                              | ł                 | 1                        | 1                        | :                        | !                        | :                        | ;                              | ł                               | ł                               |
|                                              | TEAD<br>Freq.                                                                         | N/A                             | N/A               | N/A               | N/A                              | N/A                 | N/A                  | N/A               | N/A                      | N/A                      | 8/N                      | N/A               | N/A                             | N/A                 | N/A                     | N/A                     | N/A                            | N/A                            | N/A                            | N/A              | N/A                            | N/N                            | N/A               | N/A                      | N/A                      | N/A                      | N/A                      | N/A                      | N/A                            | N/A                             | N/A                             |
|                                              | RANGE<br>Factor                                                                       |                                 | 1                 | 1                 | ;                                | 1                   | ł                    | ;                 | 16                       | ł                        | 16                       | :                 | 1                               | :                   | ł                       | 1                       | 16                             | 1                              | ł                              | ;                | ł                              | ł                              | ł                 | 8                        | 1                        | 83                       | ;                        | ;                        | 83                             | ;                               | 1                               |
|                                              | PUDA<br>Fred.                                                                         | N/A                             | N/A               | N/A               | N/A                              | N/A                 | N/A                  | N/A               | 2.2E-10                  | N/A                      | 2.2E-10                  | N/A               | N/A                             | N/A                 | N/A                     | N/A                     | 2.2E-10                        | N/A                            | N/A                            | N/A              | N/A                            | N/A                            | N/A               | 4.1E-09                  | N/A                      | 4.1E-09                  | N/A                      | N/A                      | 4.1E-09                        | N/A                             | N/A                             |
| -                                            | RANGE<br>Factor                                                                       | 98                              | ١                 | !                 | 82                               | ł                   | ł                    | ł                 | ł                        | !                        | ł                        | ¢<br>1            | 32                              | :                   | 95                      | 1                       | ł                              | 1                              | ;                              | ł                | 95                             | 95                             | 1                 | ł                        | ł                        | ţ                        | 91                       | ł                        | ;                              | ,<br>,                          | ł                               |
|                                              | PBA<br>Freq.                                                                          | 1.76-08                         | N/A               | N/A               | 1.7E-08                          | N/A                 | N/A                  | N/A               | N/A                      | N/A                      | N/A                      | N/A               | 4.2E-08                         | N/A                 | 4.26-08                 | N/À                     | N/A                            | N/A                            | N/A                            | N/A              | 4.2E-08                        | 4.26-08                        | N/A               | N/A                      | N/A                      | N/A                      | 5.56-09                  | N/A                      | N/A                            | N/A                             | N/A                             |
|                                              | ƙange<br>Factor                                                                       |                                 | ŀ                 | ;                 | ł                                | 69                  | ;                    | 1                 | !                        | ;                        | ł                        | ;                 | 1                               | 120                 | ;                       | ;                       | ł                              | 1                              | :                              | ł                | ;                              | ł                              | ł                 | !                        | 1                        | ł                        | ;                        | ł                        | 1                              | :                               | ;                               |
|                                              | ક છું                                                                                 | 4                               | A                 | Ā                 | A/                               | 80-1                | Å,                   | æ                 | æ                        | ¥                        | 4                        | æ                 | A                               | 8-                  | Å,                      | /A                      | /A                             | A/                             | ٩/                             | A/               | /A                             | A/                             | Æ                 | Æ                        | Æ                        | æ                        | æ                        | æ                        | Å,                             | < <b>I</b>                      | æ                               |
|                                              | NA)<br>Fr                                                                             | ÌÌ                              | Ì                 | ž                 | z                                | Ξ                   | ž                    | À                 | Ì                        | Ì                        | Ì                        | ž                 | Z                               | 3.98                | Ż                       | z                       | z                              | z                              | 2                              | ×                | z                              | z                              | Ż                 | Ì                        | Ì                        | ×                        | ì                        | ¥                        | ×                              | Ì                               | X                               |
|                                              | RANGE NAI<br>Factor Fri                                                               | 86 N/                           | ₩<br>             | 2                 | ł                                | <b>₩</b> .1<br>     | 2                    | ¥                 | /w<br>                   |                          | -                        | N                 | N                               | 3.95                | 1                       | ×                       | N (180                         | 80<br>N                        | 80 N                           | -                | B0 N                           | 80 N                           | /W                | N                        | ₩<br>                    | /W                       | ₩                        | N                        | 82 N                           | 82 N/                           | 82 N/                           |
|                                              | LBAD RANGE NAI<br>Fred. Factor Fri                                                    | 3.3E-0B 86 N/                   | N/A N/            | N/A N/            | N/A N                            | N/A 1.4             | N/A N/               | N/A N/            | N/A N/                   | N/A N/                   | N/A N/                   | N/A N/            | N/A N/                          | N/A 3.9E            | N/A N                   | N/A N/A                 | 4.3E-08 80 N                   | 4.3E-08 B0 N                   | 4.3E-08 80 N                   | N/A N            | 4. 3E - 08 80 N                | 4.3E-08 80 N                   | N/A N/            | N/A N/                   | N/A N/                   | N/A N/                   | N/A N/                   | N/A N/                   | 9.4E-09 82 NJ                  | 9.4E-09 82 N/                   | 9.4E-09 82 N/                   |
|                                              | RANGE LBAD RANGE NAI<br>Factor Fred. Factor Fri                                       | 3.3E-0B B6 N/                   | N/A N/            | N/A N/            | 118 N/A N                        | N/A 1.4             | N/A N/               | N/A N/            | N/A N/                   | N/A N/                   | N/A N/                   | N/A N/            | 23 N/A N/                       | N/A 3.9E            | N/A N                   | N/A                     | 4.3E-08 80 N                   | 4.3E-08 B0 N                   | 4.3E-08 80 N                   | N/A N            | 4.3E-08 80 N                   | 4.3E-08 80 N                   | N/A N/            | N/A N/                   | N/A N/                   | N/A N/                   | N/A N/                   | N/A N/                   | 9.4E-09 82 N                   | 9.4E-09 82 N/                   | 7.4E-09 82 N/                   |
|                                              | AFG RANGE LBAD RANGE NAI<br>Freq. Factor Freq. Factor Fri                             | N/A 3.3E-0B 86 N/               | N/A N/A N/        | N/A N/A N/        | 1.2E-08 118 N/A N                | N/A N/A 1.4         | N/A N/A +- N/        | N/A N/A N/        | N/A N/A N/               | N/A N/A N/               | N/A N/A N/               | N/A N/A N/        | 4.0E-08 23 N/A N/               | N/A N/A 3.9E        | N/A N/A N               | N/A N/A N               | N/A 4.3E-08 80 N               | N/A 4.3E-08 B0 N               | N/A 4.3E-08 80 N               | N/A N/A N        | N/A 4.3E-08 80 N               | N/A 4.3E-08 80 N               | N/A N/A N/        | N/A N/A N/               | N/A N/A N/               | N/A N/A N/               | N/A N/A N/               | N/A N/A N/               | N/A 9.4E-09 82 N               | N/A 9.4E-09 82 N/               | N/A 7.4E-09 B2 N/               |
|                                              | KANGE AFG RANGE LØAD RANGE NA:<br>Factor freq. factok freq. factur fri                | B6 N/A 3.3E-0B B6 N/            | N/A N/A N/        | N/A N/A N/        | 107 1.2E-08 118 N/A N            | N/A N/A 1.4         | N/A N/A +- N/        | N/A N/A N/        | 87 N/A N/A N/            | 87 N/A N/A N/            | 87 N/A N/A N/            | N/A N/A N/        | B7 4.0E-08 23 N/A N/            | N/A N/A 3.95        | 87 N/A N/A N            | 87 N/A N/A N            | 87 N/A 4.3E-06 80 N            | B7 N/A 4,3E-08 B0 N            | 87 N/A 4.3E-08 80 N            | N/A N/A N        | 67 N/A 4.3E-08 80 N            | 87 N/A 4.3E-08 80 N            | N/A N/A N/        | 73 N/A N/A N/            | 73 N/A N/A N/            | 73 N/A N/A N/            | 73 N/A N/A N/            | 73 N/A N/A N/            | 73 N/A 9.4E-09 82 N            | 73 N/A 9.4E-09 82 N/            | 73 N/A 9.4E-09 B2 N/            |
|                                              | AMAD KANGE AFG RANGE LBAD RANGE MAI<br>Freq. Factor Freq. Factor Freq. Factor Fri     | 1.6E-08 86 N/A 3.3E-08 86 N/    | N/A N/A N/A N/    | N/A N/A N/A N/    | 1.5E-08 107 1.2E-08 118 N/A N    | N/A N/A N/A 1.4     | N/A N/A N/A +- N/    | N/A N/A N/A N/    | 4.5E-UB B7 N/A N/A N/    | 4.5E-08 87 N/A N/A N/    | 4.5E-06 B7 N/A N/A N/    | N/A N/A N/A N/A   | 4.5E-08 B7 4.0E-08 23 N/A N/    | N/A N/A N/A 3.9E    | 4.5E-08 87 N/A N/A N    | 4.5E-08 87 N/A N/A N    | 4.5E-08 B7 N/A 4.3E-08 B0 N    | 4.5E-08 87 N/A 4.3E-08 80 N    | 4.5E-08 B7 N/A 4.3E-08 B0 N    | N/A N/A N/A N    | 4.5E-08 87 N/A 4.3E-08 80 N    | 4.5E-08 87 N/A 4.3E-08 80 N    | N/A N/A N/A N/    | 5.2E-09 73 N/A N/A N/    | 5.2E-09 73 N/A N/A N/A   | 5.2E-09 73 N/A N/A N/    | 5.2E-09 73 N/A N/A N/    | 5.2E-09 73 N/A N/A N/    | 5.2E-09 73 N/A 9.4E-09 82 N    | 5.2E-09 73 N/A 9.4E-09 82 N/    | 5.2E-05 73 N/A 9.4E-05 B2 N/    |
| 9<br>9 6 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 | MD. AMAD KANGE AFG RANGE LBAD RANGE MAI<br>Freq. Factor Freq. Factor Freq. Factor Fri | 12 1.6E-08 86 N/A 3.3E-08 86 N/ | 13 N/A N/A N/A N/ | [] N/A N/A N/A N/ | 13 1.5E-08 107 1.2E-08 118 N/A N | 13 N/A N/A N/A 1.41 | 13 N/A N/A N/A +- N/ | 14 N/A N/A N/A N/ | 14 4.5E-UB 87 N/A N/A N/ | 14 4.5E-08 87 N/A N/A N/ | [4 4.5E-06 87 N/A N/A N/ | 14 N/A N/A N/A N/ | 14 4.5E-08 B7 4.0E-08 23 N/A N/ | 14 N/A N/A N/A 3.95 | 14 4.5E-08 87 N/A N/A N | 14 4.5E-08 87 N/A N/A N | 14 4.5E-08 87 N/A 4.3E-08 80 N | 14 4.5E-0B B7 N/A 4.3E-0B B0 N | 14 4.5E-08 B7 N/A 4.3E-08 B0 N | 14 N/A N/A H/A N | 14 4.5E-08 87 N/A 4.3E-08 80 N | 14 4.5E-09 87 N/A 4.3E-08 80 N | 14 N/A N/A N/A N/ | 15 5.2E-09 73 N/A 9.4E-09 82 N | 15 5.2E-09 73 N/A 9.4E-09 82 N/ | 15 5.2E-05 73 N/A 9.4E-09 B2 N/ |

LEASE DESCRIPTION

2.5555.2

See notes at end of table.

Page 5 Date 20-Aug-87

### TABLE 8-34 (Continued)

# Accident Frequencies and Range Factors

| RANGE<br>Facto  | 117                               |  |
|-----------------|-----------------------------------|--|
| UNDA<br>Freg.   | <br>1.3E-08<br>1.3E-08<br>1.3E-08 |  |
| RANGE<br>FACTOR | : : :                             |  |
| TEAD<br>FREQ.   | <br>N/A<br>N/A<br>N/A             |  |
| RANGE<br>Factor | <br>: : :                         |  |
| PUDA<br>Freq.   | N/A<br>N/A<br>N/A                 |  |
| RANGE<br>FACTOR | <br>16<br>16                      |  |
| PBA<br>Fred.    | N/A<br>5.5E-09<br>5.5E-09         |  |
| RANGE<br>Factor | ; ; ;                             |  |
| NARP<br>Freg.   | N/A<br>N/A<br>N/A                 |  |
| RANGE<br>Factor | <br>82<br>82<br>82                |  |
| LBAD<br>Freq.   | <br>N/A<br>9.4E-09<br>9.4E-09     |  |
| RANGE<br>FACTOR | <br>                              |  |
| AP6<br>Frei     | <br>N/A<br>N/A<br>N/A             |  |
| RANGE<br>Factor | <br>13                            |  |
| ANAD<br>Freg.   | <br>N/A<br>5.2E-09<br>5.2E-09     |  |
| NO.             | 5 5 5<br>5                        |  |
| CEN-<br>RIO     | <br>CRVC<br>CRGC<br>CRVC          |  |

MOTES: 1. Scenarios 1-5 are per train mile; scenarios 6-15 are per exposure year.

Duration time shown for scenarios with agent releases due to detonations and spills is for spill only. Duration time for detonation is instantaneous.



Ċ

SAMPER NAMES OF DESCRIPTION NOT THE WORKS AND DESCRIPTION





) I



# TABLE 8-35 OFFSITE TRANSPORTATION - REGIONAL DISPOSAL OPTION

# Accident Frequencies and Range Factors

| FACTO           |   | 9 10   | 1       | ;     | 1       | 1      | <del>1</del> 0<br>10 | ł       | 9 10    | 9 10   | ł       | 9 10    | 9 10    | 9 10   | 9 10         | 9 10    | 9 10   | ł       | {     | ;       | 0 21    | 0 27   | 1       | 0 27         | 0 27    | 0 27         | 0 21    | 0 21    | 0      |       |
|-----------------|---|--------|---------|-------|---------|--------|----------------------|---------|---------|--------|---------|---------|---------|--------|--------------|---------|--------|---------|-------|---------|---------|--------|---------|--------------|---------|--------------|---------|---------|--------|-------|
| FREQ.           |   | 3.2E-0 | N/A     | N/N   | M/A     | N/A    | 3.2E-0               | N/A     | 3.2E-0  | 3.2E-0 | N/A     | 3.2E-0  | 3.2E-0  | 3.2E-0 | 3.2E-0       | 3.2E-0  | 3.2E-0 | N/A     | N/A   | N/A     | 4.4E-1  | 4.46-1 | N/A     | 4.46-1       | 4.4E-1  | 4.4E-1       | 1-36-1  | 1-32.1  | 3.56-1 | M / D |
| KANDE<br>FACTOR |   | ł      | 1       | ł     | ł       | ł      | ł                    | ł       | ł       | ;      | ł       | ł       | 1       | 1      | 1            | ł       | ;      | ł       | ;     | !       | ;       | ł      | 1       | ;            | ;       | 1            | 1       | 1       | 1      | ;     |
| FREQ.           |   | N/A    | N/A     | N/A   | N/A     | N/A    | N/A                  | N/A     | N/A     | N/A    | N/A     | N/A     | N/A     | N/A    | N/A          | N/A     | N/A    | N/A     | N/A   | N/A     | N/A     | N/A    | A/A     | N/A          | N/A     | N/A          | N/A     | N/A     | N/A    | N/C   |
| kande<br>Factor |   | ;      | 10      | ł     | 10      | 1      | :                    | !       | 1       | !      | 10      | ł       | ;       | ł      | 1            | ;       | ł      | 23      | ł     | 23      | ł       | ł      | 17      | ;            | ł       | ł            | 1       | 1       | 1      | ;     |
| FRED.           |   | N/A    | 3.2E-09 | N/A   | 3.26-09 | N/A    | N/A                  | N/A     | N/A     | N/A    | 3.2E-09 | N/A     | N/A     | N/A    | N/A          | N/N     | A/A    | 4.4E-10 | N/A   | 4.4E~10 | N/A     | N/A    | 4.4E-10 | N/A          | N/A     | N/A          | N/A     | A/A     | N/A    | N/3   |
| KANGE<br>FACTOR |   | !      | ;       | ł     | 1       | ;      | 01                   | Ľ       | 10      | ł      | ł       | ;       | ;       | ;      | 01           | 91      | 1      | ;       | 1     | 1       | 21      | ;      | ł       | ;            | ł       | ł            | 21      | 21      | ł      | ;     |
| FREQ.           |   | N/A    | N/A     | N/A   | N/A     | N/A    | 3.2E-09              | N/A     | 3.2E-09 | N/A    | N/A     | N/A     | N/A     | N/A    | 3.2E-09      | 3.2E-09 | N/A    | N/A     | N/A   | N/A     | 4.4E-10 | N/A    | N/A     | N/A          | N/A     | N/A          | 4.5E-10 | 4.56-10 | N/A    | M / D |
| kande<br>Factor |   | ;      | !       | :     | ;       | ;      | 1                    | 01      | 1       | 1      | ;       | !       | 1       | ļ      | ;            | ļ       | !      | ł       | ł     | 1       | 1       | ;      | 1       | ;            | ł       | !            | :       | ;       | ſ      | ſ     |
| NAAP<br>Fred.   |   | N/A    | N/A     | N/A   | N/A     | N/A    | N/A                  | 3.2E-09 | N/A     | 6/8    | N/A     | N/A     | N/A     | A/A    | N/A          | N/A     | A/A    | N/A     | N.A   | N/A     | N/A     | N/A    | N/A     | N/A          | N/Á     | N/A          | N/A     | N/A     | N/A    | N / 0 |
| KANGE<br>Factor |   | ł      | 1       | ł     | !       | ;      | ;                    | ;       | 1       | ;      | 01      | 10      | 10      | ١      | 10           | 01      | ;      | 1       | 1     | !       | 1       |        | 21      | 27           | 27      | 1            | 21      | 21      | ;      | ;     |
| FRED.           |   | N/A    | N/A     | N/A   | N/A     | N/A    | N/A                  | N/A     | N/A     | N/A    | 3.2E-09 | 3.26-09 | 3.2E-09 | M/A    | 3.2E-09      | 3.2E-09 | N/A    | N/A     | N/A   | N/A     | N/A     | N/A    | 4.4E-10 | 4.4E-10      | 4.4E-10 | N/A          | 4.56-10 | 4.5E-10 | N/Â    | N/0   |
| HANGE<br>Factor |   | ţ      | 1       | ;     | ;       | !      | 10                   | !       | ł       | :      | !       | !       | 1       | ł      | ;            | ł       | {      | }       | ł     | !       | ł       | ;      | 1       | ł            | 1       | ł            | 1       | !       | ;      | :     |
| AP6<br>Fred.    |   | N/A    | N/A     | N/A   | N/A     | N/A    | 3.2E-09              | N/A     | N/A     | N/A    | N/A     | N/A     | N/Å     | N/A    | N/A          | N/A     | N/A    | N/A     | N/A   | N/A     | N/A     | N/A    | N/A     | N/A          | N/A     | A/A          | H/A     | N/A     | N/A    | 67N   |
| KANGE<br>FACTOR |   | ł      | ł       | :     | :       | !      | 1                    | ;       | t       | }      | 1       | !       | ;       | !      | 1            | ;       | ;      | 1       | 1     | ;       | ł       | ;      | 1       | ł            | 1       | 1            | ;       | ;       | :      | ;     |
| ANAU<br>Fred.   |   | N/A    | N/A     | M/A   | N/A     | N/A    | N/A                  | N/A     | N/A     | N/A    | N/A     | N/A     | N/A     | N/A    | N/A          | N/A     | N/A    | N/A     | N/A   | N/A     | N/A     | N/A    | N/A     | N/A          | N/A     | N/A          | N/A     | NZA     | ñ/À    | N / D |
| ż               | I | ~      | n       | m     | r       | ы      | n                    | ы       | m       | m      | ~       | м       | м       | м      | м            | м       | m      | +       | *     | +       | -       | *      | -       | -            | 4       | *            | 4       | -       | u٦     | 47    |
| SCEN-<br>ARIO   |   | RCBGS  | RCDHS   | RCCGS | RCCHS   | RCINGS | RCKHS                | RCKVS   | RCMVS   | KCP65  | RCPHS   | RCPVS   | KC065   | RCQVS  | <b>KCRES</b> | RCRVS   | RCSVS  | RCDHC   | RCCEC | RCCHC   | RCHVC   | RCPGC  | RCFHC   | <b>KCPVC</b> | RCOGC   | <b>KCOVC</b> | RCKBC   | RCRVC   | RCBGF  | RCKGF |

See notes at end of table.

Real of

Ŕ

Page 2 Bate 20-Aug-87

### TABLE 8-35 (Continued)

10414-70

# OFFSITE TRANSPORTATION - KEGIONAL DISPOSAL OPTION

# Cartor

|               |     |               |                 |              |                 |               | Ĩ               | ic caent      | ianha J.        |              | aburu           |               |                 |               |                 |               |                 |
|---------------|-----|---------------|-----------------|--------------|-----------------|---------------|-----------------|---------------|-----------------|--------------|-----------------|---------------|-----------------|---------------|-----------------|---------------|-----------------|
| SCEN-<br>Ario | NO. | ANAD<br>Freq. | RANGE<br>FACTOR | APG<br>Freg. | RANGE<br>Factor | LBAD<br>Freg. | RANGE<br>FACTOR | NAAP<br>Freq. | RANGE<br>FACTOR | PBA<br>Freq. | RANGE<br>FACTOR | PUDA<br>Fred. | RANGE<br>Factor | TEAD<br>Freq. | RANGE<br>Factor | UNDA<br>FREQ. | RANGE<br>FACTOR |
|               | ł   |               |                 |              |                 |               |                 |               |                 |              |                 |               |                 |               |                 |               |                 |
| RCKHF         | רט  | N/A           | :               | 3.5E-10      | 1               | NA            | ł               | N/A           | :               | 3.5E-10      | 47              | N/A           | ł               | N/A           | ł               | 3.5E-10       | 1               |
| RCKVF         | ŝ   | N/A           | :               | N/A          | ł               | N/A           | !               | 3.56-10       | 47              | N/A          | ł               | N/A           | ł               | N/A           | ł               | N/A           | ;               |
| RCSVF         | 67  | N/A           | ł               | N/A          | ;               | N/A           | ł               | N/A           | 1               | N/A          | ł               | N/A           | ł               | N/A           | ł               | 3.56-10       | 14              |
| RCBGS         | 9   | N/A           | :               | N/A          | !               | N/A           | ł               | N/A           | ł               | N/A          | ł               | N/A           | ł               | N/A           | ;               | 1.76-11       | 20              |
| RCDHC         | 9   | N/A           | ł               | N/A          | ;               | N/A           | 1               | N/A           | 1               | N/A          | :               | 1.7E-11       | 20              | N/A           | ł               | N/A           | ł               |
| <b>KCCEC</b>  | 9   | N/A           | ;               | N/A          | 1               | N/A           | !               | N/A           | ł               | N/A          | i<br>t          | N/A           | ł               | N/A           | 1               | N/A           | ł               |
| RCCHC         | 9   | N/A           | ;               | N/A          | 1               | N/A           | 1               | N/A           | !               | N/A          | ;               | 1.7E-11       | 2               | N/A           | ł               | N/A           | ļ               |
| RCKES         | 9   | N/A           | ;               | N/A          | !               | N/N           | ł               | N/A           | !               | N/A          | ł               | N/A           | ;               | N/A           | ł               | N/A           | ł               |
| RCLHS         | \$  | N/A           | :               | 1.76-11      | 20              | N/A           | ł               | N/A           | 1               | 1.7E-11      | 20              | N/A           | ;               | N/A           | ł               | 1.76-11       | 20              |
| <b>FCKVS</b>  | •   | N/A           | ;               | N/A          | ł               | N/A           | ł               | 1.7E-11       | 20              | N/A          | 1               | M/A           | ł               | N/A           | ł               | N/A           | 1               |
| RCNVC         | \$  | N/A           | ;               | N/A          | ;               | N/A           | 1               | N/A           | !               | 1. JE-11     | 20              | N/A           | ;               | N/A           | ţ               | 1.76-11       | 20              |
| <b>RCPGC</b>  | •   | N/A           | ;               | N/A          | ł               | N/A           | 1               | N/A           | ł               | N/A          | 1               | N/A           | ;               | N/A           | 1               | 1.7E-11       | 20              |
| RCFHC         | 9   | M/A           | ł               | N.A          | 1               | 1.7E-11       | 20              | N/A           | 1               | N/A          | ;               | 1.7E-11       | 20              | N/A           | 1               | N/A           | ł               |
| KCPVC         | -0  | N/A           | 1               | N/A          | ł               | 1.76-11       | 20              | N/A           | !               | N/A          | ł               | N/A           | :               | N/A           | 1               | 1.7E-11       | 50              |
| RCGGC         | -0  | N/A           | ;               | N/A          | ł               | 1.7E-11       | 2               | N/A           | 1               | N/A          | :               | N/A           | !               | N/A           | !               | 1.76-11       | 50              |
| <b>FCQVC</b>  | -0  | N/A           | :               | N/A          | 1               | N/A           | ;               | N/A           | ł               | N/A          | !               | N/A           | ł               | N/A           | ł               | 1.76-11       | 20              |
| <b>KCKGC</b>  | -0  | N/A           | ł               | N/A          | 1               | 1.7E-11       | 20              | N/A           | ;               | 11-3/.1      | 20              | N/A           | ;               | N/A           | ł               | 1.7E-11       | 20              |
| RCRVC         | •   | N/A           | ;               | N/A          | 1               | 1.7E-11       | 20              | R/N           | ł               | 1.7E-11      | 20              | N/A           | ł               | N/A           | :               | 1.7E-11       | 20              |
| <b>RCSVS</b>  | -0  | M/A           | :               | N/A          | !               | N/A           | 1               | N/A           | ;               | N/A          | 1               | N/A           | :               | N/A           | 1               | 1.7E-11       | 2               |
| <b>RCBGF</b>  | 1   | N/A           | ;               | N/A          | 1               | N/A           | ;               | M/A           | 1               | N/A          | ł               | N/A           | 1               | N/A           | :               | 1.4E-11       | 20              |
| RCDHC         | 1   | N/A           | :               | N/A          | !               | N/A           | ;               | A/A           | 1               | N/A          | 1               | 1.4E-11       | ຊ               | N/A           | ;               | N/A           | }               |
| <b>ACCGC</b>  | 1   | N/A           | 1               | N/A          | ;               | N/A           | 1               | N/A           | ţ               | N/A          | :               | N/A           | ;               | N/A           | ;               | N/A           | }               |
| RCCHC         | 1   | N/A           | :               | N/A          | :               | N/A           | ;               | N/A           | 1               | N/A          | ł               | 1.4E-11       | 2               | N/A           | ł               | N/N           | ١               |
| <b>KCK6F</b>  | 1   | N/A           | 1               | N/A          | ;               | N/A           | ;               | N/A           | 1               | N/A          | :               | N/A           | 1               | N/A           | ł               | N/A           | 1               |
| RCAHF         | 1   | N/A           | !               | 1.4E-11      | 20              | N/A           | ł               | N/A           | ł               | 1.4E-11      | 20              | N/A           | :               | N/A           | ;               | 1.46-11       | 20              |
| <b>KCKVF</b>  | -   | N/A           | !               | N/A          | :               | N/A           | ;               | 1.4E-11       | 20              | N/A          | ł               | N/A           | 1               | N/A           | :               | N/A           | }               |
| RCNVC         | 1   | N/A           | !               | N/A          | ł               | N/A           | ;               | N/A           | :               | 1.4E-11      | 20              | N/A           | :               | N/A           | ;               | 1.46-11       | 20              |
| FCF6C         | l   | R/A           | 1               | N/A          | ;               | A/A           | :               | N/A           | ł               | N/A          | ł               | N/A           | ;               | N/A           | !               | 1.46-11       | 20              |
| <b>RCPHC</b>  | 1   | ₩/A           | 1               | N/A          | ;               | 1.4E-11       | 50              | N/A           | :               | N/A          | ;               | 1.4E-1!       | 20              | N/A           | :               | M/A           | :               |

See notes at end of table.

Ś



alinen ander ander ander G



TABLE 8-35 (Continued)

Page 3 Date 20-Aug-87

# OFFSITE TRANSPORTATION - REGIGNAL DISPOSAL OPTION

-

| RMC     RAME     RAME <thr< th=""><th>AFG<br/>AFG<br/>N/A<br/>N/A<br/>N/A<br/>N/A<br/>N/A<br/>N/A<br/>N/A<br/>N/A<br/>N/A<br/>N/A</th><th></th><th>EBAD R<br/>FREQ. F<br/>FREQ. F<br/>FREQ. F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F<br/>F</th><th>ACTOR<br/>ACTOR<br/>20<br/>20</th><th>NAAP<br/>Freg.</th><th>RANGE<br/>Factor</th><th>PBA<br/>Freq.</th><th>RANGE<br/>Factor</th><th>PUDA<br/>Freq.</th><th>RANGE<br/>Factor</th><th>TEAD<br/>Freq.</th><th>RANGE</th><th>UNDA</th><th>RANG</th></thr<> | AFG<br>AFG<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A |                   | EBAD R<br>FREQ. F<br>FREQ. F<br>FREQ. F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F<br>F | ACTOR<br>ACTOR<br>20<br>20 | NAAP<br>Freg. | RANGE<br>Factor | PBA<br>Freq. | RANGE<br>Factor | PUDA<br>Freq. | RANGE<br>Factor | TEAD<br>Freq. | RANGE  | UNDA     | RANG |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------|-------------------|-----------------------------------------------------------------------------------------------------------------------------------|----------------------------|---------------|-----------------|--------------|-----------------|---------------|-----------------|---------------|--------|----------|------|
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N N N N N N N N N N N N N N N N N N N                                            | 1 1 1 1 1 1 1 1 1 | 1.1.66-11<br>1.46-11<br>1.46-11<br>1.46-11<br>1.46-11<br>1.46-11<br>1.48<br>11/A<br>11/A<br>11/A<br>11/A                          | 3   <b>3</b> 5             |               |                 |              |                 |               |                 |               | FACTUR | FREG.    | FAC  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                  |                   | 1.4E-11<br>M/A<br>N/A<br>N/A<br>N/A<br>N/A<br>M/A<br>M/A<br>M/A                                                                   | 8   8                      | M/A           | !               | A/A          | ł               | N/A           | :               | N/A           | :      | 1.4E-11  |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N / N / N / N / N / N / N / N / N / N /                                          |                   | N/A<br>1.46-11<br>1.46-11<br>1.46-11<br>1.46-11<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                         | 18                         | N/A           | ł               | N/A          | !               | N/A           | ;               | N/N           | ;      | 1.46-11  |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                    |                   | 1.4E-11<br>1.4E-11<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                                                      | 20                         | N/A           | ;               | N/A          | ł               | N/A           | ł               | N/A           | ł      | 1.46-11  |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/N<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                    | -                 | 1.4E-11<br>N/A<br>N/A<br>N/A<br>N/A<br>N/A                                                                                        | 3                          | N/A           | ;               | 1.4E-11      | 20              | N/A           | ł               | N/A           | !      | 1.4E-11  | ~    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Q / X X X X X X X X X X X X X X X X X X                                          | 1 1 1 1           | N/N<br>N/N<br>N/N<br>N/N<br>N/N<br>N/N<br>N/N                                                                                     | 8                          | N/A           | ;               | 1.4E-11      | 8               | NUA           | ;               | N/A           | :      | 1.4E-11  | ~    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A<br>N/A<br>A/N<br>A/N<br>A/N                                                  |                   | 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4                                                                                             | ;                          | N/A           | ;               | N/A          | ł               | N/A           | 1               | N/A           | ł      | 1.46-11  | ~    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | M/M<br>N/N<br>N/A                                                                | : :               | R/N<br>R/N<br>R/N                                                                                                                 | ł                          | N/A           | 1               | N/A          | ł               | N/A           | ł               | N/A           | ł      | 4.BE-07  |      |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A<br>N/A<br>N/A                                                                | :                 | N/A<br>N/A                                                                                                                        | !                          | N/A           | ł               | N/A          | ł               | 1.5E-07       | 31              | N/A           | 1      | N/A      | ł    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A<br>N/A                                                                       |                   | N/A                                                                                                                               | ł                          | N/A           | !               | N/A          | ł               | N/A           | ł               | N/A           | ł      | N/A      | i    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                              | 1                 | N / N                                                                                                                             | ł                          | A/A           | ;               | N/A          | !               | 1.56-07       | 37              | N/A           | 1      | N/A      | i    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                  | ł                 |                                                                                                                                   | ł                          | N/A           | !               | N/A          | :               | N/A           | ;               | N/A           | ;      | N/A      | ;    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 1.1E-07                                                                          | 37                | N/A                                                                                                                               | 1                          | N/A           | l               | 3.1E-07      | 37              | N/A           | ł               | N/A           | 1      | 4.86-07  | ~,   |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                              | ł                 | N/A                                                                                                                               | ł                          | 1.7E-07       | 21              | N/A          | 1               | N/A           | ł               | N/A           | •      | N/A      | i    |
| 11<br>11<br>11<br>11<br>11<br>11<br>11<br>11<br>11<br>11<br>11<br>11<br>11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | N/A                                                                              | !                 | N/A                                                                                                                               | :                          | N/A           | ;               | 3. IE-07     | 37              | N/A           | ;               | N/A           | !      | 4.86-07  | ~    |
| 11<br>11<br>11<br>11<br>11<br>11<br>11<br>11<br>11<br>11<br>11<br>11<br>11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | N/A                                                                              | :                 | N/A                                                                                                                               | 1                          | N/A           | ł               | N/R          | ł               | N/A           | ;               | N/A           | ł      | 4.86-07  | ~,   |
| 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A                                                                              | :                 | 1.6E-07                                                                                                                           | 37                         | N/A           | ł               | N/A          | ł               | 1.5E-07       | 37              | N/A           | ł      | N/A      | ł    |
| 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A                                                                              | :                 | 1.65-07                                                                                                                           | 37                         | N/A           | ;               | N/A          | ł               | N/A           | 1               | N/A           | ł      | 4.85-07  |      |
| 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A                                                                              | :                 | L.6E-07                                                                                                                           | 37                         | N/A           | ;               | N/A          | ł               | N/A           | 1               | N/A           | ł      | 4.86-07  |      |
| 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/A                                                                              | ł                 | N/A                                                                                                                               | ł                          | N/A           | ;               | N/A          | ł               | N/A           | ł               | N/A           | 1      | 4.85-07  | -    |
| 11 11 11 11 11 11 11 11 11 11 11 11 11                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                              | 1                 | 1.65-07                                                                                                                           | 31                         | N/A           | ١               | 3. IE-07     | 37              | N/A           | ;               | N/A           | :      | 4.8E-07  | -1   |
| 11 K/A<br>12 N/A<br>12 N/A<br>12 N/A<br>12 N/A<br>12 N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | N/A                                                                              | 1                 | 1.65-07                                                                                                                           | 31                         | N/A           | ł               | 3. IE-07     | 37              | N/A           | ł               | N/A           | 1      | 4.BE-07  | ~,   |
| 12 N/A<br>12 N/A<br>12 N/A<br>12 N/A<br>12 N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                                                                              | 1                 | N/A                                                                                                                               | ł                          | N/A           | ١               | N/N          | ł               | N/A           | ;               | N/A           | ł      | 4.8E-07  | ~1   |
| 12 N/A<br>12 N/A<br>12 N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | N/A                                                                              | ł                 | N/A                                                                                                                               | 1                          | N/A           | }               | N/A          | 1               | 1.66-08       | 9               | N/A           | 1      | N/A      | ;    |
| : 12 N/A<br>: 12 N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N/A                                                                              | ;                 | N/A                                                                                                                               | ;                          | N/A           | ;               | N/A          | !               | N/A           | ł               | N/A           | :      | N/A      | 1    |
| 12 N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                                                                              | ;                 | N/A                                                                                                                               | ł                          | N/A           | ;               | M/A          | 1               | 1.65-08       | 10              | N/A           | 1      | N/A      | ;    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | N/A                                                                              | 1                 | N/Å                                                                                                                               | ;                          | N/A           | ;               | 3.4E-08      | 51              | N/A           | ł               | N/A           | :      | 5. 3E-08 | _    |
| : 12 N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | N/A                                                                              | 1                 | N/A                                                                                                                               |                            | N/A           | ł               | N/A          | ł               | N/A           | ł               | N/A           | ;      | 5.2E-08  |      |
| : 12 N/A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | N/A                                                                              | :                 | 1,85-08                                                                                                                           | 87                         | N/A           | !               | A/R          | 1               | 1.6E-vð       | 100             | N/A           | ł      | N/A      | 1    |
| IZ N/H                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | A/A                                                                              | 1                 | L. 85 - UB                                                                                                                        | 87                         | N/A           | ł               | N/A          | 1               | N/A           | ;               | N/A           | ;      | 5.2E-08  | ~    |

R. C. C. C.

J 3222 J

See rotes at end of table.

8-83

TABLE 8-35 (Continued)

Fage 4 Date 20-Aug-87

000050000000000

# DFFSITE TRANSPORTATION - REGIONAL DISPOSAL OPTION

# Accident Frequencies and Range Factors

| KANGE<br>FACTOR                                                                                | 11                        | 13                 | 86                        | 86                        | 86                 | ł                  | 86                        | ł                      | 86                 | 76                 | ł                  | ł                  | !                  | ł                  | 76                        | Ì                      | 76                 | 76                 | 1                         | 76                        | 76                        | 76                 | 16                        | 76                        | 76                 | ł                  | !                  | ł                  | 117                |
|------------------------------------------------------------------------------------------------|---------------------------|--------------------|---------------------------|---------------------------|--------------------|--------------------|---------------------------|------------------------|--------------------|--------------------|--------------------|--------------------|--------------------|--------------------|---------------------------|------------------------|--------------------|--------------------|---------------------------|---------------------------|---------------------------|--------------------|---------------------------|---------------------------|--------------------|--------------------|--------------------|--------------------|--------------------|
| UNDA<br>Freg.                                                                                  | 5.2E-08                   | 5.2E-08            | 5.46-08                   | 5.4E-08                   | 1.6E-08            | N/A                | 1.6E-08                   | N/A                    | 1. bE-08           | 1.96-10            | N/A                | N/A                | N/A                | N/A                | 1.9E-10                   | N/A                    | 1.96-10            | 1.96-10            | N/A                       | 1.96-10                   | 1.96-10                   | 1.9E-10            | 1.9E-10                   | 1.9E-10                   | 1.9E-10            | N/A                | M/A                | N/A                | 1. JE-08           |
| KANGE<br>Factor                                                                                | ł                         | ł                  | 1                         | ł                         | 1                  | ł                  | ł                         | ł                      | ļ                  | ł                  | ;                  | : •                | . ;                | ł                  | 1                         | ł                      | ł                  | ł                  | 1                         | ł                         | ł                         | 1                  | 1                         | ł                         | ł                  | ;                  | ļ                  | 1                  | :                  |
| TEAD<br>Freq.                                                                                  | N/A                       | N/A                | N/A                       | N/A                       | N/A                | N/A                | N/A                       | N/A                    | N/A                | N/A                | N/A                | N/A                | N/A                | N/A                | N/A                       | N/A                    | N/A                | N/A                | N/A                       | N/A                       | N/A                       | N/A                | N/A                       | N/A                       | R / A              | N/A                | N/A                | N/A                | N/A                |
| FACTUR                                                                                         | ł                         | ;                  | ;                         | ł                         | ł                  | ļ                  | ;                         | ł                      | ł                  | ł                  | 16                 | ł                  | 16                 | ł                  | ;                         | ł                      | ł                  | 1                  | 16                        | 1                         | :                         | 1                  | 1                         | ł                         | ł                  | 83                 | ţ                  | 83                 | :                  |
| PUDA<br>Freq.                                                                                  | N/A                       | N/A                | N/A                       | N/A                       | N/A                | N/A                | N/A                       | N/A                    | N/A                | N/A                | 2.2E-10            | N/A                | 2.2E-10            | N/A                | N/A                       | N/A                    | N/A                | N/A                | 2.2E-10                   | N/A                       | N/A                       | N/A                | N/A                       | N/A                       | N/A                | 4.16-09            | N/A                | 4.1E-09            | N/A                |
| FACTOR                                                                                         | ł                         | ł                  | B6                        | 86                        | :                  | ;                  | <b>6</b> 3                | ł                      | 1                  | ł                  | 1                  | ł                  | ł                  | 1                  | <b>B</b> 01               | ł                      | 108                | ł                  | ł                         |                           | ł                         | ł                  | 22                        | 22                        | ł                  | ł                  | ł                  | ł                  | 67                 |
| PBA<br>Freq.                                                                                   | N/A                       | N/A                | 3.4E-0B                   | 3.4E-08                   | N/A                | N/A                | 3. 3E -08                 | N/A                    | N/A                | N/A                | N/A                | N/A                | N/A                | N/A                | 6.1E-08                   | N/A                    | 6.1E-08            | N/A                | N/A                       | N/A                       | N/A                       | N/A                | 6. IE-08                  | 6.1E-08                   | N/A                | N/A                | N/A                | N/A                | 1.0E-08            |
| ACTOR                                                                                          | ;                         | ł                  | 1                         | 1                         | ;                  | ł                  | ł                         | 64                     | ţ                  | ţ                  | ţ                  | ;                  | ţ                  | ł                  | ;                         | 87                     | 1                  | }                  | ;                         | ł                         | ;                         | ;                  | 1                         | +                         | ł                  | :                  | :                  | ;                  | +                  |
| 221                                                                                            |                           |                    |                           |                           |                    |                    |                           |                        |                    |                    |                    |                    |                    |                    |                           |                        |                    |                    |                           |                           |                           |                    |                           |                           |                    |                    |                    |                    |                    |
| NAAP RG<br>Freq. Fi                                                                            | N/A                       | N/A                | N/A                       | N/A                       | N/A                | N/A                | N/A                       | I.9E-08                | N/A                       | 5.1E-08                | N/A                | N/A                | N/A                       | N/A                       | N/A                       | N/A                | N/A                       | N/A                       | N/A                | N/A                | N/A                | N/A                | N/A                |
| AANGE NAAP RG<br>Factor Freq. Fi                                                               | 87 N/A                    | N/A                | B6 N/A                    | 86 N/A                    | N/A                | N/A                | N/A                       | 1.9E-08                | N/A                       | 6.1E-08                | N/A                | N/A                | 85 N/A                    | 85 N/A                    | 85 N/A                    | N/A                | 85 N/A                    | 85 N/A                    | N/A                | N/A                | N/A                | N/A                | N/A                |
| LBAD RANGE NAAP RF<br>Freq. Factor Freq. F                                                     | 1.8E-08 87 N/A            | N/A N/A            | 1.8E-08 B6 N/A            | 1.BE-08 86 N/A            | N/A N/A            | N/A N/A            | N/A N/A                   | N/A 1.9E-08            | N/A N/A                   | N/A 6.1E-08            | N/A N/A            | N/A N/A            | 6.1E-08 85 N/A            | 6.1E-08 85 N/A            | 6.1E-08 85 N/A            | N/A N/A            | 6.1E-08 85 N/A            | 6.1E-08 85 N/A            | N/A N/A            |
| RANGE LBAD RANGE NAAP RE<br>Factor Freq. Factor Freq. F                                        | 1.8E-08 87 N/A            | N/A N/A            | 1.8E-08 B6 N/A            | 1.8E-08 86 M/A            | N/A N/A            | N/A N/A            | 73 N/A N/A                | N/A 1.9E-0B            | N/A N/A            | 74 N/A N/A                | N/A 6.1E-08            | N/A N/A            | N/A N/A            | 6.1E-08 85 N/A            | 6.1E-08 85 N/A            | 6.1E-08 85 N/A            | N/A N/A            | 6.1E-08 85 M/A            | 6.1E-08 85 N/A            | N/A N/A            |
| APG RANGE LBAD RANGE NAAP RF<br>Fred. Factor Freq. Factor Freq. F                              | N/A 1.8E-08 87 N/A        | N/A N/A N/A        | N/A 1.8E-08 B6 N/A        | N/A 1.8E-08 86 N/A        | N/A N/A N/A        | N/A N/A N/A        | 1.2E-08 73 N/A N/A        | N/A N/A 1.9E-0B        | N/A N/A N/A        | N/À N/A N/A        | 5.4E-08 74 N/A N/A        | N/A N/A 6.1E-08        | N/A N/A N/A        | N/A N/A N/A        | N/A 6.1E-08 85 N/A        | N/A 6.1E-08 85 N/A        | N/A 6.1E-08 85 N/A        | N/A N/A N/A        | N/A 6.1E-08 85 N/A        | N/A 6,1E-08 85 N/A        | N/A N/A N/À        |
| KANGE APG RANGE LBAD RANGE NAAP RE<br>Factor Frem. Factor Freq. Fa                             | N/A 1.8E-08 87 N/A        | N/A N/A N/A        | N/A 1.8E-08 B6 N/A        | N/A 1.BE-08 86 N/A        | N/A N/A N/A        | N/A N/A N/A        | 1.2E-08 73 N/A N/A        | N/A N/A 1.9E-0B        | N/A N/A N/A        | 5.4E-08 74 N/A N/A        | N/A N/A 6.1E-08        | N/A N/A N/A        | N/A N/A N/A        | N/A 6.1E-08 85 N/A        | N/A 6.1E-08 85 N/A        | N/A 6.1E-08 85 N/A        | N/A N/A N/A        | N/A 6.1E-08 85 N/A        | N/A 6.1E-08 85 N/A        | N/A N/A N/A        |
| AMAD RANGE APG RANGE LBAD RANGE WAAP RE<br>Fred. Factor Fred. Factor Fred. Factor Fred. Fi     | M/A N/A 1.8E-08 87 N/A    | N/A N/A N/A N/A    | N/A N/A 1.8E-08 B6 N/A    | N/A N/A 1.8E-08 86 N/A    | N/A N/A N/A N/A    | N/A N/A N/A N/A    | N/A 1.2E-08 73 N/A N/A    | N/A N/A N/A 1.9E-08    | N/A N/A N/A N/A    | N/A N/À N/A N/A    | N/A 5.4E-08 74 N/A N/A    | N/A N/A N/A 6.1E-08    | N/A N/A N/A N/A    | N/A N/A N/A N/A    | N/A N/A 6.1E-08 85 N/A    | N/A N/A 6.1E-08 85 N/A    | N/A N/A 6.1E-08 85 N/A    | N/A N/A N/A N/A    | N/A N/A 6.1E-08 B5 N/A    | N/A N/A 6.1E-08 85 N/A    | N/A N/A N/A N/À    |
| NO. AMAD KANGE APG RANGE LBAD RANGE NAAP RE<br>Freq. Factor Freq. Factor Freq. Factor Freq. Fi | 12 M/A N/A 1.8E-08 B7 N/A | 12 N/A N/A N/A N/A | 12 N/A N/A 1.8E-08 B6 N/A | 12 N/A N/A 1.8E-08 86 N/A | [3 N/A N/A N/A N/A | 13 N/A N/A N/A N/A | 13 N/A 1.2E-08 73 N/A N/A | 13 N/A N/A N/A 1.9E-0B | 13 N/A N/A N/A N/A | 14 N/A 5.4E-08 74 N/A N/A | 14 N/A N/A N/A 6.1E-08 | 14 N/A N/A N/A N/A | 14 N/R N/A N/A N/A | 14 N/A N/A 6.1E-08 85 N/A | 14 N/A N/A 6.1E-08 85 N/A | 14 N/A N/A 6.1E-08 85 N/A | 14 N/A N/A N/A N/A | 14 N/A N/A 6.1E-08 85 N/A | 14 N/A N/A 6.1E-08 85 N/A | 14 N/A N/A N/A N/A | 15 N/A N/A N/A N/A |

See notes at end of table.

Reserved Res

.

8-84







OFFSITE TRANSPORTATION - REGIONAL DISPOSAL OPTION

Accident Frequencies and Range Factors

| RANGE<br>Factor |   | 117     | 1       | 117      | 111          | 117     | 117      | 11      |   |
|-----------------|---|---------|---------|----------|--------------|---------|----------|---------|---|
| UNDA<br>Fred.   |   | 1.36-08 | N/N     | 1. 3E-08 | 1.36-08      | 1.36-08 | 1.36-08  | 1.35-08 | • |
| RANGE<br>FACTOR |   | :       | ł       | ł        | 1            | !       | ł        | ł       |   |
| TEAD<br>Fred.   |   | N/A     | N/A     | N/A      | N/A          | N/A     | M/A      | N/A     |   |
| RANGE<br>FACTOR |   | ;       | 83      | 1        | 1            | ł       | ;        | :       |   |
| PUDA<br>Fred.   |   | N/A     | 4.1E-09 | N/A      | N/A          | N/A     | N/A      | N/A     |   |
| RANGE<br>Factor |   | :       | ł       | 1        | ł            | :       | 67       | 67      |   |
| PBA<br>Freq.    |   | N/A     | N/A     | N/A      | N/A          | N/A     | 1.0E-08  | 1.06-08 |   |
| RANGE<br>Factor |   | ;       | 1       | ;        | ł            | ł       | ł        | !       | • |
| NAAP<br>Fred.   |   | N/A     | N/A     | M/A      | N/A          | N/A     | N/A      | A/A     |   |
| RANGE<br>Factor |   | ;       | 17      | 11       | 11           | ļ       | 2        | 11      |   |
| LBAD<br>Freq.   |   | N/A     | 6.1E-09 | 6.1E-09  | 6.1E-09      | N/A     | 6. IE-09 | 6.1E-09 |   |
| RANGE<br>Factor |   | ł       | 1       | ł        | ł            | ;       | :        | 1       |   |
| AP6<br>Freq.    |   | N/A     | N/A     | N/A      | N/A          | N/A     | N/A      | N/A     |   |
| RANGE<br>FACTOR |   | 1       | 1       | ł        | 1            | ;       | ł        | :       |   |
| ANAD<br>Freq.   |   | N/A     | N/A     | N/A      | N/A          | N/A     | N/A      | N/A     |   |
| NO.             | ł | 2       | 5       | 5        | 15           | 5       | 15       | 51      |   |
| SCEN-<br>AR 10  |   | RCPGC   | RCPHC   | RCPVC    | <b>RCQGC</b> | RCQVC   | RCRGC    | RCRVC   |   |

MOTES: 1. Scenarios 1-5 are per train mile; scenarios 6-15 are per exposure year.

 Duration time shown for scenarios with agent releases due to detonations and spills is for spill only. Duration time for detonation is instantaneous.

8-85

8.3. OFFSITE AIR TRANSPORT

The purpose of this section is to evaluate the accident sequences associated with specific air transportation transfers. The results, in terms of frequency of chemical agent release, will be combined with agent release and consequence calculations to determine the risk of the air transportation option and to compare the relative risk of the air option with other transportation alternatives.

The transfers involve transport of chemical agent munitions from both Aberdeen Proving Ground and Lexington Depot to Tooele Depot via a C-141 aircraft. Also, the analysis was carried through for a C-5 aircraft. The air flight distances planned for the air transport phases are 1540 and 2066 miles from Lexington and Aberdeen, respectively (Ref. 8-15). The specific route taken for air transport flights avoids flying directly over major population centers.

The actual number of flights to be performed during the air transport operation is classified information. However, the draft concept plan (Ref. 8-16) states that approximately 1500 flights from Lexington and about 300 flights from Aberdeen will be required for a C-141 aircraft. A C-5 aircraft would decrease the number of required flights by one-fourth.

All munitions will be transported inside an offsite transport container (OFC), designed to meet special chemical munition criteria so that it will not fail as a result of certain severe accident scenarios postulated by a panel of civilian transportation experts. The package failure thresholds are given in Section 3.3.

### 8.3.1. Procedures and Assumptions

Specific to the air transport mode, it is assumed that a crush or puncture failure cannot occur without first experiencing an impact. A B

&\*a.&\*a.&\*a.&\*a.&\*a.#\*a.#\*a.#\*a.#\*a.#\*a.#\*a.#\*

Thus, the frequency data associated with crush and puncture accidents are stated as "crush given impact" and "puncture given impact."

The hypothetical scenario of fuze-induced detonation of burstered munitions is not considered credible, (Ref. 8-6). Burster detonation from other initiators, however, is considered.

The Sandia National Laboratory data base (Ref. 8-1) regarding rates of aircraft accidents includes, as components of the data, accidents due to severe weather and midair collisions. Thus, these are not treated as separate events.

The effects of an aircraft depressurization accident are negligible and were not included in the analysis for the following reasons:

- Only projectiles, rockets, and ton containers are to be transported by air.
- Projectiles are designed to be projected into high-atmosphere (low-pressure) conditions during normal operation without breaching agent containment. Burster detonation is required to breach designed containment.
- 3. The ton containers used for transporting mustard agent contain significant voids (they are not full). Ton containers are designed to withstand significant pressure differences between the inside and outside of the package without losing structural integrity. For example, ton containers are commonly used for processing enriched uranium where the inside pressure is essentially a vacuum. Although the pressure difference is in the opposite direction in a plane depressurization accident, no threat exists for a nonleaking ton container during a depressurization scenario.

4. In the case of rockets, static pressure tests (Ref. 8-20) have been performed on rockets similar to those to be transported by air. These rockets were deburstered and defuzed so that failure by heating would occur from static pressure only. Failure occurred at pressures ranging from approximately 700 to 1000 psig and temperatures of 400° to 800°F at heatup times of 20 to 40 min. The scenario postulated is a depressurization at ambient temperatures, with the maximum threat to the agent is approximately 15 psi change in pressure. Based on this data, it is believed that a nonleaking rocket subjected to such a scenario poses no threat. EXTERNAL SUCCE SUSPECTION PROFESSION FOR

The planned air route is down the Chesapeake Bay before turning west, but it will not be over any water bodies (lake, river, ocean, etc.) which would have depths greater than the 600-ft design criterion. (Note that the maximum depth of the Chesapeake Bay is about 140 ft.) Therefore, any air accident scenario which includes package failure by immersion into the water is not credible and not included.

ということの言語

The M55 rocket is unique among obsolete chemical weapons in that the propellant is electrically initiated. It is assumed that the rocket pallets will be shielded from electromagnetic or electrostatic sources by a metal container. The OFC design can be evaluated to verify (1) that it completely shields the rockets from electromagnetic and electrostatic sources, (2) that such shielding will withstand any accident condition that would not of itself result in advertent ignition, (3) that administrative controls will be in place to control electromagnetic and electrostatic sources at accident locations, and (4) that portable shielding and grounding systems, as well as training in their use, be made available to accident response teams.

8-88

### 8.3.2. Accident Scenario Definition

An event tree was developed to more clearly define the set of postulated accident sequences starting with the top event of an air accident occuring. Figure 8-11 is the event tree that results from postulating all credible events which might result in agent release for an aircraft accident.

From the event tree, the accident sequences for the air transportation option fall into one of the following general classes or groups of initiating events:

- 1. Ground collision (severe or moderate).
- 2. A fire aboard the aircraft.

3. A ground collision (severe or moderate) with a subsequent fire.

In this report, the failure threats for the packages or packages are the same as those used by Sandia National Laboratories in their analyses of aircraft accidents (Ref. 8-1). This makes it possible to compare the relative frequency of accidents involving the different transportation modes.

The primary concern is whether the package is breached so that a leak occurs and allows the contained material to escape. Once an accident has occurred, there are five failure threats that can threaten the package and result in chemical agent release. They are:

- <u>Impact</u> Striking or being struck by an object which has no sharp projections.
- <u>Puncture</u> Striking or being struck by an object which peretrates the protective structures of the package.

8-89

| AD-A1 | 93 355<br>551F1E | CHE<br>The<br>Ca<br>D Sap | MICAL<br>DISP<br>A N I<br>E0-CDI | STOCK<br>DSAL O<br>BARSEL<br>- 15-8 | PILE D<br>F Chem<br>L et f<br>7008 d | ISPOSI<br>1. (U)<br>11. AU(<br>088815- | AL PRO<br>GA TE<br>3 87 G<br>-85-D- | GRAM R<br>Chnolo<br>A-C-18<br>8922 | ISK AN<br>GIES 1<br>563 | IALYSI:<br>INC SAI<br>F/G : | 5 OF<br>N DIEG<br>15/6.3 | <br>7/<br>NL | 13 |
|-------|------------------|---------------------------|----------------------------------|-------------------------------------|--------------------------------------|----------------------------------------|-------------------------------------|------------------------------------|-------------------------|-----------------------------|--------------------------|--------------|----|
|       |                  |                           |                                  |                                     |                                      |                                        |                                     |                                    |                         |                             |                          |              |    |
|       |                  |                           |                                  |                                     |                                      |                                        |                                     |                                    |                         |                             |                          |              |    |
|       |                  |                           |                                  |                                     |                                      |                                        |                                     |                                    |                         |                             |                          |              |    |
|       |                  |                           |                                  |                                     |                                      |                                        |                                     |                                    |                         |                             |                          |              |    |
|       |                  |                           |                                  |                                     |                                      |                                        |                                     |                                    |                         |                             |                          |              |    |
|       |                  |                           |                                  |                                     |                                      |                                        |                                     |                                    |                         |                             |                          |              |    |
|       |                  |                           |                                  |                                     |                                      |                                        |                                     |                                    |                         |                             |                          |              |    |
| 1     |                  |                           |                                  |                                     | <u> </u>                             |                                        |                                     |                                    |                         |                             |                          |              |    |



----

Fig. 8-11. Event tree for air transportation

ANK.

Ś

2222



analafa afanan istan ata bababababababab tab itab itab itab

8-90



- 3. <u>Crush</u> Subjection to structural loads which may be either highly localized or extended over a large area of the package but cannot be categorized as impact, puncture, or immersion.
- 4. <u>Immersion</u> Submersion in a liquid medium.
- 5. <u>Fire</u> Exposure to a high-temperature environment produced by combustion.

Failure threats of crush, impact, puncture, and fire are applied to the accident scenarios. Estimates of the accident consequences are made to determine the number of agent packages expected to fail, given that some probability exists that their failure thresholds were reached.

The air transport sequence designations are consistent with previous programmatic EIS risk analysis (Ref. 8-21) and are defined in Section 4.1. For this report, the XX designation is "AA" or "AB" for air transportation of agent munitions in OFCs within C-5 and C-141 aircraft, respectively.

Table 8-36 lists the accident sequences for air transport to Tooele Army Depot. Utilizing all combinations of the munitions and agent types with the five basic sequence groupings yields a total of 35 possible accident sequences (5 x 7 matrix) from both sites.

### 8.3.3. Accident Sequence Analysis

In order to evaluate the accident frequency of agent release from air transport, several quantities must be determined. These quantities include the failure threshold of the package for each threat to be investigated, the frequency that each of these failure thresholds would occur, and the frequency that each threat being investigated would occur. These quantities are then utilized to perform a logical calculation procedure as specified by Boolean algebra. This section presents





8-91



\*8.a\*6.a\*6.a\*8.a\*8.a\*8.a\*8.a\*8.a\*8.

### TABLE 8-36ACCIDENT SCENARIOS FOR AIR TRANSPORT TO TOOELE ARMY DEPOT

| AB1<br>AA1 | - | A severe ground collision involving an aircraft with munitions occurs and impact forces fail the agent package and munitions.                                                                                |
|------------|---|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| AB2<br>AA2 | - | A severe ground collision involving an aircraft with munitions<br>occurs and impact forces fail the agent package and munitions.<br>A subsequent fire occurs with a duration less than 2 h.                  |
| AB3<br>AA3 | - | A fire occurs aboard an aircraft with munitions and causes rup-<br>ture of the containment due to thermal expansion of the agent.                                                                            |
| AB4<br>AA4 | - | A moderate ground collision involving an aircraft with munitions<br>occurs and impact forces do not fail the agent package and muni-<br>tions. A subsequent fire occurs with a duration greater than<br>2 h. |
| AB5        | - | A moderate ground colligion involving an aircraft with munitions                                                                                                                                             |

AB5 - A moderate ground collision involving an aircraft with munitions
AA5 occurs causing detonation of burstered munitions and a breach of the package. For rockets, the detonation could also arise from impact induced motor ignition.



each fundamental quantity used in the calculational sequence and the results of the calculation.

### 8.3.3.1. Air Transport Data Base and Analysis.

8.3.3.1.1. <u>Fire-Related Probabilities</u>. From the SNL report (Ref. 8-1), the following quantities are of interest in fire-related air transportation accidents:

Fraction of military air accidents which involve fire = 0.35. Fraction of military air accidents which involve impact and fire = 0.22.

If it is assumed that fire accidents are independent of impact accidents, then the following quantity may be determined:

Fraction of military air accidents which involve fire only = 0.13 = 0.35 - 0.22.

Assuming that fire accidents are independent of impact accidents in air transportation is not completely valid. An estimate of the error in this assumption can be made by realizing that the fraction of all air accidents which involve impact is 0.49 and which involve fire is 0.35 (from above). If the problem were completely linear, the fraction of air accidents which involve impact and fire would be (0.49) (0.35) =0.17. But, in fact, the actual fraction of air accidents which involve fire and impact is 0.22 (from above) which demonstrates the nonlinearity of the problem. In this risk analysis, it is assumed that the relative frequencies are separable as shown above (0.13 = 0.35 - 0.22). This assumption is necessary due to the method in which the frequency data is made available. The uncertainty associated with this assumption is no worse than the uncertainty associated with the original data base itself.

Figure 8-12 (from the SNL report) indicates that the maximum fire duration for the C-141 and C-5 aircrafts is 90 and 130 min, respectively. Figure 8-12 is used to determine the failure probabilities due to fire.

Also from SNL report, the expected mean temperature from the probability distribution function for the fire temperature of an air accident fire is approximately 1850°F. Furthermore, the minimum temperature is approximately 1400°F (the minimum burn temperature of JP-4 fuel).

8.3.3.1.2. <u>Impact-Related Probabilities</u>. From SNL report (Ref. 8-1), the following quantities are of interest in impact related air transportation accidents:

Fraction of military air accidents which involve impact = 0.49. Fraction of military air accidents which involve impact and fire = 0.22.

A similar argument as before (fire-related probabilities) allows that an accident involving impact alone (without fire) may be estimated to have a frequency of:

Fraction of military air accidents which involve impact only = 0.27 = 0.49 - 0.22.

Table 8-37 shows the distribution of 149 impact accidents by military aircraft (Ref. 8-1) into landing, takeoff, and inflight. The package will be designed for a deceleration of 35 g. Analyses of potential g forces in aircraft accidents show that for impact angles less than about 10 deg, the cargo would see less than 35 g. Based on 96 military accidents, the percentage of accidents occurring at crash angles less than 10 deg are 85% for takeoff, 87% landing, and 30%



М.



Fig. 8-12. Approximate duration of fuel fire as a function of aircraft size and on-board fuel inventory



|              |    | TABL     | E 8-37   |        |            |
|--------------|----|----------|----------|--------|------------|
| DISTRIBUTION | OF | AIRCRAFT | ACCIDENT | IMPACT | CATEGORIES |

Second Second

| Flight Phase                              |    | Percentage of<br>Accidents |
|-------------------------------------------|----|----------------------------|
| Landing                                   |    | 87/149 = 0.58              |
| Takeoff                                   |    | 14/149 = 0.09              |
| Inflight                                  |    | 48/149 = 0.32              |
| Stalls at an altitude greater than 200 ft | 9  |                            |
| Inflight                                  | 26 |                            |
| Lost at sea                               | 6  |                            |
| Breakup above 5000 ft                     | 4  |                            |
| Severe and unknown                        | _3 |                            |
|                                           | 48 | 0.99                       |

HIGGOOD RECOVER FRANKING

3

d

inflight (Ref. 8-7). Thus, the fraction of accidents which are considered to fail the package is the complement of above values, i.e., 1 - 0.85 = 0.15 for takeoff.

8.3.3.1.3. <u>Puncture-Related Probabilities</u>. The SNL data base (Ref. 8-1) considered puncture from ground or water collision and from fragments, e.g., turboprop blades, which might enter the cargo compartment without collision. Approximately 6% of all military aircraft accidents could result in a cargo puncture threat. SNL did not characterize the magnitude of the cargo puncture environment, i.e., given a puncture threat, the frequency cargo will in fact experience a puncture, because 97.1% of the puncture events either involve an impact threat or the puncture threat is considered relatively benign. However, in 2.9% of the accidents, a missile passed entirely through the cargo compartment; the report did not specify whether any cargo was struck. One percent were combat related; therefore, 1.9% of military aircraft accidents of interest could be considered as causing cargo puncture in the absence of other forces.

The consequences of puncture threats are essentially the same as the consequences of impact threats; therefore, neglecting the puncture threat relative to the impact threat introduces an error of only about 2%.

8.3.3.1.4. <u>Crush-Related Probabilities</u>. The SNL data base (Ref. 8-1) reports a value of 0.05 as the fraction of military impact accidents which also involve crush. Crush is assumed as being possible only in impact accidents. The crush strength of the OFC package is specified in Table 3-1 as 520,000 lb. The SNL aircraft model predicted that crush forces onto the cargo could occur either by (1) crushes from adjacent cargo or by (2) crush due to the aircraft frame. Since only one package is aboard the C-141, a crush from adjacent cargo is not credible. On the C-5 aircraft, there are not enough packages on board to provide sufficient weight to crush a single package. In order for the aircraft itself to crush the cargo, it must be broken into parts. SNL suggests that the most likely weight of a major broken piece to be less than 80,000 lb. This is well below the OFC package crush threshold, and the crush threat to the cargo can be neglected, particularly in comparison to the impact threat.

8.3.3.1.5. Frequency of an Air Accident. Recent communication with the U.S. Air Force has established specific accident rates for the C-141 and C-5 aircraft as of May 1987. An aircraft accident is defined by the Air Force as any aircraft mishap, occurring when there is intent for flight, which results in aircraft damage. Class A accidents are currently defined as accidents producing damage in excess of \$500,000, and the data presented below are for Class A accidents. Data for other accident classes may be useful for the "detonation from undue force" event and have been requested from the Air Force. The accident rate for the C-141 aircraft, is 10 per 5,116,997 flying hours and 2,952,489 landings for the years 1971 through early 1987 (Ref. 8-12). The data presented in Table 8-38 for impact and impact plus fire accidents will be assumed to apply also to fire only accidents; thus, a separate accident rate can be determined for takeoff, inflight, and landing as:

> 3.0 x  $10^{-7}$  takeoff accidents/flight 2.0 x  $10^{-6}$  landing accidents/flight 6.2 x  $10^{-7}$  inflight accidents/hour

Assuming an average flight speed of 500 mph:

Inflight accident rate for C-141 aircraft =  $\frac{6.2 \times 10^{-7}}{\text{flying hour}} \times \frac{1 \text{ h}}{500 \text{ miles}} \times d \text{ (miles)}$ =  $(1.2 \times 10^{-9}) \text{ d}$  inflight accidents/flight

where d is the flight distance in miles.



For the C-5 aircraft, the accident rate is  $1.6 \times 10^{-5}/h$  (Ref. 8-22). Assuming that the average hours per flight for the C-5 is the same as the C-141, the following accident distribution is obtained using the data in Table 8-38:

2.5 x  $10^{-6}$  takeoff accidents/flight 1.6 x  $10^{-5}$  landing accidents/flight 5.2 x  $10^{-6}$  inflight accidents/hour

Assuming an average flight speed of 500 mph results in an inflight accident rate of  $1.0 \times 10-8$  accidents/mile.

Table 8-38 summarizes the fraction of aircraft accidents involving impact and fire. It also gives the inflight accident frequencies for specific site transfers, based on the air route mileages and estimated flight times (Ref. 8-6).

### 8.3.3.2. Failure Thresholds.

8.3.3.2.1. <u>Thermal Thresholds</u>. As stated above, all munitions are to be transported in an OFC package. One package design criterion is that after a 2 h, totally engulfing fire, the bulk agent and propellant temperatures of a M55 rocket are 250° and 400°F, respectively. However, if the fire threat is preceded by another threat, such as impact, then the 2-h value is reduced to 0.5 h. In the above temperature situation, the munition burster is on the threshold of detonation, and detonation is assumed to fail the package. Munitions without a burster might fail due to high agent pressure; however, such failure probably would not cause failure of the package. Due to the lack of specific package design information; however, the package and contained munitions (with or without burster) are assumed to fail after a 2-h fire.

Since the fuel on a C-141 aircraft can be expected to burn no longer than 90 min, no release can be expected for a fire-only accident

### SUMMARY OF AIRCRAFT ACCIDENT THREAT FREQUENCY DATA 0.49 Fraction of accidents producing impact 0.35 Fraction of accidents producing fire Fraction of accidents producing fire and impact 0.22 Fraction of accidents producing fire only 0.13 0.27 Fraction of accidents producing impact only Fraction of accidents exceeding package impact criteria Takeoff (1-0.85)0.15 Inflight (1-0.30) 0.70 Landing (1-0.87)0.13 C-5 accident rates (accidents per flight) Takeoff $2.5 \times 10^{-6}$ $2.1 \times 10^{-5}$ Inflight (APG) $1.5 \times 10^{-5}$ Inflight (LBAD) Landing $1.6 \times 10^{-5}$ C-141 accident rates (accidents per flight) $3.0 \times 10^{-7}$ $2.5 \times 10^{-6}$ Takeoff Inflight (APG) $1.8 \times 10^{-6}$ Inflight (LBAD) $1.8 \times 10^{-6}$ Landing $2.0 \times 10^{-6}$

TABLE 8-38

-5 -5 -5 -7

KULTER FRANKE

1.75555-

8-100

(probability of zero). On the other hand, the fuel on the C-5 aircraft is expected to burn for a period of up to 130 min, and the possibility of release given a fire without impact must be considered. It is assumed that enough C-5 fuel is consumed during takeoff and attaining cruising altitude that the maximum fire duration is reduced to less than 120 min; thus, the possibility of fires lasting more than 2 h is restricted to takeoff accidents only.

For accidents involving impact and fire, either type of aircraft can result in an agent release because only 30-min fire protection is available after the impact. The aircraft will be required to land at TEAD with fuel for at least 45 min of flying time; thus, the possibility of a 30-min fire is considered to exist for all phases of the flight; however, in general only 22% of all aircraft fires last at least 30 min.

Although aircraft fuel is not the only combustible material on board, only fuel fires are considered to challenge the package, i.e., an engulfing fire of 1850°F.

8.3.3.2.2. <u>Mechanical Thresholds</u>. The OFC package is being designed to withstand a deceleration of 35 g. The spectrum of aircraft impact severities is considered in two portions: one for impact severities less than 35 g which do not produce a direct impact package failure, and the second for all impact severities greater than 35 g. The projectiles are expected to fail at impact forces substantially higher than those producing failure for the package; thus, the 35-g division criterion is conservative for projectiles.

Tests with rockets packaged only in the fiberglass shipping tubes showed that impact can cause rocket motor ignition for drops from 40 ft onto a cement test pad (Ref. 8-5). Palletizing the munitions and enclosing the pallets in a package should significantly reduce the frequency of impact-induced ignitions. Without specific data for the extent of mitigation, the frequency for unpalletized rockets will be used, 0.044 ignitions per 40-ft drop. An impact less than the equivalent of a 40-ft drop can be expected in 54% of all aircraft accidents (Ref. 8-1). The accident sequence in which this parameter is being used (No. 4) starts with an impact under 35 g. Thus, the 0.44 frequency of failure applies to impacts greater than 40 ft, but less than 35 g. The fraction of accidents in this impact range is approximated as follows:

- Takeoff 0.85 less than 35 g minus 0.54 less than 40 ft results in 0.31 in the desired range.
- Inflight Conservatively assume that all of the 0.30 fraction less than 35 g result in greater than 40-ft impact.
- Landing 0.87 less than 35 g minus 0.54 less than 40 ft results in 0.33 in the desired range.

8.3.3.3. <u>Sequence Frequency Analysis</u>. The frequencies of the accident sequences listed in Table 8-36 were evaluated as described in Tables 8-39 through 8-43.

To compute the frequency of occurrence of an accident in a particular area around a particular takeoff (or landing) site, e.g., the Chesapeake Bay, multiply the takeoff (or landing) frequency from Section 8.3.3.1.5 by the fractional exposure time to the particular area. For example, if 80% of the takeoffs at APG are in the direction of the bay, one multiplies the takeoff value by 0.8. Results of the accident sequence frequencies are summarized in Table 8-44.

### TABLE 8-39AIR TRANSPORT SEQUENCE 1

AB1 - An aircraft crash occurs in which the package and munitions AA1 are subjected primarily to impact forces. The impact forces are categorized as sufficiently severe to fail the package and munitions. 

| Frequency Factors<br>to be Multiplied | Value                          | 2                    | Reference | e/Remarks |
|---------------------------------------|--------------------------------|----------------------|-----------|-----------|
| Accident rate per flight              | Varies                         |                      | Table     | 8-38      |
| Impact is the only force<br>occurring | 0.27                           |                      | Table     | 8-38      |
| Impact force fails<br>package         | Takeoff<br>Inflight<br>Landing | 0.15<br>0.70<br>0.13 | Table     | 8-38      |







### TABLE 8-40AIR TRANSPORT SEQUENCE 2

AB2 - An aircraft crash occurs in which the package and munitions
AA2 are subjected primarily to impact forces sufficient to fail the package and munitions. Fire occurs and involves agent.

| Frequency Factors<br>to be Multiplied | Valu                           | e                    | Reference | e/Remarks |
|---------------------------------------|--------------------------------|----------------------|-----------|-----------|
| Accident rate per flight              | Varies                         |                      | Table     | 8-38      |
| Impact and fire occur                 | 0.22                           |                      | Table     | 8-38      |
| Impact force fails<br>package         | Takeoff<br>Inflight<br>Landing | 0.15<br>0.70<br>0.13 | Table     | 8-38      |



### TABLE 8-41AIR TRANSPORT SEQUENCE 3

AB3 - A fire occurs on an aircraft. The package and munitions fail AA3 from thermal forces.

| Frequency Factors<br>to be Multiplied                             | Value                       | Reference/Remarks                                                                                       |
|-------------------------------------------------------------------|-----------------------------|---------------------------------------------------------------------------------------------------------|
| Accident rate per flight                                          | Varies                      | Table 8-38                                                                                              |
| Fire is the only force<br>occurring                               | 0.13                        | Table 8-38                                                                                              |
| Fire duration (2 h) suf-<br>ficient to fail package/<br>munitions | € for C-141<br>0.09 for C-5 | Insufficient<br>fuel. Takeoff<br>only (conserva-<br>tively assumes<br>fire will last<br>2 h if started) |







### TABLE 8-42AIR TRANSPORT SEQUENCE 4

AB4 - An aircraft crash occurs, but the impact forces are not suf-AA4 ficient to fail the package. Fire also occurs, and thermal forces fail the package and munitions.

| Frequency Factors<br>to be Multiplied                             | Valu                           | e                    | Reference/Remarks                                                             |
|-------------------------------------------------------------------|--------------------------------|----------------------|-------------------------------------------------------------------------------|
| Accident rate per flight                                          | Varies                         |                      | Table 8-38                                                                    |
| Impact and fire occur                                             | 0.22                           |                      | Table 8-38                                                                    |
| Impact force does not<br>fail package                             | Takeoff<br>Inflight<br>Landing | 0.85<br>0.30<br>0.87 | Table 8-38                                                                    |
| Fire duration (30 min)<br>sufficient to fail<br>package/munitions | 0.22                           |                      | Ref. 8-1, even<br>though sufficient<br>fuel available in<br>all flight phases |

### TABLE 8-43AIR TRANSPORT SEQUENCE 5

AB5 - An aircraft crash occurs, but the impact forces are not
sufficient to fail the package directly. For rockets, the impact may cause motor ignition. For burstered munitions, the undue force causes detonation.

- S.S.S.S.S. TRANSIC

ţ

| Frequency Factors<br>to be Multiplied  | Value                          | <b>e</b>             | Reference/Remarks      |
|----------------------------------------|--------------------------------|----------------------|------------------------|
| Accident rate per flight               | Varies                         |                      | Table 8-38             |
| Impact occurs                          | 0.49                           |                      | Table 8-38             |
| Impact force is not severe             | Takeoff<br>Inflight<br>Landing | 0.85<br>0.30<br>0.87 | Table 8-38             |
| Undue force detonates<br>burster       | 2.1 x 10 <sup>-</sup>          | 3                    | Ref. 8-6               |
| Impact causes rocket<br>motor ignition | 0.044                          |                      | Ref. 8-5               |
| Impact greater than<br>40 ft           | Takeoff<br>Inflight<br>Landing | 0.31<br>0.30<br>0.33 | Also less than<br>35 g |

8~107

|      | TRANSPORTATION  |
|------|-----------------|
|      | AIR             |
| 8-44 | FOR             |
| щ    | 2               |
| TABI | FREQUENCIES ( 4 |

|                 |                    |                             |                        | C-141                   |                        |                        | C-5                    |                        |
|-----------------|--------------------|-----------------------------|------------------------|-------------------------|------------------------|------------------------|------------------------|------------------------|
| Flight<br>Orig. | Sequence<br>Number | Sequence<br>Description     | Takeoff                | Inflight                | Landing                | Takeoff                | Inflight               | Landing                |
| APG             | -                  | Severe impact               | 1.2 × 10 <sup>-8</sup> | 4.7 x 10 <sup>-7</sup>  | 7.0 × 10 <sup>-8</sup> | $1.0 \times 10^{-7}$   | 4.0 × 10 <sup>-6</sup> | $5.6 \times 10^{-7}$   |
| APG             | 2                  | Severe impact + fire        | 9.9 x 10 <sup>-9</sup> | $3.8 \times 10^{-7}$    | $5.7 \times 10^{-8}$   | $8.2 \times 10^{-8}$   | 3.2 × 10 <sup>-6</sup> | 4.6 × 10 <sup>-7</sup> |
| APG             | <b>6</b> 7         | Fire during takeoff         | 0                      | 0                       | 0                      | 2.9 x 10 <sup>-8</sup> | 0                      | 0                      |
| APG             | 4                  | Moderate 1mpact + fire      | $1.2 \times 10^{-8}$   | $3.6 \times 10^{-8}$    | 8.4 × 10 <sup>-8</sup> | $1.0 \times 10^{-7}$   | $3.0 \times 10^{-7}$   | 6.7 × 10 <sup>-7</sup> |
| APG             | S                  | Moderate impact/undue force | 0                      | 0                       | 0                      | 0                      | 0                      | 0                      |
| LBAD            | 1                  | Severe impact               | $1.2 \times 10^{-8}$   | $3.4 \times 10^{-7}$    | $7.0 \times 10^{-8}$   | $1.0 \times 10^{-7}$   | 2.8 × 10 <sup>-6</sup> | 5.6 × 10 <sup>-7</sup> |
| LBAD            | 2                  | Severe impact + fire        | 9.9 x 10 <sup>-9</sup> | $2.8 \times 10^{-7}$    | $5.7 \times 10^{-8}$   | $8.2 \times 10^{-8}$   | 2.3 × 10 <sup>-6</sup> | 4.6 × 10 <sup>-7</sup> |
| LBAD            | ۳                  | Fire during takeoff         | 0                      | 0                       | 0                      | 2.9 × 10 <sup>-8</sup> | 0                      | 0                      |
| LBAD            | 4                  | Moderate impact + fire      | $1.2 \times 10^{-8}$   | 2.6 × 10 <sup>-8</sup>  | 8.4 × 10 <sup>-8</sup> | $1.0 \times 10^{-7}$   | $2.2 \times 10^{-7}$   | 6.7 x 10 <sup>-7</sup> |
| LBAD            | 5(b)               | Moderate impact/undue force | 2.6 × 10-10            | 5.6 × 10 <sup>-10</sup> | 1.8 × 10 <sup>-9</sup> | 2.2 × 10 <sup>-9</sup> | 4.6 x 10 <sup>-9</sup> | 1.4 × 10 <sup>-8</sup> |
| (a)             | er flicht.         |                             |                        |                         |                        |                        |                        |                        |

(a)Per flight. (b)For rockets, increase by a factor of 7.4. hand second second research bases by the second bases of the second bases and the second bases and

Š

8-108

### 8.4. OFFSITE MARINE TRANSPORT

The purpose of this section is to analyze the accident scenarios associated with the specific option of transporting mustard-filled ton containers from Aberdeen Proving Ground (APG), Maryland, by ship to the Johnston Atoll Army site in the Pacific Ocean where they will be disposed of along with the chemical munitions inventory currently stored on the atoll. Vaults will be used as the transport packing concept, instead of the OFCs used for rail. CONTRACT IN COURSE IN MARKING IN CONTRACT

### 8.4.1. Procedures, Assumptions, and Data

The information required to develop the frequency of accidents and the conditional probabilities is taken from the data collected on past marine transportation accidents. Data is compiled from the U.S. Coast Guard Commercial Vessel Safety File, Lloyds Weekly Casualty Reports, the National Transportation Safety Board Reports, and from individual port authority and maritime exchange records. Specific information is available for the LASH ship or ships of similar size and is used to provide the accident frequencies and conditional probabilities for the events. The computer model developed by Engineering Computer Optecnomics, Incorporated (ECO) is then used to integrate the ship specific information with all specific operating situations to obtain the overall accident frequency (Ref. 8-25).

The offsite marine transportation option is modeled by dividing the specific route into hundreds of small segments called port elements. Within each port element, the frequency and conditional probabilities are evaluated. This is done by taking the specific vessel under consideration and interacting that vessel with the situational parameters such as the actual channel configuration and geometry; the water depths, the presence, characteristics, and proximity of fixed or floating obstacles; and any operating factors under which the vessel will be operating, such as tug boat assistance or daylight operations. The results from each

8~109

port element are then aggregated to produce the overall frequency per trip.

In this model, it is the integration of the specific vessel historic incidence rates on a port element by port element basis with the interaction of the specific ship and all specific situational and operating parameters which permit the model to estimate the probability that an accident will occur. For example, if the interaction of the specific vessel with the water depth in a given port element shows water depths too deep (relative to the vessel's draft plus some margin for underkeel clearance) to have a grounding, then a zero value will result for groundings within that particular port element. As another example, if the occurrence of fixed or floating obstacles within a given port element is high, then the interaction analysis will result in an increased ramming incidence rate. On the other hand, the presence of a U.S. Coast Guard escort will tend to mitigate collision occurrences. The use of tug boats will tend to provide added controllability to a ship and thus reduce the occurrence of rammings and groundings. It is assumed that the munitions will be transported in a vault with failure thresholds as specified in Table 3-1.

### 8.4.2. Accident Sequence Definition and Analysis

The accident scenarios for the marine transportation option fall into one of the following families or groups of initiating events:

1. Collisions with other vessels, aircraft, or other moving objects.



- 2. Rammings, defined as collisions with fixed objects (i.e., piers, bridges, anchored, or moored vessels or aids to navigation) or collisions with icebergs, ice fields, or other floating objects moving with a velocity that is relatively slow compared to the striking vessel velocity.
- 3. Groundings in shallow areas.
- 4. On-board fire/explosions.
- 5. Structural failure due to heavy weather such as high winds, hurricanes, tsunamis, etc.
- 6. Aircraft crash into the marine vessel.

These initiating event families correspond to the primary or basic events. Subsequent events, such as fire, show up in the scenarios as conditional events. For example, a fire scenario may be one in which the fire, perhaps originating from spontaneous combustion, is the primary event; however, fire may also be a conditional event if it results from a collision.

Once any accident has occurred, the vault and the munition inside have five modes of failure which could be produced and result in chemical agent release. These failure modes are defined below:

- Impact striking or being struck by an object which has no sharp projections.
- 2. Puncture striking or being struck by an object which penetrates the protective structures of the container.
- 3. Crush subjection to structural loads which may be either highly localized or extended over a large area of the container but cannot be categorized as impact, puncture, or immersion.
- 4. Immersion submersion in a liquid medium.
- 5. Fire exposure to a high-temperature environment produced by combustion.

In the case of marine transportation, impact and puncture are not the dominant failure forces experienced in an accident. One reason is that for both the lighter and the ship, the cargo will be adequately braced to hold the cargo in place. The second reason is that the majority of the events are low-velocity, high-momentum events, thus the dominant failure mode is crush. Only the failure modes crush, immersion, and fire have been examined in this analysis.

The scenarios corresponding to the six initiating event families were developed using computer analyses and event trees. Figure 8-13 shows the event tree for vessel collisions. Similar trees apply to the other initiating event families.

Once the possible accident sequences were defined, they were assigned an accident scenario identification number according to the coding described in Section 4.1.

For the transport of the mistard-filled ton containers from APG, the marine transportation mode, XX, was broken down into four segments: (1) "BI" for barges in inland waters, (2) "LI" for the ship inland waters, (3) "LC" for the ship in coastal waters, and (4) "LS" for the ship at sea. The main reason for breaking the transportation route into the above segments is that it is necessary to identify not only the transportation vehicle (i.e., barge or ship) but also where the accident occurs such as near population zones, near areas of higher traffic density, or over specific ocean depths.



Ŵ

| INITIATIN     | IG EVENT               | MUNITION | FIRE<br>Avninfi        | SINKING | SEQUENCE                                 | FREQUENCY<br>PEA TRIP  | TYPE OF<br>AGENT |
|---------------|------------------------|----------|------------------------|---------|------------------------------------------|------------------------|------------------|
| CONFIGURATION | LOCATION               |          |                        |         | 2                                        |                        | RELEASE          |
| BARGE         | INLAND                 | 6.0      |                        |         | SI S | 1                      | NONE             |
|               | 5.0 × 10 <sup>-5</sup> | 1.0      | 190                    |         | BIKHSDBI                                 | ,<br>,                 | EVAPORATION      |
|               |                        |          |                        | 1~      | BIKHS802                                 | 3.1 × 10 <sup>-6</sup> | WATER ONLY       |
|               |                        |          | 0.39                   | -       | BIKHC003                                 | ÷                      | BURN             |
|               |                        |          |                        | ~ 1     | BIKHC004                                 | 2.0 × 10 <sup>-6</sup> | BURN             |
| LASH          | INLAND                 | 0.96     |                        |         | LIKHS619                                 | -                      | NONE             |
|               | 1.8 × 10-4             | 0.02     | 8.0                    | 0.95    | LIKHSBOI                                 | 2.7 × 10 <sup>-6</sup> | EVAPORATION      |
|               |                        |          |                        | 0 05    | LIKHS002                                 | 1.4 × 10 <sup>-7</sup> | WATER ONLY       |
|               |                        |          | 0 2                    | 0 95    | LIKHC003                                 | 6.7 × 10 <sup>-7</sup> | BURN             |
|               |                        |          |                        | 0.05    | LIKHCODA                                 | 3.5 × 10 <sup>-8</sup> | BURN             |
|               | COASTAL                | 96 0     |                        |         | LCKHS019                                 | -                      | NONE             |
|               | 5-01×18                | 0 02     | 08                     | 0.95    | LCKHS001                                 | 1.2 × 10 <sup>-6</sup> | EVAPORATION      |
|               |                        |          |                        | 0.05    | LCKHS002                                 | 6.5 × 10-8             | WATER ONLY       |
|               |                        |          | 0 2                    | 0 95    | LCKHC003                                 | 3.1 × 10 <sup>-7</sup> | BURN             |
|               |                        |          |                        | 0.05    | LCKHC004                                 | 8-01×91                | BURN             |
| ليبين         | SEA                    | 0 98     | ' ]<br> <br> <br> <br> |         | LSKHS019                                 | -                      | NONE             |
|               | c, 01 × 8 1            | 0.02     | 0.8                    | 0 95    | LSKHS001                                 | 2.8 × 10 - 7           | EVAPORATION      |
|               |                        |          |                        | 0 05    | 1 SKHS002                                | 1.5 × 10 · 8           | WATER ONLY       |
|               |                        |          | 0 2                    | 0 95    | LSKHC003                                 | 70×10 <sup>-8</sup>    | BURN             |
|               |                        |          |                        | 0.05    | LSKHC004                                 | 3.7 × 10 -9            | BURN             |
|               |                        |          |                        |         |                                          |                        |                  |

Event tree for marine vessel collision initiating event Fig. 8-13.

المنتخذذ

2.1.2

3

The accident sequences resulting from the analysis are listed in Table 8-45.

60,850.5

Three types of analyses, interaction, penetration, and traffic, were performed to provide input into the computer model for the offsite portion of the risk analysis.

For the interaction analysis, the specific vessel information is examined as well as information on the channel configuration, geometry, water depths, and the proximity of fixed or floating obstacles within each of the subdivisions called port elements. An example of how this is used is the case for grounding. Over one port element, a specific accident frequency exists from the data base from a specific vessel size such as a LASH vessel or lighter. This accident frequency will depend on many things, such as, the configuration of the channel and the water depth. In shallow waters, the accident frequency will obviously be higher. For deep, wide channels, the incidences of groundings is zero. As a result, the accident frequency will vary from port element to port element, and the total accident rate is therefore the aggregated probability of occurrence of a grounding accident over the entire string of port elements. Similarly, this is done for collisions and rammings.

Given the occurrence of a collision, grounding, or ramming accident, the model then examines the data base to determine the severity of the resulting damage. This is referred to as the penetration analysis. The penetration analysis provides the conditional probability of the extent of structural damage or penetration necessary to reach the packages containing the chemical munitions agent and thus, release the agent. The severity of the mechanical forces or the extent of penetration is a function of the energy transfer between the colliding bodies (i.e., ship and ship, or ship and obstacle, or ship and sea bottom), which in turn is a function of the size, speed, collision geometry, and structural characteristics of the ship and the other ship (in the case

| Initiating Event  | Fire | Sink | Barge<br>Inland | LASH<br>Inland | LASH<br>Coastal | LASH<br>Sea |
|-------------------|------|------|-----------------|----------------|-----------------|-------------|
| Collision and     | N    | N    | BI01            | LI01           | LC01            | LS01        |
| munition failure  | N    | Y    | BI02            | LI02           | LC02            | LS02        |
|                   | Y    | N    | BIO3            | LI03           | LC03            | LS03        |
|                   | Y    | Y    | BI04            | LI04           | LC04            | LS04        |
| Ramming and       | N    | N    | <b>BI05</b>     | L105           | LC05            | LS05        |
| munition failure  | N    | Y    | <b>BI06</b>     | L106           | LC06            | LS06        |
|                   | Y    | N    | BIO7            | L107           | LC07            | LS07        |
|                   | Y    | Y    | BI08            | L108           | LC08            | LS08        |
| Grounding and     | N    | N    | <b>BI09</b>     | LI09           | LC09            | LS09        |
| munition failure  | N    | Y    | BI10            | LI10           | LC10            | LS10        |
|                   | Y    | N    | BI11            | LI11           | LC11            | LS11        |
|                   | Y    | Y    | BI12            | LI12           | LC12            | LS12        |
| Heavy weather and | N    | N    | BI13            | LI13           | LC13            | LS13        |
| munition failure  | N    | Y    | BI14            | LI14           | LC14            | LS14        |
|                   | Y    | N    | <b>BI15</b>     | LI15           | LC15            | LS15        |
|                   | Y    | Y    | BI16            | LI16           | LC16            | LS16        |
| On-board fire and |      | N    | BI17            | LI17           | LC17            | LS17        |
| munition failure  |      | Y    | BI18            | LI18           | LC18            | LS18        |
| Collision(a)      | N    | Y    | BI19            | LI19           | LC19            | LS19        |
| Ramming(a)        | N    | Y    | <b>BI20</b>     | L120           | LC20            | LS20        |
| Grounding(a)      | N    | Y    | BI21            | LI21           | LC21            | LS21        |
| Heavy weather(a)  | N    | Y    | BI22            | LI22           | LC22            | LS22        |
| Aircraft crash    | Y    | N    | B123            | L123           | LC23            | LS23        |

TABLE 8-45SUMMARY OF OFFSITE MARINE TRANSPORT SEQUENCES

.

(a) No immediate munition failure in these sequences.

of collisions) or the fixed or floating obstacle (in the case of rammings) or the contour and constituency of the bottom (in the case of groundings).

The last type of analysis performed, the traffic analysis, provides the number and size distribution vessel encountered and the geometric orientation of those encounters relative to the LASH ship or lighter and the port elements in which those encounters are likely to occur over the transit. These are all derived from historic traffic data of the port, environmental factors, and the port geometry.

The computer model begins by searching both the accident data file and the traffic file for the overall area under consideration (i.e., the Chesapeake Bay from the anchorage in the vicinity of the Bush River to the Virginia Capes) and for ships of similar size and similar characteristics. It then calculates the basic event frequency. In this example it is the historic collision accident rate (i.e., collision event per transit).

2

Within each port element, the model also estimates a percentage decrease in accident rate for any controlled parameters within the system. In the illustrated case, the effect of two controlled parameters are integrated: namely, the Coast Guard escort and the tug boats. The percentage decreases in the expected accident rate for both parameters are functions of the port element channel width, depth, and geometry and port traffic (in terms of both vessel encounters and the historic traffic size distribution). In the case of the tug boat parameter, there is an additional independent variable and that is the two different modes of tug boat utilization in different port elements; i.e., providing direct, positive assistance (tied on) or trailing astern to provide assistance in the event of a propulsion or steering failure (in attendance). As can be seen from the flow diagram, the value which results is the estimated frequency of any collision accident in the port element under consideration. The estimated frequency of the collision accident is then subdivided into those with "No Release" and those with a

1 x 3 4 1 4 8 8 4

As the example shows, within each of these two major divisions of accidents, No Release and Release, the model further breaks down these two categories:

- 1. The vessel not sinking and no subsequent fire occurring.
- 2. The vessel sinking and no subsequent fire occurring.
- 3. The vessel not sinking and a subsequent fire occurring.
- 4. The vessel both sinking and having a subsequent fire occur.

In each branch, the model determines a series of conditional probabilities associated with the estimated probability for the collision event (with or without a release, as the case may be) in order to determine the estimated probabilities for the four combinations of with and without sinking and with and without a subsequent fire occurring. As an example, the branch for the probability of a release from a collision with sinking only no subsequent fire occurring, begins with a determination of the conditional probability of the LASH ship being the struck vessel in the accident. It then determines the conditional probability of the length of damage being greater than (>) some value "X". X is an input which is related to the ship's inherent survivability characteristics and is the length necessary to flood sufficient compartments along the ship to cause the ship to sink for any extent of transverse penetration. Based on both historical data and the LASH ship's loading arrangement in all three dimensions, the model next determines the conditional probability of the damage being within the cargo area, as opposed to the engine room and other noncargo hold areas. From historical data, it then determines the conditional probability of the transverse penetration being greater than (>) some value "Y". In the case of the LASH ship as loaded with the ton containers within the steel packages within the lighters, Y is the transverse distance from the outboard



side of the ship's shell plating to the first line of packages within the lighters or a distance in excess of 32 ft.

The conditional probability of transverse penetration comes from historical data and is dependent upon three factors. The first one is the striking ship's size or mass which in turn is a function of the port's traffic size distribution. The second is the striking ship's speed which in turn is a function of the port's traffic characteristics and the port element under consideration and in particular, the port element's channel depth, width, and geometry, all of which affect the speed of ships within that channel. The third and last factor is the angle of incidence between the striking ship and the struck LASH ship which in turn is a function of the channel's geometry, such as turns or intersecting channels.

The model next determines the conditional probability of no ignition which is 1 minus the conditional probability of ignition for collisions. The conditional probability of ignition for collisions is determined from historical data and the flammability characteristics of the chemical munitions agent relative to the products ignited by collisions within the accident data base.

The last portion of this branch determines the conditional probability of the water depth within four ship lengths of either side of the channel(s) in the port element being greater than (>) some value D and thus, the conditional probability of the vessel sinking to some depth, D. In this instance, D was input as 60 ft or the depth necessary to bring the water to the edge of the main deck.

The computer results are expressed as the accident frequency per trip. A limiting number of  $3.0 \times 10^{-9}$  is included in the computer model so that values less than this are truncated and only the truncated value is reported. The results of the computer output are given in Tables 8-46 and 8-47. Table 8-46 is the frequency of a lighter accident which will cause a release (R). The failure mode for each of these



# TABLE 8-46FREQUENCY OF AGENT RELEASE FOR THE LIGHTERIN THE CHESAPEAKE BAY (BUSH RIVER TO VIRGINIA CAPES)

| Scenario<br>Elements | Collision<br>Frequency<br>Per Trip | Grounding<br>Frequency<br>Per Trip | Ramming<br>Frequency<br>Per Trip | Structural<br>Frequency<br>Per Trip | Total<br>Frequency<br>Per Trip |
|----------------------|------------------------------------|------------------------------------|----------------------------------|-------------------------------------|--------------------------------|
| R/NS/NF              | (a)                                | (a)                                | (a)                              | (a)                                 | 0                              |
| R/S/NF               | $3.1 \times 10^{-6}$               | <3.0 x 10 <sup>-9</sup>            | <3.0 x 10 <sup>-9</sup>          | <3.0 x 10 <sup>-9</sup>             | $3.1 \times 10^{-6}$           |
| R/NS/F               | (a)                                | (a)                                | (a)                              | (a)                                 | 0                              |
| R/S/F                | $2.0 \times 10^{-6}$               | $<3.0 \times 10^{-9}$              | <3.0 x 10 <sup>-9</sup>          | <3.0 x 10 <sup>-9</sup>             | $2.0 \times 10^{-6}$           |
| R/ALL                | 5.1 x 10 <sup>-6</sup>             | <3.0 x 10 <sup>-9</sup>            | <3.0 x 10 <sup>-9</sup>          | <3.0 x 10 <sup>-9</sup>             | 5.1 x 10 <sup>-6</sup>         |

(a)For lighters, release (R) and no sinking (NS) are mutually exclusive. Fires (F) and no fires (NF) are also mutually exclusive.





|      | ВАҮ        |              |
|------|------------|--------------|
|      | CHESAPEAKE |              |
|      | THE        | [S]          |
|      | NI         | CAPE         |
| ~    | SHIP       | NIA (        |
| 8-4  | THE        | <b>/IRGI</b> |
| ABLE | FOR        | TO           |
| F    | ELEASE     | RIVER        |
|      | IT R       | USH          |
|      | AGEN       |              |
|      | OF         |              |
|      | FREQUENCY  |              |

| Scenario<br>Elements | Collision<br>Frequency<br>Per Trip | Grounding<br>Frequency<br>Per Trip | Ramming<br>Frequency<br>Per Trip | Fire<br>Frequency<br>Per Trip | Structural<br>Frequency<br>Per Trip | Total<br>Frequency<br>Per Trip |
|----------------------|------------------------------------|------------------------------------|----------------------------------|-------------------------------|-------------------------------------|--------------------------------|
| R/NS/NF              | 2.7 × 10 <sup>-6</sup>             | 1.7 × 10 <sup>-6</sup>             | $1.2 \times 10^{-7}$             | (a)                           | <3.0 x 10 <sup>-9</sup>             | 4.6 x 10 <sup>-6</sup>         |
| R/S/NF               | $1.4 \times 10^{-7}$               | 1.4 × 10 <sup>-7</sup>             | <3.0 × 10 <sup>-9</sup>          | (a)                           | <3.0 × 10 <sup>-9</sup>             | $2.8 \times 10^{-7}$           |
| R/NS/F               | $6.7 \times 10^{-7}$               | $3.6 \times 10^{-8}$               | 3.0 × 10 <sup>-8</sup>           | <3.0 × 10 <sup>-9</sup>       | <3.0 x 10 <sup>-9</sup>             | $7.3 \times 10^{-7}$           |
| R/S/F                | 3.5 x 10 <sup>-8</sup>             | <3.0 × 10 <sup>-9</sup>            | <3.0 x 10 <sup>-9</sup>          | <3.0 x 10 <sup>-9</sup>       | <3.0 x 10 <sup>-9</sup>             | $3.5 \times 10^{-8}$           |
| R/ALL                | 3.5 x 10 <sup>-6</sup>             | 1.9 x 10 <sup>-6</sup>             | $1.5 \times 10^{-7}$             | <3.0 x 10 <sup>-9</sup>       | <3.0 x 10 <sup>-9</sup>             | 5.6 x 10 <sup>-6</sup>         |
| (8) 51 22            | (P) as the act                     | -idant event and                   | I no fire (NF)                   | are mutually exclu            | istve. S and NS                     | denote                         |

(a) fire (F), as the accident event, and no fi sinking and no sinking; R denotes agent release.



agent releases are crush, but the probabilities of releases with sinking (S) and without sinking (NS) and for fire (F) and without fire (NF) have also been calculated because the consequences may vary as a result of these additional conditions. As shown in Table 8-46 for lighters, a release and no sinking are mutually exclusive, since a severe enough accident to cause a release will also sink the lighter. The probability for grounding, rammings, and structural damage are truncated at the  $3.0 \times 10^{-9}$  value.

Tables 8-47 through 8-49 contain the frequency of a LASH ship accident which will cause a release. These are evaluated for the bay area, the coastal area, and the high seas, respectively. Fire has been added as an initiating event, since unlike the lighter, the ship can contain an ignition source.

The last table of the computer model output, Table 8-50, is the frequency of a ship accident, but one which is not severe enough to cause an immediate release. These accidents are reported since sinking the LASH ship on the high seas may result in a rupture of the ton containers at ocean depths and allows the consequences of such an accident to be evaluated.

In addition to the accidents which were specifically model by the computer, it was requested that the scenario of an aircraft crashing into a lighter or a LASH vessel be developed. An earlier study (Ref. 8-26) for LNG ships in the Chesapeake Bay area has estimated this frequency of occurrences a 2.7 x  $10^{-9}$  accidents per ship transit. This number is based on all type aircraft for a ship in transit, and is used for the accident frequency for the LASH vessel in transit.

The aircraft crash frequencies specifically for APG are presented in Section 4.2. These are given in accidents per square mile per year. This value was used as the basis for the frequency of occurrence for the lighters. For aircraft crashes into the lighter which will be loaded

|      | AREAS     |  |
|------|-----------|--|
|      | COASTAL   |  |
|      | THE       |  |
|      | NI        |  |
| ~    | SHIP      |  |
| 8-41 | THE       |  |
| ABLE | FOR       |  |
| 5    | RELEASE   |  |
|      | AGENT     |  |
|      | OF        |  |
|      | FREQUENCY |  |

|                      | Collision              | Grounding               | Ramming                 | Fire                    | Structural              | Total                  |
|----------------------|------------------------|-------------------------|-------------------------|-------------------------|-------------------------|------------------------|
| Scenario<br>Elements | Frequency<br>Per Trip  | Frequency<br>Per Trip   | Frequency<br>Per Trip   | Frequency<br>Per Trip   | Frequency<br>Per Trip   | Frequency<br>Per Trip  |
| R/NS/NF              | 1.2 × 10 <sup>-6</sup> | 5.1 × 10 <sup>-7</sup>  | 8.0 × 10 <sup>-8</sup>  | (a)                     | <3.0 x 10 <sup>-9</sup> | 1.8 x 10 <sup>-6</sup> |
| R/S/NF               | 6.5 x 10 <sup>-8</sup> | 4.2 x 10 <sup>-8</sup>  | <3.0 x 10 <sup>-9</sup> | (8)                     | <3.0 × 10 <sup>-9</sup> | $1.1 \times 10^{-7}$   |
| R/NS/F               | $3.1 \times 10^{-7}$   | $1.0 \times 10^{-8}$    | $2.0 \times 10^{-8}$    | <3.0 × 10 <sup>-9</sup> | <3.0 x 10 <sup>-9</sup> | 3.4 × 10 <sup>-7</sup> |
| R/S/F                | $1.6 \times 10^{-8}$   | <3.0 × 10 <sup>-9</sup> | <3.0 x 10 <sup>-9</sup> | <3.0 × 10 <sup>-9</sup> | <3.0 x 10 <sup>-9</sup> | 1.6 x 10 <sup>-8</sup> |
| R/ALL                | 1.6 x 10 <sup>-6</sup> | 5.6 x 10 <sup>-7</sup>  | 10.0 x 10 <sup>-8</sup> | <3.0 × 10 <sup>-9</sup> | <3.0 × 10 <sup>-9</sup> | 2.3 x 10 <sup>-6</sup> |
| (a)Fire              | (F). As the acc        | cident event. and       | ino fire (NF) ai        | ce mitually exclu       | stve. Sand NS           | denote                 |

sinking and no sinking; R denotes agent release. alla



|                               |                                     | • -                     |                         |                         | _                       |           |
|-------------------------------|-------------------------------------|-------------------------|-------------------------|-------------------------|-------------------------|-----------|
|                               | Total<br>Frequency<br>Per Trip      | 3.8 × 10 <sup>-7</sup>  | 1.8 × 10 <sup>-8</sup>  | 8.5 × 10 <sup>-8</sup>  | 3.7 × 10 <sup>-9</sup>  |           |
| H SEAS                        | Structural<br>Frequency<br>Per Trip | <3.0 × 10 <sup>-9</sup> | <3.0 × 10 <sup>-9</sup> | <3.0 x 10 <sup>-9</sup> | <3.0 x 10 <sup>-9</sup> | 0-01      |
| SHIP ON THE HIG               | Fire<br>Frequency<br>Per Trip       | (B)                     | (a)                     | <3.0 x 10 <sup>-9</sup> | <3.0 x 10 <sup>-9</sup> | 0-01 0 01 |
| TABLE 8-49<br>Release for the | Ramming<br>Frequency<br>Per Trip    | 6.2 × 10 <sup>-8</sup>  | <3.0 × 10 <sup>-9</sup> | 1.6 × 10 <sup>-8</sup>  | <3.0 x 10 <sup>7</sup>  | 8         |
| EQUENCY OF AGENT              | Grounding<br>Frequency<br>Per Trip  | 4.3 x 10 <sup>-8</sup>  | 3.5 x 10 <sup>-9</sup>  | <3.0 x 10 <sup>-9</sup> | <3.0 × 10 <sup>-9</sup> |           |
| E E                           | Collision<br>Frequency<br>Per Trip  | 2.8 x 10 <sup>-7</sup>  | $1.5 \times 10^{-8}$    | $7.0 \times 10^{-8}$    | 3.7 × 10 <sup>-9</sup>  | r         |
|                               | Scenario<br>Elements                | R/NS/NF                 | R/S/NF                  | R/NS/F                  | R/S/F                   |           |

> S and NS denote <sup>(a)</sup>Fire (F), as the accident event, and no fire (NF) are mutually exclusive. sinking and no sinking; R denotes agent release.

> > ر مه می هو می مرکز می وارد از مرابع می موجه با مرکز می مرکز می از مرکز می از مرکز می مرکز می مرکز می مرکز می م مرکز می می مرکز می وارد می مرکز می مرکز

9. KU

|                             | 0          | ming               | rire<br>Vreauenau     | Structural              | Total<br>Frequency |
|-----------------------------|------------|--------------------|-----------------------|-------------------------|--------------------|
| uency rieque<br>Trin Per Tr | rency ried | Trin               | riequency<br>Per Trio | Fer Trip                | Per Trip           |
| 11 121 121 111              | 111        |                    | 4                     |                         |                    |
| x 10 <sup>-5</sup> 5.5 x 1  | 10-6 1.3   | x 10 <sup>-5</sup> | (B)                   | <3.0 × 10 <sup>-9</sup> |                    |

TABLE 8-50

 $1.0 \times 10^{-7}$ 

<3.0 x 10<sup>-9</sup>

(B) (a)

1.3 x 10<sup>-5</sup> <3.0 x 10<sup>-9</sup> (૧)

5.5 x 10<sup>-6</sup> 3.9 x 10<sup>-8</sup>

6.1 × 10<sup>-8</sup> 1.8 × 10<sup>-5</sup>

NR/NS/NF NR/S/NF NR/NS/F

 $0.0 \times 10^{0}$ 

| king (S) and           | usive, as are sin       | are mutually excl     | nd no fire (NF)        | ccident event, a       | (F), as the a          | (a)Fire |
|------------------------|-------------------------|-----------------------|------------------------|------------------------|------------------------|---------|
| 3.7 × 10 <sup>-5</sup> | <3.0 × 10 <sup>-9</sup> | 0.0 × 10 <sup>0</sup> | 1.3 x 10 <sup>-5</sup> | 5.6 x 10 <sup>-6</sup> | 1.8 x 10 <sup>-5</sup> | NR/ALL  |
| 0.0 × 10 <sup>0</sup>  | (ঀ)                     | ( <b>þ</b> )          | (٩)                    | ( <b>q</b> )           | ( <b>p</b> )           | NR/S/F  |
| 0.0 × 10 <sup>0</sup>  | (q)                     | (q)                   | ( <b>p</b> )           | (q)                    | (q)                    | NR/NS/F |

no sinking (NS).

(b)An accident severe enough to cause a fire would result in a release, therefore no release (NR) and fire (F) are mutually exclusive. 6

55555 Mark 655255

•

and stored near the site, the accident frequency is based on the accident rates for a 31.2 x 61.5 ft lighter with a maximum of ten lighters in the loading area. The probability is  $2.1 \times 10^{-7}$  accidents per year. This is lower than that for open storage, as expected, since there is less square mileage. This number is also used for the lighters in transit from the storage area to the LASH vessel since they will be towed in flotillas of 10 and because of the close proximity of the ship to the storage area.

The results have been formatted in the accident sequence identification format and are presented in Table 8-51. The collision accident frequencies are the first four accident scenarios. The first collision sequence is for the no sinking and no fire situation for the lighter and the three separate ship transport areas. The second collision sequence is for the sinking and no fire situation for the lighter and for the three separate ship transport areas and likewise down the list for rammings, groundings, fire, and structural damage.



アナシャ

77

## TABLE 8-51FREQUENCIES OF OFFSITE MARINE TRANSPORT ACCIDENT SEQUENCES

222222

| Sequence<br>I.D.                               | Sequence Description                                                                                           | Frequency<br>Per Trip                                                                                                                                        |
|------------------------------------------------|----------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BIKHS001 -<br>LIKHS001<br>LCKHS001<br>LSKHS001 | A collision occurs and crush forces fail agent containment.                                                    | N/A<br>2.7 x 10 <sup>-6</sup><br>1.2 x 10 <sup>-6</sup><br>2.8 X 10 <sup>-7</sup>                                                                            |
| BIKHS002 -<br>LIKHS002<br>LCKHS002<br>LSKHS002 | A collision occurs and crush forces<br>fail agent containment. Sinking<br>also occurs.                         | 3.1 x 10 <sup>-6</sup><br>1.4 x 10 <sup>-7</sup><br>6.5 x 10 <sup>-8</sup>                                                                                   |
| BIKHC003 -<br>LIKHC003<br>LCKHC003<br>LSKHC003 | A collision occurs and crush forces fail agent containment. A fire breaks out.                                 | N/A<br>6.7 x 10 <sup>-7</sup><br>3.1 x 10 <sup>-7</sup><br>7.0 x 10 <sup>-8</sup>                                                                            |
| BIKHC004 -<br>LIKHC004<br>LCKHC004<br>LSKHC004 | A collision occurs and crush forces<br>fail agent containment. A fire breaks<br>out and sinking occurs.        | 2.0 x $10^{-6}$<br>3.5 x $10^{-8}$<br>1.6 x $10^{-8}$<br>3.7 x $10^{-9}$                                                                                     |
| BIKHS005 -<br>LIKHS005<br>LCKHS005<br>LSKHS005 | A ramming occurs and crush forces fail agent containment.                                                      | N/A<br>1.2 x 10 <sup>-7</sup><br>8.0 x 10 <sup>-8</sup><br>3.7 x 10 <sup>-9</sup>                                                                            |
| BIKHS006 -<br>LIKHS006<br>LCKHS006<br>LSKHS006 | A ramming occurs and crush forces fail agent containment. Sinking also occurs.                                 | $\langle 3.0 \times 10^{-9} \\ \langle 3.0 \times 10^{-9} \rangle$ |
| BIKHC007 -<br>LIKHC007<br>LCKHC007<br>LSKHC007 | A ramming accident occurs and crush<br>forces fail agent containment. A fire<br>breaks out.                    | N/A<br>3.0 x $10^{-8}$<br>2.0 x $10^{-8}$<br>1.6 x $10^{-8}$                                                                                                 |
| BIKHC008 -<br>LIKHC008<br>LCKHC008<br>LSKHC008 | A ramming accident occurs and crush<br>forces fail agent containment. A fire<br>breaks out and sinking occurs. | $\langle 3.0 \times 10^{-9} \\ \langle 3.0 \times 10^{-9} \rangle$ |
| BIKHS009 -<br>LIKHS009<br>LCKHS009<br>LSKHS009 | A grounding accident occurs and crush forces fail agent containment.                                           | N/A<br>1.8 x 10 <sup>-6</sup><br>5.1 x 10 <sup>-7</sup><br>4.3 x 10 <sup>-8</sup>                                                                            |



**HELPOTENE** 

#### TABLE 8-51 (Continued)

| Sequence<br>I.D.                               | Sequence Description                                                                                                               | Frequency<br>Per Trip                                                                                                                                        |
|------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BIKHS010 -<br>LIKHS010<br>LCKHS010<br>LSKHS010 | A grounding accident occurs and crush<br>forces fail agent containment. Sinking<br>also occurs.                                    | $\langle 3.0 \times 10^{-9} \\ 1.4 \times 10^{-7} \\ 4.2 \times 10^{-8} \\ 3.5 \times 10^{-9} $                                                              |
| BIKHCO11 -<br>LIKHCO11<br>LCKHCO11<br>LSKHCO11 | A grounding accident occurs and crush<br>forces fail agent containment. A fire<br>breaks out.                                      | N/A<br>3.6 x $10^{-8}$<br>1.0 x $10^{-8}$<br>$\langle 3.0 x 10^{-9} \rangle$                                                                                 |
| BIKHC012 -<br>LIKHC012<br>LCKHC012<br>LSKHC012 | A grounding accident occurs and crush<br>forces fail agent containment. A fire<br>breaks out and sinking occurs.                   | $\langle 3.0 \times 10^{-9} \\ \langle 3.0 \times 10^{-9} \rangle$ |
| BIKHS013 -<br>LIKHS013<br>LCKHS013<br>LSKHS013 | Structural damage due to heavy weather occurs. Crush forces fail agent containment.                                                | N/A<br><3.0 x 10 <sup>-9</sup><br><3.0 x 10 <sup>-9</sup><br><3.0 x 10 <sup>-9</sup>                                                                         |
| BIKHS014 -<br>LIKHS014<br>LCKHS014<br>LSKHS014 | Structural damage due to heavy weather<br>occurs. Crush forces fail agent<br>containment. Sinking also occurs.                     | $\langle 3.0 \times 10^{-9} \\ \langle 3.0 \times 10^{-9} \rangle$ |
| BIKHC015 -<br>LIKHC015<br>LCKHC015<br>LSKHC015 | Structural damage due to heavy weather<br>occurs. Crush forces fail agent<br>containment. A fire breaks out.                       | N/A<br><3.0 x 10 <sup>-9</sup><br><3.0 x 10 <sup>-9</sup><br><3.0 x 10 <sup>-9</sup>                                                                         |
| BIKHC016 -<br>LIKHC016<br>LCKHC016<br>LSKHC016 | Structural damage due to heavy weather<br>occurs. Crush forces fail agent<br>containment. A fire breaks out and<br>sinking occurs. | $\langle 3.0 \times 10^{-9} \\ \langle 3.0 \times 10^{-9} \rangle$ |
| BIKHF017 -<br>LIKHF017<br>LCKHF017<br>LSKHF017 | Spontaneous fire occurs.                                                                                                           | N/A<br><3.0 x 10 <sup>-9</sup><br><3.0 x 10 <sup>-9</sup><br><3.0 x 10 <sup>-9</sup>                                                                         |
| BIKHC018 -<br>LIKHC018<br>LCKHC018<br>LSKHC018 | Spontaneous fire occurs. Sinking<br>also occurs.                                                                                   | N/A<br>$\langle 3.0 \times 10^{-9} \\ \langle 3.0 \times 10^{-9} \\ \langle 3.0 \times 10^{-9} \\ \langle 3.0 \times 10^{-9} \rangle$                        |



8-127

| INDER 0-DI (concinned | TABLE | 8-51 | (Continued) |
|-----------------------|-------|------|-------------|
|-----------------------|-------|------|-------------|

| Sequence<br>I.D.                               | Sequence Description                                                                                | Frequency<br>Per Trip                                                                                                                                        |
|------------------------------------------------|-----------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BIKHS019 -<br>LIKHS019<br>LCKHS019<br>LSKHS019 | Collision accident occurs with no<br>immediate release. Sinking also<br>occurs.                     | 3.5 x 10-6<br>5.8 x 10-7<br>2.7 x 10-7<br>6.1 x 10-8                                                                                                         |
| BIKHS020 -<br>LIKHS020<br>LCKHS020<br>LSKHS020 | Ramming accident occurs with no<br>immediate release. Sinking also<br>occurs.                       | $3.3 \times 10^{-6}$<br>$< 3.0 \times 10^{-9}$<br>$< 3.0 \times 10^{-9}$<br>$3.9 \times 10^{-8}$                                                             |
| BIKHS021 -<br>LIKHS021<br>LCKHS021<br>LSKHS021 | Grounding accident occurs with no<br>immediate release. Sinking also<br>occurs.                     | $1.6 \times 10^{-6}$<br>1.6 x 10^{-6}<br>4.7 x 10^{-7}<br>$\langle 3.0 \times 10^{-9} \rangle$                                                               |
| BIKHS022 -<br>LIKHS022<br>LCKHS022<br>LSKHS022 | Structural damage due to heavy weather<br>occurs with no immediate release.<br>Sinking also occurs. | $\langle 3.0 \times 10^{-9} \\ \langle 3.0 \times 10^{-9} \rangle$ |
| BIKHS023 -<br>LIKHS023<br>LCKHS023<br>LSKHS023 | Aircraft crashes into marine vessel.                                                                | 2.1 x 10 <sup>-7</sup><br>2.7 x 10 <sup>-9</sup><br>2.7 x 10 <sup>-9</sup><br>2.7 x 10 <sup>-9</sup>                                                         |

Ŋ,

ROUGH STATE

3



Contraction of the second seco

8-128

ч ().

The results of the uncertainty analysis indicate that the 95th percentile values may be up to: (1) 160 times higher than the median values for the truck transport scenarios, (2) 120 times higher than the median values for the rail transport scenarios, and (3) 600 times higher than the median values for the air transport scenarios. For the marine accident scenarios, the 95th percentile values are 10 times higher than the median values. NEWSON DESIGNED NAMES

Tables 8-52 through 8-56 present the error factors used. Where sufficient statistical data exist to establish the 95th percentile values, they are reflected in the smaller error factors assigned to those events which usually range from 3 to 5. Otherwise, the error factors were based on engineering judgment. The guidelines for assigning error factors presented in Section 5.5 were also applied here.

Sequence frequencies for marine transport are presented in Table 8-51. The range factor assigned to final frequency results is 10 for all sequences.

| Event No.       | Name                                                     | Probability            | Range<br>Factor |
|-----------------|----------------------------------------------------------|------------------------|-----------------|
| BE31            | Truck collision/overturn                                 | 1.4 x 10 <sup>-7</sup> | 20              |
| BE68            | Crush force generated                                    | 1                      |                 |
| BE73            | Crush force agent containment<br>(crush >50,000 lb)      | 0.1                    | 2               |
| BE60            | Impact force generated                                   | 1                      |                 |
| BE71            | Impact force fails containment                           | ε                      |                 |
| BE64            | Probe generated                                          | $1.6 \times 10^{-2}$   | 3               |
| BE67            | Probe fails agent containment<br>(V/R >100/s)            | 2.4 x $10^{-2}$        | 2               |
| BE62            | Fire generated                                           | $1.7 \times 10^{-1}$   | 2               |
| BE63            | Fire has heat and duration to detonate burster (>50 min) | 1 x 10 <sup>-6</sup>   | 50              |
| BE31A           | Truck collision/overturn<br>Same as BE31                 |                        |                 |
| BE61(R)         | Impact force sufficient to detonate rocket               | 2 x 10 <sup>-3</sup>   | 5               |
| BE52'           | Mechanical forces destroy ONC                            | $1 \times 10^{-2}$     | 3               |
| BE62A           | Fire occurs given truck<br>collision/overturn            | 7 x 10 <sup>-2</sup>   | 2               |
| BE61            | Undue force sufficient to<br>detonate burster            | 2.2 x $10^{-5}$        | 25              |
| BE75            | Thermal force fails agent<br>containment (>15 min)       | 1 x 10 <sup>-6</sup>   | 50              |
| BE31 (Aircraft) | Aircraft crash occurs at:                                |                        |                 |
|                 | APG                                                      | $1.3 \times 10^{-7}$   | 10              |
|                 | ANAD                                                     | 1.1 x 10 <sup>-9</sup> | 10              |
|                 | LBAD                                                     | $5 \times 10^{-10}$    | 10              |

#### TABLE 8-52 UNCERTAINTY DATA FOR ONSITE TRUCK TRANSPORTATION (IN ONSITE PACKAGE) ACCIDENT SEQUENCES

EXXERT DESCRIPTION DESCRIPTICON DESCRIPTICON DESCRIPTICON DESCRIPACION DESCRIPTICON DESCRI

-

1.1.1



| TABLE 8-5 | 2 (Continued) |
|-----------|---------------|
|-----------|---------------|

| Event No.      | Name                                                                       | Probability            | Range<br>Factor |
|----------------|----------------------------------------------------------------------------|------------------------|-----------------|
|                | NAAP                                                                       | 9.1 x 10 <sup>-9</sup> | 10              |
|                | PBA                                                                        | 2 x 10 <sup>-9</sup>   | 10              |
|                | PUDA                                                                       | 8.4 x 10 <sup>-9</sup> | 10              |
|                | TEAD                                                                       | $2.8 \times 10^{-10}$  | 10              |
|                | UMDA                                                                       | 1.8 x 10 <sup>-9</sup> | 10              |
| BE60           | Impact force only (no fire)                                                | 5.5 x $10^{-1}$        |                 |
| BE71           | Impact force fails containment                                             | 1                      |                 |
| BE60/62        | Impact and fire generated                                                  | 4.5 x $10^{-1}$        |                 |
| BE31 (EQ)      | Earthquake occurs (>0.5 g):                                                |                        |                 |
|                | TEAD                                                                       | $1 \times 10^{-4}$     | 10              |
|                | NAAP                                                                       | $2 \times 10^{-5}$     | 10              |
|                | Elsewhere                                                                  | 6 x 10 <sup>-6</sup>   | 10              |
| BE68           | Crush force generated                                                      | 1                      |                 |
| BE62 (EQ)      | Fire generated (Same as BE62A)                                             |                        |                 |
| BE31 (Tornado) | Tornado occurs (winds<br>>160 mph):                                        |                        |                 |
|                | TEAD, UMDA                                                                 | $3.3 \times 10^{-7}$   | 10              |
|                | PUDA, APG                                                                  | 5.6 x 10 <sup>-6</sup> | 10              |
|                | Elsewhere                                                                  | $1 \times 10^{-4}$     | 10              |
| BE31'          | Trucks caught in bad weather                                               | 0.1                    | 2               |
| BE64           | Tornado-generated missile<br>capable of failing containment<br>(>250 mph): |                        |                 |
|                | TEAD, UMDA                                                                 | 5.4 x $10^{-3}$        | 10              |
|                | PUDA, APG                                                                  | $1.8 \times 10^{-2}$   | 10              |
|                | Elsewhere                                                                  | $1.4 \times 10^{-2}$   | 10              |
| BE51           | Missile fails containment                                                  | 5.3 x 10 <sup>-5</sup> | 50              |
| BE64A          | Probe generated                                                            | $1.6 \times 10^{-2}$   | 3               |

C.

| Event No.                | Name                                                                                | Probability            | Range<br>Factor |
|--------------------------|-------------------------------------------------------------------------------------|------------------------|-----------------|
| BE31                     | Truck collision/overturn                                                            | $1.4 \times 10^{-7}$   | 20              |
| BE68                     | Crush force generated                                                               | $5 \times 10^{-2}$     | 2               |
| BE73                     | Crush force fails containment                                                       | E                      |                 |
| BE60                     | Impact force generated                                                              | $8 \times 10^{-1}$     | 1.4             |
| BE71                     | Impact force fails containment                                                      | ε                      |                 |
| BE64                     | Puncture environment occurs                                                         | $2 \times 10^{-1}$     | 2               |
| BE67                     | Probe fails agent containment<br>(0.75-in. mild steel wall<br>equivalent thickness) | 1 x 10 <sup>-2</sup>   | 2               |
| BE31 for<br>scenario VR4 | Truck accident occurs                                                               | 1.7 x 10 <sup>-7</sup> | 20              |
| BE62                     | Fire generated                                                                      | $1.7 \times 10^{-1}$   | 2               |
| BE63                     | Fire has heat and duration                                                          | E                      |                 |
| BE31A                    | Truck collision/overturn                                                            | $1.4 \times 10^{-7}$   | 20              |
| BE52'                    | Mechanical forces destroy<br>package                                                | $1 \times 10^{-2}$     | 3               |
| BE62A                    | Fire occurs given collision or overturn                                             | $7 \times 10^{-2}$     | 20              |
| BE63A                    | Fire has heat and detonation                                                        | E                      |                 |
| BE61 (R)                 | Impact force sufficient to detonate burster                                         | $2 \times 10^{-3}$     | 5               |
| BE61                     | Undue force detonation occurs                                                       | $2.2 \times 10^{-5}$   | 25              |
| BE75A                    | Thermal force fails agent containment                                               | ε                      |                 |

1. 1. 1. 1. Cath

# TABLE 8-53UNCERTAINTY DATA FOR ONSITE TRUCK TRANSPORTATION<br/>(OFFSITE PACKAGE) ACCIDENT SEQUENCES



1.111.11

### TABLE 8-53 (Continued)

| Event No.       | Name                                | Probability            | Range<br>Factor |
|-----------------|-------------------------------------|------------------------|-----------------|
| BE31 (Aircraft) | Aircraft crash occurs at:           |                        |                 |
|                 | APG                                 | $1.3 \times 10^{-7}$   | 10              |
|                 | ANAD                                | 1.1 x 10 <sup>-9</sup> | 10              |
|                 | LBAD                                | $5 \times 10^{-10}$    | 10              |
|                 | NAAP                                | 9.1 x 10 <sup>-9</sup> | 10              |
|                 | PBA                                 | $2 \times 10^{-9}$     | 10              |
|                 | PUDA                                | 8.4 x 10 <sup>-9</sup> | 10              |
|                 | TEAD                                | $2.8 \times 10^{-10}$  | 10              |
|                 | UMDA                                | 1.8 x 10 <sup>-9</sup> | 10              |
| BE60            | Impact force only (no fire)         | $5.5 \times 10^{-1}$   |                 |
| BE71            | Impact force fails containment      | 1                      |                 |
| BE60/62         | Impact and fire generated           | $4.5 \times 10^{-1}$   |                 |
| BE31 (EQ)       | Earthquake occurs (>0.5 g):         |                        |                 |
|                 | TEAD                                | $1 \times 10^{-4}$     | 10              |
|                 | NAAP                                | $2 \times 10^{-5}$     | 10              |
|                 | Elsewhere                           | 6 x 10 <sup>-6</sup>   | 10              |
| BE68            | Crush force generated               | $5 \times 10^{-2}$     | 2               |
| BE73            | Crush force fails containment       | ε                      |                 |
| BE62 (EQ)       | Fire generated (Same as BE62A)      |                        |                 |
| BE31 (Tornado)  | Tornado occurs (winds<br>>160 mph): |                        |                 |
|                 | TEAD, UMDA                          | $3.3 \times 10^{-7}$   | 10              |
|                 | PUDA, APG                           | 5.6 x 10 <sup>-6</sup> | 10              |
|                 | Elsewhere                           | $1 \times 10^{-4}$     | 10              |
| BE31'           | Trucks caught in bad weather        | 0.1                    | 2               |







|           | TABLE       8-53 (Continued)                                               |                      |                 |
|-----------|----------------------------------------------------------------------------|----------------------|-----------------|
| Event No. | Name                                                                       | Probability          | Range<br>Factor |
| BE64      | Tornado-generated missile<br>capable of failing containment<br>(>310 mph): |                      |                 |
|           | TEAD, UMDA                                                                 | $1.7 \times 10^{-4}$ | 10              |
|           | PUDA, APG                                                                  | $1.1 \times 10^{-3}$ | 10              |
|           | Elsewhere                                                                  | $1.4 \times 10^{-3}$ | 10              |
| BE51      | Missile fails containment                                                  | 3 x 10 <sup>-5</sup> | 50              |
| BE64A     | Probe generated                                                            | $2 \times 10^{-1}$   | 3               |
| BE51A     | Probe fails containment                                                    | $1 \times 10^{-2}$   | 2               |
|           |                                                                            |                      |                 |
|           |                                                                            |                      |                 |

#### TABLE 8-53 (Continued)



Janasa panana panana panana

\*\*\*\*

<u>ن</u>ک

| Event No.       | Name                                                                                   | Probability            | Range<br>Factor |
|-----------------|----------------------------------------------------------------------------------------|------------------------|-----------------|
| BE33            | Train accident occurs                                                                  | 5.5 x 10 <sup>-6</sup> | 20              |
| BE68            | Crush force generated                                                                  | $2 \times 10^{-3}$     | 2               |
| BE53            | Crush force fails containment                                                          | E                      |                 |
| BE60            | Impact force generated                                                                 | 1                      |                 |
| BE52            | Impact force fails package                                                             | $3 \times 10^{-3}$     | 3               |
| BE61            | Impact force fails munition                                                            | ε                      |                 |
| BE64            | Probe generated                                                                        | 5.9 x $10^{-4}$        | 5               |
| BE51            | Probe fails package                                                                    | 1                      |                 |
| BE62            | Fire generated                                                                         | $3.9 \times 10^{-4}$   | 20              |
| BE63            | Fire has heat and duration to detonate burster                                         | $1.6 \times 10^{-1}$   | 2               |
| BE52'           | Mechanical forces destroy<br>package                                                   | $1 \times 10^{-2}$     | 3               |
| BE63A           | Fire has heat and duration to<br>detonate burstered munitions<br>in package (>30 min): |                        |                 |
|                 | C (49 min)                                                                             | 5.5 x $10^{-1}$        | 1.5             |
|                 | P (89 min)                                                                             | $3 \times 10^{-1}$     | 2               |
|                 | M (68 min)                                                                             | $4 \times 10^{-1}$     | 1.5             |
|                 | R (10.5 min)                                                                           | $7.5 \times 10^{-1}$   | 1.4             |
| BE53            | Undue force detonates burster                                                          | 1.6 x 10 <sup>-5</sup> | 25              |
| BE75A           | Fire has heat and duration to fail nonburstered munition                               | $1.6 \times 10^{-1}$   | 2               |
| BE33 (Aircraft) | Aircraft crash occurs                                                                  | $3.1 \times 10^{-11}$  | 20              |
| BE60            | Impact-only force generated                                                            | 5.5 x $10^{-1}$        |                 |
| BE52            | Impact force fails package                                                             | 1                      |                 |

#### TABLE 8-54 UNCERTAINTY DATA FOR OFFSITE RAIL ACCIDENT TRANSPORTATION SEQUENCES

|                | TABLE 8-54 (Continued)              |                        |                 |
|----------------|-------------------------------------|------------------------|-----------------|
| Event No.      | Name                                | Probability            | Range<br>Factor |
| BE61           | Impact force fails munition         | 1                      |                 |
| BE60/62        | Impact and fire generated           | $4.5 \times 10^{-1}$   |                 |
| BE33 (EQ)      | Earthquake occurs (0.35 g):         |                        |                 |
|                | UMDA to TEAD                        | 8.2 x $10^{-4}$        | 25              |
|                | PUDA to TEAD                        | $2.5 \times 10^{-4}$   | 25              |
|                | LBAD to TEAD                        | 5.1 x $10^{-4}$        | 25              |
|                | APG to TEAD                         | $1.9 \times 10^{-4}$   | 25              |
|                | ANAD to TEAD                        | 2.4 x $10^{-4}$        | 25              |
|                | NAAP to TEAD                        | $2.2 \times 10^{-4}$   | 25              |
|                | PBA to TEAD                         | $2.7 \times 10^{-4}$   | 25              |
|                | LBAD to ANAD                        | $2.8 \times 10^{-4}$   | 25              |
|                | PBA to ANAD                         | 5.3 x $10^{-4}$        | 25              |
|                | APG to ANAD                         | $2 \times 10^{-4}$     | 25              |
|                | NAAP to ANAD                        | $3 \times 10^{-4}$     | 25              |
| BE68           | Crush force generated               | $2 \times 10^{-3}$     | 2               |
| BE33 (Tornado) | Tornado occurs (winds<br>>160 mph): |                        |                 |
|                | UMDA to TEAD                        | $3.3 \times 10^{-7}$   | 25              |
|                | PUDA to TEAD                        | $3.7 \times 10^{-7}$   | 25              |
|                | LBAD to TEAD                        | 7.3 x $10^{-5}$        | 25              |
|                | APG to TEAD                         | 6.8 x 10 <sup>-5</sup> | 25              |
|                | ANAD to TEAD                        | 7.6 x 10 <sup>-5</sup> | 25              |
|                | NAAP to TEAD                        | 6.6 x 10 <sup>-5</sup> | 25              |
|                | PBA to TEAD                         | 7.3 x 10 <sup>-5</sup> | 25              |
|                | LBAD to ANAD                        | $1 \times 10^{-4}$     | 25              |
|                | PBA to ANAD                         | $1 \times 10^{-4}$     | 25              |
|                | APG to ANAD                         | 9.3 x $10^{-5}$        | 25              |
|                | NAAP to ANAD                        | $1 \times 10^{-4}$     | 25              |
|                | 8-136                               |                        |                 |

PROPERTY AND PROPERTY AND

<. IQ **)** 757

<u>a</u>b.

100

#### TABLE 8-54 (Continued)

| Event No. | Name                                                                   | Probability            | Range<br>Factor |
|-----------|------------------------------------------------------------------------|------------------------|-----------------|
| BE64      | Missile capable of puncturing<br>and failing containment<br>generated: |                        |                 |
|           | UMDA to TEAD                                                           | $1.7 \times 10^{-4}$   | 25              |
|           | PUDA to TEAD                                                           | $3 \times 10^{-3}$     | 25              |
|           | LBAD to TEAD                                                           | $1.4 \times 10^{-3}$   | 2:,             |
|           | APG to TEAD                                                            | $1.4 \times 10^{-3}$   | 25              |
|           | ANAD to TEAD                                                           | $1.4 \times 10^{-3}$   | 25              |
|           | NAAP to TEAD                                                           | $1.4 \times 10^{-3}$   | 25              |
|           | PBA to TEAD                                                            | $1.4 \times 10^{-3}$   | 25              |
|           | LBAD to ANAD                                                           | $1.4 \times 10^{-3}$   | 25              |
|           | PBA to ANAD                                                            | $1.4 \times 10^{-3}$   | 25              |
|           | APG to ANAD                                                            | $1.4 \times 10^{-3}$   | 25              |
|           | NAAP to ANAD                                                           | $1.4 \times 10^{-3}$   | 25              |
| BE51      | Missile has orientation to fail containment                            | 5.3 x 10 <sup>-5</sup> | 50              |
| BE68      | Crush force generated                                                  | 2 x 10-3               | 2               |

1.1

8-137

in WK

| Event No. | Name                                | Probability              | Range<br>Factor |
|-----------|-------------------------------------|--------------------------|-----------------|
| A1        | Accident rate/flight                | See Section<br>8.3.3.1.5 | 10              |
| A2        | Impact only force generated         | 2.7 x $10^{-1}$          | 2               |
| A3        | Impact force fails package:         |                          |                 |
|           | Takeoff                             | $1.5 \times 10^{-1}$     | 1.5             |
|           | Inflight                            | $7 \times 10^{-1}$       | 1.5             |
|           | Landing                             | $1.3 \times 10^{-1}$     | 1.5             |
| A4        | Impact and fire occur               | $2.2 \times 10^{-1}$     | 2               |
| A5        | Fire is only force present          | $1.3 \times 10^{-1}$     | 2               |
| A6        | Fire fails package (C-141)          | ε                        |                 |
| A7        | Fire fails package (C-5)            | $9 \times 10^{-2}$       | 2               |
| A7'       | Undue force detonation              | $2.1 \times 10^{-3}$     | 25              |
| A8        | Impact causes rocket motor ignition | 4.4 x $10^{-2}$          | 5               |
| A9        | Impact greater than 40 ft:          |                          |                 |
|           | Takeoff                             | $3.1 \times 10^{-1}$     | 1.5             |
|           | Inflight                            | $3 \times 10^{-1}$       | 1.5             |
|           | Landing                             | $3.3 \times 10^{-1}$     | 1.5             |
|           |                                     |                          |                 |

TABLE 8-55 UNCERTAINTY DATA FOR OFFSITE TRANSPORTATION BY AIR ACCIDENT SEQUENCES PROSERVER RECEIPTION DESCRIPTION DE L'ANDRE D

12222

1.0 1.0 1.0 1.0

ardis arrestant harranta barara accident arresta arresta arrestan arrestan arrestan arrestan arrestan arrestan M

ville frances and the frances of the The frances of the f

|          | TABLE 8-56<br>UNCERTAINTY DATA FOR ONSITE TRUCK TRANSPORTATION<br>ACCIDENT SEQUENCES (MARINE OPTION) |                                               |                        |                |  |  |
|----------|------------------------------------------------------------------------------------------------------|-----------------------------------------------|------------------------|----------------|--|--|
|          | Event No.                                                                                            | Name                                          | Probability            | Range<br>Facto |  |  |
|          | BE31                                                                                                 | Truck collision/overturn                      | 1.4 x 10 <sup>-7</sup> | 20             |  |  |
|          | BE68                                                                                                 | Crush force generated                         | 1                      |                |  |  |
|          | BE73                                                                                                 | Crush force fails agent containment           | ε                      |                |  |  |
|          | BE60                                                                                                 | Impact force generated                        | 1                      |                |  |  |
|          | BE71                                                                                                 | Impact force fails containment                | ε                      |                |  |  |
|          | BE64                                                                                                 | Probe generated                               | $1.6 \times 10^{-2}$   | 3              |  |  |
|          | BE67                                                                                                 | Probe fails agent containment<br>(V/R >100/s) | $1.2 \times 10^{-2}$   | 2              |  |  |
| 2. A     | BE62                                                                                                 | Fire generated                                | $1.7 \times 10^{-1}$   | 2              |  |  |
| <b>.</b> | BE31A                                                                                                | Truck collision/overturn<br>Same as BE31      |                        |                |  |  |
|          | BE52'                                                                                                | Mechanical forces destroy vault               | $1 \times 10^{-2}$     | 3              |  |  |
|          | BE62A                                                                                                | Fire occurs given truck collision/overturn    | 7 x 10 <sup>-2</sup>   | 2              |  |  |
|          | BE75                                                                                                 | Thermal force fails agent containment         | E                      | ÷-             |  |  |
|          | BE31 (Aircraft)                                                                                      | Aircraft crash occurs at:<br>APG              | 1.3 x 10 <sup>-7</sup> | 10             |  |  |
|          | BE60                                                                                                 | Impact force only (no fire)                   | 5.5 x $10^{-1}$        |                |  |  |
|          | BE71                                                                                                 | Impact force fails containment                | 1                      |                |  |  |
|          | BE60/62                                                                                              | Impact and fire generated                     | $4.5 \times 10^{-1}$   |                |  |  |
|          | BE31 (EQ)                                                                                            | Earthquake occurs (>0.5 g):<br>APG            | 6 x 10 <sup>-6</sup>   | 10             |  |  |
|          | BE68                                                                                                 | Crush force generated                         | 1                      |                |  |  |

#### TABLE 8-56 UNCERTAINTY DATA FOR ONSITE TRUCK TRANSPORTATION ACCIDENT SEQUENCES (MARINE OPTION)

B

M

1.42.11.11.11.11.11.11

### TABLE 8-56 (Continued)

| Event No.      | Name                                                                       | Probability            | Range<br>Factor |
|----------------|----------------------------------------------------------------------------|------------------------|-----------------|
| BE62 (EQ)      | Fire generated (Same as BE62A)                                             |                        |                 |
| BE31 (Tornado) | Tornado occurs (winds<br>>160 mph):                                        |                        |                 |
|                | APG                                                                        | 5.6 x 10 <sup>-6</sup> | 10              |
| BE31'          | Trucks caught in bad weather                                               | 0.1                    | 2               |
| BE64           | Tornado-generated missile<br>capable of failing containment<br>(>250 mph): |                        |                 |
|                | APG                                                                        | $1.1 \times 10^{-3}$   | 10              |
| BE51           | Missile fails containment                                                  | 1.7 x 10 <sup>-5</sup> | 50              |



<u>የພາພາມ የພາມ የአሸ ພາກ "አግሎሽ አሸ አሸ አሸ አ</u>ሸ አ

#### 8.6. REFERENCES

- 8-1. Clarke, R. K., et al., "Severities of Transportation Accidents,"
   SCA-74-0001, Sandia National Laboratories, July 1976.
- 8-2. Dennis, A. W., et al., "Severities of Transportation Accidents Involving Large Packages," SAND77-0001, Sandia National Laboratories, May 1978.
- 8-3. Fischer, L. E., et al.. "Shipping Container Response to Severe Highway and Railway Accident Conditions," NUREG/CR-4829, Vols. 1 and 2, Lawrence Livermore National Laboratory, February 1987.
- 8-4. Letter, W. R. Rhyne to R. Bolig, dated April 9, 1987, "Comparison of Recent Truck/Rail Accident Data with the Data Used in the Transport Risk Analysis."
- 8-5. Rhyne, W. R., et al., "Probabilistic Analysis of Chemical Agent Release During Transport of M55 Rockets," H&R Technical Associates, Inc., M55-CD-4, H&R 255-1, September 1985.
- 8-6. "Supporting Calculations for the Risk Analysis of the Disposal of Chemical Munitions at National or Regional Sites," GA Technologies Inc., EPS-GA/DP-0092 (Issue 2), August 1987.
- 8-7. Finnegan, R. L., et al., "Preliminary Impact Speed and Angle Criteria for Design of a Nuclear Airplane Fission Product Containment Vessel," TMX-2245, National Aeronautics and Space Administration, May 1971.
- 8-8. "Transportation of Chemical Agents and Munitions: A Concept Plan," U.S. Army Office of the Program Manager for Chemical Munitions, The MITRE Corporation, Report MTR-87W00075, June 30, 1987.
- 8-9. Fraize, W., et al., "Preliminary Risk Analysis Supporting the Chemical Weapons Stockpile Disposal Program (CSDP)," The MITRE Corporation, Volume 1: Analysis, MTR-86W4-01 and Volume 2: Appendices, MTR-86W4-02, May 1987.
- 8-10. "Chemical Stockpile Disposal Concept Plan," USATHAMA, AMXTH-CD-FR-95047, March 1986.

- 8-11. PMCM meeting at Edgewood Area, October 1986.
- 8-12. Marone, V. P., Chief, Reports Branch, Directorate of Aerospace Safety, Norton Air Force Base, Letter to R. J. Robinette, H&R Technical Associates, Inc., May 18, 1987.
- 8-13. Hartman, W. F., Sandia National Laboratories, Albuquerque, New Mexico, Personal communication with W. R. Rhyne, July 1985.
- 8-14. Federal Railroad Administration, "Accident/Incident Bulletin No. 151, Calendar Year 1982," Department of Transportation, June 1983.
- 8-15. Letter from E. Hillsman, ORNL to W. R. Rhyne, H&R Technical Associates, Inc., April 15, 1987.
- 8-16. "Transportation of Chemical Agent and Munitions: A Concept Plan," U.S. Army Program Manager for Chemical Munitions, Aberdeen Proving Ground, Maryland, May 1, 1987.
- 8-17. Letter from W. R. Rhyne, H&R Technical Associates, Inc. to Dr. R. Bolig, GA Technologies, Inc., May 5, 1987.
- 8-18. Read, A., et al., "Analysis of Existing Hazardous Material Containers for Transporting Chemical Munitions," The MITRE Corporation, June 1987.
- 8-19. Letter from W. R. Rhyne to H&R Staff, "Administrative Controls -Rail and Air," File No. 307, April 27, 1987.
- 8-20. "115 mm Rocket Sensitivity Tests," AEO Project T-222 Chemical Demil Project No. 200, Ammunition Equipment Office, Tooele Army Depot, Tooele, Utah, April 1974.
- 8-21. "Probabilistic Analysis of Chemical Agent Release During Transport of Obsolete Munitions With and Without Additional Protective Packaging," H&R Technical Associates, Inc., H&R 244-4, June 1986.
- 8-22. Captain Rich, Flight Safety Branch, U.S. Air Force Inspection and Test Center, Norton AFB, California, personal communication with R. J. Robinette, May 1987.
- 8-23. "Thermal Response of M55 Rockets to Pool Fires During Transit," Southwest Research Institute for H&R Technical Associates, Inc., SWRI 8461/005, April 1985.

8-24. "FAA Statistical Handbook of Aviation, Calendar Year 1984," U.S. Department of Transportation, Federal Aviation Administration.

SAN TANK MANANY "KAGAGA TANG

- 8-25. "Risk Analysis for Barge Transportation of Ton Containers from Aberdeen Proving Ground," Engineering Computer Optecnomics, Incorporated, May 13, 1987.
- 8-26. "Risk Assessment Study for the Cove Point, Maryland LNG Facility," SAI-78-626-LJ, Science Applications, Incorporated, March 23, 1978.

#### 9. QUANTIFICATION BASES

#### 9.1. DATA BASE

#### 9.1.1. Train Accident Data

The train accident rate selected for the transportation risk analysis is from the Sandia National Laboratory (SNL) data base, based on 1972 civilian freight train data. Because the transportation of chemical munitions is a military action under strict supervision and control, a study was made to determine what, if any, reduction could be taken in the civilian train accident rate. Because of a lack of hard data, scientific judgment was used to estimate a reasonable reduction factor due to special administrative controls described in the Transportation Concept Plan (Ref. 9-1).

These administrative controls include the following actions. A pilot train will precede the munition train to ensure track integrity and to provide timely emergency response. Empty railcars will be thoroughly inspected prior to loading, and all engines and equipment will be inspected and tested before departure of the munition train. Routine enroute inspections will occur at least every 1,000 miles. A walking guard on each side of each munition car will be provided during all stops.

Some other controls are expected to be implemented also, but no credit was taken for them in reducing the train accident rate. The train speed limit will be 10 mph less than the posted limit for a given track, but never more than 50 mph. The highest quality track available (FRA Class 3 or better) will be used, consistent with the policy of avoiding highly populated areas. A standard train crew, with special training in chemical munitions, will operate the train. A railroad officer will also be present on the munition train, to act as crew leader and to serve as liaison with the Army.

Sec. 11

The train accident rate given by SNL is  $1 \times 10^{-5}$  accidents/mile (Ref. 9-2). Train accidents used to determine this rate are described by "type" of accident only: collision, derailment, and "other" (including grade-crossing collisions), and not by cause. However, data concerning civilian freight train accidents in 1982 is available from the Federal Railroad Administration (FRA) June 1983 report (Ref. 9-3), where results are reported by category (track, roadbed, and structure, mechanical and electrical, human factors, and miscellaneous factors) and by cause within each category for each accident type.

Table 9-1 shows the numbers of accidents by accident type, and the annual totals for the SNL data and the FRA data.

Because the total numbers of derailments, collisions, and "other" type accidents in 1982 do not differ greatly from those in 1972, it can be assumed that the distribution by category and cause is also similar. Therefore, the administrative controls can be applied to the 1982 accident causes data to estimate a total accident rate reduction factor for the SNL train accident rate.

The FRA data categories and the total number of accidents in each category before and after modification due to administrative controls are shown in Table 9-2.

For accidents in the track, roadbed, and structure defect category, there was not enough information to reduce the number of collisions or "other" type accidents, but it is expected that use of a pilot train will substantially reduce the number of munition train derailments. Therefore, a 95% reduction was taken for each derailment accident cause,





|      |         | Collisions | Derailments | Other | Total<br>Accidents | Total<br>Miles         |
|------|---------|------------|-------------|-------|--------------------|------------------------|
| SNL, | 1972(a) | 308        | 3880        | 465   | 4653               | $4.51 \times 10^8$     |
| FRA, | 1982(b) | 572        | 3383        | 456   | 4589               | 5.73 x 10 <sup>8</sup> |

#### TABLE 9-1 COMPARISON OF 1972 AND 1982 ACCIDENT DATA

(a) Taken from Ref. 9-2.
(b) Taken from Ref. 9-3.





......

1.1

| Category                              | Original Number | Modified Number |  |
|---------------------------------------|-----------------|-----------------|--|
| Track, roadbed and structure defects  | 1769            | 135             |  |
| Mechanical and<br>electrical failures | 796             | 597             |  |
| Human error                           | 1284            | 1284            |  |
| Míscellaneous                         | _740            | _517            |  |
| Total                                 | 4589            | 2534            |  |

## TABLE 9-21982 TRAIN ACCIDENT DATA BY CATEGORY

1.5.5
Accidents caused by mechanical and electrical failures could occur in the locomotive, the railcar, or in general. No information was available to reduce these accidents by cause. However, the predeparture inspection and routine inspections at 1,000-mile intervals allow a modest reduction, estimated at 25%, in the total number of accidents due to mechanical and electrical failures. Therefore, the original number of 796 accidents becomes 597 accidents.

No credit is taken for any reduction in accidents caused by human error, because a standard train crew will operate the train. The total number of accidents in this category remains 1284.

For accidents caused by miscellaneous factors, the following reductions were taken:

| Cause                                | Reduction %; Number | Rationale                                                          |
|--------------------------------------|---------------------|--------------------------------------------------------------------|
| Vandalism                            | 100%; 78 to 0       | Surveillance, walking<br>guard                                     |
| Interference with railroad operation | 100%; 16 to 0       | Surveillance, walking<br>guard                                     |
| Overloaded cars                      | 100%; 5 to 0        | Special loading proce-<br>dures; only two packages<br>per railcar. |
| Object on or<br>fouling track        | 100%; 30 to 0       | Pilot train                                                        |
| Equipment on or fouling track        | 100%; 11 to 0       | Pilot train                                                        |
| Snow, ice, or mud<br>on track        | 50%; 61 to 31       | Pilot train; use of best<br>track                                  |

KARANA REEKT

| Fire, not due to<br>vandalism | 100%; 24 to 0 | Already accounted for in fire probability |
|-------------------------------|---------------|-------------------------------------------|
| Hump retarder fails           | 100%; 21 to 0 | Not applicable                            |

The number of accidents in this category is reduced from 740 to 517.

The total accident reduction is from 4589 to 2534, a factor of 2534/4539 = 0.55. Therefore, this factor (0.55) is applied to the original SNL train accident rate to determine the modified train accident rate.

 $(1 \times 10^{-5})(0.55) = 5.5 \times 10^{-6}$ 

Thus,  $5.5 \times 10^{-6}$  accidents per mile is the rate chosen for use in this risk analysis. For reference purposes, the original 1982 train accident data from Ref. 9-3 are shown on Tables 9-3 through 9-6.

# 9.1.2. Onsite Truck Accident Data

The truck convoy accident data summarized here was developed by SNL (Ref. 9-2). These data represent the most comprehensive information currently available and they are commonly used for truck transportation risk analyses. Therefore, an explanation of their bases will not be presented here. A 1987 report by the Lawrence Livermore National Laboratory for the Nuclear Regulatory Commission (Ref. 9-28), describing highway accidents involving spent fuel shipping casks, was reviewed; the more recent data was found to be consistent with the SNL data (Ref. 9-2). Therefore, no changes will be made in the data used for this analysis. The SNL analyses considered five accident forces: impact, crush, puncture, fire, and immersion. Only the first four are discussed here because immersion is not considered a threat for onsite transportation.

| Cause of Accident                             | Total<br>Accidents | Collision | Derailments | Other | Rail<br>Highway<br>Crossing |
|-----------------------------------------------|--------------------|-----------|-------------|-------|-----------------------------|
| Roadbed defects                               | 109                | <u> </u>  | 108         | 1     |                             |
| Track geometry<br>defects                     | 751                | 3         | 746         | 2     |                             |
| Rail and joint<br>bar defects                 | 459                | 5         | 451         | 3     |                             |
| Frogs, switches,<br>and track appli-<br>ances | 428                | 11        | 408         | 9     |                             |
| Other way and structure                       | 17                 |           | 7           | 10    |                             |
| Signal and<br>communication<br>failures       | 5                  |           | 3           | 2     |                             |
| Subtotal                                      | 1769               | 19        | 1723        | 27    |                             |

TABLE 9.3(a) TRAIN ACCIDENTS CAUSED BY TRACK, ROADBED, AND STRUCTURE DEFECTS

(a)<sub>Taken</sub> from Ref. 9-3.

| Cause of Accident                                | Total<br>Accidents | Collision | Derailments | Other | Rail<br>Highway<br>Crossing |
|--------------------------------------------------|--------------------|-----------|-------------|-------|-----------------------------|
| Locomotive failure                               |                    |           |             |       |                             |
| Brakes                                           | 1                  |           | 1           |       |                             |
| Body                                             |                    |           |             |       |                             |
| Coupler and draft system                         |                    |           |             |       |                             |
| Truck components                                 | 6                  |           | 6           |       |                             |
| Axles and journal<br>bearings                    | 6                  |           | 6           |       |                             |
| Wheels                                           | 8                  |           | 8           |       |                             |
| Locomotives                                      | 55                 | 1         | 11          | 42    | 1                           |
| Doors                                            |                    |           |             |       |                             |
| General mechanical<br>and electrical<br>failures | 4                  | 1         | 1           | 2     |                             |
| <u>Car failure</u>                               |                    |           |             |       |                             |
| Brakes                                           | 113                | 12        | 85          | 16    |                             |
| Trailer or con-<br>tainer or flatcar             | 9                  | 1         | 5           | 3     |                             |
| Body                                             | 73                 | 2         | 65          | 6     |                             |
| Coupler and draft<br>system                      | 79                 | 13        | 58          | 8     |                             |
| Truck components                                 | 160                |           | 158         | 2     |                             |
| Axles and journal<br>bearings                    | 125                |           | 122         | 3     |                             |
| Wheels                                           | 136                | 1         | 132         | 3     |                             |
| Doors                                            | 7                  | 1         | 4           | 2     |                             |
| General mechanical<br>and electrical<br>failures | 14                 | 1         | 9           | 4     |                             |
| Subtotal                                         | 796                | 33        | 671         | 91    | 1                           |

A 10.0 SECOND DODDON DODDON DODDON DECOMPANY SECOND DODDON DODDON DODDON DODDON DODDON DODDON DODDON

# TABLE 9-4(a) TRAIN ACCIDENTS CAUSED BY MECHANICAL AND ELECTRICAL FAILURES

(a) Taken from Ref. 9-3.

CONTRACT DIVERSE SECOND ANTIMA ANTIMACIA ANTI

| Cause of Accident                              | Total<br>Accidents | Collision | Derailments | Other | Rail<br>Highway<br>Crossing |
|------------------------------------------------|--------------------|-----------|-------------|-------|-----------------------------|
| Brakes, use of                                 | 223                | 110       | 68          | 45    |                             |
| Employee, physical conditions                  | 4                  | 4         |             |       |                             |
| Flagging, fixed,<br>hand, and radio<br>signals | 47                 | 23        | 17          | 7     |                             |
| Other rules and<br>instructions                | 349                | 220       | 63          | 66    |                             |
| Speed                                          | 121                | 34        | 59          | 28    |                             |
| Switches, use of                               | 184                | 46        | 123         | 15    |                             |
| Miscellaneous                                  | _356               | _40       | <u>287</u>  | _29   |                             |
| Subtotal                                       | 1284               | 447       | 617         | 190   |                             |

# TABLE 9-5(a) TRAIN ACCIDENTS CAUSED BY HUMAN FACTORS

(a)<sub>Taken</sub> from Ref. 9-3.

Ð

Sec. 1

**NITOTOTOT** 

| Cause of Accident                                   | Total<br>Accidents | Collision | Derailments | Other | Rail<br>Highway<br>Crossing |
|-----------------------------------------------------|--------------------|-----------|-------------|-------|-----------------------------|
| Collision with high-<br>way user at crossing site   | 174                |           |             |       | 174                         |
| Emergency brake<br>application to avoid<br>accident | 0                  |           | 7           |       | 1                           |
| Vandalism                                           | 78                 | 7         | 38          | 32    | 1                           |
| Interference with railroad operation, not vandals   | 16                 | 3         | 6           | 7     |                             |
| Load shifted                                        | 30                 | 1         | 26          | 3     |                             |
| Load fell from car                                  | 6                  |           | 2           | 4     |                             |
| Overloaded car                                      | 5                  |           | 5           |       |                             |
| Improperly loaded car                               | 17                 | 1         | 16          |       |                             |
| Oversized load,<br>misrouted                        | 4                  | 1         |             | 3     |                             |
| Object on or fouling track                          | 38                 |           | 14          | 23    | 1                           |
| Equipment on or fouling track                       | 11                 | 4         |             | 7     |                             |
| Cargo tiedown<br>improperly applied                 | 4                  |           | 1           | 3     |                             |
| Overload/improper<br>load container/etc.            | 2                  |           | 2           |       |                             |
| Interaction of<br>lateral/vertical<br>forces        | 129                | 1         | 128         |       |                             |
| Failure to control<br>car speed with hand<br>brakes | 1                  |           | 1           |       |                             |
| Snow, ice, or mud on track                          | 61                 | 7         | 50          | 4     |                             |
| Fire, not due to<br>vandalism                       | 24                 |           |             | 24    |                             |
| Hump retarder failed to slow car                    | 21                 | 8         | 3           | 10    |                             |
| Switch fouled by loading chains, etc.               | 7                  |           | 5           | 2     |                             |
| Other causes                                        | <u>97</u>          | _9        | 65          | 23    |                             |
| Subtotal                                            | 740                | 43        | 372         | 148   | 177                         |

# TABLE 9-6(\*) TRAIN ACCIDENTS CAUSED BY MISCELLANEOUS FACTORS

1111 • •

(a)<sub>Taken</sub> from Ref. 9-3.

The effect of human factors on the truck accident rate is implicit in the SNL data base. If an accident occurred due to human error, it shows up in the data base just as an accident. Therefore, it is not possible to ascertain the human error contribution or to define the human error probabilities involved. No specific human reliability analysis was done for onsite transportation. Several administrative controls will be instituted, however, and these have the effect of reducing the SNL truck accident rate as shown on Table 9-7 and discussed in Section 8.1.

#### 9.1.3. Plant Accident Data

Component failure data that support all of the fault trees and event trees are presented on the following pages; references are also provided. The data used to quantify the fault tree events are also presented on the fault trees. Beta factors are used to quantify failure probabilities for identical redundant components. The beta factors are also shown on the fault trees.

The derivation of the failure rates used in this study was based on extensive review and analysis of data available in the literature. When a sufficient number of estimates (at least 10, but usually many more) was available for a component failure rate, the method described in "Reliability Engineering"\* was used to develop a nonparametric distribution of estimates. The 0.5 percentile of this distribution was used as the median of a lognormal distribution of parameter estimates. The 0.95, 0.50, and 0.05 percentiles of the nonparametric distribution were used to develop an error factor for the lognormal distribution.

When less than 10 estimates were available for a particular component, a most applicable estimate was subjectively selected to represent

\*ARINC Research Corporation, "Reliability Engineering," Prentice-Hall, Inc., 1964, p. 144.

| Munitions Vehicle<br>Accident Type       | Highway Accident<br>Rate (Per Mile) | Convoy Accident<br>Rate (Per Mile) |
|------------------------------------------|-------------------------------------|------------------------------------|
| Head-on collision                        | 4.7 x 10 <sup>-7</sup>              | 0                                  |
| Rear-end collision                       | $3.8 \times 10^{-7}$                | $3.8 \times 10^{-8}$               |
| Rear-end collision                       | 4.0 x $10^{-7}$                     | $4.0 \times 10^{-8}$               |
| Side-on into collision                   | $1.5 \times 10^{-7}$                | 0                                  |
| Side-on by other collision               | $2.3 \times 10^{-7}$                | 0                                  |
| Truck/train collision                    | $1.6 \times 10^{-8}$                | 0                                  |
| Fixed object collision                   | $4.3 \times 10^{-7}$                | $4.3 \times 10^{-8}$               |
| Overturn only                            | $1.7 \times 10^{-7}$                | $1.7 \times 10^{-8}$               |
| Subtotal (collision/<br>overturn events) | $2.47 \times 10^{-6}$               | $1.38 \times 10^{-7}$              |
| Fire only                                | $2.8 \times 10^{-8}$                | 2.8 x $10^{-8}$                    |
| Total                                    | $2.5 \times 10^{-6}$                | $1.66 \times 10^{-7}$              |

TABLE 9-7TRUCK ACCIDENT RATE(a)

(a)Probability (collision or overturn/truck accident) =

 $\frac{1.38 \times 10^{-7}}{1.66 \times 10^{-7}} = 0.831$ 

Probability (fire only/truck accident) =  $\frac{2.8 \times 10^{-8}}{1.66 \times 10^{-7}} = 0.169$ 

the median of a lognormal distribution. The error factor was also selected subjectively, but it was verified that the corresponding lognormal distribution was consistent with the other available estimates.

#### Fan Fails Off - 0.13/yr (EF = 30)

The Corps of Engineers (HND) R/M Data Base (Ref. 9-4) provides failure rates for fans (all failure modes combined) ranging from 0.9 to 9.17 per million hours. NPRD-3 (Ref. 9-5, pages 201-202) provides a range from 2 to 25 failures per million hours for fans operating under selected environmental conditions (data from GF and NS environmental codes only). Review of the failure mode descriptions in the NPRD-3 report, however, reveals that no more than about 51% of all failure events are relevant to the failure mode of interest here. Thus, the failure rate estimates from these two sources range from about 0.5 to 13 failures per million hours.

NPRDS (Ref. 9-6, pages 287-289) reports a total of 48 fan/blower failures in about 4.06 million operating hours. The failure modes described in the NPRDS report were examined to screen those that do not apply to the event of interest. This review indicates that only 23 events can be associated with the failure mode of interest. Thus, the failure rate is about  $5.7 \times 10^{-6}$ /h.

SRS (Ref. 9-7, item code 663°) provides four fan failure rate estimates ranging from 261 to 867 failures per million operating hours. These estimates were reduced by 50% to screen failure modes that do not apply to this event (the 50% reduction is based on both the NPRD-3 and NPRDS failure mode reviews).

All these sources combined provided a total of 21 failure rate estimates. These estimates were used to develop a distribution of fan failure rates, and this distribution was used to develop conservative parameters of a lognormal distribution to be used in this study.

The median and error factor developed from this distribution are 1.5 x  $10^{-5}$ /h and 30, respectively.

# <u>Motor Fails to Run</u> - 0.061/yr (EF = 20)

The Rijnmond study (Ref. 9-8, Table IX.I) suggests a failure rate range from 0.5 to 100 (median = 7, EF = 14) failures per million hours for a motor failing to run.

NPRDS (Ref. 9-6, pages 403-409) reports a total of 48 ac motor failures in about 5.2 million operating hours. The failure modes described in the NPRDS report were examined to screen those that do not apply to the event of interest, and only about 15 failures were judged applicable here. Thus, the NPRDS estimate is 2.9 x  $10^{-6}$ /operating hour.

NPRD-3 (Ref. 9-5, pages 199-201) provides a range from 0.5 to 250 failures per million operating hours under selected environmental conditions (data from DOR, GB, GF, and NS environmental codes only). Review of the failure mode descriptions in the NPRD-3 report, however, reveals that no more than about 77% of all failure events are relevant to the failure mode of interest. Thus, the failure rate range from this source is from 0.4 to 193 failures per million operating hours.

WASH-1400 (Ref. 9-9, Table III.4-2) suggests a median of 10 x  $10^{-6}$ /h with an error factor of 3. SRS (Ref. 9-7, item code 56320) provides ten failure rate estimates for electric motors ranging from 2.9 to 158 failures per million operating hours. These estimates were reduced by 54% to screen failure modes that do not apply to this event (the 54% reduction is an average of the reduction suggested by the NPRD-3 and NPRDS failure mode reviews).

All these sources combined provided a total of 29 failure rate estimates. These estimates were used to develop a distribution of motor

failure rates, and this distribution was used to develop conservative parameters of a lognormal distribution to be used in this study.

The median and error factor developed from this distribution are 7 x  $10^{-6}$ /h and 20, respectively.

#### <u>Pump Fails to Run</u> - 0.26/yr (EF = 10)

WASH-1400 reports a failure rate of 3 x  $10^{-5}$ /h with an error factor of 10. This estimate includes both the pump and the driver. SRS (Ref. 9-7, item code 69530) reports a slightly higher rate (5.4 x  $10^{-5}$ /h) that is in good agreement given the large uncertainty assumed in the WASH-1400 estimate.

NPRDS (Ref. 9-6, pages 421 through 429) reports a total of 509 (<500 GPM) pump failures in about 2.3 million operational hours. The failure modes described in the NPRDS report were examined to screen those that do not apply to the event of interest (e.g., spurious operation). This review indicates that only about 147 of the 509 events can be associated with the failure mode of interest. Thus, the failure rate is about 6.4 x 10<sup>-5</sup>/operational hour.

NPRD-3 (Ref. 9-5, page 215) reports a failure rate of 7.9 x  $10^{-5}$ / operating hour for oil pumps operating under less than ideal conditions, installed in permanent racks with adequate cooling air, and maintained by military personnel. However, the pump may occasionally be subject to shock and vibration. As for the NPRDS estimate, the failure modes described in the NPRD-3 report were reviewed, and only about 27% of the events were judged applicable to the failure mode of interest. Thus, the failure rate becomes about 2.1 x  $10^{-5}$ /operating hour.

All estimates are in good agreement with the WASH-1400 estimate, and the latter was used in this study. The error factor proposed in the WASH-1400 is also adopted here because both the NPRDS and the NPRD-3 data bases show large variations among failure rate estimates for different pumps and/or for similar pumps at different facilities.

# <u>Heater Fails Off</u> - 0.021/yr (EF = 10)

The Corps of Engineers (HND) R/M Data Base (Ref. 9-10, page 296) provides a 2.36/million hour failure rate estimate for a large (30 kw, 400 VAC), two stage heater. NPRD-3 (Ref. 9-5, page 207) provides a range from 0.4 to 3.5 failures per million operating hours for heaters operating under selected environmental conditions (data from GB environmental code only). NPRDS (Ref. 9-6, page 322) reports 18 heater failures in about 5 million operating hours. Thus, the NPRDS failure rate estimate is about 3.6 x  $10^{-6}/h$ .

The estimate from the Corps of Engineers (HND) data base (Ref. 9-4) is judged more applicable here and will be used as the median for "heater fails off" event. An error factor of 10 is assumed due to the large uncertainties associated with the applicability of these estimates to the equipment of interest. Note that all estimates are in good agreement given the large uncertainty assumed for this failure rate.

#### Loss of (Plant or Instrument) Air System - 0.016/yr (EF = 10)

NPRDS (Ref. 9-6, page 49) reports no air system failures in about 398 thousand operating hours (approximately 3.2 million calendar hours). These statistics were compiled from 24 instrument and station service air systems in U.S. nuclear power plants. The median generated from these statistics is about 1.8 x  $10^{-6}$ /operating hour (using a chi-square distribution).

There are large uncertainties regarding the similarity of the systems at this facility and the systems in the NPRDS data base, and thus, regarding the applicability of the NPRDS estimate to this facility. An error factor of 10 is judged adequate here.

# Switch, Generic--Spurious Operation - 0.015/yr (EF = 19)

A review of available data bases (Refs. 9-5 through 9-8 and Refs. 9-11 through 9-13) revealed 53 failure rate estimates for a variety of switches (e.g., pressure, temperature, etc.). These estimates were used to develop a distribution of switch failure rates, and this distribution was used to develop conservative parameters of a lognormal distribution to be used in this study.

The median of switch failure rate estimates is about 3.4 failures per million operating hours. This rate was arbitrarily reduced by 50% to represent the fraction corresponding to the failure mode of interest, i.e., "spurious operation." This reduction is believed to be conservative. The distribution of switch failure rates suggests an error factor of 19.

# <u>Controller (includes sensor, signal conditioning equipment, and control</u> <u>circuitry), Generic--Spurious Operation (high or low)</u> - 0.022/yr (EF = 12)

A review of available data bases (Refs. 9-5, 9-6, 9-8, 9-10, 9-12) revealed 19 failure rate estimates for a variety of controllers (e.g., pressure, thermostat, electronic, etc.). These estimates were used to develop a distribution of controller failure rates, and this distribution was used to develop conservative parameters of a lognormal distribution to be used in this study.

The median of the controller failure rate estimates is about five failures per million operating hours. This rate was reduced by 50% to represent the fraction corresponding to the failure mode of interest, i.e., "spurious operation" (functions without signal). This reduction is suggested in the IEEE Std. 500-1977 data base. The distribution of controller failure rates suggests an error factor of 12.

<u>Pressure Controller (includes sensor, signal conditioning equipment, and control circuitry)--Spurious Operation (high or low)</u> - 0.007/yr (EF = 12)

A review of available data bases (Refs. 9-5, 9-6, 9-10, 9-12) revealed seven failure rate estimates for pressure controllers. These estimates were used to develop a distribution of pressure controller failure rates.

The median of the pressure controller failure rate estimates is about 1.6 failures per million operating hours. This rate was reduced by 50% to represent the fraction corresponding to the failure mode of interest, i.e., "spurious operation." This reduction is suggested in the IEEE Std. 500-1977 data base.

The error factor for a generic controller, EF = 12, is adopted here for pressure controllers because the seven estimates available for pressure controller failure rates are judged insufficient to represent the spread of the distribution.

# <u>Pump Fails to Start</u> - 5.1 x $10^{-3}$ /demand (EF = 10)

WASH-1400 suggests a  $10^{-3}$ /demand probability of a pump failing to start, with an error factor of 10. This same estimate has been adopted in several other applications, including the Rijnmond study (Ref. 9-8, Table IX.I) and EGG-EA-5887 (Ref. 9-14, page 12). The WASH-1400 estimate is used in this study.

Also, a 4.1 x  $10^{-3}$ /demand probability is added to this estimate to account for cable, circuit breaker (CB), and CB control circuit faults (Ref. 9-15, Table B.5-5).

WASH-1400 suggests a  $10^{-5}/h$  (0.09/yr) estimate with an error factor of 3. A more recent study, EGG-EA-5887 (Ref. 9-39, page 18), proposes a lower,  $10^{-2}/yr$ , estimate with the same error factor. The more recent estimate is assumed for this event, but the error factor has been increased to 5 to reflect uncertainties with respect to applicability of nuclear-related data to the demilitarization facility. 1.55555777

Provensed Passasse

222222

#### Beta-Factor, Generic - 0.14 (EF = 4)

A review of available literature and data bases (Ref. 9-11 and Refs. 9-16 through 9-21) on CCFs revealed 80 beta-factor estimates for a variety of equipment (e.g., pumps, diesel generators, instrumentation and control equipment, etc.). These estimates were used to develop a distribution of beta-factor values, and this distribution was used to develop conservative parameters of a lognormal distribution to be used in this study. The median of the beta-factor estimates is about 0.14 with an error factor of 4.

#### Solenoid Valve Beta-Factor - 0.15 (EF = 4)

The event "Solenoid Valve Fails to Operate on Demand" includes a contribution from the solenoid valve itself and a contribution from the valve relay.

The generic beta-factor, 0.14, was used for the solenoid valve, and the breaker beta-factor, 0.19 (Ref. 9-17), was used for the valve relay. The overall beta-factor for this event is the average of these two betafactor estimates, weighted by their contribution to the event probability:

$$\beta = \frac{0.14 \times 10^{-3} + 0.19 \times 10^{-4}}{10^{-3} + 10^{-4}} = 0.15$$

Damper Beta-Factor - 0.14 (EF = 4)

いたという

The generic beta-factor was assumed applicable for dampers.

アンプライト

LICCONTRACTORIANS

A TOTAL REAL AND A DURING AND A D

2222222

BARRANI, RODOCCE

# Loss of Offsite Power - 0.09/yr (EF = 5)

NUREG/CR-3992 (Ref. 9-22, Table 5.1) estimated the frequency of loss of offsite power to be 0.09/yr based on industry-wide U.S. nuclear power plant data for the years 1959 through 1983. <u>This estimate was</u> <u>derived from plants with at least two offsite power connections</u> (this includes most nuclear power generating plants). An error factor of 5 is subjectively assigned to this event.

Loss of Offsite Gas Supply - 0.01/yr (EF = 10)

This is a subjective estimate.

Spurious Signal Generated by Control System - 0.014/yr (EF = 10)

A plant specific analysis (Ref. 9-23) of a digital control system indicated a 1.6 x  $10^{-6}$ /h frequency of spurious system operations resulting in a spurious signal to a <u>specific</u> component; e.g., commanding a valve to close, given appropriate inputs to the system. (This is not the total frequency of spurious system operations.) An error factor of 10 is assigned due to large uncertainties associated with the applicability of this estimate to the control system at the demilitarization facility.

#### Solenoid Valve Spuriously Closes - 0.0042/yr (EF = 10)

NUREG/CR-2770 (Ref. 9-21, page 92) estimated the frequency of motor-operated values failing to remain open to be 4.8 x  $10^{-7}$ /h. Review of the descriptions of the failure occurrences used in deriving this estimate shows that all spurious closings of values were due to command



faults where a support function fault resulted in a spurious signal to close the valve (e.g., bad switch caused closing contact to stick). Thus, since this frequency estimate does not appear to depend on the type of driver, it is judged applicable to this event.

An error factor of 10 is assigned due to large uncertainties associated with the applicability of nuclear-related data to the demilitarization facility.

# <u>Check Valve Fails to Open</u> - $10^{-4}$ /demand (EF = 5)

WASH-1400 provides a  $10^{-4}$  probability of a check value failing to open on demand, with an error factor of 3. The same estimate is proposed in EGG-EA-5887 (Ref. 9-14, page 13). NUREG/CR-2770 (Ref. 9-21, page 62) provides a 3.1 x  $10^{-7}$ /calendar hour estimate. This estimate is consistent with the WASH-1400 estimate if the value is tested monthly.

The WASH-1400 estimate is assumed for this event, but the error factor is increased to 5 to reflect uncertainties associated with the applicability of nuclear-related data to the demilitarization facility.

# <u>Control (Modulating) Valve Spuriously Opens or Closes</u> - 0.0042/yr (EF = 10)

The same estimate for "Solenoid Valve Spuriously Closes" is used here. The large uncertainty range (EF = 10) is considered sufficient to accommodate equipment variability.

Note: Spurious signals generated by the control system are not included in this estimate.

9-21

Damper Spuriously Closes - 0.0042/yr (EF = 10)

The same estimate for "Solenoid Valve Spuriously Closes" is used here. The large uncertainty range (EF = 10) is considered sufficient to accommodate equipment variability.

# <u>Pressure Controller Diaphragm Valve Fails (open or closed)</u> - 0.013/yr (EF = 10)

The Corps of Engineers (HND) R/M Data Base (Ref. 9-10, pages 1037, 1038) provides an estimate of 3 x  $10^{-6}$ /h for the frequency of failure of pressure regulation values. A 50% chance of failing either open or closed is assumed here. The assumed error factor is 10.

# Level Indicator--Spurious Operation - 0.06/yr (EF = 4)

A review of available data bases (Refs. 9-5 through 9-8, and Ref. 9-12) revealed ten failure rate estimates for level switches, level sensors, and level transmitters. These estimates were used to develop a distribution of level indicator failure rates, and this distribution was used to develop conservative parameters of a lognormal distribution to be used in this study.

The median of level indicator failure rate estimates is about 0.12 failures per operating year. This rate was arbitrarily reduced by 50% to represent the fraction corresponding to the failure mode of interest, i.e., "spurious operation." This reduction is believed to be conservative. The distribution of level indicator failure rates suggests an error factor of 4.

#### Temperature Detector--Spurious Operation - 0.095/yr (EF = 6)

A review of available data bases (Refs. 9-5 through 9-8, 9-12, and 9-13) revealed seventeen failure rate estimates for temperature switches, temperature indicators, and temperature transducers. These estimates were used to develop a distribution of temperature detector failure rates, and this distribution was used to develop conservative parameters of a lognormal distribution to be used in this study.

The median of temperature detector failure rate estimates is about 0.19 failures per operating year. This rate was arbitrarily reduced by 50% to represent the fraction corresponding to the failure mode of interest, i.e., "spurious operation"; this reduction is believed to be conservative. The distribution of temperature detector failure rates suggests an error factor of 6.

# Solenoid Valve Fails to Operate on Demand - 1.1 x $10^{-3}$ /demand (EF = 5)

The IREP data base (Ref. 9-24) proposes a  $10^{-3}$ /demand probability for this event, with an error factor of 3. The IREP estimate is adopted in this study, but the error factor has been increased to 5 to reflect the uncertainty associated with the applicability of the IREP data to the demilitarization plant equipment. Also, a  $10^{-4}$ /demand probability is added to this estimate to account for the valve relay failure to open on demand (see Relay/Breaker Fails to Operate).

#### Pressure Switch--Spurious Operation - 0.037/yr (EF = 5)

A review of available data bases (Refs. 9-5 through 9-8, 9-11, 9-13, 9-24, and 9-25) revealed thirteen failure rate estimates for a variety of pressure switches. These estimates were used to develop a distribution of pressure switch failure rates, and this distribution was used to develop conservative parameters of a lognormal distribution to be used in this study.

The median of pressure switch failure rate estimates is about 0.074 failures per operating year. This rate was arbitrarily reduced by 50% to represent the fraction corresponding to the failure mode of interest, i.e., "spurious operation"; this reduction is believed to be conservative. The distribution of pressure switch failure rates suggests an error factor of 5.

# Damper Fails to Operate on Demand - 1.1 x $10^{-3}$ /demand (EF = 10)

The same probability assumed for a solenoid valve failing to operate on demand is used here. The error factor has been increased to 10 to account for equipment differences.

# <u>Relay/Breaker Spuriously Open</u> - 8.8 x $10^{-5}$ /yr (EF = 10)

The IREP data base (Ref. 9-24, Table 5.1-1) proposes a failure rate of  $10^{-8}/h$  for loss of an electrical bus, with an error factor of 10. The loss of a bus event is dominated by failure of the supply breaker; thus the IREP estimate is used here for a relay/breaker spuriously opening.

# <u>Relay/Breaker Fails to Operate</u> - $10^{-4}$ /demand (EF = 10)

The IREP data base (Ref. 9-24, Table 5.1-1) proposes a  $10^{-4}$ /demand probability of a relay failing to operate on demand, with an error factor of 10.

# Circuit Breaker Fails to Operate - $10^{-3}$ /demand (EF = 10)

The IREP data base (Ref. 9-24, Table 5.1-1) proposes a  $10^{-3}$ /demand probability of a circuit breaker failing to operate on demand, with an error factor of 10.

# Solid State Relay Fails to Operate - 1.8 x $10^{-4}$ /demand (EF = 5)

MIL-HDBK-217D (Ref. 9-25) provides a failure rate estimate of 0.5 x  $10^{-6}$ /h for a solid state (thyristor) relay (assuming GF conditions in Table 5-2-10 and a quality factor of 5 in Table 5-2-11). This estimate

results, with an assumed monthly test scheme, in a  $1.8 \times 10^{-4}$  probability of failure on demand. The assumed error factor for this event is 5.

#### 9.1.4. Handling Accident Data

All initiating event frequency accidents, except for forklift collisions, were derived from the human reliability analysis and are discussed in Section 9.2.

The forklift collision accident frequency was derived from Ref. 9-2. In Ref. 9-2, accidents were defined to include incidents that result in fatalities, injuries, or property damage. The basic truck accident rate is 2.5 x  $10^{-6}$  accidents/mile. From Table II of Ref. 9-2, the percent of accidents leading to collisions with trucks, autos, and stationary objects and overturns is 89.35%. Table III of Ref. 9-2 also show that 50% of all accidents occur at 30 to 40 mph.

To convert the basic rate to accidents per operation, the operator's exposure time in the highway is determined. If the operator was traveling at 35 mph, the exposure time is 1.7 min.

In order to apply this information to forklift collision accidents, the following were assumed:

- The total operator exposure time during the forklift operation is 10 min. This includes the lifting of munitions from the stack, moving them to another area, and unloading them.
- 2. The time to travel from one point to another is assumed to be one-third of the total time, or 3.3 min.
- Forklift collisions will occur at speeds no greater than
  40 mph (i.e., two forklifts traveling at 20 mph).

Therefore, forklift collision accident rate is:

2.5 x 10<sup>-6</sup> x 0.893 x 
$$\frac{3.3}{1.7}$$
 = 4.3 x 10<sup>-6</sup>/operation

This median value is assigned an error factor of 10 on the basis that the data is only for 6 yr and there may be other unreported incidents more directly related to forklift operations.

Reference 9-2 also indicates that 25% of fires result from collision-type accidents. It is not evident from the data if fire from collision is directly proportional to truck speed. Our analysis assumes that it is. Therefore, we modified the data as follows:

Probability of fire =  $0.25 \times 0.29 = 0.0725$ 

where the factor 0.29 represents the percent of collisions occurring at less than 20 mph.

ANTAL AND THE SERVICE THE SAME THE SAME THE SAME AND THE SAME TH

#### 9.2. HUMAN FACTORS DATA

#### 9.2.1. Human-Error Probability Estimation - Handling Accidents

Human-error probabilities were quantified using the approach to human-error estimation described in NUREG/CR-1278 (Ref. 9-26), probabilities of human errors were estimated based on several performanceshaping factors such as munition configuration, handling operation, clothing level, and crew size. These factors are identified in the discussions that follow on the derivations of each estimate. Table 9-8 lists the error probabilities estimated for puncturing or dropping a munition based on each of these factors. These error probabilities will be incorporated into the handling scenarios as shown in the data tables in Table 6-4.

1. <u>Puncturing a munition</u>. The basis for the error estimates is taken from Section 4.4.2 of Ref. 9-27 (pages 4.4 through 4.26). This reference gives  $4 \times 10^{-5}$  as a data-based estimate of the probability of handling errors using forklifts for the rocket stockpile. This is an estimate of the likelihood of an error in forklift operation that potentially could lead to a warhead rupture while attempting to isolate a leaking rocket inside the storage igloo.

That estimate is based on conditions that do not entirely represent those assumed by this study; namely, that a threeman crew will perform all forklift operations. In this study, it is assumed that a two-man crew will perform all forklift operations--one driving the forklift and one guiding forklift and munition position from the ground. This means that the data-based estimate may not represent the probability of forklift-handling errors expected under actual conditions. Therefore, this estimate was revised to  $1 \times 10^{-4}$  to account

|       | <b>OPERATION</b> |
|-------|------------------|
|       | HANDLING         |
| 9-8   | PER              |
| TABLE | PROBABILITIES    |
|       | ERROR            |
|       | HUMAN            |

|                               |                        | Ha                     | ndling Operatio            | n for Clothing           | Type                         |                        |
|-------------------------------|------------------------|------------------------|----------------------------|--------------------------|------------------------------|------------------------|
| Turne Turne                   | Level A                | or DPE                 | Levels B,<br>(Mask, Gloves | C, and D<br>, and Boots) | Levels E<br>(Street Clothes, | and F<br>Mask Slung)   |
| For Munition<br>Configuration | Hand<br>Carry(a)       | Forklift               | Hand<br>Carry(a)           | Forklift                 | Hand<br>Carry(a)             | <b>Porklift</b>        |
| Tine Carried                  |                        |                        |                            |                          |                              |                        |
| Drop                          | 6.0 x 10 <sup>-4</sup> | 3.0 x 10 <sup>-4</sup> | 3.0 x 10 <sup>-4</sup>     | 1.5 x 10 <sup>-4</sup>   | 6.0 x 10 <sup>-5</sup>       | 3.0 x 10 <sup>-5</sup> |
| Puncture                      | VN                     | 1.0 x 10 <sup>-4</sup> | VN                         | 5.0 x 10-5               | NA                           | 1.0 x 10 <sup>-5</sup> |
| Beam Carried                  |                        |                        |                            |                          |                              |                        |
| Drop                          | NA                     | 3.0 x 10 <sup>-5</sup> | NA                         | 1.5 x 10 <sup>-5</sup>   | NA                           | 3.0 x 10-6             |
| Puncture                      | VN                     | VN                     | NA                         | NA                       | NA .                         | NA                     |
|                               |                        |                        |                            |                          |                              |                        |

(a)Hand-carry operations involve one weapon at a time.



PARTICIPATION NOT THE PARTICIPATION OF A CONTRACT PARTICI

LUCENCE DESERT STRUCT



Personal and a second

for a smaller crew. The revised estimate of  $1 \times 10^{-4}$  is the probability that one or both members of a two-man crew will err such that the forklift time is in a position to puncture a munition. (This puncture probability applies to those cases in which forklift times are used to lift munitions; it includes palletized munitions and spray tanks in overpacks.)

Another difference is that the original estimate from Ref. 9-27 (4 x  $10^{-5}$ ) was based on operations with leaking rockets. This meant that it assumes that the crew is wearing Level A protective clothing. If the same forklift operations are performed in less strenuous circumstances (i.e., if a lower level of protective clothing is worn), the error probability estimate can be lowered. Here, it has been lowered to  $5 \times 10^{-5}$  for the case of the operators' wearing partial protection (masks, gloves, and boots) and to  $1 \times 10^{-5}$  for the case of their wearing minimal protection (street clothes, with masks slung).

2. <u>Dropping a munition</u>. For palletized munitions and spray tanks in their overpacks, human-caused drops from forklifts are judged to be three times as likely as punctures caused by operating the same kind of forklift. The error-probability estimates are  $3 \times 10^{-4}$ ,  $1.5 \times 10^{-5}$ , and  $3 \times 10^{-5}$  for dropping a munition from a forklift time when wearing Level A, Level C, or Level F protective clothing, respectively.

Because of unwieldy pallet and overpacked spray tank loads, and because it is assumed that forklift-time loads are likely to be carried at higher speeds than are forklift-beam loads, the likelihood of a ton container or other beam-carried loads being dropped because of human error is judged to be an order of magnitude lower than that of a time-carried load being dropped. These are estimated to be  $3 \times 10^{-5}$ ,  $1.5 \times 10^{-6}$ , and

3 x  $10^{-6}$  for protective clothing Levels A, C, and F, respectively.

For hand-carried munitions, munition drops are estimated to be twice as likely as drops of tine-carried load from forklifts. The estimated probabilities of dropping a hand-carried munition when wearing Levels A, C, and F protective clothing are  $6 \times 10^{-4}$ ,  $3 \times 10^{-4}$ , and  $6 \times 10^{-5}$ , respectively. (Loads carried by forklift beams are never hand carried.)

These probability estimates are the likelihood of an error per handling operation. A single forklift operation may involve a single munition such as a spray tank or as many as 48 weapons on a pallet, while a single hand-carry operation will always involve only a single munition.

3. Failing to detect a leaking munition in a package. The probability of an operator's failing to detect a leak is based on his failing to monitor an OFC before opening it. The error probability is estimated as  $1 \times 10^{-3}$  based on item 9 from Table 20 through 22 of NUREG/CR-1278 (Ref. 9-26). This humanerror probability is the probability that a checker will fail to check equipment status when that status affects the checker's own safety. Since the continers are loaded elsewhere (or at least by other operators), the unloader should be cautious when handling them; he has no way to ensure a "clean" vault interior, so he will probably want to protect himself. This error estimates that the operator is likely to overlook this check on one out of every thousand vaults or transportation containers that he opens.

# 9.2.2. Human-Reliability Analysis for Plant Operations

R

The human-reliability analysis (HRA) for plant operations was conducted as an input to the plant operations internal events analysis. This section describes the scope of the HRA, the methodology used, the screening performed, and the final quantification. N\_20003X78\_\_2000314\_N/200332\_2000334\_

9.2.2.1. <u>Scope</u>. The preliminary fault-tree and event-tree models for plant operations were examined to identify human actions that had the potential to mitigate agent release. For screening, these human actions were categorized and assigned conservative human-error probabilities. Once the plant operations scenarios had been screened on the basis of frequency and consequence, the survivors were examined in greater detail to identify important human actions and to identify plant/operating system characteristics that could influence human-error probabilities. The important human actions were quantified, taking this information into account, and were integrated into the final fault-tree and event-tree models.

9.2.2.2. Methodology. Screening and final estimates of human-error probabilities were obtained by using the Technique for Human Reliability Analysis (THERP) as described in NUREG/CR-1278 (Ref. 9-26). This technique calls for identifying individual human errors and for describing the set of performance-shaping factors (PSFs) that pertain to each task situation. Usually, such descriptions are very task-, site-, and situation-specific. In this case, since there was no finished, approved human-performance system to analyze, more generic descriptions of task situations were used. That is, several assumptions about what could be realistically expected for a generic CONUS site were made, since there are, as yet, no written procedures for CONUS, no site-specific manmachine interface, no training program beyond the conceptual stage, and no finished plant design (except that for JACADS) that allows for time data to be collected. The human-reliability analysis for plant operations was based on these assumptions, which are listed in Appendix E.

9.2.2.3. <u>Screening</u>. To screen the plant-operations scenarios, generic human-error events were defined. The plant-operations logic models (fault trees and event trees) were examined to identify appropriate areas for considering the human-error contribution to release frequencies. At appropriate places on these logic models, one or more of the generic human-error events were placed, or it was determined that the human-error contribution had already been taken into account there implicitly.

Conservative human-error probabilities were estimated for each of the error events. The conservative estimates may be considered to represent the upper bound of a worst-case human-action situation. The screening human-error events are described in Table 9-9 along with the data source for each error probability. In general, the HEPs used for screening purposes are either (1) factors of 3 to 10 higher than the upper bounds reported in Ref. 9-73, (2) taken to be 1.0 or (3) assumed conservative values based on analyst experience and scientific judgment. Once these conservative values had been used in the quantitative scenario screening, more realistic human-error probabilities were estimated for the surviving scenarios.

9.2.2.4. <u>Final Quantification</u>. A preliminary draft of the event trees was examined to identify any human actions that might serve as initiators to, or mitigators of, accident scenarios. Those human actions were categorized according to the system or equipment interface dealt with by the operators. (As is usual with other risk assessments, human errors in maintenance activities were not quantified explicitly since those errors contribute to the already-estimated hardware-failure probabilities.) Table 9-10 lists those human actions in scenario-identifier order.

For final quantification, this list was grouped according to error types. Ten error types were identified that focus on: ignition, fire

# TABLE 9-9 SCREENING QUANTIFICATION FOR HUMAN-RELIABILITY ANALYSIS OF PLANT OPERATIONS

ひんしいと

| Index | Error Event                                                                                                                                                                                                    | HEP                  | Source(a)                                                                                                   |
|-------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------|-------------------------------------------------------------------------------------------------------------|
| 1     | Operator fails to respond to an alarm<br>indication. Correct response is in the<br>control room and may include taking<br>simple control action or initiating<br>emergency shutdown.                           | 1 x 10 <sup>-1</sup> | Table 20-23,<br>item 2b<br>(factor of 10<br>higher than<br>upper bound)                                     |
| 2     | Operator fails to respond to an alarm<br>indication. Correct response is outside<br>the control room, and Decontamination<br>Protective Ensemble (DPE) may be<br>required.                                     | 3 x 10 <sup>-1</sup> | (Factor of 3<br>above Index 1)                                                                              |
| 3     | Operator fails to notice a malfunction<br>or existing condition on the closed-<br>circuit TV screen. He fails to shut the<br>operation down as a result.                                                       | 5 x 10 <sup>-1</sup> | Table 20-10,<br>item 7 (upper<br>bound (or<br>Table 20-22,<br>item 4 (factor<br>of 10 above<br>upper bound) |
| 4     | Operator fails to monitor the operating<br>system. He fails to carry out a required<br>action such as closing a valve or closing<br>a blast door.                                                              | 3 x 10 <sup>-1</sup> | Table 20-6,<br>item 2 (factor<br>of 10 above<br>upper bound)                                                |
| 5     | Operator shuts down, disables, or delays<br>the operation of a safety system. This<br>could be because he misinterprets system<br>status or because the information he<br>received is incorrect or incomplete. | 1 x 10 <sup>-1</sup> | Table 20-3,<br>item 2 (by 10<br>minutes after<br>signal)                                                    |
| 6     | Operator takes action that initiates a fire or some other sequence of catastrophic events.                                                                                                                     | 1 x 10 <sup>-2</sup> | Scientific<br>judgment                                                                                      |
| 7     | Operator fails to take action to miti-<br>gate fire. He fails to close the<br>dampers.                                                                                                                         | 1.0                  | Table 20-3,<br>item 2 (by 10<br>minutes, upper<br>bound)                                                    |
| 8     | Operator fails to implement action to recover from upset condition.                                                                                                                                            | 1.0                  | Scientific<br>judgment                                                                                      |

9-33

TABLE 9-9 (Continued)

| Index | Error Event                                                                                                                                                                 | HEP                  | Source(a)                                                    |
|-------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------|--------------------------------------------------------------|
| 9     | Maintainer fails to perform tasks, to<br>perform them correctly, or to perform<br>them on time.                                                                             | 3 x 10 <sup>-1</sup> | Table 20-6,<br>item 7                                        |
| 10    | Operator fails to carry out administra-<br>tive control policy. He fails to<br>initiate a regularly scheduled action<br>or fails to follow standard operating<br>procedure. | 5 x 10-1             | Table 20-6,<br>item 1 (factor<br>of 10 above<br>upper bound) |
| 11    | Operator selects wrong component to operate.                                                                                                                                | 5 x 10-2             | Table 20-12,<br>item 2 (factor<br>of 5 above<br>upper bound) |
| 12    | Operator drops or damages munition while<br>controlling it manually, lifting or<br>carrying it with a forklift, or carrying<br>it by hand.                                  | 3 x 10 <sup>-1</sup> | Scientific<br>judgment                                       |

(a) Unless stated otherwise, all tables and item numbers refer to NUREG/CR-1278 (Ref. 9-26).

アンシン



TABLE 9-10 HUMAN-ERROR EVENTS BY SEQUENCE

<u> Esta esta e</u>

22223

| No.  | Error Events       | Area | Munition          | Sequence         |
|------|--------------------|------|-------------------|------------------|
| 1    | Conveyor Loading   | ECV  | Ton Container     | ECV-1            |
| 2    | ignition           | ECV  | Ton Container     | ECV-1            |
| 3    | Fire Suppression   | ECV  | Ton Container     | ECV-1            |
| - 4  | Ventilation System | ECV  | Ton Container     | ECV-1            |
| 5    | Conveyor Loading   | ECV  | Ton Container     | ECV-2            |
| 6    | ignition           | ECV  | Ton Container     | ECV-2            |
| 7    | Fire Suppression   | ECV  | Ton Container     | ECV-2            |
| 8    | Conveyor Loading   | ECV  | M55 Rocket        | ECV-3            |
| 9    | Conveyor Loading   | ECV  | M55 Rocket        | ECV-4            |
| 10   | Fire Suppression   | ECV  | M55 ROCKet        | ECV-4            |
| 11   | Conveyor Loading   | ECV  | M55 ROCKET        | ECV-5            |
| 12   | Fire Suppression   | ECV  | M55 HOCKet        | ECV-5            |
| 13   | Conveyor Loading   | ECV  | Mine              | ECV-5            |
| 14   | Conveyor Loading   | ECV  | Mine              |                  |
| 15   | Fire Suppression   | EGV  | MINO              |                  |
| 16   | Conveyor Loading   | ECV  |                   |                  |
| 17   | Fire Suppression   | ECV  | Mine .            |                  |
| 18   | Conveyor Loading   | ECV  | a projectile      |                  |
| 19   | Conveyor Loading   | ECV  |                   | ECV-10           |
| 20   | Fire Suppression   | ECV  | a projectile      | ECV-11           |
| 21   |                    | ECV  | a projectile      | ECV-11           |
| 22   |                    | ECV  |                   | ECV-12           |
| 20   | Conveyor Loading   | ECV  | 105-mm Projectile | ECV-12<br>ECV-13 |
| 24   |                    | FCV  | 105-mm Projectile | ECV-13           |
| 26   |                    | ECV  | 105-mm Projectile | ECV-14           |
| 27   | Fire Suppression   | ECV  | 105-mm Projectile | ECV-14           |
| 28   | Conveyor Loading   | ECV  | 105-mm Projectile | ECV-15           |
| 29   | Conveyor Loading   | ECV  | 105-mm Projectile | ECV-16           |
| 30   | Fire Suppression   | ECV  | 105-mm Projectile | ECV-16           |
| 31   | Conveyor Loading   | ECV  | 105-mm Projectile | ECV-17           |
| 32   | Fire Suppression   | ECV  | 105-mm Projectlie | ECV-17           |
| 33   | Undrained Munition | ECR  | Mine              | ECR-1DM          |
| 34   | Ventilation System | ECR  | Mine              | ECR-1DM          |
| 35   | Undrained Munition | ECR  | Mine              | ECR-2DM          |
| 36   | Ventilation System | ECR  | Mine              | ECR-2DM          |
| 37   | Undrained Munition | ECR  | Mine              | ECR-3DM          |
| 38   | Ignition           | ECR  | Mine              | ECR-3DM          |
| 39   | Fire Suppression   | ECR  | Mine              | ECR-3DM          |
| 40   | Ventilation System | ECR  | Mine              | ECR-3DM          |
| 41   | Undrained Munition | ECR  | Mine              | ECR-4DM          |
| 42   | Fire Suppression   | ECR  | Mine              | ECR-4DM          |
| 43   | Ventilation System | ECR  | Mine              | ECR-4DM          |
| - 44 | Undrained Munition | ECR  | Projectile        | ECR-1DP          |
| 45   | Ventilation System | ECR  | Projectile        | ECR-10P          |
| 48   | Undrained Munition | ECR  | Projectile        | ECR-2DP          |
| 47   | Ventilation System | ECR  | Projectile        | ECR-2DP          |
| 48   | Undrained Munition | ECR  | Projectile        | ECH-JDP          |
| 49   | FIRE Suppression   | ECR  | Projectile        |                  |
| 50   | Ventliation System | ECR  | Projectile        |                  |
| 51   | Undrained Munition | LCR  | Projectile        |                  |
| 52   | CODERESSION        |      | FF0/8G118         | 597-707          |

383

1310-1300000000000

11.48

terestication betweeter totationoo

9-35

TABLE 9-10 (Continued)

R

<u> .....</u>

| No. | Error Events        | Area | Munition        | Sequence |
|-----|---------------------|------|-----------------|----------|
| 53  | Ventilation System  | ECR  | Projectile      | EC8-40P  |
| 54  | Undrained Munition  | ECR  | Rocket          | ECR-1DR  |
| 55  | Ventilation System  | ECR  | Rocket          | ECR-1DR  |
| 56  | Undrained Munition  | ECR  | Rocket          | ECR-2DR  |
| 57  | Fire Suppression    | ECR  | Rocket          | ECR-2DR  |
| 58  | Ventilation System  | ECR  | Rocket          | ECR-2DR  |
| 59  | Undrained Munition  | ECR  | Rocket          | ECR-3DR  |
| 60  | Fire Suppression    | ECR  | Rocket          | ECR-3DR  |
| 61  | Ventilation System  | ECR  | Rocket          | ECR-3DR  |
| 62  | Undrained Munition  | ECR  | Rocket          | ECR-40R  |
| 63  | Fire Suppression    | ECR  | Rocket          | ECR-4DR  |
| 64  | Undrained Munition  | ECR  | Rocket          | ECR-5DR  |
| 65  | Ventilation System  | ECR  | Rocket          | ECR-5DR  |
| 66  | Undrained Munition  | ECR  | Rocket          | ECR-6DR  |
| 67  | Fire Suppression    | ECR  | Rocket          | ECR-6DR  |
| 68  | Ventilation System  | ECR  | Rocket          | ECR-60R  |
| 69  | Undrained Munition  | ECR  | Rocket          | ECR-7DR  |
| 70  | Fire Suppression    | ECR  | Rocket          | ECR-7DR  |
| 71  | Ventilation System  | ECR  | Rocket          | ECR-7DR  |
| 72  | Spurlous Drain      | MPB  | Bulk Container  | MP8-28   |
| 73  | Ignition            | MPB  | Bulk Container  | MP8-28   |
| 74  | Fire Suppression    | MPB  | Bulk Container  | MPB-28   |
| 75  | Ventilation System  | MPB  | Bulk Container  | MPB-2B   |
| 75  | Spurious Drain      | MPD  | Bulk Container  | MPB-38   |
| 79  |                     |      | Bulk Container  | MPB-38   |
| 70  | Ventlistice System  | MOR  | Bulk Container  | MPB-38   |
| 80  | Sourious Drain      | MPR  | Bulk Container  | MPB-38   |
| 81  | Fire Suppression    | MPB  | Bulk Container  | MPD-4D   |
| 82  | Spurious Drain      | MPB  | Bulk Containers | MPR_68   |
| 83  | Fire Suppression    | MPB  | Buik Containers | MP8_58   |
| 84  | Ventilation System  | MPB  | Buik Containers | MP8-58   |
| 85  | Spurious Drain      | MPB  | Bulk Containers | MPB-6B   |
| 86  | Fire Suppression    | MPB  | Buik Containers | MPB-6B   |
| 87  | Undrained Munition  | MPB  | Projectile      | MPB-1DP  |
| 88  | Ventilation System  | MPB  | Projectile      | MPB-1DP  |
| 89  | Undrained Munition  | MPB  | Projectile      | MPB-2DP  |
| 90  | Fire Suppression    | MPB  | Projectile      | MPB-2DP  |
| 91  | Ventilation System  | MPB  | Projectile      | MPB-2DP  |
| 92  | Undrained Munition  | MPB  | Projectile      | MPB-3DP  |
| 93  | Fire Suppression    | MPB  | Projectile      | MPB-3DP  |
| 94  | Ventliation System  | MPB  | Projectile      | MPB-3DP  |
| 95  | Conveyor Loading    | BSA  | Ton Container   | BSA-1    |
| 96  | Ignition            | BSA  | Ton Container   | BSA-1    |
| 97  | Fire Suppression    | BSA  | Ton Container   | BSA-1    |
| 98  | Undrained Munition  | BSA  | Ton Container   | BSA-2    |
| 99  | Conveyor Loading    | BSA  | Ton Container   | BSA-2    |
| 100 |                     | BSA  | Ton Container   | BSA-2    |
| 101 | FILE SUPPRESSION    | BSA  | Ton Container   | BSA-2    |
| 102 | Ventilation System  | BSA  | Ton Container   | BSA-Z    |
| 103 | Sump Fump Operation | TOX  | Agent Tank      | TOX-2    |
| 104 | FIFE SUPPression    |      | Agent lank      | 10X-Z    |

TABLE 9-10 (Continued)

| No. | Error Events        | Area | Munition                 | Sequence         |  |  |  |  |  |  |  |
|-----|---------------------|------|--------------------------|------------------|--|--|--|--|--|--|--|
| 105 | Ventilation System  | TOX  | Agent Tank               | TOX-2            |  |  |  |  |  |  |  |
| 105 | Sump Pump Operation | TOX  | Agent Tank               | TOX-3            |  |  |  |  |  |  |  |
| 107 | Fire Suppression    | TOX  | Agent Tank               | TOX-3            |  |  |  |  |  |  |  |
| 108 | Ventilation System  | TOX  | Agent Tank               | TOX-3            |  |  |  |  |  |  |  |
| 109 | Fire Suppression    | TOX  | Agent Tank               | TOX-4            |  |  |  |  |  |  |  |
| 110 | Sump Pump Operation | TOX  | Agent Tank               | TOX-5            |  |  |  |  |  |  |  |
| 111 | Fire Suppression    | TOX  | Agent Tank               | TOX-5            |  |  |  |  |  |  |  |
| 112 | Ventilation System  | TOX  | Agent Tank               | TOX-5            |  |  |  |  |  |  |  |
| 113 | Sump Pump Operation | TOX  | Agent Tank               | TOX-6            |  |  |  |  |  |  |  |
| 114 | Fire Suppression    | TOX  | Agent Tank               | TOX-6            |  |  |  |  |  |  |  |
| 115 | Tank Overfill       | TOX  | Agent Tank               | TOX-8            |  |  |  |  |  |  |  |
| 116 | Fire Suppression    | TOX  | Agent Tank               | TOX-8            |  |  |  |  |  |  |  |
| 117 | Ventilation System  | TOX  | Agent Tank               | TOX-8            |  |  |  |  |  |  |  |
| 118 | Tank Overfill       | TOX  | Agent: Tank              | TOX-9            |  |  |  |  |  |  |  |
| 119 | Fire Suppression    | TOX  | Agent Tank               | TOX-9            |  |  |  |  |  |  |  |
| 120 | Shutdown Signal     | LIC  | Ail Munitions            | L11-001          |  |  |  |  |  |  |  |
| 121 | Stop Fuel           | LIC  | All Munitions            | L12-001          |  |  |  |  |  |  |  |
| 122 | Stop Combustion     | LIC  | All Munitions            | L12-002          |  |  |  |  |  |  |  |
| 123 | Stop Fuel           | LIC  | All Munitions            | L12-005          |  |  |  |  |  |  |  |
| 124 | Shutdown Signal     | MPF  | Bulk, Projectiles        | MP1-001, MP2-005 |  |  |  |  |  |  |  |
| 125 | Shutdown Signal     | MPF  | Bulk, Projectiles        | MP2-001, MP2-003 |  |  |  |  |  |  |  |
| 126 | Stop Fuel           | MPF  | Bulk, Projectiles        | MP2-002          |  |  |  |  |  |  |  |
| 127 | Shutdown Signal     | MPF  | Bulk, Projectiles        | MP2-004          |  |  |  |  |  |  |  |
| 128 | Undrained Munition  | MPF  | Bulk, Projectiles        | MP3-001          |  |  |  |  |  |  |  |
| 129 | Shutdown Signal     | MPF  | Bulk, Projectiles        | MP3-001          |  |  |  |  |  |  |  |
| 130 | Undrained Munition  | MPT  | BUIK, Projectiles        | MP3-002          |  |  |  |  |  |  |  |
| 131 | Undrained Munition  | MPF  | Buik, Projectiles        | MP3-003          |  |  |  |  |  |  |  |
| 172 | Undrained Munition  | MPF  | Bulk, Projectiles        | MP3-004          |  |  |  |  |  |  |  |
| 133 |                     | MPE  | Bulk Projectiles         | MP4-001          |  |  |  |  |  |  |  |
| 176 | Shutdown Signal     | MOE  | Bulk Projectiles         | MP4-001          |  |  |  |  |  |  |  |
| 138 | Stop Fuel           | MDE  | Bulk Projectiles         | MP4-002, MP4-004 |  |  |  |  |  |  |  |
| 137 | Stop Compustion     | MPF  | Bulk Projectiles         | MP4-005          |  |  |  |  |  |  |  |
| 138 | Shutdown Slonel     | DES  | Bursters Bockett Miner   | MP4-005          |  |  |  |  |  |  |  |
| 139 | Shutdown Signal     | DES  | Bursters Bockets Mines   | DF2-001 DF2-003  |  |  |  |  |  |  |  |
| 140 | Stop Fuel           | DFS  | Bursters, Rockets Mines  | DF2-002          |  |  |  |  |  |  |  |
| 141 | Stop Combustion     | DFS  | Bursters, Rockets, Mines | DE2-004          |  |  |  |  |  |  |  |
| 142 | Stop Agent          | DFS  | Bursters, Rockets, Mines | DF2-006          |  |  |  |  |  |  |  |
| 143 | Shutdown Signal     | DFS  | Bursters, Rockets, Mines | DF2-005          |  |  |  |  |  |  |  |
| 144 | Fast Food           | DFS  | Bursters, Rockets, Mines | DE3-001          |  |  |  |  |  |  |  |
| 145 | Shutdown Signal     | DFS  | Bursters, Rockets, Mines | DF4-001          |  |  |  |  |  |  |  |
| 146 | Shutdown Signal     | DFS  | Bursters, Rockets, Mines | DE5-001 DE5-003  |  |  |  |  |  |  |  |
| 147 | Shutdown Signal     | DFS  | Bursters, Rockets, Mines | DF5-001 DF5-003  |  |  |  |  |  |  |  |
| 148 | Stop Fuel           | DFS  | Bursters, Rockets, Mines | DF5-002          |  |  |  |  |  |  |  |
| 149 | Shutuown Signal     | DFS  | Bursters, Rockets, Mines | DF5-002          |  |  |  |  |  |  |  |
| 150 | Stop Fuel           | DFS  | Bursters, Rockets, Mines | DF5-004          |  |  |  |  |  |  |  |
| 151 | Stop Agent          | DFS  | Bursters, Rockets, Mines | DF5-005          |  |  |  |  |  |  |  |
| 152 | Munition Counting   | DUN  | All Munitions            | DU1-001          |  |  |  |  |  |  |  |
| 153 | Munition Counting   | DUN  | All Munitions            | DU1-002          |  |  |  |  |  |  |  |
| 154 | Munition Counting   | DUN  | All Munitions            | DU1-003          |  |  |  |  |  |  |  |
| 155 | Munition Counting   | DUN  | All Munitions            | DU1-004          |  |  |  |  |  |  |  |

Kara recent second to solver the second recent recent recent recent recent recent recent recent recent real

6436

suppression, conveyor loading, munition counting, tank overfill, sump pump operation, undrained munition, furnace ventilation, ventilation system, and air compressors. Table 9-11 shows the error events, the area of the plant involved, the munition type involved, the scenario identifier, the error probability, and the error factor associated with each quantification. The data sources for the error types are described below. The data represent medians and error factors of lognormal distributions. **NYNYNYN** 

9.2.2.4.1. <u>Ignition</u>. The operator or maintainer could serve as an ignition source in some areas of the plant. For the operators, the credible cases consist of those geographical areas in which he works or traffics. These include the control room, the receiving site Unpack Area (UPA), the Instrumentation and Electric Power room (IEP), and the observation corridors. For the maintainers, these include all areas (although his entry into most areas may be limited to down times). Operators and maintainers could initiate ignition by using an ignition source in the area (e.g., by smoking or welding) or by causing sparks (e.g., by dropping a munition or other object that could create sparks). The first of these will be controlled administratively throughout the plant; the operators will only be allowed to smoke in the control room and outdoors.

For plant areas requiring the wearing of Level C or higher protective clothing, masks must be worn; this physically rules out smoking in these areas. Therefore, smoking as an initiator is credible only in the control room and in the IEP, where Levels E and D, respectively, are required. Smoking even in these areas is a failure of administrative control.

The lower bound of a failure of administrative control is 0.002 (Ref. 9-26, Table 20-6, item 1). The likelihood of a checker's failing to check something when his own safety is involved is  $1 \times 10^{-3}$  (Ref. 9-26, Table 20-22, item 9). The second value was selected as



100



1.01

# TABLE 9-11 HUMAN-ERROR EVENTS FOR FINAL QUANTIFICATION

| EF                | 10             | 0              | 2        | 2             | 2             | 0             | 2             | 5                | 0                | 2                | 0                | 2                | 2                | 9                | 10               | 01               | 9                | 2                | 2                | 0                | 01               | 0                | 10               | 0                | 0                | 10               | 0                | 0                | <u>0</u>         | 10               | 0                | 10               | 0                | 0                | 10               |
|-------------------|----------------|----------------|----------|---------------|---------------|---------------|---------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|------------------|
| blity<br>15 Min   |                |                |          |               |               |               |               | 6E-5             | 6E-5             | 6E5              | 5E-5             | 6E-6             | 5E-5             | 4E-3             | 4E3              | 4E3              | 4E-3             | 46-3             | 4E-3             |
| r Proba<br>10 Min |                |                |          |               |               |               |               | 2E-4             | 2E-4             | 2E-4             | 2E-4             | 2E-4             | 2E-4             | 15-2             | 16-2             | 1E-2             | 1E-2             | 1E-2             | 16-2             | 1E-2             | 16-2             | 1E-2             | 16-2             | 1E-2             | 1E2              | 1E-2             | 1E-2             | 1E-2             |
| Erro<br>5 Min     | epsiton        | epsiion        | 6E-4     | epsiton       | epsiton       | epsilon       | epsilon       | 56-3             | 56-3             | 56-3             | 5E-3             | 5E-3             | 5E-3             | 4E-2             | 46-2             | 4E-2             | 4E-2             | 46-2             | 4E-2             | 4E-2             | 46-2             | 4E-2             | 46-2             | 46-2             | 46-2             | 46-2             | 4E-2             | 4E-2             | 4E-2             | 46-2             | 4E-2             | 4E-2             | 4E-2             | 4E-2             | 4E-2             |
| Sequence          | MPB-28         | MPB-3B         | ECR-3DM  | BSA-2         | ECV-1         | ECV-2         | BSA-1         | TOX-4            | TOX-5            | TOX-9            | TOX-3            | TOX-8            | TOX-6            | MP8-28           | MPB-4B           | MPB-3B           | ECV-5            | ECV-4            | ECV-7            | ECR-4DM          | ECV-8            | ECR-3DM          | MPB6B            | MPB-5B           | MPB-20P          | ECR-3DP          | ECR-4DP          | MPB-3DP          | ECR-7DR          | ECR-4DR          | ECR-3DR          | ECR-6DR          | ECR-2DR          | BSA-2            | ECV-1            |
| Scenario          |                |                |          |               |               | POTAF 043     | POKAF 053     |                  |                  |                  |                  |                  |                  |                  | POKAF 051        |                  | PORAC 046        | PORAC 045        | POMVC 045        |                  | POMVC 046        |                  |                  |                  |                  | POPAC 048        |                  |                  |                  | PORAC 049        |                  | PORAC 048        |                  |                  |                  |
| Munition          | Bulk Container | Buik Container | Mine     | Ton Container | Ton Container | Ton Container | Ton Container | Agent Tank       | Bulk Container   | Bulk Container   | Buik Container   | M55 Rocket       | M55 Rocket       | Mine             | Mine             | Mine             | Mine             | Buik Containers  | Bulk Containers  | Projectile       | Projectile       | Projectile       | Projectile       | Rocket           | Rocket           | Rocket           | Rocket           | Rocket           | Ton Container    | Ton Container    |
| Area              | MPB            | MPB            | ECR      | BSA           | ECV           | ECV           | BSA           | TOX              | TOX              | TOX              | TOX              | TOX              | TOX              | MPB              | MPB              | MPB              | ECV              | ECV              | ECV              | ECR              | ECV              | ECR              | MPB              | MPB              | MPB              | ECR              | ECR              | MPB              | ECR              | ECR              | ECR              | ECR              | ECR              | BSA              | ECV              |
| Error Evente      | Ignition       | Ignition       | Ignition | l gn i t i on | Ignition      | Ignition      | Ignition      | Fire Suppression |
| No.               | -              | 2              | e        | 4             | Ð             | 9             | ~             | 80               | 8                | 01               | :                | 12               | 61               | 4                | 15               | 16               | 17               | 18               | 18               | 20               | 21               | 22               | 23               | 24               | 25               | 26               | 27               | 28               | 28               | 30               | 31               | 32               | 93               | 94               | 35               |

SCALSS TRAD

12122

 $\hat{}$ 

e.

1.1

ļ

| σ        |
|----------|
| e        |
| 3        |
| Ċ        |
|          |
|          |
| d        |
| 5        |
| -Ö       |
| 3        |
| -        |
| _        |
|          |
| 7        |
| 4        |
| <b>U</b> |
| 6-3      |
|          |
| H        |
| -        |
| <b>S</b> |
| H        |
|          |

|                    | 1      |                    |                    |                    |                    |                               |                    |                    |                    |                    |                    |                               |                    |                               |                    |                    |                    |                    |                    | _                  | _                  | _                  |                     |                     |                     | ~                   | _               | _               | -                     | -                     | _                    | -                    | ~                    | -                    | _                    | ~                    |
|--------------------|--------|--------------------|--------------------|--------------------|--------------------|-------------------------------|--------------------|--------------------|--------------------|--------------------|--------------------|-------------------------------|--------------------|-------------------------------|--------------------|--------------------|--------------------|--------------------|--------------------|--------------------|--------------------|--------------------|---------------------|---------------------|---------------------|---------------------|-----------------|-----------------|-----------------------|-----------------------|----------------------|----------------------|----------------------|----------------------|----------------------|----------------------|
| <u>لا۔</u><br>للنا |        | 2                  | 5                  | 0                  | 0                  | 2                             | 9                  | 0                  | 2                  | 2                  | 0                  | 2                             | 2                  | 2                             | 10                 | 2                  | 10                 | 0                  | 9                  | 2                  | 2                  | <u>0</u>           | AN                  | AN                  | AN                  | AN                  | 2               | õ               | 2                     | 2                     | 0                    | 10                   | 5                    | 0                    | 2                    | 01                   |
| biity              | 15 Min | 4E-3               | 4E-3               |                    |                    |                               |                    |                    |                    |                    |                    |                               |                    |                               |                    |                    |                    |                    |                    |                    |                    |                    |                     |                     |                     |                     |                 |                 |                       |                       |                      |                      |                      |                      |                      |                      |
| or Probe           | 10 Min | 1E2                | 1E-2               |                    |                    |                               |                    |                    |                    |                    |                    |                               |                    |                               |                    |                    |                    |                    |                    |                    | _                  |                    | -                   | _                   | _                   | _                   |                 |                 |                       |                       |                      |                      |                      |                      |                      |                      |
| ELL                | 5 Min  | 4E-2               | 4E-2               | 3.3E-4             | 3.3E-4             | 3.3E-4                        | 3.3E-4             | 3.3E~4             | 3.3E-4             | <b>3.3E-4</b>      | 3.35-4             | 3.36-4                        | <b>3.3E-4</b>      | 3.3E-4                        | 3.3E-4             | 3.36-4             | <b>3.3E-4</b>      | 3.3E-4             | 1.65E-5            | 1.65E-5            | 1.85E-5            | 1.65E-5            | epsilon             | epsilon             | olisde              | epsilon             | 16-3            | 1E-3            | 5E-2                  | 5E-2                  | 1.1E-4               | 1.16-4               | 1.15-4               | 1.16-4               | 1.1E-4               | 1E-2                 |
| Sequence           |        | ECV-2              | BSA-1              | ECV-13             | ECV-14             | ECV-17                        | ECV-16             | ECV-12             | ECV-15             | ECV-11             | ECV-10             | ECV-9                         | ECV-4              | ECV-5                         | ECV-3              | ECV-6              | ECV-7              | ECV-B              | ECV-1              | ECV-2              | BSA-2              | BSA-1              | DU1-001             | DU1-003             | DU1-002             | DU1-004             | TOX-8           | TOX-9           | TOX-5                 | TOX-6                 | MP3-002              | MP3-004              | MP4-001              | MP3-001              | MP3-003              | BSA-2                |
| Scenario           |        | POTAF 043          | POKAF 053          | POPAC 045          | POPAC 046          | POPAC 046                     | POPAC 045          | POPAC 044          | POPAC 044          | POPAC 046          | POPAC 045          | POPAC 044                     | PORAC 045          | PORAC 046                     | PORAC 044          | POMVC 044          | POMVC 045          | POMVC 046          |                    | POTAF 043          |                    | POKAF 053          |                     |                     |                     |                     |                 |                 |                       |                       |                      |                      |                      |                      |                      |                      |
| Munition           |        | Ton Container      | Ton Container      | 105-mm Projectile  | 105-mm Projectile  | 105-mm Projectile             | 105-mm Projectile  | 105-mm Projectile  | 105-mm Projectile  | 8º Projectile      | 8º Projectile      | 8° Projectile                 | M55 Rocket         | M55 Rocket                    | M55 Rocket         | Mine               | Mine               | Mine               | Ton Container      | Ton Container      | Ton Container      | Ton Container      | Mine                | Mine                | Mine                | Mine                | Agent Tank      | Agent Tank      | Agent Tank            | Agent Tank            | Ton Container        |
| Area               |        | ECV                | BSA                | ECV                | ECV                | ECV                           | ECV                | ECV                | ECV                | ECV                | ECV                | ECV                           | ECV                | ECV                           | ECV                | ECV                | ECV                | ECV                | ECV                | ECV                | BSA                | BSA                | MDB                 | MDB                 | MOB                 | MDB                 | TOX             | тох             | nToX                  | nToX                  | BOM                  | MOB                  | MOB                  | MOB                  | MOB                  | BSA                  |
| . Error Events     |        | S Fire Suppression | 7 Fire Suppression | 3 Conveyor Loading | 9 Conveyor Loading | <pre>O Conveyor Loading</pre> | 1 Conveyor Loading | 2 Conveyor Loading | 3 Conveyor Loading | 4 Conveyor Loading | 5 Conveyor Loading | <pre>S Conveyor Loading</pre> | 7 Conveyor Loading | <pre>3 Conveyor Loading</pre> | 9 Conveyor Loading | 3 Conveyor Loading | 1 Conveyor Loading | 2 Conveyor Loading | 3 Conveyor Loading | 4 Conveyor Loading | 5 Conveyor Loading | 3 Conveyor Loading | 7 Munition Counting | 3 Munition Counting | 9 Munition Counting | O Munition Counting | 1 Tank Overfill | 2 Tank Overfill | 3 Sump Pump Operation | 4 Sump Pump Operation | 5 Undrained Munition | 3 Undrained Munition | 7 Undrained Munition | 3 Undrained Munition | 9 Undrained Munition | 0 Undrained Munition |
| Ŷ                  |        | ĕ                  | e                  | 3                  | ē                  | ¥                             | 4                  | 4                  | 4                  | Ť                  | 4                  | 4                             | ۲                  | 4                             | 4                  | 50                 | 6                  | 9                  | 50                 | ò                  | 5                  | 5                  | 9                   | ŝ                   | 5                   | 9                   | 6               | 9               | 0                     | 6                     | 96                   | 9                    | 9                    | 8                    | 9                    | ž                    |

9-40


DOG NOW ST





ዄኯጟቔጞኯፚቘቘዸኯዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀ

| Area Munition Scenario Sequence Error<br>m TOX Agent Tank TOX-8 1E-4 |
|----------------------------------------------------------------------|
| m IUX Agent Tank TOX-8 1E-4<br>m TOX Agent Tank TOX-2 1E-4           |
| m TOX Agent Tank TCX-3 1E-4                                          |
| m TOX Agent Tank TOX-5 1E-4                                          |
| m MPB Bulk Container MPB-3B 1E-4                                     |
| m MPB Buik Container MPB-2B 1E-4                                     |
| m ECR MINe POMVC 047 ECR-2DM 1E-4                                    |
| m ECR MINe ECR-3DM 1E-4                                              |
| m ECR Mine ECR-1DM 1E-4                                              |
| m ECR Mine ECR-4DM 1E-4                                              |
| m MPB Bulk Containers MPB-5B 1E-4                                    |
| m MPB Projectile MPB-2DP 1E-                                         |
| m ECR Projectile ECR-1DP 1E-                                         |
| m ECR Projectile POPAC 047 ECR-2DP 1E-4                              |
| m MPB Projectije MPB-1DP 1E-4                                        |
| m ECR Projectile POPAC 048 ECR-3DP 1E-4                              |
| m MPB Projectile MPB-3DP 1E-4                                        |
| m ECR Projectile ECR-4DP 1E-4                                        |
| m ECR Rocket ECR-2DR 1E-                                             |
| m ECR Rocket PORAC 047 ECR-5DR 1E                                    |
| m ECR Rocket PORAC 048 ECR-6DR 1E                                    |
| m ECR Rocket ECR-70R 1E-                                             |
| m ECR Rocket ECR-1DR 1E-                                             |
| m ECR Rocket ECR-3DR 1E-                                             |
| m BSA Ton Container BSA-2 1E-                                        |
| m ECV Ton Container ECV-1 1E-4                                       |
| onFR Ton Container 727 1E-2                                          |
| IA 77 1E-                                                            |

1222

E STATE DE L'ENDE

9-41

والمراجع والمراجع

representative of this situation. Given that this failure of administrative control affects their own safety (and assuming that 30% of all operators smoke), it is estimated that  $1 \times 10^{-3} \times 3.3 \times 10^{-1} = 3.3 \times 10^{-4}$  is the probability of smoking initiating a fire.

EXECUTE ENDING

1.2.2.2.2.3

Operators or maintainers could cause sparks any time they handle a weapon or use metal tools, which they are likely to do in any area of the plant. Except for the UPA, some sort of upset would probably have to have occurred for them to be handling munitions or using tools. The likelihood of their causing sparks in such a case is the same as that of their dropping a munition during handling. The estimated probability of dropping a single munition when it is hand-carried by a two-man crew dressed in DPE was estimated as  $6 \times 10^{-4}$  in the HRA for handling scenarios as described in Chapter 8.

9.2.2.4.2. <u>Fire Suppression</u>. When a fire occurs in the UPA, the control room, the UPS, the IEP, the communications room, or the TOX, an automatic fire-suppression system should come on. If the automatic system fails to start, the operators can initiate it from the control room. He does this in response to an annunciator alarming on the panel dedicated to fire alarms (an annunciator there always indicates fire somewhere in the plant). There are probably several other annunciators alarming at the same time; we assumed six for this analysis. Item 6 from Table 20-23 (Ref. 9-26), 5 x 10<sup>-3</sup>, was used to estimate the likelihood of the operator's failing to initiate the failed automatic fire-suppression system.

If the fire-suppression system still does not respond, or if the fire is in an area of the plant that has no automatic system, the next recourse for extinguishing the fire is to isolate the room where it is burning. The operators can do this by closing the exhaust dampers for the room in question. Again, they can do this from the control room. For this analysis, we assumed that the operators' training would emphasize room isolation as the best method of fire-fighting outside of the

| SIFIE | The<br>Ca<br>D Sap | DISPO<br>A N I<br>EO-CDI | DSAL DI<br>BARSELI<br>E-IS-8 | F CHEN<br>L ET A<br>7006 D | ISPUS<br>L. (U)<br>AAA15- | GA TEI<br>3 87 GI<br>-85-D-1 | CHNOLO<br>R-C-18<br>9922 | GIES I<br>563 | NC SAI | N DIEG<br>15/6. 3 | 0 87<br>NL | 13 |
|-------|--------------------|--------------------------|------------------------------|----------------------------|---------------------------|------------------------------|--------------------------|---------------|--------|-------------------|------------|----|
|       |                    |                          |                              |                            |                           |                              |                          |               |        |                   |            |    |
|       |                    |                          |                              |                            |                           |                              |                          |               |        |                   |            |    |
|       |                    |                          |                              |                            |                           |                              |                          |               |        |                   |            |    |
|       |                    |                          |                              |                            |                           |                              |                          |               |        |                   |            |    |
|       |                    |                          |                              |                            |                           |                              |                          |               |        |                   |            |    |
|       |                    |                          |                              |                            |                           |                              |                          |               |        |                   |            |    |
|       |                    |                          |                              |                            |                           |                              |                          |               |        |                   |            |    |
|       |                    |                          |                              |                            |                           |                              |                          |               |        |                   |            |    |



use of automatic systems. Therefore, the problem is one of the operators' remembering that there is a viable solution to a fire.

The nominal diagnosis model from NUREG/CR-1278 (Ref. 9-26) was used as the basis to estimate the likelihood that the operators won't select room isolation. Since the "diagnosis" task here is fairly straightforward and since we have assumed that training will emphasize isolation as the action of choice, we used the lower bound of that curve to represent the case in which the fire is in a room without any automatic firesuppression system.

If the fire does involve one of the rooms mentioned above, the operators will likely spend at least 5 min trying to start the failed automatic system. Since the diagnosis curve is time-based, 5 min of decision time is lost early in the accident. The modified curve accounting for this, along with the curve used for the rooms without fire-suppression systems, is shown in Fig. 9-1. The results of the analysis will show that the delay in diagnosing the need for isolation is more than compensated for by having an automatic system.

If the automatic fire-suppression system (if any) does not function and if room isolation is not achieved (or if it is not achieved in time), the operators' last resort is to enter the area with the fire and fight it with the hand-held fire extinguishers that are located throughout the plant. If it is an agent fire, if DPE protective clothing is necessary to enter the area, or if burstered munitions are in the area, it is assumed that the operators will not elect to try this option; they will not fight the fire at the site in any of these cases. If the fire is in an area they can enter wearing street clothes and masks and if burstered munitions are not present, it is estimated that there is a 5 x  $10^{-2}$  probability that they will fail to try at-site fire fighting. This estimate is based on scientific judgment.



| Time        | Pr (fail to isolate by X mins) |                |  |  |
|-------------|--------------------------------|----------------|--|--|
|             | With System                    | Without System |  |  |
| by 5 mins.  | 1.0                            | 4E-2           |  |  |
| by 10 mins. | 4E-2                           | 1E-2           |  |  |
| by 15 mins. | 1E-2                           | 4E-3           |  |  |

Fig. 9-1. Probability of failure to isolate room by X min



222622

For a complicated scenario such as this, THERP suggests the use of an HRA event tree. The HRA event tree for this fire-suppression model is shown in Fig. 9-2, and the results of quantifying it are shown in Table 9-12. 9.2.2.4.3. <u>Conveyor Loading</u>. In the UPA, operators in Level C protective clothing (masks worn) unload munitions and bulk containers from pallets and/or trucks and place them onto the conveyor system that then carries the munitions and containers through the process areas. Smaller munitions such as mines, projectiles, cartridges, and M55 rockets are lifted by hand (sometimes by two operators) and placed onto the conveyors. There are metering devices that ensure proper alignment of the rockets on the conveyor and allow only a single munition at a time to enter the ECV. When hand-loading the projectiles, operators could drop the munition in the UPA. The estimated probability of dropping a single munition when it is hand-carried by a two-man crew dressed in Level C protective clothing has been estimated as  $3 \times 10^{-4}$  in the HRA for handling scenarios.

The conveyor itself has 1/2-in. high guard rails that prevent a munition's falling off the conveyor. Even if the operators load the munition crookedly, the guard rails and the metering device will orient it properly as it passes into the ECV. The only other possible error involves their loading the munition backwards. Since we assume that the operators will usually pick up the same end of each munition (at least for a time), the likelihood of their standing in the wrong position--a necessary condition for loading the munitions backwards--is very low. It has been estimated to be an order of magnitude lower than the drop probability, or  $3 \times 10^{-5}$ . The likelihood that a munition is loaded improperly by the operators such that it could drop during loading or fall off the conveyor as a result of improper loading is the sum of these two error probabilities, or  $3 \times 10^{-4} + 3 \times 10^{-5} = 3.3 \times 10^{-4}$ .



130

12222

Fig. 9-2. HRA event tree of fire suppression model



|                                         | Probability That Operators Fail to<br>Suppress the Fire (DPE required or<br>Burstered Munitions Present) |                                       |                                    |                                       |  |  |  |
|-----------------------------------------|----------------------------------------------------------------------------------------------------------|---------------------------------------|------------------------------------|---------------------------------------|--|--|--|
| Time After<br>Onset of<br>Fire<br>(min) | Agent                                                                                                    | : Fire                                | No Agent Fire                      |                                       |  |  |  |
|                                         | Automatic<br>Suppression<br>System                                                                       | No Automatic<br>Suppression<br>System | Automatic<br>Suppression<br>System | No Automatic<br>Suppression<br>System |  |  |  |
| 5                                       | $5.0 \times 10^{-3}$                                                                                     | $4.0 \times 10^{-2}$                  | $2.5 \times 10^{-4}$               | $2.0 \times 10^{-3}$                  |  |  |  |
| 10                                      | $2.0 \times 10^{-4}$                                                                                     | $1.0 \times 10^{-2}$                  | $1.0 \times 10^{-5}$               | 5.0 x $10^{-4}$                       |  |  |  |
| 15                                      | 5.0 x 10 <sup>-6</sup>                                                                                   | $4.0 \times 10^{-3}$                  | 2.5 x 10 <sup>-6</sup>             | $2.0 \times 10^{-4}$                  |  |  |  |

# TABLE 9-12 THERP QUANTIFICATION OF FIRE-SUPPRESSION MODEL



Ton containers, spray tanks, and bombs are loaded onto the conveyor using a forklift lifting beam. The estimated probability of dropping a single bulk item when a two-man crew in Level C protective clothing use a forklift with a lifting beam was estimated as  $1.5 \times 10^{-5}$  in the HRA for handling scenarios. The only other credible errors are those of loading the containers crookedly (a no-cost error given the guard rails) or backwards. Backwards loading is most likely with a ton container since its exterior profile shows no obvious fore or aft indication (except for location of the plugs). Again, the operators have separate, assigned duties during loading. Since the ton containers should be guided by one operator while the other operator drives the forklift, the likelihood of its being improperly loaded is estimated to be an order of magnitude lower than the drop probability, or 1.5 x  $10^{-6}$ . The likelihood that any kind of bulk container is loaded improperly by the operators is the sum of these two error probabilities, or  $1 \times 10^{-5} + 1 \times 10^{-5}$  $10^{-6} = 1.65 \times 10^{-5}$ .

9.2.2.4.4. Munition Counting. When munitions are unloaded in the UPA, the packing material is sent to the Dunnage Incinerator (DUN). If a munition is left in the packing material (if it is not unpacked), it will be sent as-is to the DUN, also. The operators must keep track of the pallets and barrels passing through the UPA to ensure that they are emptied before being disposed of. All pallets are unloaded completely before beginning the next pallet-unloading operation. In other words, two pallets are never partially unloaded because of their being unpacked simultaneously. Since the pallet layers must be removed to access munitions on the next layer down, it is not likely that operators will miss a palletized, unpacked munition. Also, the pallet itself does not obscure the individual munitions from view even before it has been removed. The likelihood that an operator will fail to unpack a pallet completely and send the unpacked munition to the DUN along with the dismantled pallet is negligible.



Mines are packed three to a barrel; their fuzes are packed separately but in the same barrel. There are six barrels on a pallet. Once the pallet has been dismantled, the barrels themselves must be unpacked. The barrels are inverted inside a glove box one at a time, then lifted off of the mines and the packing material. Once the barrel has been emptied, it is used to hold the discarded packing material for the trip to the DUN. For a mine to enter the DUN along with the packing material, it would have to be placed in the barrel instead of on the conveyor. Munition accountability with respect to the number processed will be checked before the dunnage is disposed of; this provides a measure of recovery should this highly unlikely event occur. The probability of a mine being fed to the DUN along with its packing material is assumed to be negligible.

9.2.2.4.5. <u>Tank Overfill</u>. When draining a bulk-agent container, the agent is transferred to an agent tank in the Toxic Cubicle (TOX). When the agent tank's capacity is reached, the process-control system should automatically halt the transfer. If the high-level sensor on the tank fails or if some other failure occurs such that the transfer is not halted, the operator who initiated the transfer can halt it manually before the tank spills over.

It should be stated in the plant's administrative-control policies (and even in the process-control logic) that a bulk container should not be drained unless its entire contents can be accepted by a single agent tank. Of the two agent tanks in the TOX, the operators could have selected (and the process-control logic could have defaulted to allow) the wrong tank to receive the agent from a bulk container. If this wrong tank has insufficient capacity to accommodate the contents of the container, TOX tank level will approach and then exceed its maximum sometime during transfer. The probability of a selection error when dealing with displays with clearly delineated mimic lines is estimated to be 5 x  $10^{-4}$  (Ref. 9-26, Table 20-9, item 1). Since this error has to occur in conjunction with a process-control failure (the probability of which is estimated to be 1 x  $10^{-3}$ ), the likelihood that the wrong tank will be selected to receive the agent is 5 x  $10^{-7}$ .

Assuming that agent is being transferred to a too-full tank, a sensor should halt the transfer at the tank's high-level setpoint. If the sensor fails, the operator (who should be monitoring the transfer intermittently) might notice the tank's high level and halt the transfer manually before a spill occurs. A typical transfer operation takes about 30 min; it is not assumed that the operator will watch the levels in the bulk container and the TOX tank for that whole period (although it is assumed that he will monitor both levels at some point since he initiated the transfer). Rather, it is assumed that he will initiate the transfer and then leave to complete other tasks while it is going on; it is also assumed that he will return to view the monitor screen periodically during the transfer to check its progress.

The estimated probability of his not noticing that the level of the TOX tank is dangerously high during the transfer operation is based on the estimated probability of an error made in reading quantitative information from an analog meter,  $3 \times 10^{-3}$  (Ref. 9-26, Table 20-10, item 1). The lower bound of  $1 \times 10^{-3}$  is used for this case to reflect better-quality reading characteristics associated with CRT analog displays. If the operator returns several times during the transfer to check the level of the TOX tank, the memory of his first reading will influence his perception of subsequent readings, so they were considered a perceptual unit. Both error probabilities are summed to estimate the total human-error contribution to this scenario. This means that  $5 \times 10^{-7} + 1 \times 10^{-3} = 1 \times 10^{-3}$ .

9.2.2.4.6. <u>Sump Pump Operation</u>. When there has been a spill in the TOX, the sump pump provides some level of mitigation. If the sump pump fails to operate following a spill, there is still a chance that the operators could start it manually from the control room. Since the spill in the TOX has already occurred when the sump pump fails, there

are probably several annunciators alarming when the sump pump alarm goes off. Assuming there are ten annunciators competing for the operator's attention,  $5 \times 10^{-2}$  (Ref. 9-26, Table 20-23, item 10) is the probability that he will fail to respond to the sump pump alarm.

9.2.2.4.7. <u>Undrained Munition</u>. There is some chance that an undrained ton container will reach the MPF, where it presents a considerable hazard. There are two points at which the operator might notice this and intervene to prevent its introduction into the MPF. The first of these is in the MPB as the container is being drained. The operator should have initiated the drain operation and should be watching for some indication that it is, in fact, taking place.

The second potential for operator intervention comes as the container leaves the BSA and is weighed before being transferred to the MPF. The operator should check the reading at the weigh station before allowing the container to continue to the MPF. The likelihood that the operator does not watch an operation that he is supposed to monitor on the CRT screen and/or the CCTV is assumed to be equivalent to his not following/using a set of written procedures. The error probability for his failing to monitor the screen(s) is  $1 \times 10^{-2}$ , taken from Table 20-6, item 3. This is used for his failing to monitor the drain operation in the MDB before the container is transported to the BSA and also for his failing to check the weight of the container as it leaves the BSA.

If the operator checks the container's weight, there is a chance that he will misread the weight on the CRT display. The probability of a misreading error when using a CART analog display is  $1 \times 10^{-3}$ (Ref. 9-26, Table 20-10, item 1, lower bound). The likelihood that the operator in neither case acts to prevent an undrained container's entering the MPF is calculated as  $(1 \times 10^{-2} \times 1 \times 10^{-2}) + 1 \times 10^{-3} =$  $1.1 \times 10^{-3}$ . 9.2.2.4.8. <u>Ventilation System</u>. Any time there is a ventilation system failure, there is some chance that the operators could effect recovery. For areas outside the furnace rooms, the operators should shut off the air supply fans within an hour of ventilation system failure. There is no direct indication that this is the needed action, so some diagnosis is involved. Using a standard diagnosis curve, the likelihood of their having failed to shut off the air supply fans by the end of an hour is estimated as  $1 \times 10^{-4}$  using Fig. 12-4 from NUREG/CR-1278 (Ref. 9-26).

9.2.2.4.9. <u>Furnace Ventilation</u>. For ventilation system failures involving the furnace rooms, the scenario is somewhat different. One train should be in service at all times. If that ventilation train fails, the operators can valve in an alternate train. This involves closing the dampers to the failed system, opening the dampers and headers to the alternate system, and starting up the alternate system. The primary ventilation system is assumed to fail at least 10 min following an initiator involving furnace shutdown; once it has failed, the operator has about 10 more minutes to complete the transfer to avoid serious consequences.

Since the ventilation system failure occurs 10 min after the furnace shutdown, the two failures do not occur "closely in time". Moreover, different operators are dedicated to monitoring the furnace and the ventilation systems. Therefore, the first-event diagnosis model (Ref. 9-26, Table 20-3, item 1) was chosen to model this event. Since the furnace shutdown is likely to lead to ventilation system failure, the operators may expect to have to deal with that problem. Because of their expectation, the lower bound of the nominal diagnosis model value, or 1 x  $10^{-2}$ , was used.

9.2.2.4.10. <u>Air Compressors</u>. Some sequences assumed a reduced capacity of the primary plant-air and instrument-air compressor because of a downstream blockage. Since the blockage does not involve the



compressor itself, no trouble alarm associated with it will sound. Instead, a low-pressure alarm for downstream will sound at some time, after which there is a 15-min period before reserve-air inventory is depleted.

The non-occurring trouble alarm would have been sufficient to cause automatic transfer to the standby compressor; since it did not alarm, the transfer must be initiated by an operator sometime in that 15-min interval. This depends on his noticing the low-pressure alarm since an operator's recognition of an annunciator means that he will respond to that annunciator. It is assumed that there would have been no other shutdowns (nor their associated alarms) for at least 15 min before the low-pressure alarm sounds, so the error estimate listed as item 1 in Table 20-23 (Ref. 9-26) was used.





## 9.3. REFERENCES

- 9-1. U.S. Army Office of the Program Executive Officer-Program Manager for Chemical Munitions, "Transportation of Chemical Agents and Munitions: A Concept Plan," Final Report by MITRE Corp., June 30, 1987.
- 9-2. Clarke, R. K., et al., "Severities of Transportation Accidents," SLA-74-001, Sandia National Laboratories, July 1976.
- 9-3. Federal Railroad Administration, "Accident/Incident Bulletin No. 151, Calendar Year 1982," Department of Transportation, June 1983.
- 9-4. Corps of Engineers (HND) R/M Data Base (Rep. No. 16), see IEEE Std. 500-1984, pages 1252 through 1254 and pages 1256 through 1259 (seven failure rate estimates).
- 9-5. Rossi, M. J., "Nonelectronic Parts Reliability Data," Report NPRD-3, prepared for Rome Air Development Center by IIT Research Institute, 1985.
- 9-6. "Nuclear Plant Reliability Data System NPRDS A02 and A03 Reports - Annual Reports of Cumulative System and Component Reliability for Period from July 1, 1974 through December 31, 1982," Reports NPRD A02/A03, prepared for the Institute of Nuclear Power Operations by Southwest Research Institute, October 1983.
- 9-7. "SRS Data Bank," National Centre of System Reliability, UKAEA, Culchetb, Warrington, WA34NE.
- 9-8. "Risk Analysis of Six Potentially Hazardous Industrial Objects in the Rijnmond Area, A Pilot Study," a Report to the Rijnmond Public Authority, D. Reidel Publishing Company, Holland, 1982.

SANTAN MANANANANAN'I NANAN'I NANANAN'I NANANAN'I NANANAN'I NANANAN'I NANANAN'I NANANAN'I NANANAN'I NANANAN'I N

- 9-9. "Reactor Safety Study, An Assessment of Accident Risks in U.S. Commercial Nuclear Power Plants," Report No. WASH-1400, U.S. Nuclear Regulatory Commission, October 1975.
- 9-10. IEEE Guide to the Collection and Presentation of Electrical, Electronic, Sensing Equipment, and Mechanical Equipment Reliability Data for Nuclear-Power Generating Station, IEEE Std. 500-1984, Wiley-Interscience, 1983.

9-11. Meachum, T. R. and C. L. Atwood, "Common Cause Fault Rates for Instrumentation and Control Assemblies: Estimates Based on Licensee Event Reports at U.S. Commercial Nuclear Power Plants, 1976-1981," Report No. NUREG/CR-3289 (EGG-2258), prepared for the U.S. Nuclear Regulatory Commission by EG&G Idaho, Inc., May 1983. ZZZZŚ WENNANY PRZZZZZ W WENNANY W

- 9-12. Humphreys, M., "Application of Reliability Techniques to Instrumentation Systems," U.K. Atomic Energy Authority Report No. SRS/GR/55, April 1982.
- 9-13. Wright, R.I., "Microprocessor Hardware Reliability," U.K. Atomic Energy Authority, Report No. SRS/GR/60, October 1982.
- 9-14. Oswald, A. J., et al., "Generic Data Base for Data and Models Chapter of the National Reliability Evaluation Program (NREP)," Report No. EGG-EA-5887, prepared for the U.S. Nuclear Regulatory Commission by EG&G Idaho, Inc., June 1982.
- 9-15. Kolb, G. J., et al., "Interim Reliability Evaluation Program: Analyses of the Arkansas Nuclear One - Unit 1 Nuclear Power Plant," NUREG/CR-2787, U.S. Nuclear Regulatory Commission, June 1982.
- 9-16. Montague, D. F., and H. M. Paula, "A Survey of B-Factor and C-Factor Applications," JBFA-LR-104-84, JBF Associates, Inc., September 1984.
- 9-17. NUREG/CR-4780 (to be published).
- 9-18. Paula, H. M. and D. J. Campbell, "Analyses of Dependent Failure Events and Failure Events Caused by Harsh Environmental Conditions," Report No. JBFA-LR-111-85, prepared for the U.S. Nuclear Regulatory Commission by JBF Associates Inc., August 1985.
- 9-19. Baranowski, P. N., et al., "A Probabilistic Safety Analysis of DC Power Supply Requirements for Nuclear Power Plants, U.S. Nuclear Regulatory Commission Report No. NUREG-0666, April 1981.

- 9-20. Atwood, C. L., "Common Cause Fault Rates for Pumps: Estimates Based on Licensee Event Reports at U.S. Commercial Nuclear Power Plants, January 1, 1972 through September 30, 1980, Report No. NUREG/CR-2098 (EGG-EA-5289) prepared for the U.S. Nuclear Regulatory Commission by EG&G Idaho, Inc., February 1983.
- 9-21. Stevenson, J. A. and C. L. Atwood, "Common Cause Fault Rates for Valves: Estimates Based on Licensee Event Reports at U.S. Commercial Nuclear Power Plants, 1976-1980," Report No. NUREG/CR-2770 (EGG-EA-5485), prepared for the U.S. Nuclear Regulatory Commission by EG&G Idaho, Inc., February 1983.
- 9-22. Battle, R. E., Collection and Evaluation of Complete and Partial Losses of Off-Site Power at Nuclear Power Plants," Report NUREG/CR-3992 (ORNL/TM-9384), prepared for the U.S. Nuclear Regulatory Commission by Oak Ridge National Laboratory, February 1985.
- 9-23. "Plant Risk Status Information Management Systems (PRISIM) Version 2.0 User's Guide," Report No. JBFA-108-86, prepared for the U.S. Nuclear Regulatory Commission by JBF Associates, Inc., October 1986.
- 9-24. Carlson, D. D., et al., "Interim Reliability Evaluation Program Procedures Guide," Report No. NUREG/CR-2728 (SAND 82-1100), prepared for the U.S. Nuclear Regulatory Commission by Sandia National Laboratories, January 1983.
- 9-25. "Military Handbook: Reliability Prediction of Electronic Equipment," Rome Air Development Center, Griffith AFB, NY, MIL-HDBK-217C, April 1979.
- 9-26. "Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Applications," NUREG/CR-1278, U.S. Nuclear Regulatory Commission, August 1983.
- 9-27. "Probabilities of Selected Hazards in Disposition of M55 Rockets," M55-CS-02, USATHAMA, September 22, 1985.
- 9-28. Fischer, L. E., et al., "Shipping Container Response to Severe Highway and Railway Accident Conditions," NUREG/CR-4829, Vols. 1 and 2, Lawrence Livermore National Laboratory, February 1987.

## 10. AGENT RELEASE CHARACTERIZATION

Section 10.1 describes the approach used in this study for analyzing the agent release for the various accident conditions. Application of the approach to the accident sequences analyzed in the collocation disposal phases is discussed in Section 10.2.

The consequences of an agent release event are strongly dependent on agent type, amount of agent release, and the mode and duration of the release. Agent dispersion and subsequent effects will be calculated in a separate study using a computer program called D2PC that embodies an analytical model for calculating agent dispersion under different meteorological conditions. Feedback from these consequence calculations helped to guide the release characterization.

#### 10.1. RELEASE ANALYSIS APPROACH AND BASES

10.1.1. Approach

The approach formulaton was aided by a systematic review of the mechanisms involved in expelling agent from its normal confinement. The first result of the systematic review was to divide the accident sequences into two groups: (1) those that occur while the agent is still present in the munitions and (2) those that occur after the agent has been separated from the munition. The first group is associated with the activities of storage, handling, and transportation, while the latter group is associated with the activities of plant operations. For the latter group, the analyses performed by Arthur D. Little for the M55 rockets (Refs. 10-1 through 10-5) were partially applicable, and similar assumptions as appropriate were made for this analysis. Additional calculations were performed in this study to determine the quantity of agent released to the environment for plant operation accidents involving munitions other than the M55 rockets.

For the accident scenarios that involve agent still confined in the munition, the agent release is dependent on the munition's mechanical and thermal failure thresholds, and the behavior of the explosives and propellants during the accident scenarios. These are discussed in the following sections. Once it was determined that the agent could be released from its normal confinement, calculations were performed to determine the amount of agent released and the possible paths by which the agent could enter the atmosphere.

# 10.1.2. Mechanical Failure Release

Munition failures result when sufficient forces are generated during accidents. A discussion of the munition failure thresholds is given in Appendix F. The failure thresholds of interest are:

- Mechanical failure of the agent containment due to impact, crush or puncture.
- 2. Detonations initiated by impact or fire.
- 3. Thermally induced hydraulic rupture of the agent containment.

10.1.2.1. <u>Impact Failure</u>. The threshold for impact failure is given in terms of velocity of impact against a nonyielding object, or the equivalent drop height. When the impact failure threshold is reached, it is assumed that the onset of failure begins. In the case of an accident involving more than one munition, e.g., a pallet drop or a truck collision, every munition does not experience the effect of impacting a nonyielding surface. At the threshold point, it is assumed that at least one munition has experienced failure. It was further assumed that the number of munitions that experience failure is a function of the kinetic energy involved in the accident. For munitions in a transportation



package, the failure threshold for both the package and the munition must be exceeded in order to cause an agent release.

The impact velocity required to initiate failure varies from 35 mph for rockets (drop height of 40 ft) to 50 mph for projectiles (drop height of 120 ft). The expected impact velocity (or drop height) for some accidents is:

| Accident Type               | Impact Velocity of<br>Drop Height                                   |  |  |
|-----------------------------|---------------------------------------------------------------------|--|--|
| Pallet drop during handling | 6 ft                                                                |  |  |
| Forklift collision          | 5 mph                                                               |  |  |
| Truck accident onsite       | 10 to 25 mph (administrative<br>control is assumed to be<br>10 mph) |  |  |
| Train accident offiste      | 50 mph                                                              |  |  |
| Aircraft crash              | >200 mph                                                            |  |  |

In view of the above, failure due to impact is not considered to be a significant contribution for handling accidents and onsite truck transportation accidents, i.e., other failure mechanisms dominate.

10.1.2.2. <u>Crush Failure</u>. Crush forces are static forces completely independent of velocity. Crush forces may arise from a vehicle overturn or from a building collapse due to an earthquake.

Crush thresholds are defined for a single munition for a pallet of munitions and for the transportation package when transportion is involved. When the crush threshold for pallets is exceeded, it was conservatively assumed that all munitions in the pallet will fail. A linear relationship for the number of units that would fail due to crush was assumed as follows:

$$n = \frac{F}{F_{o}} , \qquad (10-1)$$

where F = crush force available in the accident,

 $F_0$  = crush force threshold for the palletized munition.

At n = 1, all the munitions in one pallet have failed. The available force in an accident can be the weight of a vehicle, the weight of a building collapse, or the weight of any large object that can fall on the munitions. For those accidents involving a transportation package, the crush force available must exceed the threshold for failing both the package and the munition.

The accident scenarios that are capable of generating forces sufficiently high to produce crush involve transportation and storage where many pallets may be involved in the accident. Thus, it is possible that more than one pallet can fail. For example, the crush threshold for a rocket pallet containing 15 rockets is 43,400 lb. If the weight of an object is 100,000 lb, Eq. 10-1 predicts a failure quantity of 2.3. This corresponds to 2.3 pallets, or about 34 rockets being crushed. If the available crush force is less than the failure threshold for a single munition, then naturally, no munitions fail.

Equation 10-1 is conservative because it assumes that the total available load arising from an accident is concentrated in the most efficient way to crush the munitions. If the load was uniformly distributed over many pallets, fewer or no failures would occur.

10.1.2.3. <u>Puncture Failure</u>. The puncture threshold is defined in terms of the ratio of velocity to radius of curvature assuming the munition (or pallet) impacts an unyielding slender object or probe. Generally, the failure threshold for puncture is the lowest of the three mechanical

If the spill occurs outdoors, during handling or transport, the release analysis ends with the determination of the type and mass of liquid agent spilled and type of surface where the spill occurs. This information is sufficient input for calculation of atmospheric dispersion by the D2PC computer program. All liquid spills during handling or ground transport are assumed to occur on a hard, flat impervious surface such as level concrete or asphalt. The evaporation of the spill is calfailure thresholds. The number of failures that can occur in an accident is dependent on the number of probes present. If the puncture failure threshold is exceeded, it is assumed that one probe will fail one munition.

10.1.2.4. Liquid Spills and Evaporation. Once mechanical failure occurs, the munition agent inventory may be able to spill out on the ground or water. For fork tine punctures, the puncture is assumed to consist of a 3-in. diameter hole just below the munition centerline. The amount and time of spill is calculated to be that which can drain by gravity out of the hole. Impact, crush and probe punctures are assumed to result in the spill of the entire munitions inventory.

culated by the D2PC program by calculating the maximum puddle area and the corresponding evaporate rate. If the spill occurs to the surface of water, it is expected, based on agent density and solubility characteristics, to mix well with water or sink (depending on agent type). However, for conservatism, 5% is assumed to remain on the surface and be available for evaporation. If the release occurs underwater (e.g., after ship sinking), no agent becomes available at the surface.

If the spill occurs indoors, the release analysis in this report extends to the time dependent rates of evaporation. In general, the D2PC program was applied to calculate the evaporation rate based on the type and mass of agent spill and considering any confinement of the

liquid puddle or pool. The D2PC general equation for evaporation of a spill over a floor area corresponding to a liquid pool depth of 1/32 in. relates the time t to evaporate the entire spill inventory M (pounds) in terms of a power function of M and two coefficients a and b. The equation is

$$t = aM^{b} , \qquad (10-2)$$

where

t = time in thousands of minutes,

a, b = constant for agent GB (a = 0.79, b = 0.253),

a, b = functions of M for agents H and VX.

The area  $(ft^2)$  corresponding to the spill M (lb) and pool thickness 1/32 in. is 5.91 times M. For restricted pool areas, the equation must be modified. This equation and coefficients a and b are based on data from the Army derived from the computer program D2PC output.

For a given accident sequence the spill will generally not evaporate to completion because human intervention will mitigate the spill by covering it with foam or some other means. In such a case, an evaporation rate is calculated and applied until the time estimated for mitigation or cleanup of the spill.

From Eq. 10-2, the hourly evaporation rate is

$$m_{ev} = \frac{1}{a} M^{1-b} \frac{60 \min}{10^3 \min} , \qquad (10-3)$$

where  $m_{ev}$  has units of 1b/h. This equation applies whenever the 1/32-in. deep spill pool area, which from the agent density is about 6 ft<sup>2</sup> for each 1b of spill, is smaller than the actual confined pool

area (floor or sump). Some buildings contain floors which slope to sumps, as in the following:

PERSONAL MANAGEMENT

| Building Area | Sump Size<br>(ft) |  |  |  |  |
|---------------|-------------------|--|--|--|--|
| UPA           | 2 x 2 x 2         |  |  |  |  |
| TOX cubicle   | 4 x 5 x 3.5       |  |  |  |  |
| MHI           | 2 x 3 x 4         |  |  |  |  |
| Warehouse     | None              |  |  |  |  |
| Storage igloo | None              |  |  |  |  |

Where a sump is present, the following procedure is used to calculate evaporation. Initially, the spill is assumed to wet the entire sloped floor area. Thus, Eq. 10-3 issued for a 10 min time period without modification for pool area, unless the 1/32-in. deep pool area is larger than the actual floor area. Modification consists of limiting M in Eq. 10-3 to the mass of a 1/32-in. layer of agent over the actual floor area. After 10 min, the evaporation rate is assumed to be limited by the sump horizontal cross sectional area until the assumed mitigation/cleanup time when it drops to zero. Such limitation amounts to modifying M in Eq. 10-3 to the mass of a 1/32-in. layer in the sump.

A special case is the spill of a ton container in the MDB where the UPA sumps are too small to hold the entire inventory. In this case the overflow area is calculated based on the volume of agent in a TC and the floor slope (1/4 in. rise per linear foot).

#### 10.1.3. Detonations

The burstered munitions incorporate proven design features to preclude accidental detonation during routine handling and transportation. The impact threshold for initiating detonation, approximately 160 mph (see discussion in Appendix F), is well above the potential impact velocity for all accidents except an aircraft crash. When a munition is subjected to an impact velocity greater than the detonation threshold velocity, there is still a low probability of detonation, but it is possible. Data does not exist to develop a meaningful relationship for predicting the number of detonations that could occur given an aircraft crash into a munitions storage area or transport vehicle. This rationale is that, given a stack of munitions pallets in storage or in a transport vehicle, the munitions in the first row would absorb most of the impact energy. These munitions could detonate. The others would then be subjected to the energy of the detonations, as well as part of the energy of the aircraft crash. It is known that the detonations do not propagate, but it is assumed that many of them would rupture. This logic was applied to all the aircraft crash scenarios and a general result was reached. The conservative estimate is that:

- Fifteen percent of the munitions involved in the crash detonate.
- 2. Seventy percent of the rupture and release their agent content.
- 3. Fifteen percent are scattered but remain intact.

For impacts of burstered munitions in pallets, if a single munition detonation occurs it is assumed to rupture each surrounding munition in the pallet. A centrally located munition, which has the largest number of surrounding units, is conservatively assumed to be the one which detonates, even though it is less likely to detonate at this location than at the end. For projectiles, cartridges, and mortars, the number of adjacent munitions ruptured is five.

For rockets and mines only, the detonation of more than one munition was calculated to be credible for certain pallet impacts. In such



ALLER PROPERTY PROPERTY SUPPLY ALLERS

cases, two rockets detonate, rupturing 13 adjacent rockets. Or, three mines detonate rupturing 15 adjacent munitions.

## 10.1.4. Fire Release

Munitions subject to fire can fail due to thermally initiated detonations or due to hydraulic rupture. It is assumed that fires in direct contact with burstered munitions will be left unattended and allowed to burn until all combustible materials are consumed. Thus, bursters will detonate. Some neighboring munitions will fail due to the detonation. The failed munitions will spill combustible agent which will further fuel the fire. The fire will spread, leading to more detonations, and so on.

Tests at GA on 4.2-in. mortar projectiles and 8-in. projectiles showed that a detonation of a munition in a close packed array will cause the munitions adjacent to the detonated munition to break and spill their agent (Ref. 10-6). Other munitions not in direct view of the detonated munition were disheveled, but remained intact. Thus, one detonation is not sufficient to break all the munitions involved in the accident. A chain reaction must take place. The bursters in the neighboring munitions broken by a detonation will be subjected to more rapid heating than those of an intact munition. These bursters will detonate at a critical temperature, but it is assumed that detonation of a drained munition will not contribute to the agent release. TATATAL DAVIDATE JANGGALISANGAN DESERAN DESERAN DESERAN DESERAN DESERTATION DAVIDADE TATABADATANA

Based on the test results described above, it is inferred that all munitions in direct view of a munition detonation would be broken. In a rectangular array, typical for the munition storage configurations, this results in an agent release fraction of 1/9 due to detonation and 8/9 as a liquid spill. An irregular array, such as would exist after the first detonation, could result in a larger release fraction due to detonations. Therefore, it is assumed that 25% of the agent release is due to detonations for scenarios involving fire and detonations. It is assumed that fires involving nonburstered munitions will always be fought. However, when an accident involves a large fire, the first priority may be to contain the fire and prevent its spreading into unaffected areas. For conservatism, a large fire involving nonburstered munitions was treated as in the case for burstered munitions, i.e., all combustible materials involved in the accident are consumed. Whether burstered or nonburstered munitions are involved, large fires were assumed to be confined to one building, one railcar, or one truck, as appropriate.

Agent that is burned is basically destroyed, but the destruction is usually incomplete. A previous analysis (Ref. 10-7) indicated that the recovery of undecomposed agent from fires is 2.5% for GB and 0.2% for VX. The analysis was based on tests at Dugway Proving Ground (Refs. 10-8 and 10-9) in which a mock-up igloo with 11 pallets of rockets containing GB was allowed to burn to completion. The unburned GB vapor was measured by a grid of detectors surrounding the fire at 30 m distance and extending 30 m high. Actual test measurements were made for GB, and the results for VX were derived by extrapolation based on the boiling temperature, thermal decomposition temperature and volatility of VX relative to GB.

Although the above references provide a quantitative data point on the behavior of agent in a large fire involving an igloo or a transport vehicle, there are several reasons to increase the predicted agent release fraction for fires. These are:

1. The analytical procedure for detecting agent during the test yielded small quantities of agent distributed over a large number of detectors. The samples were analyzed by the dianisidine-peroxide method. The sensitivity of these measurements is expected to be marginal considering the short time available for sampling the gas cloud as it passed through

the detection grid. Therefore, it is possible that a significant amount of agent vapor was not detected during the test.

- 2. The rockets contain a large amount of propellant, which in turn contains its own oxidizer. The propellant burns very quickly and tends to produce a hot fire, even when the fire is limited by the amount of oxygen present. Fires involving other munitions may burn slower and at a lower temperature, which would promote a higher fraction of undestroyed agent.
- 3. In one simulated test of an igloo fire (Ref. 10-9) four rockets were launched out of the igloo. One of them traveled 1300 ft away from the igloo. None of them detonated upon impact, but they all broke open and spilled agent onto the ground. When one adds the liquid spill of the four rockets that escaped from the igloo to the 2-1/2% agent vapor recovered, the total agent release from the event is 4.9%.
- 4. The analytical extrapolation to determine the recovery fraction for VX is not documented. Further, the uncertainty of an extrapolation in a complex thermal-chemical rate process is considered to be large. Although the chemical properties of VX and GB suggest that the recovery fraction for VX should be much less than GB, the conclusion that the recovery of VX would be 6% times the recovery of GB as stated in Ref. 10-9 is viewed with skepticism. Therefore, a more conservative value of 25% was assumed for the recovery factor of VX versus GB. Similarly, the chemical properties of HD suggest that an analytical extrapolation for the recovery of HD would also be less than GB, but greater than VX. Therefore, a value of 50% was assumed for the recovery factor of HD versus GB.

In view of the above discussion, the release fraction for unburned agent GB vapor in all fire scenarios was assumed to be 10%. This provides a factor of two over the 4.9% combined liquid plus vapor measured in the test to allow for uncertainties in the test measurements and uncertainties in the liquid agent that escapes the fire. The corresponding release fractions for HD and VX are assumed to be 5% and 2-1/2%, respectively. These release fractions are not considered as over conservatism. The main conservatism arises from the assumption that all the agent inventory is involved in the fire, and no credit is taken for the possibility that the fire might be extinguished before all combustible materials are consumed.

## 10.1.5. <u>Release Duration</u>

ŗ

The accident durations assumed for this risk analysis were chosen to conservatively define a time for terminating most accidents identified in this analysis. In the scenarios involving liquid spills, the accident is terminated when the decontamination team has successfully terminated evaporation of agent vapor into the atmosphere. Army experience in handling and moving chemical munitions indicates that may of the agent spills could be cleaned up much quicker than the times assumed herein. However, since many accidents are rate events and have not occurred in the Army experience to date, conservative times for the accident durations have been applied.

The agent release for an evaporative spill is directly proportional to the release duration. Therefore, to be conservative, the release durations were estimated on the high side. The release durations assumed are:

 For agent spills occurring during handling or demilitarization operations caused by human or equipment malfunction, the release duration was assumed to be 1 h.

- 2. For agent spills involving human or mechanical error during onsite transportation, it was assumed that the accident could not be terminated as quickly as the above. Therefore, the release duration was assumed to be 2 h.
- 3. For agent release in the MDB following an accidental detonation outside the ECR, but with no fire, the release duration was assumed to be 2 h.
- For agent spills arising from an aircraft crash with no fire, the release duration was assumed to be 4 h.
- 5. For agent spills occurring during offsite transportation, it was assumed that an additional increase in decontamination time is necessary because the evaporation source may be less accessible. The accident duration for these accidents was assumed to be 6 h.
- 6. For severe external events, e.g., earthquake, tornado, airplane crash, the evaporation time was assumed to be 6 h.

Table 10-1 lists the times assumed for agent release for the accident scenarios involving fire and/or detonations. Plant operations accident scenarios are not included in the table because these accidents are mitigated by engineered safeguard features and are not covered by the discussion that follows.

The approach to deriving the assumed release durations was to group the accident scenarios with fire or detonations into sets with similar characteristics, then estimate a release time ranging from 10 min to 1 h. For accidents involving a large fire, it was assumed that all of the agent present ultimately becomes consumed or released as vapor. The conservative approach for these cases is to assume a shorter duration than expected because a given release to the atmosphere is more lethal

| Ever           | it          | Agent Release<br>Duration<br>(min) | Type of Event                                                                        |
|----------------|-------------|------------------------------------|--------------------------------------------------------------------------------------|
| Fire only - no | detonations | 10                                 | Handling vehicle collision                                                           |
|                |             | 60                                 | Aircraft crash, truck<br>collision/overturn, mete-<br>orite strike, earthquake       |
|                |             | 120                                | Train derailment, ship<br>accident                                                   |
| Fire with deto | onation     | 20                                 | Aircraft crash, truck col-<br>lision, earthquake, train<br>derailment, ship accident |
|                |             | 60                                 | Meteorite strike                                                                     |
| Detonations on | ly          | Instantaneous                      | Aircraft crash                                                                       |

•

TABLE 10-1AGENT RELEASE DURATION FOR ACCIDENTS INVOLVING FIRE AND DETONATION



when distributed over a shorter time interval. Factors which influence the choice of time periods are discussed below.

There are three possible combinations of scenarios involving fire and/or detonations:

- 1. Detonations only.
- 2. Fire and detonations.
- 3. Fire only.

10.1.5.1. <u>Detonations Only</u>. The scenarios that fall into this category involve a high velocity impact, such as an aircraft crash, or spurious detonation arising from undue forces that are part of the accident scenario, e.g., dropping a pallet. It is known that the detonations do not propagate. Therefore, the release from detonations is assumed to occur instantaneously.

10.1.5.2. Fire and Detonations. These events are associated with storage and transportation accidents. For some events, there is a source of external fuel, e.g., an airplane crash or fuel from a locomotive or truck. In these scenarios, the detonations are propagated by the fire, and concurrently the detonations allow additional munition failures that further fuel the fire. The overall result is a violent conflagration. The total duration of the accident may be an hour or more; however, for conservatism, the duration of the agent release is assumed to be 20 min. The scenarios not included in the 20-min assumption involve a meteorite strike into a storage igloo or into a temporary storage area. In this case, there is no source of external fuel, although the scenario does assume that fire is initiated, and detonations are propagated by the fire until all combustible materials are consumed. Because the meteorite fire starts out relatively localized and without external fuel, the release duration for the meteorite strike is assumed to be 1 h.

10.1.5.3. <u>Fire Only</u>. Events involving fire only occur in some handling, storage, and transportation accidents. For events associated with onsite handling the amount of agent involved in the fire is relatively small. The exposed agent is allowed to burn to completion, and the release duration is assumed to be 10 min. The accidents in this group associated with transportation involve a large source of external fuel, e.g., an airplane crash or a locomotive. In addition, these events involve large quantities of agent, but they do not involve burstered munitions. Therefore, these accidents present a less difficult situation to control than the corresponding case when burstered munitions are present. The agent release duration for these events was assumed to be 1 h. 

This section illustrates the application of the release methodology to determine agent releases for the specific accident sequences for each phase of the demilitarization process. It is not intended to encompass all sequences. Appendix I presents the agent releases for all sequences. Details of all agent release calculations are contained in the supporting calculations, Ref. 10-10.

#### 10.2.1. Handling

The procedure for analyzing agent releases during handling accidents was to first group the accident sequences according to agent release conditions or types of release. For example, there were a number of sequences resulting in liquid spill outdoors (HC5, HC7, CH10, HF1, HF7, and HC8). Table 10-2 shows the grouping results for all handling sequences. There were the following types of releases to be assessed:

- 1. Single munition rupture and spill outdoors.
- 2. Single munition rupture and evaporation indoors (in MDB, MHI, LPF, or storage igloo) or inside the package.
- 3. Burning of ruptured single munition spill outdoors.
- 4. Impact detonation of single munitions indoors.
- 5. Impact detonation and spill of munitions outdoors.
- 6. Impact detonation and spill of munitions indoors.
- 7. Fire and thermal detonation of munitions.

The agent inventory data for onsite and offsite transport containers is summarized in Table 10-3. Indoor spills are assumed to be mitigated within 1 h, so that evaporation lasts for that long. Failure

| Type of Release                                                | Single Munitions<br>Fails             | Multiple Munitions<br>In Pallet Or<br>Containers Involved <sup>(a)</sup> |
|----------------------------------------------------------------|---------------------------------------|--------------------------------------------------------------------------|
| Puncture/crash                                                 |                                       |                                                                          |
| Liquid spill                                                   |                                       |                                                                          |
| Outdoors(b)                                                    | HC5, HC7, HC8, HC10,<br>HF1, HF7      | None                                                                     |
| Evaporation                                                    |                                       |                                                                          |
| In MDB                                                         | HC32, HF2, HF8, HF9,<br>HF10          | None                                                                     |
| In MHI or LPF                                                  | HC13, CH16, HC17,<br>HC18, CH19, HC21 | None                                                                     |
| In package                                                     | HC14, HF4                             | None                                                                     |
| In storage igloo                                               | нс1, нсз, нс4                         | None                                                                     |
| Burning of agent spill                                         |                                       |                                                                          |
| Outdoors                                                       | HC2, HC6, HC9, HC20,<br>HF3           | None                                                                     |
| Impact detonation and<br>spill (if more than one)<br>(no fire) |                                       |                                                                          |
| Outdoors(b)                                                    | None                                  | HC22, HC23, HC24,<br>HC25, HF11, HF14                                    |
| Indoors                                                        | HC28, HC30, HF12                      | HC11, HC12, HC29,<br>HC31, HF13                                          |
| Fire and thermal detonation                                    | None                                  | HC26, HC27 <sup>(a)</sup> , HF5                                          |

## TABLE 10-2 GROUPING OF HANDLING SEQUENCES ACCORDING TO AGENT RELEASE CHARACTERISTICS

(a) HC27 involves inventory of offsite container: others involve one pallet.

(b)Outdoor spill release given in pounds of liquid, evaporation calculated by Mitre.


| Munition/Agent<br>Type | Munition Inventory<br>(1b) | No. Munitions<br>Per Pallet or ONC | No. Pallets<br>Per OFC |
|------------------------|----------------------------|------------------------------------|------------------------|
| Bomb                   |                            | <u> </u>                           |                        |
| GB                     | 220.0                      | 2                                  | 6                      |
| Mortar                 |                            |                                    |                        |
| Н                      | 6.0                        | 48                                 | 4                      |
| 105 cartridge          |                            |                                    |                        |
| GB                     | 1.6                        | 24                                 | 12                     |
| H                      | 3.2                        | 24                                 | 12                     |
| Ton container          |                            |                                    |                        |
| GB                     | 1500.0                     | 1                                  | 2                      |
| Н                      | 1700.0                     | 1                                  | 2                      |
| VX                     | 1600.0                     | 1                                  | 2                      |
| Mine                   |                            |                                    |                        |
| VX                     | 10.5                       | 36                                 | 3                      |
| 155 projectile         |                            |                                    |                        |
| GB                     | 6.5                        | 8                                  | 15                     |
| Н                      | 11.7                       | 8                                  | 15                     |
| VX                     | 6.0                        | 8                                  | 15                     |
| 8-in. projectile       |                            |                                    |                        |
| GB                     | 14.5                       | 6                                  | 10                     |
| VX                     | 14.5                       | 6                                  | 10                     |
| Rocket                 |                            |                                    |                        |
| GB                     | 10.7                       | 15                                 | 4                      |
| VX                     | 10.0                       | 15                                 | 4                      |
| Spray tank             |                            |                                    |                        |
| vx                     | 1356.0                     | 1                                  | 1                      |
|                        |                            |                                    |                        |

TABLE 10-3 INVENTORY DATA FOR ONSITE AND OFFSITE TRANSPORT CONTAINERS 22222





**7.** 34

**10.10** 

10-19

of the building ventilation system is a part of the definition of these sequences. The results for each of the above types of releases are summarized in Table 10-4.

## 10.2.2. Warehouse Storage Release During Earthquakes

There are three sites with stored, nonburstered munitions in warehouses. These are:

- 1. UMDA ton containers with agent HD stored in two warehouses.
- 2. NAAP ton containers with agent VX stored in one warehouse.
- 3. TEAD spray tanks with agent VX stored in two warehouses.

Only spray tanks and ton containers are stored in warehouses, none of which contain agent GB. Based on their impact characteristics, the ton containers are predicted to be able to be crushed or breached by the kinetic energy of a falling I-beam if the warehouse structure is damaged. Each I-beam has sufficient energy to crush one ton container but not two. Thus, the maximum number of ton containers crushed per warehouse is five, since there are that many I-beams in the warehouse roof. For similar reasons, the maximum number punctured is taken to be five per warehouse.

Spray tanks are stored in overpacks and, based on structural calculations, are not expected to be breached by the falling I-beams. Consequently, the mechanical breaching of spray tanks due to an earthquake is not considered a credible event. If a fire lasts beyond 30 min, spray tanks may fail due to the unsuppressed fire. Thus, for spray tanks, only one type of release is considered, namely burning of one or two warehouse inventories due to fire beyond 30 min. The release







|                                             |                                              |            |        | 4.2-In.<br>Morter | Car 10  | 5 mm<br>tridge |          | Ton Contai                                   | nere                 |
|---------------------------------------------|----------------------------------------------|------------|--------|-------------------|---------|----------------|----------|----------------------------------------------|----------------------|
| sequences                                   | Release Mechanism                            | Munitions  | 5      | H                 | 8       | æ              | 5        | H                                            | ٨X                   |
| HC13, NC16, NC17, HC10,<br>HC10, HC20       | 10 min floor evaporation<br>Sume avanoration |            | 0.71   | 2 x 10-4          | 0.02    | 1 × 10-4       | 1.07     | 4 x 10 <sup>-3</sup><br>3 x 10 <sup>-4</sup> | 5 x 10-5<br>2 x 10-6 |
| (Spill in Nat. LPP)                         | Total evaporation                            | 1          | 0.78   |                   | W       |                | 1.72     |                                              |                      |
| чс, њсе, њсе, њсе, њсго,<br>њгэ             | Burn of spill                                | 1          | 22     | 0.3               | 0.16    | 0.16           | 150      | 85                                           | 04                   |
| HCS, HC7, HC0, HC10,<br>HF1, HF7            | Outdoor apill(a)<br>(pounda of liquid)       | 1          | 220    | 6.0               | 1.6     | 3.2            | 1500     | 1700                                         | 1600                 |
| Hell Hell Hell                              | Peterst [cs vs] sees                         | x          | N N    | 5.                | 07 0    | 0.80           | <b>N</b> | AN<br>A                                      | VN                   |
| NCIL: BUIL: BULL: BULL:<br>MC11             | 10 min floor evenoration                     | : =        |        | 1 - 10-3          | 0.09    | 5 × 10-4       | V N      | NA NA                                        | N N                  |
| (Impact detonation)                         | Sump evaporation<br>Total evaporation        | . 2        | A A    | 3 × 10-4          | 0.07    | 3 x 10-4       | VN VN    | V N<br>N                                     | VN                   |
| NC14, NP4                                   | Package evaporation                          | 1          | 0.22   | 0.01              | 0.22    | 10.0           | 0.22     | 0.01                                         | v                    |
| HC28, HC30, HF12                            | Detonstion release<br>(no fire)              | 1          | Y.     | 1.50              | 0.40    | 0.80           | VN       | VN                                           | VN                   |
| HC26, HF3                                   | Detonation release                           | <b>A</b> . | NA     | 72                | 9.6     | 19.2           | VN       | NA                                           | NA                   |
| (1 pallet - thermal                         | Fire release                                 | <b>6.</b>  | 1:     | 11                | 2.9     | 2.9            | 150      | 85                                           | 40                   |
| (0)011101                                   | 2222723 TH101                                |            | t<br>t | 2                 | C. 71   | 1.22           | nct      | C                                            | •                    |
| HC32, <b>HF</b> 2, <b>HF8, HF9.</b><br>HF10 | l min evaporation inside<br>MDB              | 7          | w      | w                 | v       | w              | w        | w                                            | w                    |
| HC22, HC23, HC24, HC25,                     | Lapact detonation                            | x          | NA     | 1.5               | 0 Y . J | 0.80           | VN       | NA                                           | NA                   |
| HFII, HFI4                                  | Outdoor spill(s)                             | z          | NA     | 30                | 8.0     | 16.0           | VN       | NA                                           | NA                   |
| HC27                                        | Detonation                                   | C, X, P    | VN     | 288               | 115     | 230            | VN       | MA                                           | NA                   |
| (Offsite containers -                       | Fire release                                 |            | 264    | **                | 35      | 35             | 300      | 170                                          | 80                   |
| thermal failure)                            | Total release                                |            | 264    | 332               | 150     | 265            | 300      | 170                                          | 80                   |
| HCI, HC3, HC4<br>(Spill in storage          | Floor evaporation<br>(no sump)               | 1          | 4.26   | w.                | 0.10    | v              | 6.40     | w                                            | v                    |
| igloo)                                      |                                              |            |        |                   |         |                |          |                                              |                      |

200 • (\\\\\\\) • 2000 • 2000 • 2000 • 2000



|           | ST<br>VX          | 5 x 10-5<br>2 x 10-6<br>6                                         | 33.9                        | 1356                             | ****                                                                                    | ų                   | VN                              | NA<br>34<br>34                                      | v                                                    | VN<br>N                                       | NA<br>4 6<br>4 6                                       | v                                            | not<br>not                                                       | H = 1                                  |   |
|-----------|-------------------|-------------------------------------------------------------------|-----------------------------|----------------------------------|-----------------------------------------------------------------------------------------|---------------------|---------------------------------|-----------------------------------------------------|------------------------------------------------------|-----------------------------------------------|--------------------------------------------------------|----------------------------------------------|------------------------------------------------------------------|----------------------------------------|---|
|           | × *               | ж 10-6<br>ж 10-6                                                  | . 25                        | 0.0                              | ٥.                                                                                      |                     | .50                             | 7.5<br>.8<br>.3                                     |                                                      | 0.0                                           | 8                                                      |                                              | P = 74                                                           | others                                 |   |
|           | Rock<br>GB        | 0.07 3<br>0.07 2<br>0.14 ¢                                        | 1.07 0                      | 10.7 1                           | 5.35<br>5.97<br>5.05<br>5.05<br>5.05<br>5.05<br>5.05<br>5.05<br>5.05<br>5.0             |                     | .68 2                           | 6 . 1 . 3<br>6 . 1 . 3<br>6 . 1 . 3                 | •                                                    | .35 5.<br>39 13                               | 5 + 60<br>11 1                                         | .45 6                                        | tainer,<br>upturin                                               | For all                                |   |
|           | •11               | r 10-6                                                            | 9                           | r,                               | 53<br>10-6 0<br>10-6 0                                                                  | Ð                   | 3                               |                                                     | ~                                                    |                                               | 2 O O T                                                | 0                                            | site con<br>itions r                                             | P-15.                                  |   |
|           | 8-in.<br>Project  | 10 4 1<br>07 2 1<br>17 6                                          | 15 0.3                      | 5 14.                            | 1 2 4 9<br>1 2 4 9<br>1 2 4 9                                                           | 4<br>11             | 3 3.6                           | 8 21.<br>3 23.                                      | w                                                    | 3 3.6<br>5 72.                                | 218<br>16<br>234                                       | v<br>v                                       | in off.<br>of mun                                                | 4. and                                 |   |
|           | - 3               | 9 9 9 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1                           | 1.4                         | 14.                              | 6.0<br>0.0<br>0.0                                                                       | 0.2                 | 3.6                             | 21.<br>6.5<br>28.                                   | w                                                    | 3.6<br>72.                                    | 218<br>65<br>283                                       | 0.5                                          | llets<br>umber                                                   | •                                      |   |
| đ)        | ux vx             | 1 x 10<br>2 x 10                                                  | 0.15                        | <b>9</b> .0                      | 1.50<br>8 × 10<br>2 × 10                                                                | v                   | 1.50                            | 12.0<br>0.9<br>12.9                                 | w                                                    | 1.50<br>30.0                                  | 144<br>11<br>155                                       | ~                                            | er of pe<br>8. N = n                                             | N = 13                                 |   |
| cont İnue | M Proj            | H 10-4                                                            | . 59                        | (.t                              | .93<br>x 10-4<br>x 10-4                                                                 | 10,                 | .93                             | 3.4<br>5.9<br>6.9                                   |                                                      | 93<br>8.5                                     | 81<br>23                                               |                                              | tonating                                                         | . H = 2                                | ( |
| 10-4 (    | 155<br>155        | 0.04                                                              | 0.65 0                      | 5.5                              | 1.63 2                                                                                  | 1.22 0              | 1.63 2                          | 13.0 2<br>1.9 3<br>16.9 2                           | •                                                    | . 63 2<br>2.5 5                               | 0 4 6<br>0 4 6                                         | .24 €                                        | quid. (<br>ione de                                               | rocket                                 |   |
| TABLE     | Mine<br>VX        | 3 x 10-6<br>2 x 10-6                                              | 0.26                        | 10.5                             | 7.88<br>4 x 10-5<br>2 x 10-6<br>6                                                       |                     | 2.63                            | 95<br>7<br>102                                      |                                                      | 7.86 1<br>2363 3                              | 285<br>21<br>206<br>2<br>2                             | 0                                            | ound of 11<br>r of munit                                         | 36. <b>F</b> or                        |   |
|           | Release Mechanism | 10 min floor evaporation<br>Sump evaporation<br>Total evaporation | Burn of epill               | Outdoor apill(a)                 | Detonation release<br>10 min floor eveporation<br>Sump evaporation<br>Total evaporation | Package evaporation | Detonation release<br>(no fire) | Detonation release<br>Fire release<br>Total release | 1 min evaporation inside (<br>MDB                    | <pre>Impact detonation Outdoor spill(m)</pre> | Detonation<br>Fire release<br>Total release            | Floor eveporation<br>(no aump)               | oor spills are in terms of p<br>t, 6 = negligible. H = numbe     | <pre>4 3, N = 15, C = 3, and P =</pre> |   |
|           | sequences         | HC13, NC16, HC17,<br>HC19, HC19, HC20<br>(Spill in Mul,<br>LP7)   | HC2, HC6, HC9.<br>HC20, KF3 | HC5, HC7, HC8.<br>HC10, HF1, HF7 | HCll, HCl2, HC29,<br>HP13, HC31<br>(Impect detona-<br>tion)                             | HCIA, HFA           | HC28, HC30, HF12                | HC26, HF6<br>(1 pallet -<br>thermal failure)        | HC32, <b>HF</b> 2, <b>HF0,</b><br>H <b>F9, HF</b> 10 | HC22, HC23, HC24,<br>HC25, UP11, HP14         | HC27<br>(Offeite con-<br>tainers - thermal<br>failure) | HCI, HC3, HC4<br>(Spill in<br>storage igloo) | <pre>(a)Notes: Outd<br/>munitions in palle<br/>applicable.</pre> | For miner, M<br>and N=5.               |   |
|           |                   |                                                                   |                             |                                  |                                                                                         |                     |                                 |                                                     |                                                      |                                               |                                                        |                                              |                                                                  |                                        |   |



fraction due to unburnt VX agent in this case is 2.5%, as in other accident scenarios.

For ton containers, three release types were considered:

- Evaporation of agent spilled due to mechanical breach of one to five containers per warehouse.
- 2. Burning of agent spilled from breached containers.
- Burning of the entire inventory in the warehouse, starting at 30 min.

The evaporative release rate is not limited by the floor area, which is tens of thousands of square feet per warehouse. Thus, the evaporative release rate,  $m_{ev}$ , is given by Eq. 10-2. For 10-ton containers with agent HD, M = 17,000 lb and a  $\approx$  451 and b  $\approx$ 0.1. Thus,  $m_{ev} = 0.85$  lb/h for 10 containers. This rate of HD release is negligible. Therefore, evaporative release of spilled HD from breached munitions is negligible. For agent VX, the maximum number of breached ton containers is five. In this limiting case, M = 8000 lb and a  $\approx$  49,000, b  $\approx$  0.12. Thus,  $m_{ev} = 0.003$  lb/h for five breached containers. This rate of release is negligible.

The second and third types of releases involve burning of spilled agent from breached containers or burning of all ton containers due to a lack of fire suppression. For these cases, the release consists of the product of the appropriate inventory and the fire release fraction, F. Here, F = 0.025 for agent VX and F = 0.05 for agent HD, consistent with data described above. No credit is taken for agent vapor retention by the warehouse building, even if it is not structurally damaged by the earthquake, because it is not designed with a containment function.

As described in Section 5, an event tree was analyzed for the storage of ton containers at the UMDA and NAAP site warehouses. For the UMDA site, there were 17 release sequences with frequencies above  $10^{-10}/yr$ . Table 10-5 lists these sequences along with the information pertinent to the release calculations. For sequences in which the burning or agent spilled from breached munitions is the only release mode, a range of release is given corresponding to the range of containers breached (1 to 5 or 2 to 10). For sequences in which the nonsuppressed fire ignites the entire warehouse inventory, the number of breached containers is unimportant.

Table 10-6 presents the corresponding release results for ton containers stored at the NAAP site. Only five sequences are important since there is only one warehouse at the site. The maximum masses of agent VX released from this site are seven times lower than maximum mass releases of agent HD from UMDA.

In the event tree for spray tanks stored at the TEAD site, there were six significant sequences as given in Table 10-7. Since no spray tanks are mechanically breached, the only consequence variable is whether the unsuppressed fire is not suppressed in one or both warehouses. The releases upon burning of the entire inventory at one or both warehouses are given in Table 10-7. They are 8 to 16 times lower than the maximum release of the same agent (VX) from the NAAP site.

### 10.2.3. Plant Operation Releases

10.2.3.1. <u>Internal Events</u>. The analysis of agent release due to in-plant accidents used the same calculation models discussed above when applicable. However, many plant operations involve accidents which occur after the munition has been punched and drained. The agent releases for these events are not dependent on the munition failure

シンシンシン



| Sequence<br>ID | No. of<br>Munitions<br>Damaged | Spilled Munition<br>Agent Burns | No. Warehouses<br>In Which Entire<br>Inventory Burns | Release<br>To Atmopshere<br>(1b) |
|----------------|--------------------------------|---------------------------------|------------------------------------------------------|----------------------------------|
| SLKHF281       | 0                              |                                 | 1                                                    | 2.7 x $10^5$                     |
| SLKHF282       | 0                              |                                 | 2                                                    | 5.4 x 10 <sup>5</sup>            |
| SLKHC283       | 1-5                            | No                              | 0                                                    | ε(b)                             |
| SLKHF284       | 1-5                            | Yes                             | 1                                                    | 2.7 x $10^5$                     |
| SLKHF285       | 1-5                            | No                              | 1                                                    | 2.7 x $10^5$                     |
| SLKHF286       | 1-5                            | Yes                             | 2                                                    | 5.4 x 10 <sup>5</sup>            |
| SLKHC287       | 2-10                           | No                              | 0                                                    | ε                                |
| SLKHF288       | 2-10                           | Yes                             | 1                                                    | 2.7 x $10^5$                     |
| SLKHF289       | 2-10                           | Yes                             | 2                                                    | 5.4 x 10 <sup>5</sup>            |
| SLKHC2810      | 1-5                            | No                              | 0                                                    | ε                                |
| SLKHF2811      | 1-5                            | Yes                             | 1                                                    | 2.7 x $10^5$                     |
| SLKHF2812      | 1-5                            | Yes                             | 2                                                    | 5.4 x 10 <sup>5</sup>            |
| SLKHC2813      | 2-10                           | No                              | 0                                                    | ε                                |
| SLKHF2814      | 2-10                           | Yes                             | 1                                                    | 2.7 x $10^5$                     |
| SLKHF2815      | 2-10                           | Yes                             | 2                                                    | 5.4 x 10 <sup>5</sup>            |
| SLKHC2816      | 2-10                           | No                              | 0                                                    | ε                                |
| SLKHF2817      | 2-10                           | Yes                             | 2                                                    | 5.4 x 10 <sup>5</sup>            |
|                |                                |                                 |                                                      |                                  |

# TABLE 10-5AGENT HD RELEASES FROM TON CONTAINERS STORED IN<br/>UMDA WAREHOUSES DURING EARTHQUAKES(a)

(a) Agent inventory = 5.4 x  $10^6$  lb per warehouse, assuming warehouse is full.

(b)  $\epsilon$  = negligible (below 14 lb).



| Sequence<br>ID | No. of<br>Munitions<br>Damaged | Spilled Munition<br>Agent Burns | Entire Warehouse<br>Inventory Burns | Release<br>To Atmosphere<br>(1b) |
|----------------|--------------------------------|---------------------------------|-------------------------------------|----------------------------------|
| SLKVF261       | 0                              |                                 | Yes                                 | $7.5 \times 10^4$                |
| SLKVC262       | 1-5                            | No                              | No                                  | €(b)                             |
| SLKVF263       | 1-5                            | Yes                             | Yes                                 | 7.5 x $10^4$                     |
| SLKVC264       | 1-5                            | No                              | No                                  | :                                |
| SLKVF265       | 1-5                            | Yes                             | Yes                                 | 7.5 x $10^4$                     |
|                |                                |                                 |                                     |                                  |

# TABLE 10-6 AGENT VX RELEASES FROM NAAP WAREHOUSE TON CONTAINERS DURING EARTHQUAKES(a)

(a) Warehouse inventory =  $3 \times 10^6$  lb of VX, assuming warehouse is full.

(b) $\epsilon$  = negligible (below 0.3 lb).

| No. Warehouses<br>In Which Entire<br>Inventory Burns | Release<br>To Atmosphere<br>(1b)                                                                  |
|------------------------------------------------------|---------------------------------------------------------------------------------------------------|
| 1                                                    | $4.5 \times 10^3$                                                                                 |
| 2                                                    | 9.0 x $10^3$                                                                                      |
| 1                                                    | $4.5 \times 10^3$                                                                                 |
| 2                                                    | 9.0 x $10^3$                                                                                      |
| 1                                                    | $4.5 \times 10^3$                                                                                 |
| 2                                                    | 9.0 x $10^3$                                                                                      |
|                                                      | No. Warehouses<br>In Which Entire<br>Inventory Burns<br>1<br>2<br>1<br>2<br>1<br>2<br>1<br>2<br>2 |

# TABLE 10-7 AGENT VX RELEASE FROM SPRAY TANKS STORED AT TEAD WAREHOUSES DURING EARTHQUAKES(a)

12.22.656

CALCER OF

Þ.

(a)Agent inventory =  $1.79 \times 10^5$  lb of VX, assuming warehouse is full.

models discussed above. The bases for agent releases for these events are as follows:

- The evaporation rate for an indoor spill was calculated using the D2PC computer code (Ref. 10-11). Allowable surface area for evaporation was also calculated by D2PC for the first 10 min of the accident.
- The munition inventory in the MHI is 24 h at the design process rate.
- 3. The munition inventory in the UPA is six packages.
- 4. The maximum agent inventory in the TOX and piping is 500 gal in the collection tank, 28 gal in the piping. This inventory is assumed to be present at the time of the accident.

# 10.2.3.2. Earthquake At MDB.

### Burstered Munitions Release

There are two locations in the MDB where agent is present: the unpack area (UPA) and the TOX cubicle. The event trees for burstered munitions consider the potential scenarios leading to damage and agent release for one or more munitions in the UPA, damage and agent release of the TOX, or both. For the various seismic intensities, there were four sequences with significant frequencies of obtaining damage and



release, all involving fire in the MDB. For convenience these are summarized as follows:

| Sequence | Earthquake<br>Fails MDB | Munition<br>Puncture | TOX    | Fire<br>Suppressed |
|----------|-------------------------|----------------------|--------|--------------------|
| P033     | No                      | Not relevant         | Intact | No                 |
| P025     | Yes                     | Yes                  | Intact | Yes                |
| P026     | Yes                     | Yes                  | Intact | No                 |
| P029     | Yes                     | No                   | Intact | No                 |

Damage or failure of the MDB by the MDB by the earthquake is important since it allows release to atmosphere of any agent spill starting from time zero. Later, the MDB can fail due to nonsuppression of the fire. Other important intermediate events involve mechanical puncture and spill of a single munition during processing. Other munition failure modes such as early detonation of a single processed munition or puncture of a packed munition are screened out on the basis of low probability. Failure of the TOX, resulting in spill of the TOX agent inventory, due to the earthquake also is screened out on the basis of low probability. Both the mechanical failure mode for the TOX and the thermal failure of the TOX and piping is low probability. If the fire is not suppressed, it has the potential for failing the munitions in the UPA (entire inventory considered).

The above four sequences involve one or more combinations of two types of releases:

Sequences P026, P029, and P033 - Fire/detonation involving entire IIPA inventory.

Sequence P025 - Evaporation release of one munition inventory, or a burn release of one munition inventory.

The algorithms for calculating each of these types of release are described below.

For the first type, the agent inventory in the UPA is six packages containing one munitions pallet per package. Thus, the total inventory is the inventory of a single munition, B (in pounds of agent), times the number of munitions per pallet, C, times six. Thus,

UPA inventory =  $6 \times B \times C$  . (10-3)

Table 10-3 presents values of the single munitions inventory B and the total UPA inventory for the various burstered munitions.

The fire/detonation release is calculated by the equation,

where F is the release fraction due to incomplete burning. Here,

These values represent the estimated unburned vapor release during a fire. Consistent with other initiating events, 0.25 is taken to be the release due to detonation of some of the bursters and spraying of agent. The fire release fraction is applied to the remaining 75% of the inventory.

The other type of release consists of indoor evaporation or burning of spilled agent from one munition released directly to the atmosphere (failed MDB). The burn release is simply the munition inventory times the fire release fraction, F. The computer code D2PC is used to calculate the evaporative release. Values for the evaporative releases are presented in Table 10-8 for the various burstered munitions. Only agent GB evaporative released is significant since the releases for other agents are below threshold values for significant offsite consequences. These threshold values are 0.4 lb for agent GB, 0.3 lb for VX, and 14 lb for HD.

The evaporative releases are based on application of the evaporation data for a 6-h time period. This is the time estimated for mitigation or cleanup of the spill. For single burstered munition inventories, the 1/32-in. spill area is less than the VPA floor area. Since the floor area slopes to two 2 x 2 x 2 ft sumps, the following procedure is used.

Initially, the spill is assumed to wet the sloped floor area. Thus, the above equation is applied, without modifications due to any area restriction, for a selected 10-min time period. After that, the liquid is assumed to run down the shallow slope to one of the sumps, which is large enough to contain the entire burstered munition volume. Between 10 min and an estimated accident mitigation time of 6 h, the evaporation occurs at a rate dictated by the sump area of 4 ft<sup>2</sup>. This rate is essentially that given by Eq. 10-2 with M corresponding to the mass of liquid in a 1/32-in. layer of the sump pool, rather than the entire munition inventory. The evaporative releases between 0 and 10 min and 10 min and 6 h are summed to get the total evaporation release.

Since it is not known from the event tree analysis whether the fire engulfs the sump, the approach in this analysis is to take the maximum of the fire release and the evaporative release. Table 10-8 shows these releases. Generally, the fire release dominates.

TABLE 10-8 Agent inventories and releases

|                        |       |                |          | 4        | DENIL NE. |             |      |      |
|------------------------|-------|----------------|----------|----------|-----------|-------------|------|------|
|                        | Acont | 11PA Inventory | P026,    | P029, P( | 033       | đ           | 025  |      |
| Munition Type          | Type  | (1b)           | UPA Fire | Deton    | Total     | Evaporation | Burn | Net  |
| Burstered Munitions    |       |                |          |          | 1         |             |      |      |
| Mortar                 | Н     | 864            | 32       | 216      | 248       | v           | 0.30 | 0.30 |
| Cartridge              | GB    | 230            | 17       | 58       | 75        | 0.20        | 0.16 | 0.20 |
|                        | н     | 461            | 17       | 115      | 132       | Ŷ           | 0.16 | 0.16 |
| Mine                   | ΧΛ    | 2,268          | 43       | 567      | 610       | ę           | 0.26 | 0.26 |
| Projectile (155 mm)    | GB    | 312            | 23       | 78       | 101       | 0.23        | 0.65 | 0.65 |
| •                      | Н     | 562            | 21       | 141      | 162       | U.          | 0.59 | 0.59 |
|                        | ٨X    | 288            | ıc.      | 72       | 77        | Y           | 0.15 | 0.15 |
| Projectile (8 in.)     | GB    | 522            | 39       | 131      | 170       | 0.27        | 0.15 | 0.27 |
| I                      | ٨X    | 522            | 10       | 131      | 141       | ę           | 0.36 | 0.36 |
| Rocket                 | GB    | 963            | 72       | 241      | 313       | 0.25        | 1.10 | 1.0  |
|                        | ΧΛ    | 006            | 17       | 225      | 242       | Y           | 0.25 | 0.25 |
| Nonburstered Munitions |       |                |          |          |           |             |      |      |
| Bomb                   | GB    | 2,640          | 264      | NA       | 264       | 06.0        | 22   | 22   |
| Ton container          | GB    | 000'6          | 900      | NA       | 006       | 5.60        | 150  | 150  |
|                        | Н     | 10,200         | 510      | NA       | 510       | v           | 85   | 85   |
|                        | ٨X    | 9,600          | 240      | NA       | 240       | Y           | 40   | 40   |
| Spray tank             | ٨X    | 8,136          | 203      | NA       | 203       | Ų           | 34   | 34   |

10-32



522.5.01 (125.525.525.52)

Table 10-8 presents the calculated releases for the significant accident sequences.

In sequence P033, the building remains intact from the earthquake, so no release occurs for the initial 10 min, regardless of whether a single munition spill occurs or not. The ensuing fire is not suppressed and the UPA inventory is ignited at 10 min, resulting in a fire/ detonation release.

In sequence P025, the MDB is damaged, so that the agent spill from the single munition puncture is released to atmosphere. The fire is suppressed before additional munitions are involved. Thus, the release consists of evaporation if the fire area is not coincident with the spill area or a burn release if the fire burns the spilled agent.

In sequences PO26 and PO29, the release during the initial 10 min is small (the same as the sequence PO25). But since the fire is not suppressed, the UPA inventory is ignited and the total release becomes the (same as sequence PO33).

Table 10-8 shows that significantly large releases (75 to 610 lb) occur for sequences P033, P026, and P029. Releases for sequence P025 are small.

### Nonburstered Munitions Release

The event tree for nonburstered munitions contains three sequences with frequencies above the screening threshold of  $10^{-10}$  per year. All

of these involve earthquake-induced damage to the MDB and fire. They are as follows:

| Sequence | Munition Puncture | TOX    | Fire | Suppressed |
|----------|-------------------|--------|------|------------|
| P025     | Yes               | Intact |      | Yes        |
| P026     | Yes               | Intact |      | No         |
| P029     | No                | Intact |      | No         |

These sequences involve the same types of releases as for the burstered munitions with one exception. Nonsuppressed fire (lasting more than 10 min) for burstered munitions in the UPA involves both detonation and fire, while only fire is involved for nonburstered munitions. Also, the ignition time is 30 min for nonburstered munitions. Thus, the release algorithm is changed to:

The evaporation algorithm is similar for burstered and nonburstered munitions. Inventory algorithms are the same.

Table 10-8 presents the inventories of agents in nonburstered munitions or in the TOX. The larger inventory (over  $10^3$  lb) of the nonburstered munitions causes some special considerations for a puncture release. A puncture is interpreted to consist of a 1.5-in. diameter hole. The agent flow rate out the hole is approximately 100 lb/min, which means that the entire munition inventory spills out in about 1/4 h. In the UPA, the spill is l.mited to 2140 ft<sup>2</sup> of floor area during the initial 10 min before the liquid flows to the sump. when 379 lb of agent spills into this area, a critical pool thickness is reached, namely 1/32 in., and the evaporation rate levels off. After 10 min, the sump will be overflowed for certain munitions. The pool area is calculated based on a slope of 1/4 in. for each foot of floor space and the evaporation rate is adjusted for that area.

Results of the inventory and release calculations for nonburstered munitions are summarized in Table 10-8. The effect of fire in the UPA is found to be most important.

### 10.2.4. Transport Releases

10.2.4.1. <u>Onsite Transport Releases</u>. For onsite truck transport, each truck will carry one OFC (for rail or air transport) or one vault (for marine shipment). The agent inventory of each OFC is summarized in Table 10-3 for the various munitions. Two ton containers (agent H) are carried in the vaults.

Table 10-9 presents the truck accident release calculations for the marine transport option. Those sequences where no release values are given were screened out on the backs of low frequency. Note that no detonation releases occurs because only ton containers are involved. The only significant release sequence is VW7, involving aircraft crash, mechanical rupture, and evaporation.

Table 10-10 presents the corresponding release calculations for the air transport option. Onsite transport releases for the rail transport option are the same as for the air transport option.

10.2.4.2. <u>Offsite Transport - Air</u>. The assumptions made for agent releases during aircraft accidents are as follows:

- Given a severe impact release involving burstered munitions,
   0.15 will detonate, 0.70 will spill, and 0.15 will scatter but remain intact.
- 2. Given a severe impact release involving nonburstered munitions, all agent will spill.

RESULTS OF AGENT RELEASE FOR ONSITE TRANSPORT ACCIDENT SEQUENCES (MARINE OPTION) TABLE 10-9

| Scenario | APG Frequency           | Agent Available | Spilled<br>(1b) | Destroyed<br>(1b) | Vapor<br>(1b) | Deton <b>a</b> ted<br>(1b) | Duration<br>Time |
|----------|-------------------------|-----------------|-----------------|-------------------|---------------|----------------------------|------------------|
| VWKHSOOT | 0.0                     | 3400            | :               | :                 | 1             | •                          | :                |
| VWKHS002 | 0.0                     | 3400            | 1               | :                 | ;             | :                          | ;                |
| VWKHS003 | $2.7 \times 10^{-11}$   | 3400            | :               | !                 | ;             | ;                          | *                |
| VWKHF005 | 0.0                     | 2400            | 1700            | ;                 | ;             | ;                          | 2 h              |
| VWKHS006 | $7.2 \times 10^{-7}$    | 3400            | 2890            | ;                 | ;             | :                          | Instant          |
| VWKHS007 | $5.9 \times 10^{-8}$    | 3400            | :               | 3315              | 85            | ;                          | 20 min           |
| 000SHXMA | 0.0                     | 3400            | :               | 4<br>1            | ;             | 1                          | •                |
| OTOSHXMA | 0.0                     | 3400            | 1               | :                 | ;             | 1<br>1                     | :                |
| TIOSHXMA | 1.2 x 10 <sup>-9</sup>  | 1700            | ;               | 4<br>1            | 1             | 1<br>1                     | 2 h              |
| CTOSHXMA | 0.0                     | 3400            | ł               | ;                 | ;             | 8                          | ;                |
| VWKHS014 | 1.1 × 10 <sup>-10</sup> | 3400            | 1700            | :                 | 1             | :                          | 2 h              |
|          |                         |                 |                 |                   |               |                            |                  |



JF 1. (\*\*\*) /\*\* \*\*\*

12.01







0.516562

| Scenario | Agent<br>Available(a) | Spilled<br>(1b) | Destroyed<br>(1b) | Vapor<br>(1b) | Detonated<br>(1b) | Duration<br>Time |
|----------|-----------------------|-----------------|-------------------|---------------|-------------------|------------------|
| VAKHS001 | 3400                  |                 |                   |               |                   |                  |
| VAPGS001 | 760                   |                 |                   |               |                   |                  |
| VAPHS001 | 1404                  |                 |                   |               |                   |                  |
| VAPVS001 | 756                   |                 |                   |               |                   |                  |
| VAQGS001 | 870                   | - 0             |                   |               |                   |                  |
| VARGS001 | 645                   |                 |                   |               |                   |                  |
| VARVS001 | 612                   | ÷-              |                   |               |                   |                  |
| VAKHS002 | 3400                  |                 |                   |               |                   |                  |
| VAPGS002 | 760                   |                 |                   |               |                   |                  |
| VAPHS002 | 1404                  |                 |                   | ~ -           |                   |                  |
| VAPVS002 | 756                   |                 |                   |               |                   |                  |
| VAQGS002 | 870                   |                 |                   |               |                   |                  |
| VARGS002 | 645                   |                 |                   |               |                   |                  |
| VARVS002 | 612                   |                 |                   |               |                   |                  |
| VAKHS003 | 3400                  | 1700.0          |                   |               |                   | 2 h              |
| VAPGS003 | 760                   | 6.5             |                   |               |                   | 2 h              |
| VAPHS003 | 1404                  | 11.7            |                   |               |                   | 2 h              |
| VAPVS003 | 756                   | 6.3             |                   |               |                   | 2 h              |
| VAQGS003 | 870                   | 14.5            |                   |               |                   | 2 h              |
| VARGS003 | 645                   | 10.75           |                   |               |                   | 2 h              |
| VARVS003 | 612                   | 10.2            |                   |               |                   | 2 h              |
| VAPGC004 | 760                   |                 |                   |               |                   |                  |
| VAPHC004 | 1404                  |                 |                   |               |                   |                  |
| VAPVC004 | 756                   |                 |                   |               |                   |                  |
| VAQGC004 | 870                   |                 |                   |               |                   |                  |
| VARGC004 | 645                   |                 | 435.37            | 48.3          | 161.25            | 20 min           |
| VARVC004 | 612                   |                 | 447.5             | 11.5          | 153.0             | 20 min           |
| VAKHF005 | 3400                  |                 |                   |               |                   |                  |
| VAKHS006 | 3400                  | 3400.0          |                   |               |                   | Instant          |
| VAPGC006 | 760                   | 532.0           |                   |               | 114.0             | Instant          |
| VAPHC006 | 1404                  | 982.8           |                   |               | 210.6             | Instant          |
| VAPVC006 | 756                   | 529.2           |                   |               | 113.4             | Instant          |
| VAQGC006 | 870                   | 609.0           |                   |               | 130.5             | Instant          |
| VARGC006 | 645                   | 451.5           |                   |               | 96.75             | Instant          |
| VARVC006 | 612                   | 428.4           |                   |               | Q1 8              | Instant          |

TABLE 10-10RESULTS OF ONSITE TRANSPORT RELEASE ANALYSIS - AIR OPTION







| Scenario | Agent<br>Available(a) | Spilled<br>(1b) | Destroyed<br>(1b) | Vapor<br>(1b) | Detonated<br>(1b) | Duration<br>Time |
|----------|-----------------------|-----------------|-------------------|---------------|-------------------|------------------|
| VAKHF007 | 3400                  |                 | 3230.0            | 170.0         |                   | 20 min           |
| VAPGC007 | 760                   |                 | 513.0             | 57.0          | 190.0             | 20 min           |
| VAPHC007 | 1404                  |                 | 1000.3            | 52.7          | 351.0             | 20 min           |
| VAPVC007 | 756                   |                 | 552.8             | 14.2          | 189.0             | 20 min           |
| VAQGC007 | 870                   |                 | 587.2             | 65.3          | 217.5             | 20 min           |
| VARGC007 | 645                   |                 | 435.37            | 48.3          | 161.25            | 20 min           |
| VARVC007 | 612                   |                 | 447.5             | 11.5          | 153.0             | 20 min           |
| VAKHS009 | 3400                  |                 |                   |               |                   |                  |
| VAPGS009 | 760                   |                 |                   |               | ~~                |                  |
| VAPHS009 | 1404                  |                 |                   |               |                   |                  |
| VAPVS009 | 756                   |                 |                   |               |                   |                  |
| VAQGS009 | 870                   |                 |                   |               |                   |                  |
| VARGS009 | 645                   |                 |                   |               |                   |                  |
| VARVS009 | 612                   |                 |                   |               |                   |                  |
| VAKHS010 | 3400                  |                 |                   |               |                   |                  |
| VAPGS010 | 760                   |                 |                   |               |                   |                  |
| VAPHS010 | 1404                  |                 |                   |               |                   |                  |
| VAPVS010 | 756                   |                 |                   |               |                   |                  |
| VAQGS010 | 870                   |                 |                   |               |                   |                  |
| VARGS010 | 645                   |                 |                   |               |                   |                  |
| VARVS010 | 612                   |                 |                   |               |                   |                  |
| VAKHS011 | 3400                  | 1700.0          |                   |               |                   | 2 h              |
| VAPGS011 | 760                   | 6.5             |                   |               |                   | 2 h              |
| VAPHS011 | 1404                  | 11.7            |                   |               |                   | 2 h              |
| VAPVS011 | 756                   | 6.3             |                   |               |                   | 2 h              |
| VAQGS011 | 870                   | 14.5            |                   |               |                   | 2 h              |
| VARGS011 | 645                   | 10.75           |                   |               |                   | 2 h              |
| VARVS011 | 612                   | 10.2            |                   |               |                   | 2 h              |
| VAPGC012 | 760                   |                 | 513.0             | 57.0          | 190.0             | 20 min           |
| VAPHC012 | 1404                  |                 | 1000.3            | 52.7          | 351.0             | 20 min           |
| VAPVC012 | 756                   |                 | 552.8             | 14.2          | 189.0             | 20 min           |
| VAQGC012 | 870                   | -               | 587.2             | 65.3          | 217.5             | 20 min           |
| VARGC012 | 645                   |                 | 435.4             | 48.4          | 161.3             | 20 min           |
| VARVC012 | 612                   |                 | 447.5             | 11.5          | 153.0             | 20 min           |
| VAKHF013 | 3400                  |                 | 3315.0            | 85.0          |                   | 1 h              |

TABLE 10-10 (Continued)

Ą

Î

d

Ċ

A CALLY CALLY

| Scenario | Agent<br>Available(a) | Spilled<br>(lb) | Destroyed<br>(1b) | Vapor<br>(1b) | Detonated<br>(1b) | Duration<br>Time |
|----------|-----------------------|-----------------|-------------------|---------------|-------------------|------------------|
| VAKHS014 | 3400                  | 1700.0          |                   |               |                   | 2 h              |
| VAPGS014 | 760                   | 6.5             |                   |               |                   | 2 h              |
| VAPHS014 | 1404                  | 11.5            |                   |               |                   | 2 h              |
| VAPVS014 | 756                   | 6.3             |                   |               |                   | 2 h              |
| VAQGS014 | 870                   | 14.5            |                   |               |                   | 2 h              |
| VARGS014 | 645                   | 10.75           |                   |               |                   | 2 h              |
| VARVS014 | 612                   | 10.2            |                   |               |                   | 2 h              |
| VAPGS015 | 760                   | 32.5            |                   |               | 6.5               | Instant          |
| VAPHS015 | 1404                  | 58.5            |                   |               | 11.7              | Instant          |
| VAPVC015 | 756                   | 31.5            |                   |               | 6.3               | Instant          |
| VAQGC015 | 870                   | 72.5            |                   |               | 14.5              | Instant          |
| VARGC015 | 645                   | 623.5           |                   |               | 21.5              | Instant          |
| VARVC015 | 612                   | 591.6           |                   |               | 20.4              | Instant          |

TABLE 10-10 (Continued)

(a)From Table 1-2, "Transportation of Chemical Agents and Munitions: A Concept Plan," U.S. Army, June 15, 1987.

- 4. Given a fire only with release involving burstered munitions, 0.25 detonates and 0.75 spill. The fire release fraction (Section 10.1.4) is applied to the spilled inventory (10% for GB, 5% for H, and 2.5% for VX).
- 5. Given a fire only with release involving nonburstered munitions, all agent spills. The fire release fraction is applied to the spilled agent.
- Given a severe impact and a fire release, 0.25 detonates and 0.75 spills. The fire release fraction is applied to the spilled agent.
- 7. Given a moderate impact and a fire release, all agent spills. The fire release fraction is applied to the entire inventory.

Table 10-11 presents the agent release results in unclassified form (fraction of inventory released).

Table 10-12 presents the agent releases for marine transport accident sequences. The key parameter for evaporation releases is the fraction of agent spilled which does not sink but remains on the surface. These results are listed even for certain sequences which were screened out based on low frequency.

### 10.2.5. Uncertainties

No uncertainty analysis was performed for the agent release analysis. The releases reported are treated as conservative estimates, rather than central (e.g., median) estimates, since they are based on assumptions which are often conservative. Examples are: (1) use of

| C                  |
|--------------------|
|                    |
| - M                |
| - 81               |
| ି ପ୍ର              |
| $\mathbf{z}$       |
| i                  |
| - 23               |
|                    |
| - 21               |
|                    |
| - 3                |
|                    |
|                    |
| - 3                |
| -2                 |
|                    |
|                    |
| 1                  |
| - 01               |
| - 24               |
| - 21               |
| - 21               |
|                    |
|                    |
|                    |
|                    |
|                    |
| 2                  |
|                    |
| 1                  |
|                    |
|                    |
|                    |
|                    |
| - 9                |
| A                  |
| 4                  |
|                    |
|                    |
|                    |
|                    |
| - 41               |
|                    |
| <u>_</u>           |
|                    |
| - <sup>-</sup> - 1 |
|                    |
|                    |
|                    |
| Ϋ́                 |
|                    |
|                    |
|                    |
|                    |
|                    |
| 1                  |
| $\mathbb{N}$       |
| $\sim$             |
|                    |
|                    |

SCHOOL STREET

| Origination         Scenario         Fraction<br>Detonated         Fraction Spilled<br>or Vaporized         Fraction Bit           APG on C-141         ABKHA001         0         1.0         0           APG on C-141         ABKHA002         0         0         0           APG on C-141         ABKHA002         0         0         0           ABKHC004         0.25         0.075         0.675           ABKHC005         0         0         0           ABKHC004         0.25         0.075         0.675           ABKHC003         0         0.1         0.9           ABKHC004         0.25         0.075         0.675           ABKGC005         0         0         0           ABKGC004         0.25         0.075         0.675           ABKGC005         0         0         0         0           ABKGC005         0         0         0         0           ABKGC006         0.25         0.075         0.675           ABKGC005         0         0         0         0           ABKGC006         0         0         0         0           ABKGC005         0         0         0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                           | RELF          | ASE CONSEC | TABLE<br>QUENCES FOR  | 10-11<br>THE AIR TRANSPORT       | MODE         |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------|---------------|------------|-----------------------|----------------------------------|--------------|
| APG on C-141       ABKHA001       0       1.0       0         ABG on C-141       ABKHA002       0       0       0         ABKHC004       0.25       0.075       0.675         ABKHC005       0       0       0         APG on C-5       ABKHA001       0       1.0       0         ABKHC005       0       0       0       0         ABKHC005       0       0       0       0         ABKHC004       0.25       0.075       0.675         ABKHC005       0       0       1.0       0         ABKHC005       0       0       1.0       0         ABKHC005       0       0.1       0.9       ABKHC005       0       0         ABKHC005       0       0       1.0       0       0         ABKC005       0       0       0       0       0         ABKC7003       0       0       1.0       0       0         ABKC6005       0       0       1.0       0       9         ABKC7003       0       0       1.0       0       0         ABKC7003       0       0       0       0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                           | Origination   | Scenario   | Fraction<br>Detonated | Fraction Spilled<br>or Vaporized | Fraction Bur |
| ABKHA002         0         0         0         0           ABKHC004         0.25         0.075         0.675           ABKHC005         0         0         0           APG on C-5         ABKHC001         0         1.0         0           ABKHC004         0.25         0.075         0.675           ABKHC004         0.25         0.075         0.675           ABKHC004         0.25         0.075         0.675           ABKHC005         0         0         1.0         0           ABKHC004         0.25         0.075         0.675           ABKG003         0         0         0         0           ABKGC003         0         0         0         0           ABKG7003         0         0         0         0           ABKG7003         0         0.1         0.9         0           ABKG7003         0         0.1         0.9         0           ABKG2002         0         0         0         0           ABKG2003         0         0.1         0.9         0           ABKG2003         0         0         0         0           A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                           | APG on C-141  | ABKHA001   | 0                     | 1.0                              | 0            |
| ABKHP003         0         0         0         0           APG on C-5         ABKH002         0         0         0           APG on C-5         ABKH002         0         0         0           APG on C-5         ABKH002         0         0         0           ABKH003         0         0.1         0.9         0.675           ABKH004         0.25         0.075         0.675           ABKH005         0         0.1         0.9           ABKH004         0.25         0.075         0.675           ABKG002         0         0         0           ABKG002         0         0         0           ABKG002         0         0         0           ABKG003         0         1.0         0           ABKG004         0.25         0.075         0.675           ABKG003         0         0         0         0           ABKG003         0         0.1         0.9         0           ABKG003         0         0.1         0.9         0           ABKG003         0         0.1         0.9         0           ABK27003         0.25                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                           |               | ABKHA002   | 0                     | 0                                | 0            |
| ABKHC004         0.25         0.075         0.675           APG on C-5         ABKHA001         0         1.0         0           ABKH002         0         0         0         0           ABKH002         0         0         0         0           ABKH003         0         0.1         0.9           ABKH004         0.25         0.075         0.675           ABKH003         0         0.1         0.9           ABKH003         0         0.1         0.9           ABKH003         0         0.10         0           ABKG003         0         0         0           ABKG003         0         0         0           ABKG003         0         0         0           ABKG703         0         0.1         0           ABKG703         0         0.1         0.9           ABKG703         0         0.1         0.9           ABKG703         0         0         0           ABKG202         0         0         0           ABK2001         0.15         0.85         0           ABK2002         0         0         0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                           |               | ABKHF003   | 0                     | 0                                | 0            |
| ABKHC005         0         0         0           APG on C-5         ABKHA001         0         1.0         0           ABK         0         0         0         0           ABKHC003         0         0.1         0.9           ABKHC005         0         0.1         0.9           ABKHC004         0.25         0.075         0.675           ABKG002         0         0         0           ABKG002         0         0         0           ABKG003         0         0         0           ABKGC004         0.255         0.075         0.675           ABKGC005         0         0         0         0           ABKGC004         0.255         0.075         0.675           ABKGC005         0         0         1.0         0           ABKGC004         0.25         0.075         0.675           ABKGC005         0         0.1         0.9           ABKGC004         0.25         0.075         0.675           ABCZO05         0         0         0         0           ABKGC004         0.25         0.075         0.675           ABFZC005                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                           |               | ABKHC004   | 0.25                  | 0.075                            | 0.675        |
| APG on C-5       ABKHA001       0       1.0       0         ABG on C-5       ABKHA002       0       0       0.1       0.9         ABKHC005       0       0.1       0.9       ABKHC005       0.075       0.675         ABKC004       0.25       0.075       0.675       ABKG002       0       0         ABKC002       0       0       1.0       0       0       0         ABKC003       0       0       0       0       0       0         ABKGC004       0.25       0.075       0.675       0.675         ABKGC005       0       0       0       0       0         ABKGC004       0.25       0.075       0.675       0.675         ABKGC004       0.25       0.075       0.675       0.675         ABKGC004       0.25       0.075       0.675       0.675         ABBZ2002       0       0       0       0       0         ABBZ2002       0       0       0       0       0         ABBZ2002       0       0       0       0       0         ABPZ003       0.25       0.075       0.675       0.675                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                           |               | ABKHC005   | 0                     | 0                                | 0            |
| AFG 61 C-5       ABRIAGO2       0       0       0         ABRIAGO2       0       0       0       0         ABRIAGO2       0       0       0       0         ABRIAGO2       0       0       0       0       0         ABRIAGO2       0       0       0       0       0       0         ABRIAGO2       0       0       0       0       0       0       0         ABRIAGO2       0       0       0       0       0       0       0       0         ABRIAGO2       0       0       0       1.0       0       0       0         ABRG2003       0       0       0       0       0       0       0         ABRG2004       0.25       0.075       0.675       0.675       0.675       0.675         ABRG2003       0       0       0       0       0       0       0         ABRG2005       0       0       0       0       0       0       0         ABR22002       0       0       0       0       0       0       0       0       0       0         ABR22003       0.25                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                           | APC on C-5    | ABYUA001   | 0                     | 1.0                              | ٥            |
| ABKH7003       0       0.1       0.9         ABKH7003       0       0.1       0.9         ABKH7003       0       0.1       0.9         LBAD on C-141       ABKGA002       0       0         ABKG001       0       1.0       0         ABKG002       0       0       0         ABKG004       0.25       0.075       0.675         ABKG004       0.25       0.075       0.675         ABKG004       0.25       0.075       0.675         ABKG003       0       0       0         ABKG004       0.25       0.075       0.675         ABKG003       0       0       0       0         ABKG003       0       0       0       0         ABKG003       0       0.1       0.9       0         ABKG003       0       0.1       0.9       0         ABKG004       0.25       0.075       0.675         ABBZ000       0.15       0.85       0         ABPZ003       0       0       0         ABPZ003       0       0       0         ABPZ003       0.25       0.75       0.675 <td></td> <td></td> <td>ARKHANNO</td> <td>0</td> <td>0</td> <td>ñ</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                           |               | ARKHANNO   | 0                     | 0                                | ñ            |
| ABKHC004       0.25       0.075       0.675         ABKHC005       0       0.1       0.9         LBAD on C-141       ABKGA002       0       0       0         ABKG002       0       0       0       0       0         ABKG002       0       0       0       0       0         ABKG002       0       0       0       0       0         ABKG003       0       0       0       0       0         ABKG002       0       0       0       0       0         ABKG003       0       0.1       0       0       0         ABKG003       0       0.1       0.9       0       0         ABKG003       0       0.15       0.85       0       0         ABK26001       0.15       0.85       0       0       0         ABF26003       0       0       0       0       0       0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                           |               | VBKHEUUS   | õ                     | 0.1                              | 0.9          |
| ABKHCO05         O.12         O.13         O.9           LBAD on C-141         ABKGA001         0         1.0         0           ABKG003         0         0         0         0           ABKG003         0         0         0         0           ABKG003         0         0         0         0           ABKG004         0.25         0.075         0.675           ABKG002         0         0         0           ABKG003         0         0.1         0           ABKG003         0         0.1         0           ABKG003         0         0.1         0           ABKG004         0.25         0.075         0.675           ABKG003         0         0         0           ABKG003         0         0.1         0.9           ABKG003         0         0         0         0           ABK2C004         0.25         0.075         0.675 <t< td=""><td></td><td></td><td>ABKHCOOL</td><td>0-25</td><td>0.075</td><td>0.675</td></t<>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                           |               | ABKHCOOL   | 0-25                  | 0.075                            | 0.675        |
| LBAD on C-141       ABKGA001       0       1.0       0         ABKGA002       0       0       0       0         ABKGC004       0.25       0.075       0.675         ABKGC005       0       0       0         ABKGC005       0       0.1       0.9         ABKC001       0.15       0.85       0         ABPZC002       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC002       0       0       0         ABPZC002       0       0       0         ABPZC003       0.25                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                           |               | ABKHC005   | 0                     | 0.1                              | 0.9          |
| LBAD on C-141 ABKGA001 0 1.0 0<br>ABKGA002 0 0 0 0<br>ABKGR003 0 0 0<br>ABKGR003 0 0 0<br>ABKGR005 0 0 0 0<br>ABKGR002 0 0 0<br>ABKGR002 0 0 0<br>ABKGR003 0 0.1 0.9<br>ABKGR003 0 0.1 0.9<br>ABKGR004 0.25 0.075 0.675<br>ABDGC05 0 0 0.1 0.9<br>ABKGR002 0 0 0<br>ABFZA002 0 0 0<br>ABFZA002 0 0 0<br>ABFZC004 0.25 0.075 0.675<br>ABFZC004 0.25 0.075 0.675<br>ABFZC004 0.25 0.075 0.675<br>ABFZC004 0.25 0.075 0.675<br>ABFZC003 0 0 0<br>LBAD on C-5 AFBZA001 0.15 0.85 0<br>ABFZC003 0 0 0<br>ABFZC004 0.25 0.75 0.675<br>ABFZC005 0 0 0.1 0.9<br>LBAD on C-141 ABFZA001 0.15 0.85 0<br>ABFZC004 0.25 0.75 0.675<br>ABFZC003 0.25 0.075 0.675<br>ABFZC005 0 0 0.1 0.9<br>LBAD on C-141 ABFZA001 0.15 0.85 0<br>ABFZC005 0 0 0.1 0.9<br>LBAD on C-141 ABFZA001 0.15 0.85 0<br>ABFZC005 0 0 0.1 0.9<br>LBAD on C-141 ABFZA001 0.15 0.85 0<br>ABFZC005 0 0 0.1 0.9<br>LBAD on C-141 ABFZA001 0.15 0.85 0<br>ABFZC005 0 0 0 0<br>ABFZC005 0 0 0 0<br>ABFZC005 0 0 0.1 0.15 0.7312<br>ABFZC005 0 0 0 0<br>ABFZC003 0.25 0.01875 0.7312<br>ABFZC003 0.25 0.01875 0.7312<br>ABFZC005 0 0 0.025 0.975                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                           |               | 1          |                       |                                  | ••••         |
| ABKGA002 0 0 0 0<br>ABKGC003 0 0 0<br>ABKGC004 0.25 0.075 0.675<br>ABKGC005 0 0 0<br>LBAD on C-5 ABKGA001 0 1.0<br>ABKGC003 0 0.1 0.9<br>ABKGC004 0.25 0.075 0.675<br>ABDGC005 0 0.1 0.9<br>ABKGC004 0.25 0.075 0.675<br>ABDGC005 0 0 0.1 0.9<br>LBAD on C-141 ABPZA001 0.15 0.85 0<br>ABPZF003 0 0 0 0<br>ABPZC004 0.25 0.075 0.675<br>ABPZC004 0.25 0.75 0.7312<br>ABRZA002 0 0 0<br>LBAD on C-141 ABRZA001 0.15 0.85 0<br>ABRZA002 0 0 0<br>ABRZA002 0 0 0<br>ABRZA002 0 0 0<br>ABRZA002 0 0 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0.01875 0.7312<br>ABRZC005 0 0.01875 0.7312<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0.025 0.975                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                           | LBAD on C-141 | ABKGA001   | 0                     | 1.0                              | 0            |
| ABKGF003 0 0 0 0<br>ABKGC004 0.25 0.075 0.675<br>ABKGC005 0 0 0<br>LBAD on C-5 ABKGA001 0 1.0 0<br>ABKGC003 0 0.1 0.9<br>ABKGC004 0.25 0.075 0.675<br>ABDGC005 0 0.1 0.9<br>LBAD on C-141 ABPZA001 0.15 0.85 0<br>ABPZC003 0 0 0<br>ABPZC003 0 0 0<br>ABPZC004 0.25 0.075 0.675<br>ABPZC004 0.25 0.075 0.675<br>ABPZC005 0 0 0<br>LBAD on C-5 APBZA001 0.15 0.85 0<br>ABPZC003 0.25 0.075 0.675<br>ABPZC003 0.25 0.075 0.675<br>ABPZC004 0.25 0.75 0.675<br>ABPZC005 0 0 0<br>LBAD on C-141 ABRZA001 0.15 0.85 0<br>ABPZC003 0.25 0.75 0.675<br>ABPZC004 0.25 0.75 0.675<br>ABPZC004 0.25 0.75 0.675<br>ABPZC004 0.25 0.75 0.675<br>ABPZC004 0.25 0.71 0.9<br>LBAD on C-141 ABRZA001 0.15 0.85 0<br>ABRZC005 0 0 0<br>LBAD on C-141 ABRZA001 0.15 0.85 0<br>ABRZC005 0 0 0<br>LBAD on C-141 ABRZA001 0.15 0.85 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0<br>LBAD on C-5 ABRZA001 0.15 0.85 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0<br>LBAD on C-5 ABRZA001 0.15 0.85 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0.025 0.975                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                           |               | ABKGA002   | 0                     | 0                                | 0            |
| ABRCC004 0.25 0.075 0.675<br>ABKGC005 0 0 0 0<br>ABKGA001 0 1.0 0<br>ABKGA002 0 0 0<br>ABKGF003 0 0.1 0.9<br>ABKGC004 0.25 0.075 0.675<br>ABDGC005 0 0.1 0.9<br>LBAD on C-141 ABPZA001 0.15 0.85 0<br>ABPZC002 0 0 0 0<br>ABPZC002 0 0 0 0<br>ABPZC004 0.25 0.075 0.675<br>ABPZC004 0.25 0.075 0.675<br>ABPZC004 0.25 0.075 0.675<br>ABPZC003 0.25 0.075 0.675<br>ABPZC004 0.25 0.75 0.7312<br>ABRZC005 0 0 0<br>LBAD on C-141 ABRZA001 0.15 0.85 0<br>ABRZC005 0 0 0<br>LBAD on C-5 ABRZA001 0.15 0.85 0<br>ABRZC005 0 0 0 |                                           |               | ABKGF003   | 0                     | 0                                | 0            |
| ABRCC005       0       0       0       0         LBAD on C-5       ABKGA002       0       0       0         ABKGF003       0       0.1       0.9         ABKGC004       0.25       0.075       0.675         ABBC2005       0       0.1       0.9         ABKG2004       0.25       0.075       0.675         ABDGC005       0       0.1       0.9         LBAD on C-141       ABPZA001       0.15       0.855       0         ABPZC003       0       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC003       0       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC005       0       0       0       0         ABPZC003       0.25       0.75       0.675         ABPZC005       0       0       0         ABPZC005       0       0       0       0         ABPZC005       0       0.11       0.9       0         LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZC005       0       0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                           |               | ABKGC004   | 0.25                  | 0.075                            | 0.675        |
| LBAD on C-5 ABKGA001 0 1.0 0<br>ABKGA002 0 0<br>ABKGC003 0 0.1 0.9<br>ABKGC005 0 0.1 0.9<br>ABKGC005 0 0.1 0.9<br>ABKCC005 0 0.1 0.9<br>ABCC005 0 0.1 0.9<br>LBAD on C-141 ABPZA001 0.15 0.85 0<br>ABPZC002 0 0 0 0<br>ABPZC004 0.25 0.075 0.675<br>ABPZC004 0.25 0.075 0.675<br>ABPZC004 0.25 0.75 0.675<br>ABPZC005 0 0 0<br>LBAD on C-141 ABRZA001 0.15 0.85 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0<br>LBAD on C-5 ABRZA001 0.15 0.85 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0<br>LBAD on C-5 ABRZA001 0.15 0.85 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0.025 0.025 0.7312<br>ABRZC005 0 0.025 0.7312<br>ABRZC005 0 0.025 0.025 0.025 0.025 0.025 0.7312<br>ABRZC005 0 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.025 0.02                                                                                                  |                                           |               | ABKGC005   | 0                     | 0                                | U            |
| ABKGA002       0       0       0       0         ABKGF003       0       0.1       0.9         ABKGC004       0.25       0.075       0.675         ABBCC005       0       0.1       0.9         ABBC2001       0.15       0.85       0         ABPZA002       0       0       0         ABPZC04       0.25       0.075       0.675         ABPZC04       0.25       0.075       0.675         ABPZC04       0.25       0.075       0.675         ABPZC004       0.25       0.075       0.675         ABPZC005       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC005       0       0       0         ABPZC004       0.25       0.75       0.675         ABPZC005       0       0.11       0.9         LBAD on C-141       ABRZA002       0       0         ABRZF003       0.25       0.01875       0.7312         ABRZC005       0       0       0         ABRZF003       0.25       0.01875       0.7312         ABRZC005       0       0       0      <                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                           | LBAD on C-5   | ABKGA001   | 0                     | 1.0                              | 0            |
| ABKGF003       0       0.1       0.9         ABKGC004       0.25       0.075       0.675         ABDGC005       0       0.1       0.9         LBAD on C-141       ABPZA001       0.15       0.85       0         ABPZF003       0       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC003       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC005       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC005       0       0       0         LBAD on C-5       APBZA001       0.15       0.85       0         ABPZC004       0.25       0.75       0.675         ABPZC005       0       0       0       0         ABPZC003       0.25       0.075       0.675         ABPZC003       0.25       0.75       0.675         ABPZC003       0.25       0.75       0.675         ABPZC003       0.25       0.75       0.675         ABPZC003       0.25       0.75       0.675         ABPZC003                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                           |               | ABKGA002   | 0                     | 0                                | 0            |
| ABKGC004       0.25       0.075       0.675         ABDGC005       0       0.1       0.9         LBAD on C-141       ABPZA001       0.15       0.85       0         ABPZ002       0       0       0       0         ABPZ003       0       0       0       0         ABPZC04       0.25       0.075       0.675         ABPZC03       0       0       0         ABPZC04       0.25       0.075       0.675         ABPZC03       0       0       0         ABPZC04       0.25       0.075       0.675         ABPZ03       0.25       0.075       0.675         ABPZC03       0.25       0.75       0.675         ABPZC03       0.25       0.7312       0.7312         ABRZ03       0.25       0.01875       0.7312         ABRZ0401       0.15       0.85       0         ABRZ003       0.25                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                           |               | ABKGF003   | 0                     | 0.1                              | 0.9          |
| ABDGC005       0       0.1       0.9         LBAD on C-141       ABPZA001       0.15       0.85       0         ABPZA002       0       0       0       0         ABPZC003       0       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC005       0       0       0         LBAD on C-5       APBZA002       0       0       0         ABPZF003       0.25       0.075       0.675       0.675         ABPZC004       0.25       0.75       0.675       0.675         ABPZC003       0.25       0.075       0.675       0.675         ABPZC004       0.25       0.75       0.675       0.675         ABPZC005       0       0.11       0.9       0         LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZC003       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC003       0.25       0.01875                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 777. °                                    |               | ABKGC004   | 0.25                  | 0.075                            | 0.675        |
| LBAD on C-141 ABPZA001 0.15 0.85 0<br>ABPZA002 0 0 0<br>ABPZF003 0 0<br>ABPZC004 0.25 0.075 0.675<br>ABPZC005 0 0 0<br>LBAD on C-5 APBZA001 0.15 0.85 0<br>ABPZF003 0.25 0.075 0.675<br>ABPZC004 0.25 0.75 0.675<br>ABPZC004 0.25 0.75 0.675<br>ABPZC005 0 0.1 0.9<br>LBAD on C-141 ABRZA001 0.15 0.85 0<br>ABRZA002 0 0 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0<br>LBAD on C-5 ABRZA001 0.15 0.85 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0<br>LBAD on C-5 ABRZA001 0.15 0.85 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0.025 0.975                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                           |               | ABDGC005   | 0                     | 0.1                              | 0.9          |
| ABPZA002       0       0       0         ABPZF003       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC005       0       0       0         LBAD on C-5       APBZA001       0.15       0.85       0         ABPZC003       0.25       0.075       0.675         ABPZA002       0       0       0         ABPZC004       0.25       0.75       0.675         ABPZC005       0       0.11       0.9         LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0         ABRZC005       0       0       0         ABRZC005                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                           | LBAD on C-141 | ABPZA001   | 0.15                  | 0.85                             | 0            |
| ABPZF003       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC005       0       0       0         LBAD on C-5       APBZA001       0.15       0.85       0         ABPZC003       0.25       0.075       0.675         ABPZA002       0       0       0         ABPZC004       0.25       0.75       0.675         ABPZC004       0.25       0.75       0.675         ABPZC005       0       0.11       0.9         LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZF003       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZA002       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC005       0       0       0       0         ABRZC005       0       0.025       0.975 <td></td> <td></td> <td>ABPZA002</td> <td>0</td> <td>0</td> <td>0</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                           |               | ABPZA002   | 0                     | 0                                | 0            |
| ABPZC004       0.25       0.075       0.675         ABPZC005       0       0       0         LBAD on C-5       APBZA001       0.15       0.85       0         ABPZC002       0       0       0       0         ABPZC004       0.25       0.075       0.675         ABPZC004       0.25       0.075       0.675         ABPZC004       0.25       0.75       0.675         ABPZC005       0       0.11       0.9         LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC005       0       0       0       0         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC005       0       0       0       0         ABRZC005       0       0.25<                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                           |               | ABPZF003   | 0                     | 0                                | 0            |
| ABPZC005       0       0       0         LBAD on C-5       APBZA001       0.15       0.85       0         ABPZA002       0       0       0       0         ABPZC003       0.25       0.075       0.675         ABPZC004       0.25       0.75       0.675         ABPZC005       0       0.1       0.9         LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZC003       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0         ABRZC005       0       0       0.7312         ABRZC005       0       0.025       0.975         Note:       Z = H mistard; G GB nerve; V VX nerve       VX                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                           |               | ABPZC004   | 0.25                  | 0.075                            | 0.675        |
| LBAD on C-5 APBZA001 0.15 0.85 0<br>ABPZA002 0 0 0<br>ABPZF003 0.25 0.075 0.675<br>ABPZC004 0.25 0.75 0.675<br>ABPZC005 0 0.1 0.9<br>LBAD on C-141 ABRZA001 0.15 0.85 0<br>ABRZA002 0 0 0 0<br>ABRZF003 0 0 0<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0 0<br>LBAD on C-5 ABRZA001 0.15 0.85 0<br>ABRZA002 0 0 0 0<br>ABRZA002 0 0 0 0<br>ABRZC005 0 0 0 0<br>LBAD on C-5 ABRZA001 0.15 0.85 0<br>ABRZA002 0 0 0 0<br>ABRZF003 0.25 0.01875 0.7312<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0 0.025 0.975                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                           |               | ABPZC005   | 0                     | 0                                | 0            |
| ABPZ OOL       0.15       0.05       0         ABPZAOO2       0       0       0         ABPZFO03       0.25       0.075       0.675         ABPZCO04       0.25       0.75       0.675         ABPZCO05       0       0.11       0.9         LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC003       0.25       0.01875       0.7312         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0.025       0.975                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                           | TRAD on C-5   | ADR7 4001  | 0 15                  | 0.85                             | 0            |
| ABPZF003       0.25       0.075       0.675         ABPZC004       0.25       0.75       0.675         ABPZC005       0       0.1       0.9         LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZF003       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0         ABRZC004       0.25       0.0187       0.7312         ABRZC005       0       0       0         ABRZF003       0.25       0.0187       0.7312         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0.025       0.975         Note:       Z = H mustard; G GB nerve; V VX nerve       VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                           |               | ARP7 A002  | 0                     | 0                                | ñ            |
| ABPZC004       0.25       0.75       0.675         ABPZC005       0       0.1       0.9         LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZF003       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0         ABRZC005       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0         ABRZA002       0       0       0         ABRZC005       0       0       0         ABRZC005       0       0       0         ABRZC005       0       0       0         ABRZC005       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0.025       0.975         Note:       Z = H mustard; G GB nerve; V VX nerve       VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                           |               | ABPZFOOR   | 0.25                  | 0.075                            | 0.675        |
| ABPZC005       0       0.1       0.9         LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZA002       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZA002       0       0       0       0         ABRZA002       0       0       0       0         ABRZC005       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0.25       0.975         Note:       Z = H mustard; G GB nerve; V VX nerve       VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                           |               | ABPZC004   | 0.25                  | 0.75                             | 0.675        |
| LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZA002       0       0       0       0         ABRZF003       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0         LBAD on C-5       ABRZA001       0.15       0.85       0         ABRZA002       0       0       0       0         ABRZO05       0       0       0       0         ABRZO05       0       0       0       0         ABRZO05       0       0       0       0         ABRZO04       0.25       0.0187       0.7312         ABRZC005       0       0.25       0.01875       0.7312         ABRZC005       0       0.025       0.975         Note:       Z = H mustard; G GB nerve; V VX nerve       VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                           |               | ABPZC005   | 0                     | 0.1                              | 0.9          |
| LBAD on C-141       ABRZA001       0.15       0.85       0         ABRZA002       0       0       0       0         ABRZF003       0       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0         LBAD on C-5       ABRZA001       0.15       0.85       0         ABRZA002       0       0       0       0         ABRZC005       0       0       0       0         ABRZC005       0       0.15       0.85       0         ABRZC005       0       0       0       0         ABRZC004       0.25       0.0187       0.7312         ABRZC005       0       0.025       0.975         Note:       Z = H mustard; G GB nerve; V VX nerve       VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                           |               |            | 0.15                  | 0.05                             | <u>^</u>     |
| ABRZAUUZ       0       0       0         ABRZF003       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0         LBAD on C-5       ABRZA001       0.15       0.85       0         ABRZA002       0       0       0       0         ABRZF003       0.25       0.01875       0.7312         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0.025       0.975         Note:       Z = H mustard; G GB nerve; V VX nerve       VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                           | LBAD on C-141 | ABRZA001   | 0.12                  | 0.85                             | U            |
| ABRZF003       0       0       0         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0         LBAD on C-5       ABRZA001       0.15       0.85       0         ABRZA002       0       0       0       0         ABRZF003       0.25       0.01875       0.7312         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0.025       0.975         Note:       Z = H mustard; G GB nerve; V VX nerve       VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                           |               | ABKLAUU2   | U                     | U                                | U            |
| ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0       0         LBAD on C-5       ABRZA001       0.15       0.85       0         ABRZA002       0       0       0       0         ABRZF003       0.25       0.0187       0.7312         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0.025       0.975         Note:       Z = H mustard; G GB nerve; V VX nerve       VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                           |               | ABRZCOOL   |                       | U<br>0 01975                     | U<br>0 73135 |
| LBAD on C-5       ABRZA001       0.15       0.85       0         ABRZA002       0       0       0       0         ABRZF003       0.25       0.0187       0.7312         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0.025       0.975         Note:       Z = H mustard; G GB nerve; V VX nerve       VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                           |               | ABRZCOU4   | 0.23                  | 0.010/2                          | 0.73123      |
| LBAD on C-5 ABRZA001 0.15 0.85 0<br>ABRZA002 0 0 0<br>ABRZF003 0.25 0.0187 0.7312<br>ABRZC004 0.25 0.01875 0.7312<br>ABRZC005 0 0.025 0.975<br>Note: Z = H mustard; G GB nerve; V VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                           |               | MILLOUD    | v                     | v                                | v            |
| ABRZA002       0       0       0         ABRZF003       0.25       0.0187       0.7312         ABRZC004       0.25       0.01875       0.7312         ABRZC005       0       0.025       0.975         Note:       Z = H mustard; G GB nerve; V VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                           | LBAD on C-5   | ABRZA001   | 0.15                  | 0.85                             | 0            |
| ABRZF003         0.25         0.0187         0.7312           ABRZC004         0.25         0.01875         0.7312           ABRZC005         0         0.025         0.975           Note:         Z = H mustard;         G GB nerve;         V VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                           |               | ABRZA002   | 0                     | 0                                | 0            |
| ABRZC004         0.25         0.01875         0.7312           ABRZC005         0         0.025         0.975           Note:         Z = H mustard;         G GB nerve;         V VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                           |               | ABRZF003   | 0.25                  | 0.0187                           | 0.73125      |
| ABRZCO05         0         0.025         0.975           Note:         Z = H mustard;         G GB nerve;         V VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                           |               | ABRZC004   | 0.25                  | 0.01875                          | 0.73125      |
| Note: Z = H mustard; G GB nerve; V VX nerve                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                           |               | ABRZC005   | 0                     | 0.025                            | 0.975        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                           | Note: Z =     | H mustard; | G GB nerv             | e; V VX nerve                    | <u> </u>     |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 1. J. |               |            |                       |                                  |              |

TABLE 10-12 MARINE TRANSPORT AGENT RELEASES

|                               |                                                                                                                              | Barge                         | Lash<br>                         | Lash                             | Lash<br>. c                      |
|-------------------------------|------------------------------------------------------------------------------------------------------------------------------|-------------------------------|----------------------------------|----------------------------------|----------------------------------|
| Sequence                      | Variable                                                                                                                     | BI                            | <b>L1</b>                        | 2                                | ŝ                                |
| 001, 002<br>Collisions        | No. TCs failed<br>Inventory involved, lb<br>Evaporation release, lb<br>Fire release, lb<br>Percent failed inventory floating | 8<br>13,600<br>175<br>0<br>5  | 56<br>95,20<br>1,225<br>0<br>5   | 56<br>95,200<br>1,225<br>0<br>5  | 56<br>95,200<br>1,225<br>5       |
| 003, 004<br>Collisions + fire | No. TCs failed<br>Inventory involved, lb<br>Evaporation release, lb<br>Fire release, lb<br>Percent failed inventory floating | 8<br>13,600<br>0<br>170<br>25 | 56<br>95,200<br>0<br>1,190<br>25 | 56<br>95,200<br>0<br>1,190<br>25 | 56<br>95,200<br>0<br>1,190<br>25 |
| 005, 006<br>Rammings          | No. TCs failed<br>Inventory involved, lb<br>Evapoartion release, lb<br>Fire release, lb<br>Percent failed inventory floating | 4<br>6,800<br>87.5<br>5       | 32<br>54,400<br>0<br>5           | 32<br>54,400<br>700<br>5         | 32<br>54,400<br>700<br>5         |
| 007, 008<br>Rammings + fire   | No. TCs failed<br>Inventory involved, lb<br>Evaporation release, lb<br>Fire release, lb<br>Percent failed inventory floating | 4<br>6,800<br>0<br>25<br>25   | 32<br>54,400<br>0<br>25<br>25    | 32<br>54,400<br>0<br>25<br>25    | 32<br>54,400<br>680<br>25        |
| 009 through 012<br>Groundings | No. TCs failed<br>Inventory involved, lb<br>Evapoartion release, lb<br>Fire release, lb<br>Percent failed inventory floating | 4<br>6,800<br>0<br>0          | 68<br>115,600<br>0<br>0          | 68<br>115,600<br>0<br>0          | 68<br>115,600<br>0<br>0          |

10-42

×.

 $\langle \hat{\mathcal{N}} \rangle$ 



10.00.0000000000



TABLE 10-12 (Continued)

|                       |                                   | Barge               | Lash                  | Lash                  | Lash                  |
|-----------------------|-----------------------------------|---------------------|-----------------------|-----------------------|-----------------------|
| Sequence              | Variable                          | BI                  | LI                    | ΓC                    | LS                    |
| 013, 014              | No. TCs failed                    | -                   | 68                    | 68                    | 68                    |
| Weater damage         | Inventory involved, lb            | 1,700               | 115,600               | 115,600               | 115,600               |
|                       | Evaporation release, lb           | 21.9                | 1,488                 | 1,488                 | 1,488                 |
|                       | Fire release, lb                  | 0                   | 0                     | 0                     | 0                     |
|                       | Percent failed inventory floating | Ś                   | ŝ                     | Ś                     | S                     |
| 015, 016              | No. TCs failed                    | -1                  | 68                    | 68                    | 68                    |
| Weather damage + fire | Inventory involved, Ib            | 1,700               | 115,600               | 115,600               | 115,600               |
|                       | Evaporation release, lb           | 0                   | 0                     | 0                     | 0                     |
|                       | Fire release, lb                  | 21.3                | 1,445                 | 1,445                 | 1,445                 |
|                       | Percent failed inventory floating | 25                  | 25                    | 25                    | 25                    |
| 017, 018              | No. TCs failed                    | I                   | -                     | Ч                     |                       |
| Spontaneous fire      | Inventory involved, lb            | 1,700               | 1,700                 | 1,700                 | 1,700                 |
|                       | Evaporation release, lb           | 0                   | 0                     | 0                     | 0                     |
|                       | Fire release, lb                  | 85                  | 85                    | 85                    | 85                    |
|                       | Deton release                     | 0                   | 0                     | 0                     | 0                     |
|                       | Percent failed inventory floating | 100                 | 100                   | 100                   | 100                   |
| 019 through 022       | No. TCs failed                    | 0                   | 0                     | 0                     | 0                     |
|                       | Inventory involved, Ib            | 0                   | 0                     | 0                     | 0                     |
|                       | Evaporation release, lb           | 0                   | 0                     | 0                     | 0                     |
|                       | Fire release, 1b                  | 0                   | 0                     | 0                     | 0                     |
|                       | Percent failed inventory floating | N/A                 | N/A                   | N/A                   | N/A                   |
| 023                   | No. TCs failed                    | 560                 | 2,240                 | 2,240                 | 2,240                 |
| Aircraft              | Inventory involved, lb            | $9.5 \times 10^{5}$ | 3.8 x 10 <sup>6</sup> | 3.8 x 10 <sup>6</sup> | 3.8 x 10 <sup>6</sup> |
|                       | Evaporation release, lb           | 0                   | 0                     | 0                     | 0                     |
|                       | Fire release, lb                  | 0                   | 0                     | 0                     | 0                     |
|                       | Deton release, lb                 | 0                   | 0                     | 0                     | 0                     |

early thresholds of munition failure relative to the data (Appendix F), (2) worst-case number of adjacent munition ruptures for a munition detonation in a pallet, (3) use of maximum rather than average inventories, and (4) upper bound fire release factors, relative to the data.

ì



# 10.3. REFERENCES

- 10-1. Arthur D. Little, Inc., "Analysis of Risks Associated with Operation of Onsite M55 Rocket Demilitarization Plans," U.S. Army Toxic and Hazardous Materials Agency, M55-OD-3, May 1985.
- 10-2. Arthur D. Little, Inc., "Analysis of Risks Associated with Operation of the JACADS M55 Rocket Demilitarization Plant," U.S. Army Toxic and Hazardous Materials Agency, M55-0D-9, May 1985.
- 10-3. Arthur D. Little, Inc., "Analysis of Converted M55 Rocket Demilitarization Plant at Pine Bluff Arsenal," U.S. Army Toxic and Hazardous Materials Agency, M55-0D-10, May 1985.
- 10-4. Arthur D. Little, Inc., "Analysis of Risks Associated with Operation of the CAMDS M55 Rocket Demilitarization Plant," U.S. Army Toxic and Hazardous Materials Agency, M55-0D-11, May 1985.
- 10-5. Arthur D. Little, Inc., "M55 Rocket Separation Study," U.S. Army Toxic and Hazardous Materials Agency, M55-OD-08, November 1985.
- 10-6. "Design Development Test Results: Cryo-Pretreatment System Explosive Propagation Tests," GA Technologies Inc. Report 908644, February 1986.
- 10-7. "Black Hills Army Depot Hazard Classification Tests, USATECOM Project 5-3-0135-03," Disposition Form, STEDP-TO, Test Division, Dugway Proving Ground, Dugway, Utah, to Chief, Test Design and Analysis Office, DPG, June 3, 1966.
- 10-8. "Risk/Hazard Analysis for Hypothetical Accident/Incident at Selected Locations Along the CONUS Shipment Route for Project Red Hat," U.S. Army Munitions Command Operations Research Group Report 42, March 1970.
- 10-9. Taylor, W. T., and F. J. Massaro, "Hazard Classification Tests of Igloo Storage of GB-VX-Filled M55 Rockets," Department of the Army, Dugway Proving Ground Data Report (DPGDR)C505, USATECOM Project 5-3-0135-02, April 1966.

- 10-10. "Supporting Calculations for the Risk Analysis of the Disposal of Chemical Munitions at National or Regional Sites," GA Technologies Inc., EPS-GA/DP-0092 (Issue 2), Volumes 1, 2, and 3, August 1987.
- 10-11. Whitacre, C. G., et al., "Personal Computer Program for Chemical Hazard Prediction (D2PC), U.S. Army AMCC Report CRDEC-TR-87021, January 1987.

### 11. RESULTS

The analysis of the potential for agent release to the atmosphere from accident scenarios related to the collocation disposal option included the following major activities: (1) storage, (2) handling activities associated with the transport of munitions, (3) onsite transportation, (4) offsite transportation, and (5) plant operations associated with the demilitarization of munitions. This section discusses some of the accident probability and agent release results associated with these activities.

The results of the analysis of the various activities encompassing the collocation options cannot be presented in the same units, i.e., annual frequencies, because of the possible divulgence of classified information. This is only possible for some storage and plant operation accident scenarios. For accident scenarios related to the handling activities either at the original site, the regional site, or the national site, the unclassified portion of the probabilistic analysis is given in terms of frequency of accidents per pallet of munitions (or as a container of munitions). For onsite and offsite transportation accidents, the basic results are reported in terms of accident frequency per vehicle mile. These probabilities/unit are then multiplied by the number of handling operations or vehicle miles traveled during the stockpile disposal program.

The evaluation of the actual risk to the public and environment requires agent dispersion calculations which are not in the scope of the study reported here. Despite this limitation, the results discussed herein still provide useful insights on the contributions of the various disposal activities to the risk of an agent release. These insights are discussed below.



### 11.1. ACCIDENT SCENARIOS DURING STORAGE

The collocation alternative requires some storage of munitions in their existing location prior to transportation to the disposal site. In addition, it requires storage of munitions in offsite transport containers at the sending and receiving sites and some storage at the disposal site before movement to the demilitarization facility.

# 11.1.1. Internal Events

833353 - 5322333

There were no significant internal event initiators of accidents during storage at the disposal site before movement to the demilitarization facility. Per unit operation, forklift drop accidents occur more frequently than forklift time punctures. Also, the use of a lifting beam instead of a time leads to an order of magnitude decrease in drop frequency.

### 11.1.2. External Events

These events involve accidents caused by natural phenomena or human activity affecting munitions in storage igloos, open storage areas, holding areas, or warehouses. If these are assumed to be full of munitions, the agent inventories range up to 100, 200, 1000, and 2000 tons, respectively, for storage igloos, holding areas, open areas, and warehouses. The most frequent external accidents having significant release involve mild intensity earthquakes or small airplane crashes (order depending on site). Amounts of available agent inventories released in these events are on the order of fractions of one percent or less (munition punctures, drops, etc.).

The largest releases occur for a large aircraft crash, a meteorite strike, or a severe earthquake, especially when a warehouse (at NAAP, TEAD, or UMDA) is involved. These can result in up to 10 percent of the agent inventory released for scenarios involving a fire which has

the potential (duration) for destroying the entire inventory of an igloo or warehouse. The munitions stored in warehouses contain only VX or mustard which have much slower evaporation rates than GB and hence are not easily dispersed into the atmosphere. Thus, warehouse scenarios involving only spills are not significant risk contributors. The warehouse at UMDA has the potential for the largest release. Meteorite strike-initiated sequence median frequencies are one to two orders of magnitude lower than the aircraft crash-induced sequence frequencies. As expected, munitions stored outdoors are generally more susceptible to large aircraft crashes than those stored in warehouses or igloos, but releases are lower. Both APG and PBA have ton containers stored outdoors, and the aircraft crash probabilities at these sites are somewhat higher than at the other sites. Igloos appear to provide only minimal protection from direct crashes of large planes, but releases are an order of magnitude lower. The releases are more severe if burstered munitions are involved.



# 11.2. ACCIDENT SCENARIOS DURING HANDLING

Included in the handling analysis are (1) single munition or pallet movements by hand, forklift, or other equipment; (2) packing or unpacking pallets into transportation containers; (3) loading and unloading packages from trucks, railcars, aircraft, or barges; or (4) loading and off-loading barges into the oceanfaring vessel (LASH).

There are twice as many handling operations at the receiving sites (RDC or NDC) involving collocated munitions that are not in any transportation container. Furthermore, there are more handling operations involving munitions in onsite transport containers (ONCs) than bare munitions or those in larger offsite transport containers (OFCs).

# 11.2.1. Handling for the Rail Alternative

The results indicate that dropped munitions, whether in palletized form or not, occur more frequently than either forklift tine puncture or forklift collision accidents. In fact, the frequency of forklift collision accidents which lead to the munitions falling off the forklift is an order of magnitude lower than the drop accidents. Furthermore, the type of clothing an operator is wearing while handling these munitions influence the drop frequency value. An operator wearing Level A clothing is more likely to commit an error that would cause the munition to be dropped than when he is wearing more comfortable clothing.

The results also indicate that spray tanks (in overpacks) have relatively higher drop frequencies than other munitions. This is largely due to the assumption that spray tanks will be lifted and moved to the truck (for loading or unloading) using forklift with tines. The drop frequency using the tines is an order of magnitude higher than with the use of lifting beams.

For bare munitions, the rockets seem to be the most prone to punctures from drops or forklift tine accidents. However, the ONC or OFC itself also affects the puncture probability. Because of its weight and larger surface area, the drop of an OFC increases the munition puncture probability by about a factor of 4 to 5 (depending on the munition type and packing density) when compared to a similar drop of an ONC. However, bare munitions have higher puncture probabilities than munitions in ONCs. This observation is of course not quite evident in the final results presented because there are more handling operations involving possible drops of ONCs than bare munitions.

Bulk items that are punctured lead to larger releases than other munitions such as projectiles or rockets. Bombs are of concern because they contain GB which evaporates more readily than the other agent types. The agent vapor releases range up to 400 lb (thermal failure of all munitions in an OFC).

Within the types of handling accidents, the events designated as HC, which are related to the packaging of munitions in ONCs or OFCs and their movement from storage (sending sites) to the munitions handling igloo (MHI) (receiving sites), predominate over handling accidents related to the facility (HF). This is largely because (1) there are more handling operations involved in the HC accidents, (2) HF accidents generally involve munitions in ONCs, which provides them with some protection from puncture, and (3) HF accidents involving bare munitions occur inside the munitions demilitarization buildup (MDB) which is designed for vapor containment; hence, including the probability of a detonation which destroys the vapor containment barrier, both the frequency of a release and the release itself are relatively lower.

The frequency results for the handling accidents could not be compared with the accidents from other activities, such as plant operations, because of differences in units. To get some perspective on how they compare on a yearly basis, we can estimate the number of pallets

that could be handled based on the plant annual processing rates. For illustrative purposes we calculate the number of bomb pallets that are required to meet the annual plant processing rate as:

5.4 bombs/h x 24 h/day x 5 day/week

x 52 week/yr /2 bombs/pallet = 16,848 pallets/yr

By multiplying the HCl sequence frequency for TEAD  $(1.2 \times 10^{-7})$ pallet) with the number of pallets/yr, the annual frequency is 2.0 x  $10^{-3}$ /yr. Thus, handling accidents which lead to significant agent releases (in particular, agent GB) are dominant risk contributors because of the relatively higher annual frequency values. Of course depending on the actual munition inventory, the value of annual frequency may either increase or decrease when converted to the more meaningful per stockpile basis.

# 11.2.2. Handling for the Air Option

The accident scenarios discussed for the rail option also apply to the air option. Since the air option involves only the movement of munitions from LBAD and APG to TEAD, agent releases from 155-mm projectiles, 8-in. projectiles, rockets and ton containers are of interest. The general observations noted in the discussion of the accident frequencies for the rail option (Section S.3.2.1) also apply here. The accident release is lower for the handling of these munitions since the amounts of GB agent contained in rockets and projectiles are quite small compared to bombs.

# 11.2.3. Handling for the Marine Option

For this option, the ton containers are placed in a transportation container (vault) that is different from the OFC; hence, the handling steps are somewhat different. There are eight sequences related to handling that were identified. Sequence HW34, which involves the dropping

of a lighter by a crane while loading into or unloading from the lighter aboard ship (LASH) vessel, has a relatively high frequency of  $6.0 \times 10^{-6}$ per shipment. The structural analysis indicates that dropping of the lighter from a height of about 70 ft onto an unyielding surface of the LASH vessel could cause the crushing of several ton containers inside the lighter. The agent will be confined in the interior of the ship, and the amount of agent released to the atmosphere is small. STATES STATES STATES STATES STATES

# 11.3. ACCIDENT SCENARIOS DURING PLANT OPERATIONS

Included in the analysis for this phase are all malfunctions during agent processing/incineration within the MDB or external events affecting drained and undrained agent in the MDB, including those in the unpack area (UPA) (up to  $10^4$  lb of agent available) and munitions awaiting processing in the MHI, up to  $3 \times 10^4$  lb of agent available. After unpacking, the munitions are processed by conveyor to the burster removal area, mine punch-and-drain area, projectile mortars disassembly area, rocket and burster shearing machines, mine machine for burster removal, a bulk item drain station, a toxic cubicle (TOX) agent storage tank, furnaces for explosive deactivation, metal parts decontamination, and agent and dunnage incinerators, as appropriate.

# 11.3.1. Internal Events

Because of the engineered safety features provided in the plant design, both the frequency of release and magnitude of release associated with accidents initiated by equipment failure and human error are relatively small. Among the large number of accident scenarios analyzed, the highest frequency scenario (P052) is initiated by an inadvertent feed of an unpunched burstered munition to the dunnage incinerator  $(10^{-2}/\text{yr}$  for mines;  $5 \times 10^{-3}/\text{yr}$  for other munitions). As a result of detonation, one burstered munition inventory is released to the atmosphere as vapor (only up to 15 lb of agent).

The largest amount of agent vapor release occurs for a metal parts furnace explosion (PO44) with ventilation failure (one bulk item inventory release, up to 1700 lb). However, this scenario was assessed to have a very low frequency, around  $10^{-10}/yr$ . Another event with up to several hundred pounds of vapor release is PO48, munition detonation in the explosive containment room vestibule with subsequent fire spreading to unpacked munitions. However, this scenario also has a low frequency, around  $10^{-9}/yr$ .
#### 11.3.2. External Events

Aircraft crashes dominate the external event frequency, and there is little difference between direct and indirect crashes. The small difference is attributed to offsetting effects. Although the indirect crash has smaller conditional probabilities of failures than the direct crash, the risk model utilizes a larger target area for the indirect crash. There is very little distinction in the frequency of aircraft crashes with or without fire, since historical data indicate that there is roughly a 50 percent chance that the crash of an aircraft will involve a fire. The frequency of a crash onto the MDB is considerably larger than that for the MHI because the surface area of the MDB is more than 30 times larger than the MHI.

The frequency of large aircraft crashes is estimated to be higher at ANAD than it is for TEAD. This impacts the regional versus national collocation option. The accident scenario involving the crash of an airplane onto the outdoor agent piping system for the modified CAMDS facility at TEAD has a frequency of about  $10^{-8}/yr$  with up to 55 lb of vapor release. This scenario includes both large and small aircraft crashes. The frequency of small aircraft (including helicopters) crashes is at least two orders of magnitude higher than the frequency of large aircraft crashes at TEAD.

The frequencies of earthquake-induced accident scenarios are generally higher for TEAD than for ANAD since TEAD is located in a region more prone to earthquakes. Sequence PO33, which represents an earthquake-initiated munition fall and fire but with the MDB and TOX intact, has the highest frequency (2 x  $10^{-6}$ /yr for ANAD and 5 x  $10^{-5}$ /yr for TEAD). This sequence involves the detonation of all munitions (if burstered) in the UPA since the fire is not suppressed in this sequence.

All accident sequences related to tornadoes or meteorites were estimated to occur at frequencies of less than  $10^{-10}/yr$  and thus were screened out.

11-9

# Ŵ

Receive

TO TELEVICES

2000/1111 [Jan (Charles and States and Charles and States and

#### 11.4. ACCIDENT SCENARIOS DURING TRANSPORT

#### 11.4.1. Onsite Transportation

6663345.21

There are two truck transportation phases considered in the analysis. At the sending sites, munitions in offsite transportation packages are transported by truck to the holding area prior to loading into the train, airplane, or barge. The accidents are identified as the VR, VA, or VW (i.e., for rail, air, and water, respectively) scenarios. At the receiving sites, munitions still in offsite packages are moved to storage locations where they are removed from the offsite package and stored until they are ready for demilitarization. The accidents are also coded VR or VA. Finally, when munitions at their storage locations are ready for demilitarization, they are transferred into onsite containers and then moved by truck to the MHI. The accidents are identified as VO scenarios to distinguish between the transportation risk of using an onsite package versus an offsite package (different failure thresholds). The agent available in a truck carrying an OFC is less than 3400 lb, while up to 7000 lb is available for an ONC truck transport.

As a result of analysis for both internally initiated events (human error or equipment failure) and externally initiated events, the following conclusions were reached:

- 1. The offsite transportation package provides munitions with more protection from crush forces generated from truck accidents than the onsite package. Hence, sequences with OFC crush have insignificant accident frequency whereas scenarios with ONC crush have frequencies up to  $10^{-8}$ /truck-mile.
- 2. Both packages provide similar protection from impact forces. The results show that accident frequencies resulting in impact failure are insignificant. This is largely due to the administrative control to be imposed during truck travel which





limits truck speed to no more than 20 mph. The impact forces at this velocity are not sufficient to breach the containment.

- 3. The probability of puncture resulting from truck collision/ overturn is at least an order of magnitude higher for offsite containers than onsite containers. This results from the higher likelihood of generating a probe sufficient to puncture the container and the munition when the accident involves a large package such as the OFC.
- 4. Truck accidents which generate fires are more likely to detonate burstered munitions inside onsite packages, since they provide only a 15-min protection from an all engulfing fire (versus 2 h for the OFC). However, all these scenario frequency results are also quite low because of the administrative control for limiting the amount of fuel in the truck so as not to exceed a 10-min fire.
- 5. When rockets are involved in the accidents which generate sufficient impact forces to cause propellant ignition, there is very little distinction in the results for the two packages.
- For tornado-initiated accidents, puncture as a result of truck overturn is the dominant contributor to the sequence frequency.
- 7. Generation of undue forces during truck accidents that could cause burster detonations has a small contribution to the overall truck transportation risk.
- 8. The amount of agent spilled or burned during truck accidents resulting in the breach in containment by puncture forces generally involve the agent content of one munition. Up to 10 percent is released as vapor.

9. Both containers can fail when an aircraft crashes into the truck (VR6, VR7, V06, V07). The entire truckload is involved, and up to 10 percent is released as a vapor. Hence, aircraft crash-initiated truck accidents have the most severe consequences. It should be noted, however, that none of the accident sequences has a frequency greater than 10<sup>-7</sup>/yr.

#### 11.4.2. Offsite Transport - Rail

In this option, munitions in OFCs are transported by rail either to two regional destruction centers (RDC-ANAD or RDC-TEAD) or a single national destruction center (NDC-TEAD). The agent inventory available per railcar ranges up to 7000 lb. Results of the accident analysis indicate the following:

- Rail accident crush and impact forces are very unlikely to fail an OFC and munition inside.
- 2. The major risk contribution due to mechanical failure comes from a probe such as a railcar coupler (generated from train accidents) capable of puncturing the OFC and the munition. Munition failure frequency by puncture (RC3) is about an order of magnitude higher than train accidents which lead to fire and cause the thermal detonation or rupture of munitions (RC4 and RC5). However, the consequence (i.e., agent release) from the latter sequence is more severe.
- 3. For tornado-initiated accidents (RC14), puncture as a result of train derailment is the dominant contributor to the agent release frequency.
- 4. Aircraft crash into a train can damage the munitions (RC6 and RC7). The crash can involve one or two railcars (i.e., up to four OFCs). The largest amounts of agent released are from the bulk items (bombs, ton containers, and spray tanks). A

maximum of 10 percent of the inventory is released as vapor (up to 1400 lb). This is the largest release for rail scenarios.

#### 11.4.3. Offsite Transport - Air Option

The air transport option applies only to the movement of ton containers from APG to TEAD, and rockets and projectiles from LBAD to TEAD. Five generic sequences related to air transport were identified. These scenarios were evaluated for both the C-141 and C-5 aircrafts. There will be approximately 1500 flights from LBAD and 300 flights from APG for the C-141 aircraft. The C-5 aircraft would decrease the number of required flights by one fourth. The analysis also differentiated among accidents which occur during takeoff, while in flight, and during landing. Each flight would carry up to 3400 lb of agent inside OFCs.

The aircraft accident frequency during landing is about seven times higher than during takeoff and about three times higher than inflight accidents. However, the failure probability of the package due to impact forces is higher inflight than either takeoff or landing. If an aircraft crash occurs, the OFC and the munitions are subjected primarily to impact forces sufficient to fail the package. The accident frequencies from sequences which involve impact only are almost of the same order of magnitude as sequences which involve impact and fire (AAl versus AA20). The accident frequencies involving the C-5 aircraft are an order of magnitude higher than those for C-141 aircraft. A compensating factor is that there will be 75 percent fewer flights if the C-5 is used.

Accident scenarios involving fire of sufficient duration to fail the packages are not credible for the C-141 aircraft because of insufficient fuel available to sustain a fire of duration to fail the package containment. Accidents which lead to severe impact (AA1 and AA2; AB1 and AB2) without fire have the highest frequency and also lead to the largest amounts of agent released. For severe impact release involving burstered munitions, some of the munitions contained in the aircraft will detonate, and up to just over 400 lb will be released as vapor. For accidents involving moderate impact forces, no agent release occurs from impact alone. The moderate impact accident must be accompanied by fire to fail the package thermally.

#### 11.4.4. Offsite Transport - Marine Option

The marine option was analyzed only for the movement of ton containers filled with mustard at APG to the Johnston Atoll. There were five groups of initiating events identified. Impact and puncture are not the dominant failure forces experienced in marine accidents. The cargo will be adequately braced to hold it in place. Furthermore, most of the events are low-velocity, high-momentum events; hence, the dominant failure mode is crush. Fire, immersion, and aircraft crash events were also considered because of the large amount of agent being transported which could be involved in fire or sinking accidents.

The results indicate that:

- For the lighters in the Chesapeake Bay, collision accidents are at least three orders of magnitude more probable than either rammings or groundings.
- 2. For the LASH vessel in the Chesapeake Bay, both grounding and collision accidents are at least one order of magnitude more probable than rammings.

TANKIN DIZZAN ZAGA

3. Grounding of the LASH vessel in the coastal areas is less likely than in shallower inland waters.  For the LASH vessel in high seas, collision is still the predominant event. However, grounding results in more severe consequences. REELECTION DISTRICTION

The agent release analysis shows that collisions result in the largest number of ton containers (TCs) which fail (8) for barges, but that groundings or heavy weather damage results in the maximum number of TCs failed (68) for the LASH (except for aircraft crash, which is below the frequency screening threshold). The largest amount of agent vapor release to the atmosphere occurs for these worst events, and the amounts are not strongly dependent on whether fire occurs or not. Although a large inventory (up to 4 million 1b on the LASH) is available, no accident leads to a release of more than 0.1 percent.

#### 11.5. UNCERTAINTIES IN THE ANALYSIS

In assessing the risks associated with the CSDP alternatives, every effort was made to perform best-estimate analyses, i.e., "realistic" evaluation and quantification of the accident sequence frequencies and associated agent releases. The use of pessimistic or conservative modeling techniques or data for quantification violates the intent of the probabilistic nature of the study. Realistic modeling and quantification permits a balanced evaluation of risk contributors and comparison of alternatives. However, for realistic or best-estimate calculations, the obvious concern is the accuracy of the results. Uncertainty analysis addresses this concern. Viller

#### 11.5.1. Sources of Uncertainty

Since the event sequences discussed in Section S.3 have not actually occurred, it is difficult to establish the frequency of the sequence and associated consequences with great precision. For this reason, many parameters in a risk assessment are treated as probabilistically distributed parameters, so that the computation of sequence frequencies and resulting consequences can involve the probabilistic combination of distributions.

There are three general types of uncertainty associated with the evaluations reported in this document: (1) modeling, (2) data, and (3) completeness.

There exist basic uncertainties regarding the ability of the various models to represent the actual conditions associated with the sequence of events for the accident scenarios that can occur in the storage and disposal activities. The ability to represent actual phenomena with analytical models is always a potential concern. The use of fundamental models such as fault trees and event trees is sometimes simplistic because most events depicted in these models are treated as leading to one of two binary states: success or failure (i.e., partial successes or failures are ignored). Model uncertainties are difficult to quantify and are addressed in this study by legitimate efforts of the analysts to make the models as realistic as possible. Where such realism could not be achieved, conservative approaches were taken.

No uncertainty from oversights, errors, or omission from the models used (e.g., event trees and fault trees) is included in the uncertainty analysis results. Including these uncertainties is beyond the state-ofthe-art of present day uncertainty analysis.

The uncertainties in the assignment of event probabilities (e.g., component failure rates and initiating event frequencies) are of two types: intrinsic variability and lack of knowledge. An example of intrinsic variability is that where the available experience data is for a population of similar components in similar environments, but not all the components exhibit the same reliability. Intrinsic variations can be caused, for example, by different manufacturers, maintenance practices, or operating conditions. A second example of intrinsic variability is that related to the effects of long-term storage on the condition of the munitions as compared to their original configuration. Lack of knowledge uncertainty is associated with cases where the model parameter is not a random or fluctuating variable, but the analyst simply does not know what the value of the parameter should be. Both of these data uncertainty types are encountered in this study.

#### 11.5.2. Uncertainties

The sequence frequency results discussed in this report are presented in terms of a median value and a range factor of a probability distribution representing the frequency of interest. The range factor represents the ratio of the 95th percentile value of frequency to the 50th percentile (i.e., median) value of frequency. The uncertainty in the sequence frequency is determined using the STADIC-2 program (Ref. S-4) to propagate the uncertainties associated with each of the events in the fault trees or event trees through to the end result. Some scenarios, such as those associated with tornado missiles and lowimpact detonations have rather large uncertainties. The difficulty with tornado-generated missiles lies with the difficulty in accurately modeling the probability that the missile will be in the proper orientation to penetrate the munition and in predicting the number of missiles per square foot of wind. The difficulty with the low-impact detonations lies with the sparse amount of data available and its applicability to the scenarios of interest. In general, uncertainties tend to be large when the amount of applicable data is small and vice versa.

LEADER TO LEADER LANDER



APPENDIX A **REFERENCE LIST OF ACCIDENT SCENARIOS**  22.23

A MARINA

1-0-0-0

#### A.1. REFERENCE LIST OF ACCIDENT SCENARIOS

A reference list of accident scenarios is presented here. The list is arranged by the particular demilitarization phase with which a given scenario is associated. Accident scenarios related to storage are presented first followed by plant operations, handling, onsite transport and offsite transport. The scenarios can be identified by the coding scheme presented in Section 4 of this document. Following the scenario ID, a brief description of the accident is given along with an indication as to whether or not the scenario was considered for further analysis. The bases for scenario screening are provided in the logic model section, Section 4, of the main body of this report. 

#### STORAGE ACCIDENT SEQUENCES

Variai reessed deressed jaalaan bareen bareen berekkinderen berekkinder berekkinderessed berekkindere

| Sequence<br>ID | Sequence Description                                                                                                                                                                                             | Considered<br>for Further<br>Analysis |
|----------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------|
| SL1            | Munition develops a leak during the in-between inspection period.                                                                                                                                                | Yes                                   |
| SL2            | Munition punctured by forklift tine during leaker-handling activities.                                                                                                                                           | Yes                                   |
| SL3            | Spontaneous ignition of rocket during storage (not analyzed for lack of quantitative data).                                                                                                                      | No                                    |
| SL4            | Large aircraft direct crash onto storage area;<br>fire not contained in 30 min. (Note: Assume<br>detonation occurs if burstered munitions hit;<br>fire involving burstered munitions not con-<br>tained at all.) | Yes                                   |
| SL5            | Large aircraft indirect crash onto storage area;<br>fire not contained in 30 min. (See note in SL4.)                                                                                                             | Yes                                   |
| SL6            | Tornado-generated missiles strike the storage magazine, warehouse, or open storage area; muni-tions breached (no detonation).                                                                                    | Yes                                   |
| SL7            | Severe earthquake breaches the munitions in stor-<br>age igloos; no detonations.                                                                                                                                 | Yes                                   |
| SL8            | Meteorite strikes the storage area; fire occurs;<br>munitions breached (if burstered, detonation also<br>occurs).                                                                                                | Yes                                   |
| SL9            | Munition dropped during leaker isolation oper-<br>ation; munition punctured.                                                                                                                                     | No                                    |
| SL10           | Storage igloo or warehouse fire from internal sources.                                                                                                                                                           | No                                    |
| SL11           | Munitions are dropped due to pallet degradation.                                                                                                                                                                 | No                                    |
| SL12           | Liquid petroleum gas (LPG) infiltrates igloo/<br>building.                                                                                                                                                       | No                                    |
| SL13           | Flammable liquids stored in nearby facilities<br>explode, fire propagates to munition warehouse<br>(applies to NAAP).                                                                                            | No                                    |

a stratiska medicina statiska markatar in 1993. A statiska markatar oleh a statiska 19 19

| Sequence<br>ID | Sequence Description                                                                                   | Considered<br>for Further<br>Analysis |
|----------------|--------------------------------------------------------------------------------------------------------|---------------------------------------|
| SL14           | Tornado-induced building collapse leads to breaching/detonation of munitions.                          | No                                    |
| SL15           | Small aircraft direct crash onto warehouse or open storage yard; fire occurs; not contained in 30 min. | No                                    |
| SL16           | Large aircraft direct crash; no fire; detona-<br>tion (if burstered).                                  | Yes                                   |
| SL17           | Large aircraft direct crash; fire contained within 30 min (applies to nonburstered muni-tions only).   | Yes                                   |
| SL18           | Small aircraft direct crash onto warehouse or open storage yard; no fire.                              | Yes                                   |
| SL19           | Small aircraft indirect crash onto warehouse or open storage yard; fire contained in 30 min.           | Yes                                   |
| SL20           | Large aircraft indirect crash onto storage area;<br>no fire.                                           | Yes                                   |
| SL21           | Large aircraft indirect crash onto storage area;<br>fire contained in 30 min.                          | Yes                                   |
| SL22           | Severe earthquake leads to munition detonation.                                                        | Yes                                   |
| SL23           | Tornado-generated missiles strike the storage igloo and leads to munition detonation.                  | Yes                                   |
| SL24           | Lightning strikes ton containers stored outdoors.                                                      | Yes                                   |
| SL25           | Munition dropped during leaker isolation; muni-<br>tion detonates.                                     | Yes                                   |
| SL261          | Earthquake occurs; NAAP warehouse is intact; no ton containers damaged; fire occurs.                   | Yes                                   |
| SL262          | Earthquake occurs; NAAP warehouse is intact; ton container damaged; no fire.                           | Yes                                   |
| SL263          | Earthquake occurs; NAAP warehouse is intact; ton container damaged; fire occurs.                       | Yes                                   |

ŝ

SCALLED

818265

B. C. C. S. S.

| Sequence<br>ID | Sequence Description                                                                                                                      | Considered<br>for Further<br>Analysis |
|----------------|-------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------|
| SL264          | Earthquake occurs; NAAP warehouse is damaged; ton containers damaged; fire occurs.                                                        | Yes                                   |
| SL265          | Earthquake occurs; NAAP warehouse is damaged; ton containers; fire occurs.                                                                | Yes                                   |
| SL271          | Earthquake occurs; TEAD warehouses intact; muni-<br>tions intact; fire occurs at one warehouse.                                           | Yes                                   |
| SL272          | Earthquake occurs; TEAD warehouses intact; muni-<br>tions intact; fire occurs at two warehouses.                                          | Yes                                   |
| SL273          | Earthquake occurs; one TEAD warehouse is damaged;<br>munitions intact; fire occurs at one warehouse.                                      | Yes                                   |
| SL274          | Earthquake occurs; one TEAD warehouse is dam-<br>aged; munitions intact; fire occurs at two<br>warehouses.                                | Yes                                   |
| SL275          | Earthquake occurs; two TEAD warehouses damaged;<br>munitions intact; fire occurs at one warehouse.                                        | Yes                                   |
| SL276          | Earthquake occurs; two TEAD warehouses damaged;<br>munitions intact; fire occurs at two warehouses.                                       | Yes                                   |
| SL281          | Earthquake occurs; UMDA warehouses intact; muni-<br>tions intact; fire occurs at one warehouse.                                           | Yes                                   |
| SL282          | Earthquake occurs; UMDA warehouses intact; muni-<br>tions intact; fire occurs at two warehouses.                                          | Yes                                   |
| SL283          | Earthquake occurs; UMDA warehouses intact; muni-<br>tions in one warehouse damaged; no fire occurs.                                       | Yes                                   |
| SL284          | Earthquake occurs; UMDA warehouses intact; muni-<br>tions in one warehouse damaged; fire occurs at<br>warehouse with damaged munitions.   | Yes                                   |
| SL285          | Earthquake occurs; UMDA warehouses intact; muni-<br>tions in one warehouse damaged; fire occurs at<br>warehouse with undamaged munitions. | Yes                                   |

<u>.</u>

A-4

| Sequence<br>ID | Sequence Description                                                                                                                       | Considered<br>for Further<br>Analysis |
|----------------|--------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------|
| SL286          | Earthquake occurs; UMDA warehouses intact; muni-<br>tions in one warehouse damaged; fire occurs at<br>two warehouses.                      | Yes                                   |
| SL287          | Earthquake occurs; UMDA warehouses intact; muni-<br>tions in two warehouses damaged; no fire occurs.                                       | Yes                                   |
| SL288          | Earthquake occurs; UMDA warehouses intact; muni-<br>tions in two warehouses damaged; fire occurs at<br>warehouse with damaged munitions.   | Yes                                   |
| SL289          | Earthquake occurs; UMDA warehouses intact; muni-<br>tions in two warehouses damaged; fire occurs at<br>two warehouses.                     | Yes                                   |
| SL2810         | Earthquake occurs; one UMDA warehouse damaged;<br>munitions in one warehouse damaged; no fire<br>occurs.                                   | Yes                                   |
| SL2811         | Earthquake occurs; one UMDA warehouse damaged;<br>munitions in one warehouse damaged; fire occurs<br>at warehouse with damaged munitions.  | Yes                                   |
| SL2812         | Earthquake occurs; one UMDA warehouse damaged;<br>munitions in one warehouse damaged; fire occurs<br>at two warehouses.                    | Yes                                   |
| SL2813         | Earthquake occurs; one UMDA warehouse damaged;<br>munitions in two warehouses damaged; no fire<br>occurs.                                  | Yes                                   |
| SL2814         | Earthquake occurs; one UMDA warehouse damaged;<br>munitions in two warehouses damaged; fire occurs<br>at warehouse with damaged munitions. | Yes                                   |
| SL2815         | Earthquake occurs; one UMDA warehouse damaged;<br>munitions in two warehouses damaged; fire occurs<br>at two warehouses.                   | Yes                                   |
| SL2816         | Earthquake occurs; two UMDA warehouses damaged;<br>munitions in two warehouses damaged; no fire                                            | Yes                                   |

١

| Sequence<br>ID   | Sequence Description                                                                                                       | Considered<br>for Further<br>Analysis |
|------------------|----------------------------------------------------------------------------------------------------------------------------|---------------------------------------|
| SL2817           | Earthquake occurs; two UMDA warehouses damaged;<br>munitions in two warehouses damaged; fire occurs<br>at both warehouses. | Yes                                   |
| <u>Rail Opti</u> | on                                                                                                                         |                                       |
| SR1              | Large aircraft direct crash onto transportation containers in holding area; no fire.                                       | Yes                                   |
| SR2              | Large aircraft direct crash onto transportation containers in holding area; fire not contained.                            | Yes                                   |
| SR3              | Large aircraft direct crash onto transportation containers in holding area; fire contained.                                | Yes                                   |
| SR4              | Small aircraft direct crash onto transportation containers in holding area; no fire.                                       | Yes                                   |
| SR5              | Small aircraft direct crash onto transportation containers in holding area; fire not contained.                            | Yes                                   |
| SR6              | Small aircraft direct crash onto transportation containers in holding area; fire contained.                                | Yes                                   |
| SR7              | Tornado-generated missiles strike munitions in transportation containers in holding area; no detonation.                   | Yes                                   |
| SR8              | Tornado-generated missiles strike munitions in holding area; detonation occurs.                                            | Yes                                   |
| SR9              | Meteorite strikes munitions in transportation containers in holding area; fire occurs; detonation (if burstered).          | Yes                                   |
| Air Optio        | <u>on</u>                                                                                                                  |                                       |
| SA1              | Large aircraft direct crash onto transportation containers in holding area; no fire.                                       | Yes                                   |
| SA2              | Large aircraft direct crash onto transportation containers in holding area; fire not contained.                            | Yes                                   |
| SA3              | Large aircraft direct crash onto transportation containers in holding area; fire contained.                                | Yes                                   |

A-6

いのいれたという

でいることで

2

| Sequence<br>ID   | Sequence Description                                                                                                      | Considered<br>for Furthe<br>Analysis |
|------------------|---------------------------------------------------------------------------------------------------------------------------|--------------------------------------|
| SA4              | Small aircraft direct crash onto transportation containers in holding area; no fire.                                      | Yes                                  |
| SA5              | Small aircraft direct crash onto transportation containers in holding area; fire not contained.                           | Yes                                  |
| SA6              | Small aircraft direct crash onto transportation containers in holding area; fire contained.                               | Yes                                  |
| SA7              | Tornado-generated missiles strike munitions in transportation containers in holding area; no detonation.                  | Yes                                  |
| SA8              | Tornado-generated missiles strike munitions in holding area; detonation occurs.                                           | Yes                                  |
| SA9              | Meteorite strikes munitions in transportation<br>containers in holding area; fire occurs; detona-<br>tion (if burstered). | Yes                                  |
| <u>Marine Op</u> | otion                                                                                                                     |                                      |
| SW1              | Large aircraft direct crash cnto transportation containers in holding area; no fire.                                      | Yes                                  |
| SW2              | Large aircraft direct crash onto transportation containers in holding area; fire not contained.                           | Yes                                  |
| SW3              | Large aircraft direct crash onto transportation containers in holding area; fire contained.                               | Yes                                  |
| SW4              | Small aircraft direct crash onto transportation containers in holding area; no fire.                                      | Yes                                  |
| SW5              | Small aircraft direct crash onto transportation containers in holding area; fire not contained.                           | Yes                                  |
| SW6              | Small aircraft direct crash onto transportation containers in holding area; fire contained.                               | Yes                                  |
| SW7              | Tornado-generated missiles strike munitions in transportation containers in holding area; no detonation.                  | Yes                                  |

| Sequence<br>ID | Sequence Description                                                                   | Considered<br>for Further<br>Analysis |
|----------------|----------------------------------------------------------------------------------------|---------------------------------------|
| SW9            | Meteorite strikes munitions in transportation containers in holding area; fire occurs. | Yes                                   |
| SW10           | Large aircraft direct crash onto a flotilla of lighters; no fire.                      | Yes                                   |
| SW11           | Large aircraft direct crash onto a flotilla of lighters; fire not contained.           | Yes                                   |
| SW12           | Large aircraft direct crash onto a flotilla of lighters; fire contained.               | Yes                                   |
| SW13           | Small aircraft direct crash onto a flotilla of lighters; no fire.                      | Yes                                   |
| SW14           | Small aircraft direct crash onto a flotilla of lighters; fire not contained.           | Yes                                   |
| SW15           | Small aircraft direct crash onto a flotilla of lighters; fire contained.               | Yes                                   |
| SW16           | Large aircraft direct crash onto LASH vessel (at rest); no fire.                       | Yes                                   |
| SW17           | Large aircraft direct crash onto LASH vessel (at rest); fire not contained.            | Yes                                   |
| SW18           | Large aircraft direct crash onto LASH vessel (at rest); fire contained.                | Yes                                   |
| SW19           | Small aircraft direct crash onto LASH vessel (at rest); no fire.                       | Yes                                   |
| SW20           | Small aircraft direct crash onto LASH vessel (at rest); fire not contained.            | Yes                                   |
| SW21           | Small aircraft direct crash onto LASH vessel (at rest); fire contained.                | Yes                                   |

A-8



12.22

A A

#### OFFSITE TRANSPORT ACCIDENT SEQUENCES

| Sequence<br>ID   | Sequence Description                              | Considered<br>for Further<br>Analysis |
|------------------|---------------------------------------------------|---------------------------------------|
| <u>Main Tran</u> | sport                                             |                                       |
| BIKHS001         | A collision occurs and crush forces fail agent    | No                                    |
| LIKHS001         | containment.                                      | Yes                                   |
| LCKHS001         |                                                   | Yes                                   |
| LSKHS001         |                                                   | Yes                                   |
| BIKHS002         | A collision occurs and crush forces fail agent    | Yes                                   |
| LIKHS002         | containment. Sinking also occurs.                 | Yes                                   |
| LCKHS002         |                                                   | Yes                                   |
| LSKHS002         |                                                   | Yes                                   |
| BIKHC003         | A collision occurs and crush forces fail agent    | No                                    |
| LIKHC003         | containment. A fire breaks out.                   | Yes                                   |
| LCKHC003         |                                                   | Yes                                   |
| LSKHC003         |                                                   | Yes                                   |
| BIKHC004         | A collision occurs and crush forces fail agent    | Yes                                   |
| LIKHC004         | containment. A fire breaks out and sinking        | Yes                                   |
| LCKHC004         | occurs.                                           | Yes                                   |
| LSKHC004         |                                                   |                                       |
| BIKHS005         | A ramming occurs and crush forces fail agent      | No                                    |
| LIKHS005         | containment.                                      | Yes                                   |
| LCKHS005         |                                                   | Yes                                   |
| LSKHS005         |                                                   | Yes                                   |
| BIKHS006         | A ramming occurs and crush forces fail agent      | Yes                                   |
| LIKHS006         | containment. Sinking also occurs.                 | Yes                                   |
| LCKHS006         |                                                   | Yes                                   |
| LSKHS006         |                                                   | Yes                                   |
| BIKHC007         | A ramming accident occurs and crush forces fail   | No                                    |
| LIKHC007         | agent containment. A fire breaks out.             | Yes                                   |
| LCKHC007         | -                                                 | Yes                                   |
| LSKHC007         |                                                   | Yes                                   |
| BIKHC008         | A ramming accident occurs and crush forces fail   | Yes                                   |
| LIKHC008         | agent containment. A fire breaks out and sinking  | Yes                                   |
| LCKHS008         | occurs.                                           | Yes                                   |
| LSKHC008         |                                                   | Yes                                   |
| BIKHS009         | A grounding accident occurs and crush forces fail | No                                    |
| LIKHS009         | agent containment.                                | Yes                                   |
| LCKHS009         |                                                   | Yes                                   |
| LSKHS009         |                                                   | Yes                                   |

12222

8.50.02

A KAN TSAKASA TSAKASA TSAKASA TSAKASA TSAKASA TSAKASA TSAKASA TSAKASA TAANA TAANA TAANA TAANA TAANA TAANA TAANA

### OFFSITE TRANSPORT ACCIDENT SEQUENCES (Continued)

Ş

2

でくののようで

| Sequence<br>ID                               | Sequence Description                                                                                                            | Considered<br>for Further<br>Analysis |
|----------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------|---------------------------------------|
| BIKHS010<br>LIKHS010<br>LCKHS010<br>LSKHS010 | A grounding accident occurs and crush forces fail agent containment. Sinking also occurs.                                       | Yes<br>Yes<br>Yes<br>Yes              |
| BIKHC011<br>LIKHC011<br>LCKHC011<br>LSKHC011 | A grounding accident occurs and crush forces fail agent containment. A fire breaks out.                                         | No<br>Yes<br>Yes<br>Yes               |
| BIKHC012<br>LIKHC012<br>LCKHC012<br>LSKHC012 | A grounding accident occurs and crush forces fail<br>agent containment. A fire breaks out and sinking<br>occurs.                | Yes<br>Yes<br>Yes<br>Yes              |
| BIKHS013<br>LIKHS013<br>LCKHS013<br>LSKHS013 | Structural damage due to heavy weather occurs.<br>Crush forces fail agent containment.                                          | No<br>Yes<br>Yes<br>Yes               |
| BIKHS014<br>LIKHS014<br>LCKHS014<br>LSKHS014 | Structural damage due to heavy weather occurs.<br>Crush forces fail agent containment. Sinking<br>occurs.                       | Yes<br>Yes<br>Yes<br>Yes              |
| BIKHC015<br>LIKHC015<br>LCKHC015<br>LSKHC015 | Structural damage due to heavy weather occurs.<br>Crush forces fail agent containment. A fire<br>breaks out.                    | No<br>Yes<br>Yes<br>Yes               |
| BIKHC016<br>LIKHC016<br>LCKHC016<br>LSKHC016 | Structural damage due to heavy weather occurs.<br>Crush forces fail agent containment. A fire<br>breaks out and sinking occurs. | Yes<br>Yes<br>Yes<br>Yes              |
| BIKHF017<br>LIKHF017<br>LCKHF017<br>LSKHF017 | Spontaneous fire occurs.                                                                                                        | No<br>Yes<br>Yes<br>Yes               |
| BIKHC018<br>LIKHC018<br>LCKHC018<br>LSKHC018 | Spontaneous fire occurs. Sinking also occurs.                                                                                   | No<br>Yes<br>Yes<br>Yes               |

A-10

~~~~



y

Ķ

P

OFFSITE TRANSPORT ACCIDENT SEQUENCES (Continued)

| Sequence ID | Sequence Description | Considered for Further Analysis |
|------------------|---|---------------------------------------|
| BIKHS019 | Collision accident occurs with no immediate | Yes |
| LIKHS019 | release. Sinking also occurs. | Yes |
| LCKHS019 | • | Yes |
| LSKHS019 | | Yes |
| BIKHS020 | Ramming accident occurs with no immediate | Yes |
| LIKHS020 | release. Sinking also occurs. | Yes |
| LCKHS020 | - | Yes |
| LSKHS020 | | Yes |
| BIKHS021 | Grounding accident occurs with no immediate | Yes |
| LIKHS021 | release. Sinking also occurs. | Yes |
| LCKHS021 | | Yes |
| LSKHS021 | | Yes |
| BIKHS022 | Structural damage due to heavy weather occurs | Yes |
| LIKHS022 | with no immediate release. Sinking also occurs. | Yes |
| LCKHS022 | | Yes |
| LSKHS022 | | Yes |
| <u>Rail Tran</u> | sport | |
| RCYZW001 | A train accident involving a munitions railcar occurs and crush forces fail the agent containment. | No |
| RCYZW002 | A train accident involving a munitions railcar occurs and impact forces fail the agent containment. | No |
| RCYZW003 | A train accident involving a munitions railcar occurs and puncture forces fail the agent containment. | Yes |
| RCYZW004 | A train accident with fire occurs. Either the package insulation is torn away due to mechanical forces and the fire is able to heat the munitions inside the package, or the fire lasts long enough to cause burstered munitions in the package to detonate. Undue force created by the accident may also cause burster detonation. | Yes |

A-11

OFFSITE TRANSPORT ACCIDENT SEQUENCES (Continued)

55

222222

 \overline{R}

| Sequence ID | Sequence Description | Considered for Further Analysis |
|----------------|--|---------------------------------------|
| RCYZW005 | A train accident with fire occurs. Either the package insulation is torn away due to mechanical forces and the fire is able to heat the munitions inside the package, or the fire lasts long enough to cause thermal rupture of the munitions inside the package. | Yes |
| RCYZW006 | An aircraft crashes on a munitions railcar. No fire occurs, but impact forces lead to detona- tion and/or failure of agent containment. | Yes |
| RCYZW007 | An aircraft crashes on a munitions railcar. Fire occurs, but impact forces lead to detonation and/or failure of agent containment. | Yes |
| RCYZW008 | Combined with scenario RCYZW007. | No |
| RCYZW009 | A severe earthquake occurs involving a munitions railcar and crush forces fail the agent containment. | No |
| RCYZW010 | A severe earthquake occurs involving a munitions railcar and impact forces fail the agent containment. | No |
| RCYZW011 | A severe earthquake occurs involving a munitions railcar and puncture forces fail the agent containment. | Yes |
| RCYZW012 | A severe earthquake occurs involving a munitions railcar and subsequent fire detonates burstered munitions. | Yes |
| RCYZW013 | A severe earthquake occurs involving a munitions railcar and subsequent fire fails nonburstered munitions. | Yes |
| RCYZW014 | A tornado-generated missile leads to failure of the agent containment, or a tornado occurs, caus- ing overturn or derailment of a munitions railcar. | Yes |
| RCYZW015 | An earthquake or tornado occurs, generating undue mechanical forces which cause detonation of burstered munitions. | Yes |

6. N. S. C. C. S. C. C.

OFFSITE TRANSPORT ACCIDENT SEQUENCES (Continued)

| Sequence ID | Sequence Description | Considered for Further Analysis |
|---|---|---------------------------------------|
| <u>A</u> | ccident Scenarios for Air Transport to Tooele Army | Depot |
| ABYZA001 AAYZA001 | A severe ground collision involving an aircraft with munitions occurs and impact forces fail the agent package and munitions. | Yes |
| ABYZA002 AAYZA002 | A severe ground collision involving an aircraft with munitions occurs and impact forces fail the agent package and munitions. A subsequent fire occurs with a duration less than 2 h. | Yes |
| ABYZF003 AAYZF003 | A fire occurs aboard an aircraft with munitions and causes rupture of the compartme t due to thermal expansion of the agent. | Yes |
| ABYZCOO4 AAYZCOO4 | A severe ground collision involving an aircraft with munitions occurs and impact forces fail the agent package and munitions. A subsequent fire occurs with a duration greater than 2 h. | Yes |
| ABYZC005 AAYZC005 | A moderate ground collision involving an aircraft with munitions occurs causing a breach of the package. A subsequent fire occurs causing a breach (by detonation or thermal expansion) of the agent compartment and agent is released. | Yes |
| From Aber | deen: | |
| YZ = KH - | CM package, 1-ton package, mustard | |
| From Lexi | ngton: | |
| YZ = PH - RG - RV - PG - PV - KG - | CM package, projectiles, mustard CM package, rockets, GB nerve CM package, rockets, VX nerve CM package, projectiles, GB nerve CM package, projectiles, VX nerve CM package, 1-ton packages, GB nerve | |

ONSITE TRANSPORT ACCIDENT SEQUENCES

Ņ

SEALAN SECOND

Marana Kaassa

| Sequence | Security Description | Considered for Further |
|------------------|--|---------------------------|
| | Sequence Description | Analysis |
| VOXYZOO1 | A munitions vehicle collision/overturn occurs and crush forces fail the agent containment. | Yes |
| VOXYZ002 | A munitions vehicle collision/overturn occurs and impact forces fail the agent containment. | No |
| VOXYZ003 | A munitions vehicle collision/overturn occurs and puncture forces fail the agent containment. | Yes |
| VOXYZ 004 | A munitions vehicle accident with fire occurs, causing detonation of burstered munitions. Igni- tion of the propellant by a probe could also detonate the burster of a cartridge and the burster of a rocket could be detonated by impact- induced ignition of the rocket propellant. | Yes |
| VOXY Z005 | A munitions vehicle accident with fire occurs, causing nonburstered munitions to fail. | Yes |
| VOXYZ 006 | An aircraft crashes on a munitions vehicle. No fire occurs; impact forces fail the agent containment. | Yes |
| VOXYZ007 | An aircraft crashes on a munitions vehicle. Fire occurs, but impact forces fail the agent contain- ment. | Yes |
| VOXYZ009 | A severe earthquake occurs, causing a munitions vehicle accident and crush forces fail the agent containment. | Yes |
| VOXYZ010 | A severe earthquake occurs, causing a munitions vehicle accident and impact forces fail the agent containment. | No |
| VOXYZ 011 | A severe earthquake occurs, causing a munitions vehicle accident and puncture forces fail the agent containment. | Yes |
| VOXYZ012 | A severe earthquake occurs, causing a munitions vehicle accident and fire detonates burstered munitions. | Yes |

A-14

ONSITE TRANSPORT ACCIDENT SEQUENCES (Continued)

| Sequence ID | Sequence Description | Considered for Further Analysis |
|----------------|---|---------------------------------------|
| VOXYZ013 | A severe earthquake occurs, causing a munitions vehicle accident and fire fails nonburstered munitions. | Yes |
| VOXYZ014 | A tornado occurs, generating a missile or causing a truck overturn and mechanical forces fail agent containment. | Yes |
| VOXY2015 | A truck collision/overturn occurs generating undue mechanical forces which cause detonation of burstered munitions. | Yes |



8.5 Lat 140



2000 (III2222200) [II]22222022 [II]222222220(III)2222220)

KKKKKI DAADAA SAASAA "KAAKKA KKKKKI KKKKKI

7.7

TRUCK TRANSPORT FROM SENDING SITE TO RECEIVING SITE - OFFSITE MARINE TRANSPORT ACCIDENT SEQUENCES

| Sequence ID | Sequence Description | Considered for Further Analysis |
|-----------------|---|---------------------------------------|
| WXYZ001 | A munitions vehicle collision/overturn occurs and crush forces fail the agent containment. | No |
| VWXYZ002 | A munitions vehicle collision/overturn occurs and impact forces fail the agent containment. | No |
| VWXYZ003 | A munitions vehicle collision/overturn occurs and puncture forces fail the agent containment. | Yes |
| VWXYZ005 | A munitions vehicle accident with fire occurs, causing nonburstered munitions to fail. | No |
| VWXYZ006 | An aircraft crashes on a munitions vehicle. No fire occurs; impact forces fail the agent containment. | Yes |
| VWXYZ007 | An aircraft crashes on a munitions vehicle. Fire occurs, but impact forces fail the agent contain- ment. | Yes |
| VWXYZ009 | A severe earthquake occurs, causing a munitions vehicle accident and crush forces fail the agent containment. | No |
| VWXYZ010 | A severe earthquake occurs, causing a munitions vehicle accident and impact forces fail the agent containment. | No |
| VWXYZ011 | A severe earthquake occurs, causing a munitions vehicle accident and puncture forces fail the agent containment. | Yes |
| VWXYZ013 | A severe earthquake occurs, causing a munitions vehicle accident and fire fails nonburstered munitions. | No |
| VWXYZ014 | A tornado occurs, generating a missile or causing a truck overturn, and mechanical forces fail agent containment. | Yes |

LANKER MASSAMANS HILLONDO COPRIMIN CORECTLY MICHAELEN 29599393



TRUCK TRANSPORT FROM SENDING SITE TO RECEIVING SITE - OFFSITE AIR TRANSPORT ACCIDENT SEQUENCES

RECEIPTER REPORT RECEIPTER RECEIPTER

| Sequence ID | Sequence Description | Considered for Further Analysis |
|----------------|--|---------------------------------------|
| VAXYZ001 | A munitions vehicle collision/overturn occurs and crush forces fail the agent containment. | Yes |
| VAXYZ002 | A munitions vehicle collision/overturn occurs and impact forces fail the agent containment. | Yes |
| VAXYZ003 | A munitions vehicle collision/overturn occurs and puncture forces fail the agent containment. | Yes |
| VAXYZOO4 | Detonation of burstered munitions occurs by either (1) fire-only accident, (2) mechanical force and fire, (3) truck collision/overturn impact - induced rocket propellant ignition, or (4) truck collision/overturn - induced undue force detonation. | Yes |
| VAXYZ005 | A munitions vehicle accident with fire occurs, causing nonburstered munitions to fail. | No |
| VAXYZ006 | An aircraft crashes on a munitions vehicle. No fire occurs; impact forces fail the agent containment. | Yes |
| VAXYZ007 | An aircraft crashes on a munitions vehicle. Fire occurs but impact forces fail the agent contain- ment. | Yes |
| VAXYZ009 | A severe earthquake occurs, causing a munitions vehicle accident and crush forces fail the agent containment. | Yes |
| VAXYZO10 | A severe earthquake occurs, causing a munitions vehicle accident and impact forces fail the agent containment. | Yes |
| VAXYZ011 | A severe earthquake occurs, causing a munitions vehicle accident and puncture forces fail the agent containment. | Yes |
| VAXYZ012 | A severe earthquake occurs, causing a munitions vehicle accident and fire detonates burstered munitions. | Yes |





A-17

| AD-A1 | 93 355 | | MICAL | STOCK | PILE D | ISP05/ | AL PRO | GRAM R Chnolo | ISK AN | IALYSI INC SA | S OF N DIEG | 9/ 0 | 13 |
|--------|--------|-------|-------|-------|--------|--------|--------|------------------|--------|------------------|----------------|---------|----|
| UNCLAS | SIFIE | D SAP | EO-CD | -IS-8 | 7000 0 | AAA15- | 85-D- | 8822 | | F/G : | 15/6. 3 | NL | |
| | | | | | | | | | | | | | |
| | | | | | | | | | | | | | |
| | | | | | | | | | | | | | |
| | | | | | | | | | | | | | |
| | | | | | | | | | | | | | |
| | | | | | | | | | | | | | |
| | | | | | | | | | | | | | |
| | | | | | | | | | | | | | |



TRUCK TRANSPORT FROM SENDING SITE TO RECEIVING SITE - OFFSITE AIR TRANSPORT ACCIDENT SEQUENCES (Continued)

| VAXY2013 A severe earthquake occurs, causing a munitions No whicle accident and fire fails nonburstered munitions. VAXY2014 A tornado occurs, generating a missile or causing a truck overturn, and mechanical forces fail agent containment. VAXY2015 An earthquake or tornado occurs, generating undue Yes mechanical forces which cause detonation of burstered munitions. VAXY2016 Manual Manu | ID | Sequence Description | for Further Analysis |
|--|----------|--|-------------------------|
| YAXYZ014 A tornado occurs, generating a missile or causing is a truck overturn, and mechanical forces fail agent containment. YAXYZ015 An earthquake or tornado occurs, generating undue mechanical forces which cause detonation of burstered munitions. | VAXYZ013 | A severe earthquake occurs, causing a munitions vehicle accident and fire fails nonburstered munitions. | No |
| VAXYZ015 An earthquake or tornado occurs, generating undue Yes mechanical forces which cause detonation of burstered munitions. | VAXYZ014 | A tornado occurs, generating a missile or causing a truck overturn, and mechanical forces fail agent containment. | Yes |
| A-18 | VAXYZ015 | An earthquake or tornado occurs, generating undue mechanical forces which cause detonation of burstered munitions. | Yes |
| 4-18 | | | |
| A-18 | | | |
| A-18 | | | |
| A-18 | | | |
| A-18 | | | |
| | | | |







TRUCK TRANSPORT FROM SENDING SITE TO RECEIVING SITE - OFFSITE RAIL TRANSPORT OPTION ACCIDENT SEQUENCES

| Sequence ID | Sequence Description | Considered for Further Analysis |
|----------------|--|---------------------------------------|
| VRXYZ001 | A munitions vehicle collision/overturn occurs and crush forces fail the agent containment. | No |
| VRXYZ002 | A munitions vehicle collision/overturn occurs and impact forces fail the agent containment. | No |
| VRXYZ003 | A munitions vehicle collision/overturn occurs and puncture forces fail the agent containment. | Yes |
| VRXYZ004 | Detonation of burstered munitions occurs by either (1) fire-only accident, (2) mechanical force and fire, (3) truck collision/overturn impact-induced rocket propellant ignition, or (4) truck collision/overturn - induced undue force detonation. | Yes |
| VRXYZ005 | A munitions vehicle accident with fire occurs, causing nonburstered munitions to fail. | No |
| VRXYZ006 | An aircraft crashes on a munitions vehicle. No fire occurs; impact forces fail the agent containment. | Yes |
| VRXYZ007 | An aircraft crashes on a munitions vehicle. Fire occurs but impact forces fail the agent contain- ment. | Yes |
| VRXYZ009 | A severe earthquake occurs, causing a munitions vehicle accident and crush forces fail the agent containment. | No |
| VRXYZO10 | A severe earthquake occurs, causing a munitions vehicle accident and impact forces fail the agent containment. | No |
| VRXYZ011 | A severe earthquake occurs, causing a munitions vehicle accident and puncture forces fail the agent containment. | Yes |
| VRXYZ012 | A severe earthquake occurs, causing a munitions vehicle accident and fire detonates burstered munitions. | Yes |



A-19

. • .

TRUCK TRANSPORT FROM SENDING SITE TO RECEIVING SITE - OFFSITE RAIL TRANSPORT OPTION ACCIDENT SEQUENCES (Continued)

| Sequence ID | Sequence Description | Considered for Further Analysis |
|----------------|--|---------------------------------------|
| VRXYZ013 | A severe earthquake occurs, causing a munitions vehicle accident and fire fails nonburstered munitions. | Yes |
| VRXYZ014 | A tornado occurs, generating a missile or causing a truck overturn, and mechanical forces fail agent containment. | Yes |
| VRXYZ015 | An earthquake or tornado occurs, generating undue mechanical forces which cause detonation of burstered munitions. | Yes |







OFFSITE HANDLING - SENDING SITES ACCIDENT SEQUENCES

122223 22222

٣

11.14

Ar. 40' . 80'.

| Sequence | | Considered for Furthe |
|----------------------|---|--------------------------|
| ID | Sequence Description | Analysis |
| HC1 HA1 HW1 | Drop of bare pallet or single item at storage area. | Yes |
| HC2 HA2 HW2 | Forklift collision with short duration fire at storage area involving bare munitions. | Yes |
| НС3 НА3 НW3 | Forklift tine accident at storage area involving bare munitions. | Yes |
| HC4 HA4 HW4 | Forklift collision accident without fire at stor- age area involving bare munitions. | Yes |
| HC8 HA8 HW8 | Drop of offsite container. | Yes |
| HC9 HA9 HW9 | Collision accident with short duration fire dur- ing handling of offsite container. | Yes |
| HC10 HA10 HW10 | Collision accident without fire during handling of offsite container. | Yes |
| HC11 HA11 HW11 | Drop of bare palletized munition leads to detonation. | Yes |
| HC12 HA12 HW12 | Forklift collision accident at storage area leads to detonation. | Yes |
| HC17 HA17 HW17 | Drop of pallet containing a leaking munition dur- ing leaker isolation operations. | Yes |
| HC18 HA18 HW18 | Drop of single leaking in leakers processing facility. | Yes |









1223

165555531 CONSERVE 2855552 CONSERVED 2555555

- 20102213

OFFSITE HANDLING - SENDING SITES ACCIDENT SEQUENCES (Continued)

| Sequence ID | Sequence Description | Considered for Further Analysis |
|----------------------|---|---------------------------------------|
| HC19 HA19 HW19 | Forklift tine puncture during leaker isolation. | Yes |
| HC21 HA21 HW21 | Collision accident without fire during handling of leaking munition. | Yes |
| HC23 HA23 HW23 | Drop of munition in offsite container leads to detonation. | Yes |
| HC25 HA25 HW25 | Collision accident during munition handling in offsite container leads to detonation. | Yes |
| HC27 HA27 HW27 | Collision accident in offsite container with pro- longed fire leads to thermal detonation. | Yes |
| HC29 HA29 HW29 | Drop of pallet containing leaker leads to detonation. | Yes |
| HC30 HA30 HW30 | Drop of single leaking munition leads to detonation. | Yes |
| HC31 HA31 HW31 | Collision accident involving a leaking munition leads to detonation. | Yes |
| HC32 HA32 HW32 | Failure to detect a leak in the offsite container. | Yes |



OFFSITE HANDLING - RECEIVING SITES ACCIDENT SEQUENCES

| Sequence ID | Sequence Description | Considered for Further Analysis |
|----------------------|---|---------------------------------------|
| HC1 HA1 HW1 | Drop of bare pallet or single item at storage area. | Yes |
| HC3 HA3 HW3 | Forklift tine accident at storage area involving bare munitions. | Yes |
| HC4 HA4 HW4 | Forklift collision accident without fire at stor- age area involving bare munitions. | Yes |
| HC5 HA5 HW5 | Drop of onsite container. | Yes |
| HC6 HA6 HW6 | Forklift collision accident with short duration fire during handling of onsite container. | Yes |
| HC7 HA7 HW7 | Forklift collision without fire during handling of onsite container. | Yes |
| HC8 HA8 HW8 | Drop of offsite container. | Yes |
| HC9 HA9 HW9 | Collision accident with short duration fire dur- ing handling of offsite container. | Yes |
| HC10 HA10 HW10 | Collision accident without fire during handling of offsite container. | Yes |
| HC11 HA11 HW11 | Drop of bare palletized munition leads to detonation. | Yes |
| HC12 HA12 HW12 | Forklift collision accident at storage area leads to detonation. | Yes |

A-23


6 Levina

ES ?

OFFSITE HANDLING - RECEIVING SITES ACCIDENT SEQUENCES (Continued)

| Sequence ID | Sequence Description | Considered for Further Analysis | |
|----------------------|---|---------------------------------------|--|
| HC17 HA17 HW17 | Drop of pallet containing a leaking munition dur- ing leaker isolation operations. | Yes | |
| HC18 HA18 HW18 | Drop of single leaking in leakers processing facility. | Yes | |
| HC19 HA19 HW19 | Forklift tine puncture during leaker isolation. | Yes | |
| HC21 HA21 HW21 | Collision accident without fire during handling of leaking munition. | Yes | |
| HC22 HA22 HW22 | Drop of munition in onsite container leads to detonation. | Yes | |
| HC23 HA23 HW23 | Drop of munition in offsite container leads to detonation. | Yes | |
| HC24 HA24 HW24 | Collision accident during munition handling in onsite container leads to detonation. | Yes | |
| HC25 HA25 HW25 | Collision accident during munition handling in offsite container leads to detonation. | Yes | |
| HC26 HA26 HW26 | Collision accident in onsite container with pro- longed fire leads to thermal detonation. | Yes | |
| HC27 HA27 HW27 | Collision accident in offsite container with pro- longed fire leads to thermal detonation. | Yes | |
| HC29 HA29 HW29 | Drop of pallet containing leaker leads to detonation. | Yes | |



عار أواد أواد أو

OFFSITE HANDLING - RECEIVING SITES ACCIDENT SEQUENCES (Continued)

202000

26655555

| Sequence ID | Sequence Description | Considered for Further Analysis |
|----------------|--|---------------------------------------|
| HC30 | Drop of single leaking munition leads to | Yes |
| HA30 | detonation. | |
| HW30 | | |
| HC31 | Collision accident involving a leaking munition | Yes |
| HA31 | leads to detonation. | |
| HW31 | | |
| HC32 | Failure to detect a leak in the offsite container. | Yes |
| HA32 | | |
| HW32 | | |







FACILITY HANDLING ACCIDENT SEQUENCES

1

03-72-72 1

2333552224

ann amara maanna-ruura-ruura maraa s

| Sequence ID | Sequence Description | Considered for Further Analysis |
|----------------|--|---------------------------------------|
| HF1 | Munition pallet or container dropped during move- ment from MHI to MDB. | Yes |
| HF2 | Bare single munition dropped during handling inside the MDB. | Yes |
| HF3 | Forklift collision accident with short duration fire during handling from MHI to MDB. | Yes |
| HF4 | Forklift tine accident during handling from the MHI to MDB. | Yes |
| HF5 | Collision accident with prolonged fire during handling from MHI to MDB leads to detonation or hydraulic rupture. | Yes |
| HF7 | Collision accident without fire. | Yes |
| HF8 | Munition dropped inside the MDB. | Yes |
| HF9 | Forklift tine accident inside the MDB. | Yes |
| HF10 | Collision without fire inside the MDB. | Yes |
| HF11 | Drop of munition pallet from the MHI to MDB leads to detonation. | Yes |
| HF12 | Drop of bare single munition inside the MDB leads to detonation. | Yes |
| HF13 | Drop of palletized munition (in container) inside the MDB leads to detonation. | Yes |
| HF14 | Collision accident from the MHI to the MDB leads to detonation. | Yes |

ي.

a Langer

A-26



PLANT OPERATIONS ACCIDENT SEQUENCES - EXTERNAL EVENTS

| Sequence ID | Sequence Description | Considered for Further Analysis |
|----------------|---|---------------------------------------|
| P01 | Tornado-generated missile puncture/crush muni- tions in the MHI. | Yes |
| P02 | Tornado-generated missile detonate munitions in the MHI. | Yes |
| P03 | Tornado-generated missile puncture/crush muni- tions in the UPA. | Yes |
| P04 | Tornado-generated missile detonate munitions in the UPA. | Yes |
| P05 | Tornado-generated missile damages the agent piping system between the BDS and TOX at TEAD (bulk-only facility). | Yes |
| P06 | Meteorite strikes the MHI. | Yes |
| P07 | Meteorite strikes the UPA. | Yes |
| PO7A | Meteorite strikes the TOX. | Yes |
| P08 | Meteorite strikes the agent piping system between the BDS and TOX at TEAD (bulk-only facility). | Yes |
| P09 | Direct large aircraft crash onto the MHI; no fire. | Yes |
| P010 | Direct large aircraft crash onto the MHI; fire not contained in 0.5 h. | Yes |
| P011 | Direct large aircraft crash onto the MHI; fire contained in 0.5 h. | Yes |
| P012 | Direct large aircraft crash damages the MDB; no fire. | Yes |
| P013 | Direct large aircraft crash damages the MDB; fire not contained in 0.5 h. | Yes |
| P014 | Direct large aircraft crash damages the MDB; fire contained in 0.5 h. | Yes |
| P015 | Indirect large aircraft crash damages the MHI; no fire. | Yes |

A-27

PLANT OPERATIONS ACCIDENT SEQUENCES - EXTERNAL EVENTS (Continued)

| Sequence ID | Sequence Description | Considered for Furthe Analysis |
|----------------|--|--------------------------------------|
| P016 | Indirect large aircraft crash damages the MHI; fire not contained in 0.5 h. | Yes |
| P017 | Indirect large aircraft crash damages the MHI; fire contained in 0.5 h. | Yes |
| P018 | Indirect large aircraft crash damages the MDB; no fire. | Yes |
| PC19 | Indirect large aircraft crash damages the MDB; fire not contained in 0.5 h. | Yes |
| P020 | Indirect large aircraft crash damages the MDB; fire contained in 0.5 h. | Yes |
| P021 | Direct crash of a large or small aircraft damages the outdoor agent piping system at TEAD; no fire. | Yes |
| P022 | Direct crash of a large or small aircraft damages the outdoor agent piping system at TEAD; fire occurs and not contained. | Yes |
| P023 | Earthquake causes the munitions in the MHI to fall and be punctured.(a) | No |
| P024 | Earthquake causes munitions in the MHI to fall and detonate. ^(a) | No |
| P025 | Earthquake damages the MDB structure, munitions fall and are punctured; fire suppressed. | Yes |
| P026 | Earthquake damages the MDB structure, munitions fall and are punctured; earthquake also initiates fire; fire suppression system fails. | Yes |
| PO28A(b) | Earthquake damages the MDB structure, munitions fall and are punctured; TOX damaged; fire occurs; fire suppressed. | No |
| PO28 | Earthquake damages the MDB structure, munitions fall and are punctured; TOX damaged; fire occurs; fire suppression system fails. | No |
| P029 | Earthquake damages the MDB; munitions are intact; fire occurs: fire suppression system fails. | Yes |

15.5.5

<u>.</u> 5

A-28



5

PLANT OPERATIONS ACCIDENT SEQUENCES - EXTERNAL EVENTS (Continued)

| Sequence Description | Considered for Further Analysis | |
|---|---|--|
| Earthquake damages the MDB; munitions are intact; TOX damaged; no fire occurs.(c) | No | |
| Earthquake damages the MDB; munitions are intact; TOX damaged; fire occurs; fire suppressed. | No | |
| Earthquake damages the MDB; munitions are intact; TOX damaged; fire occurs; fire not suppressed. | No | |
| Earthquake causes munitions to fall and deto- nate; MDB breached by detonation; the TOX is intact; no fire.(c) | No | |
| Earthquake causes munitions to fall but no deto- nation occurs; the MDB is intact; the TOX is intact; earthquake also initiates fire; fire suppression system fails. | Yes | |
| Earthquake causes munitions to fall but no deto- nation occurs; the MDB is intact; the TOX is damaged; fire occurs; fire suppression system fails. | No | |
| | Sequence Description Earthquake damages the MDB; munitions are intact; TOX damaged; no fire occurs. ^(c) Earthquake damages the MDB; munitions are intact; TOX damaged; fire occurs; fire suppressed. Earthquake damages the MDB; munitions are intact; TOX damaged; fire occurs; fire not suppressed. Earthquake causes munitions to fall and deto- nate; MDB breached by detonation; the TOX is intact; no fire. ^(c) Earthquake causes munitions to fall but no deto- nation occurs; the MDB is intact; the TOX is intact; earthquake also initiates fire; fire suppression system fails. Earthquake causes munitions to fall but no deto- nation occurs; the MDB is intact; the TOX is intact; fire occurs; fire suppression system fails. | |

(c)Screened out on the basis of frequency.

ACCIDENTS FOR PLANT OPERATIONS - INTERNAL EVENTS

| Sequence ID | Sequence Description | Considered for Further Analysis No | |
|----------------|--|---|--|
| PO41 | One munition falls off the conveyor in the ECV due to a process upset or improper loading and is punctured. The spill is not cleaned up in 1 h. | | |
| P042 | One munition falls off the conveyor and deto- nates in the ECV, caused by process upset or improper loading. | Yes | |
| P043 | Same as PO41 with added fire. | No | |
| P044 | Same as PO42 with failure propagating to other munitions due to fragments. | No | |
| P045 | A process upset results in spill of agent inven- tory in ECR. | No | |
| P046 | Same as PO45 with fire. | No | |
| P047 | Same as PO45 with detonation. | No | |
| P048 | A punched munition falls off the BSA conveyor. Bulk drain station did not drain the munitions before sending it to the BSA, so that a spill occurs. | No | |
| P049 | Same as PO48 with fire. | No | |
| P050 | Large spill (contents of agent collection tank) in TOX cubicle due to pipe failure (528 gal). | No | |
| P051 | Small spill (typically less than 50 gal) in TOX cubicle due to pipe failure. | No | |
| P052 | Same as PO51 with fire. | No | |
| Other seq | uences identified are summarized in Tables A-1, A-2 | , and A-3. | |

REFERENCE REVERSE

Other sequences identified are summarized in Tables A-1, A-2, and A-3. These deal with furnace/incinerator events. The event trees corresponding to these sequences are in Section 7.1. None of the sequences in these tables was considered for detailed analysis.

A-30



the Color and the Color

State of the

TABLE A-1EVENTS CONSIDERED FOR THE DEACTIVATION FURNACE SYSTEM

| Event | Description | | |
|---------------------------------------|--|--|--|
| Stop munitions feed (DFS-SMF) | Failure on this event tree branch implies that feed of drained rockets or mines to the DFS is not discontinued, given that a shutdown signal occurs. | | |
| Ventilation system (DFS-VENT) | This branch point represents the failures of the ventilation system to provide filtered air to the DFS pump. (See Section 7.1 for the fault tree.) | | |
| Stop fuel (DFS-SFA) | Failure of this event tree branch implies that the natural gas supply line to the burner in the DFS retort is not isolated, given that a shutdown signal occurs. If ventilation to the room has failed, operator recovery is per- mitted to prevent a possible room explosion. (See Section 7.1 for the fault tree.) | | |
| Explosion does not occur (DFS-EXP) | Failure of this branch implies that a natural gas explosion has occurred in the DFS room. For the situation in which ventilation succeeds, the size of this explosion is the size of a DFS furnace explosion. For the case in which room ventilation has failed, the explosion is the size of a DFS room explosion. The probability was subjectively estimated. | | |
| Explosion contained (DFS-CONT) | Failure of this branch implies that the DFS room structure has been breached by an explo- sion. The probability was subjectively estimated. | | |



ġ

Row - REAL - PRESSEE - PRESSEE

NY SING

| TABLE A-2 | | | | | | |
|-----------|------------|-----|-----|--------|-------------|-------|
| EVENTS | CONSIDERED | FOR | THE | LIQUID | INCINERATOR | (LIC) |

| Event | Description |
|--|--|
| Ventilation system (LIC-VENT) | This branch point represents the failure of the ventilation system to provide air to the LIC room. (The fault tree is in Section 7.1.) |
| Stop agent feed (LIC-SAF) | This branch point represents both the ACS and the operator failing to shut off the agent feed and failing to recognize that the feed is not shut off. Different time periods and therefore different recovery probabilities apply for different scenar- ios. (The fault tree is in Section 7.1.) |
| Shutdown PAS (LIC-SPAS) | This branch point represents both the ACS and the operator failing to stop flow through the PAS and failing to recognize that flow continues. (The fault tree is in Section 7.1.) |
| Stop fuel to burners (LIC-SFF) | The branch point represents both the ACS and the operator failing to shut off the fuel within 15 min and failing to recogn- ize that the fuel is not shut off. This event applies to the PCC and the AB. (The fault tree is in Section 7.1.) |
| Avoid explosion (LIC-EXP) | This branch represents ignition/detonation of accumulated fuel/air or agent/air mix- tures. The probability was subjectively assigned. |
| Structure contains explosion (LIC-CONT) | This branch represents failure of the LIC room to contain an explosion. The proba- bility was subjectively assigned. |
| Stop fuel to LIC-PCC burner (LIC-SFP) | This branch point represents both the ACS and the operator failing to shut off fuel to the LIC PCC burner within 15 min and failing to recognize that the fuel is not shut off. (The fault tree is in Section 7.1.) |

A-32



TABLE A-3EVENTS CONSIDERED FOR THE METALS PARTS FURNACE (MPF)

| Event | Description | | |
|---|--|--|--|
| MPF-1 Tree | | | |
| Ventilation System (MPF-VENT) | This branch point represents the failure of the exhaust system to provide filtered air to the MPF room. (See Section 7.1 for the fault tree.) | | |
| Stop fuel (MPF-SFA) | Failure of the branch point implies that the natural gas supply to one or more burners in the MPF has not been isolated. If room ventilation has failed, operator recovery is permitted to prevent a possi- ble room explosion. (See Section 7.1 for the fault tree.) | | |
| Explosion avoided (MPF-EXP) | Failure of this branch point implies that natural gas explosion has occurred in the MPF room. For this situation in which ventilation succeeds, the size of the explosion is the size of the DFS furnace explosion. For the case in which room ventilation has failed, the explosion is the size of an MPF room explosion. The probability was subjectively estimated. | | |
| Explosion contained (MPF-CONT) | Failure of this branch point implies that the MPF room structure has been breached by the MPF explosion. The probability was subjectively estimated. | | |
| MPF-2 Tree | | | |
| Explosion does not occur (MPF-EX) | This branch point involves the undrained munition exploding in the MPF. The probability was subjectively estimated. | | |
| MPF room and vent integrity maintained (MPF-INT) | This branch point involves damage to the MPF room or vent such that agent in the room is released to the atmosphere. The probability was subjectively estimated. | | |

A-33



18 1 S at 1

APPENDIX B SENSITIVITY ANALYSIS





3.0 INTRODUCTION

Several accident scenarios were identified that could result in a significant release of agent to the environment during demilitarization operations at CONUS sites. These scenarios include:

- TOX Area Fire
- BSA Area Fire
- ECV Area Fire
- Carbon Filter Fire
- Carbon Filter Desorption
- Continued Agent Feed in Non-operating LIC
- PAS Agent Scrubbing
- Feed Full Ton Container into MPF.

Several other scenarios involving munition detonation were identified but not evaluated in favor of providing documentation for the sensitivity analyses. Results from the sensitivity analysis are described for each scenario as follows.

3.1 Results from Sensitivity Analyses

<u>3.1.1 TOX Area Fire</u>. The TOX Area fire involves the following sequence of events:

- (1) Rupture of filled 1300-gallon agent storage tank in TOX Area
- (2) Ignition of agent spill
- (3) Failure of TOX fire suppression system
- (4) Fire vaporizes agent which is vented from the TOX to the carbon filters.

Undecomposed agent can be released to the environment through the filters if the agent flow rate is sufficiently high, the filters approach saturation and/or the filter inlet gas temperature is high. The sensitivity



of the magnitude of agent released to the environment was therefore considered on the following variables:

- Residence time of volatilized agent in the TOX
- Fire size (directly related to undecomposed agent flow rate)
- Combustion efficiency (directly related to undecomposed agent flow rate)
- Capacity of carbon to absorb agent
- Gas temperature at filter inlet.

In an agent fire, heat returned to the pool of burning liquid by convective and radiative mechanisms is used to volatilize agent. Part of the volatilized agent is combusted with the remainder potentially vented from the area. Residence time in the TOX of volatilized agent that is not combusted was included in the sensitivity analysis because the fire may raise the temperature to a point where thermal decomposition of the volatilized agent could occur. As a worst case, a 1-second residence time was assumed. This is equivalent to the volatilized agent traveling a distance of about four inches prior to entering the TOX ventilation exhaust duct. Residence times of 2, 5, 10, and 14.3 seconds were also evaluated. The 14.3-second residence time is the most credible case and would involve a fire on the floor directly below the exhaust duct (5-feet above the floor). This is possible because a 1300-gallon spill of agent will fill the 500gallon sump in the TOX and completely cover the TOX floor.

As discussed in detail in the calculation summaries given in Appendix A pages A1 through A28, the fire size will be limited by the ventilation flow rate. The worst case, i.e. the largest fire, will result when the fire burns a sufficient amount of agent to reduce the oxygen concentration to the minimum level required for combustion. A second case involves a fire size equivalent to the TOX sump area. Another fire size is where the release of undecomposed agent from the TOX area reaches a maximum for a particular residence time. This fire size, calculated by trial and error, is where the agent vaporization rate is relatively high while the agent combustion rate and, in turn, the TOX temperature is sufficiently low such that thermal decomposition of agent is not appreciable. The combustion efficiencies evaluated were 50, 75 and 100 percent. It is important to note that a 100 percent combustion efficiency implies that all the agent involved in combustion is converted to CO_2 , H_2O , P_2O_5 , etc. so that the entire heating value of the agent is generated. A combustion efficiency of less than 100 percent implies that intermediate combustion products formed so that the entire heating value of the agent was not generated. Agent can be volatilized but not combusted for any combustion efficiency including 100 percent. This could occur if the part of the agent is directed away from the flames as it is volatilized.

TONESSEE TANDARD' AND ALL AND ALL ALL AND ADDRESS AND ADDRESS AND ADDRESS AND ADDRESS AND ADDRESS AND ADDRESS A

いたいししし

The capacity of the carbon to adsorb agent was varied from 0.05 lb agent/lb carbon as a worst case to 0.2 lb agent/lb carbon. The 0.2 lb capacity is still conservative when compared with the capacities of 0.37 lb HD/lb carbon, 0.298 lb VX/lb carbon, and 0.318 lb GB/lb carbon given in Reference 1. These capacities are for G210 coconut-derived, non-whetlerized activated carbon, which is similar to the activated carbon used at CAMDS.

The gas temperature at the filter inlet was varied from 100 F up to a temperature calculated from heat balances. The calculated temperature is the worst case because it does not incorporate all heat losses from the gas during traversal between the TOX Area and the filters. The rate and degree of adsorption is known to be exponentially and inversely proportional to temperature. Thus, a small increase in temperature may cause a significant decrease in adsorption efficiency.

Table 1 gives a summary of agent releases for various fire sizes and combustion efficiencies. The maximum fire duration given in Table 1 was estimated as follows. The maximum fire duration for large fires which reduce the oxygen concentration in the TOX to the minimum required for combustion is the time required for an operator to close the inlet dampers to the TOX, thereby shutting off the oxygen supply. As shown in Figure 1, approximately 15 minutes are required for a 99 percent probability that an operator will respond to close the TOX inlet dampers. This includes a 5 minute period in which the operator will attempt to start the fire protection system in the TOX. In cases where the fire size is not at a maximum, additional time is required for consumption of the oxygen remaining in the TOX after the dampers are shut. The fire will continue until the

TABLE 1. SUMMARY OF TOX AREA FIRE CALCULATIONS^(a) Fire Size and Combustion Efficiency are Varied

| Agent | Fire Size (sq. ft.) | Combustion Efficiency (%) | Time to Release)6.661 lb Agent (Min.) | Agent Released After 5 Minutes (Ibs.) | Agent Released After Maximum Fire Duration (lbs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Min) |
|----------|------------------------|---------------------------------|--|---|--|--|
| 04 | Suga (b) | 166 | 2 | 6 66 92 | 1 1959 | 22 |
| ND ND | AR | 166 | (c) | < 6 6661 | (5 565) | 15 |
| HD | Suan | 75 | 4 | 8.6828 | 6.2103 | 25 |
| HD(A) | 21 | 75 | 4 | Ø. ØØ28 | Ø.2814 | 25 |
| HD | 88 | 75 | (c) | < 0.0001 | < 5.0001 | 15 |
| HD | Supp | 58 | 34 | < 9.9601 | 6.6009 | 33 |
| HD | 121 | 50 | (c) | < 0.6061 | < 5.9091 | 15 |
| GB | Sump | 166 | 3 | 6.6630 | 2.7654 | 28 |
| GB | 51 | 150 | (c) | < 9.60 01 | < 8.0001 | 15 |
| GB | Sump | 75 | 5 | Ø. Ø167 | 6.2829 | 24 |
| GB(d) | 21 | 75 | 2 | Ø. 8229 | 6.4324 | 23 |
| CB | 68 | 75 | (c) | < 8.8961 | < 0.0001 | 15 |
| GB | Sump | 50 | 19 | 8.6682 | J . 5728 | 36 |
| GB | 103 | 50 | (c) | < 8.6601 | < 0.6001 | 15 |
| vx | Sump | 165 | (c) | (0. 88 51 | (0 .8601 | 15 |
| ٧X | 21 | 166 | (c) | < 0.0001 | < 0.0001 | 15 |
| VX(d) | 14 | 75 | > 66 | < 0.0001 | (0 .0001 | 18 |
| ٧X | Sump | 75 | > 65 | (8.960 1 | (0.0001 | 16 |
| ¥X | 28 | 75 | (c) | < 0.0001 | (0.000 1 | 15 |
| ٧X | Sump | 50 | (c) | (5.005 1 | (8 .0001 | 19 |
| ٧X | 42 | 50 | (c) | (0.969 1 | (0 .0001 | 15 |

77

5.53 V

5555555

(a) Carbon capacity = 0.05 lb agent/lb carbon, gas temperature at filter inlet calculated from heat balances. The residence time of the fire products in the TOX area = 1 second.

(b) Sump area = 20 square feet

(c) The fire does not release agent from the TOX area.

(d) Worst-case fire area/combustion efficiency combination.



SAME CANADOR COSTLAND ASSAME



DEPERTURE SERVICES SUCCESSION DEPENDENCES DE SECONDA DE SERVICES DE SECONDA DE SECONDA DE SECONDA DE SECONDA DE

| Pr (fail to isolate by X mins) | | | |
|--------------------------------|--|--|--|
| Without System | | | |
| 4E-2 | | | |
| 1E-2 | | | |
| 4E-3 | | | |
| | | | |

Figure 1. Operator Times Versus Probabilities for Failure to Close Dampers

RECEIPTER, INSURANCE

minimum oxygen level required for combustion is reached, at which point the fire is assumed to self-extinguish.

Results indicate that both fire size and combustion efficiency have a significant effect on the magnitude of agent released to the environment. The worst cases are 75 percent combustion efficiency/21 sq. ft. fire for HD and for GB. No combination of the variables allowed a significant release of VX.

Table 2 gives a summary of agent releases for various gas residence times in the TOX. The most credible residence time of 14.3 seconds results in a significantly lower agent release. The 14/75 fire size/combustion efficiency combination is the worst case for a 14.3-second residence time. This trend is explained later for the BSA area fire.

Table 3 gives a summary of agent releases for variable carbon capacities. The more credible capacity of 0.2 lb agent/lb carbon significantly reduced the amount of agent released by at least an order of magnitude.

Table 4 gives a summary of agent releases for variable gas temperatures at the filter inlet. The lower temperatures resulted in significantly lower agent releases due to the strong dependence of the adsorption rate constant on temperature.

The worst-case and most-credible-case agent releases for the TOX Area fire are given in Table 5. The most credible case was selected based on a 14.3-second residence time for the volatilized agent in the TOX, a carbon capacity of 0.05 lb agent/lb carbon (worst case), filter inlet gas temperature calculated from heat balances (worst case), and the worst case fire size/combustion efficiency combination. The worst case was as above except for a 1-second residence time. The most credible case is still very conservative because:

- The selected agent capacity of carbon is below that obtained during actual agent tests
- Filter bank inlet gas temperature will be lower than the calculated temperature when all heat losses are taken into account
- As described in the calculation summary of Appendix A, worstcase assumptions were used whenever information was unavailable.



TABLE 2. SUMMARY OF TOX AREA FIRE CALCULATIONS^(a) Residence Time of Fire Products in TOX Varied

| Agent | Fire Size/ Combustion Efficiency (sq. ft./%) | Residence Tipe (sec.) | Time to Release >9.991 ib Agent (Win.) | Agent Released After 5 Winutes (lbs.) | Agent Released After Maxieum Fire Duration (lbs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Min.) |
|-------------------|---|-----------------------------|--|---|--|--|
| HD | 28/75 | 1 | 2 | 6.6692 | Ø. 1959 | 25 |
| HD | 29/75 | 2 | 5 | Ø. 6015 | 0.1158 | 25 |
| HD | 28/75 | 5 | 7 | 6 .0065 | 0.0291 | 25 |
| HD | 20/75 | 18 | 12 | Ø. 9962 | Ø. ØØ59 | 25 |
| нд(ь) | 20 /75 | 14.3 | 19 | (5 .6661 | 0. <i>0</i> 021 | 25 |
| НС (Р) | 14/75 | 14.3 | 21 | < 0.0001 | Ø. 8958 | 31 |
| GB | 21/75 | 1 | 2 | Ø. 8229 | 6.4324 | 23 |
| C8 | 21/75 | 2 | 2 | 5.5 151 | 2.8775 | 23 |
| CB | 21/75 | 5 | 3 | 6.8853 | 0.4355 | 23 |
| CB | 21/75 | 10 | 5 | 6.0014 | 0.6581 | 23 |
| св (р) | 21/75 | 14.3 | 7 | 9.9985 | 0.0175 | 23 |
| CB(۶) | 14/75 | 14.3 | 9 | Ø. 9962 | Ø. 1613 | 29 |
| vx | 14/75 | 1 | > 65 | (1 .9991 | < 9 .9001 | 18 |
| vx | 14/75 | 2 | > 68 | < 0.0001 | < 8 .5661 | 18 |
| vx | 14/75 | 5 | > 60 | < 8.0561 | (1.555 1 | 18 |
| vx | 14/75 | 16 | > 66 | (8.995 1 | (9.85 81 | 18 |
| ٧X ^(b) | 14/75 | 14.3 | > 50 | < 0.0001 | < 0.0001 | 18 |
| VX(b) | 16/75 | 14.3 | > 68 | < 0.0001 | < 0.0601 | 21 |

(a) Carbon capacity = \$.\$5 lb agent/lb carbon. The filter inlet gas temperature calculated by heat balances. Worst case fire size/combustion efficiency combinations shown for GB and VX.

(b) Most credible residence time of the fire products in the TOX area.

TABLE 3. SUMMARY OF TOX AREA FIRE CALCULATIONS^(a)

Carbon Capacity Varied

1922222

| Agent | Fire Size/ Combustion Efficiency (sq. ft./%) | Carbon Capacity (1b agent/ 1b carbon) | Time to Release)0.001 lb Agent (Vin.) | Agent Released After 5 Winutes (lbs.) | Agent Released After Maxisus Fire Duration (1bs.) | Maxiaua Fire Duration with 99% Probability of Operator Closing Damper (Min.) |
|-------------------|---|---|--|---|--|--|
| HD | 21/75 | 0.2 | 4 | 0.0016 | 6.9214 | 25 |
| HD | 21/75 | 8.85 | 4 | Ø. 9628 | 0.2814 | 25 |
| ^{НD} (Р) | 14/75 | 0.2 | 50 | < 9.000 1 | 0.0004 | 31 |
| _{НD} (Ъ) | 14/75 | 0.05 | 21 | < 0.999 1 | Ø. 6855 | 31 |
| GB | 21/75 | 0.2 | 2 | < 8 .0997 | 8.1452 | 23 |
| C8 | 21/75 | 9.55 | 2 | 0.0229 | 6.4324 | 23 |
| с8 ^(р) | 14/75 | 0.2 | 5 | 0.0001 | 0.0025 | 29 |
| ca(_{p)} | 14/75 | 0.05 | 9 | 9.0002 | 8.1613 | 29 |
| vx | 14/75 | Ø.2 | > 60 | < 9.9951 | < 0.0001 | 18 |
| vx | 14/75 | J. J 5 | > 60 | < 0.0001 | < 0.0001 | 18 |
| ۸X _(P) | 14/75 | 9.2 | > 50 | < 5.9991 | < S.0001 | 21 |
| ۸X(p) | 14/75 | 0.85 | > 50 | < 0.0001 | (9.989 1 | 21 |

(a) Gas temperature at filter inlet calculated from heat balances. The residence time of the fire products in the TOX area = one second. Worst-case fire size/combustion efficiency combinations shown.

(b) Gas temperature at filter inlet calculated from heat balances. The residence time of the fire products in the TOX area = 14.3 seconds. Worst-case fire size/combustion efficiency combinations shown.

TABLE 4. SUMMARY OF TOX AREA FIRE CALCULATIONS^(a)

Filter Inlet Gas Temperature Varied

| Agent | Fire Size/ Combustion / Efficiency (sq. ft./%) | Filter Inlet Gas Tesperature (F) | Time to Release)0.001 lb Agent (Min.) | Agent Released After 5 Minutes (1bs.) | Agent Released After Maximum Fire Duration (Ibs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Win) |
|-------------------|---|--|--|---|--|--|
| HD | 21/75 | 100 | 18 | < \$.\$85 1 | 6.0092 | 25 |
| HD | 21/75 | 114 | 4 | < 8 .8828 | 0.2814 | 25 |
| HD(P) | 14/75 | 100 | 32 | < 0.000 1 | 6.6009 | 31 |
| н0(ь) | 14/75 | 185 | 21 | < 0.0001 | 6.0058 | 31 |
| GB | 21/75 | 100 | 9 | 8.5551 | 5 .8662 | 23 |
| G8 | 21/75 | 116 | 2 | 6.8229 | 6.4324 | 23 |
| GE(b) | 14/75 | 198 | 17 | < 0.9001 | Ø. 838 5 | 29 |
| GB ^(b) | 14/75 | 1 # 6 | 9 | 0.0002 | Ø.1613 | 29 |
| vx | 14/75 | 168 | > 68 | < 8 .0001 | < 5.0001 | 18 |
| vx | 14/75 | 126 | > 68 | < 0.0001 | < 9.0001 | 18 |
| ٧X ^(b) | 14/75 | 166 | > 60 | < 0 .0601 | (0 .6661 | 21 |
| ۸×(p) | 14/75 | 125 | > 68 | < 5.5661 | (5 .5661 | 21 |

(a) Carbon capacity = \$.85 lb agent/lb carbon. The residence time of the fire products in the TOX area = 1 second Worst-case fire size/combustion efficiency combinations shown.

(b) Carbon capacity = **5.85** Ib agent/Ib carbon. The residence time of the fire products in the TOX area = 14.3 seconds.

Worst-case fire size/combustion efficiency combinations shown.



| | | | | | | and the second |
|-------|------------------------|---------------------------------|--|---|--|--|
| Agent | Fire Size (sq. ft.) | Combustion Efficiency (%) | Time to Release)0.001 lb Agent (Win.) | Agent Released After 5 Winutes (lbs.) | Agent Released After Maxisum Fire Duration (lbs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Min.) |
| HD | 21 | 75 | 4 | 6.0628 | 5.2814 | 25 |
| GB | 21 | 75 | 2 | 6.6229 | 6.4324 | 23 |
| vx | 14 | 75 | > 68 | < 0.000 1 | < 8.9991 | 17 |

WORST CASE (a)

MOST CREDIBLE CASE (b)

| Agent | Fire Size (sq. ft.) | Combustion Efficiency (%) | Time to Release)8.661 lb Agent (Min.) | Agent Released After 5 Winutes (1bs.) | Agent Released After Maximum Fire Duration (lbs.) | Maxiaum Fire Duration with 99% Probability of Operator Closing Damper (Min.) |
|-------|------------------------|---------------------------------|--|---|--|--|
| HD | 14 | 75 | 21 | < 0.0601 | 8.9058 | 31 |
| G8 | 14 | 75 | 9 | 6.6052 | 0.1613 | 29 |
| vx | 10 | 75 | > 68 | (9.869 1 | < 0.000 1 | 21 |

(a) Carbon capacity = 0.05 lb agent/lb carbon, filter inlet gas temperature calculated from heat balance. The residence time of the fire products in the TDX area = 1 second.

(b) Carbon capacity = 0.05 lb agent/lb carbon, filter inlet gas temperature calculated from heat balance. The residence time of the fire products in the TOX area = 14.3 seconds. Worst-case fire size/combustion efficiency used. It is important to note that a spill significantly less than 500 gallons can cause the worst-case or most-credible-case agent releases to be achieved because the fire areas for these events are approximately the same as or less than the TOX sump area of 20 sq. ft.

<u>3.1.2 BSA Area Fire</u>. The BSA Area fire involves the following sequence of events:

- (1) Contents of a filled ton container are spilled on the floor in the Buffer Storage Area.
- (2) Spilled agent is ignited.
- (3) Fire vaporizes agent, which is vented from the BSA to the carbon filters.

The variables described in the TOX Area fire were evaluated for the BSA Area fire. A summary of the calculations is given in Appendix A, pages A29 through A35.

Table 6 gives a summary of agent releases during a BSA fire for various fire sizes and combustion efficiencies. The size of an agent release is most dependent on fire size. Although large fires resulted in large rates of undecomposed agent being generated, the resultant temperature in the BSA (over 1000 F in some cases) would cause significant thermal decomposition of the agent. However, in some cases the high rate of undecomposed agent being expelled from the TOX could overwhelm the carbon filters due to limitations in the adsorption kinetics. Combustion efficiency had a significant effect on agent release for all cases, with the worst case being a 100 percent combustion efficiency. The much larger agent releases in the BSA Area as compared with the TOX Area are due to the availability of more ventilation air in the BSA, thereby allowing combustion and volatilization of agent at a more rapid rate.

Table 7 gives a summary of agent releases during a BSA fire for various residence times of fire products in the BSA. The most credible residence time of 35.6 seconds is equivalent to a fire directly beneath the BSA exhaust duct. A worst-case residence time was assumed to be 1 second. For a particular residence time, the agent released is dependent upon fire size. As shown in Figure 2, the amount of agent released



B-11

TABLE 6. SUMMARY OF BSA AREA FIRE CALCULATIONS (TON CONTAINER) (a)

| Agent | Fire Size (sq. ft.) | Combustion Efficiency (%) | Time to Release >#.#81 lb Agent (Min.) | Agent Released After & Winutes (lbs.) | Agent Released After Maximum Fire Duration (1bs.) | Maxiaum Fire Duration with 99% Probability of Operator Closing Damper (Min.) |
|-------|------------------------|---------------------------------|--|---|--|--|
| HD | Sump ^(b) | 1 36 | > 58 | < 5. 556 1 | 6.60 0 4 | 62 |
| HD(g) | 18 | 196 | 1 | 13.8317 | 75.2488 | 19 |
| HD | 77 | 188 | (d) | < 0.0001 | < 0.0001 | (c) |
| HD | Suep | 75 | (d) | < 0.9001 | < 5.550 1 | 88 |
| HD | 183 | 75 | (d) | < 1 .0001 | < 0 .0001 | (c) |
| HD | Su≡p | 50 | (d) | < 5 .8001 | < 5.6901 | 127 |
| HD | 156 | 50 | (b) | < 9.8001 | < 5.6001 | (e) |
| GB | Sump | 166 | 3 | 0.0050 | Ø.3593 | 60 |
| GB(9) | 18 | 100 | 1 | 29.8980 | 168.7586 | 19 |
| GB | 66 | 190 | (d) | < 5.000 1 | < 8.0001 | (e) |
| GØ | Sump | 75 | > 60 | < 5.0001 | < 8.601 | 78 |
| GB | 89 | 75 | (d) | < 9.000 1 | < 9.6501 | (e) |
| GB | Sump | 56 | > 68 | < 8.9061 | < 0.001 | 119 |
| C8 | 135 | 50 | (b) | < 9.3681 | < 0.0001 | (1) |
| W, | Sump | 165 | > 65 | < 8.8881 | < 5.5551 | 36 |
| (a) | 11 | 196 | 3 | 6.0138 | Ø.1391 | 16 |
| ٧X | 26 | 166 | (d) | < 9.5551 | < 5.6961 | 10 |
| YX | Sump | 75 | (d) | < 0.0001 | < 8.0001 | 43 |
| VX | 36 | 75 | (d) | < 5.6001 | < 8 .8691 | 18 |
| VX | Sump | 50 | (d) | < 8 .0001 | < 5 .6001 | 79 |
| VX | 54 | 50 | (d) | < 0 .0001 | < 8.0001 | 16 |

Fire Size and Combustion Efficiency are Varied

(a) Carbon capacity = 0.05 lb agent/lb carbon, gas temperature at filter inlet calculated from heat balances. The residence time of the fire products in the BSA area = 1 second.

(b) Sump area = 4 square feet.

23574 444

(c) The fire burns to completion within 8 minutes.

(d) The fire does not release agent from the BSA area.

(e) The fire burns to completion within 7 minutes.

(f) The fire burns to completion within 8 minutes.

(g) Worst-case fire area/combustion efficiency combination.

| Agent | Fire Size/ Combustion Efficiency (sq. ft./%) | Residence Tine (sec.) | Time to Release >0.001 lb Agent (Win.) | Agent Released After 5 Minutes (1bs.) | Agent Released After Maximum Fire Duration (lbs.) | Maximum Fire Duration with 999 Probability of Operator Closing Damper (Win.) |
|-------|---|-----------------------------|--|---|--|--|
| HD | 18/109 | 1 | 1 | 13.8317 | 75.2488 | 19 |
| HD | 18/100 | 2 | 1 | 18.6869 | 53.5485 | 19 |
| HD | 18/189 | 5 | 1 | 5.1578 | 20.6459 | 19 |
| HD | 18/100 | 18 | 1 | Ø.9675 | 4.7516 | 19 |
| HD | 18/108 | 35.6 | 7 | 5.8987 | 8.0035 | 19 |
| HD | 10/100 | 35.6 | 1 | . 4758 | 6.8254 | 29 |
| GB | 18/198 | 1 | 1 | 29.8986 | 168.7586 | 19 |
| GB | 18/156 | 2 | 1 | 23.9972 | 133.3433 | 19 |
| GB | 18/188 | 5 | 1 | 12.7584 | 68.9815 | 19 |
| GB | 18/196 | 10 | 1 | 4.7153 | 24.2835 | 19 |
| GB | 18/166 | 35.6 | 1 | Ø. 8382 | Ø. 1983 | 19 |
| GB | 11/166 | 35.6 | 1 | 2.2634 | 28.8833 | 26 |
| vx | 11/100 | 1 | 3 | 0.0138 | 6.1391 | 16 |
| ٧X | 11/155 | 2 | 3 | 0.0161 | 8.8941 | 16 |
| vx | 11/186 | 5 | 3 | Ø.8644 | 0.0350 | 16 |
| vx | 11/150 | 18 | 5 | 6.6613 | 6.0092 | 18 |
| vx | 11/188 | 35.6 | > 6# | < 0.8661 | < 0.600 1 | 16 |
| vx | 7/188 | 35.6 | > 5# | < 0.0001 | < 9.00 1 | 28 |

TABLE 7. SUMMARY OF BSA AREA FIRE CALCULATIONS (TON CONTAINER) (a) Residence Time of Fire Products in BSA Varied

(a) Carbon capacity = 0.05 lb agent/lb carbon. The filter inlet gas temperature calculated from heat balances. Worst-case fire size/combustion efficiency combinations shown.









77777554

KCC STAN

Exercise.





increases as the fire size increases, reaches a maximum, and then falls to zero. As the fire size is increased the amount of volatilized agent that is not combusted increases in proportion to the fire size. However, increasing the fire size causes the temperature in the area of the fire to increase such that thermal decomposition becomes significant. Because thermal decomposition is exponentially related to fire size through temperature, the amount of undecomposed agent decreases as the fire size increases. These trends are illustrated in Figure 3.

The fire size which gives the maximum agent release decreases as the residence time increases, as shown in Figure 2. This is because as the residence time is increased the amount of undecomposed agent released decreases for a particular fire size. Thus, smaller fires which result in a lower temperature and hence, lower degree of thermal decomposition, would favor generation of more undecomposed agent than larger fires. It is important to note that the worst-case fire size/combustion efficiency was calculated by trial and error for each worst-case and most-credible-case residence time for the TOX, BSA and ECV fire scenarios.

Table 8 gives a summary of agent release during a BSA fire for various carbon capacities. The carbon capacity has only a slight effect on the amount of agent released within the range of capacities evaluated. This is because the high temperature of the gases entering the filters makes adsorption unfavorable.

Table 9 gives a summary of agent releases during a BSA fire for various gas temperature at the filter inlet. The lower gas temperature had a significant effect on agent release amounting to a reduction of between three and five orders of magnitude.

The worst-case and most-credible-case agent releases for the BSA Area fire scenarios are given in Table 10. The most credible case was based on a 35.6-second residence time for the volatilized agent in the BSA, a carbon capacity of 0.05 lb agent/lb carbon (worst case), a filter inlet gas temperature calculated from heat balances (worst case), and the worst-case fire size/combustion efficiency combination. The worst case is as above except for a 1-second residence time. The most credible case is still very conservative for similar reasons to those given in the TOX area fire section.

B-15



TIGGERAR MERCEASE BESSESSE DOUDLESS DISTURD

Figure 3. Variance of Thermal Decomposition of Agent and Generation Rate of Undecomposed Agent with Fire Area for the BSA Fire

| | - | | | | _ | |
|-------------------|---|---|--|---|--|--|
| Agent | Fire Size/ Combustion Efficiency (sq. ft./%) | Carbon Capacity (1b agent/ 1b carbon) | Time to Release)0.001 lb Agent (Win.) | Agent Released After 5 Winutes (lbs.) | Agent Released After Maximum Fire Duration (lbs.) | Maxisum Fire Duration with 99% Probability of Operator Closing Damper (Min.) |
| HD | 18/195 | . 2 | 1 | 12.7931 | 61 . 2925 | 19 |
| HD | 18/190 | 0.05 | 1 | 13.8317 | 75.2488 | 19 |
| _{HD} (b) | 10/100 | 0.2 | 1 | 0.4213 | 3.9583 | 29 |
| HD (b) | 10/100 | Ø.05 | 1 | 6 .4758 | 6.8254 | 29 |
| GB | 18/166 | ●.2 | 1 | 26.8897 | 128.6018 | 19 |
| GB | 18/198 | 0.05 | 1 | 29.8980 | 168.7586 | 19 |
| GB (b) | 11/166 | Ø.2 | 1 | 1.9434 | 15.7784 | 26 |
| G8 (b) | 11/188 | Ø. 85 | 1 | 2.2634 | 28.8833 | 26 |

8.9163

8.6138

< 0.0001

< 8.8881

8.8666

6.1391

< 0.0001

< 0.0001

16

16

20

20

DESCRIPTION DESCRIPTIO

22222

TABLE 8. SUMMARY OF BSA AREA FIRE CALCULATIONS (TON CONTAINER) (a)

Carbon Capacity Varied

(b) Same as in (a) except the residence time of the fire production in the BSA area = 35.8 seconds.

the BSA area = 1 second. Worst-case fire size/combustion efficiency combinations shown

(a) Gas temperature at filter inlet calculated from heat balances. The residence time of the fire products in

3

3

> 68

> 66

٧X

٧X

vx (b)

vx (b)

11/100

11/168

7/188

7/188

1.2

8.85

1.2

0.65

TABLE 9. SUMMARY OF BSA AREA FIRE CALCULATIONS (TON CONTAINER) (a)

| Agent | Fire Size/ Combustion Efficiency (sq. ft./%) | Filter Inlet Gas Temperature (F) | Time to Release >0.601 lb Agent (Min.) | Agent Releases After 5 Minutes (Ibs.) | Agent Released After Maximum Fire Duration (lbs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Win.) |
|---------------|---|--|--|---|--|--|
| HD | 18/196 | 166 | 14 | < 0.0601 | 8.0074 | 19 |
| HD | 18/199 | 179 | 1 | 13.8317 | 75.2488 | 19 |
| HD (b) | 10/100 | 100 | 38 | (8.906 1 | 6. 566 9 | 29 |
| HD (b) | 10/100 | 143 | 1 | 0.4758 | 6.8254 | 29 |
| C8 | 18/196 | 155 | 7 | 0.0003 | 0.6973 | 19 |
| C8 | 18/196 | 183 | 1 | 29.8988 | 168.7586 | 19 |
| CB (b) | 11/188 | 160 | 17 | < 0.000 1 | 8.0144 | 26 |
| GB (b) | 11/188 | 152 | 1 | 2.2634 | 28.8833 | 26 |
| X X | 11/100 | 185 | > 64 | < 8.0001 | < 0.0001 | 15 |
| vx | 11/166 | 288 | 3 | 0.6138 | 0.1391 | 16 |
| AX (p) | 7/100 | 155 | > 65 | (8.8661 | < 8 .0001 | 25 |
| vx (p) | 7/100 | 174 | > 66 | (5.865 1 | < 0.0001 | 20 |

Filter Inlet Gas Temperature Varied

(a) Carbon capacity = 0.05 lb agent/lb carbon. The residence time of the fire products in the BSA area = 1 second Worst-case fire size/combustion efficiency combinations shown.

(b) Same as in (a) except the residence time of the fire products in the BSA area = 35.6 seconds.

andra Diana Tanàna

TABLE 10. BSA AREA FIRE WORST CASE/MOST CREDIBLE CASE AGENT RELEASES

| Agent | Fire Size (sq. ft.) | Combustion Efficiency (%) | Time to Release)0.001 lb Agent (Nin.) | Agent Released After 5 Minutes (lbs.) | Agent Released After Maxiaum Fire Duration (Ibs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Win.) |
|------------|------------------------|---------------------------------|--|---|--|--|
| HD | 18 | 100 | 1 | 13.8317 | 75.2488 | 19 |
| CB | 18 | 190 | 1 | 29.8985 | 168.7586 | 19 |
| v x | 11 | 186 | 3 | 6.6138 | 0.1391 | 16 |

WORST CASE (a)

MOST CREDIBLE CASE (b)

| Agent | Fire Size (sq. ft.) | Coebustion Efficiency (%) | Time to Release >0.061 lb Agent (Min.) | Agent Released After 5 Minutes (1bs.) | Agent Released After Maximum Fire Duration (lbs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Min.) |
|-------|------------------------|---------------------------------|--|---|--|--|
| HD | 10 | 199 | 1 | 0.4758 | 6.8254 | 29 |
| GB | 11 | 155 | 1 | 2.2634 | 28.8833 | 26 |
| VX | 7 | 188 | > 68 | (0.9001 | < 0.0001 | 26 |

(a) Carbon capacity = 0.05 lb agent/lb carbon, filter inlet gas temperature calculated from heat balances. The residence time of the fire products in the BSA area = 1 second.

(b) Carbon capacity = 0.05 lb agent/lb carbon, filter inlet gas temperature calculated from heat balances. The residence time of the fire products in the BSA area = 35.6 seconds.

It is important to note that the worst-case agent release would involve a TC that gradually leaks agent rather than a ruptured TC that spills the entire contents at once. The size of the fire following ignition of spilled agent from a leaking TC may be at the worst-case conditions depending upon the leak rate and spill configuration. However, ignition of the spill from a ruptured TC would probably cause an initial large fire that, because of thermal decomposition, releases an insignificant amount of agent to the environment. This large fire would rapidly consume agent and decrease in size until it is restricted to the sump at which time low levels of agent would be released to the environment. The fire would rapidly pass through the zone where large amounts of undecomposed agent are generated. As an approximation, agent released from a fire in the case of a ruptured TC can be taken as being equivalent to a sump fire for the entire fire duration.

<u>3.1.3 ECV Area Fire</u>. The ECV Area fire involves the following sequence of events:

 Contents of a filled ton container are spilled on the floor in the Explosive Containment Vestibule. The location assumed is given in Appendix A. N. COOCH

(2) Spilled agent is ignited.

(3) Fire vaporizes agent which is vented from the ECV area to the carbon filters.

The variables described in the TOX area fire were evaluated for the ECV area fire. A summary of the calculations is given in Appendix A, pages A36 through A41.

Table 11 gives a summary of agent releases during an ECV fire for varying fire size and combustion efficiency. As in the case of the BSA Area fire, both the fire size and combustion efficiency have a significant effect on the amount of agent released. The worst cases are 100 percent combustion efficiency/11 sq. ft. fire size for HD and GB. No significant VX releases were observed for any combination of fire size and combustion efficiency.

Table 12 gives a summary of agent releases during an ECV fire for various residence times of volatilized agent in the ECV. The most credible

TABLE 11. SUMMARY OF ECV AREA FIRE CALCULATIONS (TON CONTAINER) (a)

54.51.54F.55

Š

c.n

| Agent | Fire Size (sq. ft.) | Combustion Efficiency (%) | Time to Release >5.551 lb Agent (Min.) | Agent Released After 5 Winutes (lbs.) | Agent Released After Maximum Fire Duration (ibs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Win.) |
|-------------------|------------------------|---------------------------------|--|---|--|--|
| HD | Sump (b) | 166 | > 66 | < 5.6651 | < 5.555 1 | 30 |
| _{HD} (c) | 11 | 100 | 4 | 8.6822 | 8.8258 | 16 |
| HD | 48 | 100 | (e) | < 9.6661 | (5.66 51 | 16 |
| HD | Sump | 75 | > 60 | < 5.5601 | < 5.995 1 | 39 |
| HD | 64 | 75 | (•) | < 0.0201 | < 5.90 91 | 19 |
| HD | Sump | 58 | > 68 | < 0.0001 | < 9.000 1 | 69 |
| HD | 96 | 50 | (*) | < 0. 550 1 | < 9 .0601 | 19 |
| G8 | Sump | 166 | > 68 | < 5.0001 | < 5.66 51 | 30 |
| 68 ^(c) | 11 | 166 | 2 | 6.6118 | 6.1788 | 16 |
| G8 | 41 | 155 | (e) | < 8.0091 | < 5.0001 | 18 |
| GB | Sump | 75 | > 66 | < 5.0001 | < 0.0001 | 37 |
| G8 | 55 | 75 | (•) | < 8.500 1 | < 0.0 0 91 | 10 |
| C8 | Sump | 58 | > 68 | < 0.0001 | (5.555) | 53 |
| C 8 | 84 | 50 | (*) | < 0.0001 | < 9.6661 | (b) |
| vx | Sump | 188 | > 65 | < 8.900 1 | (0.6691 | 18 |
| VX (c) | 7 | 196 | > 88 | < 0.0001 | < 0.995 1 | 14 |
| YX | 17 | 166 | (•) | < 0.0001 | < 8.805 1 | 15 |
| YX | Sump | 75 | > 68 | < 0.0001 | < 9.0001 | 23 |
| VX. | 22 | 75 | (•) | < 0.0001 | < 5 .9901 | 10 |
| ٧X | Sump | 50 | > 68 | < 9.0051 | < 5 .6661 | 37 |
| ٧X | 34 | 58 | (e) | < 0.0001 | (0.000 1 | 19 |

Fire Size and Combustion Efficiency Varied

(a) Carbon capacity = 0.05 lb agent/lb carbon, gas temperature at filter inlet calculated from heat balances. The residence time of the fire products in the ECV area = 1 second.

(b) Sump area = 4 square feet.

(c) Worst-case fire size/combustion efficiency combination.

(d) The fire burns to completion within 18 minutes.

(e) The fire does not release agent from the ECV area.

TABLE 12. SUMMARY OF ECV AREA FIRE CALCULATIONS (TON CONTAINER)^(a)

| Agent | Fire Size/ Combustion Efficiency (sq. ft./%) | Residence Time (sec.) | Time to Release)8.661 lb Agent (Win.) | Agent Released After 5 Minutes (lbs.) | Agent Released After Maximum Fire Duration (lbs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Min.) |
|-------------------|---|-----------------------------|--|---|--|--|
| HD | 11/188 | 1 | 4 | < 8.8822 | Ø. 8288 | 16 |
| HD | 11/169 | 2 | 5 | 5.661 4 | Ø.Ø116 | 16 |
| HD | 11/100 | 5 | 9 | S. 0004 | Ø. ØØ29 | 16 |
| HD | 11/100 | 10 | 26 | < 0.000 1 | e. 6885 | 16 |
| HD | 11/158 | 21.1 | > 65 | < 8 .8651 | < 8.6561 | 16 |
| _{НD} (Ь) | 7/165 | 21.1 | 45 | < 0.0001 | 5 . 5 651 | 21 |
| C8 | 11/188 | 1 | 2 | 0.0118 | 5 .1785 | 16 |
| GB | 11/195 | 2 | 3 | J . 55 83 | 9.1071 | 16 |
| 68 | 11/188 | 6 | 3 | 0.0833 | J.J 313 | 16 |
| C8 | 11/185 | 16 | 6 | Ø. 8815 | 8.6671 | 16 |
| C8 | 11/168 | 21.1 | 21 | 8.6661 | 0.6867 | 16 |
| _{СВ} (b) | 7/166 | 21.1 | 19 | < 8 .5661 | 8.6616 | 22 |
| vx | 7/160 | 1 | > 60 | < 9.9091 | (9 .6001 | 14 |
| vx | 7/168 | 2 | > 66 | < 0.0001 | (5 .6951 | 14 |
| ٧X | 7/196 | 3 | > 60 | (0.666 1 | < 9.666 1 | 14 |
| ٧X | 7/196 | 10 | > 66 | (0.9901 | < 0. 999 1 | 14 |
| vx | 7/180 | 21.1 | > 60 | (8 . 566 1 | (0.0551 | 14 |
| ۲x ^(b) | 5/100 | 21.1 | > 6# | (5.966 1 | (5.6661 | 16 |

Residence Time of Fire Products in ECV Varied

(a) Carbon capcity = 0.05 lb agent/lb carbon. The filter inlet gas temperature calculated from heat balances. Worst-case fire size/combustion efficiency combination shown for the 1-second residence time.

(b) Same as in (s) except the worst-case fire size/combustion efficiency combination shown for the 21.1-second residence time.

residence time of 21.1 seconds is equivalent to a fire directly beneath the ECV exhaust duct. A worst-case residence time was assumed to be 1-second.

Table 13 gives a summary of agent releases during an ECV fire for variable carbon capacities. The carbon capacity has a significant effect on agent release. However, the amount of agent release was not directly proportional to the carbon capacity, but varied from about a two-fold to a ten-fold reduction in agent release as the carbon capacity was increased four-fold.

Table 14 gives a summary of agent releases during an ECV fire for various gas temperatures at the filter inlet. The lower gas temperature generally caused a reduction in the amount of HD and GB released by about two orders of magnitude.

The worst-case and most-credible-case agent releases for the ECV Area fire scenario are given in Table 15. The most credible case was based on a 21.1-second residence time for the volatilized agent in the ECV, a carbon capacity of 0.05 lb agent/lb carbon (worst case), a filter inlet gas temperature calculated from heat balances (worst case), and the worst-case fire size/combustion efficiency combination. The worst case was as above except for a 1-second residence time. The most credible case is still very conservative for reasons similar to those given in the TOX Area fire section.

<u>3.1.4 Carbon Bed Fire</u>. Two possible scenarios were considered for ignition of the carbon filter beds -- ignition from an entrained spark and spontaneous ignition. In the former scenario, a spark from a fire in the TOX, ECV, BSA or other area is entrained in the exhaust gases entering the filter banks. This would not cause a fire in the carbon bed because the pre-filter and HEPA filter, located upstream of the carbon beds, would stop the spark. These filters are composed of noncombustible fiberglass. The fiberglass would not achieve the melting temperature and allow the spark to pass through during any of the scenarios evaluated.

In the second scenario, the hot gases exhausted from a fire in the TOX, ECV, BSA, or other area or from a failure in the LIC ductwork, allowing exhaust gases from the operating LIC to enter the LIC room would heat the carbon bed. Based on the configuration of a CAMDS-type carbon bank, the



B-23

TABLE 13. SUMMARY OF ECV AREA FIRE CALCULATIONS (TON CONTAINER)^(a)



Carbon Capacity Varied

| Agent | Fire Size/ Combustion Efficiency (sq. ft./%) | Carbon Capacity (16 agent/ 16 carbon) | Time to Release >0.661 lb Agent (Min.) | Agent Released After 5 Winutes (lbs.) | Agent Released After Maximum Fire Duration (Ibs.) | Maxiaum Fire Duration with 99% Probability of Operator Closing Damper (Min.) |
|-------------------|---|---|--|---|--|--|
| HD | 11/160 | ●.2 | 4 | Ø.0016 | 8.5691 | 16 |
| HD | 11/188 | 8.85 | 4 | 8.9522 | Ø. 8258 | 16 |
| _{НD} (b) | 7/166 | Ø.2 | > 65 | (0 .0001 | (5.50 51 | 21 |
| _{НD} (ь) | 7/100 | 5.05 | 45 | < 0 .6081 | 6.8001 | 21 |
| GB | 11/100 | 0.2 | 3 | 0.0072 | 5 .5438 | 16 |
| GB | 11/188 | 6.05 | 2 | 8.5118 | 6.1788 | 16 |
| G8 (P) | 7/188 | 0.2 | 42 | < 8.9681 | 6.6864 | 22 |
| G8 (b) | 7/166 | 6.05 | 19 | < 0.6981 | 5.561 6 | 22 |
| vx | 7/100 | 0.2 | > 68 | < 0.0001 | < 9.9001 | 14 |
| vx | 7/169 | Ø. 85 | > 68 | < 5.6661 | (0 .6001 | 14 |
| vх (р) | 5/188 | 0.2 | > 68 | (0 .0081 | < 8 .6001 | 16 |
| vx (p) | 5/189 | 0.05 | > 68 | < 8 .9951 | < 0.6001 | 16 |

(a) Gas temperature at filter inlet calculated from heat balances. The residence time of the fire products in the ECV area = 1 second. Worst-case fire size/combustion efficiency combinations shown.

(b) Same as in (a) except the residence time of the fire production in the ECV area = 21.1 seconds.

| Agent | Fire Size/ Combustion Efficiency (sq. ft./%) | Filter Inlet Gas Temperature (F) | Time to Release >0.001 lb Agent (Win.) | Agent Released After 5 Minutes (lbs.) | Agent Released After Maximum Fire Duration (Ibs.) | Maxiaum Fire Duration with 99% Probability of Operator Closing Daeper (Min.) |
|-------------------|---|--|--|---|--|--|
| HD | 11/186 | 196 | 29 | < 5.900 1 | 9.6 5 91 | 15 |
| HD | 11/190 | 115 | 4 | 6.6622 | 0.0208 | 16 |
| _{НD} (Ь) | 7/186 | 188 | 59 | < 0.6691 | (0.000 1 | 21 |
| _{НD} (ь) | 7/100 | 103 | 45 | < 0.0001 | < 9 .5881 | 21 |
| GB | 11/100 | 188 | 16 | < 0.0001 | 6.6613 | 16 |
| CB | 11/100 | 117 | 2 | 6.5618 | ₫.1789 | 16 |
| G8 (P) | 7/166 | 100 | 32 | < 5 .5961 | Ø. 9692 | 22 |
| GB (b) | 7/168 | 185 | 19 | < 0.0001 | < 9.0061 | 22 |
| vx | 7/100 | 166 | > 68 | (9.685 1 | < 8.6061 | 14 |
| vx | 7/160 | 128 | > 68 | < 5.686 1 | < 9 .9991 | 14 |
| AX (p) | 5/100 | 150 | > 60 | < 9 .6661 | (8.6661 | 16 |
| ۸X (p) | 5/100 | 117 | > 68 | (0 .0001 | < 0 .0061 | 16 |

TABLE 14. SUMMARY OF ECV AREA FIRE CALCULATIONS (TON CONTAINER)^(a)

Filter Inlet Gas Temperature Varied

(a) Carbon capcity = 0.05 lb agent/lb carbon. The residence time of the fire products in the ECV area = 1 second. Worst-case fire size/combustion efficiency combinations shown.

(b) Same as in (a) except the residence time of the fire products in the ECV area = 21.1 seconds.
TABLE 15. ECV AREA FIRE WORST CASE/MOST CREDIBLE CASE AGENT RELEASES

TORST CASE (a)

| Agent | Fire Size (sq. ft.) | Coebustion Efficiency (%) | Time to Release)8.861 15 Agent (Min.) | Agent Released After 5 Minutes (1bs.) | Agent Released After Maximum Fire Duration (Ibs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Min.) |
|-------|------------------------|---------------------------------|--|---|--|--|
| HD | 11 | 199 | 9 | < 0.000 1 | 0.9102 | 16 |
| C8 | 11 | 166 | 7 | Ø.\$862 | S. 8895 | 16 |
| vx | 7 | 166 | > 68 | < 8.6661 | < 9.5561 | 14 |

MOST CREDIBLE CASE^(b)

| Agent | Fire Size (sq. ft.) | Combustion Efficiency (%) | Time to Release >8.661 lb Agent (Nin.) | Agent Released After 5 Minutes (lbs.) | Agent Released After Maxisum Fire Duration (lbs.) | Maximum Fire Duration with 99% Probability of Operator Closing Damper (Min.) | |
|-------|------------------------|---------------------------------|--|---|--|--|--|
| HD | 11 | 166 | 11 | < 9.6691 | 0.6035 | 16 | |
| CB | 11 | 196 | 8 | (0.0001 | 6.6149 | 16 | |
| XY | 7 | 180 | > 60 | < 0.0051 | < 0.00 1 | 16 | |

(a) Carbon capacity = 0.05 ib agent/ib carbon, filter inlet gas temperature calculated from heat balances. The residence time of the fire products in the ECV area = 1 second.

(b) Carbon capacity x = 0.2 lb agent/lb carbon, filter inlet gas temperature calculated from heat balances. The residence time of the fire products in the ECV area x one second.



minimum ignition temperature of the carbon was estimated to be 230 F (See Appendix A, pages A42 through A44). Raising the temperature of the carbon beds to 230 F or more could cause an ignition if sufficient time is allowed. To determine the sensitivity of temperature/time on carbon ignition, the worst-case filter inlet gas temperatures from the TOX, BSA and ECV Area fires were evaluated. The results, given in Table 16, indicate that spontaneous ignition is unlikely because of the short exposure periods of the carbon filter to elevated temperatures. No other scenarios for potential carbon ignition were identified.

<u>3.1.5 Agent Feed to Nonoperating LIC</u>. This scenario involves the following sequence of events:

- Shutdown of LIC burners/combustion air blowers while continued agent feed into the hot, but nonoperating LIC
- (2) Closure of the LIC exhaust damper, thereby isolating the LIC from the PAS
- (3) Vaporization of agent fed into the LIC as a result of contact with the hot refractory lining. There is a slight pressure buildup in the LIC until agent is vented into the LIC room, probably through the combustion air blower. The exhausted agent is then transported to the filter system via the ventilation.

The amount of agent released versus length of time that agent is fed into the nonoperating LIC was calculated. An agent flow rate into the LIC at a constant rate of 17.5 lb/min for HD and GB and 11.7 lb/min for VX was assumed as a worst case for the calculation. The previous fire scenarios indicated that the filter inlet gas temperature had a significant impact on the amount of agent released. As such, the temperature of the agent exhausted from the LIC was varied by changing the amount of refractory inside the LIC that is used to vaporize and heat the agent. These calculations are given in Appendix A, pages A45 through A49. Results of the calculations, given in Table 17, indicate that an operator has about 33 minutes to stop the agent feed into the nonoperating LIC before the agent release exceeds 0.001 lb.

| Scenario | Agent | Maximum T at Filters, °F | Time to Ignition of Activated Carbon | Fire Duration in Scenario, min. |
|-------------------------|-------|--------------------------------|---|---------------------------------------|
| TOX Fire ^(a) | HD | 154 | (b) | 76 min |
| TOX Fire | GB | 151 | (b) | 77 min |
| TOX Fire | VX | 148 | (b) | 187 min |
| BSA Fire ^(a) | HD | 305 | 80 min | 8 min |
| BSA Fire | GB | 296 | 85 min | 6 min |
| BSA Fire | vx | 287 | 100 min | 17 min |
| ECV Fire ^(a) | HD | 167 | (b) | 14 min |
| ECV Fire | GB | 164 | (b) | 12 min |
| ECV Fire | vx | 162 | (b) | 29 m in |
| LIC(c) | A11 | 230 | >9 hrs | |

TABLE 16.TIME REQUIRED FOR SPONTANEOUS IGNITION
OF CARBON DUE TO HEATING

(a) All worst-case values given here

(b) Below minimum temperature required for ignition.

(c) LIC/AB exhausts into LIC area.



| Agent | Fraction of LIC Refractory that Heats Agent | Time to Release >0.001 lb Agent (Min.) | Agent Released After 5 Minutes (lbs.) | Agent Released After 20 Minute (lbs.) |
|-------|---|--|---|---|
| HD | 1.0 ^(a) | 33 | < 0.0001 | < 0.0001 |
| HD | 0.1 | (b) | < 0.0001 | < 0.0001 |
| GB | 1.0 | 33 | < 0.0001 | < 0.0001 |
| GB | 0.1 | (b) | < 0.0001 | < 0.0001 |
| vx | 1.0 | > 60 | < 0.0001 | < 0.0001 |
| ٧X | 0.1 | (b) | < 0.0001 | < 0.0001 |

TABLE 17. AGENT RELEASE FROM CARBON FILTERS WHILE CONTINUED AGENT FEED INTO NON-OPERATING LIC

 (a) 1.0 implies that the entire inner layer of high conductivity refractory (4-1/2-inch thick) within the volatilization chamber (52 inches ID by 7-ft. ht.) is available to volatilize agent fed into the LIC.

(b) The refractory cools to below the boiling point before <0.001 lb. is released.



B-29

<u>3.1.6 Carbon Filter Desorption</u>. A telephone conversation with Dr. Gerry Wood of the Air Purification Branch at the Chemical Research Development and Engineering Center revealed a general lack of agent desorption data. The desorption process cannot yet be modeled by empirical correlations. However, in qualitative terms, desorption may be insignificant. A report was cited (Reference 1) in which no GB or GD was desorbed after purging a carbon filter for 30 days at ambient temperature.

<u>3.1.7 PAS Agent Scrubbing</u>. The potential for agent removal in the PAS quencher was evaluated. The LIC PAS was used as the basis for the calculations. The calculations are given in Appendix A, pages A50 through A56.

The equations used to estimate agent scrubbing efficiency in the quencher indicated a strong dependence on the droplet size emitted from the quencher spray nozzle. Based on designed flow rates, the nozzles in the quencher should result in a median particle diameter of 1000 microns or less. A diameter of 4000 microns (worst case) and 100 microns (optimistic case) were also evaluated. The effect of gas residence time in the quencher was also evaluated ranging from 2.0 seconds (worst case) to 4.0 seconds (optimistic) as well as the 2.9 second designed residence time.

The results of the calculations are summarized in Table 18. The worst case agent removal efficiency (4000 micron particle size, 2.0 second residence time) was about 50 percent while the most optimistic (100 micron particle size, 4.0 second residence time) was over 99.999 percent. The most credible removal efficiency (1000 microns particle size, 2.9 second residence time) was 68.7 percent.

<u>3.1.8 MPF/Full TC</u>. The MPF accident that was evaluated involves inadvertent processing of a full TC in the MPF. It was assumed that the MPF burners would remain in operation after the TC was placed in the MPF (i.e., plant personnel were unaware that a full TC was placed in the MPF). Several scenarios were evaluated for this accident. Scenario 3 is considered to be the worst case.

In scenario 1, the agent volatilizes from the TC through punched holes at a rate dependent on the MPF burners heat duty. Sufficient area is

| Liquid Particle Size (Microns) | Residence Time of Gas in Quencher (Sec.) | Removal Efficiency (percent) |
|--------------------------------------|--|---------------------------------|
| 4000 | 2.0 | 49.7 |
| 4000 | 2.9 ^(a) | 63.1 |
| 4000 | 4.0 | 74.7 |
| 1000 ^(b) | 2.0 | 54.2 |
| 1000 | 2.9 ^(a) | 68.7 |
| 1000 | 4.0 | 80.5 |
| 100 | 2.0 | 98.7 |
| 100 | 2.9 ^(a) | 99.94 |
| 100 | 4.0 | > 99.999 |

TABLE 18. AGENT REMOVAL IN LIC QUENCHER

Ų

EXERCISE DURING BARANCE PRIMARY 1994

(a) Designed gas residence time.

(b) Typical median particle size from spray nozzle operating at flow rates specified on design drawings.



100 miles

available in the punched holes so as to prevent over-pressurization of the TCs. Assumptions used in scenario 1 calculations include:

- A single TC placed in the MPF inadvertently
- Agent burns in the TC but container does not rupture
- Combustion-quench air at 3690 lb/hr in MPF
- Agent is at 120 F when placed in the MPF
- MPF operates at 1600 F

• Thermal input to TC is 1,745,953 Btu/hr (Radiation and Convection).

The calculations are given in Appendix A, pages A58 through A63. The agent flow rates from the MPF to the afterburner resulting from scenario 1 are shown in Table 19. The "agent not combusted", shown in Table 19, represents the amount of agent in 1b/min not combusted in the MPF under stoichiometric conditions. These values are reasonable considering the fact that the MPF was designed to burn only residual agent on various metal parts and one TC. The agent not combusted in the MPF will flow into the afterburner via the MPF exhaust flow and will be thermally decomposed there if the afterburner continues to function, normally with a 2-second residence time for MPF exhaust. As such, no significant agent release to the environment would result during this scenario. Also, as described in the calculations (Appendix A), an agent vapor/air explosion should not be possible due to the limited amount of oxygen available in the MPF.

In scenario 2, the TC would rupture when heated in the MPF due to over-pressurization. The contents of the container would be ejected to the floor of the MPF. All of the agent is not vaporized instantly but, rather, the vaporization rate is dependent on the rate of heat transfer by conduction from the refractory to the agent. Assumptions used in scenario 2 calculations are as follows:

- The agent does not vaporize instantly and it is concentrated on the floor area
- Heat transfer to agent is primarily by conduction through the floor refractory

B-32



TABLE 19. TIMES TO VAPORIZE AGENT FROM ONE TON CONTAINER PLACED IN MPF

| Agent | Time (min) | Mass Liq (1b) | | Mass Vapor (1b) | Vapor Flow (cfm) | Mass of Agent Comb (lb/min) | Mass of Agent not Combusted (lb/min) to Afterburner |
|-----------------|---------------|---------------------|------|-----------------------|------------------------|---|--|
| HD | | | | | | | |
| Boil (411 F) | 6 | | 1700 | 0 | 0 | 0 | 0 |
| Sat Vap | 15.8 | 3 | 0 | 1700 | 968 | 8.64 | 99.0 |
| GB | | | | | | | |
| Boil (316 F) | 4.2 | | 1600 | 0 | 0 | 0 | 0 |
| Sat Vap | 12.3 | 3 | 0 | 1600 | 1353 | 7.0 | 124 |
| VX | | | | | | | |
| Boil (568 F) | 9.4 | | 1500 | 0 | 0 | 0 | 0 |
| Sat Vap | 16.0 | 5 | 0 | 1500 | 486 | 4.0 | 86.4 |

- TAXABLE MASSION
- 4.5-in of refractory with high thermal conductivity contributes to heat flux into the liquid
- Agent spills at the boiling point
- Thermal conductivity of the 4.5-in refractory slab is 2.6 Btu/hr-ft-F
- Average slab temperature is 1600 F.

The calculations are given in Appendix A, pages A67 through A77. The agent flow rates from the MPF to the afterburner, summarized in Table 20, indicate that no significant agent release to the environment would occur during this scenario. These flows should be easily combusted in the afterburner since the residence time will be higher than normal without full MPF combustion exhaust. Since the flow capacity for the 24-in-diameter duct is approximately 2500 scfm at the nominal 2 iwg pressure differential between the MPF and afterburner, there will be no pressure rise in the MPF at these conditions.

In scenario 3, the TC ruptures and the entire contents are instantly vaporized. The agent flow rates from the MPF to the afterburner, the afterburner destruction efficiencies (the afterburner was assumed to flame out due to the large spike of agent vapor) and the amounts of agent released to the environment are given in Table 21. The calculations are given in Appendix A, pages A76 through A90. This scenario assumes, as a worst case, that the entire agent is vented through the afterburner. However, because of the over-pressure resulting from the vaporization of the agent, the MPF fume containment would be compromised, thereby expelling agent into the MPF area. Table 22 indicates that over-pressures that would likely cause MDB structural failure can occur if as little as one-fourth of the contents of a TC were expelled to the MPF room in this manner. Because two of the MPF walls are located adjacent to the outside, essentially all of the agent involved could be released to the environment. Any combination of variables could result in a significant agent release to the environment due to the large over-pressures. Although not quantitatively estimated, scenario 3 could result in the essentially instantaneous release of hundreds of pounds of point to the environment.

| Agent | Mass Agent Vapor | Volume Agent | Time | Ventilation Rate |
|-------|------------------|--------------|-------|------------------|
| | (1bm) | (cu ft) | (min) | (cfm) |
| HD | 1700 | 14070 | 16 | 879 |
| | 850 | 7517 | 8.2 | 916 |
| | 425 | 3896 | 4.1 | 950 |
| vx | 1500 | 7568 | 9.7 | 780 |
| | 750 | 3996 | 4.8 | 832 |
| | 375 | 2058 | 2.4 | 858 |
| GB | 1600 | 14952 | 14.2 | 1053 |
| | 800 | 8008 | 7.1 | 1128 |
| | 400 | 4156 | 3.5 | 1187 |

TABLE 20. AGENT VAPORIZATION RESULTING FROM AGENT SPILLS ON HOT MPF FLOOR

100. Ő

Instantaneous Vaporization of Agent in MPF Table 21.

3. Analysis of Agent Breakthrough to AfA Instantaneous Agent Volitization in MPF No Agent Pooling Scenario No. Assumptions:

Entire Refractory Volume Contributes to Vaporization (258 cu Average Refractory Temperature 740 F

| Parameters | ₽ | GB | ۸۷ |
|---|---|--|--|
| Ta (F) Mass Rgent (lbm) Worst Case M4 (lb/mole) R (ft-lbf/lbmole-R) Volume (cu ft) Pressure Rise (psia) Density @Ta (lb/cu ft) | 707 1700 1590 1545 1545 1545 174, 3120 0, 171656 | 708 1600 140.1 1545 768 186.3500 186.3500 | 717 267.4 1545 1545 768 768 768 768 768 768 768 768 768 768 |
| Volume Req Vent (cfm) Vent Flow Inst Flash(cfm) CF Pressure Flow to MPH Exhaust (cfm) Actual Flow (acfm) Time to Purge Agent (sec) | 9175.576 5044545. 100.4037 84962.02 204168.4 2.696472 | 9809.240 5424744. 107.3376 93113.15 223872.5 22628970 | 4855.310 2452386. 53.12923 47406.10 114514.1 2.543952 |
| Mass of Agent to AFB (lbm) Calculation of AFB Breakthrough AFB Temp Volume AFB in cu ft 55 | 1575.047 for AFB Fla | 1468.750 me Out | 1386.886 |

0.153402 0.139901 0.273503 611.2651 1079.514 1211.773 0.611906 0.265011 0.126263 Mass Breakthrough in AFB (1bm) Destruction Efficiency (%) 1

Residence Time (t) sec

1 1

Calculation for AFB Breakthrough AFB Operating @ 1800 F

| JFB Temp (F) 18 | 00 | | |
|---|------------|-------------|----------|
| Jolume AFB in cuft 5 | 22 | | |
| <pre>APF Exhaust @1600 F ACFM 126</pre> | 3 0 | | |
| JFB Exhaust @1000 F ACFM 156 | 20 | | |
| actual Flow to MPF (acfm) | 205945.4 | 225649.5 1 | 16291.1 |
| lass of Agent To AFB (lbm) | 1575.047 | 1460.250 1 | 306.086 |
| Residence Time (sec) | 0.152079 | 0.1.00249-0 | 0.269373 |
| lass Breakthrough to AFB (1bm) | C | Ð | c |
| Jestruction Efficiencey (2) | - | - | - |

Mass Breakthrough to AFB (16m) Destruction Efficiencey (2)

EVAN TEREDER TEGESSEE DEREREN




| MPF Area P Rise (psig) | 18.04 | 11.07 | 5.37 | 19.36 | 11.87 | 6.04 | 16.38 | 10.40 | 0.04 40 |
|--------------------------------------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|------------|------------|
| Vent Rate for No P rise (cfm) | 4,810,075 | 2,953,613 | 1,484,152 | 5,163,885 | 3,166,191 | 1,610,899 | 4,367,917 | 2,780,890 | 1,423,940 |
| Gas Volume @ ambient P (cc ft) | 12754 | 57283 | 45038 | 75702 | 20002 | 10001 | 69069 | 55844 | 44536 |
| MPF Area Trup. (F) | 6E9 | 460 | 286 | 675 | 484 | BOR | 040 0 | 404 404 | 290 |
| Agent Gas Volume (cu ft) | 13969 | 7464 | 3869 3 | 15187 | 8044 | 4151 | 04E2 | 026E | 2602 |
| MPF Exit Temp. (F) | 1931 | 4041 | 1524 | 1362 | 1470 | 1532 | 1 335 | 1400 | 1525 |
| Total Wright (155) | 1700 | 850 | 425 | 1600 | 800 | 400 | 1500 | 750 | 375 |
| Agent Lee | 0H | | | 613 | 60 | 60 | x > | x > | ×> |

B-37

3.2 Summary/Conclusions

Sensitivity analyses were performed for several accident scenarios involving relatively large quantities (i.e. over 100 pounds) of agent. A summary of agent releases for the accident scenarios evaluated are given in Table 23. Other conclusions are as follows:

- Insufficient information is available to quantify desorption of agent from carbon filters.
- Between 50 (worst-case) and 99.999 (most-credible-case) percent removal efficiencies of agent are anticipated in the PAS quencher.

3.3 References

 Morrison, R. W.; Rogers, C. L.; Grue, R. C.; and Hiob, G. D.; "Effect of Relative Humidity on the Performance of ASC Carbon in the Removal of Chemical Agents", CRDC-TR-86012, February, 1986.

| | Wors | t-Case | Most_Credible-Case |
|--|----------------------|---|------------------------------------|
| Scenario | Agent Agent | Release (lbs) | Agent Release (lbs) |
| TOX Area Fire | HD GB | 0.2814 | 0.0050 |
| BSA Area Fire(b) | VX HD GB | <pre>< 0.0001 75.2408 168.7586</pre> | < 0.0001 6.8254 28.8833 |
| BSA Area Fire(c) | VX HD | 0.1391 0.0004 | < 0.0001 |
| ECV Area Fire | VX HD | <pre>0.3593 < 0.0001 0.0288 0.1202</pre> | 0.0001 |
| Carbon Filter Fire | AD AD AD | <pre>0.1788 < 0.0001 (d) </pre> | < 0.0016 < 0.0001 (d) |
| Agent Feed to non- | VX HD GB | (d) (d) < 0.0001 < 0.0001 | (d) (d) < 0.0001 < 0.0001 |
| Feed Full TC into MPF (Scenarios 1 and 2) | VX HD GB | <pre> 0.0001 < 0.0001 < 0.0001 < 0.0001 </pre> | < 0.0001 |
| Feed Full TC into MPF (Scenario 3) | VX HD GB VX | <pre> 0.0001 100 100 100 100 100 </pre> | |
| | - | • • • | |

TABLE 23. SUMMARY OF AGENT RELEASES FROM ACCIDENT SCENARIOS EVALUATED IN THE SENSITIVITY ANALYSES

(a) Agent releases for the fire scenarios are for the maximum fire duration.

(b) The agent releases given here are for a leaking TC.

(c) The agent releases given here are for a ruptured TC and assumes agent release from a sump fire.

(d) Ignition of the carbon is not anticipated in any of the evaluated scenarios.

AND A CONTRACTOR OF A CONTRACTOR



APPENDIX C STRUCTURAL ANALYSIS 

Sec. Sec.

C.1. STRUCTURAL ANALYSIS

This appendix summarizes the structural analysis methodology used to determine failure thresholds and probabilities for munitions and structures. Supporting calculation for the results used in this study can be found in Ref. C-1.

C.1.1. PUNCTURE

This section addresses two types of munition puncture: (1) puncture due to dropping a munition; and (2) forklift puncture.

C.1.1.2. Puncture Due to Drop

The probability P_F of a munition puncturing on impact with a probe depends on the type of munition, the number of probes to which a dropped munition is exposed, and the geometry of the probe. This probability is computed from the following:

 $P_F = P_B \times PLL \times PD \times A_s$

where $P_B =$ probe density (number of probes per square foot of surface area),

PLL = an admissible probability value for probe length to diameter
 ratio,

PD = an admissible probability value for probe diameter,

The number of probes per square foot of surface area (P_B) is based on engineering judgment. It is assumed that the igloo is clean and that objects that could be potential probes are not likely to be left in the igloo. Therefore, one probe per igloo (i.e., one probe per 2160 ft²) was assumed for igloo storage. For all other storage areas, a probe density of one per 1000 ft² was assumed. In the general working area, loading docks, etc., it is assumed that the potential for probes will be much more likely than in an igloo. Probes such as posts, tools, rocks, or chunks of steel are possible; therefore, one probe per 100 ft² is assumed for the general working area. In the UPA during an earthquake, it is assumed that the earthquake could generate additional probes by causing objects to fall onto the floor; therefore, one probe per 50 ft² is assumed for the UPA during an earthquake.

The PLL term in the above expression represents the probability that the probe has a length-to-diameter ratio (L/D) which is less than that which would cause buckling failure of the probe without penetration of the dropped munition but greater than that corresponding to a probe length which is insufficient to penetrate the munition. Probe dimensions (diameter and L/D) were treated statistically and the minimum probe length for penetration was calculated for each munition.

The PD term in the above expression represents the probability that the diameter of the probe is less than or equal to the maximum that could penetrate the munition but greater than a minimum diameter corresponding to the compressive strength of the probe. The maximum diameter of the probes which could penetrate through the munition wall is determined from

$$D_{\rm u} = \frac{(W \times H)^{0.667}}{672 t}$$



where D_u = maximum probe diameter (in.), W = weight of munition/pallet (lb), H = drop height (ft), t = munition thickness (in.).

These expressions are taken from Ref. C-2.

The munition area vulnerable to probe penetration (A_S) was determined assuming a maximum probe length of 2 in. This term was calculated for each munition/pallet configuration of interest and reflects the number of munitions involved in each handling operation. Thus, if more than one munition were being handled, the vulnerable area of each munition was multiplied by the actual number of munitions involved in the handling event.

C.1.1.2. Forklift Tine Puncture

For forklift time puncture, the munitions are at rest and the probe (the forklift time) is the moving object. This makes calculating the munition vulnerability simpler since the mass of the moving object (the forklift) and the shape of the probe (the time) are the same for all munitions. The only variable is the munition thickness. Since the puncture energy is proportional to the thickness of the munition, the relative puncture resistance of the munitions is simply the ratio of the thicknesses.

The probability P of a forklift time puncture of the munitions was assumed to be governed by

$P = P_1 \star P_2 \star N ,$

where P_1 = the probability that a munition is struck by a forklift time per pallet operation,

• • •

 P_2 = the probability that the munition is punctured given that the forklift time strikes the munition,

N = number of handling operations.

The critical puncture velocity V_c (in ft/s) was determined from

$$V_c = \frac{64}{W} (672 \text{ Dt})^{3/2}$$

where W = weight of the forklift (1b),

- D = equivalent diameter of the forklift tine (in.),
- t = munition wall thickness (in.).

C.1.2. WIND-GENERATED MISSILES

The probability of a wind-generated missile rupturing a munition is the product of two probabilities: (1) the probability of having a wind of sufficient velocity to generate a missile that can rupture a munition and (2) the probability that the missile hits the munitions in an orientation that will rupture the munition.

C.1.2.1. <u>Required Wind Velocity</u>

The wind velocity required to generate a missile that can penetrate a munition is computed as follows:

 The missile velocity required to penetrate the munition is computed using the equation (Ref. C-2):

$$V_{\rm m} = 0.682 \quad \frac{64}{W} (672 \ Dt)^{3/2}$$

where V_m = the penetration velocity (mph), W = the weight of the missile (lb), D = the equivalent missile diameter (in.), t = the wall thickness of the munition (in.).

Each munition was evaluated for two critical missiles: a 10-ft section of 3-in. pipe and a 13.5-in. diameter utility pole. In addition to penetration, the utility pole was evaluated to determine the velocity required to crush the munition.

バン

 The missile velocity required to penetrate the storage structure was also computed using the following equation (Ref. C-2).

a de de la contra d

For concrete structures:

$$V_{\rm g} = 1000 \frac{f_{\rm c} T D^{1.8}}{427 W} 0.75$$

- where T = thickness of concrete element to be just perforated (in.),
 - W = weight of missile (lb),
 - D = diameter of missile (in.),
 - V_s = striking velocity of missile (fps),
 - $f_c = compressive strength of concrete (psi).$

For steel structures:

$$V_s = 0.682 \quad \frac{64}{W} (672 \text{ DT})^{3/2}$$

,

3. The missile velocity required to penetrate both the munition and structure is computed using the following equation which is based on summing the energies required to penetrate the munition and structure separately:

$$\mathbf{v} = \sqrt{\mathbf{v}_{\mathrm{m}}^2} + \mathbf{v}_{\mathrm{s}}^2$$

where V_m = velocity required to penetrate the munition, V_s = velocity required to penetrate the structure.

4. The probability of the required wind occurring was based on functional data for each site.



C.1.2.2. Probability of Hitting and Rupturing the Munition

Given a sufficient wind, the probability that a missile hits and ruptures a munition was computed from:

 $P = P_d P_o D A$,

- where $P_d =$ probability that the direction of missile travel is nearly perpendicular to the target,
 - P_o = probability that the missile is oriented to penetrate (i.e., not tumbling or going sideways),
 - D = number of missiles per unit area,
 - A = area of target.

Values for P_d , P_o , and D are difficult to evaluate and are not available from the literature. Consequently, the values used for the analysis were computed based on engineering judgment. These values were selected to give a "best estimate" of the overall probability. The following is a discussion of these assumptions.

The missile velocity must be nearly perpendicular to the wall of a structure or munition in order for the missile to penetrate. The further the missile strikes from an angle which is perpendicular, the less likely that the missile will penetrate. As the angle deviates from the perpendicular, the effective thickness of munition increases proportionally to the reciprocal of the cosine of the angle (where the angle is measured from the perpendicular); thus, a higher missile velocity (which has a lower probability of occurring) is required for penetration. In addition, the missile is more likely to ricochet at higher angles. Based on engineering judgment, it is estimated that if the missile velocity is more than 30 deg off from perpendicular, the missile will not penetrate. This yields a value of 0.17 for P_d .

The missile velocity must be aligned along the missile axis in order for the missile to penetrate. In other words, the missile must move like an arrow rather than tumbling or going sideways. Of the two missiles analyzed, it was found that it is more important that the pipe be aligned properly than the utility pole because of the larger impact area of the utility pole. For this reason, it was assumed that the velocity must be aligned within 5 deg of the axis of the pipe and within 10 deg of the axis of the utility pole. These assumptions resulted in values for P_0 of 0.004 for the pipe and 0.015 for the utility pole. The path of the tornado is generally from 1/8 to 3/4 of a mile wide (Ref. C-3). For this analysis, it was assumed that the tornado is 1/2 mile wide and that it carries one utility pole and 10 iron pipes. It was further assumed that the pipes are evenly distributed to a height of 50 ft and the utility pole at a height of 20 ft (Ref. C-4 indicates the maximum heights for pipes is 100 ft and for utility poles is 50 ft which indicates that our assumption is conservative). Therefore, the number of missiles per square foot of wind (D) is 7.6 x 10^{-5} for pipes and 1.9 x 10^{-5} for utility poles.

The target area is different for each scenario and depends on the number of munitions involved and the storage configuration (see Ref. C-1).

The product of P_d , P_o , and D is approximately 5.0 x 10^{-8} for both the pipes and utility pole.

C.1.3. EARTHQUAKE AND WIND FAILURE OF UBC DESIGNED STRUCTURES

C.1.3.1. Strength Factor of Safety

The Uniform Building Code (UBC) ensures that structures are designed with a factor of safety. This factor of safety varies depending on the type of structure, materials used and components selected. For earthquake and wind loads, this factor of safety ranges from 1.3 to 1.6 for concrete structures designed to ultimate design strength principals and from 2.6 to 3.0 for concrete and steel structures designed to working stress methods. For the risk analyses in this report, it is assumed that the factor of safety will be 1.3 for concrete structures (since the CONUS structures are being designed to ultimate strength) and 2.6 for the steel structures.

C.1.3.2. Wind Loads

For UBC-designed concrete structures such as the MDB, wind does not govern the design of the main structural components. The MDB is a rigid concrete moment resisting framed and shear wall structure and will fail under seismic conditions only. For the steel structures such as the bulk agent warehouses, the wind governs the design in most cases. Wind loads will fail the walls of the structure before the structure will collapse. Since the stresses in a structure due to winds are proportional to the square of the wind velocity, a wind velocity which is 1.6 (square root of the 2.6 factor of safety on strength) times greater than the design wind load can be expected to fail the walls of the steel structure.

C.1.3.3. Earthquake Loads

The Applied Technology Council (ATC), which is associated with the SEAOC, presents a set of curves that can be used to estimate the probability of an earthquake, which exceeds a specific g-level, occurring

anywhere in the U.S. (Ref. C-5). These curves are shown in Section 4.2. Each curve represents a seismic map area which is similar to the seismic zones used by the UBC. The ATC divided the country into seven seismic map areas (1-7). The UBC uses five seismic zones (0-4). Reference C-5 contains maps showing the seismic map areas. These maps color code the seismic map areas, and, consequently, have not been reproduced for this report since a black and white reproduction would not be helpful. The maps show that APG, ANAD, LBAD, PBA, UMDA, and PUDA are in seismic map area 2; NAAP is in seismic map area 3; and TEAD is in seismic map area 5. **NAME OF CONTRACT**

Section 4.2 presents the seismic risk curves for seismic map areas 2, 3, 5, and 7.

The earthquake g-level that will fail a structure depends on four principal factors: (1) the design g-level, (2) the strength factor of safety, (3) the dynamic amplification in the structure, and (4) the ductility of the structure. The dynamic amplification factor reduces the factor of safety, and the ductility increases the factor of safety. The dynamic amplification factor has been conservatively estimated at 2.3 based on a referenced analysis (Ref. C-6). Ductility factors are estimated to be in the range of 2.5 to 3.5 for concrete structures with shear walls and from 3.5 to 5.0 for steel structures. For this analysis, 2.5 was used for concrete walls and 3.5 was used for steel-walled structures. Based on these factors, a UBC structure with concrete walls was assumed to fail at an earthquake g-level that is approximately 1.4 times the design g-level, and a UBC structure with steel walls was assumed to fail at a g-level that is approximately 4.0 times greater than the design g-level.

For UBC designed structures with concrete walls in Seismic Zone 3 (design g-level of 0.14), the expected failure g-level is 0.4 g. Due to the uncertainty of the analysis, there is a probability that the structure will survive larger earthquakes or will fail during smaller

earthquakes. Consequently, the following probabilities of failure have been assumed:

A 0.3-g earthquake has a 0.1 probability of producing failure.
 A 0.4-g earthquake has a 0.5 probability of producing failure.
 A 0.5-g earthquake has a 0.9 probability of producing failure.
 A 0.6-g earthquake has a 1.0 probability of producing failure.

LEVELES TEASERS

The failure g-levels for Seismic Zone 2 are half of the g-levels for Seismic Zone 3 since the design g-level for Seismic Zone 2 (0.07 g) is half the design g-level for Seismic Zone 3 (0.14 g).

For UBC designed structures with steel walls in Seismic Zone 2 (the warehouses at NAAP and UMDA), the following probabilities of failure have been assumed:

A 0.2-g earthquake has a 0.1 probability of producing failure.
 A 0.3-g earthquake has a 0.5 probability of producing failure.
 A 0.4-g earthquake has a 0.9 probability of producing failure.
 A 0.5-g earthquake has a 1.0 probability of producing failure.

C.1.4. EARTHQUAKE FAILURE OF NRC-DESIGNED STRUCTURES

The TOX cubicle, tank, and piping system will be designed to Nuclear Regulatory Commission (NRC) standards for nuclear power plants. In summary, this will involve the following:

- Seismic experts will determine the "maximum credible earthquake" that can occur at TEAD based on the seismic history of the area and the proximity of earthquake faults. This "maximum credible earthquake" will be selected as the safe shutdown earthquake (SSE) to be used as the design earthquake for the TOX at all eight sites.
- 2. The TOX will be analyzed for the SSE using finite-element time-history computer programs.
- 3. The TOX will be constructed to NRC standards.

Since the design g-level has not yet been determined, an SSE g-level had to be assumed with the intent to ensure that the TOX will withstand relatively high g-forces. For this risk analysis, it was conservatively assumed that the TOX will be designed for a 1-g SSE. FYZZZZY AL RECEVEN

Since the TOX will be designed for no failures in the event of a SSE, an earthquake larger than the SSE will be required to produce a failure. Since the NRC seismic design requirements are quite different from the UBC seismic requirements, the methodology used to determine failure g-levels for the UBC structures does not apply to NRC-designed structures. Based on GA's experience in seismic design of nuclear power plants, it was estimated that an earthquake which is twice the SSE will have a 0.5 probability of either rupturing the TOX tank/piping system or breaching the TOX wall. There is a possibility that the TOX will survive larger earthquakes or that a smaller earthquake will cause a

C-12



failure. Consequently, the following probabilities are selected for the rupture of the TOX storage tank and for the breaching of the TOX walls:

- 1. A 1.8-g earthquake has a 0.1 probability of producing failure.
- 2. A 2.0-g earthquake has a 0.5 probability of producing failure.
- 3. A 3.0-g earthquake has a 0.9 probability of producing failure.
- 4. A 4.0-g earthquake has an ~1.0 probability of producing failure.

C.1.5. METEORITES

The probability of a meteorite penetrating a munition can be estimated from:

$$P = F (f_i + f_s) A S$$

where F = frequency of meteorite strikes per square foot of area,

- f_i = fraction of the striking meteorites which are iron meteorites
 and can penetrate the target,
- f_s = fraction of the striking meteorites which are stone meteorites and can penetrate the target,
 - A = area of target,
 - S = fraction of the target area which must be impacted to rupture a munition or bulk agent container (spacing factor).

The frequency of meteorite strikes for meteorites 1.0 lb or greater is $0.4 \ge 10^{-13}/\text{ft}^2$ (Ref. C-7). For small meteorites (a ton or less), stone meteorites are approximately 10 times more common than iron meteorites (Ref. C-8). However, iron meteorites are more dense and tend to have higher impact velocities, and consequently, represent a significant portion of the total meteorites that can rupture munitions. The size distribution of both iron and stone meteorites striking the earth surface was estimated from the data presented in Refs. C-7 and C-8.

The size of the meteorite required to penetrate a munition or munition and structure was computed using the equations presented in Ref. C-2. The impact velocity was computed based on the data presented





in Ref. C-8, which gives impact velocities for a series of large meteorites. These data were plotted and extrapolated to estimate the velocities for the smaller meteorites. For the smallest stone meteorites, the extrapolation yields impact velocities which were less than their terminal velocities. In these cases the terminal velocities are used. LΛ.



536-57 S S S

C.1.6. AIRCRAFT CRASH

The probabilities used in the analysis of crashes involving aircraft takeoffs and landings were obtained by modifying Table C-1, which was taken from Ref. C-9. The following modifications were made to this table:

- 1. U.S. air carrier (commercial) crash probabilities between 5 and 8 miles from the end of the runway were increased from 0.0 to 0.14 x 10^{-8} which is equal to the probability for crashes between 8 and 9 miles from the end of the runway.
- The probabilities for USN/USMC were averaged with the probabilities for USAF to obtain probabilities for military aircraft in general.
- 3. The probabilities for crashes of military aircraft at distances which are 5 to 10 miles from the runway were assumed to be the same as for U.S. commercial air carriers.
- 4. The general aviation probabilities for crashes which are 5 to 10 miles from the end of the runway are assumed to be five times greater than U.S. air carrier probabilities.
- 5. Helicopter crash probabilities were assumed to be twice the probabilities for general aviation.

Tables C-2 through C-17 summarize the input data that were used to calculate the annual probabilities of both small and large aircraft crashes at each of the eight sites. The effective areas of the crash sites are summarized in Table C-18.



| Distance From End of Runway (miles) | Probability (x 10 ⁸ of a Fatal Crash per Square Mile per Aircarft Movement ^(a) | | | | | | | |
|---|---|------------------|----------|------|--|--|--|--|
| | U.S. Air Carrier | General Aviation | USN/USMC | USA | | | | |
| 0-1 | 16.7 | 84.0 | 8.3 | 5.7 | | | | |
| 1-2 | 4.0 | 15.0 | 1.1 | 2.3 | | | | |
| 2-3 | 0.96 | 6.2 | 0.33 | 1.1 | | | | |
| 3-4 | 0.68 | 3.8 | 0.31 | 0.42 | | | | |
| 4-5 | 0.27 | 1.2 | 0.20 | 0.40 | | | | |
| 5-6 | 0 | NA | NA | NA | | | | |
| 6-7 | 0 | NA | NA | NA | | | | |
| 7-8 | 0 | NA | NA | NA | | | | |
| 9-9 | 0.14 | NA | NA | NA | | | | |
| 9-10 | 0.12 | NA | NA | NA | | | | |

TABLE C-1 AIRCRAFT CRASH PROBABILITIES NEAR AIRPORTS(a)

(a)Reference C-9.

| | APG |
|-------|----------|
| | AT |
| C-2 | AIRPLANE |
| TABLE | LARGE |
| | 4 |
| | OF |
| | ASH |

| | _ | | |
|-----------------|---------------------|---|--|
| | AL AVIATION C1 P | AL AVIATION C2 F .06-09 3.06 .46-07 0.06 | |
| | CENER CENER | GENER N GENER 5.2=+81 7 6.0=+99 8 | |
| AT APG | ٩ | P 8.6⊕-87 8.6⊕+66 | |
| C-2 Lirplane | MILITARY | MILITARY C2 1.40-09 7.00-08 | |
| TABLE (| AIRWAYS | AIRPORTS N 1.00+002 0.00+002 | |
| LASH OF A | ٩ | P - 08 8.0++08 | r e e y |
| 5 | COMMERCIAL C1 | COMMERCIAL C2 1 1.40-09 0 1.70-07 | |
| | Z | 1.20° | 7888 90 77 77 77 77 77 77 77 77 77 77 77 77 77 |
| | ROUTE WIDTH | MILES TO SITE 8 1 | r ights pr tv of a cr tv of a cr tv of a cr |
| | ROUTE NONE | AIRPORT PHILLIPS AAF WEIDE AAF | N = Number of C1= Probebili C2= Probebili P = Probebili |
| | | C-18 | |

6404046464666

5553 NY 357

5.55.55

1





| 400 | | | | | |
|-----|--------------------|-----------------------------------|---|--|--|
| | | ALL P 1.50-08 | ALL P 2.29-07 1.19-03 1.19-03 | | |
| | | 8 P 8 - 6 8 | 4 7 7 16-67 16-63 101AL | | |
| | | AVIATIO 1 -89 4. | AVIATIO 2 -09 1. -07 1. | | |
| | | ENERAL C 1 2.00 | ENERAL C 1 7.00 3 8.40 | | |
| | | 68 C 9 Z 1 . 5 Z 1 . 5 Z | 2.4.2 9.4.2 1.3 9.40 | | |
| | APG | 5 - 63 | 7 - 6 6 6 6 | | |
| | NE AT | 6 ≻. 0 | × 688 2.⊌ 4.6 | | |
| | C-3 VIRPLAI | MILITAR C1 2.00-0 | MILITAR C2 1.40-0 7.00-0 | | |
| • | TABLE A SMALL / | AIRWAYS 8.4001 | 2.44-01 2.44-01 4.04-02 | | |
| | RASH OF 1 | P. Ge-10 | P 70-68 3.00+06 | L R | |
| | 0 | IERCIAL Ci 00-10 2 | ERCIAL C2 40-09 70-07 | | |
| | | COMM 91 4. | COMM 51 1. | | |
| | | Z. | Z • • 2 0 0 7 0 | E = = | |
| | | ROUTE WIDTH 5 | WILES TO SITE 8 | 2 2 2 2 2 2 2 2 2 2 2 2 2 2 | |
| | | ROUTE - | AIRPORT PHILLIPS AAF WEIDE AAF | N = Number of C1= Probability C2= Probability P = Probability | |
| | | | C 19 | | |

TABLE C-4 Crash of a large airplane at anad

`_*&`_`&`_*&`_#a`_#a`_#a`_#a`_#a`_#a

ate afficiente afficiente a

1.1.1

| | ALL P 3.09-06 1.29-06 3.79-06 | ALL P 0.00-00 7.90-00 | |
|---------|--|-------------------------------------|--|
| | GENERAL AVIATION N C1 P 5.00+03 2.00-09 1.30-00 4.00+03 2.00-09 6.70-07 0.00+00 2.00-09 0.00+000 | GENERAL AVIATION N C2 P TOTAL | |
| | P 1.386 4.667 3.786 | ٩ | |
| | MILITARY C1 2.00-09 2.00-09 2.00-09 | MILITARY C2 | |
| AIRWAYS | N 6.0+03 2.4+03 7.3+03 | AIRPORTS N | |
| | 5.00-07 1.30-07 0.00+000 | ٩ | |
| | :0MMERCIAL C1 4.00-10 4.00-10 | .OMMERCIAL C2 | |
| | 6 + 6 9 4 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 | z | |
| | ROUTE WIDTH 8 12 4 | MILES TO SITE | |
| | ROUTE J14-52 V16 IR69 | AIRPORT NONE | |

N = Number of flights per year Cl= Probability of a crash per mile C2= Probability of a crash per sq. mile P = Probability of a crash per sq. mile per year ĨĴŶŢĨŎŶŢŎŶŢĊŎĬŢĬŎĬĨŎĬŢŎŶŢĬŎŶŢĿŎŶŢĿŎŶŢŶŎĬŢŎŶŢĬŎĬŢŎĬŢŎĬŢŎĬŢŎĬŢŎĬŢŎĬŢ

N.

•

4.4.4.4.4.4.4.4.4.4







ŊġĊġŊġĊġĊġĊġĊġĊġĊġĊġĊġĊġĊġĊġĊġĊġĊ ; ;

TABLE C-5 CRASH OF A SMALL AIRPLANE AT ANAD

| | ÄLL | 6.8++86 1.2+-85 8.8++88 | | ALL P 6.00000 | 1.20-05 |
|---------|------------------|---|---------|-----------------------------------|---------|
| AIRWAYS | GENERAL AVIATION | N C1 C1 C2) 80+86 2.30-89 8.60+86) 90+84 2.80-89 1.20-85) 80+88 2.80-89 8.80+88 | | GENERAL AVIATION N C2 P | TOTAL |
| | WILITARY | R C1 6.89+66 2.69-69 6.69+66 6.89+66 2.69-69 6.89+66 8.89+86 2.69-69 8.89+66 | IRPORTS | MILITARY N C2 P | |
| | TE COMMERCIAL | TH N CI P 8 8.80+00 4.00-18 8.80+08 12 8.80+08 4.00-19 8.00+08 1 8.00+08 4.00-19 8.00+08 | | S COMMERCIAL Te n commercial p | |
| | ROUT | ROUTE WID1 J14-62 WID1 V18 1769 | | MILES Mone to Sit | |

N = Number of flights per year Cl= Probability of a crash per mile C2= Probability of a crash per sq. mile P = Probability of a crash per sq. mile per year (\mathbf{h}, \mathbf{h})

C~21
| | | ALEXANDER DESERVED TRANSPORT |
|--------------------------------|-----|--|
| | | a (1) / 27/27 Automatical States and the Color States of the states of t |
| P = Probability of a crash per | ¢\$ | |



| | ALL P-6 1.66-6 2.66-6 | ALL P. 6 4.6 4.6 | | |
|--------------------|---|-------------------------------------|---|--|
| | CENERAL AVIATION N C1 P 2.64+03 2.04-09 6.34-07 2.64+03 2.04-09 4.24-07 8.04+00 2.04-09 8.04+06 | GENERAL AVIATION N C2 P TOTAL | | |
| AT LBAD | Р 6.36-67 4.26-07 2.06-06 | ٩ | | |
| C-6 IRPLANE | MILITARY C1 2.09-09 2.09-09 2.09-09 | WILITARY C2 | | |
| TABLE A LARGE A | AIRWAYS N 2.56+03 2.56+03 4.06+03 AIRPORTS | z | | |
| RASH OF / | P 2.56-07 1.76-07 0.0€+00 | ه. | L B D | |
| Ö | :OMMERCIAL C1 4.00-10 4.00-10 4.00-10 | COMMERCIAL C2 | | |
| | C 6 6 6 6 3 6 6 6 6 3 6 6 6 6 3 6 6 6 6 3 6 6 6 6 | z | | |
| | ROUTE WIDTH 12 12 | WILES TO SITE | 1 | |
| | LUTE 3 3 MBING RUN | RPORT NE | H = = = = = = = = = = = = = = = = = = = | |
| | 2945 | 19 22 | ZÜÜL | |
| *.* | | | | |

10 N N N

~~~~







1.0.1.0.1.0.1.0.1

# TABLE C-7 CRASH OF A SMALL AIRPLANE AT LBAD

|         | PLL             | 0.00+00    | 8.00+00                                | 1.80-07             | ALL<br>P<br>0.00000 | 1.80-07 |
|---------|-----------------|------------|----------------------------------------|---------------------|---------------------|---------|
|         | TION            | 8.8++86    | 8.84+00<br>8.84+08                     | 1.687               | TION                | TOTAL   |
|         | ERAL AVIA       | 2.009      | 2.00-09                                | 2.60-89             | ERAL AVIA<br>C2     |         |
|         | N GEN           | 0.0+00     | 80+00.0<br>80+08                       | 4.0+02              | r<br>Gen            |         |
|         | ٩               | 8.8++66    | 0.0●+00<br>0.0●+00                     | 6.0+-08             | ٩                   |         |
|         | MILITARY<br>C1  | 2.0-09     | 2.00-09                                | 2.8+-09             | MILITARY<br>C2      |         |
| AIRWAYS | Z               | 8.6+98     | 6.0++00<br>6.0++00                     | 2.40+02<br>AIRPORTS | z                   |         |
|         | ٩               | 8.6+66     | 8.8••08                                | 2.80-08             | ٥.                  |         |
|         | DMMERCIAL<br>C1 | 4.00-10    | 4.00-10                                | 4.00-10             | OMMERCIAL<br>C2     |         |
|         | z               | 8 . 8 + 88 | 0.000000000000000000000000000000000000 | 4.0.02              | U<br>Z              |         |
|         | ROUTE           | 60 (       | 4                                      | 8                   | MILES<br>TO SITE    |         |
|         | BOUTE           | 5          | J43<br>BOMBING RUN                     | ŧ                   | AIRPORT<br>None     |         |

N = Number of flights per year Clæ Probability of a crash per mile C2= Probability of a crash per sq. mile P = Probability of a crash per sq. mile per year

C-23

Ash. Doogoo sa siddaada ahaa daadaa madaa madaada . LUCIUM RICESS







| ALL<br>P<br>1.56-05<br>1.56-06<br>9.06-07<br>7.20-07                                                                                   | ALL<br>P<br>0.00+00<br>4.60-06                                 |
|----------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------|
| GENERAL AVIATION<br>N CI P<br>2.56+03 2.06-09 6.36-07<br>2.56+03 2.06-09 5.06-07<br>2.06+03 2.06-09 5.06-07<br>2.06+03 2.06-09 4.06-07 | GENERAL AVIATION<br>N C2 P<br>0.0⊕+08 7.0⊕-09 0.0⊕+00<br>T0TAL |
| 6.30-67<br>6.30-67<br>3.60-67<br>2.40-67                                                                                               | 9<br>9<br>9                                                    |
| MILITARY<br>C1<br>2.00-09<br>2.00-09<br>2.00-09<br>2.00-09                                                                             | MILITARY<br>C2<br>1.40-89                                      |
| AIRWAYS<br>N<br>2.56+03<br>2.56+03<br>1.2+03<br>1.2+03                                                                                 | AIRPORTS<br>N<br>0.00+00                                       |
| COMMERCIAL P<br>C COMMERCIAL P<br>C C C C C C<br>C C C C C C C<br>C C C C C                                                            | COMMERCIAL<br>N C2 P<br>8.8+88 1.4+-89 8.8+80                  |
| ROUTE<br>WIDTH<br>8<br>8<br>8<br>10                                                                                                    | MILES<br>TO SITE<br>B                                          |
| ROUTE<br>J73<br>J80<br>V171<br>V434                                                                                                    | AIRPORT<br>Rowe                                                |

N = Number of flights per year Cl= Probability of a crash per mile C2= Probability of a crash per sq. mile P = Probability of a crash per sq. mile per year



₹.



|         | ALL<br>P<br>0.00000<br>0.00000<br>0.00000<br>0.00000<br>1.00000<br>7.00000                                                              | ALL<br>P<br>7.89-96<br>2.39-85                                 |
|---------|-----------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------|
|         | GENERAL AVIATION<br>N C1 P<br>0.00+00 2.00-09 0.00+00<br>0.00+00 2.00-09 0.00+00<br>3.56+04 2.00-09 0.700-00<br>3.56+04 2.00-09 7.00-06 | GENERAL AVIATION<br>N C2 P<br>1.80+83 7.80-89 7.80-86<br>T0TAL |
| AIRWAYS | MILITARY<br>N C1<br>C9.00+000 2.00-009 0.00+00<br>0.00+000 2.00-099 0.00+00<br>0.00+000 2.00-099 0.00+00<br>0.00+000 2.00-099 0.00+000  | AIRPORTS<br>Military<br>N C2<br>B.00+00 1.40-09 B.00+00        |
|         | COMMERCIAL P<br>N C1 P<br>0.00+000 4.00-10 0.00+00<br>0.00+000 4.00-10 0.00+00<br>0.00+000 4.00-10 0.00+00<br>0.00+00 4.00-10 0.00+00   | COMMERCIAL COMMERCIAL R<br>N C2 P<br>8.0+00 1.4-99 8.0+00      |
|         | ROUTE<br>ROUTE WIDTH<br>173 WIDTH<br>180 8<br>2180 8<br>2171 8<br>2171 8<br>210                                                         | AIRPORT TO SITE<br>Rowe                                        |

N = Number of flights per year Cl= Probability of a crash per mile C2= Probability of a crash per sq. mile P = Probability of a crash per sq. mile per year

1999 T 1999 T 1999 T 1999 SAMAN TAMANA TANANG T

C-25

155555 2555777 KARAT BERRENA TRACERAT ANALAT BERRENA

Š.

NANAYANAYANA MANAYA

TABLE C-10 CRASH OF A LARGE AIRPLANE AT PBA

Þ

| ALL              | 1.50-06                 | ALL<br>P<br>0.000000       |
|------------------|-------------------------|----------------------------|
| GENERAL AVIATION | 2.5e+83 2.0e-89 6.3e-87 | GENERAL AVIATION<br>N C2 P |
| ٩                | 6.307                   | ٩                          |
| MILITARY         | 2.809                   | WILITARY<br>C2             |
| AIRWAYS          | 2.5.403                 | AIRPORTS<br>N              |
| ٩                | 2.5-07                  | ٩                          |
| COMMERCIAL<br>C1 | 4.0-10                  | COMMERCIAL<br>C2           |
| Z                | 5.0+03                  | Z                          |
| ROUTE            | <b>0</b> 0              | WILES<br>TO SITE           |
| ROUTE            | J42                     | A I RPORT<br>NONE          |

1.50-06

TOTAL

N = Number of flights per year Cl= Probability of a crash per mife C2= Probability of a crash per sq. mile P = Probability of a crash per sq. mile per year



Presses a

Contraction of the

## TABLE C-11 CRASH OF A SMALL AIRPLANE AT PBA

|         | ALL<br>P<br>6.00+80<br>1.80-07                                                   |          | ALL<br>P<br>0.00+00        | 1.8+-67 |
|---------|----------------------------------------------------------------------------------|----------|----------------------------|---------|
|         | GENERAL AVIATION<br>N C1 P<br>0.00+00 2.00-09 0.00+00<br>4.00+02 2.00-09 1.00-07 |          | GENERAL AVIATION<br>N C2 P | TOTAL   |
|         | P<br>6.6e+00<br>6.6e-08                                                          |          | ٩                          |         |
|         | MILITARY<br>C1<br>2.00-09<br>2.00-09                                             |          | MILITARY<br>C2             |         |
| AIRWAYS | N<br>8.00+000<br>2.40+02                                                         | AIRPORTS | Z                          |         |
|         | P<br>8.8●+88<br>2.8●-68                                                          |          | ٩                          |         |
|         | 0MMERCIAL<br>C1<br>4.00-10<br>4.00-10                                            |          | DIMMERCIAL<br>C2           |         |
|         | 0<br>8.6+96<br>4.6+62                                                            |          | z                          |         |
|         | ROUTE<br>WIDTH<br>8<br>8                                                         |          | WILES<br>TO SITE           |         |
|         | ROUTE<br>J42<br>-                                                                |          | A I R P OR T<br>NONE       |         |

N = Number of flights per year Cl= Probability of a crash per mile C2= Probability of a crash per sq. mile P = Probability of a crash per sq. mile per year C-27



• 23:22:23

sq. mile per year Ē a crash per Ë. N = Number of fligh Cl= Probability of d C2= Probability of d P = Probability of d

> . N. -

6.9<del>-0</del>5 TOTAL

|                 | ROUTE          | J       | COMMERCIAL |         |          | MILITARY |                  | GENE    | RAL AVIAT | NOIL    | ALL             |
|-----------------|----------------|---------|------------|---------|----------|----------|------------------|---------|-----------|---------|-----------------|
| ROUTE           | WIDTH          | z j     | ចុ         | مہ ز    | <b>z</b> | ម        | د<br>ا<br>ا      | z,      | ចុ        | ٩       | ا<br>م          |
| J28             | 90             | 1.00+03 | 4.00-10    | 5.0e-08 | 5.04+02  | 2.00-09  | 1.30-07          | 5.0e+82 | 2.809     | 1.30-07 | 3.867           |
| 117             | 10             | 1.0.+03 | 4.0-10     | 4.0-08  | 5.0++02  | 2.0-09   | 1.007            | 5.0++02 | 2.809     | 1.007   | 2.40-07         |
| V10-244         | 80             | 4.0.402 | 4.0-10     | 2.6-08  | 2.40+82  | 2.009    | 6. <i>8</i> e-08 | 4.0+02  | 2.009     | 1.807   | 1.807           |
| V19-83          | 80             | 4.0.402 | 4.0-10     | 2.0e-08 | 2.40+82  | 2.009    | 6.0+-08          | 4.0     | 2.809     | 1.8-07  | 1,807           |
| V81             | 10             | 4.0.402 | 4.0-10     | 1.6e-Ø8 | 2.40+02  | 2.609    | 4.808            | 4.00+02 | 2.009     | 8.008   | 1.4-07          |
| V389            | 18             | 4.8.482 | 4.8e-18    | 1.608   | 2.4++02  | 2.809    | 4.808            | 4.0.00  | 2.009     | 8.008   | 1.40-07         |
|                 |                |         |            |         |          |          |                  |         |           |         |                 |
|                 |                |         |            |         | AIRPORTS |          |                  |         |           |         |                 |
| ATRPORT         | WILES<br>NILES | 2       | COMMERCIAL | ٩       | 2        | MILITARY | ٩                | GENEI   | RAL AVIAT | NOI     | ٩٢              |
| PUEBLO MEMORIAL | 8              | 9.1++63 | 1.40-09    | 1.30-05 | 9.1.4.03 | 1.469    | 1.3-05           | 4.6+03  | 7.889     | 3.20-05 | 5.8 <b>-</b> 05 |

TABLE C-12 CRASH OF A LARGE AIRPLANE AT PUDA

AIRWAYS



 $\langle \hat{\mathbf{x}} \rangle$ 



| ALL<br>P<br>6.00-00<br>6.00-00<br>1.70-06<br>1.70-06<br>1.40-06<br>1.40-06                                                                                                                                                    | ALL<br>P<br>9.86-95<br>1.86-64                                 |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------|
| GENERAL AVIATION<br>N C1 P<br>0.00+00 2.00-09 0.00+00<br>7.00+03 2.00-09 1.7-00<br>7.00+03 2.00-09 1.7-00<br>7.00+03 2.00-09 1.4-00<br>7.00+03 2.00-09 1.4-00                                                                 | GENERAL AVIATION<br>N C2 P<br>1.40+84 7.80-89 9.80-85<br>T0TAL |
| MILITARY<br>N CI<br>B.Be+BB 2.Be-B9 8.Be+BB<br>B.Be+BB 2.Be-B9 8.Be+BB<br>G.Be+BB 2.Be-B9 8.Be+BB<br>G.Be+BB 2.Be-B9 8.Be+BB<br>B.Be+BB 2.Be-B9 8.Be+BB<br>B.Be+BB 2.Be-B9 8.Be+BB                                            | AIRPORTS<br>Military<br>N C2<br>0.00+00 1.40-09 0.00+00        |
| CONNERCIAL N CONNERCIAL N CI E P<br>6.86+86 4.86-16 6.86+86<br>6.86+86 4.86-16 6.86+86<br>6.86+86 4.86-16 6.86+86<br>6.84+86 4.86-16 6.86+86<br>6.84+86 4.86-16 6.86+86<br>6.84+86 4.86-16 6.84+86<br>6.84+86 4.86-16 6.84+86 | COMMERCIAL P<br>N C2 P<br>0.00+00 1.40-09 0.00+00              |
| ROUTE ROUTE<br>J28 WIDTH<br>J28 J17 10<br>J17 10<br>V18-244 8<br>V19-83 10<br>V81 10<br>V389 10                                                                                                                               | MILES<br>AIRPORT TO SITE<br>PUEBLO MEMORIAL 8                  |

AIRWAYS

N = Number of flights per year Cl= Probability of a crash per mile C2= Probability of a crash per sq. mile P = Probability of a crash per sq. mile per year 1

C-29

| inio<br>por year                                                                                                                              |  |
|-----------------------------------------------------------------------------------------------------------------------------------------------|--|
| N = Number of flights per year<br>Cle Probability of a crash per mi<br>C2= Probability of a crash per ac<br>P = Probability of a crash per ac |  |



للمنددة



|                    | L67                                                                |                                   |                                                                         |  |
|--------------------|--------------------------------------------------------------------|-----------------------------------|-------------------------------------------------------------------------|--|
|                    | 3.60<br>ALI<br>3.60                                                |                                   |                                                                         |  |
|                    | SENERAL AVIATION<br>5<br>82 2.64-69 2.64-67                        | GENERAL AVIATION<br>C2 P<br>TOTAL |                                                                         |  |
|                    | 2 <b>8</b> .                                                       | z                                 |                                                                         |  |
| AT TEAD            | P<br>1.20-07                                                       | <b>e</b>                          |                                                                         |  |
| C-14<br>IRPLANE    | MILITARY<br>C1<br>2.84-09                                          | MILITARY<br>C2                    |                                                                         |  |
| TABLE (<br>LARGE A | AIRWAYS<br>N<br>4.80002                                            | AIRPORTS<br>N                     |                                                                         |  |
| ASH OF A           | - 5<br>- 5<br>- 5<br>- 5<br>- 5<br>- 5<br>- 5<br>- 5<br>- 5<br>- 5 | · •                               | L<br>B<br>X                                                             |  |
| G                  | COMMERCIAL<br>N C1<br>3.00+02 4.00-10                              | COMMERCIAL<br>N C2                | Year Aile<br>Per Aile<br>Per aq. Mile Per                               |  |
|                    | ROUTE<br>WIDTH<br>8 8                                              | WILES<br>TO SITE                  | of flights per<br>lity of a crash<br>lity of a crash<br>lity of a crash |  |
|                    | ROUTE<br>V267                                                      | A I RP OR T<br>NONE               | N = Number<br>Cle Probabi<br>C2= Probabi<br>P = Probabi                 |  |
|                    | C-30                                                               |                                   |                                                                         |  |

| , | - | 3 | 0 |  |
|---|---|---|---|--|
|---|---|---|---|--|

NAN TI KING ANA AN

TABLE C-15 CRASH OF A SMALL AIRPLANE AT TEAD

| ALL<br>P<br>3.6e-06                                                                              | ALL<br>8.00-00<br>3.50-06           |
|--------------------------------------------------------------------------------------------------|-------------------------------------|
| GENERAL AVIATION<br>N C1 P<br>1.40+04 2.00-09 3.50-06                                            | GENERAL AVIATION<br>N C2 P<br>TOTAL |
| 9<br>- 00<br>- 00                                                                                | ٩                                   |
| MILITARY<br>C1<br>2.00-09                                                                        | MILITARY<br>C2                      |
| AIRWAYS<br>N<br>8.00+00                                                                          | AIRPORTS<br>N                       |
| P<br>₽<br>0.0⊕                                                                                   | ٩                                   |
| :OMMERCIAL<br>C1<br>4.8=-18                                                                      | C2 COMMERCIAL C2                    |
| U<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩<br>₩ | z                                   |
| ROUTE<br>WIDTH<br>8                                                                              | WILES<br>TO SITE                    |
| ROUTE<br>V267                                                                                    | A I RPORT<br>NONE                   |

N = Number of flights per year Cl= Probability of a crash per mile C2= Probability of a crash per sq. mile P = Probability of a crash per sq. mile per year 22222

C-31

|                                                                                                                | *****     |                  |             |
|----------------------------------------------------------------------------------------------------------------|-----------|------------------|-------------|
| 8                                                                                                              |           |                  |             |
|                                                                                                                |           |                  |             |
|                                                                                                                |           |                  |             |
|                                                                                                                |           |                  |             |
| State of the second |           |                  |             |
|                                                                                                                |           |                  |             |
|                                                                                                                |           |                  |             |
|                                                                                                                |           | 4 <b>~ 8864</b>  | ₹ 8. 9      |
|                                                                                                                |           | <b>*</b> * * * * | • •         |
| 5                                                                                                              |           |                  |             |
|                                                                                                                |           | 866 <b>8</b>     | <u> </u>    |
|                                                                                                                |           |                  | z• ₽        |
|                                                                                                                |           |                  | 110         |
|                                                                                                                |           |                  | VIN         |
|                                                                                                                |           |                  | C, ×        |
|                                                                                                                |           | <b>A</b> 4996    | ERAL        |
| ×.                                                                                                             |           | W mmmg           | W.          |
|                                                                                                                |           |                  | z           |
|                                                                                                                |           |                  |             |
|                                                                                                                |           |                  | `           |
|                                                                                                                | <         |                  | •           |
| 88                                                                                                             | Q         | 6                | _           |
|                                                                                                                | ວ<br>ພ    |                  | 0.          |
| <b>孩</b>                                                                                                       | EA.       | ** \$\$ * \$     |             |
|                                                                                                                | NE        | ≻_6666           | 2 KK        |
|                                                                                                                | 6<br>PLA  |                  | E           |
|                                                                                                                |           | <b>H</b>         | NIL         |
|                                                                                                                | ິ<<br>ພິພ |                  | TS          |
| 8                                                                                                              | BLI       |                  | BO Z        |
| K                                                                                                              | ₹ Y       | 71 VI            | VIR         |
| R.                                                                                                             | ~         |                  |             |
|                                                                                                                | 04        |                  |             |
|                                                                                                                | H         |                  | •           |
| R                                                                                                              | RAS       |                  |             |
|                                                                                                                | ប         | 2 8888           | . र         |
| S.                                                                                                             |           |                  |             |
| κ.                                                                                                             |           |                  |             |
|                                                                                                                |           | Ö                | CON         |
|                                                                                                                |           |                  | _           |
|                                                                                                                |           |                  | 2           |
|                                                                                                                |           | <b>***</b>       |             |
|                                                                                                                |           |                  | ш           |
|                                                                                                                |           | 29               | LES<br>SITI |
|                                                                                                                |           | M.               | He He       |
|                                                                                                                |           |                  | F           |
| K                                                                                                              |           |                  |             |
| 8                                                                                                              |           |                  | _           |
|                                                                                                                |           | ш 🛱              | DRT         |
| 8                                                                                                              |           |                  | L KP        |
| R.                                                                                                             |           | 29222            | ZX          |
| 8                                                                                                              |           |                  |             |
| 88                                                                                                             |           | (                | 2-32        |
| þ.                                                                                                             |           |                  |             |
| 5                                                                                                              |           |                  |             |
| 8                                                                                                              |           | <b>.</b>         |             |
|                                                                                                                |           |                  |             |

its per year a crash per mile a crash per sq. mile a crash per sq. mile per year N = Number of flights C1= Probability of a C2= Probability of a P = Probability of a



ቔዾኯዿቘጟኯዹጟዀቔጞኯዸቘጟኯዸጟዀቔኯኯዸዀቔቜዀቔጞዀቔጞዹዄዀጜቔኯኇዀጜኇዀጜቜዀጜቜዀጜኇቚጜኇዀጜኇዀጜኇ ዸ







TABLE C-17 CRASH OF A SMALL AIRPLANE AT UMDA

| ALL<br>P<br>6.84488<br>6.84488<br>1.24-95<br>0.84488                                                                                   | ALL.<br>P<br>0.00+00<br>1.20-05     |  |
|----------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------|--|
| GENERAL AVIATION<br>N C1 P<br>8.00+00 2.00-09 8.00+00<br>9.00+04 2.00-09 8.00+00<br>7.00+04 2.00-09 8.00+00<br>8.00+00 2.00-09 8.00+00 | GENERAL AVIATION<br>N C2 P<br>TOTAL |  |
| 0000<br>0000<br>0000<br>00000<br>00000<br>000000                                                                                       | <b>e</b> .                          |  |
| MILITARY<br>C1<br>2. <b>69-99</b><br>2.69-69<br>2.69-69<br>2.69-69                                                                     | MILITARY<br>C2                      |  |
| AIRWAYS<br>B . B . B<br>B . B . B<br>B . B . B . B<br>B . B .                                                                          | AIRPORTS<br>N                       |  |
| P. 80.00<br>9.80.00<br>9.90.00<br>9.90.00<br>9.90.00                                                                                   | ۹.                                  |  |
| OMMERCIAL<br>C1<br>4.00-10<br>4.00-10<br>4.00-10<br>4.00-10                                                                            | DWMERCIAL<br>C2                     |  |
| 2 0 0 0 0<br>9 0 0 0<br>2 0 0 0 0<br>0 0 0 0<br>0 0 0 0                                                                                | J                                   |  |
| ROUTE<br>WIDTH<br>12<br>12<br>6                                                                                                        | WILES<br>TO SITE                    |  |
| ROUTE<br>J54<br>J20<br>V4<br>VR1354                                                                                                    | AIRPORT<br>NONE                     |  |

N = Number of flights per year Cl= Probability of a crash per mile C2= Probability of a crash per aq. mile P = Probability of a crash per aq. mile per year

فككفك

ESSI® EESSI PRESS

1.5

 $(\cdot, \cdot)$ 

C-33

1026-023 118. 414 424 152-444 4

C NY



TABLE C-18 EFFECTIVE AREAS OF CRASH SITES(a)

(a)Units of area is square miles.

## Ś, Ś

- C.1.7. REFERENCES
- C-1. "Supporting Calculations for the Risk Analysis of the Disposal of Chemical Munitions at National or Regional Sites," GA Report EPS-GA/DP-0092, August 1987.
- C-2. "Design of Structures for Missile Impact," Bechtel Power Corporation, BC-TO-9A, September 1974.
- C-3. "Standard for Estimating Tornado and Extreme Wind Characteristics at Nuclear Power Sites," ANSI/ANS-2.3-1983, 1983.
- C-4. Coats, D. W., "National Phenomena Hazards Modeling Project: Extreme Wind/Tornado Hazard Models for Department of Energy Sites," UCRL-53526, Lawrence Livermore National Laboratory, February 1984.
- C-5. "Tentative Provisions for the Development of the Seismic Regulations for Buildings," Applied Technology Council, ATC 3-06, NBS SP-510, NSF 78-8, June 1978.
- C-6. "Structural Analysis and Design of Nuclear Plant Facilities," American Society of Civil Engineers, Manuals and Reports on Engineering Practice No. 58, 1980.
- C-7. Solomon, K. A., K. C. Erdmann, and D. Okrent, "Estimate of the Hazards to a Nuclear Reactor from the Random Impact of Meteorites," Nuclear Technology, Vol. 25, January 1975.
- C-8. Blake, V. E., "A Prediction of the Hazards from the Random Impact of Meteorites on the Earth's Surface," Sandia Laboratories SC-RR-86-838, December 1968.
- C-9. "USNRC Standard Review Plan," Section 3.5.1.6, "Aircraft Hazards," U.S. Nuclear Regulatory Commission, NUREG-0800, July 1981.



APPENDIX D SITE INFORMATION 

THE REPORT OF THE PERSON OF THE PERSON OF THE

### D.1. SITE INFORMATION

This appendix discusses the location and characteristics of the eight CONUS sites where chemical munitions are stored and provides a brief description of the storage areas. Figure D-1 shows the general location of the eight sites. The site characteristics discussed included recorded earthquake activity and aircraft patterns in the vicinity.

### D.1.1. ABERDEEN PROVING GROUND

As shown in Figs. D-2 and D-3, the Aberdeen Proving Ground (APG) is located in Harford County, Maryland near the head of the Chesapeake Bay.

APG is a Test and Evaluation Command (TECOM) installation within U.S. Army Materiel Command (AMC). The main activities/mission of APG include testing and evaluating vehicles, munitions, and other combat hardware. A major tenant activity, the Chemical Research, Development, and Engineering Center (CRDEC), is located at APG.

APG is comprised of two general areas, the Aberdeen Area and Edgewood Area. The Edgewood Area is situated adjacent to the town of Edgewood in the southwestern part of Harford County. There have occurred in the vicinity of the APG site 48 recorded earthquakes of Modified Mercalli Intensity (MMI) levels from I to VII, as summarized in Table D-1.

The chemical storage area at APG is located in the northeast corner of the Edgewood Area. The Chemical Agent Storage Yard (CASY) is an open area encompassing approximately 5 acres and is situated along the Bush



Location of chemical agents and munitions in the U.S. Fig. D-1. 

| AD-A1 | 93 355<br>SSIFIE | CHE<br>The<br>CA<br>D SAF | MICAL<br>DISP<br>A W I<br>PEO-CDI | STOCK<br>DSAL O<br>BARSEL<br>E-IS-8 | PILE D<br>F Chem<br>L et r<br>7006 d | ) I SPOSI<br>I (U)<br>IL. AU(<br>)AAA15- | AL PRO<br>Gr Te<br>3 87 Gi<br>85-D-1 | GRAM R<br>Chnolo<br>R-C-19<br>9022 | ISK AN<br>GIES 1<br>563 | IALYSI:<br>NC SAI<br>F/G : | 5 OF<br>N DIEG<br>L5/6.3 | 0 10/<br>NL | 13 |
|-------|------------------|---------------------------|-----------------------------------|-------------------------------------|--------------------------------------|------------------------------------------|--------------------------------------|------------------------------------|-------------------------|----------------------------|--------------------------|-------------|----|
|       |                  |                           |                                   |                                     |                                      |                                          |                                      |                                    |                         |                            |                          |             |    |
|       |                  |                           |                                   |                                     |                                      |                                          |                                      |                                    |                         |                            |                          |             |    |
|       |                  |                           |                                   |                                     |                                      |                                          |                                      |                                    |                         |                            |                          |             |    |
|       |                  |                           |                                   |                                     |                                      |                                          |                                      |                                    |                         | و<br>                      | u 0                      |             |    |
|       |                  |                           |                                   | 2                                   |                                      | н н<br>-                                 |                                      |                                    |                         | الو در                     |                          |             |    |
| ä.    |                  |                           |                                   |                                     | *                                    | ~**<br>*                                 | 1                                    | 1997<br>- 199                      | ليست                    |                            | 1                        |             |    |
|       |                  |                           |                                   |                                     |                                      |                                          |                                      |                                    |                         |                            |                          | 10 A        |    |
|       |                  |                           |                                   |                                     |                                      |                                          |                                      |                                    |                         |                            |                          |             |    |



2.5

2.0



TABLE D-1 EARTHQUAKES IN THE VICINITY OF THE APG SITE (Ordered By Distance From Site)

ł

ALLESS ALLESS ALLESS ALLESS ALLESS ALLESS ALLESS ALLESS ALLESS ALLESS ALLESS ALLESS ALLESS ALLESS ALLESS ALLES

| Year  | Month | Day | Location     | MMI                                        | Distance<br>from Site<br>(km) |
|-------|-------|-----|--------------|--------------------------------------------|-------------------------------|
| 1883  | 3     | 11  | 39.5N. 76.4W | v                                          | 14                            |
| 1883  | 3     | 12  | 39.5N. 76.4W | V                                          | 14                            |
| 1883  | 3     | 12  | 39.5N. 76.4W | ĪTT                                        | 14                            |
| 1883  | 3     | 12  | 39.5N. 76.4W | v                                          | 14                            |
| 1030  | 6     | 22  | 39.5N. 76.6W | ŤTT                                        | 28                            |
| 1030  | 11    | 18  | 39.5N, 76.6W | TV                                         | 28                            |
| 1030  | 11    | 26  | 39.5N, 76.6W | v                                          | 28                            |
| 1930  | 11    | 01  | 39.1N. 76.5W | TV                                         | 38                            |
| 1030  | 11    | 01  | 39.1N. 76.5W | ттт                                        | 38                            |
| 1906  | 10    | 13  | 39.2N. 76.7W | TTT                                        | 41                            |
| 1910  | 04    | 24  | 39.2N, 76.7W | TTT                                        | 41                            |
| 1758  | 04    | 25  | 38.9N. 76.5W | v                                          | 58                            |
| 1876  | 01    | 30  | 38.9N. 76.5W | •                                          | 58                            |
| 1070  | 07    | 16  | 30 ON 76 2W  | v                                          | 58                            |
| 109/0 | 04    | 10  | 30 ON 76 3W  | v                                          | 58                            |
| 109/  | 04    | 23  | 30 ON 76 3W  | v                                          | 58                            |
| 1010  | 01    | 23  | 30 6N 77 AU  | тт                                         | 56                            |
| 1020  | 02    | 24  | 39 ON 76 7W  | **                                         | 65                            |
| 1020  | 10    | 24  | 30.9N, 70.7W | VΤ                                         | 66                            |
| 1005  | 02    | 00  | 10 ON 76 2W  | ▼⊥<br>TV                                   | 67                            |
| 1007  | 0.5   | 03  | 40.0N, 76.3W | 1 V<br>T T                                 | 67                            |
| 1071  | 04    | 14  | 40.0M, 70.3W | 11<br>T17                                  | 60                            |
| 1071  | 12    | 14  | 39./M, /3.0W | 1 V<br>TV                                  | 69                            |
| 1072  | 01    | 27  | 20 7N 75 6U  | 1 V<br>T V                                 | 60                            |
| 1972  | 01    | 02  | 2017N 75 6W  | 1 V<br>T 17                                | 60                            |
| 1072  | 01    | 03  | 20 7N 75 6U  | 1 V<br>T 17                                | 60                            |
| 19/2  | 01    | 07  | 39./N, /3.0W | 1 V<br>TV                                  | 69                            |
| 19/2  | 01    | 22  | 39./N, /3.0W | 1 V<br>TV                                  | 69                            |
| 19/2  | 01    | 23  | 39./N, /3.0W |                                            | 69                            |
| 19/2  | 01    | 23  | 39./N, /3.0W | 11                                         | 69                            |
| 19/2  | 02    | 11  | 39./N, /3.0W | v                                          | 69                            |
| 19/2  | 02    | 11  | 39./N, /3.0W | <b>T 1 1</b>                               | 69                            |
| 19/2  | 00    | 14  | 39./N, /3.0W | τv                                         | 69                            |
| 19/2  | 06    | 14  | 39./N, /3.0W | <b>T17</b>                                 | 69                            |
| 19/4  | 04    | 20  | 39./N, /3.0W | 11                                         | 71                            |
| 1007  | 03    | 00  | 40.0N, 76.0W | v                                          | 71                            |
| 1007  | 10    | 10  | 40.0N, 70.0W | Ŧ17                                        | 71                            |
| 10/1  | 10    | 10  | 39.0N, 75.5W | 11                                         | 72                            |
| 10/3  | 03    | 20  | 39.2N, /3.3W | V<br>                                      | 72                            |
| 1902  | 03    | 10  | 39.0M, 77.1W | 111                                        | 72                            |
| 1902  | 03    | 11  | 39.0N, //.1W | +<br>+ + + + + + + + + + + + + + + + + + + | 72                            |
| 1003  | 11    | 17  | 37.0N, //.1W | 1<br>77                                    | 12                            |
| 1003  | 12    | 10  | 37.0N, /3.0W | v                                          | 13                            |
| 1971  | 10    | 12  | 30 7N 75 FT  | <b>WT T</b>                                | 15                            |
| 10/1  | 10    | 10  | 20 6N 77 0   | V 1 1<br>7 7 7                             | /0                            |
| 1002  | 03    | 10  | 30 6N 77 0   | 111<br>777                                 | 80                            |
| 1002  | 03    | 01  | 30 6N 77 01  | 111                                        | 0U                            |
| 1003  | 01    | 01  | 37.0N, //.2W | <u>+</u> ++                                | 80                            |
| 1302  | 01    | 01  | 39.0N, //.2W | II                                         | 80                            |

Data provided by the National Geophysical Data Center, NOAA.



River. The storage yard consists of a central aisleway of finished concrete and the ton containers are secured over a gravel surface. There are two buildings in the CASY that are used to store equipment. Only mustard-filled ton containers are stored at APG and they are stored outdoors in accordance with AMC regulations.

AVAILATIN DESCRIPTION SECTORS AND ALL TRADUCT SECTORS SECTORS

The airspace above the Edgewood area of APG is continuously restricted (Restriction No. R-4001A). Permission to fly at altitudes above 10,000 ft from midnight to 7:00 AM may be requested 24 hr in advance. The Weide Army Air Field (AAF) is located within a mile of the storage area. It has a 4600-ft runway which is used by a general aviation flying club and an Air National Guard helicopter unit located at Weide AAF. The Army estimates that there are approximately 2600 general aviation operations (takeoffs/landings), 7200 helicopter operations, and 800 small fixed-wing military operations per year at Weide. There are no large aircraft operations.

Phillips AAF is located approximately 8 miles to the northeast. It has three runways. The longest is 8000 ft. The Army indicates that the edges of the approach and holding patterns for Phillips are more than 2 miles north of the storage area. Therefore, they are not considered a threat to the storage area per the guideline of Ref. D-3.

There are three other airports located in the area. Baltimore Airpark is approximately 8 miles to the west and has one 2200-ft runway. Martin State Airport is located 8 miles to the southeast. It has three runways. The longest is 7000 ft. The largest airport in the area is Baltimore Washington International Airport which is 26 miles southwest of Aberdeen. Its longest runway is 9500 ft. There are two low altitude federal airways (V378 and V499) that pass approximately 8 miles from the storage area. The closest high altitude jet routes (J42-8 and J40) are approximately 14 miles from the storage area. These airports and airways are not expected to present a significant threat to the storage



いたちょう かんたい かんしょう かんしょう かんしょう かんかん かんかん かんしょう ひょう かんしょう かんしょう たいしょう いんかん アングングング かんしょう 日本学校 かんかん かんかん



area because of the distances involved and because the storage area is protected by the restricted airspace. (22.22.23)

8.5.2.2.8

8 Pak Vak 6.8 4ak 1



### D.1.2. ANNISTON ARMY DEPOT

As shown in Figs. D-4 and D-5, the Anniston Army Depot (ANAD) is located within Calhoun County in northeast Alabama adjacent to Fort McClellan, another active U.S. Army installation. ANAD is a major supply, stock distribution, and storage depot for general and strategic material, equipment, and supplies, including ammunition. Its functions also include maintenance and disposal activities associated with ammunition supply and storage, such as ammunition preservation, demilitarization, surveillance and training.

The chemical storage area at ANAD is located along the northeastern edge of the installation. The chemical storage area is divided into two adjacent areas, G-block and C-block. The ANAD chemical munition stockpile consists of all munition types except bombs and spray tanks. Munitions are stored in 40-ft, 60-ft, and 80-ft igloos. All 40-ft and 60-ft igloos are equipped with a single door, while all 80-ft igloos are equipped with a double door. The igloos are well maintained with no evidence of chronic structural problems. All igloos were re-waterproofed in 1984. The re-waterproofing involved removing the earthen covering over the igloo and sealing the concrete surface with tar. The earthen cover was then replaced to specifications.

The stockpile of chemical munitions stored at ANAD includes 105-mm cartridges, 4.2-in. mortars, 155-mm and 8-in. projectiles, 115-mm rockets, land mines, and ton containers. Documentation indicates that all of the 105-mm projectiles are stored in the cartridge configuration, packaged two cartridges per box. All munitions are stored in their standard configurations in accordance with AMC regulations.

As shown in Table D-2, five earthquakes of MMI levels V to VII have occurred in the vicinity of the ANAD site in this century.



Fig. D-4. Alabama state map showing the location of ANAD



404<u>54089</u>

ORNL-DWG 85-131058



a a start

Fig. D-5. County map showing the location of ANAD



. . . .

| Year | Month | Day | Location                       | Epicentral Intensit;<br>(MMI) |  |  |
|------|-------|-----|--------------------------------|-------------------------------|--|--|
| 1916 | 10    | 18  | Irondale, AL<br>33.5N, 86.2W   | VII                           |  |  |
| 1927 | 6     | 16  | Scottsboro, AL<br>34.7N, 86.0W | V                             |  |  |
| 1931 | 5     | 5   | Cullman, AL<br>33.7N, 86.6W    | V to VI                       |  |  |
| 1939 | 5     | 4   | Anniston, AL<br>33.7N, 85.8W   | v                             |  |  |
| 1975 | 8     | 28  | Northern, AL<br>33.8N, 86.6W   | VI                            |  |  |

## TABLE D-2 EARTHQUAKES IN THE VICINITY OF THE ANAD SITE(a) (Chronological Listing)

(a)Earthquakes within a 50- to 60-mile radius of the Anniston site, abstracted from Table 2.5-2, Clinch River Breeder Reactor Plant Preliminary Safety Analysis Report. Source: Ref. D-1.

The airspace above the chemical munition storage area at the ANAD is unrestricted. The airspace just north and northeast of the chemical storage area is restricted continuously to 24,000 ft (Restriction number R-2102). The area just west of the chemical munition storage area is restricted up to a 5000-ft level from 7:00 AM to 6:00 PM Monday through Friday (Restriction number R-2101).

BUGGERS BEESSEN

The closest major airfields are Anniston and Talladega, both of which are approximately 8 miles from the chemical munition storage area. Anniston has a 7000-ft runway and can accept aircraft as large as a Lockheed C-141. Air traffic flying in and out of Anniston must stay to the south of the depot (Ref. D-1). Talladega has a 6000-ft runway. It has handled Lockheed C-130s but cannot accept C-141s. Air traffic coming out of Talladega must stay west of the depot (Ref. D-1). Consequently, the edge of the flight path in and out of Anniston and out of Talladega is at least 2 miles from the storage area.

To the east and north of the city of Anniston, there are two small airports and a heliport, the closest of which is 8 miles from the storage area. Air traffic from these airports is not a significant threat to the storage area since there is 3 miles of restricted airspace between these airports and the storage area.

There is one low altitude federal airway (V18) which passes 6 miles south of the storage area and one high altitude jet route (J14-52) which passes directly above the storage area. The high altitude jet route is the preferred jet route for ai traffic between Atlanta and Denver (Ref. D-2). Military training route IR69 passes over the storage area and then returns three miles south of the storage area.

A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A S

### D.1.3. LEXINGTON-BLUE GRASS ARMY DEPOT

As shown in Figs. D-6 and D-7, the Lexington-Blue Grass Army Depot (LBAD) is located in Madison County, south of Richmond, Kentucky. The primary mission of the depot is to operate a general supply and ammunition depot activity providing for the receipt, storage, issue, maintenance, demilitarization, and disposal of assigned commodities.

The chemical munition storage area at LBAD is located in the north central half of the Blue Grass facility. The chemical munition stockpile at LBAD consists of 8-in. projectiles, 155-mm projectiles, and M55 rockets. These munitions are stored in 89-ft oval-arch igloos. Seventy-five percent of the igloos were waterproofed in 1982. The procedure involved removing the earth covering the igloo to apply a layer of tar, and then replacing the earthen cover.

Table D-3 summarizes earthquake activity in the vicinity of the LBAD site.

LBAD airspace is not restricted. There are three small airfields in the vicinity of the depot: Madison County Airport, Berea Richmond Airfield, and Galla Airfield. Madison County Airport is approximately 9 miles from the storage area. At the Madison County Airport, there is a civilian flight school which operates light aircraft, ranging from single engine light planes up to twin engine aircraft. The flight school uses two training areas near the depot, one to the north and the other to the east. The Madison County airport has a 4000-ft runway. The Berea Richmond Airfield is approximately 6 miles from the storage area and can support only light aircraft on its 2400-ft grass strip runway. Galla is a small, private airfield 12 miles east of the storage area. The air traffic from these airports over the storage area is not expected to be a significant hazard.





141.741.941.941.941.94

ORNL-OWG 85-1116583



Fig. D-7. Madison county map showing the location of LBAD



ላ እስለ የሆኑ እና እር እር እር እር እር እር እር እ

53355

| Year | Month | Day | Location                        | Epicentral Intensity<br>(MMI) |
|------|-------|-----|---------------------------------|-------------------------------|
| 1779 | 1     | 1   | Kentucky<br>38.0N, 84.0W        | Unknown                       |
| 1834 | 11    | 20  | Northern KY<br>37.0N, 86.0W     | v                             |
| 1933 | 5     | 28  | Maysville, KY<br>38.7N, 83.7W   | V                             |
| 1954 | 1     | 1   | Middlesboro, KY<br>36.6N, 83.7W | VI                            |
| 1968 | 12    | 11  | Louisville, KY<br>38.0N, 85.5W  | v                             |
| 1974 | 6     | 4   | Kentucky<br>38.6N, 84.77W       | V (est)                       |
| 1976 | 1     | 19  | Kentucky<br>36.88N, 83.82W      | VI                            |
| 1979 | 11    | 9   | NE Kentucky<br>38.42N, 82.88W   | V (est)                       |
| 1980 | 6     | 27  | Kentucky<br>38.17N, 83.91W      | VII                           |
| 1980 | 8     | 2   | Kentucky<br>37.99N, 84.92W      | III                           |
| 1980 | 8     | 22  | Kentucky<br>37.99N, 84.92W      | III                           |

## TABLE D-3 EARTHQUAKES IN THE VICINITY OF THE LBAD SITE(a) (Chronological Listing)

(a)Earthquakes within a 50- to 60-mile radius of the Lexington-Blue Grass Site, abstracted from Table 2.5-2, Clinch River Breeder Reactor Plant Preliminary Safety Analysis Report. Source: Ref. D-1.

D-16

والمحمد والمتحدين المراجع والمراجع والمراجع

There is a U.S. Air Force radar bombing/scoring detachment stationed at the LBAD with frequent flights (10 to 11 aircraft per day) of Air Force B-52, F-4, and F-111 aircraft at low altitudes (750 and 3000 ft). The flights are active from 11:30 AM to 3:30 PM and from 6:00 PM until midnight every day. They fly at 750 ft under visual flight rules and at 2000 to 3000 ft under instrument rules with a visual observer. Generally, they make three simulated bombing runs per flight at distances at least 2 miles away from the chemical exclusion area. Per the guidelines of Ref. D-3, this is not expected to be a significant problem. ZUG RUZURA MATAZA PERSONA BOOKER

### D.1.4. NEWPORT ARMY AMMUNITION PLANT

The Newport Army Ammunition Plant (NAAP) is located in west central Indiana, west of Indianapolis, as shown in Figs. D-8 and D-9. NAAP is operated by Mason & Hangar. The mission of NAAP is to (1) manufacture explosive and chemical materials, (2) fill chemical munitions, and (3) to store chemical munitions. Items 1 and 2 are currently inactive, while item 3 involves the activities associated with storage of VX chemical agent ton containers. TRACES STRAD

The chemical storage area at NAAP includes a single storage warehouse (Building 144) that is used to house VX ton containers. The storage building is approximately 79 ft wide and 279 ft long. The walls and roof of the building are of heavy gauge corrugated sheet metal, supported by steel beams.

The warehouse is in an exclusion area adjacent to the former VX production facility. The grounds within the exclusion area are all concrete or macadam covered surfaces. There are several large storage tanks that were used to store agent which are located along the southeast side of the warehouse. These storage tanks are currently empty. A 409-ft tall flash tower is located 450 ft to the east of Building 144. The flash tower was utilized during production of VX to burn several flammable gas by-products. Just outside the exclusion area, approximately 560 ft to the east of Building 144, is the site of a natural gas metering station. Natural gas was distributed to the production plant and to the area boiler from this point. Several empty storage vessels are located approximately 350 ft from the nearest ton containers outside the exclusion area. These tanks were used in conjunction with the former VX production facility. These tanks are to remain empty during the demilitarization campaign.

Table D-4 summarizes earthquake activity in the vicinity of the NAAP site.



Fig. D-8. Indiana state map showing the location of NAAP

D-19


Fig. D-9. County map showing the location of NAAP

Ē

131313131313131

D-20

ORNL-DWG 85-16894



The airspace at NAAP is not restricted. The only airport within a 10-mile radius of the plant is a private airstrip (Rowe) with a 2600-ft runway located 8 miles west of the plant. The nearest public airport is Clinton which is approximately 12 miles south of the plant. Low altitude federal airway V171 passes 2 miles east of the storage area and airway V434 passes 5 miles north of the storage area. High altitude jet routes J80 and J73 cross over the storage area.





| Year | Month | Day | Location      | MMI | Distance<br>from Site<br>(km) |
|------|-------|-----|---------------|-----|-------------------------------|
| 1909 | 9     | 27  | 39.5N, 87.4W  | VII | 41                            |
| 1921 | 3     | 14  | 39.5N, 87.5W  | IV  | 41                            |
| 1903 | 12    | 31  | 40.0N, 87.9W  |     | 42                            |
| 1974 | 11    | 25  | 40.3N, 87.4W  | II  | 48                            |
| 1906 | 7     | 13  | 39.7N, 86.8W  |     | 57                            |
| 1906 | 8     | 13  | 39.7N, 86.8W  | IV  | 57                            |
| 1984 | 8     | 29  | 39.3N, 87.2W  | v   | 58                            |
| 1978 | 2     | 16  | 39.8N, 88.23W |     | 68                            |
| 1984 | 7     | 28  | 39.2N, 87.1W  | V   | 78                            |

# TABLE D-4EARTHQUAKES IN THE VICINITY OF THE NAAP SITE<br/>(Ordered By Distance From Site)

19.94.96.96.96.96.96.91.96.91

X

Data provided by the National Geophysical Data Center, NOAA.



#### D.1.5. PINE BLUFF ARSENAL

As shown in Figs. D-10 and D-11, the Pine Bluff Arsenal (PBA) is located southeast of Little Rock, Arkansas and northwest of the city of Pine Bluff, Arkansas. The primary missions include storage of conventional and chemical munitions, destruction of nontoxic chemicals, and production of smoke munitions, white phosphorus projectiles and other incendiary devices. Future responsibilities include demilitarization of the BZ stockpile and production of binary chemical munitions.

The chemical storage area at PBA is located in the northwestern section of the installation. The following munitions are stored at PBA: 4.2-in. mortar projectiles, M55 rockets, land mines, and ton containers. All munitions except ton containers are stored in 80-ft igloos. Ton containers containing mustard agent are stored outdoors in a fenced area within the chemical storage area. The ton containers are strapped to railroad rails and stacked one high per AMC regulations.

Table D-5 summarizes earthquake activity in the vicinity of the PBA site.

PBA airspace is not restricted. The closest important airfield, Grider Field, is about 16 miles southeast of the chemical munition storage area. There are three smaller airfields which are closer (10 to 14 miles). Because of the relatively significant distances from airfields, PBA is not considered to have a significant hazard due to airfield operations.

Grider handles approximately 115 aircraft movements per day, seven days a week. About 95% of this traffic is corporate/civilian, and the remainder is military. The runway at Grider Field is 6,000 ft and can occasionally accommodate commercial 727 and military C-141 aircraft. Low altitude federal airways V74, V305, and V16 pass within 7, 10, and 11 miles, respectively. High altitude airway J42 passes over the site.

D-23



2023





でスジナシ

There is a helipad onsite about 2 miles away from the chemical munition storage area boundary. The flight frequency was estimated to be 30 or less flights a month (Ref. D-1).



F11111110 122222044 125555997



| Year | Month | Day | Location      | Epicentral Intensity<br>(MMI) |
|------|-------|-----|---------------|-------------------------------|
| 1911 | 3     | 31  | 33.8N, 92.2W  | VI                            |
| 1918 | 10    | 4   | 34.7N, 92.3W  | v                             |
| 1930 | 11    | 16  | 34.3N, 92,8W  | v                             |
| 1939 | 6     | 19  | 34.1N, 93.1W  | V                             |
| 1967 | 6     | 4   | 33.5N, 90.8W  | VI                            |
| 1967 | 6     | 29  | 33.5N, 90.8W  | v                             |
| 1969 | 1     | 1   | 34.3N, 92.6W  | VI                            |
| 1974 | 2     | 15  | 33.9N, 93.0W  | v                             |
| 1974 | 12    | 13  | 34.5N, 91.8W  | v                             |
| 1978 | 9     | 23  | 33.6N, 91.89W | v                             |
| 1982 | 1     | 21  | 35.1N, 92.2W  | v                             |
| 1982 | 1     | 24  | 35.2N, 92.2W  | v                             |
| 1982 | 2     | 24  | 35.1N, 92.2W  | v                             |
| 1982 | 3     | 1   | 35.1N, 92.2W  | v                             |
| 1983 | 1     | 19  | 35.1N, 92.2W  | v                             |

## TABLE D-5 EARTHQUAKES IN THE VICINITY OF THE PBA SITE(a) (Chronological Listing)

(a)Earthquakes within a 100 mile (160 km) radius of the Pine Bluff site as provided by the National Geophysical Data Center, NOAA. Records believed to be duplicates are reported only once. Source: Ref. D-1.



#### D.1.6. PUEBLO DEPOT ACTIVITY

The Pueblo Depot Activity (PUDA) is under the command of the Tooele Army Depot. As shown in Figs. D-12 and D-13, the installation lies east of the city of Pueblo, Colorado and north of the Arkansas River. The mission of PUDA facilities is to operate a reserve storage and maintenance function providing for (1) limited receipt, storage, and issue of assigned commodities; (2) depot maintenance of assigned commodities; (3) limited maintenance of facilities to prevent deterioration of the ammunition stockpile; (4) operation of a calibration service for an assigned geographical area; (5) demilitarization and disposal of deteriorated explosives and munitions; (6) ammunition surveillance; (7) small arms clipping and linking; (8) operation of the Function/Trace Test Range; and (9) missile maintenance/production.

The chemical storage area at PUDA is located in the northeast corner of the depot in the G-block storage area. The following munitions are stored at PUDA: 155-mm projectiles, 105-mm cartridges and projectiles, and 4.2-in. mortar projectiles. All munitions are stored in 80-ft igloos.

Table D-6 summarizes earthquake activity in the vicinity of the PUDA site.

The airspace at the PUDA is not restricted. There is a private airport (Youtsey) a few miles south of the depot. The nearest public airport is Pueblo Memorial which is located 6 miles west of the boundary of the depot. This airport has four runways, the longest being 10,500 ft. Pueblo Memorial is used as a training airport for both commercial and military aircraft. Low altitude federal airways V10, V19, V81, V83, V244, and V389 all pass within a few miles of the depot, as do high altitude jet routes J17 and J28.



1.222.222

Fig. D-12. Colorado state map showing the location of PUDA

04W - 9M4 83- 1488



initian and an

D-29



Fig. D-13. County map showing the location of PUDA



| (Ordered By Distance From Site) |       |     |               |     |                               |
|---------------------------------|-------|-----|---------------|-----|-------------------------------|
| Year                            | Month | Day | Location      | MMI | Distance<br>from Site<br>(km) |
| 1963                            | 11    | 13  | 38.3N, 104.6W | IV  | 22                            |
| 1870                            | 12    | 4   | 38.5N, 104.0W | VI  | 37                            |
| 1955                            | 11    | 28  | 38.2N, 103.7W | IV  | 58                            |
| 1925                            | 2     | 18  | 38.2N, 105.1W | IV  | 67                            |
| 1888                            | 10    | 23  | 38.1N, 105.2W | IV  | 78                            |

لتككك

TABLE D-6 EARTHQUAKES IN THE VICINITY OF THE PUDA SITE (Ordered By Distance From Site)

Data provided by the National Geophysical Data Center, NOAA.



#### D.7. TOOELE ARMY DEPOT

1773-14 Cook

1070144

The Tooele Army Depot (TEAD) is located in north central Utah southwest of Salt Lake City as shown in Figs. D-14 and D-15. The Army Depot consists of two separate areas, North and South. The chemical agent storage and demilitarization operations are located in the South Area. The mission of TEAD is to operate a supply depot providing for receipt, storage issue, maintenance and disposal of assigned commodities; and to operate other facilities such as the Chemical Agent Munitions Disposal System (CAMDS).

The chemical storage area at TEAD is located in the center of the south area. There are storage magazines, warehouse buildings, and several storage yards within the chemical agent exclusion area. The storage magazines include both 89-ft oval-arch magazines and 80-ft igloo magazines. M55 rockets, 155-mm and 8-in. projectiles, 105-mm cartridge projectiles, 4.2-in. mortar projectiles, GB and VX ton containers, M23 land mines, and weteye bombs are stored in the 80-ft igloos. MC-1 bombs, 155-mm and 105-mm projectiles are stored in the 89-ft oval-arch magazines. Ton containers containing mustard are stored outdoors. The two warehouse buildings currently are used to store VX spray tanks packaged inside TMU-28/B storage and shipping containers.

The warehouse buildings are flat-roofed, single-story structures approximately 188 ft long, 179 ft wide, and 16 ft high. Details of construction are shown in Army Corps of Engineers Drawing 201-25-65. The side walls of the buildings are single piece precast concrete panels 6 in. thick, 16 ft high, with widths varying around 30 ft. The roof is of corrugated sheet metal, supported by a steel beam support structure and steel box beam vertical support columns. The main beams are W24 x 68 steel I-beams with unsupported spans of about 30 ft. Open trusses are used to span between the main beams.

D-32



Fig. D-14. Utah state map showing the location of TEAD

D-33



الانتنا



Table D-7 summarizes earthquake activity in the vicinity of the TEAD site.

The airspace over the TEAD South Area is not restricted but pilots are requested (for reasons of national security) to avoid flying below 6400 ft over this area for a radius of 3 nautical miles (3.5 statute miles).

Tooele Municipal Airport is the nearest airport to the site. It is located 14 miles north of the site and is not expected to present a significant hazard.

There are two low altitude federal airways in the vicinity of the TEAD South Area: V257, three miles to the west, and V253, 17 miles to the northeast. High altitude airways are not considered a hazard for this site.

There is a helipad located near the administrative building approximately 3 miles from the chemical munition storage area. The helipad is used infrequently. The number of flights per month is estimated to be 15.

| الكنيك ا |
|----------|
|----------|

1000 - 1555.

2020

٧1

0 C A A

G 10

-7

0.0.0.0.0

| Year | Month  | Day     | Location      | Epicentral Intensity<br>(MMI) |
|------|--------|---------|---------------|-------------------------------|
| 1853 | 12     | 1       | 39.7N, 111.8W | V                             |
| 1876 | 3      | 22      | 39.5N, 111.5W | VI                            |
| 1880 | 9      | 17      | 40.8N, 112.0W | v                             |
| 1884 | 11     | 10      | 40.8N, 111.9W | VIII                          |
| 1894 | 1      | 8       | 39.7N, 113.4W | v                             |
| 1894 | 6      | 8       | 39.9N, 113.4W | v                             |
| 1894 | 7      | 18      | 41.2N, 112.0W | VII                           |
| 1899 | 12     | 13      | 41.0N, 112.0W | v                             |
| 1900 | 8      | 1       | 39.8N, 112.2W | VII                           |
| 1906 | 5      | 24      | 41.2N, 112.0W | v                             |
| 1909 | 11     | 17      | 41.7N, 112.2W | v                             |
| 1910 | 5      | 22      | 40.8N, 111.9W | VII                           |
| 1914 | 4      | 8       | 41.2N, 111.6W | V                             |
| 1915 | 7      | 15      | 40.3N, 111.7W | VI                            |
| 1915 | 7      | 30      | 41.7N, 112.1W | v                             |
| 1915 | 8      | 11      | 40.5N, 112.7W | v                             |
| 1915 | 10     | 5       | 40.1N, 114.0W | v                             |
| 1916 | 2      | 5       | 40.0N, 111.7W | v                             |
| 1920 | 9      | 18      | 41.5N, 112.0W | VI                            |
| 1920 | 9      | 19      | 41.5N, 112.0W | VI                            |
| 1920 | 11     | 20      | 41.5N, 112.0W | VI                            |
| 1934 | 3      | 12      | 41.5N, 112.5W | VIII                          |
| 1934 | 4      | 14      | 41.5N, 112.5W |                               |
| 1934 | 5      | 6       | 41.7N, 113.0W |                               |
| 1938 | 7      | 9       | 40.5N, 111.6W | v                             |
| 1938 | 6      | 30      | 40.5N, 111.6W | VI                            |
| 1943 | 2      | 22      | 40.4N, 111.8W | VI                            |
| 1947 | 3      | /       | 40.5N, 111.6W | V                             |
| 1949 | 3      | /       | 40.5N, 111.6W | v                             |
| 1950 | 5      | 8       | 40.0N, 111.5W | V                             |
| 1951 | 0      | 12      | 40.2N, 111.4W | V<br>                         |
| 1952 | y<br>5 | 28      | 40.3N, 111.5W | V                             |
| 1955 | 2      | 24<br>1 | 40.5N, 111.5W | V1                            |
| 1955 | 5      | 10      | 40.5N, 111.6W | v                             |
| 1958 | 2      | 13      | 40.4N, 111.6W | V                             |
| 1958 | 11     | 29      | 40.5N, 111.5W | VI<br>VI                      |
| 1958 | 12     | 20      | 40 5N 112 5W  | V<br>V                        |
| 1958 | 12     | ⊥<br>2  | 40 5N 112.JW  | ¥<br>\7                       |
| 1961 | 4      | 16      | 30.1N 111 50  | ¥<br>\$7 T                    |
| 1962 | 9      | 5       | 40.7N. 112.0W | VT                            |
| 1963 | 7      | 7       | 39.6N. 111 OU | VT<br>VT                      |
| 1963 | ,<br>7 | ,<br>9  | 40.0N. 111.2W | ₹1                            |
| 1963 | 7      | 10      | 39.9N. 111.4W | v                             |
| 1965 | 5      | 11      | 41.0N. 111.5W | •                             |

## TABLE D-7 EARTHQUAKES IN THE VICINITY OF THE TEAD SITE<sup>(a)</sup> (Chronological Listing)

D-36

| Year | Month | Day | Location      | Epicentral Intensity<br>(MMI) |
|------|-------|-----|---------------|-------------------------------|
| 1966 | 5     | 23  | 39.2N, 111.4W |                               |
| 1967 | 2     | 16  | 41.3N, 113.3W | v                             |
| 1967 | 9     | 24  | 40.7N, 112.1W | v                             |
| 1967 | 12    | 7   | 41.3N, 111.7W | v                             |
| 1968 | 1     | 16  | 39.3N, 112.2W | v                             |
| 1968 | 11    | 17  | 39.5N, 110.9W | v                             |
| 1969 | 5     | 23  | 39.0N, 111.8W | v                             |
| 1970 | 4     | 14  | 39.6N, 110.7W | v                             |
| 1970 | 10    | 25  | 39.1N, 111.3W | v                             |
| 1972 | 10    | 1   | 40.5N, 111.3W | VI                            |
| 1972 | 10    | 16  | 40.4N, 111.0W | v                             |
| 1973 | 7     | 16  | 39.1N, 111.5W | v                             |
| 1977 | 11    | 28  | 41.3N, 111.6W | v                             |
| 1978 | 2     | 28  | 40.7N, 112.2W | v                             |
| 1978 | 3     | 9   | 40.7N, 112.0W | VI                            |
| 1978 | 3     | 13  | 40.7N, 112.0W | v                             |
| 1980 | 5     | 24  | 39.9N, 111.9W | v                             |
| 1981 | 2     | 20  | 40.3N, 111.7W | v                             |
| 1981 | 5     | 14  | 39.4N, 111.0W | v                             |
| 1983 | 10    | 8   | 40.7N. 111.9W | VI                            |

TABLE D-7 (Continued)

(a)Earthquakes within a 100-mile radius of TEAD as provided by the National Data Center, NOAA. Records believed to be duplicated are reported only once. Source: Ref. D-1.

#### D.1.8. UMATILLA DEPOT ACTIVITY

The Umatilla Depot Activity (UMDA) is under the command of TEAD. As shown in Figs. D-16 and D-17, the installation is located in Umatilla and Marrow Counties in northeastern Oregon, near the south shore of the Columbia River, west of Hermiston, Oregon. UMDA's mission is to operate a reserve storage depot activity under the command of TEAD providing care and preservation for and minor maintenance of assigned commodities.

Sector S

The storage area is located at the northern edge of the installation. Eighty-foot igloo magazines and warehouses are used to store the chemical munition stockpile of 155-mm and 8-in. projectiles, M55 rockets, M23 land mines, bombs, spray tanks, and ton containers. Warehouses are used to store ton containers containing mustard agent. The magazines are spaced 400 ft apart.

The warehouses are butler type buildings connected by a roof with a steel structure and aluminum siding (single sheet). The two buildings are defined as transitory structures, approximately 154 ft wide (total for both buildings) and 300 ft long.

Table D-8 summarizes earthquake activity in the vicinity of the UMDA site.

The UMDA airspace is not restricted. The nearest active airfield to the Umatilla site is Hermiston Municipal Airport approximately 12 miles from the depot. With one 4000-ft runway, its capabilities aro limited to aircraft up to the size of corporate jets. The Tri-Cities Airport in Pasco, Washington, with a maximum runway length of 7700 ft, is approximately 30 miles from the depot. In general, it does not handle military aircraft. There is also a paved runway on the UMDA site capable of handling small aircraft up to the size of a Beech U-21 light utility aircraft. The nearest military airfields are in Spokane, Washington; Moses Lake, Washington; and Mt. Home, Idaho.



D-39

A DECEMBER OF THE REAL OF T



actives back add



| Year | Month | Day | Location             | Epicentral Intensity<br>(MMI) |
|------|-------|-----|----------------------|-------------------------------|
| 1893 | 3     | 5   | Umatilla, OR         | VI                            |
| 1918 | 11    | 1   | 46.7N, 119.5W        | V to VI                       |
| 1921 | 9     | 14  | Dixie, WA            | V to VI                       |
| 1924 | 1     | 6   | Walla Walla, WA      | IV                            |
| 1924 | 1     | 6   | Milton Weston, OR    | V                             |
| 1924 | 5     | 26  | Walla Walla, WA      | IV                            |
| 1926 | 4     | 23  | Walla Walla, WA      | IV                            |
| 1936 | 7     | 15  | 46.0N, 118.5W        | VII                           |
| 1936 | 7     | 18  | 46.0N, 118.3W        | v                             |
| 1936 | 7     | 20  | Freewater, OR        | IV                            |
| 1936 | 8     | 4   | 45.8N, 118.6W        | v                             |
| 1936 | 11    | 17  | Walla Walla, WA      | III                           |
| 1937 | 2     | 9   | Walla Walla, WA      | IV                            |
| 1937 | 6     | 4   | Walla Walla, WA      | IV                            |
| 1938 | 8     | 11  | Milton, OR           | IV                            |
| 1938 | 10    | 27  | Milton, OR           | IV                            |
| 1944 | 9     | 1   | Walla Walla, OR      | IV                            |
| 1945 | 9     | 22  | Walla, Walla, OR     | IV                            |
| 1951 | 1     | 7   | McNary, OR           | V                             |
| 1959 | 1     | 20  | Milton-Freewater, OR | V                             |
| 1959 | 11    | 9   | Heppner, OR          | IV                            |
| 1971 | 10    | 25  | 46.7N, 119.5W        | IV                            |

## TABLE D-8 EARTHQUAKES IN THE VICINITY OF THE UMDA SITE(a) (Chronological Listing)

2

Earthquakes within a 50- to 60-mile radius of the Umatilla site, abstracted from Table 2.5-2, UNI-M-90, "N Reactor Updated Safety Analysis Report," United Nuclear Industries, Inc., February 28, 1978. Source: Ref. D-1.





D-41

The Medium Attack Tactical Electronic Warfare Wing bombing range is located 10 miles to the southwest of UMDA chemical munitions exclusion area. This area is a restricted airspace (Restriction numbers R-5701, R-5704, R-5706) in which the Navy holds bombing exercises. Grumman A-6 aircraft, in groups of four, fly about 14 sorties during the day and ten sorties at night, five days a week, dropping inert 25-1b bombs and, occasionally, 500- to 1000-1b inert bombs. Per the guidelines of Ref. D-8, this is not considered a significant threat. There are two low altitude federal airways in the general area of the depot: V-4 and V-112. Three high altitude airways (J-16, J-20, and J-54) cross within 6 miles of the depot toward Pendleton, Oregon.

The installation provides limited maintenance to preclude deterioration of facilities and retains limited shipping and receiving capabilities.

SAVET EXCERCE TRANSPORT SAVET

## D.1.9. REFERENCES

- D-1. Science Applications International Corporation, "Probabilities of Selected Hazards in Disposition of M55 Rockets," U.S. Army Toxic and Hazardous Materials Agency, M55-CS-2, November 1985.
- D-2. Jeppesen, "United States High Altitude Enroute Charts," U.S. (HI) 1-5, March 1986.
- D-3. "Aircraft Hazards," U.S. Nuclear Regulatory Commission Standard Review Plan 3.5.1.6, NUREG-0800, Rev 2, July 1981.

D-43



APPENDIX E (Deleted)

<u> 283 - 2999 - 2000 - 2000 - 20093</u>







## APPENDIX F MUNITION FAILURE THRESHOLDS



No.6

#### F.1. MUNITION FAILURE THRESHOLDS

The munition stockpile is comprised of 11 different munition types. This appendix contains a description of the physical characteristics of each munition type, a description of their existing storage configurations, and a description of the munition failure thresholds that are important for quantifying the agent release associated with each accident scenario. The failure thresholds discussed herein are the thresholds for accidental burster detonation, the thermal threshold for hydraulic rupture of the agent compartment, and the mechanical failure thresholds which lead to failure of the agent compartment.

#### F.1.1. DESCRIPTION OF CHEMICAL MUNITIONS

The chemical stockpile is presently made up of the following munitions:

 8-in. artillery projectiles. The 8-in. projectiles are filled with the nerve, agent either GB or VX. They are stored without fuzes, but they may be stored with or without bursters. The 8-in. projectiles are stored on wooden pallets with six rounds per pallet.

EXERCISE LONGER BESSES.

- 2. 155-mm artillery projectiles. The 155-mm projectiles may contain GB, VX, or mustard. They are stored without fuzes, but they may be with or without bursters. The 155-mm projectiles are stored on wooden pallets with eight rounds per pallet.
- 3. 105-mm artillery rounds. The rounds are filled with either mustard or GB. The rounds may be stored as bare projectiles

F~1

on wooden pallets, with 24 rounds per pallet, and with 2 pallets butted together and secured with steel banding, or as cartridges in fiber tubes, with two tubes in a wooden field box, and with either 12 or 15 boxes unitized on a skid based wooden pallet. The cartridges include burster, fuze, cartridge case and propellant.

4. 4.2-in. mortar projectiles. All are filled with mustard agent. The mortars may be stored with burster, fuze, and propellant in fiber tubes, with two tubes in a wooden field box, with either 36 boxes on a wooden pallet, or 24 boxes on a wooden skid base. The mortars may also be stored without burster and fuze in wooden pallets.

- 5. M23 land mines. All land mines are filled with VX. The mines are burstered, and are packaged three to a steel drum. Mine activators and fuzes are packaged separately in the same drum. Twelve drums are contained on a wooden pallet.
- 6. M55 rockets. The M55 rockets are filled with either GB or VX. The rockets are equipped with fuzes and bursters which contain explosives. Propellant is also built into the motor of the rocket. The rocket casing is made of aluminum which may slowly react with nerve agent to form hydrogen gas. Pressure buildup in some of the rockets has caused a leakage problem.

The rockets are individually packaged in fiberglass shipping tubes with metal end caps. Fifteen containers with rockets are packed on a wooden pallet.

7. MC-1 750-1b bombs filled with GB. The MC-1 bombs are stored without explosive components on wooden pallets with two bombs per pallet.

. . .

------

- 8. MK-94 500-1b bombs filled with GB. The MK-94 bombs are stored without explosive components in individual MK-410 storage and shipping containers.
- 9. MK-116 (Weteye) 600-1b Navy bombs filled with GB. These bombs are stored without explosive components in individual MK-398 storage and shipping containers.
- 10. TMU-28/B airborne spray tanks filled with VX. They were designed for releasing chemical agent from slow-traveling, low-flying aircraft. The spray tanks are stored in individual CNU-77/E23 storage and shipping containers.
- 11. Ton containers. A large fraction of the chemical stockpile is stored in bulk form in cylindrical steel containers referred to as ton containers. The ton containers may contain GB, VX, or mustard. The ton containers are not palletized, but are banded together in clusters.

Drawings and photographs of each of the above munitions are shown in Figs. F-1 through F-35.

During transportation of the munitions, either to an onsite disposal facility or an offsite disposal facility, the munitions are placed in a protective shipping container or package. The shipping package has not yet been designed, but criteria for the structural and thermal protection to be provided during munition transport are defined in Ref. F-1.



WWWW NNEWS

111111 2125 C

شندندنان

| LENGTH         | 35.1 in.               |
|----------------|------------------------|
| DIAMETER       | 8 in.                  |
| TOTAL WI.      | 199 lb.                |
| AGENT          | GB                     |
| AGENT WT.      | 14.5 16.               |
| PUZE           | None                   |
| BURSTER        | M83                    |
| explosive      | Сощр В                 |
| EXPLOSIVE WT.  | 7.0 15.                |
| SUPP. CHARGE   | 0.3 1b. TNT            |
| PROPELLANT     | None                   |
| PROPELLANT WT. | N/A                    |
| PRIMER         | None                   |
| QD/SCC         | 84                     |
| PACKAGING      | 6 rounds/wooden nallet |

PROJECTILE, 8 INCH, GB, M426

Fig. F-1. Projectile, 8-in., GB, M426



Fig. F-2. Eight-inch projectiles are stored on wooden pallets, six rounds to a pallet

DC. Do. A



2222223

22.22.23

KANARA -

Koosa - America - Konstant - Koosa - Koosa

| LENGTH         | 26.7 in.               |
|----------------|------------------------|
| DIAMETER       | 155em                  |
| TOTAL WT.      | 100 15.                |
| AGENT          | GB                     |
| AGENT WT.      | 6.5 lb.                |
| FUZE           | None                   |
| BURSTER        | M37                    |
| Explosive      | Tetrytol               |
| EXPLOSIVE WT.  | 2.75 lb.               |
| SUPP. CHARGE   | 0.3 lb. Tetrytol       |
| PROPELLANT     | None                   |
| PROPELLANT WT. | N/A                    |
| PRIMER         | None                   |
| QD/SCG         | 84                     |
| PACKAGING      | 8 rounds/wooden pallet |

## PROJECTILE, 155mm, GB, M121

Fig. F-3. Projectile, 155-mm, GB, M121



P.N

•

| LENGTH         | 26.7 in.               |
|----------------|------------------------|
| DIAMETER       | 155mm                  |
| TOTAL WT.      | 100 15.                |
| AGENT          | GB                     |
| AGENT WT.      | <b>6.5</b> 1b.         |
| FUZE           | None                   |
| BURSTER        | M71                    |
| EXPLOSIVE      | Comp B4                |
| EXPLOSIVE WT.  | 2.45 16.               |
| SUPP. CHARGE   | 0.3 lb. Tetrytol       |
| PROPELLANT     | None                   |
| PROPELLANT WT. | N/A                    |
| PRIMER         | None                   |
| QD/SCG         | 84                     |
| PACKAGING      | 8 rounds/wooden pallet |

PROJECTILE, 155mm, GB, M121A1

Fig. F-4. Projectile, 155-mm, GB, M121A1



\*\*\*\*\*\*

| LENGTH         | 26.7 in.               |
|----------------|------------------------|
| DIAMETER       | 155aa                  |
| TOTAL WT.      | 100 1Ъ.                |
| AGENT          | GB                     |
| AGENT WI.      | 6.5 lb.                |
| FUZE           | None                   |
| BURSTER        | N37                    |
| EXPLOSIVE      | Tetrytol               |
| EXPLOSIVE WT.  | 2.75 15.               |
| SUPP. CHARGE   | 0.3 1b. INT            |
| PROPELLANT     | None                   |
| PROPELLANT WT. | N/A                    |
| PRIMER         | None                   |
| QD/SCG         | 8.                     |
| PACKAGING      | 8 rounds/wooden pallet |

# PROJECTILE, 155mm, GB, M122

Fig. F-5. Projectile, 155-mm, GB, M122

F-8





LENGTH DIAMETER TOTAL WT. AGENT AGENT WT. FUZE BURSTER EXPLOSIVE EXPLOSIVE WT. SUPP. CHARGE PROPELLANT PROPELLANT WT. PRIMER QD/SCG PACKAGING 26.7 in. 155mm 100 lb. VX 6.0 lb. None M71 Comp B4 2.45 lb. 0.3 lb. Tetrytol None N/A None 84 8 rounds/wooden pallet

PROJECTILE, 155mm, VX, M121A1

Fig. F-6. Projectile, 155-mm, VX, M121A1

1

F-9



Se .

i de la comp

| LENGTH         | 26.8 in.               |
|----------------|------------------------|
| DIAMETER       | 155mm                  |
| TOTAL WT.      | <b>99 1b.</b>          |
| AGENT          | H                      |
| AGENT WT.      | 11.7 15.               |
| FUZE           | None                   |
| BURSTER        | M6                     |
| EXPLOSIVE      | Tetrytol               |
| EXPLOSIVE WT.  | .41 1b.                |
| PROPELLANT     | None                   |
| PROPELLANT WT. | N/A                    |
| PRIMER         | None                   |
| QD/SCG         | 54                     |
| PACKAGING      | 6 rounds/wooden pallet |

PROJECTILE, 155mm, H, M110

Fig. F-7. Projectile, 155-mm, H, M110

(v. v


<u>ᡧᡕ᠖ᠼ᠔ᠼ᠔ᠼᡘᡧᠼᢟᡀᢌ᠄ᢗᡀᡵᢟᡆᠼᡧᡖᡀᢞᢛ᠘ᢞᢛ᠘ᡷᠲᠥᡷᡎᡧᠼᢤᢨᡇᢤᡛᢙᢟᡆᢤᡘᡀᡀᡱᡜᡀᡀᡀᡀᡘᠼᡭᠼᡘᡱᢤᡘᡭ</u>

LENGTH 26.8 in. DIAMETER 155mm TOTAL WT. 95 1b. AGENT HD AGENT WT. 11.7 15. None BURSTER MG EXPLOSIVE Tetrytol EXPLOSIVE WT. .41 15. PROPELLANT None PROPELLANT WT. N/A PRIMER None QD/SCG 54 PACKAGING 6 rounds/wooden pallet

PROJECTILE, 155mm, HD, M104

FUZE

Fig. F-8. Projectile, 155-mm, HD, M104



**٦**.

U

v c

Fig. F-9. 155-um projectiles are stored on wooden pallets with eight rounds per pallet



LENGTH DIAMETER TOTAL WT. AGENT AGENT WT. FUZE BURSTER EXPLOSIVE EXPLOSIVE WT. PROPELLANT QD/SCG PACKAGING

RUDUCAS AND A CAR

B

3

NXX

Ş

16.0 in. 105mm 32 lb. GB 1.63 lb. M508 M40, M40Al Tetrytol(M-40) Comp B(M40A) 1.12 lb. Removed 5A 24 projectiles/wooden pallet S.1557 3322 3

Note: Projectile is stored with and without fuze and burster. Fuze cavity of unfuzed unburstered projectile is sealed by a closing plug.

PROJECTILE, 105mm, GB, M360

Fig. F-10. Projectile, 105-mm, GB, M360



**n n n n n n n** n n

) a 30

Ŷ

ŝ

| ENGTH         | 21.0 in.                     |
|---------------|------------------------------|
| LAMETER       | 105aa                        |
| TOTAL WT.     | 32 1b.                       |
| GENT          | HD                           |
| GENT WI.      | 3 16.                        |
| UZE           | PD M51A5, M57                |
| URSTER        | M5                           |
| XPLOSIVE      | Tetrytol                     |
| EXPLOSIVE WT. | 0.51 15.                     |
| ROPELLANT     | Removed                      |
| PACKAGING     | 24 projectiles/wooden pallet |
|               |                              |

PROJECTILE, 105mm, HD, M60

Fig. F-11. Projectile, 105-mm, HD, M60



Fig. F-12. 105-mm artillery rounds stored in one of the two acceptable configurations







l

1222

KANNA PANANA

1223323

----

DTTDDDA TOADDA LAAAAA LAAAAA TAADAAA TAADAAA TAADAAA TAADAAA

1.1.1

| 21.0 in.                |
|-------------------------|
| 4.2 in.                 |
| 25 lb.                  |
| HD                      |
| 6.0                     |
| M8                      |
| ML4                     |
| Tetryl                  |
| .14 1b.                 |
| Removed                 |
| M2 8A2                  |
| 54                      |
| 24 rounds/wooden pallet |
|                         |

CARTRIDGE, MORTAR, 4.2 INCH, HD, M2/M2A1

Fig. F-15. Cartridge, mortar, 4.2-in., HD, M2/M2A1

PROPELANT BURSTER HD BURSTER WELL STRIKER NUT KGNITION CARTIFIDGE OBTURATING MECHANISM BODY in in in it.

- SCHOOL BOOK

፟ዸኇ፟ዾቔጞዹቔኇዀቔጞዺቘኇ፟ዺጜጚ፟፟፟፟ቘቒጞዾቔጞ፟፟፟፟፟፟፟፟፟፟፟ቘጞዾዀጞ፝ዹቘጜ፟ዀ፟፟፟ዀዀዀጞዄ፟፟፟ቘጜ፟ኯቜጞዾቜኇ፟ኯቜጜዄዄጞኯቔ

እናለ

21.0 in. LENGTH 4.2 in. DIAMETER 25 1b. TOTAL WT. ĦD AGENT 6.0 AGENT WT. M8 FUZE <u>M14</u> BURSTER EXPLOSIVE Tetry1 EXPLOSIVE WT. .14 1b. Removed PROPELLANT M28A2 PRIMER QD/SCG 54 24 rounds/wooden pallet PACKAGING

CARTRIDGE, MORTAR, 4.2 INCH, HD, M2/M2A1

Fig. F-16. Cartridge, mortar, 4.2-in., HD, M2/M2A1



| LENGTH        |
|---------------|
| DIAMETER      |
| TOTAL WT.     |
| AGENT         |
| AGENT WI.     |
| FUZE          |
| BURSTER       |
| EXPLOSIVE     |
| EXPLOSIVE WT. |
| PROPELLANT    |
| PRIMER        |
| QD/SCG        |
| PACKAGING     |

P

21.0 in. 4.2 in. 25 lb. HT 5.8 lb. M51A5 M14 Tetryl .14 lb. Removed M28A2 5A 24 rounds/wooden pallet

CARTRIDGE, MORTAR, 4.2 INCH, HT, M2/M2A1

Fig. F-17. Cartridge, mortar, 4.2-in., HT, M2/M2A1







Fig. F-18. 4.2-in. mortars are stored in fiber tubes with two tubes per wooden box

# 4.2 INCH MORTAR PALLET







And a business of the second model of the second 


NUMBER OF STREET

53-5-5-5-5-5-F

6000000

The second second

| 5 in.             |
|-------------------|
| 13.5 in.          |
| 23 15.            |
| XX                |
| 10.5 15.          |
| M603              |
| M38               |
| Comp B            |
| .8 15.            |
| None              |
| N/A               |
| N/A               |
| 54                |
| 3 mines/steel dru |
|                   |

### MINE, 2 GALLON, VX, M23

Fig. F-20. Mine, 2-gal, VX, M23



Fig. F-21. M23 mines are stored in drums with three mines per drum



1555555 TRANSING

**UNANC** 

LENGTH 78.0 in. DIAMETER 115mm 57 1b. TOTAL WT. AGENT GB AGENT WT. 10.7 15. FUZE M417 BURSTER M34, M36 EXPLOSIVE Comp B EXPLOSIVE WT. 3.2 1b. PROPELLANT M28 PROPELLANT WT. 19.3 PRIMER M62 QD/SCG 5▲ PACKAGING 15 rounds/wooden pallet

ROCKET, 115mm, GB, M55

Fig. F-22. Rocket, 115-mm, GB, M55

F-25

Note: Stored in firing tube with fins folded toward the axis.



\*\*\*\*\*

| LENGTH         | 78 in.    |
|----------------|-----------|
| DIAMETER       | 11.5em    |
| TOTAL WT.      | 56 1b.    |
| AGENT          | <b>VX</b> |
| AGENT WT.      | 10.0 1    |
| FUZE           | M417      |
| BURSTER        | M34, M    |
| EXPLOSIVE      | Comp B    |
| EXPLOSIVE WT.  | 3.2 15    |
| PROPELLANT     | M67       |
| PROPELLANT WT. | 19.3 1    |
| PRIMER         | M62       |
| OD/SCG         | SA        |
| PACKAGING      | 15 rou    |

VX 10.0 1b. M417 M34, M36 Comp B 3.2 1b. M67 19.3 1b. M62 5A 15 round/wooden pallet

Note: Stored in firing tube with fins folded toward the axis.

ROCKET, 115mm, VX, M55

Fig. F-23. Rocket, 115-mm, VX, M55



830125-12

Fig. F-24. M55 rockets are stored in their shipping tubes with 15 rockets housed in a wooden crate

## **M 55 ROCKET PALLET**



115-mm M55 rocket pallet containing 15 rockets in individual

fiberglass tubes (15/pallet)

Fig. F-25.



| ءک | -10 a |
|----|-------|
|    | •     |
| ÷  |       |
| ~  | -14   |

| 50 in.                |
|-----------------------|
| 16 in.                |
| 725 16.               |
| GB                    |
| 220 15.               |
| None                  |
| None                  |
| None                  |
| N/A                   |
| None                  |
| N/A                   |
| None                  |
| 88                    |
| 2 bombs/wooden pallet |
|                       |

BOMB, 750 LB., GB, MC-1

Fig. F-26. Bomb, 750-1b, GB, MC-1

ŝ.

ROUTENT BUILD RECEASE



Fig. F-27. MC-1 750-1b bomb pallet with two bombs

2 1 Anda • •

ķ



Researces Contractions

Sw:

| 89 in.        |
|---------------|
| 11 in.        |
| 441 1b.       |
| GB            |
| 108           |
| None          |
| None          |
| None          |
| N/A           |
| None          |
| None          |
| 1 bomb/pallet |
|               |

Fig. F-28. Bomb, 500-1b, GB, MK 94-0



10.000.00

Fig. F-29. MK-94 bombs are stored individually in MK-410 storage and shipping containers



LENGTH DIAMETER TOTAL WT. AGENT AGENT WT. FUZE BURSTER EXPLOSIVE EXPLOSIVE WT. PROPELLANT PRIMER PACKAGING

86 in. 14 in. 525 lb. GB 384 lb. M904E2 None None N/A None Stored in a metal shipping and storage container

Fig. F-30. MK-116 Mod O bomb (Weteye) with M990 D fuze installed





840099-158

### Fig. F-31. MK-116 bombs are stored individually in MK-398 storage containers



------



| LENGTH         | 185 in.                |
|----------------|------------------------|
| DIAMETER       | 22.5 in.               |
| TOTAL WT.      | 1935 lb.               |
| AGENT          | XX                     |
| AGENT WT.      | 1356 lb.               |
| FUZE           | None                   |
| BURSTER        | None                   |
| EXPLOSIVE      | None                   |
| EXPLOSIVE WT.  | N/A                    |
| PROPELLANT     | None                   |
| PROPELLANT WT. | N/A                    |
| PRIMER         | None                   |
| QD/SCG         | 84                     |
| PACKAGING      | 1 tank/steel container |
|                |                        |

Fig. F-32. Tank, spray, VX, TMU-28/B





13. 119. 149. 149. 148. 148. 148

ale states and sold and

840099-87

A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A S







LENGTH DIAMETER TOTAL WT. AGENT AGENT WT. FUZE BURSTER EXPLOSIVE EXPLOSIVE WT. PROPELLANT PROPELLANT WT. PRIMER QD/SCG PACKAGING

81.5 in. 30.1 in. 3100 1b.; 2900 1b.; 3000 1b. ED GB XX 1700 1500 1600 None None None N/A None N/A None 84 None

### TON CONTAINER

Fig. F-34. Ton container



Fig. F-35. Ton containers store chemical agents in bulk form



### F.1.2. BURSTER DETONATION THRESHOLD

Stimuli which can initiate detonations in high explosives include (Ref. F-2):

- Shock initiation.
- Impact initiation.
- Thermal initiation.
- Friction.
- Static electric discharge.

High explosives can always be detonated by sufficiently strong shock waves because that is their mode of initiation in normal use. By design, burster reaction initiated by either friction or static electric discharge is considered incredible. In addition, secondary high explosives are relatively insensitive to shock and impact initiation for safety in use, transportation, and storage. Nevertheless, accidental detonation of munitions is considered credible when the munitions are subjected to undue force arising from an accident. A measure of the sensitivity of the munitions to accidental impact is indicated by the Susan test. In this test, the ignition point of the high explosive is determined as a function of impact velocity. Given the explosive confinement designed into the munition, ignition can be interpreted as leading to a violent explosion. According to Ref. F-3, the threshold velocity for ignition is about 180 ft/s (123 mph) for COMP B-3 and 235 ft/s (160 mph) for TNT. COMP B-3 and TNT are major components of the munition bursters. These velocities are well above any credible impact velocity arising from the accident scenarios considered herein except the aircraft crash. However, spontaneous, or unexplained detonations have been known to occur. Therefore, the possibility of a detonation is evaluated for those accidents which may introduce an undue force as part of the accident scenario.

### F.1.3. THERMAL FAILURE THRESHOLDS

The thermal failure threshold is defined as the time to fail the agent compartment when the munition is enveloped by fire. In the case of burstered munitions, including those which are also packaged with propellant, the thermal threshold may be a violent detonation. For non-burstered munitions, failure occurs by rupturing the agent-containing vessel because of internal pressure buildup associated with the addition of heat. The thermal failure thresholds for the various munition types were determined by analysis (Refs. F-4 and F-5). They are shown in Table F-1. Two fire scenarios were considered: (1) direct heating of a munition by an 1850°F fire and (2) indirect heating of a munition whereby the fire heats a 1/4-in. steel plate positioned 6 in. from the munition. The air space between the plate and the munition is considered static with heat transfer occurring by conduction and radiation.

このからですが、「とくなくない」」「ないたちない」

As shown in Table F-1, the results indicate that burster detonation occurs before hydraulic rupture. When subjected to direct exposure to a fire, rockets can detonate in as little as 4 min, and cartridges and projectiles in 6.5 min. A significant increase in exposure time is generally predicted for an indirect fire. This would correspond to the munitions in an uninsulated steel overpack such as a rocket sport, or the vault container to be used for offsite transportation. The corresponding times to reach detonation temperature are 10.5 min for rockets, 75 min for cartridges, and 89 min for projectiles.

The nonburstered munitions are subject only to hydraulic rupture when enveloped by fire. The predicted exposure time to reach failure (Table F-1) is typically about 30 min for direct exposure to fire and typically more than 2 h for indirect exposure.

|                                                                | Direct Exposure            | Indirect Exposure              |
|----------------------------------------------------------------|----------------------------|--------------------------------|
| Cartridges(a)                                                  |                            |                                |
| Burster detonation<br>Hydraulic failure<br>Propellant ignition | 6.5 min<br>11 min<br>6 min | 75 min<br>>2 h<br>49 min       |
| Projectiles(a)                                                 |                            |                                |
| Burster detonation<br>Hydraulic failure                        | 6.5 min<br>12 min          | 89 min<br>>2 h                 |
| Bomb(a)                                                        |                            |                                |
| Hydraulic failure<br>Ton containers(a)                         | 35 min                     | >2 h                           |
| Hydraulic failure<br>Spray tank <sup>(a)</sup>                 | 30 min                     | >1 h                           |
| Hydraulic failure<br>Mine(a)                                   | >2 h                       | >2 h                           |
| Burster detonation<br>Rocket(c)                                | 16 min(b)                  | 68 min                         |
| Buster detonation<br>Propellant ignition<br>Hydraulic failure  | 4 mín<br>5 mín<br>7 mín    | 10.5 min<br>13.7 min<br>12 min |

### TABLE F-1 CALCULATED THERMAL FAILURE THRESHOLDS

(a)One-dimensional calculation with radiation heat transfer.

(b)For individual mine (not in drum), based on test data from Ref. 5-11.

(c)Multi-dimensional calculation with convection and radiation heat transfer.

### F.1.4. MECHANICAL FAILURE THRESHOLDS

Limited information was available from other studies to define the munition mechanical failure thresholds. H&R Technical Associates reported both calculated and test results relevant to the M55 rockets (Ref. F-5). In addition, H&R Technical Associates calculated the mechanical failure thresholds for other munitions (Ref. F-6). The results of the calculated crash, impact, and puncture failure thresholds are shown in Table F-2. The results of impact tests available at the start of the risk analysis are summarized in Table F-3. The results of additional impact tests performed in July 1986 are discussed in a subsequent section.

The crush threshold is defined as the static load required to deform the munitions beyond their yield strength. Two crush threshold values are presented in Table F-2, one for axial load and another for a side load. The calculation of the axial, or end crush threshold of a single bare munition assumes that the crushing force is applied parallel to the axis against the end of the munition and that the force is equally distributed over the munition cross section. The weakest portion of the munition cross section is assumed to be the portion of the agent compartment with the thinnest wall. The side crush of a bare munition was calculated based on the assumption that the crushing force applies perpendicular to the axis against the side of the munition and that the force is equally distributed along the length of the munition. The wall thickness is assumed to be uniform along the wall. For the calculation of the end and side crush thresholds of a packaged munition, the smallest end of a pallet was chosen to be crushed on a surface. This assumes a perfectly planar fit between the pallet and its crushing surface. The pallet is also assumed to be resting on a perfectly inelastic massive surface.

The impact threshold is defined as the velocity of impact against an unyielding surface which will deform the munition beyond its failure



ENDELEXE ACCULATION AND A CONTRACTOR





\*. 64\*. 14\*. 14\*. 14\* . 64\* at

1.10° . 10° . 10

### TABLE P-2 CRUSH, IMPACT, AND PUNCTURE CALCULATION RESULTS FOR CHEMICAL MUNITIONS

| Municion Type                                                         | Axial<br>Crush<br>Force(A)<br>(1b) | Axial<br>Impact<br>Velocity(3)<br>(fpe/mph) | Axial<br>Impact<br>Height<br>(ft) | Side<br>Crush<br>Force(C)<br>(1b) | Side<br>Impact<br>Velocity(D,AE)<br>(fps/mph) | Side<br>Impact<br>Height(AE)<br>(ft) | Forklift<br>Puncture<br>Velocity(E)<br>(fps/mph) | Road ACC<br>Puncture<br>Velocity(F)<br>(fps/mph) | Rail ACC<br>Puncture<br>Velocity(G)<br>(fps/mph) |
|-----------------------------------------------------------------------|------------------------------------|---------------------------------------------|-----------------------------------|-----------------------------------|-----------------------------------------------|--------------------------------------|--------------------------------------------------|--------------------------------------------------|--------------------------------------------------|
| Weapons Containing Agent, Fuze, and Burster                           |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| <pre>Propellant-Assembled(Z,X)</pre>                                  |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| <pre>l15-mm M55 rocket in M441 S/L tube (wt 67 lb)</pre>              | 40'09                              | 57/39                                       | 50                                | 20,600(1)                         | 41/28                                         | 26                                   |                                                  |                                                  |                                                  |
| Palletized weapons - 15/pallet -<br>(wc 1,350 1b)                     | 608,000                            | 49/34                                       | 38( VV)                           | (Г)007'E7                         | 13/09                                         | ( <b>vv</b> )E                       | 10/1                                             | 2/01                                             | <b>%</b>                                         |
| 4.2-in. M2/M2Al cartridge (wt 25 lb)                                  | 152,000                            | 180/123                                     | 503                               | 18,400                            | 63/43                                         | 61                                   |                                                  |                                                  |                                                  |
| Special H55 pallet calculation(AF)                                    | ł                                  | ł                                           | 1                                 | 112 ,000(AF)                      | 51/35(AF)                                     | 40(VE)                               | 1                                                | 12/8(AF)                                         | 25/17(AP)                                        |
| Palletized in wooden boxes = 2/box -<br>24 boxes/pallet (wt 1,700 lb) | 7,300,000                          | 149/102                                     | 345                               | 110,000                           | 18/12                                         | S(AA)                                | 2/01(K)                                          | 5/04(K)                                          | 7/05(K)                                          |
| 105-eem M60/M360 cartridge (wt 32 lb)                                 | 279,000                            | 216/147                                     | 726                               | 61,600                            | 69/101                                        | 158                                  |                                                  |                                                  |                                                  |
| Palletized in wooden boxes - 2/box -<br>15 boxes/pallet (wt i,880 lb) | 8,370,000                          | 155/105                                     | 371                               | <b>306 ,</b> UM                   | 30/20                                         | 14( <b>AA</b> )                      | 4/03(K)                                          | 14/09(K)                                         | 20/14(K)                                         |
| Weapons Containing Agent, Burster                                     |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| Assembled - Exposed(2)                                                |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| 155-and MU4/N110/M121/M121A1/N122<br>projectile (wr 99 lb)            | 621,000                            | 183/125                                     | 522                               | 155,000(L)                        | 79/54                                         | 97                                   |                                                  |                                                  |                                                  |
| Palletízed weapons - 8/pallet -<br>(wt 832 lb)                        | 4,960,000                          | 178/122                                     | 497                               | 230,000(L)                        | 39/26                                         | 24 ( AA )                            | 5/04                                             | 20/14                                            | 25/17(AF)                                        |
| 8-in. M426 projectile (wt 199 lb)                                     | 1,170,000                          | 178/121                                     | 167                               | (J)000'181                        | 70/48                                         | 76                                   |                                                  |                                                  |                                                  |
| Palletized veapons - 6/pallet -<br>(wr 1,253 lb)                      | 7,030,000                          | 174/118                                     | 468                               | 361,000(L)                        | 39/27                                         | 24( <b>AA</b> )                      | <del>6</del> /04                                 | 15/10                                            | 22/15                                            |
| 750-1b MC-1 bomb (wt 725 1b)                                          | 1,290,000                          | 98/66                                       | 148                               | 54,500(M)                         | 40/27(N)                                      | 25                                   |                                                  |                                                  |                                                  |
| Palletized weapons - 2/pallet -<br>(wc 1,575 lb)                      | 2,580,000                          | 64/64                                       | 137                               | 54,500(M)                         | 27/19(N)                                      | 11(A)                                | 6/03                                             | 8/05                                             | 10/01                                            |
| Assembled in Containers(Y,Z)                                          |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| 500-Ib HK 94 bomb (wt 440 Ib)                                         | 981,000                            | 109/75                                      | 186                               | 35,800(M)                         | 30/20(0)                                      | 14                                   |                                                  |                                                  |                                                  |
| Containerized weapons - 1/container -<br>(wt 530 lb)                  |                                    |                                             |                                   |                                   |                                               |                                      | 4/03(K)                                          | 16/11(K)                                         | 23/16(K)                                         |
| MX-116 bomb (wt 562 1b)                                               | 224,000                            | ££/87                                       | 53                                | 8,160(L)                          | (0)6/E1                                       | 3                                    |                                                  |                                                  |                                                  |
| Containerized weapons - 1/container -<br>(wt 851 lb)                  | 486,850                            | 55/38                                       | 47 ( AA )                         | 14,900(L)                         | 22/15(ų)                                      | 8( <b>AA</b> )                       | 3/02(R)                                          | 8/06(R)                                          | 10/07(R)                                         |

tina nama nagan sa sa

NAM BOOGON DAVATON ANGONA DAVATON BOOTAN DEGRAM AGEORY AGEORY AGEORY AGEORY AGEORY AGEORY AGEORY AGEORY AGEORY

TABLE P-2 (Continued)

| Munition Type                                          | Axial<br>Crush<br>Force(A)<br>(1b) | Axial<br>Impact<br>Velocity(B)<br>(fps/mph) | Axial<br>Impact<br>Height<br>(ft) | Side<br>Crush<br>Porce(C)<br>(1b) | Side<br>Impact<br>Velocity(D,AE)<br>(fps/mph) | Side<br>Impact<br>Height(AE)<br>(ft) | Forklift<br>Puncture<br>Velocity(E)<br>(fps/mph) | Road ACC<br>Puncture<br>Velocity(F)<br>(fps/mph) | Rail ACC<br>Puncture<br>Velocity(G)<br>(fps/mph) |
|--------------------------------------------------------|------------------------------------|---------------------------------------------|-----------------------------------|-----------------------------------|-----------------------------------------------|--------------------------------------|--------------------------------------------------|--------------------------------------------------|--------------------------------------------------|
| deapons Containing Agent, Fuze and Burster             |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| Unassembled(2)                                         |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| H23 mine (wt 23 1b)                                    | 513,513                            | 131/89                                      | 266                               | 111                               | 60/61                                         |                                      | (M)                                              | (M)                                              | (M)                                              |
| Containerized weapons - 3/drum -<br>(wr 115 lb)        | 88,593                             | 64/44                                       | 64                                | 18,000(AG)                        | 30/20(q)                                      | 14( <b>AA</b> )                      | ( <b>n</b> )10/1                                 | 2/01(V)                                          | 3/02(V)                                          |
| Palletized drums - 12/pallet -<br>(wt 1,337)           | 531,558                            | 46/31                                       | 33(AA)                            | 72,000(AG)                        | 18/12(Q)                                      | 5( <b>AA</b> )                       |                                                  |                                                  |                                                  |
| Meapons Containing Agent - in<br>Containers(Y,Z)       |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| THU-28 spray cank (1,935 lb)                           |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| Containerized weapons - 1/container -<br>(wt 6,000 lb) | 390°000                            | 55/38                                       | 47                                | 900 <sup>+</sup> 65               | 25/17(T)                                      | ( <b>vv</b> )01                      | 10/01(U)                                         | 10/07(U)                                         | 12/08(U)                                         |
| Shipping Containers - Agent(2)                         |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| Type E (wt ~],000 lb)                                  | 969,000                            | 42/28                                       | 27                                | 11,900                            | 9/6(N)                                        | 1                                    | 2/01                                             | 2/02                                             | 3/02                                             |
| Type A, D (wt ~3,000 lb)                               | 1,510,000                          | 52/35                                       | 42                                | 54,000(AG)                        | 14/10(N)                                      | 3( AG )                              | 3/02                                             | 60/4                                             | ٤٥/٤                                             |
| Overpacked container (wr ~9,000 lb)                    | 2,014,560                          | 35/24                                       | 19                                | 12,780(AB)                        | 7/5(AC)                                       | -                                    | 7/05( <b>AD</b> )                                | 5/03(AD)                                         | (TV) 70/L                                        |
| Weapons Containing Agent                               |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |
| 105- <b>mm</b> M60/M360 projectile (wt 32 lb)          | 279,000                            | 216/147                                     | 726                               | 61,200                            | 101/69                                        | 158                                  | 4/03                                             | 14/09                                            | 20/14                                            |
| Pailetized weapons = 24/pallet =<br>(wt 799 lb)        | 6,690,000                          | 212/145                                     | 698                               | 245,000                           | 41/28                                         | 26(AA)                               |                                                  |                                                  |                                                  |
|                                                        |                                    |                                             |                                   |                                   |                                               |                                      |                                                  |                                                  |                                                  |

Unless otherwise noted, the calculational model is a simple pipe crushed by an axial force bearing on the pipe annulus.

Both the weapon and the pallet or container are assumed to free fall No credit is given for shock absorbing effects of pallet or packing materials. to failure. × æ

Unless otherwise noted, the calculational model is a simple pipe with closed ends: the crush force is distributed across the crest of one side with a reaction force at each end creating a bending moment. The weakest side of the pallet was chosen for crush force application where one pallet side was larger than another. ់

No credit is given for shock absorbing effects of pailet or packing materials. Both the weapon and the pailet or container are assumed to free fall to failure. The pailet is assumed to land on its weakest side. .

A forklift with a mass of 5000 lb requires this velocity to puncture the weapon with a tine. No shock absorbing effects or pallet sliding considered. . س





PROCESS PROPERTY

122222









### TABLE F-2 (Continued)

- Inside a truck body, a metal rod 1-1/2 in. diameter requires this velocity to be driven into the weapon with the weight of one pallet behind it. .
- Inside a railcar, a metal rod 3 in. diameter requires this velocity to be driven into the weapon with the weight of one pailet behind it. త
- The MSS is close fitted into a shipping/launch tube which is included in the models. The rocket weighs 57 lb; 10 lb are added for the S/L tube. ÷
- I. The force required to crush the 40 in. of warhead is used.
- The force required to crush the saddle supports into the warhead portion of the rockets is used. ;
- The wooden bex or aluminum container are assumed to provide negligible protection to the munition. 3
- L. The crush force is assumed to bear on only 2/3 of the side of the munition.
- M. The crush force is assumed to bear on only 1/2 of the side of the munition.
- N. A slde wall deformation of 4 in. is assumed to be required for failure.
- 0. A side wall deformation of 2 in. is assumed to be required for failure.
- P. The aluminum container is assumed to provide no protection from crush, impact, puncture.
- Q. A side wall deformation of the container of 5 in. is assumed to be required for failure.
- R. The puncture probes must travel 4 in. into the container for failure.
- S. The TMU-28 well is assumed to be much weaker than the container wall.
- T. A side wall deformation of the container of 12 in. is assumed to be required for failure.
- U. The puncture probes must travel 20 in. into the container for failure.
- V. The puncture probes must travel 3 in. into the container for failure.
- The puncture force required to puncture the container is sufficient to puncture the munition also. ÷
- Some assembled cartridge versions of the M60/M360 are assumed to have been cannibalized for fuzes and propellent. 2
- Y. Fuzes or explosive cutters are assumed to have been removed.
- 2. Unless otherwise noted impact and puncture deformations of 1 in. cause failure.
- These heights represent perfect impact into a totally massive surface with no energy absorption by the weapon packaging. It is assumed for veloci-ties below about 50 fps (39 ft equivalent height), energy absorbtion by weapon packaging and the surface will dominate the energy of the collision. In these cases test data are of more value. ¥.
- AB. This represents a crush force applied over 16 in. of the thin wall portion of the container.
- AC. A side wall defomation of 7 in. is assumed to be required for failure.
- AD. The puncture probes must travel 5 in. into the container for failure.
- calculated impact failure threshold data listed were superceded by test data. Therefore, the calculated values are shown for reference only. The AE.
- A multi-dimensional crush and impact analysis and a sophisticated puncture calculation were performed only for the M55 pallet. AF.

AG. Based on test data.

TABLE F-3 TEST DATA CORRESPONDING TO EFFECTS OF IMPACT

1 4.4 4.4 A.4 4.4 4.4 8.4 A.4 A.4 A.4 A.4 A.4 A.4 A.4 A.4 A.4

| Munitions<br>nfigurations | Number<br>Tested | Components<br>Present                 | Test<br>Dates       | Test<br>Procedure                                                                                                      | Test<br>Results                                                                          |
|---------------------------|------------------|---------------------------------------|---------------------|------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------|
| idges                     |                  | Z                                     | lo test da          | ta available for this report                                                                                           | IJ                                                                                       |
| sctiles                   |                  |                                       |                     |                                                                                                                        |                                                                                          |
| )-mm<br>) packaging)      | 1                | Burster                               | 1975(a)             | <ul> <li>25 10-ft drops on steel</li> <li>plate and concrete for:</li> <li>Base</li> <li>Side</li> <li>Nose</li> </ul> | No failures for base/<br>side drops, agent com-<br>partment failure in<br>lith nose drop |
|                           |                  |                                       | 1061 ( <b>8</b> )   | 40-ft nose drons on steel                                                                                              | No failures recorded                                                                     |
| n.<br>, nerkeetne)        | 30               | burster                               | TOAT                | plate and concrete                                                                                                     |                                                                                          |
| parcas 1116/              | 30               | Burster                               |                     | 40-ft side drops on steel<br>plate and concrete                                                                        |                                                                                          |
|                           | 30               | Burster                               |                     | 40-ft base drops on steel<br>plate and concrete                                                                        |                                                                                          |
|                           |                  |                                       |                     |                                                                                                                        |                                                                                          |
| 1                         | 2                | Burster                               | 1955(a)             | 6-ft side drop on steel                                                                                                | No failures recorded                                                                     |
| -Ib bomb<br>packaging)    | 7                | One with<br>burster only,<br>one with | 1955(a)             | off side drop on steel<br>plate and concrete                                                                           | No failures recorded                                                                     |
|                           | 7                | burster and<br>fuze<br>Fuze/booster   | 1971 <sup>(b)</sup> | Dropped from plane at<br>387-ft and 280-mph onto<br>concrete                                                           | Hit at 285-mph, bounced<br>88-ft high, no fail-<br>ures recorded                         |

F-46






1,920,542,624

TABLE F-3 (Continued)

| Munitions<br>Configurations  | Number<br>Tested       | <b>Components</b><br><b>Present</b> | Test<br>Dates | Test<br>Procedure                                                 | Test<br>Results                     |
|------------------------------|------------------------|-------------------------------------|---------------|-------------------------------------------------------------------|-------------------------------------|
| Containers                   |                        |                                     |               |                                                                   |                                     |
| TMU-28B                      | 7                      | Simulant                            | 1973(a)       | 10-ft side drop on                                                | No failures recorded                |
| spray tank<br>(no packaging) | 1                      | Simulant                            | 1966(a)       | concrete<br>10-ft side drop on                                    | No failures recorded                |
|                              | 1                      | Simulant                            | 1968(a)       | concrete<br>10-ft side drop on                                    | No failures recorded                |
| Type D ton                   | 1                      | Simulant                            | 1964(c)       | concrete<br>40-ft drop on steel (end)                             | Leakage                             |
| container<br>(an anaborian)  |                        |                                     |               | 40-ft drop on steel                                               | Major leakage<br>Major leakage      |
| (IIO packagilig)             | -                      |                                     |               | (JJ-deg angle)<br>40-ft drop on steel (side)                      |                                     |
|                              | -                      |                                     |               | 6-ft drop on concrete<br>(end, corner, side)                      | No leakage                          |
| Mines                        |                        |                                     |               |                                                                   |                                     |
| M23 mine only                | 3<br>(8 drops<br>each) | No burster                          | 1958(d)       | 6-ft drop on steel plate<br>and concrete (side,<br>top, end edge) | Visible leak on last of<br>24 drops |
| M23 mine only<br>(prototype) | 5<br>(2 drops<br>each) | Side and<br>central<br>burster      | 1960(d)       | 6-ft drop on steel plate<br>and concrete                          | No leaks                            |
| M23 mine only<br>(prototype) | 30                     | Inert<br>bursters                   | (þ)0961       | 3-ft drop on gravel road<br>at 30 mph                             | One mine had trace leak             |

agentfilled

F-47

. . .

<u>,</u>

| st<br>ults                       | id small                                         |         | eets dis-<br>le firing<br>ced and<br>arhead bent<br>ould not be<br>agent leaked<br>ant funct-           | royed"<br>s all<br>[4 of 15<br>maged, no<br>cage or<br>: function.                 | \$ |
|----------------------------------|--------------------------------------------------|---------|---------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------|----|
| Te<br>Res                        | Two mines ha<br>leaks                            |         | Both end she<br>lodged, or<br>tube crack<br>enclosed w<br>so that co<br>fired, no<br>or propell<br>ion. | Pallet "dest<br>scattering<br>rockets, l<br>rockets de<br>agent leak<br>propellant |    |
| (Continued)<br>Test<br>Procedure | 4-ft drop on concrete<br>slab                    |         | 40-ft drop onto concrete<br>(nose 30 deg below hori-<br>zontal)                                         | 40-ft accidental drop<br>onto steel cargo deck,<br>nose down                       | Č  |
| ABLE F-3 (<br>Test<br>Dates      | 1960 <sup>(d)</sup>                              |         | 1964(e)                                                                                                 | N/A(a)                                                                             |    |
| T<br>Components<br>Present       | Side and<br>central<br>bursters<br>M120 bursters |         | Agent simu-<br>lant                                                                                     | 114                                                                                |    |
| Number<br>Tested                 | 41                                               |         | -                                                                                                       | -                                                                                  |    |
| Munitions<br>Configurations      | M23 mine only<br>(prototype)                     | Rockets | Complete pallet                                                                                         | Complete pallet                                                                    | ~  |
|                                  |                                                  |         | F-48                                                                                                    |                                                                                    |    |







A BOOM AND A DOOL ON THE SALES





| Munitions<br>Configurations | Number<br>Tested | Components<br>Present | Test<br>Dates | Test<br>Procedure                              | Test<br>Results                         |
|-----------------------------|------------------|-----------------------|---------------|------------------------------------------------|-----------------------------------------|
| Weteye Bomb                 |                  |                       |               |                                                |                                         |
| Bomb in ship-               | 5                | Agent simu-           | 1965(f)       | 40-ft drop onto steel                          | For both end and side                   |
| ping container              |                  | lant                  |               | plate embedded in con-<br>crete (and and elde) | drop, the shipping<br>container seal no |
|                             |                  |                       |               |                                                | longer effective, bomb                  |
|                             |                  |                       |               |                                                | nose bent, no agent                     |
|                             |                  |                       |               |                                                | leakage.                                |
|                             |                  |                       |               |                                                |                                         |
| ( <sup>a)</sup> Reference F | -7.              |                       |               |                                                |                                         |
| (b)Reference F              | -8.              |                       |               |                                                |                                         |
| (c)Reference F              | -9.              |                       |               |                                                |                                         |
| (d)Reference F              | -10.             |                       |               |                                                |                                         |

LEAD FIRE

F-49

(e)Reference F-11. (f)Reference F-12.

A A A A A A A

^

point. The end (axial) and side impact forces on single and palletized munitions were originally determined analytically and whenever possible, supported with test data. Sufficient drop test information was available on the M55 rocket pallets (Table F-3) to determine that simple analyses were not adequate; therefore, multidimensional, nonlinear analyses were conducted (Ref. F-5). This resulted in defining the impact failure threshold as a 40-ft drop height for M55 rocket pallets rather than 3-ft as calculated by simple analysis shown in Table F-2. Therefore, the calculated impact failure thresholds for the other munitions were also considered to be overly conservative, and additional tests were performed at DPG to better define the impact failure threshold for the various munitions. These are discussed in a subsequent section.

The puncture threshold is defined in terms of a ratio of velocity to radius of curvature of the puncture object assuming that the munition (or the pallet) impacts an unyielding slender object. If there is more than one protective barrier, (e.g., mines packaged in drums), the threshold is the velocity required to puncture all the barriers. The puncture failure threshold was determined by calculating the force required to cause material failure with a slender object. During handling operations, munition puncture failures will most likely be caused by forklift tines. The puncture velocity was calculated based on a typical 5000-1b forklift. The munitions are assumed to be in their stored or shipped configuration, as appropriate. Wooden and aluminum containers are assumed to provide no protection to a probe. Some material deformation is also assumed and is consistent with the assumptions made for crush failure threshold calculations. Based on the SNL data base, the calculated truck accident puncture velocity assumed a 3/4-in. radius probe, while the railroad accident puncture velocity assumed a 1.5-in. radius probe. These probe sizes are considered the most probable for truck and rail accidents. In each case, the most likely object capable of acting as a probe was considered to be a trailer/railcar coupler.

F-50

### F.1.5. ADDITIONAL TEST DATA FOR MECHANICAL FAILURE (IMPACT) THRESHOLDS

In the risk analysis, the objective is to determine the probability that a munition will fail and release agent to the environment. Early Army tests, however, were designed to verify that properly packaged muitions would withstand certain guideline loads rather than to determine the point at which a failure would occur. A summary of various impact tests on chemical munitions is given in Table F-3. The results in Table F-3 indicate that the calculated failure thresholds for impact shown in Table F-2 are too conservative. The one-dimensional mechanical calculations appear to be reasonable for puncture and crush failure, but the modeling is not sufficiently sophisticated to consider the impact energy absorption of the wood, cardboard, and styrofoam protective packaging or the load spreading capability of the shipping configuration. (Multidimensional calculations were performed only on the M55 rocket.) To determine the impact failure thresholds of munitions more accurately, tests were conducted in July 1986 on mines, ton containers, cartridges, and projectiles at DPG. The test results are summarized in Table F-4 and are discussed below. All munitions contained the appropriate quantity of agent simulant. All drops were onto a 10- x 10- x 1-ft concrete slab reinforced with standard bar and angle strips of steel. For some tests, the pad also had a special hard concrete surface.

Two drop tests were conducted with 30-gal drums, each containing three M23 mines. The first drop was from a height of 60 ft such that the side of the drum impacted the cement; substantial leakage of the simulated agent resulted. For a second drop, at 45 ft and in the side orientation, no failures occurred. Figure F-36 shows the three mines after they had been removed from the drum dropped 45 ft. Note that the side of each mine was deformed.

Five ton containers were dropped in eight tests. The first ton container was dropped from three heights, (15, 30, and 40 ft) in a side orientation. After the first test, the ton container was rotated

F-51

| <ul> <li>2 Mine drums containing Non-burstered</li> <li>1 45 ft drop, side orientation</li> <li>1 60 ft drop side orientation</li> <li>6 Ton containers, no</li> <li>8 Simulant filled</li> <li>9 of 1 15, 30, and 40 ft, side</li> <li>overpack</li> <li>3 40 ft drop, side orientation</li> <li>4 0 ft drop, side orientation</li> <li>2 40 ft drop, 1 pallet, side</li> <li>105-mm projectiles</li> <li>8 Non-burstered</li> <li>1 60 ft drop, 1 pallet, side</li> <li>1 60 ft drop, 1 pallet, impact</li> <li>2 Pallets of six 155-mm Non-burstered</li> <li>2 Pallets of six 155-mm Non-burstered</li> <li>2 Pallets of 44.2 in. Non-burstered</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | Munitions Configuration                         | Components<br>Present            | No. of<br>Tests | Test Description                                            | Test Results                                                       |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------|----------------------------------|-----------------|-------------------------------------------------------------|--------------------------------------------------------------------|
| 1       60 ft drop side orientation         6 Ton containers, no       Simulant filled       3 of 1       15, 30, and 40 ft, side         overpack       3       40 ft drop, side orientation         3       40 ft drop, side orientation         2       40 ft drop, side orientation         6 Pallets of 15 M360       Non-burstered       1       60 ft drop, 1 pallet, side         105-mm projectiles       simulant filled       1       60 ft drop, 1 pallet, side         105-mm projectiles       simulant filled       1       60 ft drop, 1 pallet, impact         each       1       60 ft drop, 1 pallet, impact       1         9       6 ft drop, 1 pallet, impact       1       60 ft drop, 1 pallet, impact         9       6 ft drop, 1 pallet, impact       0       60 ft drop, 1 pallet, impact         1       60 ft drop, 1 pallet, impact       0       60 ft drop, 1 pallet, impact         2       Pallets of six 155-mm       Non-burstered       1       60 ft drop, 1 pallet, impact         2       Pallets of six 155-mm       Non-burstered       1       60 ft drop, 1 pallet, impact         2       Pallets of 48 4.2 in.       Non-burstered       1       60 ft drop, 1 pallet, impact         2       Pallets of 48 4.2 in.       Non-burstered                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 2 Mine drums containing<br>three M23 mines each | Non-burstered<br>simulant filled | 1               | 45 ft drop, side orientation                                | Deformation in side of<br>each mine, no leaks.                     |
| <ul> <li>6 Ton containers, no</li> <li>6 Ton containers, no</li> <li>8 Simulant filled</li> <li>9 of 1 15, 30, and 40 ft, side</li> <li>9 orientation</li> <li>3 40 ft drop, side orientation</li> <li>2 40 ft drop, 1 pallet, side</li> <li>105-mm projectiles</li> <li>1 60 ft drop, 1 pallet, side</li> <li>1 60 ft drop, 1 pallet, impact</li> <li>2 Pallets of six 155-mm</li> <li>1 60 ft drop, 1 pallet, impact</li> <li>2 Pallets of six 155-mm</li> <li>1 60 ft drop, 1 pallet, impact</li> <li>2 Pallets of six 155-mm</li> <li>1 60 ft drop, 1 pallet, impact</li> <li>2 Pallets of 4,2 in. Non-burstered</li> <li>1 60 ft drop, 1 pallet, impact</li> <li>2 Pallets of 48 4.2 in. Non-burstered</li> <li>3 60 ft drop, 1 pallet, impact</li> <li>4 60 ft drop, 1 pallet, impact</li> <li>4 6 6 ft drop, 1 pallet, impact</li> <li>4 6 6 ft drop, 1 pallet, impact</li> <li>5 7 6 ft drop, 1 pallet, impact</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                 |                                  | 1               | 60 ft drop side orientation                                 | Major leak                                                         |
| 340 ft drop, side orientation240 ft drop, side orientation240 ft drop, i pallet, side105-mm projectilessimulant filled105-mm projectilessimulant filled105-mm projectilessimulant filled160 ft drop, 1 pallet, impact160 ft drop, 1 pallet, impact2Pallets of six 155-mm2Pallets of six 155-mm2Pallets of 44rop, 1 pallet, impact2Pallets of six 155-mm2Pallets of 44rop, 1 pallet, impact2Pallets of 44rop, 1 pallet, impact2Pallets of 44rop, 1 pallet, impact2Pallets of 44rop, 1 pallet, impact160 ft drop, 1 pallet, impact2Pallets of 44rop, 1 pallet, impact2Pallets of 48 4.2 in. Non-burstered160 ft drop, 1 pallet, impact160 ft drop, 1 pallet, impact1on corner of nose end1on corn                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 6 Ton containers, no<br>overpack                | Simulant filled                  | 3 of 1          | 15, 30, and 40 ft, side<br>orientation                      | Leak from 4 in. crack<br>after the third drop.                     |
| 240 ft drop at 45 deg angle6 Pallets of 15 M360Non-burstered160 ft drop, 1 pallet, side105-mm projectilessimulant filled0ft drop, 2 pallets bandedeach160 ft drop, 1 pallet, impacteach160 ft drop, 1 pallet, impact160 ft drop, 1 pallet, impact160 ft drop, 1 pallet, impact160 ft drop, 1 pallet, impact2 Pallets of six 155-mmNon-burstered12 Pallets of six 155-mmNon-burstered12 Pallets of 48 4.2 in.Non-burstered12 Pallets of 4.2 in.Non-burstered12 Pallets of 48 4.2 in.Non-burstered12 Pallets of 4.2 in.Non-burstered12 Pallets of 48 4.2 in.Non-burstered12 Pallets of 60 ft drop, 1 pallet, impact2 Pallets of 8 4.2 in.Non-burstered140 ft drop, 1 pallet, impact40 ft drop, 1 pallet, impact41 mortars each140 ft drop, 1 pallet, impact41 mortars each140 ft drop, 1 pallet, impact40                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                 |                                  | e               | 40 ft drop, side orientation                                | No leaks                                                           |
| <ul> <li>6 Pailets of 15 M360 Non-burstered 1 60 ft drop, 1 pallet, side orientation 105-mm projectiles simulant filled orientation</li> <li>each each 1 60 ft drop, 2 pallets banded together, side orientation</li> <li>1 60 ft drop, 1 pallet, impact on nose edge of projectiles</li> <li>2 Pallets of six 155-mm Non-burstered 1 60 ft drop, 1 pallet, impact on corner of nose end projectiles each simulant filled 1 60 ft drop, 1 pallet, impact on corner of nose end more projectiles each simulant filled 2 Pallets of 48 4.2 in. Non-burstered 1 60 ft drop, 1 pallet, impact on corner sech simulant filled 1 60 ft drop, 1 pallet, impact on corner sech simulant filled 2 Pallets of simulant filled 1 60 ft drop, 1 pallet, impact on corner sech simulant filled 1 60 ft drop, 1 pallet, impact on corner sech simulant filled 1 60 ft drop, 1 pallet, impact on corner sech simulant filled 1 60 ft drop, 1 pallet, impact ot mortars each simulant filled 1 60 ft drop, 1 pallet, impact ot mortars each simulant filled 1 60 ft drop, 1 pallet, impact ot mortars each simulant filled 1 60 ft drop, 1 pallet, impact ot mortars each simulant filled 1 60 ft drop, 1 pallet, impact ot mortars each simulant filled 1 60 ft drop, 1 pallet, impact ot mortars each simulant filled 1 60 ft drop, 1 pallet, impact ot mortars each simulant filled 1 60 ft drop, 1 pallet, impact ot mortars each simulant filled 1 60 ft drop, 1 pallet, impact 1 60 ft drop, 1 pallet, 1 60 ft drop, 1 pallet, 1 60 ft drop, 1 pallet, 1 f</li></ul> |                                                 |                                  | 2               | 40 ft drop at 45 deg angle                                  | No leaks                                                           |
| each160 ft drop, 2 pallets banded<br>together, side orientation160 ft drop, 1 pallet, impact<br>along edge160 ft drop, 1 pallet, impact<br>on nose ends of projectiles2Pallets of six 155-mm2Pallets of six 155-mm160 ft drop, 1 pallet, impact<br>on corner of nose ends2Pallets of six 155-mm2Pallets of six 155-mm2Pallets of six 155-mm3Simulant filled40560 ft drop, 1 pallet, impact<br>on corner of nose end<br>along edge2Pallets of 48 4.2 in. Non-burstered<br>mortars each2Pallets of 48 4.2 in. Simulant filled<br>mortars each2Pallets of 48 4.2 in. Non-burstered<br>mortars each                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 6 Pallets of 15 M360<br>105-mm projectiles      | Non-burstered<br>simulant filled | 1               | 60 ft drop, l pallet, side<br>orientation                   | No leakage from any of<br>the tests. Typically,                    |
| 160 ft drop, 1 pallet, impact<br>along edge160 ft drop, 1 pallet, impact160 ft drop, 1 pallet, impact<br>on nose ends of projectiles2Pallets of six 155-mm2Pallets of six 155-mm160 ft drop, 1 pallet, impact<br>on corner of nose end2Pallets of six 155-mm2Pallets of six 155-mm2Pallets of six 155-mm160 ft drop, 1 pallet, impact<br>along edge2Pallets of six 155-mm2Pallets of function2Pallets of 48 4.2 in.160 ft drop, 1 pallet, impact<br>on corner of nose end<br>on corner of nose end2Pallets of 48 4.2 in.160 ft drop, 1 pallet, impact<br>on corner of nose end2Pallets of 48 4.2 in.160 ft drop, 1 pallet, impact<br>on corner of nose end2Pallets of 48 4.2 in.160 ft drop, 1 pallet, impact<br>on corner of nose end160 ft drop, 1 pallet, impact160 ft drop, 1 pallet, impact150 ft drop, 1 pallet, impact                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | each                                            |                                  | ı               | 60 ft drop, 2 pallets banded<br>together, side orientation  | pallet was shattered, but<br>projectiles showed little<br>effects. |
| <ul> <li>1 60 ft drop, 1 pallet, impact<br/>on nose ends of projectiles</li> <li>2 Pallets of six 155-mm Non-burstered</li> <li>1 60 ft drop, 1 pallet, impact<br/>on corner of nose end</li> <li>2 Pallets of six 155-mm Non-burstered</li> <li>1 60 ft drop, 1 pallet, impact</li> <li>2 Pallets of 48 4.2 in. Non-burstered</li> <li>2 Pallets of 48 4.2 in. Non-burstered</li> <li>3 60 ft drop, 1 pallet, impact</li> <li>2 Pallets of 48 4.2 in. Non-burstered</li> <li>3 60 ft drop, 1 pallet, impact</li> <li>3 7 60 ft drop, 1 pallet, impact</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                 |                                  | 1               | 60 ft drop, l pallet, impact<br>along edge                  |                                                                    |
| <ul> <li>2 Pallets of six 155-mm Non-burstered 1 60 ft drop, 1 pallet, impact on corner of nose end projectiles each simulant filled along edge</li> <li>2 Pallets of 48 4.2 in. Non-burstered 1 60 ft drop, 1 pallet, impact on corner of nose end nortars each simulant filled 1 60 ft drop, 1 pallet, impact</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                 |                                  | 1               | 60 ft drop, l pallet, impact<br>on nose ends of projectiles |                                                                    |
| <pre>2 Pallets of six 155-mm Non-burstered 1 60 ft drop, 1 pallet, impact<br/>projectiles each simulant filled along edge<br/>1 60 ft drop, 1 pailet, impact<br/>on corner of nose end<br/>mortars each simulant filled 1 60 ft drop, 1 pallet, impact</pre>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                 |                                  | 1               | 60 ft drop, l pallet, impact<br>on corner of nose end       |                                                                    |
| <pre>1 60 ft drop, 1 pailet, impact<br/>2 Pallets of 48 4.2 in. Non-burstered 1 60 ft drop, 1 pallet, impact<br/>mortars each simulant filled along edge</pre>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 2 Pallets of six 155-mm<br>projectiles each     | Non-burstered<br>simulant filled | 1               | 60 ft drop, l pallet, impact<br>along edge                  | No leaks. Projectiles<br>showed only slight                        |
| <pre>2 Pallets of 48 4.2 in. Non-burstered 1 60 ft drop, 1 pallet, impact<br/>mortars each simulant filled along edge</pre>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                 |                                  | 1               | 60 ft drop, l pailet, impact<br>on corner of nose end       | effects.                                                           |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 2 Pallets of 48 4.2 in.<br>mortars each         | Non-burstered<br>simulant filled | 1               | 60 ft drop, l pallet, impact<br>along edge                  | No leaks. Boxed gen-<br>erally broken, but no                      |
| I bU IT Grop, 1 PALLET, 1MPACT<br>on corner of nose end                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                 |                                  | 1               | 60 ft drop, l pallet, impact<br>on corner of nose end       | evidence of munition<br>damage.                                    |

TABLE F-4 JULY 1986 DROP TESTS

CARLONNAL C

F-52



Fig. F-36. Land mines after 45-ft side drop in a 30-gal drum

about 90 deg, and after the second test, the cylinder was rotated about 45 deg. The first two tests produced flat spots along the length of the cylinder. On the third drop of the first container, leakage of the agent simulant was observed from a 4-in. long crack on the inside of the protective cylindrical apron in the vicinity of the head weld. Since the crack appeared to emanate from the flat spot created from a prior drop, it was postulated that the failure occurred because of the multiple drops experienced by the cylinder. The postulate was confirmed by three more drops of three separate cylinders, all in a side orientation from 40 ft: no leakage occurred. Figure F-37 shows the flat spot, about 6-in. wide, created by a typical 40-ft drop in a side orientation. Two additional cylinders were dropped from 40 ft, but at a 45 deg angle. The protective apron was bent but no leakage occurred. Figure F-38 shows the deformed apron.

Two pallets of 4.2-in. mortars were dropped from 60 ft, the highest drop height possible with the crane that was used. The orientation of the first drop was such that the edge of the pallet, along the length of the munition, initially impacted the cement. No deformation of the munition itself occurred and no leakage was observed, although most of the wooden boxes were broken open and some of the cardboard tubes were damaged. The munitions were removed (at least partially) from the four cardboard tubes that were the most damaged and stacked in the midst of the undisturbed remnants of the pallet (Fig. F-39). In a second test from 60 ft, the pallet was oriented so that the corner (with the nose of the munition) initially struck the cement. Similar damage to the pallet dunnage occurred, but the munition itself was undamaged.

Six pallets of M360 105-mm projectiles were dropped in five tests, all from 60 ft: (1) a single pallet oriented to strike the side containing the fewest munitions (three); (2) two pallets banded together and oriented to strike the side containing the fewest munitions; (3) a single pallet oriented so that the pallet edge along the length of the munition would initially impact the cement; (4) a single pallet oriented

TANK AN AND AND AND A

| ľ        | AD-A19 | 93 355<br>SIFIE | CHE<br>THE<br>CA<br>D SAP | MICAL<br>DISPO<br>A N E<br>E0-CDE | STOCK<br>SAL O<br>ARSELI<br>- I S-8 | PILE D<br>F CHEM<br>ET A<br>7006 D | ISPOSF<br>(U)<br>L. AUG<br>AAA15- | IL PRO<br>GA TE<br>87 Gi<br>85-D-( | GRAM R<br>Chnolo<br>R-C-19<br>B022 | ISK AN<br>GIES 1<br>563 | IALYSI<br>INC SAI<br>F/G : | 5 OF<br>N DIEG<br>15/6.3 | 0 11/<br>NL | 13 |
|----------|--------|-----------------|---------------------------|-----------------------------------|-------------------------------------|------------------------------------|-----------------------------------|------------------------------------|------------------------------------|-------------------------|----------------------------|--------------------------|-------------|----|
|          |        |                 | حزم                       |                                   | 202                                 |                                    | 411-                              | 1                                  |                                    |                         |                            |                          |             |    |
|          |        |                 |                           |                                   |                                     |                                    |                                   |                                    |                                    |                         |                            |                          |             |    |
|          |        |                 |                           |                                   |                                     |                                    |                                   |                                    |                                    |                         |                            |                          |             |    |
|          |        |                 |                           |                                   |                                     |                                    |                                   |                                    |                                    |                         |                            |                          |             |    |
|          |        |                 |                           |                                   |                                     |                                    |                                   |                                    |                                    |                         |                            |                          |             |    |
|          |        |                 |                           |                                   |                                     |                                    |                                   |                                    |                                    |                         |                            |                          |             |    |
|          |        |                 |                           |                                   |                                     |                                    |                                   |                                    |                                    |                         |                            |                          |             |    |
| $\Delta$ | ·      |                 | _                         |                                   |                                     |                                    |                                   |                                    |                                    |                         | _                          |                          |             |    |







Fig. F-38. Ton container after a 40-ft drop at a 45-deg angle

K

૽ૺ૱ૼ૱ૡૻ૱ૡૻ૱ૼ૱ૻ૱ૼૼ૱





Fig. F-39. 4.2-in. mortar shells after a 60-ft drop from a pallet

so that the 15 nose ends initially impact the cement; and (5) a single pallet oriented so that the corner of the pallet containing the nose of a munition would initially impact the cement. The last test produced the most damage to the munition, but no leakage occurred. Figure F-40 shows that the worst damage was a slightly deformed nose end.

Two pallets of 155-mm projectiles were dropped from 60 ft. One was oriented so that the edge of the pallet along the munition length impacted first and the other so that the corner of the pallet with the projectile nose initially impacted. The munitions generally were undamaged except for the paint and some bruising of the brass rotating band. For the corner drop, the nose ring of the munition in the corner was broken as shown in Fig. F-41.

## F.1.5.1. Basis for Selection of Impact Failure Thresholds

The drop test data clearly demonstrated that the calculated failure thresholds are extremely conservative. The drop tests were able to provide a more realistic estimate of the impact failure threshold for rockets, mines, and ton containers. However, the tests were limited to a drop height of 60 ft and no failures or severe damage were observed for cartridges and projectiles. Thus, the actual failure thresholds for these "stronger" munitions could not be established directly from tests. For these munitions, and also bombs and spray tanks, the impact failure threshold was inferred by scaling analytical results using scaling factors obtained from test data on similar munitions.

<u>Rocket</u>. Two rocket pallet drops have occurred from a height of 40 ft (Table F-3); neither produced failure, although in one case the nose of one rocket was severely bent, indicating that the failure threshold for the worst orientation may not be much higher. In addition, conservative calculations indicated failure at 40 ft. Thus, 40 ft was selected as the failure threshold.





TELES TELES





Fig. F-41. 155-mm projectile after a 60-ft drop from a pallet

<u>Mine</u>. Two tests with individual drums of mines resulted in agent containment failure at a drop height of 60 ft and no failure at 45 ft (Table F-4). The mine body deformation at the 45-ft drop height, however, indicated that other drops at 45-ft, or even slightly less, could produce failure. Thus, 45 ft was selected as the failure threshold. Due to the energy absorption capability of the styrofoam packaging, it was judged that the effect of palletizing the drums is negligible.

<u>Ton Container</u>. A prior test produced failure for a 40-ft side drop, and a 40-ft, 45-deg drop (Table F-3). The more recent tests produced no failures for three side drops and two 45-deg drops from 40 ft, using five different ton containers (Table F-4). Failure did occur in one ton container for a side drop from 40 ft after it had already been dropped from 15 and 30 ft. Thus, the failure threshold was selected as 40 ft. The analytical estimate was 3 ft. A scale factor of 13 is obtained between the analytical estimate and the test data.

<u>Bomb</u>. In two prior tests, an MC-1 750-1b bomb was dropped from a plane traveling at a height of 387 ft and a speed of 285 mph (Table F-3). The bomb impacted a concrete runway at an average terminal velocity of 283 mph; the height of the first bounce averaged 88 ft. No leakage of the agent simulant occurred. The impact orientation of the bomb was not given in the test report; however, the vertical component was estimated as 105 mph. The equivalent drop height corresponding to 105 mph is 368 ft. It was assumed that the effect on the bomb more closely resembled a pure axial load rather than a pure side load. The analytical estimate for an axial load was 148 ft (Table F-2).

The bomb is similar to a ton container, and hence the scaling factor of 13 obtained for the ton container for a side load will be used to estimate the failure threshold for a side load on the bomb. The analytical estimate for a side impact load was 25 ft (Table F-2). Hence, the failure threshold for the bomb can be estimated to be 325 ft (25 x 13).

F-61

<u>105-mm</u>, <u>155-mm</u>, <u>and 8-in. Projectiles</u>. Two types of projectiles (105 mm and 155 mm) were dropped from 60 ft (Table F-5) with no observed failures. The M110, 155-mm projectile has a calculated failure threshold of 24 ft (Table F-2). Thus, an apparent scaling factor of at least 60/24 = 2.3 exists between the calculated and experimental failure thresholds. Test limitations precluded dropping projectiles from heights greater than 60 ft; hence, a scaling factor was used to get a more realistic failure threshold. The scale factor of 13 obtained for the ton container was used to determine the failure threshold of projectiles. A failure threshold of 312 ft (24 x 13) was obtained for the 155 mm projectiles. The projectile representative munition is the M426, 8-in. projectile which also has a calculated failure threshold of 24-ft, but no tests were performed with 8-in. projectiles. Hence, the failure threshold for the 155 mm was used as an approximate failure threshold (312 ft) for the 8-in. projectile.

<u>4.2 in. Mortars</u>. Palletized cartridges were calculated to fail at a drop height of 5 ft (Table F-2). In the test, cartridges were dropped from a height of 60 ft (Table F-4), the maximum height permitted by test limitations. There were no deleterious effects on the munitions, only the dunnage was affected. If a scaling factor of 13 is used, an estimated drop height of 65 ft (13 x 5) is obtained. Since no damage occurred at 60 ft, a value of 65 ft is too low. This is partly due to conservative analytical estimate (5 ft) when energy absorption due to dunnage was omitted. The cartridge is weaker than the bomb or the projectile, but should have a failure threshold greater than 60 ft. Hence, in the absence of any other data a mean value (180 ft) between the projectile and test data of 60 ft will be used as an approximate failure threshold for the cartridges (312 + 60/2).

<u>Weteye Bomb</u>. Data reported in the Weteye Final Environmental Impact Statement (FEIS) indicate that the bomb in its shipping container did not fail but was severely damaged for drop tests from 40 ft for



|               | IN SHIPPING CONFIC                       | GURATION |                |
|---------------|------------------------------------------|----------|----------------|
| Munition      | Failure Threshold<br>Drop Height<br>(ft) | Basis    | Scaling Factor |
| Rocket        | 40                                       | (a), (b) |                |
| Mine          | 45                                       | (a)      |                |
| Ton container | 40                                       | (a)      |                |
| Bomb          | 325                                      | (c)      | 13             |
| Cartridge     | 180                                      | (a), (d) |                |
| Projectile    | 312                                      | (c)      | 13             |
| Weteye        | 40                                       | (a)      |                |
| Spray tank    | 50                                       | (c)      | 5              |

# TABLE F-5ESTIMATED IMPACT FAILURE THRESHOLD FOR MUNITIONSIN SHIPPING CONFIGURATION

(a)<sub>Test</sub> data.

(b)Analytical data.

(c)Scaled analytical data.

(d)Limited data available; mean of test data and projectile estimate.

either side or end orientation (Table F-3). The corresponding calculated side drop failure threshold was 8 ft (Table F-2). Thus, the test data show that the failure threshold is at least five times the calculated value.

Spray Tank. The analytical failure estimate for the spray tank was 10 ft (Table F-2). No tests were performed on the spray tank; however, the spray tank in its shipping container is similar to the Weteye bomb in its shipping container. Thus, the scaling factor obtained for the Weteye bomb was used to estimate the failure threshold for the spray tank. A failure threshold of 50 ft (10 x 5) was obtained for the spray tank.

#### F.1.6. REFERENCES

- F-1. "Transportation of Chemical Agents and Munitions, A Concept Plan," report dated June 30, 1987, prepared by the MITRE Corporation.
- F-2. Whitney, M. G., et al., "A Manual to Predict Blast and Fragment Loadings from Accidental Explosions of Chemical Munitions Inside an Explosion Containment Structure," Southwest Research Institute, Report No. SWRI-6714, April 1983.
- F-3. Dobratz, B. M., "Explosives Handbook, Properties of Chemical Explosive and Explosive Simulants," LLNL, March 16, 1981.
- F-4. Rhyne, W. R., Letter to Dr. Rick Bolig dated October 20, 1986.
- F-5. Rhyne, W. R., et al., "Probabilistic Analysis of Chemical Agent Release During Transport of M55 Rockets," H&R Technical Associates, Inc., M55-CD-4, H&R 255-1, September 1985.
- F-6. Rhyne, W. R., Letter to Dr. C. A. Bolig, March 31, 1986.
- F-7. Solomon, I., et al., "Hazards Associated with the Movement of Chemical Munitions and Containers from Dugway Proving Ground to Tooele Army Depot, South Area," Annex L to OPLAN DTS, May 1976.
- F-8. Taylor, W. T., et al., "Simulant Phase 1, Test of MC-1 Bomb, Volume I, Final Report," Test 69-14, Deseret Test Center, Fort Douglas, Utah.
- F-9. Final Second Supplement to FEIS, Transportation of Chemical Material, Operation RMT, USAMDRC, Alexandria, Virginia, 1981.
- F-10. U.S. Army Toxic and Hazardous Materials Agency, "Appendix A, Tests of Chemical Munitions," Handout, April 1984.
- F-11. "Hazard Classification Tests for Storage and Handling of GB- and VX-filled Chemical Ammunition, M55 and M23," U.S. Army Test and Evaluation Command, DPGTR-380, May 1964.
- F-12. Final Environmental Impact Statement, Transportation of Chemical Materiel, Operation RMT, USAMDRC, Alexandria, Virginia, 1977.

F-65



DOCOMPANY AND A VALLEY

# APPENDIX G DEMILITARIZATION ACTIVITIES



<u>ند ا</u>

#### G.1. DEMILITARIZATION ACTIVITIES

As noted in Section 3.1, the steps in the demilitarization process were grouped into five major activities for the risk analysis: storage, handling, onsite transportation by truck, offsite transportation by rail, and demilitarization operations. Each of these activities, as well as decommissioning, is discussed in detail in the sections which follow.

#### G.1.1. STORAGE

Safe storage of the chemical munitions is required up to the time they are processed in a demilitarization facility. It is assumed that the current storage arrangement will continue until a process facility or facilities are ready for operation. Large-scale movement of chemical munitions must take place within the constraints of the program schedule, plant operating schedules, logistical limitations of transport operations, availability of storage facilities at the NDC or RDC, and in compliance with safety and regulatory requirements of transport.

Munition movement sequences are generally planned to coincide with plant disposal sequences. Munitions would be moved as storage space is created by disposal plant operations. Ideally, movement would be accomplished in advance of disposal operations to ensure that stocks are in place and that plant operations are not delayed.

For the purposes of this risk analysis, it has been assumed that storage basically occurs at the original storage site. However, handling activities were accounted for at both the sending and receiving sites as though the munitions were placed into a similar storage facility at the receiving site. Storage of chemical munitions is governed by the general safety guidelines of AMC-R 385-100 (Ref. G-1). Specific regulations for the storage of GB and VX are given in DARCOM-R 385-102 (Ref. G-2), and in DARCOM-R 385-31 (Ref. G-3) for mustard types H, HD, and HT. In accordance with these regulations, it was assumed that the munitions are stored as follows:

- Magazines or structures used for the storage of agent-filled items are in specially designated areas. The structures have floors and floor surfacing that can be decontaminated.
- 2. Munitions that contain explosives are stored in igloo magazines. The igloos are spaced according to hazard class and the quantity of explosives that the igloo is permitted to hold.
- 3. Munitions and bulk containers containing GB or VX, but containing no explosives, are stored in igloo magazines except VX ton containers are in warehouses at NAAP, and VX spray tanks are stored in warehouses at TEAD.
- 4. Munitions containing mustard, but containing no explosives, are stored in igloos or other approved structures. Bulk containers containing mustard are also stored outdoors at APG, PBA, and TEAD. Mustard-filled bulk containers stored outdoors are secured on metal supports and positioned over crushed stone, gravel, or porous earth surfaces to minimize atmospheric contamination in the event of leakage.

5. Munitions in storage are packaged, stacked, and arranged in accordance with instructions set forth in Army regulations and approved AMC drawings and directives. The methods for stack-ing provide adequate ventilation. Aisles are maintained so

G-2



that units in each stack can be inspected, inventoried, and removed for shipment or surveillance tests.

- 6. The ends of ton containers are kept freshly painted and rustfree to enhance visual detection of agent leakage at valves and plugs. Shipping bonnets are removed from ton containers in storage to facilitate inspection for leakage. If a leaking container is found, the leak is repaired, or the contents are transferred into a new container.
- 7. Work performed in magazines and storage areas is limited to the types permitted in Chapter 18 (Storage of Explosives and Ammunition) of AMC-R 385-100.
- 8. Leaking munitions are encapsulated in specially provided containers until disposition is accomplished.

Three types of storage magazines are currently in use: igloo magazines (in 40-, 60-, and 80-ft lengths), 80-ft Stradley magazines, and 89-ft oval-arch magazines. While size and design details differ, they are all earth-covered, arched-roof structures designed to protect their contents from the blast and shrapnel effects of a potential detonation of a neighboring magazine. For this risk analysis, except as noted for specific accident scenarios, the structural characteristics of all the storage magazines are represented by the 80-ft igloo magazine. General design characteristics of the 80-ft igloo magazines are listed below (Ref. G-4):

1. The minimum compressive strength of the concrete used in igloo construction is 2500 psi.

5 - - 5 5

- 3. The minimum thickness of the exposed concrete front face of the igloo is 18 in.
- 4. The minimum thickness of the earth cover is 24 in. at the crown of the arch. The earth cover has a maximum slope of two horizontal units to one vertical unit and is stabilized by establishing a controlled vegetation cover such as grass, or by mechanical means appropriate to the local soil conditions and climate.

PASSOCIAL ASSOCIATION INCORPORT IN 2019-17

と言うとうないと言語でなっている。

- 5. The igloo is designed to prevent water ingress. Preventative measures include membrane waterproofing, a perforated drain system along foundation footings, interior floor slope and gutters, and slope of the concrete entry apron away from the front of the igloo.
- 6. Passive ventilation is provided in the form of louvered vents in the front concrete face of the igloo and a single ventilator stack penetrating the earthen cover at the rear of the igloo. The stack ventilator is designed to prevent backdrafts.

Fusible links are provided in the vents to close the ventilation path in the event of a fire.

7. Single or double doors, which open outward, are provided in the front face of the igloo. Double doors create an opening measuring 8 by 8 ft. A reinforced concrete "King Tut" block is provided in front of each door as a security device. The



block weighs approximately 5000 lb and rests on a post embedded in the concrete apron in front of each igloo; a forklift is required to remove the block from in front of the igloo door. In addition, the doors are padlocked shut with highsecurity locks.

- 8. A lightning protection system is provided.
- 9. No electric power system is permanently installed in the igloos; however, an electrical junction box is provided on the outside front face of each igloo.
- 10. No fire fighting system is installed in or near the igloos; however, depot fire fighting teams are located within a few minutes response time from the storage locations. In addition, all nonelectric vehicles are required to carry fire extinguishers when operating in or near the ammunition storage areas. Also, while personnel are operating in the igloos, one or more decon trucks carrying a large supply of water is kept on standby immediately outside the igloo. This water supply can be used for emergency fire fighting if required.

11. An intruder alert system is installed in all igloos.

Warehouses are in use at three sites to store bulk containers. The size and construction of the warehouses are different at each of the three sites. Descriptions of the warehouses are provided in the discussion of site-specific data in Appendix D.

Any munitions in open storage (mustard-filled ton containers) are stored in configurations specified in AMC drawings, but are otherwise unprotected from the elements.



Detailed information on pallet configurations is given in the Continued Storage Risk Analysis report (Ref. G-5).

#### G.1.1.1. Activities Associated with Storage

こうごう とんな いたんたたた いたち ちょうししょう

The activities associated with munition storage consist of surveillance and maintenance of the stored munitions, surveillance and maintenance of the storage facilities, and inventory of stored munitions. It is assumed that all surveillance will be accomplished in accordance with IAW SB 742-1300-94-1 (Ref. G-6). Three types of inspections are conducted; these are periodic inspections (PI), safety in storage inspections (SSI), and storage monitoring inspections (SMI).

Periodic inspections are cyclical inspections of the munitions for deterioration or nonstandard conditions. Periodic inspections are conducted at 2-yr intervals on all chemical munitions, unless conditions warrant more frequent inspection. (PI does not apply to munitions in demilitarization accounts.)

Safety in storage inspections are periodic inspections of unserviceable, nonrepairable munitions and munitions in demilitarization accounts, conducted to assure that the munitions are safe for continued storage, handling, and demilitarization. Visual inspections are supplemented by propellant stability testing. Lots that are considered potentially hazardous are inspected no less frequently than the intervals specified for PI. Lots determined to be nonhazardous may have their SSI intervals extended, but the extended interval may not exceed twice the PI interval.

Storage monitoring inspections are performed on chemical agent munitions, containers of bulk chemical agents, and containerized munitions specifically to detect leakers and any other visual defects. Frequency of SMI is as required by technical instructions for the specific item. At a minimum, all storage facilities (magazines, warehouses, etc.) are inspected at quarterly intervals. The inspections consist of both internal and external visual examinations. Other than appropriate protective clothing and flashlights, no special equipment is required. No moving or restacking of pallets is involved. The inspections address the following:

- 1. Exterior
  - Structural integrity.
  - Condition of storage area.
  - Vegetation control.
  - Clear of dried debris.
  - Firebreaks cleared.
  - Adequacy of earthen cover.
  - Condition of doors and ventilators.
  - Correct type of fusible link on vents.
  - Lightning protection system.
  - Condition of service roads.

### 2. Interior

- Condition of munitions.
- Compliance with storage drawings.
- Lot segregation.
- Stability of pallet stacks.
- Adequacy of aisles.
- Absence of unauthorized materials or equipment.
- Containers are not damaged.
- Presence of proper records.





<u> 1999–99 - 1997–1998 - 1998–1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998</u>

LUCULTURE PORTERE

Evidence of termites, rodents, water leakage, or other nonstandard conditions.

Visits to each of the chemical storage sites by the members of the analysis team indicate that the condition of the storage facilities with respect to the above characteristics has been excellent. Only minor repairs for water leakage on igloos have been required.

ACCURATE A

An enhanced storage monitoring program is in place for the rockets, some of which have experienced vapor leaks. Typically, the inspection involves a three- or four-man team and consists of walking the aisles between the stacks of pallets and making an initial visual inspection for observable signs of agent leakage. Lighting for the storage monitoring inspection is provided by powerful hand-held flashlights. If signs of leakage are found at any time during the inspection, masks are donned and the area is cleared. Following visual inspection, a munition is selected at random for air sampling of the interior of the shipping and firing tube. Sampling is accomplished in Level B or Level A protective clothing (see Table G-1). The inspection procedure involves no moving or restacking of pallets (unless a leaker is found). All equipment is located on a self-contained cart, which is rolled into the igloo by hand.

Ton containers that are stored in igloos or warehouse buildings are inspected for leakage quarterly (Ref. G-6). Ton containers stored in the open are also required to be inspected quarterly (Ref. G-7). A number of these containers (primarily ton containers with GB) have experienced severe corrosion of the brass fill and drain valves, and some have experienced corrosion in the area of the threaded plugs installed in the container ends. The current plan is to replace the brass valves with stainless steel valves on all GB ton containers. The same degree of corrosion has not been associated with agents other than GB. The corrective procedure for containers containing those agents has been to replace the corroded valves or plugs. This is accomplished with



# TABLE G-1 PERSOMAL PROTECTIVE CLOTHING AND EQUIPHENT(=)

|                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                     | Protection                                                                                                                                                                               | Level                                                                                                                                                                                                   |                                                                                                                                             |                                                                                                                                                               |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------|
| A (GB and VX)                                                                                                                                                                                                          | B (GB and VX)                                                                                                                                                                                                                                                                       | C (GB only)                                                                                                                                                                              | D (VY only)                                                                                                                                                                                             | E (cl of VX)                                                                                                                                | r (Gh or VX)                                                                                                                                                  |
|                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                     | Protective (                                                                                                                                                                             | Clothing                                                                                                                                                                                                |                                                                                                                                             |                                                                                                                                                               |
| Suit, TAP(b) (HJ)<br>Coveraila,<br>Coveraila,<br>Estigues or pro-<br>tective liner<br>Hood, TAP (HJ)<br>Butyl boots with<br>affety toe, TAP<br>(H2A!)<br>Butyl Bloves (HJ<br>Undershirt<br>Drawers<br>Sochs<br>aeties) | Coveralla or fatigues<br>Hood, TAP (MJ for M9<br>mask) or M6A2 for M17<br>mask)<br>Butyl boots with<br>safety tos, TAP<br>(H2A1)<br>Butyl apron, attend-<br>ing below top of<br>boots (M2)<br>Butyl glowe ast)<br>Underahitt<br>Drawers<br>Socka<br>Mask-worn (M9 or<br>M17 sarisa) | Coveralle or<br>fatigues<br>Buryl boots with<br>asfety toe, TAP<br>(H2A1)<br>Buryl gloves (M3,<br>H4, or glove aet)<br>Undershirt<br>Drawers<br>Socks<br>Maak-worn (M9<br>or N17 aetlee) | Coveralle of<br>fatigues<br>Buryl boots with<br>aafety too, TAP<br>(R2A1)<br>Aucyl gloves (M3,<br>M4, or glove ast)<br>Undershitt<br>Undershitt<br>Drawers<br>Socka<br>Hah, alung (M9 of<br>H17 aerias) | Coveralls or<br>facigues<br>Safaty abos (1f<br>required)<br>Buryl gloves (M1,<br>M4, or glove<br>M4, or glove<br>M17 aeriee)<br>M17 aeriee) | Streat attire<br>Haak, alung<br>position (MB at<br>M17 astiae)                                                                                                |
| Area of spilled<br>agent or liquid<br>contamination<br>Storage operations<br>Sampling<br>operations<br>Maintenance<br>operations<br>fire fighting/<br>incident control                                                 | Area of suspected<br>agent or agent<br>vapors<br>First entry monitoring<br>of outside storage<br>area<br>area<br>frained emergency<br>personnel responding<br>to an accident<br>Loeding and charging<br>the M9 or M12 decon-<br>teminating apparetus                                | Conditions<br>Immadiate opera-<br>ting area whara<br>auspected contami-<br>neted items or<br>equipment are<br>preasent<br>Ho contact with<br>contantiated items<br>is required           | Required<br>Immediate area of<br>outside operations<br>where suspected<br>items or equipment<br>are present<br>No contact with<br>contaminated items<br>is required                                     | Mora by observers<br>or supervisors of<br>operation and<br>laboratory<br>personnel                                                          | Morm by visitors.<br>casuals, super-<br>visory, or opera-<br>tions control<br>perconnal in are<br>where hazardous<br>atored of in<br>clean operating<br>areas |

(a)This table presents a brief summary of the data presented in Ref. G-4. (a)

(b)TAP - toxicological agent protective.

÷.,

(c)Only for operation when G&/VX containers are handled.

the container filled with agent. While implementing these procedures, Level A protective clothing is worn by all personnel in the immediate vicinity. The procedures involve removing the leaking container from its storage igloo and lifting the container onto a special fixture which will permit the container to be tilted from a horizontal to a vertical orientation. The lifting operation is accomplished with an electric forklift, using an Ml lifting bar which is specifically designed to lift a ton container in a horizontal position by engaging both ends of the container with self-locking hooks. Once the container is placed in the fixture, it is tilted to the vertical orientation with the valve end pointing up. The leaking valves are removed, the threads in the container are recut, as required, and a new valve is installed in its place.

PAN'N' MARKE WARKE ANNUA DAVANA ANALY

BULLING

Visual examination of the ton containers also reveals the degree of rusting that the containers are experiencing. Specific criteria for allowable rusting are given in SB 742-1 (Ref. G-6). In general, the ton container will be placed in condition Code E and scheduled for derusting and repainting if any of the following occur:

- Minor rust on the ends of the container exceeds 25% of the container surface.
- Sufficient rust exists in the vicinity of the values to hinder the detection of agent leakage.
- 3. Rust or corrosion on the cylindrical surface of the container has progressed to the point of a scaly, granular, or flaked condition, accompanied by definite pitting or etching of the material.



4. Rust or corrosion has progressed to the point where the identification markings on the container are threatened to be rendered illegible.

#### G.1.2. ONSITE TRANSPORTATION

Transport of munitions on military reservations is, essentially, the movement of these munitions between an interim storage area and an onsite railhead or disposal facility. Generally, this movement is characterized by locating a transport vehicle at the loading apron, loading the transport vehicle, traveling to an unloading station, and unloading the vehicle. For the NDC and RDC alternatives, onsite transportation includes transportation between the storage igloo and an onsite railhead at both the shipping and receiving end.

At APG and NAAP, munitions will be moved from storage to the railhead by forklift. At all other storage sites, chemical munition movement will take place using either an enclosed trailer or a stake and platform trailer. These trailers are designed for ease of cargo handling with a forklift/pallet system.

The enclosure trailer is similar to any van-type trailer with the addition of a roller-conveyor unit in the trailer floor. The use of the trailer permits forklift loading of pallets through the end of the van body, minimizing forklift travel. Once in the trailer, the pallet is manually rolled to its position and secured. The trailer is unloaded in reverse at its destination. The use of the enclosed trailer in this manner provides some basic thermal and mechanical protection of the munition pallets and leakage containment by the van body.

The stake and platform trailer is a large trailer designed for side loading. With the sides removed, pallets can be loaded directly onto all points of the trailer floor by forklift. The use of either the enclosed trailer or the stake and platform trailer, negates the requirement for special loading ramps for forklifts.

Once the transport vehicle is loaded, the pallets are secured to prevent the load from shifting during movement. Blocking devices, which fit into the roller-conveyor, are used to secure pallets in the enclosed

G-12

trailer. Wooden chocks and blocks are spiked into the stake and platform trailer to secure the pallets.

Movement of munitions will take place within the chemical munition exclusion area on existing and/or newly constructed roads. Specific road conditions vary from site to site. At some sites, the roads are essentially flat; at others, the roads are hilly with steep grades. The road surface itself also varies in condition and type. In addition, obstacles such as utility poles are present at some of the reservations, while others have none. The immediate surrounding terrain also varies in each case from sandy and flat to firm clay with ravines.

Equipment to mitigate the effects of a transport accident are present with the munition transporter. This equipment includes fire fighting and decontamination equipment that is fully manned and ready.





#### G.1.3. OFFSITE TRANSPORTATION

Three basic methods have been considered for transporting M55 rockets within the United States: military aircraft, military truck convoy, and munition trains. A probabilistic analysis of agent release (Ref. G-8) resulting from transportation accidents involving these rockets indicated that munition trains were the preferred mode for offsite transportation. This analysis also evaluated a packaging configuration referred to as an offsite transport container (OFC) for shipping the M55 rockets. The current disposal plan (Ref. G-11) indicates that M55 rockets, M23 land mines, 4.2-in. mortars, and 105-mm cartridges are to be transported offsite in an OFC. The first two munitions are especially hazardous because they are relatively thin-walled and contain explosives. The latter two items are considered hazardous because they are fuzed and shipped with propellants and explosives. Because they have the potential to release a large quantity of agent, ton containers are also assumed to be shipped in the OFC.

The OFC technology builds on the transport package developed for the U.S. Department of Energy for protection of radioactive materials during transport (Ref. G-9). The OFC is designed to provide for the containment of agent vapors with crush and puncture resistance and insulation from fire. The rail version of the OFC weighs approximately 48,000 lb and can hold 15 pallets of M55 rockets. Two rail OFCs can fit on a flatbed railcar.

Rail convoys would consist of two separate trains: an 18-car pilot train consisting of support cars for personnel and backup emergency response cars, and the munition train consisting of 50 munition cars and 18 support cars. A third train of unspecified length may be present for the purpose of carrying emergency medical supplies for response to an accident in the event of a catastrophic agent release. The munition trains would carry sufficient equipment and personnel to provide security to the cargo, to control accidental agent release, and

G-14
to provide treatment to personnel accompanying the movement. A typical makeup of munition trains is shown in Tables G-2 and G-3.

26666644

こしし ノンソノ

Munition cars would be loaded and unloaded inside the chemical exclusion areas at the shipping and receiving depots. This requires extension of a rail spur into the exclusion areas capable of holding 50 cars (flatbed cars for OFCs and standard boxcars for palleted projectiles) and/or extension of the areas to surround existing rail sidings. The rail facility would be engineered during preparation of a detailed transportation plan. To permit the entire train to be prepared for movement in 7 days, a relatively long loading dock would be required, allowing the simultaneous loading of 5 to 10 railcars.

Trains would move continuously, day and night, at an average speed of about 15 mph (maximum speed 35 mph) over routes chosen to bypass large population centers. Specific routes will be established with the rail carriers as part of a detailed transportation plan. Rail beds will be inspected and repaired or upgraded as necessary, prior to and during the movement campaign.

Munition trains will be accompanied by relays of surveillance helicopters during daylight hours. All grade crossings and overpasses will be guarded by civilian police or military personnel during use. Train movements will be carefully coordinated in advance with appropriate state and federal emergency response forces along the route.

To minimize temporary storage in rail cars, the rate for transporting munitions from the existing CONUS sites to the demilitarization site should match the plant process rate. The plant process rates and the optimum transportation requirements for the NDC and RDC alternatives are shown in Tables G-4 and G-5, respectively.

| Item                             | Munition Train | Pilot Train |
|----------------------------------|----------------|-------------|
| Munition cars                    | 50             | 0           |
| Tanker                           | 1              | 1           |
| Decontamination                  | 1              | 1           |
| Pullman                          | 0              | 6           |
| Diner                            | 0              | 1           |
| Guard and support personnel cars | 10             | 1           |
| Ambulance                        | 0              | 1           |
| Communications support           | 2              | 1           |
| Ramp                             | 0              | 1           |
| Spare trailer                    | 0              | 2           |
| Spare tractor                    | 4              | 3           |
| Laboratory car                   | _1             | _1          |
| Total                            | 69             | 19          |
|                                  |                |             |

#### TABLE G-2 MUNITION TRAIN REQUIREMENTS(a)

(a) The requirements for the medical supply train have not yet been defined and therefore were not included in this table.



Sec. 64. 65

#### TABLE G-3 PERSONNEL ACCOMPANYING RAIL MOVEMENT

| Туре                                                                                                                                                   | Number |
|--------------------------------------------------------------------------------------------------------------------------------------------------------|--------|
| Guards for munitions cars                                                                                                                              | 100    |
| Guards for munitions train (other than munitions cars)                                                                                                 | 15     |
| Guards for pilot train                                                                                                                                 | 27     |
| Support personnel for munition train (medics; drivers for trucks, ambulances, forklifts, crane, and heavy equipment; mechanics; radio operators; etc.) | 38     |
| Support personnel (as above) for pilot train                                                                                                           | 35     |
| Command and control team for munition train                                                                                                            | 10     |
| Technical escort personnel                                                                                                                             | 18     |
| Total                                                                                                                                                  | 243    |



|                     | Process Rate(a,c) | No. of Mu  | initions   | Delivery Rate |
|---------------------|-------------------|------------|------------|---------------|
| Munition            | wk                | Railcar    | Train      | Trains/wk     |
| Rocket              | 14,400            | 450(b)     | 30,000     | 0.48          |
| 4.2-in. mortar      | 62,640            | 1,056(b)   | 52,800     | 1.18          |
| 105-mm cartridge    | 71,280            | 600(b)     | 30,000     | 2.38          |
| 155-mm projectile   | 36,720            | 1,088      | 54,400     | 0.68          |
| 8-in. projectile    | 14,760            | 516        | 25,800     | 0.57          |
| Mine                | 20,160            | 1,080(b)   | 54,000     | 0.37          |
| Ton container       |                   |            |            |               |
| GB<br>VX            | 360<br>240        | 144<br>144 | 600<br>600 | 0.60<br>0.40  |
| Mustard             | 420               | 144        | 600        | 0.70          |
| MC-1 750-1b<br>bomb | 1,548             | 144        | 7,200      | 0.21          |

### TABLE G-4NATIONAL DESTRUCTION CENTER TRANSPORTATION REQUIREMENTS

(a)Basis: 120 operating hours per week at the average throughput of JACADS facilities. The national site employs two JACADS bulk plants as well as three standard JACADS plants.

(b) Shipped in a CAMPACT.

(c)Reference G-11.



|                   | Process<br>Per Wee | s Rate<br><sub>ek</sub> (a,c) | Number of<br>Munitions | Delive:<br>Trains p | ry Rate,<br>per Week |
|-------------------|--------------------|-------------------------------|------------------------|---------------------|----------------------|
| Munition          | Tooele             | Anniston                      | Per Train              | Tooele              | Annistor             |
| Rocket            | 9,600              | 4,800                         | 30,000(b)              | 0.32                | 0.16                 |
| 4.2-in. mortar    | 41,760             | 20,880                        | 52,800 <sup>(b)</sup>  | 0.80                | 0.40                 |
| 105-mm cartridge  | 47,520             | 23,760                        | 30,000(b)              | 1.58                | 0.79                 |
| 155-mm projectile | 24,480             | 12,240                        | 54,000                 | 0.45                | 0.23                 |
| 8-in. projectile  | 9,840              | 4,920                         | 25,800                 | 0.38                | 0.19                 |
| Mine              | 13,440             | 6,720                         | 54,000(b)              | 0.25                | 0.12                 |
| Ton container     |                    |                               |                        |                     |                      |
| GB<br>VX          | 216<br>144         | 144<br>96                     | 600(b)<br>600(b)       | 0.36<br>0.24        | 0.24<br>0.16         |
| Mustard           | 252                | 168                           | 600(b)                 | 0.42                | 0.28                 |
| MC-1 750-1b bomb  | 1,032              | 516                           | 7,200                  | 0.14                | 0.07                 |

### TABLE G-5REGIONAL DESTRUCTION CENTERS TRANSPORTATION REQUIREMENTS

(a)Basis: 120 operating hours per week at the average throughput of JACADS facilities. Regional sites employ one each JACADS and bulk plants at ANAD with two JACADS and one bulk plant at TEAD.

(b) Shipped in CAMPACT.

(c)<sub>Reference</sub> G-11.



Special munition trains will be used for rail transportation. Each munitions train will be preceded by a pilot train. The munitions train is configured so that the cars are divided into groups with buffer cars containing inert material between the groups. Five cars containing inert material are placed between the locomotives and the first munitions car. The composition of typical munitions and pilot trains is shown in Table G-6. The number of ammunition cars was assumed to be 70 for the analyses in this report, with about 50 support cars.

The effect of human factors on the train accident rate (Section 9.2) is implicit in the SNL data base. If an accident occurred due to human error, it shows up in the data base just as an accident. Therefore, it is not possible to ascertain the human error contribution or to define the human error probabilities involved. No specific human reliability analysis was done for rail transportation.

Extensive administrative controls will be in effect during rail transportation, however. These controls are described briefly below. For all rail transport, a 10-person command and control team will be located at both the shipping and receiving sites, and another will accompany the munitions train. A technical escort team will also accompany the munitions during transport. It is assumed that emergency response to a train accident by the pilot train will be within 1 h, with any accident agent release contained within 6 h.

Each transport vehicle will only carry one type of chemical agent (e.g., all VX or all GB) and one kind of munition at a time. The munitions train may transport more than one munition type but only one agent type per trip. The vehicles will be inspected before each planned movement.

| Type of Railcar       | Escort Train | Munition Train |
|-----------------------|--------------|----------------|
| Munition (loaded)     | 0            | 70             |
| Munition (empty)      | 5            | 0              |
| Buffer                | 0            | 33(b)          |
| Tank                  | 1            | 1(c)           |
| Decontamination       | 2            | 2(c)           |
| Passenger             | 3            | 0              |
| Guard                 | 0            | 7(d)           |
| Support equipment     | 4            | (c)            |
| Container overpack(e) | 2            | 0              |
| Radio support         | 1            | 2              |
| Laboratory            | 1            | 1              |
| Medical               | 1            | 0              |
| Command               | _0           | <u>1</u>       |
| Totals                | 20           | 117            |
|                       |              |                |

#### TABLE G-6 MAKEUP OF THE MUNITION AND ESCORT TRAIN IN A RAIL CONVOY(a)

(a)Reference 8-12.

Cab

(b)Five cars between the command car engines and the first munition car. Two cars between cars carrying Army personnel and munition cars.

(c) Will be used as buffer cars.

(d)Guard cars will be interspersed among the munition cars. They are likely to be modified passenger cars.

(e)Overpack containers will be carried on 89-ft flatcars.

The munition and pilot trains are limited to a maximum speed of 50 mph and will stop only for engine changes, crew changes, and munition car inspections. At 1000-mile intervals, the train will stop for equipment checks; cars will be inspected and monitored concurrently.

#### G.1.4. HANDLING ACTIVITIES

The following paragraphs describe onsite facility handling activities as they are presently defined for the JACADS facility. Unless another reference is specifically cited these descriptions were taken from the JACADS final design description (Ref. G-10). When it is apparent that differences are required for handling activities at other processing sites, these differences are described. Although the risk analysis did not address specific accident scenarios involving the handling of leakers, normal handling procedures for leaking munitions as described in Ref. G-10 have also been presented.

One condition that may vary from site to site and possibly from igloo to igloo within a site is the relative levels of the pavement inside and outside the entrance to the igloos. Because of differences in floor/ramp level inside and outside the igloo, munitions that are being transported from an igloo are placed on a pad outside the storage igloo to be picked up and loaded onto the transport truck by another forklift. The forklift used outside the igloo may be either electric, gasoline, or LPG powered.

Standing operating procedures exist for continuous monitoring and periodic inspections to identify and isolate leakers of all munition configurations. When preparing for munition removal from an igloo, it is particularly important to identify and isolate leaking munitions so that they may be decontaminated, overpacked, and segregated until processed in a separate campaign. To do this, munitions other than ton containers and spray tanks must be removed from their pallets and handled separately. (Ton containers and spray tanks are always handled singularly.) Normally, no lifting equipment, other than an electric forklift truck, is available for handling single munitions.

When a leaking munition is removed from a pallet, a nonleaking munition of the same configuration and lot number is normally inserted

in place of the leaker to keep the pallets fully populated. In this way there is only one broken pallet in a given munition lot.

At the demilitarization facility, munitions arrive in their normal packaging units either from the MHI or directly from the storage igloo. From the MHI, munitions will be delivered by forklift directly into the elevator and then to the UPA. Munitions coming directly from the storage igloos to the MDB will be transported by a flatbed munition truck or a munition van. On arrival at the MDB, a forklift will be used to unload munitions from the truck and place them in the elevator.

#### G.1.4.1. Projectiles and Mortars

Each of the 105-mm M60 and M360 cartridges are currently stored in a fiber tube container, with two fiber tube containers per wooden box, and 12 or 15 boxes per pallet. Each 4.2-in. M2/M2A1 mortar cartridge is stored in a fiber tube container, with two fiber tube containers per wooden box, and 24 wooden boxes per pallet.

For the purpose of this study, it was assumed that there will be a special area, separate from the demilitarization facility, where cartridges will be unpacked, the propellant and ignition cartridge removed, and the projectiles repacked in a configuration of 24 munitions per pallet. The mortar propellants which are removed will be placed into 4-in. diameter, 18-in long, thin metallic pipes. The ends of these pipes will be capped with plastic lids. These tubes with propellant and cartridge cases with primers will be fed to the deactivation furnace system (DFS) through the mine and rocket transport conveying system in a separate campaign. This approach is similar to that for the JACADS plant. However, the U.S. Army is also considering other approaches to be used at the CONUS sites for removal of propellant from these cartridges.

Projectiles are strapped directly to wooden storage/shipping pallets. The pallets contain either twenty-four 105-mm projectiles, eight 155-mm projectiles, or six 8-in. projectiles.

#### G.1.4.2. Rockets

Each M55 rocket is encased in a fiberglass shipping and firing tube that has aluminum nose and tail closures. Fifteen rocket tubes are strapped onto a wooden storage/shipping pallet. Rocket pallets are moved in their as-stored configuration to or from the railhead. At the railhead, the pallets are placed in an OFC.

At the NDC or RDC, rocket pallets are placed in SPORTS and into onsite containers (ONC) prior to being transported to the MDB. Each transport truck will carry up to four ONCs with 15 rockets per ONC. The rocket pallets and ONCs are handled inside the storage igloos by electric forklift trucks.

#### G.1.4.3. Mines

Mines are packed three to a drum along with three M603 fuzes and three M1 activators. Mine pallets (12 drums per pallet) are moved by forklift to the ONC for transport to the MHI. From the MHI, another forklift is used to transfer mine pallets in the ONC into the MDB, and subsequently to the UPA.

#### G.1.4.4. Bulk Items

MC-1 750-1b bombs are stored two-to-a-pallet while the MK-94 500-1b bombs are not palletized. Ton containers are not palletized in storage. They are moved by forklift but are placed onto the forklift using an M-1 or similar type lifting beam. Spray tanks are stored without pallets in customized containers. For this analysis, it is assumed that spray tanks are normally handled using forklifts.

#### G.1.5. MUNITIONS DEMILITARIZATION

The proposed NDC site for the disposal of the total CONUS stockpile would be located at TEAD. The NDC site would consist of five facilities for processing the munitions and bulk agents, each sized for the JACADS process rates.

Three of these facilities would be identical to the JACADS facility with capability for processing munitions and bulk agent. Two JACADSadapted facilities would process bulk agent only. The current CAMDS facility will be modified to incorporate the JACADS process equipment, and would comprise one of the two bulk process lines. The second bulk facility would be a new JACADS-adapted bulk facility. Multiple JACADStype facilities would be utilized to minimize the required time for design, procurement, and construction, thus maximizing the time period remaining for disposal operations.

The plan for the RDC is to provide two regional disposal facilities, one at TEAD, and the second at ANAD. The RDC at TEAD would consist of three facilities: two JACADS-type facilities for processing munitions and bulk agent plus a modified JACADS-adapted CAMDS facility to process bulk agent only. The Anniston RDC would consist of two facilities: one JACADS-type facility for processing munitions and bulk agents and one JACADS-adapted bulk-only facility.

#### G.1.5.1. Baseline Technology

The demilitarization facility evaluated in this study is based on the JACADS process and consists of an integrated munition-handling system to process all of the different types of munitions and agents. All demilitarization will be performed in the MDB, which houses the UPA, rocket and mine punch-and-drain machine, projectile mortar disassembly machine, rocket and burster shearing machine, mine machine for booster removal, a bulk drain station (BDS) to punch/drain bulk items, agent

transfer equipment, a toxic cubicle (TOX) for agent storage tanks, and furnaces for explosive deactivation, agent incineration, metal parts decontamination, and dunnage incineration. All furnaces will have afterburners to ensure complete agent destruction. Each furnace has its own pollution abatement system (PAS).

Revisions to the JACADS design will be necessary for site adaptation (Ref. G-12). Some of the revisions are listed below:

- Equipment weather enclosures will be added for all process equipment which will be located outdoors, i.e., pollution abatement system (PAS), brine reduction area (BRA), and bulk chemical storage (BCS).
- 2. All fuel burning equipment, ducts, and fans will be resized for higher altitude and different fuel, where applicable.
- Rooms will be resized to provide any additional space necessary to accommodate the changes noted above.
- The structural design for the building and equipment supports will be evaluated and revised, if required, to meet higher seismic loads.
- 5. Refrigerated plant air dryers will be changed to desiccant type to prevent water condensation in outdoor piping during winter operations.

A simplified schematic diagram of the process is shown in Fig. G-1. The demilitarization process for each group of munitions is described below (Ref. G-12).

#### G.1.5.2. <u>Projectiles and Mortars</u>

These munitions (in ONCs) will be examined for leakers in the unloading area. Nonleaking munitions will be unloaded and transferred



Live:

5

ر ا



by elevator to the UPA located on the second floor, where they will be removed from the pallets by personnel wearing Level D protective clothing. They will then be loaded manually on an input tray conveyor, taken to the explosive containment vestibule, and then moved through airlocks and blast gates to the explosive containment room (ECR). All dunnage resulting from the unpacking operation will be burned in the dunnage incinerator.

Inside the ECR, the projectile will be automatically placed on the projectile/mortar disassembly machine (PMD) turntable for removal of explosive components. The burster will then be conveyed to the burster size reduction machine (BSR) and fed by gravity through a discharge chute with double isolation valves into the deactivation furnace system (DFS). The fuze will be moved by conveyor to the DFS for incineration. The projectile will be probed to verify burster removal and placed on a conveyor from the ECR and leading to the multipurpose demilitarization machine (MDM) in the munitions processing bay. A pick-and-place robot will pick up a projectile from the pallet and place it on a rotating table. Here, the burster well will be extracted from the projectile and a tube will be inserted into the projectile to remove the liquid agent by suction and convey it to a storage tank in the toxic cubicle. If the burster well is stuck or welded in place, a milling station on the MDM rotating table will cut off the top of the burster well to allow its removal.

Agent collected in the TOX will be incinerated in a liquid incinerator (LIC). The drained projectiles will be placed on a tray and conveyed into the waiting munitions lift car, which descends to the first floor to transfer the tray to a charge car for introduction into the metal parts furnace (MPF). The MPF will thermally decontaminate the drained projectiles to a 5X level.\*





#### G.1.5.3. Rockets

M55 rockets will arrive at the MDB in ONC containers. Only ONCs verified to be free of leaking rockets will be unloaded in the package unloading facility. Operators wearing Level D protective clothing will manually remove individual rockets, feed them through a munition metering system to the explosive containment vestibule (ECV), then into the ECR. The rockets will be automatically punched and drained at the rocket drain station (RDS) in the ECR. Agent will be drained from the rocket by pump suction and collected in the TOX for subsequent incineration in the LIC. Once drained of agent, the punched rockets will be conveyed to the rocket shear machine (RSM), which will shear the rockets into the required number of pieces. The separated sections fall by gravity into the feed chute leading to the DFS, which is located on the first floor of the MDB. An interlock will ensure that only one of the two blast gates in the feed chute is open at any given time.

If there are leaking rockets stored in an ONC, it will not be opened in the UPA, but will be conveyed directly to the ECV where operators wearing demilitarization protective ensemble (DPE) suits will open the ONC and manually unload each rocket onto the feed table feeding the conveyor. They will then be processed in the same way as nonleakers.

#### G.1.5.4. Mines

Pallets of nonleaking mine drums will be removed from the ONCs in the package unloading facility. Mine drums (three mines in a drum) will be unloaded from their pallet in the UPA and placed, unopened, on the drum conveyor entering a mine glove box (MIG) in the ECV. An operator wearing protective clothing will open the drum in the glove box and remove the mines. The activators and fuzes that have been packed in the drums will be placed in a cardboard container and conveyed to the DFS chute. The arming plug will also be removed. A mine will be conveyed into an ECR, where it will be automatically punched and drained of

agent. The agent will be drained from the mine by pump suction and pumped to the TOX for subsequent incineration in the LIC.

While in the ECR, the mine will be placed automatically in the mine machine (MIN) to punch out the booster. The mine body and booster are dropped into the DFS.

#### G.1.5.5. Bulk Items

Bombs, ton containers, and spray tanks in ONCs will be moved from the MHI by forklift and unpacked at the package unloading facility where an elevator will be used to transfer the munitions to the UPA which is located on the second floor of the JACADS plant. For the bulk only plants, the UPA will be located on the ground floor. A forklift will move the bulk item to the UPA for pallet removal and subsequent transfer to tray assemblies on the input conveyors. Spray tanks will be removed from their shipping containers in the UPA and transferred to tray assemblies on the input conveyor. Unpalletized bulk items, such as ton containers, will be placed directly on tray conveyors. The trays will be conveyed to the bulk drain station (BDS), which is equipped with a large punch and an agent pump and removal tube. The punch will produce a hole in the top of the bulk item, and the removal tube will be inserted through the hole to allow removal of the liquid agent, which will be transferred to the TOX by agent pipe lines. The tray containing the drained bulk item will be transported to the munitions lift car, which descends to the first floor to discharge the tray to the buffer storage conveyor and into the MPF. Residual agent will be burned in the MPF.

#### G.1.6. DECOMMISSIONING

After the existing stockpile of lethal chemical agent and munitions at each site has been destroyed, the demilitarization facility will be decommissioned. The activities for cleanup and closure of the destruction facilities, as discussed in Chemical Stockpile Disposal Plan (Ref. G-11), are as follows:

- 1. Decontamination of the MDB and laboratory.
- 2. Disposal of all solid wastes and residues.
- 3. Certification of the plant and site as nontoxic.

The first step in the cleanup operation will be the removal of all equipment not required for the decommissioning effort from the noncontaminated areas of the facility. The contaminated portions of the building and the contaminated destruction equipment will be washed with an aqueous decontamination solution. When the washing operations are complete and the level of decon verified, the surrounding areas will be tested and monitored to verify that any vapor concentrations of agent are within allowable limits. The equipment will be disassembled for thermal decontamination. The building itself will be tested or monitored to verify that any vapor concentrations are within allowable limits.

The furnaces used for decontamination of the munitions will be maintained in place and used for the decontamination of process equipment as long as possible. Final decontamination of the remaining furnace and supporting equipment could be accomplished in a transportable furnace brought to the site.

After all necessary decontamination, disassembly, and demolition, all solid waste and residue resulting from the decommissioning will be disposed of. Materials that cannot be certified for other uses will be disposed of at approved hazardous waste landfill sites. The decontami-



nated plant and site will be monitored and tested to ensure that no residual toxic agent is present. After monitoring has been completed and monitoring samples satisfactorily analyzed, the plant will be certified closed. DESERVE RECEIPTION

18.20 A



#### G.1.7. REFERENCES

Sec. Sec.

- G-1. "Safety Manual," Department of the Army, AMC-R 385-100, August 1985.
- G-2. "Safety Regulations for Chemical Agents GB and VX," Department of the Army, DARCOM-R 385-102, May 1982.
- G-3. "Safety Regulations for Chemical Agents H, HD, and HT," Department of the Army, DARCOM-R 385-31, April 1979.
- G-4. Science Applications International Corporation, "Probabilities of Selected Hazards in Disposition of M55 Rockets," U.S. Army Toxic and Hazardous Materials Agency, M55-CS-2, November 1985.
- G-5. "Risk Analysis of the Continued Storage of Chemical Munitions," GA Technologies Inc., GA-C18564, December 1986.
- G-6. "Toxic Munitions and Bulk Storage GB, VX, H, HT, HD: Surveillance and Leakage Test Procedures Ammunition Surveillance and Safety-In-Storage Procedures," Department of the Army, SB 742-1300-94-1, June 1972.
- G-7. "Ammunition Surveillance Procedures (Draft)," Department of the Army, SB 742-1, November 1985.
- G-8. Rhyne, W. R., et al., "Probabilistic Analysis of Chemical Agent Release During Transport of M55 Rockets," H&R 255-1, H&R Technical Associates, Inc., September 1985.
- G-9. "Research and Development Services for Mechanical Process Development/Laboratory Studies in Support of the Munition/Agent Process Development Program," Volume 1, Book 2, GA Technologies Inc., GA-A16891, November 1982.
- G-10. The R. M. Parsons Company, "JACADS Final Design Description," Task E-2, March 1985.
- G-11. "Chemical Stockpile Disposal Plan," U.S. Army Toxic and Hazardous Materials Agency, Draft Report, AMXTH-CD-FR-85047, March 1986.

"<u>15222222</u>"

11222111

G-12. The R. M. Parsons Company, "JACADS Final Design Analysis Narrative (Sections 2 and 3)," April 1985.







APPENDIX I TABULATED ACCIDENT SEQUENCE RESULTS KUM EEEEN IN NOOT 


ļ

#### I.1. TABULATED ACCIDENT SEQUENCE RESULTS

ANALY USED TO THE PARTY TO THE PARTY OF

ALALANSA REPORT REPORT REPORT

Land State Production

The following subsections give the accident sequence results for long term storage, interim storage, handling, plant operations, onsite transport, and offsite transport of munitions.

I.1.1. LONG TERM STORAGE

The following tables list the accident results for long term storage for munitions at existing sites.







File: COMSTOR.MK1 date21-Aug-87 page 1

# STORAGE ACCIDENTS - Ifrequency waits given at battem of table) FOM NUMMITIONS AT EXISTING SITES

|     |            |               |             |                   |                 | ¥           | ident Fr    | elueuc i e: |                 |            |                 |               |                 |            |                |           |                  | -             | lant Ave         | Ini -              | fiel nased      |                  |
|-----|------------|---------------|-------------|-------------------|-----------------|-------------|-------------|-------------|-----------------|------------|-----------------|---------------|-----------------|------------|----------------|-----------|------------------|---------------|------------------|--------------------|-----------------|------------------|
|     | , AhAA     | RAME<br>FACTO |             | RAMEI<br>P FACTOR |                 | 2 Z<br>2 Z  | MER<br>C DR | THE T       | RMIGE<br>FACTOR | P M        | RANGE<br>Factor | PUBA<br>Finco | RANKE<br>Factor | IKG<br>1   | PANE<br>FACTOR |           | FACTOR<br>FACTOR | MENT<br>MMIL. | LINS.<br>SPILLED | L IS.<br>Reionated | LBS.<br>Entited | BURATION<br>Time |
| · • | veloos -   | leak dur      |             | teen-tak          | ections         | ber i bd.   | :           |             |                 |            |                 |               |                 |            | •              |           |                  |               |                  |                    |                 |                  |
|     | 1          | 4             |             | ę                 | ,               | A/4         | ,           | A/M         | •               | <b>N/N</b> | •               | A/A           | •               | 2-35-1     | 1.0€+01        | 1.55-04 1 | . 06 +01         | 1.495+05      | •                | '                  | 4. 005 +01      | Po Mi            |
| -   | 2.8        | ·J 1.06•      | /W 10       | 5                 | -               | N/A         | •           | A/A         | •               | W/N        | ,               | 1.06-04       | 1.0€+01         | 7.06 04    | 1.06+01        | <b>V</b>  |                  | 6.91E+04      | •                | •                  | 4.906+00        | 18 S             |
| _   | 2.8        | -07 1.06.     | 10          | e.                | ,               | N/A         | ,           | N/N         | •               | N/A        | •               | W/W           | •               | 7.06-04    | 1.05+01        | N/N       | •                | 1.446.404     | •                | •                  | 1.406+00        | I MY             |
| -   | 2.86-      | -0) 1.06-     | N 10        | A)                | •               | N/A         | ,           | N/N         | ,               | N/A        | •               | 1.06-06       | 1.0€+01         | N/N        | •              | W/W       | •                | 2.926+04      | ,                | '                  | 3.006+00        | 14 9             |
| -   | 1          | A'            |             | ۲.                | -               | N/A         | •           | 8/8         | •               | A.M        | :               | N/N           | ,               | 1. % -     | 1.04.401       | N/N       | ı                | 2.076+05      | '                | '                  | 9.00€+01        | TO MY            |
| -   | - 3, %-    | ·30.1 40      | 01 10       | Å,                | ,               | <b>H</b> /A | ,           | N/A         | •               | M/A        | •               | N/N           | ,               | W/W        | •              | A/A       | ,                | 1.446+05      | '                | '                  | 2.136+01        | VM 06            |
| -   | 1          | 4             | Ť.          | V.                | -<br>,          | N/A         | ·           | N/A         | •               | 5.96-06    | 1.0€+01         | R/A           | •               | 5.96-04    | 1.05+01        | R/A       | -<br>-           | LASS.         | 9.005+01         | '                  | •               | 90 DAY           |
| -   | 1          | đ             | 36 5        | -06 1.06+         | -               | R/A         | •           | N/A         | •               | N/N        | •               | N/N           | •               | N/A        | •              | N/N       |                  | LASS.         | 1.005+00         | '                  | '               | N.               |
| -   | 1          | 4             | - N         | e,                | -               | N/A         | ,           | N/A         | ·               | A/M        | •               | N/A           | •               | N/A        | •              | 5.96-06 1 | 10+30.           | 5.316+06      |                  | •                  | 2.136+01        | S MY             |
|     | A .        | 4             | *           | ¥,                | -<br>,          | M/A         |             | A/A         | •               | A/N        | •               | A/A           | •               | 5.96-06    | 1.06+01        | N/N       | •                | 2.216+05      | 1                | •                  | 1.026-01        | Po BAT           |
| -   |            | A.            | ×           | A.                | -<br>,          | N'A         | ,<br>,      | .96-96 1.   | .0E+01          | M/N        | •               | K/A           | •               | N/N        | •              | R/A       | •                | 2.796+06      | •                | '                  | 1.825-01        | PO DAY           |
| -   | - 36.6     | -06 1.06+     | 10          | A.                | -               | R/A         | ŗ           | <b>A</b> /A | •               | 1.16-06    | 1.0€+01         | N/N           | •               | 2.55-06    | 10+30-1        | 3.16-04 1 | 10+30.           | 3.846+04      | ,                | ı                  |                 | NM 04            |
| -   | - 16 -     | -06-1.06-     | ті<br>П     | Ą                 | -               | N/A         | ,           | N/A         | •               | N/A        | '               | A/A           | ,               | 8.16-05    | 1.05+01        | 4.25-05 1 | 10+30.           | 4.72E+04      | '                | •                  | NEG.            | PO BAY           |
|     | ÷.4        | -06 1.0E+     | Ĥ IO        | A,                | - 9.34          | 10.1 60-3   | 10+3        | N/A         | ,               | A/A        | •               | 5.06-06       | 1.0€+01         | 8. 1E-05   | 1.0£+01        | N/N       | •                | 1.216+05      | •                | •                  |                 | 90 BAY           |
| -   | **         | 66 1.06       | ýi ()       | é                 | 9.3             | E-06 1.0    | E+01        | N/A         | •               | N/N        | •               | N/A           | •               | 8. IE-05   | 1.0€+01        | 4.25-05 1 | 10+30.           | 6. 20E+04     | '                | •                  |                 | 100 BAT          |
| -   | - 36 -     | · 1.06 •      | а<br>10     | ¥,                | - 9.3           | E-06 1.0    | E+01        | <b>M/A</b>  | •               | N/N        | •               | N/N           | •               | 8.15-05    | 1.05+01        | 1 20-32.4 | 10+30.           | 1.005 +01     | '                | •                  |                 | TO DAT           |
| -   | <b>*</b>   | 4             | 1<br>1      | A)                | - <b>-</b><br>, | N/A         |             | A/M         | •               | N/N        | •               | W/W           | •               | 0. IE-05   | 1.0€+01        | 4.25-05 1 | 10+30.           | 4. 00E +04    | ,                | •                  | a<br>H          | 90 MI            |
| -   | 6. IE-     | -12 1.05-     | я́<br>Į     | A'                | н.<br>Н. А.     | E-05 1.0    | 10+3        | A/M         | ,               | 9.16-07    | 1.0£+01         | N/A           | •               | 1. XE-01   | 1.0€+01        | 1.16-31.1 | 10+30            | 1.046+04      | •                | '                  | 1.075+01        | 71 BMY           |
|     | 6. If-     | ·10.1 ču      | <b>N</b> 10 | é                 | × -             | 10.1 20-3   | E+01        | A/A         | ,               | 9.16-07    | 1.0£+01         | N/N           | •               | 1. 36-03   | 1.05+01        | 1.16-04 1 | 10+30.           | 3.706+04      | •                | •                  | 2. 7uE -02      | 90 DAY           |
|     | <b>W</b> . | 4             | ÷           | , A               | -               | N/A         | •           | N/A         | •               | N/A        | •               | N/A           |                 | N/N        | •              | 9.16-05 1 | 10+30.           | 1.346+04      | •                | •                  | 3               | 90 DAY           |
| -   |            |               | <u>ءَ</u>   | Á,                | -               | N/A         | ,           | N/A         | •               | N/A        | •               | <b>N/A</b>    | •               | 9.86-45    | 1.01.01        | #/#       | •                | 1.836+05      | ,                | •                  |                 | TAG 06           |
| a   | nctured .  | by forthi     | ft time du  | string Leaks      | t handly        | ing activ   | vities.     |             |                 |            |                 |               |                 |            |                |           |                  |               |                  |                    |                 |                  |
| ~   | 1          | -             | - 10        | ×.                | -               | N/A         | ,           | N/A         | ,               | N/A        | •               | M/A           | •               | 5.26-04    | 1. 36 +01      | 5.26-96 1 | 10+35            | 1.405+32      | •                | •                  | 4.266+00        | 19 - F           |
| ~   | · 4.4E-    | 45 1.36+      | (H 10       | ¥,                |                 | N/A         | •           | K/A         | •               | N/A        | •               | 4.46-05       | 1.36+01         | 4.46-05    | 1.36+01        |           |                  | 2.00€+02      | •                |                    | 1. 305-03       | Ĩ                |
| 2   | 1.16       | 45 1.3F+      | A 10        | Å.                | -               | N/A         | ,           | A/N         | •               | N/A        | ,               | A/N           | •               | 1.16-05    | 1.3£+01        | B/8       | •                | 1.806+01      | '                | '                  | 1.096-01        | *                |
| 2   | -1.16-     | 45 1.36+      | 01 IN       | đ,                | -               | 4/4         | ,           | R/A         |                 | A/A        | •               | 1.1E-05       | 1.36+01         | <b>N/A</b> | •              | A/N       | •                | 1.406+01      |                  | ,                  | 1.306-03        | Ŧ                |
| ~   |            | đ             | ¥           | ( <b>A</b>        | -               | N/A         | •           | N A         | •               | A/A        | •               | N/A           |                 | 0.06+00    | •              | A/N       | ,                |               |                  |                    |                 |                  |

See notes at and of table.

100 No. 100 No. 100 No. 100 No. 100 No. 100 No. 100 No. 100 No. 100 No. 100 No. 100 No. 100 No. 100 No. 100 No.

ENA EXEMPENSE EXECT

4

I-3

NA N

File: CONSTOK.UK1 date21-Aug-07 page 2

A Charles and a contraction

STORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EIISTIMS SITES

| 20-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10-00<br>10<br>10<br>10-00<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>1 |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 4.006-01<br>9.706-01<br>6.706-01<br>1.606-02<br>1.606-02                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| 10 X 11 10 X 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 10-37.1<br>10-37.1<br>10-37.1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 1. 26-95<br>1. 26-95<br>1. 26-95<br>1. 26-95<br>1. 26-95<br>1. 26-95                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| N/A<br>N/A<br>2.25-04<br>3.25-04                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 1. 36 •01<br>1. 36 •01<br>1. 36 •01<br>1. 36 •01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 6.06-05<br>6.06-05<br>8.06-05<br>3.26-04<br>3.26-04                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 4 4 4 4 4 4<br>2 2 2 2 2 2 2 2 2 2 2 2 2 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 1. 3 <b>E •01</b><br>1. 3 <b>E •01</b><br>1. 3E •01<br>1. 3E •01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 6. 06 -05<br>6. 06 -05<br>11/A<br>3. 26 -04<br>3. 26 -04                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| ~~~~~                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |

See notes at end of table.

<u> 1999 - 1999 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997</u>

STALES NEWS

<u>ر</u> الم

I-4



N





File: CONSTOR.MK1 date21-Aug-07 page 3

STORAGE ACCIDE

| tablel       |             |
|--------------|-------------|
| at bottom of |             |
| ts given     | 6 SITES     |
| equency uar  | AT EXISTIM  |
| DENTS - (Fr  | SMUTTINUM A |

| E LBAD RANGE NAAF RANGE<br>R Freg Factor Freg Factor |
|------------------------------------------------------|
|                                                      |
| - M/M - 1' 04 - 12 1' 04 - 12 1' 04 -                |
| - N/A - N/A -                                        |
| - N/A - N/A                                          |
| - N/A - N/A -                                        |
| - N/A - N/A                                          |
| - N/A - N/A                                          |
| - W/A - W/A -                                        |
| - N/A - N/A                                          |
| · 1.3E-10 1.0E+U1 M/A                                |
| - N/A - N/A                                          |
| - N/A - N/A                                          |
| - 1.3E-10 1.0E+01 N/A                                |
| - M/A - M/A                                          |
| - N/A - N/A                                          |
| - 1.3E-10 1.0E+UI N/A                                |
| - N/A - N/A                                          |
| - N/A - N/A                                          |
| - N/A - N/A                                          |
| - N/A - N/A                                          |
| - N/A - N/A                                          |
| · 1.3E-10 1.0E+01 N/A                                |
| - N/A - N/A                                          |
| - N/A - N/A                                          |
| - 1.36-10 1.0E+01 M/A                                |
| - N/A - N/A                                          |
|                                                      |
| ; fire not contained in 30 minutes                   |
| - N/A - N/A                                          |

See notes at end of table.

BEERE STATES NEW STATES CONTRACT

File: CONSTOR.UN1 date21-Aug-87 page 4

Sec. 2

SIGMAGE MCCIDENIS - ifrequency while given at bottom of table) From munitions at filting cites

|              |               |        |                  |                 |             |                 |          | Accident        | Fr equencia         | ĩ               |               |                 |              |         |                  |           |             |                | æ               | gent Avai | ) has eided         | le le ssed      |                  |
|--------------|---------------|--------|------------------|-----------------|-------------|-----------------|----------|-----------------|---------------------|-----------------|---------------|-----------------|--------------|---------|------------------|-----------|-------------|----------------|-----------------|-----------|---------------------|-----------------|------------------|
| SCENNE       | 9             | 9<br>9 | AMAD             | RANGE<br>FACTOR | AP6<br>FRED | RANGE<br>FACTOR | L BAD    | RANGE<br>FACTOR | KAAP<br>Free        | RANGE<br>FACTOR | P MA<br>F REQ | RANGE<br>FACTOR | PUDA<br>FREG | FACTOR  | IEA<br>Free      | FACTOR    | FRED        | ANNE<br>FACTOR | MGENT<br>AVAIL. |           | L DS.<br>DE TOMATED | LGS.<br>Emitted | BURATICK<br>TINE |
| 57.165 183.  | 181           | 5      | A/A              |                 | A VA        | ,               | 4/4      | ı               | N/N                 |                 | A/M           |                 | N/A          |         | 2. X -12         | 1.36+01   | N/N         |                | 1. 496 +05      | •         |                     | 1.496+04        | Ħ                |
| SUDIC 160    | đ             | 5      | 5.76-11          | 1.36+01         | 4 <u>2</u>  | ,               | N/N      | •               | R/A                 | •               | A.A           | ı               | N/A          | ,       | N.N              |           | B/A         |                | 1.576+04        | '         | B. 93£ +03          | 1.34E+03        | 20 NIM           |
| SLIDHC (B)   | )<br>B        | 'n     | 11-36.2          | 10:35-11        | N/A         | ,               | N/A      | •               | <b>B</b> / <b>B</b> | •               | N/N           | •               | 4.46-10      | 1.36+01 | 2.76-12          | 1.36+01   | N/N         | •              | 1. 956 +04      | •         | 1.246+04            | 1.846+03        | 20 NIN           |
| SLIDHIC (189 | <b>JG</b>     | ~      | N/A              | •               | N/N         | ,               | N/A      | •               | N/A                 | •               | N/N           | •               | A/A          | •       | 2. 76-12         | 1.3£+01   | N/N         |                | 1. 91E+04       | •         | 1.73€+04            | 2.59€+03        | 20 NIN           |
| SLCGC (40    | <b>(</b> 10)  | ŝ      | 5.76-11          | 1.36+01         | N/N         | •               | N/A      | •               | <b>M/A</b>          | •               | W/W           | •               | A/A          | ,       | A.M              | •         | N/N         |                | 7.206+03        | •         | 1.805+03            | 5.40€+02        | 20 NIN           |
| SLCGC 180'   | 191           | 'n     | 5.96-11          | 1.36+01         | N/N         | •               | A/A      | •               | N/N                 | •               | N/N           | •               | A/A          | •       | 2.76-12          | •         | N/N         | •              | 1. 79€+03       | •         | 2.45€+03            | 7.34E+02        | 20 AIM           |
| SLCGC (89'   |               | ŝ      | N/A              | •               | M/A         | •               | N/A      | •               | N/N                 | •               | A/N           | •               | W/W          | •       | 2.76-12          | •         | N/A         | •              | 1.466-04        | '         | 3.45€+03            | 1.095+03        | 20 NIN           |
| SLCHE 160    | <b>.</b>      | 5      | 5.76-11          | 1. 36+01        | A/A         | •               | M/A      | •               | A/A                 |                 | W/W           | •               | N/N          | •       | N/N              | ۰         | N/N         |                | 1.44€+04        | '         | 3.40€+03            | 5.40€+02        | ZO NIN           |
| SLCHC 180'   |               | *7     | 5.96-11          | 1.36+41         | N/N         | •               | N/A      | •               | A/A                 | 1               | N/N           | •               | 4.46-10      | 1.36+01 | N/A              | 1. XE +01 | N/N         |                | 1.965+04        | •         | 4.906+03            | 7.346+02        | 20 NIN           |
| SLCHC 189    |               | ŝ      | R/N              | '               | N/N         | •               | N/N      | ·               | N/A                 | '               | N/A           | •               | N/N          | •       | N/A              | 1.3E+01   | N/N         |                | 10+326-51       | •         | 7.306+03            | 1.095+03        | 20 MIN           |
| SLKGF 180    | Î             | ŝ      | <b>R/A</b>       | '               | R/A         | •               | N/A      | •               | N/A                 | •               | A/A           | •               | N/A          | •       | 2. 76-12         | 1.36401   | N/N         |                | 20+310.1        | •         | •                   | 2.07E+04        | ۹.               |
| SLENE 140    | 16.           | n      | 5.75-11          | 1.36+01         | N/N         | •               | N/A      | ٠               | N/N                 | •               | N/N           | •               | N/A          | ,       | N/N              | •         | N/A         | •              | 1.466+05        | •         | •                   | 7.31E+03        | Ξ.               |
| SLAHF (OPE   | â             | ş      | N/A              |                 | 2.16-09     | 1.0€+01         | N/A      | ,               | N/N                 | •               | 7.96-09       | 1.0€+01         | A/N          | •       | 2.76-12          | 1.05+01   | N/A         | ਰ<br>'         | .A55.           | •         | •                   | 4. BUE+04       | 91.              |
|              |               | ~      | N/N              | •               | N/A         | •               | N/A      | •               | #/#                 | •               | N/N           | •               | A/A          | •       | <b>W</b> /W      | •         | 1.46-00 1.  | .16+31.        | 5. 39€ +04      | •         | ,                   | 2.49€+05        | 9                |
| SLEVE (BO'   | [ <b>6</b> ]) | ŝ      | N/N              | •               | N/A         | •               | N/N      | •               | A/A                 | •               | N/N           | •               | A/N          | •       | 2. <b>TE-1</b> 2 | 1.36+01   | N/A         | ,              | 2.216+05        | •         | •                   | 5.52£+03        | ۳.               |
| Surve (m)    |               | ŝ      | R/R              | •               | A/A         | •               | N/A      |                 | 3.86-09             | L. IE+0I        | <b>W/W</b>    | •               | A/A          | •       | N/N              | •         | <b>N/N</b>  | , ,            | 2.996+06        | •         | `                   | 7.446+04        | £                |
| SLINC 160    | 3             | *7     | 5. Xf - 11       | 1.36+01         | <b>A</b> /M | •               | N/N      | •               | 6/N                 | •               | N/A           | •               | N/A          | •       | A/M              | •         | N/A         | ,              | 7.72E+04        | •         | 4.806+03            | 5.10€+02        | 20 MIN           |
| SLIPVE 180   | Ē             | 5      | 5.96-11          | 1.36+01         | N/N         | •               | N/N      | •               | A/A                 | •               | 1.16-11       | 1. XE •01       | N/N          | •       | 2. 12-12         | 1. X +01  | 1. IE-10 1. | . 16+3C.       | 10+346-51       | ,         | 9.446+03            | 1.236+02        | 20 NIN           |
| 091 39475    | 3             | ~      | 5. 78 - 11       | 1.36+01         | NA          | •               | N/N      | •               | N/A                 | •               | N/A           | •               | N/N          | •       | N/A              | •         | R/A         | - 1            | 3.59€+04        | •         | 8.97E+03            | 2.49€+03        | 20 MIN           |
| SLPGC 180    | Ĵ             | 5      | 5.16-11          | 1.36+01         | N/N         | •               | R/A      | •               | N/A                 | •               | R/N           | •               | N/N          | •       | 2. 76-12         | 1. 36 401 | 1.16-10 1.  | 36+01 5        | 5.02E+04        | •         | 1.266+04            | 3.776+03        | 20 NIN           |
| 51.PSC 189   | Ĵ             | 'n     | A/M              | •               | N/N         | •               | N/A      |                 | N/N                 |                 | M/A           | •               | A/N -        | •       | 2.76-12          | 1.36+01   | N/N         |                | 1.725+04        | '         | 1.486+04            | 5.04£+03        | 20 AIN           |
| 51.PMC 140   | Ĵ             | 'n     | 5. 76 - 11       | 1. 36 • 01      | 8/8         | •               | N/A      | •               | A/4                 | •               | <b>W/A</b>    | •               | N/N          | •       | N/N              | •         | #/#         | -              | 1.446+04        | •         | 1.416+04            | 2.426+03        | 20 N.M           |
| SLPHC (BU    | Ĵ             | 5      | 11-36-2          | 1. 36 • 61      | N/N         | •               | N/A      | •               | N/N                 | ,               | N/N           | •               | 4.4E-10      | 1.3£+01 | 2.76-12          | 1.3£+01   | N/N         |                | 9.045+04        | '         | 2.246+04            | 3.39€+03        | 20 NIN           |
| SLPHC 189    | Ĵ             | ŝ      | R/A              | •               | N/N         | •               | 3.46-11  | 1.36+01         | 8/8                 | •               | N/N           | •               | N/N          | •       | 2.76-12          | 1. 3E +01 | N/N         | ,              | 1.216+05        | •         | 3.026+04            | 4.536+03        | 20 AIN           |
| 51.PVC 160   | 161           | **     | 5. <i>7</i> E-11 | 1. XE+01        | 4/¥         | •               | R/A      | •               | <b>A</b> / <b>A</b> | ·               | A/M           | ٠               | N/N          | •       | N/N              | •         | N/N         | '              | 3. JIE +04      | •         | <b>B. 206 +0</b> 3  | 4.21E+02        | 20 NIN           |
| 981 JAA'15   | Ĵ             | 'n     | 5.96-11          | 1.36+01         | <b>W</b> /W | •               | N/A      | •               | A/A                 | •               | N/N           | •               | N/N          | •       | 2.76-12          | 1.36.01   | 1.16-10 1.  | 32+01 4        | 1,416+04        | '         | 1.146+04            | 8.496+02        | ZO NIN           |
| SLPVC 109    | đ             | ÷      | W/W              | •               | N/A         | •               | 3. # -11 | 1.36+01         | N/A                 | •               | N/N           | '               | N/N          | '       | 2.76-12          | 1.36.101  | N/N         | •              | 1.206+01        | •         | 1.33€+04            | 1.16€+03        | 20 NIN           |
| 21.065       | â             | n      | 5. 76 - 11       | 1. XE +01       | N/N         | •               | N/N      |                 | A/N                 | •               | A/N           | ,               | W/W          | •       | N/N              | •         | N/N         | '              | 5, 296 +04      | •         | 8.225+03            | 2.47E+03        | 20 NIN           |

1

Y. - 1 Υ.

5551 LANSS

<u>ন</u>্থিন মূল

kard secon "and " second beau

See notes at end of table.







MANY AND A

File: CONSTOR.MK1 date21-Aug-87 page 5

# SIORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUNITIONS AT EIISTING SITES

| - 2 |
|-----|
| - ē |
| . 2 |
| ž   |
| •   |
|     |
| _   |
| ~   |
| 2   |
| -   |
|     |
|     |
| 3   |
| Ę   |

| $ \begin{array}{c c c c c c c c c c c c c c c c c c c $                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |    |          |              |                 |              |                 | -             | kcident Fi      | a rouente :  |                     |             |                 |               |                 |                   |                    |            |               | -               | Agent Avai      | lable and         | Released        |                                        |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----|----------|--------------|-----------------|--------------|-----------------|---------------|-----------------|--------------|---------------------|-------------|-----------------|---------------|-----------------|-------------------|--------------------|------------|---------------|-----------------|-----------------|-------------------|-----------------|----------------------------------------|
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |    | G        | Anað<br>Freg | KANGE<br>FACTOR | AP6<br>Frego | RANGE<br>Factor | L BAD<br>FRED | RANGE<br>Factor | NAAP<br>FRED | RANGE<br>Factor     | PBA F       | RAMGE<br>FACTOR | PUDA<br>FIRED | RANGE<br>FACTOR | TEAB R            | AARGE<br>Actor     | UNDA 1     | ANGE<br>ACTOR | AGENT<br>Avail. | LBS.<br>SPILLED | LDS.<br>Detomated | LBS.<br>Emitted | DURAT I DH<br>T 1 NE                   |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 1  | 1        |              |                 |              | •               |               |                 |              |                     |             |                 |               |                 |                   |                    |            |               |                 |                 |                   |                 |                                        |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 2  | ŝ        | 5.96-11      | 10+32.1         | A/A          | •               | N/A           | ,               | 4/N          |                     | N/A         | •               | N/A           | ,               | P. 7E-12 1.       | JE+01 1            | .i 61-31.  | 10+3r         | 1.595+04        | 4               | 1.156+04          | 3.45€+03        | 20 NIN                                 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 2  | \$       | M/N          |                 | A/N          | ,               | 3.46-11-1     | . 36 +01        | A/A          | ,                   | N/A         |                 | N/N           | •               | 2.76-12 1.        | 10+ <u>3</u> E     | N/A        | ·             | b. 00E+04       | •               | 1.50E+04          | 4.50€+03        | 20 NIN                                 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | â  | -        | N/A          |                 | N/N          |                 | N/A           | •               | M/A          |                     | M/A         | •               | N/A           |                 | A/A               | •                  | A/A        | •             | 3.296+04        | ·               | 8.22E+03          | 6.176+02        | 20 MIM                                 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | æ  | \$<br>•  | N/A          | •               | N/A          |                 | N/A           |                 | N/A          | •                   | M/A         |                 | N/A           | ,               | P. 7E-12 1.       | 36+01 1            | .1 01-31.  | 36+01         | 1.595+04        | •               | 1.156+04          | B. 61E+02       | 20 MIN                                 |
| (B)         5 $3.71 + 11$ $1.11 \times 10$ $MA$ $1.16 + 11$ $1.16 + 11$ $1.16 + 11$ $1.16 + 11$ $1.16 + 11$ $2.16 + 11$ $2.16 + 11$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$ $3.16 + 10$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 3  | •<br>-   | M/A          | •               | M/A          | •               | N/A           | ,               | N/A          |                     | M/A         |                 | N/A           | ,               | 2.76-12 1.        | 36+01              | M/A        | •             | 6.00E+04        | •               | 1.50E+04          | 1.13£+03        | 20 MIN                                 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | ਡ  | ~        | 5. 7E-11     | 1.35+01         | A/A          | •               | A/N           |                 | N/A          | •                   | A/N         | •               | N/A           | •               | N/A               |                    | A/A        | ,             | 2.126+04        | '               | 5.306+03          | 1.59€+03        | 20 MLM                                 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | đ  | 5        | 5.96-11      | 1. 3€+01        | A/A          | •               | 4/4           |                 | N/A          |                     | 1.16-11 1.  | 36+01           | N/A           | •               | P. 7E-12 1.       | 1 10+ X            | .1 01-31.  | 36+01         | 2.826+04        | •               | 7.066+03          | 2. I 2£ +03     | 20 MIM                                 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | ŝ  | 5        | M/A          |                 | N/N          | •               | 3.46-11 1     | . 36 • 01       | M/A          | •                   | A/A         | ,               | N/A           |                 | 2.76-12 1.        | 3E+01              | N/A        |               | 1.04E+04        | •               | 1.016+04          | 3.03£+03        | 20 MIN                                 |
| 6L:       5       5.KE-11       1.KF-10       1                                                                                                                                                                                                                    | ŝ  | 5        | 5. XE - LI   | 1. 35 +01       | A / A        |                 | N/A           |                 | N/A          |                     | A/A         | •               | N/A           | •               | N/A               | •                  | A/A        | •             | 1.986.04        | •               | 4.956+03          | 3. 71E+02       | 20 MIN                                 |
| (a)       5       N/A       -       N/A       -       N/A       -       1.6F-11       1.2F-01       1.7F+02       7.0F+02       7.0F+02<                                                                                                                                                                                                                             | ŝ  | ~<br>-   | 5.96-11      | 1.35.01         | 414          | •               | 47 <b>4</b>   | ,               | 4 / H        | ,                   | 1.11-31.1   | X +01           | W/W           | ,               | 2.7E-12 1.        | 10+30              | .1 01-3L.  | 36+01         | 2.44E+04        | '               | 6.60E+03          | 4.95€+02        | 20 NIN                                 |
| 61       5       NA       E       NA       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F       F <td>j,</td> <td>•••</td> <td>N/A</td> <td>•</td> <td>8/8</td> <td>•</td> <td>3.46-11</td> <td>. 36 + 01</td> <td>M/A</td> <td>•</td> <td>N/A</td> <td>•</td> <td>A/A</td> <td>,</td> <td>R-12 1.</td> <td>. 3E+01</td> <td>N/A</td> <td>1</td> <td>3.786+04</td> <td>•</td> <td>9.456+03</td> <td>7.095+02</td> <td>20 MIN</td>                                                                                                                                                            | j, | •••      | N/A          | •               | 8/8          | •               | 3.46-11       | . 36 + 01       | M/A          | •                   | N/A         | •               | A/A           | ,               | R-12 1.           | . 3E+01            | N/A        | 1             | 3.786+04        | •               | 9.456+03          | 7.095+02        | 20 MIN                                 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 3  | ~<br>~   | A/M          |                 | A/A          |                 | N/A           | •               | A/N          | 1                   | N/A         | •               | N/A           | •               | N/A               | •                  | .1E-10 1.  | 3E+01         | 1.366+04        | •               | •                 | 3. 39£+02       | Ĩ                                      |
| add generated alsolies strite the storage againet, aarehouse, or open storage area; aunitions breached ino defanation.       a. N.A       v.A       <                                                                                                                                                                                                                                                                                                                            |    | ŝ        | <b>8/8</b>   | •               | M/M          | •               | N/A           | ı               | A/A          | •                   | A/A         | •               | M/A           | •               | . 0E-10 1.        | .1E+01             | N/A        | •             | 1.836+05        | •               | •                 | 4.586+03        | ١<br>٣                                 |
| b         N/A         ·         2.554601         6 486         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         · </td <td>ğ</td> <td>a gene</td> <td>rated aiss</td> <td>ales strik</td> <td>e the stor</td> <td>rage sagar</td> <td>ine, wireh</td> <td>ouse, or t</td> <td>pen stor.</td> <td>ide area;</td> <td>6401 £1083</td> <td>1 breached</td> <td>I ino deti</td> <td>mation).</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> | ğ  | a gene   | rated aiss   | ales strik      | e the stor   | rage sagar      | ine, wireh    | ouse, or t      | pen stor.    | ide area;           | 6401 £1083  | 1 breached      | I ino deti    | mation).        |                   |                    |            |               |                 |                 |                   |                 |                                        |
| b       4.86-12       9.46-01       1/4       >       1/36-02       6/48         b       1.176-12       9.46-01       1/4       >       1/36-02       1/36-02       6/48         b       1.176-12       9.46-01                                                                                                                                                                                                                                                                                                                                                                                                                              |    | -0       | N/A          | •               | N/A          | ,               | N/A           | •               | N/A          | •                   | N/A         |                 | N/A           | '               | i. SE-15 9.       | 1 10+34            | .26-15 9.  | 10+31         | 4.40£+02        | •               | •                 | 2.566+01        | 11111111111111111111111111111111111111 |
| b       4.86-12       9.46-01       1/4       -       1/4       -       1/06-00       4.86         R1       9.48-12       9.46-01       1/4       -       1/4       -       1/06-00       4.86         R1       b       4.86-12       9.46-01       1/4       -       1/4       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       - <td></td> <td>-0</td> <td>4, 85-12</td> <td>9.4E+01</td> <td>N/A</td> <td>,</td> <td>N/A</td> <td></td> <td>N/A</td> <td>•</td> <td>N/A</td> <td>-</td> <td>. ZE-13 9.</td> <td>10+3</td> <td>. BE-15 9.</td> <td>4E+01</td> <td><b>X/A</b></td> <td>,</td> <td>2.886+02</td> <td>•</td> <td>•</td> <td>1.30E-02</td> <td>¥ 9</td>                                                                                                                                               |    | -0       | 4, 85-12     | 9.4E+01         | N/A          | ,               | N/A           |                 | N/A          | •                   | N/A         | -               | . ZE-13 9.    | 10+3            | . BE-15 9.        | 4E+01              | <b>X/A</b> | ,             | 2.886+02        | •               | •                 | 1.30E-02        | ¥ 9                                    |
| b       4.86-12       9.46-10       k/A       k/A       5.76-13       8.46       k/A       6.06+01       c       1.306-02       6.48         (b)       b       k/A       k/A       k/A       k/A       k/A       1.366-03       c       1.316+01       6.44         (b)       k/A                                                                                                                                                                                                                                                                                                                                                                                              |    | 4        | 4, 86-12     | 9.4E+01         | 4 / H        | '               | N/A           | •               | A/A          | ,                   | A/A         | •               | A/M           | •               | i. BE-15 9.       | 10+3t              | N/N        | ,             | 1.805+01        | •               | •                 | 1.09€+00        | 9119                                   |
| B1       b       N/A       -       N/A       -       N/A       -       1.306+03       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |    | •        | 4.86-12      | 9.4E+01         | N/A          |                 | N/A           | ,               | N/A          | •                   | N/A         | ,<br>,          | 1.26-13 9     | .4E+01          | R/A               |                    | N/A        |               | 9.405+01        | '               | •                 | 1.306-02        | 6 H 9                                  |
| B1       b       1.72-12       9.46-01       N/A       -       N/A       -       1.706+03       -       -       -       6 446-01       6 446         b       N/A       -       6.46-11       9.46-01       N/A       -       1.726-12       6.46-01       N/A       -       1.706+03       -       -       -       6 446-01       6 446         b       N/A       -       N/A       -       N/A       -       N/A       -       1.726-15       4.66-01       1.706-03       -       -       -       -       6 446         6.1       b       N/A       -       N/A       -       N/A       -       1.706-03       -       -       1.446-01       6 446         6.1       b       N/A       -       N/A       -       N/A       -       1.706-03       -       1.446-01       6 446       -       -       1.446-01       6 446       -       1.446-01       6 446       -       1.446-01       6 446       -       1.446-01       6 446       -       1.446-01       6 446       -       1.446-01       6 446       -       1.446-01       6 446       -       1.446-01       -       1.446-01       <                                                                                                                                                                                                                                                                                                                                                                                       | đ  | •        | W/W          | •               | N/A          | •               | N/A           |                 | N/A          |                     | A/A         | •               | M/A           | ,               | 2.4E-15 9.        | 10+31 <sup>-</sup> | R/A        | ,             | 1.50E+03        | '               | 1                 | 3.71E+01        | 9 H                                    |
| b         N/A         -         6.6E-11         9.4E-10         N/A         -         N/A         -         1.2E-12         9.4E-01         N/A         -         1.4E-01         A is           6.1         b         N/A         -         N/A         -         N/A         -         1.70E+03         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -                                                                                                                                                                                                                                                                                                                                                                    | ŝ  | <b>9</b> | 1. ZE - 12   | 9.46+01         | N/A          | •               | M/A           |                 | M/A          |                     | N/A         | ,               | R/A           | •               | N/A               | •                  | N/4        | ,             | 1.70€+03        | '               | •                 | 1.44E-01        | 6 HE                                   |
| b         N/A         I/A                                                                                                                                                                                                                                                                                          | -  | 4        | A/A          | •               | 6.6E-11      | 9.4E+UI         | M/A           | •               | A/A          | •                   | 9.96-10 9.  | 10+3+           | - N/A         | •               | . 21-32.          | 10+34              | N/A        | ,             | 1.70€+03        | 1.70€+03        | •                 | ,               | ۹<br>۲                                 |
| 6L:       b       N/A       -       N/A       -       N/A       -       1.40€+03       -       -       -       1.40€+03       -       -       1.40€+03       -       -       1.40€+03       -       -       1.40€+03       -       -       1.40€+03       -       -       -       1.40€+03       -       -       -       1.40€+03       -       -       -       1.40€+03       -       -       -       1.40€+03       -       -       -       1.40€+03       -       -       -       -       1.40€+03       -       -       -       -       -       1.40€+03       -       -       -       -       -       1.40€+03       -       8.48€+12       -       -       -       1.40€+03       -       -       -       -       -       1.40€+03       5.46€+03       -       -       -       -       -       -       -       -       -       -       1.40€+03       -       8.48€+12       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       <                                                                                                                                                                                                                                                                                                                                                                                                                      |    | 4        | N/A          | •               | A/N          | •               | A/A           |                 | N/A          | •                   | R/A         | •               | N/A           | •               | N/N               | -                  | .96-13 9.  | 10+34         | 1.70€+03        | •               | •                 | 1.44E-01        | ¥<br>1                                 |
| b         N/A         -         N/A         -         N/A         -         N/A         -         N/A         -         1.60E+03         5 we           6         4.8E-12         9.4E+01         N/A         -         N/A         -         1.60E+03         5 we           6         4.8E-12         9.4E+01         N/A         -         N/A         -         1.50E+03         5 we           6         4.8E-12         9.4E+01         N/A         -         N/A         -         N/A         -         -         -         -         2.30E+03         6 we           6         4.8E-12         9.4E+01         N/A         -         N/A         -         N/A         -         5.6E+05         6.6E+01         0.75         -         -         -         2.30E+03         6 we           6         4.8E-12         9.4E+01         N/A         -         N/A         -         N/A         -         2.30E+02         0.8E+01         0.8E+15         0.4E+01         0.75         0.4E+01         0.75         0.4E+01         0.8E+15         0.4E+01         0.75         0.4E+01         0.75         0.4E+01         0.8E+12         0.4E+01         N/A         -                                                                                                                                                                                                                                                                                                        | 3  | <b>9</b> | M/A          | •               | A.A          |                 | R/A           | ,               | A/A          | ,                   | N/A         | •               | N/A           | ,               | 2.4E-15 9.        | 4E+01              | N/A        | •             | 1.40€+03        | '               | •                 | 1.40£-03        | ¥ 9                                    |
| 6       4.86-12       9.46-01       N/A       N/A       N/A       -       2.306-04       6.16         6       4.86-12       9.46-01       N/A       -       N/A       -       1.35-14       9.46-01       5.86-15       9.46-01       3.86-16       -       -       -       2.306-04       6.16         6       4.86-12       9.46-01       N/A       -       N/A       -       N/A       -       5.66-10       6.16       1.47-01       5.66-10       -       -       5.66-10       6.16         6       4.86-12       9.46-01       N/A       -       4.46       5.66-15       4.60       5.66-10       -       -       -       2.206-12       6.16         6       4.86-12       9.46-01       N/A       -       1.4       -       5.86-15       9.46-01       5.86-15       9.46-01       -       -       2.206-12       2.306-10       5.86       1.46-01       5.86-15       9.46-01       -       -       -       2.206-12       2.306-10       5.86-15       9.46-01       5.86-15       9.46-01       5.86-15       9.46-01       5.86-15       -       -       2.206-10       2.206-10       2.306-10       -       - <t< td=""><td></td><td>•</td><td>N/A</td><td></td><td>N/A</td><td>•</td><td>N/A</td><td>-</td><td>. XE-10 9.</td><td>10+31</td><td>A/N</td><td></td><td>N/A</td><td></td><td>N/A</td><td></td><td>A/A</td><td>,</td><td>1.40€+03</td><td>•</td><td>'</td><td>1.406-03</td><td>۹<br/>۲</td></t<>                                                            |    | •        | N/A          |                 | N/A          | •               | N/A           | -               | . XE-10 9.   | 10+31               | A/N         |                 | N/A           |                 | N/A               |                    | A/A        | ,             | 1.40€+03        | •               | '                 | 1.406-03        | ۹<br>۲                                 |
| 6 4.86-12 9.46-01 N/A N/A N/A N/A N/A N/A N/A N/A N/A 5.866-15 9.46-01 5.66-15 9.46-01 4.726-04 - 5.5666-00 6 46<br>6 4.86-12 9.46-01 N/A 4.86-12 9.46-01 N/A - N/A - 3.26-13 9.46-01 5.66-15 9.46-01 N/A - 1.216-05 - 2.206-12 6 4<br>6 4.86-12 9.46-01 N/A - 4.86-12 9.46-01 N/A - N/A - N/A - 5.86-15 9.46-01 5.86-15 9.46-04 5.206-04 6 44                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |    | •        | 4.86-12      | 9.45+01         | N/A          |                 | M/A           | ,               | N/A          | <del>ب</del> ت<br>ا | 1. XE-12 9. | 46+01           | A/A           | •               | . 35-14 9.        | 10-31              | .86-15 9.  | 10+34         | 3.866+04        | •               | •                 | 2.306-04        | 9 H                                    |
| b 4.06-12 9.46-01 N/A - 4.06-12 9.46-01 N/A - N/A - 3.206-13 9.46-01 5.06-15 9.46-01 N/A - 1.216-05 - 2.206-12 à N<br>b 4.06-12 9.46-01 N/A - 4.06-12 9.46-01 N/A - N/A - N/A - 5.06-15 9.46-01 5.06-16 6.206-04 - 2.206-04 å N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |    | -4       | 4.86-12      | 9.4E+01         | N/A          | •               | N/A           | •               | N/A          | •                   | N/A         | •               | N/A           | ,               | .86-15 9.         | 10+34              | .8 21-38.  | 4E+01 4       | 6.72E+04        | ,               | •                 | 5.60€+00        | ¥<br>9                                 |
| b 4.18f 12 9.46f-01 M/A - 4.86f-12 9.46f-01 M/A - M/A - 5.86f-15 9.46f-01 5.86f-15 9.46f-01 5.206f-04 2.206f-04 5.46f                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |    | -0       | 4.06-12      | 9.46+01         | A/A          | ÷               | 4.86-12.9     | . 4E+01         | N/A          | •                   | N/A         | -               | 1.26-13 9     | 10+3+           | . <b>BE-15</b> 9. | 10-31              | 8/8        | •             | 1.216+05        | '               | •                 | 2.206-02        | 筆 1                                    |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |    | •        | 4.86 -12     | 10+3+.9         | N/A          | •               | 4.86-12 9     | .46+01          | N/A          |                     | N/A         | ٠               | W/W           | ,               | 6.86-15 9.        | 46+01 5            | .86-15 9.  | 10+34         | 6.20E+04        | •               | •                 | 2.206-04        | 9<br>1<br>1<br>1                       |

See notes at end of table.

<u> 1888 - 1888 - 1888 - 1888 - 1888 - 1888 - 1888</u>

N. T. Strate

Ξ

kan taata taata kaal

.

بعوارت بمرجع ومحاج وتعاويتهن

File: CONSTOR.WK1 date21-Aug-87 page 6

K.

STORMSE ACCIDENTS - (Frequency waits given at bottom of table) FOR MUNITIONS AT EXISTING SIFES

| 3 |  |  |
|---|--|--|
|   |  |  |
|   |  |  |
| ļ |  |  |
| 5 |  |  |
|   |  |  |
|   |  |  |
|   |  |  |
| Ē |  |  |
|   |  |  |
|   |  |  |
| • |  |  |
| ł |  |  |
| • |  |  |
|   |  |  |

|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |        |               |             |             |                 |                | Accident F.     | r equenci e  | -               |               |                 |              |                 |              |                 |               |              | ~               | igent Avas      | lable and         | Rel eased    |                                                                                                  |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------|---------------|-------------|-------------|-----------------|----------------|-----------------|--------------|-----------------|---------------|-----------------|--------------|-----------------|--------------|-----------------|---------------|--------------|-----------------|-----------------|-------------------|--------------|--------------------------------------------------------------------------------------------------|
| A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A         A                                                                                                                                                                                                                                                                                                                                                                                                                                                      | - :    | D. AWAD       | FANGE       | APS<br>FREQ | RANGE<br>FACTOR | L BAD<br>F REQ | RANGE<br>Factor | MAAP<br>FRED | RANGE<br>Factor | PLA F         | RANGE<br>Factor | PUDA<br>FREQ | RANGE<br>Factor | TEAD<br>FREQ | RANGE<br>FACTOR | UNDA<br>FRED  | <b>NUCE</b>  | AGENT<br>AVAIL. | LISS.<br>LILLED | LBS.<br>Detowated | LIS.<br>LIS. | DURATION<br>Time                                                                                 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |        | - <b>1</b>    | 12 9.46+01  | <b>1</b> 14 |                 | 4.12           | 10.10           | 4/1          | ,               | 4/1           | ,               |              |                 |              | 104.34          | 0 51-30 S     | 101          | 10-200          |                 | •                 | 100100       | 9                                                                                                |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |        |               |             | A.4         |                 |                |                 |              |                 |               | •               |              |                 |              |                 |               |              | DAG LOA         |                 |                   | - 30C C      | 9                                                                                                |
| No.         No. <td></td> <td></td> <td>12 0 46+01</td> <td></td> <td></td> <td>1 05.17</td> <td>0 46 401</td> <td></td> <td>•</td> <td>1 0C-11 0</td> <td>40.401</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>046404</td> <td></td> <td>•</td> <td>1 61CLAN</td> <td>E 9</td>                                                                 |        |               | 12 0 46+01  |             |                 | 1 05.17        | 0 46 401        |              | •               | 1 0C-11 0     | 40.401          |              |                 |              |                 |               |              | 046404          |                 | •                 | 1 61CLAN     | E 9                                                                                              |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |        |               | 10.11.0     |             |                 |                |                 |              | •               |               |                 |              | ,               |              |                 |               |              |                 |                 | •                 | 1.336.700    | E :                                                                                              |
| $\mathbf{k}$                                                                                                                                                                                                                                                                                                                                                                                                           |        |               | 10.4.6.71   |             |                 | 9.06-12        | Y. 41 + UI      | H/H          | •               | 1. 11 - 11 9. | . +E +01        |              | •               |              |                 | 5. HE - 15 %  | 10+34        | . 786 +04       | •               | •                 | 2.80E-04     | ¥ •                                                                                              |
| $\mathbf{k}$                                                                                                                                                                                                                                                                                                                                                                                                           | â      | P #//         | '<br>       | N/N         | ,               | A/A            | •               | N/A          | •               | A/A           | •               | N/A          | ,               | N/A          | ,               | 3, 56 - 15 9. | 1 10+3+      | . 366 +04       | '               | •                 | 1.406-03     | ¥<br>9                                                                                           |
| Rescripture bractical transform         MA         1.46-04 1.5601         Additation         2.546-01         Additation           7         3.66-01         3.64         1.46-04 1.5601         7.66-01         3.66-01         3.66-01         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66         4.66 <t< td=""><td></td><td><b>6 R</b>/À</td><td>•</td><td>N/N</td><td>•</td><td>N/A</td><td>•</td><td>#/#</td><td>,</td><td>N/A</td><td>•</td><td>N/N</td><td>•</td><td>1. ZE-13 9</td><td>·#E+01</td><td>N/A</td><td>•</td><td>CO+3KB.</td><td>•</td><td>•</td><td>I.40E-03</td><td>또<br/>1</td></t<> |        | <b>6 R</b> /À | •           | N/N         | •               | N/A            | •               | #/#          | ,               | N/A           | •               | N/N          | •               | 1. ZE-13 9   | ·#E+01          | N/A           | •            | CO+3KB.         | •               | •                 | I.40E-03     | 또<br>1                                                                                           |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Ke ear | thquake br    | reaches the | auni tions  | In storage      | e igloo; n     | o detonatio     | DAS.         |                 |               |                 |              |                 |              |                 |               |              |                 |                 |                   |              |                                                                                                  |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |        | T N/A         | ,<br>,      | A/A         | •               | M/M            |                 | A/A          | •               | W/W           | ·               | N/A          | •               | 1.66-06 1    | . JE+01         | 7.06-08 1.    | 3E+01        | . 496 +05       | '               | •                 | 2.546+01     | à tà                                                                                             |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |        | 7 3.06-0      | 10-3E-01    | N/A         | •               | M/M            |                 | M/A          | ,               | N/A           | •               | 3.06-08 1    | 10+32.          | 7.06-07 1    | . JE +01        | A/A           | -0           | 40+316-"        | '               | '                 | 7.806-03     | ¥                                                                                                |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |        | 7 7.05 0      | 10 1.36 +01 | N/A         | ŀ               | N/A            | •               | M/A          | •               | A/A           | •               | N/A          | •               | 1.46-07 1    | . JE +01        | N/A           | ,            | 101 391-        | •               | •                 | 4.53E-01     | Ŧ ~                                                                                              |
| R.1       W.A       W.A       W.A       W.A       W.A       L.1.E-03       L.2.075405       L.2       C.3.715401       A MB         R.1       7       4.46-07       1.35401       W.A       -       W.A       -       1.464-05       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |        | 7 7.05-0      | 19 1.36+01  | N/A         | •               | N/A            | •               | N/A          | •               | N/A           | •               | 7.06-09 1    | 10+37.          | A/A          | ı               | N/A           | - 3          | 10+326"         | '               | •                 | 2.076-03     | ¥<br>-9                                                                                          |
| 8.1       7       4.65-07       1.156-01       k/A       5       N/A       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05       5       1.465-05                                                                                                                                                                                                                                                                                                                                                                                     | Ĩ      | 3 N 4         | •           | A S M       | '               | 81W            | •               | 4/X          | •               | N/N           | •               | A/M          | •               | 1.1E-U5 1    | 10·X.           | N/A           | ,            | .076+05         | '               | •                 | 3.716+01     | 9<br>9                                                                                           |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 9F)    | 1 4.66-6      | 07 1.36+01  | N/N         | •               | N/A            | •               | N/A          | •               | N/A           | •               | M/A          | •               | N/A          | •               | N/A           |              | .466+05         | •               | •                 | 1.445-01     | 9<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |        | 7 N.A         | •           | 0.0E+00     | •               | N/A            | •               | N/A          | •               | 0.0€+00       | •               | N/A          | ,               | 0.0€+00      | •               | N/N           | <b>1</b> 0 · | ASS.            | '               | '                 | •            | '                                                                                                |
| 8.)       7       N/A       -       N/A       -       N/A       -       1.16-05       1.154-01       N/A       -       1.066-05       Å       N/A       -       1.066-05       Å       N/A       -       1.066-05       Å       -       1.066-05       -       1.066-05       -       1.066-05       -       1.066-05       -       1.066-05       -       1.066-05       -       1.066-05       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -<                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |        | 7 N/A         | •           | N/A         | •               | N/A            | ,               | N/A          | •               | A/A           | •               | N/A          | •               | N/A          |                 | M/A           | ŝ            | . 396 +06       |                 |                   | •            | •                                                                                                |
| 7       N(A       -       N(A       -       N(A       -       N(A       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | ŝ      | 7 N/A         | •           | N/A         | •               | N/A            | •               | N/A          | •               | <b>N/A</b>    | •               | A/A          | •               | 1.16-05 1    | 36+01           | M/A           | ~<br>·       |                 | ,               | •                 | 1.606-03     | 塗っ                                                                                               |
| 7       1.8E-08       1.3E+01       K/A       K/A       K/A       K=08       1.3E+01       1.8E+08       1.3E+01       1.9E+01       1.7E+01       1.8E+08       1.3E+01       1.8E+08       1.3E+01       1.8E+08       1.2E+01                                                                                                                                                                                                                                                                                         |        | 7 N/A         | •           | N/A         | •               | M/N            |                 | N/A          |                 | N/A           | •               | N/A          | ,               | N/N          | ,               | N/A           | - 7          | 90+366 -        | ,               | •                 | •            | •                                                                                                |
| 7       0.66+00       k/A       k/A <td< td=""><td></td><td>7 1.86-0</td><td>10+321-80</td><td>N/A</td><td></td><td>M/A</td><td>•</td><td>N/A</td><td>•</td><td>1.86-08 1.</td><td>.3E+01</td><td>N/A</td><td>•</td><td>(.1E-07 1</td><td>10+31</td><td>1.06-00 1.</td><td>3€+01 ]</td><td>. 846+04</td><td>•</td><td>'</td><td>1,006-04</td><td>9¥ 9</td></td<>                                                                                                                               |        | 7 1.86-0      | 10+321-80   | N/A         |                 | M/A            | •               | N/A          | •               | 1.86-08 1.    | .3E+01          | N/A          | •               | (.1E-07 1    | 10+31           | 1.06-00 1.    | 3€+01 ]      | . 846+04        | •               | '                 | 1,006-04     | 9¥ 9                                                                                             |
| 7 0.06+00 • N/A • 0.06+00 • N/A • N/A • 0.06+00 • 0.06+00 • 0.06+00 • 1.216+03 • • • • • • • • • • • • • • • • • • •                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |        | 7 0.6E+U      | . 2         | N/N         | •               | N/A            |                 | N/A          | •               | N/A           | ,               | N/A          | •               | 0.0€+00      | •               | 0.05+00       | -0           | - 726+04        | •               | •                 | •            | •                                                                                                |
| 7       0.0E+00       k/A       0.0E+00       k/A       0.0E+00       k/A       2.0E+00       k/A       k/A <td></td> <td>7 0.06+0</td> <td>-<br/>9</td> <td>N/A</td> <td>•</td> <td>0.0€+00</td> <td></td> <td>N/A</td> <td>•</td> <td>N/A</td> <td>'n</td> <td>0.0E+00</td> <td></td> <td>0.0€+00</td> <td></td> <td>W/A</td> <td>-<br/>,</td> <td>.216+05</td> <td>•</td> <td>•</td> <td>•</td> <td>•</td>                                                                                                                                              |        | 7 0.06+0      | -<br>9      | N/A         | •               | 0.0€+00        |                 | N/A          | •               | N/A           | 'n              | 0.0E+00      |                 | 0.0€+00      |                 | W/A           | -<br>,       | .216+05         | •               | •                 | •            | •                                                                                                |
| 7 0.06+00 - N/A - 0.06+00 - N/A - N/A - N/A - 0.06+00 - 0.06+00 - 0.06+04                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |        | 7 0.05.0      | · 2         | N/N         |                 | 0.0€+00        |                 | A/A          |                 | N/A           | ,               | - N/A        | •               | 0.0£+00      | ,               | 0.06+00       | -0<br>1<br>1 | . 205 +04       | •               | •                 | •            | •                                                                                                |
| 7 N.A - 0.0E+00 - 0.0E+00 - 6.00E+04                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |        | 7 0.0E+0      | - 0         | N/N         | •               | 0,0€+00        |                 | N/A          | •               | N/A           | 'n              | A/M          | •               | 0.0€+00      | •               | 0.0E+00       | 1            | 006 +04         | •               | '                 | •            | •                                                                                                |
| 7 9.7E-08 1.5E+01 N/A · 9.7E-08 1.XE+01 N/A · 9.7E-08 1.XE+01 4.04E+04 · - 2.46E+00 6 MR<br>7 9.7E-08 1.5E+01 N/A · 9.7E-08 1.XE+01 N/A · 7.7E-08 1.XE+01 3.76E+04 · - 1.00E-04 6 MR<br>64 7 N/A · 0.0E+00 · 0.0E+04 · · · 1.00E-04 6 MR<br>64 7 N/A · 0.0E+00 · 1.36E+04 · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |        | 7 N/A         | •           | A/A         |                 | A/A            | •               | N/A          | ,               | N/N           | •               | N/A          | •               | 0.0€+00      | ,               | 0.454.00      | - <b>9</b>   | . 00E +04       | •               | •                 |              | •                                                                                                |
| ) 9.1E-08 1.1E+01 N/A - 9.1E-08 1.1E+01 N/A - 9.1E-08 1.1E+01 9.1E-08 1.1E+01 3.78E+04 - 1.00E-04 6 MB<br>64) 7 N/A - N/A - N/A - N/A - N/A - N/A - 0.0E+00 - 0.0E+00 - 1.14E+04                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |        | 7 9.7E-0      | 10 1.35+01  | N/A         |                 | 9.7E-08        | I.X •01         | A/M          | ,               | 9.76-00 1.    | 3E+01           | N/A          | ,               | 2.1E-06 1    | 36+01           | 1.7E-00 1.    | 36+01 4      | . 04E+04        | •               | •                 | 2.686+00     | <b>A</b> 146                                                                                     |
| 64) 7 N/A - N/A - N/A - N/A - N/A - N/A - 0.06+00 - 0.06+00 - 1.36+04                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |        | ) 9. JE -U    | 10+31.1 80  | N/N         | •               | 9.76-08        | 1.36+61         | 8/A          | ,               | 9.76-08 1.    | 36+01           | N/N          | ,               | 2. 1E-06 1   | .JE+01          | P. 7E-06 L.   | 36+01 3      | .786+04         | •               | •                 | 1. 00E-04    | 9<br>9                                                                                           |
| 7 K.A . N/A N/A . N/A . N/A . N/A . N/A . 1,825+05                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 3      | 7 N/A         | •           | N/A         | •               | N/A            | •               | N/A          | •               | N/A           |                 | N/A          | ,               | 0,0£+00      |                 | 0.0E+00       | •            | 101 305 .       | •               | '                 | '            | •                                                                                                |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |        | 7 N.A         | •           | N/A         | •               | N/A            | •               | N/A          |                 | N/N           | •               | N/A          | •               | N/A          |                 | N/A           |              | 20+368.         | •               | •                 |              | •                                                                                                |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |        |               |             |             |                 |                |                 |              |                 |               |                 |              |                 |              |                 |               |              |                 |                 |                   |              |                                                                                                  |

see notes at end of table.

ني:

I-8







File: CONSTON.UKI date21-Aug-B? page 7

SIOMAGE ACCIDENTS - (Frequency waits given at bottom of table) FOR munitions at Elistims SITES

|                |            |                  |                 |              |                 |                     | Accident F      | r equenci e | r               |            |                 |              |                 |                    |                   |                 |                | -               | igent Aven      | i de set           | ki ezset        |                 |
|----------------|------------|------------------|-----------------|--------------|-----------------|---------------------|-----------------|-------------|-----------------|------------|-----------------|--------------|-----------------|--------------------|-------------------|-----------------|----------------|-----------------|-----------------|--------------------|-----------------|-----------------|
| SCENNEIO       | H          | Free Contraction | RANGE<br>FACTOR | <b>X</b>     | RANGE<br>FACTOR |                     | RANKE<br>Factor | a da        | RANGE<br>FACTOR | A P        | RANKE<br>FACTOR | PURA<br>FIEG | RANKE<br>FACTOR | - LEVE             | RANKE<br>Finction |                 | ANNE<br>Factor | AGENT<br>AMAIL. | LIS.<br>SPILLED | LAS.<br>Interveted | LAS.<br>Entited | NURATIO<br>TINE |
|                | 1          |                  |                 |              |                 |                     |                 |             |                 |            |                 |              |                 |                    |                   |                 | -              |                 |                 |                    |                 |                 |
| 300            | -          | A/N              | '               | A/M          | •               | N/N                 | •               | N/A         | •               | <b>V/N</b> | •               | W/W          |                 | 6. XE-12 2.        | . 16+31.          | 6. X-12 2       | 1 10+31.       | 1.4%6+05        | •               | ,                  | 1.496+04        | -               |
| SL BMC         |            | 4.75-12          | 2.46+01         | N/A          | •               | <b>A</b> / <b>N</b> | •               | A/A         | •               | N/A        | •               | 4.7E-12 2    | 10-34           | 1. JE-12 2.        | 10-37             | <b>U/N</b>      | -              | 1.916-04        | •               | 1.736+04           | 2.5% +03        | 20 MII          |
| SI CEC         | 89         | 6. 7E-12         | 2.46+01         | N/N          | •               | A/A                 | •               | N/A         | •               | A/A        | ı               | N/A          | •               | 1. X-12 2          | 10-31.            | W/W             |                | 1.446+04        | •               | 3.45£+03           | 1.046+03        | 20 M            |
| SLCHC          | -          | 6. 7E-12         | 2.46+01         | N/A          | •               | A/H                 | •               | N/N         | •               | A/A        | •               | 4. X-12 2    | 10-31           | W/W                | •                 | N/A             |                | 1.926+04        | ۰               | 7.306+03           | 1.016+03        | 11 R            |
| SLKGF (16L)    | 8          | A/A              | ,               | A/A          | •               | A.A                 | ,               | N/A         | •               | <b>N/N</b> | •               | N/A          |                 | 6. 7E-12 2.        | 10-31             | <b>V/I</b>      |                | 2.07E+05        | •               | •                  | 2.07E+04        | -               |
| SLUF (GL)      | -          | 6. X12           | 2.06.01         | N/N          | •               | <b>8</b> /8         | •               | N/N         | •               | N/A        | •               | N/A          | •               | N/N                | •                 | <b>N/A</b>      | •              | 1.446+05        | •               | •                  | 7.31E+03        | -               |
| SLINE (OPEN)   | 8          | N/A              | •               | 1.26-11      | 1.75+01         | N/N                 | ,               | N/A         | •               | 1.26-11    | 1.76+01         | A/A          | •               | 1.11-32-11         | .75+01            | N/N             | đ              | ASS.            | •               | •                  | 4. BKE +01      | -               |
| Succession and | 8          | R/A              | •               | N/N          | •               | N/A                 | ,               | A/A         | •               | N/N        | •               | N/N          | •               | N/A                | •                 | 1.46-10 2       | 2 10+37.       | 5.396+06        | •               | •                  | 2.416+05        | -               |
| SLEVE (181)    | -          | W/W              | •               | N/A          | ı               | A/A                 |                 | <b>N/A</b>  | •               | N/N        | •               | N/N          |                 | 1. XE-12 2.        | 10+34.            | N/N             |                | 2.216-05        | •               | •                  | 5.52E+03        | -               |
| SLEVE (MI)     | 8          | A/X              | •               | N/A          | •               | R/A                 | •               | 1.06-09 2   | 10+34.1         | N/A        | •               | N/A          | •               | N/A                | ,                 | N/N             |                | 2.996+04        | ,               | •                  | 7.466.04        | =               |
| SLINC          |            | 6. 7E-12         | 2.66+01         | N/A          | •               | A.A                 | •               | <b>N/N</b>  | •               | 6. 7E-12   | 2.46+01         | <b>N/N</b>   | •               | 6. 7E-12 2         | 10+39             | 4. 7E-12 2      | 10-34.         | 3.846+04        | •               | 9.446+03           | 7.2XE+02        | 20 M            |
| 34 PGC         | <b>ce</b>  | 6.7E-12          | 2.66+01         | M/A          | •               | N/A                 | ,               | N/A         | •               | N/N        | •               | M/A          |                 | 1. X-12 2.         | 10+34.            | 4. 7E-12 2      | 10-31          | 1.725.04        | •               | 1.486+01           | 5. 04E+03       | <b>20 M</b>     |
| SLPHC          | 8          | 6. 7E -12        | 2.66+01         | N/N          | •               | 6. 7E-12            | 2.66+01         | N/A         | ,               | N/A        | •               | 6. 7E-12 2   | . 66+01         | h. 7E-12 2.        | 10+31.            | R/A             | •              | 1.216+05        | •               | 3.02E+04           | 4.536+05        | 20 M.I          |
| SL PVC         | 60         | <b>1.7E-12</b>   | 2.46+01         | N/N          | •               | 4. 7E-12            | 2.6E+U1         | N/N         | •               | N/N        | ,               | N/N          |                 | 6. JE-12 2.        | 10-34.            | 4. JE-12 2      | 4 10+34.       | 1.206+04        | ,               | 1.55£+04           | 1.146+03        | 20 MII          |
| 20.000         | 8          | 6. 7E-12         | 2.46+01         | N/N          | •               | 6. <i>X</i> E-12    | 2.66+01         | W/W         | •               | N/N        |                 | 8/8          | •               | h. 7E-12 2.        | 10+34.            | 6.7E-12 2       | . 16-34.       | 1. 00E +01      | •               | 1.56.5             | 4.506+03        | 20 MI           |
| SL DVC         | <b>a</b>   | N/A              | •               | A/A          | •               | N/A                 | •               | #/#         | •               | N/N        | •               | N/N          | •               | <b>1.7E-12 2</b> . | 10+31.            | 6. XE-12 2      | 10-37"         | 1.006+04        | 1               | 1.566+01           | 1.13€+03        | 20 MU           |
| SLREC          | 8          | 6. XE-12         | 2.46+01         | N/N          | •               | 6. X -12            | 2.4E+01         | N/A         | •               | 4. 7E-12   | 2.46+01         | <b>V</b> /8  |                 | h. K-12 2.         | 10-31             | 6. JE-12 2      | 10-31          | 1.0£+01         | •               | 1.016+34           | 3.03E+03        | 20 NII          |
| SLEVC          | 89         | 6. 7E -12        | 2.46+01         | N/N          | •               | 6. XE-12            | 2.46+01         | #/#         | •               | 4.76-12    | 2.4E+01         | ۹/۹          |                 | 1. X-12 2.         | 10-31.            | 6. XE-12 2      | . 10+34.       | 1.78644         | •               | 9.456+03           | 7.0% +02        | 20 MLI          |
| SL SWF (16L)   | •          | N/A              | •               | N/A          | •               | N/N                 | ,               | N/A         | •               | N/N        | •               | W/W          | •               | R/N                |                   | 6. <b>TE-12</b> | 1.16+34.       | 1. Mi 44        | 1               | •                  | 3.39€+02        | -               |
| Su Svf Imm     | <b>00</b>  | A/A              | •               | A/A          | •               | N/A                 | •               | N/N         | •               | N/N        | 1               | N/N          | •               | 1.76-09 2.         | 10-37.            | K/N             | •              | 1.036+05        | •               | '                  | 4.566.43        | -               |
| SL9 - Runti    | on dro     | the duri         | ng leaker       | 1501 at 1 on | activities      |                     |                 |             |                 |            |                 |              |                 |                    |                   |                 | •              |                 |                 |                    |                 |                 |
| SLINC          | <b>0</b> * | N/A              | •               | N/N          | •               | <b>N</b> /N         | •               | A/A         | •               | N/A        | •               | N/A          |                 | 6.4E-07 1.         | . 10+3L.          | 6.4E-07 1       | - 10+3C.       | 1.446+02        | •               | •                  | 4.246+00        | -               |
|                | •          | 4. 12-07         | 1.36+01         | <b>N/A</b>   | •               | 4/h                 | •               | U/N         | •               | W/W        | •               | 1.14-35.1    | 10-3.           | 1.56-07 1.         | .XE+01            | N/N             |                | 1.000-02        | •               | •                  | 1.366-03        | -               |
| SLCGC          | •          | 9.86-00          | 1.36.01         | N/N          | •               | N/A                 | •               | #/W         | •               | N/A        | •               | N/N          | •               | P. IK-00 1.        | .3641             | N/N             | •              | 10+300.1        | •               | •                  | 1.016-01        | -               |
| SI CHC         | •          | 9.06-08          | 1. 32 +01       | A/A          | •               | <b>W</b> /W         | •               | N/N         | •               | N/N        | •               | 1.8-31.1     | 10+37.          | R/A                | •                 | A/A             | •              | 1.405+01        | •               | •                  | L. 36€ +3       | -               |
| SLAGC          | •          | N/A              | •               | N/N          | •               | N/A                 | •               | N/N         | •               | N/A        | •               | W/W          |                 | 1.16-07 1.         | .XE+01            | V.H             | -              | 1.50£+03        | •               | •                  | 6.406+00        | -               |
| SLUCC<br>SLUCC | •          | 1.96-07          | 1.36+01         | 1.96-07      | 1.3£+01         | A/N                 | •               | N/N         | •               | 1.96-07    | 1.16.01         | <b>W</b> /W  | •               | 1.96-97 1.         | 10+37             | 1.16-07 1       | 1 10+3K.       | 1.706+03        | •               | •                  | 2.566-02        | -               |
| SLIVE          | •          | N/N              | •               | N/N          | •               | N/A                 | •               | 1.96-07 1   | . 3£ +01        | N/N        | •               | R/A          | •               | 1.96-07 1.         | 16+3(.            | <b>A/A</b>      | •              | 1.405+93        | •               | •                  | 2.706-04        | -               |

See notes at ead of table.

1

File: CONSTOR.MK1 date21-Aug-B7 page 8

SIORAGE ACCIDENTS - Ifrequency units given at bottom of table)

| r  |    |
|----|----|
|    |    |
| 5  |    |
| •  |    |
| i  |    |
|    |    |
|    | 8  |
|    | Ξ  |
|    | ŝ  |
|    | 49 |
|    | Ξ. |
| ł. | 5  |
| -  | ¥  |
|    | Ξ. |
| 1  | -  |
| ۲  |    |
|    | 52 |
|    | 8  |
|    | =  |
|    | Ξ. |
|    | 5  |
|    | -  |
| 1  | 窶  |
|    | Ξ. |
|    |    |
|    |    |
|    |    |
|    |    |

| 3         |              | 2          | AMA<br>Fred<br>F | RANGE<br>FACTOR | APG<br>FREQ | RANGE<br>FACTOR | L BMB       | RANGE<br>FACTOR | FIRED       | RANGE<br>Factor | PBA<br>Free | RANGE<br>FACTOR | PUDA<br>FREG | RANGE<br>FACTOR | TEA<br>Free       | FACTOR    |                     | FACTOR   | MEAT<br>Avail. | LINS. | LIS.<br>In TomAtes | LIK.           |   |
|-----------|--------------|------------|------------------|-----------------|-------------|-----------------|-------------|-----------------|-------------|-----------------|-------------|-----------------|--------------|-----------------|-------------------|-----------|---------------------|----------|----------------|-------|--------------------|----------------|---|
|           |              | :          |                  |                 |             |                 |             |                 |             |                 |             |                 |              |                 |                   |           |                     |          |                |       |                    | 1              |   |
| SLINC     |              | •          | 3.66-07          | 1.36+01         | N/N         | •               | N/A         | •               | M/A         | •               | 3.66-07     | 1. 3€+01        | A/A          | •               | 3.46-07           | 1. 36 401 | 3.46-07             |          | 3.766+02       | •     | •                  | S              | - |
| 51,760    |              | •          | 0.0E+00          | •               | N/N         | •               | N/A         | '               | N/A         | •               | N/N         | •               | A/A          | •               | 0.0€+00           | •         | 0.0£+00             | •        | 5.2KH          | •     | •                  | •              |   |
| SUPAC     |              | •          | 0, 0£+00         | •               | A/N         | •               | 0.0€+00     | •               | N/A         | •               | A/N         | ı               | 0.0E+00      | •               | 0.0£+00           | •         | M/A                 | •        | 9.346.01       | •     | •                  |                |   |
| N N       |              | œ          | 0.05+00          | •               | N/A         | •               | 0.06+00     | •               | A.A         | '               | N/A         | ,               | AVA          | •               | 0.06+00           | •         | 0.05+00             | ,        | 1.066-401      | •     | •                  |                |   |
| 31.060    |              | •          | 0.0€+00          | •               | N/N         | •               | 0.0€+00     | •               | N/A         | •               | N/A         | •               | N/N          | •               | 0.0£+00           | •         | 0.06+00             | •        | B. 706+01      | •     |                    | •              |   |
| SLIMC     |              | •          | N/N              | •               | N/A         | •               | R/A         |                 | N/N         | •               | A/#         | •               | M/A          | •               | 0.0£+00           | •         | 0.06 +00            |          | 8.766+01       |       | •                  |                |   |
| SLIBEC    |              | •          | 1.55-06          | 1.3£+01         | N/N         | •               | 1, 56 -06   | 1.3€+01         | N/A         | ,               | 1.56-06     | 1.36+01         | N/N          | •               | 1.5%              | 1.36-01   | 8-37-1              | 1.36+01  | 1.466 +02      | •     |                    | 1. 46-01       | - |
| SLAVC     |              | •          | 1.56-04          | 1. 36+01        | N/N         | •               | 1.56-06     | 1.3£+01         | N/N         | •               | 1.56-06     | 1.36+01         | R/A          | •               | <b>8-35-1</b>     | 1.36+01   | 1.5-2               | 1. 16+01 | 1.566-02       | •     |                    | 19<br>19<br>19 | - |
| SIL SWC   |              | •          | N/N              | •               | N/N         | •               | N/A         | •               | N/A         | •               | N/N         | •               | A/A          | •               | 3.46-04           | 1.3£+01   | 3.46-04             | 1.36.01  | I. Jaf +03     |       |                    | 2 706 -04      | - |
| 515       | Seall a      | urcraf     | t direct         | crash pate      | P BAL PHONE | se or open      | storage     | vard: fire      | not cont    | at bants        | 30 einut    | ž               |              |                 |                   |           |                     |          |                |       |                    |                |   |
| SLAFF I   | <b>(1</b> )  | 2          | N/A              | ,               | N/N         | . 1             | N/A         | •               | N/A         | •               | N/A         | •               | N/A          | •               | 0.0£+00           | •         | A/N                 | •        | 2.076+05       | •     |                    |                |   |
| Sund a    | Ĵ            | 2          | 0.0£+00          | •               | N/N         |                 | A/A         | •               | N/A         | •               | N/A         | •               | N/N          | •               | A/A               | •         |                     | •        | 1.46E+05       | ٠     | •                  |                |   |
| SLANF I   | <b>P</b> (H) | 2          | N/N              | •               | 1.46-05     | 1.0€+01         | A/A         | •               | N/N         | •               | 5.46-07     | 1.06+01         | R/A          | ı               | 1.46-07           | 1.06+01   | <b>N/N</b>          |          | L MGG.         | •     | •                  | 5.166-03       | Ř |
| a and     | Ĩ.           | 51         | A/N              | •               | N/A         | •               | A/M         | •               | A/A         | •               | N/N         | •               | N/N          | •               | A/A               | •         | 16- <b>34</b> - 0   | 1.06+01  | 5. 316 44      | ,     | •                  | 2.4%+05        | 2 |
| SLKWF (   |              | 5          | A/M              | •               | N/A         | ,               | A/A         | ,               | A/A         | •               | N/A         | •               | N/N          | •               | 0.116+00          | •         | <b>4</b> /8         | •        | 2.216+05       | •     | •                  | •              |   |
| SLIKVE II | Î            | 2          | A/A              | •               | A/A         | ,               | N/A         | •               | 8. IE-09    | 1.0£+01         | N/N         | •               | N/N          | •               | N/N               | •         | <b>V</b> / <b>I</b> | •        | 2.9K +4        | •     | •                  |                | 3 |
| SLSWF 1   | (B)          | 2          | N/N              | •               | N/N         | •               | N/A         | •               | N/N         | •               | N/N         | •               | N/N          | •               |                   | •         |                     | '        | 1. 18 ee       | ,     | •                  | •              |   |
| SLSWF I   | ŝ            | 2          | 8/8              | •               | N/A         | •               | A/A         | •               | N/N         | •               | W/W         | •               | N/N          | •               | 1. <del>1</del> 8 | 1.06.01   | <b>4</b> /1         | •        | 1.8%+05        | •     | •                  | 2.016+05       | * |
| 9119      | Large        | Incerat    | It durect        | Crash; no       | fire. (bu   | urstered a      | - SUOIT INN | detonate        |             |                 |             |                 |              |                 |                   |           |                     |          |                |       |                    |                |   |
| SLIPEC IE | 191.0        | 91         | N/A              | •               | N/A         | •               | A/A         | ,               | A/A         | •               | N/N         | •               | N/N          | •               | 1.26-11           | 1.0€+01   | 5.06-10             | 1.0€+01  | 1.111111       | •     |                    | 5.126-01       | • |
| SLIPEC II | 181.6        | 16         | N/A              | •               | A/A         | r               | N/A         | ٠               | Ψ/ <b>N</b> | •               | A/M         | •               | . N/A        | •               | 1.36-11           | 1.0€+01   | <b>V</b> / <b>I</b> | :        | 1.4% +05       | •     | •                  | 5.128.401      | - |
| SUPPC 1   | () [BL)      | <b>9</b> ] | 2.06-10          | 1.0€+01         | A/N         | •               | N/N         | •               | N/A         | •               | N/N         | •               | N/N          | •               | A/A               | •         | A/A                 | •        | 3.576+04       | '     | 7.146+02           | 2.001-01       | • |
| SLIPHC (6 | 0. 191 .0    | 1          | 2.46-10          | 10+30.1         | N/A         | •               | N/A         | •               | N/A         | •               | N/N         | •               | 2.06-09      | 10+30.1         | 11-32-1           | 1.0€+01   | N/N                 | •        | 10-354.1       | •     | 9.916+02           | 2.046-41       | - |
| SU BHC 16 | 19. IGU)     | 16         | A/A              | 1               | N/N         | •               | N/N         | •               | N/A         | •               | N/A         | •               | N/N          | •               | 11-36-11          | 1.46+01   | <b>N/N</b>          | •        | 6.91E+44       | •     | 1. 18 to 3         | 2.0HE-01       | - |
| SLCGC 14  | 0 161)       | 16         | 2.06-10          | 1.0€+01         | A/N         | •               | A/A         | •               | M/A         | •               | N/A         | •               | N/A          | •               | N/N               | •         | W/W                 | •        | 7.206+03       | •     | 1.446+02           | 5. 12 to 1     | - |
| SLCGC 16  | 191.0        | 91         | 2.46-10          | 10+30-1         | N/A         | ,               | N/A         | •               | N/N         | •               | N/N         | •               | N/N          | •               | II-32             | •         | A VI                | •        | 9.716+03       | '     | 1.946+02           | 5.126-01       | • |
| SLCBC 16  | 19° 16L)     | 4          | A/M              | •               | N/N         | ı               | N/N         | •               | N/N         | •               | N/N         | •               | N/A          | •               | 1.16-21           | •         | <b>V/I</b>          | •        | 1.446+04       | •     | 2.926.42           | 5.125+01       | • |
| SLONC 12  | 10. 181)     | 9          | 2.06-10          | 1.0€+01         | N/A         | •               | N/A         | •               | N/A         | •               | N/N         | •               | N/N          | •               | W/W               | •         | W/W                 | •        | 1.46.01        | •     | 2.866+62           | 2.005-01       | - |

See mates at ead of table.

2203633

4









file: CBMSIGN.MK1 date21-Aug-87 page 9

ţ ALC NONTO STORAGE

| i table) |       |
|----------|-------|
| bottoe c |       |
| 14       |       |
| a i      | SITES |
| units    |       |
| Juency   | N EII |
| · (Free  |       |
| ENTS -   |       |
| ACC 1    | Ű.    |
|          |       |

|                          |                     |      |        |             |          |             |       | -                   | kcuéent F       | requests   | Ŧ               |            |         |             |                  |          |           |            |                | -               | <b>Nyan</b> t Avai | lale ad         | blasse         |                 |
|--------------------------|---------------------|------|--------|-------------|----------|-------------|-------|---------------------|-----------------|------------|-----------------|------------|---------|-------------|------------------|----------|-----------|------------|----------------|-----------------|--------------------|-----------------|----------------|-----------------|
| N335                     |                     | 2    | FIRED  |             |          | a 2 .       |       | ERED<br>FIELD       | RANGE<br>FACTOR | FIED F     | RANKE<br>FACTOR | A B        | FACTOR  | Aux<br>Pres | FACTOR<br>FACTOR |          | FACTOR    |            | PARK<br>FACTOR | AGENT<br>MMAIL. | chs.               | LBS.<br>Demated | LIS.<br>Enited | NURATIO<br>Time |
|                          | : :                 |      |        | 9<br>-<br>- |          |             |       |                     |                 | 1          |                 |            |         | 5           |                  | 1        |           |            |                |                 |                    |                 |                |                 |
|                          |                     | =    | 2.84-  | -10 1.0E    |          | 5           | •     |                     | •               |            | •               |            | •       | 2.01-07     | 1.01.101         |          | 1.05.00.1 |            | •              |                 | 1                  | 2.94 124.5      | 2. 00E-01      | -               |
|                          | 9                   | 11   | 2      | •           | *        | N.          | •     | R/A                 | •               | <b>N/N</b> | •               | N/N        | ı       | W/W         | •                |          | 1.05+01   | N/N        | •              | 2.926+44        | •                  | 5.84£+02        | 2.006-01       | -               |
| SLAFE (                  |                     | 1 0  |        | ę           | -        | V.          | •     | N/A                 | •               | W/W        | •               | N/N        | •       | N/N         | •                | 1.25-11  | 1.46+01   | <b>R/A</b> | •              | 2.076+05        | •                  | •               | 5.126+01       | Ŧ               |
| SLUC (F                  | 9                   | 3    | 2.06-  | 10 1.06     | 10.      |             | •     | N/A                 | •               | W/W        | •               | N/N        | •       | N/N         | •                | N/N      | ١         | W/W        | •              | 1.466+05        | •                  | •               | 2.006-01       | Ŧ               |
| SLANS ID                 | (iiii)              | 1    | 2      |             | - 1.35   | -04         | 10+3¢ | M/A                 | ,               | A/A        | •               | 9.46-09    | 1.0€+01 | W/W         | •                | 4. 32-09 | 1.06+01   | N/N        | ני<br>י        | LASS.           | 3.40£+05           | 1               | •              | -               |
| SLUG (B                  | ŝ                   | -    | 2      |             | <b>.</b> |             | ,     | W/W                 | ,               | N/A        | •               | N/A        | '       | N/N         | •                | R/N      | •         | 1.45-00    | 10+30.1        | 5.396+04        | 3.466-65           | •               | •              | <b></b>         |
| SLKWC I                  | 9<br>9              | 1    | N.     | æ           |          |             | ,     | A/A                 | •               | A/A        | •               | N/A        | •       | A/A         | •                | 1.26-11  | 1.0€+01   | N/N        | •              | 2.216+05        | 1                  | '               | 2.146-03       | -               |
| SANS (III                | 2                   | 4    | N.     |             | <b>1</b> | N.          | ۰     | N/A                 | ı               | 2.06-09 1  | 10+30.          | R/A        | •       | N/N         | •                | A/A      | •         | N/N        | ,              | 10-346-2        | 3.206+05           | 1               | •              | *               |
| SLIME 14                 | - 0<br>- 10<br>- 10 | 1    | 2.06-  | 10 1.UE     | -01 W    |             | ,     | A/A                 | •               | W/W        | 4               | N/A        | ,       | A/A         | •                | N/A      | •         | R/A        | '              | 2.726+04        | 1                  | 1.346+03        | 2.166-03       | -               |
| SLINC (9                 | 9                   | 2    | 2,46-  | 30-1 01-    | 101      |             | •     | N/A                 | •               | N/A        | •               | 3.06-11    | 1.06+01 | W/W         | •                | 11-32.1  | 1.0€+01   | 5.06-10    | 1.06+01        | 3.846+04        | •                  | 1.93E+03        | 2.146-03       | Ŧ               |
| SLPSC IA                 | 9                   | 1 19 | 2.06-  | 10 1.06     | -01 W    |             | •     | A/A                 | •               | <b>N/A</b> | •               | N/N        | '       | N/N         | •                | N/N      | •         | N/N        | •              | 3.596+04        | •                  | 7.186+02        | 5.126+01       | Ŧ               |
| SLPGC (0)                | .e                  | 1 1  | 2.66-  | 10 1.06     | 101      |             | ,     | <b>M/A</b>          | •               | A/N        | •               | N/N        | •       | N/N         | •                | 11-32.1  | 1.06+01   | 5.06-10    | 1.06+01        | 5.02£+04        | •                  | 1.00E+03        | 5.126+01       | -               |
| 8) J9475                 | 9                   | 2    | A .    | ÷           | #<br>1   |             | ,     | N/A                 | •               | A/N        | •               | N/N        | •       | N/A         | •                | 1.3E-11  | 1.0€+01   | N/N        | •              | 4.725+04        | •                  | 1.346+03        | 5.12£+01       | -               |
| SLPHC (4)                | - Q                 | 1    | 2.06-  | 10 1.06     | •01 H    | A/I         | •     | A/A                 |                 | A/A        | ·               | N/A        | •       | N/A         | •                | A/N      | •         | N/N        | •              | 4.465+04        | •                  | 1.296+03        | 2.00E-01       | =               |
| SLFIC 18                 | 91                  | 1 2  | 2.K    | -10 1.06    | •01 N    | N.A.        | ,     | N/A                 | •               | N/A        | •               | N/A        | •       | 2.06-09     | 1.05+01          | 11-32-11 | 1.0€+01   | N/A        | •              | 9.065+04        | •                  | 1.816+03        | 2.006-01       | Ŧ               |
| 81 JHJ 18                | 9. 16               | 1    |        |             | -        | NA.         | •     | N/A                 | 1.06+01         | N/A        | •               | <b>8/8</b> | •       | N/A         | •                | 1.36-11  | 10+30.1   | N/N        | •              | 1.216+05        | •                  | 2.426+03        | 2.00E-01       | Ŧ               |
| SLPMC 14                 | .e                  | 3    | 2.06-  | 10 1.06     | 101      | ¥/1         | •     | N/N                 |                 | N/A        | •               | N/N        | •       | N/N         | •                | A/A      | •         | N/N        | •              | 3.316+04        | •                  | 6.42E+02        | 2.146-03       | =<br>+          |
| SLIPPC (B)               | 9                   | 1 19 | - 2.K- | 10 1.06     | -10-     | 5           | •     | A/A                 | •               | N/A        | •               | W/W        | •       | N/A         | •                | 11-32-11 | 1.0E+01   | 5.06-10    | 1.0€+01        | 4.46+04         | '                  | 9.276+02        | 2.16E-03       | Ŧ               |
| SLMC 10                  | 9.16                | 11   | 1      |             | <b>a</b> |             | ,     | 1.46-10 1           | 1.06+01         | M/A        | •               | N/N        | •       | N/N         | •                | 11-36-11 | 1.0€+01   | N/N        | •              | 4.205+04        | •                  | 1.246+03        | 2.146-03       |                 |
| 19 J98 X                 | 9                   | 1    |        | 10 1.06     | 101 W    | N.N         |       | <b>A/A</b>          | •               | R/A        | •               | N/N        | •       | N/N         | •                | N/N      | •         | N/N        | •              | 3.29£+04        | •                  | 4. SBE+02       | 5.121+01       | -               |
| SLIDEC 18                | 5                   | 3    | 2.4E-  | 10 1.05     | W 10.    | NA.         |       | N/N                 | •               | A/N        | •               | N/A        | '       | AVN-        | '                | 11-32.1  | 1.0£+01   | 5.06-10    | 1.06+01        | 4.596+04        | •                  | 9.19€+02        | 5.126+01       | -               |
| SLOGC 18                 | 9 . E               | 11   | 1      | ٩.          | -        |             | ,     | 1.46-10 1           | 1.0€+01         | N/N        | •               | N/N        | •       | N/N         | . 1              | 1.32-11  | 1.05.401  | N/N        | •              | 6.000444        | •                  | 1.206+03        | 5.126+01       | Ŧ               |
| SLIPHC 14                | .0                  | 1    | 2.05   | 10 1.06     | •01 N    |             | •     | A/A                 | ,               | A.A        | •               | N/N        | '       | N/N         | 1                | N/N      | •         | N/N        | •              | 3.296+04        | •                  | 4.566+02        | 2.146-03       | Ŧ               |
| SLIPHC 18                | <u>9</u> .0         | 1    | - 2.K- | -30°1 01-   | W 104    | N.A.        | •     | A/A                 | •               | N/A        | •               | N/A        | •       | N/N         | •                | 11-32.1  | 1.0E+01   | 5.06-10    | 10+30'1        | 4.5% +04        | •                  | 9.196+02        | 2.166-03       | -               |
| SLIPPC 18                | . e                 | 1 1  |        |             | -        | V.          | •     | A/N                 |                 | N/N        | •               | N/A        | '       | <b>V/R</b>  | •                | 1.36-11  | 1.05+01   | N/N        | •              | 4- 00E +04      | •                  | 1.20E+03        | 2.166-03       | -               |
| 4) 1990 (*<br>2) 1990 (* | 0.16                | 11   | 2.06-  | 10 1.06     | -01 R    |             | ,     | <b>4</b> / <b>8</b> | •               | N/A        | •               | N/N        | •       | N/A         | •                | A/N      | •         | N/N        | •              | 2.126+04        | '                  | 1.046+03        | 5.125-01       | Ŧ               |
| SUBSC (B                 | 9.0                 | 1 1  | 2.66-  | -10 1.0E    | •01 II   | <b>W</b> /1 | •     | N/N                 | •               | A/A        | ,               | 5.06-11    | 1.0£+01 | A/A         | '                | 1.25-11  | 1.05+01   | 5.06-10    | 10+30.1        | 2.826+04        | •                  | 1.41E+03        | 5.126+01       | -               |
| SLINGC 18                | 9<br>1              | 11   | ì      | •           | -        |             | ,     | 1.46-10 1           | 10+30.          | N/A        | ,               | W/W        | •       | A/A         | •                | 11-32.1  | 1.06+01   | N/N        | •              | 4.046+04        | •                  | 2.02E+03        | 5.126+01       |                 |

See notes at end of table.

I-11

File: COMSTOR.UKI date21-Aug-87 page 10

## STORAGE ACCIDENTS - IFrequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

| Mark         Mark <th< th=""><th></th><th></th><th></th><th></th><th></th><th>-</th><th>Accident Fi</th><th>r equenci e</th><th>÷</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>-</th><th>lgent Avai</th><th>  bas elde!</th><th>Rel eased</th><th></th></th<>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                         |       |            |             |                 | -               | Accident Fi     | r equenci e | ÷               |             |                 |              |                 |              |                 |                |                       | -               | lgent Avai       | bas elde!         | Rel eased       |          |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------|-------|------------|-------------|-----------------|-----------------|-----------------|-------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|----------------|-----------------------|-----------------|------------------|-------------------|-----------------|----------|
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ANAB KANG<br>Freq Facto | FALLE |            | APG<br>FREQ | RANGE<br>FACTOR | L BAB<br>F IEEO | RANGE<br>FACTOR | FRED        | RANGE<br>Factor | PBA<br>Freq | RANGE<br>Factor | PUDA<br>FREG | RANGE<br>Factor | TEAD<br>Fred | RANGE<br>Factor | LINDA<br>FIELO | <b>NAME</b><br>FACTOR | AGENT<br>AVAIL. | LING.<br>SPILLED | LBS.<br>Detomited | LUS.<br>Emitted | NURAT IC |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                         | !     | :          |             |                 |                 |                 |             |                 |             |                 |              |                 |              |                 |                |                       |                 |                  |                   |                 |          |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 2.06-10 1.01            | 3     | ē          | N/N         | •               | N/N             | •               | N/A         | •               | N/A         | •               | N/N          | •               | N/N          | •               | N/N            | •                     | 1.906+04        | •                | 9.90€+02          | 2.146-03        | ÷        |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 2.46-10 1.06            |       | 10.        | N/N         | ,               | N/A             |                 | N/A         |                 | 5.06-11 1.  | 10+30.          | N/A          | •               | 1.11-32.1    | 10.30.1         | 5.06-10 1      | 10+30                 | 2.44E+04        | •                | 1.326+03          | 2.146-03        | Ŧ        |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N/A                     |       | ٠          | N/A         | •               | 1.46-10         | 1.0E+01         | N/A         | •               | N/A         | •               | N/A          | •               | 1. JE-11 1   | 1.06+01         | N/N            | •                     | 3.706+04        | •                | 1.896+03          | 2.146-03        | Ŧ        |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | M/A                     |       | •          | N/N         | •               | N/A             | •               | N/A         | ,               | N/A         | •               | N/N          | •               | N/A          | •               | 5.06-10 1      | 10+30.                | 1.366+04        | •                |                   | 2.166-03        | -        |
| (b) 1/100 (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 10) (1 + 1 | M./ A                   |       | 1          | N/A         | •               | N/A             | ,               | N/A         | •               | N/A         |                 | 8/8          | •               | 1.01-30.1    | 1.06+01         | N/A            | •                     | 1.836+05        | 2.716+05         | •                 | •               | Ŧ        |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Ht durect cr.           | 5     | 1) iyse    | ire contai. | ned within      | 30 minute       | es. (Applie     | rs to non   | -burstere       | d eunitie   | (y ino se       |              |                 |              |                 |                |                       |                 |                  |                   |                 |          |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N/A                     |       | ,          | 3.7E-13     | 1.0€+01         | A/N             |                 | N/A         | 1               | 2.65-12 1.  | .06+01          | N/A          | •               | 1.26-12 1    | 10+30.1         | N/A            | ם<br>י                | LASS.           | •                | '                 | 1.706+04        | 1H 95    |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N/A                     |       | '          | N/A         | •               | N/A             | ,               | N/A         | •               | W/W         | •               | N/N          | •               | A/A          | •               | 3.86-12 1      | 10+30                 | 5. 39€+06       | '                | •                 | 1.706+04        | 30 ML    |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N/A                     |       | •          | A/A         | •               | N/N             | •               | 5.66-13 1   | 10+30.          | N/A         | •               | N/A          | •               | N/N          | •               | N/A            |                       | 2.996+06        | •                | '                 | B. 00E+03       | 30 MI    |
| 1.3.1.0.01       with markhause or open storage parts ino fire.       2.3.25.4.01       1.1.1.1.0.01       1.1.1.1.0.01       2.3.25.4.01       -       -       4.4         -       1.4.4       -       1.4.4       -       1.4.4       -       1.4.5.1       2.3.25.4.01       -       -       4.4         -       1.4.4       -       1.4.4       -       1.4.4       -       1.4.5       2.5.25.4.01       -       -       4.4         -       1.4.4       -       1.4.4       -       1.4.4       -       1.4.5       2.5.25.4.01       -       -       4.4         -       1.4.4       -       1.4.4       -       1.4.4       -       1.4.4       -       1.4.4       -       4.4         -       1.4.4       -       1.4.4       -       1.4.4       -       1.4.4       -       1.4.4       -       -       4.4       -       4.4       -       4.4       -       4.4       -       4.4       -       4.4       -       4.4       -       4.4       -       4.4       -       4.4       -       4.4       -       1.2       -       4.4       -       -       1.2       -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | R/A                     |       | ,          | N/A         |                 | N/A             | •               | N/A         | •               | N/A         | •               | <b>N/N</b>   | •               | 1.36-13.1    | 1.06+01         | N:N            | ,                     | 1.835+05        | •                | •                 | 6.786+03        | I W OS   |
| -       2.06-05       WA       -       WA       -       U.A       -       L.R07       L.66-01       WA       -       L.R07       L.66-01       WA       -       L.R07       L.66-01       WA       -       L.R07       L.66-01       WA       -       L.R07       L.66-01       S.756-04       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -       -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Ht Sirect               |       | crash ont  | to wirehow  | ise or open     | storage 1       | yard; no fi     | ire.        |                 |             |                 |              |                 |              |                 |                |                       |                 |                  |                   |                 |          |
| ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·         ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N/N                     |       | •          | 2.06-05     | 1.0E+01         | N/N             | •               | N/N         |                 | 6.95-07 1.  | 10+30.          | N/N          | ۱               | 1.76-07 1    | 1.0E+01         | W/W            | ď                     | LASS.           | 2.556+04         | •                 | '               | -        |
| T         NA         F         NA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | R/N                     |       | •          | A/A         | •               | N/N             | •               | N/N         | ,               | N/A         | •               | N/N          | •               | N/N          | ,               | 1.1E-08 1      | 10+30                 | 5.396+04        | 2.55€+04         | •                 | •               | <b>T</b> |
| · NA       · NA       · NA       · NA       · 1.8E-06       1.6E-06       I.A       · 1.8E605       2.0E+06       · 4         ****       0.10 - 07       1.3E - 01       NA       · NA       · 1.4E - 06       1.6E - 01       I.A       · 1.8E - 05       2.0E - 03       · 4         ***       1.0 - 07       1.3E - 01       NA       · NA       · 1.1E - 08       1.3E - 01       1.4E - 01       1.4E - 03       3.0E       1.4E - 03       3.0E - 03                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | R/N                     |       | •          | N/A         | •               | A/A             | •               | 1.06-08.1   | 10+30.          | A/A         | ı               | R/A          | •               | N/A          | •               | N/A            | •                     | 2.996+04        | 2.40€+04         | •                 | •               | Ŧ        |
| 3:5h Onto warehouse or open storage yard; fire contained in 30 anates.       3:0k - 01 1.35 + 01 1.35 + 01 1.35 + 01 1.35 + 01 1.35 + 01 1.36 + 01 3.36 + 01 1.35 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01 3.36 + 01         | N/A                     |       | •          | N/N         | •               | N/N             |                 | N/A         | •               | N/A         | •               | N/N          | •               | 1.96-00 1    | 1.0€+01         | 8/8            |                       | 1.836+05        | 2.03E+04         | •                 | •               | Ŧ        |
| - J.(i-0)       I.X.       - N/A       - I.(i-0)       J.X.       - C.(ASS.        - I.206.403       30 MI         - WA       - WA       - WA       - WA       - WA       - J.(i-0)       J.X.       - I.206.403       30 MI         - WA       - WA       - WA       - WA       - WA       - J.06-10       J.X.6401        - 1.206.403       30 MI         - WA       - WA       - WA       - WA       - WA       - WA       - J.06-10       J.X.6401                          -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ft durect               |       | Lrash ont  | to warehou  | ise or open     | storage :       | yard; fire      | containe    | d 10 30 et      | inutes.     |                 |              |                 |              |                 |                |                       |                 |                  |                   |                 |          |
| VM     NM     NM     NM     NM     NM     NM     -     1.266-101     3.061     1.266-101     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061     3.061                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | N/A                     |       | •          | 3.06-07     | 1.3£+01         | N/A             | •               | N/A         |                 | 1.16-08 1.  | .36+01          | #/#          | •               | 2.76-09 1    | 1.3£+01         | N/A            | а<br>'                | LASS.           | •                | •                 | 1.206+03        | 30 M     |
| -       U/A       -       U/A <td< td=""><td><b>H</b>/A</td><td></td><td>•</td><td>W/W</td><td></td><td>N/N</td><td></td><td>N/A</td><td></td><td>N/A</td><td>,</td><td>N/N</td><td>•</td><td>N/A</td><td>•</td><td>3.06-10 1</td><td>36+01</td><td>5.39€+06</td><td>'</td><td>•</td><td>1.206+03</td><td>IN OS</td></td<>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | <b>H</b> /A             |       | •          | W/W         |                 | N/N             |                 | N/A         |                 | N/A         | ,               | N/N          | •               | N/A          | •               | 3.06-10 1      | 36+01                 | 5.39€+06        | '                | •                 | 1.206+03        | IN OS    |
| -         U/A         -         U/A <thu a<="" th="">         -         U/A         -<!--</td--><td>N/A</td><td></td><td>•</td><td>N/A</td><td>•</td><td>N/A</td><td>,</td><td>1.5£-10 1</td><td>. J£ +01</td><td>N/A</td><td>•</td><td>N/N</td><td>•</td><td>A/A</td><td>•</td><td>N/A</td><td>,</td><td>2.996+04</td><td>•</td><td>•</td><td>4.00€+02</td><td>20 11</td></thu>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N/A                     |       | •          | N/A         | •               | N/A             | ,               | 1.5£-10 1   | . J£ +01        | N/A         | •               | N/N          | •               | A/A          | •               | N/A            | ,                     | 2.996+04        | •                | •                 | 4.00€+02        | 20 11    |
| crash outo storage area; no fire.       u/A       -       J.X=12 1.X=01       1.4E-10 1.19E+05       -       -       1.38E+02       4 H         -       U/A       -       U/A       -       U/A       -       J.X=12 1.X=01       1.4E-10 1.19E+05       -       -       1.38E+02       4 H         -       U/A       -       U/A       -       U/A       -       U/A       -       1.4E-10 1.19E+05       -       -       -       1.38E+02       4 H         X=01       U/A       -       U/A       -       U/A       -       U/A       -       1.3E+12 1.2E+01       U/A       -       0.00E+00       1.06E+02       4 H         X=01       U/A       -       U/A       -       U/A       -       U/A       -       1.06E+01       1.06E+02       -       -       1.05E+02       4 H         X=01       U/A       -       U/A       -       U/A       -       U/A       -       0.00E+00       1.06E+00       1.06E+01       0.00E+00       1.06E+00       1.06E+00       1.06E+00       1.06E+00       1.06E+01       0.00E+00       1.06E+00       1.06E+00       1.06E+00       1.06E+00       1.06E+01       0.00E+00       1.06E+00 <td>N/A</td> <td></td> <td>•</td> <td>M/A</td> <td>•</td> <td></td> <td>•</td> <td>A/A</td> <td>•</td> <td>N/A</td> <td>•</td> <td>8/8</td> <td>•</td> <td>3.06-10 1</td> <td>1.36+01</td> <td>N/A</td> <td></td> <td>1.836+05</td> <td>•</td> <td>•</td> <td>5.046+02</td> <td>30 MI</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N/A                     |       | •          | M/A         | •               |                 | •               | A/A         | •               | N/A         | •               | 8/8          | •               | 3.06-10 1    | 1.36+01         | N/A            |                       | 1.836+05        | •                | •                 | 5.046+02        | 30 MI    |
| ·         I/A         ·         ·         ·         ·         ·         ·         ·         ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ft indired              | ~     | ct crash o | mto stora   | in tease abi    | o fire.         |                 |             |                 |             |                 |              |                 |              |                 |                | •                     |                 |                  |                   |                 |          |
| ·         I/A         ·         · <td>A/N</td> <td></td> <td>•</td> <td>N/A</td> <td></td> <td>A/A</td> <td>ı</td> <td>N/3</td> <td>•</td> <td>N/A</td> <td>•</td> <td>N/A</td> <td>•</td> <td>3.36-12-1</td> <td>1.3€+01</td> <td>1.46-10 1</td> <td>10+37</td> <td>1.19€+05</td> <td>'</td> <td>,</td> <td>1.30€+02</td> <td>Ţ</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | A/N                     |       | •          | N/A         |                 | A/A             | ı               | N/3         | •               | N/A         | •               | N/A          | •               | 3.36-12-1    | 1.3€+01         | 1.46-10 1      | 10+37                 | 1.19€+05        | '                | ,                 | 1.30€+02        | Ţ        |
| XF+01     N/A     -     N/A     -     N/A     -     N/A     -     N/A     -     0.065-00     3.105-02     4       XF+01     N/A     -     N/A     -     N/A     -     N/A     -     0.065-00     3.105-02     4       XF+01     N/A     -     N/A     -     5.46-10     1.35-01     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00     -     4.055-00<                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N/N                     |       | •          | N/N         | •               | A/A             | •               | A/A         | ,               | M/A         | •               | N/A          | 1               | 3.46-12 1    | 1.36+01         | N/N            | ,                     | 1.49€+05        | •                | •                 | 1.73€+02        | ÷        |
| XF-01     N/A     N/A     N/A     -     N/A     -     4,005 +00     4,295 +01     -     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00     4,005 +00 </td <td>7.06-11</td> <td></td> <td>1. 36 + 01</td> <td>N/N</td> <td>•</td> <td>A/A</td> <td>•</td> <td>N/A</td> <td>•</td> <td>N/N</td> <td>•</td> <td>W/W</td> <td>•</td> <td>N/A</td> <td>•</td> <td>W/W</td> <td></td> <td>3.576+04</td> <td>•</td> <td>6. 00E+00</td> <td>3.106-02</td> <td>÷</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 7.06-11                 |       | 1. 36 + 01 | N/N         | •               | A/A             | •               | N/A         | •               | N/N         | •               | W/W          | •               | N/A          | •               | W/W            |                       | 3.576+04        | •                | 6. 00E+00         | 3.106-02        | ÷        |
| - N/A - N/A - N/A - N/A - N/A - 3.46-12.1.35-01 N/A - 4.915-04 - 4.005-00 5.995-02 4.16<br>35-01 N/A - N/A - N/A - N/A - N/A - 0.01.945-00 1.945-00 4.8<br>35-01 N/A - N/A - N/A - N/A - 1.35-12 - N/A - 7.205-03 - 1.405-00 2.475-00 4.1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 1.36-11                 |       | 1.36+01    | A/A         | •               | A/A             | ŀ               | A/M         | •               | <b>N/N</b>  | •               | 5.46-10 1.   | .36+01          | 3. JE-12 1   | 10+3(.)         | N/N            | •                     | 1.95£+04        | '                | 4. J0E+00         | 4.296-02        | -        |
| 15-01 и/а - 1,205-03 - 1,405+00 1,945+00 4,1<br>15+01 и/а - и/а - и/а - и/а - 3.35-12 - и/а - 9.795+03 - 1,405+00 2,4,15+00 4,1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N/N                     |       | •          | A/A         | •               | A/N             | •               | #/#         | •               | A/A         | '               | N/A          |                 | 3.46-12 1    | 1. JE +01       | A/A            |                       | 1.916+04        | •                | 4.005+00          | 5.996-02        | Ŧ        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 7.06-11                 |       | 1.3£+01    |             | •               | N/N             | •               | A/A         | ,               | N/N         | •               | W/W          | •               | R/A          | •               | N/N            | ,                     | 7.206+03        | •                | 1.40€+00          | 1.946+00        | -        |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 7. 36-11                | _     | 10+35.     | N/A         | •               | A/N             | •               | N/A         | ,               | N/N         | 1               | R/A          | •               | 3.36-12      | •               | N/A            | •                     | 9.796+03        | •                | 1.40€+00          | 2.476+00        | Ę        |

See notes at end of table.



DISTREEMEN DISTING INTERNET

، ، ، ، ، ، کړې

(<u>16</u>)

Γ.



61666600

File: CONSTOR.UK1 date21-Aug-B7 page 11

SIORAGE ACCIDENTS ~ IFrequency units given at bottom of table) FOR MUNITIONS AT EXISTING SITES

DURATION TINE

.....

| tased      | LUS.                 | 00+386              | 346-02                                                             | 106-02             | 746-02      | 306+01      | 296-01   | '          | 00E+00   | 246-03   | 146-02   | M-318    | 256-03    | 246+01   | 736+01   | 325+01    | 64E -02  | 496-02     | 6 <b>8E-</b> 02 | M)-349.    | 195-04    | 10-389    | 001-380     | 106+00   | 27E+00    | 916-04    | 66E-04    | 10E-04    | 176+01     |
|------------|----------------------|---------------------|--------------------------------------------------------------------|--------------------|-------------|-------------|----------|------------|----------|----------|----------|----------|-----------|----------|----------|-----------|----------|------------|-----------------|------------|-----------|-----------|-------------|----------|-----------|-----------|-----------|-----------|------------|
| and Rel    | NED EN               |                     | -00<br>-00<br>-00<br>-00<br>-00<br>-00<br>-00<br>-00<br>-00<br>-00 | +00 J.             | E+00 4.     | 'n          |          | ,          | - 2.     | - 2.     |          | E+01 8.  | E+01 1.   | E+00 1.  | E+00 1.  | E+00 2.   | E+01 4.  | E+01 6.    | E+01 B.         | E+00 4.    | E+00 b.   | E+00 B.   | E+01 5.     | E+01 7.  | E+01 9.   | E+01 1.   | E+01 2.   | E+01 3.   | E+01 1.    |
| ulable     | 1.0%<br>No.134       | 1.40                | 3.20                                                               | 3.20               | 3.206       |             |          | _          |          |          |          | 1.051    | 1.051     | - 6.50   | . 4.50   | 6.50      | 1.178    | . 1.17     | 1.1.1           | - 6.00     | 6.00      | - A.00    | . 1.45      | 1.451    | 11.151    | 1.45      | 1.454     | . 1.45    | 1.07       |
| Agent Ava  | LBS.<br>SPILLED      |                     | •                                                                  | '                  | '           |             | •        | 4.805+04   | ,        | •        | •        | •        | •         | •        |          | •         | •        | •          | ,               | •          | •         |           | •           | •        |           | ·         | •         |           | ,          |
|            | MGENT<br>AVAIL.      | 1.446+04            | 1.44E+04                                                           | 1.966+04           | 2.926+04    | 2.07E+05    | 1.466+05 | LASS.      | 5.39€+06 | 2.21E+05 | 2.996+06 | 2.726+04 | 3.865-04  | 3.596+04 | 5.02E+04 | 6.72E+04  | 6.466+04 | 9.04E+04   | 1.216+05        | 3.316+04   | 4.4E+04   | 6.20E+04  | 3.296+04    | 4.596+04 | 6.00E+04  | 3.296+04  | 4.596+04  | 6.00E+04  | 2.126+04   |
|            | RANGE<br>FACTOR      |                     | •                                                                  | •                  | •           | •           | •        | ŗ          | 1. 1E+01 | •        | •        | •        | 1.36+01   | •        | 1.36+01  | •         | •        | '          | ł               | •          | 1. 3€+01  | ;         | •           | 1.3E+01  | •         | •         | 1.36+01   | •         | •          |
|            | UNDA<br>FRED         |                     |                                                                    | W/W                | A/A         | N/N         | N/A      | N/A        | 1.75-08  | N/N      | N/A      | N/N      | 1.4E-10   | N/A      | 1.45-10  | N/A       | N/N      | A/A        | N/N             | N/N        | 1.46-10   | N/A       | N/H         | 1.46-10  | N/A       | N/A       | 1.46-10   | N/A       | N/A        |
|            | RANGE<br>FACTOR      |                     | •                                                                  | 1.36+01            | 1. 36+01    | 1.36+01     | •        | 1.0E+01    | •        | 1.36+01  | ,        | •        | 1.36+01   | •        | 1.3£+01  | 1. JE +01 | '        | 1.3£+01    | 1.3€+01         | •          | 1. 3E+01  | 1.3€+01   | •           | 1.36+01  | 1. 3E+01  | •         | 1. 3E +01 | 1. JE +01 | ı          |
|            | TEA <b>B</b><br>FRED | 1.46-12             |                                                                    |                    | N/N         | 3. 36-12    | N/A      | 3.56-09    | R/A      | 3. 36-12 | M/A      | N/N      | 3.36-12   | N/N      | 3. 36-12 | 3.4E-12   | 8/8      | 3. XE-12   | 3.46-12         | N/A        | 3. 36-112 | 3.4E-12   | N/N         | 3.36-12  | 3.46-12   | A/N       | 3. 36-12  | 3.4E-12   | N/A        |
|            | RANGE<br>Factor      |                     | •                                                                  | 1.3€+01            | •           | 4           | •        | •          | •        | •        | •        | ,        | ٠         | •        | •        | ,         | ۰        | 1.36+01    | •               | •          | '         | •         | •           | ,        | '         | •         | •         | •         | •          |
|            | PUDA<br>Fred         | N/A                 |                                                                    | 5.4E-10            | N/N         | N/A         | A/N      | N/N        | N/N      | N/A      | AUN.     | N/N      | N/A       | N/A      | N/N      | N/A       | N/N      | 5.46-10    | A/N             | N/N        | N/N       | N/A       | N/A         | N/N      | N/A       | N/A       | N/A       | N/A       | A/A        |
|            | RANKE<br>Factor      |                     | •                                                                  | 1                  | •           | •           | •        | 1.06+01    | •        | •        | ,        | •        | 1. XE+01  | •        | •        | •         | '        | •          | •               | •          | ,         | •         | •           | •        | ,         | •         | '         | •         | •          |
|            | P <b>BA</b><br>FREQ  | V.                  |                                                                    | N/N                | N/A         | N/A         | N/A      | 9.76-09    | A/M      | N/A      | 474      | N/A      | 1.46-11   | N/A      | W/W      | N/N       | N/A      | A/H        | R/A             | A/M        | N/N       | N/N       | A/A         | N/A      | A/N       | N/A       | N/A       | N/A       | ¥/¥        |
| ε          | RANGE<br>FACTOR      | ,                   | •                                                                  | '                  | •           | •           | •        | •          | •        | '        | 1.16.01  | •        | •         | •        | •        | •         | •        | •          | ,               | •          | ı         | •         | •           | ,        | •         | •         | •         | •         | ,          |
| Fr equenci | FRED                 | A/4                 |                                                                    | N/N                | A/N         | 8/8         | N/A      | <b>N/A</b> | N/A      | N/A      | 1.16-09  | N/A      | A/N       | N/A      | A/A      | N/A       | N/A      | A/N        | N/A             | N/A        | N/A       | N/N       | N/A         | N/A      | N/A       | N/A       | 4/N       | N/N       | N/N        |
| Accident   | RANGE<br>FACTOR      |                     | •                                                                  | •                  | •           | ,           | •        | •          | •        | '        |          | •        | •         | •        | •        | •         | •        | •          | 1.36+01         | •          | •         | 1. 36 +01 | •           | '        | I.3E+01   | •         |           | •         | •          |
|            | LING<br>Free         | <b>V</b> /1         |                                                                    | A M                | N/N         | V/N         | N/A      | R/A        | A/M      | N/N      | A.A      | N/A      | N/A       | #/#      | A/M      | N/A       | N/N      | N/A        | 4.25-11         | N/A        | N/A       | 4.25-11   | M/A         | N/A      | 11-32.4   | N/A       | A/M       | A.A       | R/N        |
|            | RANGE<br>FACTOR      | •                   | '                                                                  | •                  | •           | '           | •        | 1.06+01    | •        | '        | •        | ,        | •         |          | •        | •         | •        | •          | •               | ,          |           | •         | •           | •        | •         | •         | •         | ,         |            |
|            | ar<br>Bag            | A.4                 |                                                                    | A/M                | <b>W</b> /W | N/N         | A/A      | 2.66-09    | N/N      | N/A      | M/A      | N/A      | A/A       | N.A      | N/A      | A/M       | R/A      | M/M        | N/A             | N/A        | M/M       | N/A       | M/A         | N. A     | N/A       | N A       | N/A       | N/A       | N/A        |
|            | RANGE<br>FACTOR      |                     | 1.36+01                                                            | 1.36+01            | · ·         | •           | 1.3£+01  | •          | •        | •        | •        | 1.36+01  | I. XE +01 | 1.35+01  | 1.36+01  | •         | 1.31+01  | 1. X •01   | •               | 1. 36 • 01 | 1. 3£+01  | •         | 1.35+61     | 1.35 +01 |           | 1. 36 +01 | 1.35-01   |           | 1. 32 •01  |
|            | ANAS<br>FRED         | <b>4</b> / <b>1</b> | 7.05-11                                                            | 7.36-11            |             | <b>W</b> /W | 7.06-11  | N/A        | N/A      | N/A      | 8/8      | 7.06-11  | 7.36-11   | 7.06-11  | 7. X-11  | N/A       | 7.06-11  | 7.36 11    | 8/N             | 7.06-11    | 7. 36-11  | A/M       | 7.06-11     | 1.36-11  | A/A       | 7. 06 11  | 7. JE -11 | 6/H       | 7.06-11    |
|            | ġ                    |                     | 2                                                                  | : 2                | 2           | 2           | 2        | 2          | 20       | 20       | 20       | 20       | 20        | 2        | 2        | 2         | 2        | 2          | 2               | 2          | 2         | 20        | 50          | ዳ        | 20        | 20        | 20        | 20        | 29         |
|            | 10                   | 19                  | â                                                                  | đ                  | â           | â           | ġ        | -          |          | 3        |          | Ĵ        |           |          | <b>B</b> | ĝ         | Ĵ        | <b>1</b> 6 | æ               | ġ          | 9         | ŝ         | â           | Ĵ        | <b>16</b> | 3         | 161       | Ĵ         | ( <b>1</b> |
|            | SCENNE               | 8                   |                                                                    | 1 8<br>1<br>1<br>1 | 8<br>       | ŝ           | 3        | S 10PE     | (HE)     | 00 J     | 1        | . (PO.   | 1 (80     | C 160    | (B)      | .681 3    | С 160    | С<br>В     | 80<br>0         | ر ۱۵۵      | ر<br>بو   | £ 189     | C 160       | 9<br>9   | 68) 3     | ť 180     | 98)<br>J  | 199       | 99<br>U    |
|            |                      |                     | 33                                                                 | 3                  | 3           | 31.5        | 3        | 100        | 10 M     | SLAVI    | 112      | € K      | SL MV     | 5. PS    | 2, 76    | 3, 2,     | 22       | 3          | £ 3             | 23         | 2         | 23        | 98 15<br>15 | 3        | 38.75     | 2.92      | S B       | 85        | St #6      |

¥ 爭

۹

¥ 쯀 Ŧ 筀

÷

¥

\*\*\*

ÉDOR DE ESTERE ESTEREN ESTEREN ESTEREN ESTEREN DE ESTERE DE ESTERE DE ESTERE ESTERE DE ESTERE DE ESTERE ESTERE

file: CBMST@D.MLI Aute21-Aug-07 page 12

STORAGE ACCIDER:5 - (Frequency units given at bottom of table)

| • | Si tes  |  |
|---|---------|--|
|   | 9011511 |  |
|   | UN SHE  |  |
|   |         |  |
|   | ä       |  |

See notes at end of table.

and the second secon

in the

I-14







IN TAXABLE AND

File: COMSTOK.MM. | date21-Aug-87 page 13

States and the states of the

SIORAGE ACCIDENTS - (Frequency units given at bottom of table) FOR munilions al Elistime SiTES

|                                                                                                  |                                                                                                |                                                                                     |                                                                                       |                                                                                 |                                                                     |                                                            |                                               |                                     | Accident F             | r suanba i   |     |             |                 |              |                 |              |                 |                  |                | -               | lgent Ava       | ulatie and                   | fiel oased      |                  |
|--------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------|---------------------------------------------------------------------------------|---------------------------------------------------------------------|------------------------------------------------------------|-----------------------------------------------|-------------------------------------|------------------------|--------------|-----|-------------|-----------------|--------------|-----------------|--------------|-----------------|------------------|----------------|-----------------|-----------------|------------------------------|-----------------|------------------|
| NO. AMAD RANGE APG RANGE LEAD RANGE MAAP RANG<br>File Falton Freg Falton Freg Falton Freg Falton | NO. AMAD RANGE APG RANGE LEAD RANGE MAAP RANG<br>Freq Factor Freq Factor Freq Factor Freq Fact | L. AMAD RANKE APG RANGE LAAD RANGE MAAP RANG<br>Freg Falton Freg Facton Freg Factor | D RANGE APG RANGE LAAD RANGE MAAP RANG<br>3 Falton Freg Facton Freg Factor Freg Facto | CANGE APG RANNGE LEND RANGE NAAM RANN<br>Ktom Freg Factom Freg Factor Freg Fact | APG RANGE LEAD RANGE MAAP RANG<br>Freg Factor Freg Factor Freg Fact | RANGE LEAD RANGE MAAF RANG<br>Factor freg factor freg fact | LEAD RANGE MAAF RANG<br>Freg factor freg fact | RANGE MAAF RANG<br>Factor freq fact | HAAP RANG<br>Freq Fact | FACT<br>FACT |     | PBM<br>FREQ | RANGE<br>Factor | PUDA<br>Frig | RANGE<br>Factor | TEAD<br>FRED | RANGE<br>Factor | LINGA<br>FREG    | ANGE<br>FACTOR | AGENT<br>Avail. | 185.<br>SPILLED | LBS.<br>Detomater            | LBS.<br>Emitted | DURATION<br>TINE |
|                                                                                                  |                                                                                                |                                                                                     | ··· ····· ····· ······ ······ ······ ····                                             |                                                                                 |                                                                     |                                                            |                                               |                                     | 0.06.440               | :            | : ' |             |                 | ¥,           |                 | V.           |                 | 4/1              |                | 2. 996 - DA     | ;               |                              |                 | ,                |
|                                                                                                  |                                                                                                |                                                                                     |                                                                                       |                                                                                 |                                                                     |                                                            |                                               |                                     |                        |              |     | 1 AC. AD    | 10.01           |              |                 | 10-31        | 1 45 401        | 1 05-00 2        | 16+01          | 1 945-04        | ;               | 3, 156 401                   | 2.556-04        | 9H 9             |
|                                                                                                  |                                                                                                |                                                                                     |                                                                                       |                                                                                 |                                                                     |                                                            |                                               |                                     |                        |              |     | 10-20-V     |                 |              |                 |              | 10-11-1         | 1 75-00 2        | 10-31          | 175.04          | ;               | A. 565 AD                    | 2 BAG +00       | 9                |
|                                                                                                  |                                                                                                |                                                                                     |                                                                                       |                                                                                 |                                                                     |                                                            |                                               |                                     |                        |              | •   |             |                 |              |                 |              |                 |                  |                |                 |                 |                              |                 |                  |
| 22 4.7E UP 2.6E+01 N/A 4.7E-09 2.6E+01 N/A                                                       | 22 4.7E UP 2.6E+01 M/A 4.7E-09 2.6E+01 M/A                                                     | • 4,7E UP 2,6E+01 M/A 4,7E-09 2,6E+01 M/A                                           | UP 2.6E+01 M/A 4.7E-09 2.6E+01 M/A                                                    | .66+01 M/A 4.7E-09 2.66+01 M/A                                                  | N/A 4,7E-09 2,6E+01 N/A                                             | 4 /E -09 2.6E+01 N/A                                       | 4./E-09 2.6E+01 N/A                           | 2.6E+01 N/A                         | N/N                    |              | •   | A/M         | •               | 4.75-09      | 2.66 +01        | 1.06-07      | 2.66+01         | A/A              | ,              | 1.21E+05        | :               | 1.1/6+01                     | 7-76-1          | ž                |
| 22 4.75-09 2.66401 N/A - 4.76-09 2.66401 H/A                                                     | 22 4.75-09 2.66401 N/A - 4.76-09 2.66401 N/A                                                   | . 4.76-09 2.66401 N/A 4.76-09 2.66401 N/A                                           | -09 2.46+01 N/A 4.76-09 2.66+01 N/A                                                   |                                                                                 | H/A - 4,76-09 2,66+01 H/A                                           | - 4, 7E-09 2, 6E+01 H/A                                    | 4.7E-09 2.6E+01 N/A                           | 2.6E+U1 N/A                         | A/H                    |              | •   | N/A         | •               | N/A          | •               | 1.06-07      | 2. b£ +01       | 4.7E-09 2        | 10+39.         | e. 20E +04      | :               | 6.00E+00                     | 1.056-04        | 9<br>1<br>1<br>9 |
| 22 3.4E-09 2.6E+01 N/A 5.4E-09 2.6E+01 N/A                                                       | 22 3.4E-09 2.4E+01 N/A 5.4E-09 2.4E+01 N/A                                                     | 3.46-09-2.66+01 NrA 5.46-09-2.66+01 N/A                                             | -09 2.46+01 N/A 5.46-09 2.66+01 N/A                                                   | 6E+01 N/A 5.4E-09-2.6E+01 N/A                                                   | N/A 5.46-09 2.46+01 N/A                                             | - 5, 4£-09 2, 6£+01 N/A                                    | 5.4E-09 2.6E+01 N/A                           | 2.6E+01 N/A                         | N/A                    |              | •   | N/A         | •               | N/A          | •               | 7. bf(18     | 2.6€+01         | 3.4E-09 2        | 10+39          | 6.00E+04        | :               | 1.45€+01                     | 2.80£+00        | 9<br>19          |
| 22 NiA - NIA - NIA - NIA                                                                         | 22 N/A - N/A - N/A                                                                             | NIA - NIA - NIA                                                                     | ALA - NIA - NIA - NIA                                                                 | Ala - Ala - Ala -                                                               | N/A - N/A - N/A                                                     | - N/A - N/A                                                | N/A - N/A                                     | - N/A                               | B/B                    |              | •   | N/N         | 1               | N/N          | ı               | 90.34.(      | 2.66+01         | 3.4E-09 2        | 10+39.         | 6. UDE 111      | ;               | 1.456+01                     | 1.052-04        | <b>坐</b> 9       |
| 22 5. 4E -09 2. 66+01 N/A 5. 9E -09 2. 66+01 N/A                                                 | 22 5.4E-09 2.6E+01 N/A 3.9E-09 2.6E+01 N/A                                                     | 5. 4E -09 2. 6E+01 N/A 3.9E-09 2.6E+01 N/A                                          | -09 2.66+01 M/A 3.9E-09 2.66+01 M/A                                                   | 66+01 N/A 3.9E-09 2.6E+01 N/A                                                   | H/A 3.9E-09 2.6E+01 H/A                                             | 3.9E-09 2.6E+01 M/A                                        | 3.9E-09 2.6E+01 N/A                           | 2.6E+01 N/A                         | A/A                    |              | •   | 3.96-09     | 2.66+01         | A/A          | •               | 8.96-08      | 2.66+01         | 3.96-09 2        | 10139.         | 4.046.04        | ł               | 2.146+01                     | 5.806+00        | ¥.               |
| 22 3, 46-09 2, 66+01 M/A 7, 96-09 2, 66+01 M/A                                                   | 22 3, 95-09 2, 66+01 N/A 7, 96-09 2, 66+01 N/A                                                 | 3, 46 - 09 2, 66 + 61 M/A 3, 96 - 09 2, 66 + 01 M/A                                 | -UY 2.66+01 N/A 7.96-UP 2.66+01 N/A                                                   |                                                                                 | W/A 2.9E-09 2.6E+01 N/A                                             | - 1.9E-09 2.6E+01 N/A                                      | 1.9E-09 2.6E+01 N/A                           | 2.6E+01 N/A                         | A/M                    |              | •   | 3.96-09     | 2.66+01         | A/A          | •               | B. 9E -08    | 2.46+01         | 3.96-09 2        | 10+39.         | 3. 7BE+04       | ;               | 2.00€+01                     | 2.00K-E         | 9<br>9           |
| 22 N/A N/A N/A N/A                                                                               | 22 h/A . N/A . N/A . N/A                                                                       |                                                                                     |                                                                                       |                                                                                 |                                                                     |                                                            | N/A - N/A                                     | - N/A                               | R/A                    |              | ľ   | A/A         | •               | N/A          | •               | A/A          | •               | 0.0€+00          | •              | 1.366+04        |                 | •                            |                 | •                |
| 22 N/A N/A N/A N/A                                                                               | 22 N/A - N/A - N/A - N/A                                                                       |                                                                                     |                                                                                       |                                                                                 |                                                                     |                                                            | R/A - A/A                                     | 8/M -                               | M/A                    |              | •   | N/N         | ,               | N/N          | 1               | 0.0€+00      | •               | N/A              | •              | 1.836+05        | •               |                              |                 | •                |
| sado generated missiles strike the slorage igion and cause munition detonal:                     | ado generated missiles strike the storage igloo and cause munition detonal:                    | werated asssiles strike the storage igloo and cause munition detonat:               | aissiles strike the storage igloo and cause munition detonati                         | les strike the storage igion and cause munition detonat:                        | ce the storage igloo and cause munition detonat                     | orage igloo and cause munition detonat                     | a and cause munition detonat                  | se aunition detonal                 | a detonat              |              |     |             |                 | 1            |                 |              |                 | :                |                |                 |                 |                              |                 |                  |
| 23 N/A N/A - N/A N/A                                                                             | 2.3 N.A. N.A. N.A. N.A.                                                                        |                                                                                     | U/2 · U/2 · U/2 · U/2                                                                 |                                                                                 |                                                                     |                                                            |                                               | W/W                                 | W/W                    |              | •   | A/A         | •               | N/N          | •               | 0.04+00      | •               | 0. <b>0</b> + 00 | •              |                 |                 |                              |                 | '                |
| 23 3.4E-13 9.9E+01 NJA NJA NJA                                                                   | 23 3.4E-13 9.9E+01 NJA - NJA - NJA                                                             | 1 3.4E-13 9.9E+01 NJA NJA NJA                                                       | -15 9.9E+01 N/A N/A N/A                                                               | 95+01 B/M - B/M - B/M                                                           |                                                                     |                                                            | N.A - N/A                                     | A/A -                               | A/A                    |              | i.  | N/A         | •               | 2.26-14      | 9.96+01         | 3.26-16      | 9.95+01         | N/A              | •              | 4.91E+04        | •               | . b.00E+0C                   | 6.50E-U         | ¥ -9             |
| 23 3,46-13 9,96+01 N/A - N/A - N/A                                                               | 23 3.46-15 9.96+01 N/A - N/A - N/A                                                             | 1 3.4E-15 9.9E+01 N/A - N/A - N/A                                                   | -15 9.9E+01 N/A - N/A - N/A                                                           | .95+01 N/A - N/A - N/A                                                          | 2/2 · 2/2 · 2/2                                                     | - N/A - N/A                                                | N/A - N/A                                     | - N/A                               | N/A                    |              | •   | N/A         | •               | N/N          | •               | 3.26-16      | 9.9€+01         | <b>W/W</b>       | •              | 1.466+04        | ,               | 1.605+0(                     | 5.456-0         | ¥ 9              |
| 23 3.4E-13 9.9E+01 N/A N/A N/A                                                                   | 23 3.46-13 9.96+01 N/A - N/A - N/A                                                             | 1 3.46-13 9.9E+01 N/A - N/A - N/A                                                   | -13 9,9E+01 N/A - N/A - N/A                                                           | 9E+01 N/A N/A N/A                                                               | N/A - N/A - N/A                                                     | - N/A - N/A                                                | A/M - A/M                                     | A/M -                               | N/N                    |              | ı   | A/A         | ı               | 2.26-14      | 9.96+01         | N/A          | •               | N/A              | •              | 2.926+04        |                 | - 3.20£+00                   | P-305-9         | ¥.               |
| 23 N/A N/A N/A N/A                                                                               | 23 N/A N/A N/A N/A                                                                             |                                                                                     | 14 - N/A N/A N/A                                                                      | - N/A - N/A - N/A                                                               | N/A N/A N/A                                                         | N/A N/A                                                    | N/A - A/A                                     | A/N ·                               | N/A                    |              | •   | <b>B</b> /4 | •               | N/N          | •               | 0.06 400     | •               | N/A              | •              | 2.07E+05        | •               |                              |                 | •                |
| 23 0.0€+00 - N/A - N/A - N/A                                                                     | 23 0.06+00 - M/A - M/A - M/A                                                                   | : 0.0E+00 - N/A - N/A - N/A                                                         | •00 - N/A - N/A - N/A                                                                 | - N/A - N/A - N/A                                                               | A/A - A/A - A/A                                                     | - E/A -                                                    | N/A - N/A                                     | - N/A                               | N/A                    |              | •   | M/M         | •               | N/A          | •               | A/A          | •               | A/A              | '              | 1.466+05        | •               |                              |                 | •                |
| 23 N/A - U.VE+00 - N/A - N/A                                                                     | 23 N/A - U.VE+00 - N/A - N/A                                                                   | M/A - N/A - N/A - N/A                                                               | /A - U.VE+00 - N/A - N/A                                                              | - U.VE+00 - N/A - N/A                                                           | U.VE+D0 - N/A - N/A                                                 | - N/A - N/A                                                | N/A - N/A                                     | - N/A                               | N/A                    |              |     | 0.0€+00     | •               | N/N          | •               | 0.0E+00      | •               | N/A              | '              | LASS.           |                 |                              |                 | •                |
| 23 N/A - N/A - N/A - N/A                                                                         | 23 N/A - N/A - N/A - N/A                                                                       | . M/A - M/A - M/A                                                                   | 1A - N/A - N/A - N/A                                                                  | - N/A - N/A - N/A                                                               | N/A - N/A - N/A                                                     | - N/A - N/A                                                | N/A - N/A                                     | - N/A                               | N/A                    |              | •   | N/A         | •               | N/N          | •               | N/A          | •               | 0.0£+00          | •              | 5.39€+06        |                 | •                            |                 |                  |
| 23 Nr.A - Nr.A Nr.A Nr.A                                                                         | 23 NJA - NJA - NJA NJA                                                                         | 1. N.A . N/A . N/A                                                                  |                                                                                       | A14 A14 - A14 -                                                                 | N/A - N/A N/A                                                       | - N/A N/A                                                  | M/A M/A                                       | #/#                                 | A/A                    |              | •   | ¥/¥         | 1               | N/A          | •               | 0.0£+00      | •               | 8/8              | ;              | 2.216+05        |                 |                              |                 | •                |
| 25 R/A - N/A - 0.0E+00                                                                           | 22 III/A - N/A - 0.0E+00                                                                       | · N/A - N/A - 0.05-00                                                               | /A - M/A - N/A 0.0€+00                                                                | M/A - M/A 0.0E+00                                                               | M/A - 0.0E+00                                                       | - N/A 0.0E+00                                              | M/A 0.0E+00                                   | 0.05 •00                            | 0.05+00                |              | •   | N/A         | •               | N/N          | •               | N/A          | •               | N/A              | •              | 2.996+06        | ·               | •                            |                 | •                |
| 21 3.4E-13 9.9E+U1 N/A N/A N/A                                                                   | 21 2.4E-13 9.9E+UI N/A N/A N/A                                                                 | 1 3.4E-13 9.9E+U1 M/A M/A M/A                                                       | -13 9.9E+UI M/A - M/A - M/A                                                           | .9E+J1 N/A - N/A - N/A                                                          | N/A - N/A - N/A                                                     | - N/A - N/A                                                | N/A - N/A                                     | - N/A                               | N/A                    |              | •   | 1.06-13     | 9.96+01         | A/N          | •               | 7.46-16      | 9.96 +01        | 4.06-16 9        | 10+34.         | 3.866404        | •               | 3.154 101                    | 2.556-0         | 9¥<br>•9         |
| 23 3.4E-15 9.9E+01 N/A - N/A N/A                                                                 | 23 3.4E-13 9.9E+01 M/A M/A N/A                                                                 | : 3.4E-13 9.9E+01 M/A - M/A M/A                                                     | -13 9.9E+01 N/A - N/A N/A                                                             | 9E+01 N/A - N/A N/A                                                             | N/A - N/A N/A                                                       |                                                            | A/A A/A                                       | R/N                                 | N/A                    |              | •   | <b>M</b> /A | •               | N/A          | •               | 3. 26 - 16   | 9.96+01         | 4.06-16 9        | 10+34.         | 6.72E+04        | •               | <ul> <li>b.50€+00</li> </ul> | 2.806+01        | * .              |
| 23 3.46-13 9.96+01 N/A 3.46-13 9.56+01 N/A                                                       | 23 3.46-13 9.96+01 N/A 3.46-13 9.56+01 N/A                                                     | 3 3.4E-13 9.9E+01 N/A - 3.4E-13 9.5E+01 N/A                                         | -13 9,96+01 N/A 3,46-13 9,56+01 N/A                                                   | .96+01 N/A 3.46-13 9.56+01 N/A                                                  | N/A - 3.4E-13 9.5E+U1 N/A                                           | - 3,4E-13 9,5E+U1 N/A                                      | 3,4E-13 9,5E+U1 N/A                           | 9.5E+U] N/A                         | A/N                    |              | '   | N/A         | •               | 2.26-14      | 9.96+01         | 3. 26 - 16   | 9.95+01         | N/A              | ۱              | 1.21E+05        |                 | 1.176+01                     | 0-350 T         | ¥.               |
| 23 3.4E-15 9.9E+01 N/A 3.4E-13 9.9E+01 N/A                                                       | 23 3.46-15 9.96+01 N/A 3.46-13 9.96+01 N/A                                                     | 1 3.4E-15 9.9E+01 N/A 3.4E-13 9.9E+01 N/A                                           | -15 9.9E+01 N/A 3.4E-13 9.9E+01 N/A                                                   | 9E+01 N/A 3.4E-13 9.9E+01 N/A                                                   | N/A 3.4E-13 9.9E+01 N/A                                             | - 3, 4E -13 9, 9E +01 N/A                                  | 3.4E-13 9.9E+01 N/A                           | 9.9E+01 N/A                         | N/A                    |              |     | N/A         | •               | N/N          | •               | 3.26-16      | 9.95+0]         | 4.06-14 9        | 10+36.         | 4.20E+04        |                 | - 6.00E+0(                   | 0-1.056-0       | ¥ 9              |
| 23 2.46-13 9.96+01 N/A 3.46-13 9.96+01 N/A                                                       | 23 2.4E-13 9.9E+01 N/A 5.4E-13 9.9E+01 N/A                                                     | 1 1.4E-13 9.9E+01 N/A 3.4E-13 9.9E+01 N/A                                           | -13 9.9E+01 M/A - 3.4E-13 9.9E+01 N/A                                                 | .9E+01 N/A - 3.4E-13.9.9E+01 N/A                                                | H/A - 3.4E-13 9.9E+(1 H/A                                           | - 3,4E-13 9,9E+(i] N/A                                     | 3.4E-13 9.9E+(1 N/A                           | 9.9E+(i] N/A                        | N/A                    |              | •   | N/A         | •               | A/A          | '               | 3.26-16      | 9.96+01         | 4.06-16 9        | 10+36.         | 6.00E+04        |                 | - 1.456+0                    | 2.806+0         | ¥ 9              |
| LIS N/A - N/A - N/A - N/A                                                                        | C N/A - N/A - N/A - N/A                                                                        |                                                                                     | ra - N/A - N/G - N/H                                                                  | - N/A - N/A - N/A                                                               | N/A - N/A - N/A                                                     | 118 - 118 -                                                | N/6 - N/H                                     | . <b>N</b>                          | N, L                   |              | •   | A / N       |                 | N/A          | •               | 3. 26-16     | 9.96+01         | 4.06-16 9        | 10+34.         | 6.06E+04        | •               | - 1.456+01                   | 1.056-0         | 9<br>1           |

See motes at end of table.

IN SALES

22222

I-15
File: CONSTON.MXI date21-Aug-07 page 14

STORAGE ACCIDENTS - ifrequency units given at bottom of table) FOR Mumiltons AT Existing SITES

505652

12222223

|                |             |               |                 |               |                 | -4          | kcident F       | r aquenc s e |                 |              |                 |                     |        |             |         |             |          | hjen           | t Availal        | le and R        | leased    |                  |
|----------------|-------------|---------------|-----------------|---------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|---------------------|--------|-------------|---------|-------------|----------|----------------|------------------|-----------------|-----------|------------------|
| SCENAR IO      | 2           | ANAD<br>FISED | KANGE<br>FACTOR | APG<br>FRED   | RANGE<br>FACTOR | LIBAD       | RANGE<br>FACTOR | FRED         | RANGE<br>FACTOR | Pika<br>FREQ | RANGE<br>FACTOR | PUDA<br>FRED        | And C  | TEA0 R      |         | LINDA RA    |          | AIL. SPI       | 82.<br>11 10 100 | LBS.<br>Tomated | LDS. D    | URAT ION<br>TINE |
|                | ;           | :             |                 |               |                 |             | 1               |              |                 |              |                 | 1                   | •      |             |         |             |          | :              | •                |                 |           | !                |
|                | 3           | ·             | Y. YE +U1       |               | •               | · · · · · · | . ****          |              |                 | 1.11-12 \$   |                 | <b>4</b> / <b>8</b> |        |             |         |             |          | 5.7            |                  |                 |           | ž                |
| SLIPC<br>SLIPC | :2          | 3.46-13       | 9.96+01         | N/N           | •               | 3.46-13 9   | 10+34.          | N/A          |                 | 1.16-12 9    | 10+36.4         | 8/8                 |        | . K - IS 9. | 9E+01 4 | 10.9 41-30. | E-01 3.  | 101-301        |                  | 10-300          | 10-380"   | ¥<br>9           |
| SA SVS (16L)   | 23          | N/A           | •               | N/A           |                 | A/A         | •               | N/N          |                 | N/A          | •               | N/A                 | •      | N/N         |         | 00+30"      | <u>.</u> | 10-36          | •                | •               | •         | •                |
| STSVS (MH)     | 23          | N/A           | •               | N/N           | •               | N/A         | •               | A/A          | •               | M/A          | •               | N/N                 | •      | 00+30.      | •       | W/W         | -<br>-   | 50+3KI         | •                | •               | •         | 1                |
| SL 24 - L1gh   | taing st    | rites ton     | container       | s stored o    | but door s.     |             |                 |              |                 |              |                 |                     |        |             |         |             |          |                |                  |                 |           |                  |
| SLANS (OPEN)   | ~           | A.M           | •               | 1.46-10       | 1.0€+01         | N/A         |                 | N/A          | •               | 5.16-10 1    | 10+30.1         | A/A                 |        | .1 01-34.   | 06+01   | N/N         | -        | 106+03 1.7     | 0E+03            | •               | •         | 2 148            |
| SLZ5 - Num     | tions dr    | opped dur     | ing leaker      | 1 sel at 1 on | i eunitio       | n detonate  | ÷               |              |                 |              |                 |                     |        |             |         |             |          |                |                  |                 |           |                  |
| SL MGC         | អ           | N/A           | •               | N/N           | •               | M/A         | •               | N/A          | •               | W/W          | •               | N/A                 |        | 00+30.      | ,       | . 06+00     | -        | 10E+02         | •                | ,               | •         | •                |
| SL DHC         | 2           | 1.7E u7       | 2.46+01         | A/A           |                 | N/A         | •               | N/A          | •               | A/M          | •               | 1.76-07 2           | 10134  | .76-07 2.   | 10+34   | N/N         | - 2.1    | <b>10E</b> +02 | - <b>a</b>       | 00+300          |           | 2 148            |
| SLCGC          | 5           | 8.96-08       | 2.4E+01         | A/A           | •               | A/A         | •               | A/A          | •               | #/#          | •               | <b>NIA</b>          | ,      | 1.96-08 2.  | 10-34   | N/N         | ÷        | 10+ 30I        |                  | 00+307          | 10-351-01 | 2 🏨              |
| SLCMC          | 2           | 8. 91 - 08    | 2.6€+01         | N/N           | ,               | 8/W         | ,               | N/A          | •               | N/A          | •               | 8.96-08 2           | 10+34. | A/N         | ,       | N/A         |          | 105-01         | ,<br>,           | 20€+00          |           | ∰<br>~           |
| SLIEG          | 52          | M/A           | •               | N/A           | '               | N/A         | •               | N/N          | •               | A/A          |                 | N/N                 | •      | 00+30"      | •       | N/A         |          | 50E+03         | •                | •               | •         | '                |
| SLINC          | 2           | 0.0€+00       | •               | 0.01+00       | •               | N/N         |                 | N/N          | •               | N/A          | •               | N/N                 |        | 00+30.1     | ,       | .06+00      | -        | 70€+03         | •                | •               | •         | •                |
| SLEVE          | ដ           | <b>8</b> /8   | •               | N.N           | •               | A/A         |                 | 0.0€+00      | •               | A/A          | •               | N/N                 | ,      | 00+30"      | •       | N/A         | •        | LOE +03        | •                | •               | •         | •                |
| SLINC<br>SLINC | 2           | 1.36-07       | 2.66+01         | A.N           | •               | A/A         | •               | M/A          | •               | 1. 36-07 2   | 10+34.1         | N/A                 | •      | . XE-07 2.  | 1 10+39 | .36-07 2.6  | E+01 3.  | 70E +02        | +i<br>           | 10+351          |           | 2 149            |
| 31 PGC         | 22          | 3.26-08       | 2.66+01         | N/A           | •               | N/N         | ,               | N/N          | •               | <b>A</b> /N  | •               | N/N                 | 1      | i.ZE-08 2.  | PE+01 3 | .26-08 2.6  | E+01 5.  | 20E+01         | - <b>đ</b>       | S0E+00          | 00+308-1  | 2 ¥R             |
| SLPIE          | 22          | 3.26-08       | 2.46+01         | N/N           | •               | 3. 25 -08 2 | 10+34.1         | N/N          |                 | N/A          | •               | 3. 2E-08 2          | 10+34. | . 26-08 2.  | 10+39   | N/N         | 6        | 10+342         | -                | 176+01          | . 05£ -02 | 2 HB             |
| SLPVC          | <b>ب</b> تا | 3.26-08       | 2.46+01         | N/A           | •               | 5.26-06 2   | 10+34.          | N/N          | •               | N/N          | •               | A/A                 | ,      | L.2E-08 2.  | 10+39   |             | E+01 4.  | 10+30          | -                | 00+300          | 10-350    | 2.66             |
| SL.06C         | 23          | 1. ZE - 08    | 2.6£+01         | N/N           | •               | 3.26-08 2   | 10-34.5         | <b>M/A</b>   | •               | A/A          | •               | N/N                 | •      | . 26-08 2.  | 4E+01 3 | .26-09 2.6  | E+01 8.  | 10+301         | -                | 10+351          | 00+308-1  | 2 14             |
| SLAVC          | ង           | A/M           | •               | N/N           | •               | N/A         | ı               | W/W          | •               | R/N          | •               | N/N                 |        | ZE-09 2.    | 4E+01 3 | . 26-08 2.6 | E+01 B.  | 10E+01         | -                | 10+351-         | 10-350    | 2 ¥              |
| SLINGC         | <u>ب</u>    | 5.76-08       | 2.06+01         | N/N           | ,               | 5.76-08 2   | 10+34.5         | N/N          | •               | 5.76-08 2    | 1.46+01         | N/N                 | •      | .709 2.     | 10+34   | .76-00 2.4  | E+01 1.4 | 106+02         | - 2              | 10+3+1          | . BOE+00  | 2 HB             |
| SLRVC          | 2           | 5.76-08       | 2.4E+01         | N/A           | •               | 5. 76-00 2  | 10+34.5         | N/A          | •               | 5. 7E-00 2   | 7.46+01         | N/N                 | •      | .7E-08 2.   | 10+34   | .76-08 2.6  | -1 10+3  | 50E +02        | - 2              | 10+300          | 10-380"   | 5 H              |
| SL SVC         | 2           | N/A           | •               | N/A           | ,               | N/N         | •               | R/N          | •               | R/A          | •               | 8/B                 |        | 00+30"      | ,       | .06+00      | <br>י    | 56E+03         | ł                | •               | •         | '                |

NDTES:

See notes at end of table.











File: COMSTOR. MAI date21-Aug-87 page 15

#### 570RAGE ACCIDENTS - (Frequency units given at bottom of table) FOR MUMITIONS AT EXISTING SITES

|            | AT LON                   | 1 |             |
|------------|--------------------------|---|-------------|
| _          | E BUE                    | : |             |
| Keiezsed   | L <b>9</b> 5.<br>Emittei |   |             |
| lable and  | LBS.<br>De tomated       |   |             |
| Agent Avai | LBS.<br>SPILLED          |   |             |
|            | AGENT<br>AVASL.          |   |             |
|            | RAMEE<br>Factor          |   |             |
|            | UNDA<br>Free             |   |             |
|            | RANGE<br>FACTOR          |   |             |
|            | TEAD<br>Free             |   |             |
|            | RANGE<br>FACTOR          |   |             |
|            | PUDA<br>Freg             |   |             |
|            | RANGE<br>Factor          |   |             |
|            | PBA<br>FREQ              |   |             |
| 4          | Range<br>Finctor         |   |             |
| Frequenc   | FRED                     | : |             |
| Accident   | FANGE<br>FACTOR          | : |             |
|            | L 840<br>F RE2           |   | tion year   |
|            | RANGE<br>FACTOR          | : | s per mai   |
|            | AF6<br>Freq              |   | are events  |
|            | FAL TOR                  |   | , I otherad |
|            | ANAD<br>FREQ             |   | 1s tor st   |
|            | ,<br>M                   | ł | ינא מטין    |
|            | LCEMARIO                 |   | 1. Frequen  |

- ... Frequency units for scenarios 2, 9, and 25 are events per leaker.
- <sup>2</sup> Frequency units for scenarios 4, 5, 8, 15 through 21, and 23 are events set storage unit-year ligido or marehouse). For ton containers stared outdoors, frequency units for scenarios 8 and 24 are events per cluster-year of ton containers (15 TC/cluster).
- Agent release for SLEMS 1 (open) assumes putdoor spill pato a purous surface.
- 5. Frequency units for scenarios 7 and 22 are events per year.

ŧ

PAGE 1 21-Åug-87

STORAGE EARTHQUAKE - MAREHOUSES

| (PER YEAR)<br>Accident frequencies |
|------------------------------------|
| LB40 RANGE NA<br>Freq Factor Fr    |
| N/A N/A 1.                         |
| N/A N/A 9.                         |
| 1 W/W W/W                          |
| , N/A N/A 3.                       |
| N/A N/A 1.4                        |
| N/A N/A N/                         |
| N/A N/A N/I                        |
| N/A N/A N/                         |
| N/A N/A N/                         |
| N/A N/A N/                         |
| N/A N/A N/A                        |
| N/A N/A N/I                        |
|                                    |
| N/N N/N N/N                        |
| N/A N/A N/A                        |
|                                    |
| W/A N/A W/                         |
|                                    |
| N/A N/A N/I                        |
| N/A N/A N/A                        |
| N/A N/A N/A                        |
| N/A N/A N/A                        |
| N/A N/A N/                         |
| N/A K/A N/A                        |
| N/A N/A N/I                        |







 $\mathbf{X}$ 



The second



21-Aug-67 PAGE 2

STORAGE EARTHQUAKE - MAREHOUSES

|                    |            | RATION<br>TIME  | <b>筆筆筆</b><br>                           |
|--------------------|------------|-----------------|------------------------------------------|
|                    | RELEASED   | NITTED M        | . 4E +05<br>MEGL<br>. 4E +05             |
|                    | IBLE AND   | DNATED E        | сч сч<br>1 г г г                         |
|                    | NT AVAILI  | NS. L           |                                          |
|                    | AGE        | 26 L<br>MIL: 26 | 16 +06<br>16 +06<br>16 +117              |
|                    |            | ICR AV          | 10 10 10 10 10 10 10 10 10 10 10 10 10 1 |
|                    |            |                 | -10 2.7<br>-05 5.7                       |
|                    |            | FRE             | 3.66<br>5.66<br>1.16                     |
|                    |            | RANGE<br>FACTOR | 8/8<br>8/8<br>8/8                        |
|                    |            | TEAD<br>Fred    | 6/1<br>6/1                               |
| NSE S              |            | RANGE<br>Factor | 4 4 4<br>1 1 2<br>1 1 2                  |
| THE WARFH          |            | PUDA<br>FREG    |                                          |
| SOENTS IN          |            | RANGE<br>Factor | 4/1<br>4/1                               |
| NDUCED AC(         |            | PBA<br>Fred     | А/И<br>А/И<br>А/И                        |
| A THRUNKE - 1      |            | RANGE<br>FACTOR |                                          |
| - El<br>(PER YEAR) | REQUENCIES | HAAP<br>Freq    | 8/8<br>8/8<br>8/8                        |
| STORAGE            | ACCIDENT F | RANGE<br>Factor | €/1<br>₹/1                               |
|                    |            | LIND            | 8/8<br>8/8                               |
|                    |            | PANGE<br>FACTOR | R/N<br>R/N<br>R/N                        |
|                    |            | SA<br>Sa        | 4/N<br>4/N                               |
|                    |            | RANGE<br>FACTOR | N/N<br>N/A<br>N/A                        |
|                    |            | MAA)<br>Freed   | #/#<br>#/#<br>#/#                        |
|                    |            | 9               | 2015<br>2185<br>2185                     |
|                    |            | SCENNARIO       | **************************************   |

20



#### I.1.2. INTERIM STORAGE

The following tables list the accident results for interim storage of munitions.

EXERCISE SERVICE EXERCISE EXCERCISE EXCERCISE

K333,2333





Same and the second 
**(**•

file: STOKHOLD.WKI Page 1 Date 21-Aug-87

INTERIM STORAGE SCEMARIOS - MATTOMAL/REGIONAL DISPOSAL DF110NS (PER YEAR)

| SCENNRIO<br>19 | W             | <b>MAD</b>          | RANGE<br>FACTOR | APS         | RANGE<br>FACTOR | LBAD       | RANGE<br>Factor       | WAAP    | RANGE<br>Factor | V84      | RANGE<br>Factor | PUDA       | RANGE<br>ACTOR | TEAD<br>F  | RANGE<br>ACTOR | ed MU                  | RANGE<br>Factor | AGENT<br>IVALLABLE | LBS.<br>SPILLED E | LBS.<br>Ketanated | LBS. D<br>Emitted | URATION<br>TINE |
|----------------|---------------|---------------------|-----------------|-------------|-----------------|------------|-----------------------|---------|-----------------|----------|-----------------|------------|----------------|------------|----------------|------------------------|-----------------|--------------------|-------------------|-------------------|-------------------|-----------------|
|                | ;             |                     |                 |             |                 |            | -<br>-<br>-<br>-<br>- |         |                 |          |                 |            |                |            |                |                        |                 |                    |                   |                   |                   |                 |
| Large air      | rcraft        | crash o             | nta cont        | at ner 5; 1 | no fire         |            |                       |         |                 |          |                 |            |                |            |                | :                      | :               | 2                  |                   |                   |                   | -               |
| SKIBES         | _             | N/A                 | :               | N/A         | :               | K/A        | :                     | A/A     | ;               | N/A      | ;               | N/A        | ;              | 3. 6E - 10 | 2              | 1.55-09                | 2               | 3.76+05            | 3.1E+05           | :                 | ;                 | Ĩ               |
| SPONC          | -             | 7.86-09             | 0               | N/A         | :               | N/A        | ;                     | N/A     | ;               | N/A      | ;               | 5.96-08    | 9              | 3.66-10    | 2              | N/A                    | :               | 1.66+05            | 1. IE+05          | 2.4E+04           | ;                 | Ħ               |
| SACGC          | -             | 1.86-09             | 2               | N/A         | :               | R/A        | ;                     | A/A     | ;               | N/A      | ;               | N/A        | ;              | N/A        | :              | N/A                    | :               | 6. SE +04          | 1.56+04           | 9.7E+03           | :                 | Ĭ               |
| SRCHC          | -             | 7.85.09             | 01              | 8/8         | :               | N/A        | ;                     | N/A     | ;               | A/A      | ;               | 5.96-08    | 2              | 3.6E-10    | 01             | A/A                    | 1               | 1. 36+05           | 9.06+04           | 1.96+04           | ;                 | Ĩ               |
| SR1.65         | -             | N/A                 | :               | M/A         | ;               | 47#        | ;                     | N/A     | :               | M/A      | ;               | A/A        | ;              | M/A        | ;              | M/6                    | :               | 4.26+05            | 3.46+05           | :                 | ;                 | £               |
| SRKHS          | 1             | 7.86-09             | 01              | 5. 36-10    | 91              | N/A        | ;                     | N/A     | :               | 1.56-09  | 9               | A/A        | :              | 3.66-10    | 2              | 1.35-08                | 9               | 4. BE +05          | 4.05+05           | :                 | ;                 | £               |
| SREVS          | -             | N/A                 | ;               | A/M         | :               | R/A        | :                     | 4.4E-09 | 2               | M/A      | ;               | N/A        | ;              | 3.66-10    | 9              | R/A                    | ;               | 4. SE + 05         | 3.86+95           | ;                 | :                 | Ĩ               |
| SRRVC          | -             | 7.85-09             | 01              | A/A         | :               | R/A        | :                     | N/A     | ;               | 1.56-09  | 9               | N/A        | ;              | 3.6E-10    | 10             | 1.56-08                | 2               | 1.66+05            | 1. 1E+05          | 2.4E+04           | :                 | RHI             |
| SRPGC          | -             | 7.86-09             | 9               | N/A         | :               | N/A        | ;                     | A/A     | :               | N/A      | ;               | NIA        | ;              | 3.6E-10    | 01             | 1.5E-08                | 2               | 1.16+05            | 7.46+04           | 1.65+04           | ;                 | H:              |
| SRPHC          | 1             | 7.86-09             | 9               | N/A         | :               | 4.56-09    | 2                     | N/A     | ;               | N/A      | ;               | 5.96-08    | 2              | 3.6E-19    | 01             | NrA                    | :               | 2.0€+05            | 1.4£+05           | 2.9E+04           | ;                 | Ŧ               |
| SAFVC          | -             | 60-3 <b>8</b> °,    | 01              | N/A         | :               | 4.56-09    | 01                    | A/M     | ;               | A/A      | :               | N/A        | 1              | 3.6E-10    | 2              | 1. 5E - ng             | 9               | 1.0€+05            | 7.1E+04           | 1.55+04           | :                 | Ĩ               |
| SPREC          | -             | 7.86-09             | 2               | N/A         | ;               | 1.56-09    | 2                     | N/A     | ;               | N/A      | ;               | K/A        | :              | 3.66-10    | 0              | 1. 3E -08              | 2               | 1.2€+05            | B. 3E +04         | 1.85+01           | ;                 | <u>۲</u>        |
| SROVC          | -             | 8/A                 | ;               | <b>M/A</b>  | :               | N/A        | :                     | N/A     | ;               | N/A      | ł               | N/A        | :              | 3.6E-10    | 10             | 1.56-08                | 9               | 1.2£ +05           | 8.5E+04           | 1.8€+04           | :                 | Ŧ               |
| SPREC          | -             | 7. BE -09           | 10              | R/A         | ;               | 4.56-09    | 2                     | N/A     | ;               | 1.36-09  | 01              | M/A        | 1              | 3.66-10    | 01             | :. <b>3</b> E -09      | 2               | 9. JE+04           | 6. JE +04         | 1. 36+04          | ÷                 | Ŧ               |
| SRRVC          | -             | P. BE - 09          | 10              | M/A         | :               | 4, 56 - 09 | 61                    | ¥/¥     | :               | 1.56-09  | 10              | A/A        | ;              | 3.66-10    | 2              | 1.56-08                | 01              | B. 4E+04           | 5.96+04           | 1. JE+04          | ;                 | ¥H              |
| SASVS          |               | #/#                 | ł               | N/A         | ;               | N/A        | ;                     | N/A     | 1               | M/A      | ;               | N/A        | ;              | 3.66 10    | 10             | 1. 5E - 0 <del>8</del> | 9               | 1. 96 +05          | 1. 6£ +05         | :                 | !                 |                 |
| Large air      | tiero.        | Crash D             | nto cont        | 15 Jaure    | fire not        | contas.ed  |                       |         |                 |          |                 |            |                |            |                |                        |                 |                    |                   |                   |                   |                 |
| SRBG           | 2             | <b>M</b> / <b>M</b> | :               | A/M         | ;               | M/A        | :                     | M/A     | :               | N/A      | :               | A/A        | ;              | 1.96-10    | 01             | 7.96-09                | 2               | 3. 7E+05           | ;                 | ;                 | 3. 7E+04          | £               |
| SPDHC          | •<br>~        | 1. 25 - 09          | 0               | M/A         | :               | M/A        | :                     | N/A     | ;               | A/A      | ;               | 3. IE -08  | 2              | 1.9€-10    | 9              | N/A                    | ;               | 1.6€+05            | ;                 | 1.05+04           | 6.0E+03           | ZOMIN           |
| SRCGC          | <b>4</b><br>~ | 4.25-09             | 9               | R/A         | ;               | N/A        | :                     | R/A     | ;               | N/A      | ;               | A/A        | ;              | N/A        | :              | A/A                    | :               | 6.5E+04            | :                 | 1.65+04           | 4.8E+03           | 20M1N           |
| SRCHC          | ÷             | 1.25-09             | 2               | N/A         | :               | A/A        | :                     | A/A     | :               | N/A      | :               | 3.15 08    | 9              | 1.96-10    | 6              | N/A                    | :               | 1. 3E +05          | :                 | 3.25+04           | 1.86+03           | 20HIN           |
| 33 48S         | 2             | <b>R/A</b>          |                 | N/A         | :               | N/A        | ;                     | A/A     | :               | N/A      | ;               | #/#        | !              | N/A        | :              | M/A                    | :               | 4. 2E + 05         | :                 | :                 | 1. 25+04          | <b>9</b>        |
| 14.25S         | ~-            | 4.25-09             | 2               | 2.86-10     | 5               | R/A        | :                     | N/A     | ;               | 7.96-10  | <u>e</u>        | N/A        | ł              | 1.96-10    | 2              | 7.96-09                | 2               | 4. BE + 05         | :                 | :                 | 2.4E+04           | ŝ.              |
| 54.45          | 2             | A/A                 | ;               | N/A         | :               | R/A        | ł                     | 2.45-09 | 0               | A/A      | ;               | 4/H        | :              | 1,96-10    | ŝ              | N/A                    | ;               | 4.56+05            | ;                 | ;                 | 1. IE+04          | Ť               |
| DVMR2          | ~             | 1.25-39             | 01              | N/A         | ;               | N/A        | ;                     | M/A     | :               | 7.91-11) | 6               | N/A        | :              | 1.96-10    | 2              | 1.96-09                | 2               | 1.66+05            | :                 | 1.05+01           | 3.0E+03           | ZONIN           |
| SRPEC          | <b>4</b><br>~ | 4. 25 - 09          | 0               | A/A         | ;               | N/A        | ;                     | M/A     | ;               | #/#      | :               | N/A        | :              | 1.96-10    | 2              | 1.95-09                | 0               | 1.16+05            | :                 | 2.76+04           | B. 2E+03          | 20414           |
| ¥a#S           | -<br>         | 1.25-09             | 9               | N/A         | ;               | 2.46-09    | 2                     | N/A     | ;               | N/A      | :               | 3. 16 - 08 | 10             | 1.96-10    | 01             | N/A                    | :               | 2.0E+05            | :                 | 4.96+04           | 7.46+03           | ZONIN           |
| 34.645         | 5             | 4. ZE -09           | 2               | N/A         | :               | 2.46-09    | 9                     | A/A     | 1               | R/R      | :               | A/A        | :              | 1.96-10    | 51             | 7.96-09                | 9               | 1.06+05            | :                 | 2.56+04           | 1.96+03           | 20H1N           |

File: STORHOLD.WKL Page 2 Date 21-Aug-87

INTERIM STORAGE SCEMARIOS - MATEONAL/REGIONAL DISPOSAL UPTIONS (PER YEAR)

100

| SEEMARI(<br>ID | .CW 0      | ANAP<br>F           | RAMGE<br>Factor | 9d¥         | RANGE<br>FACTOR | LBAD       | RANGE<br>Factor | MAAP    | RANGE<br>Factor | PBA        | RANGE<br>Factor | PUDA       | RANGE<br>FACTOR | IEAD       | RANGE<br>ACTOR | UNDA F     | RANGE<br>ACTOR | ABENT<br>AVALLABLE | 185.<br>SPILLER | LBS.<br>DETOMATED | LBS. D<br>EN117ED | IRATION<br>11ME |
|----------------|------------|---------------------|-----------------|-------------|-----------------|------------|-----------------|---------|-----------------|------------|-----------------|------------|-----------------|------------|----------------|------------|----------------|--------------------|-----------------|-------------------|-------------------|-----------------|
|                |            |                     |                 |             |                 |            |                 |         |                 |            |                 |            |                 |            |                |            |                |                    |                 |                   |                   |                 |
| SKOGC          | ~          | 4. 25 - 09          | 2               | N/A         | :               | 2.46-09    | 01              | N/A     | :               | N/A        | ;               | A/N        | ł               | 1.96-10    | 9              | 7.96-09    | 2              | 1.26+05            | ;               | 3.0€+04           | 9. IE+03          | 20111           |
| SPQVC          | •          | A/A                 | ŧ.              | N.A         | :               | 8/8        | į               | N/A     | ;               | A/A        | :               | N/N        | ;               | 1.96-10    | 9              | 7.96-09    | 10             | 1.26+05            | :               | 3. 05 +1)4        | 2.3E+03           | ZOMLN           |
| SRRGC          | 2          | 1. 25 -09           | 01              | 8/N         | ÷               | 2.46-09    | 01              | N/A     | ;               | 7.9E-10    | 2               | N/A        | ;               | 1.96-10    | 10             | 1.96-09    | 9              | 9.0E+04            | i               | 2. ZE +0.4        | 6.7E+03           | ZONIN           |
| SRAVC          |            | 4.26-09             | 2               | A/A         | ;               | 2.45-09    | 01              | A/N     | ;               | 7.9E-10    | 9               | A/A        | ;               | 1.96-10    | 10             | 7.96-09    | 61             | B. 4E+04           | :               | 2.16+04           | 1.66+03           | 20MIM           |
| 34:585         |            | N/ P                |                 | A/A         | ;               | 4/4        | :               | A/A     | ;               | A/A        | ;               | N/A        | ;               | 1.95-10    | 01             | 7.96-09    | 0              | 1.95+05            | ļ               | 1                 | 4.76+03           | Ŧ               |
| rarge at       | Ireraft    | Crash or            | nto cont        | 310615      | ire cont        | tained     |                 |         |                 |            |                 |            |                 |            |                |            |                |                    |                 |                   |                   |                 |
| 54.86F         | -          | 12 × 2              | ;               | A'A         | :               | 4/A        | :               | N/A     | :               | N/A        | :               | A/N        | :               | 01-30-1    | Ξ              | 4. 3E-09   | Ξ              | 3.76+05            | ;               | ;                 | 3.16+04           | ЯнI             |
| SADHC          | ~          | 2.26-09             | Ξ               | N/A         | :               | N/A        | :               | N/A     | ;               | <b>M/A</b> | ;               | 1, 7E - 08 | 11              | 1.06-10    | 11             | N/A        | ł              | 1.65+05            | ;               | :                 | :                 |                 |
| SPCEC          | ,          | 2.2E-09             | Ξ               | N/A         | ;               | N/A        | ;               | A/A     | :               | N/A        | ;               | M/A        | :               | A/A        | ;              | ¥/4        | ļ              | 6.5E+04            | :               | :                 | :                 | :               |
| SPCHC          | r,         | 2. 25 -09           | Ξ               | N.A         | ;               | N/A        | ;               | N/A     | :               | N/A        | ;               | 1.75-08    | =               | 1.06-10    | Ξ              | N/P        | :              | C0+3C.1            | ;               | :                 | 4                 | ;               |
| Str 6F         | 15         | 4                   | :               | 4/W         | ;               | NIG        | :               | N/A     | ;               | ¥.4        | ;               | 4:4        | :               | 4/H        | :              | M/ F       | ;              | 1.76+05            | ;               | ;                 | 3.66+04           | ¥1.             |
| 14 des         | <b>F</b> 1 | 2. 2E - 09          | Ξ               | 1.56 -10    | =               | M/A        | :               | N/A     | ;               | N/A        | ;               | A.A        | :               | 1.06-10    | н              | 4.36-05    | Ξ              | 4.8E+05            | :               | ;                 | 7.0E+04           | EHT.            |
| Skii VE        | 5          | ų, N                | :               | N.A         | :               | M/A        | ÷               | 1.36-09 | Ξ               | N.A        | ;               | N/A        | :               | 1.06-10    | Ξ              | A/#        | ;              | 4. 5£ +05          | :               | :                 | 9.56+03           | Ŧ               |
| CAMAS          | ~          | 2.26-09             | Ξ               | N/A         | ;               | A/A        | !               | N/A     | :               | 1.36-10    | H               | N/A        | :               | 1.05-19    | Ξ              | 4.3E-09    | Ξ              | 1.66+05            | ;               | ;                 | :                 | ;               |
| 587.60         | ~          | 2.25 09             | Ξ               | A.N         |                 | N/A        | ;               | N/A     | ,               | A/A        | ;               | N/A        | :               | 1.0€-10    | Ξ              | 4.35-09    | Ξ              | 1.1£+05            | ;               | ;                 | :                 | :               |
| 3HiddS         | m          | 2.26 09             | ::              | N/A         | ;               | 1.35-09    | =               | N/A     | ÷               | N/A        | ;               | 1. 7E-09   | =               | 1.06-10    | Η              | N/A        | ;              | 2.06+15            | :               | ;                 | ;                 | ;               |
| SRPVC          | •7         | 2.26-09             | Ξ               | A, M        | ÷               | 1.36-09    | Ξ               | N/A     | :               | N/A        | ;               | N/A        | ï               | 01-30'1    | Ξ              | 4. JE - 09 | =              | 50+30-1            | ;               | :                 | ;                 | :               |
| SROGE          | m          | 2.26-09             | Ξ               | N/A         | ;               | 1.36-09    | Ξ               | A/A     | ;               | N/A        | ;               | N/A        | ;               | 1.06-10    | Ξ              | 4. 3E - 09 | Ξ              | 1.26+05            | :               | ;                 | :                 | :               |
| SROVC          |            | N, A                | ;               | 4, N        | ;               | M/A        | :               | A/A     | ;               | N/A        | ;               | N/A        | ;               | 01-30-1    | Ξ              | 4.36-09    | =              | 1.25+05            | ;               | ;                 | :                 | ;               |
| SRRGC          | ы          | 2.7E-03             | Ξ               | N/A         | ,               | 1. 36 - 09 | Ξ               | N/A     | :               | 4. 3E - 10 | Ξ               | N/A        | :               | 1, 05 - 10 | Ξ              | 4. XE - 09 | =              | 9,06+04            | ;               | ;                 | ;                 | ÷               |
| SRRVC          | ~          | 2.21-09             | 11              | A:A         | ;               | 1.36-09    | Ξ               | N/A     | ;               | 4.36-10    | Ξ               | M/A        | :               | 1.0E-10    | =              | 4. 3E -09  | Ξ              | 8.4E+04            | 1               | ;                 | ;                 | ;               |
| SRSVF          | ••         | N / P               | ;               | N/A         | ;               | A/A        | ;               | N/A     | ;               | M/A        | :               | 47.W       | ;               | 1.0E-10    | =              | f. JE-09   | Ξ              | 1.96+95            | :               | :                 | 4.01+03           | ₽¥F             |
| te lles        | ircraft    | Crash or            | ito cont        | I Der S:    | no fire         |            |                 |         |                 |            |                 |            |                 |            |                |            |                |                    |                 |                   |                   |                 |
| SKBGS          | •          | K A                 | ;               | 414         | :               | 43         | ;               | A/#     | ;               | A/A        | :               | N/A        | ÷               | 1.4E-08    | 01             | 1.26-08    | 10             | 3. 7E+05           | 2. ZE +03       | ;                 | :                 | ĨĦ              |
| SRIPHC         | -          | 1.26-08             | 9               | N/A         | :               | N/A        | :               | M/A     | ;               | N/A        | ;               | 9.96-08    | 0               | 1.4E-08    | 01             | N/A        | ;              | 1.65+05            | ;               | 1.7€+02           | 9. IE +02         | 8H1             |
| SRCBC          | •          | 1.26-08             | 0               | M/A         | :               | N/A        | :               | M/A     | ;               | N/A        | :               | A/A        | :               | N/A        | ;              | N/A        | ;              | 6. SE + U4         | ;               | 6.95+01           | 3. 26+02          | ЯН.             |
| SKCHC          | •          | BO 37.1             | 10              | A/A         | :               | N/A        | :               | A/A     | ;               | N/A        | ;               | 90~36"6    | 2               | 1.45-08    | 01             | N/A        | ;              | 1. JE +05          | :               | 1.4E+02           | 5.55+07           | θH.             |
| SRF 65         | -          | <b>A</b> . <b>H</b> | :               | <b>K</b> /A | :               | N/N        | :               | N/A     | :               | M/A        | ;               | N/A        | :               | N/A        | :              | N/A        | ;              | 4. 2E + 05         | 2.6€+03         | ;                 | :                 | 9HI             |
| CREWS          | -          | 1 26-08             | 01              | 1.75-04     | 10              | 4 / 4      | ;               | N/N     | !               | 1.16-07    | 10              | 8/8        | :               | 1.4F-08    | 01             | 1.25-08    | 01             | 4. BF+05           | 2.96+03         | ;                 | ;                 | ۲.              |

SALAN MANAGAN SALA

A REAL PRODUCED - NOVEMBER 1855 NOVEMBER 1855 NOVEMBER 1855 NOVEMBER 1855 NOVEMBER 1855 NOVEMBER 1855 NOVEMBER

2 \*\*\*. • •



File: SICMMOLD.MKI Page 3 Date 21-Aug-87

INTERIN STORAGE SCEMATIOS - MATIONAL/REGIOMML DISPOSAL DATIONS (PER YEAR)

| SCENNEIO AG<br>10 | -         | FACTOR    | 9 <b>.</b> W | RANGE<br>FACTOR | (IN)       | FACTOR | 3       | RANGE<br>FACTOR | M          | RANGE<br>Factor | WERL     | RANGE<br>FACTOR | TEAD      | RANGE<br>ACTOR | ŧ          | RANGE<br>FACTOR | AGENT<br>AVAILABLE | US.<br>SPILLED | LOS.<br>Detomated | LDS.<br>ENLITED | DURATION<br>11/K |
|-------------------|-----------|-----------|--------------|-----------------|------------|--------|---------|-----------------|------------|-----------------|----------|-----------------|-----------|----------------|------------|-----------------|--------------------|----------------|-------------------|-----------------|------------------|
|                   | 1         |           | !            |                 |            |        |         |                 |            |                 |          |                 |           |                |            |                 |                    |                |                   |                 |                  |
| STEVS             | N/N 1     | ;         | N/A          | :               | <b>4/8</b> | :      | 2.36-08 | 10              | N/N        | :               | R/A      | :               | 1.46-06   | 2              | N/N        | :               | 4. 96 + 05         | 2.76+03        | 1                 | ;               | Ĩ                |
| THINC T           | -32.1     | 2         | N/N          | ;               | N/A        | ;      | W/W     | :               | 1. IE-07   | 2               | M/N      | :               | 1.46-08   | 2              | 1.25-00    | 2               | 1.46+05            | :              | 1.7€+02           | 7.96+02         | Ĩ                |
|                   | 1.24      | 90        | N/N          | :               | <b>W/W</b> | :      | A/A     | :               | <b>M/A</b> | :               | N/A      | :               | 1, 45 -08 | 2              | 1.25-00    | 2               | 1.16+05            | ;              | 1.2.02            | 5.56+02         | Ŧ                |
| UNARS             | 1.25-     | 01<br>80  | N/A          | :               | 1.16-10    | 9      | R/A     | :               | N/N        | :               | 9, 96-08 | 9               | 1.45-08   | 0              | 8/8        | :               | 2.0E+05            | ;              | 2. I£+02          | 9.86+02         | Ĩ                |
| P 2000            | 1.1.      | 90        | A/M          | ;               | 1.86-10    | 2      | W/W     | :               | R/A        | :               | W/W      | :               | 1.46-08   | 2              | 1.25-46    | 2               | 1.06+05            | :              | 1. IE+02          | 5.0E+02         | ۹.               |
| P 299672          | 1.26-     | 90        | <b>W/W</b>   | :               | 1.06-10    | 2      | N/A     | :               | N/N        | :               | A/N      | :               | 1. 4£ -08 | 2              | 1.71-08    | 2               | 1.25+05            | ;              | 1. XE +02         | <b>4. IE+02</b> | Ĩ                |
| SROVC             | I N/A     | :         | N/A          | ;               | N/A        | :      | R/A     | ;               | N/N        | :               | N/A      | :               | 1.46-00   | 9              | 1.26-00    | 2               | 1.26+05            | :              | 1. XE+02          | 4.16+02         | Ĩ                |
| SPREC             | 1.21      | 01<br>80  | A/N          | ;               | 1.05-10    | 2      | A/A     | :               | 1.16-07    | 2               | R/A      | :               | 1.46-08   | 2              | 1.26-00    | 2               | 9.06+04            | ;              | 9.65+01           | 4.56+02         | Ĩ                |
| SRRVC             | 1.25-     | 90        | N/A          | ;               | 1.86-10    | 2      | N/A     | :               | 1.1E-07    | 2               | M/A      | :               | 1.45-08   | 2              | 1.75-00    | 9               | 8.4E+04            | :              | 9.0€+01           | 4.25+02         | Ĩ                |
| ersvs             | 1 N/A     | ;         | N/N          | :               | A/A        | :      | A/A     | :               | A/A        | :               | N/A      | ;               | 1.4E-08   | 2              | 1.25-08    | 2               | 1.96+05            | 1.26+03        | :                 | ;               | W.               |
| Small Arrer       | raft cras | h onto co | stainers;    | fire not        | contained  | -      |         |                 |            |                 |          |                 |           |                |            |                 |                    |                |                   |                 |                  |
| Steel             | 5 RVA     | ;         | <b>N/A</b>   | :               | N/N        | ;      | #/#     | :               | <b>W/A</b> | 1               | N/N      | :               | 1.55-09   | 9              | 1. 36 - 09 | 2               | 3.75+05            | :              | :                 | 3. 76+04        | Ĩ                |
| Setter            | 5 1.36-   | 60        | N/A          | :               | N/A        | :      | R/A     | ;               | W/W        | ;               | 1.16-00  | 2               | 1.36-09   | 2              | N/N        | ;               | 1.45+05            | :              | 4° 0E +04         | 4.00403         | 20MIN            |
|                   | 5 1.36-   | 01 10     | A/A          | :               | A/A        | :      | A/A     | :               | N/A        | :               | N/N      | ł               | N/A       | ;              | N/N        | :               | 6.30.01            | ;              | 1.46+04           | 4.06+03         | 20414            |
| SHORE             | 5 1.36-   | 01 F0     | K/N          | ł               | A/A        | ;      | N/A     | :               | W/W        | ;               | 1.16-08  | 2               | 1.56-99   | 2              | A/A        | ;               | 1. X +05           | :              | 3. 26+04          | 4.86+03         | 20414            |
| 5 10 HS           | 5 N/A     | ;         | W/W          | ł               | N/A        | ł      | N/N     | :               | W/W        | ł               | N/N      | ;               | N/N       | ;              | #/#        | ;               | 4. ZE -05          | :              | ;                 | 4.26+04         | Ĩ                |
|                   | 5 1.36-   | 60        | B. XE -0     | 1 10            | N/A        | ;      | A/A     | :               | 1. Z£ -00  | 2               | N/A      | ;               | 1. 56 -09 |                | 1.36-09    | 2               | 4.86+05            | ;              | ;                 | 2.4€+04         | Ĩ                |
|                   | 5 N/A     | ;         | N/N          | ;               | N/A        | ;      | 2.46-09 | 2               | N/A        | :               | A/N      | ;               | 1.55-09   | 2              | N/N        | ;               | 1.56+05            | ;              | ;                 | 1.16+04         | £                |
|                   | 5 1.36-   | 9<br>80   | M/M          | ;               | N/A        | ;      | #/#     | ;               | 1.26-00    | 9               | N/N      | ;               | 1.56-09   | 2              | 1. X -9    | 2               | 1.6€+03            | ;              | 4.05+94           | 3.06+03         | 20M1M            |
| Surger 3          | 1.36-     | 04 10     | N/A          | :               | N/N        | :      | N/N     | :               | N/N        | ;               | N/A      | :               | 1.56-09   | 2              | 1. X-09    | 9               | 1.16+05            | :              | 2.76+04           | 0.7£+03         | 20111            |
| Judges            | -x-       | 01 60     | <b>N/N</b>   | !               | 11-36-11   | 01     | #/#     | ;               | N/A        | :               | 1.16-08  | 2               | 1.55-09   | 2              | R/A        | :               | 2.06+05            | :              | 1. 16 +01         | 7.4€+03         | NINO2            |
| Steve             | 5 1.76-   | 01 60     | A/N          | ;               | 11-34-1    | 01     | N/N     | :               | N/A        | ;               | N/A      | ;               | 1. 36 -09 | 9              | 1.36-99    | 2               | 1.06+05            | :              | 2.56+01           | 1.9€+03         | NEWS 2           |
| 20005             | 3.1.8     | 0         | A/N          | :               | 11-36-11   | 01     | A/A     | :               | N/A        | ;               | R/A      | ;               | 1.56-99   | 2              | 1.36-09    | 2               | 1. 21+05           | :              | 3.06+04           | 9. IE+03        | 20M1#            |
| 1 24645           | S N/A     | ;         | A/M          | :               | A/M        | :      | N/A     | :               | <b>W/A</b> | ;               | N/A      | ;               | 1.56-09   | 2              | 1. 36 -09  | 2               | 1.25+05            | :              | 3. 8 - 2          | 2.3€+03         | 20M1K            |
| SPREC             | 5 1.35-   | 01 10     | N/N          | :               | 11-36-11   | 01     | A/A     | :               | 1.26-00    | 2               | A/A      | :               | 1.56-09   | 2              | 1.36-09    | 9               | 9.06+04            | ;              | 2.26+04           | <b>6.</b> 7E+03 | 20HLIN           |
| - JANR            | 5 1.X-    | 01 60     | N/N          | :               | 11-36-11   | 2      | N/A     | ;               | 1.26-09    | 2               | A/H      | ;               | 1.56-09   | 91             | 1. 36-09   | 01              | 8.46+04            | ;              | 2.16+04           | 1.46+03         | 20M1N            |
| SASWF             | 5 R/A     | :         | A/A          | :               | A/A        | ;      | N/N     | :               | N/A        | :               | N/A      | ;               | 1.55-09   | 9              | 1.35-09    | 9               | 1.96+05            | :              | :                 | 4.7E+03         | Ĩ                |
| Seall aircr       | raft cras | h onto co | ntainers;    | fire cont       | taned      |        |         |                 |            |                 |          |                 |           |                |            |                 |                    |                |                   |                 |                  |
| 35445             | 6 N/A     | :         | A/A          | ;               | N/A        | ;      | N/A     | ;               | A/A        | :               | N.A      | ;               | 1.06-08   | 2              | 8.56-09    | 9               | 3. 76 • 05         | ;              | :                 | 2.26+02         | <u>۳</u>         |
|                   |           |           |              |                 |            |        |         |                 |            |                 |          |                 |           |                |            |                 |                    |                |                   |                 |                  |

File: STORMOLD.MK1 Page 4 Date 21-Aug-87

INTERIM STORAGE SCEMMRIOS - MATIOMAL/REGIONAL DISPOSAL OPTIONS IPER YEAN

| SCENARIO    |             | FACTOR              | 5.W        | RANGE<br>Factor | (BAB)      | RANGE<br>FACTOR | <b>2</b> 44 | RANGE<br>Factor | <b>16</b> 4 | RANGE<br>Factor | FUDA    | RANGE<br>Factor | TEAD<br>F | RANGE<br>Actor | -          | RANGE<br>Factor / | AGENT<br>IVA ILABLE | LBS.<br>SPILLED | L DS.<br>Detonate | LOS.<br>D ENTTED  | DURAT I<br>TENE |
|-------------|-------------|---------------------|------------|-----------------|------------|-----------------|-------------|-----------------|-------------|-----------------|---------|-----------------|-----------|----------------|------------|-------------------|---------------------|-----------------|-------------------|-------------------|-----------------|
|             |             |                     |            |                 |            |                 |             |                 |             |                 |         |                 |           |                |            |                   |                     |                 |                   |                   |                 |
| HON         | 6 5.6E-0    | 01<br>•             | R/A        | ;               | A/H        | ;               | W/W         | :               | M/A         | :               | 7.16-08 | 01              | 1.05-08   | 9              | 8/4        | ;                 | 1 46+05             | ;               | :                 | :                 | ;               |
| SINCEC      | 6 5.6E-0    | 01 6                | N/N        | :               | N/A        | :               | A/A         | ;               | A/4         | ;               |         | :               |           | :              |            | ;                 | 10-31 T             | ;               | :                 | :                 |                 |
| SPERE       | 6 5.6E-0    | 6                   | N/A        | ł               | N/A        | ;               |             | ;               | A/4         | ;               | 7.15-00 | 9               | 1 05-00   | 5              |            | ;                 | TELOS               | : :             | : :               | : :               | •               |
| SREE        | A N/A       | ;                   | N/N        | :               | N/A        | ;               |             | ;               |             | :               |         | :               |           | 2              |            | ;                 |                     |                 | ł                 | 11.11             |                 |
| SREE        | 6 5.6E-0    | 9 10                | 5.56-06    | 9               | N/N        | ;               |             | :               | - 42 - 60   | 9               |         | ; ;             | 1 05-00   | 5              | 85.40      | 1                 | Co. 27 - 1          | :               | :                 | 10.10.1           |                 |
| SRAVE       | 6 N/A       | ;                   | N/A        | :               | ¥          | :               | 1.66-08     | 01              |             | 2               |         | :               | B0-30 1   | 2 9            | U. JC - V7 | <u>ا</u>          |                     | ; ;             | : 1               | 1.96401           |                 |
| SRIVC       | 6 5.6E-0    | <b>0</b> 1 <b>6</b> | 8/8        | ;               | N/A        | :               | N/A         | :               | 90-38''     | 9               |         | ;               | 1.06-08   | 2 5            | 8 56 - 09  | =                 | 1 46405             | ; ;             | ; ;               | 0.91.40           | Ĕ.              |
| 28065       | 6 5.6E-0    | 6                   | M/A        | 1               | N/A        | ;               | R/A         | ;               | N/N         | :               | N/A     | ;               | 1.05-08   | 2 2            | 8.55-09    | : 9               | 1.15+05             | :               | ;                 | ;                 | :               |
| SRPHC       | 6 5.6E-0    | 01 6                | N/A        | ;               | 1.36-10    | 9               | A/A         | :               | A/A         | :               | 7.16-08 | 10              | 1.05-08   | : =            | N/A        | :                 | 2. af +05           | ;               | :                 | ;                 | :               |
| SRFVC       | 6 3.6E-0    | 9 10                | N/A        | ł               | 1.36-10    | 10              | N/A         | ;               | N/N         | :               | W/W     | :               | 1.06-08   | : 9            | 9.56-09    | 01                | 1.06+05             | ;               | ;                 | :                 | :               |
| SROGC       | b 5.6E-0    | 6                   | W/W        | :               | I. JE - 10 | 2               | N/N         | :               | N/A         | :               | N/N     | ;               | 1.06-08   | 9              | 9. 3E - 09 | 9                 | 1. 2E+05            | ;               | ;                 | ;                 | !               |
| SPONC       | 6 N/A       | :                   | M/A        | :               | R(A        | :               | <b>8</b> /8 | ;               | R(N         | ;               | A/A     | ;               | 1.06-08   | 9              | B. 5E - 09 | 10                | 1.26+05             | :               | ;                 | ;                 | ;               |
| SAREC       | 6 5.66-0    | °<br>₽              | W/W        | ;               | 1.36-10    | 9               | M/A         | ;               | 7.0€-08     | 9               | N/A     | ;               | 1.05-08   | 9              | 8.56-09    | 2                 | 9.06+04             | ;               | ;                 | ;                 | ;               |
| SRRVC       | 6 5.6E-0    | 01                  | N/A        | ;               | I. 3E-10   | 2               | N/N         | ;               | 7. BE-08    | 2               | A/A     | :               | 1.06-08   | 2              | 9.56-09    | 9                 | B. 4E+04            | ;               | ;                 | ;                 | ;               |
| ASAP2       | 6 W/A       | ;                   | A/#        | ;               | <b>W/A</b> | ;               | N/A         | :               | N/A         | ;               | N/A     | ;               | 1.06-08   | 9              | 8.56-09    | 01                | 1.9€+05             | ;               | ;                 | 2.96+01           | Ť.              |
| .Tornado-ge | merated al- | ssiles per          | letrate ci | ontau ner s     | ; no deton | ation           |             |                 |             |                 |         |                 |           |                |            |                   |                     |                 |                   |                   |                 |
| SRINGS      | 7 8/8       | :                   | N/A        | :               | N/A        | ;               | N/N         | ;               | A/A         | ;               | N/A     | :               | 9.65-16   | 76             | 9.4E-16    | 76                | 3.76+05             | 1               | ;                 | 1.85+00           | 97              |
| SHORS       | 7 1.51-1    |                     | N/A        | ;               | N/N        | 1               | A/A         | :               | M/A         | :               | 6.ZE-13 | 8               | 5.56-15   | 86             | M/A        | ;                 | 1. 66+05            | :               | ;                 | R 16-07           | 9               |
| SRC65       | 7 9.IE-1    | *                   | N/A        | ;               | A/N        | ;               | #/#         | :               | N/A         | :               | N/A     | 1               | 3.26-15   | 5              | N/A        | ;                 | 6.56+04             | ;               | 1                 | 1.86+00           | Ĩ               |
| SECIES      | 7 9.15-1.   |                     | R/A        | :               | N/A        | :               | N/A         | ;               | #/¥         | :               | 3.86-13 | 5               | M/A       | ;              | N/A        | :                 | 1.3€+05             | ;               | :                 | B. 3£-02          | ž               |
| SBM65       | N/N /       | :                   | R/A        | ;               | H/A        | ;               | M/A         | ;               | N/A         | ł               | W/N     | ;               | A/A       | :              | N/A        | ;                 | 4.ZE+05             | ;               | :                 | 1.BE+00           | ž               |
| Sitting in  | 7 2.4E-1.   | *                   | 7.8E-14    | 5               | N/A        | ;               | N/A         | :               | 2.46-12     | 2               | R/A     | ;               | 5.06-16   | 2              | 5.06-16    | 5                 | <b>6.86</b> +05     | 1               | :                 | 8.3E-02           | ۹.              |
| 5455        | 7 8/4       | :                   | N/A        | :               | M/A        | :               | 2.4E-12     | 2               | #/#         | :               | N/A     | 1               | 5.06-16   | 76             | N/A        | :                 | 1.56+05             | ;               | ;                 | 1.96-07           | ž               |
| SAMAS       | 1-36-1      | 8                   | N/A        | :               | N/A        | 1               | N/A         | :               | 11-37.11    | 2               | N/A     | ;               | 4.96-15   | 5              | 1.96-15    | 56                | 1.6€+05             | ;               | ;                 | 1.06-03           | ž               |
| SHPES       | 1-38.1      | 5                   | A/A        | ;               | #/#        | :               | A/A         | :               | N/A         | :               | N/A     | :               | 2.76-15   | 2              | 2.76-15    | 14                | 1. IE+05            | ;               | :                 | 1.85+00           | ۲.              |
| SHORES      | 1-36-1      | 5                   | N/A        | :               | 7.06-12    | 2               | N/A         | :               | N/A         | :               | 3.ZE-13 | 5               | 2.7E-15   | 5              | N/A        | :                 | 2.0€+05             | ;               | ;                 | <b>B.</b> 3E - 02 | ۹.              |
| SKPVS       | 1 36 1      | 5                   | R/A        | :               | 7.06-12    | 2               | N/A         | :               | N/A         | :               | N/A     | 1               | 2.76-15   | 8              | 2.76-15    | 5                 | 1.06+05             | ;               | ;                 | 1.05-03           | ž               |
| Skoes       | 1.96.1      | 5                   | N/N        | :               | 7.86-12    | 6               | N/A         | ;               | N/A         | ;               | #/#     | ;               | 2.7E-15   | 2              | 2.76-15    | 5                 | 1.25+05             | ;               | ;                 | 1.8€+00           | Ű.              |
| SMBAS       | A/N 2       | :                   | N/A        | :               | ¥/4        | :               | N/N         | ł               | N/A         | :               | M/A     | :               | 2.7E-15   | 5              | 2.7E-15    | 56                | 1.26+05             | ;               | :                 | t.0F-03           | Ĩ               |

SALATING SECT PRODUCT NUMBER SECTOR







1.1.1

L'ALS.

INTERIM STOPAGE SEEMARIOS - MATIONAL/REGIONAL DISPOSAL OFTIONS (PER YEAR)

| DURATIC    | Ĕ             | 2HB              | Э¥С      | ЭЩ.      |             | 997<br>2 | ۲.        | EHZ      | ۶.      | ۴.      | ž       | ۶,         | Ħ         | 248      | ۶.      | Э́Е,     |           | Ĩ       | 20M1       | 20ME           | ZONIA      | μ.       | Ē         | ۴.         | ZONE     | 20M14      | 20M1     | 20111      | 11HOZ    | 200110     |
|------------|---------------|------------------|----------|----------|-------------|----------|-----------|----------|---------|---------|---------|------------|-----------|----------|---------|----------|-----------|---------|------------|----------------|------------|----------|-----------|------------|----------|------------|----------|------------|----------|------------|
| L95.       |               | 1.86+00          | 1.0E-03  | 1.0E-03  |             | ;        | ;         | ;        | :       | 1       | ;       | ;          | ;         | :        | ;       | 1        |           | 2.46+02 | 10+32-1    | 3.56+01        | 3.56+01    | 3.0€+02  | 1.7E+02   | 8.0E+01    | 2. IE+01 | 5.86+01    | 5.36+01  | 1.46+01    | 6.5E+01  | 1.46+01    |
| LBS.       | 94.1 UMA 16.0 | ;                | ;        | :        |             | 6.0€+00  | 1.65+00   | 3.2£+00  | 0.05+00 | 0.0E+00 | 0.0E+00 | 3.26+01    | 6. SE +00 | 1.2E+01  | 6.0E+00 | 1.5€+01  |           | ;       | 2.9€+02    | 1.26+02        | 2. 3£ +02  | ;        | ;         | :          | 2.8E+02  | 2.06+02    | 3.5€+02  | 1.86+02    | 2.2£+02  | 2.26+02    |
| LBS.       |               | ;                | :        | ;        |             | 3.06+01  | 8.06+00   | 1.45+01  | 1.15+03 | 3.26+01 | 5.86+01 | 3.0€+01    | 7.36+01   | 7. 36+01 | 6.2E+02 | 5.0€+02  |           | ;       | ;          | ;              | :          | ;        | ;         | ;          | :        | ł          | ;        | :          | ;        | ;          |
| AGENT      |               | 9.0E+04          | B. 4E+04 | 1.9€+05  |             | 1.65+05  | 6. SE +01 | 1. JE+05 | 1.6€+05 | 1.1E+05 | 2.0E+05 | 1.0E+05    | 1.26+05   | 1.26+05  | 9.05+04 | 8. 4E+04 |           | 3.76+05 | 1.6E+05    | 6.5E+04        | 1.3€+05    | 4. ZE+05 | 4. BE +05 | 4, 56+05   | 1.66+05  | 1.1E+05    | 2.06+05  | 1.06+05    | 1.26+05  | 1. 25+05   |
| RANGE      |               | 46               | 5        | 5        |             | ł        | ;         | ;        | \$      | 66      | :       | 66         | 66        | 8        | 5       | \$6      |           | 26      | ;          | ;              | :          | ;        | 36        | ł          | 26       | 26         | ;        | 56         | 26       | <b>3</b> 6 |
| MDA        |               | 6.2E-15          | 6.25-15  | 1.76-15  |             | N/A      | N/N       | N/A      | 1.86-15 | 1.06-15 | N/A     | 1.06-15    | 1.06-15   | 1.06-15  | 6.6E-15 | 6.6E-15  |           | 2.06-10 | N/A        | N/N            | N/N        | A/A      | 1.95-10   | A/A        | 2.06-10  | 2.1E-10    | #/#      | 2.1E-10    | 2.16-10  | 2.16-10    |
| 39MP2      | 5             | 76               | 5        | \$       |             | 8        | 8         | ;        | 8       | 5       | \$      | 8          | £         | 8        | 3       | 2        |           | 97      | 36         | ;              | 26         | ;        | 56        | ł          | 26       | 26         | 36       | <b>3</b> 8 | 26       | <b>5</b>   |
| TEAD       |               | 6.25-15          | 6.25-15  | 1, 75-15 |             | 5.96-15  | 3.56-15   | N/A      | 1.86-15 | 1.06-15 | 51-30-1 | 1.06-15    | 1.06-15   | 1.06-15  | 6.4E-15 | 6.6E-15  |           | 2.06-10 | 1.86-10    | N/A            | 2. ZE - LO | A/A      | 1.96-10   | N/N        | 2.06-10  | 2. IE-10   | 2. IE-10 | 2. IE-10   | 2. IE-10 | 2. IE-10   |
| RANGE      |               | :                | ;        | :        |             | 8        | :         | <b>:</b> | ;       | ţ       | 5       | ł          | ;         | ;        | :       | ;        |           | :       | 26         | ;              | 26         | ;        | ł         | ł          | :        | 1          | 26       | 1          | ;        | :          |
| MUDA       |               | N/A              | N/A      | A/A      |             | 2.36-13  | N/N       | 1.4E-13  | A/A     | N/A     | 3.5E-13 | N/N        | N/N       | N/N      | A/A     | N/A      |           | N/N     | 1.86-10    | N/N            | 2.26-10    | N/A      | N/N       | N/A        | W/#      | A/A        | 2.16-10  | N/A        | K/A      | N/A        |
| RANGE      |               | 5                | 76       | :        |             | :        | :         | ł        | 8       | :       | ;       | ;          | :         | ;        | 94      | 76       |           | ;       | :          | :              | :          | ;        | 26        | ;          | 24       | ;          | ;        | ;          | 1        | ;          |
| P84        |               | I. <i>7</i> E-11 | 1.76-11  | M/R      | 12          | N/A      | . V/N     | N/N      | 4.96-12 | W/W     | N/N     | N/N        | A/A       | N/A      | 1.8%-11 | 1.86-11  |           | N/N     | A/N        | N/N            | R/A        | N/N      | 1.96-10   | N/A        | 2.0E-10  | N/A        | N/A      | N/A        | N/A      | N/A        |
| RANGE      | FACTOR        | :                | ;        | ;        | for rockel  | :        | ;         | :        | ;       | ;       | :       | :          | ;         | ;        | :       | :        |           | ł       | :          | :              | :          | ł        | :         | <b>9</b> 2 | ;        | :          | ;        | :          | :        | !          |
| a a        |               | N/A              | A/A      | N/A      | ignition    | N/N      | N/A       | N/N      | N/A     | N/A     | N/A     | N/N        | N/A       | N/A      | N/A     | N/A      |           | N/N     | N/N        | 8/8            | N/N        | N/A      | N/A       | 1.96-10    | N/A      | N/A        | N/A      | N/A        | N/A      | #/#        |
| RANGE      |               | 5                | 8        | :        | t lor motor | :        | ;         | :        | ;       | :       | 4       |            | 5         | :        | 5       | 5        |           | ;       | :          | ;              | :          | :        | :         | :          | ;        | 1          | 54       | 26         | 26       | :          |
| LBAD       |               | 11-76-11         | 11-3/-11 | N/N      | detonation  | R/A      | ť.        | A/A      | N/A     | A/A     | 2.96-12 | 2.46-12    | 2.96-12   | N/N      | 11-38.1 | 1.06-11  |           | N/N     | R/A        | N/A            | N/N        | N/A      | N/N       | A/A        | 4/H      | N/N        | 2. iE-lo | 2.16-10    | 2.16-10  | N/A        |
| RANGE      | FACTOR        | ;                | ;        | :        | tainers:    | •        | :         | :        | :       | :       | :       | ;          | ;         | ;        | ;       | ;        |           | :       | ;          | ;              | ł          | ;        | 26        | ;          | 1        | ;          | 1        | :          | 1        | ;          |
| APS        |               | A/A              | N/N      | 4/1      | rate con    | N/A      | W/W       | N/A      | N/A     | A/A     | N/N     | <b>W/W</b> | N/A       | #/#      | N/A     | N/A      | 2         | A/H     | N/A        | N/A            | N/A        | N/A      | 1.96-10   | N/A        | N/A      | N/N        | A/A      | N/N        | N/A      | A/A        |
| RANGE      |               | 36               | 5        | ;        | le penet    | 8        | 66        | 66       | 66      | 56      | ţ,      | F          | 66        | ;        | 8       | \$       | olding a  | ;       | <b>3</b> 6 | <b>3</b> 6     | 26         | ;        | 92        | ;          | 26       | <b>2</b> 6 | 2P       | 36         | 36       | ;          |
| Ŷ          |               | . <i>1</i> 1-37. | .7E-11   | N/N      | M 01551     | .66-12   | .4E-12    | .4E-12   | .96-12  | .96-12  | .96-12  | .96-12     | .91-39.   | N/N      | :I-38.  | .BE-11   | es the h  | N/A     | .86-10     | 0I- <b>3</b> Z | .25-10     | N.       | .96-10    | N/N        | .01-30   | . 16-10    | . 15-10  | .16-10     | . 16-10  | N/N        |
| ġ          | 1             | 1                | -        | ~        | mer at      | 9<br>9   |           | 8        | -       | 8       | 8       | 8          | 8         |          | <br>    | -        | strik     |         | •          | 6              | 9          | •        | -         | •          | 9        | 9          | 6        | 9          | сч<br>8- | •          |
| SCENARIO I | 9             | SRR65            | SKRVS    | SRSVS    | Tor nado-ge | SRDHC    | SREEC     | SPICHE   | SRIVC   | SRPEC   | SRPHC   | SRPVC      | SROGC     | SROVC    | SRRGC   | SRRVC    | Heteorite | SREEF   | SROHC      | SACGC          | SPCHC      | Sex of   | SRKIE     | SRKVF      | SREVC    | SRPGC      | SRPHC    | SRPVC      | SROGE    | SROVC      |

- - -

\_ \_ \_ \_ \_ \_ \_

STATES ...

EX.

File: STORHOLD.NKI Page & Date 21-Aug-87

INTERIM STORAGE SCERARIDS - NATIONAL/REGIONAL DISPOSAL OPTIONS (PER VEAR)

1

| 5                  | ; | Ξ          | Ē          | ~          |
|--------------------|---|------------|------------|------------|
| DURAT              | 1 | 1 20K      | 1 20M      | Ξ          |
| LBS.<br>LBS.       |   | 4.86+0     | 1.1E+0     | 3.46+0     |
| LDS.<br>Detomated  |   | 1.66+02    | 1.5€+02    | :          |
| LBS.<br>LBS.       |   | :          | ;          | ;          |
| AGENT<br>Available |   | 9.06+04    | B. 4E+04   | 1. 9E+05   |
| RANGE              |   | 26         | 36         | 26         |
| MDA                |   | 1.96-10    | 1.96-10    | 3.26-10    |
| RANGE<br>actor     |   | <b>3</b> 8 | 26         | <b>3</b> 6 |
| TEAD               |   | 1.96-10    | 1.96-10    | 3.25-10    |
| RANGE<br>Factor    |   | ł          | ;          | :          |
| PUDA               |   | A/A        | N/A        | N/A        |
| RANGE<br>Factor    |   | 36         | 2 <b>6</b> | :          |
| PBA                |   | 1.96-10    | 1.96-10    | W/A        |
| RANGE<br>Factor    |   | :          | ;          | :          |
| <b>AAAM</b>        |   | N/A        | A/A        | N/A        |
| RANGE<br>Factor    | 1 | 2          | 26         | :          |
| L MO               |   | 1.96-10    | 1.96-10    | N/A        |
| RANGE<br>Factor    |   | 1          | ;          | ;          |
| 94PS               |   | R/A        | N/A        | M/A        |
| KANGE<br>KCTOR     |   | <b>9</b> 2 | 26         | :          |
| - II<br>OVMAN      |   | 91-36      | . 91-10    | N/A        |
| ÷                  | : | •          | •          | ۰          |
| SCENARIO<br>LD     |   | SPAREC     | SRRVC      | SRSVC      |
|                    |   |            |            |            |

ž

11.11.1.255.555.1

S.C. 333.54

I-28





3.7

File: STORHOLD.WK1 Fage 1 Date 21-Aug-87

Interim Storage Barge Only levents/yr)

| SCENARIO NO.    |         |      | APG              | RANGE     | AGENT   | LBS.     | LBS.      | LBS.    | DURATION |
|-----------------|---------|------|------------------|-----------|---------|----------|-----------|---------|----------|
|                 |         |      |                  | FACTOR AV | AILABLE | SPILLED  | DETONATED | ENTTED  |          |
|                 |         |      |                  |           |         |          |           |         |          |
| Large aircraft  | crash   | onto | holding area;    | no fire   |         |          |           |         |          |
| I SHXHS         |         |      | 2.06-11          | 10        | 238000  | 202300.0 | ł         | ł       | IHR      |
| Large aircraft  | crash   | onto | holding area;    | fire not  | contain | Ed.      |           |         |          |
| SWKHF 2         |         |      | 1.1E-11          | 10        | 23B000  | 1        | ł         | 11900.0 | 1HR      |
| Large aircraft  | crash   | onto | holding area;    | fire con  | tained  |          |           |         |          |
| SWKHF 3         |         |      | 5.8E-12          | 11        | 238000  | 1        | ł         | 10115.0 | 1H<br>H  |
| Small aircraft  | crash   | onto | holding area;    | no fire   |         |          |           |         |          |
| SUKHS 4         |         |      | 3.0E-07          | 01        | 238000  | 2890.0   | ł         | ł       | 1HR      |
| Small aircraft  | crash   | onto | holding area;    | fire not  | contain | Eđ       |           |         |          |
| SWKHF 5         |         |      | 3.2E-08          | 11        | 238000  | ł        | ł         | 11900.0 | IHR      |
| Small aircraft  | crash   | onto | holding area;    | fire con  | tained  |          |           |         |          |
| SWKHF 6         |         |      | 2. IE-07         | 10        | 238000  | ł        | 1         | 144.5   | H<br>H   |
| Tornado-genera  | ted mis | sile | penetrates va    | ult       |         |          |           |         |          |
| SWKHC 7         |         |      | 2.5E-14          | 46 .      | 238000  | ł        | !         | 0.1     | ZHR      |
| Meteorite stril | kes vau | t    |                  |           |         |          |           |         |          |
| SMKHF 9         |         |      | 1.06-10          | 26        | 238000  | :        | 1         | 85.0    | IHR      |
| Large aircraft  | crash   | onto | lighter; no f    | ire       |         |          |           |         |          |
| SWKHS 10        |         |      | 7. <i>T</i> E-12 | 10        | 95200   | 1        | ł         | 1040.0  | 24HR     |
| Large aircraft  | crash   | onto | lighter; fire    | not cont. | ained   |          |           |         |          |
| SIKH 11         |         |      | 4. IE-12         | 10        | 95200   | ;        | ;         | 4962.3  | 1HR      |
| Large aircraft  | crash   | onto | lighter; fire    | containe  | -       |          |           |         |          |
| SMKHF 12        |         |      | 2.2E-12          | 11        | 95200   | :        | ł         | 202.3   | IHR      |
| Small aircraft  | crash   | onto | lighter; no f    | ire       |         |          |           |         |          |
| SWKHS 13        |         |      | 1.2E-07          | 10        | 95200   | 1        | ł         | 3.1     | 1HI      |
| Small aircraft  | crash   | onta | lighter; fire    | not cont  | ained   |          |           |         |          |
| SWKHF 14        |         |      | 1.2E-08          | 11        | 95200   | ;        | ;         | 4760.0  | 3HI      |
| Small aircraft  | crash   | onto | lighter; fire    | containe  |         |          |           |         |          |
| SWKHF 15        |         |      | 8.2E-08          | 10        | 95200   | ł        | ;         | 680.0   | 3HI      |
| Large aircraft  | crash   | onto | ship; no fire    |           |         |          |           |         |          |
| SWKHS 16        |         |      | 3. 3E-11         | 01        | 3808000 | ;        | ;         | 350.0   | 8H8      |
| Large aircraft  | crash   | onto | ship; fire no    | t contain | pa      |          |           |         |          |

100000000 COLORIS COLORIS Sectory's SERVICE STRUCTURE 

0000

File: STORHOLD.WK1 Fage 2 Date 21-Aug-87

Interia Sinteria Storage Barge Unly levents/yr)

| SCENNE          | RIO NO.        |        |         | •<br>•    | APG                | FACTOR         | ABENT<br>AVATLABLE | LBS.   | LBS.<br>DETONATED | LBS.<br>ENITTED | DURATION<br>TIME |
|-----------------|----------------|--------|---------|-----------|--------------------|----------------|--------------------|--------|-------------------|-----------------|------------------|
| SIKIF           | 11             |        |         |           | 11-38.             | 10             | 3808000            | 1      | ł                 | 60520.0         | IHR              |
| Large<br>Sukh   | aircraft<br>18 | crash  | onto sl | hìp:<br>9 | fire co<br>.5E-12  | ntained<br>11  | 3808000            | a<br>I | *                 | 3400.0          | AR               |
| Small<br>SMKHS  | aircraft<br>19 | crash  | onto sl | hip;      | no fire<br>.8E-07  | 10             | 3808000            | :      | 1                 | 19.3            | 2HR              |
| Small<br>SWKHF  | aircraft<br>20 | cr ash | onto s  | hip:      | fire no<br>i.1E-08 | it conta<br>11 | i ned<br>3808000   | ;      |                   | 57375.0         | 1HR              |
| Saal 1<br>SWKHF | aircraft<br>21 | crash  | onto s  | hips      | fire`ca<br>5.4E-07 | ntained<br>10  | 3808000            | ;      | ł                 | 255.0           | 1HR              |



Personal states

PERSON DESCRIPTION DESERVED





1.10<sup>1</sup>.11<sup>1</sup>.11<sup>1</sup>.11

File: STORHOLD.MKi Page I Date 21-Aug-87

INTERIN SIGNAGE AIR OFIION (EVENTS/YR)

| Scenario | ž i    | W          | RANGE<br>FACTOR | AP6               | RANGE<br>FACTOR | LBAD      | RANGE<br>Factor | <b>A</b> M | RANGE<br>FACTOR | Vđ   | FACTOR<br>FACTOR | MUN | RANGE<br>Factor | TEAD       | RANGE<br>Factor | MDA | RANGE<br>Factor | AGENT<br>AVAILABLE | 1937.<br>SPILLED | LOS.<br>Detomated | LBS. 1<br>Entted | DURATION<br>TIME |
|----------|--------|------------|-----------------|-------------------|-----------------|-----------|-----------------|------------|-----------------|------|------------------|-----|-----------------|------------|-----------------|-----|-----------------|--------------------|------------------|-------------------|------------------|------------------|
|          |        |            |                 |                   |                 |           |                 |            |                 |      |                  |     |                 |            |                 |     |                 |                    |                  |                   |                  |                  |
| Large as | rcraft | crash c    | mta cont.       | a istante         | in fire         |           |                 |            |                 |      |                  |     |                 |            |                 |     |                 |                    |                  |                   |                  |                  |
| SAMES    |        | M/A        | ;               | 6.0E-10           | 2               | N/N       | :               | N/A        | ;               | N/A  | ;                | N/A | :               | 3. 36 - 09 | 9               | N/A | :               | 21000              | 43350.0          | 1                 | ;                | ۹.               |
| SIPHC    | -      | <b>K/A</b> | :               | N/N               | ;               | 3.26-09   | 9               | N/A        | ;               | 8/8  | :                | R/A | :               | 3. 3£-09   | 2               | W/W | 1               | 21060              | 14742.0          | 3159.0            | :                | Щ.               |
| SAPVC    | -      | N/N        | ;               | N/N               | ;               | 3.26-09   | 2               | N/A        | ;               | N/N  | ;                | N/N | :               | 3. 36 -09  | 9               | 8/8 | ;               | 10800              | 7560.0           | 1620.0            | ;                | Ħ                |
| SARGC    | -      | N/N        | ;               | N/A               | ;               | 5. 26-09  | 2               | N/A        | :               | N/N  | :                | A/A | ;               | 3. 36-09   | 9               | A/N | ;               | 13050              | 9135.0           | 1957.5            | ;                | Ĩ                |
| SARGC    |        | N/A        | ;               | #/#               | ;               | 3.26-09   | 9               | N/N        | :               | A/H  | ;                | N/N | ;               | 3.35-09    | 01              | A/A | ;               | 9630               | 6741.0           | 1444.5            | ;                | Ĩ                |
| SARVE    |        | N/A        | ;               | K/A               | :               | 3.2E-09   | 2               | N/N        | ;               | R/A  | :                | 878 | ;               | 3. 36 -09  | 10              | A/M | ;               | 9006               | 6300.0           | 1350.0            | ;                | ž                |
| Large an | reraft | crash o    | nto cont.       | inters; f         | ire not         | contained |                 |            |                 |      |                  |     |                 |            |                 |     |                 |                    |                  |                   |                  |                  |
| SIRCHE   | 2      | N/A        | ;               | 3. 26-10          | 2               | N/A       | ;               | N/N        | :               | M/A  | :                | N/A | 1               | 1.86-09    | 0               | N/A | :               | 51000              | ł                | :                 | 2550.0           | Ŧ                |
| SAPHC    | ۰.     | K/A        | ;               | N/A               | :               | 1.76-09   | 2               | R/A        | ł               | N. A | :                | N/N | ;               | 1.86-09    | 9               | N/A | ;               | 21060              | ;                | 5265.0            | 789.8            | ZOMIM            |
| SAP VC   | 7      | N/A        | :               | N/A               | 1               | 1.76-09   | ŝ.              | N/N        | ;               | #/#  | ;                | N/A | ;               | 1.85-19    | 9               | N/A | :               | 10800              | :                | 2700.0            | 202.5            | ZONIN            |
| SADGC    | 2      | N/A        | ;               | N/A               | ;               | 1.76-09   | 2               | N/A        | ;               | N/N  | :                | A/M | ;               | 1.85-09    | e               | N/A | ;               | 13050              | ;                | 1262.5            | 9.819            | 20HIN            |
| SARGC    | ډ،     | R/A        | :               | N/A               | ;               | 1.75-09   | 2               | A/M        | ;               | N/A  | 1                | M.A | :               | 1, 86 - 09 | 2               | N/A | :               | 9630               | ;                | 2407.5            | 722.3            | ZONIN            |
| SARVC    | 2      | M/A        | :               | N/A               | :               | 1.76-09   | 9               | N/A        | ;               | #/#  | ;                | N/A | ;               | 1.85-09    | 01              | N/A | :               | 0006               | :                | 2250.0            | 168.8            | 20MIN            |
| Large au | rcraft | Crash o    | nto conti       | ILINEY 5: F       | ire cont        | an ned    |                 |            |                 |      |                  |     |                 |            |                 |     |                 |                    |                  |                   |                  |                  |
| 544      | ~      | N/A        | ;               | 1.75-10           | Ξ               | N/A       | ;               | R/A        | ;               | N/A  | !                | K/A | ;               | 9.46-10    | =               | N/A | :               | 51000              | ;                | ;                 | 2167.5           | Ť.               |
| SAPHC    | *7     | N/A        | ;               | N/A               | :               | 9.16-10   | =               | <b>K/A</b> | ;               | N/N  | ;                | N/A | 1               | 9.46-10    | н               | #/# | :               | 21060              | :                | :                 | ;                | ;                |
| SAPVC    | ~      | N/A        | ;               | N/A               | ;               | 9.16-10   | Ξ               | A/A        | 1               | N/A  | :                | R/A | ;               | 91-31-6    | Ξ               | N/A | ;               | 1080               | ;                | :                 | :                | :                |
| SADGC    | • •    | N/A        | ;               | A/A               | :               | 9.16-10   | =               | M/A        | ;               | N/A  | :                | N/A | ;               | 9.46-10    | 11              | 4/M | :               | 13050              | ;                | :                 | ;                | ;                |
| SARGC    | ~      | A/N        | :               | M/A               | ;               | 9.16-10   | Ξ               | N/A        | ;               | N/A  | :                | N/A | ;               | 9.4E-10    | Ξ               | N/A | ;               | 9630               | :                | :                 | :                | ;                |
| SARVC    | •7     | A/A        | ;               | M/A               | 1               | 9.16-10   | =               | A:A        | ;               | 818  | :                | R/H | ;               | 9.45-10    | Ξ               | N/A | ;               | 9006               | ;                | ;                 | :                | ;                |
| Small an | rcraft | crash o    | ato conta       | n is auti         | o fire          |           |                 |            |                 |      |                  |     |                 |            |                 |     |                 |                    |                  |                   |                  |                  |
| SAKHS    | •      | N/A        | ;               | B. 3E - 07        | 2               | N/A       | :               | A/A        | :               | N/A  | :                | N/A | ;               | 1.51-09    | 10              | N/A | :               | 51000              | 2890,0           | ;                 | ;                | 3HI              |
| SAPHC    | -      | N/A        | :               | A/A               | :               | 1.96-11   | 10              | N'N        | ;               | N/A  | :                | N/A | ;               | 1.56-09    | 9               | N/A | 1               | 21060              | 982.8            | 210.6             | :                | ŧ                |
| SAPVC    | •      | N/A        |                 | A/A               | ;               | 11-36.1   | 0               | N/A        | ;               | N/A  | ;                | N/A | ;               | 1.56-09    | 10              | N/A | :               | 10800              | 504.0            | 108.0             | :                | 1.<br>1          |
| SADGC    | -      | 4.4        | :               | A/H               | :               | 11-36-11  | 01              | M/A        | :               | N/A  | :                | A/M | :               | 1.56-09    | 9               | N/A | ;               | 13050              | 609.0            | 130.5             | i                | £                |
| SARGC    | 4      | R/A        | :               | 4/4               | 1               | 11-34.1   | 0               | N/A        | ł               | N/A  | :                | N/A | :               | 1.56-09    | 01              | A/A | ;               | 9630               | 449.4            | - 96              | ŗ                | Ē                |
| SARVC    | -      | R/A        | ;               | A/A               | :               | 1.96-11   | 0               | N/A        | ;               | N/A  | :                | A.A | :               | 1.56-09    | <u>:</u>        | N/A | :               | 0006               | 4.20.0           | 9.06              | ;                | 1HR              |
| Small an | rcraft | crash o    | nto conta       | Iners; f          | ire not         | contained |                 |            |                 |      |                  |     |                 |            |                 |     |                 |                    |                  |                   |                  |                  |
| Sign in  | ŝ      | A.A        | :               | <b>8. B£</b> - 08 | Ξ               | A/A       | !               | N/A        | :               | N/A  | :                | N/A | :               | 1,66-10    | Ξ               | 4/W | ;               | 21000              | :                | ;                 | 2550.0           | 3HL              |
|          |        |            |                 |                   |                 |           |                 |            |                 |      |                  |     |                 |            |                 |     |                 |                    |                  |                   |                  |                  |

| Scentri        | 2       | <b>CAMA</b> | RANGE      | AP6         | ANNEE     | LIND            | RANGE      | <b>AW</b>   | RANGE       | <b>984</b> | RANGE  | Publ       | RANGE  | TEAD     |       | MGHO       | RANGE  | NGENT     | <u>8</u> | LB5.        | Э            | NURATION   |
|----------------|---------|-------------|------------|-------------|-----------|-----------------|------------|-------------|-------------|------------|--------|------------|--------|----------|-------|------------|--------|-----------|----------|-------------|--------------|------------|
|                |         |             | FACTOR     |             | FACTOR    |                 | FACTOR     |             | FACTOR      |            | FACTOR |            | FACTOR |          | ACTOR |            | FACTOR | AVAILADLE | SPILLED  | DETONATED   |              | TINE       |
|                | *       | A/A         | :          | <b>1</b> /1 | :         | 2.06-12         | 3          | <b>8</b> /8 | :           | 8/8        | ;      | 8/8        | ;      | 1.66-10  | Ξ     | N/N        | ;      | 21060     | ;        | 5265.0      | <b>189.8</b> | ZONIN      |
| SAPVC          | 5       | VI          | :          | A/H         | :         | 2.06-12         | =          | A / N       | :           | W/W        | ł      | V/N        | ;      | 1.66-10  | Ξ     | N/A        | ;      | 10800     | ;        | 2700.0      | 202.5        | ZONIN      |
| SHORE          | -       | N/N         | :          | R/A         | :         | 2.06-12         | Ξ          | N/N         | :           | W/W        | 1      | N/A        | :      | 1.66-10  | Ξ     | R/A        | :      | 13050     | ;        | 3262.5      | 978.8        | 20HIN      |
| Samer          | 'n      | N/N         | :          | A/A         | !         | 2.06-12         | Ξ          | N/N         | ;           | A/A        | :      | N/A        | ł      | 1.66-10  | Ξ     | A/A        | :      | 9630      | ;        | 2407.5      | 722.3        | 20M1N      |
| SARVC          | -       | N/N         | ;          | N/A         | :         | 2.06-12         | Ξ          | A/N         | :           | N/A        | :      | N/A        | :      | 1.66-10  | Ξ     | M/A        | ;      | 9006      | ţ        | 2250.0      | 168.B        | 20M1N      |
| Seall a        | Ircrafi | t crash     | onto cont  | STARTS!     | fire cont | anned           |            |             |             |            |        |            |        |          |       |            |        |           |          |             |              |            |
| <b>SACT</b>    | -0      | N/A         | :          | 5. 96-07    | 2         | N/A             | :          | N/A         | :           | N/A        | ;      | A/A        | :      | 1.16-09  | 2     | R/A        | ;      | 21000     | ;        | ţ           | 14.5         | 248        |
| SAPRC          | -9      | N/A         | ł          | A/A         | ;         | 11-34-11        | 9          | N/A         | :           | N/A        | :      | 8/8        | :      | 1.16-09  | 2     | N/A        | ;      | 21060     | ;        | ;           | ;            | ;          |
| SAPVC          | •       | N/A         | ;          | N/A         | :         | 11-91-11        | 01         | A/A         | :           | N/A        | :      | N/A        | :      | 1.16-09  | 2     | N/A        | ;      | 10800     | ;        | :           | ;            | :          |
| SARGEC         | -0      | N/N         | ;          | N/A         | ;         | 1.46-11         | 10         | N/A         | :           | N/A        | ;      | N/A        | ;      | 1.16-09  | 9     | N/A        | ;      | 13050     | :        | !           | ;            | ł          |
| SARGC          | -0      | A/N         | :          | N/A         | ;         | 1.46-11         | 10         | N/A         | :           | N/A        | :      | R/A        | ;      | 1.16-09  | 2     | N/A        | ;      | 9630      | ;        | :           | ;            | ;          |
| SARVC          | •       | A/N         | ;          | N/A         | :         | 1.4E-11         | 01         | N/A         | ;           | M/A        | :      | <b>M/A</b> | :      | 1.16-09  | 9     | N/A        | :      | 9000      | ;        | ;           | ;            | ;          |
| <b>Tornado</b> | -dener  | ated and    | tsiles pen | etrate ci   | ontainers | it no deton     | ation      |             |             |            |        |            |        |          |       |            |        |           |          |             |              |            |
| Saures         |         | A/N         |            | 8.4E-14     | 5         | N/A             | :          | N/A         | ;           | N/A        | :      | N/A        | :      | 5.06-16  | 96    | N/A        | ;      | 51000     | !        | :           | 0.1          | 244        |
| SHANS          | ~       | N/N         | :          | N/A         | ;         | 8.4E-13         | 16         | N/A         | :           | N/A        | ;      | N/A        | ;      | 2. TE-15 | 5     | N/A        | ;      | 21060     | ;        | :           | 0.1          | ۶,         |
| SALAS          | 1       | N/A         | :          | N/A         | :         | 8.4E-13         | 5          | N/N         | :           | A/A        | ;      | N/A        | 1      | 2.76-15  | 94    | M/N        | :      | 10800     | ;        | :           | 0.0          | 2HR        |
| SADES          | ~       | N/A         | :          | N/A         | ;         | 8.4E-13         | 5          | N/A         | :           | N/A        | ;      | N/A        | :      | 2.7E-15  | 96    | A/A        | !      | 13050     | :        | :           | 1:8          | 9 <b>9</b> |
| SARES          | 1       | N/A         | :          | N/A         | :         | 1.86-12         | 2          | N/A         | :           | N/N        | :      | R/A        | :      | 6.ZE-15  | 94    | N/A        | ;      | 9630      | ;        | :           | 1.8          | 患          |
| SVIEWS         | ~       | R/A         | :          | R/A         | :         | 1.86-12         | 94         | N/A         | :           | N/A        | :      | N/A        | ;      | 6.25-15  | 56    | A/A        | ł      | 9006      | ;        | :           | 0.0          | 248        |
| Tornade        | -dener  | ated my     | ssile pene | itrate co   | ntainers; | detonatio       | in ter est | or igniti   | on for rock | ets)       |        |            |        |          |       |            |        |           |          |             |              |            |
| Surric         | •       | A/A         | :          | N/A         | 1         | 3. IE-13        | 8          | N/A         | 1           | A/A        | :      | N/A        | ;      | 1.06-15  | 8     | <b>N/N</b> | :      | 21060     | 5.5      | 11.7        | :            | Ж,         |
| <b>SAPVC</b>   | -       | N/A         | :          | N/A         | :         | 3. 16-13        | 8          | N/A         | :           | A/M        | :      | N/A        | :      | 1.06-15  | 8     | N/R        | :      | 10800     | 30.0     | <b>6</b> .0 | ;            | ан<br>Х    |
| SAMBIC         | 80      | N/A         | ;          | 4/W         | :         | 3.16-13         | 8          | W/A         | ;           | N/N        | :      | N/A        | :      | 1.06-15  | 8     | N/A        | :      | 13050     | 72.5     | 14.5        | ;            | ۲¥         |
| SAME           | -       | N/A         | :          | N/A         | ;         | 1.96-12         | 5          | N/N         | :           | N/A        | ;      | <b>N/A</b> | :      | 6.65-15  | 8     | A/A        | ;      | 9630      | \$20.6   | 21.4        | ;            | Ĩ          |
| SURVE          | 80      | A/A         | :          | A/A         | :         | 1.96-12         | 5          | N/A         | :           | A/A        | ;      | N/A        | :      | 6.6E-15  | 2     | <b>A/A</b> | :      | 9006      | 580.0    | 20.0        | ;            | <u>ال</u>  |
| Netrori        | te str  | Lites the   | e holding  | ar e a      |           |                 |            |             |             |            |        |            |        |          |       |            |        |           |          |             |              |            |
|                | o       | N/A         | ;          | 2.06-11     | 26        | N/A             | ł          | R/N         | ;           | N/N        | ł      | N/N        | ;      | 2.06-11  | 26    | N/A        | :      | 21000     | ;        | ;           | 170.0        | Ĩ          |
| SAPHC          | •       | N/A         | :          | R/A         | ;         | 2.26-11         | 26         | M/N         | ;           | N/N        | ;      | N/N        | :      | 2.26-11  | 26    | N/N        | :      | 21060     | :        | 351.0       | 52.7         | ZONIN      |
| SAPPC          | e.      | N/A         | :          | R/A         | ;         | 2.26-11         | 26         | A/M         | :           | N/N        | :      | N/N        | :      | 2.26-11  | 26    | N/N        | :      | 10800     | :        | 180.0       | 13.5         | 20M [ N    |
| SADEC          | •       | N/N         | ;          | N/N         | ł         | 2. <b>A-1</b> 1 | <b>5</b> 8 | N/A         | ;           | A/A        | :      | N/A        | 1      | 2.26-11  | 26    | 8/8        | :      | 13050     | :        | 217.5       | 63.3         | 20M1W      |

ŚŻ

1.5.5.5.1 2.5.5.5.1

37.772X-

5555

 $\hat{\mathbf{N}}$ 

File: STOMMOLD.UKI Page 2 Date 21-Aug 87

INTERIM STORAGE ALR OPTION (EVENTS/YR)

11011

2.° # ] #]

 $i_{i}$ 

MARCHARCERARCE

00000000

41:e: STORMOUD.UNI Page 3 Date 21-Aug-87

INTERIM STORAGE AIR OPTION (EVENTS/YR)

| RATION          | H I          | 20MIN<br>20MIN     |
|-----------------|--------------|--------------------|
| BS.             |              | 48.2<br>11.3       |
| L155.           | DETONATED EN | 160.5<br>150.0     |
| LBS.            | SPILLED      | 11                 |
| AGENT           | AVAILABLE    | 9630<br>9000       |
| RANGE           | FACTOR       | 1:                 |
| MDA             |              | N/N<br>N/A         |
| RANGE           | ACTOR        | 2 2                |
| TEAD            |              | 2.0€-11<br>2.0€-11 |
| RANGE           |              | 11                 |
| PUBA            |              | 8/8<br>8/8         |
| RANGE           |              | ;;                 |
| <b>VII</b> d    |              | R/A<br>R/A         |
| RANGE           |              | 11                 |
| <b>AW</b>       |              | N/N<br>N/N         |
| RANGE<br>Factor |              | 26<br>26           |
| LBAD            |              | 2.0E-11<br>2.0E-11 |
| RANGE<br>Factor |              | ;;                 |
| 946             |              | 8/8<br>8/8         |
| RANGE<br>FACTOR |              | ;;                 |
| GMAD            |              | 8/8<br>8/8         |
| 9               | !            | • •                |
| Scenars         |              | SARKE              |

**JSSAMARIE** 

84. 888884 NASAN 88884 NA



10.010.000

I.1.3. HANDLING

The following tables list the accident results for handling of munitions.







**6**47.





500 C 20

COCOCCUSE

File: WLIEGST.UK1 17-Aug-97 Poge 1

#### HANNLING ACCIDENTS - REGIONAL PROCESSING OFTION - PER FALLET

Accident Frequencies and Range Factors

Agent Available and Released

| SCE          | 8  |          | Panel P     | 5          | <b>CANE</b> | 3           |           |         |          | The second |           | MON      | RANGE     | 15 AD     | <b>NAMES</b> |            | <b>RANKS</b> | 1650     | S     | S        |           | DURATI |
|--------------|----|----------|-------------|------------|-------------|-------------|-----------|---------|----------|------------|-----------|----------|-----------|-----------|--------------|------------|--------------|----------|-------|----------|-----------|--------|
| ME 10        |    | <u>B</u> | FACTOR      | FIELD      | FACTOR      | FINEG       | FACTOR    | JAC O   | FACTOR   | FIRED      | ACTOR     | FIED     | FACTOR    | FIRE      | ACTOR        | FRED       | FACTOR       | WAILABLE | BILLE | ETOMATED | ENTTED    | Mil.   |
|              | ;  | -        |             |            |             |             |           |         |          |            |           |          |           |           |              |            |              |          |       |          |           |        |
| NC DOC       | -  | 1 8/4    | :           | A/H        | :           | A/A         | ;         | N/A     | ;        | N/A        | 1         | N/N      | ;         | 1.26-07   | 1.36+01      | 6.16-00    | 1.3E+01      | 440.0    | ;     | :        | 4.36+00   | ۹.     |
| KINC         | -  | 1 3.26-1 | 1. 3.401    | N/A        | ł           | N/N         | :         | N/A     | 1        | R/N        | ;         | 1.46-98  | 1.32401   | 3.26-00   | 1. 36+01     | N/N        | ;            | 200.0    | :     | ;        | 1.36-03   | Ĩ      |
| NCCRC        | -  | 1 5.76-0 | # 1.X.01    | N/N        | ;           | A/A         | 1         | N/N     | :        | W/W        | ;         | N/N      | 1         | 5.76-09   | 1.36+01      | N/N        | ;            |          | ;     | ;        | 1.16-01   | E      |
| NCOK         | -  | 1 3.16-0 | 1.36+01     | A/A        | :           | N/N         | :         | N/N     | :        | <b>V</b>   | :         | 2.86-09  | 1.36+01   | 5.76-04   | 1.36+01      | W/W        | :            | 74.8     | :     | :        | 1. 36-03  | Ĩ      |
| HCKGC        | -  |          | ;           | W/W        | <b>,</b> !  | A/A         | ;         | N/A     | ;        |            | ;         | N/N      | ;         | 4.06-00   | 1.36401      | N/N        | ł            | 1500.0   | ł     | ;        | 6.4E+00   | Ŧ      |
| HCKHC        | -  | 9-30-4   | 1.1.1.1     | 2.05-06    | 1.36+01     | N/N         | ;         | N/N     | :        | 2.06-00    | 1.3E+01   | N/N      | :         | 1.06-08   | 1.3€+01      | 2.06-08    | 1.36+01      | 1700.0   | ;     | :        | 2.56-02   | Ŧ      |
| HCKVC        | -  | 1 1.06-0 | 1.36-01     | N/N        | :           | <b>N/N</b>  | :         | 2.06-00 | 1. 36+01 | <b>V/</b>  | ;         | N/N      | :         | 4.06-00   | 1.36+01      | N/N        | :            | 1600.0   | ;     | ;        | 2.76-04   | Ĩ      |
| HCMC         | -  | 1 2.4E-0 | 1.36+01     | <b>W/W</b> | ;           | N/A         | :         | N/N     | :        | 1.75-00    | 1. XE +01 | N/N      | :         | 2.46-00   | 1.36+01      | 1.25-00    | 1. 35 +01    | 378.0    | ;     | ;        | 1.76-05   | Ĩ      |
| 3943N        |    | 1 0.06+0 | ;           | N/N        | ;           | 8/8         | ;         | R/A     | ;        | AIN.       | ;         | W/W      | :         | 0.0€+00   | :            | 0.06+00    | ſ            | 52.0     | ;     | ;        | :         | :      |
| MCPMC        | -  | 1 0.05+4 | :           | N/N        | ;           | 0.0€+00     | ;         | N/A     | ł        | N/N        | ;         | -0°+30°0 | 1         | 0.0£+00   | :            | N/A        | :            | 93.4     | ;     | ;        | ;         | ;      |
| NCMC         | -  | 1 0.05+0 | :           | N/N        | ;           | 0.0€+00     | :         | N/N     | ;        | N/N        | :         | N/N      | :         | 0.0€+00   | :            | 0.0E+00    | :            | 48.0     | ;     | ;        | :         | ł      |
| HCBEC        | -  | 1 0.05+0 | :           | K/A        | ;           | 0.0€+00     | :         | N/A     | 1        | A/A        | :         | K/A      | :         | 0.0£+00   | :            | 0.0€+00    | :            | 87.0     | ;     | :        | ł         | :      |
| HCENC        | -  | N/N      | :           | N/N        | ;           | N/A         | :         | N/N     | :        | W/W        | ;         | N/N      | :         | 0.0€+00   | :            | 0.0£+00    | :            | 87.0     | ;     | :        | :         | :      |
| HCR6C        | -  | 1 9.36-0 | 1.XE+01     | W/W        | ;           | 4. R 00     | 1. 36 +01 | N/N     | :        | 1. IK-0    | 1. XE +01 | N/N      | :         | 9.55-00   | 1.32+01      | 4. 86 - 00 | 1. 36+01     | 160.5    | 1     | ;        | 4. X -01  | Ē      |
| HCINC        | -  | 1 9.55-0 | B 1.3E+01   | <b>N/N</b> | :           | 4. IK -00   | 1. X +01  | N/N     | :        | 4. IE - M  | 1.3£+01   | N/N      | :         | 9.56-00   | 1.36+01      | 4.86-00    | 1.35+01      | 150.0    | ;     | :        | 1.46-05   | Ē      |
| HESVC        | -  | N/N 1    | ;           | <b>W/W</b> | :           | N/N         | ;         | N/N     | :        | N/N        | :         | N/N      | ;         | 1.05-04   | 1.36+01      | 5. 26-07   | 1. 3E+01     | 1354.0   | ;     | :        | 2.76-04   | 9      |
| HCKIN        |    | 5 N/A    | ;           | 5.26-10    | 3. IE +01   | A/A         | ;         | N/N     | 1        | 5.26-10    | 3.1E+01   | N/A      | ;         | 5.26-10   | 3. IE+01     | W/W        | :            | 1700.0   | :     | ;        | 0.3£+01   | IO NU  |
| HCINGC       | -1 | 5 N/A    | :           | N/N        | ;           | N/N         | ;         | N/A     | :        | N/A        | ;         | N/N      | :         | 2.46-07   | 1. 36+01     | 1.36-07    | 1.3€+01      | 440.0    | :     | ;        | 4. 3£+00  | Ē      |
| Ne Ne        | -7 | 5 7.46-0 | 7 1.36+01   | R/A        | 1           | N/N         | :         | N/A     | ;        | R/A        | ;         | 3.7E-07  | 1. XE +01 | 7.46-07   | 1. XE +01    | N/N        | ;            | 286.0    | ;     | ł        | 1. 36 -03 | Ĩ      |
| HCCBC        | -1 | 5 1.86-0 | 7 1. JE +01 | N/N        | 1           | A/A         | ł         | N/N     | ;        | N/A        | ;         | N/N      | ł         | 1.86-07   | 1.36+01      | R/A        | :            | 4.8%     | :     | :        | 1.16-01   | Ĩ      |
| MCH          | -1 | 5 1.86-0 | 7 I.X +01   | N/A        | :           | <b>4</b> /2 | ;         | N/N     | ;        | N/N        | ł         | 80-36-8  | 1.36+01   | 1.05-07   | 1.36+01      | N/N        | ;            | 74.0     | :     | :        | 1. JE -03 | ۹.     |
| HCIIINC      | -1 | 5 1.16-0 | 4 1.Xt+01   | N/N        | ;           | N/N         | :         | N/N     | ;        | 7.1E-07    | 1.XE+01   | A/A      | :         | 1.46-06   | 1.3£+01      | 7.16-07    | 1. 3E +01    | 378.0    | :     | ;        | 1.76-05   | Ĩ      |
| NCPGC        | -1 | 3 1.06-0 | 7 1.XE+01   | 2          | :           | A/A         | ;         | N/N     | ;        | N/N        | :         | N/N      | :         | 1.06-07   | 1.36+01      | 5.06-08    | 1. 3£ +01    | 52.0     | :     | ;        | 2.4E-01   | Ē      |
| HCHC<br>HCHC | -1 | 3 1.06-0 | V 1.X.01    | A/A        | :           | 5.06-00     | 1. JE +01 | N/A     | :        | N/N        | ;         | 5.0E-08  | 1.36+01   | 1.05-07   | 1.36+01      | N/A        | ;            | 93.6     | :     | :        | 2. IE-03  | 5      |
| Ж            | ~7 | 3 1.06-0 | 7 1. XE+01  | R/R        | :           | 2, S, S     | 1. XE +01 | N/N     | 1        | N/N        | :         | N/N      | ;         | 1.0E-07   | 1.3E+01      | 5.06-08    | 1. 3E+01     | 48.0     | :     | :        | 2. IE -05 | Ē      |
| HC OGC       | -7 | 3 1.06-0 | 1.3E+01     | N/N        | ;           | 5.06-08     | 1.3£+01   | N/A     | ł        | W/W        | ł         | N/N      | 1         | 1.06-07   | 1.XE+01      | 5.06-08    | 10+31.1      | 0.70     | ;     | :        | 5.46-01   | Ĩ      |
| HCBNC        | -1 | 5 N/A    | :           | N/N        | :           | <b>N/N</b>  | :         | N/A     | :        | N/N        | ł         | N/N      | ;         | 1.0E-07   | 1.3E+01      | 5.0E-08    | 1.36+01      | 87.0     | :     | :        | 2. IE-05  | 2      |
| NCREC        | -1 | 5.XE-0   | 10-37.1 \$  | N/N        | :           | 2.65-04     | 1.3£+01   | N/A     | ł        | 2.46-06    | 1.3E+01   | N/N      | ;         | 5.36-04   | 1.3£+01      | 2.46-06    | 1.3£+01      | 140.5    | :     | ;        | 4.56-01   | Ē      |
| HCIIVC       | -  | 5.36-0   | 6 L. 3E+01  | N/N        | :           | 2.66-04     | 1.36+01   | N/N     | ;        | 2.46-06    | I.X +01   | N/A      | :         | 5. JE -06 | 1. 3E+01     | 2.6E-06    | 1.3E+01      | 150.0    | :     | ;        | 1.65-05   | Ĩ      |

Filer HRURESST.MC1 17-Aug-87 Page 2

CLAIN ADA NA

HANDALING ACCIDENTS - REGIONAL PROCESSING OPTION - PER PALLET

| r) nosed    | LDS BURATION<br>ITTED TIME |   |            | 100-10<br>100-10 | 3E-03 114 | 1E-01 [148 | 新して<br>第二<br>第二<br>第二<br>第二<br>第二<br>第二 | #+00 IM         | 5E-02 118 | 7E-04 1HR           | H 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | :        | :      | 1       | 1       | 1       |            | R-5 18        | 2-41 LH | <b>¥</b>    | <b>9</b>     | <b>E</b>    | <u>۲</u>   | <b>H</b>   | Ē :     | <u></u>  | <u>۳</u>  | <u></u>   | <u>.</u>  | 1           |  |
|-------------|----------------------------|---|------------|------------------|-----------|------------|-----------------------------------------|-----------------|-----------|---------------------|-----------------------------------------|----------|--------|---------|---------|---------|------------|---------------|---------|-------------|--------------|-------------|------------|------------|---------|----------|-----------|-----------|-----------|-------------|--|
| lable and A | LDS<br>Stonated En         |   | :          | -<br>-<br>-      | <br>-     | <br>1      | -<br>1                                  | - <b>a</b><br>1 |           | ين<br>۱             |                                         | :        | ;      | :       | ;       | ;       | -          | -             |         | :           | ;            | ;           | :          | 1          | :       | ;        | ;         | ;         | ;         | 1           |  |
| gent Avail  | LINS<br>SPILLED M          |   | ł          | 1                | 1         | 1          | ;                                       | :               | 1         | ł                   | :                                       | 1        | :      | :       | ;       | 1       | ł          | ł             | :       | 2.X +02     | 4.05+00      | 1.46+00     | 3.26+00    | 1.56+03    | 1.76+03 | 1.46+03  | 1.06+01   | 6-1X-19   | 1.ZE+01   | 6. K + M    |  |
| •           | AVAILABLE<br>AVAILABLE     |   | 1334.0     | 440.0            | 200.0     | 7.35       | 74.8                                    | 1500.0          | 1790.0    | 1400.0              | 378.0                                   | 52.0     | 9-64   |         | 07.0    | 17.0    | 160.5      | 126.0         | 1354.0  | 440.0       | 200.0        | <b>1</b> .8 | 74.8       | 1500.0     | 1700.0  | 1400.0   | 379.0     | 52.0      | 93.4      | 48.0        |  |
|             | RANKE<br>FACTOR            |   | 1. 26 401  | 1.36+01          | ;         | :          | 1                                       | :               | 1. 36+01  | ł                   | 1.32+01                                 | :        | ;      | 1       | 1       | 1       | 10+3(1)    | 1.36+01       | 1.32.01 | :           | 1            | :           | ł          | :          | :       | ;        | 1         | 1         | :         | ł           |  |
|             | Nen<br>Line                |   | 1.36-07    | 1.76-09          | N/N       | N/N        | N/N                                     | N/A             | 1.26-09   | N/N                 | 1-36-1                                  | 0, 15+00 | N/N    | 0.05+00 | 0.45-00 | 0.05+00 | 3. IE-09   | 3.16-41       | 2.76-08 | N/N         | W/W          | N/N         | R/A        |            | W/H     | W/W      | N/N       | W/W       |           | N/N         |  |
|             | FACTOR<br>FACTOR           |   | 7 1.36+01  | 9 1.35+01        | 9 1.X +01 | 1.1.144    | 1 1. XE+01                              | 1. 1. 10.       | 1.36401   | 1.32.01             | 9 1.35+01                               | :        | :      | :       | :       | 1       | 9 1. 26-01 | 9 1.26401     | 1.36+01 | 9 1.JE+01   | 9 1.32401    | 0 1.3€+01   | 0 1.35.01  | 1.1441     | 1.1.1.1 | 10-37.1  | 9 1.3E+01 | 9 1.26401 | 9 1.36+01 | 9 1. 26 401 |  |
|             | TEA0<br>Fined              |   | 6.1E-0     | 3.46-0           | 1 1.46-0  | 3.16-1     | 1 3.9E-1                                | 10-31-1         | 1.46-0    | 1-#-0               | 1.46-0                                  | 0-36-0   | 0.06+0 | 0-96-0  | 0.05+0  | 0-30-0  | 6.26-0     | 4-32.4        | 5.4E-0  | 0-X-9       | 4.36-0       | 1-32.1      | 7.26-11    | 1.36-0     | 1.36-0  | 1.36-0   | 0-39-8    | 1.15-0    | 1. IE-0   | 1, LE-0     |  |
|             | FINE                       |   | :          | 1                | 10 1.XH   | :          | II 1.XH                                 | ł               | ;         | :                   | :                                       | ł        | :      | 1       | 1       | t       | ł          | 1             | 1       | 1           | 1            | :           | 1          | :          | 1       | :        | :         | 1         | :         | :           |  |
|             | PURM<br>Fired              | - | A/A        | N/N              | -90-1     | N/N        | 2.06-                                   | N/N             | A/N IO    | A/H                 | 01 II/N                                 | A/A      | ¢.0E+  |         | N/N     | N/N     | A/M 10     | 01 N/A        | N/N     | <b>1</b> /1 | N/N          | N/N         | A/N        | N/N        | W/W     | A/A      | N/N       | N/N       | N/N       | N/N         |  |
|             | RANKE<br>FACTOR            |   | :          | 1                | 1         | 1          | :                                       | :               | ₩ 1.X+    | ŀ                   | -10 1.X.                                | I        | ;      | t       | :       | 1       | 1. I.      | H 1.K.        | 1       | 1           | :            | 1           | 1          | 1          | :       | ł        | 1         | ł         | 1         | 1           |  |
| kters       |                            | - |            |                  | N/N       | N/N        | N/N                                     | A/N             | 4. X      | V/N IO              | -<br>-                                  |          | N/N    | N/N     | N/N     | N/N     | 3.16-      | 3.16-         |         |             | N/N          | W/W         | A/A        |            | N/N     | N/N      | N/N       | A/N       | N/N       |             |  |
| d Range Fi  |                            |   | ;          | +                | 1         | ۱<br>-     | :                                       | -               | 1         | H- 5-               | 1                                       | :        | 1      | 1       | 1       | 1       | :          | 1             | :       | +           | ;            | 1           | 1          | 1          | :       | 1        | 1         | 1         | 1         | 1           |  |
|             | ₩.                         |   | A/M        | N/N              |           | N.         | N/N                                     | N/N             | N/N       | 7.7                 | 1/1                                     | 1/1      | R/A    | N.      |         | N/N     | I N/N      | I NV          | N.      | N/N         | N/N          |             | N/N        | N/N        | S       | N/N      | N/N       |           | N.        | N/N         |  |
| it Frequen  | RANK<br>FACTOR             | İ | :          | 1                | 1         | I          | :                                       | ;               | :         | ł                   | ł                                       | 1        | ;      | :       | :       | 1       | H 1.XH     | 4 L.XH        | 1       | 1           | ł            | ł           | :          | t          | :       | :        | ;         | 1         | ;         | ;           |  |
| Acciden     |                            |   | <b>W/H</b> | N/N              | N/N       | N/N        | A/A                                     | A/A             | N/N 10    | N/N                 | N/N                                     | N/N      | 0.06+0 | 0.05+1  | 9-90.6  | W/N     | 3. 16-0    | 3.16-4        | A/N     | <b>8/8</b>  | W/W          | W/W         | A/A        | R/A        | A/N     | W/W      | N/N       | N/N       | N/N       | W/W         |  |
|             | FACTOR                     |   | :          | ł                | 1         | I          | <b>,</b> †                              | :               | M 1.XH    | ;                   | :                                       | ł        | :      | :       | :       | 1       | :          | 1             | :       | 1           | :            | 1           | ;          | 1          | ;       | ł        | 1         | 1         | ;         | 1           |  |
|             |                            |   | A/N        | 8/8              | N/N I     | N/N I      | N/N                                     | A/N             | -31-9     | NN I                | A/M                                     | A/A      | A/A    | N/N     | A/N     | N/N     | 8/8        |               | N/N     |             |              |             | N/N        | <b>N/N</b> | VII I   | N.       |           | N/N       | 8/8       | W/W         |  |
|             | FACTOR<br>FACTOR           |   | :          | :                | 1. 3440   | 1.244      | 1.36+01                                 | :               | 1.X.0     | B 1.X+0             | 9 1. XE +0                              | ;        | :      | 1       | ;       | l       | 9 L.XHI    | 9 1.264       | 1       | ł           | 1.26.01      | 0 1.3440    | 0 1. 16 40 | I          | 1.1.1.  | 1. 1. 1. | 1.1.1440  | 1.12.40   | 1. 32 +01 | 1.14.40     |  |
|             |                            |   |            | N/N              | 1.46-0    | 3.96-1     | 3.96-1                                  | A/A             | 1.46-0    | 1. <del>1</del> . – | 1.46-0                                  | 0.05+0   | 0.06+8 | 0.06+0  | 0.06+0  | N/A     | 6.25-0     | <b>6.75-0</b> | A/H     | A/A         | <b>Р</b> Ж.4 | 7.26-1      | 7.26-1     | N/A        | 1.36-0  | 1.36-4   | 9-34.8    | 1.16-0    | 1.16-0    | 1.15-0      |  |
|             | SCEN- DP. NO.<br>Meio      |   | NCSWC 3    | HCINE +          | KCMC 4    | HCCBC 4    | HCDIC 1                                 | NCKRC 4         | HCKKC 4   | ACING 4             | NCINC +                                 | NCMC +   | ICMC 4 | +<br>WX | NCOSC 4 | HCBNC 1 | + HORE     | HCINC 4       | HESNC + | HCINGS 5    | HCINS 5      | NCC65 5     | HCDIS S    | HCKES 5    | HCURS 5 | NCIVS 5  | ICINS 5   | NCPGS 5   | KCPHS 5   | HCPVS 5     |  |

MARKANNA NYZZA RYVNA RYVNA

anala sanan

<u> I KARAN</u> INANA I KANAN

Karden -

2





Page 3 11-hug-07 Filer MUREST.MCI NAMELING ACCIDENTS - REGIONAL PROCESSING OFTION - PER PALLET

|          |        |         |               |             |                | Acciden:    | t Frequenci | I pur su | lange Facto | 5          |                  |              |                 |               |                 |            |                | -                  | Agent Ava | e aldeli        | nd Release    | 7            |
|----------|--------|---------|---------------|-------------|----------------|-------------|-------------|----------|-------------|------------|------------------|--------------|-----------------|---------------|-----------------|------------|----------------|--------------------|-----------|-----------------|---------------|--------------|
| SCEN- OI | 2<br>4 |         | MME<br>FACTOR |             | RANK<br>Factor | L NA        | FACTUR      | ANN CONT | RANGE       |            | FACTOR<br>FACTOR | PUM<br>Fired | RANKE<br>FACTOR | TEAD<br>Fired | RANGE<br>FACTOR |            | RAME<br>FACTOR | AGENT<br>AMAILABLE |           | LPS<br>BETOMATE | LIS<br>DITTED | IIIK<br>IIIK |
|          | :      |         |               |             |                |             |             |          |             |            |                  |              |                 |               |                 |            |                |                    |           |                 |               |              |
| HEBNS    | n      | N/N     | :             | N/N         | ;              | N/N         | :           | A/N      | ;           | W/W        | ;                | N/N          | ;               | 1.16-09       | 1. JE +01       | N/N        | ;              | 07.0               | 1.58.41   | ;               | ł             | Ĩ            |
| HCINES   | 'n     | 1.96-09 | 1. X +01      | N/N         | 1              | NVA.        | :           | R/N      | :           | <b>N</b>   | :                | N/N          | ;               | 1.95-09       | 1.36.41         | N/A        | 1              | 16.5               | 1.16+01   | ł               | :             | Ĩ            |
| NCINS    | 5      | 1.95-09 | 1.36+01       | 8/8         | 1              | N/N         | :           | -        | I           | . W/N      | :                | N/N          | :               | 4.96-09       | 1.36401         | N/N        | :              | 150.0              | 1.05+01   | ;               | 1             | Ĩ            |
| NCSVS    | 'n     | N/A     | :             | N/N         | 1              | N/N         | :           | N/N      | :           | N/N        | :                | N/A          | ;               | 1.46-07       | 1. X 401        | N/N        | ł              | 1356.0             | 1.4£+03   | :               | :             | Ĩ            |
| NC DGF   | -      | N/N     | :             | A/A         | <b>`</b> ¦     | N/N         | 1           | N/N      | :           |            | 1                | A/H          | :               | 11-32.9       | 3. 16+01        | N/N        | :              | 440.0              | ;         | ;               | 2.26.01       | IN NI        |
| HC DIF   | •      | 1.96-10 | 3.16+01       | A/A         | :              | A/A         | 1           | N/N      | !           | N/N        | :                | N/N          | :               | 1.96-10       | 3. 16+01        | N/A        | ;              | 200.0              | :         | ;               | 3.06-01       | 10 MI        |
| HCCGF    | -      | 0.0€+00 | :             | N/N         | ;              | N/A         | :           | N/N      | :           | N/N        | ł                | N/A          | :               | 0.0€+00       | :               | N/N        | :              | 4. M               | :         | ;               | :             | :            |
| NCCHE    |        | 0.0E+00 | ;             | A/A         | 1              | N/N         | ;           | N/N      | :           | A/A        | 1                | N/N          | :               | 0.06+00       | :               | N/A        | ;              | 74.8               | ;         | :               | :             | :            |
| HCK&F    | -0     | N/A     | :             | N/N         | ÷              | <b>N/N</b>  | :           | N/N      | 1           | <b>N/A</b> | :                | <b>N/A</b>   | :               | 2.16-10       | 3.16+01         |            | ;              | 1500.0             | ;         | 1               | 1.56.42       | IN OL        |
| HCKH     | •      | 2.16-10 | J. 1E+01      | N/N         | ;              | A/A         | :           | N/A      | :           | N/N        | 1                | N/A          | ;               | 2. IE-10      | 3. 16+01        | A/A        | :              | 1700.0             | :         | :               | B. 5£+01      | 10 M         |
| NCKVF    |        | 2.15-10 | 3. IE+01      | R/N         | :              | N/N         | ;           | N/N      | :           | N/A        | :                | N/A          | :               | 2.16-10       | 3. IE+01        | N/A        | :              | 1600.0             | :         | ;               | 1.06+01       | 10 MI        |
| HCIEVE   |        | 2.76-10 | 3.16+01       | R/A         | :              | A/N         | ;           | N/A      | ;           | A/N        | ;                | N/A          | ;               | 2.76-10       | 3.16+01         | N/A        | :              | 378.0              | :         | :               | 2.66-01       | IN OI        |
| HCP6F    | •      | 0.0£+00 | :             | <b>W/A</b>  | ;              | N/N         | 1           | A/A      | ;           | A/A        | :                | N/A          | ł               | 0.0£+00       | :               | N/A        | :              | 52.0               | 1         | ;               | :             | :            |
| HCPW     | 4      | 0.0E+00 | :             | N/N         | ;              | N/N         | :           | N/N      | ;           | N/N        | :                | N/A          | :               | 0.06+00       | :               | N/A        | ;              | 93.4               | 1         | :               | ;             | :            |
| NCPVF    | •      | 0.0E+00 | ;             | R/A         | ;              | N/N         | :           | N/A      | ;           | N/N        | :                | N/N          | ;               | 0.0£+00       | :               | N/N        | ;              | <b>10</b> .0       | :         | ;               | :             | ;            |
| HC OCL   | •      | 0.0€+00 | ł             | A/A         | ;              | N/N         | 1           | R/A      | :           | A/A        | !                | N/N          | ;               | 0.05+00       | ;               | N/A        | :              | 87.0               | 1         | ;               | :             | 1            |
| HCOVE    | -      | N/N     | ł             | W/W         | :              | N/N         | :           | N/N      | ;           | N/N        | ;                | N/N          | :               | 0.0E+00       | :               | N/N        | :              | 87.0               | ;         | ;               | :             | ;            |
| HCINGF   | 4      | 1.46-10 | 3. IE +01     | N/A         | :              | N/N         | 1           | N/N      | ł           | A/A        | ;                | A/N          | ;               | 01-39.1       | 3. IE +01       | N/N        | ;              | 160.5              | :         | ;               | 1.16+00       | 10 M         |
| HCRVF    | -0     | 1.46-10 | 3.16+01       | N/N         | ;              | N/N         | :           | N/N      | ;           | N/N        | ;                | N/N          | :               | 1.46-10       | 3. IE+01        | N/N        | ;              | 150.0              | ł         | ł               | 2.55-01       | IO NII       |
| HCSVF    | -      | N/N     | 1             | N/N         | ;              | N/N         | ;           | A/H      | 1           | N/N        | ;                | <b>K</b> /N  | ł               | 3.96-09       | 3. IE+01        | N/A        | ;              | 1354.0             | ł         | :               | 3.4E+01       | 10 MI        |
| NC D65   | ~      | N/N     | :             | N/N         | :              | N/A         | ;           | N/N      | ;           | N/N        | :                | N/A          | 1               | 3.76-09       | 1.3£+01         | <b>W/W</b> | ł              | 440.0              | 2.7£+02   | :               | ;             | Ĩ            |
| HC BHS   | ~      | 1.16-00 | 1.3£+01       | N/N         | :              | N/N         | :           | N/N      | :           | M          | :                | N/A          | ;               | 1.16-00       | 1.36+01         | N/N        | ł              | 200.0              | 6.00+00   | :               | 1             | Ĩ            |
| HCCBS    | ~      | 0.06+00 | :             | R/N         | :              | N/A         | :           | N/N      | :           | W/W        | 1                | N/N          | :               | 0.0E+00       | ;               | N/A        | ;              | +.<br>197          | ;         | :               | ;             | ;            |
| NCONS    | -      | 0.0E+00 | :             | A/A         | ;              | N/A         | ;           | #/#      | ;           | N/N        | 1                | N/A          | :               | 0.05 >00      | 1               | N/A        | ;              | 74.8               | ;         | :               | 1             | 1            |
| MCK6S    | ~      | N/N     | :             | N/A         | :              | N/N         | :           | R/A      | :           | W/W        | :                | N/N          | :               | 1.26-00       | 1.36+01         | N/A        | ;              | 1500.0             | 1.5£+03   | :               | 1             | Ē            |
| NCICHS   | -      | 1.25-09 | 1. 36 +01     | <b>1</b> /1 | ;              | N/N         | ;           | N/N      | :           | W/W        | ;                | N/A          | :               | 1.26-08       | 1.36+01         | N/A        | ;              | 1700.0             | 1.76+03   | ;               | 1             | Ē            |
| NCKVS    | -      | 1.26-00 | 1.3£+01       | N/N         | :              | #/#         | ;           | W/W      | :           | N/N        | 1                | N/N          | ;               | 1.25-08       | 1.3£+01         | N/N        | :              | 1400.0             | 1.46+03   | ;               | :             | Ŧ            |
| MCIIIVS  | ~      | 1.46-08 | 1.3E+01       | <b>A</b> /A | I              | <b>V</b> /R | :           | N/A      | :           | A/A        | :                | W/W          | :               | 1.46-00       | 1. X +01        | A/A        | 1              | 378.0              | 1,0£+01   | :               | :             | Ē            |
| HCP65    | ~      | 0.05+00 | :             | N/N         | :              | N/N         | ł           | N/A      | ;           | N/N        | ;                | N/A          | :               | 0.0£+00       | ;               | R/A        | :              | 52.0               | 1         | :               | ;             | ;            |

Pape 4 17-Aug-07 Filer MURESS.MU 1317W WCLINEWLZ - BEBIORNY ADOCESSING ODIION - 5EN PAYTEL

1 4.1 V. \*. 61".

ろんというな

|                  |        |         |                 |            |                 | Accident     | Frequenci       | in mi      | ange Factor     | z               |                 |            |                 |              |                 |            |               | đ          | pent Avai       | lable and         | Re] eased         |                                  |
|------------------|--------|---------|-----------------|------------|-----------------|--------------|-----------------|------------|-----------------|-----------------|-----------------|------------|-----------------|--------------|-----------------|------------|---------------|------------|-----------------|-------------------|-------------------|----------------------------------|
| SCCI- DI<br>Milo | ġ.     |         | RANKE<br>Factor | £ ģ        | RANGE<br>FACTOR | LINN<br>Free | RANNE<br>Factor | FINE F     | RANGE<br>FACTOR | N G             | AANGE<br>Factor | FIER       | RANGE<br>Factor | TEAB<br>FINE | RANGE<br>Factor |            | NAME<br>ACTOR | MAILANLE : | LINS<br>ULLED 1 | LIG<br>ETOMATED ( | LINS IN<br>MITTED | 10<br>10<br>11<br>11<br>11<br>11 |
|                  | i<br>I |         |                 |            |                 |              |                 |            |                 |                 |                 |            |                 |              |                 |            |               |            |                 |                   |                   |                                  |
| KUNK             | ~      | 0.05.00 | ;               | N/N        | :               | 2            | !               | N/N        | :               | N/N             | 1               | N/N        | ;               | 0.05+00      | :               | N.         | ł             | 93.4       | ;               | ;                 | ;                 | ;                                |
| RCMS             | -      | 0.0E+00 | 1               | N/A        | ;               | N/N          | :               | W/W        | ł               | N/N             | :               | N/N        | :               | 0,0€+00      | :               | N/N        | ł             | 48.0       | ł               | ;                 | :                 | ;                                |
| NC WES           | ~      | 0.05+00 | :               | N/N        | ;               | <b>1</b> /1  | 1               | R/A        | ł               | <b>N/N</b>      | ;               | N/N        | :               | 0,0£+00      | ;               | <b>N/N</b> | ;             | 87.0       | ;               | ;                 | ;                 | ;                                |
| NCINS            | ~      | N/N     | ł               | A/A        | !               | 5            | •               |            | ۱               | N/N             | 1               | V.W        | ł               | 0.0£+00      | ;               | V/I        | ţ             | 87.0       | ł               | ;                 | 1                 | :                                |
| <b>ACM65</b>     | ~      | 9.46-01 | 1. 32 -01       | N/A        | <b>`</b> ¦      | A/A          | :               | W/W        | ł               | , vi            | ı               | N/N        | ;               | 9.46-09      | 1. 36 +01       | N/N        | ;             | 160.5      | 1. iE +01       | :                 | :                 | H.                               |
| ACINS            | •      | 9.45-09 | 1. 35 +01       | A/A        | ;               | N/A          | 1               | A/A        | :               | A/A             | 1               | N/N        | 1               | 9.66-09      | 1.36+01         | W/W        | ;             | 130.0      | 1.05-01         | :                 | ;                 | Ē                                |
| HCSVS            | ~      | N/A     | ł               | A/A        | ;               | R/A          | ;               | N/N        | 1               | N/N             | :               | N/N        | :               | 2.36-07      | 1. XE+01        | <b>V/R</b> | ;             | 1354.0     | 1.4£+03         | 1                 | 1                 | Ŧ                                |
| <b>HCINES</b>    | -      | N/N     | !               | N/N        | ;               | N/N          | ;               | A/N        | 1               | #/#             | ;               | N/N        | ł               | 2.36-08      | 1.36-01         | 2.35-08    | 1. XE+01      | 10540.0    | 2. XE +02       | :                 | :                 | Ē                                |
| HC DHS           | -      | 1.25-06 | 1.32.01         | W/W        | ł               | A/N          | ;               | N/N        | ł               | <b>V</b>        | ;               | 1. X -08   | 1. JE +01       | 1.25-08      | 1.3£+01         | N/N        | ł             | 1101.0     | 6.05+00         | :                 | !                 | Ē                                |
| HEC65            | -      | 1.0C-09 | 1.3£+01         | W/W        | ł               | N/N          | ł               | N/N        | 1               | N/N             | ;               | N/N        | 1               | 8. AE -01    | 1.36+01         | N/N        | 1             | 1843.2     | 1.65+00         | ;                 | ł                 | Ē                                |
| HCCHS            | on,    | 8.06-09 | 1.35+01         | W/W        | ;               | N/N          | 1               | N/N        | 1               | N/N             | 1               | E. KO      | 1. X 401        | 1. JE-99     | 1. X +01        | R/A        | :             | 1.111      | 5. ZE +00       | 1                 | ł                 | Ē                                |
| HCK65            | •      | N/N     | ł               | W/W        | ł               | N/N          | ł               | <b>N/N</b> | 1               | W/W             | ł               | N/N        | 1               | 2.26-00      | 1. 3E+01        | N/N        | ł             | 12000.0    | 1.5£+03         | :                 | ;                 | Ĩ                                |
| HCKHS            | -      | 2.25-08 | 1.14-01         | 2.26-08    | 1.15+01         | N/N          | ;               | N/A        | ;               | 2.26-00         | 1.3£+01         | N/N        | :               | 2.26-00      | 1. XE+01        | 2.25-00    | 1. X +01      | 13400.0    | 1.76+03         | :                 | :                 | ۹.                               |
| NCKVS            | -      | 2.26-08 | 1, 3E+01        | N/N        | ł               | W/W          | ;               | 2. XE-00   | 10-12-1         | <b>N/A</b>      | ı               | N/N        | :               | 2.26-00      | 1.14-101        | N/N        | :             | 12800.0    | 1.46+03         | ;                 | :                 | Ĩ                                |
| HCIINS           | -      | 1.25-00 | 1. JE +01       | R/A        | ł               | R/A          | :               | N/N        | ;               | 1.26-00         | 1.16+01         | <b>N/N</b> | :               | 1.26-00      | 10-31.1         | 1.26-00    | 1. X +0!      | 0.4534.0   | 10+34-1         | :                 | :                 | Ē                                |
| HCPGS            | 80     | 1.06-08 | 1. 3£+01        | ¥/W        | ł               | R/A          | :               | N/N        | ;               | N/N             | 1               | N/N        | 1               | 1, 06-01     | 1.36.401        | 1.05-00    | 1.3£+01       | 3120.0     | 1. SE +00       | ;                 | :                 | Ē                                |
| ICPIS            | ~      | 1.05-00 | 1.35.401        | N/N        | ł               | 1.05-08      | 1.3£+01         | N/N        | :               | N/N             | ;               | 1.05-00    | 1.16+01         | 1.06-00      | 1.3£+01         | N/N        | ł             | 5414.0     | 1.25+01         | :                 | ;                 | Ē                                |
| HCPVS            | æ      | 1.0£-00 | 1. 3E+01        | N/N        | ł               | 1.06-00      | 1.12+01         | N/N        | 1               | N/N             | :               | #/#        | ł               | 1.06-98      | 1.3£+01         | 1.05-08    | 1.3£+01       | 2880.0     | 6. 0E +00       | :                 | ł                 | Ē                                |
| HC06S            | æ      | 1.0E-08 | 1.36+01         | N/N        | ł               | 1, 06-08     | 1.3E+01         | N/A        | :               | <b>W/W</b>      | 1               | N/N        | :               | 1.06-08      | 1.3£+01         | 1.06-00    | 1.3£+01       | 3480.0     | 1.5£+01         | :                 | 1                 | Ű.                               |
| HCBVS            | æ      | N/N     | 1               | N/N        | ł               | 4.4          | :               | N/N        | ł               | R/A             | ł               | N/A        | :               | 1.05-08      | 1.3E+01         | 1.05-00    | 1.3£+01       | 3480.0     | 10-35-1         | ;                 | ł                 | Ŧ                                |
| HCR6S            | -      | 1.36-00 | 1. 3E+01        | A/A        | ;               | 1.05-00      | 1.32+01         | N/N        | 1               | 1.3%-00         | 1. 32 +01       | W/W        | 1               | 1.36-08      | 1.JE+01         | 1.16-00    | 1.3E+01       | 2541.0     | 1.16+01         | :                 | ;                 | Ŧ                                |
| HCRVS            | -      | 1.35-06 | 1. XE +01       | N/N        | ;               | <b>1</b>     | 10-37.1         | N/N        | 1               | 1.36-06         | 1. J. 1. 101    | W/W        | ;               | 1.36-08      | 1. XE +01       | 1.35-00    | 1.3£+01       | 2400.0     | 1.05.01         | ;                 | ;                 | Ĩ                                |
| SNS H            | -      | A/A     | ł               | N/N        | ł               | <b>1</b> /1  | ł               | N/N        | ;               | <b>N/N</b>      | ;               | N/A        | :               | 8-35-09      | 1.36+01         | 8°-35'8    | 1.3£+01       | 5424.0     | 1.46+03         | ;                 | 1                 | Ĕ.                               |
| ik, def          | •      | A/A     | :               | N/N        | ;               | N/A          | :               | N/N        | ;               | R/A             | 1               | N/A        | ;               | B.4E-10      | 3. IE +01       | 8.45-10    | 3.16+01       | 10560.0    | ;               | ;                 | 10+32             | NUN QI                           |
|                  | •      | 5.4E-10 | 3. IE+01        | N/N        | ٢               | N/A          | ;               | N/N        | ;               | <b>V</b> /N     | 1               | 5.4E-10    | 3. IE+01        | 5.46-10      | 3. 1E+01        | N/N        | 1             | 0.8084     | ;               | 1                 | - 10-30           | O NIN                            |
| HCCBF            | •      | 2.4E-10 | 3. 16+01        | N/A        | ١               | N/N          | :               | N/N        | :               | R/N             | ;               | N/A        | 1               | 2.46-10      | 3. IE+01        | N/A        | ł             | 1843.2     | ;               | 1                 | 10-31             | O NIM                            |
| ¥CO#             | •      | 2.6E-10 | 3. IE+01        | <b>N/A</b> | ł               | N/A          | ;               | N/A        | ł               | W/W             | ;               | 2.4E-10    | 3. IE+01        | 2.46-10      | 3. IE+01        | N/A        | 1             | 3484.4     | ;               | :                 | 10-39.            | NIN O                            |
| HCKGF            | •      | R/A     | ;               | N/N        | ł               | N/N          | :               | N/N        | ;               | V.              | ł               | A/A        | ;               | 6.96-10      | 3. IE+01        | N/N        | :             | 12000.0    | 1               | ;                 | 56+02             | NIN O                            |
| HCKHE            | •      | 6.9E-10 | 3. 1E+01        | 6.95-10    | 3. IE+01        | W/W          | ;               | N/A        | ;               | <b>6.9</b> E-10 | 3. IE+01        | N/N        | ;               | 6.9E-10      | 3. IE+01        | 61-36-6    | 3, iE+01      | 13600.0    | ;               | -                 | 10+35             | NIN O                            |

ALAL THE PERSON AND ALLER AND ALLER AND ALLER AND ALLER AND ALLER AND ALLER AND ALLER AND ALLER ALLER ALLER ALL

22



F





S ate Files HOLNEDST.HK1

### NAMMA, ING. ACCINENTS - NEGLONAL, PODCEDENG, OPTION - PER PALLET

|                  |            |           |            |        | Accide       | at Frequenci | I           | Ruge Facts | 5          |             |            |                 |               |           |                   |               | -            | Ĩ        | i <b>-</b> 14- i |             | -             |
|------------------|------------|-----------|------------|--------|--------------|--------------|-------------|------------|------------|-------------|------------|-----------------|---------------|-----------|-------------------|---------------|--------------|----------|------------------|-------------|---------------|
| 201- 00-<br>1919 | 12 8<br>12 |           | <b>S H</b> |        |              |              |             |            | ž          | Read in the |            | NAME<br>For the | TEN           |           |                   |               | AND THE REAL |          |                  |             |               |
|                  |            |           |            |        |              |              | !           |            | !          |             |            |                 |               |           |                   |               |              |          |                  |             |               |
|                  | -          | E-10 3.1E | VA II      | :      |              | I            | - 24-4      | • 3.IE+01  | N/N        | 1           | M          | ŧ               | 6. YE-10      | 3. IE+01  | 7/1               | :             | 1200.0       | 1        | :                | 4. IE +01   | 10 MEM        |
| NCINF            | 5.3        | E-10 3.1E | V/II III   | ł      |              | ł            |             | 1          | 5. X-10    | 3. IE+01    | <b>V/8</b> | 1               | 5. 36-10      | 3. 16+01  | 5. X-10           | 3. IE+01      | 4534.0       | 1        | ;                | 2.46-01     | III OI        |
| NC PGF           | 9 1.2      | E-10 3.1E | V/II IO    | ;      | W/N          | 1            |             | :          | 5          | ;           | N/N        | ;               | 1.26-10       | 3. IE+01  | 1.25-10           | 3. IE+01      | 3120.0       | ł        | :                | 6-X-9       | II II         |
| NCHE             | 9 1.2      | E-10 3.1E | V/II III   | 1      | -1.4         | H 3.1Emi     |             | t          | N          | 1           | 1.25-10    | 3.16+01         | 1.26-10       | 3.16+01   | A/N               | :             | SHE.0        | ı        | 1                | 5. 11-11    | 10 BII        |
| NCPVF            | 9 1.2      | E-10 3.1E | 41 IVA     | .1     | -1.7         | 10 3. IE+01  | A/M         | 1          | 5          | 1           | N/N        | 1               | 1.26-10       | 3. IE+01  | 1.76-10           | 3. IE+01      |              | 1        | ;                | 1.15-11     | III II        |
| NC OBF           | 1.2        | E-10 3.16 | AVII 19+   | 1      | - 77 - 1     | 10 J. 16+01  | 8/8         | ł          | N/N        | ł           | <b>A/N</b> | 1               | 1.75-10       | 3. IE+01  | 1.76-10           | 3. IE+01      | 148.0        | I        | :                | <b>8</b> +₩ | IIN OI        |
| HCBVF            | •          | :         | R/A        | 1      | A/A          | ;            | W/W         | :          | W/W        | 1           | N/N        | ł               | 1.2.10        | 3. IE+01  | 1.26-10           | 3. IE+01      | 1.00 V       | ;        | :                | 3.6.41      | II UII        |
| <b>ACREF</b>     | 9 5.4      | E-10 3.1E | W/H IO-    | 1      | 5.45         | 10 3.16-01   | A/A         | 1          | 5.46-10    | 3. JE+01    | N/N        | :               | 5.46-10       | 3. IE+01  | 5.46-10           | 3.16+01       | 23M.1        | 1        | 1                | 1.15+44     | IL NI         |
| NCRVF            | 9 5.6      | T-10 3.1E | ALL BA     | :      | 5.66-        | 10 J.IE+01   | N.N         | :          | 3.46-10    | 3.16+01     | N/N        | 1               | 5.46-10       | 3. 1E+01  | 5.46-10           | 3. IE+01      | 244.0        | 1        | ;                | 2.26-01     |               |
| HCSWE            | ) iii (    |           |            | 1      |              | 1            | W/W         | :          | 2          | I           | W/W        | ł               | 2.76-10       | 3. IE+01  | 2.76-10           | 3.16+01       | 5424.0       | t        | :                | 3.46.41     |               |
| NCINES           | 10 11/     | •         | W.M        | :      | 5            | ;            | N/N         | 1          | <b>V</b> N | ł           | N/N        | ;               | 1.16-00       | 1. 36+01  | 1.16-00           | 1.32.01       | India.a      | 2. Z +02 | :                | ;           | 1             |
| <b>NCIMS</b>     | 10 4.9     | E-M 1.X   | A/H IM     | ;      | VN .         | 1            | N/N         | 1          | N/N        | 1           | 4-36-04    | 1. 36+01        | 4°-34-9       | 1. X +01  | N/N               | :             | 1. M. I      | 6- W -9  | :                | 1           | Ē             |
| NCC65            | 10 3.4     | E-M 1.3E  | AVII ION   | :      | W/W          | 1            | N/N         | 1          | N/N        | ł           | N/N        | ł               | 3. # -0       | 1. X +01  | N/N               | 1             | 1043.2       | 1.4:40   | :                | 1           | 2             |
| NCO45            | 10 3.4     | E-00 1.X  | -01 N/A    | 1      | VIII .       | :            | 8/8         | :          | W/H        | 1           | 3.45-09    | 1. 2            | 3.45-09       | 1. X +01  | W/W               | ł             |              | 3.26.40  | ł                | 1           | Ē             |
| NCXES            | 10 N/      | •         | AVN .      | :      |              | :            | <b>V</b>    | :          | N/I        | ł           | N/N        | 1               | 0. K01        | 1.3£+01   | N/N               | 1             | 12000.0      | 1.56+03  | ;                | :           | Ĩ             |
| NCINS N          | 10 1.6     | N-1 M-3   | 11 B.B.    | 371 64 | AVII IO-     | I            | A/A         | ı          | 8. M19     | 100 H       | Y.         | :               | <u>е</u><br>Ж | 1. 26 441 | 8. M-9            | 1. X +        | 12440.0      | 1.76+45  | 1                | 1           | 2             |
| NCIVS            | 10 8.8     | E-0 1.X   | V/N 10+    | 1      |              | :            |             | N 1.5441   | N/N        | :           |            | ł               | 8. KE-01      | 1.32+01   | A/A               | 1             | 1200.0       | 1.46.45  | ;                | :           | Ĩ             |
| <b>INCINE</b>    | 10 4.8     | X'I 4-3   | A/H 10+    | 1      | N/N          | 1            | <b>V</b> /# | :          | 1-H-H      | 1. 36 +01   | N/N        | ł               | 4. K9         | 1, 35 +01 | 6. Ki - M         | <b>一字第</b> "二 | 4534.0       | 1.65-01  | ł                | t           | <b>E</b>      |
| NC 422           | 10 1.4     | E-09 1.X  | AN IN      | ;      | N/N          | :            |             | 1          | <b>N/</b>  | ł           | 2          | :               | 1.46-09       | 1. 36 +01 | 1.46-91           | 1.16+01       | 3120.0       | ₽-W-9    | :                | :           | Ĩ             |
| NCM5             | 1.1        | 31 6-3    | AVII IO-   | 1      | -146-        | M 1.XH       | M           | ł          | N/N        | 1           | 1.45-09    | I. X 🍋          | 1.65-09       | 1. X +01  | <b>1</b> /1       | 1             | 5414.0       | 1.25.01  | ;                | :           | 2             |
| REMS             | 11 01      | E-01 1.X  | A/H 10+    | •      | - <b>1.6</b> | M 1.EH       | A/A         | :          | N/N        | :           | R/A        | ;               | 1.45-09       | 1.32.401  | 1.6-9             | 1. X + 1      |              | 4.05+00  | :                | :           | Ē             |
|                  | 10 1.4     | E-M 1.X   | AVII 104   | 1      |              | M 1. K-1     | R/A         | 1          | <b>N/N</b> | 1           | N.         | I               | 1.46-01       | 1. X + 1  | \$<br>#<br>#<br>1 | 1.14.41       |              | 1.14.41  | ł                | 1           | Ĩ             |
| NCINS            | 10 11/     | -         | A/A        | ;      | M            | :            | A/H         | ;          | N/N        | :           | N/N        | :               | 1.66-09       | 1. X +01  | 1.65-19           | 1.14.40       | <b>1</b>     | 1.94.91  | :                | ;           | Ĩ             |
|                  | 10 7.2     | H.I. 10-3 | A/H 104    | 1      | 7.8-         | M 1.X.401    | N/N         | ł          | 7.25-01    | I.¥.4]      | N/A        | 1               | 7.75-94       | 1.36401   | 7.25-09           | I. X tol      | 241.0        | 1.16+01  | :                | :           | Ē             |
| HCINS            | 10 7.2     | 3.1 10-3  | V/N 10+    | 1      | -1.1         | P 1. X+1     | <b>V</b>    | 1          | 1.25-49    | 1. 26 401   | ٩/٩        | :               | 7.26-09       | 1. 3E+01  | 7. X-1            | 1. 36 +01     | 2404.4       | 1.66.401 | ;                | :           | Ĩ             |
| HESAS            | 10 11/     | •         | W/W        | ;      | -            | :            | N/N         | 1          | W/W        | 1           | N/N        | :               | N-W-1         | <b>1</b>  | \$-W.1            | 1. 14 401     | 5424.0       | 1.46.45  | ł                | 1           | <u>.</u>      |
| X MC             | 11 5.8     | E-09 2.46 | +01 N/N    | ;      | N/N          | 1            | A M         | ;          | N/N        | 1           | 2.96-09    | 2.66+01         | 5.86-09       | 2.46+01   | N/N               | ;             |              | 1        | 6- W - O         | 1. X - 1    | Ē             |
| MCCBC            | 11 2.9     | E-00 2.4E | WA IN      | ł      |              | :            |             | I          | M          | :           | R/A        | I               | 2.96-09       | 2.66+01   | N/N               | I             | T.           | 1        | 1.15+00          | 1.45-01     | <b>1</b>      |
| NCON             | 11 2.9     | E-09 2.46 | A/H 10+    | ;      | <b>W</b>     | 1            | A/A         | ;          | N/N        | :           | 1.46-09    | 2.46+01         | 2.96-09       | 2.46+01   | <b>N</b> /N       | :             | 74.8         | :        | 3.26+00          | ₩.8         | <b>E</b><br>- |

SSEED S

Page à 17-Aug-07 Files NOLNESSI.NKL HANDLING ACCIDENTS - REGIONAL PROCESSING OFTICH - PER PALLET

) )

|                |         |                     |                 |             |                 | Accident       | Frequenci         | I           | ange Facto      | 5                |               |             |               |                |               |                  |          | 4     | igent Avai | i ale ad         | Released             |                  |
|----------------|---------|---------------------|-----------------|-------------|-----------------|----------------|-------------------|-------------|-----------------|------------------|---------------|-------------|---------------|----------------|---------------|------------------|----------|-------|------------|------------------|----------------------|------------------|
| SCEN- 0        | ۴.<br>۳ | NIN DE              | RANGE<br>Factor | ¥ į         | RANKE<br>Factor | LINA           | RANKE<br>Facility |             | RANGE<br>Factor |                  | ANNE<br>ACTOM | Puen P      | ANNE<br>ACTOR | FREG           | AMME<br>ACTOR |                  | A CLOR   | MENT  |            | L DG<br>DELANTED |                      | MATION<br>Antion |
|                | 1       |                     |                 |             |                 |                |                   |             |                 |                  |               |             |               |                |               |                  |          |       |            |                  |                      |                  |
| HCIIIVE        | =       | 4. 36-01            | 2.66+01         | R/N         | :               | A/A            | :                 | W/W         | ;               | 2.26-99          | 2.66+01       | N/N         | 1             | 4.35-09        | 2.46+01       | 2.26-09          | 2.46-01  | 378.0 | !          | 3.26+01          | S-12.1               | Ē                |
| HCPSC          | Ξ       | 9.46-10             | 2.66+01         | R/A         | ;               | A/A            | 1                 | N/N         | 1               | V/               | 1             | W/W         | 1             | 9.46-10        | 2.46+01       | 4.86-10          | 2.46-01  | 52.0  | :          | 6.56+00          | 1.16-01              | Ħ                |
| <b>N</b><br>N  | Ξ       | 1.46-10             | 2.66+01         | N/N         | ;               | 4.66-10        | 2.46+01           | N/N         | ;               | Z                | ;             | 4. JE-10    | 2.46+01       | 9.46-10        | 2.4£+01       | N/N              | 1        | 93.4  | ;          | 1.25+01          | N-30'                | Ĩ                |
| HCPVC<br>HCPVC | 3       | 9.46-10             | 2.46+01         | N/A         | ١               | 4.K-10         | 2.46+01           | A/A         | ı               | M                | ł             | W/W         | :             | 9.46-10        | 2.45+01       | 4. <b>IE</b> -10 | 2.46+01  |       | :          | 4.05+00          | 1. 05-15             | =                |
| HC BIGC        | =       | 7.26-10             | 2.46+01         | N/N         | <b>`</b> ¦      | 3.46-10        | 2.46+01           | N/N         | ;               | , va             | ł             | A/N         | ;             | 7.76-10        | 2.46+01       | 3.46-10          | 2.46+01  | 87.0  | 1          | 1.5£+01          | 10-21.1              | Ŧ                |
| NCBVC          | Ξ       | <b>N/A</b>          | ;               | N/N         | :               | <b>N</b> N     | 1                 | R/N         | ;               | N/N              | :             | N/N         | :             | 7.26-10        | 2.66+01       | 3.46-10          | 2.46+01  | 0.70  | I          | 10-35-1          | 1. MQL               | ۹                |
| <b>NCREC</b>   | Ξ       | 1.8-1               | 2.46+01         | N/N         | ١               | 9.06-10        | 2.46-01           | N/N         | ;               | 9.06-10          | 2.46+01       | N/N         | :             | 1.16-09        | 2.45+01       | 9.06-10          | 2.46+01  | 140.5 | 1          | 2. LE+01         | 1. 16+00             | <b>4</b>         |
| HCINC          | =       | 1.05-09             | 2.4E+01         | <b>N/N</b>  | ł               | 9, 06-10       | 2.46+01           | R/A         | 1               | 9.06-10          | 2.46+01       | N/A         | ł             | 1.06-09        | 2.46+01       | 9.06-10          | 2.4E+01  | 150.0 | ;          | 2.06+01          | 14-18-15<br>15       | <b>H</b>         |
| <b>KOKC</b>    | 2       | 4. IE-10            | 2. LE +01       | N/N         | :               | A/N            | :                 | N/N         | ;               | A/A              | 1             | 2. IE-10    | 2.46+01       | 4.16-10        | 2.46+01       | N/N              | :        | 200.0 | 1          | 4.06+00          | 1. H-01              | 5                |
| NCORE          | 2       | 2.16-10             | 2.46+01         | N/N         | ł               | <b>N/N</b>     | ;                 | N/N         | 1               | ş                | ł             | N.          | 1             | 2. IE-10       | 2.4E+01       | N/N              | ;        | 1.2   | ł          | 1.46+00          | 16-31-1              |                  |
|                | 2       | 2. IE-10            | 2.46+01         | <b>N/A</b>  | 1               | N/N            | :                 | A/N         | :               | N/N              | ł             | 1.96-10     | 2.6EM1        | 2.16-10        | 2.16+01       | N/N              | ł        | 74.8  | ;          | 3.ZE+00          | 1. M                 | <b>E</b> -       |
| <b>NCINC</b>   | 2       | 3. IE-10            | 2.46+01         | <b>V</b>    | ۱               | N.N            | ł                 | N/N         | ;               | 01-X-1           | 2.46+01       | N/R         | ł             | 3.16-10        | 2.4E+01       | 1.56-10          | 2.4E+01  | 378.0 | 1          | 3.25-01          | 1.25-55              | 1                |
| NCPSC          | 2       | 11-36-9             | 2.4E+01         | N/N         | ł               | R/N            | ł                 | N/N         | 1               | A/N              | :             | N/N         | :             | 6.96-11        | 2.46+01       | 3.46-11          | 2.46+01  | 52.0  | t          | 4. SE +00        | 1.15-01              | <b>E</b>         |
| <b>KCMC</b>    | 2       | 4.16-11             | 2. JE +01       | W/W         | 1               | 3.46-11        | 2.46.441          | M           | 1               | A/A              | 1             | 3.46-11     | 2.46+01       | 11-31-1        | 2.46+01       | N/N              | ;        | 13.4  | I          | 1.26+01          | 1.8.4                | 9                |
| KCPVC          | 2       | 4.16-11             | 2.46+01         | N/N         | ١               | 3.46-11        | 2.46+01           | N/N         | 1               | R/A              | ;             | R/A         | 1             | <b>6.9E-11</b> | 2.66+01       | 3.4E-11          | 2.46+01  | ŧ     | 1          | 4.06+00          | 1. M-45              | <b>9</b>         |
|                | 2       | 5.26-11             | 2.46+01         | W/W         | 1               | 2.46-11        | 2,46+01           | <b>W</b> /W | ł               | N/N              | ł             | N/N         | I             | 11-3C.2        | 2.16+01       | 2.16-11          | 2.66+41  | 87.0  | 1          | 1.56-01          | 1.7-11               | 9<br>1           |
| KONC           | 2       | N/N                 | 1               | N/A         | ;               | W/R            | ł                 | M           | ł               | N/N              | 1             | R/A         | 1             | 5.26-11        | 2.4E+01       | 2.46-11          | 2.4E+01  | 87.0  | ;          | 1.56+01          | #<br>₩ 1             | 9                |
| <b>FCRE</b>    | 2       | 1.3.10              | 2.46+01         | N/N         | 1               | 1-2.9          | 2.46+01           | T.          | ;               | 11-35-11         | 2.4E+01       | N/N         | ł             | 1.36-10        | 2. JE +01     | 11-35-9          | 2.6E+01  | 140.5 | 1          | 2. IE+01         |                      |                  |
| RUME           | 2       | 1. 3%-10            | 2.46+01         | N/N         | :               | 1-3.1          | 2.46+01           | N/N         | ;               | 11-X-9           | 2.46+01       | N/N         | ;             | 1. JE-10       | 2.66+01       | 11-35-9          | 2.6E+01  | 120.0 | 1          | 2.0€+01          | 5. M - PL            | <b>9</b>         |
|                | 1       | <b>N/N</b>          | 1               | <b>4</b> /4 | ;               | N/N            | ł                 | N/N         | 1               | N/N              | ;             | N/A         | ł             | å.0E-13        | 2.46+01       | 6.06-13          | 2.46+01  | 440.0 | 1          | :                | 1. H-01              | Ĩ                |
| HCHRC<br>HCHRC | 17      | 4.0E-13             | 2.46+01         | 4/4         | 1               | <b>6.0E-13</b> | 2.46+01           | 8/8         | :               | 4. OE-13         | 2.46+01       | K/A         | !             | 4.0E-13        | 2.46+01       | 6.06-13          | 2.46+01  | 14.5  | ;          | ł                | 1.4.41               |                  |
| HCINC          | 1       | 4.06-13             | 2.46+01         | N/N         | 1               | 6. HE-13       | 2.46+01           | 5           | ;               | 6.0E-13 .        | 2.4E+01       | W/          | :             | 6.0E-13        | 2.4E+01       | 4.06-13          | 2.46+01  | 159.0 | ;          | 1                | 14-14-1<br>1-14-14-1 | Ĩ                |
| K DGC          | 2       | <b>V</b> / <b>I</b> | ;               | <b>V</b> /  | 1               | N.N            | 1                 | <b>W</b> /W | 1               |                  | :             | <b>4</b> /8 | :             | 1.26-12        | 2.46+01       | 1.26-12          | 2.4E+01  | 220.0 | ;          | :                | , <b>K</b> → I       | Ē                |
| NC/NGC         | 2       | 1.X-12              | 2.65.401        | <b>N</b> /N | ł               | 1.75-12        | 2.46+01           | W/W         | ł               | 1.76-12          | 2.4€+01       | Y.          | :             | 1.21-12        | 2.46+01       | 1.26-12          | 2.4E+01  | 19.7  | ;          | :                | 1.45-01              | Ē                |
| NCINE          | =       | 1.76-12             | 2.4E+01         | N/N         | ;               | 1.26-12        | 2.6€+0}           | V.          | 1               | 1.X-12           | 2.46+01       | #/#         | :             | 1.*-12         | 2.4E+01       | 1.7.12           | 2.46+01  | 10.0  | ;          | 1                | 1-2-1                | 5                |
| NC NGC         | 5       | <b>4</b> /4         | :               | 21          | ;               | N/A            | :                 |             | ł               | N/N              | ł             | N/N         | 1             | 2              | 3.16+01       | 2.4E-15          | 3.16+01  | 440.0 | ;          | ;                | 1.16-01              | Ĩ                |
| NC NGC         | =       | 2.46-11             | 3. IE+01        | N/N         | ;               | 2.46-14        | 3.16+01           | N/N         | ;               | 2. <i>H</i> E-14 | 3. IE+01      | W/W         | ;             | 2.44-74        | 3. IE+01      | 2.4E-14          | 3.16+01  | 160.5 | 1          | ;                |                      | £                |
| KUNC           | •       | 2,46-11             | 3. IE+01        | N/N         | 1               | 2.46-11        | 3. IE+01          | 2           | ł               | 2.4E-14          | 3.16+01       | R/3         | 1             | 2.46-14        | 10+31.1       | 2.46-14          | 3. JE+01 | 150.0 | 1          | :                | 19-12-1              | Ĩ                |
| HC DOC         | 2       | 8/8                 | :               | 8/8         | ;               | 11/1           | ;                 | N/N         | ;               | N/N              | :             | N.N         | ;             | 3.4E-10        | 10+31 °°      | 3.4E-10          | 3. 1E+01 | 220.0 | :          | :                |                      | Ē                |

KANDA HERERA DOCONG DOCONALA

~~~







NUMERING ACCIDENTS - REGIONAL PROCESSING OFTICH - PER PALLET

| | | | | | | Accident | Frequenci | Ī | tanga Facta | 2 | | | | | | | | | ĭ | ilate an | i Reinese | _ |
|--------------|-----|----------|-------------------|---------------------|-----------------------|-------------|-----------|-------------|-------------|-------------|----------|-------------|---------|-----------------|----------|-------------|----------|------------|------------------|--------------------|-----------|----------|
| | i - | | Rental Ficting | ŝ | MARK FACING | | | 32 | | 22 | | Į | | | TACTOR N | | | MEIT | | L 16 Netrinated | | 1141 |
| | | | | | | | | | | | | | | | | | | | | | | |
| | 2 | 3.16-11 | 1 3.IE+01 | N. | : | 3. IE-10 | 3. IE+01 | 5 | ; | 3.16-10 | 3.1641 | 2 | ł | 3. 16-10 | 3. 16+01 | 3. IE-10 | 3. IE+01 | 10.7 | ł | ; | ₩÷ | ŝ |
| HCRVC | 2 | 3. IE-1 | 1 3.1E+01 | NN. | ł | 3.16-10 | 3.16+01 | N | ł | 3.16-18 | 3. iE+41 | W/W | 1 | 3.16-10 | 3. IE+41 | 3.16-10 | 3.16+01 | 10.0 | ŧ | : | 1.701 | |
| | 2 | 1.75-11 | 2.46+01 | A/A | 1 | W /# | ł | A/A | : | N/N | : | V. | : | 1. K-11 | 2.46.41 | W/R | : | 20.0 | 3.66-01 | 4.65.40 | : | E |
| NCOL | 2 | 1. %-11 | 2.46+01 | W/W | 1 | N | ł | N | ł | M | 1 | M | 1 | 1. %.1 | 2. (E+0) | 5 | 1 | , a | 0-W-0 | 1.46400 | : | Ĩ |
| NCDI | 2 | 1.3.4 | 2.46+01 | W/W | ۰, | 8/8 | : | N/N | 1 | Ň | 1 | 8/8 | : | 4.X-II | 2.66+01 | W/W | ; | 74.1 | 1.16-01 | 3.26+00 | 1 | 2 |
| NEINC | 2 | 11-35.9 | 2.46401 | N/N | 1 | N/N | : | N/N | ł | 2 | 1 | | : | 11-35.4 | 2.46-01 | W/W | ł | 1 . | 3. 26+02 | 3.26+01 | 1 | |
| NCPBC | 2 | 1.46-11 | 2.46+01 | #/# | 1 | N/N | : | A/N | : | N/N | 1 | VI | : | 11-34-11 | 2.46+01 | N/A | : | 22.0 | 3.26+01 | 6.56+00 | : | |
| JICH | 2 | 1.46-11 | 2.46+01 | 5 | ; | N/N | ; | N/N | 1 | 5 | 1 | R/A | ł | 11-34-11 | 2.4641 | 8/8 | ł | 13.6 | 5. 16+41 | 1.25+01 | 1 | Ĩ |
| NCM. | 2 | 11-39-11 | 2.46+01 | N/N | : | N/N | : | N/N | 1 | N/N | : | N.N | ; | 11-34-11 | 2.46+01 | W /W | 1 | ÷. | 3.0€+01 | 4.161-00 | : | Ĩ |
| | 2 | 1.16-11 | 2.46.41 | 42 | : | 1 /1 | ; | N/N | : | W/W | 1 | 2 | ł | 1.16-11 | 2.46+01 | N | 1 | 87.0 | 7. 16+01 | 1.55.01 | ł | Ĩ |
| NC INC | 2 | V | 1 | V. | 1 | N/N | : | N/N | ; | N /N | ; | N/N | : | 1.15-11 | 2.46+01 | N/N | ; | 0.70 | 7. 2441 | 1.36.41 | : | E |
| | 22 | 2.76-11 | 2.46+01 | N/N | 1 | A/A | : | N/N | ł | N/N | : | 2 | ŧ | 2. T -11 | 2.46+01 | N/N | ł | 14.5 | 1.46+02 | 2. IE+01 | I | Ĩ |
| NCIN | z | 2. 76-11 | 2.46+01 | V | ; | N/N | 1 | N/N | : | 8/8 | 1 | W /W | : | 2.76-11 | 2.46+01 | 2 | 1 | 126.0 | 1. X +02 | 2.06+01 | : | Ē |
| NC INC | 23 | 2.3E-11 | 2.46.41 | A N | ; | 1 /1 | 1 | N/N | ; | W | : | 2.36-11 | 2.4E+01 | 2.X-11 | 2.46+01 | N/N | : | 1152.0 | 3.06+01 | 6.0E+00 | 1 | Ĩ |
| NCCRC | 5 | 3.56-11 | 2.46.401 | V | ; | N/N | : | N/N | ; | N/N | ł | N/N | : | 3.36-11 | 2.46+01 | V. | ; | 110.1 | 8. IE + I | 1.46+00 | ł | Ĩ |
| NCOK | 2 | 3.56-11 | 2.46+01 | 5 | ; | W/W | ł | N/N | 1 | 1 /1 | ł | 3.56-11 | 2.46+01 | 3.56-11 | 2.4E+01 | N/N | : | 9-124 | 1.16-01 | 3.26+00 | 1 | E |
| KCRVC | 23 | 1. 3.1 | 2.46+01 | N/N | 1 | N/N | ; | N/N | : | 11-32-11 | 2.6E+01 | V /I | ł | 1.36-11 | 2.46+01 | 11-37-11 | 2 46401 | 1134.0 | 3.5€+02 | 3.25+01 | ; | Ē |
| NCME | 2 | 11-34-1 | 2.46+01 | W / W | 1 | N/A | ł | N/N | : | A/A | 1 | N/N | 1 | 1.46-11 | 2.46+01 | 1.46-11 | 2.46+01 | 786.0 | 3. 26+01 | 6.35+00 | ł | Ĩ |
| ж. | 2 | 1.46-11 | 2.65+01 | 4 /1 | : | 1.46-11 | 2.46+01 | R/A | : | N/N | ; | 1.46-11 | 2.4E+01 | 1.46-11 | 2.46+01 | e/a | : | IH. | 5.16+01 | 10+32-1 | ; | Ĩ |
| HCPVC | 2 | 1.46-11 | 2.46+01 | V | ۱ | 11-31-1 | 2.46+01 | N/N | ł | W/W | ł | N/A | : | 1.46-11 | 2.46+01 | 11-34-11 | 2.46+01 | 120.0 | 3.46+01 | 6-0E+00 | 1 | |
| | 2 | 7.26-12 | 2.4641 | W/W | : | 7.26-12 | 2.66.41 | N/N | 1 | ž | ł | V.I | ŧ | 7.26-12 | 2.46+01 | 7.25-12 | 2.4E+01 | 870.0 | 7. 35 +01 | 1.56-01 | 1 | Ĩ |
| HCENC | 2 | N/N | 1 | Š | ł | N/N | 1 | N/N | ł | . WA | 1 | A/A | 1 | 7.26-12 | 2.46+01 | 7.25-12 | 2.46+01 | 870.0 | 7. 36 +01 | 1.56+01 | : | Ē |
| | 23 | 7.26-12 | 2.46+01 | N/N | ; | 7.25-12 | 2.46+01 | N/N | : | 7.25-12 | 2.4£+01 | N/N | ; | 7.26-12 | 2.46+01 | 7.26-12 | 2.46+01 | M2.0 | 1.4€+02 | 2.15+01 | : | Ē |
| HCINC | 2 | 7.76-12 | 2.46+01 | R/A | 1 | 7.26-12 | 2.6E+01 | W /W | : | 7.26-12 | 2.4€+01 | N/N | : | 7.26-12 | 2.46+01 | 7.26-12 | 2.6£+01 | 0.044 | 1.X+02 | 2.0€+01 | : | Ē |
| ¥194 | 2 | 11-37-1 | 2.46+01 | V /8 | ; | N /N | : | N/N | I | N | ł | ž | 1 | 4.21-11 | 2.46+01 | M | : | 20.0 | 3.06+01 | <u>4. 05 +00</u> | ; | Ē |
| HCCBC | 3 | 3.16-11 | 2.46+01 | V / | 1 | 5 | 1 | Ň | 1 | | ł | N/N | 1 | 3. IE - 11 | 2.4€+01 | 1 /1 | 1 | | 8.05+00 | 1.45+00 | ; | 1 |
| ¥CH | 5 | 3. IE-11 | 2.16+01 | N | : | N/N | : | N/N | 1 | N/N | ł | N/N | : | 3.16-11 | 2.4€+01 | N/N | t | 74.8 | 1.46+01 | 3.25.00 | ł | Ĩ |
| NCINC | 2 | 11-39-1 | 2.46+01 | 5 | I | I V | : | N/N | : | NN N | ł | N/N | 1 | 4.46-11 | 2.46+01 | N/N | ; | 378.0 | 3.56+02 | 3.26+01 | : | Ĩ |
| NCPGC | 5 | 1.06-11 | 2. 4E +01 | N/A | ; | N/N | : | | I | N/ | ; | N/N | 1 | 11-30-11 | 2.4E+01 | N/N | : | 52.0 | 3.26+01 | 6.35+00 | ł | Ĩ |

Pape 1 Files NOLNEGSL, MCI 17-Aug-07

HAMBLING ACTINENTS - REGIONAL PROCESSING OPTION - PER PALLET

| | | | | | | Accident | Frequenci | 1 | lange Facti | ĩ | | | | | | | | - | lyent Ave | ilable an | Relevan | |
|----------------|----------|-----------|---------|---------------------|-------------|-------------|-------------|-----|--------------|-------------|-------------|---------------------|---------------|----------|-------------|----------|-------------|-----------|-----------|------------|---------|----------|
| | 6. 10 | ł | Junic | ş | BANG | 3 | JANK | 3 | Janua | Ĩ | Junu | ¥2 | Jan vu | TEND | JANN | W | MANO | NGENT | 9 | s | 8 | DURAT 20 |
| 24 | | FINED | FACTOR | FIEB | FACTOR | FRED | FACTOR | | FACTOR | FINE | FACTON | FINER | FINC TOR | FINE | FACTOR | FINER | FACTOR | ANAILANLE | B'ILLED | IE TOWATED | ENTTER | |
| - | | | | | | | | | | | | | | | | | | | | | | |
| NCNC NCNC | ~ | 11-30-11 | 2.46+01 | N/N | ł | A/A | ; | N/N | ; | VI | ł | A/N | 1 | 1.06-11 | 2.46+01 | 8/8 | 1 | 93.4 | 5.16+01 | 10+32.1 | ; | H. |
| HEME | * | 11-30-11 | 2.46+01 | N/N | 1 | N/N | : | N/N | 1 | N /N | 1 | A/N | ; | 1.06-11 | 2.66+01 | N/A | 1 | 48.0 | 3, 66 +01 | 4.0€+00 | ł | |
| NC BEC | 24 | 1.76-12 | 2.46+01 | A/A | ; | 8/8 | 1 | 1/1 | 1 | N/N | : | A/A | : | 7.78-12 | 2. JE +01 | N/N | ł | 0.70 | 7. 16+01 | 10+35-1 | ; | Ē |
| HCDVC | 2 | N/N | 1 | N/N | I | W/W | 1 | - | ł | 5 | : | A/N | ł | 7.76-12 | 2.46+01 | N/N | ł | 07.0 | 7. 16+01 | 1.56+01 | 1 | 1 |
| NCNSC | 71 | 11-36-11 | 2.46+01 | A/A | , ! | W/W | : | N/N | 1 | . WA | ; | V / | : | 11-34-11 | 2.46+01 | A/A | : | 160.5 | 1.46+02 | 2.16-01 | 1 | Ĩ |
| ICINC | 5 | 1.%-11 | 2.46+01 | N/A | : | X/N | ł | N/N | 1 | N/N | ł | W/W | 1 | 1.96-11 | 2.46+01 | N/A | 1 | 150.0 | 1. 1.402 | 2.0E+01 | : | Ĩ |
| NCNC NCNC | R | 11-31-11 | 2.46+01 | R/N | : | R/A | : | N/N | ; | | : | 11-31.1 | 2.46+01 | 1.7E-11 | 2.46+01 | N/N | 1 | 1152.0 | 3.06+01 | 4.06+00 | ; | Ξ. |
| NCCR | 2 | 2.56-11 | 2.46+01 | ¥/# | 1 | N/N | ł | N/N | : | | ł | 8/8 | ł | 2.56-11 | 2.6€+01 | N/N | 1 | 110.1 | 0, 0E +00 | 1.45+00 | 1 | Ĩ |
| HCCHC HCCHC | 2 | 2.96-11 | 2.46+01 | N/N | 1 | W/W | 1 | N/N | : | N/N | : | 2.56-11 | 2.46.01 | 2.36-11 | 2.4E+01 | R/A | 1 | 1.154 | 1.4£+01 | 3.26+00 | ; | Ē |
| NCH C | ĸ | 9.36-12 | 2.66-41 | ¥/¥ | 1 | 4/H | ŧ | N/N | 1 | 9.XE-12 | 2.46+01 | 8/8 | I | 9.36-12 | 2.46+01 | 9.3E-12 | 2.46+01 | 1134.0 | 3.56+42 | 3.26+01 | 1 | Ĩ |
| NC PGC | 2 | 11-30-1 | 2.46+01 | N/N | ; | N/N | : | N/ | : | A/A | : | N/A | : | 1.06-11 | 2.46+01 | 11-30-1 | 2.4€+01 | 700.0 | 3.26+01 | 6.5E+00 | ; | Ē |
| NCMC | 2 | 11-30-1 | 2.46+01 | A/M | : | 11-30-1 | 2.46+01 | R/A | ; | R/H | 1 | 1.4-11 | 2.66+01 | 130.1 | 2.66+01 | N/N | ; | 1404.0 | 5. IE +01 | 1.26441 | 1 | Ħ |
| JACON | 2 | 11-30-11 | 2.46+01 | A/A | 1 | 130.1 | 2.46+01 | N/N | : | A /A | : | V | 1 | 1.06-11 | 2.6€+01 | 11-30-11 | 2.46+01 | 720.0 | 3.06+01 | 4.0€+00 | : | Ē |
| HC86C | 2 | 5.X-12 | 2.46+01 | A/A | ; | 5.26-12 | 2.46+01 | N/N | ł | 1 /1 | 1 | N/A | : | 3.26-12 | 2.46+01 | 5.26-12 | 2.66+01 | 879.9 | 1.35+01 | 10+35-1 | ł | Ţ |
| HCBNC | R | A/H | ; | V | 1 | N/A | ł | N/N | ł | V /8 | ł | #/# | I | 5.26-12 | 2.4E+01 | 5.26-12 | 2.4£+01 | 870.0 | 7.36+01 | 1.5€+01 | ł | Ē |
| HC166C | 2 | 5.26-12 | 2.4E+01 | N/N | ł | 5.76-12 | 2.46+01 | #/# | 1 | 5.26-12 | 2.46+01 | N/A | 1 | 5.26-12 | 2.66+01 | 5.X-12 | 2.66+01 | M2.0 | 1.4€+02 | 2.16+01 | 1 | 1 |
| KCINC | 2 | 5.21-12 | 2.46+01 | N/N | ; | 3.X-12 | 2.46+01 | N/N | : | 5.26-12 | 2.66+01 | N/A | : | 3.26-12 | 2.46+01 | 5.26-12 | 2.65+01 | 10.041 | 1. 36 +02 | 2.0€+01 | : | Ĩ |
| NC NOL | 2 | A/A | 1 | N/N | ł | N/N | ł | N/A | 1 | V | 1 | N/N | ł | 0.05+00 | ; | N/A | 1 | 440.4 | ; | 1 | 1 | ł |
| | 2 | 0,05+00 | ; | N/N | 1 | W/W | ł | N.N | ; | A VA | ł | N/A | ; | 0.0E+00 | ; | W/W | ; | 200.0 | t | ; | 1 | ł |
| HCCBC | 2 | 0, 05 +00 | 1 | A/A | 1 | N/N | ł | N/N | : | | ; | A/A | 1 | 0.06+00 | ł | N/A | ; | 4. BY | ł | 1 | 1 | ł |
| HCCK | 7 | 0,0€+00 | : | N/N | ł | N/A | ł | N/N | ł | | ; | A / N | ł | 0.0E+00 | ł | N/A | : | 74.8 | ; | ; | ; | ł |
| HCKGL | 2 | 8/8 | ! | N/N | ł | W /W | ł | N/N | 1 | 1 /1 | 1 | K/N | ł | 0.0E+00 | ı | N/A | ; | 1500.0 | ł | 1 | t | ł |
| HCH. | * | 0,0E+00 | : | N/N | ł | 8/8 | ł | N/N | 1 | U /N | ; | N/N | 1 | 0.65+00 | ; | N/N | : | 1700.0 | : | ١ | : | ł |
| HCIVE | 2 | 0.05+00 | : | 1 /1 | : | N/N | ł | W/W | 1 | 2 | 1 | A/N | 1 | 0.0E+00 | ; | N/N | ; | 1600.0 | ł | 1 | 1 | ; |
| NCH | 38 | 0.05+00 | : | V / H | ; | W/W | ł | N/N | ł | N/N | I | N/N | 1 | 0.05+00 | ł | R/A | : | 378.0 | ; | ł | ł | ; |
| 3943H | * | 0.05+00 | : | | ŧ | N/N | ł | N/N | 1 | - | ł | N/A | 1 | 0.0E+00 | ١ | N/A | ł | 52.0 | 1 | ١ | ; | 1 |
| жж | 38 | 0.05+00 | ; | W/W | ; | W/W | ł | W/W | ; | W/W | : | R/N | 1 | 0.0E+00 | ١ | N/A | ŧ | 13.4 | : | : | ; | : |
| Ň | 2 | 0.65+04 | 1 | A /A | ł | N/A | ł | N/A | I | | 1 | A/A | 1 | 9.05+00 | ł | N/N | ł | | ł | ١ | 1 | 1 |
| HC INCC | 2 | 0.05+00 | 1 | N/N | ł | N/A | : | N/N | ; | N/A | 1 | R/A | : | 0.05+00 | ; | N/A | 1 | 87.0 | : | ; | 1 | : |











Filet MRLREGST.HKL 17-Aug-07 Page 9

NAMBALING ACCIDENTS - REGIONAL PROCESSING OPTION - PER PALLET

| | | | | | | Accident | Frequencia | Ĩ | unge Facti | F | | | | | | | | ¥ | i int | ilalle a | d Release | _ |
|---------------|----------|----------|---------------|-------------|--------|------------|-----------------|-------------|----------------|----------------|-----------------|---------------|-----------------|---------------|-----------------|------------|-----------------|--------|-----------|------------------|----------------|----------|
| AREO . | 2 2 | THE | ANNE ACTOR | s S | FICTOR | | Read Processing | | TANK FACTOR | 뤁녩 | INNEE FACTOR | PURA FIRED | RANGE FACTOR | TEN) Fried | RANGE FACTOR | UNION | RANKE FACTOR | MENT | | LDS JETOMATET | LDS BRITTED | TINK I |
| | | | •. | | | | | | | | | | Į | | | | | | | | | |
| HCBVC | 2 | N/N | 1 | 4 /8 | 1 | N/N | : | N/N | ł | 5 | : | N/N | : | 0.0£+00 | ł | N/N | ; | 00 | 1 | ; | 1 | ; |
| HC INGC | 36 | 0.0E+00 | : | N/N | : | N/N | ł | N/N | ١ | 8/R | ł | N/A | ; | 0.0E+00 | : | W/W | ; | 140.3 | 1 | ł | 1 | : |
| HERVE | 38 | 0.0€+00 | ; | N/A | : | N/N | ; | 4 /8 | 1 | W/W | : | N/A | 1 | 0.0€+00 | : | N/ | : | 154.0 | 1 | ; | : | ; |
| HESVE | 36 | N/A | ł | N/N | 1 | N/A | ł | V/I | 1 | N/N | : | N/N | t | 0.06+00 | : | N/N | : | 1354.0 | ŧ | : | ; | ł |
| HC16F | 27 | A/A | : | R/A | ١, | W/W | 1 | N/N | ; | M | : | N/N | 1 | 0,0£+00 | 1 | 0.0E+00 | 1 | 2640.0 | 1 | ; | : | : |
| FIDE | 21 | 0.0£+00 | ; | N/A | ; | N/N | 1 | N/N | 1 | 8/8 | ; | 0.0£+00 | 1 | 0.0€+00 | ł | N/A | : | 1152.0 | : | ; | 1 | 1 |
| NCCEC | 27 | 0.05+00 | ; | N/A | 1 | R/N | ŧ | 8/8 | : | N/N | ; | A/A | : | 0.0E+00 | : | 8/8 | : | 440.0 | : | ; | ł | : |
| NCONC | 27 | 0.0€+00 | : | N/N | ; | N/N | ; | A/A | 1 | N/N | 1 | 0.0€+00 | : | 0.0£+00 | ; | N/A | I | 1.124 | 1 | ; | 1 | ; |
| NCK6F | 27 | R/A | ; | N/N | ; | N/N | ; | A/N | ; | #/# | : | N/N | ; | 0.0€+00 | ; | R/A | 1 | 3000.0 | ł | ł | 1 | ł |
| NC/OF | 27 | 0.0E+00 | : | 0,00+30,0 | : | N/N | ł | A/H | 1 | 0.05400 | ł | W/W | : | 0.06+00 | ; | 9.06+90 | 1 | 2400.0 | ł | ł | ł | ł |
| HCKVF | 27 | 0.0E+00 | : | A/A | ł | N/N | ; | 0.05+00 | 1 | N/A | ; | N/A | : | 0.0E+00 | : | W/W | ; | 3200.0 | ; | ; | 1 | : |
| HEIRVE | 27 | 0.0£+00 | ł | N/N | ; | N/A | ł | N/N | ; | 0,06+00 | : | N/N | ł | 0.0E+00 | : | 0,05+00 | : | 1134.0 | ł | : | ł | ; |
| HCPGC | 23 | 0.0E+00 | ; | A/A | ; | N/N | : | N/N | ; | N/N | ; | | : | 0.05+00 | ł | 0.0€+00 | : | 780.0 | ł | : | ; | ; |
| KCPHC | 21 | 0.06+00 | ; | N/A | : | 0,0£+00 | ł | W/W | ; | 8/8 | 1 | 0.05+00 | 1 | 0.05+00 | ; | N/N | 1 | 1404.0 | ł | ; | ł | : |
| NCPVC | 27 | 0.0€+00 | ; | N/N | : | 0.06+00 | : | R/N | ; | N/N | ; | A/N | 1 | 0.0E+00 | 1 | 0.06+00 | ! | 720.0 | ; | : | : | 1 |
| NC OBC | 27 | 0.0£+00 | ł | N/N | : | 0.06+00 | 1 | N/N | : | R/A | ł | N/N | 1 | 0.0E+00 | : | 0.0£+00 | ł | 670.0 | ł | ; | ! | : |
| HCBVC | 23 | A/A | : | R/R | 1 | W/W | ; | N/N | : | N/N | ! | N/N | : | 0.0E+00 | : | 0.0€+00 | : | 870.0 | : | ; | : | ; |
| HCRGC | 23 | 0.0€+00 | ł | M/A | ; | 0.0E+00 | ł | R/A | ; | 0.06+00 | : | N/N | 1 | 0.06+00 | : | 0.0€+00 | 1 | 442.0 | : | : | ; | ł |
| HCRVC | 11 | 0.0E+00 | : | R/A | 1 | 0.0E+00 | ; | A/1 | : | 0.0E+00 | : | N/N | : | 0.0E+00 | : | 0.05+00 | : | 600.0 | ţ | ł | ; | ; |
| HC SVF | 2 | N/N | ; | W/W | : | N/A | ł | V /I | 1 | A/A | : | ¥/¥ | ; | 0.0£+00 | 1 | 0.0£+00 | : | 1354.0 | ţ | : | ; | 1 |
| HCRSC | ۶, | 60-30.9 | 2.66+01 | N/N | : | 9.05-09 | 2.46+01 | N/N | : | 1,05-01 | 2.6E+01 | R/A | : | 9.0E-09 | 2.46+01 | 9.06-09 | 2.65+01 | 140.5 | ; | 2.16+01 | 7. IE-02 | ۳ |
| NCRVC | 5 | 9.06-99 | 2.6E+01 | 1 /4 | : | 9.06-09 | 2.45+01 | N/N | ; | 9.06-09 | 2.46+01 | N/N | t | 9.06-09 | 2.46+01 | 9.05-09 | 2.4£+01 | 150.0 | ł | 2.06+01 | 1.06-03 | 9 - |
| HCREC | 8 | 1.85-08 | 2.46+01 | N/N | : | 1. 85-00 | 2.46+01 | N/N | ſ | 1. IE-0 | 2.66+01 | A/A | ; | 1.16-00 | 2.4E+01 | 1. W-00 | 2.66+01 | 10.7 | ł | 1.15+01 | ; | ISMI |
| HCRVC | 8 | 1.85-06 | 2.46+01 | N/A | ł | 1.85-06 | 2.6€+01 | W /# | ł | 1. M -8 | 2.6E+01 | N/N | : | 1.06-08 | 2.46+01 | 1.05-00 | 2.65+01 | 10.0 | ŧ | 1.0€+01 | : | INST |
| HCRGC | 5 | 11-35-11 | 2.6E+01 | N/N | : | 1.36-10 | 2.6€+01 | N/N | : | 1.3E-10 | 2.6€+01 | N/N | : | 6.56-11 | 2.46+01 | 1.36-10 | 2.4E+01 | 642.0 | 1.4€+02 | 2.16+01 | : | £ |
| HERVE | 5 | 4.56-11 | 2.6E+01 | N/N | ; | 1.36-10 | 2.66+01 | N/N | ł | 1.36-10 | 2.66+01 | N/N | : | 11-35'9 | 2.46+01 | 1.36-10 | 2.4E+01 | 400.0 | 1. XE +02 | 2.0€+01 | ; | |
| HCBGS | 22 | N/N | ł | N/N | ; | N/N | : | N/N | 1 | N/N | : | R/N | ; | 1.06-03 | 1.05+01 | 1.06-03 | 1.06+01 | 220.0 | ł | ; | 193 | ; |
| HCR65 | 2 | 1.0E-03 | 1.0€+01 | N/N | i | L. 06-03 | 1.0€+01 | N/N | ł | 1.06-03 | 1.0£+01 | N/N | 1 | 1.06-03 | 1.06+01 | 1.06-03 | 1.0€+01 | 10.7 | : | ; | WE BI | ; |
| HCRVS | 22 | 1.0E-03 | 1.0€+01 | N/N | : | 1.06-03 | 1.0€+01 | W/W | I | 1.06-03 | 1.0€+01 | N/N | ; | 1.06-03 | 1.0€+01 | 1.06-03 | 1.06+01 | 10.0 | : | : | ME GL | ; |

3

T. T. YUNUNU

Province of

111111111111

Korrersel by research

C) Gerren

Contine Contine Contine

Page 1 Date 21-Aug-87

102722J

ONSITE TRANSPORTATION - BARGE

Scenario Frequencies and Range Factors

Agent Available and Released

| SCENARIO | NO. | AP6 Freq | RANGE Factor | AGENT AVAIL | LBS. Spilled | LBS. DETONATED | LBS. ENITTED | DURATION TIME |
|----------|-----|-------------|-----------------|----------------|-----------------|-------------------|-----------------|------------------|
| VWI HS | 1 | 0.0E+00 | | 3400.0 | | | | |
| VWKHS | 2 | 0.0E+00 | | 3400.0 | | | | |
| VWKH3 | 3 | 2.7E-11 | 26 | 3400.0 | 1.7E+03 | | | 2 HRS |
| VWKHE | 5 | 0.0E+00 | | 3400.0 | | | | |
| vili HS | 6 | 2.3E-07 | 20 | 3400.0 | 2.9E+03 | | | 1 HR |
| Vwr.hF | 7 | 1.9E-07 | 20 | 3400.0 | | | 8.52+01 | 20 MIN |
| VWKHS | ۶ | 0.0E+00 | | 3400.0 | | | | |
| VWKHS | 10 | 0.0E+00 | | 3400.0 | | | | |
| VHEHS | 11 | 1.2E-09 | 14 | 3400.0 | 1.7E+03 | | •• | 2 HRS |
| VWKHE | 13 | 0.0E+00 | | 3400.0 | | | | |
| VWKHE | 14 | 1.18-09 | 24 | 3400.0 | 1.7E+03 | | | 2 HRS |

6. C. L. C. N. A



10.2





File: HFKE6.WM.L Fage 1 Date 20-Aug-87

REGIONAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Mandling Operations (MF, Events per Pallet or Container)

Agent Available and Released

| | SCENARIO | GANA | RANGE | ŝ | RANGE | LBAD | RANGE | avi | RANGE | đ | RANGE | PUDA | RANGE | TEAD | RANGE | MON | RANGE | MGENT | ŝ | 50 | 2 | BURATION |
|--------------|----------|------------|-----------|------|--------|-------|--------|------------|--------|------|--------|------|--------|------------|----------|------|--------|-----------|-------------|------------|-----------|----------|
| | NUMBER | FRED | FACTOR | FREQ | FACTOR | FINED | FACTOR | FINED | FACTOR | FREQ | FACTOR | FRED | FACTOR | FREQ | FACTOR | FREQ | FACTOR | AVAILADLE | SPILLED 1 | JET CHATED | ENITIED | 11HE |
| | | | | | | | | | | | | | | | | | | | | | | |
| HF DES | _ | 0.0E+00 | ; | N/A | ; | N/A | : | N/N | : | N/A | ; | N/N | : | 4, 25-09 | 1.36+01 | N/A | ł | 440.0 | 220.0 | : | ; | lhr |
| HF DHS | | 4.4E-09 | 1.36+01 | N/A | ; | N/A | : | N/N | : | A/M | ; | N/N | : | 4.25-09 | 1.36+01 | N/N | 1 | 200.0 | 6. 0 | ; | ; | 1hr |
| HFC6S | - | 4.86-10 | 1.3£+01 | N/N | : | N/N | : | N/A | : | N/A | ; | N/A | : | 4.81-10 | 1. 36+01 | A/A | : | 48.0 | ١.6 | ; | : | IN. |
| HFCHS | - | 01-30.0 | 1.36+01 | N/A | : | N/A | 1 | N/N | ; | A/A | ; | A/N | : | 4.06-10 | 1.36+01 | N/A | : | 96.0 | 3.2 | ; | : | IN. |
| HFEGS | - | 0.0€+00 | ; | N/A | ÷ | N/A | : | A/N | ; | R/A | 1 | N/A | : | 8.46-09 | 1.36+01 | A/A | : | 1500.0 | 1500.0 | ; | : | ł |
| HF HS | - | 8.86-09 | 1.3€+01 | N/A | : | N/A | ; | N/N | ; | N/A | ; | N/A | ; | 8°-98 | 1.35+01 | N/A | ! | 1700.0 | 1700.6 | 1 | ł | F |
| HFKVS | - | 8.86-09 | 1. 3£ +01 | N/A | ; | N/N | ; | A/A | ; | N/A | ; | N/A | ; | 8.46-09 | 1.3E+01 | N/A | ł | 1400.0 | 1600.0 | ; | 1 | J. |
| HERVS | - | 5.06-09 | 1.36+01 | N/A | ; | A/N | : | N/N | : | N/A | ; | N/N | ; | 5.86-09 | 1.36+01 | A/A | ; | 378.0 | 10.5 | ; | ; | ž |
| HF P65 | - | 7.21-10 | 1.3€+01 | N/A | ; | A/M | : | N/A | : | N/A | 3 | N/N | : | 7.26-10 | 1.36+01 | N/N | ; | 52.0 | è.5 | : | : | Ņ |
| HEPHS | - | 7.25-10 | 1. 3E+01 | 4/N | : | W/W | : | N/N | 1 | W/W | 1 | N/N | : | 7.26-10 | 1.36+01 | N/N | : | 93.6 | 11.7 | ; | ; | ž |
| HFPVS | 1 | 7.25-10 | 1. 3E+01 | N/A | : | A/A | : | N/A | : | N/A | ; | N/N | ; | 7.26-10 | 1.3E+01 | N/A | : | 48.0 | 6.0 | ; | ; | Å |
| HF 265 | - | 7.25-10 | i. 3€+01 | N/A | : | M/M | : | N/A | : | N/A | ; | N/N | : | 7.2E-10 | 1.35+01 | N/A | : | 87.0 | 14.5 | : | : | Ä |
| HEBVS | - | 0.0€+00 | : | NIA | : | N/N | : | N/N | ; | N/A | ; | N/A | ; | 7.25-10 | I. 3E+01 | N/A | : | 87.0 | 14.5 | ; | ; | ł |
| HFR65 | | 3. ZE - 09 | 1. 3E+01 | A/A | : | A/A | : | N/A | : | N/A | : | A/N | : | 3.26-09 | 1.3£+01 | N/A | ; | 160.5 | 10.7 | : | : | Ň |
| HFRVS | - | 3.26-09 | 1.36+01 | N/N | ; | N/A | : | N/N | ; | N/A | : | A/A | : | 3.26-09 | 1.36+01 | N/A | ; | 150.0 | 10.0 | : | : | ł |
| HFSVS | - | 0.0E+00 | : | A/A | : | N/N | : | N/N | : | A/A | : | N/N | : | 1.06-04 | 1.35+01 | N/N | : | 1356.0 | 1356.0 | : | : | h |
| MF 265C | ~ | 0.0€+00 | ; | N/A | : | N/A | : | N/A | ł | R/N | : | N/A | ł | 1.06-16 | 3.16+01 | N/A | ; | 220.0 | ; | ł | HE GL | ; |
| ¥ PK | 2 | 0.0€+00 | | A/A | : | R/A | ; | N/A | : | N/A | ; | A/A | 1 | 0.0E+00 | | N/A | ; | 6.0 | ; | ; | ; | : |
| NFCGC | 2 | 0,05+00 | | A/N | ; | A/A | : | A/A | : | N/A | : | N/A | : | 0.0€+00 | | N/A | ; | 1.6 | : | : | : | ; |
| NFCHC | 2 | 0.0€+00 | | 4/H | : | A/A | : | AVA | ; | 4/H | ; | N/A | ; | 0.06+00 | | N/A | : | 3.2 | : | : | ; | ; |
| HFK6C | 2 | 0.0€+00 | ; | N/A | : | N/N | : | N/N | : | N/A | ; | N/A | ; | B. 4E - 16 | 3. 16+01 | N/A | ; | 1500.0 | : | ; | NE BL | ; |
| FEE | 2 | 8.4E-16 | 3. 16+01 | N/A | : | N/N | ÷ | N/N | : | ₩/₽ | : | N/A | ; | B.4E-16 | 3. IE+01 | N/A | : | 1700.0 | ; | ; | 19 | ; |
| HFKVC | 2 | 8.4E-16 | 3. 15+01 | N/A | : | R/N | ; | N/A | : | N/N | . 1 | N/A | ; | 8.4E-16 | 3. IE+01 | N/A | : | 1600.0 | ł | ; | 163W | ; |
| HFIIVC | 2 | 2.46-17 | 3. IE+01 | N/A | ; | A/A | ; | N/A | 1 | N/A | : | N/A | : | 2.46-17 | 3. IE+01 | N/N | : | 0.5 | : | ; | жe. | 1 |
| HFPBC | 2 | 0.0€+00 | | N/N | ; | N/N | ; | N/N | ł | N/A | : | N/A | ; | 0.0£+00 | | N/A | : | 6.5 | : | ; | ; | ; |
| HEPHC | 2 | 0.0E+00 | | A/A | : | A/A | ; | A/A | ; | N/N | ; | N/A | ; | 0.0E+00 | | A/A | ; | 11.7 | : | ; | : | : |
| HEPVC | 2 | 0.0E+00 | | N/N | ; | A/A | : | N/A | : | N/A | : | A/A | ; | 0.0E+00 | | N/A | ; | 6.0 | : | ; | : | : |
| 598.44 | 2 | 0.0€+00 | | N/N | : | N/A | : | N/N | : | A/H | : | N/A | : | 0.0€+00 | | N/A | : | 14.5 | : | : | : | : |
| HEDVS | 2 | 0.0E+00 | ; | M/A | : | N/N | ł | N/A | ; | A/A | 1 | N/A | : | 0.016+00 | | M/A | ; | 14.5 | ; | ; | ; | : |

File: WFAEG.WKI Page 2 Date 20-Aug-07

REGEDMAL COLLOCAL, ON OPTION - FACILITY NANDLING

Accident Frequencies for Facility Mandling Operations (MF) (Events per Pallet or Container)

Agent Available and Released

| | SCENAR10 | QVIIV | RANGE | 9 . W | RANGE | T BAL | RANGE | ł | RANGE | M | RANGE | Puba | XUNC | TEND | INNEE | 5 | WHEE | AGENT | 50 | 18 | | URAT 10H |
|--|----------|-----------------|----------|--------------|--------|------------|--------|------------|--------|------------|--------|-------|-------------|------------|----------|------|-------------|----------|---------|-----------|-----------|----------|
| | NUMBER | FREG | F ACTOR | FRED | FACTOR | FRED | FACTOR | FIED | FACTOR | FRED | FACTOR | 1 KEO | TACTOR | | AC TOR | a a | AC108 | WAILABLE | SPILLED | DETONATEB | ENLITED | ¥ |
| HE REC | 2 | 1.96-16 | 3, 16+01 | W/W | ł | W/W | 1 | V/N | ; | N/N | : | N/N | ; | 1.06-14 3 | 16+91 | N/N | ; | 10.7 | ; | 1 | ж6 | : |
| HERVE | ~ | 1.86-16 | 3. 16+01 | N/A | ; | A/N | 1 | A/N | 1 | A/A | ! | N/N | : | 1.86-16 3 | IE+01 | N/N | : | 10.0 | : | : | ж. | : |
| HFSVC | 2 | 0.0€+(0 | ; | N/A | : | N/A | : | A/A | : | N/A | ; | N/A | : | 4.56-15 3 | 10+31.1 | A/A | ; | 1356.0 | ; | : | ж. С | : |
| 1961 | | 0.0E+00 | ; | A/A | ; | N/A | : | N/A | : | A/A | : | N/N | : | 3.16-11 3 | i. 1E+01 | A/A | ; | 440.0 | : | : | 2.206+01 | losia |
| HF DHF | 5 | 9.4E-11 | 3. 16+01 | A/A | ï, | N/A | : | N/N | ; | A/A | : | N/A | ; | 9.46-11 3 | (, IE+01 | N/N | : | 288.0 | ; | : | 10-300.3 | 10ein |
| 18C6F | • | 0.0€+00 | | N/N | : | N/A | : | N/A | : | N/N | ; | N/N | ; | 0, 0E+00 | | N/A | ł | 48.0 | ; | ; | 1 | : |
| HCH | 5 | 0.0€+00 | | N/A | : | N/N | : | N/N | : | N/A | : | N/N | 1 | 0,00+30.0 | | N/N | ; | 96.0 | : | ; | 1 | : |
| 152.44 | - | 0.0€+00 | ; | N/A | ; | N/A | 1 | N/A | : | #/# | ; | | ; | 1.01-30.1 | 10+31.1 | 414 | ; | 1.000.1 | : | ; | | 10min |
| 10 A A A A A A A A A A A A A A A A A A A | • | 1.06-10 | 3. IE+01 | N/A | ; | N/A | ; | N/N | : | A/A | : | N/A | : | 1.05-10 3 | I. IE+01 | N/A | ; | 1700.0 | : | ; | 1. 50E+01 | Idein |
| HEXVE | • | 1.06-10 | 3. IE+01 | A/A | ; | N/N | : | N/A | ; | N/N | : | N/N | : | 1.05-10 3 | 6. IE+01 | N/N | ! | 1600.0 | ; | : | 10+300.1 | 10mm |
| HEAVE | • | 1.46-10 | 3. 16+01 | N/N | : | N/A | ; | N/N | ; | N/N | : | N/A | 1 | L.4E-10 3 | i. 1E+01 | N/A | : | 378.0 | ; | ; | 2.63E-01 | 10aia |
| 19 d H | r | 0.06+00 | | A/A | ; | N/A | : | N/A | ł | N/N | : | N/N | : | 0.0E+00 | | N/A | : | 52.0 | : | ł | ; | : |
| HE PAE | 5 | 0.06+00 | | M/A | ; | N/N | ; | 8/8 | : | A/H | : | N/N | ; | 0.0£+00 | | N/N | : | 93.6 | ; | 1 | 1 | ; |
| HEPVE | n | 0,00+00 | | N/A | ; | N/N | ; | N/N | ; | N/A | : | N/N | ; | 0.0£+00 | | N/A | : | 48.0 | ; | 1 | : | : |
| 190 H | ~ | 0.0E+00 | | N/A | ; | N/N | : | N/N | ; | N/N | : | N/N | ; | 0.0E+00 | | N/A | : | 87.0 | ; | ; | ; | : |
| HEAVE | 5 | 0.0E+00 | ; | N/A | ł | A/H | : | N/N | : | A/N | ; | A/A | ; | 0.0E+00 | | N/A | : | 87.0 | ; | ; | : | ; |
| 194 AN | •• | B. JE-11 | 3. IE+01 | N/A | ł | N/N | : | N/N | ł | N/A | ; | N/A | ; | B. 1E-11 3 | . IE+01 | N/A | ; | 160.5 | : | ; | .07E+00 | 10ei n |
| 14.94 | ы | 8.16-11 | 3.16+01 | N/N | : | N/N | : | A/N | ; | A/A | ; | N/A | ! | B. IE-11 2 | 10+31.1 | N/N | ; | 150.0 | 1 | : | 2.56E-01 | lûnin |
| HESVE | ŗ | 0.0E+00 | ; | N/N | ; | N/N | : | N/A | ; | A/H | : | N/N | ; | 2.06-09 3 | i. 1E+01 | N/A | ł | 1356.0 | ; | ; | 10+365.1 | 10ain |
| HESVS | - | 0,0€+00 | ; | N/A | ł | N/N | 1 | N/A | ; | N/A | : | N/A | ; | 3. IE-07 | | N/A | : | 1356.0 | ; | ; | 1. DOE-05 | ł |
| HF 86F | ŝ | 0.0E+00 | 1 | N/A | : | N/N | 1 | N/N | ; | A/A | : | N/A | : | 0.0E+00 | | N/A | : | 288.0 | ł | : | : | : |
| FDFC | 5 | 0.0E+(0 | | N/N | ; | N/N | : | A/N | ; | A/A | ł | N/N | ; | 0.0E+00 | | N/N | ; | 280.0 | : | 1 | 1 | ; |
| WFC8C | 'n | 0.0€+00 | | N/A | ł | N/N | 1 | N/A | : | N/N | ; | N/A | 1 | 0.0E+00 | | N/A | : | 40.0 | ; | : | ; | : |
| HFCHC | • | 0.0E+00 | | N/A | ł | N/N | ; | W/W | 1 | W/W | : | N/A | ; | 0.06+00 | | N/N | : | 96.0 | ł | ; | 1 | ł |
| Frict | 'n | 0,0E+00 | ; | R/A | ţ | N/N | ; | N/N | : | N/N | : | N/N | : | 0.0£+00 | | N/N | : | 1500.0 | ; | ; | ; | ł |
| 年六年 | 'n | 0,00+00 | | R/R | ; | W/W | ł | N/N | ł | N/N | : | N/N | ; | 0.0E+00 | | 11/A | ; | 1700.0 | ; | ; | ; | ł |
| HEKVE | s | 0.0€+00 | | N/A | ; | N/N | ; | N/A | ; | N/A | ; | A/A | ; | 0.0E+00 | | N/A | : | 1600.0 | 1 | ; | ; | 1 |
| HFIRIC | 5 | 0.0€+(0 | | N/A | ţ | N/N | ł | N/A | ; | N/N | : | N/A | ; | 0.05+00 | | N/N | ; | 378.0 | ; | ł | ł | 1 |
| 19d JH | 'n | 0.0€+00 | | N/A | ; | N/A | ł | N/A | : | N/A | 1 | A/A | ; | 0.0E+00 | | N/N | : | 52.0 | : | ; | ; | : |



.1





File: HFREG.WK1 Page 3 Date 20-Aug-87

REGIONAL COLLOCATION DPTION - FACILITY HAMDLING

Accident Frequencies for Facility Handling Operations (MF) (Events per Fallet or Container)

홂

Agent Available and Released

| | SEEMAR10 | ANA | PANGE | 9 9 | RANGE | LBAD | RANGE | MAP | RANGE | PBM | RANGE | PUDA | RANGE | TEAD | RANGE | MON | RANGE | AGENT | 587 | ŝ | 5 | DURAT |
|---------|----------|------------|----------|-------------|--------|-------------|--------|--------------|--------|------|--------|-------|--------|-----------------|-----------|------|--------|-----------|---------|----------|---------|----------|
| | NUMBER | FRED | FACTOR | FRED | FACTOR | FRED | FACTOR | FRED | FACTOR | FREQ | FACTOR | FRED | FACTOR | FRED | FACTOR | FRED | FACTOR | AVAILABLE | SPILLED | DETONATE | D ENTTE | WI C |
| | | | | | | | | | | | | | | | | | | | | | | |
| 3Hd HL | 5 | 0.0€+00 | | W/W | ; | N/A | ; | N/A | : | A/A | ; | N/A | : | 0,0€+00 | | A/N | : | 93.6 | : | : | ; | : |
| HEPVC | ŝ | 0.0€+00 | | A/N | ; | N/A | ; | N/N | : | A/A | : | N/A | ; | 0.0E+00 | | R/A | : | 48.0 | : | : | ; | ; |
| HF 86C | 'n | 0.0E+00 | | N/N | ; | A/N | : | N/A | : | M/A | ; | N/A | : | 0.0€+00 | | N/A | : | 87.0 | : | : | ; | : |
| JAB35 | ŝ | 0.0€+00 | : | N/N | ; | R/A | : | A/A | ; | N/A | ţ | N/A | ; | 0.06+00 | | A/A | ; | 87.0 | ; | : | ; | : |
| HF RGC | 5 | 0.0E+00 | | 8/8 | ţ | A/4 | : | A/N | ; | 61N | ; | N/A | : | 0.0E+00 | | N/N | : | 140.5 | : | 1 | : | ; |
| Злезн | 'n | 0.0€+00 | | A/N | : | N/A | ; | R/A | ; | N/N | : | N/A | ł | 0.0E+00 | | M/A | ; | 150.0 | : | ł | ; | ; |
| Hc 2.1t | 'n | 0.0E+00 | ; | A/A | : | A/M | ; | N/N | : | A/A | ļ | N/A | ; | 0.0E+00 | | N/A | : | 1356.0 | : | ; | ; | ; |
| P.C 865 | r., | 0.0E+00 | : | N/N | : | A/N | : | M/A | ; | A/A | ; | N/A | ! | B. 36-10 | 1.3£+01 | N/A | : | 440.0 | 220.0 | : | : | ž |
| SHJ JH | ~ | 2.55-09 | 1.35+01 | A/A | ; | 4 /R | : | A/A | ; | N/A | ; | #/# | ; | 2.56-09 | 1.31.401 | 8/8 | ; | 288.0 | 9.Q | : | ; | Æ |
| HF C65 | 7 | 0.01+00 | | R/A | : | A/A | : | A/H | : | N/N | : | A/# . | ; | 0.01400 | | N/A | ; | 48.0 | : | ; | ; | 1 |
| HE CHS | ~ | 0.06+00 | | N/A | : | 4 . R | ; | 4/H | : | A/N | ; | N/A | ; | 0.0E+00 | | M/A | ! | 96.0 | ; | : | ; | ; |
| Hi 165 | 1 | 0.064.00 | | A/A | : | ₩.₩ | ; | A/A | : | 8/N | : | N/A | : | 2.76-09 | 1. 3E +01 | N/A | : | 1500.0 | 1500.0 | ; | : | ihr |
| HEHE | •. | 2. 7E 09 | 1.35+01 | 4 / N | : | A . W | | 4 ' N | : | N/A | : | N/A | ł | 2.7E-09 | 1.36+01 | N/A | ; | 1700.0 | 1700.0 | : | : | In . |
| SAIJH | r | 2.75-03 | 1. 36+01 | A.M | : | M/A | : | A/N | ÷ | 4, N | ; | R/A | ; | 2.7E-09 | 1. JE+01 | N/A | : | 1600.0 | 1600.0 | ; | 1 | 14r |
| HF INS | 5 | 3.66-09 | 1.36+01 | A/A | ; | 4/N | : | A/A | ; | A SH | : | N/A | : | 3.66-09 | 1. 3£ •01 | N/A | ; | 378.0 | 30.5 | : | : | lhr I |
| HF F65 | 2 | 0.0E+00 | | U .1 | : | M/A | : | 11/ 2 | : | N/A | ; | N/A | : | 0.01+00 | | K/A | : | 52.0 | ; | ! | ł | ; |
| HEFHS | 2 | 0.0E+00 | | N/A | ; | A/M | ; | 8/8 | í | N/A | ; | N.N. | | 0.01+00 | | N/A | : | 93.6 | ; | ; | : | : |
| SV93H | 1 | 0.06+00 | | 4/# | ; | N/A | : | A/M | ; | N/A | : | N, A | : | 0.01+20 | | N/A | : | 46.0 | ; | : | ; | ; |
| HF 065 | 7 | 0.0E+U0 | | N/A | : | 8 / N | ; | N/A | • | U/# | : | N/A | : | 0.06+00 | | N/A | : | 87.0 | : | ; | 1 | ; |
| HF QVS | 7 | 0.05+00 | ; | N/A | : | A/N | : | K/A | : | M/M | : | A/N | : | 0.01+00 | | N/A | ; | 87.0 | : | : | : | : |
| HFR6S | 2 | 2.2E-U9 | 1. 36+01 | N,A | | N/A | : | N/A | | N/A | • | 4 · 2 | | 2.26-39 | 1.35+01 | N/A | ; | 160.5 | 10.7 | : | : | 141 |
| HFRVS | 1 | 2.2E-09 | 1.36+01 | A/N | ; | N/A | : | 4/W | ; | A/6 | • | ¥/¥ | • | 2.26-09 | 1.36+01 | 4/4 | ; | 150.0 | 10.0 | ; | 1 | Ъ |
| HF SVS | ٢ | 0.01+00 | : | #/# | ; | A/A | : | N/A | | N/A | ; | N/A | ; | 5.26-08 | I. 3E+01 | N/A | : | 1,756.0 | 1356.0 | : | : | N. |
| HF BGC | æ | 0.0E+00 | ; | N/A | : | R/N | ; | A/A | : | M/4 | ; | N/A | ; | 5.36-18 | 3.1E+01 | N/A | ; | 220.0 | ; | 1 | ₩ E | : |
| | 8 | 5. XF - 18 | 3.16+01 | N/A | : | 8/# | : | N/A | : | A/N | 1 | N/A | : | 5. 36 - 18 | 3. IE+01 | M/A | : | 6.0 | ; | ; | NE BU | : |
| HF CGC | æ | 6 OE 19 | J. IE+01 | N/A | ; | N/A | ; | 8/8 | : | A/M | ; | A/A | ; | 6.0E-19 | 3. 16+01 | A/A | : | 1.6 | : | : | ₩E6 | : |
| MCHC | æ | 6.0E-19 | 3. IE+01 | N/A | ; | N/A | ; | N/A | ł | A.M | ; | N/A | ; | 61-30.9 | 3. 16+01 | N/N | ; | 3.2 | : | ; | 39 | : |
| HFF6C | 89 | 0.0E+00 | : | N/A | : | N/A | : | N/A | ; | A/A | : | N/A | ; | 1.16-17 | 3. IE+01 | A/A | : | 1500.0 | ; | : | ж6L | : |
| Jerge | 8 | 1. IE-17 | 3. 16+01 | N/A | : | N/A | : | N/A | : | A/A | ; | N/A | ; | 1. IE -17 | 3. IE+01 | A/A | : | 1700.0 | : | ; | IE BL | : |

:

I-49

-C-040-

File: WFRE6.WK1 Page 4 Date 20-Aug-B7

REGIONAL COLLOCATION OPTION - FACILITY NANDLING

Accident Frequencies for Facility Mandling Operations (MF) (Events per Fallet or Container)

Agent Available and Released

| | SCENAR 10 | ANAD | RANGE | APS | RANGE | LBAD | RAMGE | AMM | RANGE | PBA | RANGE | PUDA | RANGE | TEAD | RANGE | ADA | RANGE | AGENT | Sel | SAL | 587 | DURATION |
|--------|-----------|------------|-------------|---------------------|--------|------|--------|------|--------|-------------|--------|------|---------|----------|----------|------|--------|-----------|---------|-------------|----------|----------|
| | NUMBER | FRED | FACTOR | FRED | FACTOR | FIED | FACTOR | FREQ | FACTOR | FRED | FACTOR | FRED | FAC TOR | FRED | FACTOR | FRED | FACTOR | AVAILABLE | 191114S | DETONATED | EN117ED | Ĕ |
| | | | | | | | | | | | | | | | | | | | | | | |
| fyvc | - | 1.16-1) | 3.16+01 | W / M | 1 | N/A | ; | W/W | : | A/M | ; | A/A | : | 1.16-17 | 3. IE+01 | N/N | : | 1600.0 | ; | ; | 183 | ; |
| FINC | 80 | 7.25 19 | 1 3. TE+01 | N/A | : | N/A | 1 | A/A | ; | A/A | : | M/A | : | 1.21-11 | 3. IE+01 | R/A | ; | 10.5 | ; | ł | ME GL | ; |
| F 6C | 80 | 9.06-19 | 3. 16+01 | N/N | : | N/A | ; | A/N | ; | N/A | ; | N/A | : | 9.06-19 | 3.16+01 | N/A | ; | 6.5 | : | : | ÆGL | ; |
| FPHC | 60 | 9.06-19 | 3. IE+01 | A/N | ; | N/A | ; | N/N | 1 | A/A | ; | N/A | 1 | 9.01-19 | 3. IE+01 | N/A | : | 11.7 | : | : | NE GL | ; |
| FPVC | 80 | 9.06-19 | J. 1E+01 | A/A | i, | N/N | : | N/A | : | N/N | : | N/A | : | 9.116-19 | 3. IE+01 | N/A | ; | 6.0 | : | : | ж.61 | ; |
| F 06C | 60 | 9.06-19 | 3.16+01 | M/A | : | N/A | : | N/A | : | N/A | : | N/A | ; | 9.06-19 | 3. IE+01 | N/N | ; | 14.5 | ; | ; | ME BL | ; |
| FOVC | 8 | 0.0€+00 | 1 | N/A | ; | N/A | : | N/A | ; | A/A | : | N/A | ; | 9.0E-19 | 3. IE+01 | N/A | ; | 14.5 | ; | : | NEGL | ; |
| FREC | 80 | 4. IE-18 | 3. 16+01 | A/A | : | A/A | ; | M/N | ; | N/A | ; | A/M | : | 4. 16-18 | 3. IE+01 | N/A | : | 10.7 | : | ; | ж61 | ; |
| FRVC | 8 | 4, 15 - 10 | 3.16+01 | M/A | : | N.N | ; | W/W | ; | A/A | : | A/A | ; | 4. 15-18 | 3.16+01 | N/A | : | 10.0 | : | : | MEGL | : |
| FSVC | œ | 0.0E+00 | ; | N/A | ; | A/A | : | M/A | : | M /A | ; | NIA | ; | 1.26-15 | 3.16+01 | M/A | ; | 1356.0 | : | : | ж Ю | ; |
| FSVC | ۵r | 0.0E+00 | : | N/A | ; | N/A | : | A/M | : | N/A | : | A/A | : | 7.76-16 | 3. IE+01 | A/N | : | 1356.0 | : | ; | 1.356-04 | Ŧ |
| F 06C | 10 | 0.0€+00 | : | A/A | 1 | A/N | ; | N/A | : | M/M | : | N/N | ; | 4.3E-19 | 3. IE+01 | N/A | ; | 440.0 | : | 1 | ME GL | ; |
| FDHC | 91 | 1. 35-16 | 3.16+01 | A/N | ; | N/A | : | A/A | : | W/W | : | 8/8 | ; | 1. 36-18 | 3. IE+01 | N/A | : | 288.0 | : | ; | 183H | : |
| FCGC | 10 | 0.0€+00 | | 4/W | ; | N/A | ; | N/N | ; | A/M | ; | N/A | ; | 0.0€+00 | | A/N | ; | 48.0 | ; | ; | жer | ; |
| FCHC | 10 | 0.0E+00 | - | N/N | : | N/A | ; | N/N | : | A/A | : | A/A | ; | 0.0€+00 | | N/A | : | 96.0 | ; | : | NE BI | : |
| FKGC | 10 | 0.06+00 | ; | A/A | : | N/A | : | N/A | : | N/A | ; | A/A | : | 1.46-18 | 3. 16+01 | M/A | : | 1500.0 | ; | ; | NE BL | ; |
| FXHC | 10 | 1.4E-18 | 1 3.16+01 | N/N | ł | A/A | ł | N/N | : | R/A | : | N/A | : | 1.4E-18 | 3.1E+01 | N/N | ; | 1700.0 | ; | : | NE GL | ; |
| FKVC | 01 | 1.46-18 | 1 J. IE +01 | N/N | ; | A/A | : | A/N | ; | M/A | : | N/A | : | 1.46-18 | 3.16+01 | N/A | : | 1600.0 | ; | ; | ЯW | : |
| FINC | 10 | 1. 96 - 18 | 3.16+01 | N/A | ł | N/A | ; | N/N | ł | N/A | ł | N/A | : | 1.96-18 | 3. JE+01 | N/A | ; | 378.0 | : | ; | ME BL | : |
| FPGC | 10 | 0.0E+00 | - | 8/8 | ; | N/A | ; | N/A | ł | N/A | : | N/A | ; | 0.0E+00 | | N/A | ; | 52.0 | : | ; | щer | ; |
| FPHC | 10 | 0.0£+00 | - | A/A | ; | A/N | ; | N/A | ; | N/A | ; | N/A | ; | 0.01400 | | N/A | : | 93.6 | : | : | ¥E0. | : |
| FPVC | 91 | 0.0€+00 | + | A/A | ; | N/A | ł | N/A | ; | N/A | : | 8/H | ; | 0.0E+00 | | N/A | ; | 48.0 | ; | ; | MEG. | ł |
| F 06C | 10 | 0.0£+00 | _ | N/N | ; | 8/8 | ; | N/N | : | N/A | 1 | N/4 | : | 0.0E+00 | | N/A | ; | 87.0 | : | ; | ME GL | : |
| FDVC | 01 | 0.0€+()0 | 1 | N/N | : | N/A | : | A/N | : | N/A | 1 | N/A | : | 0.01+00 | | N/A | ; | 87.0 | ; | : | 18 W | : |
| FRGC | 2 | 1.16-18 | 3. IE+01 | N/A | : | N/N | : | N/N | : | N/A | : | N/A | 1 | 1.16-18 | 3. 1E+01 | N/N | : | 160.5 | : | : | щ | ; |
| IF RVC | 10 | 1. JE-18 | 3.16+01 | N/A | : | 4/N | ; | A/M | ; | R/N | : | N/A | 1 | 1.16-18 | 3. IE+01 | N/A | : | 150.0 | : | ł | Щ. Ш | : |
| E SVC | 91 | 0.0€+00 | ; | N/A | : | A/N | ; | A/N | ; | R/A | : | N/A | : | 2.76-17 | 3. IE+01 | N/A | ; | 1356.0 | ; | ; | ME BI | : |
| FDHC | = | 5.86-11 | : | A/N | : | N/N | : | N/N | ; | M/A | ; | N/A | ; | 5.86-11 | 2.46+01 | A/A | : | 200.0 | 30.0 | b. 0 | : | 1hr |
| FC6C | Ξ | 2.96-11 | 2.65+01 | A/M | ; | A/A | ; | W/W | ; | A/M | : | W/W | : | 2.9E-11 | 2.66+01 | N/A | : | 48.0 | 8.0 | 1.6 | ; | Ä |

I-50

арххар







RESIDNML COLLOCATION OPTION - FACILITY HANDLING

| Agent Available and Released |
|----------------------------------|
| (Events per Pallet or Container) |
| ty Nandling Operations (NF) |
| ent Frequencies for Facilit |
| ξ. |

78. N.

| | SCENARIO | | RAMES | 34 | RANGE | | PANET | - | PANES | 1 | Baue | PUDA | RANGE | 15 AD | RANGE | A CHER | PANCE | AGF KI | 8 | S | 547 1 | DIRATION |
|-------------------|----------|----------|-----------|------|--------|------------|--------|------------|--------|------|-------------|------|--------|----------|----------|--------|--------|--------------|---------|----------|----------|----------------|
| | NUMBER | FRED | FACTOR | FREQ | FACTOR | FREG | FACTOR | FIED | FACTOR | FREE | FACTOR | FRED | FACTOR | FREQ | FAC TOR | FIED | FACTOR | AVAILABLE | SPILLED | ETONATED | ENTTER | 11mm |
| | | | 1 | | | | | | | | | | | | | | | | | | | ļ |
| MCHC | Η | 2.96-11 | 2.66+01 | N/A | ; | N/N | ; | N/N | 1 | N/N | ι | N/N | ; | 2.96-11 | 2.6E+01 | W/W | : | 96.0 | 16.0 | 3.2 | ; | ĥ |
| HFINC | Ξ | 1.36-11 | 2.6 +01 | N/A | : | N/A | : | K/N | : | N/A | : | N/N | : | 4.36-11 | 2.66+01 | N/A | ; | 378.0 | 157.5 | 31.5 | ; | -II- |
| NJ L BC | = | 9.65-12 | 2.6€+01 | N/A | ; | A/A | : | N/N | ; | N/A | ; | N/A | ł | 9.46-12 | 2.4E+01 | N/A | : | 52.0 | 32.5 | 6.5 | ; | J. |
| HPHC | = | 9.46-12 | 2.46+01 | N/N | ; | N/N | : | N/A | ; | N/A | ۲ | N/N | 1 | 9.6E-12 | 2.46+01 | N/A | ; | 93.6 | 58.5 | 11.7 | ; | Ä |
| HFPUC | Ξ | 9.65-12 | 2.46+01 | N/N | ł | #/# | : | N/N | : | N/N | ; | N/A | : | 9.46-12 | 2.6E+01 | N/N | ; | 48.0 | 30.0 | 6.0 | ; | Sar |
| 1985. | = | 7.26-12 | 2.66+01 | R/A | ł | N/A | ; | A/N | ; | N/N | ; | N/N | : | 7.26-12 | 2.6E+01 | N/A | ; | 87.0 | 72.5 | H.5 | 1 | Pr. |
| HEOVE | Ξ | 0.0€+00 | 1 | N/A | ; | N/A | : | N/A | : | N/A | ţ | A/A | : | 7.25-12 | 2.6E+01 | N/A | ; | 67.0 | 72.5 | 14.5 | 1 | Ņ |
| HFRGC | Ξ | 1.66-11 | 2.6€+01 | N/N | : | N/N | : | N/N | : | R/A | ; | N/A | ; | 1.86-11 | 2.6E+01 | N/A | ; | 140.5 | 139.1 | 21.4 | ; | Ä |
| HFRVC | Π | 1.86-11 | 2.46+01 | N/A | : | N/N | ; | A/N | ; | N/A | ţ | ¥/¥ | : | 11-36-11 | 2.4E+01 | A/A | : | 150.0 | 130.0 | 20.0 | ; | Ihr |
| いま | 21 | 3.06-10 | 2.46+01 | A/M | ; | N/N | ; | N/N | : | N/A | ; | N/A | : | 3.06-10 | 2, 6E+01 | N/A | : | 6.0 | : | 6.0 | ; | INSTANT |
| ¥506C | 12 | 3.06-10 | 2.66+01 | N/A | ; | W/W | : | N/N | ; | N/A | ţ | N/A | ; | 3.05-10 | 2.65+01 | N/N | : | 1.6 | ; | 1.6 | ; | INSTANT |
| FCFC | 12 | 3.06-10 | 2.66+01 | A/N | ; | N/A | ; | N/A | ; | A/N | ţ | N/N | : | 3.06-10 | 2, 6E+01 | N/A | ; | 3.2 | ; | 3.2 | 1 | INSTANT |
| HFINC | 12 | 3.06-10 | 2.46+01 | N/A | ; | A/A | : | N/N | : | N/A | : | N/A | ; | 3.0E-10 | 2.66+01 | N/A | ; | 10.5 | : | 10.5 | ; | INS AN I |
| 394 # | 12 | 3.06-10 | 2.6€+01 | N/A | ; | K/N | ; | N/A | ; | N/A | ł | N/A | : | 3.06-10 | 2.6E+01 | N/A | : | 6.5 | : | 6.5 | ; | INSTANT |
| HFPHC | 12 | 3.06-10 | 2.6€+01 | N/A | : | N/A | ; | N/A | ; | A/N | 1 | N/A | ł | 3.06-10 | 2.4E+01 | N/N | : | 11.7 | : | 11.7 | : | INS TANT |
| HEPVC | 12 | 3.06-10 | 2.66+01 | M/A | : | N/A | : | N/A | : | A/A | ł | N/A | : | 3.0E-10 | 2.6E+01 | K'N | ; | b .0 | ; | 6.0 | : | INS FANT |
| HF 26C | 12 | 3.06-10 | 2.6€+01 | #/# | ; | M/M | : | N/N | : | A/A | , | N/A | 1 | 3.06-10 | 2.6E+01 | R/A | : | 14.5 | : | 11.5 | : | INSTANT |
| HE BAC | 12 | 0.0€+00 | : | 8/K | : | N/N | : | #/# | : | N/A | : | N/A | ; | 3.06-10 | 2.4E+01 | N/A | ; | 14.5 | : | 11.5 | : | INSTANT |
| HF RGC | 12 | 3.06-10 | 2.6E+01 | N/A | : | N/N | : | N/A | ; | N/A | 1 | A/A | ! | 3.06-10 | 2.46+01 | N/N | : | 10.7 | : | 10.7 | ; | INSTANT |
| HEBVC | 12 | 3.06-10 | 2. b£+01 | 4/8 | ; | 4/N | ł | N/A | 1 | A/A | ٢ | N/A | 1 | 3.06-10 | 2.6E+01 | N/A | ; | 10.0 | ; | 10.0 | ; | [NS TAK [|
| JHD JH | 13 | 7.25-11 | 2.66+01 | N/N | : | N/A | : | N/N | ; | N/A | ; | N/A | ; | 7.26-11 | 2.6E+01 | A/M | ; | 288.0 | : | 6.0 | . 706-03 | lbr |
| #C6C | :: | 3.66-11 | 2.66+01 | A/A | ; | W/W | ; | N/A | : | A/N | : | N/A | : | 3.46-11 | 2.6E+01 | N/A | ÷ | 48.0 | ; | 1.6 | 10-301 | lir I |
| HCK H | 13 | 3.6E-11 | 2. éf +0ł | N/A | ; | K/A | : | A/A | : | N/A | ; | 4/N | : | 3.66-11 | 2.6E+01 | N/N | ; | 96.0 | : | 3.2 | .60E-03 | 1hr |
| HF MVC | 13 | 5.46-11 | 2.4E+01 | A/A | ; | N/N | ; | N/A | : | N/A | : | N/A | ; | 5.4E-11 | 2.4E+01 | A/A | ; | 378.0 | { | 31.5 | 20-305 | Å |
| 394 JH | 11 | 1. 26-11 | 2.6E+01 | N/A | ; | N/N | : | A/N | ; | A/M | ł | N/A | : | 11-32-11 | 2.6E+01 | N/N | : | 52.0 | : | 6.5 | 10-304. | ĥ |
| HE PHC | : | 1.26-11 | 2.6£+01 | N/A | : | N/A | ; | ¥/¥ | : | N/N | ł | N/A | : | 1.21-11 | 2.4E+01 | A/A | : | 93.6 | ; | 11.71 | . 906-03 | Ä |
| JA JH | 2 | 1.25-11 | 2.6E+01 | #/# | : | W/W | 1 | A/A | : | N/A | ţ | N/A | : | 1.26-11 | 2.46+91 | N/N | : | 48 .0 | ł | 0.0 | 90-300 | ji. |
| 14 06C | : | 9.06-12 | 2.46+01 | A/A | : | N/A | ; | A/A | : | N/A | ٢ | R/A | : | 9.06-12 | 2.6E+01 | N/N | ; | 87.0 | : | 14.54 | 10-305 | Ŀ, |
| HERVC | = | 0.05+00 | ; | N/N | ; | N/A | ; | N/N | : | A/A | ; | N/N | : | 9.06-12 | 2.66401 | N/N | ; | 87.0 | ; | 1.5 | 405-05 | -Pr |

KARREN DARA DARA KARAN MAN

10.20 1150

File: WFKE6.WHI Page & Date 20-Aug-87

REGIONAL COLLOCATION DPTION - FACILITY MANDLING

| | | | | | | Accident | Frequenci | es for Fi | icility Ha | ndling (| herations | 5 (H) 0 | Events per | · Pailet i | or Contain | . | | æ | Jent Avail | able and | Kelease | _ |
|---------|----------------------|--------------|-----------------|-------------|-----------------|----------|-----------------|---------------|-----------------|-------------|-----------------|--------------|-----------------|----------------|-------------------------|----------|-----------------|----------------------|---------------------------|--------------------|--------------|------------------|
| | SCENNAR 10 NUMBER | AKAD FREQ | RANGE Factor | AP6 Fred | RANGE Factor | LBAD | RANGE Factor | ILAAP FREQ | RANGE Factor | PBA Fred | RANGE Factor | PUDA FRED | RANGE Factor | TEND Freq | AMMGE ACTOR | UNDA - | RANGE Factor | AGENI AVAILADLE S | L IS PPILLED DE | L BS FTOMATED E | LIS ATTED | NRAT 10 T 1ME |
| | | | | | | | | | | | | | | | | | | | • | | | |
| HF REC | 13 | 2.26-11 | 2.6€+01 | N N | : | A/N | ; | A/A | : | N/A | : | N/A | ł | 2.26-11 | 10+34.1 | N/N | ; | 160.5 | : | 21.4 9. | 10-308 | Ņ |
| HERVC | 1 | 2.26-11 | 2.6€+01 | N/N | ; | A/A | ; | N/A | ; | A/A | ; | N/N | : | 2.26-11 | 1.66+01 | N/N | ; | 150.0 | : | 20.0 2. | 20-306 | ł |
| HF DHC | = | 1.3E-14 | 2.6€+01 | W/W | ; | A.A | ; | NIA | ; | N/N | ; | 4/4 | : | 1.38-34 | 2.8E+01 | ų/ų | : | 200.0 | 30.0 | 6.0 | ; | Ä |
| HFCGC | 1 | 6.66-15 | 2.4E+01 | A/A | ; | N/N | ! | A/A | : | N/A | : | N/A | ; | 6.46-15 | 2.46+01 | A/A | : | 48.0 | 8.0 | 1.6 | ; | Ä |
| HECK | 1 | 6.66-15 | 2.66+01 | A/M | ł | N/A | : | N/A | : | N/N | : | #/# | : | 6.46-15 | 2. 6 E + 0 I | N/N | : | 96.0 | 16.0 | 3.2 | ł | ł |
| HERVE | = | 9.96-15 | 2.66+01 | N/A | ; | N/A | ; | N/A | : | N/N | : | N/A | ; | 9.96-15 | 2.46+01 | #/# | : | 378.0 | 157.5 | 31.5 | ; | ł |
| HF PGC | 2 | 2.26-15 | 2.66+01 | A/N | ; | A/A | : | N/N | : | N/N | : | N/A | ; | 2.26-15 | 1.6E+01 | N/A | ; | 52.0 | 32.5 | 6.5 | : | ¥ |
| THE PHC | = | 2.26-15 | 2.66+01 | M/A | : | A/A | : | N/A | : | 8/N | ; | N/N | ; | 2.26-15 | 2.65+01 | N/N | : | 93.6 | 58.5 | 11.7 | : | ž |
| HFPVC | - | 2.28-15 | 2.66+01 | N/N | : | #/# | ; | A/A | : | N/A | : | N/N | : | 2.26-15 | 2. 6E+01 | ¥/¥ | ; | 48.0 | 9.9 | 6.0 | ; | H |
| HF DGC | Ξ | 1.7E-15 | 2.66+01 | A/A | : | N/A | : | A/N | ; | N/A | ; | N/N | ; | 1.76-15 | 2.4E+01 | N/A | ; | 87.0 | 72.5 | 14.5 | ; | H |
| HF DVC | 2 | 0.0€+00 | ; | N/A | ; | N/N | : | N/N | : | A/A | ; | N/N | ; | 1.76-15 | 7.4E+01 | N/A | : | 87.0 | 72.5 | 14.5 | ; | ž |
| HF REC | Ξ | 4.16-15 | 2.46+01 | N/N | ; | A/A | ; | N/N | : | N/A | : | N/N | : | 4.16-15 | 2. 6E + 0 1 | #/# | ; | 160.5 | 139.1 | 21.4 | ; | ž |
| NFRVC | ± | 4, IE-15 | 2.6E+01 | N/A | : | N/A | ; | A/A | : | N/A | : | N/A | ; | 1.16-15 | 2.65+01 | W/W | : | 150.0 | 130.0 | 20.0 | : | ž |
| | | | | | | | | | | | | | | | | | | | | | | |

Recession of

6.53.55

al Mah

.







1.1.1.1.1.1.1

File: HONINAT.NKI, Page 1 Date 18-Aug-87

HANDLING ACCIDENTS - NATIONAL PT ZESSING OPTION - PER PALLET OR CONTAINER

| | | | | | | NCC1 BELL | Frequencial | r an Ki | wite ractor | | | | | | | | | £ | | i able and | Nel eased | |
|------------------|----------|------------------|---------------|-----------------------|-----------------|-----------|-----------------|--------------|-----------------|---------|----------------|------------------|---------------|------------------|----------------|----------------|----------------|-------------------|-----------------|------------------|-----------------|-----------------|
| SCEN- OP MRIO | Ś | ANA. | ANGE ACTOR | APS FRED | RANGE Factor | LBAD | RANGE FACTOR | KAAP FRED | RANGE Factor | PBA F | RANGE ACTOR | PUDA 6 Fred F | AMPE ACTOR | TEAD 1 FRED 6 | AMINE ACTOR | FRED | RAMEE ACTOR | AGENT WAILABLE | LINS SPILLED | LDS Detomated | L INS DA | URATION 11NE |
| | | | | | | | | | | | | | | | | | | | | | | |
| HC BGC | • | N/A | : | N/N | ; | N/N | : | N/N | ; | N/N | ; | A/A | ; | 1.25-07 | 1.36+01 | 6.1E-00 | 1.3€+01 | 640°8 | ; | ; | 4. 36 +00 | Ŧ |
| KDKC | - | 80-39 | 1. XE +01 | N/A | ļ | N/N | : | K/A | : | V. | 1 | 1.66-08 | 1.3£+01 | 3.26-08 | 1.36+01 | N/A | ; | 2 M .0 | ł | : | 1.36-03 | Ē |
| NCCEC | - | , BE -09 | 1. JE +01 | N/A | ; | N/N | 1 | N/N | ; | N/N | : | N/N | ţ | 5.76-09 | 1. XE +01 | N/A | ; | . . | : | ÷ | 1.15-01 | ž |
| HICHE | - | 0-38° | 1.36+01 | N/N | : | N/N | ; | A/A | ; | A/A | ; | 2.86-09 | 1.3€+01 | 5.76-09 | 1.3£+01 | N/N | : | 74.4 | ; | ; | 1. X-03 | Ĩ |
| HCKSC | - | A/A | ; | N/A | ١ | N/N | ; | A/M | : | . W/W | ; | A/A | ; | 4.06-08 | 1.3€+01 | A/A | ; | 1500.0 | ; | : | 6.46+00 | Ĩ |
| HCKHC | - | 90-30 | 1. JE+01 | 2.06-08 | 1.36+01 | N/N | 1 | N/A | : | 2.06-08 | 1.3£+01 | R/N | : | 90-30°t | 1.36+01 | 2.06-08 | 1.3€+01 | 1700.0 | : | ; | 2.5E-02 | Ĩ |
| HCKVC | - | N/A | : | N/N | ; | A/A | ; | 2.05-08 | 1.3£+01 | N/N | 1 | N/A | : | 1.05-08 | 1.3£+01 | N/N | ; | 1400.0 | : | ; | 2.76-04 | ۳. |
| HCMVC | - | . 2E - 08 | 1. XE +01 | N/A | ; | N/N | ; | N/N | ł | 1.25-08 | 1.3€+01 | N/A | ł | 2.4E-08 | 1. 3E+01 | 1.25-00 | 1.3£+01 | 378.0 | : | ; | 1.76-05 | ۹. |
| JS4DM | 1 | .06+00 | ; | W/W | ; | N/A | 1 | N/N | ; | N/N | : | N/A | : | 0.06+00 | : | 0.06+00 | ; | 52.0 | : | : | ; | : |
| HCPHC | • | 00+30" | ! | N/N | : | 0.0£+00 | : | N/N | 1 | N/A | Ŧ | 0.0E+00 | 1 | 0.06+00 | : | W/W | ; | 93.4 | 1 | ; | : | : |
| MCPVC | • | 06+30.1 | : | N (N) | ; | 0.06+00 | ; | R/R | : | N/N | : | A/M | : | 0.0£+00 | : | 0.0€+00 | ; | 98.0 | : | ; | : | i |
| HC 06C | - | 00+30" | : | N/A | 1 | 0.0E+00 | 1 | N/A | : | N/A | : | N/A | ; | 0.06+00 | 1 | 0.0€+00 | : | 87.0 | ; | : | ; | : |
| HCDVC | - | N/A | ; | N/A | ; | A/A | 1 | N/A | ; | N/A | : | #/# | ; | 0.0€+00 | 1 | 0.0E+00 | ; | 07.0 | : | ; | : | : |
| HCRGC | - | 86-98 | 1.3£+01 | N/A | 1 | 4, 85-08 | 1.36+01 | N/A | ; | 4.86-08 | 1. JE +01 | N/N | ; | 9329 | 1. 3E +01 | 90-39.4 | 1.3E+01 | 160.5 | ; | : | 4.56-01 | Ĩ |
| HCRVC | - | 1. BE -08 | 1.36+01 | ¥/¥ | : | 4.85-08 | 1.3£+01 | N/A | ; | 4.86-08 | 1.3€+01 | ٩/٣ | ł | 90-3S-6 | 1.36+01 | 4.66-08 | 1.36+01 | 150.0 | 1 | ; | 1.65-05 | ۹. |
| HCSVC | - | ¥/¥ | ; | K/A | ; | N/A | : | N/A | ; | N/N | : | N/N | ţ | 1.05-06 | 1. JE +01 | 5. ZE -07 | 1.36+01 | 1356.0 | : | : | 2.75-04 | ٩. |
| HUKHL | 2 | N/A | : | 5.26-10 | 3. IE+01 | N/A | : | N/A | ł | 5.2E-10 | 3. IE +01 | N/A | 1 | 5.26-10 | 3. IE+01 | N/A | : | 1700.0 | : | : | B.SE+01 | IO NIN |
| HCB6C | 'n | W/W | ; | N/N | ; | W/W | ; | N/A | : | A/M | 1 | N/A | 1 | 2.46-07 | 10+3(1 | 1.36-07 | 1.35+01 | 440.0 | : | : | 4. 36 +00 | 941 |
| FLAC | | <i>7</i> E-07 | 1.3€+01 | N/A | ; | N/A | ; | N/A | ; | N/N | ł | 3.76-07 | 1. JE +01 | 7.46-07 | 1.36+01 | N/N | ; | 200.0 | ; | : | 1. XE -03 | Ŧ |
| HCCBC | ۰ ۳ | 1. 9E - 08 | 1.36+01 | W/W | ; | N/N | 1 | N/A | ; | N/A | 1 | N/A | ; | 1.86-07 | 1. XE +01 | N/A | : | 38. F | : | ; | 1.1E-01 | Ш. |
| HCCHC | 2 | . 96 - 08 | I. 3£ +01 | N/A | ; | W/W | : | N/N | : | N/A | ; | 80-36-B | 1. XE •01 | 1.86-07 | 1. 3E+01 | N/A | ; | 74.8 | ; | ; | 1. X -03 | Ŧ |
| HCINC | 2 | . 15-07 | 1.3€+01 | W/W | ; | N/A | ; | N/N | 1 | 7.16-07 | 1.3£+01 | N/A | ; | 1.46-06 | 10+35-1 | 7.1E-07 | 1.3€+01 | 378.0 | ; | ; | 1.75-05 | ١. |
| HCPGC | 5 | · 0E - 08 | I. JE +01 | N/N | ; | A/A | : | #/# | ; | N/N | ; | M/A | ; | 1.0E-07 | 1.3E+01 | 5.06-08 | 1.3€+01 | 52.0 | : | ; | 2.4E-01 | Ŧ |
| HCPHC | 5 | 90 - 30 * | 1.36+01 | 4 /4 | : | 5.06-08 | 1. 36 +01 | A/4 | : | N/N | ; | 5.06-08 | 1.3£+01 | 1.06-07 | 1.3€+01 | N/A | : | 93.6 | ; | ; | 2. IE -03 | Ē |
| HCPVC | ~ | 80-30 | 1.35+01 | N/N | : | 5, 06 -09 | 1.36+01 | N/A | : | M/A | ; | N/A | : | 1.05-07 | 1. 3E+01 | 5.06-08 | 1. XE +01 | 49.0 | 1 | ; | 2. IE -05 | ۹. |
| HCDEC | 5 | . 05-08 | I. XE +01 | A/A | : | 5.06-08 | 1.3€+01 | N/A | ł | N/A | ł | N/A | : | 1.0E-07 | 1.36+01 | 5.06-08 | 1.3E+01 | 87.0 | : | ; | 5.65-01 | |
| HCBVC | - | R/A | : | N/A | : | W/W | : | N/A | ł | N/A | ; | N/N | ; | 1.06-07 | 1.3E+01 | 5.0E-08 | 1. XE +01 | 87.0 | ; | : | 2. IE -05 | Ĩ |
| HCRGC | 3 2 | . 65-06 | 1.3£+01 | N/N | : | 2.46-06 | 1.35+01 | N/A | ; | 2.6E-06 | 1.36+01 | N/A | 1 | 5. 3E - 06 | 10+ X. I | 2.6E-06 | 1.3€+01 | 140.5 | ; | : | 1.5E-01 | Ш. |
| HCRVC | 1 2 | · 66 - 06 | 1.XE+01 | N/N | | 2.65-06 | 1. JE +01 | R/A | ł | 2.4E-06 | 1.35+01 | N/A | 1 | 5. 3E - C& | 1.3E+01 | 2.46-06 | 1.3€+01 | 150.0 | : | ; | 1.66-05 | Ĕ. |

| Γ | AD-A1: | 93 355 551F1E | Che The Ca D Sap | MICAL DISPO A N I E0-CDI | STOCK DSAL O BARSEL E-IS-8 | PILE D F Chem L et r 7008 d | ISPOS 1. (U) 1. AU AAA15- | AL PRO GA TE 3 87 Gi -85-d-1 | GRAM R Chnolo A-C-18 0022 | ISK AN GIES 1 563 | ALYSI NC SAU F/G 1 | 5 OF 1 DIEGO 15/6. 3 | , 12/ | 13 |
|---|--------|------------------|---------------------------|-----------------------------------|-------------------------------------|--------------------------------------|------------------------------------|---------------------------------------|------------------------------------|-------------------------|--------------------------|----------------------------|-------|----|
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | _ | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |


File: WELMAT.AKI, Page 2 Date 10-Aug-87

HANDLING ACCIDENTS - NATIONAL PROCESSING OPTION - PER PALLET ON CONTAINER

| | | | | | Accident | Frequenci | a ha r | nge Factor | - | | | | | | | | • | ym t Ava | ilable and | l Rel essed | |
|------------------|--------------|--------------|-------------|-----------------|---------------|-----------------|---------------------|-----------------|------------|---------------|------------|---------------|----------------|----------------|-----------|----------------|-------------------|-----------------|------------------|-------------------|----------------|
| 9058- 07 Ario | 로분 모 고 | 10 RANGE | AFG Biff | RANGE FACTOR | L DAN Freq | RANGE Factor | INAR Fred | RANGE Factor | | MARK ACTOR | PUON | AMBE ACTOR | TEND I Free | ANNEE ACTOR | L HIDA | ANDER ACTOR | AGENT VALLABLE | 105 SP11E0 | LBS Detonated | LOS DA Enitted | RATION 11ML |
| | 1 | | | ! | | | | | | | | | | | | | | | | | |
| NCSVC | 3 N/A | - | N/N | : | N/N | : | N/A | : | N/N | : | 8/8 | ; | 6.1E-07 | 1.36+01 | 1.56-07 | 1.36+01 | 1354.0 | ; | : | 2.76-04 | ۹. |
| HENGE | | : | A/M | ; | N/N | ; | N/N | ; | A/H | ; | N/N | : | 3.4E-09 | 1. JE +01 | 1.76-09 | 1.36+01 | 440.0 | ; | : | 4. 32 +00 | Ē |
| NCINC | 4 8.06 | -10 1. XE+0 | N/A | : | N/N | ; | N/N | ; | N/N | : | 0.06-10 | 1.3€+01 | 1.46-09 | 1.36+01 | N/A | : | 280.0 | : | ; | 1.36-03 | Ē |
| HCCBC | 4 2.06 | 1.36+01 | | ł | N/N | : | N/A | : | A/A | ; | N/N | ; | 3.96-11 | 1.3£+01 | N/A | : | у н. т | : | ! | 1. IE -01 | ĩ |
| NCINC | 4 2.06 | -11 1.3E+01 | N/N | ١ | N/N | : | N/N | : | . W/N | : | 2.06-11 | 1.3£+01 | 3.96-11 | 1.36+01 | N/N | ; | 74.8 | ; | : | 1.36-03 | Ĩ |
| NCKEC | | : | N/N | ł | N/N | ; | N/N | ; | N/A | ; | N/N | ; | 1.45-08 | 10+32.1 | N/N | ; | 1500.0 | : | : | 1.76-45 | Ĩ |
| HCLHC | 1.2 | -04 1.3E+0 | 1 6.7E-04 | 1.3£+01 | N/N | ; | N/N | : | 4.75-09 | 1.3E+01 | N/A | ; | 1.46-08 | 1. JE +01 | 7.25-09 | 1. XE+01 | 1700.0 | : | : | 2.4E-01 | Ĩ |
| NCIVIC | - N/ | : | A/N | : | N/N | ; | 7. 25-09 | 1.36+01 | N/N | : | N/N | : | 1.46-08 | 10+37.1 | N/A | : | 1440.0 | : | : | 2. IE-03 | Ĩ |
| HENVE | 4 9.% | -10 1' XE+0[| W/W | ; | N/N | : | N/N | ; | 4.96-10 | 1.36+01 | N/N | ; | 1.4E-09 | 1.35+01 | 6.9E-10 | 1.3E+01 | 378.0 | ; | : | 2.16-05 | £ |
| HCPSC | 4 0.0 | - 8- | N/N | ł | N/N | ł | A/A | 1 | N/A | ; | N/N | ; | 0.0€+00 | | 0.0E+00 | : | 52.0 | : | 1 | ł | : |
| HCMC | 4 0.05 | 00+ | N/N | 1 | 0.0E+00 | ; | N/N | : | N/N | ; | 0.0£+00 | : | 0.0€+00 | | N/A | ; | 93.6 | ; | 1 | : | : |
| HCPVC | 4 0.05 | | R/A | : | 0.06+00 | : | N/N | ł | N/N | ł | N/A | ; | 0.05+00 | | 0.06+00 | ; | 46.0 | : | : | : | ; |
| HCD6C | 4 0.05 | | N/N | : | 0.06+00 | : | N/N | ł | N/N | : | W/W | ; | 0.0£+00 | | 0.0E+00 | : | 87.0 | : | : | ; | ; |
| HCEVC | - N/ | : | A/A | : | A/A | 1 | W/W | ; | A/A | ; | N/N | : | 0.0£+00 | | 0.0£+00 | ; | 0.70 | ; | : | t | ; |
| HCREC | 4 3.IE | -04 1°X 0- | N/N | : | 3. IE-09 | 1.36+01 | N/N | ł | 3.15-09 | 1.35+41 | N/N | 1 | 4.25-09 | 1. 36 441 | 3.16-09 | 1. 35 +01 | 160.5 | 1 | : | 10-35-01 | £ |
| HCRVC | 4 3.E | -04 1° 3E+0 | N/N | ; | 3.16-09 | 1. XE+01 | N/N | 1 | 3. IE - 09 | 1.36+01 | N/N | ; | 4.25-09 | 1.36.401 | 3. IE -09 | 1.36+01 | 150.0 | : | : | 1.45-05 | Ē |
| NESVC | • | • | N/A | : | N/A | ţ | V / I | 1 | N/N | : | A/A | ; | 5.45-08 | 1.36+01 | 2.76-00 | 1. JE +01 | 1354.0 | : | : | 2.76-04 | Ĩ |
| HCBGS | S N/ | : | N/N | : | N/N | : | N/A | : | N/A | 1 | N/N | ł | 4.36-09 | 1.3£+01 | W/H | ; | 440.0 | 2.26+02 | : | ; | Ē |
| HC BHS | 5 11/1 | • | N/N | ; | R/A | ł | N/N | : | N/N | 1 | N/N | 1 | 4. 36-09 | 1.3E+01 | N/N | ; | 2 m. 0 | 6.0E+00 | ; | ; | Ĩ |
| NCCES | 2 11/2 | - | N/N | : | N/N | ; | W/W | : | N/N | .1 | N/N | 1 | 7.26-10 | 1. 36 +01 | W/W | : | 7.92 | 1.46+00 | ; | : | Ĩ |
| HCCNS | 2 12 | | N/N | : | N/N | ; | V/I | ; | N/A | ; | W/W | : | 7.26-10 | 1.3£+01 | N/N | ; | 74.0 | 3.26+00 | ; | : | Ĩ |
| HERES | 2 | : | N/N | ; | N/N | : | N/N | 1 | N/A | 1 | N/N | 1 | 1.36-00 | 1.3E+01 | N/N | ł | 1500.0 | 1.5€+03 | ; | ł | Ĩ |
| HCIHS | 2 11/1 | 1 | #/# | : | N/A | 1 | N/A | ł | N/N | : | N/A | : | 1.35-08 | 1.3E+01 | N/N | 1 | 1700.0 | 1.7E+03 | : | : | Ē |
| HCICVS | 2 11/1 | : | | : | R/A | 1 | N/N | ; | N/A | ; | N/A | ; | 1.35-06 | 1.5-31 | N/ | ; | 1600.0 | 1.46+03 | : | : | ŧ |
| NCIINS | 2 8/1 | : | N /N | : | R/A | ł | N/N | : | N/N | : | N/N | : | 8.46-09 | 1.32.01 | N/N | : | 378.0 | 1.05-01 | 1 | : | Ĩ |
| MCP65 | 5 N/ | : | N/N | : | V | : | N/N | ł | N/N | ł | M/A | 1 | 1.16-00 | 1.36+01 | N/N | 1 | 52.0 | 6, 5£ +00 | : | : | Ĩ |
| HCPAS | 5 N/A | - | V /R | : | N/N | : | N/A | : | W/W | ł | N/A | : | 1.16-09 | 1.36+01 | R/B | ł | 13.4 | 1.25+01 | ; | ; | ŧ |
| HCPVS | 5 11/1 | : | N/A | 1 | N/A | : | A/N | : | N/N | ; | N/N | 1 | 1.16-09 | 1.32+01 | R/A | ; | €. ₩ | 6.05+00 | : | : | Ĩ |
| NC BOS | 5 11/4 | : | S | 1 | N/N | ł | N/N | 1 | N/N | : | N/N | ; | l. lE-0 | 1.32.401 | N/N | : | 87.0 | 1.5441 | ; | : | Ē |

10 XXXXX

I-54







file: HOLMAT.MKI, Page 3 Date 18-Aug-87

NAMOLING ACCIDENTS - INTIONAL PROCESSING OPTION - PER PALLET OR CONTAINER

| | | | | | | Accident | Frequenci | es and N | ange facto | 5 | | | | | | | | | Agent Ava | iilable a | d Reienser | _ |
|------------------|----|--------------|-----------------|-------------|-----------------|---------------|-----------------|----------|-----------------|-------------|-----------------|-----|-----------------|----------|-----------------|--------------|-----------------|--------------------|--------------------|------------------|---------------|------------------|
| SCEN- DP Neto | 8 | ANNA Free | RANGE FACTOR | APS FRED | RAMGE Factor | LIMD FIRED | RANGE Factor | TRED T | RANGE Factor | PIM Free | RANGE FACTOR | PUM | RANGE Factor | TEND | RANGE Factor | UNDA Freg | RANGE FACTOR | AGENT AVAILADLE | SPILLED SPILLED | LDS Detomatei | LDS ENITED | NURATION 1116 |
| | | | | | | | | | | | | | | | | | | | | | | |
| NCDVS | n | | ; | 4/H | ł | | ; | S | : | | : | | ŧ | 1.16-09 | 1. XE +01 | V/R | : | 87.0 | 10+X-1 | 1 | 1 | Ĩ |
| HCRGS | ŝ | N/A | : | N/N | ; | N/N | : | N/N | : | N/N | : | N/N | ł | 4.96-09 | 1.36+01 | N/N | 1 | 160.5 | 1. IE+01 | ; | ; | Ĩ |
| NCRVS | s | N/N | : | N/A | : | N/N | : | N/A | ł | A/A | ł | N/A | : | 4.96-09 | 1.3E+01 | N/A | : | 150.0 | 1.0E+01 | ; | ; | Ĩ |
| HESVS | 5 | N/N | ; | N/A | : | N/N | : | N/A | : | N/A | ł | N/N | : | 1.4E-07 | 1.3£+01 | N/N | : | 1356.0 | 1.4E+03 | : | ; | Ĩ |
| HCBGF | • | N/N | : | A/A | ٢ | R/A | ; | N/N | ; | . W/N | ; | N/N | 1 | 4.2E-11 | 3.16+01 | N/N | ł | 440.0 | ; | : | 2.26+01 | IO NIN |
| HC DHF | - | N/N | : | N/A | : | N/N | 1 | N/A | : | N/A | ł | N/A | : | 1.96-10 | 3.16+01 | N/N | : | 298.0 | : | : | 3.06-01 | NIN 01 |
| HCC6F | - | N/N | ; | N/A | ; | N/N | ; | N/A | 1 | N/N | ; | N/N | ł | 0.0E+00 | : | N/N | ; | 38°.4 | : | : | ; | ; |
| HCCHF | 9 | N/N | ; | N/N | : | A/A | ; | N/N | 1 | N/N | 1 | N/A | : | 0.0E+00 | ł | N/N | : | 76.8 | ł | ! | ; | 1 |
| HCK6F | • | N/N | : | N/A | : | N/N | ; | N/A | : | N/N | : | N/A | : | 2.16-10 | 3. IE+01 | N/N | : | 1500.0 | ; | : | 1.5€+02 | IO NIN |
| HCKHF | - | N/A | ; | A/A | ł | N/N | ; | N/N | : | N/N | ł | N/N | ; | 2.1E-10 | 3.16+01 | N/N | : | 1700.0 | : | 1 | 8.SE+01 | IO NIN |
| HCKVF | -9 | N/A | : | N/A | ; | N/N | ; | N/N | ; | 8/N | 1 | 4/1 | : | 2. JE-J0 | 3. IE+01 | N/N | : | 1600.0 | ; | : | 4.06+01 | 10 MIN |
| HCHVF | -0 | N/A | : | N/A | : | N/A | ; | N/A | : | N/A | ; | N/N | 1 | 2.76-10 | 3. IE+01 | A/A | ; | 378.0 | : | ; | 2.66-01 | IO NIN |
| HCPGF | 9 | N/A | : | A/M | : | N/N | : | N/N | : | N/N | I | N/A | : | 0.0E+00 | : | N/A | : | 52.0 | ; | : | ; | : |
| HCPHF | 9 | N/A | : | A/A | ; | N/A | 1 | N/A | : | N/A | 1 | N/A | ۱ | 0.0£+00 | ; | A/A | ; | 93.6 | ; | ; | : | : |
| HCPVF | 4 | R/A | ; | 8/8 | 1 | N/N | : | N/A | ; | A/M | : | A/4 | : | 0.0E+00 | : | N/N | : | 48.0 | ł | ; | : | : |
| HCDGF | 9 | N/N | : | N/N | ; | N/A | : | A/A | ; | A/A | : | N/N | : | 0.06+00 | ; | N/N | ; | 87.0 | ; | ; | : | : |
| HCEVF | -0 | N/N | ; | N/N | : | N/N | ; | N/A | : | N/N | ; | A/A | 1 | 0.0£+00 | : | A/N | : | 87.0 | ; | ; | ł | : |
| HCR6F | 4 | N/N | : | N/A | ; | N/N | : | N/A | : | A/N | : | N/A | : | 1.65-10 | 3.16+01 | N/A | ł | 160.5 | : | : | 1.1E+00 | IO NIN |
| HERVE | -0 | N/N | : | N/N | ; | W/W | : | N/A | ; | 6/9 | ; | N/A | ; | 1.66-10 | 3. IE+01 | N/A | ł | 150.0 | ; | ; | 2.56-01 | IO NIN |
| NCSVF | • | N/A | : | N/A | ; | 8/8 | ; | N/N | ; | A/A | ; | N/A | : | 3.96-09 | 3.16+01 | N/A | 1 | 1356.0 | : | ; | 3.46+01 | IO NIN |
| HCB6S | - | N/A | ; | A/A | ; | N/N | ; | N/N | 1 | A/N | : | N/A | : | 3.7E-09 | 1.3£+01 | N/N | ; | 440.0 | 2.25+02 | ; | ; | ¥. |
| HCBHS | 1 | N/A | : | N/A | : | N/A | ; | N/N | : | N/A | ; | N/A | ; | 1.16-00 | 1.3£+01 | N/A | ; | 288.0 | 6. 0E +00 | 1 | : | ١. |
| HCC65 | - | N/N | : | M/A | : | N/N | ; | W/W | ; | N/A | ; | N/A | : | 0.0E+00 | ; | A/H | : | 1 .95 | : | ; | ł | : |
| HICHS | • | W/W | 1 | N/A | ; | N/N | ; | N/A | 1 | N/N | ; | N/A | : | 0.0E+00 | : | N/A | : | 74.8 | ; | 1 | ; | : |
| HCKGS | ~ | A/A | ; | A/A | ; | 8/N | ; | N/N | 1 | W/W | : | N/A | : | 1.2E-00 | 1.36+01 | N/A | ; | 1500.0 | 1.5€+03 | : | : | Ĩ |
| NCKNS | - | A/A | ; | A/A | ; | N/A | : | N/A | : | N/N | ł | N/A | : | 1.26-08 | 1. JE+01 | N/A | : | 1700.0 | 1.7E+03 | ł | : | E. |
| HCKVS | ~ | N/A | : | N/A | ; | N/N | ; | N/N | ł | N/A | : | N/A | : | 1.2E-00 | 1.3£+01 | A/N | : | 1400.0 | 1.46+03 | : | : | ٩. |
| HCRVS | • | N/A | : | N/N | ; | N/A | ł | N/A | ł | N/N | ; | N/A | : | 1.65-08 | 1.3€+01 | N/A | : | 378.0 | 1.0E+01 | ł | ł | Ħ |
| HCF6S | ~ | A/A | ; | N/A | ; | N/N | : | A/A | ; | N/A | : | A/A | 1 | 0.0E+00 | ; | N/A | : | 52.0 | ; | : | : | ; |

ᢂᢂᢣᠴᢣᠪᢣᠵᢣᢣᠽᢦᡈᡊᡢᠫ᠕ᡛᢣᠺᡈᠧᠴᡪᡦᡊᡛᡵᡘᠼᠺᠧᡄᡪ᠋ᠧᡊᠽᠧᠧᠧᠽᡧᡗᠺᡚᡞᠿᡵᡭᡬᢓᡘᠼᡬᡵᢗᢓᡭᡚᡭᡚᡭᠸᡬᠼᡗᢑᡭᠧᡵᡭᠼᡭᠧᡭᠸᡭᢑᡭᢑᡭᢑᡭᢌᡭᢌᡭ

a sea a sea a sea a sea a sea a sea a sea a sea a sea a sea a sea a sea a sea a sea a sea a sea a sea a sea a s A

File: HDLMAI.MKI, Page 4 Date 18-Aug-87

HANDLING ACCIDENTS - NATIONAL PROCESSING OPTION - PER PALLET OR CONTAINER

ALC ALS BAT BAT BAT

| | | | | | | Acci dent | Frequenci | es and R. | inge Facto. | 2 | | | | | | | | - | agent Ava | ulable an | d Released | |
|------------------|------------|------------|-----------------|-------------|-----------------|---------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|-----------------|--------------------|-----------------|------------------|------------|-----------------|
| SCEN- DI ARID | P. #0. | FREG | RANGE Factor | APG Fred | RANGE FACTOR | L BAD Freq | RANGE Factor | NAAP Freg | RANGE Factor | PBA Free | RAMGE Factor | PUDA Freq | RANGE Factor | TEAD Fred | RANGE Factor | LMDA FRED | RANGE FACTOR | AGENT AVAILABLE | LIBS SPILLED | LBS Detonated | LBS [| URATION TINE |
| | ; | | | | | | | | | | | | | | | | | | | | | |
| HCPHS | ~ | N/A | ł | N/A | ; | N/N | : | A/# | : | N/A | ţ | R/N | : | 0.01.00 | ; | A/A | : | 93.6 | : | ; | : | ; |
| HCFVS | 1 | N/A | : | R/A | : | N/A | : | N/A | : | N/A | ; | N/A | ł | 0.06+00 | ; | N/A | : | 6.0. | i | ; | ; | ; |
| HCGGS | • | 4,4 | : | 4/W | : | A/A | ; | N/A | : | N/A | ; | N/A | ; | 0.0E+00 | ; | N/A | ; | 87.0 | ; | ; | ; | ; |
| SVECH | • | A/A | : | #/# | ; | M/A | : | A/M | ; | R/A | ; | M/A | ; | 0.0E+00 | : | N/A | ; | 87.0 | ; | ; | ÷ | ; |
| HCRES | ^ | M/A | : | N/A | ì | N/A | ; | N/A | : | N/A | ; | M/A | ; | 9.65-09 | 1. XE+01 | R/A | ; | 160.5 | 1.16+01 | : | ; | 8HI |
| HCRVS | ~ | K/A | ; | M/A | : | N/A | ; | N/A | 1 | A/A | : | N/A | 1 | 9.66-09 | 1.3€+01 | N/A | ; | 150.0 | 1.05+01 | ; | ; | ž |
| HCSVS | ^ | N/A | ; | N/A | ; | N/A | ; | M/A | : | N/A | ; | 8/8 | ; | 2. 3E 07 | 1.36+01 | N/A | ł | 1356.0 | 1.45+03 | : | ; | ۹. |
| HCINGS | 8 | N/N | : | N/A | ; | N/A | : | N/A | : | N/A | : | A/N | : | 2. 3E - 00 | 1.3E+01 | 2.35-00 | 1.36+01 | 10540.0 | 2.26+02 | : | : | Ŧ |
| HCDMS | | 1, 25-00 | 1. JE +01 | N/A | ; | A/N | ; | N/A | : | 8/N | ; | 1.25-08 | 1.3£+01 | 1.25-08 | 1. 36+01 | A/A | ; | 91608.0 | 6.0€+00 | ; | ; | <u>en</u> |
| HCC6S | æ | 8.86-09 | 1.35+01 | N/A | ; | N/A | : | N/N | ; | W/W | ; | N/N | ; | 8°-38 | 1. 3E+01 | N/A | ; | 1843.2 | 1.46+00 | ; | : | Ŧ |
| HECHS | 80 | 8°.05-09 | 1.X +01 | 8/8 | : | N/A | ; | N/A | : | N/A | ; | 8.0E-09 | 1.3£+01 | 8.86-09 | 1.3£+01 | M/A | : | 3686.4 | 3.26+00 | : | ; | ٩. |
| HCF65 | æ | A/A | ; | H/A | ; | N/A | : | N/A | : | N/A | ; | M/A | ; | 2.26-08 | 1.35+01 | N/A | : | 12000.0 | 1.56+03 | ; | : | ۹. |
| HCHI | 8 | 2.25-08 | 1. JE +01 | 2.25-08 | 1. JE +01 | R/A | : | N/A | ; | 2.2E-08 | 1.36+01 | N/A | : | 2.26-09 | 1.3£+0) | 2.25-08 | 1.3€+01 | 13400.0 | 1.7€+03 | ; | : | Ŧ |
| HCKVS | œ | N/A | ; | N/A | : | N/A | ł | 2.2E-00 | 1.3€+01 | R/A | : | N/N | ; | 2.25-08 | 1.36+01 | N/A | : | 12900.0 | 1.46+03 | : | : | Ŧ |
| HEIRVS | 8 | 1.2E-08 | 1.35+01 | N/A | ; | N/A | ; | N/N | ; | 1.25-08 | 1.35+01 | N/N | ; | 1.26-08 | 1.36+01 | 1.25-08 | 1. 3€ •01 | 4534.0 | 1.06+01 | : | ; | Ŧ |
| HCP65 | 80 | 1, 05-08 | 1.3€+01 | R/A | 1 | N/A | : | 8/N | : | A/A | ; | N/A | : | 1.05-06 | 1.35-01 | 1.06-08 | 1. 3£ +01 | 3120.0 | 6. 3E+00 | : | ; | Ŧ |
| HCPHS | 8 | 1.06-09 | 1.36+01 | N/A | ; | 1.06-00 | 1. 3E+01 | N/A | ; | N/A | : | 1.05-08 | 1.3£+01 | 1.05-08 | 1.36 +01 | N/A | ; | 5416.0 | 1.26+01 | : | : | £ |
| HCPVS | 80 | 1. of - 08 | 1.36+01 | A/A | ł | 1.0E-08 | 1.36+01 | N/A | : | N/A | : | N/A | ; | 1.05-08 | 1.36+01 | 1.05-08 | 1.36+01 | 2990.0 | 6.05+00 | ; | : | ŧ. |
| HCDES | 6 0 | 1.0E-08 | 1. JE +01 | N/A | : | 1.06-08 | 1.36+01 | N/A | ţ | N/A | : | N/A | ; | BO-30.1 | 1. 3E +01 | 1.06-08 | 1.3% +01 | 3480.0 | 10+35-1 | : | ; | Ŧ |
| HCOVS | 80 | N/N | ; | N/A | ; | N/N | : | A/A | : | M /A | ; | N/A | : | 1.06-08 | 1. X +01 | 1.06-08 | 1. 36 +01 | 3480.0 | 1.56+01 | ; | ł | œ. |
| HCR6S | æ | 1. JE-08 | 1. 3E +01 | N/A | ; | 1.36-08 | 1.3€+01 | N/N | ; | 1.35-00 | 1.36+01 | A/A | : | 1. 35 -08 | 1.3€+01 | 1.36-09 | 1. X +01 | 2544.0 | 1.1E+01 | : | : | Ŧ |
| HCRVS | 80 | 1.35-08 | 1.36+01 | N/A | ; | 1.36-00 | 1.3E+01 | N/A | 1 | 1. 36 -08 | 1. 36 +01 | N/A | : | 1.36-00 | 1.36+01 | 1.36-00 | 1.3£+01 | 2400.0 | 1.06+01 | : | : | Ŧ |
| HCSVS | • | N/A | ; | M /M | ; | M/A | : | 8/W | : | #/# | ; | N/N | : | 8°-35-08 | 1. X +01 | 90-35.B | 1.3£+01 | 5424.0 | 1.46+03 | : | : | Ē |
| HC B6F | • | N/N | 1 | N/N | ; | K/A | : | N/A | 1 | A/A | ; | N/A | ; | 8.65-10 | 3. IE+01 | 8.4E-10 | 3. 1E+01 | 10560.0 | ł | : | 2.26+01 | 10 MIN |
| HE HE | a - | 5.4E-10 | 3. IE +01 | N/A | ; | N/A | : | N/A | ; | N/A | : | 5.4E-10 | 3. IE+01 | 5.4E-10 | 3.1E+01 | N/A | ; | 4408.0 | ; | : | 3.06-01 | IO NIN |
| HCC6F | • | 2.65-10 | 3. IE +01 | N/A | : | N/A | : | A/N | ; | N/A | : | A/A | 1 | 2.4E-10 | 3. IE+01 | N/N | ; | 1043.2 | : | ; | 10-37'1 | IO NIN |
| HELEN | • | 2.6E-10 | 3. IE +01 | N/N | : | #/# | ; | M/A | ; | N/N | ; | 2.66-10 | 3.16+01 | 2.6E-10 | 3. IE+01 | A/A | : | 3484.4 | ; | : | 1.45-01 | IO NIN |
| HCKGF | • | N/A | ; | N/N | : | N/A | ; | M/A | : | N/N | ; | N/A | : | 6.95-10 | 3. IE +01 | N/N | : | 12000.0 | ł | : | 1.5€+02 | 10 NIN |
| HCKHF | • | 9.96-10 | 3. IE +01 | 6.96-10 | 3. IE+01 | N/N | : | N/N | ; | 6.9E-10 | 3. IE +01 | N/A | ; | 6.96-10 | 3. IE+01 | 6.96-10 | 3. IE+01 | 13400.0 | : | : | 8.5£+01 | 10 RIM |





ᡩ᠋ᢓᢢᢓᢣ᠋ᠫᡊᢓᢢᢓ᠅ᢓᢣᡷᡧᡬᠯᡷᢓᠵᢓᢢᢓᢣᢓᢢᢓᡸᡘᢓᡸᢓᠧᡬᢓ<u>ᡷᢄᢣ᠋ᡷ᠘ᢤᢄᡔᠺ᠅ᠵ᠘ᡷ᠘ᡷ᠘ᡷ᠘ᡷ᠘ᡷ᠘ᡷ᠘ᡷᠺᡘᠼᡘᠼᡘᠼᡘᠼᡘᡊᠼᡘᡘᡊᡘᡊᡘᡘᡘᡘᡘᡘᡘᡘᡘᡘᡘᡘᡘ</u>

SCOCCE

Koncur



File: MitMillitti, Page 5 Bate 10-Aug-87

NUMBLING ACCIDENTS - INTIONAL PROCESSING OPTION - PER PALLET OR CONTAINER

| | | | | | | Accident | Frequencie | 2 14 2 | nge Factor | | | | | | | | | | TAN Junit | | i Rejese | _ |
|--------------------------|---------|----------------|---------------|------------|-----------------|----------|-----------------|------------|-----------------|-------------|---------------|--------------|-----------------|--------------|-----------------|--------------|-----------|-----------|-------------------|-------------------|-----------|-----------------|
| 100 100 100 100 | ₹ | 2 E 2 S | 38 E | SA BU | RANKE FACTOR | | ANNUE Factor | FREE | RANGE FACTOR | | AMME ACTOR | PUBA Free | RANKE Factor | TEAD FRED | RANGE FACTOR | UNDA FRED | FACTOR | NVALLABLE | SULLED SPILLED | LJIS Detomited | LING | NUMATION 11K |
| | ; ; | • | | ĺ, | | | | | | | | | | | | | | | | } | | |
| NCKVF | à r | 4 | ; | N/N | ; | W/W | 1 | 6.96-10 | 3.16+01 | W /W | : | V.N | ; | 4.76-10 | 3. IE+01 | A/A | : | 12806.0 | : | ; | 1.05+01 | 10 NIN |
| HCMF | 1.5.1 | K-10 J | 10+31 | N/N | : | N/N | ; | 8/8 | 1 | 3. XE-10 | 3.16+01 | N/N | : | 5.36-10 | 3. 16+01 | S. 3E-10 | 3.16+01 | 4534.0 | : | ; | 2.46-01 | IO NIN |
| NCPGF | 9 1.2 | 16-10 J | 16+31 | N/N | ; | N/N | : | N/N | ; | N/N | : | N/N | : | 1.26-10 | 3. 16+01 | 1.25-10 | 3. 1E+01 | 3120.0 | ; | : | 14.33 | 10 818 |
| RUM | • 1.2 | K-10 J. | 16+31 | N/N | ; | 1.25-10 | 3.16+01 | A/N | : | W/W | : | 1.26-10 | 3. IE+01 | 1.26-10 | 3. IE+01 | | : | 5414.0 | : | : | 5. M-01 | 10 NIN |
| KCMF | • | £-10 3. | 16+31 | N/N | 2 | 1.26-10 | 3.1E+01 | A/A | : | | ; | | : | 1.26-10 | 3.16+01 | 1.26-10 | J. 1E+01 | 200.0 | : | ; | 19-35-1 | IO NIN |
| HC06F | 9 1.2 | 16-10 J | 16+91 | N/A | : | 1.25-10 | 3.16+01 | N/N | : | V /I | ; | | : | 1.26-10 | 3.1E+01 | 1.26-10 | 3. (£+0) | 3480.0 | : | : | 1.56.66 | IO NIN |
| HEBVF | Ì | e. | : | N/N | : | N/N | : | W/W | : | 4 /# | ; | N/N | ; | 1.25-10 | 3.16+01 | 1.25-10 | 3. IE+01 | 3440.0 | : | : | 3.45-01 | 10 NIN |
| NCINGF | 4 S.4 | K-10 3 | 10+31 | N/A | : | 5.46-10 | 3.16+01 | N/A | ; | 5.46-10 | 3. IE+01 | N/N | : | 5.66-10 | 3. IE+01 | 5.46-10 | 3.16+01 | 240.0 | ; | : | 1.16+00 | IO NIN |
| NCRVF | 9 5.4 | ¥-10 3. | 10+31. | A/N | ; | 5.46-10 | 3. JE +01 | N/A | ; | 3. AE-10 | 3. 16+01 | W/W | : | 5.66-10 | 3, 16+01 | 5.66-10 | 3. 16+01 | 2409.0 | : | ; | 2.56-01 | IO NIX |
| HCSVF | Ì | ę | ; | N/N | i | A/A | : | A/A | : | A/A | ; | 8/8 | : | 2.7E-10 | 3. IE+01 | 2.76-10 | 3.16+01 | 5424.0 | : | ; | 3.46+01 | 10 NIN |
| HCINES | 10 11/ | é | ; | W/W | : | N/A | : | A/M | : | N/N | : | A/N | ; | 1.16-00 | 1.36+01 | 1.16-00 | 1.36+01 | 10560.0 | 2.26+02 | : | ; ; | Ĩ |
| HC BKS | 10 4.9 | 1 | 3E+01 | K/A | ; | N/A | : | N/N | ; | A/A | 1 | 4-36-9 | 1.36+01 | 60-36-9 | 1.3€+01 | N/A | : | 1600.0 | 6.0£+00 | 1 | ; | ŧ |
| MCC6S | 10 3.4 | 1 60-31 | JE +01 | N/N | : | N/A | ; | N/N | : | N/N | : | W/W | ; | 3.4E-09 | 1.3E+01 | N/A | : | 1043.2 | 1.45+00 | ; | : | Ĩ |
| NCON | 10 3.4 | 1E-09-1. | 10+ X | N/A | : | N/N | ; | N/N | ; | A/A | : | 3.46-09 | 1.35+01 | 3.4E-09 | 1.3€+01 | N/N | ; | 1.486.4 | 3. 7£+00 | : | ; | Ĩ |
| MCK6S | 10 11/ | é | ; | W/W | : | R/A | : | A/N | ; | N/A | ; | N/N | : | 8.06-09 | 1.3E+01 | N/N | : | 12000.0 | 1.56+03 | ; | ; | <u>.</u> |
| HE HE | 10 8.6 | ₩-8 1. | 10+ X. | 0.96-09 | 1.36+01 | N/N | ł | R/A | 1 | 8.86-09 | 1.3E+01 | N/N | ; | 8.86-09 | 1.36+01 | 9.86-09 | 1.3€+01 | 13600.0 | 1.7£+03 | ; | : | Ĩ |
| HCIVS |) 10 | é, | : | W/W | : | N/N | ; | 8°-98 | 1.3£+01 | N/N | ; | N/N | : | 8. 8E - 09 | 1.36+01 | N/N | : | 12800.0 | 1.46+03 | : | 1 | U |
| HEINS | 10 4.8 | ¥-8 | 10+32 | 8/8 | : | N/N | ; | R/A | 1 | 6. DE -09 | 1.3€+01 | N/N | ; | 60-38-9 | 1.3€+01 | 6.05-09 | 1. XE +01 | 6534.0 | 1.0€+01 | : | ; | Ħ |
| HCPGS | 1.1 | £-09 1. | XE +01 | N/A | : | N/A | ; | #/# | ; | A/H | ; | N/N | ; | 1.45-09 | 1.3£+01 | 1.46-09 | 1.3E+01 | 3120.0 | 6.5E+00 | : | : | ١. |
| HCPHS | 10 1.6 | 1 69-34 | 10+ 3E | N/N | : | 1.46-09 | 1.3E+01 | M/A | : | N/N | : | 1.66-09 | 1. JE +01 | 1.66-09 | 1.3€+01 | N/N | 1 | 5616.0 | 1.25+01 | : | ; | Ĩ |
| HEPVS | 10 1.6 | - 8-3 | X +01 | A/A | ; | 1.66-09 | 1. X +01 | K/N | : | N/N | ; | N/N | ; | 1.66-09 | 1.3£+01 | 1.46-09 | 1.3£+01 | 2000.0 | 6.0E+00 | : | : | Ĩ |
| NCINGS | 10 1.4 | | 36+01 | N/N | : | 1.46-09 | 1.3£+01 | N/A | : | W/W | ; | N/N | : | 1.66-09 | 1.3€+01 | 1.46-09 | 1.36+01 | 3480.0 | 1.55+01 | : | ; | Ĩ |
| HERVS | i k | | ; | N/N | ; | N/N | : | N/N | : | W/W | ; | N/N | ; | 1.45-09 | 1.3€+01 | 1.46-09 | 1.3£+01 | 3480.0 | 1.5£+01 | : | ; | Ŧ |
| HCINGS | 10 7.2 | 1 9-3 1 - 9 | 10+3C | N/N | : | 7.26-09 | 10+32.1 | #/# | 1 | 7.26-09 | 1. X +01 | N/A | 1 | 7.2E-09 | 1. XE+01 | 7.25-09 | 1.36+01 | 2349.0 | 1.16+01 | ; | : | Ŧ |
| HEINS | 10 7.2 | - 10-3i | 10+X | 8/8 | : | 7.26-09 | 1.38+01 | N/A | ; | 7.25-09 | 1. XE +01 | N/N | ; | 7.25-09 | 1. JE+01 | 7.26-09 | 1.3%+01 | 2400.0 | 1.0£+01 | : | ; | ۹. |
| NCSVS | 10 | ę | ; | N/N | 1 | R/A | ; | W/A | : | #/# | ł | N/A | 1 | 3.55-09 | 1. 36+01 | 3.56-09 | 1. 35+01 | 5424.0 | 1.46+03 | ; | ; | ¥. |
| KCBHC | 11 2.9 | £-09 2. | 10+ 39 | B/A | : | A/A | : | N/N | ; | N/N | i | 2.96-09 | 2.65+01 | 5. BE - 09 | 2.6€+01 | A/A | ; | 200.0 | : | 6.0E+00 | 1.35-03 | Ť. |
| Nicec | | 1E-09 2. | 10+39 | N/A | : | N/A | 1 | A/A | : | N/N | ; | N/N | ; | 2.96-09 | 2.4€+01 | A/A | : | 1.1 | : | 1.66.00 | 1.46-01 | ¥. |
| HCCHC | | (E-00 2. | 10+ 39 | N/A | : | N/N | ; | N/A | ; | N/A | ; | 1.4E-09 | 2.6E+01 | 2.96-09 | 2.4€+01 | M/A | ; | 74.8 | : | 3. 26+00 | 8. 0E -04 | Ŧ |

File: MCMMI.UKI, Page & Date 18-Aug-87

MANDLING ACCIDENTS - MATLOMML PROCESSING OPTION - PER PALLET OR CONTAINER

ý

102X 001

| | | | | | | Accident | Frequencia | | lange Facto | 5 | | | | | | | | - | Agent Ava | sijale an | i Reieos o d | _ |
|---------|-------------|---------------|---------|-------------|--------------|-----------------|-------------|------------|--------------|---------|-------------|---------|-------------|----------|-----------|----------|-----------|-------------|-----------|-----------------|-------------------------|---------------|
| SCEF B. | ŝ | | ¥ | ¥ | TANKE | M | NAME | | Janua | M | Here | M | Jane | IEN | | MM | | AGENT | 8 | 91 | 8 | UNATION |
| | | | | | | | | | | | | | | | | | | | | | | |
| 1CINC | - | 2 8 -X | 101.31 | | : | 1 /1 | 1 | | 1 | 2.76-00 | 2.45+01 | | 1 | 1.35-00 | 7.45.41 | 2.76-00 | 7.45 401 | 171.0 | : | 3.75 +01 | 1. X-95 | 9 |
| HCrec | = | | 10-31 | | ; | | : | | 1 | | : | N/N | ۱ | 9.46-10 | 2.4E+01 | 4.06-10 | 2.46+01 | 52.0 | ; | 90-35-4 | 1.16-01 | 9 |
| KCMC | = | -10 2 | 10-31 | N/N | 1 | 4. IK-10 | 2.46+01 | W/W | ; | W/W | ; | 4.06-10 | 2.65-01 | 9.46-10 | 2.46+01 | N/N | 1 | 13.4 | ; | 1.75+01 | 1.06-04 | 9 |
| HOMC | = | | 10-39 | 1 /1 | : | 1.06- 10 | 2.66+01 | W/W | 1 | N/N | ł | N/N | ; | 9.46-10 | 2.46+01 | 4. ME-10 | 2.46+01 | 48.0 | ł | 4.05+00 | 1.06-05 | H |
| NCBEC | = 3 | .46-10 2 | 10+37 | R/A | ٢ | 3.46-10 | 2.66+01 | R/N | : | . V/II | : | N/A | : | 7.2E-10 | 2.46+01 | 3.46-10 | 2.6E+01 | 0.70 | : | 1.56+01 | 1.76-01 | Ŧ |
| HERVE | = | N/N | ; | | : | R/A | ł | N/A | : | A/N | ; | N/A | 1 | 7.26-10 | 2.6E+01 | 3.46-10 | 2.4E+01 | 87.0 | ; | 10+35-11 | 90-30.9 | 9 |
| NCINGC | = | .0E-10 2 | 10-31 | N/N | ; | 9,06-10 | 2.66+01 | N/N | ; | 9.06-10 | 2.4E+01 | N/N | ; | 1.85-09 | 2.6E+01 | 9.06-10 | 2.46+01 | 140.5 | : | 2. LE+01 | 1.0€+00 | £ |
| HERVE | = | .06-10 2 | 10+39 | K/N | ١ | 9.0E-10 | 2.46+01 | N/N | : | 9.06-10 | 2.66+01 | N/A | : | 1.16-09 | 2.46+01 | 9.0E-10 | 2.46+01 | 150.0 | ł | 2.06+01 | 5.06-06 | E. |
| NCDHC | 12 2 | IE-10 2 | 10+39 | N/N | ; | R/N | : | N/A | ; | N/N | : | 2.1E-10 | 2.66+01 | 4.16-10 | 2.45+01 | N/N | ; | 200.0 | : | 4.06+00 | 1.36-03 | E |
| NCOSC | 12 | .01-30 | 10+31 | ۲. | ; | N/N | : | N/A | : | W/W | • | N/N | ; | 2. IE-10 | 2.46+01 | N/N | : | 1.2 | 1 | 1.46+00 | 1.46-91 | ¥ |
| HCDK | 1 1 | .01-30. | 10+31 | ¥/H | ; | N/N | : | N/N | ; | N/A | ł | 1.06-10 | 2.46+01 | 2. IE-10 | 2.4E+01 | N/N | ; | 74.8 | 1 | 3.26+00 | 8. K - Ol | # |
| MENVC | 12 | .56-10 2 | 10+31 | W/W | ; | R/A | ; | N/N | ; | 1.36-10 | 2.6E+01 | N/N | ; | 3.16-10 | 2.6E+01 | 1.5E-10 | 2.46+01 | 370.0 | ; | 3.26+01 | 4. X-05 | Ŧ |
| NCPGC | 12 | .46-11 2. | 10-31 | N/N | ; | N/N | : | N/N | 1 | N/N | 1 | N/N | ; | 11-36.4 | 2.6E+01 | 3.46-11 | 2.6E+01 | 52.0 | ; | 6.36+00 | 1.15-01 | Ť. |
| JH4JW | 23 | .46-11 2 | 10+39. | N/N | 1 | 3.46-11 | 2.46+01 | N/N | ; | N/N | : | 3.46-11 | 2.4£+01 | 11-36.6 | 2.6E+01 | R/A | : | 93.4 | : | 10+32.1 | 7.05-04 | 1 |
| HCPVC | 2 | .46-11 2 | 10+39 | N/A | ; | 3.46-11 | 2.46+01 | N/N | ; | N/N | ; | N/N | ! | 6.96-11 | 2.6€+01 | 3.4E-11 | 2.6E+01 | H .0 | 1 | 6.0E +00 | 1. 05-95 | Ŧ |
| HC GEC | 12 2 | .46-11 2 | 10-31 | #/# | : | 2.46-11 | 2.46+01 | N/A | : | N/N | : | W/H | 1 | 5.2E-11 | 2.6E+01 | 2.46-11 | 2.46+01 | 0.70 | 1 | 1.56+91 | 1. 76-01 | E |
| HCENC | 2 | #/# | 1 | R/A | ١ | N/N | ł | N/A | ; | N/N | 1 | W/W | ; | 5.2E-11 | 2.46+01 | 2.46-11 | 2.46+01 | 0.70 | 1 | 1.56+01 | 10-31.4 | E - |
| NCNGC | 12 6 | .SE-11 2 | 10+39. | W/W | : | 11-3C-11 | 2.6€+01 | N/N | : | 11-35.4 | 2.6E+01 | W/W | 1 | 1.3%-10 | 2.45+01 | 11-3C-11 | 2.4E+01 | 140.5 | ; | 2. IE+01 | 1.0€+00 | 5 |
| NCINC | - - - | 56-11-2 | 10+34. | | ï | 11-36-91 | 2.6€+01 | N/N | 1 | 4.56-11 | 2.66+01 | | : | 1.3£-10 | 2.66+01 | 6.5E-11 | 2.66+01 | 150.0 | : | 2.0€+01 | 5. 66-46 | ¥ |
| HCMCC | 2 | N/N | 1 | ¥/¥ | ١ | W/W | : | N/N | : | N/N | ; | W/W | ; | 4.0E-13 | 2.6E+01 | 6.0E-13 | 2.4E+01 | 440.0 | ; | ; | 7.86-01 | Ĩ |
| MCREC | • | .0E-13 2. | 10+31 | N/A | ; | 4.06-13 | 2.46+01 | N/A | ; | 6.0E-13 | 2.6E+01 | N/A | ł | 4.0E-13 | 2.4€+01 | 6.06-13 | 2.46+01 | 140.5 | : | : | 1.46-01 | Ĩ |
| NCIPYC | 4 | .06-13 2. | 10+39 | R/A | : | 6.0E-1 3 | 2.46+01 | N/A | : | 6.0E-13 | 2.6E+01 | N/N | ł | 6.0E-13 | 2.65+01 | 4.0E-13 | 2.6£+01 | 150.0 | ł | : | 4. 794 | Ē |
| HCDEC | = | N/N | 1 | 1 /1 | ; | N/A | : | N/N | : | N/N | 1 | N/N | ; | 1.26-12 | 2.65+01 | 1.28-12 | 2.6€+01 | 220.0 | ł | : | 7.06-01 | Ē |
| MCINGC | 8 | .26-12 2 | 10-39 | N/N | ; | 1.21-12 | 2.4E+01 | N/A | ; | 1.26-12 | 2.46+01 | N/N | 1 | 1.21-12 | 2.46+01 | 1.21-12 | 2.46+01 | 10.7 | : | 1 | 1.46-01 | - |
| HCRVC | | .212 2. | 10+39 | N/N | ; | 1.26-12 | 2.4E+01 | N/N | : | 1.2£-12 | 2.4E+01 | N/N | ; | 1.26-12 | 2.6£+01 | 1.25-12 | 2.46+01 | 10.0 | ; | 1 | 4. 75-04 | Ĩ |
| HC BGC | • | N/N | : | N/N | } | W/W | ; | W/W | : | #/# | ; | N/N | ; | 2.66-15 | 3. 1E+01 | 2.46-15 | 3. 16+01 | 440.0 | ł | : | 7.16-01 | Ĩ |
| HCRGC | 19 2. | . 66-14 3. | . IE+01 | N/N | 1 | 2.66-14 | 3. IE +01 | N/N | 1 | 2.46-14 | 3. IE +01 | N/N | 1 | 2.66-11 | 3. IE +01 | 2.66-14 | 3. IE+01 | 160.5 | : | : | 10-31-1 | Ĩ |
| HCRVC | 19 2. | " FI-39" | 1E+01 | K/A | ł | 2.6E-14 | 3. IE+01 | N/A | 1 | 2.46-14 | 3. IE+01 | N/A | : | 2.66-14 | 3.16+01 | 2.46-14 | 3. IE+01 | 150.0 | ; | ; | 4.75-04 | 2 |
| NC DOC | ≈ | NA | 1 | #/# | ; | N/A | 1 | W/W | : | ¥/# | ł | W/W | : | 3.4E-18 | 3. 16+01 | 3.4E-1B | 3. IE •01 | 220.0 | ; | : | 7.05-01 | Ē |

non hand hill bound a start of a start of the start of th

Person ANNER TRUE WAY

ፚኇፚኇፚኯፚኇዄኇዄጛዄዀዀዀዀዀዀዀዀዀዀዀ







NAMALING ACCIDENTS - NATIONAL PROCESSING OFTION - PER PALLET ON CONTAINER

Accident Frequencies and Range Factors

Agent Available and Released

| bunation 11/K | | Ĩ | Ĩ | Ĩ | ۹. | Ē | Ĩ | Ĩ | Ĩ | Ĩ | Ĩ | Ĩ | Ē | Ĩ | Ĩ | Ē | Ħ | ŧ, | Ĩ | ij | Ē | 2 | Ē | Ē | 2 | Ē | Ē | Ē | Ē | Ē |
|---------------------------|---|------------|----------|----------------|----------|----------|------------|------------|------------|-------------|------------|---------|----------|--------------|---------|----------|--------------|----------|----------------|----------|----------|-----------|-----------|----------------|------------------|----------|----------|----------|---------|---------|
| LDS Entted | | 1.46-01 | 1.76-06 | : | : | ; | : | : | ; | : | : | ; | ł | : | : | : | : | : | : | : | : | ; | : | ; | : | ; | : | 1 | : | ; |
| LIIS ETOMATED | | 1 | : | 4.06+00 | 1.46+00 | 3.2£+00 | 3.26+01 | 4.5£+00 | 1.26+01 | 4.0E+00 | 1.52401 | 1.5€+01 | 2. IE+01 | 2.0E+01 | 4.05+00 | 1.45+00 | 3.25+00 | 3.26+01 | 6.5E+00 | 10+32.1 | 4.05+00 | 10-35-1 | 1.5€+01 | 2.16+01 | 2.0€+01 | 4.0€+00 | 1.611 | 3.26+00 | 3.26+01 | 6.3E+00 |
| | | ł | : | 3.0E+01 | B. 0E+00 | 1.45+01 | 3.5£+02 | 3.26+01 | 5. EE +01 | 3.0E+01 | 7.36401 | 7.3E+01 | 1.4€+02 | 1.3€+02 | 3.0€+01 | B. 0E+00 | 1.4£+01 | 3.5£+02 | 3.26+01 | 5.86+01 | 3.0€+0? | 7.3€+01 | 7.X +01 | 1.46+02 | 1. X +02 | 3.06+01 | B. 0£+00 | 1.66+01 | 3.5£+02 | 3.25+01 |
| AGENT | | 10.7 | 10.0 | 2 10. 0 | 7.17 | 74.8 | 378.0 | 52.0 | 13.4 | 8 .0 | 87.0 | 0.10 | 140.5 | 156.0 | 1152.0 | 1.11 | 4.124 | 1134.0 | 700.0 | 1404.0 | 720.0 | 870.0 | 870.0 | 472-0 | 6. MJ | 0.M | 4. 1 | 74.0 | 378.0 | 52.0 |
| ANDE ACTOR | | 3.16+01 | 3.1E+01 | : | ; | : | ; | ; | 1 | : | : | 1 | ; | : | : | : | 1 | 2.6E+01 | 2.46+01 | 1 | 2.6E+01 | 2.66+01 | 2.46+01 | 2.6E+01 | 2.4E+01 | : | : | I | : | : |
| UNIA LAED | | 3.16-18 | 3.16-10 | • | W/W | | W/W | N/A | W/W | N/A | R/A | N/N | N/N | N/N | N/A | A/A | N/A | 11-36-11 | 11-31-11 | A/N | 11-34-11 | 7.26-12 | 7.21-32.1 | 7.26-12 | 7.26-12 | N/A | N/N | N/N | N/N | N/N |
| ACTOR | | 3.16+01 | 3.16+01 | 2.46+01 | 2.46+01 | 2.46+01 | 2.45+01 | 2.46+01 | 2.46+01 | 2.46+01 | 2.66+01 | 2.66+01 | 2.46+01 | 2.46+01 | 2.46+01 | 2.46+01 | 2.46+01 | 2.46+01 | 2.6E+01 | 2.46+01 | 2.4E+01 | 2.46+01 | 2.46+01 | 2.46+01 | 2.66+01 | 2.45+01 | 2.46+01 | 2.46+01 | 2.46+01 | 2.46+01 |
| TEAN R | | 3. IE-10 | 3.16-10 | 1.76-11 | 1.36-11 | 11-36-11 | 11-35-11 | 1.4E-11 | 11-34-11 | 11-34-11 | 11-31-1 | 1.16-11 | 2.76-11 | 2.76-11 | 2.36-11 | 3.56-11 | 3.56-11 | 1.38-11 | 11-39-11 | 11-31-11 | 1.46-11 | 7.25-12 | 7.25-12 | 7.26-12 | 7.26-12 | 11-37.9 | 3.16-11 | 3. IE-II | 11-31-1 | 11-30.1 |
| ANDE ACTOR | | ł | ; | ; | ; | : | ; | : | : | ; | : | : | ; | : | 2.45+01 | : | 2.66+01 | : | ł | 2.46+01 | : | ; | 1 | ; | : | : | ł | ; | : | : |
| | | W/W | N/N | N/N | N/N | A/A | N/N | N/N | N/A | N/A | N/A | N/N | V. | N/N | 2.36-11 | N/N | 11-35.5 | W.W | N/N | 11-31-11 | N/A | N/N | N/A | N/N | #/# | N/N | N/N | N/A | W/W | N/N |
| MER KCTON | | 3. IE +01 | 3. 16+01 | ; | ł | : | : | ; | ; | ; | : | ł | ; | 1 | : | ; | : | 2.66+01 | ; | ; | : | : | ; | 2.46+01 | 2.44.401 | : | 1 | : | : | : |
| | | 3. IE-10 | 3.16-10 | W/W | V/I | W/W | N/N | N/N | A/A | N/N | N/N | N/A | W/W | A/A | VI | N/A | R/A | 11-36-11 | N.N | N/N | N/N | R/N | Va | 7.26-12 | 7.26-12 | N/N | N/N | N/N | e/e | N/N |
| N N | | : | ł | : | ; | ; | ! | : | ; | : | : | : | ; | : | : | : | ; | 1 | 1 | : | : | 1 | ł | : | ł | ÷ | : | ; | : | 1 |
| | | N/A | N/A | W/W | W/W | N/N | A/A | N/N | N/N | N/A | N/N | N/N | N/A | W/W | W/W | N/N | H/A | N. | N/N | N/N | N/N | N/N | N/N | N/N | N/N | W/W | N/N | N/N | N/N | N/N |
| | | 10+31.5 | 3. 1E+01 | ; | 1 | : | ; | : | : | : | : | : | ; | ł | ; | ; | ; | ; | ł | 2.4£+01 | 2.45+01 | 2.46+01 | ; | 2.46+01 | 2.66-01 | ; | ; | ; | : | ; |
| | | 11E-10 | 11-31.1 | N/N | N/N | W/W | N/N | N/N | N/N | N/ | N/N | N/N | V/R | N/N | N/N | R/A | V. | W/W | V / | 11-34-1 | | . 21-32.1 | N/N | . 21-32.1 | . 21-32. | ×. | N/N | N/N | N.N | N/A |
| | | ; | 1 | ; | ; | ١ | : | : | : | 1 | : | ; | : | ; | ; | ł | : | ; | ; | ; | : | ; | : | 1 | ; | ; | ł | ; | : | ; |
| 8 2 8 2 | | 8/8 | N/N | N/N | R/A | N/N | N/N | 8/8 | N/A | R/A | - | N/N | N/N | N/N | N/A | N/A | N/N | N/N | N/A | 8/8 | W/W | N/N | AVA | N/N | N. | N/N | N/N | N/N | N/A | 8/8 |
| | | (, 1E+01 | l, IE+01 | ; | 1 | : | ; | : | ; | : | : | ; | ; | : | 1.46+01 | 10+34-1 | 10+34-1 | 10+34-0 | 10+39" | 10+34-1 | 10+34.1 | 10-31- | ; | 10+31. | 10-39.1 | ; | ; | ; | ; | ; |
| | | 1E-10 | 11-11 | 1/1 | 1/1 | N.A | 1/1 | N.A | N.A | N.N | | 14 | 1.4 | 1/1 | 36-11 2 | 5 11-35 | S-11-35 | 3 II-3 | 2 II-3F | 4E-11-34 | 11-34 | X-12 2 | 1/A | Z-12 2 | X-12 2 | N.A. | 2 | | 1 | 1/1 |
| | : | 21 3. | 21 3. | 22 | 2 | 2 | 22 | 22 | 22 # | 22 | 22 | 22 | 22 | 22 | 23 2. | 23 3. | 23 3. | 23 1. | 23 1. | 23 1. | 23 1. | 23 7. | 53 | 23 7. | 23 7. | 51 W | 3 | 1 | * | × |
| 5054- 8 7 19810 | | NCNEC | NCINC | N D N C | NCCOC | NCINC | NCIIINC | ICTEC | ENC | KCPVC | NC INCL | KCINC | ICING | NCINC | | NCCEC | HCCHC | HCIINC | 154CH | JNLJN | HEPVE | NC DEC | HCOVC | NCNEC NCNEC | HCIINC HCIINC | X | ACCEC | NCNC | HCIEVC | KCPGC |

File: HRUMAT.UKI, Page B Date 18-Aug-87

HAMDALING ACCIDENTS - NATIONAL PROCESSING OPTION - PER PALLET OR CONTAINER

| SEEF. U. MAIL FREB FALL ANIO FREB FALL FREB FALL ANIO FREB FALL FALL FALL ANIO 23 MAI ANIO 23 23 ANIO 23 23 ANIO 23 ANIO 23 ANIO 23 | APG R/A | RANKE Control | W | ; | | | | AMOR | VIN | | | NAIS | | PANCE | NGENT | | ä | ¥ | |
|--|-------------|------------------|----------|------------------|-------------|------------------|-------------|---------|------------|---------|----------|-------------|-------------|---------|-------------|-----------|--------------------|---------|----------------|
| NCNC 74 NA NA NCNC 74 NA NA NCNC 74 NA NA NCNC 74 NA NA NCNC 74 NA NA NCNC 74 NA NA NCNC 74 NA NA NCNC 74 NA NA NCNC 74 NA NA NCNC 74 NA NA NCNC 75 2.55-11 2.46-01 NA NCNC 75 2.56-11 2.46-01 NA NCNC 75 1.06-11 2.46-01 NA NCNC 75 </th <th></th> <th></th> <th>FREG</th> <th>FACTOR FACTOR</th> <th></th> <th>FACTOR FACTOR</th> <th></th> <th>AC 108</th> <th></th> <th>FACTOR</th> <th></th> <th>ACTOR</th> <th>3</th> <th>FACTOR</th> <th>NALLADLE</th> <th>9111E</th> <th>LUSS DE TOMATED</th> <th>ENTITED</th> <th>MATION 11ME</th> | | | FREG | FACTOR FACTOR | | FACTOR FACTOR | | AC 108 | | FACTOR | | ACTOR | 3 | FACTOR | NALLADLE | 9111E | LUSS DE TOMATED | ENTITED | MATION 11ME |
| NEME 24 NA NEWE 24 NA NA NEWE 24 NA NA NEWE 24 NA NA NEWE 24 NA NA NEWE 24 NA NA NEWE 24 NA NA NEWE 23 NA NA NEWE 25 1.25 1.26 NA NEWE 25 2.55 1.26 NA NEWE 25 1.06 1.26 NA NEWE 25 1.06 1.26 NA NEWE 25 1.06 1.06 NA NEWE 25 NA 2.46 | R/R R/R | | | | | | | | | | | | | | | | | | |
| HCMC 73 HCM 74 74 HCMC 23 HCM 74 74 74 HCMC 23 HCM 74 74 74 74 HCMC 23 HCM 75 HCM 74 74 74 HCMC 23 HCM 25 LSE-11 2.46-01 101 HCMC 25 2.55 L.56-11 2.46-01 101 101 HCMC 25 2.55 L.56-11 2.46-01 101 101 HCMC 25 1.06-11 2.46-01 10 | R/A | ; | R/A | ; | N/N | : | N/N | ; | R/A | ; | 1.06-11 | 2.46+01 | R/N | : | 13.4 | 5.0€+01 | 1.25+01 | : | Ĩ |
| HORE 23 V/A V/A HCRU 23 V/A V/A HCRU 24 V/A V/A HCRU 23 V/A V/A HCRU 23 V/A V/A HCRU 23 V/A V/A HCRU 23 V/A V/A HCRU 25 2.52-11 2.46-01 V/A HCRU 25 2.52-11 2.46-01 V/A HCRU 25 1.06-11 2.46-01 V/A | | ; | R/A | ł | N/N | : | W. | : | W/W | ; | 1.06-11 | 2.46+01 | 8/8 | ; | 48.0 | 3.06.401 | 4.06+00 | : | ٩. |
| NCMC 23 NA NA NCMC 23 NA NA NA NCMC 23 L/T-L1 2.46-01 NA NA NCMC 25 L/T 2.46-01 NA NA NCMC 25 L/T 2.46-01 NA NA NCMC 25 L/T 2.46-01 NA NA NCMC 25 L/T <td>N/A</td> <td>:</td> <td>A/A</td> <td>:</td> <td>N/N</td> <td>:</td> <td>۲.</td> <td>;</td> <td>N/A</td> <td>;</td> <td>7.76-12</td> <td>2.46+01</td> <td>A/N</td> <td>:</td> <td>87.0</td> <td>7.3€+01</td> <td>1.56+01</td> <td>:</td> <td>Ĩ</td> | N/A | : | A/A | : | N/N | : | ۲. | ; | N/A | ; | 7.76-12 | 2.46+01 | A/N | : | 87.0 | 7.3€+01 | 1.56+01 | : | Ĩ |
| NCR6 24 NA NO NCPUC 24 NA NO NCDNC 24 NA NO NCDNC 25 1.04 NO NCDNC 25 2.55-11 2.46-01 NO NCNC 25 2.55-11 2.46-01 NO NCNC 25 1.06-11 2.46-01 NO NCNC 25 NA NO | N/N | : | N/N | 1 | R/N | : | W/W | : | N/N | : | 7.76-12 | 2.46+01 | N/N | : | 87.0 | 7. 36+01 | 1.56.01 | : | Ĩ |
| HCINC 23 1, 71 2, 460 1, 71 HCINC 25 1, 2, 71 1, 2, 460 1, 10 HCINC 25 2, 25 11 2, 460 1, 10 HCINC 25 2, 25 11 2, 460 1, 10 HCINC 25 2, 25 1, 2, 460 1, 10 1, 10 HCINC 25 2, 25 1, 2, 460 1, 11 1, 10 1, | N/N | ٢ | N/N | 1 | N/N | : | . W/M | : | N/N | ; | 11-34.1 | 2.46+01 | R/A | : | [60.5 | 1.4€+02 | 2.16+01 | : | Ĩ |
| HOME 25 1.75-11 2.46-01 NO HCORE 75 2.55-11 2.46-01 NO HCORE 75 2.55-11 2.46-01 NO HCME 75 1.06-11 2.46-01 NO HCME 75 1.06 1.06 NO NO HCME 75 1.06 1.06 1.06 NO HCME 75 1.06 1.06 1.06 NO HCME 76 1.06 1.06 1.06 NO NO < | N/A | ; | N/N | ; | R/A | : | A/A | ; | A/A | : | 1.96-11 | 2.66+01 | R/A | : | 159.0 | 1. 其+02 | 2.06+01 | ; | Ŧ |
| HCGG 25 2.56-11 2.66-01 101 HCMC 25 2.56-11 2.66-01 101 HCMC 25 1.06-11 2.66-01 101 HCMC 25 1.06-11 2.66-01 101 HCMC 25 1.06-11 2.66-01 101 HCMC 25 1.06-11 2.66-01 101 HCMC 25 5.76-12 2.66-01 101 HCMC 25 8.1/A 0.01 HCMC 25 8.1/A 0.01 HCMC 26 8. | A/A | : | N/N | ; | N/A | : | N/N | ; | 11-36-11 | 2.46+01 | 11-37.1 | 2.45+01 | N/N | : | 1152.0 | 3.0€+01 | 4.00+00 | : | ŧ |
| HCHC 25 2.36-11 2.46-01 10 HCHC 25 9.36-12 2.46-01 10 HCHC 25 1.06-11 2.46-01 10 HCHC 25 1.06-11 2.46-01 10 HCHC 25 1.06-11 2.46-01 10 HCHC 25 5.76-12 2.46-01 10 HCHC 25 5.76-12 2.46-01 10 HCHC 25 10 10 HCHC 25 10 10 HCHC 26 10 10 HCHC | N/A | : | N/A | : | 8/8 | : | N/N | ; | N/N | : | 11-35-2 | 2.66 401 | A/A | ; | 1.61 | B. 6E+00 | 1.46+00 | : | ŧ. |
| NEMC 75 7.36-61 100 NEPRC 75 1.06-11 2.46-61 101 NEDRC 75 5.76-12 2.46-61 101 NEDRC 75 1.06 101 101 NEDRC 75 1.06 101 101 NEDRC 26 101 101 101 NEDRC 26 101 101 | N/N | ; | N/N | ł | W/W | : | V. | : | 2.56-11 | 2.46+01 | 2.56-11 | 2.66+01 | V. | 1 | \$21.6 | 1.46+01 | 3.26+00 | : | Ĩ |
| HCPRC 25 1.06-11 2.46-01 101 HCPVC 25 5.76-12 2.46-01 101 HCPVC 26 101 101 101 1 | A/A | : | W/W | ł | N/A | : | 9. X-12 | 2.46+01 | 8/8 | ; | 9.36-12 | 2.46+01 | 9. 36-12 | 2.46+01 | 1134.0 | 3.5€+02 | 3. X +01 | : | 2 |
| NEMC 25 1.06-11 2.46-01 NI NEPMC 25 1.06-11 2.46-01 NI NEDMC 25 5.26-12 2.46-01 NI NEDMC 25 5.26-12 2.46-01 NI NEDMC 25 5.26-12 2.46-01 NI NEMC 25 5.27-12 2.46-01 NI NEMC 25 NI NI NEMC 25 NI NI NEMC 25 NI NI NEMC 26 NI NII NEMC | N/A | ; | N/N | ; | 1 /1 | : | N/N | ; | N/A | ; | 11-30-11 | 2.46+01 | 11-30"1 | 2.46+01 | 780.0 | 3.26+01 | 6.5£+00 | ; | Ē |
| HEVIC 25 1.06-11 2.46-01 101 HEVIC 25 5.76-12 2.46-01 101 HEVIC 25 5.76-12 2.46-01 101 HEVIC 25 5.76-12 2.46-01 101 HEVIC 25 5.76-12 2.46-01 101 HEVIC 25 101 101 HEVIC 25 101 101 HEVIC 26 101 101 101 HEVIC 26 101 101 101 101 101 101 101 101 101 10 | A/A | ; | 1.06-11 | 2.46+01 | N/N | : | A/A | 1 | 1.06-11 | 2.46+01 | 1.01-11 | 2.46-01 | W/W | ; | 1404.0 | 5.66+01 | 1.25.401 | : | Ē |
| NGMC 25 5.26-12 2.46-01 VI NCMC 25 5.76-12 2.46-01 VI NCMC 25 1/A | A/A | : | 11-30-1 | 2.46-01 | N/N | ; | R/A | : | N/A | : | 11-30-1 | 2.46+01 | 11-30.1 | 2.46+01 | 720.0 | 3.05+01 | 1.05-00 | ; | Ĩ |
| HCPPC 25 11/1 | N/N | ; | 5.26-12 | 2.4€+01 | N/N | ; | N/N | ; | - | : | 5.26-12 | 2.46+01 | 5.26-12 | 2.46+01 | 070.0 | 7. 16+01 | 1.5€+01 | ; | Ē |
| HCMC 25 5.75-12 2.45-01 VI HCMC 25 5.75-12 2.45-01 VI HCMC 25 1.75-12 2.45-01 VI HCMC 26 VIA 11 VIA HCMC 26 VIA 11 VIA 11 VIA HCMC 26 VIA 11 VIA 11 VIA HCMC 26 VIA 11 VIA | N/N | : | N/A | ł | N/N | : | N/N | : | N/N | ; | 5.26-12 | 2.46+01 | 3.26-12 | 2.46+01 | 870.0 | 7. XE +01 | 1.56+01 | ; | Ħ |
| HCMC 25 5.75-12 2.46-01 40 HCMC 25 4.44 11 14 HCMC 26 4.44 11 14 HCCMC 26 4.44 11 14 14 14 14 14 14 14 14 14 14 14 1 | A/A | ; | 5.26-12 | 2.46+01 | N/N | 1 | 5.26-12 | 2.46+01 | 8/8 | ł | 5.21-12 | 2.45+01 | 5.26-12 | 2.46+01 | M2.0 | 1.4€+02 | 2.16+01 | : | H. |
| KOM 25 1/4 KOM 26 1/4 1 KOM 26 1/4 1 1 KOM 26 1/4 1 1 1 KOM 26 1/4 1 1 1 1 KOM 26 1/4 1 <t< th=""><td>A/A</td><td>:</td><td>5. 26-12</td><td>2.46+01</td><td>R/R</td><td>:</td><td>5.25-12</td><td>2.46+01</td><td>A/A</td><td>;</td><td>5.26-12</td><td>2.46+01</td><td>S. 26-12</td><td>2.66+01</td><td>6.00.0</td><td>1. 36+02</td><td>2.06+01</td><td>;</td><td>Ĩ</td></t<> | A/A | : | 5. 26-12 | 2.46+01 | R/R | : | 5.25-12 | 2.46+01 | A/A | ; | 5.26-12 | 2.46+01 | S. 26-12 | 2.66+01 | 6.00.0 | 1. 36+02 | 2.06+01 | ; | Ĩ |
| ACMC 26 10 ACCAC 26 10 11 | W/W | ; | N/A | ; | R/A | 1 | R/R | : | 8/8 | : | 0.0[+00 | : | W/W | : | 440.0 | : | : | : | ; |
| HCCR 25 U/A 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | R/A | ; | E/A | ; | #/# | ; | N/N | ; | R/A | ; | 0.0£+00 | : | W/W | : | 9.82 28 | : | : | : | ; |
| HCHC 23 11/4 12 HCHC 24 11/4 11/4 11/4 11/4 11/4 11/4 11/4 1 | W/W | : | 8/8 | ; | R/A | : | A/# | ; | N/N | ; | 0.0€+00 | ; | N/N | ; | 4.9X | 1 | ; | : | 1 |
| NG56 25 1/A 1 1/A NC167 25 1/A 1 1/A 1 NC167 25 1/A 1 1/A 1 1/A NC167 26 1/A 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | W/W | : | 8/8 | ; | N/A | : | R/N | ; | N/N | : | 0.0£+00 | : | 1 /1 | : | 74.0 | ; | : | : | : |
| HETHE 26 K/A V/A HETHE 26 K/A V/A HETHE 26 K/A V/A HETHE 26 K/A V/A HETHE 26 K/A V/A | | : | N/A | : | M/A | 1 | V. | ; | N/A | ; | 0.05+04 | ; | R/A | : | 1500.0 | : | : | : | ; |
| HERVE 26 H/A H/A HERVE 26 H/A H/A HERVE 26 H/A H/A HERVE 26 H/A H/A | N/N | : | N/N | ; | W/W | : | V | : | W/W | : | 0.06+00 | : | N/N | ; | 1700.0 | 1 | 1 | ; | ; |
| NCIPC 24 N/A N/A NCPGC 24 N/A N/A NCPNE 24 N/A N/A | 8/8 | : | 8/8 | ; | N/N | ł | 1 /1 | ; | N/A | : | 0.0€+00 | : | N/N | ; | 164.0 | ; | : | ; | ; |
| NCPSC 26 N/A N/A NCPMC 24 N/A N/A | 8/8 | ; | | : | N /N | ; | N. | : | N/A | : | 9.0£+00 | ŧ | N/N | : | 378.0 | ; | : | : | : |
| NCPHC 74 N/A N/A | N/N | : | R/A | ; | #/# | ; | | ; | K/A | ŧ | 0.05+06 | ; | 8/8 | ; | 52.0 | 1 | 1 | ; | : |
| | R/A | ; | N/N | : | #/# | 1 | U/N | : | N/A | ; | 0.05+00 | : | A/M | : | 13.4 | ; | ; | : | : |
| NCPVC 26 N/A N/A | 1 /1 | ł | A/A | ; | N/N | ; | A.M | : | N/N | ; | 0.05+00 | : | N/N | : | 1 .0 | ; | { | 1 | ; |
| MCB6C 26 M/A M/A | W/W | 1 | 8/8 | : | N/N | : | N/A | : | N/N | ; | 0, 05+90 | : | 8/8 | : | 07.0 | : | ; | : | : |

SSN 85



D.C.





HHMOLING ACCIDENTS - MATTOMAL PROCESSING OPTION - PER PALLET ON CONTAINER

| | | | Accudent | t Frequenc | The and | tange Fact | | | | | | | | | - | igent Avai | Liable and | faleset | |
|---|---------------------|-----------------|----------|----------------|---------------------|-----------------|-------------|-----------------|---------|--------|--------------|-----------------|------------|---------|-------------------|-------------------|-----------------|----------|----------------|
| | APG Fixed | RANKE Factor | | RANK FACTOR | | RANGE FACTOR | | RANGE Factor | PUIM | FACTOR | TEAD Free | RANGE Factor | | NCTOR | AGENT MALLADLE | LINS SPILLED I | LPS DEVINTED | LDS D | MATION TINE |
| | | ! | | | | | | | | | | | | | | | | | |
| | N/N | ; | N/N | ; | R/A | : | N/A | ł | #/# | : | 0.0E+00 | : | N/A | ; | 87.0 | ; | ; | : | ; |
| | N/N | : | N/N | 1 | | : | 5 | ł | N/N | : | 0.0£+00 | ł | N/N | : | 140.5 | : | ; | ; | ; |
| | A/A | : | R/A | ; | H/N | : | W.W | : | A/A | : | 0.06+00 | ; | N/A | : | 150.0 | ł | : | : | ; |
| | N/N | : | N/N | : | W/W | : | N/N | ; | W/W | : | 0.0£+00 | ł | N/N | : | 1336.0 | : | ł | : | : |
| | U /N | : | N/N | : | N/N | ; | N/N | : | N/N | : | 0.0£+00 | : | 0.0€+00 | : | 2640.0 | : | : | : | : |
| | N/N | ; | W/W | : | A/A | : | V/I | ; | 0.0€+00 | : | 0.0£+00 | ł | N. | : | 1152.0 | ; | : | ; | : |
| | N/N | ; | N/N | : | V / H | : | W/W | ; | Ņ | 1 | 0.06+00 | : | U/N | : | 1 .041 | ; | ; | ; | ; |
| | N/N | ; | | : | N/N | : | N/N | ; | 0.05+10 | 1 | 0.0€+00 | ł | N/N | : | 1.124 | 1 | : | : | ; |
| | 1 /1 | ; | N/N | : | N/N | : | N/N | : | A/A | : | 0.0€+00 | ; | R/A | ł | 3000.0 | ; | ; | ; | 1 |
| | 0.06+ | : | W/# | ; | R/A | : | 0.05+#0 | ; | N/N | ľ | 0.0£+00 | ; | 0.0£+00 | ; | 3400.0 | : | : | : | ; |
| | N/N | ł | N/N | 1 | 0.06+0 | : | A/A | ; | N/A | ; | 0.0€+00 | t | A/A | : | 5200.0 | : | ; | : | ; |
| | V / I | ; | N/N | : | 4 / 1 | ; | 0.06+90 | 1 | N/N | : | 0.0E+00 | : | 0.05+00 | 1 | 1134.0 | : | 1 | : | ; |
| | A/A | ; | N/N | : | 4 /1 | ; | R/A | 1 | N/N | ; | 0.0£+00 | : | 00+30.0 | : | 700.0 | : | : | ; | : |
| | N/N | : | 0.00+0 | : | N/N | 1 | N/N | ; | 0.0E+00 | ; | 0,0E+00 | : | N/N | : | 1404.0 | : | ; | : | : |
| ! | A/H | : | 0.00+00 | : | N/N | : | N/N | ; | N/A | : | 0.0E+00 | : | 0.06+00 | : | 720.0 | ł | : | : | : |
| | W/W | ; | 0.06+00 | : | 8 /8 | : | N/N | ł | W/W | : | 9.0£+00 | ł | 0.0£+00 | ; | 879.0 | : | 1 | ; | ; |
| | N/N | ; | W/W | : | N/N | ; | N/N | : | N/N | : | 0.00+00 | ł | 0.05+00 | ł | 870.0 | ł | ; | ł | : |
| | W/W | ; | 0.0E+00 | : | W/N | ł | 0.0£+00 | : | N/N | : | 0.0E+00 | ł | 0.05+20.0 | ; | 642.0 | ; | : | ; | : |
| • | A/H | ; | 0.06+00 | : | A/A | : | 0.00+00 | : | R/A | : | 0.0€+00 | : | 0.0£+00 | ; | 6.004 | : | : | : | : |
| : | N.N | : | N/N | : | N/N | : | V /I | : | N/A | : | 00+30-0 | ; | 0.0€+00 | : | 1354.0 | : | ; | : | : |
| 3 | N N/N | : | 9.06-04 | 3 2.65+01 | N/N | : | 9.06-09 | 2.46+61 | N/N | : | 9.06-09 | 2.6€+01 | 9.06-09 | 2.66+01 | 160.5 | 1 | 2. IE+01 | 7. IE-02 | |
| : | NI N/A | : | 9.06-01 | 1 2.46+01 | B/A | : | 9.06-09 | 2.46+01 | N/N | : | 9.06-09 | 2. 6E+01 | 9.06-09 | 2.46+01 | 150.0 | : | 2.06+01 | 1.06-03 | ¥ - |
| 3 | N. N.A | ; | 1.86-00 | 1 2.46+01 | NIA | : | 1.85-98 | 2.46+01 | N/A | : | 1.06-09 | 2.65 401 | 1.06-30.1 | 2.66+01 | 10.7 | : | 1.11.01 | : | ISHI |
| Ξ | NI 1/4 | ; | 1.95-00 | 1 2.46+01 | V/R | : | 1. K-b | 2.6€+01 | N/N | : | 1.8E-08 | 2.66+01 | 1.85-00 | 2.46+01 | 10.0 | : | 10-30.1 | : | ISINI |
| ž | DI N/A | : | 1.36-10 | 3 2.4E+01 | R/A | : | 1.32-10 | 2.66+01 | N/N | ÷ | 1.3E-10 | 2.6E+01 | 1.36-10 | 2.66+01 | 642.0 | 1.4€+02 | 2.16+01 | : | Ħ |
| 3 | N/N IO | ; | 1.36-10 | 3 2.46+01 | N/N | : | 1.3E-10 | 2.46+01 | N/N | : | 1.36-10 | 2.66+01 | 1.36-10 | 2.46+01 | 6.00.0 | 1. XE+02 | 2.0€+01 | ; | ž |
| | N/N | ; | N/N | : | N/N | ! | N/N | ; | 8/8 | ; | 1.06-03 | 1.06+01 | 1.05-01 | 1.0€+01 | 220.0 | ; | : | 8 | : |
| 2 | N R/A | ; | 1.06-01 | 1.06+01 | R/A | : | 1.06-03 | 1.0€+01 | N/N | : | 1.06-03 | 10+30.1 | 1.06-03 | 1.05+01 | 10.7 | : | : | щ. | : |
| - | DE N/A | ; | 1.06-01 | 1.06+01 | N/N | 1 | 1.0E-03 | 1.05+01 | N/A | ; | 1.05-03 | 10+30.1 | 1.06-03 | 10+30.1 | 10.0 | : | ; | Ж. В | : |

22.23

22.22.22

I-61

File: HEMAT.MK1 Page 1 Date 20-Aug-87

MATIONAL COLLOCATION OFTICH - FACILITY NANDLING

Accident Frequencies for Facility Nandiing Operations (NF) (Events per Pallet or Container)

Agent Available and Released

| | SCENAALG NUMBER | ANNA FIREQ | RAMGE Factor | AP6 FRED | RANGE FACTOR | UNG | RANGE FACTOR | MAAP Free | RANGE Factor | PAN FREQ | nange Factor | PUOA FREQ | RANNGE FACTOR | TEND NAME Freq Factor | unda Fieg | RANGE FACTOR | AGENT AVATLABLE | 105 SP1LLED | L DS De tonated | LINS UN | URAT 10H T INE |
|----------|--------------------|---------------|-----------------|-------------|-----------------|---------------------|-----------------|--------------|-----------------|-------------|-----------------|--------------|------------------|--------------------------|--------------|-----------------|--------------------|----------------|--------------------|---------|-------------------|
| | | | | | | | | | | | | | | | | | | | | | |
| 596.2 | - | N/N | ; | 8/8 | : | N/N | : | R/A | : | N/A | : | N/N | : | 4. 26-09 1. 36+0 | A N/A | ; | 440.0 | 220.0 | : | ; | ĥ |
| # BHZ | | N /N | : | N/N | : | N/A | ; | R/A | : | N/N | : | N/N | ; | 1.26-09 1.36+0 | I N/A | ; | 200.0 | 4.0 | : | ; | ¥ |
| HFC65 | | N/N | : | W/W | : | N/A | ; | N/A | ; | A/A | : | N/N | : | 1.06-10 1.36+0 | I N/A | : | 48 .0 | 1.4 | : | ; | ¥ |
| MECHS | | 4/H | ; | W/W | : | A/A | ; | A/A | : | N/A | : | N/A | : | 4.0E-10 1.3E+0 | I N/A | ; | 94.0 | 3.2 | ł | ; | ž |
| HFKGS | • | | : | | : | A / N | ; | N/N | : | A/A | : | N/N | ; | B.4E-09 1.3E+0 | I N/A | : | 1500.0 | 1500.0 | ; | ; | J. |
| R K K | | V /N | : | | į | W/W | : | W/W | ; | A7A | : | R/A | ; | 0.46-09 1.36+0 | 1 N/N | : | 1200.0 | 1700.0 | ; | ; | J. |
| HEKVS | - | W/W | : | N/A | : | N/N | ; | N/N | ; | W/W | ; | N/N | : | 8.46-09 1.35+0 | I N/N | ; | 1600.0 | 1400.0 | ; | ; | lir I |
| THE REAL | | W/W | : | N/A | : | A/A | ; | A/A | ; | N/N | : | W/W | : | 5.86-09 1.36+0 | A/M F | 1 | 378.0 | 10.5 | ; | ; | 1hr |
| 594 JH | | A/M | : | N/N | : | ¥/# | ; | R/N | : | N/N | : | N/N | : | 7.2E-10 1.3E+0 | 1 N/A | : | 52.0 | 6.5 | ; | : | 1hr |
| HEPNS | - | R/A | : | R/8 | : | 8/8 | : | K/A | 1 | A/A | ; | 8/N | ; | 7. 25-10 1. 3E+0 | 1 N/N | : | 93.4 | 11.7 | ; | ; | IN. |
| HE PVS | | A/M | : | A/M | : | N/A | ; | A/8 | : | V.N | : | A/N | : | 7. 2E-10 1. 3E+0 | I N/N | : | 48.0 | 4.0 | ; | ; | lhr |
| HF D65 | | 4/8 | ; | 8/N | ; | W/W | ; | N/N | : | W/W | ; | N/N | ł | 7.26-10 1.36+0 | I N/N | ; | 97.0 | 14.5 | ; | ; | lir. |
| HF DVS | | N/A | ; | R/A | : | N/A | : | N/N | : | V/H | ; | N/N | ; | 7.26-10 1.36+0 | 1 N/N | ; | 07.0 | 14.5 | ; | ; | ¥ |
| HF RGS | - | N/N | ; | N/N | : | 8/N | 1 | W/W | ; | A/A | 1 | N/N | ; | 3.26-09 1.36-0 | 1 N/A | : | 160.5 | 10.7 | : | ; | IN. |
| MERVS | - | N/A | : | 4/H | ; | N/N | ; | N/N | ; | W/W | : | A/A | ; | 3.26-09 1.36+0 | 1 N/A | ; | 150.0 | 10.0 | : | ; | J. |
| HE SVS | | 8/8 | 1 | A/A | : | A/A | ; | N/N | ; | W/W | ; | A/N | ; | 1.06-04 1. 36+0 | - | ; | 1356.0 | 1356.0 | : | ; | ž |
| ₩.NeC | ~ | N/A | : | A/N | ; | A/A | 1 | A/N | : | W/W | : | N/A | ; | 1.06-16 3.16+0 | 1 N/A | ; | 220.0 | : | ; | щ. Ш | ; |
| ¥ PC | 2 | 8/8 | ; | A/A | : | N/N | ţ | A/N | : | W/W | ł | N/A | ; | 0.05+00 | M/N | : | 6.0 | ; | ; | ļ | ; |
| NFC GC | ~ | N/A | ; | N/A | : | N/A | : | N/A | ; | M/A | : | N/A | ; | 0.0€+00 | N/A | ; | 1.6 | : | ; | ; | ; |
| NFCHC | 2 | A/A | : | N/A | : | N/N | : | N/N | : | A/A | : | N/A | : | 0.06+00 | | : | 3.2 | : | ; | : | : |
| HFKBC | 2 | A/A | : | A/A | : | A/A | ; | R/A | : | 6/6 | : | N/A | ; | 8.46-14 3.16+0 | 1 N/A | ; | 1500.0 | ; | ; | Щ. Ш | : |
| THE RE | 7 | N/A | ; | A/M | ; | A/A | : | N/A | : | N/A | : | A/A | : | B.4E-14 3.1E+0 | 1 8/4 | ; | 1700.0 | ; | ; | ж Ю | ; |
| HFRVC | 2 | N/N | : | A/N | : | N/N | : | R/A | : | N/N | : | M / A | ; | 0.46-14 3.16+0 | I N/A | : | 1400.0 | : | ; | 10 10 | : |
| HERVE | 2 | N/A | ; | M/A | : | A/N | : | N/N | : | A/A | : | N/N | : | 2.46-17 3.16+0 | I BVA | : | 10.5 | : | : | HE BI | : |
| 19.6C | 7 | N/N | ; | A/N | ; | N/N | : | A/A | ; | A/A | : | N/A | ; | 0.0€+00 | N/A | ; | 6.5 | : | 1 | ; | : |
| JH JH | 2 | A/H | : | N/N | ; | N/N | 1 | A/M | : | W/W | : | A/A | : | 0,06+00 | A/N | : | 11.7 | ; | ; | : | 1 |
| HF PVC | 2 | N/N | ; | N/N | : | N/N | ; | N/A | : | R/A | : | 8/8 | ÷ | 0.06+03 | N/N | : | b.0 | ; | : | : | : |
| HF 065 | 2 | N/N | ł | N/A | : | N/N | ; | N/N | ; | N/A | : | N/N | ł | 0,06+00 | A.M. | 1 | 14.5 | : | ; | : | ; |
| HEBVS | 2 | N/A | : | N/N | : | N/N | ; | N/N | : | N/N | : | N/A | : | Ú. DE +Gi) | A/N | ; | 14.5 | : | 1 | : | ; |



. مرجد 1-62







S.S.L. M. Callin

File: HFMA1.UKI Page 2 Date 20-Aug-87

MATIONAL COLLOCATION OPTION - FACILITY HANDLING

Accident Frequencies for Facility Mandling Operations (MF) (Events per Pallet or Container)

Agent Available and Released

| | SCENNEID | QMNA | KANGE | 9 8 | RANGE | LBAD | RANGE | Ì | RANGE | M | RANGE | PIDA | PANG | 1EAB | ANG? | YOUN | TANK? | AGENT | SI I | 8 | S | |
|---------|----------|-------|--------|------------|--------|-------------|--------|------------|--------|-------------|--------|------------|--------|-------------|----------|-------|-------|-----------|---------|--------------|-----------|--------|
| | NUMBER | FIED | FACTOR | FRED | FACTOR | F REQ | FACTOR | F RE O | FACTOR | FRED | FACTOR | F REQ | FACTOR | FRED | ACTOR | FINED | ACTOR | AVAILABLE | 0311145 | DETOMATED | ENTTED | 311 |
| | | | | | | | | | | | 1 | | | | | | | | | | | |
| | 7 | N/N | ł | W/W | ; | N/N | ; | N/N | : | 4/1 | : | N/N | : | 1.86-14] | 1. 1E+01 | A/A | : | 10.7 | : | : | ۲ W | ; |
| HERVE | 2 | N/N | ; | N/A | ; | N/N | 1 | R/N | ; | N/N | : | A/A | ; | 1.06-14] | IE+01 | N/A | : | 10.0 | ; | 1 | ME BL | : |
| HF SVC | 2 | #/# | ; | N/N | ; | N/N | ; | N/N | : | N/A | : | N/A | 1 | 4.56-15 3 | 1. IE+01 | N/A | : | 1354.0 | ; | 1 | ж. | ; |
| | n | N/N | ł | N/N | : | A/A | ; | N/N | ł | N/A | : | A/A | ; | 3.16-11 3 | 10+31-1 | N/N | : | 440.0 | ; | ~ | 206+01 | 10an |
| 15 H | • | N/N | : | N/N | ; | N/N | : | N/A | ; | N/N | ! | N/N | ; | 9.4E-11 3 | 10+31 | W/W | ; | 288.0 | ; | ; | 10-300. | 1001 |
| 14CS1 | • | W/W | ; | W/W | ÷ | N/N | ; | N/N | 1 | N/N | ; | K/A | ; | 0.0€+00 | | N/A | ; | 48.0 | ; | : | : | ; |
| HCH. | 5 | N/N | : | N/N | ; | N/A | ; | N/A | : | N/A | ; | N/A | : | 0, UE + 00 | | A/A | ; | 96.0 | ; | ; | : | : |
| HFX6F | - | #/# | : | A/A | : | A/A | ; | A/A | : | N/A | : | N/N | : | 1.06-10] | . 1E+01 | N/A | ł | 1500.0 | ; | ; | 50E+02 | 10ei e |
| H.L.H. | - | N/A | ! | N/N | ; | N/N | : | N/A | : | N/A | : | N/A | 1 | 1.06-10 3 | . 16+01 | N/A | ; | 1700.0 | ; | - | 10+3(5) | 10010 |
| 51 K | • | 4/H | ; | N/A | : | N/A | ; | N/A | ; | #/# | : | M/A | ; | 1.06-10 3 | . 16+01 | N/A | ; | 1600.0 | ; | ; | 10+300. | 10era |
| HF INVE | 5 | A/A | ! | A/A | ; | #/# | 1 | A/A | ; | N/N | ; | N/A | : | 1.4E-10] | . 1E+01 | N/A | : | 378.0 | : | ~ | 10-366. | lúaj n |
| 14 P 65 | ~ | N/A | ; | A/A | : | N/A | 1 | A/A | : | A/M | ; | N/A | ; | 0.05+06 | | W/W | ; | 52.0 | ; | ; | ; | ; |
| HE PHE | 5 | A/N | ; | N/A | ; | N/N | : | N/A | : | N/A | ; | N/A | : | 0.0€+00 | | N/A | : | 93.6 | ł | ; | ł | : |
| HEPVE | • | M/A | : | A/A | : | N/A | ; | N/A | ; | A/A | ; | 11.10 | : | 0.0€+00 | | N/N | : | 48.0 | ; | ; | ; | : |
| 190 ± | • | A/A | ; | A/A | ; | A/A | ; | #/# | : | N/A | ; | K/A | ; | 0.0£+00 | | N/N | ; | 87.0 | : | : | : | : |
| NF QVF | 5 | N/A | : | N/A | ; | A/A | : | N/A | : | N/A | 1 | N/A | : | 0.0€+00 | | N/A | : | 87.0 | : | ; | ; | ; |
| 192 H | • | M/A | ; | 8/N | ; | N/A | ; | M/A | ; | A/M | : | N/N | 1 | 8. IE-JI 3 | . 16+01 | N/A | : | 160.5 | : | : | .07E+00 | 10ein |
| HFRUF | 5 | N/A | i | A/A | ; | A/N | ; | N/A | 1 | N/A | : | 4/H | ; | 8. IE-11 3 | 10+31. | N/A | : | 150.0 | : | ~ ; | 10-305 | 10010 |
| HF SVF | ~ | N/A | : | A/4 | ; | A/A | ; | N/A | ; | N/A | : | N/A | : | 2.06-09 3 | . 16+01 | N/A | : | 1356.0 | ; | : | 10+365. | 1001 H |
| ₽5vS | - | N/A | : | N/A | ; | ¥/¥ | : | R/A | ł | N/A | ; | N/A | ; | 3. IE -07 1 | .36+01 | N/A | : | 1356.0 | ; | م | . 00E -05 | Ä |
| HF R6F | 5 | N/A | ; | 4/W | ; | A/A | : | A/A | : | N/A | : | A/A | ; | 0.0£+00 | | #/# | ; | 200.0 | ; | : | ; | : |
| HF DHC | • | A/A | : | A/A | 1 | N/A | : | N/A | : | N/A | ; | W/A | : | 0.0€+00 | | A/N | : | 288.0 | : | ; | : | ; |
| HF C PC | ŝ | N/A | : | A M | ; | A.N | : | W/W | : | N/A | : | M/A | ; | 0.0E+00 | | N/A | ; | 48.0 | ; | : | ; | : |
| HECHC | 5 | N/A | : | A/A | ; | N/A | ; | A/A | ; | M /A | ; | M/A | ; | 0.0€+00 | | N/A | ; | 96.0 | : | : | : | ; |
| HE C | • | N/A | : | W/W | ; | N/A | : | N/N | ; | N/A | : | N/A | : | 0. 0E+60 | | #/# | ; | 1500.0 | : | : | - | ; |
| ž ž | ŗ | R/A | : | R/N | ; | N/N | ; | A/A | : | N/A | : | ¥/¥ | : | 0.06+00 | | #/# | : | 1700.0 | ; | ł | : | ; |
| HF. VF | 2 | A/R | ; | N/N | ; | #/# | ; | N/A | : | N/A | ; | K.A | ; | u.0£+0ù | | M/A | ; | 1600.0 | : | ; | : | : |
| HE LC | S | 4 / ¥ | ; | 4 / R | ; | N /N | : | A/A | ; | N/A | : | 8/N | ; | U. OE+60 | | M/A | : | 378.0 | : | : | ; | : |
| HFF6C | ŝ | N/A | ! | 4/H | 1 | 8/8 | ; | M/A | ; | N/A | ; | N/A | : | 00+30.0 | | #/# | : | 52.0 | : | ; | : | : |

I-63

File: WEMMATUMMI Page J Date 20-Aug-87

NATIONAL COLLOCATION OPTION - FACILITY HANDLING

| | | | | | | Accident | Frequencia | s for Fa | cility Nac |) fui (pi | berations | (HE) (E | vents per | Pallet or | Containe | 5 | | - | Agent Avi | ntakle ar | d Release | y |
|----------|----------------------|---------------------|-----------------|------------------|-----------------|-------------|-----------------|----------|-----------------|-----------|-----------------|--------------|-----------------|----------------------|----------|-------|------|--------------------|--------------------|--------------------|----------------|----------|
| 1 | SCENAM 10 INUMBER | ANN Pres | RANGE FACTOR | APG Free F | RANGE FACTOR | LIMD | RANGE FACTOR | 3 E | RANGE Factor | Per o | RANGE Factor | PUDA Fred | RAMGE Factor | TEAD RAI Fred Fac | 39 | MON R | ANDE | AGENT NVALLADLE | SPILLED SPILLED | L 65 Dù tomatei | LBS ENLITED | DURATION |
| 3 | | | ; | | ; | | : | | : | | 1 | 4,1 | 1 | | | | 1 | 1 | 1 | | : | |
| | | | : | | : : | | : ; | | ; | | : ; | | : : | 0.06+00 | | | : : | | : : | : : | : : | : : |
| MF DGC | | V / I | : | | ; | | ; | | : | | ; | | 1 | 0.0€+00 | | | : | 0.70 | ; | ; | ; | ; |
| HF OVC | ~ | R/H | : | | ; | N/N | : | 8/N | 1 | N/N | ; | M/A | ł | 0.0€+00 | | N/N | : | 87.0 | ; | ; | : | : |
| HF REC | 'n | N/N | : | N/N | : | A/N | ; | 8/8 | : | W/W | ; | N/A | : | 0.0€+00 | | N/N | ; | 140.5 | : | : | ; | : |
| MF RVC | • | A/A | ; | A/H | : | A/A | ; | #/# | ; | A TH | ; | N/N | ; | 0.06+00 | | N/A | ; | 150.0 | ; | : | ; | : |
| HE SVF | • | N/A | : | N/N | ; | N/N | : | A/A | : | N/N | 1 | N/A | : | 0.06+00 | | N/A | ; | 1354.0 | ; | : | : | ; |
| 16 B65 | ~ | 4/2 | 1 | N/N | : | A/A | : | A/A | : | N/N | : | N/N | ; | B. 36-10 1. | 10+31 | N/A | ; | 440.0 | 220.0 | ; | : | N |
| ¥ NG | ~ | N/N | : | R/B | : | A/N | : | N/A | : | 8/8 | : | N/A | : | 2.56-09 1. | 16+01 | N/A | : | 286.0 | 6 .0 | ; | ; | Ä |
| 14 C65 | ~ | 8/8 | ; | A/N | : | N/N | : | N/N | ł | W/W | ; | N/N | ; | 0.0E+00 | | A/N | : | 48.0 | : | : | ; | : |
| #CKS | ~ | A.A | ; | A/A | : | N/A | : | N/A | : | #/# | ; | #/# | ; | 0.01+00 | | #/# | : | 94.0 | : | : | : | : |
| HFIGS | ~ | N/N | ; | N/N | ; | N/N | ; | A/A | : | N/A | : | N/A | : | 2.76-09 1.3 | 10+31 | N/N | ; | 1500.0 | 1500.0 | ; | : | lir. |
| SH 1 SH | ~ | A/A | : | N/N | : | N/N | : | N/A | ; | 8/8 | : | N/N | : | 2.76-09 1.1 | 10+31 | N/A | : | 1700.0 | 1700.0 | : | : | je. |
| HEXVS | ~ | 8/8 | : | N/N | ; | N/N | ł | N/N | ; | A/A | : | K/N | ; | 2.76-09 1.3 | 10+31 | N/A | : | 1400.0 | 1600.0 | ; | : | line . |
| SVE # | • | N/N | : | N/N | : | R/N | ; | N/A | ; | N/N | : | A/A | : | 3.66-09 1.] | 10+31 | A/A | ; | 378.0 | 14.5 | ; | : | lhr |
| MF P65 | ~ | #/# | : | N/N | : | N/A | ; | N/N | : | N/N | : | N/A | ; | 0.0€+00 | | N/A | : | 52.0 | : | : | : | : |
| 92.¥ | ' | 4 · N | : | A/N | : | A/A | ; | N/A | : | A/A | : | A/A | : | 0.0€+00 | | M/A | ; | 93.6 | ; | : | : | ; |
| SA JH | ~ | A / A | : | 8/N | ; | N/A | ; | A/H | : | N/A | : | M/A | ; | 0.0€+00 | | N/A | : | 48.0 | : | ; | : | : |
| | 1 | A.A | : | 4/# | : | A/A | ; | N/N | ; | N/N | ; | N/A | ; | 0.0€+00 | | N/A | : | 87.0 | : | : | : | ; |
| HE BVS | ~ | N/A | ; | A/N | : | 4/H | ; | N/A | : | N/N | ; | N/A | ; | 0.06+00 | | N/A | : | 07.0 | : | : | : | : |
| | ~ | M/ A | : | A/H | : | A/A | ; | W/W | 1 | N/A | : | N/A | : | 2.26-09 1.3 | 10+35 | N/A | : | 140.5 | 10.7 | : | ł | 1% |
| HE RVS | ~ | #/# | ; | R/A | ; | A/N | ; | N/N | ; | 8/8 | : | A/A | ; | 2.26-09 1. | 10+31 | N/A | : | 150.0 | 10.0 | : | : | 1hr |
| ME SVS | ~ | N/N | : | N/A | : | N/N | ; | N/A | : | N/A | : | N/N | : | 5.26-00 1.3 | 10+3 | N/A | ; | 1354.0 | 1356.0 | ; | : | jų. |
| H RC | - | A/N | : | N/A | : | A/H | ; | N/A | ł | N/A | ; | N/N | ; | 5. 36-18 3.1 | 10+31 | N/N | : | 220.0 | : | ; | NE CI | : |
| Ц. Ж. | - | N/N | : | A/N | : | 8/8 | ; | R/N | : | ₩/₩ | ; | N/A | 1 | 5.36-18 3.1 | 10+31 | R/A | ; | 6.0 | : | : | NE GL | : |
| AFCRC | • | A/A | : | N/N | ; | N/H | ; | N/A | : | N/N | : | A/A | : | 6.06-19 3.1 | 10+31 | W/W | : | • | : | 1 | ME FL | : |
| MECHE | • | W/W | ; | A/A | : | N/N | ; | N/A | : | #/# | ; | N/N | : | 6.06-19 3.1 | 10+3) | N/A | : | 3.2 | : | ; | щ Ш | : |
| HF K CC | • | N/N | : | W/W | : | N /N | : | N/N | : | N/A | : | N/N | ; | 1.16-17 5.1 | 10+3 | N/N | ł | 1500.0 | : | : | ж. М | : |
| Ш. | - | W/W | : | N/N | ; | N/N | ; | #/# | : | W/W | : | N/N | ; | I.IE-17 3.I | 10+3 | A/A | ; | 1700.0 | : | : | NE BI | ; |





NATIONAL COLLDCATION OPTION - FACILITY HANDLING

Agent Available and Keleased

| | | | | | | Accident | fr equencie | s far Fi | scility Ha | 8 Guilbr | per at Lons | (HF) (E | vents per | fallet c | x Containt | ī | | ~ | gent Avai | lable and | Kel eased | |
|--------------|---------------------|---------------|------------------|---------------------|-----------------|----------------|-----------------|-------------|-----------------|---------------|-----------------|--------------|-----------------|---------------|-------------|--------------|---------------|--------------------|-----------|------------------|--------------|--------|
| | SCEMAR LO NUMBER | ANAD F RED | RANGE Falitor | AP6 FRED | KANGE FACTOR | L BAD F REQ | RANGE Factor | FRED | RANGE FACTOR | P BA F REQ | RANGE Factor | PUDA FREQ | RANGE Factor | TEND F | AMIGE ACTOR | MDA 9 998 | AMEE ACTOR | AGENT AVATLABLE | SFILLED D | LBS DETONATED | LIPS D | 11/100 |
| 1 | | | | | | | | | | | | | | | | | | | | | | |
| 14 5 AL | 6 4 | | : | | : : | | : : | | : | | : : | | : : | 1 26-10 | 1 15 401 | | : : | 1.0001 | : : | : : | 6 a | : 1 |
| | b a | | : | | | | | | : : | | | | : : | 0 11211 | 10.21 | | : : | | : | ; | 6 a | ; |
| NC DIE | i , a | | : : | | | | ; ; | | : : | | : : | | : : | 0 01 - 30 - 1 | . 16401 | | : : | | : | : : | i J | ; ; |
| E P C | , ac | 4/H | : | | : | | ; | | : | | : | N/A | : | 9.06-19 3 | 16+01 | | ; | 6.9 | ; | ; | i di W | ; |
| HF 06C | . 60 | N/P | : | A/N | ; | N/A | ; | N/A | ; | N7A | ; | N/A | ; | 9.06-19 | I. IE+01 | N/N | ; | 6.61 | ; | 1 | ц Ш | ; |
| HF QVC | 80 | N/A | : | A/4 | : | N/A | : | N/A | : | A/A | : | A/A | ; | 9.06-19 3 | 1. IE+01 | N/A | : | 14.5 | : | : | | ; |
| HE REC | œ | 414 | : | 4/# | ! | N/N | ; | N/A | ; | A/N | : | N/A | : | 4, 16-18 | S. IE+01 | N/A | : | 10.7 | : | ; | ME GL | : |
| JANN | 80 | A.N | ; | N/A | : | N/A | ; | N/A | ; | N/A | : | N/A | : | 4.16-18 | 10+31.8 | N/A | ; | 10.0 | : | : | ₩EG. | ; |
| 4F SVC | 8 | 878 | : | N/A | : | 4. M | ; | A/A | : | N/A | ł | 4/H | ; | 1.26-15 | 1. IE+01 | A/A | : | 1356.0 | ; | : | а Ш | 1 |
| HF SVC | D | N/A | : | #/¥ | : | A/M | ; | N/A | ; | A/A | : | N/A | : | 7.76-16 | 1.16.01 | M/A | : | 1356.0 | ł | | 10-351 | N |
| HF BGC | 10 | N/A | : | N/A | : | N/A | : | N/A | : | A/M | : | N/A | : | 4. XE-19 | 1. 15+01 | N/A | ; | 440.0 | ; | : | ₩. 61 | : |
| HF DHC | 10 | N/A | : | A.N | ; | M/A | ; | N/A | ; | N/A | ; | #/# | : | 1.34-18 | 1. IE •01 | N/A | : | 200.0 | ; | : | 163 1 | : |
| HFC BC | 91 | N/A | : | N/A | : | N/A | ; | N/A | ; | N/A | : | N/N | : | 0.06+00 | | A/N | : | 48.0 | ; | ; | 16 W | : |
| 볼 C HC | 0 | N/A | : | N/A | : | A/A | ; | N/A | : | A/N | : | N/A | ; | 0.0€+00 | | N/A | ; | 96.0 | : | ; | 16 W | : |
| HFLOC | 10 | N/A | ; | N/A | ; | N/A | ; | A/A | ; | A/M | ; | K/A | 1 | 1.4E-18 1 | 1. 1E+01 | N/A | ; | 1500.0 | : | ; | ж Ю | : |
| FFERC | 10 | N / N | ; | N/A | : | A/A | : | N/A | : | N/A | : | N/N | : | 1.46-19 | 5. IE +01 | N/A | ; | 1760.0 | : | ; | Щ. Ш | : |
| HERVE | 10 | N/A | • | N/A | : | N/A | : | R/A | : | 8/8 | : | N/A | : | 1.46-10 | 1. IE •01 | N/A | : | 1600.0 | ; | ; | ₩.GL | : |
| Dom SH | 9 | N/A | : | A.A | ; | N/A | ; | A/N | ; | N/A | ; | A/A | : | 1.91-39.1 | 10+31.3 | #/# | : | 378.0 | ; | : | 16 W | ; |
| 18 PSC | 10 | N/A | ; | N/A | : | A/A | ; | N/A | ; | R/A | : | N/N | ; | 0.0£+00 | | N/A | : | 52.0 | ; | ; | ۲ ۲ | : |
| HF PHC | 10 | N/A | : | 8/8 | | N/N | : | A/A | : | N/A | ; | N/A | : | 0°.0E+00 | | N/N | : | 93.6 | ; | ; | щeс | ; |
| HE PVC | 11 | 8/8 | ; | M : A | : | N/N | : | N/N | : | N/N | : | N/A | ; | 0.0€+00 | | N/A | : | 18.0 | ; | : | 103 | : |
| 190 F | 9 | N/A | ; | 8/N | : | A/N | : | N/A | ; | A/A | ; | N/A | : | 0.(ft+0u | | A/A | ; | 87.0 | ; | : | щ. Ш | : |
| HERVC | 3 | N/A | : | 8/8 | ; | A/N | : | A/N | : | A/N | : | N/A | ; | 0.0€+00 | | N/A | : | 87.0 | | ; | a W | ; |
| HF RGC | 10 | N/A | ; | A/A | : | #/# | : | N/A | : | #/# | ; | W/N | ; | 1.16-18 | 1. IE +01 | #/# | ; | 140.5 | : | : | ж Ю | : |
| HERVE | 16 | A/A | ; | N/A | ; | N/N | : | N/A | 1 | N/N | ; | A/A | 1 | 1.16-18 | 1.16+61 | #/# | : | 150.0 | ; | : | a M | ; |
| HF SVC | 9 | N/A | : | N/A | ; | N/A | : | N/A | ; | A/A | ł | #/# | ; | 2. 11 17 | 10+31.1 | #/# | ÷ | 1354.0 | : | ; | 16 W | ; |
| HE BHC | Ξ | R/N | : | A/N | : | N/A | ł | N/A | ; | A/A | : | N/A | ; | 5.06-11 | 2.46+01 | N/A | ; | 286.0 | 0.0 | 6.9 | : | ž |
| ## CPC | = | ¥/H | ; | N/N | : | N/A | ; | 1 /1 | : | N/A | ; | N/N | : | 2.96-11 | 10+39.5 | N/A | : | 48.0 | 9.0 | • | : | ž |

90

12 K

÷,

3

Page 5 Date 20-Aug-87 Files WMMT.WC NAFICHAL COLLOCATION OPTION - FACILITY HANDLING

| | | | | | | Accident | Fr equenci e | s for Fa | culuty Nam | d ing D | per at i ons | (H) (E | reals per | Pallet or (| Container | - | | £ | ent Avai | lable and | Rejest | - |
|--------------|--------------------|---------------|-----------------|-------------|-----------------|------------|-----------------|-------------|-----------------|-------------|-----------------|---------------------|-----------------|-----------------------|-----------|------------|----------|---------------------|----------|-------------|----------|------------------|
| | SCENAR10 NUMBER | ANAD FIRED | RANGE FACTOR | AP6 FRED | RANGE FACTOR | THE LAN | RANGE FACTOR | FREQ | RAMGE Factor | PIM Freg | RANGE FACTOR | PUDA | RANGE Factor | TEAD AME Fred Fact | 55 26 | | ALLER AL | AGENT VALLADLE S | | LBS | LIIS | DURATION 11NE |
| NE DAC | = | | ; | | 1 | | 1 | | ; | | f | | : | 2.96-11 2.66 | 10+ | | 1 | 94.0 | 14.0 | 5.2 | ; | ł |
| THE REAL | : = | | ; | | ; | W/W | ł | N/N | : | W/W | ; | V / I | ; | 4.36-11 2.64 | 10 | 5 | : | 378.0 | 137.5 | 21.5 | ; | ž |
| 394 M | : = | N/N | ; | N/N | ; | | ; | N/A | ; | W/W | ; | | 1 | 9.46-12 2.6 | 10+ | N. | : | 52.0 | 32.5 | 6.5 | ł | ik. |
| HF FRC | - | N/N | ; | N/A | : | N/A | ; | N/A | : | W/W | 1 | N/N | ł | 9.66-12 2.66 | 10+3 | K / | ! | 93.4 | 5.15 | H7 | ; | M |
| JV-1 SH | = | N/N | : | A/W | : | N/N | ; | N/N | : | N/A | ; | N/N | ; | 9.66-12 2.66 | I 10+3 | A/1 | ; | 48.0 | 30.0 | 6 .0 | ; | ikr |
| 198 HK | Ξ | N/N | ; | N/A | ÷ | A/N | : | N/N | ; | N7A | ; | N/A | 1 | 7.25-12 2.4 | 10-3 | ¥. | : | 87.0 | 2.51 | 14.5 | ; | ji ji |
| HF BVC | = | N/N | : | A/A | ; | A/H | : | N/A | ; | N/N | ; | N/N | : | 7.26-12 2.66 | 101 | K / | : | 87.0 | 72.5 | 14.5 | ; | j. |
| NF RGC | 11 | R/N | ; | A/M | : | N/A | 1 | N/N | : | N/A | ; | N/A | : | 1.96-11 2.64 | 10+3 | N.N | ; | 160.5 | 139.1 | 21.4 | ; | ľ. |
| HF RVC | = | N/N | 1 | A/A | : | N/A | ; | N/A | : | A/A | { | A/H | : | 1.86-11 2.66 | 1013 | N.N | ; | 150.0 | 150.0 | 20.02 | ; | Ihr |
| ¥ DNC | 21 | A/K | ; | N/A | : | N/A | (| A/N | ; | N/A | ; | N/A | : | 3.06-10 2.4 | 10+3 | N.N | ; | 6.0 | ; | 6.0 | ; | INSTANT |
| HF C GC | 12 | A/A | ; | M/A | : | N/N | : | 8/8 | ; | N/N | : | A/H | : | 3.06-10 2.6 | N 10-3 | A/4 | ; | • | ; | 1.6 1.6 | ; | INS (AN) |
| HECK | 12 | N/N | : | N/A | : | M/A | { | N/N | ; | A/H | ; | N/A | ł | 3.06-10 2.61 | 10-3 | N.N | : | 3.2 | : | 3.2 | ; | INSTANT |
| HE RIVC | 21 | M/A | : | N/A | ; | N/N | : | N/A | ; | N/N | ; | A/A | ; | 3.06-10 2.66 | 1013 | | : | 10.5 | : | 10.5 | : | INSTANT |
| 3943H | 12 | A/N | ; | N/A | ; | N/N | { | N/A | : | R/A | : | N/N | ; | 3.0E-10 2.6I | 10+3 | N.N | ; | 6.5 | ; | 6.5 | ÷ | INSTANT |
| HF PKC | 12 | N/A | ; | A/A | ; | N/N | 1 | N/N | ; | W/W | ; | A/M | ł | 3.06-10 2.66 | 101 | N.N | ; | 11.7 | : | 11.7 | ; | THAT 2H |
| HEPVC | 12 | N/N | : | N/N | : | A/N | : | N/N | : | N/N | ; | N/N | ł | 3.06-10 2.6 | 1013 | 4/1 | ; | b.0 | : | 6.0 | : | INSIANI |
| NF QGC | 12 | N/N | : | N/A | ; | N/N | ł | N/A | : | 8/8 | ; | N/N | ł | 3.06-10 2.6 | 1013 | | ; | 14.5 | ; | 14.5 | : | INSTANT |
| HF DVC | 12 | A/A | ; | N/A | : | A/N | ; | N/N | ; | R/A | ; | N/N | : | 3.06-10 2.6 | 10+3 | | ; | 14.5 | ; | 11.5 | ; | THE REL |
| NF REC | 12 | N/N | ; | N/N | ; | A/N | ; | N/N | ; | R/A | : | N/N | ; | 3.06-10 2.61 | 10+3 | N.N | ; | 10.7 | ; | 10.7 | : | INVISE) |
| JAPRAC | 12 | R/R | : | N/A | : | A/N | : | N/N | ; | A/A | ; | N/N | : | 3.06-10 2.61 | 10+3 | S | ; | 10.0 | ; | 10.0 | ! | THE THE |
| ž Š | | N/N | ; | N/A | ! | N/N | ; | N/N | ; | 8/8 | ; | N/N | ; | 7.21-32.1 | 10.3 | | ; | 288.0 | ; | 6.01 | . 706-05 | J. |
| HFC6C | 13 | N/N | ; | R/A | : | N/N | ; | W/W | ; | N/A | ; | N/A | ł | 3.46-11 2.61 | 1 10+3 | N.A | ; | 48.0 | ł | 1.63 | .706-01 | ž |
| MENC | 11 | N/N | : | A VI | ; | A/N | ; | N/A | ; | N/N | ; | N/N | ; | 3.46-11 2.46 | 10+3 | ¥/ | ; | 96.0 | ; | 5.21 | .60E-03 | ž |
| HF INC | 11 | N/N | ; | #/# | : | 8/8 | : | W/W | ; | A/A | : | N/A | : | 5.46-11 2.41 | 10-3 | 1/# | ; | 378.0 | ; | 31.5 3 | 306-90 | lir. |
| MF PGC | 13 | N/N | : | A/N | : | N/N | ; | A/A | ; | R/A | : | 11/1 | : | 1.21-31.1 | 10+3 | | : | 52.0 | : | 6.5 3 | 10-306. | ihr |
| JH4 JH | :: | N/N | : | A/H | : | N/N | : | U /H | ; | R/A | ; | N/N | ; | 1.21-11-2.4 | 1013 | K) | : | 93.6 | ; | 11.71 | .906-03 | ih. |
| JNJJH | 13 | A/H | : | A/A | : | N/N | : | N/A | : | W/W | ; | N/N | ; | 1.2E-11 2.4 | 10-3 | K.N | : | 1 .0 | ; | 6.01 | .106-05 | Ĩ |
| HF DEC | 51 | N/A | ; | 4/8 | ; | N/N | ; | #/# | ; | N/A | ; | 8/8 | ; | 9.06-12 2.4 | 1 10-3 | N. | ; | 87.0 | ; | 14.54 | 10-305. | lir I |
| HF BVC | :: | N/N | ; | N/A | ; | N/N | ; | W/W | ; | ¥/¥ | ; | W/W | ; | 9.06-12 2.61 | E+01 E | ¥ | ; | 87.0 | ; | 14.5 1 | 105-05 | Ň |











2

Ç

Page 6 Bate 20-Aug-87 File: MML.W.

INFIDING COLLOCATION OPTION - FACILITY INNUMERING

| | | | | | | Accident | Fr equence | tes for F | scility A | p eilbe | üperatıon: | 1 (#2) 1 | Events per | r Pallet | or Contain | er) | | - | igent Avai | at the | l Release | _ |
|--------------|----------|-------------|---|----------|---|----------|------------|-----------|-----------|----------------|-----------------|------------|-----------------|----------|-----------------|---------------|----------------|--------------------|------------|------------------|-----------|----------|
| | SCENARIO | 1 | | 2 | | | | | | PIN | RANGE Facior | FUEN | RANGE FACTOR | TEND | RANGE FACTOR | LINGA FRED | NAME FACTOR | AGENT AVAILABLE | 91165 | LIS Ketomated | | ACT I DA |
| | | | | | | ! | | | | | | | | | | | | | | | | |
| NF RGC | 5 | | : | | : | A/N | ; | A/N | ; | 8/8 | : | N/A | : | 2.26-11 | 2.66+01 | N/N | ł | 140.5 | : | 21.4 | 10-308.4 | Ņ |
| HENC | = | 8/8 | ; | A/N | ; | W | : | N/A | ; | R/N | : | N/A | : | 2.26-11 | 2.46+01 | N/N | ł | 156.0 | ; | 20.0 | 2.906-05 | Ņ |
| | Ξ | | ; | N/N | : | A/H | : | N/N | ; | W/W | : | N/N | ; | 1.36-14 | 2.46+01 | N/N | 1 | 200.0 | 8°.0 | 0' 1 | ; | 1Nr |
| MFCSC | = | N/N | ; | A/N | ; | W/W | : | N/N | : | N/A | : | A/N | ; | 6.46-15 | 2.66+01 | N/N | ; | 49.0 | | 1:1 | ; | ¥ |
| MCHC | = | 1 /1 | : | A/A | : | A.M | : | A/N | ; | N/A | : | N/N | : | 6.46-15 | 2.66+01 | A/A | ; | 94.0 | 16.0 | 3.2 | 1 | N |
| HF RVC | 2 | N/N | : | N/N | ÷ | A/M | : | A/N | ; | N/A | : | N/N | : | 9.96-15 | 2.4E+01 | N/A | 1 | 378.0 | 157.5 | 31.5 | : | F |
| 1943H | = | A/N | : | N/N | : | N/N | : | A/N | : | R/A | : | A/N | ; | 2.26-15 | 2.6€+01 | N/A | ; | 52.0 | 32.5 | 6.5 | : | ĥ |
| HFF | = | R/N | : | A/8 | : | A/N | : | N/A | ; | M/A | : | A/H | : | 2.26-15 | 2.46+01 | 4/1 | : | 93.4 | 5.3 7 | 11.7 | ; | ł |
| HF VC | = | N/N | : | A / H | : | A/A | : | R/A | ; | #/# | : | R/A | : | 2.26-15 | 2.46+01 | A/A | ! | 48.0 | 30.0 | 6.0 | ; | W |
| 1981 | = | A/R | : | A/N | : | A/A | : | A/M | : | A/N | ; | N/A | ; | 1.76-15 | 2.46+01 | N/A | : | 07.0 | 72.5 | 14.5 | : | ž |
| AF BVC | • | A/N | : | A.M | : | W/W | : | 4/H | : | A/N | : | ¥/¥ | : | 1.76-15 | 2.66+01 | N/N | ; | 87.0 | 72.5 | 14.5 | : | ľ. |
| MF RIG. | = | R/A | : | A/N | : | A/N | : | N/A | : | N/N | : | A/A | : | 4.16-15 | 2.46+01 | N/N | ; | 140.5 | 139.1 | 21.4 | : | ľ. |
| JA BAC | • | R/8 | : | R N | : | A/N | ; | N/N | : | N/A | : | N/A | : | 4.16-15 | 2.4E+01 | N/N | : | 150.0 | 130.0 | 20.0 | : | ł |

 \mathbf{L}

22.22

1-67

PAGE I DATE 21-Aug-87

DHSITE DISPOSAL OPTION (PER PALLET OR CONTAINER)

| | | | | | | | Accide | at Frequ | MCIPS 6 | or Onsite | Handling |) Operatio | (OH) Suc | | | | | ~ | gent Avai | ilable and | Released | |
|--------------|-----------|----------------|-----------------|-------------|-----------------|----------------|-----------------|------------|------------------|------------|----------|--------------|----------------|--------------|-----------------|------------|------------------|-------------------|--------------|--------------------------|------------|-----------------|
| | SCENAR (O | ANAD F REQ | RANGE Factor | AP6 FREQ | RANGE Factor | FREE | RANGE FACTOR | FRED | INNUSE INCTOR | PIM I | Interest | Pubh Pubh | ANNUE ACTOR | TEAD FREG | RANKE FACTOR | L MON | RANGE ACTOR A | AGENT MAILABLE | LISS LISS | L BS Setomated | LIDS DA | MAT ION TINE |
| | | | | | | | | | | | | | | | | | | | | | | |
| HORE: | - | N/N | ; | N/A | ; | 8/8 | : | A/# | : | N/N | : | N/N | ; | 90-31-9 | 1.3£+01 | 6.1E-08 | 1.36+01 | 440.0 | ; | 1 | 1. 36 + 00 | ١. |
| HOOHC | | 1.46-08 | 10+37-1 | N/N | ł | A/N | ; | N/N | ; | A/A | ; | 1.66-08 | 1. 35 +01 | 1.66-08 | 1.3£+01 | N/A | : | 200.0 | ł | ; | 1. 35-01 | Ĩ |
| HOCEC | - | 2.86-09 | 1.36+01 | A/A | ; | A/N | ; | N/A | : | R/N | : | N/N | ; | 2.86-09 | 1.36+01 | N/A | : | . B . | ; | : | 1.16-01 | Ĩ |
| HOCHC | | 2.26-09 | 1. 36+01 | R/A | ; | R/A | 1 | 8/8 | ; | N/N | 1 | 2.86-09 1 | 10+31.1 | N/N | ; | W/W | : | 74.8 | : | ł | . 36-03 | ٩. |
| HOKEC | - | N/N | : | A.M | ; | A/N | ; | N/N | : | N/N | : | W/W | 1 | 2.06-08 | 1.36+01 | N/N | : | 1500.0 | 1 | : | .4E+00 | Ē |
| HOKHC | - | 2.06-08 | 1.3£+01 | 2.0E-08 | 1. XE +01 | V/N | ; | N/N | ; | 2.06-00 1 | 10-36-01 | | : | 2.06-08 | 1.36+01 | 2, % -08 | 1.36+01 | 1700.0 | : | 1 | 20-35-02 | 1 |
| HOK VC | - | N/A | : | N/A | ; | N. | : | 2.06-00 | 10+3E. | N/N | : | N/N | ; | 2.06-09 | 10.3E+01 | #/# | : | 0.0061 | : | ; | 2.76-04 | Ħ |
| HORVC | - | B0-37.1 | 1.3£+01 | N/N | ; | N/N | ; | N/N | : | 1.06-37-1 | 1.36+01 | N/N | : | 1.25-00 | 1.36+91 | 1.25-08 | 1.3£+01 | 378.0 | : | ; | 1. 76-05 | Ŧ |
| HOPEC | - | 0.0€+00 | : | A/A | ; | A/A | 1 | N/N | : | N/N | : | N/A | : | 0.0E+00 | : | 0.0E+00 | 1.36+01 | 52.0 | ; | : | : | : |
| HOPHC | - | 0.0€+00 | : | N/N | ; | 0.0E+00 | ; | R/N | : | A/A | : | 0.0E+00 | : | 0.116+00 | ; | R/A | : | 93.6 | ; | : | : | ; |
| NOPVC | - | 0.0€+00 | ; | N/N | : | 0.0€+00 | ; | A/M | : | A/N | ; | A/M | 1 | 0.0E+00 | 1 | 0,05+00 | : | 48.0 | ; | ; | : | 1 |
| H006C | ~ | 0.0€+00 | : | N/A | 1 | 0.0E+00 | ł | A/N | ł | R/N | : | N/N | : | 0.0E+00 | : | 0.0E+00 | : | 07.0 | : | ł | : | : |
| HORVE | - | A/A | : | R/A | ł | N/N | : | N/A | ; | A/A | : | N/A | ; | 0.0E+00 | : | 0.0£+00 | ; | 87.0 | : | : | : | : |
| HOREC | - | 4. 0E -00 | 1.36+01 | R/A | ł | 1.0E-0B | 1.3€+01 | A/A | ! | 1 8-31 | 1.36+01 | N/N | : | 4.0E-09 | 1.36+01 | 4, 8E-08 | 1.36+01 | 160.5 | : | ; | 1.5E-01 | H |
| HORVC | - | 4.86-06 | 1.35+01 | A/A | ; | 90-36't | 1.36+01 | N/A | : | 4. HE-00 1 | 1. 3£+01 | N/A | : | 1.86-08 | 1.36+01 | 90-36.4 | 1.36+01 | 150.0 | : | ; | \$0-39" | Ĩ |
| NOSVC | - | N/A | ; | N/A | 1 | A/M | ł | N/N | ; | R/N | ! | N/N | : | 5.2E-07 | 1.36+01 | 5. ZE-07 | 1.35+01 | 1336.0 | : | ; | 2.76-04 | Ŧ |
| #CK# | 2 | #/# | ; | 5.26-10 | J. 1E+01 | A/A | ; | #/# | : | 5.26-10 3 | 1. IE+01 | N/A | 1 | 5.25-10 | 3. IE+01 | A/N. | : | 1700.0 | ; | ; | 10+35.0 | NIN O |
| HORIGE | 2 | N/A | : | R/A | : | A/N | : | A/A | : | R/N | : | N/N | ł | 1.3E-07 | 1.36+01 | 1.36-07 | 1. JE+01 | 440.0 | : | : | 1.36+00 | Ŧ |
| HOME | • | 3.76-07 | 10+35-11 | A/A | ; | W/W | ; | #/# | ; | N/N | : | 3.76-07 | 36+01 | 3.7E-07 | 1.35+01 | A/H | ; | 288.0 | : | ; | . JE - 01 | Ĩ |
| HOCEC | *7 | 8.95-08 | 1.36+01 | N/N | ; | W/W | ł | N/N | ; | N/N | 1 | N/N | ł | 8.96-08 | 1.3E+01 | N/N | : | 38.4 | ; | ; | 10-31. | Ħ |
| HOCHE | • | 80-36.8 | 1.3£+01 | A/A | ; | N/A | : | W/A | : | N/A | ; | 8.96-06 | 10+32.1 | N/N | ; | N/N | : | 76.8 | : | ; | . XE-03 | ¥. |
| JANGH | r | 7. IE-07 | 1.36+01 | N/A | : | #/# | : | N/N | : | 7.IE-07 1 | 1.35+01 | A/A | : | 7.1E-07 | 1.3E+01 | 7.16-07 | 1.36+01 | 378.0 | ł | ; | 20-34 | Ŧ |
| HOPEC | 2 | 5.06-08 | 1.36+01 | A/N | ; | N/N | : | N/A | : | N/A | : | N/N | : | 5.06-08 | 1.3E+01 | 5.0E-08 | 1.36+01 | 52.0 | : | ÷ | .46-01 | Ĩ |
| KOPIC | 2 | 5.06-08 | 1.3£+01 | N/A | 1 | 5.06-08 | 1.35+01 | N/A | ; | R/A | : | 5.06-00 | 1.36+01 | 5.06-08 | 1.3E+01 | N/A | ; | 93.6 | : | ; | 2. IE-03 | ١. |
| HOPVC | 2 | 5.06-08 | 1.36+01 | N/A | ; | 5.06-06 | 1.3€+01 | | ; | N/N | ł | A/A | ! | 5.06-08 | 1.3E+01 | 5.06-08 | 1.36+01 | 48.0 | ; | ; | Co-31.1 | Ŧ |
| HOGEC | • | 5.06-08 | 1.3E+01 | N/A | ; | 5.06-08 | 1.36+01 | A/N | ł | N/A | ł | A/A | : | 5.06-08 | 1.35+01 | 5.06-06 | 1.36+01 | 87.0 | ; | ; | 10-39.1 | ١. |
| HOGVC | • | A/N | : | R/N | : | W/W | : | A/8 | : | N/A | 1 | N/A | ; | 90-30.5 | 1.3E+01 | 5.06-08 | 1.36+01 | 87.9 | ; | ; | 20-31-3 | Ŧ |
| HOREC | 'n | 2.66-06 | 1.3£+01 | N/N | : | 2.65-06 | 1. 3E+01 | N/N | ; | 2.66-06 1 | 10+35.1 | A/A | ; | 2.6E-06 | 1.36+01 | 2.6E-06 | 1.36+01 | 160.5 | : | ; | 10-35.1 | ۲. ۱ |
| HORVC | ~ | 2.6E-06 | 1.36+01 | W/W | 1 | 2.6E-06 | 1.3E+01 | A/N | ; | 2.66-06 1 | 10+3£ -1 | N/A | ł | 2.6E-06 | 1.36+01 | 2.66-06 | 1.36+01 | 150.0 | ; | ; | - 9E - 05 | ۲. |

01.0.1.0.1.0.1.0.1.0.1.0.1.0

5)

22233



X



PAGE 2 DATE 21-Aug-87

ONSITE DISPOSAL OPTION (PEA PALLET OR CONTAINER)

| (10ms |
|-----------------|
| 0er 2 |
| G U 1 NG |
| |
| Onsit |
| ş |
| Aumont - |
| 2 |
| ACC den |

Agent Available and Released

| | SCENNEIO | GMA | RANGE | APS | RANGE | EAB. | RANGE | MAR | RANGE | Ma | NAME | PUBA R | ANGE | TEAD R | NACK. | a wown | MIGE | AGENI | 587 | ŝ | 8 | KURAT CON |
|--------|----------|------------|----------|-----------|---------|----------|----------|---------|---------|-----------|-------------|-----------|--------|------------|---------|------------|---------|----------|-----------|-----------------|----------|-------------|
| | WHEEK | FREQ | FACTOR | FREQ | FACTOR | FRED | FACTOR | FIRED | FACTOR | FRED | ACTOR | FREQ F | ACTOR | FRED F | ACTOR | FRED | ACTOR A | VAILABLE | SPILLED 1 | ETOMATED | ENLITED | 11 1 |
| | | | | | | | | | | | | | | | | | | | | | | |
| JASOM | - | N/N | ; | A/N | ; | A/M | ; | N/A | ; | N/A | : | A/N | ; | 1.46-07 1 | . 3E+01 | 1.65-07 | 1.36+01 | 1356.0 | 1 | ; | 2.76-04 | Æ |
| HOBEC | - | A/A | ; | N/A | : | N/N | ; | N/A | : | N/N | : | A/N | : | 1.76-09 1 | . 3E+01 | 1. 7E-09 | 1.35+01 | 440.0 | ; | ; | 4.36+00 | Ĩ |
| HODHC | * | 8.06-10 | 1.36+01 | N/A | : | A/A | ; | M/A | : | #/# | 1 | 8.06-10 1 | 10+35 | 1.06-10 1 | .35+01 | N/N | : | 200.0 | ; | 1 | 1.35-03 | Ŧ |
| NOCEC | • | 2.06-11 | 1.36+01 | A/A | ; | N/A | ; | N/N | : | W/W | ł | W/W | ; | 2.06-11 1 | .X +01 | E N | : | 38.4 | 1 | 1 | 1.16-01 | Ĩ |
| HOCHC | • | 2.0E-11 | 1.35+01 | N/A | ; | R/A | : | A/N | : | N/N | : | 2.06-11 1 | 10+X. | W/W | ; | V/H | ; | 76.8 | ; | ł | 1.36-03 | Ŧ |
| MOKEC | - | A/A | : | N/N | : | A/H | : | N/N | 1 | N/N | : | . W/W | 1 | 1.12-21.1 | .3E+01 | W/W | ; | 1500.0 | ; | ; | 6.46+00 | £ |
| HORMC | • | 7.25-09 | 1.3E+01 | 6. JE -09 | 1.3£+01 | , ILA | ł | N/N | ſ | 6.7E-09 1 | 1.36+01 | A/H | 1 | J. 7E-09 1 | .3E+01 | 1.26-09 | 1.36+01 | 1700.0 | : | : | 2.5E-02 | ۹. |
| MORVC | • | A/A | : | N/N | ; | N/N | ; | 1.76-09 | 1.32+01 | N/N | : | N/N | : | b. 7E-09 1 | .35+01 | A/M | ; | 1600.0 | ; | : | 2. 7E-04 | Ŧ |
| HOMVC | • | 6.9E-10 | 1.36+01 | A/A | ; | A/N | ; | A/A | ; | 6.96-10 | 1.3£+01 | N/N | ; | 1 01-36-9 | .3E+01 | 6.96-10 | 1.36+01 | 378.0 | ; | ; | 1.76-05 | Ĩ |
| HOPEC | • | 0.0E+00 | : | N/N | ; | N/N | ; | N/N | ; | N/N | ; | A/H | : | 0.0E+00 | ; | 00+30.0 | ; | 52.0 | : | 1 | ; | ; |
| CHACH | • | 0.0€+00 | : | N/N | ; | 0.0€+00 | ; | N/A | 1 | 8/N | : | 0.0€+00 | 1 | 0.06+00 | ; | N/N | ; | 93.6 | ; | ł | ł | : |
| JUPVE | - | 0.0€+00 | ; | N/N | 1 | 0.06+00 | ; | N/A | : | N/N | : | A/A | ! | 3.06+00 | 1 | 3,06+00 | ; | 48.0 | : | ; | ; | : |
| HOGEC | • | 0.0€+00 | : | N/N | ; | 0.0E+00 | ; | R/A | : | N/N | ; | A/M | ; | 0.06+00 | ł | 00+30" | : | 87.0 | : | ; | 1 | : |
| HODVC | • | A/N | : | N/N | ; | R/A | ; | N/N | : | N/N | : | #/# | 1 | 0.06+90 | ; | 0.0E+00 | ! | 87.0 | ; | ! | 1 | ł |
| HOREC | • | 3.16-09 | 1. 36+01 | N/A | : | 3. 16-09 | 1.3E+01 | N/A | : | 3.16-09 | 1.36+01 | A/N | : | S. IE-09 1 | .36+01 | 1. 1E-09 | 1.3E+01 | 160.5 | ; | ; | 4.56-01 | Ŧ |
| HORVC | • | 3. IE-09 | 1.36+01 | N/N | : | 3, IE-09 | 1.36+01 | N/A | ; | 3. 16-09 | I. 3E+01 | N/N | 1 | 3, 16-09 1 | 10+X. | S. 1E-09 | 1.36+01 | 150.0 | ; | ; | 1.65-05 | ٩. |
| HOSVC | • | N/N | ; | N/N | ; | N/N | ; | N/A | ; | R/A | ; | A/N | ; | 2.76-09 1 | .3E+01 | 2.7E-08 | 1.35+01 | 1356.0 | : | ; | 2.76-04 | Ĩ |
| 590M | 'n | N/N | ; | N/N | : | A/N | ; | A/N | ; | N/N | : | N/N | ; | 1.46-08 | .36+01 | J. JE -09 | 1.36+01 | 440.0 | 2.26+02 | ; | ; | ۲. |
| SHOOH | 5 | 4.4E-09 | 1.36+01 | A/H | : | N/N | ; | N/N | : | N/N | : | 4.46-09 1 | .3E+01 | 1.46-09 1 | .3E+01 | N/N | ; | 288.0 | 6.05+00 | ; | 1 | ž |
| HOCES | 'n | 5.66-10 | 1.3E+01 | N/N | ţ | N/N | ; | N/N | ł | M/A | : | N/A | 1 | 5.66-10 1 | | N/N | ! | 38.4 | 1.66+00 | ; | : | Ĕ. |
| HOCHS | 5 | 5.66-10 | 1.3E+01 | N/N | : | A/N | ; | N/A | ; | A/N | : | 5.66-10 1 | .JE+01 | A/A | ; | A/4 | : | 76.8 | 3.26+00 | ; | ; | £ |
| HOY.65 | 'n | N/A | : | N/N | ; | N/N | : | N/N | ; | N/N | ; | M/A | ; | 5.06-09 1 | | N/N | ; | 1500.0 | 1.56+03 | î | : | 9¥. |
| NUKIS | ŝ | 5.06-08 | 1.36+01 | 5.06-08 | 1.3E+01 | N/A | ; | R/A | ł | 5.06-08 | 10+32.1 | 4/# | : | 5. UE-08 1 | .35+01 | 5. 0E-08 | 1.36+31 | 1700.0 | 1.7€+03 | : | ; | £ |
| SA.40H | r | N/A | ; | N/N | ; | N/N | : | 5.06-08 | 1.3£+01 | A/N | ł | A/M | ; | 5.05-08 | . 35+01 | N/A | 1 | 1600.0 | 1.6€+03 | { | ł | an a |
| SANDH | 'n | 3.4E-09 | 1.35+01 | N/N | : | A/A | ; | N/A | ; | 3.46-09 | 1.36+01 | N/A | ; | 3.45-09 1 | . 3E+01 | 3.4E-09 | 1.36+01 | 378.0 | 1.0€+01 | ; | : | <u>AN</u> |
| HOP 65 | 5 | 1.16-09 | 1.36+01 | A/N | : | N/N | ; | R/N | : | N/A | : | N/A | ; | 1.16-09 1 | | 1.16-09 | 1.3E+01 | 52.0 | 6.5E+00 | ł | : | Ě. |
| NOFHS | *7 | 1. IE -09 | 1.36+01 | R/A | : | 1.16-09 | 1. 36+01 | N/N | ł | N/A | ; | 1.16-09 | ; | 1.16-09 1 | .35+01 | N/A | : | 93.6 | 1.26+01 | 1 | : | ۹. |
| SAUDH | r | 1.1E-09 | 1.3£+01 | N/N | ; | 1. IE-09 | 1.3€+01 | R/A | ; | R/N | : | N/A | 1 | 1.16-09 | .3€+01 | 1.16-09 | : | 48.0 | 6.05+01) | ł | : | £ |
| SODOH | *7 | 1. 15-09 | 1. 36+01 | N/N | : | 1.16-09 | 1.3E+01 | N/A | ł | A/A | 1 | N/N | ; | 1.1E-09 1 | .36+01 | 1.16-09 | : | 87.0 | 10+35-1 | ; | ; | ÷. |

PAGE 3 DATE 21-Aug-67

R

DISITE DISPOSAL SPTION (PER PALLET DA CONTAINER)

| ŝ |
|----------------|
| Operations |
| Mandi I ne |
| r Desite |
| Frequencies fo |
| Accident. |

Agent Available and Released

| | SCENARIO | ł | | 54 | | ļ | Mark | ł | Ĭ | Ē | Mark | Ĩ | NAME | | Xamu | - WOWN | TANKE | AGENT | 51 | | 8 | HURAT I OH |
|----------------|----------|---------|------------|-------------|----------|-------------|-------------|-----------|----------|------------|-------------|------------|-------------|-----------|------------|--------------------------------|--------------|-------------|----------------|----------|---------|------------|
| | NUMBER | | FACTOR | FREG | FACTOR | | FACTOR | E MER | | 1460 | FACTOR | BR6 | FACTOR | FRED | FACTOR | | FACTOR | AVAILABLE | 311165 | DETONATE | ENLITED | ¥. |
| | ~ | | : | 4 /1 | : | 4 /8 | : | | ľ | 8/8 | ; | A/4 | ; | 1.16-09 | 1. XE+01 | 1.16-09 | 1 | 87.0 | 1.56+01 | : | ! | Ē |
| SSMOW | ~ | 1.46-08 | 1. 36 • 01 | A/N | ; | 1. 1 | 1.36+01 | | ; | 1.46-00 | 1.36+01 | N/N | ; | 1.45-00 | 1. 36+01 | 1.45-00 | 1.3£+01 | 160.5 | 1.16+01 | ł | ; | Ē |
| SVIIDH | ~ | 1.46-08 | 1.3£+01 | A/N | : | 1.45-00 | 1.36+01 | A/A | : | 1.46-90 | 1.36+01 | N/N | : | 1.45-00 | 1.36+01 | 1.45-08 | 1.36+01 | 150.0 | 1.06+01 | : | : | Ē |
| SV20H | • | N/N | : | A/A | : | A/A | ; | A/A | ; | A/N | : | A/N | ; | 1.46-04 | 1.36+01 | 1.46-06 | 1.3£+0[| 1354.0 | 1.46+03 | ł | ; | Ē |
| 1980H | -9 | N/A | ; | N/N | ; | A/W | : | A VI | ł | A/H | ; | N/N | ; | 11-32.9 | 1.16+01 | 6.25-11 | 3.16+01 | 440.0 | : | ; | 2.26+01 | IO NIN |
| Jan | • | 1.91-10 | 3.16+01 | R/A | : | A/N | : | A/B | ; | N/N | : | 1.96-10 | 3. IE+01 | 1.96-10 | 3.16+01 | #/# | : | 2007 O | : | : | 3.06-01 | IO NIN |
| NOCEF | -0 | 0.0€+00 | : | N/A | : | N/N | : | A/N | : | #/# | • | A/A | : | 0.0€+00 | : | A/N | ; | 38.4 | : | : | ; | : |
| | • | 0.0€+00 | : | A/H | ; | A/N | : | A/A | : | A/N | : | 0.0£+00 | ; | A/A | : | N/A | : | 76.8 | ; | : | : | : |
| 10 K BL | • | N/A | ; | R/A | : | A/A | : | R/A | : | A/N | : | R/A | ; | 2.16-10 | 3.16+01 | A/H | : | 1500.0 | : | : | 1.5€+02 | IO NIN |
| E Ser | 4 | 2.16-10 | 3. 16+01 | 2.16-10 | 3. 16+01 | A/A | : | A/M | : | 2. IE-10 | 3.16+01 | A/H | : | 2. IE-10 | 3. 16+01 | 2.1E-10 | 3.16+01 | 1700.0 | : | : | 8.56+01 | IO RIM |
| HOKVF | 4 | R/A | : | N/A | : | A.M | ; | 2. 1E-10 | 3. IE+01 | ¥/# | : | A/A | : | 2.16-10 | 3. 16+01 | N/N | ; | 1600.0 | : | ; | 4.0€+01 | 10 RIM |
| NOW | • | 2.7E-10 | 3. IE+01 | A/A | : | N/N | : | A/N | ; | 2.76-10 | 3.16+01 | R/A | : | 2.76-10 | 3. 16+01 | 2. XE-10 | 3. IE+01 | 378.0 | ; | ł | 2.66-01 | NJN 01 |
| 33-0H | • | 0.06+00 | : | N/A | : | A/A | ; | N/N | : | N/N | ; | A/N | ; | 0.0E+00 | : | 0.0E+00 | ; | 52.0 | ; | ; | : | : |
| H-10H | 4 | 0.0E+00 | | A/A | ; | 0.0E+00 | : | N/A | ; | 8/8 | : | 0.0€+00 | : | 0.06+00 | : | N/A | ; | 13.6 | ; | ł | : | : |
| HOPVE | -0 | 0.0€+00 | : | N/N | : | 0.0€+00 | ; | A/N | : | 4/H | : | N/N | ; | 0.0£+00 | ÷ | 0. OE +00 | : | H . | ; | : | ; | : |
| HODEE | -9 | 0.0€+00 | ; | N/N | ; | 0.0£+00 | : | A/N | : | A/N | : | N/N | : | 0.00+30.0 | : | 0.0E+00 | : | 0.78 | ł | : | ; | : |
| HOGVF | 4 | N/A | : | | : | A/N | : | A/N | ; | R/A | : | N/N | ; | 0.0E+00 | : | 0.0E+00 | : | 87.0 | : | : | ; | : |
| HOREF | 4 | 1.46-10 | 3.16+01 | N/N | ; | 1.66-10 | 1.16+01 | A/A | ; | 1.46-10 | 3. IE+01 | N/N | : | 1.46-10 | 3. 16+01 | 1.46-10 | 3.16+01 | 160.5 | ; | ; | 1.16+00 | IO NIN |
| HORVE | -0 | 1.66-10 | 3. IE+01 | A/N | ; | 1.66-10 | 3.16+01 | A/N | ; | 1.46-10 | 3. 16+01 | N/N | ; | 1.46-10 | 3. IE • 01 | 1.46-10 | 3.16+01 | 150.0 | ; | ; | 2.56-01 | In NIN |
| HOSVE | -0 | N/N | ; | N/N | ; | N/N | ; | A/A | ; | A/H | : | A/N | : | 3, 16-09 | 3. IE+01 | 5. 16 -01 | 3, 16+01 | 1356.0 | 1 | : | 3.46+01 | 10 NIN |
| 59 0 0H | ~ | N/A | ; | N/N | 1 | A/N | ; | N/N | ; | R/A | ; | R/N | : | 1.25-09 | 1. X •01 | 1.25-09 | 1.36+01 | 140.0 | 2.26+02 | : | ; | Ĩ |
| 100H | 2 | 3.7E-09 | 1.3E+01 | N/N | : | A/A | : | A/N | : | N/A | : | 3. 76-04 | 1.3%+01 | 3.76-09 | I. X +01 | A/N | : | 200.0 | 6.0E+00 | : | ; | Ŧ |
| HOCES | ~ | 0.0E+00 | ; | R/N | ; | A/N | ; | N/N | : | A/A | ; | R/R | ; | 0.0£+00 | : | N/A | ; | 1 .6 | : | : | ; | ; |
| HOCKS | 1 | 0.0€+00 | ; | N/N | ł | R/A | ł | R/N | ; | A/N | ; | 0. EE +00 | ; | 8/8 | : | N/N | : | 76.8 | : | : | : | : |
| HOKES | ~ | R/A | : | N/N | : | A/A | ; | | ; | R/A | ; | | : | 4.16-01 | 1.36+01 | W/W | ; | 1500.0 | 1.56+03 | : | : | Ĩ |
| FOR | • | 4.15-09 | 1.3€+01 | 4.1E-09 | 1.3£+01 | A/N | : | R/A | ; | 4, IE-99 | 1. 36 • 01 | | : | 4.16-09 | 1. X • • I | 4. IE -01 | 1.36+01 | 1700.0 | 1.76+03 | ; | : | Ŧ |
| HOKVS | 7 | N/A | 1 | N/N | : | A/N | ; | 4. IE -09 | 1.3£+01 | R/A | ; | A/N | : | 4.16-09 | 1.36-01 | N/N | ; | 1400.0 | 1.46+03 | : | : | Ŧ |
| SVM0H | 1 | 5.4E-09 | 1.35+01 | N/N | ; | N/N | : | A/N | : | 5.46-04 | 1. % +01 | | : | 5.#-0 | 1. X • 01 | 5. K . 6 | 1.36+01 | 378.0 | 10+30.1 | : | : | Ŧ |
| 5940H | 1 | 0.0€+00 | ł | A/A | ; | A/N | 1 | | ; | N/N | ; | A/B | : | 0.6+00 | : | 0.0E+00 | ; | 52.0 | ; | ; | ; | ; |

000000

5555555 - 7775

TANKER RECENCE RECENCE AND RECENCE

TSSEE





10.00



DUSITE BISPOSAL OPTION (PER PALLET OR CONTAINER)

| 물 |
|-------------|
| Operations |
| Mand! 1 ng |
| Desite |
| Ē |
| Frequencies |
| Accident |

Agent Available and Released

| | SCENNING Immeria | ANNA Free | RANGE FALTION | 8 0 8 | RANGE Far Top | L DAD Erse | RANGE FACTOR | HAAP C BEG | PANGE LAFTIN | PIM | RANGE | PUDA | ANNUE ANT THE | TEAB R | | | | AGENT Mari and | | L DS Metromaten | LOS I | URATION |
|--------|---------------------|--------------------|------------------|---------------------|------------------|-----------------|-----------------|---------------------|-----------------|---------|---------|------------|------------------|-------------|--------------------|-----------------|---------|-------------------|---------|--------------------|----------------|-------------|
| | | | | | 5 | | | | | | | | | | 5 | | | | | | | |
| SHADH | 1 | 0.0£+00 | : | A/H | : | 0.0E+00 | : | A/N | : | | ł | 0, 0E +00 | : | 0.06+00 | : | A/A | ; | 93.6 | ; | : | : | ; |
| SAJOH | 1 | 0.0€+00 | : | 1 /1 | ; | 0.06+00 | ; | N/N | : | | : | R/N | ; | 0.05+00 | ; | 00+30. | ; | 48.0 | ; | ł | ; | ; |
| 5980+ | • | 0.0€+00 | : | N/N | : | 0.0€+00 | ; | A/A | : | A/A | 1 | N/N | : | 0.05+00 | : | 00+30 | : | 87.0 | ; | : | 1 | ł |
| SVOON | ~ | A/A | : | N/N | ; | A/N | : | A/N | ; | N/N | ; | N/N | : | 0.0E+00 | : | 06+00 | : | 87.0 | ; | : | : | ; |
| SBIOM | ~ | 3.26-09 | 1.3£+01 | A/A | : | 3. 26 - 09 | 1.36+01 | A/N | : | 3.26-09 | 1.36+01 | N/N | 1 | 3.26-09 1. | 36+01 3 | - 3E - 09 | 1.3€+01 | 160.5 | 1.16+01 | ; | ; | Ħ |
| SVIIDH | 1 | 3.26-09 | 1. X +01 | A/N | ; | 3.26-09 | 1. 3£+01 | N/N | ; | 3.26-09 | 10+3(:1 | R/A | : | 3.26-09 1. | 36+01 3 | 60-3Z | 1.36+01 | 150.0 | 10+30.1 | : | : | ۹. |
| SVSOH | ~ | A/M | ; | R/A | ; | NI N | : | A/A | : | R/N | : | N/N | ; | 7.76-00 1. | .3E+01 7 | . 75 -08 | 1.36+01 | 1336.0 | 1.4E+03 | ; | ; | ۴. |
| JHODH | Ξ | 2, 96-09 | 2.6€+01 | R/N | : | A/A | : | N/N | : | A/A | ; | 2.96-09 | 2.46+01 | 2.96-09.2 | 10+39 | N/N | : | 286.0 | ; | 6.06+00 | L. 3E -03 | ۴. |
| HOCEC | = | 1.45-09 | 2.46+01 | N/N | : | A/N | ; | V | : | N/N | ; | W/W | : | 1.46-09 2 | 10+39 | A/N | : | 38.4 | ; | 1.46+00 | 1.46-01 | and a |
| HOCH | = | 1.46-09 | 2.46+01 | A/N | ł | A/A | ; | N/N | ; | R/A | : | 1.46-99 | 2.46+01 | A/A | ; | N/A | : | 74.8 | ; | 3.26+00 | B. 06-04 | ۹. |
| JVNDH | = | 2.25-09 | 2.46+01 | N/A | : | R/A | : | N/N | : | 2.26-09 | 7.66+01 | N/N | ; | 2. 7E-09 2. | 10+34 | 50-32° | 2.46+01 | 378.0 | ; | 3.25+01 | 4.25-05 | £ |
| HOP GC | Ξ | 4. BE -10 | 2.46+01 | N/N | : | A/N | : | 4 / 8 | ; | N/N | ; | N/N | : | 4.86-10 2 | 10+39 | 01- 36 - | 2.46+01 | 52.0 | ; | 6.5£+00 | 1.1E-01 | H H1 |
| HOP HC | Ξ | 4.8E-10 | 2.66+01 | N/N | : | 4.66-1 0 | 2.66+01 | N/N | : | A/A | : | 6.0E-10 2 | 10+19. | 4.0E-10 2. | 10+39 | A/A | : | 93.6 | ; | 1.2E+01 | 7.06-04 | ٩. |
| JV90H | H | 4, 66 - 10 | 2.46+01 | A/N | 1 | 4.86-10 | 2.66+01 | N/N | : | N/N | : | N/N | : | 4.#E-10 2. | 10-31 | .01-30. | 2.4E+01 | 48.0 | : | 6.0E+00 | 1.06-05 | ١. |
| 1980H | Ξ | 3.46-10 | 2.66+01 | N/N | : | 3.46-10 | 2.46+01 | N/N | ; | A/A | ; | R/N | ; | 3.4E-10 2. | 66+01 3 | .46-10 | 2.46+01 | 87.0 | ; | 1.5€+01 | 1.76-01 | ۴I |
| JADOH | Ξ | A/A | : | #/# | ; | A/A | : | N/N | ; | A/N | : | A/N | : | 3.46-10 2. | 6E+01 3 | .66-10 | 2.66+01 | 87.0 | ; | 1.5€+01 | 6.0E-06 | ٩. |
| HURBC | = | 9.06-10 | 2.46+01 | A/A | : | 1.06-10 | 2.66+01 | A/N | : | 9.06-10 | 2.65+01 | N/N | ; | 9.06-10 2. | 6 10+39 | 01-30. | 2.6€+01 | 160.5 | ; | 2. IE+01 | 1.06+00 | Ĩ |
| HORVC | Ξ | 9.06-13 | 2.46+01 | A / A | : | 9, 0E -10 | 2.66+01 | #/# | : | 9.06-10 | 2.46+01 | A/A | ; | 9.06-10 2 | 10-39 | 01-30 | 2.66+01 | 150.0 | ; | 2.06+01 | 5.0E-06 | H. |
| JH00H | 12 | 2.16-10 | 2.46+01 | A/A | : | A/A | : | A/N | ; | N/N | : | 2. IE-10 2 | 10+34.1 | 2.16-10 2. | 10+39 [.] | N/N | ; | 288.0 | ; | 6.0E+00 | 1.3E-03 | ŧ |
| HOCEC | 21 | 1.06-10 | 2.6€+01 | A/A | : | A/N | : | | : | N/N | 1 | N/N | ; | 1.01-10 2. | 10+39 | A/A | ; | 38.4 | ļ | 1.66+00 | 1.65-01 | Ŧ |
| HOCK | 12 | 1-30-10 | 2.66+01 | W/W | : | A/M | : | A/H | ; | R/A | : | 1.06-10 2 | 10+34.1 | N/N | : | A/M | ; | 76.8 | ; | 3.26+00 | B. 0E-04 | Ŧ |
| JAMOH | 2 | 1.36-10 | 2.46+01 | A/N | : | F/H | : | N/N | : | 1.36-10 | 2.46+01 | 8/8 | : | 1.56-10 2. | 1 10+39 | ol-3C. | 2.46+01 | 378.0 | ; | 3.26+01 | 4.25-05 | Ē |
| 1990H | 12 | 11- 34 -5 | 2.6€+01 | W /W | ; | N/A | : | N/N | ; | N/N | : | 8/8 | : | 3.46-11 2. | 6 10+39 | .46-11 | 2.6E+01 | 52.0 | ; | 6.56+00 | 1.16-01 | ۴. |
| JH40H | 21 | 3.46-11 | 2.4E+01 | A/A | : | 3.46-11 | 2.66+01 | A/N | : | N/N | ; | 3.46-11 | 10+39.3 | 3.46-11 2. | 6E+01 | N/N | ; | 93.6 | : | 10+32-1 | 7.05-04 | Ē |
| 10PVC | 12 | 3.46-11 | 10-39.5 | W/W | : | 3.46-11 | 2.66+01 | W/W | : | N/A | ; | N/A | : | 3.46-11 2. | 96+01 3 | H-34. | 2.66+01 | 48.0 | ; | 6.0E+00 | 1.05-05 | œ. |
| 1900H | 12 | 2.46-11 | 2.46+01 | A N | : | 2.46-11 | 2.66+01 | A/N | : | N/N | : | A/N | : | 2.66-11 2. | 66 +01 2 | 11-39. | 2.65+01 | 87.0 | : | 1.56+01 | 1.76-01 | Ŧ |
| HORVE | 21 | N/N | ; | A/N | : | N/N | ; | N/N | ; | N/N | ; | N/N | ; | 2.6E-11 2. | 6E+01 2 | 11-39. | 2.66+01 | 87.0 | ; | 1.56+01 | 6.0E-06 | Ē |
| DBAGH | 2 | 6. 3 E - 11 | 2. d€ +01 | N/A | ; | 4.5E-11 | 2.66+01 | #/# | ; | 4.35-11 | 2.46+01 | A/A | : | 4. SE-11 2. | PE +01 - 1 | 11-35. | 2.66+01 | 160.5 | ; | 2. JE+01 | 1.0£+00 | Ħ |
| HORVE | 12 | 6.56-11 | 2.46+01 | ¥/# | ; | 6. X - 11 | 2.6E+01 | N/A | ; | 11-35.4 | 2.46+01 | N/A | : | 6.5E-11 2. | 9 10-39 | 11-35 | 2.6E+01 | 150.0 | ; | 2.9€+01 | 5.06-06 | Ĩ |

.....

N. A. C. C. C.

. . . .

. . . .

PAGE 5 DATE 21-Aug-87

ŝ . for flacted Vanding ident Ers

ONSITE DISPOSAL OPTION (PER PALLET OR CONTAINER)

1.10°.245,245.4

| | | | | | | | Accide | mt Frequ | - HI I HI | for Bmsitt | e Kanél se | g Operatu | (DH) SUD | | | | | | Agent Av | allable ac | od Rei easu | 2 |
|------------|--------------------|------------|-----------------|---|-----------------|--------------|-----------------|----------|------------------|-------------|-----------------|-----------|-----------------|--------------|-----------------|------------------|-----------------|--------------------|--------------------|------------------|-------------|------------------|
| | SCENARIO Number | AIINO | RANGE Factor | a angla a Baga angla | PANKE Factor | L MO L MO | RANGE Factor | ANN BURN | RANGF. FACTOR | PIA Free | RANGE FACTOR | PUBA | RANGE Factor | TEAD Fred | RANGE FACTOR | UNDA | RANKE FACTOR | AGENT AVAILABLE | SPILLED SPILLED | LDS DETOMATEI | LISS (| DURATION TINE |
| | 1 | ; | | | | | | | | | | | | | | | | | | | | |
| | 3 | | 10+34.2 | | ; | | : | | : | | ; | 8. /E -] | 2.65+01 | 8. /1-11 | 2.66+01 | | : | 288.0 | 3.0E+01 | 9.06+00 | ; | Ē |
| HOCEC | 22 | 1.36-11 | 2.66+01 | 8/8 | ; | A/H | : | N/N | : | W/W | : | N/N | ; | 11-36-11 | 2.66+01 | N/N | : | 38.4 | 8.0E+00 | 1.6€+00 | ; | Ŧ |
| HOCHC | u | 1.36-11 | 2.66+01 | N/N | : | A/N | : | A/A | : | A/A | : | 4. X-11 | 2.65+01 | A/N | ; | N/N | ; | 76.8 | 1.46+01 | 3.ZE+00 | : | Ĩ |
| HOMVC | ដ | 6. 36 -11 | 2.66+01 | N/N | ; | A/N | : | N/N | ; | 6.5E-11 | 2.46+01 | A/A | : | 6.36-11 | 2.6E+01 | 6.35-11 | 2.46+01 | 378.0 | 3.56+02 | 3.76+01 | ; | 9 |
| HOPEC | :: | 1.46-11 | 2.6E+01 | A/N | ; | A/N | : | N/N | ; | A/A | ł | N/N | : | 1.46-11 | 2.6E+01 | 1.46-11 | 2.66+01 | \$2.0 | 3.25+01 | 6.5E+00 | ; | 9 |
| HOH | 22 | 1.46-11 | 2.66+01 | N/N | : | 11-34-11 | 2.46+01 | N/N | ; | A/A | 1 | 1.46-11 | 2.66+01 | 1.46-11 | 2.6E+01 | N/N | : | 93.6 | 5.86+01 | 1.25+01 | ; | 9 |
| HOFVC | 53 | 1.46-11 | 2.46+91 | N/N | ; | 1.46-11 | 2.66+01 | A/A | ; | 8/N | : : | N/N | : | 11-34-1 | 2.65+01 | 1.46-11 | 2.66+01 | 48.0 | 3.06+01 | 6.0E+00 | ł | Ĩ |
| HODEC | ន | 11-31.1 | 2.6E+01 | A.A | ; | 1.11.11 | 2.66+01 | N/A | ; | N/A | ; | N/N | : | 1.16-11 | 2.46+01 | 1.16-11 | 2.66+01 | 87.0 | 7.36+01 | 1.36+01 | ; | 9 |
| HORVE | 22 | A/M | : | N/A | : | N/N | : | A/H | : | N/N | ; | N/N | ; | 1.16-11 | 2.6E+01 | 1.1E-11 | 2.6E+01 | 87.0 | 7.36+01 | 1.56 +01 | ; | 9 |
| HOREC | ដ | 2.75-11 | 2.66+01 | M/A | ; | 2.76-11 | 2.66+01 | N/A | ; | 2. 16-11 | 2.66+01 | N/A | ; | 2.75-11 | 2.6E+01 | 2. <i>T</i> E-11 | 2.6E+01 | 140.5 | 1.46+02 | 2.16+01 | : | Ĩ |
| JVPDH | H | 2.76-11 | 2.46+01 | R.N | : | 2.76-11 | 2.46+01 | A/A | : | 2.76-11 | 2.66+01 | N/A | ; | 2.76-11 | 2.6E+01 | 2.75-11 | 2.65+01 | 150.0 | 1. 36+02 | 2.0€+01 | ; | 9 |
| HOH. | 54 | 6.X-11 | 2.6E+01 | M/N | : | N/N | ; | A/N | : | N/A | ; | 6.25-11 | 2.6€+01 | 6. ZE-11 | 2.4E+01 | N/N | ; | 288.0 | 3.06+01 | 6.0E+00 | ; | E. |
| NOCEL | 7. | 3. 16 - 11 | 2.04+01 | A/N | ; | N/A | : | 4/N | ; | N/A | ; | N/A | : | 3. IE-11 | 2.66+01 | N/A | : | 38.4 | 8.9£+00 | 1.46+00 | ; | Ĩ |
| HOCH | 7 . |]. IE 11 | 2.66+01 | N . A | ; | A/A | : | N/N | ; | N/A | ; | 3.16-11 | 2.66+01 | N/A | ; | N/A | : | 76.8 | 10+39.1 | 3. ZE+00 | : | ٩. |
| JANGH | 7 . | 1. df - 11 | 2.65+01 | 4/8 | ; | N/A | : | N/A | ; | 4.66-11 | 2.6E+01 | N/A | 1 | 4.66-11 | 2.6E+01 | 4,66-11 | 2.6€+01 | 378.0 | 3.56+02 | 3. 2E+01 | ! | ۹. |
| 35aGH | . | 1.06-11 | 2.46+01 | N/A | ; | W / W | : | N/N | ; | R/A | ; | A/N | ; | 11-30.1 | 2.6E+01 | 1.06-11 | 2.66+01 | 52.0 | 3.26+01 | 6.5£+00 | : | ۹. |
| HOFHC | 7. | 1.0% 11 | 2.66+01 | N/A | : | 1.06-11 | 2.46+01 | A/N | ; | R/M | : | 1.06-11 | 2.6E+01 | 11-30-1 | 2.66+01 | A/A | : | 93.6 | 5.0€+01 | 1.26+01 | ; | Ĩ |
| 34304 | 7 . | 11-30' | 2.66+01 | A/M | : | 11-06-11 | 2.66+01 | R/A | : | M/A | ; | N/A | : | 11-30.1 | 2.46+01 | 1.0E-11 | 2.6E+01 | 48.0 | 3.06+01 | 6.0€+00 | : | 9 <u>+</u> |
| HODEC | 2 | 7.75-12 | 2.66+01 | R . N | : | 7.76-12 | 2.46+01 | N/N | ; | N/A | ; | N/A | ; | 7.76-12 | 2.66+01 | 7.7E-12 | 2.66+01 | 87.0 | 7. 36+01 | 1.56+01 | ; | Ŧ |
| JADOH | ₹. | ¥/₩ | : | N/A | : | A/A | : | N/A | ; | N/A | ; | N/A | ; | 1.78-12 | 2.6E+01 | 7.76-12 | 2.65+01 | 67.0 | 7.36+01 | 1.5£+01 | ; | Ĩ |
| N 396C | 2 | 1.91 | 2.66+01 | N/A | ; | 1.96.1 | 2.46+01 | N/A | : | 11-36.1 | 2.6E+01 | N/# | ; | 11-36-11 | 2.6E+01 | 1.96-11 | 2.6E+01 | 160.5 | 1.46+02 | 2.16+01 | ; | ۹. |
| HPRVC | 2 | 1.96.1 | 2.66+01 | N/A | ; | 11-36-11 | 2.46+01 | A/M | i | 11-36.1 | 2.65+01 | N/A | ; | 11-36-1 | 2.6E+01 | 1.96-11 | 2.6E+01 | 150.0 | 1.3E+02 | 2.0€+01 | : | Ť |
| HCBEE | <u>ب</u> | A/A | : | N,A | ; | A/N | : | A/N | | N/A | ; | N/A | : | 0.0E+00 | : | 0. 0E +00 | ; | 440.0 | : | : | : | : |
| HEDHC | 9 . | 0.0€+00 | : | A/N | ; | A/H | ł | A/A | ; | N/A | ; | 0.0E+00 | : | 0.0E+00 | ; | M/A | ; | 289.0 | : | : | : | : |
| HOCEC | 2 F | 0.0E+0c | : | NIA | ; | A/M | ; | A/A | ; | N/A | ł | A/A | : | 0.0E+00 | : | M/N | ; | X8.4 | : | : | ; | ; |
| HOCHC | 26 | 0.05+00 | : | N/A | • | N/N | | R/A | : | N/N | ; | 0.0E+00 | ; | N/A | ; | N/N | : | 76.8 | ; | : | : | : |
| HOK 6F | 26 | A/A | | A/A | ; | N/A | : | A/N | ; | A/# | ; | N/N | : | 0.0E+00 | ł | N/N | ; | 1500.0 | ; | : | ; | : |
| ¥ AN | 2 9 | 0.96+30 | ; | 0.0E+00 | ; | A'A | ; | A/A | ; | 0.0£+00 | : | N/N | : | 0.0E+00 | ; | 0.0€+00 | ; | 1700.0 | ; | ţ | : | : |
| 5 | 5a | A/A | : | #/# | ; | A/A | : | 0.0€+00 | : | N/A | ; | A/N | : | 0.0€+00 | ; | N/A | : | 1600.0 | : | ; | ; | ; |







ONSITE DESPOSAL OPTION (PER PALLET OR CONTAINER)

| ŝ |
|-------------|
| Operations |
| Hand! I ng |
| Desite |
| ţ |
| Frequencies |
| Accident |

Agent Available and Beleased

| SCENAR | A MAD | FACTOR | REG FREG | EANER FACTO | | Indice Factor | 3 2 | RANKE FACTOR | A C | FACTOR | 12 | FACTOR | TENB FRED | TACTOR FACTOR | UMDA Freg | | ANNIL MALE | 51110 51110 | LIS DETOMATEI | ENTTED | |
|--------|-----------|--------|-------------|----------------|---------|------------------|------------|-----------------|-----------|--------|---------|--------|--------------|------------------|----------------|---|------------|----------------|------------------|--------|---|
| | | | | | | | | | | | | | | | : | | ļ | | | | |
| . 4 | 26 0.06+0 | : | 18 | : | A/H | ; | N/N | : | 0-00-00-0 | : | N/N | : | 0.0E+00 | : | 0.05+08 | : | 37.0 | : | ļ | : | ; |
| , • | 26 0.06+0 | : | 1 | - | R/A | : | A/A | 1 | N/N | : | A/H | ; | 0.05+00 | : | 0.0E+00 | ; | 52.0 | ; | ; | ; | ; |
| | 26 0.06+0 | : | Ì | | 0.05+00 | : | A/N | ; | R/R | : | 0.05+00 | ł | 0,0£400 | : | A/N | : | 93.6 | : | ; | : | : |
| .4 | 26 0.06+0 | : | / H | - | 0.0€+00 | 1 | A/N | 1 | | ł | A/A | : | 0.06+40 | ł | 9, 0E +00 | : | . | : | ; | : | : |
| , 4 | 26 0.06+0 | : | 1 | | 0.06+00 | ; | W/W | ; | A/M | : | A/M | : | 0.0E+00 | ; | 0.0E+00 | : | 17.4 | : | ; | : | : |
| | 26 N/ | : | | | A/H | : | A/A | 1 | A/H | 1 | A/8 | : | 0.05+40 | 1 | 9, 95, 40 | ; | 07.0 | : | ; | ; | ; |
| | 26 0.0E+0 | : | N. | - | 0.0€+00 | ; | W/W | 1 | 0.05+00 | : | A/N | : | 0.0€+00 | : | 0.05+00 | : | 160.5 | : | ; | : | ; |
| .4 | 26 0.06+0 | : • | 1 | • | 0.06+00 | : | N/A | ; | 0.0€+00 | : | A.M | : | 0,0£+00 | 1 | 0.05+00 | ; | 150.0 | : | ; | ; | ; |
| | 26 11/ | : • | 1 | : | A/A | : | N/N | ; | N/N | i | A/A | : | 0,0£+00 | 1 | 0.0E+00 | ; | 1334.4 | : | : | : | ; |

 $\sim \sim \sim$

21-Aug-87 Page 1

| ATOLL | | |
|-------------|----------|---|
| IOHNIC [OH | | |
| 2 | | |
| 3 | | |
| FROM / | | |
| Ť. | | j |
| E | | 1 |
| 2 | Ł | - |
| ē | Ē | |
| ã | ĩ | 3 |
| 5 | 5 | Ì |
| 5 | يت. ح | - |
| 8 | Ĩ | į |
| N. | Ĩ | ľ |
| 2 | E | 1 |
| ž | ž | |
| | | |
| 4 | 7 | |
| | Ŧ | |

| | | | | | | £ | | | | | | 1.000 | | | | | | | 4 | at Avail | able and | Released | |
|-------|------|--------------|--------|--------|----------|--------|-------------|-----------|-------------|--------|------------|-------------|-------------|---------|--------|-----------|-------|---------|-----------|-----------|----------|----------|----------|
| | | | | | | | | Dent Pres | duencies | | | | | | | | | į | f | | | | 2 |
| Î | 3 | | G | 4 | ĝ | (11) | ŧ | 9 | Ĵ. | Ĵ | ŧ | Î | ê | ē | â | Ē | 6 | e | 2 | (× | | 3 | |
| -REN- | 1 | 2 | | SAME. | 500 | RANGE | (BAD | RANGE | ł | RANGE | ĩ | NAME | a MD | L 39M | EAD RA | ㅋ | NU NU | ¥ ۲ | | 2 | S | 5 | URAT LUN |
| DIN | | . <u>.</u> . | 838 | FACTOR | FRED | FACTOR | 560 | FACTOR | FREQ | FACTOR | FRED | AC FOR | FREG | ACTOR F | REG FA | ت 1961 | EG FA | TOP AVA | ILANCE SI | ,(LLED DE | TOWATED | ENTTEB | ШШ Ж |
| | ÷ | ; | | | : | | | | | | | | | 1 | | 1 | | : | | • | | | |
| | | | | ; | 90° 00 (| | | : | | : | | ; | 1 /1 | - | A/ | | • | ; | 1700 | 1700 | : | : | i hr |
| | | | | | | | | 1 | | : | | : | | : | • | - | | : | 1700 | ; | ; | B. SE+01 | 10 ein |
| | | | | : | VI-71. | | | | | 1 | | ; | | : - | | | | : | 1200 | 1700 | ; | : | - Pr |
| | | • | | ; | - 1/ - 0 | | | : | | ; | | | | | | | | : | 9021 | 1700 | ; | : | ž |
| | | 'n | | ; | - H - C | 17.8 | | : | | : | | ; | | | | | | | | | ; | 10725 0 | 2 |
| | | - | #/# | : | 3.16-10 | 31.1 | W /W | ; | | ; | | ; | | ; | | | E | 1 | 00/1 | | ; | 1 . Y . | E . |
| ST WE | | - | 4/H | : | 3.96-09 | 12.8 | N/N | : | N.A | : | 1 | ; | N/N | - | ¥ / | = | < | : | 8 | 8 | : | : | N |
| X | 6 | 1 | 4/H | ; | 0.06+00 | : | N/N | ; | 8 /8 | ; | N/N | : | N/N | : | R/1 | = | ¥. | : | 0021 | : | ; | : | * |
| ¥. | | | V. | ł | 6.06-66 | 0.01 | N/N | ; | A/A | : | 4/8 | : | #/# | : | ¥. | - | • | ; | 1700 | : | : | 1. IE+00 | ¥ |
| | | | | | | | | | | | | | | | | | | | | | | | |
| | , | | | | | | | | | | | | | | | | | | | | | | |
| | Free | i Re | bulala | Site | | i July | /311/3510 | | 61566 | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | | | | | | | | | | |

| IS DURATION TEB TINE | 8 | H. | .00 10HIN | | : | Ĩ |
|-------------------------|---|--------------|-----------|----------------|----------------|---------|
| LDS LI Ketomatenenti | = | | 2 | | : | |
| LDS ESPILLED | € | 1700.0 | | 1700.0 | ; | 1700.0 |
| AGENT AVAILABLI | ŝ | 1700.0 | 1700.0 | 1700.0 | : | 1700.0 |
| RANGE FACTOR | 9 | 12.8 | 31.1 | 12.8 | ; | 10.01 |
| LPEQ L | Ę | 2.96-09 | 3. 16-10 | 3, 96-09 | A/A | 6.0E-06 |
| ġ | ē | m | 4 | ~ | 3 8 | |
| 8 | Ξ | | | | | |
| SCEN- Meio | • | HAKIN | HINKINS | Heik is | HARTING | NH MH |

2022/228 [] 52222222 [622

72.22.22

12222233







MIEI8-Ang-07 PAGE

NUMERTING ACCIDENTS - COLLOCATION PROCESSING OFTION (AIR TRANSPORT) (PER CONTAINER OR PALLET)

| LANDE HAAP MANDE PAN LACTOR FINED FACTOR FINED |
|---|
| |
| |
| N/A N/A |
| N/A N/A |
| V/I - V/I - |
| 1.36+01 N/A B/A |
| 1.3E+01 N/A N/A |
| N/A N/A |
| 1 |
| 1.X+01 N/A N/A |
| 1.3E+01 N/A N/A |
| 1.32+01 N/A N/A |
| 1.3E+01 N/A N/A |
| A/A A/A |
| 8/8 8/8 |
| N/A N/A |
| 8/A 8/A |
| 1. X+01 N/A N/A |
| 1.3E401 F'A A'A |
| 8/8 8/8 |
| N/A N/A |
| A/N A/N |
| N/A N/A |
| N/A H/A |
| R/A R/A |
| 11/A 11/A |
| 8/A 8/A |
| N/A N/A |
| 8/8 8/4 |
| N/A N/A |

MTE18-Aug-07 PAGE2

NAMELING ACCIDENTS - COLLOCATION PROCESSING OPTION (AIR FRANSPORT) (PER CONTAINER OR PALLET)

| | ; |
|--|--------|
| | |
| | d Bean |
| | |
| | |
| | |
| | - |

| | | | | | | | K CLI | ent Fr | Ne sa Cheek | l Range | factors | | | | | | | | | Agent Avai | late and R | elessed |
|----------------|----|-------------|-----------------|------------|------------------|--------|------------------|------------|-----------------|-------------|-----------------|--------------|-----------------|-----------------|----------------|---------------|-----------------|--------------------|----------------|------------------|------------------|------------------|
| 900 - 10 10 | 2 | | RANKE Factor | £ £ | nange Factor | | Numb Facility | | RANGE Factor | A S | RANGE FACTON | PUIM Freq | RANGE FACTOR | | NAME Factor | UNIA Linea | nnige Factor | AGENT AVAILABLE | 108 SPILLED | LDS Hetomated | L96 Emitted | NUMATION 11NE |
| • | | | | | | | | | | | | | | | | | | | | | | |
| HARVE | 1 | N /N | : | N/N | : | A/A | : | A/N | : | A/H | : | N/N | : | 1.46-10 | 3. IE+01 | N/N | : | 150.0 | : | : | 2.55-01 | 10 NIN |
| SHOWN | 1 | W/W . | ; | N/N | : | N/N | : | N/N | : | N/A | ł | W/W | : | 4. IE-09 | 1.3E+01 | N/N | : | 1700.0 | 1.7E+03 | : | : | 1 |
| SHAM | î | N/N | ; | N/A | : | N/N | : | A/A | ; | N/A | ì | N/N | ; | 0.0€+00 | : | 1/1 | ; | 93.4 | ; | 1 | 1 | ; |
| SAMI | î | | : | N/N | ; | N/N | : | A/M | ; | V / | ; | N/A | ; | 0. AE +00 | : | N/A | ; | 18 .0 | : | 1 | ; | ; |
| HADES | ^ | W/N | ; | N/N | ,: , ' | N/N | : | N/N | : | N. | : | W/W | : | 0.0€+00 | : | W/W | : | 87.0 | ; | : | : | : |
| HARGS | ` | W/W | : | A/A | : | N/A | : | A/4 | : | #/# | : | N/N | : | 3.26-09 | 1.3£+01 | N/N | : | 160.5 | 1.16+01 | ł | ; | ± |
| HARVS | î | W/W | ; | A/A | ; | N/N | : | N/N | : | N/A | ; | N/N | : | 3.26-09 | 1.36+01 | N/N | ł | 150.0 | 1.4€+01 | : | ; | ¥ - |
| HAL IS | | W/W | ; | 3. 35 -00 | 1.38+01 | N/N | : | 8/8 | : | N/A | ; | N/N | : | 3. 35-00 | 1.3£+01 | W/# | : | 3400.0 | 1.75+03 | ; | ; | ¥ - |
| SHANN | 60 | A/M | : | N/N | : | 1.56-0 | 0 1.3E+01 | A/A | : | N/A | ; | A/H | : | 1.56-08 | 1.3£+01 | N/N | ; | 1404.0 | 1.25+01 | ; | ; | ∰ |
| SAMM | | N/N | ; | N/N | ; | 1.55-0 | 0 1.36+01 | A/N | ; | N/A | : | W /# | : | 1.56-06 | 1.36+01 | N/N | ; | 720.0 | 4.00+00 | 1 | ; | ¥ |
| NINGS | æ | W/W | ; | A/N | : | 1.5E-0 | 6 1.XE+01 | N/N | : | W/W | : | A/H | : | 1.55-00 | 1.3£+01 | R/A | : | 870.0 | 1.5€+01 | 1 | : | ¥ |
| HARGS | | A/H | ; | N/N | ; | 1.95-0 | 0 1.3£+01 | A/N | : | N/N | : | N/N | ; | 1.16-00 | 1.36+01 | N/N | ; | 642.0 | 1.16+01 | ſ | : | ¥ - |
| HARVS | æ | W/N | : | N/A | : | 1.96-0 | 0 1.XE+01 | A/N | : | N/N | : | N/N | 1 | 1.76-08 | 1.3£+01 | N/N | ; | 600.0 | 1.0€+01 | ł | ; | 9 - |
| HAR H | • | W/W | : | 1.06-09 | 3. IE+01 | N/N | ; | N/A | 1 | N/N | ; | N/N | : | 1.06-09 | 3. IE+01 | N/N | : | 3400.0 | 1 | ; | 0.5£+0] | 10 1111 |
| MAN | • | N/N | ; | N/A | : | 1.96-1 | 0 3.16+01 | N/A | : | N/N | ; | N/N | : | 1.96-10 | 3.16+01 | N/N | ; | 1404.0 | ; | ſ | 5. BE-01 | IO NEN |
| HAPVE | • | - | : | N/N | ; | 1-36-1 | 0 3.16+01 | R/A | ; | N/N | ł | R/N | ł | 1.96-10 | 3. IE+01 | W/W | : | 720.0 | 1 | ſ | 1.56-01 | IO NIN |
| HADGF | • | | 1 | N/N | : | 1-36-1 | 0 3.16+01 | N/N | ; | N/N | : | N/N | : | 1.96-10 | 3.16+01 | A/N | : | 870.0 | : | ; | 1.5£+00 | NIN OI |
| HARGE | • | N/N | 1 | N/N | : | 1-3t-1 | 0 3.1E+01 | A/N | ł | N/A | ; | N/A | ; | 0.4E-10 | 3.16+01 | N/N | : | 64 2.0 | : | ; | 1.16+00 | 10 MIN |
| HARVE | • | | ; | N/N | ; | 1-3t-B | 0 3.16+01 | R/N | : | N/A | ; | N/A | ; | 0. 4E-10 | 3. IE+01 | N/N | ; | 600.0 | ; | 1 | 2.56-01 | 10 NIN |
| HAK'NS | 9 | N N | : | 89-35 T | 1.3£+01 | N/N | ; | N/N | ; | V / | 1 | N/A | : | 1.36-08 | 1.3E+01 | N/N | : | 3400.0 | 1.7€+03 | 1 | : | ¥ |
| SHAMH | 9 | N /N | : | N/N | : | 2.46-0 | 9 1.36+01 | N/N | 1 | R/8 | ; | N/N | ; | 2.46-09 | 1.36+01 | N/A | ł | 1404.0 | 1.25+01 | ſ | ; | ¥ - |
| SVARH | 5 | #/# | : | H/A | ; | 2.46-0 | 9 I.X +01 | N/A | 1 | N/A | ; | N/N | ; | 2.46-09 | 1.36+01 | N/N | 1 | 720.0 | 6.0E+00 | 1 | ł | ¥ ~ |
| SBOWN | 2 | N/N | : | N/A | ; | 2.46-0 | 9 1.35+01 | N/A | : | A/H | 1 | N/N | ; | 2.45-09 | 1.36+01 | N/N | : | 870.0 | 1.55+01 | 1 | ; | ¥ - |
| HARGS | 2 | W/W | ł | N/N | : | 1.16-0 | B 1.3E+01 | N/N | ; | W/W | ł | N/N | ; | 1.16-00 | 1. JE +01 | N/N | ; | 642.0 | 1.16+01 | : | ; | 9 - |
| RARVS | 2 | W/W | : | R/A | : | 1.16-0 | B 1.3E+01 | N/N | : | 4 /# | ; | N/A | : | 1.15-08 | 1.36+01 | N/N | ł | 600.0 | 1.0£+01 | 1 | : | ¥ |
| HAPHC | Ξ | W/W | ! | N/A | 1 | 1.86-1 | 0 2.66+01 | N/A | 1 | W/W | ; | N/N | 1 | 9.46-10 | 2.66+01 | N/N | : | 93.4 | ; | 1.76+01 | 7.06-04 | ¥ - |
| HAPVC | Ξ | N/N | : | N/N | : | 1.86-1 | 0 2.6€+01 | N/N | ł | #/# | : | N/N | ; | 9.46-10 | 2.46+01 | N/N | ; | 48.0 | ; | 4.0€+00 | 50-3 4 .1 | ₩ |
| HADGC | = | A.A | : | N/N | : | 3.66-1 | 0 2.46+01 | N/N | : | R/N | ł | N/A | ; | 7.26-10 | 2.65+01 | N/A | : | 87.0 | ; | 1.56+01 | 1.76-01 | ¥ |
| HARFE | Ξ | N/N | : | N/N | ! | 9.06-1 | 0 2.6E+01 | N/N | ; | N/N | ; | N/A | ; | 1. BE - 09 | 2.6€+01 | N/N | : | 160.5 | : | 2. IE+01 | 1.0€+00 | £ |

3

1-76

22.20



ballE10-Aug-07 PAGE3

NAMOLING ACCINENTS - COLLOCATION PROCESSING OFIION IAIA TAANSPORT) (PER CONTAINER OR PALLET)

Accident Frequencies and Range Factors

Agent Available and Released

| ND AMAG RANKE APS RANKE LAA | MAD RANGE APS RANGE LEA Kee Factor Free Factor Fre | INDUE APS RANGE LBA | APS RANGE LIA | FACTOR FIRE | A R | | FACTOR | | RANGE Factor | ¥ ¥ | FACTOR | PUM | RANGE FACTOR | TEND Fred | RANGE FACTOR | | RANKE Factor | AGENT | 1901 190 | L 95 DE TOMATES | LDS Entited | DURATION TINE |
|----------------------------------|---|---------------------------|---------------------------|-----------------------|-----------------------|---|--------|-----|-----------------|------------|--------|-----|-----------------|--------------|-----------------|--------|-----------------|--------|-------------|--------------------|----------------|------------------|
| | | | | | | | | | | | | | | | | | | | | | 2 | 9 |
| 11 M/M | | | | | | | | : | | | : : | | | | 1 3 FEF | | | | 1 | 1.75.1 | | 9 |
| 12 N/A N/A 35-06-14 25-065 N/A | 14 16/14 35-00-11 21-00-16/14 14 16/14 3.100-11 2.100-00 16/14 | | | 3.45-11 2.45401 N/A | 3.46-11 2.46401 M/A | 2. ME-01 IVA | | ; ; | | | : | | : | | 2.46.0 | | 1 | | : | A. 06+00 | | . <u>.</u> |
| 12 N/A 2.46-11 2.46-01 N/A | (A 11/A 2.46-11 2.46-01 11/A | N/A 2.46-11 2.46401 N/A | IVA 2.46-11 2.46401 IVA | 2.46-11 2.46+01 N/A | 2.46-11 2.46+01 N/A | 2.4E+01 N/A | | ; | | W/H | : | | ; | 5.26-11 | 1 2.6640 | | : | 87.0 | ; | 1.5€+01 | 1.76-01 | ¥ |
| 12 N/A N/A 6.5E-11 2.6E+01 N/A | 14 11/A 6.56-11 2.66+01 11/A | K/A 6,56-11 2,66+01 N/A | H/A 7 6.56-11 2.66-01 H/A | 6.5E-11 2.6E+01 N/A | 6.56-11 2.66+01 N/A | 2.66+01 N/A | W/W | • | : | i'a | 1 | A/A | : | 1.36-16 | 0 2.66+0 | 1 8/4 | : | 160.5 | ; | 2. IE+01 | 1.06+00 | ¥ |
| 12 N/A N/A 6.56-11 2.66-01 N/A | 1A 16/A 6.56-11 2.66+01 10/A | K/A 6.56-11 2.66401 N/A | IKA 6.56-11 2.66401 N/A | 4.5E-11 2.4E+01 N/A | 4.5E-11 2.6E+01 N/A | 2.4E+01 N/A | R/N | | 1 | A/A | : | A/M | ; | 1-36-10 | 0 2.46+0 | N N/A | : | 150.0 | ł | 2.06+01 | 5.06-01 | Ŧ |
| 17 N/A N/A 4.06-13 2.46+01 N/A | 1A 11/A 4.0E-13 2.6E+01 11/A | N/A 6.0E-13 2.6E+01 N/A | N/A 6.0E-13 2.6E+01 N/A | 6.0E-13 2.6E+01 N/A | 4.0E-13 2.4E+01 N/A | 2.46+01 N/A | W/W | | ; | A/A | 1 | N/N | : | 4.06-1 | 3 2.4640 | AN I | 1 | 160.5 | 1 | ł | 1.45-01 | ¥ - |
| 17 N/A N/A 6.06-13 2.46+01 N/A | ra Bra 6.06-13 2.66-01 N/A | B/A 6.06-13 2.46+01 N/A | N/A 6.06-13 2.66+01 N/A | 6.0E-13 2.4E+01 N/A | 4.0E-13 2.4E+01 N/A | 2.4E+01 N/A | N/N | - | : | 8/8 | ł | N/N | ; | 9-96-1 | 3 2.46+0 | N/N E | : | 150.0 | ! | 1 | 9-27-9 | ¥ |
| IB N/A N/A 1.25-12 2.66+01 N/A - | /A N/A 1.26-12 2.66-01 N/A - | N/A 1.26-12 2.66-01 N/A | N/A 1.26-12 2.66+01 N/A - | 1.26-12 2.46+01 N/A | 1.2E-12 2.6E+01 N/A - | 2.4E+01 N/A - | - VI | • | , | N/A | 1 | N/N | : | 1-32-1 | 2.66+0 | AVA I | ł | 10.7 | 1 | : | 1.45-01 | ¥ |
| IB N/A K/A 1.26-12 2.46+01 N/A - | ra Nra 1.26-12 2.46401 Nra - | N/A 1.26-12 2.46+01 N/A | HZA 1.26-12 2.46401 HZA - | 1.2E-12 2.4E+01 N/A | 1.26-12 2.46+01 N/A - | 2.4E+01 N/A - | - V/I | ' | | V | ; | W/W | : | 1-27-1 | 2 2.46+0 | N NA | t | 10.6 | ; | 1 | 4.75-01 | Ŧ |
| 19 N/A N/A 2.66-10 3.16+01 N/A | ra m/n 2.66-10 3.16+01 m/n | M/A 2.6E-14 3.1E+01 N/A | M/A 2.66-10 3.16+01 N/A | 2.66-14 3.16+01 N/A | 2.6E-14 3.1E+01 N/A | 3.1E+01 N/A | W/W | : | | N/N | ; | V/I | : | 2.46-14 | 4 J. IE+0 | N 8/A | : | 160.3 | : | ; | 1.45-01 | ¥ ~ |
| 19 N/A N/A 2,46-14 3,16+01 N/A | ra N/A 2.66-14 3.16+01 N/A | N/A 2.46-14 3.16+01 N/A | N/A 2.46-14 3.16+01 N/A | 2.66-14 3.16+01 N/A | 2.6E-14 3.1E+01 N/A | 3.1E+01 N/A | V/R | ; | | N/N | ł | N/N | I | 2.46-1 | 4 J. IE+0 | N N/N | 1 | 150.0 | 1 | 1 | 4.76-0(| ¥ - |
| 21 N/A N/A 3.16-19 3.16401 N/A | /A N/A 3,16-10 3,16+01 N/A | N/A 3.16-19 3.16+01 N/A | N/A 3.16-19 3.16+01 N/A | 3.16-10 3.16+01 N/A | 3.16-18 3.16+01 N/A | 3. IE+01 N/A | | : | | N/N | ; | N/N | : | 3. IE-11 | 1 3.1E+0 | N N/N | 1 | 160.5 | 1 | : | 1.45-01 | ¥ |
| 21 N/A N/A 3.16-18 3.16-01 N/A - | /A N/A 3.16-18 3.16+01 N/A - | N/A 3.16-18 3.16+01 N/A - | N/A 3.16-18 3.16+01 N/A - | 3.16-18 3.16+01 N/A - | 3.16-18 3.16+01 N/A - | 3.1E+01 N/A - | - V/N | ' | | A/N | : | R/N | ł | 3. IE-11 | 1 3.IE+0 | N N/N | ł | 150.0 | : | 1 | 4.76-06 | ¥ - |
| 22 N/A N/A N/A N/A | /A B/A B/A B/A | 8/A 8/A 8/A | R/A B/A R/A | #/A #/A | N/A N/A | N/N | #/# | | : | N/N | ; | W/W | 1 | 1-#-1 | 1 2.66+0 | N N/N | : | 93.6 | 5.840 | 1.25+01 | : | ¥. |
| 22 N/A N/A N/A N/A | /A N/A N/A N/A | N/A N/A N/A | N/A N/A N/A | N/A N/A | N/A N/A | TVN | N.N | | : | N/N | : | N/N | : | 1-#-1 | 1 2.66+0 | N N/A | : | 48.0 | 3.06+0 | 1 6.05+00 | : | ¥. |
| 22 N/A N/A N/A N/A | /A N/A N/A N/A | N/A N/A N/A | N/A N/A N/A | N/A N/A | N/A N/A | N/A | N/N | | ; | A/N | : | N/N | ; | I.IE-1 | 1 2.46+0 | N N/N | ; | 87.0 | 7.36+9 | 1.5£+01 | 1 | £ |
| 22 N/A N/A N/A N/A | (A N/A N/A N/A | N/A N/A N/A | N/A N/A N/A | N/A N/A | N/A N/A | N/A | N/N | | ; | 8/8 | : | A/A | ; | 2. 1-1. | 1 2.66+0 | A/N E | ; | 160.3 | 1.46+0 | 2 2.IE+01 | : | £ |
| 22 N/A N/A N/A N/A | /A N/A N/A N/A | N/N N/A N/A | | N/A N/A | N/N N/N | N/A | N/N | | ; | 8/8 | ; | A/H | ; | 2.76-1 | 1 2.46+0 | N 8/A | : | 150.0 | 1.34.0 | 2 2.06+01 | : | ÷. |
| 23 N/A N/A 2.26-11 2.66-01 N/A | /A N/A 2.2E-11 2.4E+01 N/A | N/A 2.26-11 2.46+01 N/A | N/A 2.26-11 2.46+01 N/A | 2.2E-11 2.4E+01 N/A | 2.2E-11 2.4E+01 N/A | 2.4E+01 N/A | R/A | | ; | N/N | : | N/N | ; | 2.21-1 | 1 2.66+0 | N N/N | ł | 1404.0 | 5.86+0 | 1.25+01 | : | ۲. |
| 23 H/A N/A 2.2E-11 2.4E+01 N/A | ra N/A 2.2E-11 2.4E+01 N/A | N/A 2.2E-11 2.4E+01 N/A | N/A 2.2E-11 2.4E+01 N/A | 2.2E-11 2.4E+01 N/A | 2.2E-11 2.4E+01 N/A | 2.4E+01 N/A | N/A | | : | N/N | : | N/A | : | 2.26-1 | 1 2.66+0 | N N/A | 1 | 720.0 | 3.0E+0 | 1 6.05+00 | 1 | ۹. ۳ |
| 23 W/A W/A L.IE-11 2.46401 W/A | (A N/A 1,1E-11 2.4E+01 N/A | N/A 1,1E-11 2.4E+01 N/A | N/A 1,1E-11 2.4E+01 N/A | 1,1E-11 2.4E+01 N/A | 1.1E-11 2.4E+01 N/A | 2.46+01 N/A | N/N | | : | N/N | : | N/A | ł | 1.16-1 | 1 2.46+0 | N N/N | ; | 870.0 | 7.3£+0 | 1.55+01 | 1 | Ĩ |
| 23 N/A N/A 1.1E-11 2.6E+01 N/A - | ra N/A 1.1E-11 2.66401 N/A - | N/A 1.16-11 2.66+01 N/A - | N/A 1.16-11 2.66+01 N/A - | 1.1E-11 2.6E+01 N/A - | 1.1E-11 2.6E+01 N/A - | 2.6E+01 N/A - | - V.N | ' | , | A/N | ; | A/A | ł | 1.1E-1 | 1 2.66+0 | N/N I | : | 642.0 | 1.46+0 | 2.1E+01 | ! | ¥ |
| 23 N/A N/A 1,16-11 2,46+01 N/A | /A N/A 1,16-11 2,46+03 N/A | N/A 1,1E-11 2.4E+01 N/A | N/A 1.16-11 2.46+01 N/A | 1,16-11 2,46+01 N/A | 1.1E-11 2.4E+01 N/A | 2.4E+01 N/A | N/N | : | | A/N | : | R/A | ; | 1.1E-1 | 1 2.66+0 | N N/N | ; | 600.0 | 1.36+0 | 2 2.06+01 | : | 또 - |
| 24 N/A N/A N/A N/A | N/A N/A N/A N/A | N/A N/A N/A | H/A H/A H/A | K/A K/A | N/A N/A | N/N | H/A | ; | | N/N | : | N/N | ; | 1-30-1 | 1 2.66+0 | N 11/A | ; | 93.6 | 5.86+0 | 1.25+01 | ; | Ŧ |
| 24 N/A N/A N/A N/A | N/N N/N N/N N/N | N/N N/N N/N | V/N V/N V/N | N/N N/N | N/A N/A | W/W | N/A | : | | N/N | : | A/N | : | 1.06-1 | 1 2.66+0 | 11 N/A | ; | 48.0 | 3,00+0 | 1 6.0€+00 | + | ¥. |
| 24 N/A N/A N/A N/A - | (A K/A K/A K/A - | H/A H/A H/A - | - N/A N/A N/A - | | N/A N/A | · • • • • • • • • • • • • • • • • • • • | - W/W | ć | • | A/N | : | N/A | : | 7-31-1 | 2 2.4E+0 | N N/A | 1 | 87.0 | 7.3£+0 | 1 1.5£+01 | : | ¥ - |
| 24 N/A N/A N/A N/A | - N/A N/A N/A N/A - | N/A N/A N/A - | N/A N/A N/A - | W/W W/W | - N/N N/N - | - N/N | | , | | A/A | : | A/A | ; | 1-36-1 | 1 2.66+0 | ANN IN | ; | 140.5 | 1.45+0 | 2 2.1E+01 | ; | ¥ - |
| 24 N/A N/A N/A N/A | 1A N/A N/A N/A - | N/A N/A N/A - | - N/A N/A N/A - | N/A N/A | N/A N/A | · • • • • • • • • • • • • • • • • • • • | | | ; | A/A | ; | N/N | : | 1-36-1 | 1 2.6E+0 | N/N IO | ; | 150.0 | 1. 35+0 | 2 2.0€+01 | : | ¥. |

<u>SONT EEEEEEE DAARAAT KAGOOT KAGOOT KAGOONTAE EEKKESITTAA MATERATAA DAARAAT DAARAAT DAARAAT TAARAATIDA </u>

1-77

N. C. C. C. C. C.

INTELS-Aug-ST PAGE

NAMELING ACCIMENTS - COLLOCATION PROCESSING OFFICM (AIR FRANKPORT) (PER CONTAINER OR PALLET)

| | | | | | | | Accidi | mt Fre | New Cres an | d Runge F | actors | | | | | | | | | New Inda | Ale and a | el estad |
|-------|----|---------------------|-----------------|------------|-----------------|-------------------|-----------------|-------------|-----------------|--------------|-----------------|--------------|-----------------|--------------|-----------------|---------------|-----------------|--------------------|--------------|--------------------|----------------|------------------|
| N KC | 2 | MIN Free | RANGE FACTOR | See Bai | RANGE FACTOR | LINN Free Linn | RANGE FACTOR | ANN Birg | RANGE Factor | PINA Free | RANGE Factor | PUBA FREQ | RANGE Factor | TEAD FRED | RANGE FACTOR | UNIA Linea | RANGE FACTOR | MGENT MARILABLE | ennes Sen | L DS DE TOMATED | LIS ENTITED | DURATION TINE |
| | | | | | | | | | | | | | | | | | | | | | | |
| ж. | 2 | N/N | : | 8/8 | : | 1.5%-11 | 2.46+01 | W/W | 1 | 8/8 | ; | N/N | : | 11-36-11 | 2.46+01 | N/N | ; | 1404.0 | 5.86+01 | 1.25.41 | ł | ¥ - |
| JA | 2 | N/A | : | N/A | : | 1.56-11 | 2.46+01 | N/N | ł | W/W | : | N/N | ; | 1.58-11 | 2.6€+01 | N/N | 1 | 720.0 | 3, 66+01 | 6.05+00 | 1 | ¥- |
| | 2 | N/N | : | N/A | : | 7.76-12 | 2.46+01 | N/N | ; | A/A | ł | A/A | : | 7.7E-12 | 2.46+01 | N/N | : | 0.0 | 7.3€+01 | 1.55.01 | : | 1 |
| HARGE | 2 | N/N | : | N/N | 1 | 7.76-12 | 2.46+01 | V. | 1 | A/A | ; | N/N | ; | 1.76-12 | 2.4€+01 | A/A | 1 | 642.0 | 1.46+02 | 2.16+01 | : | 9 |
| HARVE | 2 | A / B | ; | A/A | .¦ | 1.75-12 | 2.46+01 | N/N | ; | A/A | ; | N/N | : | 7.76-12 | 2.46+01 | N /N | : | 6.003 | 1. JE +02 | 2.0€+01 | : | ¥ |
| AN AN | 36 | N/A | : | N/N | ; | N/N | : | N/N | ; | | ; | #/# | ł | 0.0£+00 | : | N/A | : | 1700.0 | ; | : | ; | 1 |
| HAPHC | 26 | W/W | : | A/A | : | N/N | : | A/N | : | N/N | ; | A/M | : | 0.0£+00 | ; | N/N | : | 93.4 | ł | ; | ! | ; |
| JAANH | * | N/A | : | N/A | : | N/A | ; | N/N | ; | A/A | ; | N/N | ; | 0.06+00 | ł | W/W | ; | 10 .0 | ; | : | ; | ; |
| TAGEC | 26 | N/A | ; | M/A | ; | A/A | ; | N/A | : | N/N | ; | N/A | ; | 0.0£+00 | ; | N/N | ; | 07.0 | : | ; | ; | : |
| HAREC | 26 | N/N | ; | N/N | ; | R/A | ł | N/N | ; | N/A | i | N/N | : | 0.0E+00 | 1 | A/A | 1 | 140.5 | ł | : | ; | ; |
| HARVC | * | N/N | : | N/A | ÷ | N/N | ; | N/N | : | N/N | ; | N/N | : | 0.0€+00 | ; | A/N | ; | 150.0 | : | : | : | ł |
| H MH | 27 | N/N | : | 0.0€+00 | ; | N/A | : | N/N | 1 | N/N | 1 | N/N | : | 0.0E+00 | ; | N/N | ; | 3400.0 | ; | : | : | : |
| JHJH | 27 | N/A | : | N/N | 1 | 0.0£+00 | ; | N/N | : | N/N | : | A/N | : | 0.0E+00 | ; | N/N | ; | 1404.0 | ; | : | ; | ł |
| HAPVC | 27 | N/N | : | N/A | ; | 0.0£+00 | ; | N/N | : | 8/8 | ; | N/N | ł | 0.0£+00 | ł | N/N | 1 | 720.0 | : | 1 | : | ; |
| HADGC | 2) | N/N | : | A/A | ; | 0.05+00 | : | N/N | ; | N/N | ; | N/N | ; | 0.0E+00 | ; | N/A | : | 870.0 | : | ł | : | ; |
| HARGC | 2) | N/A | : | A/A | ; | 0,06+90 | : | N/A | ; | N/N | : | N/N | ; | 0.0€+00 | ł | N/N | : | 642.0 | ; | ; | 1 | ł |
| HARVC | 2) | N/A | : | N/A | : | 0.0€+00 | ; | N/N | ; | N/A | ; | N/A | : | 0.0E+00 | : | R/A | ; | 600.0 | ; | ; | : | ; |
| HARGC | 62 | 8/8 | ; | N/A | : | 9.05-09 | 2.46+01 | N/N | ; | N/N | : | N/A | ; | 9.06-09 | 2.6€+01 | N/N | ; | 160.5 | : | 2. IE+01 | 1.0€+00 | ¶ |
| MARVE | ۶. | R/A | ł | N/A | ; | 9.06-09 | 2.6€+01 | N/N | : | N/N | : | N/A | : | 9.05-09 | 2.66+01 | N/N | ; | 150.0 | ! | 2.0€+01 | 5.06-06 | 9 - |
| HARGC | 8 | N/A | ; | A/A | ; | 1.86-98 | 2.46+01 | N/N | : | A/N | ; | N/A | ; | 1.85-09 | 2.6€+01 | A/A | ; | 10.7 | ; | 1. IE +01 | ; | 1991 |
| HARVC | 8 | N/N | ł | N/N | ; | 1.86-09 | 2.6€+01 | N/A | ; | A/N | ; | N/A | : | 1.86-09 | 2.6E+01 | N/N | ; | 10.0 | ; | 1.0£+01 | ł | 1991 |
| HARGC | 5 | N/A | ; | 4/4 | ł | 11-35-9 | 2.46+01 | N/N | : | N/A | ÷ | N/A | : | 6.56-11 | 2.66+01 | N/N | ; | 140.5 | ; | 2. JÉ +01 | 1.0£+00 | Ŧ |
| HARVC | 5 | A/A | : | K/A | : | 4.56-11 | 2.46+01 | N/N | : | N/N | : | N/A | : | 6.55-11 | 2.6€+01 | N/A | ; | 150.0 | : | 2.0€+01 | 5.06-04 | ¥ |
| HARGS | 32 | W/W | ; | N/A | : | 1.05-03 | 1.05+01 | A/A | ; | N/A | ; | 8/8 | ; | 1.06-03 | 1.0€+01 | A/A | ; | 642.0 | ; | 1 | ц Ш | ; |
| HARVS | 22 | N/A | : | #/# | : | 1.05-03 | 1.05+01 | N/N | : | N/N | ; | 4/H | : | 1.06-03 | 1.06+01 | N/N | : | 400.0 | : | { | 3 | |



RAM TOMAR SPANN DIMINAL BRAREN POINTAN REPORT OF THE TAMAR

لمندغر والماما والم

2



I.1.4. PLANT OPERATIONS

The following tables list the results for internal and external accidents during plant operations.











00

File: FulCCLiwri, 20-Aug-87 PASE1

PLART UPERATIONS COLLOCATION Median Accident Frequency (Fer Year)

S. X.Y.

PLANT OPERATIONS COLLOCATION Median Accident Frequency (Fer Year)

| SENSES. | ND. | ANAD RDC Fred | RANDE Faltor | FREQ | FACTOR | TDTAL Available | LBS. SPILLED | LBS. Detonated | LBS. Entred | DURATION TIME | TEAD NDC Freq | RANGE Factor | TOTAL Availab | 165. SP1LLED | LBS. Detonated | LES. DI Enitted | JRAT ION T I ME |
|---------------|------------|-----------------------|-----------------|---------------------|--------------|--------------------|-----------------|-------------------|----------------------------|--|---|---|------------------|--|---|--|---|
| | 0 ado | aéneratèd | alissile | puncture/cr | ush munit | tons in the | HI. | 1 | | 0 5 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 | 6 6 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 | > 5 6 6 6 6 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 | | | 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 | 4 1 4 4 4 1 1 1 1 1 1 1 | |
| Fi Bû, | | 3/6 | ; | 1.56-16 | 5 | 7040.0 | ; | 1 | 5.30E-01 | 2 HK | 1.5E-16 | 94 | 7040.0 | ; | ; | 5.30E-01 | 2 HK |
| | - | d.v€-13 | ŧ. | 1.36-15 | 94 | 46.08.0 | ; | ; | 2.40E-02 | 2 HK | 1.36-15 | 94 | 4608.0 | ł | : | 2.40E-02 | 분 건 |
| 1997 - | • | 2.16-13 | \$5 | 4.36-16 | † 6 | 614.4 | : | ; | 5.30E-01 | 2 HR | 4.36-16 | 94 | 614.4 | 1 | 1 | 5.30E-01 | 2 HR |
| # | | 1-15-13 | 15 | 4.36-15 | 1 6 | 1228.8 | : | : | 2.40E-02 | 3 HK | 4.36-16 | 16 | 1228.8 | ; | 1 | 2.40E-02 | 2 張 |
| | | 11/2 | ł | 2. 0E-16 | ¥6 | 2400.0 | ; | : | 1.605+00 | 2 HR | 2.0E-16 | 94 | 24000.0 | ; | 1 | 1.60E+00 | 2 HK |
| 'ī | | 1.: [-13 | 5 | 2.0E-16 | 10 | 27200.0 | : | : | 3.00E-01 | 2 HR | 2.0E-16 | 1 6 | 27200.0 | ; | ; | 3.00E-01 | 2 HK |
| •• | - | 1.16-13 | 46 | 3.0E-16 | Ţ | 25600.0 | ; | : | 3.60E-04 | 2 HR | 2.CE-16 | 94 | 25600.0 | ; | 1 | 3. DùE-04 | 2 HK |
| | | 5.15 13 | 7 5 | 1.36-15 | F 5 | 6048.0 | : | 1 | 2.00E-04 | 2 HR | 1.56-15 | 44 | 6048.0 | ; | ; | 2. 00E -04 | ~ 至~ |
| • | | | 7 5 | 5.76-16 | 45 | 9.32 E | ! | ; | 5.3uE-01 | 2 HK | 5.7E-16 | 64 | 832.0 | 1 | ; | 5.306-01 | 2 1 K |
| .' | • • | | 7.5 | 5. E-16 | | 1477.0 | : | : | 2.4uE-02 | 2 HK | 5.7E-16 | 64 | 1497.6 | : | ļ | 2.406-02 | 2 HK |
| بە: بە: | | :: 'H | .7 | 5 76 10 | . | 756.0 | : | : | 2.00E-04 | 2 HR | 5.76-16 | 94 | 768.0 | : | ; | 2. 00E-04 | 2 HK |
| | •-• | 11 22 23 | 3 | 5.86 is | ; | 1372.0 | ţ | ; | 5.30E-01 | 2 HR | 6.8E-16 | 94 | 1392.0 | : | ł | 5.30E-01 | 3.H.S |
| • | | Ĩ | • | tet le | ** | 1.52.6 | ; | ; | 2.00E-04 | 2 HR | 6.8E-16 | 56 | 1392.0 | : | 1 | 2.00E-04 | ∂H6 |
| · · · · · | | 1. H. L | 7 | 1.46 15 | ホ | 1563. | 1 | : | 10-302.3 | 2 HR | 1.46-15 | 8 6 | 2568.0 | : | ł | 5.306-01 | 2 HK |
| | | t. 11 13 | 4 17 | 1.35.1 | • • | 3400°0 | • | ; | 2. Jule - 04 | 2 HR | 1.46-15 | 46 | 2400.0 | ł | ; | 2. vùE - v4 | HH. |
| с. С | | a R | t. | 3. /E - 15 | 46 | 21496.0 | 1 | : | 3, 00E-04 | 2 HR | 3.7E-16 | 94 | 21696.0 | ; | 1 7 | 3. JOE - 04 | がない |
| | -515- | -génerated | al 1551 e | detinate mu | 1.111005 1 | s the AHL. | | | | | | | | | | | |
| 1 | ••• | 1.76-1 | 44 | . 15.16 | 65 | to.8.0 | ; | 6. ňŋ£+0ů | 8.00E-03 | 2 북 | 2.7E-16 | 66 | 9.608.0 | ł | 6.00E+00 | 8. 00E-03 | 3 HF |
| 1 | r 4 | 4.55 14 | ÷. | 5.16 17 | 3 | 514.4 | i | 1. 60E+00 | 1.60£-01 | 2 HG | 9. IE-17 | 65 | 614.4 | ; | 1.50E+00 | 1.605-01 | 2 HŘ |
| | | 1 37. F | 5. 1. | 9.16-17 | 65 | 1228.8 | ł | 3.20E+00 | 4.00E-03 | 2 HR | 9.16-17 | 65 | 1228.8 | ; | 5. 20E+00 | 4.00E-03 | 3 FE |
| | | | 55 | cl-3 | 66 | 60 48. 0 | : | 3. 15E+01 | 2.59E~01 | 2 祇 | 3.2E-16 | 56 | 6.148.0 | ; | 3.156+01 | 2.59E-01 | いた |
| 1991 | · . | t. tí·14 | 60 | 1.16-15 | 53 | 832.U | ł | 6.508+00 | 5.001-03 | 2 HR | 1.2E-16 | 66 | 832.0 | ; | 6.50E+U0 | 5.0116-03 | 2 HK |
| Qu 201 | C 4 | ÷1-3-19 | 15 | 1. ĬĔ~15 | 65 | :497.6 | ; | 1.17E+01 | 00E-05 | 2 1 5 | 1.26-16 | 66 | 1497.6 | ; | 1.176+01 | 4.005-05 | 2 HK |
| 14.40 A | ~ | 6. E 14 | 6, | 1.76-16 | 65 | 768.0 | 1 | 6.00E+00 | 4.356-01 | S HR | 1.26-16 | 65 | 768.Ŭ | ; | e. 00E+00 | 4. 38E-ut | H C |
| | • • | 6. nE - 14 | 6 | 1.56-16 | 5 | 1392.0 | ; | 1.45E+0L | 5. 0)E-05 | 2 HR | 1.5E-16 | 66 | 1392.0 | ł | 1.456+01 | 5. UUE-05 | ÷۳ |
| | ~1 | H/H | ; | 1.55-16 | 55 | 1392.0 | ; | 1.456+01 | 1.67E-01 | N HF | 1.56-16 | 65 | 1392.0 | ; | 1.45£+01 | 7.675-01 | 2 바 |
| 19404 | r 1 | 1.96-13 | έt | 3. <i>i</i>)E - 1a | 55 | 2568.0 | ! | 2.14E+01 | 6.00E-05 | 2 뷰 | 3.0E-16 | 66 | 25e8.ù | ł | 2.14E+01 | 6.00E-05 | 5 ¥ |
| 242 | - 1 | : 1 - 35 - 1 | 51 | 3. vE-16 | 55 | 2400.Û | : | 2.006401 | N/6 | 2 HR | 3.06-16 | 66 | 2400.0 | ; | 2.00E+U1 | N/A | 2 HK |
| F1: - 130 | -9165 | -jenerat=1 | al 155 le | pusctureitr | rush sunt | 1065 16 the | UPA. | | | | | | | | | | |

C

File: FLICOL.WA1, 20-Aug-87 PAGE2

| | - |
|------------------------------|---------------------------------|
| | YEAR |
| PLANT DFERATIONS COLLOCATION | IEDIAN ACCIDENT FREQUENCY I FER |
| | |

PLANT OPERATIONS COLLOCATION Median Accident Frequency (Per Year)

| | ; | | | | | | | | | | | | | | | | | | | | | | | | | | | | | | |
|----------|-----------|-----------|----------|------------|-------------|----------|----------|-------------|-----------|----------------|-------------|------------|------------|----------|----------|------------|---------|--------------|------------|----------|----------|-----------|-------------|------------|------------|------------|----------|----------|----------|----------------|----------|
| DURATION | | 1 HR | 1 HR | 1 HK | I HR | 1 1 | - 天 | 一 表 | JH 1 | 1 3 | 1 HK | 1 HÊ | 1 HK | - 美 - | 1 HK | 1 HR | 1 HK | | 2 HK | 2 HR | 2 HK | 2 HK | 2 HK | 2 HR | 2 HK | 2 氏 | 2 HK | 2 HF | 5 HB | | 1 HR |
| 165. | ENTITED | 5. 30E-01 | 2.406-02 | 5.30E-01 | 2.40E-02 | 1.44E+00 | 7.00E-02 | NEGL | 2.00E-04 | 5.306-01 | 2.40E-02 | 2.0ijE-04 | 5.30E-01 | 2.00E-04 | 5.30E-01 | 2.00E-04 | NEGL | | NEGL | 1.32E-01 | KEBL | NEGL | 2.316-01 | NEBL | KEGL | 10-301.1 | NEGL | 7.196-01 | NEGL | | : |
| LBS. | DETGNATED | ł | ; | : | ł | ; | ł | 1 | ; | 1 | 1 | ! | : | ł | 1 | ł | ; | | 6.00E+00 | 1.60E+00 | 3.20E+00 | 3.15E+01 | 6.50E+00 | 1.17E+01 | 6.00E+00 | 1.45E+01 | 1.45€+01 | 2.146+01 | 2.00E+01 | | ł |
| LBS. | PILLED | ! | ł | : | ; | ; | ł | ; | ł | ; | ł | : | ł | ; | ; | ; | ł | | ł | ł | ł | ł | : | ł | ! | 1 | ł | 1 | ł | | 48E+02 |
| TOTAL | AVRILAB | 2640.0 | 1728.0 | 230.4 | 460.8 | 9000.0 | 10200.0 | 9600.0 | 2268.0 | 312.0 | 561.6 | 288.0 | 522.0 | 522.0 | 963.0 | 900.0 | 8136.0 | | 1728.0 | 230.4 | 460.8 | 2268.0 | 312.0 | 561.6 | 288.0 | 522.0 | 522.0 | 963.0 | 900.0 | | יי ו |
| RANGE | FACTOR | 44 | 5 | 5 | F6 | 8 | 2 | 6 | 84 | 44 | 5 | 5 | \$ | 8 | £ | 1 5 | 54 | | 5 | 5 | 94 | 5 | \$ 6 | 1 6 | 8 6 | 8 6 | 5 | 94 | 94 | | 54 |
| TEAD NDC | FRED | 2. IE-15 | 7.96-15 | 2.7E-15 | 2.7E-15 | 3.6E-15 | 3.6E-15 | 3.66-15 | 9.25-15 | 4. IE-15 | 4.16-15 | 4, IE-15 | 4.1E-15 | 4.16-15 | 8.56-15 | 8.5E-15 | 1.56-14 | | B.4E-16 | 2.9E-16 | 2.9E-16 | 9.BE-16 | 4.46-16 | 4.4E-16 | 4.4E-16 | 4.4E-16 | 4.4E-ib | 4.46-16 | 4.4E-16 | facility). | 2.3E-11 |
| DURATION | 11ME | ¥ - | 9¥ - | HH - | 王 天 一 | 1 15 | 1 HS | 1 HR | 1 HR | 1 HR | 1 HR | HH 1 | 1 HR | œ | 1 HK | 1 HR | 1 HR | | 2 HR | 2 茶 | 2 HR | 2 HR | 2 HR | 2 HR | 2 HS | 2 HR | 2 HR | 2 HK | 2 HR | CAD (bulk-only | 1 HR |
| LBS. | ENTTED | 5. 306-01 | 2.40E-02 | 5.306-01 | 2.40E-02 | 1.44E+00 | 7.00E-02 | NEGL | 2.00E-04 | 5. JùE-01 | 2.40E-02 | 2.00E-04 | 5.30E-01 | 2.00E-04 | 5.306-01 | 2.00E-04 | NEGL | | NEGL | 1.32E-01 | NEGL | NEBL | 2.31E-01 | NEBL | NEGL | 4.10E-01 | NEGL | 7.19E-01 | MEGL | TOX at TE | 1 |
| L85. | DETORATED | 1 | ; | 1 | : | : | 1 | ; | ; | ł | I I | ł | : | ł | ; | ; | : | | 6.00E+00 | 1.60€+00 | 3.20€+00 | 3.15E+01 | 6.50E+00 | 1.17E+01 | 6.00E+00 | 1.456+01 | 1.45E+01 | 2.14E+01 | 2.00E+01 | he 60S and | : |
| L85. | SPILLED | ; | ; | ; | ł | ; | : | : | : | ; | 1 | ; | ; | ; | ; | 1 | ł | | ł | : | : | ł | ł | ł | ł | ł | 1 | ł | ; | between t | 5.4BE+02 |
| TOTAL | AVAILABLE | 2540.0 | 1728.0 | 239.4 | 460.8 | 9000.0 | 10200.0 | 9600.0 | 2268.0 | 312.0 | 561.6 | 289.0 | 522.0 | 522.0 | 963.0 | 900.0 | 8136.0 | the UFA. | 1728.0 | 230.4 | 460.8 | 2268.0 | 312.0 | 561.6 | 288.0 | 522.0 | 522.0 | 963.0 | 900.0 | ng systee | 1 |
| RANGE | FACTOR | 5 | 94 | 54 | 49 | 46 | 54 | \$ £ | 64 | 64 | \$ 5 | ¥5 | 1 6 | 46 | £6 | 46 | 46 | ittions in | 94 | 44 | 64 | 94 | 54 | 1 6 | 94 | 94 | 5 | 4 | 5 | agent pip | 54 |
| TEAD RDC | FREQ | 2. IE-15 | 7.96-15 | 2.7E-15 | 2.76-15 | 3.66-15 | 3.6E-15 | 3. éE-15 | 9.2E-15 | 4.15-15 | 4. IE-15 | 4. IE-15 | 4. IE-15 | 4.16-15 | 8.5E-15 | 8.5E-15 | 1.56-14 | detonite mun | 8.4E-16 | 2.96-16 | 2.96-16 | 9.86-16 | 4.46-16 | 4.4E-16 | 4.4E-16 | 4.45-16 | 4.4E-16 | 4.4E-16 | 4.46-16 | camages the | 2. JE-11 |
| RANGE | FACTOR | ; | 54 | 4 5 | † 5 | ! | 94 | 1 6 | 96 | 64 | 94 | 1 6 | t 5 | 1 | 46 | 15 | ł | aissile (| 1 6 | 15 | 46 | 84 | 46 | 5 | 16 | 94 | : | 5 | 46 | aissile (| : |
| ANAD RDC | FAEQ | N/A | 6.6E-12 | 2.3E-12 | 2. 3E-12 | ĥ/Å | 3.4E-12 | 3.sE-12 | 6. UE-13 | 3. SE - 12 | 3.5E-12 | 3.56-12 | 3.5E-12 | N/A | 7.16-12 | 7.1E-12 | N/A | generated | 7.0E-13 | 2.46-13 | 2.4E-13 | 1. 3E-12 | 3. /{-13 | 3.76-13 | 3. 78-13 | 3. 7E-13 | N/A | 7.56-13 | 00+30°0 | generated | Ψ/Ψ |
| NJ. | | •~ | r-1 | · ^ | 5 | ** | • • | 17 | ~ | •7 | 5 | r | m | | ~ | * 7 | • - | -oper | - | -+ | 4 | 4 | •• | •† | •* | - | 4 | - | - | -ope | v٦ |
| SLENARIO | 1.0. | F 066C | Բսնեն | FOCGC | FUCH | 13 60 | FURHC | 201105 | FORM | FLF6C | Fuffic | F0F :C | FCC6C | 50203 | F US6 | 36-52-5 | JABOJ | iG1 - TGri | FUDHC | FOCOL | FUCHL | FúnvC | FOFSC | PL-HC | JA20: | F006C | POORC | P.)\$6C | FUEVC | FOS - Tarc | r 0465 |



2







| | INE INE INE INE INE INE |
|-----------|--|
| | IOW PEK YEAR LBS. DUR HITTEO T |
| | S COLLOCAT B COLLOCAT E BS. LBS. |
| | DFERATIONS Cident Fre Luss. Pilled Dei |
| | PLANT MEDIAN AC Total Availab |
| | RANGE Factor |
| | TEAD NDC FREQ 0.06+00 |
| | DURATION TIME 1 HR |
| 1989, 198 | LBS. |
| | L6S. Det conated |
| | L6S, sP1LLED 1 6.38E+02 |
| |) TOTAL AVAILABLE |
| | FER YEAR RANGE Factor |
| | 553 563 Seculoca Frequency 1 Freq 0.0E-00 |
| | 9-87 PAI 9-87 PAI 1 Operation Accloent Range Facior |
| | IFI, 20-Au Flân Median Freu Fieu |
| 83A | 22 HC |

11 A Y

ilo tes:

1-83

Frequency unit = events/operating year
 Scenarios 5 applies only to the TEAD bulk-only facility

10 M 10 M 10 M

10.00

File: PLTCOL.NR1, 20-Aug-87 PA6E1

| | | PLAN Nedian | IT OFERATI. Accident 1 | ons colloca Frequency (| NTION Fer yea | R) | | | | | | | PLAN | T OPEKAT Accident | IONS COLLOCA | ATION I PER YES | 1K) |
|--------------------|----------|-------------------|---------------------------|----------------------------|------------------|--------------------|-----------------|-------------------|-----------------|---------------------|------------------|-----------------|------------------|----------------------|-------------------|--------------------|-----------------|
| SCENARIO 1.5. | NO. | ANAD RDC Freq | RANGE FACTOR | TEAD KOC Freq | RANGE FACTOR | TOTAL Available | LBS. Spilled | LBS. Detonated | LBS. Enitted | BURAT ION T I NE | TEAD NDC Freq | RANGE Factor | TOTAL Availab | LBS. SPILLED | LBS. Detonated | LBS. I Emitted | URATION TIME |
| | | | | | 5 6 1 1 | | | | | | | | | | | | |
| FU6 - Net COLOC | -Tupa: | të strikes MAA | the AHL. | 1 15-15 | 76 | 7040 | 1 | ; | 7 046407 | 9 | 1 46-15 | 24 | 7040 0 | ł | ; | 7 A45407 | 87 1 |
| POCHI POCHI | • • | 4 DE-14 | 40 | 0 95-14 | 3 2 | 0.070/ | : | 1 156+03 | 1 776+02 | 20 MIN | 0 BF-14 | 3 2 | 1408.0 | 1 | 1.156+03 | 1.736+07 | 20 MUM |
| FOLGE | - | 6.0E-16 | 2 | 6.0F-16 | 59 59 | 614.4 | 1 | 1.546+02 | 4.616+01 | 20 MIN | 6,0E-16 | 25 | 614.4 | ł | 1.546+02 | 1.61E+01 | ZO NIN |
| FOCHC | - 0 | 6.0E-16 | 2 | 6.0E-16 | 3 | 1228.8 | ţ | 3.07E+02 | 4.61E+01 | 20 MIN | 6.0E-16 | 58 | 1228.8 | 1 | 3.07E+02 4 | 1.61E+01 | 20 NIN |
| FOX6F | -0 | N/A | : | 2.06-15 | 36 | 24000.0 | ; | ; | 2.40E+03 | H HE | 2.06-15 | 26 | 24000.0 | ; | : | 2.40E+03 | 1 HK |
| FORAF | -0 | 2.0E-15 | 26 | 2.06-15 | 26 | 27200.0 | 1 | : | 1.36E+03 | 1 HK | 2.0E-15 | 26 | 27200.0 | ; | ! | L.36E+03 | 1 HR |
| ji t0j | -0 | 2. JE-15 | 3 9 | 2.0E-15 | 26 | 25600.0 | ł | 1 | 6.40E+02 | 1 HR | 2.06-15 | 26 | 25600.0 | ł | - | 5.40E+02 | 1 1 1 |
| FUNC | 4 | 1.5E-15 | 25 | 1.56-15 | 26 | 6048.0 | 1 | 1.51E+03 | 1.13E+02 | 20 MIN | 1.56-15 | 26 | 6048.0 | ; | 1.51E+03 | 1.13E+02 | 20 NIN |
| 29404 | 40 | 4.6E-16 | 26 | 4.6E-16 | 26 | 832.0 | ł | 2.08E+02 | 6.24E+01 | 20 MIN | 4.6E-16 | 26 | 832.0 | ł | 2.08E+02 (| 5.24E+01 | 20 NIN |
| FOFHC | 9 | 4.66-16 | 26 | 4.6E-16 | 92 | 1497.6 | 1 | 3.74E+02 | 5.62E+01 | ZO MIN | 4.6E-16 | 26 | 1497.6 | : | 3.74E+02 | 5.62E+01 | 20 MIN |
| JV1GP | • | 4.6E-16 | 2b | 4.6E-16 | 36 | 768.0 | 1 | 1.92E+02 | 1.44E+01 | 20 MIN | 4.66-16 | 26 | 768.0 | : | 1.92E+02 | 1.44E+01 | 20 NIN |
| FOMGC | -0 | 4.ó£-16 | 26 | 4.6E-16 | 26 | 1392.0 | | 3.486+02 | 1.04E+02 | 20 MIN | 4.66-16 | 26 | 1392.0 | 1 | 3.486+02 | 1.04E+02 | 20 MIN |
| FJUVC | -0 | A/A | ; | 4.6E-16 | 26 | 1392.0 | 1 | 3.486+02 | 2.61E+01 | 20 MIN | 4.6E-16 | 26 | 1392.0 | ł | 3.40E+02 | 2.61E+01 | 20 NIN |
| FCRBC | 9 | 2.16-15 | 2 b | 2. IE-15 | 26 | 256 8. Ú | ; | 6.42E+02 | 1.93E+02 | 20 MIN | 2. IE-15 | 26 | 2568.0 | 1 | 6.42E+02 | 1.936+02 | 20 NIN |
| PCFVC | -0 | 3.4E-15 | 2 6 | 2. IE-15 | 5 9 | 2460.0 | ł | 6.00E+02 | 4.50E+01 | 20 MIN | 2. IE-15 | 26 | 2400.0 | 1 | 6.00E+02 | 1.50E+01 | 20 MIN |
| FGSVF | Ŷ | N/A | ł | 3.4E-15 | 2 b | 21696.0 | ; | ł | 5.42E+02 | 또 또 - | 3.46-15 | 26 | 21696.0 | ; | ; | 5.42E+02 | ¥ ~ |
| F07 - flet | eorit | te strikes | the UFA. | | | | | | | | | | | | | | |
| FOBGF | 2 | A/A | ; | 2.9E-12 | 36 | 2640.0 | 1 | 1 | 2.64E+02 | H H | 2.9E-12 | 26 | 2640.0 | ł | 1 | 2.64E+02 | 3H 1 |
| FODHC | 2 | 2.0E-12 | 36 | 2.0E-12 | 2 6 | 1728.0 | ; | 4.32E+02 | 6.48E+01 | 20 MIN | 2.0E-12 | 26 | 1728.0 | 1 | 4.32E+02 | 5.4BE+01 | 20 NIN |
| F0C6C | ~ | I. IE-12 | 3 P | 1.16-12 | 26 | 230.4 | ; | 5.76E+01 | 1.73E+01 | 20 HIN | 1. IE-12 | 26 | 230.4 | 1 | 5.766+01 | 1.736+01 | 20 MIN |
| FUCHC | ~ | I. IE-12 | 26 | 1.16-12 | 26 | 460.8 | ł | 1.15E+02 | 1.736+01 | 20 NIN | 1.15-12 | 26 | 460.8 | ł | 1.15€+02 1 | 1.736+01 | ZO NIN |
| PCkGF | 1 | N/A | : | 4.0E-12 | 36 | 9000.0 | ; | ! | 9.00E+02 | 1 HK | 4.05-12 | 26 | 9000.0 | ł | ; | 0.00E+02 | UH H |
| 単語に | ~ | 4.06-12 | 26 | 4.0E-12 | 28 | 10200.0 | ; | ; | 5.10£+02 | E . | 4.0E-12 | 26 | 10200.0 | 1 | : | 5. 10E+02 | 1 1 |
| POK VE | 1 | 4.0E-12 | 36 | 4.6E-12 | 26 | 9600.0 | ł | ł | 2.40E+02 | 1 150 | 4.0E-12 | 26 | 9600.0 | 1 | : | 2.40E+02 | ž – |
| FONUC | ~ | 2.96-12 | 36 | 2.96-12 | 9 2 | 2268.0 | ł | 5.67E+02 | 4.25E+01 | 20 NIN | 2.96-12 | 26 | 2268.0 | 1 | 5.67E+02 | 1.25E+01 | ZO NIN |
| PCPSC | 1 | 8.85-13 | 26 | 8.86-13 | 26 | 312.0 | ł | 7.805+01 | 2.34E+01 | 20 NIN | 8.06-13 | 56 | 312.0 | ł | 7.80€+01 2 | 2.34E+01 | 20 NIN |
| FUPHC | r~ | B.8E-13 | 26 | B. BE - 13 | 26 | 561.6 | : | 1.40E+02 | 2.11E+01 | 20 MIN | 8.6E-13 | 26 | 561.6 | ; | 1.40E+02 2 | 2.11E+01 | 20 MIN |
| F0F.XC | ۰. | 6.86-13 | 26 | 8.86-13 | 38 | 288.0 | ł | 7.20E+01 | 5.40E+00 | 20 MIN | 8.8E-13 | 26 | 268.0 | ł | 3. 206+01 | . 40E+00 | ZO NIN |
| 10000 | ~ | 8.85-13 | 56 | 8.86-13 | 26 | 522.0 | ł | 1.31E+02 | 3.92E+01 | 20 MIN | B. 8E-13 | 26 | 522.0 | : | 1.3:6+02 3 | 1.926+01 | 20 NIN |



্র জাস

1.1.1.1.1.1.1.1

No. of the second second second second second second second second second second second second second second s

. 52222

22





PLANT DPERATIONS COLLOCATION

| I I? 5.48E+01 10 MIN 5.0E-11 1? 5.19E+01 10 MIN 1 1? 5.19E+01 10 MIN 5.0E-11 1? 5.19E+01 10 MIN 1 1? 1.27E+01 10 MIN 5.0E-11 1? 1.27E+01 10 MIN | NT FREQUEN RF TEAD F FREQ R FREQ R 4.06 26 4.06 26 3.46 26 3.46 26 3.46 26 3.46 26 3.46 26 3.46 26 3.46 26 10,000 | MCY (FER Y RDC RANKE Q FACTOR FACTOR E-13 26 E-13 26 E-12 26 E-13 26 E-13 26 E-13 26 F-13 26 F-13 26 R9 5ystee be ¹ | AK) TOTAL AVALLABLE SI AVALLABLE SI 522.0 963.0 9136.0 8136.0 15.1 15.1 15.1 | PfLLED | LBS. DETOMATED 1.31E+02 2.41E+02 2.25E+ | LBS. LBS. 9.795400 7.225401 1.695401 2.035402 2.035402 1.646400 9.535-01 3.785-01 4.0017 44 | URATION TTRE 20 MEM 20 MEM 20 MEM 1 HR 1 HR 1 HR 1 HK 1 HK | TEAD NDC FREQ 8.86-13 4.06-12 6.76-12 3.46-13 3.46-13 | RANGE FACTOR 26 26 26 26 26 26 26 | MEDIAN A TUTAL Availar 522.0 963.4 963.4 963.4 19.1 15.1 15.1 | CCLOENT | FREQUENCY DETOMATED DETOMATED 1.316+02 2.416+02 2.256+02 2.256+02 2.55+02 2.55+02 2.55+02 2.55+02 2.55 | (PER YEI LBS. 1 ENLITED 9.795+00 7.225+01 1.655+01 1.655+01 1.656+00 1.646+00 1.646+00 1.646+00 | IK) URANION 20 MIN 20 MIN 20 MIN 20 MIN 1 HR 1 HR 1 HR |
|---|--|--|---|--------|---|---|---|---|---|--|---------|---|--|---|
| 1 17 3.19E+01 10 MIN 3.0E-11 17 3.19E+01 10 MIN 1 17 1.27E+01 10 MIN 3.0E-11 17 1.27E+01 10 MIN | | -11 11- | ; | ł | 1 | 5.48E+01 | NIN OJ | 3.06-11 | 17 | 1 | : | ł | 5.486+01 | 110 01 |
| 1 17 1.27E+01 10 HIN 3.0E-11 17 1.27E+01 10 HIN | | -11 12 | 1 | 1 | ; | 3, 196+01 | IO MIN | 3.06-11 | 11 | 1 | ł | 1 | 3.19E+01 | IÙ MIN |
| | | 11 17 | ł | 1 | ł | 1.27E+01 | IO NIN | 3.0E-11 | 17 | 1 | 1 | 1 | 1.27E+01 | IÙ NIN |

I-85

Hútes:

Frequency unit = events/operating year
 Scenarios 8 applies only to the TEAD bulk-only facility

File: PLICOL. Hk1, 20-Aug-B7 PAGE1

CHOROLOGIC

| | DURATION |
|----------------------------|----------|
| | LBS. |
| | LBS. |
| | LBS. |
| - | TOTAL |
| ITTON Per year | RANGE |
| DNS COLLOCA Frequency i | TEAD RDC |
| T OPERATI | RANGE |
| PLAN | , RDC |
| ¥ | S. |

PLANT DPERATIONS COLLOCATION Median Accident Frequency (Per Year)

| ENARIO I 1.D. | NU, ANAD RDC Freq | RANGE FACTOR | TEAD RDC Freq | RANGE Fåctor | TOTAL AVAILABLE | LBS. SPILLED | LBS. Detonated | LBS. Emitted | DURAT I DN T I ME | TEAD NDC FREQ | RANGE FACTOR | TOTAL LB: AVAILAB SPILI | ED DETONATE | LBS. D ENITED | DURATION TIME |
|------------------|----------------------|-----------------|------------------|-----------------|--------------------|--------------------|-------------------|-----------------|----------------------|------------------|-----------------|----------------------------|-------------|------------------|-----------------------|
| | • | | | | | | | | | | | | | | |
| - Úrec | ct large airi | craft crasl | h onto the i | MHI; no f | lre | | | | | | | | | | |
| S940 | 9 N/A | ; | 1.2E-11 | 9 | 7040.0 | 2.11€+03 | 1 | 1 | £ 9 | 11-32-11 | 2 | 7040.0 2.11E | 10 | ! | 9 HG |
| ODHC | 9 2.65-10 | 01 | 1.2E-11 | 9 | 4608.0 | 1.15€+03 | 2.30E+02 | ; | 9 HE | 1.2E-11 | 2 | 4608.0 1.15E | 03 2.30E+(| | 6 RK |
| UCSC | 9 2.6E-10 | 10 | 1.2E-11 | 01 | 614.4 | 1.54E+02 | 3.076+01 | 1 | 6 148 | 11.2E-11 | 2 | 614.4 1.54E | -02 3.67E+(| | 9 MK |
| DCH. | 9 2.66-10 | <u>01</u> | 1.2E-11 | 9 | 1228.8 | 3.076+02 | 6.14E+01 | : | 8H 9 | 1.26-11 | 10 | 1228.8 3.07E | 02 6.14E+(| : | ¥王 9 |
| 046S | 9 N/A | ; | 1.25-11 | 9 | 24060.0 | 7.206+03 | ; | 1 | 6 HR | 1.26-11 | 10 | 24000.0 7.2064 | · · · | 1 | 8 H |
| On HS | 9 2.6E-10 | 10 | 1.26-11 | 0 | 27200.0 | B. 16E+03 | 1 | 1 | 6 NR | 1.2E-11 | 10 | 27200.0 B.16E | 10 | 1 | A HK |
| 01 VS | 9 2.6E-10 | 9 | 1.26-11 | 0 | 25600.0 | 7.686+03 | : | 1 | 6 HR | 1.26-11 | 9 | 25600.0 7.686 | | 1 | 9 1년 |
| DHVC | 9 2.6E-10 | 01 | 11-32.1 | 10 | 6048.0 | 2.42E+03 | 3.02E+02 | ! | 6 HR | 1.2E-11 | 9 | 6048.0 2.42E4 | 03 3.02E+(| | 6 HK |
| 01-61 | 9 2.6E-10 | 2 | 1.26-11 | 10 | 832.0 | 2.08E+02 | 4.16E+01 | 1 | 6 HG | 1.25-11 | 10 | B32.0 2.086 | +02 4.16E+(| - | 6 HR |
| JFHC | 9 2.6E-10 | <u>0</u> 1 | 1.2E-11 | 10 | 1497.6 | 3.746+02 | 7.49E+01 | ; | 6 HR | 11.25-11 | 91 | 1497.6 3.746 | 02 7.496+0 | : | 6 HF |
| OFVC | 9 2.6E-10 | 10 | 1.26-11 | 10 | 768.0 | 1.92E+02 | 3.84E+01 | : | 6 HR | 1.2E-11 | 01 | 768.0 1.925 | -02 3.84E+(| | ₩ 1 1 1 1 |
| 066C | 9 2.6E-10 | 10 | 1.2E-11 | 10 | 1392.0 | 3.48E+02 | 6.96E+01 | : | 6 HR | 1.25-11 | 9 | 1392.0 3.4864 | 02 6.96E+(| ! | 6 HK |
| JUPVC | 9 N/A | ; | 1.26-11 | 10 | 1392.0 | 3.48E+02 | 6.96E+01 | : | 9 HE | 1.26-11 | 9 | 1392.0 3.4864 | 02 6.96E+(| | 6 HK |
| DESC | 9 2.6E-10 | 10 | 11-32-11 | 0 | 2568.0 | 1.03E+03 | 1.286+02 | 1 | 6 HH | 1.25-11 | 10 | 2568.0 1.03E | 03 1.286+6 | | 9 HY |
| UKVC | 9 2.6E-10 | 2 | 1.2E-11 | 10 | 2400.0 | 9.60E+02 | 1.20E+02 | ł | 6 HR | 1.2E-11 | 10 | 2400.0 9.606 | W2 1.20E+(| | ~ 또 ~ |
| 3505 | 9 N/A | 1 | 1.26-11 | 91 | 21696.0 | 6.51E+03 | ł | 1 | 6 HR | 11-32-11 | 10 | 21696.0 6.5IE4 | | : | 9 HE |
| ó - Dire | ect large and | rcraft cra | sh anto the | NHI; fir | e not conta | ined in Ù . | 5 hours | | | | | | | | |
| ÜBGF 1 | 10 N/A | : | 9.86-12 | 10 | 7040.0 | : | 1 | 7.04E+02 | 9 1 - | 9.06-12 | 01 | 7040.0 | : | - 7.046+02 | ¥ - |
| ODHC 1 | 10 2.2E-16 | 2 | 9.86-12 | 01 | 4608.0 | 1 | 1.15E+03 | 1.73E+02 | 20 MIN | 9.86-12 | 9 | 4608.0 | 1.156+0 | 1.73€+02 | ZO NIN |
| 0C6C 1 | 10 2.2E-10 | 9 | 9.86-12 | ÷. | 614.4 | ; | 1.54E+02 | 4.61E+01 | 20 MIN | 9.86-12 | 2 | 614.4 | 1.54E+(| 2 4.61E+01 | 20 MIN |
| 0CHC 1 | 10 2.2E-10 | 10 | 9.05-12 | 9 | 1228.8 | : | 3.07E+02 | 4.61E+01 | 20 MIN | 9.86-12 | 2 | 1228.8 | 3.07E+C | 10+319.4 Z | 20 NIN |
| 0¢6f 1 | 10 N/A | ; | 9.86-12 | 9 9 | 24000.0 | ; | 1 | 2.40E+03 | 1 148 | 9.86-12 | 2 | 24000.0 | : | - 2.40E+03 | , HK |
| CAHF 1 | 16 2.2E-10 | 10 | 9.86-12 | 9 | 27200.0 | 1 | 1 | 1.36E+03 | 9¥ - | 9.06-12 | 9 | 27200.0 | : | - 1.366+03 | 1 1 1 |
| ÜK VF 1 | 10 2.26-10 | 10 | 9.86-12 | 01 | 25600.0 | 1 | ł | 6.40E+02 | ÷. | 9.86-12 | 10 | 25600.0 | • | - 6.40E+02 | ¥ - |
| DRVC 1 | 10 2.2E-10 | 91 | 9.86-12 | 10 | 6048.0 | ; | 1.51E+03 | 1.136+02 | 20 MIN | 9.86-12 | 9 | 604B.0 | 1.516+0 | 3 1.13E+02 | 20 MIN |
| 0F6C 1 | 10 2.26-10 | 9 | 9.8E-12 | 9 | 832.0 | ł | 2.08E+02 | 6.24E+01 | 20 NIN | 9.86-12 | 2 | B 32.0 | 2.08E+(| 12 6.24E+01 | 20 NIN |
| DFHC 1 | 10 2.2E-10 | 0 | 9.86-12 | 10 | 1497.6 | : | 3.74E+02 | 5.62E+01 | 20 MIN | 9.8E-12 | 10 | 1497.6 | 3.74E+0 | 2 5.62E+01 | ZO NIN |
| DFVC 3 | 10 2.2E-10 | 2 | 9.86-12 | 2 | 768.0 | : | 1.92E+02 | 1.44E+01 | 20 MIN | 9.86-12 | 2 | 768.0 | 1.926+(| 10+341.1 20 | 20 NIN |
| 100C 1 | 10 2.26-10 | 9 | 9.86-12 | 9 | 1392.0 | ! | 3.48€+02 | 1.04E+02 | 20 MIN | 9.86-12 | 9 | 1392.0 | 3.484 | 2 1.04E+02 | ZÙ MIN |

1-86





PLANT OFERATIONS COLLOCATION MEDIAN ACCIDENT FREQUENCY (FER YEAR)

PLANT OPERATIONS COLLOCATION MEDIAN ACCIDENT FREQUENCY (PER VEAR)

| RATION | TINE | 20 NIN | 20 NIN | 20 NIN | L HK | | 1 HR | 1 HK | - Fr | 1 HK | 1 HR | | 8H 8 | A HR | 6 HR | 6 HF | 6 HK | Å HÅ | 9 HK | 6 HK | 8 H.R | 6 HF | 6 I K | 9 HY | 6 HR | Å HR | 6 HK | 6 HR | | 1 HR | 20 NIN |
|----------|------------|-------------|-------------|-------------|---------|------------|----------|----------|---------------|------------|-------------|--------------|-----------|-----------|----------|----------|-----------|------------|-----------|-----------|-----------|----------|------------------|----------|-------------------|----------|----------|-----------|-------------|----------|-------------|
| LBS, DU | NITED | 51E+01 | P3E+02 | 10+30S | 12E+02 | | 11E+02 | 20E+02 | 38E+02 | 92E+02 | 63E+02 | | ; | ! | 1 | ł | : | ; | ; | ; | ; | ; | ; | ; | ; | 1 | ; | ; | | 56E+02 | 10+355 |
| 195. | TONATED E1 | .486+02 2.6 | .426+02 1.9 | .00E+02 4.5 | - 5. | | 2.1 | 7.1 | - | | 1. | | : | .59€+02 | .466+01 | .91E+01 | : | : | ł | .40E+02 | .686+01 | .42E+01 | .32E+01 | .836+01 | . B.XE +01 | .44€+02 | . 35£+02 | : | | 2.6 | .32E+02 6.5 |
| LBS. | FILLED DE | r | -9 | 9 | ; | | ł | 1 | ; | 1 | : | | 22E+03 | 22E+03 2 | 73E+02 3 | 36E+02 6 | 52E+03 | 52E+03 | 02E+03 | 60E+03 3 | 30E+02 4 | 06E+02 B | 126+02 4 | 776+02 7 | 766+02 7 | B6E+02 1 | 41E+02 1 | BOE+03 | | ; | - |
| TOTAL | AVAILAB S | 1392.0 | 2568.0 | 2400.0 | 21696.0 | | 7040.0 | 24000.0 | 27200.0 | 25600.0 | 21696.0 | | 2656.4 2. | 1747.1 1. | 246.8 1. | 479.9 3. | 9016.4 7. | 10219.1 8. | 9615.1 8. | 2283.1 1. | 328.4 2. | 580.7 4. | 303.1 2. | 538.4 3. | 537.1 3. | 979.3 6. | 915.1 6. | 8151.1 6. | | 2656.4 | 1747.1 |
| RANGE | FACTOR | 10 | 01 | 10 | 01 | | 13 | 13 | 13 | 13 | = | | 10 | 10 | 2 | 10 | 2 | 10 | 01 | 10 | 2 | 1ċ | 91 | 10 | 10 | 10 | 10 | 10 | | 01 | 1Ū |
| TEAD NDC | FRED | 9.8E-12 | 9.8E-12 | 9.86-12 | 9.86-12 | | 3.36-15 | 3.36-15 | 3.36-15 | 3.36-15 | 3.3E-15 | | 3.56-10 | 3.56-10 | 3.56-10 | 3.56-10 | 3.5E-10 | 3.5E-10 | 3.5E-10 | 3.56-10 | 3.5E-10 | 3.5E-10 | 3.56-10 | 3.SE-10 | 3.56-10 | 3.5E-10 | 3.5E-10 | 3.56-10 | | 2.96-10 | 2.96-10 |
| RATION | []NE | 20 MIN | 20 MIN | 20 NIN | - HE | | 1 H |) HK | E E | Ŧ. | 1 HR | | 6 HR | 6 HR | 6 HR | 6 HK | 6 HR | 9 E | 6 HR | 6 HR | 6 HR | 6 HR | 6 HR | 6 HR | 6 1 ,6 | 6 HK | 6 HR | 6 HR | | HK | 20 MIN |
| LBS. DU | MITTED | . 61E+01 | . 93E+02 | 10+305 | .42E+02 | | 2.11E+02 | .20E+02 | 08E+02 | . 92E+02 | . 63E+02 | | ł | : | 1 | 1 | ; | 1 | : | 1 | ł | ; | ł | 1 | ; | ł | ł | 1 | | . 66E+02 | |
| LBS. | ETONATED E | 3.486+02 2 | 6.426+02 | 6.00E+02 4 | : | rs | : | 1 | 1 | : | : | | ; | 2.59E+02 | 3.46E+01 | 6.91E+01 | ; | ; | 1 | 3.40E+02 | 4.68E+01 | B.42E+01 | 4.32E+01 | 7.83E+01 | 7.81E+01 | 1.44E+02 | 1.35E+02 | ł | 0.5 hours | 1 | 4.326+02 4 |
| LBS. | SPILLED D | ; | : | : | : | in 0.5 hou | ; | ł | ; | ; | : | | 2.22E+03 | 1.22E+03 | 1.73€+02 | 3.366+02 | 7.526+03 | 8.52E+03 | 8.02E+03 | 1.60E+v3 | 2. 30E+02 | 4.06E+02 | 2.12E+02 | 3.77E+02 | 3.76E+02 | 6.86E+02 | 6.41E+02 | 6.80E+U3 | itained in | ; | ; |
| TOTAL | AVAILABLE | 1392.0 | 2568.0 | 2400.0 | 21696.0 | contained | 7040.0 | 24000.0 | 27200.0 | 25600.0 | 21696.0 | o fire | 2656.4 | 1747.1 | 246.8 | 419.9 | 9016.4 | 10219.1 | 9415.1 | 2283.1 | 328.4 | 580.7 | 303.1 | 539.4 | 537.1 | 979.3 | 915.1 | 8151.1 | ire not con | 2656.4 | 1747.1 |
| RANGE | FACTOR | 10 | 91 | 10 | 10 | NH1; fire | 13 | 13 | 13 | 11 | 1 | the MDB; n | 10 | 10 | 2 | 2 | 91 | 9 | 2 | 1Ú | 2 | 10 | 9 | 3 | 10 | 10 | 2 | 2 | the MDB; f | = | 10 |
| TEAD RDC | FREG | 9.86-12 | 9.96-12 | 9.86-12 | 9.86-12 | h onto the | 3. 3E-15 | 3. 36-15 | 3. 36 - 15 | 3. 36 - 15 | 3. 36-15 | th damages 1 | 3.56-10 | 3.56-10 | 3.5E-10 | 3.56-10 | 3.5E-10 | 3.5E-1u | 3.5E-10 | 3.56-10 | 3.56-10 | 3.56-10 | 3.5E - 1u | 3.56-10 | 3.56-10 | 3.5E-10 | 3.56-10 | 3.56-10 | h damages (| 2. 3E-14 | 2.96-10 |
| RANGE | FACTOR | ; | 10 | 2 | ; | craft cras | : | ; | 13 | 11 | ; | craft cras | : | 10 | 10 | 10 | 1 | 61 | 10 | 91 | l. | 10 | 10 | 10 | : | ē | 10 | : | craft cras | 4 | 16 |
| ANAD REC | FKE0 | A/A | 2.26-10 | 2.26-10 | N/A | large airc | N/A | N/A | 7.36-14 | 7.36-14 | M /A | large airt | N/A | 7.7E-u3 | 7.7E-09 | 1.75-69 | N/A | 7.76-09 | 7.76-09 | 7.75-09 | 7.7E-09 | 7. 7E-U9 | 7.76-09 | 7.76 04 | N/A | 7.76-09 | 7.76-49 | R/N | large airc | N/A | 6. 3E - UY |
| ж. | | - | 2 | 2 | Ξ | rect | = | Ξ | = | Ξ | Ξ | rect | 12 | 1 | Ľ | 멅 | 2 | 12 | 12 | 12 | 71 | 12 | 12 | 12 | 17 | 2 | 71 | <u>с</u> | rect | - | |
| SCENNK10 | 1.0. | 1400-1 | Füküt | F01-1 | FÚSVE | FÚLL - DL | F066F | 1046F | まう | PUN VE | FOSVE | F012 - 01 | FuildS | FÜGHE | FUCG | 5 GC HL | FùrdS | FUNHS | F ()+ vS |)/W0-4 | F ÙF uC | FOFHC | f Út ví | 13261 | FUNC | F 0F 6C | FÜLVÍ | 10505 | f01: - Di | H UPIGF | 1HJ9.4 |

A SUSSED REPORT NAMES AND DEPARTURE DEPARTOR

File: FLTCOL.WK1, 20-Aug-87 PAGE3

A A A CAR

PLANT OPERATIONS COLLOCATION

8

| | | MEDIAN | ACCIDENT | FREQUENCY (| PEK YEA | њ) | | | | | | | NEDIAN AI | CCIDENT (| FREQUENCY (| PER VEA | 2 |
|-------------------|----------|------------------|-----------------|------------------|-----------------|--------------------------|--------|-------------------|-----------------|------------------|------------------|-----------------|------------------|-------------------|---------------------|----------------------|-----------------|
| . Емњ 10 1. D. | нб. | ANÀU HOC Frea | RANGE FACTOR | TEAD KDC Fred | RANGE FACTOR | TOTAL L AVAILABLE SPI | BS. | LES. Detonated | LBS. Emitted | DURATION TIME | TEAD NDC FREQ | RANGE FACTOR | TOTAL Availab | LBS. SPILLED 1 | LBS. Detonated 1 | LBS. DI Emitted | URATION TIME |
| եսլն։ - | 1 | 6.3£-09 | · · | 2.96-10 | e. | 246.B | | 5.766+01 | 1.855+01 | 20 MIN | 2.96-10 | 10 | 246.8 | ; | 5.76E+01 1 | 10+35 8 . | 20 MIN |
| F JÍ HL | 1 | و، 3٤ - ٩ | ÷ | 2. 4E - 11 | 19 | 479.9 | ! | 1.156+02 | 1.805+01 | 20 MIN | 2.96-10 | 10 | 479.9 | 1 | 1.15E+02 1 | 10+308. | 20 MIN |
| FÛKU: | 2 | 17N | : | 2.46-10 | ÷ | \$+: 9. 4 | ł | ; | 9.02E+02 | 1 HR | 2.95-10 | 01 1 | 9016.4 | ; | 6 | .02E+02 | 1 HK |
| ч сь н. | <u> </u> | 6.3E-J9 | 10 | 2.91-14 | 14 | 10219.1 | 1 | ; | 5.11E+02 | 1 HE | 2.96-10 | 10 | 10219.1 | ; | 3 - | .116+02 | 4¥ - |
| 5.424 | | 6. 3E - 09 | -1 | 2.96-19 | 10 | 9615.1 | ; | : | 2.40E+02 | HR - | 2.96-10 | 01 | 9615.1 | ł | - 2 | .40E+02 | 34 1 |
| FünvC | :1 | a. 3£ -03 | 2 | 2.96-10 | 16 | 2283.1 | ; | 5.67E+02 | 4.286+01 | 20 MIN | 2.96-10 | 10 | 2283.1 | : | 5.67E+ú2 4 | . 286+01 | 20 HIN |
| r JFG. | 1 | 6. JE -04 | 2 | 2.96-10 | 10 | 328.4 | : | 7.806+01 | 2.46E+01 | 20 MIN | 2.96-10 | 10 | 328.4 | ; | 7.80€+01 2 | .46E+01 | 20 NIN |
| H fn 1 | 1 | 6. 3f - Už | - | 2.9E -10 | | 580.7 | ; | 1.40E+02 | 2.18€+01 | 20 MIN | 2.96-10 | 01 | 580.7 | ; | 1.40€+02 2 | . 18E+01 | 20 MIN |
| , Iu | 2 | 6. 3f 199 | ÷ | 2.96-10 | 3 | 303.1 | ; | 1.205+01 | 5.686+00 | 20 MIN | 2.9E-10 | 10 | 303.1 | ; | 7.20E+01 5 | .686+00 | 20 MIN |
| ի ննալ | | e. 3t (9 | 16 | 2.96-10 | 2 | 538.4 | 1 | 1.316+02 | 4.04E+01 | ZO MIN | 2.96-10 | 10 | 539.4 | ţ | 1.31E+ù2 4 | .04E+01 | 20 MIN |
| - 00 - | 1 | Ч. N | | 2.96-10 | 10 | 537.1 | ; | 1.31E+02 | 1.01E+01 | 20 MIN | 2.96-10 | 10 | 537.1 | ; | 1.31E+02 1 | .016+61 | 20 MIN |
| i ukuč | 2 | 6.3E V9 | 10 | 2.96-10 | 10 | 979.3 | ; | 2.41E+02 | 7.356+01 | 20 MIN | 2.9E-10 | 01 | 979.3 | : | 2.41E+u2 7 | . 356 +01 | ZU MIN |
| 1.64 | :2 | 6. Ji 109 | - | 2. 36 - 10 | 10 | 915.1 | : | 2.25E+02 | 1.72E+01 | 20 MIN | 2.96-10 | 9 | 915.1 | 1 | 2.25E+02 1 | .726+01 | 20 MIN |
| FüsvE | 1 | A. A. | ; | 2.96-10 | 10 | 8151.1 | ; | 1 | 2.04E+02 | HR I | 2.96-10 | 10 | 8151.1 | ; | 2 | .04E+02 | I HK |
| 614 - Di | rect | large alri | craft cra | ish damages | the MUB; | fire contained | 11 0.5 | hours | | | | | | | | | |
| + 0;kGF | Ξ | A : N | ; | 9.76-14 | 2 | 2656.4 | ; | ; | 2.30E+02 | 30 MIN | 9.75-14 | 13 | 2656.4 | ; | 2 2 | .30E+02 | 30 HIN |
| for a f | = | N:A | ; | 9.76-14 | 1 | 9.116.4 | ł | ; | 7.60E+02 | 30 MIN | 9.76-14 | 13 | 9016.4 | ; | 1 | .605+02 | JU NIN |
| j ti ti ti | = | 2. IL 12 | 11 | 9.76-14 | - | 10219.1 | ; | • | 4. JuE+02 | 30 MIN | 9.75-14 | 13 | 10219.1 | : | + | . 30€+02 | JO NIN |
| Fur VE | Ξ | 2.1E 12 | 1 | 9.76-14 | 11 | 9615.1 | ; | ; | 2.03E+02 | 30 MIN | 9.76-14 | 13 | 9615.1 | ; | - 2 | .03E+02 | 30 MIN |
| 1.501 | 1 | 4'B | : | 9.75 -14 | :3 | 8151.1 | ! | : | 1.72E+02 | JO NIN | 9.75-14 | 13 | 8151.1 | ł | | . 72E+02 | JU NIN |
| 11 - 510 | hdirec | ct large ai | ircraft c | rash dasage | 's the MHI | ; no fire | | | | | | | | | | | |
| Fulfes | 5 | N Å | ; | 2.96-12 | 11 | 7040.0 | : | 0.00E+00 | 6.30€+00 | 1 hr | 2.9E-12 | 13 | 7040.0 | 1 | 0.00E+00 6 | .30E+00 | l hr |
| FUDHL | 2 | 6. JE - 11 | : | 2.96-12 | 11 | 4608.0 | ; | 0.00E+00 | SN | 1 hr | 2.96-12 | 13 | 4608.0 | : | 0. 00E+00 | SN | Ч Н |
| fŨĨĠĈ | 15 | 6.36-11 | :1 | 2.96-12 | 13 | 614.4 | ł | 0.00£+00 | 1.256-01 | 1 hr | 2.96-12 | 13 | 614.4 | : | 0.00E+00 1 | . 25£ -01 | l hr |
| FOLHE | 5 | 6. 36-11 | <u></u> | 2.96-12 | 2 | 1228.8 | 1 | 0,00E+0 | SN | 1 hr | 2.96-12 | 13 | 1228.8 | ! | 0.00E+00 | SM | L hr |
| F 04 65 | 5 | A ' N | 1 | 2.96-12 | 1 | 24660.0 | : | 0.001+00 | 6.30€+00 | 1 hr | 2.96-12 | 13 | 24000.0 | ł | 0.00E+00 6 | . 306+00 | l hr |
| FOAHS | 15 | 6.35-11 | 13 | 2.96-12 | :: | 27200.0 | ; | 0.00E+00 | NS. | 1 hr | 2.9E-12 | 13 | 27200.0 | ; | 0.00E+00 | SN | 1 hr |
| 5 1 1 V 2 | Υ. | 6. 31-11 | 13 | 2.96-12 | 11 | 25600.0 | ; | 0.001 100.0 | SN | 1 hr | 2.96-12 | 11 | 25600.0 | ; | 0. 00E+00 | MS | l hr |
| FGMVL | £ | 6. 35 -11 | 2 | 2.46-12 | 1 | 6048.0 | : | 3.15E+01 | SN | i hr | 2.96-12 | 13 | 6048.0 | 1 | 3.156+01 | SN | l hr |

A CONTRACTOR

a second and second and second in any size and second similar second second and second and second second second

6

4

-- 0.00E+00 2.80E-01

B32.0

=

2.96-12

1 hr

-- 0.00E+00 2.60E-01

832.0

:

2.96-12

2

6. JE - J I

5

F UF 6(



File: FLTCOL.Mkl, 20-Aug-87 PAGE4

<u>aanaan na sa</u>

PLANT OPERATIONS COLLOCATION MEDIAN ACCIDENT FREQUENCY (PER YEAR)

PLANT OPERATIONS COLLOCATION MEDIAN ACCIDENT FREQUENCY (PER YEAR)

| SCENARIO I R | ġ | ANAD KDC | RANGE | TEAD ROC | RANGE | TOTAL And to America | LBS. | LBS. Detanaten | LBS. CMITTED | DURATION TIME | TEAD NDC CDED | RANGE | TOTAL AUATI AN | LBS. | L BS. Netrimater | LDS. D | XURAT JON T I NE |
|-----------------|---------|-----------------------|-----------|-------------|------------|-------------------------|-----------|-------------------|-----------------|------------------|-------------------|-------|-------------------|------|---------------------|----------|---------------------|
| | | - MEN | | - KER | | | אוורנה | UE IUMA CU | | | 1 JUL | | | | | | 3811 |
| JHJOJ | 5 | 6.36-11 | 13 | 2.96-12 | 13 | 1497.6 | : | 0.00E+00 | SN | - 14 | 2.9E-12 | 13 | 1497.6 | 1 | 0.00E+00 | Si | 1 hr |
| POPVC | 5 | 6. 36-11 | 11 | 2.96-12 | 13 | 768.0 | 1 | 0.00E+00 | SN | 1 hr | 2.9E-12 | 13 | 768.0 | : | 0.00E+00 | SH | l hr |
| J900 ł | 2 | 6. 3E ⁻ 11 | 11 | 2.96-12 | : | 1392.0 | ; | 0.00E+00 | 4.53E-01 | 1 hr | 2.9E-12 | 13 | 1392.0 | : | 0.00E+00 | 4.536-01 | - hr |
| FUQUC | 51 | N.A | ; | 2.96-12 | 13 | 1392.0 | ! | 0.00£400 | SN | 1 hr | 2.96-12 | 11 | 1392.0 | 1 | 0.00E+00 | SH SH | l hr |
| F 0F6C | 12 | 6. 35-11 | = | 2.9E-12 | 11 | 2568.0 | ! | 2.14E+01 | 5.656-01 | 1 hr | 2.96-12 | 13 | 2568.0 | 1 | 2.14E+01 | 5.65E-01 | 14 |
| FGKVÛ | 2 | 6. JE-11 | 13 | 2.96-12 | 13 | 2400.0 | : | 2.00E+01 | SN | l hr | 2.9E-12 | 11 | 2400.0 | : | 2.006+01 | NS: | 1 hr |
| FOSVS | 15 | N/A | : | 2.96-12 | 11 | 21696.0 | 1 | 0.00E+00 | SN | 1 hr | 2.96-12 | 13 | 21696.0 | ł | 0.00E+00 | SH | 1 hr |
| F016 - In | dire | ct large a | urcraft c | rash danage | is the NHI | ; fire not ci | ontained | in 0.5 hou | Ir S | | | | | | | | |
| F066F | 9 | N/A | : | 2. 36-12 | 11 | 7040.0 | ; | 1 | 7.04E+02 | ÷. | 2. 36-12 | : | 7040.0 | : | : | 7.04E+02 | - 190 |
| FODHC | 16 | 5.26-11 | 13 | 2.4E-12 | 13 | 4408.0 | 1 | 1.156+03 | 1.73E+02 | 20 NIM | 2.4E-12 | 13 | 1608.0 | 1 | 1.15€+03 | 1.73E+02 | 20 MIN |
| FGCGC | 16 | 5.26-11 | 11 | 2.4E-12 | 11 | 614.4 | 1 | 1.54E+02 | 4.61E+01 | 20 MIN | 2.4E-12 | 13 | 614.4 | ! | 1.54E+02 | 4.61E+01 | 20 MIN |
| FOCHC | 16 | 5.2E-11 | : | 2.4E-12 | 11 | 1228.8 | 1 | 3.07E+02 | 4.61E+01 | 20 NIN | 2.4E-12 | 13 | 1228.8 | : | 3.07E+02 | 4.61E+01 | 20 MIN |
| Fük6f | 16 | N/A | ; | 2.36-12 | 11 | 24000.0 | ł | ; | 2.40E+03 | œ. | 2.36-12 | 13 | 24000.0 | : | 1 | 2.40E+03 | 9 1 1 |
| FORHE | 16 | 5. IE - 11 | 11 | 2.36-12 | :1 | 27200.0 | 1 | 1 | 1.36E+03 | 5 | 2.11-12 | 11 | 27200.0 | : | 1 | 1.366+03 | ž – |
| FUKVF | 16 | 5. IE-11 | 11 | 2.3E-12 | :1 | 25600.0 | : | 1 | 6.40E+02 | 1 HB | 2.3E-12 | 2 | 25600.0 | : | ; | 6.40E+02 | ۲. ۲ |
| FONYC | 16 | 5.2E-11 | 11 | 2.4E-12 | :: | 6048.0 | : | 1.51E+03 | 1.13E+02 | 20 NIN | 2.4E-12 | = | 6048.0 | : | 1.51E+03 | 1.13€+62 | 20 MIN |
| FOPEC | 2 | 5.26-11 | 11 | 2.4E-12 | 13 | 832.0 | 1 | 2.08E+02 | 6.24E+01 | 20 NIN | 2.4E-12 | 13 | 832.0 | : | 2.00£+02 | 6.24E+01 | 20 NIN |
| FOFHC | 16 | 5.2E-11 | = | 2.4E-12 | = | 1497.6 | : | 3.74E+02 | 5.62E+01 | 20 MIN | 2.4E-12 | :1 | 1497.6 | 1 | 3.74E+02 | 5.62E+01 | 20 NIN |
| FOFVC | 16 | 5.26-11 | 13 | 2.4E-12 | 13 | 768.0 | ; | 1.92E+02 | 1.44E+01 | 20 NIN | 2.4E-12 | 13 | 768.0 | ; | 1.92E+02 | 1.44E+01 | 20 NIN |
| 19001 | 16 | 5.2E-1I | 11 | 2.4E-12 | = | 1392.0 | : | 3.486+02 | 1.04E+02 | 20 MIN | 2.4E-12 | :1 | 1392.0 | : | 3.40€+02 | 1.04E+02 | 20 MIN |
| FUGVC | 16 | N/A | ; | 2.4E-12 | 11 | 1392.0 | 1 | 3.486+02 | 2.61E+01 | 20 MIN | 2.4E-12 | 13 | 1392.0 | ; | 3.48€+02 | 2.61E+01 | 20 NIN |
| FÜRGC | 16 | 5.26-11 | = | 2.46-12 | 13 | 2568.0 | : | 6.42E+02 | 1.936+02 | 20 MIN | 2.4E-12 | 11 | 2568.0 | 1 | 6.42E+02 | 1.93E+02 | 20 MIN |
| FORVC | 16 | 5.26-11 | 11 | 2.4E-12 | = | 2400.0 | 1 | 6.00E+02 | 4.50E+01 | 20 MIN | 2.4E-12 | 13 | 2400.0 | ; | 6.00E+02 | 4.50E+01 | 20 MIN |
| FOSUF | 16 | N/A | : | 2.3E-12 | 1 | 21696.0 | 1 | : | 5.42E+02 | H H | 2.3E-12 | 13 | 21696.0 | : | ; | 5.42E+02 | 1 HK |
| F017 - In | idi rev | ct large a | urcraft c | rash damage | s the MHI | ; fire contai |) ul paul |).5 hours | | | | | | | | | |
| FUBBF | 17 | N/A | : | 8.0E-16 | 16 | 7040.0 | ! | 1 | 2.20E+01 | 30 ain | B. 0E - 16 | 16 | 7040.0 | 1 | ; | 2.20€+01 | 30 ein |
| f uix GF | 2 | N/A | ; | 8.06-16 | 16 | 2400.0 | : | : | 1.50E+02 | 30 min | B. 0E - 16 | 16 | 24000.0 | : | : | 1.50E+02 | 30 BIN |
| FOK NF | 17 | 1.86-14 | 16 | 8.0E-16 | 16 | 27200.0 | : | : | 8.50€+01 | 30 min | B.0E-16 | 16 | 27200.0 | : | 1 | 8.50€+01 | 30 min |
| FGA VF | 1 | 1.86-14 | 16 | B.0E-16 | 16 | 25600.0 | : | : | 4.005+01 | 30 min | B.0E-16 | 16 | 25600.0 | : | ; | 4.00E+01 | 30 ain |
| FOSVF | 1 | N/A | : | 8.0E-1o | 16 | 21696.0 | ; | : | 3.39E+01 | 30 mm | 8.06-16 | 16 | 21696.0 | : | ł | 3.39€+01 | 30 BIR |

File: FLTCOL.NK1, 20-Aug-87 PAGE5

| | | PLAN NEDIAN | IT OPERATI Accident | FREQUENCY (| VTION Pek yeai | 2 | | | | | | | PLANI NEDIAN A | CCIDENT F | DNS COLLOC | ATION (PER YEA | Î |
|-----------|-------|-------------------|------------------------|-----------------------|-------------------|--------------|-----------|------------|----------|------------------|------------|--------|-------------------|-----------|------------|-------------------------|----------------|
| SCENARIO | R. | ANAD RDC | RANGE | TEAD RDC | RANGE | TOTAL | 185. | LBS. | 1.95. | DURATION | TEAD MDC | RANGE | TOTAL | LDS. | LBS. | LIBS. | URATION |
| - E. | | FRED | FACTOR | FREQ | FACTOR | AVAILABLE | SPILLED | DETOMATED | ENITTED | 1 INE | FRED | FACTOR | AVAILAB | 1 0311145 | DETOWATED | ENTIED | |
| F018 - 1a | direc | t large a | ircraft c | rash da na ge: | s the MDB: | no fire | | | | | | | | | | | |
| F086S | 8 | N/A | : | 4.0E-10 | = | 2640.0 | ; | 1 | 6.10E+00 | 8¥ 1 | 4.06-10 | = | 2640.0 | : | 1 | 6.10E+00 | ž. |
| FOGHC | 8 | B. BE - 09 | Π | 4.0E-10 | Π | 1728.0 | ; | : | SN | ۹. | 4.06-10 | Ξ | 1728.0 | ; | ; | SN | ۲. ۲ |
| F0C6C | 8 | 8.85-09 | Π | 4.05-10 | Ξ | 230.4 | 1 | ļ | 1.04E-01 | œ | 4.0E-10 | 11 | 230.4 | : | : | 1.04E-01 | 또 - |
| FOCHC | 8 | 8.8 E-09 | Ξ | 4.0E-10 | Ξ | 460.8 | ł | 1 | SN | H | 4.0E-10 | = | 460.8 | : | : | SN | 1 HK |
| F 04 6S | 8 | N/N | ł | 4.0E-10 | Ξ | 9000.0 | ; | 1 | 6.80E+00 | Ť. | 4.0E-10 | = | 9000.0 | : | : | 6.805+00 | 1 NG |
| F-OA HS | 81 | 8.66-09 | Π | 4.05-10 | 11 | 10200.0 | i | 1 | SN | ÷ | 4.0E-10 | Ξ | 10200.0 | : | : | ŝ | - 2 |
| F-04 VS | 18 | 8.BE-09 | Ξ | 4. ŬE-10 | 11 | 9600.0 | ; | ł | SN | ۲. | 4. ÚE-10 | Ξ | 9600.0 | : | ; | W2 | 1 15 |
| FONUC | 18 | 8.85-09 | 11 | 4.0E-10 | Π | 2268.0 | ; | 3.15E+01 | SH | Ť. | 4.0E-10 | Ξ | 2268.0 | 1 | 3.15E+01 | SN | 1 15 |
| F0F6C | 8 | 8.85-09 | Ξ | 4.05-10 | Ξ | 312.0 | ; | ; | 1.30E-01 | 또 - | 4.06-10 | Ξ | 312.0 | ; | 1 | 1.30E-01 | ЭШ Ш |
| F OF HC | 10 | 8.8E-09 | Ξ | 4.05-10 | 11 | 561.6 | 1 | ; | SN | 1 1 1 1 | 4.0E-10 | 11 | 561.6 | 1 | : | SN | 1 HK |
| FDFVC | 8 | 8.85-09 | Π | 4.05-10 | Ξ | 288.0 | ; | : | SN | Ŧ | 4.0E-10 | Ξ | 288.0 | 1 | 1 | SN | ¥ |
| F006C | 8 | B.BE-09 | н | 4.05-10 | 11 | 522.0 | 1 | : | 1.99E-01 | Ť. | 4.05-10 | Η | 522.0 | 1 | : | 10-366.1 | ЭH П |
| FOGVC | 99 | K/A | 1 | 01-30.4 | = | 522.0 | 1 | ł | SN | 1 #2 | 4.06-10 | Ξ | 522.0 | ; | ! | SH | 1 HR |
| FORGC | 8 | 8.86-09 | Π | 4.0E-10 | 11 | 963.0 | 1 | 2.14E+01 | 5.44E-01 | Ě. | 4.0E-10 | Π | 963.0 | 1 | 2.14E+01 | 5.446-01 | L HK |
| FORVE | 8 | 8.8E-ù9 | = | 4.0E-10 | Π | 900.0 | ł | 2.00E+01 | SM | F | 4.05-10 | Ξ | 900.0 | ł | 2.00E+01 | SN | I HK |
| FOSVS | 18 | N/A | ; | 4.0E-10 | Ξ | 8136.0 | 1 | ! | SN | E | 4.0E-10 | Ξ | 8136.0 | 1 | 1 | SN | 9H 1 |
| f019 - In | direc | t large a | ircraft c | rash danage | s the MOB | i fire not c | contained | 1n 0.5 hou | Irs | | | | | | | | |
| FOLGF | 61 | N/A | ! | 3.3E-10 | Ξ | 2640.0 | ; | ! | 2.64E+02 | 99 - | 3.3E-10 | Ξ | 2640.0 | ; | 1 | 2.64E+02 | 1 HR |
| FODHC | 19 | 7.25-09 | 11 | 3. 3E-10 | Ξ | 1728.0 | ; | 4.32E+02 | 6.48E+01 | 20 NIN | 3.36-10 | = | 1728.0 | : | 4.32E+02 (| 6. 4 8E + 01 | 20 MIN |
| F0C6C | 19 | 7.26-09 | Ξ | 3.36-10 | Ξ | 230.4 | 1 | 5.76E+01 | 1.73€+01 | 20 NIN | 3.36-10 | Ξ | 230.4 | : | 5.76E+01 | 1.736+01 | 20 MIN |
| FOCHC | 61 | 7.2E-09 | Π | 3.36-10 | Π | 460.8 | : | 1.15E+02 | 1.73E+01 | 20 MIN | 3. 3E - 10 | = | 460.8 | ; | 1.15€+02 | 1.73E+01 | 20 MIN |
| FONGF | 6 | N/A | : | 3.3E-10 | 11 | 9000.0 | ; | 1 | 9.00E+02 | - HR | 3.3E-10 | Ξ | 9000.0 | ; | 1 | 9.00E+02 | I HÊ |
| FOHF | 6] | 7.1E-49 | Ξ | 3.3E-10 | Η | 10200.0 | ; | ! | 5.10E+02 | EH - | 3.3E-10 | = | 10200.0 | ; | ; | 5.10E+02 | 、天 |
| F04 VF | 61 | 7.16-09 | Ξ | 3.36-10 | = | 9400.0 | ł | : | 2.40E+02 | 1 #5 | 3.3E-10 | Ξ | 9600.0 | 1 | 1 | 2.40E+02 | 1 HR |
| F-DMVL | 19 | 7.26-09 | Ξ | 3.3E-10 | Π | 2268.0 | ; | 5.676+02 | 4.256+01 | 20 MIN | 3.3E-10 | Ξ | 2268.0 | : | 5.67E+02 | 4.256+01 | 20 NIN |
| FUF6C | 19 | 7.25-09 | Ξ | 3. 36 - 10 | Ξ | 312.0 | : | 7.80€+01 | 2.34E+01 | 20 MIN | 3.36-10 | Ξ | 312.0 | : | 7.80€+01 | 2.34E+01 | 20 NIN |
| JH 30-1 | 19 | 7.25-09 | Ξ | 3. 3E-10 | = | 561.6 | 1 | 1.40E+02 | 2.11E+01 | 20 MIN | 3.36-10 | = | 561.6 | 1 | 1.40€+02 | 2.11E+01 | 20 MIN |
| FOFVE | 19 | 7.7E-09 | Ξ | 3.3E-10 | = | 289.Ù | i | 7.20E+01 | 5.40E+00 | 20 MIN | 3.36-10 | Ξ | 268.0 | : | 7.20E+01 | 5.40E+00 | 20 MIN |
| J970.j | 19 | 7.2E-09 | 11 | 3. 3E - 10 | 11 | 522.0 | 1 | 1.316+02 | 3.92E+01 | 20 MIN | 3.3E-10 | Π | 522.0 | ł | 1.31E+02 | 3.92E+01 | 20 MIN |

AND ADDREAM THE ADDREES AND ADDREES AND ADDREES AND ADDREES AND ADDREES S.








KANCENERS AND

File: FLTCOL.Mk1, 20-Aug-87 PAGE6

<u> Charles and Charles and Charles and Charles and Charles and Charles and Charles and Charles and Charles and C</u>

| | 10M | | NIN O | NIN O | 王 | | NIN O | O NIN O | NIN O | NIN O | 0 NIN | | Ĩ | 1 HK | 1 HR | | 0 NIN | NIN O | 0 HIN |
|----------------------------|---------------------|----------|----------|----------|----------|--------------|----------|----------|----------|----------|----------|-------------|----------|----------|----------|-------------|----------|----------------|-----------------|
| YEAR) | DURA D 111 | ~ • | 1 2 | 1 2 | 2 | | 2 2 | 2 | ۳ ۶ | 1 3 | 1 3 | | ! | ! | ! | | - | 1 1 | - |
| PER | LBS. Emitte | . 79E+0 | 1.22E+0 | i.69E+0 | 2.03E+0 | |). 70E+0 | 9.60E+0 | 2.706+0 | 1.00E-0 | 1.00E-0 | | • | , | • | | 5.485+0 | 5. 19E+0 | I.27E+0 |
| NIS COLLOCA | LBS. Jetonated | 1.31E+02 | 2.416+02 | 2.25E+02 | 1 | | 1 | 1 | 1 | | 1 | | : | : | 1 | | 1 | 1 | 1 |
| OPERATIC CIDENT F | LBS. | 1 | : | : | ; | | : | 1 | : | 1 | 1 | | 4BE+02 | 38E+02 | 07E+02 | | ; | 1 | 1 |
| PLANT - MEDIAN ACI | TDTAL Availab Si | 522.0 | 963.0 | 900.0 | 8136.0 | | 2640.0 | 9000.0 | 10200.0 | 9600.0 | 8136.0 | | 548.0 5. | 638.0 6. | 507.0 5. | | 548.0 | 638.0 | 507.0 |
| | RANGE Factor | = | Π | = | Ξ | | = | = | | Ξ | = | | 2 | 9 | 01 | | 9 | 91 | 01 |
| | TEAD NDC Freq | 3.3E-10 | 3.3E-10 | 3.36-10 | 3.36-10 | | 1. IE-13 | 1. IE-13 | 1.IE-13 | 1. IE-13 | 1.16-13 | | 1.06-08 | 1.05-09 | 1.06-08 | | 8.26-09 | 8.25-09 | B. 2E-09 |
| | DURATION Time | 20 MIN | 20 MIN | 20 MIN | £ | | 30 MIN | | ÷ | E HE | i He | curs | NIN OI | 10 MIN | IO NIN |
| | LBS. Enitted | 9.79E+00 | 7.22E+01 | 1.69E+01 | 2.03E+02 | | 9.70E+00 | 9.60£+00 | 2.70E+00 | 7.005-01 | 7.006-01 | id; no fire | : | ; | 1 | 10; fire oc | 5.486+01 | 3.19E+01 | 1.276+01 |
| | LBS. Detonated | 1.31E+02 | 2.416+02 | 2.25E+02 | 1 | .5 hours | ; | : | 1 | : | : | itee at TEA | : | : | 1 | itee at TEA | : | ł | ! |
| | LBS. Spilled | : | ; | : | : | ained in 0 | : | ł | ; | : | ł | piping sys | 5.40E+02 | 6.38E+02 | 5.07E+02 | piping sys | : | 1 | ł |
| 2 | TDTAL AVAILABLE | 522.0 | 963.0 | 900.0 | 8136.0 | fire cont | 2640.0 | 9006.0 | 10200.0 | 9600.0 | 8136.0 | door agent | 548.0 | 638.0 | 507.0 | door agent | 548.0 | 638.0 | 507.0 |
| TION FER YEAU | RANGE FACTOR | = | П | == | Ξ | s the MDB | = | : | = | = | 1 | s the out | 9 | 9 | 61 | s the out | 10 | 10 | 01 |
| ons colloca Frequency i | TEAD RDC Freq | 3.36-10 | 3.3E-10 | 3.36-10 | 3. 3E-10 | rash danage: | 1.16-13 | 1.16-13 | I. IE-13 | 1.16-13 | 1. IE-13 | rash daaage | 1.0E-08 | 1.05-08 | 1.0E-08 | rash damage | B. 2E-09 | B.2E-09 | 8. 2E-09 |
| T OFERATI Accident | RANGE FACIOR | : | Ξ | 11 | : | urcraft c | : | ; | • | • | : | ircraft c | ; | ł | : | ircraft c | ; | 1 | : |
| PLAN MEDIAN | ANAD RDC Freq | N/A | 7.25-09 | 7.26-09 | N/A | t large a | N/A | N/A | 2.46-12 | 2.4E-12 | N/A | r small a | N/A | N/A | N/A | ir small a | N/A | N/A | N/A |
| | .DM | 61 | 61 | 61 | 51 | direc | 3 | 20 | 2 | 20 | 3 | r qe o | 2 | 5 | 21 | irge o | .2 | 5 | 32 |
| | SCENAR10 1. b. | FOOVC | FORGE | FORVC | FOSVE | F020 - 1a | FOI66F | F0+6F | F0 Hf | F.0k VF | FUSVF | P021 - La | FOA6S | FOAHS | FOAVS | F022 - La | FOAGS | FOAHS | FOAVS |

I-91

Nates:

Frequency unit = events/operating Year
 Scenarios 21 and 22 apply only to the TEAD bulk-only facility

File: FLTCDL.Wk1, 20-Aug-87 FA6E1

 \mathbf{x}

| | | FLANT Median ai | UPERATIN CLIDENT R | DNS COLLOCA FREQUENCY (| rion Per yea | بر | | | | | | | PLANT Median A | r opekatio Accident fi | NS COLLOCA Reduency (| ATION (PER YEA | - X |
|------------------|-------------|--------------------|-----------------------|----------------------------|-----------------|--------------------|-----------------|-------------------|-----------------|------------------|----------------------|-----------------|-------------------|---------------------------|--------------------------|--------------------|----------------------------|
| SCENARIO 1.D. | NO. A | ANAO ROC Ffee I | KANGE Factor | TEAD KDC Freq | RANGE FACTOR | TOTAL Available | LBS. Spilled | LBS. Detonated | LBS. Emitted | DURATION TIME | TEAD NDC Fred | RANGE Factor | TOTAL Availab | LBS. SPILLED D | LBS. Etonated | LBS. D Enited | UKATION TIME |
| F3.5 - Ea | rthaua | ule danage: | the MOL | 9 structure. | . eunitio | s fall t ou | incture: 1 | fire supare | ssed | | | | | | | 8 | , , , , , , |
| FübGC | 8 | N/A | • | 1.96-07 | 1 | 2640.0 | . 1 | | 2.20E+01 | 6 HS | 1.95-07 | 7 | 2640.0 | : | 3 | 2.20E+01 | 6 HF |
| FURAC | К | NĔĠĹ | ; | NEGL | ; | 1728.0 | ł | : | 3.005-01 | 6 HR | NEGL | 1 | 1728.0 | 1 | - | S. 00E-01 | 6 H.F |
| FULLE | ž | NEUL | ; | NEGL | ; | 230.4 | ! | : | 2.00E-01 | 6 HR | NEGL | ; | 230.4 | 1 | - 7 | 2.00E-01 | 6 HR |
| FUCHČ | 23 | NEGL. | ł | HEGL | ł | 46Ú.B | 1 | : | 1.60E-01 | 9 ¥ | NEGL | ; | 460.8 | ; | : | 10-309.1 | 6 天 6 |
| F0+61 | ŝ | N/A | ł | 1.6E-Üb | 7 | 9000.0 | 1 | 1 | 1.50E+02 | 6 HR | 1.6E-06 | 7 | 9000.0 | : | 1 | L. 50E+02 | 6 HF |
| Farm | 5 | 7.1E-08 | 1 | 1.6E-Üb | F*- | 10200.0 | ł | ; | 8.50E+01 | 6 H | I.6E-06 | 1 | 10200.0 | 1 | 8 | 3.50E+01 | 6 HK |
| FC: VC | <u>ان</u> | 7.1E-08 | 1 | 1.65-05 | 1 | 9600.0 | 1 | : | 4.00E+01 | 6 HR | 1.6E-06 | 2 | 9600.0 | ; | - | I, 00E+01 | A HK |
| FURVE | 53 | 2. JE-09 | -0 | 5.06-08 | | 2068.0 | ł | : | 2.60E-01 | 6 HE | 5.0E-08 | 7 | 2268.0 | ł | - 3 | 2.60E-01 | 6 HK |
| ruf60 | 53 | NEGL | 1 1 | NEGL | ! | 312.0 | ; | : | 6.50E-01 | 9 HR | NEGL | : | 312.0 | ; | - | 5. 50E-01 | 6 HK |
| FUFHL | 13 | NEGL | ; | NEGL | : | 501.6 | 1 | 1 | 5.906-01 | 6 HR | NEGL | : | 561.6 | : | 1 | 5.906-01 | 6 HK |
| FuFVC | ÷7 | NEGL | ! | NEGL | : | 208.0 | ; | ; | 1.50E-01 | 6 HR | NEGL | 1 | 268.0 | ł | : | 1.506-01 | 6 HK |
| 10304 | ង | NE GL | 1 | NEGL | ; | 522.0 | 1 | : | 2.70E-01 | 6 폱 | NEGL | ł | 522.0 | ł | 1 | 2.7ùE-01 | 圣 9 |
| FOLVO | 5 | N/A | ł | NEGL | ; | 522.0 | 1 | ł | 3.60E-01 | 6 HR | NEGL | ; | 522.0 | ł | | 5.60E-01 | 9 HK |
| FURGE | 13 | 1.56-08 | Ŷ | 3.3E-07 | 1 | 963.0 | : | 1 | 1.10E+00 | 6 HR | 3.3E-07 | ~ | 963.0 | 1 | 1 | 1.10E+00 | 6 HR |
| FORVC | <u>ال</u> ا | 1.56-08 | 9 | 3.36-07 | 1 | 900.0 | 1 | : | 2.50E-01 | 6 HR | 3.3E-07 | 1 | 900.0 | 1 | - 2 | 2.50E-01 | A HK |
| FOSVC | 13 | 3. 3E-07 | 1 | B.4E-06 | 7 | 8136.0 | : | I | 3.40E+01 | 6 HR | B. 4E-06 | 7 | 8136.0 | 1 | - | 3.40E+01 | 6 HK |
| r026 - ča | rthqua | ike damage: | s the KGI | 8 structure, | , suntio | ns fall & pu | incture; (| ear thquake | initiates | fire; fire | suppression system f | ails. | | | | | |
| FübüC | <u>2</u> 6 | N/A | : | 60-31.6 | 21 | 2640.0 | 1 | : | 2.64E+02 | 우 폱 9 | 6.1E-09 | 13 | 2640.0 | ł | - 1 | 2.64E+02 | 6 浜 |
| F ÜCHC | .9 | HE BL | ; | NEGL | : | 1728.0 | : | 4.326+02 | 6.48E+01 | 6 폱 | NEGL | ł | 1728.0 | ; | 4.32E+02 6 | 5.48E+01 | A HR |
| Fücec | 92 | NE GL | ; | NEGL | : | 230.4 | : | 5.76E+01 | 1.73E+01 | 6 HR | NEGL | : | 230.4 | 1 | 5.766+01 1 | 1.73E+01 | 6 丧 |
| FOCHC | 2 | NEGL | : | NEGL | ; | 460.8 | : | 1.15E+02 | 1.73E+01 | 9 분 | NEBL | ! | 460.8 | ; | 1.15E+02 1 | 10+361.1 | 6 HR |
| Fir GC | 26 | N/H | ; | 4.96-08 | = | 9000.0 | 1 | ; | 9.00E+02 | 6 HR | 4.96-08 | 11 | 9000.0 | i | 1 | 7.00E+02 | 6 HR |
| FUNK | 93 | 1.86-09 | 11 | 4.95-08 | 13 | 10200.0 | : | : | 5.10E+02 | 6 HR | 4.96-08 | 13 | 10200.0 | : | רש | 5.10E+02 | 6 HR |
| FGEVC | 2 | 1.86-09 | 11 | 4.9 E-1/9 | 13 | 9600.0 | ; | 1 | 2.40E+02 | 6 HE | 4.96-08 | 13 | 9600.0 | ; | : | 2.40E+02 | 6 HR |
| FUTVC | 2 | NEGL | ; | NEGL | : | 2268.0 | : | 5.67E+02 | 4.256+01 | 6 轰 | MEBL | : | 2268.0 | : | 5.67E+02 4 | I. 25E+01 | 6 HR |
| FOFEC | 2 | NEGL | ; | NEGL | : | 312.0 | ; | 7.80E+01 | 2.34E+01 | 6 玉 | NEGL | ı | 312.0 | ; | 7.80E+01 2 | 2.34E+01 | 6 HR |
| PJFHC | <u>7</u> 9 | NEGL | ; | NEGL | : | 561.6 | : | 1.40E+02 | 2.116+01 | 6 HR | NEGL | ł | 561.6 | ł | 1.40E+02 2 | 2.11E+01 | 6 HR |
| FJFVC | 9 | NEGL | 1 | NEGL | ; | 288.0 | : | 7.20E+01 | 5.40E+00 | 6 HR | NEGL | ; | 289.0 | 1 | 7.20E+01 5 | 5.40E+0ù | 6 HR |
| F0091 | 26 | HEBL | ł | NEGL | : | 522.0 | : | 1.31E+02 | 3.92E+01 | 6 H | NEGL | : | 522.0 | ł | 1.316+02 3 | 5. 92E+01 | 9 H¥ |





DURAT I ON TIME -0 PLANT OPERATIONS COLLOCATION MEDIAN ACCIDENT FREQUENCY I PER YEAR 1.40E+02 2.11E+01 7.20E+01 5.40E+00 1.31E+02 3.92E+01 LBS. Emitted -- 2.64E+02 1.31E+02 9.79E+00 2.41E+02 7.22E+01 -- 5.106+02 -- 2.40E+02 1.31E+02 9.79E+00 2.03E+02 5.10E+02 2.40E+02 2.25E+02 1.69E+01 2.036+02 2.64E+02 4.326+02 6.486+01 5.7bE+01 1.73E+01 -- 9.00E+02 5.e7E+02 4.25E+01 7.80E+01 2.34E+01 2.41E+02 7.22E+01 2.25E+02 1.69E+01 4.32E+02 6.48E+01 5.766+01 1.736+01 1.15E+02 1.73E+01 -- 9.00E+02 5.47E+02 4.25E+01 1.15E+02 1.73E+01 ession system fails. LBS. LBS. SPILLED DETONATED 1 : 1 ÷ 1 1 1 1 + + + + 1 1 1 : : : ł : : 1 ł ł 1 1 1 1 1 1 1 TDTAL 8136.0 2268.0 312.0 561.6 288.0 522.0 522.0 522.0 963.0 900.0 460.8 9000.0 10200.0 9600.0 963.0 900.0 ire supp 9000.0 8136.0 2640.0 1728.0 2640.0 1728.0 230.4 460.8 10200.0 9600.0 2268.0 230.4 RANGE FACTOR <u>9999999999999999</u> fire; 1 = = = 1 2 2 2 1 1 1 2 TOX is intact; earthquake unitia NEGL 1.0E-08 1.0E-08 2.7E-07 2.26-05 2.26-05 2.26-05 2.26-05 2.26-05 2.26-05 2.26-05 2.26-05 2.2E-05 2.2E-05 2.2E-05 2.2E-05 2.2E-05 2.2E-05 N/A 4.BE-05 4.BE-05 2.2E-05 2.2E-05 2.2E-05 4.85-05 N/A N/A **BE-05** N/A TEAD NOC FREQ DURATION) 변 8 뚶 뚶뚶 £ fails. Ħ 뚶 € H 1 1 1 £ ¥ Æ 또 똪 TIME 뚶 -0 9.79E+00 7.22E+01 1.69E+01 2.03E+02 on system 1.736+01 1.736+01 1.736+02 5.106+02 5.106+02 2.406+02 4.256+01 2.346+01 2.346+01 2.116+01 5.406+00 3.926+01 9.796+00 2.64E+02 1.69E+01 6.46E+01 1.73E+01 5. 10E+02 6.48E+01 7.22E+01 2.64E+02 1.73E+01 9.00E+02 2.40E+02 2.03E+02 4.25E+01 ENLITED the MDB is intact; the LBS. -- 1.31E+02 9 -- 2.41E+02 7 -- 2.25E+02 1 -- 2.25E+02 1 -- 2.25E+02 1 -- 2 1.31E+02 • 31E+02 TOTAL LOS. LDS. Available Spilled Detomated 1.31E+02 2.41E+02 2.25E+02 5.76E+01 1.15E+02 --2.41E+02 4.32E+02 5.75E+01 1.15E+02 --7.20E+01 2.25E+02 ł ł 5.67E+02 ł 7.80E+01 1.40E+02 4.32E+02 5.67E+02 ł 1 1 ÷ OCCUFS: ition occurs; 522.0 963.0 900.0 8136.0 460.8 9000.0 10200.0 9500.0 2268.0 312.0 561.6 288.0 522.0 522.0 fire a 2640.0 1729.0 230.4 963.0 900.0 8136.0 2640.0 1728.0 230.4 460.8 2268.0 9000.0 5600.0 0200.0 MEDIAN ACCIDENT (PEQUENCY (FER YEAR) are inf RANGE FACTOR 2222222222222222222222 : = = = 1 2 2 2 1 1 1 8 3 PLANT OFERATIONS COLLOCATION but euni ti ons 2. 26 - 05 2. 15 - 05 2. 27 - 05 2. 27 - 05 2. 26 - 05 2. 26 - 05 2. 26 - 05 2. 26 - 05 2. 26 - 05 2. 26 - 05 2. 26 - 05 2. 26 - 05 2. 26 - 05 2. 26 - 05 2. 27 - 05 to fall f NE6L 1.0E-08 1.0E-08 2.7E-07 N/A 4.8E-05 4.8E-05 4.8E-05 N/A N/A N/A 6
 N/A
 N/A

 2.6
 4.0E-10
 11

 A.0E-10
 11
 11

 F05vC
 2.6
 4.0E-10
 11

 F05vC
 2.6
 N/A
 -

 F05vC
 2.6
 N/A
 -

 F05vC
 2.6
 N/A
 -

 F05vC
 2.9
 N/A
 -

 F05vC
 2.9
 N/A
 -

 F04vC
 29
 186-07
 9
 2.7

 79
 7.6E-97
 7
 9
 2.7
 TEAD ROC FRED **.**8 Ount tons RANGE Factor ł 18881118 ¢ FOSI - Earthquake causes FOEGC 33 N/A 1. 7E-06 1. 7E-06 1. 7E-06 7.86-07 7.86-07 7.86-37 7.8E-07 7.5E-37 7.8E-07 7.6E-07 7.86-07 7.66-07 R/N N/A A/A A/A A/A 7E-Cb ANAD RUC FREQ ₽. 8888888888 ***** SUEMARIO FGAVC FOFEC r Jf Hù F.F.C F0060 F00VC FORGE FOSVC FÚCHC FOYGC FOYHC FOYHC FUCEC FUCHC FGHC FÜKVÜ FGAVC F04.5C 10116 FüDHL

뚶퐃픛똪

픉 폱 ≌ 훉

푼 푼 풋 푼 푼

똪똪 ₩

File: FLTCOL.WK1, 20-Aug-87 FAGES

| () 8 | JURAT I CN TIME | 6 HR | 6 HR | 6 HF | è Hƙ | 6 HF | 6 祇 | 6 浜 | 6 HR | |
|-----------------------------|----------------------|-----------|----------|----------|----------|----------|----------|------------|----------|--|
| ATION FER YEA | LBS. C | 2.34E+01 | 2.11E+01 | 5.40E+00 | 3.92E+01 | 9.79E+00 | 7.226+01 | 10+369.1 | 2.03E+02 | |
| ons colloc Frequency | L6S. Detokated | 7.80€+01 | 1.40€+02 | 7.20E+01 | 1.31E+02 | 1.31E+02 | 2.41E+02 | 2.25E+02 | 1 | |
| OPERATI CIDENT | LBS. PILLED | 1 | 1 | 1 | ; | ł | 1 | 1 | 1 | |
| PLANT MEDIAN AC | TOTAL AVAILAB S | 312.0 | 561.6 | 268.0 | 522.0 | 522.0 | 963.0 | 900.0 | 8136.0 | |
| | RANGE Factor | 30 | 2 | 2 | 20 | 20 | 20 | 20 | 1 | |
| | TEAD NDC Freq | 4.86-05 | 4.86-05 | 4.8E-05 | 4.66-05 | 4.86-05 | 4.86-05 | 4.86-05 | N/A | |
| | DURATION TIME | 至 9 | 6 HR | 6 HK | 6 HS | 6 HR | 6 HR | 6 HR | 6 HR | |
| | LBS. Enitted | 2.34E+01 | 2.11E+01 | 5.406+00 | 3.92E+01 | 9.79E+0ù | 7.22E+01 | 1.69E+01 | 2.03E+02 | |
| | LBS. Detonated | 7.80E+01 | 1.40E+02 | 7.20E+01 | 1.31E+02 | 1.31E+02 | 2.41E+02 | 2.25E+02 | 1 | |
| | LBS. PILLED | | 1 | 1 | ; | 1 | ; | ł | : | |
| () | TDTAL Available S | 312.0 | 561.6 | 288.0 | 522.0 | 522.0 | 963.0 | 900.0 | 8136.0 | |
| FER YEA | RANGE FACTOR | 20 | 20 | 20 | 20 | 20 | 30 | 2Ú | 1 | |
| DNS COLLOCA) Freduency (| TEAD KDC Freq | 4.8E-((5 | 4.8E-05 | 60-38.1 | 4.6E-05 | 4.8E-05 | 4.85-05 | 4.8E-05 | N/N | |
| C OFERATI CCIDENT | RANGE Factor | 20 | 20 | 20 | 20 | 1 | 20 | 20 | ł | |
| PLAN HEDTAN | ÂNÂD RDC Freq | 1. 7E -06 | 1.7E-06 | 1.7E-06 | 1.7E-06 | A.A | 1.76-06 | 1.7E-06 | N/A | |
| | NŬ. | 11 | R | 5 | 2 | 5 | 12, | 11 | 5 | |
| | SCENARIO I.D. | F0F6C | FOFHC | FUFVC | F005C | 37604 | FUKGC | PORUC | 04504 | |

Kütes:

I-94

1. Frequency unit = events/operating year

L.

DATE21-Aug-87 PAGE1 PLTOPSOS

PLANT OPERATIONS INTERNAL INITIATING EVENTS RDC OPTION MEDIAM ACCIDENT FREQUENCY (PER FACILITY-YEAR)

FLANT OPERATIONS INTERNAL INITIATING EVENTS NDC OPTION MEDIAN ACCIDENT FREQUENCY (PER FACILITY-YEAR)

| | SCENA NUMBE | ANAD RDC Fred | RANGE Factor | TEAD RDC Fred | RANGE Factor | TEAD NDC Fred | RANGE Factor | AGENT AVAIL | LBS SPILLED | LRS Detonated | LBS Emítted |
|-----|----------------|------------------|-----------------|------------------|-----------------|------------------|-----------------|----------------|----------------|------------------|----------------|
| | | | | | | 3 6 8 9 | | ļ | | | |
| | Ħ | 3.3E-10 | 4.85+01 | 3.3E-10 | 4.8E+01 | 3.3E-10 | 4.8E+01 | 262.5 | 1 | 1 | 1.01E+02 |
| 13 | 1 | 3. 3E-10 | 4.8E+01 | 3.3E-10 | 4.8E+01 | 3.3E-10 | 4 .8E+01 | 262.5 | ł | ł | 1.01E+02 |
| | Ŧ | 3. 3E-10 | 4.8E+01 | 3.3E-10 | 4.8E+01 | 3.3E-10 | 4.8E+01 | 175.5 | ł | 1 | 1.80E+00 |
| പ | 42 | N/A | : | 9.96-09 | 3.7E+01 | 9, 9E -i)9 | 3.76+01 | 11.0 | ; | 1 | 2.20E+01 |
| പ | 42 | 9.9E-09 | 3.7E+01 | 9.95-09 | 3.7E+01 | 9.95-09 | 3.7E+01 | 21.6 | ł | ł | 2.16E+01 |
| ى | 42 | 9.9E-ù9 | 3.7E+01 | 9.9E-19 | 3.7E+01 | 9.9E-09 | 3.7E+01 | 5.8 | ł | ł | 5.80E+00 |
| പ | 42 | 9.9E-09 | 3.7E+01 | 9.9E-09 | 3.7E+01 | 9.95-09 | 3.7E+01 | 11.5 | ł | ł | 1.15E+01 |
| ပ္ပ | 42 | N/A | ł | 9.9E-09 | 3.7E+01 | 9.9E-09 | 3,7E+01 | 75.0 | ł | ł | 3.75E401 |
| ല | 42 | 9.96-09 | 3.7E+01 | 9.9E-09 | 3.7E+01 | 9.95-09 | 3.7E+01 | 85.0 | ł | ł | 4.25E+01 |
| ų | 42 | 9.9E-09 | 3. 7E+01 | 9.9E-09 | 3.7E+01 | 9.9E-09 | 3.7E+01 | 80.0 | ł | 1 | 4,00E+01 |
| ų | 42 | N/A | : | N/A | ; | N/A | 1 | ; | ł | ! | ; |
| ير | 42 | 9.9E-09 | 3.7E+01 | 9.9E-09 | 3. 7E+01 | 9.9E-09 | 3. 7E+01 | 15.6 | ł | ł | 1.56E+01 |
| ي | 42 | 9.9E-09 | 3.7E+01 | 9.9E-09 | 3.7E+01 | 9.95-09 | 3.7E+01 | 28.1 | ł | ł | 2.81E+01 |
| ب | 42 | 9.98-09 | 3.7E+01 | 9.9E-09 | 3.7E+01 | 9.95-09 | 3.7E+01 | 14.4 | ; | 1 | 1.44E+01 |
| പ | 42 | 9.9E-09 | 3.7E+01 | 9.9E-09 | 3.7E+01 | 9.96-09 | 3.7E+01 | 19.6 | ţ | ł | 1.96E+01 |
| പ | 42 | N/A | ł | 9.9E-09 | 3.7E+01 | 9, 9E - 09 | 3.7E+01 | 19.6 | ł | 1 | 1.96E+01 |
| പ | 42 | N/A | 1 | N/A | ł | N/A | ! | 1 | ţ | 1 | ł |
| പ | 42 | N/A | 1 | N/A | ł | N/A | ! | ; | ł | ł | ; |
| رب | 42 | N/A | ł | 9.9E-09 | 3.7E+01 | 9.96-09 | 3.7E+01 | 67.8 | ١ | 1 | 3.39E+01 |
| പ | 54 | N/A | ł | 1.6E-09 | 4.1E+01 | 1.6E-09 | 4.1E+01 | 220.0 | ł | 1 | NEGL |
| ں | 24 | N/A | 1 | 2.3E-10 | 4.1E+01 | 2.3E-10 | 4.1E+01 | 1500.0 | ł | ł | NEGL |
| ې | 43 | 2.7E-10 | 4. IE+01 | 2.7E-10 | 4. IE+01 | 2.7E-10 | 4. iE+01 | 1700.0 | ł | ł | NEGL |
| ى | 43 | 1.5E-10 | 4.1E+01 | 1.5E-10 | 4. JE+01 | 1.5E-10 | 4. 1E+01 | 1600.0 | ł | 1 | NEGL |
| പ | £\$ | N/A | : | 1.8E-10 | 4.1E+01 | 1.8E-10 | 4.1E+01 | 1356.0 | ļ | ; | NEGL |
| പ | 44 | N/A | ; | 1.6E-10 | 4.1E+01 | 1.6E-10 | 4.1E+01 | 220.0 | ; | ; | 2.20E+02 |
| د | 44 | N/A | ; | 2. JE-10 | 4.1E+01 | 2.3E-10 | 4.1E+01 | 1500.0 | ; | ł | 1.50E+03 |
| പ | 44 | 2.7E-10 | 4.1E+01 | 2.7E-10 | 4.1E+01 | 2.7E-10 | 4.1E+01 | 1700.0 | ł | 1 | 1.70E+03 |
| ب | 4 | 1.5E-10 | 4. IE+01 | 1.5E-10 | 4.1E+01 | 1.56-10 | 4.1E+01 | 1600.0 | ł | : | 1.60E+03 |
| പ | 4 | N/A | ; | 1.8E-10 | 4, 15+01 | 1.86-10 | 4.1E+01 | 1356.0 | 1 | 1 | 1.36E+03 |

12 NIN

÷

12 MIN **12 NIN** 12 MIN 12 MIN 12 MIN

ł 1

ł ł

IS MIN IS MIN

DURATION

...... **JHI**

15 NIN 12 MIN 12 MIN

12 HIN 12 MIN NIN ZI 12 NIN 12 NIN 122223

2 MIN

2 MIN 2 MIN 2 MIN

2 NIN

1 1 1

I-95

DATE21-Aug-87 PAGE2 PLT0PS0S

RDC OPTION MEDIAN ACCIDENT FREQUENCY (PER FACILITY-YEAR) PLANT OPERATIONS INTERNAL INITIATING EVENTS

| | CLEND | ANAN RNC | EANGE | TEAN DAG | PANGE | TEAN MAC | BANCE | ACENT |
|--------------|-------|----------------|------------------|----------|----------|----------|----------|--------|
| | NUMBE | FRED | FACTOR | FRED | FACTOR | FRED | FACTOR | AVAIL |
| | | | | | | | | |
| POKGF | 45 | N/A | 1 | 4.0E-09 | 1.4E+01 | 4.0E-09 | 1.4E+01 | 1500.0 |
| POKHF | 45 | 4.05-10 | 1.4E+01 | 4.05-10 | 1.4E+01 | 4.0E-10 | 1.4E+01 | 1700.0 |
| POKVF | 45 | 4.06-10 | 1.4E+01 | 4.0E-10 | 1.4E+01 | 4.0E-10 | 1.4E+01 | 1600.0 |
| FODHC | 46 | 9.0E-09 | 2.6E+01 | 9.0E-09 | 2.6E+01 | 9.0E-09 | 2.6E+01 | 6.0 |
| POCEC | 46 | 1.0E-08 | 2.6E+01 | 1.06-08 | 2.6E+01 | 1.0E-08 | 2.6E+01 | 1.6 |
| POCHC | 46 | 1.0E-08 | 2.6E+01 | 1.0E-08 | 2.6E+01 | 1.0E-08 | 2.6E+01 | 3.2 |
| POHVC | 44 | 4.0E-07 | 2.6E+01 | 4.0E-07 | 2.6E+01 | 4.0E-07 | 2.6E+01 | 10.5 |
| J9d0d | 46 | 6.0E-07 | 2.6E+01 | 6.0E-07 | 2.6E+01 | 6.0E-07 | 2.6E+01 | 6.5 |
| POPHC | 46 | 6.0E-07 | 2.6E+01 | 6.0E-07 | 2.6E+01 | 6.0E-07 | 2.6E+01 | 11.7 |
| POPVC | 46 | 6.0E-07 | 2.6E+01 | 6.0E-07 | 2.6E+01 | 6.0E-07 | 2.6E+01 | 6.0 |
| POGGC | 46 | 3.0E-07 | 2.6E+01 | 3.0E-07 | 2.6E+01 | 3.0E-07 | 2.6E+01 | 14.5 |
| POOVC | • | N/A | ; | 3.0E-07 | 2.6E+01 | 3.0E-07 | 2.6E+01 | 14.5 |
| PORGC | 46 | 1.5E-07 | 2.7E+01 | 1.5E-07 | 2.7E+01 | 1.5E-07 | 2.7E+01 | 10.7 |
| PORVC | 46 | 1.5E-07 | 2.7E+01 | 1.5E-07 | 2.7E+01 | 1.5E-07 | 2.7E+01 | 10.0 |
| PODHC | 14 | 8. IE-11 | 3. IE+01 | 8. IE-11 | 3. 1E+01 | 8. 1E-11 | 3.1E+01 | 6.0 |
| POCEC | 11 | 9.0E-11 | 3. IE+01 | 9.06-11 | 3.16+01 | 9.0E-11 | 3. 1E+01 | 1.6 |
| POCHC | 14 | 9.06-11 | 3. iE+01 | 9.06-11 | 3. IE+01 | 9.0E-11 | 3.1E+01 | 3.2 |
| PONUC | 14 | 3.6E-09 | 3, 16+01 | 3.6E-09 | 3. JE+01 | 3.6E-09 | 3. IE+01 | 10.5 |
| P0P6C | 11 | 5.4E-09 | 3. IE+01 | 5.4E-09 | 3. 1E+01 | 5.4E-09 | 3. IE+01 | 6.5 |
| DHODA | 14 | 5.4E-09 | 3. IE+01 | 5.4E-09 | 3.1E+01 | 5.4E-09 | 3. IE+01 | 11.7 |
| PUPUC | 14 | 5.4E-09 | 3. IE+01 | 5.4E-09 | 3. IE+01 | 5.4E-09 | 3. 1E+01 | 6.0 |
| P006C | 47 | 2. TE-09 | 3. IE+01 | 2.7E-09 | 3.1E+01 | 2.7E-09 | 3. IE+01 | 14.5 |
| JAOOd | 4 | N/A | : | 2.7E-09 | 3.1E+01 | 2.7E-09 | 3. JE+01 | 14.5 |
| PORGC | 47 | 1.5E-07 | 2.7E+01 | 1.5E-07 | 2.7E+01 | 1.56-07 | 2.7E+01 | 10.7 |
| PDRVC | 14 | 1.5E-07 | 2. TE+01 | 1.5E-07 | 2.7E+01 | 1.56-07 | 2.7E+01 | 10.0 |
| PODHC | 48 | 9.0E-12 | 3.3E+01 | 9.06-12 | 3.3E+01 | 9.0E-12 | 3.3E+01 | 1728.0 |
| POCGC | 48 | 1.0E-11 | 3. 3E+01 | 1.0E-11 | 3.3E+01 | 1.06-11 | 3. 3E+01 | 230.4 |
| POCHC | 48 | 1.0E-11 | 3. JE +01 | 1.06-11 | 3. 3E+01 | 1.0E-11 | 3. 3E+01 | 460.8 |
| POHVC | 48 | 4.05-10 | 3. 3E+01 | 4.06-10 | 3.3E+01 | 4.06-10 | 3. 3E+01 | 1268.0 |

FLANT OFERATIONS INTERNAL INITIATING EVENTS Mdc option median accident frequency (per facility-year)

NIN 901 II4 NIN BO NIN

5.03E+01 2.97E+01 1.91E+01

ł

DURATION

LBS

Sa Sa

LBS

11ME

ENITTED

DETONATED

SPILLED NIN 09 AIN 05

1.00E-01

1.00E-01 .005-01

3.00E-03

ł

NEGL

NEGL

1 1

ł 1 1

1

NIN 09 NIN 09

3.00E-01 B.00E-03

6.00E-01

1.00E+00

ł

NEGL

NEBL

1 1 1 1 I

1

NIN 09 NIN 09 NIN 09 **NIN 09**

2.90E+00

2.40E+00

NEGL

2.60E+00

1

5.00E-01

00+300.1 9.00E-01

1

NEGL NEGL

1

Ň

3

NEGL

5.00E-01

20 MIN 20 HIN

1.50E-01

1.00E-01 1.00E-01

ł 1 1

NEGL

1

1.506-01

ł

20 MIN 20 HIN

6.00E-01

6.00E-01 1.00E+00 5.00E-01

1

5.00E-01

1

1

ł

NEGL

NEGL

| 46 | 46 | 46 | 46 | 46 | 4 | 46 | 46 | ; |
|-------|--------------|-------|-------|-------|-------|-------|-------|---|
| POMUC | J9d0d | POPHC | POPVC | PODEC | POOVC | PORGC | PORVC | |



20 MIN 20 MIN 20 MIN 20 MIN 20 MIN 20 MIN

1.00E+00

.00E+00 9.00E-01 2.16E+02

1 1

2.00E-01

3.20E+01

1.20E+00

2.60E+00

ł 1

1.406-01

3.005-01

2.60E+00

20 MIN 20 MIN 20 MIN

1.72E+01 4.30E+01

. 15E+02 5.67E+02

> ł 1

1.72E+01

5.80E+01

•







DATE21-Aug-87 PAGE3 PLTOPSOS

PLANT OPERATIONS INTERNAL INITIATING EVENTS Roc option median accident frequency (per facility-year)

PLANT OPERATIONS INTERNAL INITIATING EVENTS NDC OPTION MEDIAN ACCIDENT FREQUENCY (PER FACILITY-VEAR)

| | SCENA NUMBE | ANAD RDC Freq | RANGE Factor | TEAD RDC Freq | RANGE Factor | TEAD NDC Freq | RANGE Factor | AGENT AVAIL | LBS SPILLED | LBS Detonated | LBS Emitted | DURATION TIME |
|--------|----------------|------------------|-----------------|------------------|------------------|------------------|-----------------|----------------|----------------|------------------|----------------|------------------|
| | | | | 8 | | | | | | | | |
| OPGC | 48 | 6.0E-10 | 3.3E+01 | 6.0E-10 | 3.3E+01 | 6.0E-10 | 3.3E+01 | 312.0 | ł | 7.90E+01 | 2.36E+01 | 20 MIN |
| OPHC | 4 8 | 6.0E-10 | 3. JE+01 | 6.0E-10 | 3.3E+01 | 6.0E-10 | 3, 3E+01 | 561.6 | 1 | 1.42E+02 | 2.15E+01 | 20 MIN |
| J.\dD, | 4 8 | 6.0E-10 | 3. JE+01 | 6.0E-10 | 3.3E+01 | 6.0E-10 | 3. 3E+01 | 288.0 | 1 | 7.25E | 5.10E+00 | 20 MIN |
| 0860 | 48 | 3.0E-10 | 3, 3E+01 | 3.0E-10 | 3. 3E+01 | 3.0E-10 | 3.3E+01 | 522.0 | ł | 1.34E+02 | 4.02E+01 | 20 MIN |
| JV90' | 48 | 4/2 | 1 | 3.0E-10 | 3. 3E+01 | 3.0E-10 | 3.3E+01 | 522.0 | ł | 1.34E+02 | 1.03E+01 | 20 MIN |
| PORGC | 48 | 1.5E-09 | 3.2E+01 | 1.5E-09 | 3. 2E+01 | 1.5E-09 | 3.2E+01 | 963.0 | ; | 2.42E+02 | 7.30E+01 | 20 MIN |
| PORVC | 48 | 1.5E-09 | 3.2E+01 | 1.5E-09 | 3.2E+01 | 1.5E-09 | 3.2E+01 | 900.0 | ł | 2.26E+02 | 1.72E+01 | 20 MIN |
| PODHC | 49 | 3.0E-06 | 3. 1E+01 | 3.0E-06 | 3. IE+01 | 3.0E-06 | 3. JE+01 | 6.0 | ł | 6.00E+00 | ł | INSTANT |
| POCGC | 49 | 4.0E-06 | 3. IE+01 | 4.0E-06 | 3. IE+01 | 4.0E-06 | 3. 1E+01 | 1.6 | 1 | 1.40E+00 | 1 | INSTANT |
| POCHC | 49 | 4.0E-0b | 3. IE+01 | 4.0E-06 | 3.1E+01 | 4.0E-06 | 3.1E+01 | 3.2 | 1 | 3.20E+00 | ł | INSTANT |
| POMUC | 49 | 4.0E-09 | 3.7E+01 | 4.0E-09 | 3.7E+01 | 4.0E-09 | 3.7E+01 | 10.5 | ł | 1.05E+01 | ł | INSTANT |
| POPGC | 49 | 2.0E-06 | 3, 1E+01 | 2.0E-06 | 3. 1E+01 | 2.0E-06 | 3. 1E+01 | 6.5 | ł | 6. 50E+00 | ł | INSTANT |
| POPHC | 49 | 2.0E-0b | 3. IE+01 | 2.0E-06 | 3. 1E+01 | 2.0E-06 | 3. 1E+01 | 11.7 | ł | 1.17E+01 | ł | INSTANT |
| POPVC | 49 | 2.0E-06 | 3. IE+01 | 2.0E-06 | 3.1E+01 | 2.0E-06 | 3. 1E+01 | 6.0 | ł | 6. 00E+00 | ł | INSTANT |
| P086C | 43 | 8.0E-07 | 3. 1E+01 | B. (E-1)7 | 3. 1E+01 | 8.0E-07 | 3.1E+01 | 14.5 | ł | 1.45E+01 | : | INSTANT |
| POQVC | 46 | R/N | : | 8.0E-07 | 3. 1E+01 | 8.0E-07 | 3. IE+01 | 14.5 | ł | 1.45E+01 | ł | INSTANT |
| PORGC | 49 | 5.0E-07 | 3.4E+01 | 5.0E-07 | 3.4E+01 | 5.0E-07 | 3.4E+01 | 10.7 | ł | 1.07E+01 | 1 | INSTANT |
| PORVC | 49 | 5.0E-07 | 3.4E+01 | 5.0E-07 | 3.4E+01 | 5.0E-07 | 3.4E+01 | 10.0 | 1 | 1.00E+01 | ł | INSTANT |
| PGDHC | 50 | 3.0E-08 | 3.7E+01 | 3.0E-08 | 3.7E+01 | 3.0E-08 | 3.7E+01 | 6.0 | 1 | 6.00E+00 | ł | INSTANT |
| POCEC | 50 | 4°)E-08 | 3. 7E+01 | 4.0E-08 | 3.7E+01 | 4.0E-08 | 3.7E+01 | 1.6 | 1 | 1.60E+00 | 1 | INSTANT |
| POCHC | 0 17 | 4.0E-08 | 3.7E+01 | 4.0E-08 | 3.7E+01 | 4.0E-08 | 3.7E+01 | 3.2 | 1 | 3.20E+00 | 1 | INSTANT |
| PONC | 3 | 4.0E-11 | 3.7E+01 | 4.05-11 | 3.7E+01 | 4.0E-11 | 3.7E+01 | 10.5 | ł | 1.05E+01 | ł | INSTANT |
| P0F6C | 50 | 2.0E-08 | 3.7E+01 | 2.0E-08 | 3. 7E+01 | 2.0E-08 | 3.7E+01 | 6.5 | 1 | 6.50E+00 | 1 | INSTANT |
| POPHC | 50 | 2.0E-08 | 3. 7E+01 | 2.0E-08 | 3. 7E+01 | 2.0E-08 | 3.7E+01 | 11.7 | ł | 1.17E+01 | ł | INSTANT |
| P0PVC | 20 | 2.0E-08 | 3.7E+01 | 2. ùE-08 | 3. 7E+ů1 | 2.0E-0B | 3.7E+01 | 6.0 | ł | 6.0E+00 | 1 | INSTANT |
| P006C | 33 | 8. 0E-09 | 3.7E+01 | 8.0E-09 | 3.7E+01 | 8.0E-09 | 3.7E+01 | 14.5 | : | 1.45E+01 | ! | INSTANT |
| POQVC | 20 | N/A | ; | 8.0E-09 | 3.7E+01 | 8.0E-09 | 3.7E+01 | 14.5 | : | 1.45E+01 | ; | INSTANT |
| PORGC | ŝ | 5.0E-07 | 3.4E+01 | 5.0E-07 | 3.4E+01 | 5.0E-07 | 3.4E+01 | 10.7 | ł | 1.07E+01 | 1 | INSTANT |
| PGRVC | 20 | 5.0E-07 | 3.4E+i)] | 5.0E-07 | 3 . 4E+01 | 5.0E-07 | 3.4E+01 | 10.0 | 1 | 1.00E+01 | : | INSTANT |

| | NUNBE | FRED | FACTOR | FREQ | FACTOR |
|-------|-------|----------|----------|----------|----------|
| | | | | | |
| POKGF | 51 | N/A | 1 | 4.0E-09 | 1.4E+01 |
| POKHF | 51 | 4.0E-09 | 1.4E+01 | 4. 0E-09 | 1.4E+01 |
| POKVF | 51 | 4.0E-09 | 1.4E+01 | 4.0E-09 | 1.4E+01 |
| PODHC | 52 | 4.4E-03 | 5. 7E+01 | 4.4E-03 | 5. 7E+01 |
| PDCGC | 52 | 5.0E-03 | 5.7E+01 | 5.0E-03 | 5.7E+01 |
| POCHC | 33 | 5.0E-03 | 5. 7E+ûl | 5.06-03 | 5.7E+01 |
| PONVC | 52 | 1. IE-02 | 5.7E+01 | 1.1E-02 | 5.7E+01 |
| D9404 | 5 | NEGL | ł | NEGL | ł |
| POPHC | 25 | NEGL | ł | NEGL | 1 |
| POPVC | 52 | NEGL | ł | NEGL | 1 |
| PODGC | 23 | NEGL | 1 | NEGL | ł |
| 200VC | 52 | N,A | ł | NEGL | ; |
| PORGC | ដ | 1.6E-03 | 5.7E+01 | 1.6E-03 | 5. 7E+01 |
| PORVC | 5 | 1.6E-03 | 5.7E+01 | 1.6E-03 | 5.7E+01 |

| | | DURATION TIME | 61.4 NIN | 69.3 MIN | 34.7 NIN | INSTANT | INSTANT | INSTANT | INSTANT | INSTANT | INSTANT | INSTANT | INSTANT | INSTANT | INSTANT |
|------------|----------------------------|------------------|----------------------|----------|----------|------------------|----------|----------|----------|----------|----------|----------|----------|----------|----------|
| | Y-YEAR) | LBS Enitted | 2.90E+01 | 1.80E+01 | 5.B0E+00 | ; | ł | ł | ł | ; | ł | ł | ł | ł | ł |
| | ENTS ER FACILIT | LBS Detonated | ; | 1 | ; | 6.00E+00 | 1.60E+00 | 3.20E+00 | 1.05E+01 | 6.50E+00 | 1.17E+01 | 6.00E+00 | 1.45E+01 | 1.456+01 | 1.07E+01 |
| | AFING EVE Jency (P | LBS PILLED | ł | ! | ! | ł | ; | 1 | ł | ł | ; | 1 | | ł | ł |
| | RNAL INITI | AGENT | 1500.0 | 1700.0 | 1600.0 | 6.0 | 1.6 | 3.2 | 10.5 | 6.5 | 11.7 | 6.0 | 14.5 | 14.5 | 10.7 |
| | TIONS INTE | RANGE FACTOR | 1. (E+01 | 1.4E+01 | 1.4E+01 | 5. <i>7</i> E+01 | 5.7E+01 | 5.76+01 | 5.7E+01 | ; | ł | ł | 1 | 1 | 5. 7E+01 |
| | PLANT OPERA NDC OPTION | TEAD NDC Fred | 4.05-09 | 4.0E-09 | 4.0E-09 | 4.46-03 | 5.0E-03 | 5.0E-03 | 1.1E-02 | NEGL | NEGL | NEGL | NEGL | HEGL | 1.6E-03 |
| | ACILITY-YEAR) | RANGE Factor | 1.4E+01 | 1.4E+01 | 1.4E+01 | 5.7E+01 | 5.7E+01 | 5.7E+01 | 5.7E+01 | 1 | 1 | 1 | ł | 1 | 5. 7E+01 |
| | g Events V (Per F | TEAD RDC Freq | 4.0E-09 | 4.0E-09 | 4.0E-09 | 4.4E-03 | 5.0E-03 | 5.06-03 | 1.1E-02 | NEGL | NEGL | NEGL | NEGL | NEGL | 1.6E-03 |
| SOSA | . INITIATIN IT FREQUENC | RANGE Factor | : | 1.4E+01 | 1.4E+01 | 5.7E+01 | 5.7E+01 | 5. 7E+01 | 5.7E+01 | ł | ł | ; | ; | ł | 5.7E+01 |
| PAGE4 PLTC | IS INTERNAL An Accide) | ANAD RDC Freq | A/A | 4°0E-04 | 4.0E-09 | 4.4E-03 | 5.0E-03 | 5.0E-03 | 1. IE-02 | NEGL | NEGL | NEGL | NEGL | A.A | 1.6E-03 |
| 78-97 | PERATION Ion Medi | SCENA NUMBE | 51 | 5 | 51 | 52 | 52 | ទ | 22 | 52 | 25 | 52 | 52 | 52 | 5 |
| DATE21- | PLANT O RDC OPT | | POK6F | POKHF | POKVF | PUDHC | POCGC | POCHC | PONVC | D0404 | POPHC | POPVC | POBGC | P004/C | PORGC |
| | | | | I | 98 | | | | | | | | | | |





211111111

PUNCTUR RECEASE

لعدين

a start and a start of a

ኯፚኯኯ፟ፚኯኯቘኇኯ፼ዹጜቔኯጚዸኯኯቔኯዄቔኯጟቝኇቔዹጟቝኯኯቚኯቔቚ፟ዀቚ፟ዀጜዀ፟ዀቘ፟ዀ፟ዄዀ፟ዀ፟ዀ፟፟፟፟፟፟ዀዀጚኯቒቚዄዸ፟ቚ

The following tables list the accident results for onsite transport of munitions.

L [NS33337] [NS3

Ţ

5678003C

ኯፚኯዀኯፚኯዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀ

F

. 1975 -







NN N N

ONSITE TRANSPORTATION - ONSITE PACKAGE - COLLOCATION OPTION INDVEMENT FROM STORAGE TO GENIL FACILITY IN ONSITE PACKAGE Scenario Frequencies and Range Factors

Agent Available and Released

| | DURATION TIME | | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HKS | 2 HRS | 2 HRS | 2 HPS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | SAH 1 |
|---|-------------------|---|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|------------|-------------|------------|-----------|--------------|-----------|-----------|------------|-----------|-----------|-----------|-----------|------------------|-----------|-----------|-----------|---------------------|---------------------|-----------|
| | LBS. I Emitted | | ; | ; | 1 | : | 1 | ; | 1 | ; | : | 1 | ł | : | 1 | ; | ; | { | : | ; | 1 | : | ł | : | : | : | : | : | : | : | : |
| | LBS. Etcnated | | 1 | 1 | : | 1 | ! | ; | ; | ł | 1 | : | ł | 1 | ; | ł | 1 | 1 | ; | 1 | : | : | t | ; | ; | ł | ; | : | : | : | ; |
| | LBS. SPILLED D | | 2.2E+02 | 6.0E+00 | 1.6E+00 | 3.2E+00 | 1.5E+03 | L. 7E+03 | L.6E+03 | 10+30.1 | 65E+00 | 1.2E+01 | 6.0E+00 | 1.5E+01 | 1.5E+01 | 1.16+01 | 1.úE+01 | 1.46+03 | 3.5E+02 | 2.2E+02 | 6.0E+00 | 1.6E+00 | 3.2E+00 | 1.5€+03 | 1.7E+03 | 1.66+03 | 1.0E+01 | 6.5E+00 | 1.2E+01 | 6. UE+CU | 1.56+01 |
| : | AGENT | | 1760.0 | 1152.0 | 153.6 | 307.2 | 6000.0 | 6800.0 | 6400.0 | 1512.0 | 208.0 | 574.4 | 192.0 | 348.0 | 348.0 | 642.0 | 60.09 | 2712.0 | 1392.0 | 1760.0 | 1152.0 | 153.6 | 307.2 | 6000.0 | 6600.0 | 6400.0 | 1512.0 | 208.0 | 374.4 | 192.0 | 348.0 |
| | RANGE Factor A | | 1 | ł | ; | : | ł | : | ; | 1 | : | : | : | ; | : | ; | ; | : | ; | : | ; | 1 | ; | : | 1 | : | : | 1 | ; | : | ; |
| | UNDA Freg | | N/A | R/A | A/A | N/A | N/A | N/A | N/A | R/N | N/A | A/A | N/A | N/A | N/A | N/A | A/N | N/A | ₹/N | N/A | N/A | A/A |
| | RANGE FACTOR | | 8 2.2E+01 | B 2.2E+01 | 8 2.2E+01 | B 2.2E+01 | B 2.2E+01 | B 2.2E+01 | 8 2.26+01 | 8 2.2E+01 | 8 2.2E+01 | 8 2.2E+01 | 8 2.2E+u1 | 8 2.26+01 | 8 2.2E+01 | 8 2.26+01 | 8 2.2E+01 | 8 2.25+01 | 8 2.2E+01 | 1 2.6E+01 | 1 2.6E+01 | 1 2.66+01 | 1 2.66+01 | 1 2.66+01 | 1 2.6E+01 | 1 2.6E+01 | 1 2.66+01 | 1 2.6E+01 | 1 2.6E+UI | 1 2.6E+Vi | 1 2.56+)1 |
| | TEAD Freq | | 1.4E-0 | 1.4E-0 | 1.46-0 | 1.46-0 | 1, 4E-0 | 1.46-0 | 1.4E-Ù | 1.4E-0 | 1.45-0 | 1.4E-0 | 1.46-0 | 1.46-0 | 1.4E-0 | 1.4E) | 1.46-0 | 1.46-0 | 1.4E-0 | 5.46-1 | 5.4E-1 | 5.45-1 | 5.45-1 | 5.46-1 | 5.46-1 | 5.46-1 | 5.46-1 | 5.46-1 | 5.4E-1 | 5.46-1 | 5.4-1 |
| | RANGE Factor | | 1 | | ł | ; | I | : | 1 | 1 | ; | : | ł | ; | : | : | : | : | ! | 1 | : | 1 | ! | : | : | : | ! | : | : | ł | 1 |
| | PUDA Freq | | N/A | N/N | N/A | N/A | N/A | N/A | M/A | N/A | N/A | E/N | N/A | H/A | N/A | N/A | 8/N | N/A | N/A | N/A | M/A | N/A | R/A | A/A |
| | RANGE FACTOR | | : | : | : | ; | : | : | : | 1 | ; | : | ; | ; | ; | : | : | : | : | ; | ; | : | : | ; | 1 | 1 | : | 1 | 1 | ; | : |
| | PBA Freq | | A/M | N/A | N/A | N/A | N/A | A/A | N/A | A/A | N/A | N/A | NIA | N/A | N/A | A/M | A/A | N/A | N/A | 8/N | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | R/A | N/A |
| | RANGE FACTOR | , | ł | 1 | : | ; | 1 | ; | ł | 1 | : | : | ; | ; | 1 | : | 1 | : | ; | : | : | ; | : | : | : | 1 | ł | : | 1 | : | ; |
| | NAAP Freg | | N/A | N/A | N/A | N/A | N/A | A/A | A/A | N/N | N/A | N/A | K/A | A/A | N/A | N/A | N/A | N/A | N/A | 4/¥ | N/A | N/A | R/N | A/A | A/N | A/A | N/A | N/6 | K/A | A/A | R/A |
| | RANGE Factor | | 1 | ; | : | 1 | ! | ; | 1 | : | : | ; | : | 1 | 1 | : | ; | 1 | ; | ; | : | ! | ; | : | : | : | : | : | ; | ł | 1 |
| | LBAD Freq | | N/A | N/A | A/N | N/A | N/A | A/A | E/N | N/A | N/A | N/A | N/A | A.A | N/A | 4/N | н/н | N/A | A/A | A/A | e/N | N/A | N/A | e/N | N/A | N/A | N/A | A/A | A/A | A/A | 4/H |
| | RANGE Factor | | ! | : | ; | ; | : | 1 | 1 | : | : | : | ; | ; | : | : | : | : | ; | : | : | ; | : | : | : | ; | : | ; | ł | : | : |
| | AP5 Free | | A/A | A/A | N/A | N/A | A/A | N/A | N/A | A/A | N/A | N/A | A/A | M /A | A/A | N/A | R/A | N/A | N/A | N/A | N/A | N/A | 4/H | N/A | N/6 | N/A | A/A | N/A | e/N | N/R | A/N |
| • | RANGE Factor | | | 2.25+01 | 2.26+01 | 10-32.2 | 1 | 2.2E+01 | 2.26+01 | 2.26+01 | 2.2E+01 | 2.2E+01 | 10+32-2 | 2.25+31 | ; | 12.2E+01 | 12.2E+01 | 1 | : | 1 | 2.65+01 | 2.6E+01 | 2.4E+01 | : | Ĵ. 6E+) | 2.6E+01 | 2.46+01 | 10+36-1 | 2.6E+01 | 2.65+91 | 2.65+01 |
| | ANAD Freq | | A/M | 1.46-06 | 1. H-90 | 1.46-06 | N/A | 1.46-08 | 1.46-08 | 1.46-06 | 1.4638 | 1.45-06 | .45-36 | 1.46-06 | M/A | 1. #5-26 | <u>}(-</u> 3 | A/A | A'A | A/A | 5. 46-1 | 5.46-11 | 5. +E-1 | 41K | 1 -11 | 5.46-11 | 5.46-11 | Ω.#-1 | - 1 -4-1 | 5. f f-1 | 5. 16-11 |
| | ż | ! | | - | - | - | ~ | - | - | - | - | | - | - | | - | | - | - | ۴ ٦ | m | •7 | ~ | *1 | m | 17 | ** | - | ** | ** | • • |
| | SCEN- | | VOBES | SHOD/ | VOCES | VOCHS | VOKES | NDKHS | SV YOV | NURVS | VOPES | VOPHS | SAJOA | 20802 | SVOCV | VORGS | VORVS | SASCA | VONGS | V0865 | SHOD! | VOCES | VOCHS | SOUGE | VOKHS | VOLVS | SANDA | VOFSS | SH4DA | SVODV | V0065 |

See nutes at end of table.

I-101

5.5

File: CMEITFEG.MANI Fage 2 Date 19-Aug-87

OMSITE TRANSPORTATION - OMSITE FATKAGE - CULGGATION OFTION (ROVEMENT FROM STORAGE TO DEWIL FACILITY IN OMSITE PACKAGE) Scematic Frequencies and Kange Factors

Agent Available and Released

| URATION Tine | - | | 1 HAS | 2 HRS | 2 HIRS | 2 HRS | 20 MIN | ZO HIN | 20 HIN | 20 NIN | Zo NIN | 20 NIN | 20 MIN | 20 MIN | 20 HIN | 20 NIN | ZO MIN | Ŧ | 補し | 王 | Ŧ | 9 4 | ۹. | 2 HRS | Sim 1 | 2 HRS | 2 HKS | 2 HRS | 2 HRS | 2 HES |
|-------------------|---|------------------------------|----------|---------|---------|------------|---------|---------|-------------|---------|-------------|---------|---------|---------|----------|----------|----------|---------|---------|----------|------------|------------|---------|---------|----------|---------|---------|----------|----------------|-----------|
| LOS. D Emitted | | 1 | ; | : | : | : | 4.36+01 | 1.2E+01 | 1.2E+01 | 4.56+02 | 2.66+02 | 1.2E+02 | 2.86+01 | 10+39.1 | 1.46+01 | 3.6E+00 | 2.46+01 | 1.8€+02 | 6.05+02 | 3.46+02 | 1.65+02 | 6.85+01 | 1.45+02 | : | ł | : | : | ; | ; | ; |
| LBS. ETOMATED | | ł | : | ; | ; | ; | 2.9€+02 | 3.86+01 | 7.76+01 | 3.0£+02 | 5. 2E +01 | 9.4E+01 | 4.86+01 | 8.75+01 | 8. 7E+01 | 1.66+02 | 1.5€+02 | ; | ; | ; | ; | ; | ; | ; | 1.7€+02 | 2.36+01 | 4.66+01 | ł | 1 | 1 |
| LBS. | | 1.76.01 | 1. 16+01 | 10+30-1 | l.4E+03 | 3.5£+02 | : | ; | ; | : | ; | : | ; | : | : | : | : | ł | : | ; | 1 | ; | ; | 1.56+03 | B. 1E+02 | 1.16+02 | 2.25+02 | 5. IE+03 | 5.86+03 | 5.46+03 |
| AGENT VAILABLE | | 148.0 | 642.0 | 600.0 | 2712.0 | 1392.0 | 1152.0 | 153.6 | 307.2 | 1512.0 | 203.0 | 374.4 | 192.0 | 348.0 | 348.0 | 642.6 | 600.0 | 1760.0 | 6000.0 | 680.0 | 6400.0 | 2712.0 | 1392.0 | 1760.0 | 1152.0 | 153.6 | 307.2 | 6000.0 | 6 800.0 | 6400.0 |
| RANGE Factor | | 1 | : | : | t | : | : | : | ; | ; | ; | ł | : | ; | : | ; | ; | ; | ; | ; | ł | ł | : | : | : | : | ł | ; | : | : |
| UNDA Fred | | 4/N | A/A | N/A | N/A | N/A | R/A | A/M | N/A | A/A | R/A | N/A | N/A | N/A | N/A | K/A | e/w | N/A | N/A | A/A | R/A | N/A | M/A | A/A | N/A | N/A | N/A | N/A | A/M | #/# |
| RANGE FACTOR | | · | 2.6E+01 | 2.66+01 | 2.6E+01 | 2.6€+01 | 1.46+02 | 1.4E+02 | 1.4E+02 | 1.46+02 | 1.46+02 | 1.46+02 | 1.4E+02 | 1.4€+02 | 1.46+02 | 3. 3£+01 | 3. 3E+01 | 1.6€+02 | 1.66+02 | 1.6E+02 | 1.6E+02 | 1.6E+02 | 1.6E+02 | 2.0€+01 | 2.0€+01 | 2.0€+01 | 2.0E+01 | 2.0€+01 | 2.0€+01 | 2. OE +01 |
| TEAD Freq | : | - - - - - - - - - - - | 5.46-11 | 11-34.2 | 5.4E-11 | 5.46-11 | 3.06-12 | 3.06-12 | 3.0E-12 | 3.05-12 | 3.0E-12 | 3.06-12 | 3.06-12 | 3.06-12 | 3.0E-12 | 2.2E-10 | 2. 2E-10 | 2.86-14 | 2.BE-14 | 2.8E-14 | 2.06-14 | 2.86-14 | 2.86-14 | 1.66-10 | 1.66-10 | 1.66-10 | 1.66-10 | 1.66-10 | 1.66-10 | 1.66-10 |
| RANGE FACTOR | | ; | ; | ł | : | ; | ; | ; | ; | : | ; | : | : | : | : | 1 | 1 | ; | ; | 1 | 1 | ł | 1 | 1 | 1 | : | ł | : | ł | 1 |
| PUDA FREQ | | 4/1 | 4/W | N/A | M/A | K/A | A/A | A/N | M /A | A/A | N/ 6 | N/A | N/A | N/A | M/A | N/A | A/A | N/A | A/N | 6/N | A/A | N/A | N/A | A/N | A/A | N/A | A/A | A/A | A/A | N/A |
| RANGE Factor | | 1 | ; | ł | : | : | ; | ł | : | 1 | 1 | ł | : | : | ; | : | : | 1 | : | : | : | : | 1 | 1 | : | : | 1 | ; | : | : |
| PBA FREQ | | H/H | N/A | A/A | M/M | N/A | N/A | N/N | N/A | N/A | N/A | M/A | N/A | N/A | N/A | N/A | K/A | M/A | A/A | N/A | N/A | N/A | N/A | N/A | N/N | N/A | N/A | N/A | W/A | N/A |
| RANGE Factor | | : | : | 1 | ; | ; | : | ; | : | : | ! | ł | ; | 1 | ; | : | : | : | ; | ł | ł | 1 | : | : | : | : | ; | : | ; | 1 |
| NAAP FRED | 4 | H/H | N/A | N/A | N/A | A/A | N/A | A/A | A/A | N/A | N/A | ₿/₿ | N/A | K/A | N/A | N/A | A/A | A/M | #/# | N/A | N/N | N/A | A/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A |
| RANGE FACTOR | | : | : | 1 | 1 | : | ; | ł | ; | 1 | : | 1 | : | ; | : | ł | : | : | ; | : | 1 | 1 | : | ; | { | ł | : | : | ſ | : |
| LBAD FRED | | ¥/# | N/A | N/A | N/A | R/A | N/A | R/A | N/A | A/N | 6/N | K/A | M/A | M/A | N/A | 4/W | N/A | A/A | R/A | R/A | N/A | N/A | N/A | A/A | N/N | A/A | N/A | N/A | N/A | N/A |
| KANGE FACTOR | | ; | ł | 1 | ; | ; | : | ; | ; | ; | ; | ; | ; | ; | ; | ; | : | : | : | : | : | ; | : | ; | : | : | : | ; | : | ; |
| AP6 FRED | 1 | | N/A | N/A | N/A | A/M | W/A | N/A | N/A | N/A | N/A | N/A | A/N | A/M | N/A | W/W | N/A | N/A | N/A | A/A | K/A | 4/4 | N/A | M/A | A/N | A/A | N/A | R/R | N/A | N/A |
| RANGE Factor | | 1 | 2.66+01 | 2.6E+01 | : | ł | 1.46+02 | 1.46+02 | 1.4€+02 | 1.4E+02 | 1.46+02 | 1.4E+02 | 1.4E+02 | 1.46+02 | ; | 3.36+01 | 3. 36+01 | ; | : | 1.6E+02 | 1.6€+02 | : | ł | : | 2.0€+01 | 2.0E+01 | 2.0E+01 | : | 2.0E+01 | 2.0€+01 |
| ANAD Freq | 1 | | 5.4E-11 | 5.4E-11 | A/A | A/A | 3.0E-12 | 3.0E-12 | 3.06-12 | 3.0E-12 | 3.06-12 | 3.06-12 | 3.06-12 | 3.06-12 | A/A | 2.86-10 | 2.86-10 | N/A | N/A | 2.86-14 | 2.86-14 | N/A | N/A | N/A | å. 0E-10 | 6.0E-10 | 6.0E-10 | A/A | 6.0E-10 | 6.0E-10 |
| : ا <u>ن</u> | | 'n | • 7 | •7 | m | •~; | + | - | * | - | - | - | - | 4 | - | - | - | 67 | 67 | 67 | w') | 6 7 | 47 | • | Ł | 4 | -13 | ٩ | 40 | • |
| SCEN- | | | VORES | VORVS | SV2DV | VOMES | VOBHC | VJC6C | VICHC | CONVC | V0P6C | VOFIE | 20900 | VDQGC | DAGCA | VOK6C | VORVE | VOBGF | VQ# 65 | VDNHF | VOKVF | VOSVF | VONGF | V0865 | CIHCO.A | VOCEC | VICHC | VOK65 | SHADA | VDKVS |

See notes at end of table.



1 . . .



.





A. C.

File: CWSITREG.KK1 Fage 3 Date 19-Aug-87

والمراجع والمراجع والمراجع

DWSITE FRANSPORTATION - OWSITE PACKAGE - COLLOCATION OPTION INDVEMENT FROM STORAGE TO DENLL FACILITY IN DNSITE PACKAGE) Scenario Frequencies and Range Factors

Agent Available and Released

| | DURATION TIKE | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HES | 2 HRS | 2 HRS | HR. | 20 NIN | ZO NIN | ZO NIN | ¥. | ¥ :- | 1 1 1 | 26 RIN | 20 NIN | ZO NIN | 20 NIN | 20 RIN | ZO NIN | 20 NIN | 20 NIN | 년 동 | ۲. ا | 2 HRS | 2 HRS |
|---|--------------------|----------------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|------------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-------------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|------------|------------|
| | LBS. Emitted | 1 | ; | ; | : | ; | : | ; | ŧ | ł | 1 | 1.85+02 | 4.35+01 | 1.25+01 | 1.25+01 | 6.0E+02 | 3.4E+02 | 1.6E+02 | 2.8E+01 | 10+39-1 | 1.45+01 | 3.6E+00 | 2.6E+01 | 6.5E+00 | 4.8E+01 | 1.16+01 | 6.85+01 | 1.4E+02 | 1 | ; |
| | LBS. Detonated | 2.3E+02 | 3. IE+01 | 5.65+01 | 2.96+01 | 5.2E+01 | 5.2E+01 | 9.6E+01 | 9.05+01 | 1 | ł | ; | 2.9E+02 | 3.BE+01 | 7.7E+01 | ; | ł | ; | 3.86+02 | 5.2E+01 | 9.4E+01 | 4.8E+01 | 8.7E+01 | 8.75+01 | 1.6E+02 | 1.5£+02 | ł | ; | ł | ł |
| | 1 3371145 - S87 | 1.1E+03 | 1.5E+02 | 2.4E+02 | 1.3€+02 | 2.4E+02 | 2.4E+02 | 4.56+02 | 4.26+02 | 2.3£+03 | 1.26+03 | ; | ; | 1 | ; | ; | ł | ; | : | ; | ; | : | ; | ł | ; | ; | : | : | 2.2E+02 | 6.0E+00 |
| | AGENT AVATLABLE | 1512.0 | 208.0 | 374.4 | 192.0 | 348.0 | 348.0 | 642.0 | 600.0 | 2712.0 | 1392.0 | 1.8€+03 | 1.2E+03 | 1.5E+02 | 3. IE+02 | 6.0E+03 | 6.BE+03 | 6.4E+03 | 1.5E+03 | 2.1E+02 | 3.7E+02 | 1.9E+02 | 3.5E+02 | 3.5E+02 | 6.45+02 | 6.0E+02 | 2.7E+03 | 1.4E+03 | 1760.0 | 1152.0 |
| | RANGE Factor | 1 | ; | ; | ł | ; | 1 | 1 | ł | 1 | ; | ł | : | ł | 1 | ; | ; | 1 | 1 | : | ! | ; | ; | ; | : | : | ł | ; | 1 | 1 |
| | UNDA Freq | N/A | R/A | N/A | N/A | A/A | A/A | N/A | N/A | N/A | N/A | N/A | A/N | N/A | N/A | N/A | N/A | A/A | M/A | N/A | R/A | N/A | K/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| | RANGE FACTOR | 0 2.0E+01 | 0 2.06+01 | 0 2.0E+01 | 0 2.06+01 | .0 2.0E+01 | 0 2.05+01 | 0 2.0E+01 | 0 2.06+01 | 0 2.0E+01 | 10+31.1 S | 5 1.IE+01 |
| | TEAD Freq | 1.66-1 | 1.65-1 | 1.6E-1 | 1.66-1 | 1.66-1 | 1.6E-1 | 1.66-1 | 1.6E-1 | 1.66-1 | 1.66-1 | 1.3E-1 | 1.3E-1 | 1-35-1 | 1. 3E-1 | 1.36-1 | 1.36-1 | 1.32-1 | 1.35-1 | 1.3E-1 | 1.35-1 | 1. 3E-1 | 1.35-1 | 1.36-1 | 1.35-1 | 1.3E-1 | 1.3E-1 | 1.36-1 | 1.66-0 | 1.05-0 |
| | RANGE Factor | ; | ; | 1 | : | 1 | ł | : | : | ; | ł | 1 | : | 1 | : | 1 | ; | 1 | : | : | ł | ł | : | : | ł | ; | ; | 1 | ł | 1 |
| | FUDA Freq | N/A | N/A | K/A | N/A | N/A | N/A | N/A | A/A | A/A | N/A | N/A | N/A | N/A | N/A | N/A | M/A | R/A | N/A | N/A | R/A | N/A | A/A | N/A |
| | RANGE FACTOR | : | : | : | : | : | ; | : | : | : | ; | ; | : | 1 | : | : | : | ; | : | 1 | : | ł | : | : | ; | : | : | : | ; | ł |
| | PBA Freq | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | K/R | H/H | N/A | N/R | N/A | M/A |
| | RANGE FACTOR | ł | 1 | 1 | ; | ł | : | 1 | ; | : | : | ; | : | : | : | : | 1 | : | : | : | : | : | ; | : | ; | ; | : | ; | ł | : |
| | NAAP Freq | N/A | A/A | N/A | N/A | N/A | N/A | K/A | N/A | A/A | N/A | R/A | A/N | N/A | N/A | N/A | N/A | R/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | N/A | R/A |
| | RANGE FACTOR | ; | ; | ; | : | : | : | ; | : | : | ł | 1 | 1 | ; | ; | : | ; | ł | ; | ł | : | ; | 1 | : | ł | : | í | : | ; | : |
| | LBAD Freq | A/A | N/A | 8/N | N/A | N/A | N/A | N/A | 8/N | N/A | N/A | N/A | A/A | R/N | N/A | N/A | N/A | N/A | N/A | N/A | Ψ/N | N/A | A/A | N/A | N/A | A/M | NZA | N/A | N/4 | ۲/X |
| | RANGE Factor | ł | ; | ; | ; | ; | : | : | : | : | : | ; | ; | : | : | : | : | : | ; | ; | 1 | ; | ł | ł | ţ | ; | : | ; | ; | 1 |
| | AP6 FREQ | N/A | N/A | N/N | N/A | N/A | N/A | N/A | ħ/Ħ | N/A | NZA | N/A | N/A | N/A | N/A | N/A | N/N | R/A | N/A | N/A | N/A | N/A | N/A | N/A |
| • | RANGE Factor | 2.0E+01 | 2.06+01 | 2.0E+01 | 2.0E+01 | 2.0E+01 | ; | 2.0E+01 | 2.0E+01 | 1 | ; | : | 2.0E+01 | 2.0E+01 | 2.0E+01 | ; | 2.0E+01 | 2.0E+01 | 2.0E+01 | 2. CE+01 | 2.05+01 | 2.0E+01 | 2.0E+31 | ; | 2.0E+01 | 2.0E+01 | : | : | ; | 1.1£+01 |
| | ANAD FREQ | 0 E-1 0 | 06-10 | 01-30" | eE-10 | . CE-10 | N/A | 1.0E-10 | .0E-10 | N/A | N/A | N/A | 1.9E-10 | °I-36" | . 96-10 | N/A | . 95-10 | .96-10 | . 96-10 | .9E-10 | . 96-10 | 01-36") | .96-10 | N/A | . 96-10 | . 96-10 | N/A | R/A | N/A | . 0E-07 |
| | 2 | 9 | ¢ ¢ | 4 | ¢ ¢ | 9 | • | ę | -0 -0 | o | 43 | | * | ~ | - | r | 2 | | ~ | 7 | 7 | 1 | | r- | - | 4 | ~ | - | o - | -13 0- |
| | SCEN- Ario | JANOA | VOPEC | VOFHC | VOPVC | VDQGC | VODVC | VORGC | VJRVC | SV2DV | VDMGS | V086F | VIDHC | 7930A | VICHC | VOX6F | YOY HE | VOKVF | 20HUC | 70760 | VIPHC | VDPVC | 7006C | VODVC | VORGC | VORVC | VOSVF | V046F | V0865 | SHODA |

See notes at end of table.

File: ONSITREG.MK1 Page 4 Date 19-Aug-87

CHRIFE TRANSFORTATION - ONSITE PACKAGE - CCLLUCATION OFTION INDVEMENT FROM STORAGE TO DEMIL FACILITY IN ONSITE PACKAGE) Scenario Frequencies and Range Factors

| - | DURAT JON T ME | 7 1100 | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HKS | 2 HRS | 2 HRS | 2 HKS | 2 HRS | 2 HRS | Z HRS | 2 HRS | 2 HKS | 2 HRS | 6 HRS | 6 HRS | 6 HRS | 6 HRS | 6 HRS | 6 HRS | 6 HRS | 6 HRS | 6 HRS | 6 HRS | 6 HRS | 6 HES | 6 HRS | 6 H45 |
|------------|--------------------|----------|-----------------|--------------|---------|---------|---------|---------|----------|---------|----------|---------|----------|------------|-----------|---------|-----------|---------|---------|---------|---------|---------|---------|---------|---------|------------|----------|---------|---------|----------|
| Rel ease | LBS. Entited | 1 | 1 | ł | ; | : | ł | ; | : | : | ; | : | t | ; | : | ; | ; | : | ł | : | : | ; | ł | ; | : | ; | ; | : | ; | ; |
| able and | LBS. | 1 | : | ; | ; | : | : | : | ; | ; | ; | ; | 1 | ; | : | ł | ; | : | ; | ; | ł | ł | ; | ł | 1 | 1 | ; | ; | ; | 1 |
| gent Avail | LBS. | TE ADO | 3. 2E+00 | 1.56+03 | 1.7E+03 | 1.6E+03 | 10+30.1 | 6.5E+00 | 1.2E+01 | 6.0E+00 | 10+35-1 | 1.5E+01 | (. 1E+0(| 1,0E+01 | 1.4E+03 | 3.5E+02 | 2 2E + 02 | 5.0E+00 | 1.65+00 | 3.2E+00 | 1.56+03 | 1.76+03 | 1.65+03 | 1.0E+01 | 6.5E+00 | 1.2E+01 | 6.0E+00 | 1.56+01 | 1.56+01 | 1.15+01 |
| đ | AGENT AVALLABLE | 7 151 | 307.2 | 4000.0 | 6800.0 | 6400.0 | 1512.0 | 208.0 | 374.4 | 192.0 | 348.0 | 348.0 | 642.0 | 600.0 | 2712.0 | 1392.0 | 1760.0 | 1152.0 | 153.6 | 307.2 | 6000.0 | 6800.0 | 6400.0 | 1512.0 | 208.0 | 274.4 | 192.0 | 348.0 | 348.0 | 542.0 |
| | RANGE FACTOR | 1 | 1 | 1 | ł | 1 | ł | í | ſ | : | ; | ; | ; | 1 | { | ; | ł | ł | : | { | ł | ł | ; | ; | ; | ; | ; | ; | ł | ; |
| | UMDA FREQ | 4/4 | | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | A/A | N/A |
| | RANGE FACTOR | | 1.15+01 | 1. IE+01 | 1.1E+01 | 1.1E+01 | 1.1E+01 | 1.1E+01 | 1. 1E+01 | 1.1E+01 | 1. IE+01 | 1.1E+01 | 1.1E+01 | 1.1E+01 | 1. IE +01 | 1.1E+01 | 1.4E+01 | 1.4E+01 | 1.4E+01 | 1.4E+01 | 1.4E+01 | 1.4E+01 | 1.4E+01 | 1.4E+01 | 10+34.1 | 1.4E+01 | 1,4E+01 | 1.45+01 | 1.4E+01 | 1.4E+01 |
| | TEAD Freq | 20-01 | 1.05-05 | 1.05-05 | 1.05-05 | 1.0E-05 | 1.0E-05 | 1.0E-05 | 1.0E-05 | 1.0E-05 | 1.0E-05 | 1.0E-05 | 1.0E-05 | 1.0E-05 | 1.0E-05 | 1.05-05 | 3.96-08 | 3.96-06 | J.9E-08 | 3.96-06 | 3.9E-08 | 3.9E-08 | 3.9E-08 | 3.95-08 | 3.96-08 | 3.96-08 | 3.95-03 | 3.95~08 | 3.9E-08 | 3. 96-08 |
| | RANGE Factor | 1 | ł | ł | 1 | : | ł | ł | ł | 1 | ; | ; | ; | ł | ł | ł | 1 | ł | ; | ; | ł | ; | ; | 1 | ; | ; | ; | ł | ; | ; |
| | PUDA FREQ | 6/4 | | M/A | N/A | N/A | N/A | N/A | R/A | A/A | A/A | N/A | N/A | N/A | N/A | A/A | A/A | A/A | N/A | A/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | 4/N | N/A |
| | RANGE FACTOR | ! | 1 | ; | : | ł | ; | 1 | 1 | : | ł | 1 | ; | ; | 1 | ; | : | ; | : | : | ; | : | ; | : | : | ł | : | : | ł | ; |
| | PBA I | 6/N | N/A | N/A | N/A | N/A | N/A | A/A | N/A | K/A | K/A | N/A | N/A | N/A | R/8 | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A |
| | RANGE Factor | ; | ; | } | ; | ; | : | ; | 1 | ļ | ; | 1 | ł | 1 | ł | ; | 1 | ; | ; | ; | ; | ; | ; | 1 | : | 1 | ; | 1 | ł | : |
| or s | NARP FREQ | 414 | | N/A | N/A | M/A | N/A | N/A | A/N | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | A/M | N/A | N/A | Ψ/N | N/A | K/A | K/A | A/A | N/A | A/A |
| Range Fact | RANGE Factor | 1 | ! | 1 | ; | : | ; | ; | ; | : | 1 | ; | 1 | : | 1 | ١ | : | 1 | : | ; | : | ł | ! | : | ; | 1 | : | : | ł | : |
| pue sa | LBAD Freq | E/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | N/A | A/A | A/A | K/A | N/A | N/A | N/A | R/A | A/A | N/A | N/A | R/N | N/A | A/N | N/A |
| frequenci | RANGE FACTOR | 1 | ; | ; | ; | ; | ; | ; | ; | ; | ; | ; | : | ; | ; | : | 1 | : | ł | ; | : | : | ; | ; | : | ; | : | : | ; | ; · |
| Cenario | AP6 Free | 9/ M | A/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N.A | N/A | N/A | N/A | N/A | N/A | N/A | R/8 | N/A | N/A | N/A | N.A | N/A |
| S | RANGE FACTOR | 1 15 401 | 1.16+01 | 1 | 1.16+01 | 1.1E+01 | 1.15+01 | 1.1E+01 | 1.16+01 | 1.16+01 | :. !E+0! | ł | 10+31.1 | 1.16+01 | : | : | ; | i.4E+01 | 10+3t.: | 1.45+01 | : | 1.4E+01 | 1.46+01 | 1.45+01 | 1.4E+01 | 1.4E+01 | i. 4E+01 | 1.45+01 | ; | 1.45+01 |
| | ANAD FREG | 05-07 | 6-01 6.0E-07 | N/A | 6.0E-07 | 6.0E-07 | 6.0E-07 | 6.0E-07 | 6.0E-07 | 6.0E-07 | 6.UE-07 | N/A | 6.UE-07 | 6.0E-07 | N/A | M/A | N/A | 1.46-09 | 3°4E-09 | 2.4E-09 | 878 | 2.4E-09 | 2.46-09 | 2.4E-09 | 2.4E-09 | 2.45-03 | 2.45-09 | C-4E-03 | £/3 | 2.4E-05 |
| | 3 | a | | م | • | œ | • | e. | æ | -دن | - - | • | \$ | с - | • | 6- | Ξ | Ξ | :: | = | 11 | = | = | = | Ξ | Ξ | Ξ | = | =: | Ξ |
| | SCEN- | CELEC | VOCHS | VDK65 | SHUDA | VDEVS | SVIICA | VOPES | SHjûn | SA JON | V0955 | SABCA | VOKAS | SVAC | SV2CV | VONES | V0955 | SHODA | \$830V | 2430% | VOLGS | VOKHS | 201-25 | SVHON | ·0F65 | SHdOt | SV-JCA | 10065 | SNOCA | :::DESS |

See notes at end of table.

デジン

•

: •

I-104







Sector (Sector)

File: OKSITRE6.MK1 Page 5 Date 19-Aug-87

ONSITE TRANSPORTATION - DNSITE PACKABE - COLLOCATION OPTION (NOVEMENT FACH STORAGE TO DENIL FACILITY IN ONSITE PACKAGE) Scenario Frequencies and Range Factors

Agent Available and Released

| DURAT I DN T I ME | 9 HKS | 6 HRS | 9 HRS | 20 MIN | 20 HIN | ZÙ MIN | 20 MIN | 20 MIN | 20 NIN | 20 NIN | ZC NIN | 20 NIN | 20 HIN | 20 MIN | HK I | 1 1 1 | Ť. | 1 H | 1 HR | 1 #5 | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS |
|----------------------|---------|--------------|---------|-----------|----------|-----------|----------|----------|------------|----------|----------|----------|----------|----------|---------|-----------------|----------|----------|-----------|----------|-----------|----------|----------|----------|---------|-----------|----------|----------|---------|
| LBS. Emitted | : | : | : | 4.36+01 | 1.2E+01 | 1.26+01 | 2.8E+01 | 1.6E+01 | 1.4E+01 | 3.6E+00 | 2.6E+01 | 6.5€+00 | f.8E+01 | 1.16+01 | 1.8E+02 | 6.0E+02 | 3.4E+02 | 1.46+02 | 6. BE +01 | 1.4E+02 | 5.36-01 | 2.4E-02 | 5.3E-01 | 2.4E-02 | : | : | ; | 2.06-04 | 5.3E-01 |
| DETONATED | ; | 1 | : | 2.9E+02 | 3.8€+01 | 7. 7E+01 | 3.8E+02 | 5. 2E+01 | 9.4E+01 | 4.BE+01 | B. 7E+01 | 8. 7E+01 | 1.6E+02 | 1.5€+02 | 1 | : | : | ł | : | ; | ; | ł | 1 | 1 | : | : | : | : | : |
| LDS. | 1.0€+01 | 1.4€+03 | 3.5E+02 | ; | : | 1 | ; | ł | ł | : | ł | ; | 1 | ; | ; | ; | 1 | ł | ; | ; | : | ł | ł | 1 | 9.0€+02 | 1.0E+03 | 9.6E+02 | : | : |
| AGENT AVAILABLE | 600.0 | 2712.0 | 1392.0 | I. 2E+C3 | 1.5€+02 | 3. IE+02 | 1.56+03 | 2.1E+02 | 3. 7E+02 | 1.9£+02 | 3.5E+02 | 2.5E+02 | 6, 4E+02 | 6. 0E+02 | 1.86+03 | 6. 0E+03 | 6.8E+03 | 6.4E+03 | 1.4£+03 | 0.0E+00 | 1760.0 | 1152.0 | 153.6 | 307.2 | 6000.0 | 6800.0 | 6400.0 | 1512.0 | 208.0 |
| RANGE Factor | ; | : | 1 | ; | ; | 1 | ł | ; | : | 1 | : | 1 | 1 | 1 | 1 | : | : | : | : | ; | : | ; | : | ł | 1 | ; | ; | : | : |
| UNDA FREQ | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | M/M | N/A | N/N | K/A | N/A | N/A | A/A | N/N | A/A | N/A | N/A | N/A | N/A | N/A | N/N | K/A | N/A | N/A | N/A | N/A |
| RANGE FACTOR | 1.46+01 | 11.4E+01 | 1.46+01 | 10+38.8 i | 8.6E+01 | 10+38.8 i | 8.8€+01 | 8.8E+01 | 10+38.8 | 10+38.6 | 8.8E+01 | 8.85+01 | 2.0€+01 | 2.0E+01 | 1.0€+02 | 1.0E+02 | 1.0E+02 | 1.0€+02 | 1.0E+02 | 1.0€+02 | 1. JE +01 | 1.3E+01 | 1.36+01 | 1.35+01 | 1.3E+01 | 1.3E+01 | 1.3E+01 | 1.3E+01 | 1.3E+01 |
| TEAD Freq | 3.96-08 | 3.9E-08 | 3.96-08 | 7. IE-12 | 7. IE-12 | 7. IE-12 | 7. IE-12 | 7. IE-12 | 7. IE-12 | 7. HE-12 | 7. IE-12 | 7.1E-12 | 2.0E-07 | 2.0E-07 | 7.1E-12 | 7.15-12 | 7. JE-12 | 7. IE-12 | 7. IE-12 | 7. IE-12 | 3.3E-09 | 3. 36-09 | 3. 36-09 | 3. 3E-09 | 3.36-09 | 3. 3E -09 | 3. 3E-09 | 3.36-09 | 3.36-09 |
| RANGE FACTOR | : | : | 1 | ! | ł | : | ł | ; | ł | 1 | 1 | 1 | ł | 1 | : | 1 | ł | : | : | ł | : | : | : | : | i | ł | ; | 1 | ; |
| PUDA FRED | A/A | N/A | N/A | N/A | N/A | N/A | N/A | N/N | N/A | N/A | N/A | A/A | N/A | N/A | N/A | N/N | A/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| RANGE Factor | : | ; | : | : | 1 | ł | ; | : | : | : | : | ; | : | : | ; | : | : | : | : | : | : | : | 1 | : | : | ; | ł | 1 | : |
| PBA Freq | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/R | N/A | N/A | R/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| RAMBE Factor | ł | : | : | : | 1 | 1 | ł | 1 | ł | ; | ļ | : | 1 | : | : | : | ; | ; | ł | : | 1 | : | ł | : | ; | : | 1 | ; | : |
| NAAP Free | N/N | 8/N | A/A | A/M | N/A | N/A | A/N | N/A | N/A | A/A | N/A | N/A | N/A | N/A | A/A | N/A | A/A | N/A | N/A | N/A | N/A | N/A | N/A | K/A | N/A | N/A | N/A | N/A | ¥/A |
| RANSE FACTOR | ! | 1 | : | 1 | 1 | : | ; | : | : | : | : | : | : | : | : | 1 | 1 | : | ł | ł | 1 | ; | : | 1 | 1 | : | : | : | : |
| LBAD FREG | #/# | A/A | N/A | N/A | N/A | M/A | N/A | N/A | H/A | A/A | N/A | A/A | N/A | N/A | N/A | M/A | N/A | N/A | N/A | N/A | N/A | K/A | N/A | N/A | N/A | R/A | R/A | N/A | N/A |
| RANGE FACTOR | : | ; | : | 1 | ; | 1 | : | 1 | ; | : | : | : | ; | : | ; | 1 | : | 1 | 1 | 1 | : | ; | : | ; | ł | 1 | : | : | : |
| AP6 FRED | N/A | A/A | R/N | N/A | N/A | N/A | A/A | A/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | A/A | A/A | N/A | N/A | R/N | N/A | N/A | N/A | N/A | N/A | N/A | R/A | A/A |
| RANGE Factor | 1.46+01 | : | ; | 1.0E+32 | 1.0E+02 | 1.0E+02 | 1.0£+02 | 1.0€+02 | 1.0E+02 | 1.06402 | 1.0€+02 | : | 1.7E+01 | 10+37.1 | I | 1 | 1.15+02 | 1. 15+02 | : | 1 | | 1.0€+01 | 1.05+01 | 1.0€+01 | : | 1.05+01 | 1.0E+01 | 10+30.1 | 1.0E+01 |
| ANAD FREQ | 2.4E-09 | N/A | N/A | 4.26-13 | 4.2E-13 | 4.2E-13 | 4.2E-13 | 1.25-13 | 4.2E-13 | 1.26-13 | 4.25-13 | A/A | 1.25-06 | 1.26-06 | N/A | N/A | 4.25-13 | 4.2E-13 | N/A | N/A | N/A | 1. 15-06 | 1.16-06 | 1.1E-06 | N/A | 1. 1E-06 | 1.1E-06 | 1. IE-06 | 1.1E-04 |
| щ. | Ξ | = | = | 2 | 1 | ដ | 2 | 5 | 2 | 2 | 13 | Ľ | 9 | 9 | 11 | 12 | 2 | = | = | 12 | 2 | = | 2 | ± | = | = | = | 1 | = |
| SCEN- | VORVS | VCSVS | VDMGS | VIDHC | VOCEC | VOCHC | VONVC | VOPGC | VOPHC | 705407 | VOREC | VOOVE | VORGC | VORVE | VOBGC | VOKEC | VOKHC | VORVC | 24201 | 70460 | 10660 | 04003 | VOCGC | 79540 | VOLES | SHIGA | VOLVS | VORVC | VOPEC |

TARGARA DESERVATING SANDA DESERV

See notes at end of table.

.

÷.,

· · ·

1.1

File: ENGLTTEG.MF1 Fage & Cate 19-Aug-57

| COLLOCATION OFTIGN | IN CNSITE PACHAGE) | |
|--|--|-------------------------------|
| ONSITE TRANSPORTATION - ONSITE PACKAGE - | INCVEMENT FROM STOKAGE TO DEMIL FACILITY | Frequencies and Kange Factors |

| Ð | DURATION TIME | | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HKS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HRS | 2 HPS | 2 HKS | 2 HRS | 2 HRS |
|------------|--------------------|-----------------------|----------|---------|---------|----------|----------|----------|----------|----------|----------|----------|---------|---------|---------|----------|----------|----------|---------|----------------|---------|
| l Release | LBS. Emitted | | 2.4E-02 | 2.0E-04 | 5.3E-01 | 2.0E-04 | 5.36-01 | 2.0E-04 | 1 | 5.3E-01 | 1 | ; | ; | 1 | ; | 1 | 1 | 1 | ; | ł | : |
| lable and | LBS. Ietonated | | 1 | 1 | ł | : | : | : | ; | 1 | 6.0€+00 | 1.6E+00 | 3.2E+00 | 3.2E+01 | 6.5E+00 | 1.2E+01 | 6.0E+00 | 1.5€+01 | 1.56+01 | 2. IE+01 | 2.0E+01 |
| igent Avai | LBS. | | 1 | ! | ; | ; | 1 | ; | 5.2E-15 | 1 | 3.0E+01 | B. UE+00 | 10+39.1 | 1.5E+03 | 3.2E+01 | 5.8€+01 | 3.0£+01 | 7.35+01 | 7.36+01 | 6.2E+02 | 5.BE+02 |
| | AGENT AVALLABLE | 3 4 5 5 5 | 374.4 | 192.0 | 348.0 | 348.0 | 642.0 | 600.0 | 2712.0 | 1392.0 | 1152.0 | 153.6 | 307.2 | 1512.0 | 208.0 | 374.4 | 152.0 | 348.0 | 348.0 | 642.0 | 600.0 |
| | RANGE Factor | | ; | ; | 1 | : | ; | ł | : | ; | : | : | ; | : | 1 | ł | : | : | 1 | ł | : |
| | UNDA Freg | | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | K/A | N/A |
| | RANGE FACTOR | | 1.36+01 | 1.3E+01 | 1.36+01 | 1.JE+01 | 1.3E+01 | 1.36+01 | 1.3E+01 | 1.3E+01 | 5. 1E+01 | 5. IE+01 | 5.1E+01 | 5.16+01 | 5.16+01 | 5. 16+01 | 5. IE+01 | 5. IE+01 | 5.1E+01 | 5. IE +01 | 5.1E+01 |
| | TEAD Freq | | 3. 3E-09 | 3.36-09 | 3.36-09 | 3.36-09 | 3. 36-09 | 3. 3E-09 | 3. 3E-09 | 3. JE-09 | 2.26-09 | 2.2E-09 | 2.2E-09 | 2.2E-09 | 2.2E-09 | 2.26-09 | 2.2E-09 | 2.2E-09 | 2.26-09 | 2.2E-09 | 2.2E-09 |
| | RANGE FACTUR | | ; | ł | : | ; | ł | 1 | : | ; | 1 | : | ; | : | ; | : | 1 | ; | 1 | ł | ١ |
| | P104 | | N/A | N/A | N/A | N/A | N/A | K/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| | RANGE Factor | | 1 | ł | : | 1 | ł | ; | ; | : | : | : | : | : | : | 1 | ; | 1 | : | : | 1 |
| | F84 FRE0 | | N/A | N/A | K/A | N/A | A/A | N/A | N/A | K/A | N/A | A/A | N/A | M/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| | RANGE Factor | | 1 | : | : | : | 1 | ł | ; | : | : | 1 | : | : | ł | ; | ł | ; | : | : | 1 |
| :or 5 | NAAP Freg | | 1/N | A/A | K/A | N/A | N/A | N/A | A/N | A/A | A/A | R/N | N/A | A/M | N/A | N/A | N/A | N/A | N/A | N/A | A/A |
| Kange Facl | RANGE Factor | | : | ł | ; | ; | ; | ł | ; | ; | : | ł | : | ; | : | : | ; | : | ł | : | 1 |
| tes and | L BAD F R E G | | N/A | N/N | N/A | N/A | N/A | N/A | A/A | K/A | N/A | N/A | N/A | N/A | R/A | N/A | A/A | R/A | K/A | N/A | N/A |
| Frequenc | FANGE FACTOF | | ; | ; | ł | ł | ; | ! | ; | ; | ; | : | : | ; | ; | ; | : | ; | ; | ; | ; |
| SCENAL 10 | AP6 Free | | R/A | K/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | A/A | N/A | N/A |
| 0. | RANGE Factor | | 1.06+01 | 10+30-1 | 10+30-1 | 1 | 1.0E+01 | 1.0E+01 | ł | | 4.2E+01 | 4.26+01 | 4.26+01 | 4.25+01 | 4.2E+01 | 4.2E+01 | 4.2E+01 | 4.2E+01 | ; | 4.26+01 | 4.2E+01 |
| | ANA) FREE | | 1.16-06 | 1.15-06 | 1.16-06 | N/A | 1. IE-06 | 1.16-06 | K/A | N/A | 2.46-09 | 2.45-09 | 2.46-09 | 2.4E-09 | 2.46-09 | 2.4E-09 | 2.4E-09 | 2.4E-09 | £/M | 2.4E-09 | 2.4E-09 |
| | ż | • | 1 | 1 | 1 | . | = | | <u>"</u> | . | 2 | 13 | :: | 23 | 2 | 2 | 2 | 13 | 5 | 23 | 22 |
| | SCEN- AR 10 | | VOFHC | COPVC | V026C | VOCVC | VORGC | VDRVC | V05VS | VONEC | VODHC | VCCGC | VJCHC | VDRVC | VOPGC | VOPHC | JV90VC | VOREC | V00VC | VORSC | VICING |

У.

12.70

NDTES: 1. Scenarios 1–5 are per truck mile; scenarios 6–15 are per exposure year. 2. Duration time shown for scenarios with agent releases due to both detonations and spills is for spills only. Duration time for detonation is instantameous. 3. Scenarios 4 and 15 frequencies are multiplied by a factor to account for new undue force value.

200 D

CONCELENCE CONCERNENCE
للسلاما فلم

122222222

shita.

I-106



File: DMSITMEE.MA1 Page 1 Date 19-Aug-87

| 0110 | |
|----------------|------------|
| NSPOSAL | PACLAGE) |
| INT COLL | I DEFSLIE |
| ŧ | |
| - REGIONAL | A LOL B |
| 8 | 2 |
| TRANSPORTAT | (NOVENENT) |
| 311500 | |

| | | | | KENAL10 | Frequenc | in al | Range Fact | er s | | INVERC | | | | | ž | | | • | ient frai | lale ad | Ael ease | _ |
|---------|----|----------|---------|---------|----------|---------|------------|---------|----------|-------------|----------|---------|---------|-----------|-------------|------------|-----------------|-------------------|-----------------|-----------------------|-----------|-------------|
| -1130S | ź | | RANGE | 94V | RAISE | | RANGE | | RANGE | MB d | RANGE | PUDA | RANGE | 1EAB | MMER | Wen | ANGE . | AGENT MACHINE | LINS. | LINS. Etherator | LIS. | |
| | 1 | | | | | | | | | JMC | | | | | | | | | | | | |
| VIENCES | - | A/A | ; | A/A | ; | R/A | 1 | N/N | : | N/N | 1 | N/A | ; | 2.06-10 | 2.26+01 2 | . BE-10 2 | 10+32.1 | 2.4£+03 | 2. 2E+02 | ; | ; | 2 |
| VRBHS | - | 2.86-10 | 2.26+01 | N/A | : | N/A | { | N/N | : | A/A | ł | 2.06-10 | 2.26+01 | 2.06-10 | 2.26+01 | N/N | ł | 1.26+45 | 6.00+00 | 1 | ł | <u>چ</u> |
| VRCuis | ~ | 2.86-10 | 2.26+01 | A/A | : | N/A | ; | N/A | ; | N/N | : | N/A | : | 2.06-10 | 2.26+01 | V/ | ł | 4,46+02 | 1.45+00 | : | ; | 2 (19) |
| VRCHS | ~ | 2.06-10 | 2.26+01 | N/A | ; | N/A | : | N/A | : | N/A | ; | 2.86-10 | 2.26+01 | 2.86-10 | 2.26+01 | #/# | : | 9. ZE +02 | J. ZE+00 | : | ; | ¥ ~ |
| VAR65 | - | A/A | 1 | A/A | ; | A/A | 1 | N/N | : | A/A | : | N/A | ! | 2.86-10 | 2. ZE +01 | N/A | 1 | 3.0€+03 | 1.5£+03 | : | 1 | <u>چ</u> |
| VRKHS | ~ | 2.86-10 | 2.26+01 | 2.86-10 | 2. ZE+01 | N/N | { | N/A | : | 2.06-10 | 10+32°2 | N/A | ; | 2.86-10 | 2.ZE+01 | 2.06-10 2 | 10+32. | 3.4€+03 | 1.76+03 | ; | : | 2 19 |
| VRKVS | ~ | 2.81-10 | 2.2E+01 | N/A | ; | N/A | ; | 2.86-10 | 10+32.21 | A/A | : | A/A | : | 2.01-30.2 | 2.26+01 | A/A | : | 3.2£+03 | 1.46+03 | : | ţ | ₩ 2 |
| VRIIVS | ~ | 2.06-10 | 2.26+01 | N/A | : | N/A | ; | N/A | : | 2.06-10 | 10+32.2 | N/A | ; | 2.86-10 | 2.26+01 | 2.96-10 2 | 10+32.1 | 1. 1E+03 | 1.06+01 | ; | ; | <u>۲</u> |
| VRP6S | - | 2.86-10 | 2.26+01 | N/A | : | A/A | : | N/4 | i | N/A | : | ALA | : | 2.86-10 | 2.26+01 | . 86-10 2 | . 26+01 | 7.86 +02 | 4.36+00 | : | ; | 2 更 |
| VRPHS | 5 | 2.86-10 | 2.2E+UI | A.A | : | 2.86-10 | 2.25+01 | N/A | ; | R/A | ; | 2.8E-10 | 2.26+01 | 2.86-10 | 2.26+01 | N/A | : | 1.4€+03 | 1.2£+01 | : | ; | 2 180 |
| VAPVS | ~ | 2.06-10 | 2.2E+01 | 8/8 | : | 2.86-10 | 2.26+01 | N/A | : | N/N | : | N/A | ; | 2.86-10 | 2.26.01 2 | .86-10 2 | . ZE+01 | 7.26+02 | 4.0€+00 | 1 | ; | 2 180 |
| VRGGS | ~ | 2.86-10 | 2.2E+01 | N/A | ; | 2.86-10 | 2.26+01 | N/A | : | N/N | ł | N/A | : | 2.86-10 | 2.26+01 | 2.0E-10 2 | 10+32.1 | B. 7£+02 | 1.56+01 | : | ; | 2 第 で |
| VROVS | ~ | N/A | : | N/A | ; | N/A | { | N/A | : | N/A | : | A/A | ł | 2.86-10 | 2.2E+01 2 | .86-10 2 | 10+32. | 8 . 7E +02 | 1.5€+01 | : | ; | 2 HES |
| VRKGS | ~ | 2.86-10 | 2.26+01 | N/A | : | 1.86.10 | 2.2E+01 | N/A | ; | 2.86-10 | 2.2E+01 | N/A | { | 2.06-10 | 2. 2E+01 2 | 2.0E-10 2 | 10+32. | 6.4E+02 | 1.16+01 | : | ; | 2 #65 |
| VRRVS | ~ | 2.86-10 | 2.2E+01 | N/A | : | 2.86-10 | 2.2E+u1 | M/N | : | 2.06-10 | 12.2E+01 | N/A | : | 2.06-10 | 2. 10+35.5 | | .25+01 | 6. OE +02 | 10+30-1 | ; | ; | 2 19 |
| WRSVS | ~ | #/# | ; | 8/8 | : | 8/8 | : | N/N | : | N/N | : | N/A | ; | 2.96-10 | 2. 2E+01 | L. BE-10 2 | | 1.4E+03 | 1.46+03 | ; | ł | 2 19 |
| VRDHC | - | 3. 46-12 | 2.46+01 | M.A | ; | N/A | ; | N/A | : | N/A | : | 3.06-12 | 2.6€+01 | 3.06-12 | 2.6E+01 | W/W | 1 | 1.26+03 | ; | 2.4€+02 | 4.36+01 | So NI |
| VRCGC | - | 3.0E-12 | 2.66+01 | N/A | : | N/A | : | N/A | : | 8/8 | : | A/A | ; | 3.06-12 | 2.66+01 | N/N | : | 4.65+02 | : | 1. Z£ +02 | 3.56+01 | 20 MI |
| VICHC | - | 3.06-12 | 2.6E+01 | N.A | : | N/ N | : | N/A | 1 | A/A | : | 3.0E-12 | 2.66+01 | 3.06-12 | 2.6E+01 | N/N | ; | 9. ZE +02 | ; | 2. 3E + 02 | 3.56+01 | 2 |
| VRIIVC | + | 3. u£-12 | 2.6E+01 | #/# | : | A/A | 1 | N/A | : | 3.06-12 | 2.66+01 | M/A | 1 | 3.06-12 | 2.66+01 3 | 1.06-12 2 | 10+39" | 1, 16+03 | ; | 2. 8E +02 | 2. IE+01 | 20.01 |
| VRFGC | + | 3.06-12 | 2.6E+01 | N/R | ; | A/N | : | N/A | : | N/A | ; | A/A | 1 | 3.06-12 | 2.46+01 3 | 0E-12 2 | . b E+01 | 7.8€+02 | : | 2.0€+02 | 5.00+01 | 20 NJ |
| VRPHC | - | 3.06-12 | 2.6E+01 | N/A | : | 3.06-12 | 2.6E +UI | N/A | ; | N/A | : | 3.06-12 | 2.6E+01 | 3.06-12 | 2.66+01 | N/A | ; | 1.46+03 | ; | 3.5£+02 | 5. 36+01 | 2 |
| VRPWC | + | 3.06-12 | 2.66+01 | N/A | : | 3.06-12 | 2.6€+01 | N/A | : | N/N | : | M/A | ſ | 3.06-12 | 2.66+01 3 | 1.06-12 2 | 10+39" | 7.25+02 | , ; | 1.8€+02 | 1.4€+01 | 20 M |
| VRUEC | - | 3.06-12 | 2.66+01 | N/A | : | 3.0E-12 | 2.6E+U1 | N/A | ; | R/A | : | N/A | : | 3.06-12 | 2.66-01 | 06-12 2 | 10+39- | 8.76+02 | ; | 2.ZE+02 | 4.36.01 | 20 N |
| VRINC | • | M/A | ; | N/A | : | M/A | : | N/A | : | N/N | ł | M/A | : | 3.06-12 | 2.66+01 3 | i. 0E-12 2 | 10+39" | B . 7E+02 | : | 2. XE +02 | 1.45+01 | 20 MII |
| VARGC | - | 2.26-10 | 2.6E+01 | N/A | : | 2.26-10 | 2.6E+01 | N/A | 1 | 2.26-10 | 2.66+01 | W/W | : | 2.25-10 | 2.66+01 | 1.25-10 2 | 10+39" | 4.46+02 | ; | 1.46+02 | 4. BE +01 | 20 111 |
| VRAVC | - | 2.26-10 | 2.4E+01 | A/A | : | 2.2E-10 | 2.66+01 | N/A | 1 | 2.26-10 | 2.66+01 | N/A | ł | 2.25-10 | 2.46+01 2 | . 25-10 2 | 10+39- | 6. 0E+02 | ; | 1.56+02 | 1. IE+01 | ZO NE |
| VRBGS | -0 | N/A | ; | A/A | ; | N/A | : | N/A | : | N/A | ; | N/A | : | 1.66-10 | 2.06+01 | . 06-99 2 | 10+30" | 2.6£+03 | 2.26+03 | : | ; | ∰ ~ |
| VKDHC | 4 | 6.05-10 | 2.06+01 | M/A | ; | R/A | : | A/A | ; | A/A | : | 4.76-09 | 2.0€+01 | 01-34.1 | 2.06+01 | #/¥ | ; | 1.26+03 | B. 1E+02 | 1.7€+02 | ; | ₹ ¥ |

See notes at end of table.

ESTERNE SERVICE RECOMMENDED FOR THE FORM THE RECOMMENDED FOR THE FULL FOR THE FORM

.)

I-107

File: OMSITAE6.MALI Page 2 Date 19-Aug-87

| OPT 1 ONS | | |
|-------------------|-----------|--|
| DISPOSAL | PACKAGE) | |
| MATIONAL | I OFFSLIE | |
| Ĵ | = | |
| Ĩ | 8 | |
| REGU | 9 | |
| <u>.</u> | ł | |
| 2 | 2 | |
| TRANSPORTA | (NOVENENT | |
| 311SM | | |

Factor

Range

and

Scenario Frequencies

ł ż

3

Ngent Available and Released

BURATION TIME

| SU) | ENLITE | | ; | : | : | ; | : | : | : | : | : | | | : | : | : | 2.66+0 | 4. 32.0 | 3.5£+0 | 3.56+0 | 3.06+0 | 1.76+0 | B. UE+0 | 2.15+0 | 5.86+0 | 5.36+0 | 1.46+0 | 6.Xto | 1.66+0 | 4.BE+0 | 1.16+0 |
|---|--|---|---------------------------|---------------------------|-----------------|--|-------------------------------------|--|--------------------------|---------------------------------------|---------------------------------------|---------------------------------------|-----------------|---|---|-----------------|-----------------|----------------------------|----------------------------|------------------------|-----------------|--|--|------------------------------------|----------------------------|--|--|--|-----------------|--|--|
| LIS. | DETONATEL | | 6.9€+01 | 1.4€+02 | : | : | : | 1.7€+02 | 1.26+02 | 2. IE+02 | 1.1E+02 | 1.3€+02 | 1.3£+02 | 9.66+01 | 9.0£+01 | : | : | 2.9€+02 | 1.26+02 | 2. XE+02 | ; | : | ; | 2.6€+02 | 2.0€+02 | 3. SE +02 | 1.0€+02 | 2. XE +02 | 2.2£+02 | 1.46+02 | 1.56+02 |
| LIIS. | SPILLED | | 3.26+02 | 4.SE+02 | 2.6E+03 | 2.9€+03 | 2.7E+03 | 7.9€+02 | 5.5£+02 | 9.0€+02 | 5.0€+02 | 6. IE+02 | 6. 1E+02 | 4.56+02 | 4.26+02 | 1.26+03 | : | ; | ; | 1 | : | ; | : | ; | ` ; | ; | : | : | ; | 1 | ; |
| AGENT | AVAILABLE | | 4.6£+02 | 9.26+02 | 3.0€+03 | 3.4£+03 | J. ZE +03 | 1. IE+03 | 7.0€+02 | 1.4€+03 | 7.Z£+02 | B. 7E+02 | 8. 7E+02 | 6.4E+02 | 6. 0E+02 | 1.46+03 | 2.66+03 | 1.26+03 | 4.46+02 | 9.26+02 | 3.0£+03 | 3.46+03 | 3. ZE +03 | 1.16+03 | 1.0€+02 | 1.4£+03 | 7.26+02 | B. 7E+02 | 8.76+02 | 6.4E +02 | 6.0E+02 |
| RANGE | FACTOR | | ; | 1 | ł | 2.06+01 | : | 1 2.0€+01 | 2.06+01 | : | 1 2.06+01 | 3.06+01 | 1 2.06+01 | 2.06+01 | 3.06+01 | 3 2.06+01 |) 2.0E+01 | : | ; | : | : | 0 2.06+01 | : | 10+30.5 (| 0.2.0E+01 | : | 3.0€+01 | 0 2.06+01 | 10+30.2 | 0 2.0€+01 | 3 2.06+01 |
| MUN | FREG | | 8/8 | A/A | N/N | 1.05 05 | A/A | 1.06-09 | 1.06-09 | N/N | 1.06-09 | 1.66-05 | 1.06-09 | 1.06-01 | 1.06-09 | 1.06-0 | 8.56-10 | A/N | N/A | N/A | N/A | 8.56-10 | N/A | 8.56-10 | 0.5£-10 | N/N | 8.36-10 | 9.32-10 | 8.56-10 | 8.5E-1(| B. S 10 |
| RANGE | FACTOR | | 10 2.06+01 | 10 2.06+01 | 10 2.0E+01 | 10 2.0E+01 | 10 2.0€+01 | 10 2.0€+01 | 10 2.0€+01 | 10 2.06+01 | 10 2.06+01 | 10 2.0E+01 | 10 2.0€+01 | 10 2.0€+01 | 10 2.06+01 | 10 2.06+01 | 10 2.06+01 | 10 2.0E+u1 | 10 2.06+01 | 10 2.06+01 | 10 2.0E+01 | ·10 2.0E+01 | 10 2.06+01 | 10 2.06+01 | 10 2.0E+01 | 10 2.0€+01 | 10 2.0€+01 | 10 2.06+01 | 10 2.06+01 | 10 2.06+01 | 10 2.06+01 |
| TEAD | FREE | | - 9 | 1.66- | 1.46- | 1.66 | -39-1 | - 96-1 | 1.66- | -39.1 | - 96 - | -36- | - 96- | , K | -'8'-' | 1.66 | н. Н. Ж. | -1.15 | -x- | 1.8 | 1.36- | н. Ж | 1.35- | 1.35 | 1.36- | -1.8 | -X.1 | 3 | -32.1 | Ж | н. Н |
| RANGE | FACTOR | | : | 2.06+01 | ; | ł | ; | ; | ; | 2. UE+01 | ; | : | ; | : | : | : | : | 1 2.06+0 | ; | 2.06+0 | : | : | : | : | ; | 12.06+01 | ; | 1 | : | : | : |
| PUDA | FREG | | N/N | 4.76-09 | A/N | N/A | N/A | M/A | N/A | 1.76-05 | A/N | N/A | 6/R | A/A | A/A | ¥74 | N/N | 3.96-0 | N/N | 3.96-0 | A/A | A/H | A/A | A/A | N/A | 3.96-09 | K/A | N/N | R/3 | N/N | A/N |
| | | | | | | | | | | | | | | | | | | | | | | - | | _ | | | | | | | |
| RANGE | FACTOR | | ; | : | ; | 09 2.06+01 | : | 19 2.0€+01 | : | ; | : | ; | ; | J9 2.0E+01 | 19 2.0E+01 | ; | ; | : | ; | : | : | 10 2.06+0 | : | 10 2.0E+01 | : | ; | ; | ; | : | 10 2.0E+01 | 10 2.06+01 |
| PBA RANGE | FREG FACTOR | | N/A | W/W | N/A | 1.1E-09 2.0E+01 | N/N | 1.16-09 2.06+01 | N/A | ¥/N | N/N | N/A | W/W | 1. 1E-U9 2. 0E+01 | 1.16-09 2.06+01 | NIA | N/A | N/A | N/A | N/A | ¥/W | 9. IE-10 2. 0E+0 | V/# | 9. IE-10 2.0E+01 | N/A | W/W | N/A | W/M | V/N | 9. IE-10 2. 0E+01 | 9.1E-10 2.0E+01 |
| RANGE PBA RANGE | FACTOR FREQ FACTOR | / | W/W | W/W | 11/1 | 1.16-09 2.06+01 | 0 2.0E+01 N/A | 1.1E-09 2.0E+01 | N/A | N/N | N/A | ·· W/N ·· | W/W | I. IE-UP 2.0E+01 | 1.16-09 2.06+01 | 8/8 | N/A | W/W | N/A | 8/8 | W/W | 9. IE-10 2. 0E+0 | 0 2.0E+u1 N/A | 9, IE-10 2, 0E+01 | N/A | N/N | N/N | 4/# | W/N | 9.16-10 2.06+01 | 9.1E-10 2.0E+01 |
| MAAP RANGE PBA RANGE | FREG FACTOR FREG FACTOR | | N/A N/A | N/A N/A | N/A N/A | N/A 1.1E-09 2.0E+01 | 5.06-10 2.06+c1 N/A | N/A 1.16-09 2.06+01 | N/A N/A | N/A N/A | N/A N/A | N/A N/A | N/A N/A | N/A 1.1E-U9 2.0E+01 | N/A 1.1E-U9 2.0E+01 | N/A N/A | N/A N/A | N/A N/A | N/A N/A | N/A N/A | N/A N/A | N/A 9.16-10 2.06+0 | 4.1E-10 2.0E+01 N/A | N/A 9,1E-10 2,0E+01 | N/A N/A | W/A N/A | N/A N/A | N/A N/A | W/A W/A | N/A 9, IE-10 2, 0E+01 | M/A 9.1E-10 2.0E+01 |
| RANGE MAAP RANGE PBA RANGE | FACTOR FREG FACTOR FREG FACTOR | | M/A N/A | N/A N/A | N/A N/A | N/A 1.1E-09 2.0E+01 | 5.06-10 2.06+01 N/A | N/A 1.16-09 2.06+01 | N/A N/A | 2.0E+UI N/A N/A | 2.0E+01 N/A N/A | 2.0E+01 N/A N/A | N/A N/A | 2.06+01 N/A 1.16-09 2.06+01 | 2. JE+UI N/A 1. IE-U9 2. OE+01 | N/A N/A | N/A N/A | N/A N/A | N/A N/A | N/A N/A | N/A N/A | N/A 9.16-10 2.06+0 | 4,16-10 2.06+01 M/A | N/A 9, IE-10 2, 0E+01 | N/A N/A | 2.0E+U1 N/A N/A | 2.0E+01 N/A N/A | 2.0E+01 N/A N/A | N/A N/A | 2.0E+U1 N/A 9.1E-10 2.0E+U1 | 2.06+01 N/A 9.1E-10 2.0E+01 |
| LBAD RANGE MAAP RANGE PBA RANGE | FREG FACTOR FREG FACTOR FREG FACTOR | | 4/A 4/A 1/A | N/A N/A N/A | N/A N/A N/A | N/A N/A 1.1E-09 2.0E+01 | N/A 5,06-10 2,06+01 N/A | N/A N/A 1.16-09 2.06+01 | N/A N/A N/A | 2.7E-1V 2.0E+UI N/A N/A | 2.7E-10 2.0E+01 N/A N/A | 2.7E-10 2.0E+01 N/A N/A | N/A N/A N/A | 2.7E-10 2.0E+01 N/A 1.1E-09 2.0E+01 | 2.7E-10 2.JE+U1 N/A 1.1E-U9 2.0E+01 | NIA NIA NIA | N/A N/A N/A | N/A N/A N/A | N/A N/A N/A | M/A N/A N/A | N/A N/A N/A | N/A N/A 9.1E-10 2.0E+0 | N/A 4,16-10 2.06+01 N/A | N/A N/A 9.16-10 2.06+01 | N/A N/A N/A | 2.3E-10 2.0E+01 N/A N/A | 2.3E-10 2.0E+01 N/A N/A | 2.3E-10 2.0E+01 N/A N/A | N/A N/A N/A | 2.3E-10 2.0E+U1 N/A 9.1E-10 2.0E+U1 | 2.3E-10 2.0E+01 N/A 9.1E-10 2.0E+01 |
| KANGE LIBAD RANGE NAAP RANGE PBA RANGE | FACTOR FREG FACTOR FREG FACTOR FREG FACTOR | | #/A #/A #/A | N/A N/A N/A | N/A N/A N/A | 2.0E+01 N/A N/A 1.1E-09 2.0E+01 | N/A 5,06-10 2,06+01 N/A | M/A N/A 1.16-09 2.06+01 | N/A N/A N/A | 2.7E-1U 2.0E+UI N/A N/A | 2.7E-10 2.0E+01 N/A N/A | 2.76-10 2.06+01 N/A N/A | N/A N/A N/A | 2.7E-10 2.0E+01 N/A 1.1E-09 2.0E+01 | 2.7E-10 2.0E+U1 N/A 1.1E-U9 2.0E+01 | N/A N/A N/A | N/A N/A N/A | N/A N/A N/A | N/A N/A N/A | N/A N/A N/A | N/A N/A N/A | 2.06+01 N/A N/A 9.16-10 2.06+0 | N/A 4,16-10 2.06+01 N/A | N/A N/A 9,1E-10 2,0E+01 | N/A N/A N/A | 2,3E-10 2.0E+01 N/A N/A | 2.3E-10 2.0E+01 N/A N/A | 2,3E-10 2,0E+01 N/A N/A | N/A N/A N/A | 2.3E-10 2.0E+UI N/A 9.1E-10 2.0E+UI | 2.36-10 2.06+01 N/A 9.1E-10 2.06+01 |
| APG KANGE LUAD RANGE NAAP RANGE PUA RANGE | FREG FACTOR FREG FACTOR FREG FACTOR | | N/A N/A N/A N/A | N/A N/A N/A N/A | N/A N/A N/A N/A | 7.2E-08 2.0E+01 N/A N/A 1.1E-09 2.0E+01 | N/A N/A 5,06-10 2,06+01 N/A | N/A N/A N/A 1.16-09 2.06+01 | N/A N/A N/A N/A | N/A 2,7E-1v 2.0E+u1 N/A N/A | N/A 2.7E-10 2.0E+01 N/A N/A | N/A 2.76-10 2.06+01 N/A N/A | B.A B/A B/A B/A | M/A 2.7E-10 2.0E+01 N/A 1.1E-U9 2.0E+01 | N/A 2.7E-10 2.0E+U1 N/A 1.1E-09 2.0E+01 | N/A N/A N/A N/A | N/A N/A N/A N/A | N/A N/A N/A N/A | N/A N/A N/A N/A | N/A N/A N/A N/A | N,A N/A N/A N/A | 5.9E-UB 2.0E+01 N/A 9.1E-10 2.0E+0 | N/A N/A 4,16-10 2,06+01 N/A | N/A N/A N/A 9.1E-10 2.0E+01 | N/A N/A N/A N/A | N/A 2.3E-10 2.0E+01 N/A N/A | N/A 2.3E-10 2.0E+01 N/A N/A | N/A 2.3E-10 2.0E+01 N/A N/A | N/A N/A N/A N/A | N/A 2.3E-10 2.0E+U1 N/A 9.1E-10 2.0E+U1 | N/A 2.36-10 2.06+01 N/A 9.16-10 2.06+01 |
| RANGE APG KANGE LBAD RANGE MAAP RANGE PBA RANGE | FACTOR FREE FACTOR FREE FACTOR FREE FACTOR FREE FACTOR | | 0 2.06+01 M/A M/A M/A M/A | 0.2.uE+01 N/A N/A N/A N/A | N/A N/A N/A N/A | 10 2.0E+01 7.2E-08 2.0E+01 N/A N/A 1.1E-09 2.0E+01 | U 2.0E+01 N/A N/A 5.0E+U 2.0E+U N/A | 10 2.06+01 M/A N/A N/A 1.16-09 2.06+01 | 02.06+01 N/A N/A N/A N/A | 0 2.0E+U1 N/A 2.7E-1U 2.0E+U1 N/A N/A | U 2.UE+01 N/A 2.7E-10 2.0E+01 N/A N/A | 0 2.UE+01 N/A 2.7E-19 2.0E+01 N/A N/A | N/A N/A N/A | U 2.0E+01 M/A 2.7E-1U 2.0E+01 M/A 1.1E-U9 2.0E+01 | U 2. UE+UI N/A 2.7E-IU 2. UE+UI N/A 1.1E-U9 2.0E+01 | N/A N/A N/A N/A | N/A N/A N/A N/A | 10.2.0E+01 N/A N/A N/A N/A | 10 2.0E+01 N/A N/A N/A N/A | 10 2.0E+UI N/A N/A N/A | N/A N/A N/A N/A | N 2.0E+01 5.%E-UB 2.0E+01 N/A 9.1E-10 2.0E+0 | 10 2.0E+01 N/A N/A 4.1E-10 2.0E+01 N/A | 10 2.uE+01 N/A N/A 9.1E-10 2.0E+01 | .u 2.uE+01 N/A N/A N/A N/A | 10 2.UE+01 N/A 2.3E-10 2.0E+U1 N/A N/A | 10 2.0E+01 N/A 2.3E-10 2.0E+01 N/A N/A | 10 2.0E+01 N/A 2.3E-10 2.0E+01 N/A N/A | N/A N/A N/A N/A | 10 2.0E+UI M/A 2.3E-10 2.0E+UI N/A 9.1E-10 2.0E+UI | 10 2.0E+01 N/A 2.3E-10 2.0E+01 N/A 9.1E-10 2.0E+01 |

notes at end of table.

ž

1 15 20 11 15 20 11 15 1 15 20 11 15 20

2.66.92 4.35.01 3.56.91 3.56.91 3.56.91 3.56.91 1.76.92 1.76.91 5.56.91 5.56.91 1.56.91 1.66.91 1.66.91 1.16.91 1.16.91



0



|--|

| M/A 1.3 |
|-------------------|
| |
| N/A |
| 4/H 4/H 4/H |
| |
| |

See otes at end of table.

e ...

7

File: OMSITAE6.MK1 Page 4 Date 19-Aug-87

ONSITE TRAMISPORTATION - REGIONAL AND MATIONAL DISPOSAL OPTIONS (NOVENENI TO AND FROM RAIL IN OFFSITE PACKAGE)

| • |
|----------|
| |
| |
| • |
| _ |
| |
| |
| - |
| 3 |
| 2 |
| |
| • |
| <u> </u> |
| - |
| - |
| - |
| - |
| 2 |
| - |
| - |
| - |
| 2 |
| 5 |
| ÷. |
| - |
| |
| |
| |

1.010.

appendi serveran andresser appendenti serversa anternas V

| <u>e</u> |
|-----------------------|
| K/A K/A 1.1E-09 |
| |
| |
| |
| ; |
| |

See notes at end of table.

•

I-110





| | | | RATION 11ME | 5 EF | |
|----------|-----------------------|-----------------------|------------------|------------------------|---|
| | | el eased | LBS. DU | : : | |
| | | | LDS. Mater Ei | E+01 E+01 | |
| | | t Availad | LDS. | E+02 2.1 | |
| | | unday. | | +02 6.2 | |
| | | | WALLA . | 44 | |
| | | | RANGE FACTOR | 0 5.0£+01 | |
| | | 1 Der S | | 1-34-1 1-34-1 | |
| | | POSAL OFT | FACTOR | 9 5.06 +01 | |
| | | SILE PAC | | 2. X -0 | |
| | | AND NATI | FACTOR | : : | |
| c∂x ● | | REGIONNI. FROM RA | FUDA | 8/8 8/8 | e value. |
| · · · | | ACION - AT TO AND | RANGE FACTOR | 5.56-01 | adue for c |
| | | FRANSPOR) (HOVENE) | FIED | 3. 7E-10 5. 7E-10 | attons tartons tareus. or new u |
| | | 045116 | RANGE FACTOR | : : | uure year 15 linstand 2004 440 2004 4 |
| | | tors | | N/A N/A | ere to to to to to to to to to to to to to |
| | | sey abues | RANGE FACTOR | 5. 56 eul | - 15 - 15 - 16 - 161 - 16 - 161 - 16 - 16 - 16 - |
| | | la si | L DAD | 3. 76-10 3. 76-10 | emarics (agent tron tran 15 M/A. uitiplice |
| | (#- 87 | Fr equenc | FACTOR | : ; | arte; st beros at beros at tis anno es are a |
| | bite 1 1 4 | Scenar 10 | AP6 FREQ | 8/8 8/8 | er truch pills out frequenci |
| | | | RANGE FACTOR | 5.56 •01 5.56 •01 | e she y mout to for s sposal 0 f and 15 |
| | | | ANN DI LI | 3. 7E -10 3. 7E -10 | marios 1 ration ti spilta tional Di marios 4 |
| | : QUELITA | | 9 | 52 | 25 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 |
| | 516 2 | | | VRRV | |
| \$ | | | | | |
| | | | | | |
| | | | I - | 111 | |

マテン

1.7.1.5.6

File: ONSITBRG.WK1 Page 1 Date 19-Aug-87

ONSITE TRANSPORTATION - AIR

Scenario Frequencies and Range Factors

Agent Available and Released

| | | AP6 | RANGE | LBAD | RANGE | TEAD | RANGE | AGENT | LBS. | LBS. | LBS. | DURATION |
|----------|-----|---------|---------|---------|---------|---------|---------|---------|---------|-----------|---------|----------|
| SCENARIO | NO. | FFEQ | FACTOR | FREQ | FACTOR | FREQ | FACTOR | AVAIL | SPILLED | DETONATED | EMITTED | TIME |
| | | | | | | | | | | | | |
| VAKHS | 1 | 0.0E+00 | | N/A | | 0.0E+00 | | 3.4E+03 | | | | |
| VAPHS | i | N/A | | 0.0E+00 | | 0.0E+00 | | 1.4E+03 | | | | |
| VAPVS | 1 | N/A | | 0.0E+00 | | 0.0E+00 | | 7.6E+02 | | | | |
| VAGGS | 1 | N/A | | 0.0E+00 | | 0.0E+00 | | 8.7E+02 | | | | |
| VAR6S | 1 | N/A | | 0.0E+00 | | 0.0E+00 | | 6.5E+02 | | | | |
| VARVS | 1 | N/A | | 0.0E+00 | | 0.0E+00 | | 6.1E+02 | | | | |
| VANHS | 2 | 0.0E+00 | | N/A | | 0.0E+00 | | 3.4E+03 | | | | |
| VAPHS | 2 | N/A | | 0.0E+00 | | 0.0E+00 | | 1.4E+03 | | / | | |
| VAPVS | 2 | N/A | | 0.0E+00 | | 0.0E+00 | | 7.6E+02 | | | | |
| VAQGS | 2 | N/A | | 0.0E+00 | | 0.0E+00 | | 8.7E+02 | | | | |
| VARGS | 2 | N/A | | 0.0E+00 | | 0.0E+00 | | 6.5E702 | | | | |
| VARVS | 2 | N/A | | 0.0E+00 | | 0.0E+00 | | 6.1E+02 | | | | |
| VANHS | 3 | 2.8E-10 | 2.2E+01 | N/A | | 2.9E-10 | 2.2E+01 | 3.4E+03 | 1.7E+03 | | | 2 HRS |
| VAPHS | 3 | N/A | | 2.8E-10 | 2.2E+01 | 2.8E-10 | 2.2E+01 | 1.4E+03 | 1.2E+01 | | | 2 HRS |
| VAPVS | 3 | N/A | | 2.8E-10 | 2.2E+01 | 2.8E-10 | 2.2E+01 | 7.2E+02 | 6.0E+00 | | | 2 HRS |
| VAQGS | 3 | N/A | | 2.8E-10 | 2.2E+01 | 2.8E-10 | 2.2E+01 | 8.7E+02 | 1.5E+01 | | | 2 HRS |
| VAFBS | 3 | NZA | | 2.8E-10 | 2.2E+01 | 2.8E-10 | 2.2E+01 | 6.4E+02 | 1.1E+01 | | | 2 HRS |
| VARVS | 3 | N/A | | 2.8E-10 | 2.2E+01 | 2.8E-10 | 2.25+01 | 6.0E+02 | 1.0E+01 | | | 2 HRS |
| VAPHC | 4 | N/A | | 3.0E-12 | 2.6E+01 | 3.0E-12 | 2.6E+01 | 1.4E+03 | | 3.5E+02 | 5.3E+01 | 20 MIN |
| VAPVC | 4 | N/A | | 3.0E-12 | 2.6E+01 | 3.0E-12 | 2.6E+01 | 7.2E+02 | | 1.8E+02 | 1.4E+01 | 20 MIN |
| VH05C | 4 | N/A | | 3.0E-12 | 2.6E+01 | 3.0E-12 | 2.6E+01 | 8.7E+02 | | 2.2E+02 | 6.5E+01 | 20 MIN |
| VARGC | 4 | N/A | | 2.2E-10 | 2.6E+01 | 2.2E-10 | 2.6E+01 | 6.4E+02 | | 1.6E+02 | 4.8E+01 | 20 MIN |
| VASVC | 4 | N/A | | 2.2E-10 | 2.6E+01 | 2.2E-10 | 2.6E+01 | 6.0E+02 | | 1.5E+02 | 1.1E+01 | 20 MIN |
| VANHE | 5 | 0.0E+00 | | N/A | | 0.0E+00 | | 3.4E+03 | | | | |
| VHEHS | 6 | 7.2E-08 | 2.0E+01 | N/A | | 4.8E-09 | 2.0E+01 | 3.4E+03 | 2.9E+03 | | | 2 HR |
| V423C | 6 | A/K | | 2.7E-10 | 2.0E+01 | 4.8E-09 | 2.0E+01 | 1.4E+03 | 9.8E+02 | 2.1E+02 | | 2 HR |
| VAFVE | ć | N/A | | 2.7E-10 | 2.0E+01 | 4.8E-09 | 2.0E+01 | 7.2E+02 | 5.0E+02 | 1.1E+02 | | 2 HR |
| VADGC | 6 | N/A | | 2.7E-10 | 2.0E+01 | 4.8E-09 | 2.0E+01 | 8.7E+02 | 6.1E+02 | 1.3E+02 | | 2 HR |
| VARGE | 6 | N/A | | 2.7E-10 | 2.0E+01 | 4.8E-09 | 2.0E+01 | 6.4E+02 | 4.5E+02 | 9.6E+01 | | 2 HR |

File: ONSITBRG.WK1 Page 2 Date 19-Aug-07

ONSITE TRANSFORTATION - AIR

Scenario Frequencies and Range Factors

| SCENARIO | NO. | APG Freq | RANGE Factor | LBAD Freq | RANGE Factor | TEAD FREQ | RANGE Factor | AGENT AVAIL | LBS. Spilled | L9S. DETONATED | LBS. EMITTED | DURATION Tine |
|----------|----------|-------------|-----------------|----------------|-----------------|--------------|-----------------|----------------|-----------------|-------------------|-----------------|------------------|
| USEUC | | N/A | | 2 75-10 | 2 0E+01 | 4 8F-09 | 2 0E+01 | 6 0E+02 | 4 2F+02 | 9 0E+01 | | 2 118 |
| UALUE | 7 | 5 0C_AD | 2 05201 | 1.7C IV | 2.02.01 | 4.0C 07 | 2.02.01 | 7 AE+07 | | | 1 75+02 | 1 48 |
| VADUC | , | J.76-VD | 2.001 | 0/8 5.75-10 | 5 (EL01 | 1 0E_00 | 2.05.01 | 1 46403 | | 7 55+02 | 5 35101 | 20 MIN |
| VACUE | , | N7 M | | 2.35-10 | 2.00101 | 4.VE-07 | 2.001 | 1.96103 | | 1 0E+02 | 1 ACINI | 20 NIN 20 NIN |
| VARYL | 1 | N/H | | 2.35-10 | 2.00101 | 4.00-01 | 2.00701 | 1.20192 | | 1.00102 | 1.46701 | 29 NIN 20 MIN |
| VHUDL | 1 | N/A | | 2.35-10 | Z.UETUI | 4.05-07 | 2.00101 | 0./2702 | | 2.20702 | 0.JETUI | 20 MIN |
| VANGE | <u>′</u> | N/A | | 2.3E-10 | 2.02+01 | 4.02-09 | 2.06+01 | 6.4E+U2 | | 1.65+02 | 4.8E+01 | 20 min |
| VARVE | / | N/A | | 2.3E-10 | 2.0E+01 | 4.0E-09 | 2.0E+01 | 6.0E+02 | | 1.56+02 | 1.16+01 | 20 MIN |
| VARHS | 9 | 0.0E+00 | | N/A | | 0.08+00 | | 3.4E+U3 | | | | |
| VAFHS | 9 | N/A | | 0.0E+00 | | 0.0E+00 | | 1.4E+03 | | | | |
| VAPVS | 9 | N/A | | 0.0E+00 | | 0.0E+00 | | 7.2E+02 | | | | |
| VAGGS | 9 | N/A | | 0.úE+00 | | 0.0E+00 | | 8.7E+02 | | | | |
| VARGS | 9 | N/A | | 0.0E+00 | | 0.0E+00 | | 6.4E+02 | | | | |
| VARVS | 9 | N/A | | 0.0E+00 | | 0.0E+00 | | 6.0E+02 | | | | |
| VAEHS | 10 | 0.0E+00 | | N/A | | 0.0E+00 | | 3.4E+03 | | | | |
| VAPHS | 16 | N/A | | 0.0E+00 | | 0.0E+00 | | 1.4E+03 | | | | |
| VAPVS | 10 | N/A | | 0.0E+00 | | 0.0E+00 | | 7.2E+02 | | | | |
| VAQGS | 10 | N/A | | 0.0E+00 | | 0.0E+00 | | 8.7E+02 | | | | |
| VARGS | 10 | N/A | | 0.0E+00 | | 0.0E+00 | | 6.4E+02 | | | | |
| VARVS | 10 | N/A | | 0.0E+00 | | 0.0E+00 | | 6.0E+02 | | | | |
| VAFHS | 11 | 1.2E-08 | 2.2E+01 | N/A | | 2.08-07 | 2.2E+01 | 3.4E+03 | 1.7E+03 | | | 6 HRS |
| VAPHS | 11 | N/A | | 1.2E-08 | 2.2E+01 | 2.0E-07 | 2.2E+01 | 1.4E+03 | 1.2E+01 | | | 6 HRS |
| VAPVS | 11 | N/A | | 1.2E-08 | 2.2E+01 | 2.0E-07 | 2.2E+01 | 7.2E+02 | 6.0E+00 | | | 6 HRS |
| VARSS | 11 | N/A | | 1.2E-08 | 2.25+01 | 2.0E-07 | 2.2E+01 | 8.7E+02 | 1.5E+01 | | | A HRS |
| VARGS | 11 | N/A | | 1. 2E-08 | 2.2E+01 | 2.0E-07 | 2.2E+01 | 6 4E+02 | 1.1E+01 | | | A HRS |
| VARVS | 11 | N/A | | 1 2E-08 | 2 2E+01 | 2 0E-07 | 2 2E+01 | 5 0E+02 | 1 05+01 | | | 4 HPS |
| VAPHE | 12 | N/A | | C 0E+00 | | 0 05+00 | | 1 4E+03 | | 3 5E+02 | 5 35101 | 20 MTM |
| VEPUE | 17 | N/4 | | 0.0E+00 | | 0.05+00 | | 7 05+60 | | 1 BE+02 | 1 45+01 | 20 MIN |
| V4080 | 12 | N/A | | 0.6E+00 | | 0.05+00 | | 9.7E+02 | | 2 25+02 | A 50401 | 20 111 |
| VARGE | 17 | N/A | | 9 KE-09 | 2 06+01 | 1 AE-07 | 2 0E+01 | 6 4E+02 | | 1 45+02 | 4 8E+01 | 20 MIN |

Agent Available and Released

File: ONSITBRG.WK1 Page 1 Date 19-Aug-87

ONSITE TRANSFORTATION - AIR

Scenario Frequencies and Range Factors

AP6 RANGE LBAD RANGE TEAD RANGE AGENT LBS. LBS. LBS. DURATION SCENARIO NO. FREQ FACTOR FREQ FACTOR FREQ FACTOR AVAIL SFILLED DETONATED EMITTED TIME VARVC N/A 9.6E-09 2.0E+01 1.6E-07 2.0E+01 6.0E+02 12 --1.5E+02 1.1E+01 20 MIN ---VANHE 13 0.0E+00 --N/A --0.0E+00 --3.4E+03 --1.7E+02 1 HR --VANHC 14 1.1E-09 2.4E+01 N/A --6.8E-11 2.5E+01 3,4E+03 ----2.5E-01 2 HR VAPHC ---2 HR 14 N/A --2.1E-08 2.5E+01 6.8E-11 2.5E+01 1.4E+03 --2.5E-01 VAPVC 6.8E-11 2.5E+01 7.2E+02 ----2.1E-03 14 N/A --2.1E-08 2.5E+01 2 HR VAGGC 14 N/A 2.1E-08 2.5E+01 6.8E-11 2.5E+01 8.7E+02 ----5.58+00 2 HB --VARGC 14 N/A 2.1E-08 2.5E+01 6.8E-11 2.5E+01 6.4E+02 ----5.5E+00 2 HR --VARVC ----2 HR 14 2.1E-08 2.5E+01 6.8E-11 2.5E+01 6.0E+02 2.1E-03 N/A --V4PHC 3.7E-10 5.5E+01 2.2E-09 5.8E+01 1.4E+03 5.8E+01 1.2E+01 2 HR 15 N/A ------VARVE 15 N/A --3.7E-10 5.5E+01 2.2E-09 5.8E+01 7.2E+02 3.0E+01 6.0E+00 --2 HR VAQGC 15 N/A --3.7E-10 5.5E+01 2.2E-09 5.8E+01 8.7E+02 7.3E+01 1.5E+01 --2 HR VARGE 15 N/A --3.7E-10 5.5E+01 2.2E-09 5.8E+01 6.4E+02 6.2E+02 2.1E+01 --2 HR VARVC 15 N/A 3.7E-10 5.5E+01 2.2E-09 5.8E+01 6.0E+02 5.8E+02 2.0E+01 --2 HR --

Agent Available and Released

12:22:22

I-114



I.1.6. OFFSITE TRANSPORT

The following tables list the accident results for offsite transport of munitions.







Fage 1 Date 20-Aug-87

OFFSITE TRANSPORTATION - REGIONAL DISPOSAL OPTION

Accident Frequencies and Range Factors

Agent Available and Released

| IRAT ION TIME | | 6 HRS | 6 HRS | : | 6 HKS | ; | 6 HKS | 9 HKS | 6 HES | 6 HKS | 6 HRS | 6 HKS | 6 HKS | 6 HRS | S3H 9 | 6 HRS | ł | 0 ein | ! | O ain | O BIN | 0 0 10 | 0 e la | 0 BIN | UTU () | 0 BLD | 1) B LF | Ú 810 | 2 HRS | ; |
|--------------------------|---|---------|----------|--------|---------|-------|--------------|----------|--------------|--------|---------------|----------|----------|---------|----------|------------|---------|----------|-------|----------|--------------|---------------|---------------|----------|-----------|----------|----------------|----------|---------|-------------|
| LBS. DA | | ; | ; | : | ; | ł | : | 1 | ; | 1 | ; | 1 | : | ; | ; | ; | 1 | 3E+01 | 1 | 5E+01 | 16+01 | E +01 | 36+01 2 | 4E+01 2 | 5E+01 2 | 96+01 | 10+38 | 10+31 | .6E+02 | : |
| LBS. DNATED EP | | : | : | 1 | 1 | : | ; | ; | : | ; | : | ; | ; | : | 1 | ; | ; | 9E+02 4. | ; | 3E+02 3. | BE+02 2. | 0E+02 5. | 5E+02 5. | BE+02 1. | 2E+02 6. | 2E+02 1. | 6E+02 4. | 5E+02 1. | 5. | ; |
| LBS. | | 2E+02 | 06+00 | ł | 2E+00 | : | 7E+03 | BE+03 | 10+30 | 5E+00 | 2E+01 | 00+30 | 10+3S | 2E+01 | 10+31 | 10+30 | : | : | ! | | ; ; | 2 | ··· | | | 2. | ; | | : | 1 |
| SENT SENT SELE SPI | | :+03 2. | E+03 6.0 | +02 | :+03 3. | 103 | :+03 I. | E+03 1.1 | E+03 1.0 | 103 b. | E+03 L. | :+03 6.1 | :+03 1. | 103 1.1 | 1.1 20+3 | +03 1.6 | 103 | •03 | +02 | 101 | · •01 | | 103 | •03 | · · · · · | · . | •01 | +01 | | 101 |
| R AVAIL | | 5.3 | 2.3 | 9.26 | 1.8 | 6.0 | 6.8 | 1.2 | 2.3 | 1.61 | 2.8 | 1.46 | 1.7 | 1.76 | 1.36 | 1.26 | 2.7 | 2.38 | 9.26 | 1.8 | 2.3 | 1.6 | 2.8 | 1.46 | 1.7 | 1.76 | 1.3 | 1.26 | 5.3 | 6.Q |
| FACTO | | 2 10 | 1 | 1 | : | ł | 9 10 | : | e S | 9 | ł | 2 | 9 | 2 | 2 | 2 | 50 | : | 1 | ; | 21 | 12 (| ! | 17 (| 12 (| | 21 | 12 | ÷ | : |
| UNDA Freq. | | 3.26-0 | N/A | N/A | N/A | N/A | 3.2E-0 | N/A | 3.2E-0 | 3.2E-0 | N/A | 3.25-0 | 3.2E-0 | 3.2E-0' | 3.2E-0 | 3.26-0 | 3.2E-0 | N/A | N/A | N/A | 4.46-1 | 4. IE-10 | N/N | 4.46-1(| 4.46-11 | 4.46-10 | 1.55-10 | 4.56-10 | 3.56-10 | N/A |
| RANGE FACTOR | | ; | ł | ł | 1 | 1 | ł | ł | ! | ! | ł | ł | ; | ! | ł | ł | ! | ; | ł | ł | ł | ł | ; | ; | 1 | ! | 1 | ł | 1 | ! |
| TEAD Freg. | | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| RANGE FACTOR | | ł | 2 | ł | 01 | 1 | ł | ł | ; | ; | 01 | ł | ł | ł | ; | : | ł | 23 | ł | 23 | : | ; | 27 | 1 | ; | ł | ; | 1 | ; | : |
| PUDA Freq. | | 8/N | 3.2E-09 | N/A | 3.2E-09 | N/A | N/A | N/A | A/A | N/A | 3.2E-09 | N/A | N/A | N/A | N/A | N/A | N/A | 4.4E-10 | N/A | 4.4E-10 | N/A | N/A | 4.4E-10 | N/A | e/N | 8/N | N/A | A/A | N/A | N/A |
| RANGE FACTOR | | ; | 1 | ł | ł | ł | 01 | 1 | 2 | 1 | ł | ; | ; | : | 01 | 2 | ł | ł | ł | : | 21 | ł | ; | : | ; | ; | 21 | 21 | ! | : |
| PBA Freq. | | N/A | N/A | N/A | N/A | N/A | 3.26-09 | N/A | 3. ZE-09 | N/A | A/A | N/A | N/A | N/A | 3.2E-ù9 | 3. 2E-09 | N/A | N/A | N/A | N/A | 4.4E-10 | N/A | N/A | 8/N | N/A | E/N | 1.56-10 | 1.56-10 | N/A | N/A |
| RANGE Factor | | : | ; | ; | ; | ; | 1 | 01 | 1 | ł | 1 | ; | ł | : | ; | ; | ; | : | ; | ; | ; | ; | 1 | : | : | ł | : | 1 | ; | : |
| KAAP Freq. | | N/A | N/A | N/A | N/A | N/A | N/A | 3.26-09 | A/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | R'.N | N/A | A/A | N/A | N/A | A/A | N/A | N/A | N/A |
| RANGE FACTOR | | ł | ; | 1 | 1 | 1 | 1 | ł | ł | ł | 01 | 01 | 01 | : | 2 | 01 | : | ; | ł | 1 | ł | | 21 | 27 | 51 | ; | 21 | 51 | 1 | ; |
| LBAD Free. | | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | 3.2E-09 | 3.2E-09 | 3, 2E-ù9 | N/A | 3.2E-09 | 3.2E-09 | N/A | N/A | N/A | N/A | N/A | N/A | 4.4E-10 | 4.46-10 | 4.46-10 | N/A | 4.5E-10 | 4.5E~10 | e/N | A/A |
| RANGE FACTOR | | ; | ł | ł | ! | ; | 2 | ! | ; | ١ | 1 | ł | 1 | : | : | ; | 1 | ; | ł | ł | : | 1 | : | ; | ! | ! | ; | : | : | ; |
| AFG Freg. | | M/A | A/A | M/N | N/A | N/A | 3.26-09 | M/N | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/N | N/A | N/A | N/A | N/A | R/N | N/G | A/A | N/A | N/A | N/A | N/A | N/A | A/A |
| KÄNGE Factor | | ł | : | : | : | ; | ; | ł | ; | 1 | ; | ; | ; | ; | : | ; | ; | ł | ; | ; | ! | ! | : | ; | 1 | 1 | : | ; | : | ; |
| ANAD Freq. | | R/A | N/A | N/A | N/A | A.A | N/A | A/N | 8/K | H/H | 8/F | 8/8 | 8/ÿ | R/N | ₩/₩ | A/A | N/A | R/N | N/A | 4/H | N/A | H/H | 4. H | R/A | N/A | A/A | ₩/Ĥ | N/Ĥ | N/A | Ч/ н |
| Ę. | ł | ~ | *7 | *7 | *7 | A-) | *7 | ~ | r 7 | • • | ~ | *7 | 57 | *") | ~ | ר י | • 7 | 4 | + | 7 | - | + | + | - | - | - | - | v | 67 | 5 |
| SCEN- AKIÙ | | RC 855 | REDHS | KCL 65 | RCCHS | 60103 | FCNHS | KC; vš | EURVS | £CF65 | FICFHS | KLF4E | FLUGS | 6:0:3 | FCR65 | ECE.VS | r (CS's | FL CHU | FC.GC | FCCHC | FCRVC | FCF GL | RCFHC | FCF VC | 96034 | 35834 | RCRBU | ACFVC | -itef | F.C. GF |

101 101

See notes at end of table.

I-117

Fage 2 Date 20-Aug-87

OFFSITE TRANSFORTATION - REGIONAL DISPOSAL OPTION

| | | Î |
|--|--|---|
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |

| | | | | | | | 4 | iccident | Fr equea | cies and | ƙange | Factors | | | | | | - | Agent Ava | ilable and | l Keleáser | _ |
|---------------|----------|---------------|-----------------|--------------|-----------------|---------------|-----------------|---------------|-----------------|----------------|----------------|---------------|-----------------|---------------|-----------------|---------------|-------------------|-------------------|-----------|-------------------|-------------------|------------------|
| SCEN- | ND. | ANAD FREQ. | RANGE Factor | APG Freq. | RANGE FACTOR | LBAD Fred. | RANGE FACTOR | NAAP Freg. | RANGE Factor | PBA Freu. 1 | KANGE ACTOR | FUDA FRED. | RANGE Factor | TEAD Freq. | RANGE Factor | UNDA FREQ. | RANGE Factor # | AGENT VAILASLE | LBS. | LBS. DETONATED | LBS. (Emitted | IURATION Time |
| | | | | | : | | | | | | ! | | | • | | ł | ; | | | | | 5311 |
| KOAF | u" | A/A | ; | 3.56-10 | - | A/M | : | N/A | 1 | 3.56-10 | 1 | N/N | ł | A/A | ; | S. 3E-10 | ¥ | 6. BL +US | ; | : | 1. /±+02 | Z HKS |
| PCi VF | υJ | R/A | ! | N/A | : | N/A | ; | 3.56-10 | 47 | N/A | ł | N/A | ! | N/A | ļ | N/A | ! | 7.2E+03 | 1 | ł | 9.úE+ù1 | 2 HKS |
| êÛS.F | чŋ | e/N | 1 | N/A | : | N/A | ł | N/A | ł | N/A | ! | N/A | ; | N/A | ł | 3.56-10 | 4 | 2.76+03 | : | : | 3.46+01 | 2 HKS |
| ki būs | n | N/A | ÷ | A/A | ł | N/A | ; | H/A | ; | N/A | ; | N/A | ; | N/A | ł | 1.7E-11 | 20 | 5. 3E +03 | 9.0E+03 | ; | ; | 6 HRS |
| RCDHC | ربہ ا | 4. N | : | N/A | ; | N/A | ; | N/A | 1 | N/A | ! | 1.7E-11 | 20 | N/A | ł | N/A | ; | 2.3E+03 | 3.2E+03 | 6.95+02 | ; | 6 HKS |
| 90104 | ٥ | 4/H | ; | N/A | : | N/A | ; | M/A | ; | N/A | ; | N/A | ł | N/A | ; | N/A | 1 | 9.26+02 | ; | ł | ; | 1 |
| FLÜHÙ | ٩ | R/Å | ; | N/A | ; | N/A | i | N/A | ; | ĥ/ĥ | ; | 1.76-11 | 20 | N/A | ; | N/A | : | 1.66+03 | 2.6E+03 | 5.5E+02 | ! | 6 HKS |
| 6.155 | ø | N/A | ; | N/A | ; | N/A | : | N/A | ; | N/A | ; | N/A | ł | N/A | : | N/A | 1 | 6.0E+03 | : | ; | ł | ; |
| KCHIS | ы | 67N | : | 1.75-11 | <u>2</u> 0 | N/A | ; | N/A | ł | 1.75-11 | 20 | N/A | 1 | N/A | 1 | 1.76-11 | 20 | 6.8E+03 | 1.2E+04 | ; | ; | 6 HKS |
| 6-14 VS | ų | 4/4 | : | N/A | ; | N/A | 1 | 11-36-11 | 20 | N/A | | N/A | ; | N/A | ł | N/A | ł | 7.2E+03 | 1.25+04 | ł | 1 | 6 HRS |
| KCRNC | ¢ | A.A | ; | A/A | ; | N/A | ; | N/A | : | 1.7E-11 | 20 | N/A | ł | N/A | ; | 1.7E-11 | 20 | 2. JE+U3 | 3.2E+03 | 6. BE +U2 | 1 | 6 HKS |
| FLFut | 4 | N/A | : | N/A | ! | N/A | 1 | N/A | ; | N/A | ł | N/A | ; | N/A | 1 | 1.7E-11 | 20 | 1.6E+03 | 2.26+03 | 4.7E+02 | : | 6 HRS |
| FUFH Û | -0 | A/A | ł | N/A | ; | 11-37.1 | 20 | N/A | ł | N/A | ł | 1.76-11 | 20 | A/A | ł | N/A | : | 2.8E+03 | 3.9E+03 | E. 4E+02 | ł | 6 HKS |
| SCF C | Ð | M /A | 1 | N/A | ; | 1.76-11 | 20 | M/A | ł | N/A | ł | N/A | ; | N/A | ł | 11-37.1 | 20 | 1.46+03 | 2.0E+03 | 4.36+02 | ; | 6 HKS |
| St Pol | 4 | N/A | : | N/A | ; | 1.7E-11 | 20 | N/A | ; | N/A | ł | N/A | ł | N/A | 1 | 1.76-11 | 20 | 1.7E+03 | 2.4E+03 | 5.26+02 | : | 6 HKS |
| ĥĈĹŀ | 0 | N/ń | : | N/A | : | A/A | 1 | A/A | : | N/A | 1 | N/A | ł | N/A | : | 11-37.1 | 20 | 1.7€+03 | 2.4E+03 | 5.2E+U2 | : | 6 HR5 |
| 19414 | 'n | A/A | ; | M/6 | : | 1.7E-11 | 20 | N/A | ł | 1.76-11 | 20 | N/A | ; | N/A | ł | 1.7E-11 | 20 | 1.36+03 | 1.86+03 | 3.9E+u2 | ł | 6 HKS |
| ۰C£ JC | • | A.A | ; | 4/2 | ł | 11-37.1 | 20 | 4/N | : | 1.7E-11 | 20 | N/A | ; | N/A | 1 | 11-37.1 | 20 | 1.2E+03 | 1.76+03 | 3.6E+02 | ; | 6 HKS |
| 6Vc5VS | • | 4 '¥ | ; | A/A | : | H/A | ! | N/A | ł | N/A | ; | N/A | 1 | N/A | ; | 1.7E-11 | 20 2 | 2.7E+03 | 4.6E+03 | ; | : | S'YH 9 |
| FUBGF | ^ | H.C.N. | : | N/A | : | N/A | 1 | N/A | : | N/A | ; | N/A | : | N/A | ł | 1.4E-11 | 20 | 5. 3E+03 | ; | : | 1. IE+03 | 1 HK |
| FLL-AC | ~ | A/A | ÷ | K/3 | ; | N/A | ; | N/A | : | N/A | ł | 1.46-11 | 20 | N/A | ; | A/A | : | 2. JE+03 | ; | 1.2E+03 | 1.7E+02 | 20 MIN |
| KUCE | - | N /A | ! | 6/N | 1 | 4/4 | ; | N/A | 1 | N/A | ł | N/A | ; | N/A | ; | M/A | 1 | 9.26+02 | ł | ł | : | ; |
| ALCHU | •- | N.A | ; | R/N | ; | N/A | ; | N/A | ł | N/A | ł | 1.46-11 | 20 | N/A | ł | N/A | : | 1.86+03 | ! | 9.2E+02 | 1.4E+02 | 20 NIN |
| £C⊧ŭF | ~ | M/A | : | R/A | : | N/A | : | N/A | : | N/A | ł | N/A | : | N/A | ł | N/A | : | 6.0E+03 | : | : | : | i |
| fill HF | Ľ | 6/W | ; | 1.46-11 | 20 | N/A | ; | N/A | ; | 1.4E-11 | 20 | N/A | : | N/A | 1 | 1.46-11 | 20 | 6.8E+03 | 1 | ; | 6.8E+02 | I HK |
| RC: VF | - | A/A | ; | R/A | : | N/A | 1 | 1.4E-11 | 20 | N/A | ! | N/A | ; | N/A | 1 | N/A | ; | 7.2£+03 | ; | ; | 3.6E+02 | - XK |
| PCRVC | • | M/A | : | N/A | : | N/A | ; | N/A | : | 1.4E-11 | 20 | N/A | ł | N/A | 1 | 1.46-11 | 20 | 2.36+03 | ; | 1.1E+03 | B. 5E+01 | 20 NIN |
| KCPGC | ~ | R / N | ! | N/A | : | N/A | ! | N/A | ; | N/A | ; | H/A | ; | N/A | ł | 1.46-11 | 2 | 1.6E+03 | : | 7.86+02 | 2. 3E+02 | 20 NIN |
| ŔŨFHŰ | - | N/A | ; | N/A | 1 | 1.4E-11 | 20 | N/A | : | H/A | ; | 1.4E-11 | 20 | N/A | ł | N/A | ; | 2.8E+03 | ; | 1.4E+U3 | 2. IE+02 | 20 NIN |
| | | | - 14- 4 | | | | | | | | | | | | | | | | | | | |
| | ŝ | | | | | | | | | | | | | | | | | | | | | |

A state of a last state of a last









lo.





うううつけて

Fage 3 Date 20-Aug-87

OFFSITE TRANSPORTATION - REGIONAL DISPOSAL OPTION

Accident Frequencies and Range Factors

₹.

| Released |
|----------|
| bre |
| able |
| Avail |
| Agent |

| DURATION Time | 20 MIN | 20 MIN | NIN DZ | 20 MIN | 20 MIN | ₩ 1 | 6 HKS | 6 HKS | ł | 6 HRS | ł | 6 HKS | 6 HKS | 6 HKS | 6 HKS | A HRS | 6 HKS | 6 HKS | 6 HKS | 6 HFS | 6 HKS | 6 HKS | 20 NIN | ; | 20 MIN | 20 MIN | ZÙ MIN | 20 NIN | ZU NIN |
|-------------------|--------------|-----------------|----------|---------------|----------|---------|------------------------|--------------|---------|---------|---------|--------------|---------|--------------|----------------|--------------|---------|---------|---------|----------|----------|---------|----------|---------|---------|-----------|---------|----------|----------|
| LBS. Emitted | 5.4E+01 | 2.6E+02 | 6.5E+01 | 1.9E+02 | 4.56+01 | 1.4E+02 | ł | 1 | ł | ; | ; | ; | ł | ł | ł | ł | ; | ; | : | ; | ; | 1 | 4.3E+01 | ; | 3.56+01 | 2. IE+01 | 5.8E+01 | 5.36+01 | 1.46+01 |
| LBS. DETORATED | 7.26+02 | 8. 7E+02 | 8.7E+02 | 6.4E+02 | 6. 0E+02 | ł | : | 1 | 1 | ł | ; | ł | ; | : | : | ł | ł | ; | ł | 1 | ł | ł | 2.9E+02 | ; | 2.3E+02 | 2.8E+ù2 | 2.0E+02 | 3.5E+02 | 1.8E+02 |
| LBS. SPILLED I | 1 | ł | ; | 1 | ł | 1 | 2.2E+02 | 6.0E+0i) | ŀ | 3.2E+00 | ; | 1.7E+03 | 1.86+03 | 1.0€+01 | 6.5E+00 | 1.2E+01 | 6.00+00 | 1.56+01 | 1.5E+Ùl | 1.1E+01 | 1.0E+01 | 1.4E+03 | ł | 1 | ł | ł | ł | ; | ; |
| AGENT VAILABLE | 1.4E+03 | 1.7E+03 | 1.7E+03 | 1. 3E+03 | 1.2E+03 | 2.7E+03 | 5. 3E+03 | 2.36+03 | 9.2E+02 | 1.6E+03 | 6.0E+03 | 6.6E+03 | 7.2E+03 | 2.3E+03 | 1.6E+03 | 2.8E+03 | 1.4E+03 | 1.7E+03 | 1.7E+03 | 1. 3E+03 | 1.2E+03 | 2.7E+03 | 2. 3E+03 | 9.2E+02 | 1.85+03 | 2.3E+03 | 1.6E+03 | 2.86+03 | 1.4E+03 |
| IANGE Actor A | 20 | 50 | 20 | 20 | 20 | 2 | 37 | ł | 1 | 1 | ł | 31 | 1 | 37 | 31 | : | 37 | 31 | 37 | 37 | 31 | 37 | ; | 1 | 1 | 11 | 73 | } | 73 |
| LINDA FREQ. | 11-31-1 | 1.4E-11 | 11-31-11 | 1.46-11 | 1.46-11 | 1.46-11 | 4.85-07 | N/A | A/A | 6/N | N/A | 4.8E-07 | N/A | 4.8E-07 | 4.BE-07 | N/A | 4.85-07 | 4.8E-07 | 4.8E-07 | 4.85-07 | 4.85-07 | 4.8E-07 | A/A | N/A | N/A | 5.3E-08 | 5.2E-08 | N/A | 5. 2E-08 |
| RANGE Factor | ; | ł | 1 | ł | ł | ł | ł | ł | ł | ł | ١ | 1 | ł | 1 | ł | ; | ţ | : | ł | ; | ł | ł | ł | 1 | ; | ł | 1 | ł | ł |
| TEAD Fred. | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| RANGE Factor | ł | 1 | 1 | 1 | 1 | ł | ł | 37 | ł | 37 | ł | ł | ł | ; | ł | 37 | ł | : | ! | ł | ł | ł | 10 | ł | 10 | : | : | 100 | 1 |
| PUDA Freg. | A/A | N/A | N/A | N/A | N/A | N/A | N/A | 1.5E-07 | N/A | 1.56-07 | N/A | N/A | A/A | N/A | N/A | 1.5E-07 | N/A | N/A | N/A | N/A | N/A | N/A | 1.6E-08 | N/A | 1.6E-08 | N/A | N/A | 1.65-08 | N/A |
| RANGE Factor | ł | ł | 1 | 20 | 20 | ł | ł | ł | ! | 1 | ł | 37 | ł | 37 | ł | ! | ł | 1 | ł | 31 | 37 | ł | ł | ł | 1 | 51 | ! | ł | 1 |
| PBA Freq. | N/A | N/A | N/A | 11-31.1 | 11-31-1 | N/A | N/A | N/A | N/A | N/A | N/A | S. IE-07 | N/R | S. 1E-07 | N/A | N/A | N/A | N/A | ĥ/ĥ | 5.1E-07 | S. 1E-07 | N/A | H/A | N/A | N/A | S. 4E -08 | N/A | 4/H | K/A |
| RANGE Factor | ; | 1 | 1 | ; | 1 | ; | 1 | ł | ł | ; | ; | ; | 37 | ; | ļ | ! | ; | 1 | ł | 1 | 1 | 1 | ; | ! | 1 | 1 | 1 | : | ł |
| NARP Fred. | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | 1.7E-ù7 | N/A | N/Å | N/A | N/Å | N/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A | A/A | N/A | N/A | N/A |
| RANGE FACTOR | 20 | 50 | ; | 2 | 20 | ; | ł | 1 | ł | 1 | ł | 1 | : | ; | : | 12 | 31 | 5 | ł | 37 | 37 | ! | ; | ; | ; | 1 | | 87 | 87 |
| L BAD Fred. | 1.46-11 | 1.4E-11 | N/A | 1.45-11 | 1.46-11 | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | 1.6E-07 | 1.6E-07 | 1.6E-07 | N/A | l.6E-07 | 1.65-07 | N/A | N/A | H/H | K/A | A/A | H/A | 1.86-118 | 1.85-08 |
| RANGE FACTOR | ł | : | ; | ; | ; | ! | ; | ł | ł | ł | : | 37 | ; | 1 | : | : | : | 1 | : | : | ; | ; | ; | ; | 1 | ! | 1 | ; | : |
| APG Fred. | N/A | A/A | N/A | N/A | N/A | R/A | N/A | N/A | N/A | N/A | N/A | 1.16-07 | N/A | R/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| RANGE FACTOR | : | : | : | ; | ; | ; | ; | 1 | ; | ; | ; | ; | ; | 1 | ; | ! | 1 | ! | ł | : | 1 | ! | : | : | : | ł | ; | 1 | ; |
| ANAD Fred. | N/A | 4/R | N/A | N/A | Ψ/N | A/A | E/N | N/A | N/A | N.A | 6/N | A/A | N/A | A/A | R/N | A/A | N/6 | N/A | N/A | N/A | R/A | N/À | N/A | H/A | ħ/À | N/A | N/A | N/A | R/N |
| RD. | ٢ | ~~ | 1 | ۳. | 1 | | 11 | Ξ | Ξ | Ξ | Ξ | = | Ξ | Ξ | Ξ | = | Ξ | Ξ | Ξ | Ξ | Ξ | Ξ | Ľ | 2 | 2 | <u> </u> | 2 | 9 | 2 |
| 50EA- 6F10 | SV104 | FU260 | SCOVE | K LRBL | ACRVC | ĒŪĠVF | K Ü b 65 | RUDHS | kůc65 | RCCHS | KUK 65 | FU HS | 57. 13Y | ECNVS | F CF6S | ECFHS | 54-309 | 5.093.3 | FCGVS | F.CFGS | RURVS | RCSVS | RCDHC | FC:GC | RUCHE | FURVE | 5CPGC | FERC | R.JFVC |

See notes at end of table.

SASI TAATAA MUUUUU SUUSSA DAADAA KUUUUU DAADAA KUUUUUU DAADAA KUUUUUU DAADAA KUUUUUU DAADAA KUUUUU

N . Y . I

Faye 4 Bate 20-Aug-8?

UFFSITE TRANSPORTATION - REGIONAL DISPOSAL OPTION

Accident Fr

| 1 | | |
|---|--|--|
| 4 | | |
| ŝ | | |
| à | | |
| 1 | | |
| ; | | |
| į | | |
| | | |
| | | |

| | | | | | | | æ | ccident. | Frequen | pue saco | Range | Factors | | | | | | - | Agent Ava: | ilable and | i Released | |
|--------------------|--------------|---------------|------------------|--------------|-----------------|-----------------|-----------------|-----------------|-----------------|------------------|-----------------|---------------|-----------------|---------------|-----------------|-------------------|------------------|--------------------|-------------------|-------------------|-------------------|----------------------|
| 306A- 46.0 | 2 | ANHU Freg. | RANGE Factur | AFG Freq. | RANGE FACTOR | L BAD F REQ. | RANGE Factor | NAAP Freg. 1 | RANGE Factor | P5A FREG. 5 | KANGE Factor | PUDA FREE. | RANGE Factor | TEAD Fred. | RANGE Factor | UNDA F Freq. F | RANGE Actor I | AGENT AVAILAGLE | LBS. SPILLED I | LBS. Detunated | LBS. (Emitted | JUKAT I ON I I ME |
| | | | 1 1 1 1 | | | 60 Je | | | | | , , , | | | | | - JE DO | 11 | 1073L 1 | | 7 JE 407 | 10+35 7 | NIM OC |
| Laure Foot | 12 | 4/H | ; ; | 4/2 7/ N | : : | 1.05-V0 1/6 | È ! | 1/11 0/11 | : : | 4/H | : : | 1 () N | ; | 4/4 4/4 | i | 5.26-08 | : = | 1.76+03 | ; | 2. 2E+02 | 1.66+01 | 20 MIN |
| والمناز والمناز | : 1 | | ; | | ; | 1.8f-u8 | 86 | н. н И/А | : | 3.45-08 | 86 | 4/N | ; | N/A | ł | 5.46-08 | 99 90 | 1.36+03 | ; | 1.66+02 | 4.86+01 | 20 HIN |
| 1997. 1997. | : 2 | 4'R | ; | N/1 | ; | 1. 86-08 | 86 | N/A | : | 3.46-08 | - 98 9 | N/A | ; | N/A | 1 | 5.46-08 | 90 90 | 1.2E+ù3 | ; | 1.56+02 | 1.16+01 | 20 NIN |
| 6.537 | :::: | 4/2 | ł | E/N | : | N/A | ; | N/A | : | N/A | : | N/A | ; | N/A | ; | 1.6E-08 | 86 | 5.36+03 | | ; | 2.6E+02 | 2 HKS |
| RUN OF | | N/1 | : | NZA | ; | NIA | ţ | R/N | 1 | N/A | ; | N/A | ; | N/A | ; | N/A | ; | 6.ÚE+ÙJ | : | ; | ; | ÷ |
| 5-1 HF | 2 | N/A | ; | 1.26-08 | 13 | N/A | ; | M/A | ; | 3.3E-0B | 29 | 4/H | ; | N/A | : | 1.66-08 | 86 | 6.8E+03 | ł | ł | 1.7E+02 | 2 HRS |
| 10101 | 1 | († .) 12 | ; | 4/N | ; | N/A | ; | 1.96-08 | 64 | N/A | ; | N/A | ; | N/A | : | HIH | ł | 7.2E+U3 | ; | : | 9.06+01 | 2 HRS |
| FC5./F | :1 | 87.A | ; | A/A | ; | e/N | ; | N/A | ł | N/A | : | N/A | ; | N/A | { | 1.6E-08 | 98 | 2.7E+03 | 1 | ł | 3.46+01 | 2 HKS |
| 359.14 | = | 4/5 | ; | N/A | i | N/A | ; | N/A | ; | NZA | : | N/A | ; | A/A | ; | 1.5E-10 | 76 | 5. 3E+03 | ! | ; | 5.5[+0) | 8H 9 |
| NUC4C | . | 6/N | ļ | N/A | ! | R/N | ; | N/A | : | H/H | : | 2.2E-10 | 16 | N/A | { | N/A | ļ | 2.3E+03 | ł | ; | 2.5E-01 | AH & |
| 50035 | 1 | 1.2 | : | N/A | : | N/A | ł | N/A | : | N/A | 1 | N/A | ; | N/A | 1 | N/A | 1 | 9.26+02 | ; | ł | : | 1 |
| k.CH(| *1 | 년/성 | ; | N/A | 1 | N/A | ; | R/N | ł | N/A | ; | 2.2E-10 | 15 | N/A | ł | N/A | ; | 1.86+03 | ; | 1 | 2.5E 01 | 6 HF |
| 6.660 | Ξ | N/H | ; | N/A | : | R/A | ! | N/A | ; | N/A | ł | N/A | ; | A/A | ł | N/A | ! | 6. 0E +03 | ; | ; | ; | ; |
| SH 40 A | 1 | N/H | ; | 5.4E-08 | 41 | N.A | : | N/A | ł | 6.1E-08 | 108 | N/A | ; | N/A | { | 1.96-10 | 76 | 6.86+03 | 1 | : | 2.56-01 | 6 HF |
| Ect 1.C | = | ц, л | i | 6/N | : | N/A | ! | 6. IE -08 | 87 | A/A | 1 | N/A | ; | N/A | ł | N/A | ł | 7.2E+03 | ; | ł | 2. IE-03 | γHĘ |
| ACRVC | = | 11 / N | ł | 6/N | ł | R/A | ! | K/A | ł | 6.1E-08 | 108 | N/A | ł | N/A | ł | 1.96-10 | 76 | 2.3E+03 | ł | 1 | 2.1E-VJ | Å HE |
| Fifei | ÷ | 879 | 1 | N/A | ł | A/A | | N/A | ; | N/A | ł | A/A | ; | N/A | ; | 1.96-10 | 9 <i>i</i> | 1.6€+03 | ; | ; | 5.56+00 | 6 HK |
| FUCERC | : | Η, ê | ; | N/A | 1 | e.1E-ù8 | 59 | к/ А | ł | N/A | ; | 2.2E-10 | 16 | N/A | ; | N/A | ; | 2.85+03 | ł | ł | 2.56-01 | ьнК |
| 3.424 | = | e'n | ; | N/A | ł | 6.1E-UB | 8 5 | R/N | { | N/A | 1 | H/N | ; | N/A | ł | 1.96~10 | 16 | 1.4€+03 | ; | ł | 2.16-03 | 6 KF |
| k, gdC | = | R/A | ļ | N/A | ł | 6.1E-08 | 5 | N/A | ł | A/A | ł | N/A | ; | N/A | 1 | 1.96-10 | 76 | 1.7E+03 | ł | ł | 5.5E+00 | å HK |
| 50 0 ~C | <u>۲</u> | н. Б | ; | N/A | 1 | N/A | ; | N/A | ł | N/A | ł | N/A | ; | N/A | ł | 1.9E-10 | 76 | 1.7E+03 | ; | 1 | 2.1E-03 | 6 Hĥ |
| F.řĒÛ | 1 | N.A | ł | N/A | : | 6.15-36 | 85 | N/A | ł | 6.1E-08 | 25 | N/A | ; | N/A | ł | 1.96-10 | 76 | 1.3E+03 | ; | ; | 5.56+00 | A HR |
| ii CF .(C | <u>.</u> | N/H | 1 | N.'A | ; | 5.1E-0B | 8 | N/A | ; | 6.1E-08 | 25 | N/A | ; | N/A | : | 1.9E-10 | 16 | 1.26+03 | ; | : | 2. IE-03 | ₽ ¥¥ |
| F.CyC | = | й/Х | ł | ₩/Ĥ | ł | N/A | ; | N/A | ! | N/A | ł | N/A | ; | N/A | : | 1.9E-10 | 76 | 2.7E+03 | ł | ł | 2. IE-03 | Å HŔ |
| ьûрнu | 2 | N - L | ; | N/A | ; | N/A | ; | N/9 | ł | N/A | 1 | 4. IE-09 | 83 | N/A | ł | N/A | ; | 2.3E+03 | 3.0E+01 | 6. (E+ù0 | 1 | 6 HKS |
| r:::C | 5 | H .H | : | N/A | ł | N/R | : | N/A | ŀ | N/A | ł | N/A | ; | N/A | { | N/A | 1 | 9.26+02 | ; | ; | ł | 1 |
| ACCA L | ~ <u>.</u> 3 | N/4 | ; | e/N | ł | N/A | 1 | N/A | ł | N/A | ł | 4.1E-09 | 83 | N/A | ł | N/A | 1 | 1.8€+03 | 10+39.1 | 3.2E+00 | ; | 6 HRS |
| ы(п'л | 5 | H/À | : | N/A | : | 4/4 | : | N/A | ; | 1,06-08 | 67 | ₩/Ĥ | ; | N/A | ł | 1.3E-08 | 117 | 2.36+03 | 1. 1E+03 | 3.26+01 | : | 6 HRS |

KULLER RECEIPTING REPAIRS

630

See notes at end of table.









Fäge 5 Date 24-Aug-87

OFFSITE TRANSPORTATION - REGIONAL DISPOSAL OFTION

| ę | DURATION Tike | | 6 HRS | 6 HES | 6 HKS | 6 HKS | 5 HK 9 | 6 HRS | 6 HKS |
|------------|-------------------------|---|---------------|------------|-----------|---------|---------|---------------|--------------|
| l ƙeleasi | LBS. Emitted | | ł | ; | ; | 1 | ł | ł | ; |
| ilable and | LBS. Detonated | | 6.5E+00 | 1.26+01 | 6.0E+00 | 1.5E+01 | 1.56+01 | 1.1E+01 | 1.0E+01 |
| Agent Avai | LBS. Spilled i | | 3.2E+01 | 5.86+01 | 3.0E+01 | 7.3E+01 | 7.36+01 | 6.2E+02 | 5.0E+01 |
| | ABENT VALLABLE | | 1.6E+03 | 2.8E+03 | 1.46+03 | 1.7E+03 | 1.7E+03 | 1.3£+03 | 1.2E+03 |
| | range Factor a | | 117 | : | 117 | 117 | 117 | 117 | 117 |
| | UNDA FREQ. | | 1.35-08 | N/A | 1.3E-08 | 1.3E-08 | 1.36-08 | 1.3E-08 | L. 3E-08 |
| | RANGE Factor | | ł | ł | ł | ł | 1 | ; | ł |
| | TEAD Freq. | | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| | RANGE Factor | | ł | 8 3 | : | ł | : | ; | ; |
| factors | FUDA Fred. | | N/A | 4,16-09 | N/A | N/A | N/A | N/A | R/A |
| d Range | RANGE Factor | | 1 | ł | ł | 1 | ł | 67 | 67 |
| cies an | FBA Freq. | | N/A | N/A | N/A | N/A | N/A | 1.0E-08 | 1.0E-08 |
| Frequen | RANGE FACTOR | | ; | ; | : | ł | ; | : | ł |
| kcident | NAAP Freg. | | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| | KANGE Fåltor | | 1 | 11 | 11 | 11 | : | 12 | 11 |
| | L BAD Free. | | N/A | 6.1E-09 | 6. IE -09 | 6.1E-09 | N/A | 6.1E-09 | 6.1E-09 |
| | RANGE FACTOR | | : | ; | 1 | 1 | : | ; | ļ |
| | APG Freq. | | N/A | N/A | N.A | R/A | A/A | N/A | N/A |
| | ràn ge Fâctor | | ; | 1 | ! | : | ; | ; | ł |
| | ANAD Fred. | | N/A | ₿/Å | N/A | e/H | N/A | 6/H | E/N |
| | NO. | i | 5 | 5 | 5 | 2 | 5 | <u>5</u> | <u>1</u> |
| | -019 -1910 | • | RCF GC | ECPAC | KČF∜C | AC 261 | ŔĽŨVŨ | KCF aC | FURVE |

Müffö: 1. Scenarios 1-5 are per train mile; scenarios 6-15 are per exposure year.

 Buration time shown for scenarios with agent releases due to detonations and spills is for spill only. Buration time for detonation is instantaneous. I-121

Date 20-Aug-87 Fage 1

ġ

SCEN-Ario

~

DFFSITE TRANSPORTATION - NATIONAL DISPOSAL OPTION

Accident Frequencies and Range Factors

Agent Available and Keleased

DURATION Time

LBS. Emitted

6 HRS 6 HRS 6 HRS 6 HRS 6 HRS

1 : : : 6 HFS

25H 25H 25H 25H

: : 1 1 1

ΞŦ

1

HPS

6 HES 5 HES 6 HES UT# ()] 20 ein

4.4E-10 4, 4E-10 4, 4E-10 4, 4E-10

RCF65 RCC45 RCC45 RCF45
4, 4E-10 4, 4E-10 4, 4E-10 4.45-10

| | μ. | | | | | | | | | | | | | | | | | - | *7 | ۴ ٦ | 64 | ŝ | 5 | | -0 | - | - | | C 4 |
|--------------|-----------|---------|----------|---------|---------|---------|---------|-------------|----------|---------|---------|----------|---------|---------|-------------|---------|----------|----------|---------|------------|---------|---------|----------|-------------|---------|---------|---------|----------|------------|
| LBS. | DETONATED | 1 | 1 | 1 | ł | ; | ł | ; | ; | ; | : | ! | : | ; | ł | ; | ; | 2°4E+02 | 1.2E+02 | 2.35+02 | 2.85+02 | 2.0E+02 | 3, SE+02 | 1. BE +() 2 | 2.2E+02 | 2.2E+02 | 1.65+02 | 1.56+02 | : |
| LBS. | SPILLED | 2.2E+02 | 6.0E+00 | 1.6E+00 | 3.2E+00 | ł | 1.7E+03 | 1.8E+03 | 1.0E+01 | 6.5E+00 | 1.2E+01 | 6.0E+00 | 1.56+01 | 1.5E+01 | 1.1E+01 | 1.0E+01 | 1.46+03 | ł | 1 | ; | ; | 1 | : | ł | : | ; | ł | ł | ; |
| AGENT | AVAILAPLE | 5.3E+03 | 2. JE+03 | 9.2E+02 | 1.8E+03 | 6.0E+03 | 6.8E+03 | 7.2E+03 | 2.3E+1)3 | 1.6E+03 | 2.8E+03 | 1.4E+(13 | 1.7E+03 | 1.7E+03 | 1. JE +1) J | 1.25+03 | 2.7E+03 | 2. 3E+03 | 9.2E+02 | 1.8E+03 | 2.3E+03 | 1.6E+03 | 2.BE+03 | 1.4E+03 | 1.7E+03 | 1.7E+03 | 1.3E+03 | 1. 2E+03 | 5.3E+03 |
| RENGE | FACTOR | 9 | ł | ; | ; | ļ | 10 | ł | 01 | 9 | ł | 10 | 10 | 01 | 9 | 2 | 01 | ł | ł | ł | 21 | 27 | ; | 27 | 27 | 27 | 21 | 5 | 47 |
| UNDA | FREQ. | 3.2E-09 | N/A | N/A | N/A | N/A | 3.26-09 | N/A | 3.2E-09 | 3.2E-09 | N/A | 3.2E-09 | 3.2E-09 | 3.2E-09 | 3.26-09 | 3.2E-09 | 3. ZE-09 | N/A | N/A | N/A | 4.4E-10 | 4.4E-10 | N/A | 4.46-10 | 4.4E-10 | 4.4E-10 | 4.5E-10 | 4.5E-1) | 3.56-10 |
| RANGE | FACTOR | 1 | ł | ł | : | ł | ; | 1 | 1 | ţ | 1 | ; | ł | ; | ; | ł | ł | ł | ! | ł | ł | ł | ; | ; | ł | ł | ; | ł | t + |
| TEAD | FREQ. | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | M/A | N/A | N/A | A/A | N/A | H/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | N/A | N/A |
| RANGE | FACTOR | : | 10 | : | 91 | ! | ł | ! | ; | ; | 91 | ł | 1 | ł | 1 | ł | ł | 13 | ; | 23 | ! | ł | 21 | ! | ł | ; | : | ł | 1 |
| FUDA | FREQ. | N/A | 3.2E-09 | N/A | 3.2E-09 | N/A | N/A | N/A | N/A | N/A | 3.2E-09 | N/A | N/A | N/A | N/A | N/A | N/A | 4.4E-10 | N/A | 4.4E-10 | N/A | N/A | 4.4E-10 | N/A | N/A | N/A | N/A | N/A | A/A |
| RANGE | FACTOR | : | ł | 1 | ; | ł | 10 | ; | 9 | 1 | 1 | ł | ļ | ; | 01 | 2 | ł | 1 | ! | 1 | 21 | ł | 1 | ; | 1 | ! | 21 | 21 | ; |
| PBA | FREQ. | N/A | N/A | N/A | N/A | N/A | 3.26-09 | N/A | 3.26-09 | N/A | N/A | N/A | N/A | A/A | 3.2E-09 | 3.2E-09 | N/A | N/A | N/A | N/A | 6.4E-10 | N/A | N/A | N/A | N/A | A/A | 1.5E-LO | 1.5E-10 | N/A |
| PANGE | FACTOR | : | ł | ; | ; | : | ; | 9 | : | ļ | ; | ł | ł | ł | 1 | ; | 1 | 1 | ; | ł | ł | 1 | ; | : | ; | ; | ł | 1 | ; |
| NAAP | FREQ. | ¥/¥ | N/A | N/A | N/A | N/A | N/A | 3.2E-09 | N/A | N/A | A/A | N/A | N/A | N/A | A/A | N/A | A/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| RANGE | FACTOR | : | ł | ł | : | : | : | 1 | ; | ! | 01 | 01 | 01 | 1 | 01 | 10 | ł | ł | ł | ł | ł | | 27 | 27 | 27 | ł | 21 | 21 | ł |
| LEAD | FREQ. | N/A | N/A | N/A | N/A | N/A | N/A | #/ # | N/A | N/A | 3.2E-09 | 3. 2E-09 | 3.2E-09 | N/A | 3.2E-09 | 3.2E-09 | N/A | N/A | N/A | N/A | N/A | N/A | 4.4E-10 | 4.4E-10 | 4.4E-10 | N/A | 1.56-10 | 4.5E-10 | N/A |
| RANGE | FACTOR | | 1 | : | 1 | ! | 10 | ! | ! | ; | ; | 1 | ł | 1 | 1 | 1 | ł | ! | ; | 1 | ł | ł | ł | ! | 1 | ! | 1 | 1 | ; |
| AF6 | FREQ. | N/A | N/A | N/A | N/A | A/A | 3.2E-09 | 8/N | N/A | N/A | N/A | N/A | N/A | A/N | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| NGE | ĕ | | 01 | 10 | 9 | ; | 9 | : | 2 | 2 | 01 | 9 | 2 | 1 | 2 | 2 | ; | 23 | 23 | 23 | 21 | 27 | 11 | 11 | 12 | : | 71 | 21 | ł |
| R | FAC | | | | | | | | | | | | | | | | | | | | | | | | | | | | |

20 min

20 ain

20 MIG 20 **6**16 20 MG 20 Fin 20.616

4. 36 - 91 3. 56 - 61 3. 56 - 61 3. 56 - 61 2. 16 - 91 5. 36 - 61 1. 46 - 91 1. 46 - 91 1. 46 - 91 1. 16 - 91 1. 16 - 91 1. 16 - 91 2. 66 - 02 2. 76 - 02 2. 76 - 0

20 BLC 20 M D 2 H65 2.865

÷ 1 ÷ 1 1

.

N/A N/A A/A N/A N/A A/A ¢/N N/A

A / **N** /

.

N/A N/A A/N

A/A A'A

4.5E-10

N/A

3.5E-10

: :

1 1

3.56-10

4

3.56-10

N/A

1.76+02

ł

ł 1

6.0E+03 6.8E+03

; 7

3.5E-19 A/A

See notes at end of table.

TANK TANK

SPACED SERVERSED DESCRIPTION SERVERSED SERVERSED SERVERSED SERVERSED SERVERSED SERVERSED SERVERSED SERVERSED S

I-122

Fage 2 Date 20-Aug-87

•

F

.



OFFSITE TRANSPORTATION - NATIONAL DISPOSAL OPTION

| UN TACL. THUUN TACE. THUUN TACE. THUUN NUTLAL ALLE ALLE ALLE ALLE ALLE ALLE ALLE | 20 M// 1 | | | | £ ! | 2 | илек. и | 7.000 7.000 7.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 8.000 | глев. гамцик глев. гамцик N/A N/A N/A br>N/A | The function The function The function The function | TALE: TALUA TALE: TALUA TALE: N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A N/A N/A N/A 1.7E-11 20 N/A N/A N/A N/A N/A 1.7E-11 20 N/A 1.7E-11 1.7E-11 20 N/A 1.7E-11 1.7E-11 20 N/A 1.7E-11 | Theo. Theo. Theo. Theo. Theo. 5 N/A N/A N/A 6 N/A N/A N/A 6 1.7E-11 20 N/A N/A 6 1.7E-11 20 N/A N/A 6 1.7E-11 20 N/A N/A 7 N/A N/A 8 1.7E-11 20 N/A 8 1.7E-11 20 N/A 9 1.7E-11 20 N/A N/A 6 1.7E-11 20 N/A N/A 6 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A 1.7E-11 20 1.7E-11 20 N/A 1.7E-11 20 1.7E-11 20 N/A 1.7E-11 20 |
|---|--|---|----------------------------|---|--|---|---|---|---|--|--|---|
| N/A N/A | 1 20 1 20 1 1 1 20 1 1 1 20 1 1 1 1 20 1 1 1 1 20 1 1 1 1 1 1 1 1 1 1 | N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A | | , | | | NIA NIA NIA NIA NIA NIA NIA NIA NIA NIA | N/A N/A | N/A | | N/A | 5 N/A N/A N/A 5 N/A N/A N/A 6 1.76-11 20 N/A N/A 7 N/A N/A N/A 6 1.76-11 20 1.76-11 20 N/A 6 1.76-11 20 N/A N/A 7 N/A N/A N/A 6 1.76-11 20 N/A N/A 6 1.76-11 20 N/A N/A 7 1.76 1.76 20 N/A 6 1.76-11 20 N/A 1.76 7 1.76-11 20 N/A 1.76 6 1.76-11 20 N/A 1.76 7 1.76-11 20 N/A 1.76 |
| N/A N/A 1.7E-11 20 5.3E+03 9.0E+03 1.7E-11 20 N/A N/A 2.3E+03 3.2E+03 6.9E+02 M/A N/A N/A 2.3E+03 3.2E+03 5.9E+02 N/A N/A 1.4 2.3E+03 2.8E+02 1.7E-11 20 N/A N/A 1.8E+03 3.5E+02 N/A N/A N/A 5.5E+02 0 N/A N/A 1.7E-11 20 6.8E+03 0 N/A N/A 5.5E+04 0 N/A N/A 7.2E+04 | | N.A N.A N.A N.A N.A N.A N.A N.A N.A N.A | | | · · · · · · · · · · · · · · · · · · · | | NIA NIA NIA NIA NIA NIA NIA NIA NIA NIA | | N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A 1.7E N/A N/A N/A 1.7E 1. N/A N/A N/A N/A 1.7E 1. 20 N/A N/A 1.7E 1. 20 N/A | | N.A. N.A. N.A. ITE-II 20 N/A N/A N/A ITE-II 20 I.TE-II 20 N/A N/A ITE-II 20 I.TE-II 20 N/A N/A ITE-II 20 I.TE-II 20 N/A N/A ITE-II 20 N/A N/A N/A I.TE-II 20 | N/A |
| 1.7E-11 20 N/A N/A 2.3E+03 3.2E+03 6.9E+02 N/A N/A 9.2E+02 1.3E+03 2.6E+02 1.7E-11 20 N/A N/A 1.8E+03 2.6E+02 N/A N/A 1.8E+03 2.6E+03 5.5E+02 N/A N/A 1.7E-11 20 4.7 0 N/A N/A 5.6E+03 1.7E-11 20 6.8E+03 1.2E+04 1.8 N/A 1.7E+11 20 1.8 N/A 1.7E+03 1.2E+04 | | N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A | | = | N N N N N N N N N N N N N N N N N N N | | N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A 1.7E-11 20 | | N/9 N/1 N/3 N/4 N/3 N/3 N/4 N/3 N/3 N/3 N/3 N/3 N/3 N/3 N/3 N/3 N/3 N/3 N/3 N/3 N/3 N/4 N/3 N/3 N/4 N/3 N/3 N/4 1.7E-11 20 N/3 N/6 1.7E-11 20 N/3 | 20 N/A U/A N/A 20 N/A N/A N/A 20 N/A N/A N/A N/A N/A N/A N/A N/A N/A 20 1.76-11 20 N/A N/A N/A N/A N/A 20 N/A N/A N/A 20 N/A N/A N/A 20 N/A N/A N/A 20 N/A 1.76-11 20 N/A 20 N/A 1.76-11 20 N/A | 1. 7E-11 20 N/A U/A N/A 1. 7E-11 20 N/A N/A N/A 1. 7E-11 20 N/A N/A N/A 1. 7E-11 20 1. 7E-11 20 N/A N/A 1. 7E-11 20 1. 7E-11 20 N/A N/A 1. 7E-11 20 1. 7E-11 20 N/A N/A 1. 7E-11 20 1. 7A N/A N/A 1. 7E-11 20 N/A 1. 7E-11 20 N/A 1. 7E-11 20 N/A 1. 7E-11 20 N/A | b 1.7E-11 20 N/A /A N/A N/A |
| W/A N/A 9.2E+02 1.3E+03 2.8E+02 1.7E-11 20 N/A N/A 1.8E+03 2.8E+02 N/A N/A 1.8E+03 2.6E+03 5.5E+02 N/A N/A 6.0E+03 0 N/A 1.7E-11 20 6.8E+03 1 A 1.7E-11 20 6.8E+03 1.2E+04 1 N/A N/A 1.2E+04 | 1 1 20 1 1 20 1 1 20 1 1 1 1 20 1 1 1 20 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | N/A N/A N/A N/A 1.76-11 N/A N/A N/A N/A N/A N/A N/A N/A | | | N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A | | N/A N/A N/A N/A | | N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A | 20 N/A N/A N/A 20 N/A N/A N/A N/A N/A N/A N/A N/A N/A 20 1.7E-11 20 N/A N/A N/A N/A N/A 20 N/A N/A N/A 20 N/A N/A N/A 20 N/A N/A N/A 20 N/A 1.7E-11 20 N/A | I.7E-I1 20 N/A N/A N/A I.7E-I1 20 N/A N/A N/A N/A N/A N/A N/A I.7E-I1 20 1.7E-11 20 N/A N/A I.7E-11 20 1.7E-11 20 N/A N/A N/A N/A N/A N/A I.7E-11 20 1.7E-11 20 N/A N/A I.7E-11 20 N/A 1.7E-11 20 N/A I.7E-11 20 N/A 1.7E-11 20 N/A I.7E-11 20 N/A 1.7E-11 20 N/A | 6 1.7E-11 20 N/A N/A N/A b 1.7E-11 20 N/A N/A N/A b N/A N/A N/A N/A b 1.7E-11 20 1.7E-11 20 N/A N/A b 1.7E-11 20 1.7E-11 20 N/A N/A b 1.7E-11 20 N/A 1.7E-11 20 N/A b 1.7E-11 20 N/A 1.7E-11 20 N/A |
| 1.7E-11 20 N/A N/A 1.8E+03 2.6E+03 5.5E+02 N/A N/A N/A 6.0E+03 0 N/A N/A 1.7E-11 20 6.8E+03 1.2E+04 N/A N/A N/A 7.2E+03 1.2E+04 | | N/A N/A 1.76-11 N/A N/A N/A N/A N/A N/A N/A N/A | 8 | - | N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A | | N/A N/A N/A N/A | | N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A | 20 N/A | 1.7E-11 20 N/A | b 1.76-11 20 N/A N/A N/A b N/A N/A N/A N/A b 1.76-11 20 1.76-11 20 N/A N/A c N/A N/A N/A N/A c N/A N/A N/A N/A b 1.76-11 20 N/A N/A N/A b 1.76-11 20 N/A N/A N/A b 1.76-11 20 N/A 1.76-11 20 N/A d 1.76-11 20 N/A 1.76-11 20 N/A d 1.76-11 20 N/A 1.76-11 20 N/A |
| N/A N/A N/A 6.0E+03 0 N/A N/A 1.7E-11 20 6.8E+03 1.2E+04 N/A N/A N/A 7.2E+03 1.2E+04 | | N/A 1.75-11 N/A 1.76-11 N/A N/A N/A N/A N/A N/A N/A N/A 1.75-11 | | | N/A N/A N/A N/A N/A N/A N/A N/A N/A | | N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A I.7E-II 20 N/A I.7E-II 20 N/A I.7E-II 20 N/A | | N/A N/A N/A N/A 1.7E-11 20 N/A N/A N/A 8/A N/A N/A 8/A N/A N/A N/A N/A N/A N/A N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A | | N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A 1.7E-11 20 N/A | b N/A |
| 0 N/A N/A I.7E-11 20 6.8E+03 1.2E+04 N/A N/A N/A 7.2E+03 1.2E+04 | 20 1.1 20 | 1.76-11 N/A I.76-11 N/A N/A N/A N/A N/A N/A N/A N/A 1.76-11 | : 8 : : : : : : : : | | M/A 1.7E-11 N/A N/A N/A N/A N/A N/A | N/A 1.76-11 N/A N/A 20 N/A 20 N/A 20 N/A 20 N/A | N/A N/A K/A 1.76-11 N/A N/A N/A N/A 1.76-11 20 N/A 1.76-11 20 N/A 1.76-11 20 N/A | 20 N/A N/A N/A I.7E-11 N/A N/A N/A N/A I.7E-11 20 N/A I.7E-11 20 N/A N/A N/A | 1.7E-11 20 N/A N/A N/A K/A 1.7E-11 N/A N/A N/A N/A N/A N/A N/A 1.7E-11 20 N/A | 29 1.76-11 20 N/A N/A N/A N/A 1.76-11 20 N/A N/A N/A 20 N/A N/A N/A 20 N/A N/A N/A 20 N/A 1.76-11 20 N/A 20 N/A 1.76-11 20 N/A 20 N/A 1.76-11 20 N/A | 1.7E-11 20 1.7E-11 20 N/A N/A N/A N/A N/A 1.7E-11 1.7E-11 20 N/A N/A 1.7E-11 1.7E-11 20 N/A N/A N/A N/A N/A N/A N/A N/A N/A 1.7E-11 20 N/A | b 1.7E-11 20 1.7E-11 20 1.7E-11 20 N/A |
| N/A N/A N/A 7.2E+03 1.2E+04 | | N/A 1.76-11 N/A N/A N/A N/A N/A N/A 1.76-11 | ⁸ : : : : : : : | | 1.75-11 N/A N/A N/A N/A N/A N/A N/A | 1.76-11 N/A N/A 20 N/A 20 N/A 20 N/A 20 N/A 20 N/A | K/A 1.7E-11 K/A N/A N/A N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A | K/A I.7E-II K/A K/A N/A N/A I.7E-II 20 N/A I.7E-II 20 N/A N/A N/A | N/A H/A I.7E-II N/A N/A N/A N/A N/A N/A N/A 1.7E-II 20 N/A | N/A N/A 1.76-11 20 N/A N/A N/A 20 N/A N/A N/A 20 N/A 1.76-11 20 N/A 20 N/A 1.76-11 20 N/A 20 N/A 1.76-11 20 N/A | N/A N/A N/A I.7E-II I.7E-II 20 N/A N/A I.7E-II I.7E-II 20 N/A N/A N/A I.7E-1I 20 N/A I.7E-II 20 N/A I.7E-II 20 N/A I.7E-II 20 N/A I.7E-II 20 N/A I.7E-II 20 N/A | c N/A N/A I/TE-11 20 N/A 1/T/T T D N/A D D D <thd< th=""> <thd< th=""> <thd< th=""> D<!--</td--></thd<></thd<></thd<> |
| | 20 1.1 20 1.1 20 1.1 20 20 1.1 20 20 20 20 20 20 20 20 20 20 20 20 20 | 1.76-11 N/A N/A N/A N/A N/A N/A N/A N/A N/A | | | N/N A/N A/N A/N A/N A/N A/N | 20 N/A 20 N/A 20 N/A 20 N/A 20 N/A 20 N/A 20 N/A | N/A N/A N/A N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A | N/A N/A N/A N/A 1.76-11 20 N/A 1.76-11 20 N/A N/A N/A | N/A N/A N/A N/A N/A N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A N/A N/A | 20 N/A N/A N/A 20 N/A N/A N/A 20 N/A 1.7E-11 20 N/A 20 N/A 1.7E-11 20 N/A 20 N/A 1.7E-11 20 N/A | 1.7E-11 20 N/A N/A N/A N/A 1.7E-11 20 N/A N/A N/A 1.8E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A | 6 1.7E-11 20 N/A N/A N/A 6 1.7E-11 20 N/A N/A N/A b 1.7E-11 20 N/A 1.7E-11 20 N/A 6 1.7E-11 20 N/A 1.7E-11 20 N/A 6 1.7E-11 20 N/A 1.7E-11 20 N/A |
| 0 N/A N/A 1.7E-11 20 2.3E+U3 3.2E+03 b.BE+02 | | N/A N/A N/A N/A N/A N/A 1.7E-11 | : : : : : : | | N/A N/A A/A A/A A/A | 20 N/A 20 N/A 20 N/A 20 N/A 20 20 | N/A N/A 1.7E-L1 20 N/A 1.7E-L1 20 N/A 1.7E-L1 20 N/A N/A N/A | N/A N/A 1.7E-L1 20 N/A 1.7E-L1 20 N/A 1.7E-L1 20 N/A N/A N/A | N/A N/A N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A N/A N/A | 20 N/A N/A N/A 20 N/A 1.7E-11 20 N/A 20 N/A 1.7E-11 20 N/A 20 N/A 1.7E-11 20 N/A | 1.7E-11 20 N/A N/A N/A N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A | b 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7F-11 20 N/A 1.7F-11 20 N/A 1.7F-11 20 N/A </td |
| N/A N/A 1.7E-11 20 1.6E+03 2.2E+03 4.7E+02 | 1.75 1.77 1.77 1.77 1.77 1.77 1.77 1.77 | N/A N/A N/A N/A N/A N/A 1.7E-11 | : : : : : | | N/A N/A N/A A/A | 20 N/A 20 N/A 20 N/A 20 N/A | 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A N/A N/A | 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A N/A N/A | N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A N/A N/A | 20 N/A 1.7E-11 20 N/A 20 N/A 1.7E-11 20 N/A 20 N/A 1.7E-11 20 N/A | 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A | b 1.7E-11 20 N/A 1.7E-11 20 N/A b 1.7E-11 20 N/A 1.7E-11 20 N/A b 1.7E-11 20 N/A 1.7E-11 20 N/A |
| 1.7E-11 20 N/A N/A 2.8E+03 3.9E+03 B.4E+02 | | N/A N/A N/A N/A 1.7E-11 | | | N/A N/A N/A | 20 N/A 20 N/A N/A | 1.7E-11 20 N/A 1.7E-11 20 N/A N/A N/A | 1.7E-11 20 N/A 1.7E-11 20 N/A N/A N/A | N/A 1.7E-11 20 N/A N/A 1.7E-11 20 N/A N/A N/A N/A | 20 N/A 1.7E-11 20 N/A 20 N/A 1.7E-11 20 N/A | 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.7E-11 20 N/A 1.6.11 20 N/A 1.7E-11 20 N/A | b 1.7E-11 20 N/A 1.7E-11 20 N/A b 1.7E-11 20 N/A 1.7E-11 20 N/A |
| N/A N/A 1.7E-11 20 1.4E+03 2.0E+03 4.3E+02 | | N/A N/A 1.7E-11 | : : : | | N/A N/A | 20 N/A | 1.7E-11 20 N/A N/A N/A | 1.7E-11 20 N/A N/A N/A | N/A 1.7E-11 20 N/A N/A N/A N/A | 20 N/A 1.7E-11 20 N/A | 1.76-11 20 N/A 1.76-11 20 N/A ata bia bia bia | 6 1.7E-11 20 N/A 1.7E-11 20 N/A |
| N/A N/A 1.7E-11 20 1.7E+03 2.4E+03 5.2E+02 | | N/A 1.7E-11 | : : | | A/A | N/N | N/A N/A | N/A N/A | N/G N/A N/A | | 1110 N/A N/A N/A | |
| N/A N/A 1.7E-11 20 1.7E+03 2.4E+03 5.2E+02 | 20 20 21 20 20 20 20 20 20 20 20 20 20 20 20 20 | 11-37-11 | ł | | | 412 | | | | N/G N/A N/A | | 5 N/A N/A N/A N/A |
| 0 N/A N/A I.7E-11 20 I.3E+03 I.8E+03 3.9E+02 | 20 1 20 | | ; | | N/A | ZU R/A | 1.7E-11 20 N/A | 1.7E-11 20 N/A | N/A 1.7E-11 20 N/A | 20 N/A 1.7E-11 20 N/A | 1.7E-11 20 N/A 1.7E-11 20 N/A | \$ 1.7E-11 20 N/A 1.7E-11 20 N/A |
| 0 N/A N/A 1.7E-11 20 1.2E+03 1.7E+03 3.6E+02 | ≥≥: :: | 1./11 | : | | N/A | 20 N/A | 1.7E-11 20 N/A | 1.7E-11 20 N/A | N/A 1.7E-11 20 N/A | 20 N/A 1.7E-11 20 N/A | 1.7E-11 20 N/A 1.7E-11 20 N/A | e 1.7E-11 20 N/A 1.7E-11 20 N/A |
| N/A N/A 1.7E-11 20 2.7E+03 4.6E+03 | /N | N/A | 1 | | N/A | N/A | N/A N/A | N/A N/A | N/A N/A N/A | N/A N/A N/A | N/A N/A N/A N/A | 6 N/A N/A N/A N/A |
| N/A N/A 1.4E-11 20 5.3E+03 | | N/A | ł | | N/A | N/A | N/A N/A | N/A N/A | N/A N/A N/A | N/A N/A N/A | M/A N/A N/A N/A | 7 M/A N/A N/A N/A |
| 1.4E-11 20 N/A N/A 2.3E403 1.2E403 | 1.4E | N/A | ł | | N/A | N/A | N/A N/A | N/A N/A | N/A N/A N/A | 20 N/A N/A N/A | 1.4E-11 20 N/A N/A N/A | 7 1.4E-11 20 N/A N/A N/A |
| N/A N/A N/A 9.2E+02 4.6E+02 | 87 | N/A | ł | | N/A | N/A | N/A N/A | N/A N/A | N/A N/A N/A | 20 N/A N/A N/A | 1.4E-11 20 N/A N/A N/A | 7 1.4E-11 20 N/A N/A N/A |
| 1.4E-11 20 N/A N/A 1.8E+03 9.2E+02 | 3-1 | N/A | : | | N/A | N/A | N/A N/A | N/A N/A | N/A N/A N/A | 20 N/A N/A N/A | 1.4E-11 20 N/A N/A N/A | 7 1.4E-11 20 N/A N/A N/A |
| N/A N/A N/A 6.0E+03 | N/ | N/A | : | | N/A | N/A | K/A N/A | N/A N/A | N/A N/A N/A | N/A N/A N/A | N/A N/A N/A N/A | 7 N/A N/A N/A N/A |
| 0 N/A N/A 1.4E-11 20 6.8E+03 4 | 20 N/. | 1.46-11 | 1 | | N/A | N/A | N/A N/A | 20 N/A N/A | 1.4E-11 20 N/A N/A | 20 1.4E-11 20 N/A N/A | 1.4E-11 20 1.4E-11 20 N/A N/A | 7 1,4E-11 20 1.4E-11 20 N/A N/A |
| N/A N/A N/A 7.2E+03 | N/ | N/A | ຊ | | 1.4E-11 | 1.4E-11 | N/A 1.4E-11 | N/A 1.4E-11 | N/A N/Á 1.4E-11 | N/A N/A 1.4E-11 | N/A N/A N/A 1.4E-11 | 7 N/A N/A N/A 1.4E-11 |
| 0 N/A N/A 1.4E-II 20 2.3E+03 1.1E+03 I | 20 N/. | 1.4E-11 | : | | N/A | N/A | N/A N/A | N/A N/A | N/A N/A N/A | 20 N/A N/A N/A | 1.4E-11 20 N/A N/A N/A | 7 1.4E-11 20 N/A N/A N/A |
| N/A N/A 1.4E-11 20 1.6E+03 7.8E+02 2 | NV | N/A | ł | | N/A | N/A | N/A N/A | N/A N/A | N/A N/A N/A | 26 N/A N/A N/A | 1.4E-11 20 N/A H/A N/A | 7 1.4E-11 20 N/A H/A N/A |
| 1.4E-11 20 N/A N/A 2.6E+03 1.4E+03 3 | 1.45 | N/A | : | | N/A | 20 N/A | 1.4E-11 20 N/A | 1.4E-11 20 N/A | N/A 1.4E-11 20 N/A | 20 N/A 1.4E-11 20 N/A | 1.4E-11 20 N/A 1.4E-11 20 N/A | 7 1,4E-11 20 N/A 1,4E-11 20 N/A |
| N/A N/A 1.4E-11 20 1.4E+03 7.2E+02 5 | N/ | E/N | ł | | N/A | 20 N/A | 1.4E-11 20 N/A | 1.4E-11 20 N/A | K/A 1.4E-11 20 N/A | 20 N/A 1.4E-11 20 N/A | 1.4E-11 20 K/A 1.4E-11 20 N/A | 7 1.4E-11 20 K/A 1.4E-11 20 N/A |
| N/A N/A 1.4E-11 20 1.7E+03 6.7E+02 3 | N. | N/A | ; | | N/A | 20 N/A | 1.4E-il 20 N/A | 1.4E-il 20 N/A | N/H 1.4E-11 20 N/A | 20 N/H 1.4E-11 20 N/A | 1.4E-11 20 N/H 1.4E-11 20 N/A | 7 1.4E-11 20 N/H 1.4E-11 20 N/A |

.....

. And in . I-123

Fage 3 Date 20-Aug-87

GFFSITE TRAMSPORTATION - NATIONAL DISPOSAL OFTION

Accident Frequencies and Range Factors

Agent Available and Released

こうてい かんかん 一部のよう しょう

| NO | 쏉 | ł | X | z | z | | S. | ទួ | ş | ş | | 50 | Ņ | 'n | ŝ | ι, Γ | s | ŝ | S | Ş, | çı | ş | æ | × | z | z | z | z | z | ¥ | z | æ |
|-------------|----------|---|---------|-------------------|---------|---------|--------------|---------|----------|--------------|----------|------------------|--------|------------------|---------|---------|---------|---------|---------|---------|------------------|---------|-------------|--------------|---------|------------|---------------|---------|--------------|----------|---------|------------|
| DURATI | Ξ | ļ | 20 MI | 20 MI | 20 MI | ¥ | -9 | 9 1 | 4 9 | à Hĥ | ; | 9 1 1 1 | 4 9 | 9 1 1 1 | 9 H | ~ ₹ | 6 HF | 6 HF | 9 H 9 | ·9 또 | 9 1 1 1 | 6H 9 | 20 MI | 20 MI | 20 HI | 1N 02 | 20 MI | 20 NI | 20 MI | Z0 NI | 2 | 20 NI |
| TBS. | C3111N3 | | 6.5E+01 | 1.96+02 | 1.5E+01 | 1.4E+02 | ; | ; | ł | ; | ; | ; | ł | ; | ł | ł | 1 | ł | : | ł | ! | ; | 1.35+01 | 10+32.5 | 3.56+01 | 2.1E+01 | 5.BE+01 | 5.36+01 | 1.46+01 | 6.5E+01 | 1.6E+01 | 10+38.1 |
| LBS. | IONATED | | .7E+02 | .4E+02 | .0E+02 | ! | ł | ! | ł | ; | ; | ; | ; | 1 | ł | ; | ; | ł | ; | ł | ; | ; | 95+02 | 2E+02 | 3E+02 | .BE+02 | . 0E+02 | SE +02 | . BE + U2 | .2E+02 | . 2E+02 | . b€ +û2 |
| LES. | LLED BE | | | ه ! | - 9 | ! | E+02 | 00+3I | E+00 | E+00 | ļ | 'E+03 | E+03 | 10+3 | E+00 | 10+3 | E+00 | E+01 | 10+3 | E+01 | E+01 | £+03 | - 2 | <u>ن</u> ـــ | | | - 2 | n 1 | | - | ~i · | - |
| = | E SFI | : | - 2 | | | | 3 2.7 | 3 6.(| 1.6 | 3.3.2 | | 3 1.7 | I 1.8 | 3.1.0 | 3 6.5 | 3 1.2 | 13 b.0 | 33 1.5 | 3 1.5 | 1.1 | 3 1.0 | 3 1.4 | | ≃ | - | - | ي ۲ | | ۲ | ~ | - | 5 |
| AGED | AVAILABL | ļ | 1.7E+(| 1.35+(| 1.2E+(| 2.7E+(| 5. 3E+(| 2. 3E+(| 9. 2E +(| 1.8£+(| 6.0E+(| 6.8E+(| 7.2E+0 | 2. 3E+(| 1.6E+0 | 2.66+0 | 1.45+0 | 1.76+0 | 1.76+0 | 1.3E+C | 1.2E+C | 2.7E+(| 2. 3E+(| 9.2E+C | 1.85+0 | 2. 3E+(| 1.6E+(| 2.BE+0 | 1.4E+(| 1.7E+0 | 1.76+6 | 1. SE+G |
| RANGE | FAC TOR | | 20 | 20 | 20 | 20 | 37 | ł | ł | ł | ł | 37 | ; | 37 | 37 | ! | 37 | 37 | 37 | 37 | 37 | 27 | ! | ł | ; | 11 | 13 | i | 1, | 13 | 52 | 86 |
| ADAU | FREQ. | | 1.46-11 | 1.46-11 | 1.46-11 | 1.4E-11 | 4.65-07 | N/A | N/A | N/A | N/A | 4.86-07 | N/A | 4.BE-07 | 4.86-07 | N/A | 4.8E-07 | 4.66-07 | 4.86-07 | 4.86-07 | 4.BE-07 | 4.BE-07 | N/A | N/A | N/A | 5.36-08 | 5.26-08 | N/A | 5.2E-08 | 5.2E-08 | 5.2E-08 | 5.4E-08 |
| RANGE | FACTOR | | ł | ł | ł | 1 | ł | ; | 1 | ! | 1 | ; | 1 | ! | ; | 1 | 1 | ł | 1 | ł | ł | ł | ł | ; | ł | ł | 1 | ! | ; | 1 | : | 1 |
| TEAD | FREQ. | | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | A/A | K/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | н/А | N/A |
| RANGE | FACTOR | | ł | ; | ; | ł | ł | 37 | ; | 37 | 1 | ł | 1 | ł | ; | 37 | 1 | ŧ | I | : | { | ſ | 3 | ł | 10 | 1 | { | 100 | ; | í | 1 | ; |
| FUDA | FREQ. | | N/A | NA | N/A | N/A | N/A | .56-07 | N/A | .56-07 | N/A | N/A | N/A | N/A | N/A | .56-07 | N/A | k/A | N/A | N/A | N/A | N/A | .6E-08 | N/A | .66-08 | N/A | N/A | . 6E-08 | N/A | N/A | N/A | N/A |
| ANGE | ACTOR | | 1 | 20 | 20 | 1 | ł | - | ! | - | ł | 37 | ł | 37 | : | ; | ! | ł | ; | 37 | 37 | ; | : | ł | - | 61 | ; | - | ł | : | ; | 86 |
| YEA K | KEQ. F | | N/A | +E-11 | 45-11 | N/A | N/A | N/A | N/A | N/A | N/A | 6E-07 | N/A | 6E -07 | H/A | N/A | N/A | N/A | R/A | 6E-07 | 6E - 07 | N/A | N/A | N/A | N/A | 7E-08 | н/а | N/A | N/A | 14/A | N/A | B0-3/ |
| ANGE F | ACTOR F | | ; | | | ł | ; | ; | 1 | ł | ; | | 37 | + | : | 1 | ł | ; | : | | | ; | ; | ł | ; | | ; | ł | ; | ; | : | |
| iaP R. | E9. F | 1 | I/A | (/A | t/A | (7Å | 1/A | (/A | I/A | (/A | 1/A | NA | E-07 | t/A | I/A | I/A | I/A | 1/A | I/A | I/A | I/A | I/A | 1/A | I/A | I/A | 1/4 | I/A | N.A | I/A | ۲.A | I/A | I/A |
| EE RE | IOK FF | | - | 20 6 | 20 | - | - | - | - | - | - | - | | - | - | 37 | 37 | 37 | | 37 | 27 | - | | | - | - | - | 5 | - 3 | 5 | - | 49 49 |
| KAN | . FaC | - | | 11 | Ξ | _ | | _ | | _ | _ | | | _ | | 01 | 01 | 01 | | 01 | 01 | _ | - | | | | | - 80 | 1 80 | | | 3 |
| L BAD | FREG | | N/A | 1.46- | ÷. | N/4 | A/N | 4/H | 8/N | N/A | N/A | N. F | N.A | A/N | A/A | 3.6 | 3. CE- | 3. G | A/A | 3.05- | 3.05- | A/A | €/N | N.A | R/A | N/A | A / A | 3.16- | з З | 3. 3 | A/N | 3. 36- |
| RANGE | FACTOR | | : | ! | ! | ; | 1 | ! | ! | ł | ; | 37 | ł | ! | ; | ; | ł | : | ; | ! | 1 | ł | : | 1 | 1 | ; | ; | ; | ; | ł | : | : |
| 4PG | FKEQ. | | A/N | N/6 | N/A | N/A | N/A | N/6 | N/A | N/A | N/A | 1.1E-07 | N/A | N/A | A/A | N/A | N/A | N/A | R/N | N/A | R/N | R/N | A/A | R/A | N/A | N/A | N/A | e/N | A/N | N/à | N/A | N/A |
| ANGE | ALTOR | | ; | 20 | 2 | ; | : | 11 | 37 | 15 | ; | 2 | ł | 37 | 31 | 12 | ŝ | | ; | 5 | I. | ; | 96 | 95 | 3.6 | <u>105</u> | 1-0 | 100 | 0.4 | 100 | ; | θb |
| 0946 | .KED. | - | N/A | 11-34.1 | 1.46-11 | ₩/Ĥ | e/n | 1.46-07 | l.4E-u7 | 1.46-07 | 878 8 | 1.45-07 | N/A | 1.45-07 | 1.45-07 | 1.46-37 | 1.46-07 | 1.46-07 | H/H | 1.46-07 | 1.45-07 | A/A | 1. 5E - (:B | 1. 5£ - 08 | 9r-3r.i | 1. b£ -06 | 90-35-1 | 1.56-08 | 1.5£-08 | 1.5£-ii8 | 4/X | 1. e£ -1)8 |
| жС. , | - | ; | 1 | ~ | ~ | 1 | Ξ | = | Ξ | Ξ | Ħ | 11 | 11 | = | Ξ | = | = | Ξ | = | = | Ξ | Ξ | <u>_</u> | <u>_</u> | 2 | 1 | <u> </u> | - | 2 | 2 | | |
| - 8305 | ĥĥ | | FCGVC | FCSuC | ACEVC | AC50F | KLEGS | FLDHS | FCLES | FCCH5 | 59 (19 | Filt HS | SV 10 | RCAVS | FLF 65 | FCFHS | ELF 25 | 6.Eus | ດເບັນ | Fi KUS | FCFVS | 515.5 | ECCHC | Filist | RLCHC | FCAVE | FCF GC | FCFHC | FLFVC | KCG6C | FULL | ACF 30 |



See nutes at end of table.

TANGGON TANGGON ANALAN SANANG KANANAN KANANAN KANANA MANANAN MANANAN MANANAN MANANAN MANANAN TANANAN TAN







\$14<u>1</u>410141<u>8</u>14154154154<u>1</u>549154<u>15</u>4154

Fage 4 Date 20-Aug-87

UFFSITE TRANSFORTATION - NATIONAL DISPOSAL OFTION

Accident Frequencies and Range Factors

Agent Available and Released

| DURATION Time | | 20 MIN | 2 HRS | ; | 2 HKS | 2 HRS | 2 HKS | Å HRS | S HKS | 6 HKS | 6 HRS | ; | 6 HKS | SAH 6 | 6 HKS | 6 HKS | 6 HKS | S H 9 | 6 HRS | 6 HRS | 6 HRS | 6 HKS | A HRS | 6 HKS | 6 HKS | 6 HKS | 6 HES | 6 HKS | 6 HKS | 6 HFS | 6 HRS |
|-------------------|---|----------|----------|---------|--------------|---------|--------------|--------------|-------------|---------|---------|----------|---------|---------|-----------|---------|----------|----------|---------|----------|---------|--------------|----------|----------|----------|---------|-----------|-------------|--------------|-----------|------------|
| LBS. Enitted | | 1. 1E+01 | 2.6E+02 | 1 | 1.7E+02 | 9.0E+01 | 3.46+01 | 5.5E+00 | 2.5E-01 | 3.2E+00 | 2.5E-01 | ; | 2.5E-01 | 2.1E-03 | 2. IE-03 | 5.56+00 | 2.5E-01 | 2. IE-03 | 5.56+00 | 2. IE-03 | 5.56+00 | 2. IE-03 | 2. IE-03 | ł | ; | { | ; | : | ; | 1 | ł |
| LBS. Detonated | | 1.5E+02 | : | 1 | ; | 1 | ł | ł | 1 | ł | 1 | ł | ; | : | 1 | ; | ; | ; | ! | : | 1 | : | ; | 6.0E+00 | 1.6E+00 | 3.2E+00 | 3.26+01 | 6.5E+00 | 1.26+01 | 6. UE +U0 | 1.56+01 |
| 195. SPILLED | | ł | : | 1 | 1 | ł | 1 | 1 | 1 | ł | 1 | 1 | : | 1 | ; | : | ! | : | ; | ; | 1 | : | : | 3.0£+01 | 8. UE+OU | 1.6E+0! | 1.16+03 | 3.2E+01 | 5.8E+01 | 3.0E+01 | 7.36+01 |
| AGENT VAILABLE | | 1.2E+03 | 5. 3E+03 | 6.0E+03 | 6.8E+03 | 7.2E+03 | 2.7E+03 | 5. 3E+03 | 2.3E+03 | 9.2E+02 | 1.8€+03 | 6. 0E+03 | 6.8€+03 | 7.2E+03 | 2. 3E +03 | 1.6E+03 | 2.8E+03 | 1.4E+03 | 1.7E+03 | 1.7E+03 | 1.36+03 | 1.2E+03 | 2.7E+03 | 2. 3E+u3 | 9.2E+02 | 1.8€+03 | 2.3E+03 | 1.6€+03 | 2.6E+03 | 1.4E+03 | 1. 7E+i)3 |
| ARNGE Actor 4 | | 86 | 80 | ; | 89 | 1 | 68 | 76 | 1 | ! | ł | : | 76 | 1 | 76 | ۶۲ | : | 76 | 76 | 76 | 76 | 76 | 76 | ; | : | ; | 117 | 117 | ; | 117 | 117 |
| UNDA F | | 5.4E-08 | 5.2E-08 | N/A | 5. IE-12 | N/A | 5.2E-08 | 1.95-10 | N/A | N/A | N/A | N/A | 1.96-10 | N/A | 1.96-10 | 1.96-10 | N/A | 1.96-10 | 1.96-10 | 1.96-10 | 1.95-10 | 1.9E-10 | 1.96-10 | R/A | N/A | N/A | 1.35-08 | L. 3E-08 | N/A | 1.35-08 | 1.36-08 |
| RANGE Factor | | ł | ł | ł | { | ł | ł | 1 | 1 | ł | 1 | ł | ; | ł | ł | ; | ł | ł | ł | ł | 1 | 1 | ١ | ł | ł | ł | ł | ١ | ١ | 1 | ; |
| TEAD Freq. | | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | M/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A |
| RANGE Factor | | ł | 1 | ł | ł | 1 | ł | ł | 16 | ł | 4 | ł | ; | 1 | : | ł | 16 | ! | ł | ; | 1 | ; | ł | 83 | ; | 83 | ; | ; | 63 | ł | : |
| PUDA Freq. | | N/A | N/A | N/A | N/A | N/A | N/A | N/A | 2.2E-10 | N/A | 2,26-10 | N/A | N/A | N/A | N/A | N/A | 2.26-10 | N/A | N/A | N/A | N/A | N/A | N/A | 1.1E-09 | N/A | 1.16-09 | NA | N/A | 1.16-09 | N/A | N/A |
| RANGE Factor | | 86 | ł | ł | 82 | 1 | ł | 1 | 1 | 1 | 1 | ł | 95 | 1 | 35 | : | 1 | 1 | ł | 1 | 35 | 95 | : | ; | ; | ; | 16 | 1 | ! | : | : |
| PBA Freq. | | 1.7E-0B | N/A | N/A | 1.76-08 | N/A | N/A | N/A | N/A | N/A | N/A | N/A | 4.26-08 | N/A | 4.2E-08 | N/A | N/A | N/A | N/A | N/A | 4.2E-08 | 4.2E-ù8 | N/A | N/A | N/A | #/# | 5.5E-09 | N/A | N/6 | N/A | N/A |
| KANGE Factor | | ; | ł | ł | ł | 69 | ţ | ł | ł | ł | ł | ł | ; | 120 | ţ | ł | ; | ; | ł | 1 | ł | ł | ; | ; | ; | ; | 1 | ł | : | 1 | 1 |
| NAAP Freq. | | N/A | N/A | N/A | N/A | 1.4E-08 | N/A | N/A | N/A | N/A | N/A | A/A | N/A | 3.9E-08 | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/R |
| RANGE Factor | | 86 | ł | ; | ł | ł | ł | 1 | 1 | ; | 1 | ; | ł | ł | ł | | 60 | 69 | B | ; | 90 | 80 | ł | ; | ; | 1 | ; | 1 | 82 | 82 | 28 |
| LBAD Freq. | | 3.3E-08 | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | N/A | R/N | N/A | N/A | N/A | 4. 3E-08 | 4.3E-0B | 4.3E-ù8 | N/A | 4.3E-08 | 4.36-08 | N/A | 8/N | N/A | N/A | N/A | N/A | 9.4E-09 | 9.4E-09 | 9.4E-ú9 |
| KANGE Factor | | ! | ł | ł | 118 | ! | ł | ; | : | ł | ł | 1 | 23 | ł | 1 | ł | ; | ; | ; | ; | 1 | ! | ! | ; | : | ł | ; | ł | : | ł | 1 |
| AP6 Freq. | | N/A | N/A | N/A | 1.2E-08 | N/A | N/A | N/A | N/A | H/N | N/A | N/A | 4.0E-08 | N/A | N.A | N/A | A/A | N/A | N/A | R/N | N/A | N/Â | N/A | A/A | N/A | P./A | ₽/N | H .A | N/A | N/A | ₽/N |
| RANGE Factor | | 66 | ; | ł | 107 | 1 | ł | ł | 87 | 87 | 87 | ł | 87 | ł | 87 | 87 | 87 | (B | 87 | ł | 87 | 87 | ł | 11 | 22 | 73 | 73 | ٤: | 13 | 73 | 73 |
| ANAD Fred. | | 1.46-08 | N/A | N/A | 1.5E-08 | A:A | N/A | A/A | 4.56-08 | 4.5E-08 | 4.5E-0B | R/A | 4.5E-08 | N/A | 4.5E-08 | 4.56-08 | 4.5E-08 | 4.56-08 | 4.5E-ùē | H/A | 4.5E-08 | 4.56-08 | N.A. | 5.2E-09 | 5.26-09 | 5.2E-09 | 5. 2E -03 | 5.26-09 | 5. 2E - () 9 | 5. 2E-09 | 5. 2E-09 |
| NŪ. | ł | 2 | 5 | 5 | 5 | 5 | <u>:</u> | ÷ | * | +1 | = | <u>+</u> | ± | *1 | ± | ÷ | ± | Ξ | * | 5 | ŧ | * | 1 | 5 | ŝ | 5 | 5 | 5 | ŝ | 2 | <u>u''</u> |
| SCEN- 5 | | FCRVC | 5C66F | RCI GF | FCEHF | FCI VF | RESVE | FCB35 | CLAC | RCCEC | RCEHC | 50 GC | RURC | RCI VC | AC YUC | ECF6C | KEFLC | RCFVC | ĥŭũđC | RCOVC | KC KGC | FCRVC | F_SVC | RC0HC | FCCGC | KC:HC | SCHAC | RCFEC | FLFHC | ECPSC | FC DEGC |

See notes at end of table.

Page 5 Date 20-Aug-87

DFFSITE TRANSPORTATION - NATIONAL DISPOSAL OPTION

| Factors |
|---------------|
| Range |
| and |
| fr equenci es |
| Accident |

| eased |
|-------|
| fel |
| and |
| lable |
| Avai |
| Agent |

| DURAT I ON | TINE | | 6 HRS | A HKS | A UEC | |
|------------|-----------|-------|---------|----------|-----------|-----------|
| L85. | ENTTED | | : | : | ł | |
| LBS. | ETONATED | | 1.5€+01 | 1 15+01 | 10.30 | 1. 05. 01 |
| LBS. | SPILLED 0 | | 7.36+01 | A 76+07 | | 3.VETU1 |
| AGENT | ALLABLE | | 1.7E+03 | 1 16 401 | | 1.22703 |
| RANGE | FACTOR AV | | 117 | : = | | 111 |
| UNDA | FREQ. | | 1.35-00 | 1 76-00 | | 1.36-08 |
| RANGE | FACTOR | | ; | | : | : |
| TEAD | FREQ. | | M / A | | | N/N |
| RANGE | FACTOR | | ١ | | : | ł |
| PUDA | FREQ. | | N / N | | N/N | N/A |
| GANGE | FACTOR | | | 1 | 16 | 16 |
| pgq | FRFQ. | | | N/N - | 5.56-04 | 5. 56-09 |
| QANGE | FACTOR | | | ł | ; | 1 |
| NAAD | 6660 | | | E/H | K/A | N/A |
| 6.ANCC | FALTOR | | | : | 62 | 82 |
| 1 640 | CCC0 | | | W/A | 9.46-09 | 9.45-09 |
| DANCC | CALTOD | | | ł | ; | ! |
| 104 | R DCD | | | A/A | N/A | N/A |
| | RANDE | | | ł | 11 | 13 |
| | RARU | PREM. | | N/A | 5. 2E-09 | 5.26-09 |
| 9 | ž. | | ; | 5 | 67 - 1 | 12 |
| | - SEE | ANIC | | RCOVE | REAC | REEVC |
| r _ ' | 12 | 6 | | | | |

NGTES: 1. Scenarios 1-5 are per train mile; scenarios 6-15 are per exposure YEAr.

Furation time shown for scenarios with agent releases due to detonations and spills is for spill only. Ouration time for detonation is instantaneous.

HICKNOOD

「こうでした」とない。「「こうとうないない」を言いていていていた。「こうちょうない」を言いていたが、「なる」とないないです。

لمستعد والمحاطية والمحاطية والمحاطية والمحاطية والمحاطية والمحاطية والمحاطية والمحاطية والمحاطية والمحاطية والم

222222222

للللك والمسالين

. . . .



File: OFSITREG.WK1 Page 1 Date 27-Jul-87

OFFSITE TRANSFORTATION - BARGE

Agent Available and Released

REAL REAL

STATES IN

S

| SCENARIO | NC. | FREDUENCY PER TRIP | RANGE Factor | AGENT AVAILABLE | L B S. Spilled | LBS. DETONATED | LBS. Enitted | DURATION TIME |
|----------|-----|-----------------------|-----------------|--------------------|--------------------------|-------------------|-----------------|------------------|
| RIKHS | 2 | 3.09E-06 | 10 | 95200 | | | 175.4 | 74HR |
| BIKHE | 4 | 2.00E-06 | 10 | 95200 | | | 170.0 | 1HR |
| BIKHS | 6 | 3.00E-09 | 10 | 75200 | | | 87.7 | 24HR |
| BIKHF | 8 | 3.00E-09 | 10 | 95200 | | | 17.0 | 1HR |
| BIKHS | 10 | 3.00E-09 | 10 | 95200 | | | | |
| BIKHS | 12 | 3.00E-09 | 10 | 95200 | | | | •• |
| BIKHS | 14 | 3.00E-09 | 10 | 95200 | | | | |
| BIKHS | 16 | 3.00E-09 | 10 | 95200 | | | | |
| BIKHS | 19 | 3.53E-06 | 10 | 95200 | | | | |
| BIKHS | 20 | 3.53E-06 | 10 | 95200 | | | | |
| BIKHS | 21 | 1.64E-06 | 10 | 75200 | | | | |
| BIKHS | 22 | 3.00E-09 | 10 | 95200 | | | | |
| BIKHS | 23 | 2.06E-07 | 10 | 95200 | | | 8500. 0 | 1HP |



File: DFSITREG.WK1 Page 1 Date 27-Jul-87

OFFSITE TRANSPORTATION ~ SHIP INLAND

Agent Available and Released

C.

| SCENARIO | NO. | FREQUENCY PER TRIP | RANGE FACTOR | AGENT AVAILABLE | L 95. Spilled | L BS. DETONATED | LBS. ENITTED | DURATION TIME |
|----------|-----|-----------------------|------------------------|--------------------|-------------------------|---------------------------|-----------------|------------------|
| | | | ****** | | | | ******* | |
| LIKHS | 1 | 2.678-06 | 10 | 3.81E+06 | | | 1.23E+03 | 24HR |
| LIKHS | 2 | 1.418-07 | 10 | 3.81E+06 | ~- | | 1.23E+03 | 24HR |
| LIKHC | 3 | 6.68E-07 | 10 | 3.81E+06 | | | 1.19E+03 | 1HR |
| LIKHF | 4 | 3.53E-08 | 10 | 3.81E+06 | | | 1.19E+03 | 1HR |
| LIKHS | 5 | 1.19E-07 | 10 | 3.81E+06 | | ~- | 7.02E+02 | 24HR |
| LIKHS | 6 | 3.00E-09 | 10 | 3.81E+06 | | | 7.02E+02 | 24HR |
| LIKHF | 7 | 2.98E-08 | 10 | 3.81E+06 | | | 6.80E+02 | 1HR |
| LIKHS | 8 | 3.00E-09 | 10 | 3.81E+06 | | | 6.80E+02 | 1HR |
| LIKHS | 9 | 1.77E-06 | 10 | 3.81E+06 | | | ~- | |
| LIKHS | 10 | 1.43E-07 | 10 | 3.81E+06 | | | | |
| LIKHS | 11 | 3.61E-08 | 10 | 3.81E+06 | | | ** | |
| LIKHS | 12 | 3.00E-09 | 10 | 3.81E+06 | | | ~~ | |
| LIKHS | 13 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LIKHS | 14 | 3.00E-09 | 10 | 3.81E+06 | | | | ** |
| LIKHS | 15 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LIKHS | 16 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LIKHF | 17 | 3.00E-09 | 10 | 3.81E+06 | | | 8.50E+01 | IHR |
| LIKHF | 18 | 3.00E-09 | 10 | 3.B1E+06 | | | 8.50E+01 | 1HR |
| LIKHS | 19 | 5,84E-07 | 10 | 3.81E+06 | | | | |
| LIKHS | 20 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LIKHS | 21 | 1.61E-06 | 10 | 3.81E+06 | | | | |
| LIKHS | 22 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LIKHS | 23 | 2.70E-09 | 10 | 3.81E+06 | | | 6.05E+04 | ihr |

ULIANDAR KENDIKE NUT KENDING NUT KENDING
File: OFSITREG.WK1 Page 1 Date 27-Jul-87

OFFSITE TRANSPORTATION - SHIP COASTAL

Agent Available and Released

| | | FREQUENCY | RANGE | AGENT | LBS. | LBS. | LBS. | DURATION |
|----------|-----|-----------|--------|-----------|---------|-----------|-----------|----------|
| SCENARIO | NO. | PER TRIP | FACTOR | AVAILABLE | SPILLED | DETONATED | EMITTED | TIME |
| | | | *** | | | | * | |
| LCKHS | 1 | 1.24E-06 | 10 | 3.81E+06 | | | 1.23E+03. | 24HR |
| LCKHS | 2 | 6.51E-08 | 10 | 3.81E+06 | | | 1.23E+03 | 24HR |
| LCKHF | 3 | 3.09E-07 | 10 | 3.81E+06 | | | 1.19E+03 | 1HR |
| LCKHF | 4 | 1.63E-08 | 10 | 3.81E+06 | | | 1.19E+03 | 1HR |
| LCKHS | 5 | 7.97E-08 | 10 | 3.81E+06 | | | 7.02E+02 | 24HR |
| LCKHS | 6 | 3.00E-09 | 10 | 3.81E+06 | | | 7.02E+02 | 24HR |
| LCKHF | 7 | 1.99E-08 | 10 | 3.81E+06 | ~- | | 6.80E+02 | 1HR |
| LCKHF | 8 | 3.00E-09 | 10 | 3.81E+06 | | | 6.80E+02 | 1HR |
| LCKHS | 9 | 5.13E-07 | 10 | 3.81E+06 | | | | |
| LCKHS | 10 | 4.16E-08 | 10 | 3.81E+06 | | | | |
| LCKHS | 11 | 1.05E-08 | 10 | 3.81E+06 | | | | |
| LCKHS | 12 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LCKHS | 13 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LCKHS | 14 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LCKHS | 15 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LCKHS | 16 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LCKHF | 17 | 3.00E-09 | 10 | 3.81E+06 | | | 8.50E+01 | 1HR |
| LCKHF | 18 | 3.00E-09 | 10 | 3.81E+06 | | | 8.50E+01 | 1HR |
| LCKHS | 19 | 2.70E-07 | 10 | 3.81E+06 | | | | •• |
| LCKHS | 20 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LCKHS | 21 | 4.67E-07 | 10 | 3.81E+06 | | | | |
| LCKHS | 22 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LCKHS | 23 | 2.708-09 | 10 | 3.81E+06 | | | 6.05E+04 | 1HR |





File: OF5ITREG.WK1 Page 1 Date 27-Jul-87

OFFSITE TRANSPORTATION - SHIP HIGH SEAS

Agent Available and Released

| | | FREQUENCY | RANGE | AGENT | LBS. | LBS. | LBS. | DURATION |
|----------|-----|-----------|--------|-----------|---------|-----------|----------|----------|
| SCENARID | NO. | PER TRIP | FACTOR | AVAILABLE | SPILLED | DETUNATED | ENTITED | 1145 |
| | | | | | | | | |
| LSKHS | 1 | 2.79E-07 | 10 | 3.81E+06 | | | 1.23E+03 | 24HR |
| LSKHS | 2 | 1.47E-08 | 10 | 3.81E+06 | | | 1.23E+03 | 24HP |
| LSKHF | 3 | 6.98E-08 | 10 | 3.81E+06 | | | 1.19E+03 | 188 |
| LSKHF | 4 | 3.67E-09 | 10 | 3.81E+06 | | | 1.19E+03 | 1 H R |
| LSKHS | 5 | 3.67E-09 | 10 | 3.81E+06 | | | 7.02E+02 | 24HR |
| LSKHS | 6 | 3.00E-09 | 10 | 3.81E+06 | | | 7.02E+02 | 24HR |
| LSKHF | 7 | 1.56E-08 | 10 | 3.81E+06 | | | 6.80E+02 | 1HR |
| LSKHF | 8 | 3.00E-09 | 10 | 3.81E+06 | | | 6.808+02 | 1HR |
| LSKHS | 9 | 4.33E-08 | 10 | 3.81E+06 | | | | |
| LSKHS | 10 | 3.51E-09 | 10 | 3.81E+06 | | | | |
| LSKHS | 11 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LSKHS | 12 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LSKHS | 13 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LSKHS | 14 | 3.00E-09 | 10 | 3.B1E+06 | | | | |
| LSKHS | 15 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LSKHS | 16 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LSKHF | 17 | 3.00E-09 | tů | 3.81E+06 | | | 8.50E+01 | 1HR |
| LSKHF | 18 | 3.00E-09 | 10 | 3.81E+06 | | | 8.50E+01 | 1HR |
| LSKHS | 19 | 6.10E-08 | 10 | 3.81E+06 | ** | | | |
| LSKHS | 20 | 3.94E-08 | 10 | 3.81E+06 | | | | |
| LSKHS | 21 | 3.00E-09 | 10 | 3.81E+06 | | | | |
| LSKHS | 22 | 3.00E-09 | 10 | 3.81E+06 | ** | | | |
| LSKHS | 23 | 2.70E-09 | 10 | 3.81E+06 | | | 6.05E+04 | 1HR |

I-130

OFFSITE TRANSPORTATION - AIR LEG FOR C141 AIRCRAFT

Agent Available and Released

| CCENADIO | NU | APS | RANGE | LBAD | RANGE | AGENT | LBS. | LBS. | LBS. | DURATION |
|----------|------------|----------|---------|----------|-------|-------|----------|----------|----------|----------|
| | | razz. | THE TUR | | | | | | | |
| ABKHS | 1T | 1.20E-08 | 11 | N/A | | 3400 | 2.89E+03 | | | 24 HR |
| ABPHC | 17 | N/A | | 1.20E-08 | 11 | 1498 | 1.05E+03 | 2.25E+02 | | 24 HR |
| ABPVC | 11 | N/A | | 1.205-08 | 11 | 768 | 5.38E+02 | 1.15E+02 | | 24 HR |
| ABORC | ίŢ | N/A | | 1.208-08 | 11 | 1044 | 7.31E+02 | 1.57E+02 | | 24 HR |
| ABRVC | 11 | N/A | | N/A | | 1044 | 7.31E+02 | 1.57E+02 | | 24 HR |
| ABRGC | 11 | N/A | | 1.20E-08 | 11 | 542 | 4.49E+02 | 9.60E+01 | | 24 HR |
| ABRVC | 11 | N/A | | 1.202-08 | 11 | 600 | 4.20E+02 | 9.00E+01 | | 24 HR |
| ABKHF | 21 | 9.90E-09 | 11 | N/A | | 3400 | | | 1.7E+02 | 1 HR |
| ABPHC | 21 | N/A | | 9.908-09 | 11 | 1498 | | 3.74E+02 | 5.6E+01 | 20 MIN |
| ABPVC | 2T | N/A | | 9.90E-09 | 11 | 768 | | 1.72E+02 | 1.4E+01 | 20 MIN |
| ABBEC | 21 | N/A | | 9.90E-09 | 11 | 1044 | | 2.61E+02 | 7.8E+01 | 20 MIN |
| ABQVC | 2T | N/A | | N/A | | 1044 | | 2.61E+02 | 2.0E+01 | 20 MIN |
| ABRGC | 21 | N/A | | 9.905-09 | 11 | 642 | | 1.60E+02 | 4.8E+01 | 20 MIN |
| ABRVC | 21 | N/A | | 9.90E-09 | 11 | 600 | | 1.50E+02 | 1.1E+01 | 20 MIN |
| ABKHE | 31 | 0.00E+00 | | N/A | | 3400 | | | | |
| ABPHC | 3T | N/A | | 0.00E+00 | | 1498 | | | | |
| ABPVC | 31 | N/A | | 0.00E+00 | | 768 | | | | |
| ABGSC | 31 | N/A | | 0.00E+00 | | 1044 | | | | |
| ABOVC | 31 | N/A | | N/A | | 1044 | | | | |
| ABRGC | 31 | N/A | | 0.00E+00 | | 642 | | | | |
| ABRVC | 31 | N/A | | 0.00E+00 | | 600 | | | | |
| ABKHE | 4T | 1.208-08 | 13 | N/A | | 3400 | | 0.0CE+00 | 1.70E+02 | |
| ABPHC | 4T | N/A | | 1.20E-08 | 13 | 1498 | | 3.74E+02 | 5.60E+01 | |
| ASPVC | 4T | N/A | | 1.20E-08 | 13 | 763 | | 1.92E+02 | 1.40E+01 | |
| ABOOC | 4T | N/A | | 1.208-08 | 13 | 1044 | | 2.61E+02 | 7.80E+01 | |
| ABQVC | 41 | N/A | | N/A | | 1044 | | 2.615+02 | 2,00E+01 | |
| ABREC | 4 T | N/A | | 1.20E-08 | 13 | 642 | | 1.60E+02 | 4.B0E+01 | |
| ABRVC | 4T | N/A | | 1.20E-09 | 13 | 600 | | 1.50E+02 | 1.10E+01 | |
| ABKHE | 5T | 0.00E+00 | | N/A | | 3400 | | | | |
| ABPHC | 5T | N/A | | 2.61E-10 | 57 | 1498 | 5.80E+01 | 1.20E+01 | | 24 HR |
| ABPVC | 51 | NIA | | 2.61E-10 | 57 | 768 | 3.00E+01 | 6.00E+00 | | 24 HR |
| ABQGC | 57 | N/A | | 2.61E-10 | 57 | 1044 | 7.20E+01 | 1.40E+01 | | 24 HR |
| ABOVC | 5T | N/A | | N/A | | 1044 | 7.205+01 | 1.40E+01 | | 24 HR |
| ABRGE | 5T | N/A | | 1.88E-09 | 57 | 542 | 5.998+02 | 4.30E+01 | | 24 HE |
| ABRVC | 51 | N/A | | 1.885-09 | 57 | 600 | 5.602+02 | 4.00E+01 | | 24 HR |
| ABKHS | 1F | 4.70E-07 | 11 | N/A | | 5400 | 2.89E+03 | | | 24 HR |
| ABPHC | 1F | N/A | | 3.40E-07 | 11 | 1499 | 1.05E+03 | 2.25E+02 | | 24 HR |
| ABPVC | 1F | N/A | | 3.40E-07 | 11 | 768 | 5.38E+02 | 1.15E+02 | | 24 48 |
| ABQGC | 1F | N/A | • | 3.40E-07 | 11 | 1044 | 7.31E+02 | 1.57E+02 | | 24 HR |
| ABQVC | ١F | N/A | | N/A | | 1044 | 7.31E+02 | 1.578+02 | | 24 HR |
| ABROC | 1F | N/A | | 3.40E-07 | 11 | 642 | 4.49E+02 | 9.60E+01 | | 24 HR |
| ABRVC | IF | N/A | | 3.402-07 | 11 | 600 | 4.20E+02 | 9.09E+01 | | 24 HR |
| ABKHE | 2F | 3.80E-07 | 11 | N/A | | 3409 | | | 1.75+02 | ЧЕ |
| ABPHC | 25 | N/A | | 7 80E-07 | 11 | 1499 | | 3.74E+01 | 5.52-01 | 20 MIN |
| ABPVC | 2F | N/A | | 2.808-07 | 11 | 768 | | 1.528+02 | 1.45+91 | 20 MIN |
| A8060 | 25 | N/A | | 2.80E-07 | 11 | 1044 | | 2.61E+02 | 7.8E+01 | 20 MIN |
| AFGVC | 2F | N/A | | N/4 | | 1044 | | 2.61E+02 | 2.0E+01 | 20 MIN |
| APRSC | 2F | N/A | | 2.898-07 | 11 | 542 | | 1.605+02 | 4.8E+01 | 20.518 |







SCORESS- BUSSESSE SECURICE ACCESSES SEPARATE SCORESSES

OFFSITE TRANSPORTATION - AIR LES FOR C141 AIRCRAFT

Agent Available and Released

| SCENARIO | NC. | AP6 Freq. | RANGE Factor | LBAD FREQ. | RANGE Factor | AGENT AVAILABLE | LBS. SPILLED | LBS. DETENATED | LBS. EMITTED | DURATION TIME |
|----------|--------------|-------------------|-----------------|---------------|-----------------|--------------------|-----------------|-------------------|-----------------|------------------|
| ABRVC | 2F | N/A | | 2.B0E-07 | 11 | 600 | | 1.50E+02 | 1.1E+01 | 20 MIN |
| ABKHF | 3F | 0.00E+00 | | N/A | | 3400 | | | | ** |
| ABPHC | 3F | N/A | | 0.002+00 | | 1498 | | | | |
| ABPVC | 3F | N/A | | 0.00E+00 | | 768 | | | | |
| ABQSC | 3F | N/A | | 0.00E+00 | | 1044 | | | | |
| ABOVC | 3F | N/A | | N/A | | 1044 | | | | |
| ABRGC | 3F | N/A | | 0.00E+00 | | 642 | | | | |
| ABRVC | 3F | N/A | | 0.002+00 | | 600 | | | | |
| ABKHE | 4F | 3.60E-08 | 13 | N/A | | 3400 | | 0.00E+00 | 1.70E+02 | |
| ARPHC | 4F | N/A | | 2.60E-0B | 13 | 1499 | | 3.74E+02 | 5.60E+01 | |
| ARPVC | 4F | N/A | | 2.60E-08 | 13 | 768 | | 1.975+07 | 1.40E+01 | |
| ABOSC | 45 | N/A | | 2.60F-08 | 13 | 1044 | | 7. 61E+07 | 7.802+01 | |
| AROVC | AF | N/A | | N/A | | 1044 | | 7 61E+07 | 2 005+01 | |
| ARRSC | 45 | N/A | | 2 605-08 | 13 | 647 | | 1 605+02 | 4 805+01 | |
| ADDUC | 46 | N/A | | 2.000 00 | 13 | 140 | | 1 505+07 | 1 105+01 | |
| ADEVUC | - TF - 55 | 0.00 5 400 | | 2.00C-V0 | 1.3 | 7.100 | | 1. JVE+V2 | 1.102-01 | |
| ADDUC | JF | V.CUETUU N/A | | 5 405-10 | 67 67 | 1400 | 5 005101 | 1 705+01 | | |
| ADDUC | JF EE | R/ B M / A | | 5 405-10 | J/ 57 | 07P1 | 7.005.01 | 1.200401 | | 24 10 |
| ADOCC | - GP EF | 11/H N/A | | 5.40E-10 | 5/ | 00 | 3.00EFUL | 5.00E300 | | 24 HR |
| ADDUC | | N/H | | 3.40E-10 | 2/ | 1044 | 7.200401 | 1.495+91 | | 24 118 |
| ASUVL | 31 | N/R | | N/R | | 1044 | 7.208+01 | 1.402+01 | | 24 HK |
| ABKOL | 10 | N/A | | 3.885-99 | 3/ | 642 | 3.99E+02 | 4.30E+01 | | 24 HK |
| ABRVC | - 5F | N/A | | 3.88E-09 | 57 | 600 | 5.60E+02 | 4.00E+01 | | 24 HR |
| ABKHS | 11 | 7.00E-08 | 11 | N/A | | 3400 | 2.89E+03 | | | 24 HR |
| ABPHC | 11 | N/A | | 7.00E-08 | 11 | 1498 | 1.05E+03 | 2.25E+02 | | 24 HR |
| ABPVC | 1L | N/A | | 7.00E-08 | 11 | 768 | 5.38E+02 | 1.15E+02 | | 24 HR |
| ABGGC | 11 | N/A | | 7.00E-08 | 11 | 1044 | 7.31E+02 | 1.57E+02 | | 24 HR |
| ABBAC | 1L | N/A | | N/A | | 1044 | 7.31E+02 | 1.57E+02 | | 24 HR |
| ABREC | IL | N/A | | 7.00E-08 | 11 | 642 | 4.49E+02 | 9.60E+01 | | 24 HR |
| ABRVC | 11 | N/A | | 7.00E-08 | 11 | 600 | 4.20E+02 | 9.90E+01 | | 24 HR |
| ABYHF | 2L | 5.708-08 | 11 | N/A | | 3400 | | | 1.7E+02 | 1 HR |
| ABPHC | 2L | N/A | | 5.70E-08 | 11 | 1498 | | 3.74E+02 | 5.6E+01 | 20 MIN |
| A89VC | 2L | N/A | | 5.70E-08 | 11 | 768 | | 1.92E+02 | 1.4E+01 | 20 MIN |
| A806C | 2L | N/A | | 5.70E-08 | 11 | 1044 | | 2.61E+02 | 7.8E+01 | 20 MIN |
| ABQVC | 2L | N/A | | N/A | | 1044 | | 2.61E+02 | 2.0E+01 | 20 MIN |
| ABREC | 21L | N/A | | 5.70E-08 | 11 | 642 | | 1.50E+02 | 4.8E+01 | 20 MIN |
| ABRVC | 2L | N/A | | 5.70E-08 | 11 | 600 | | 1.50E+02 | 1.1E+01 | 20 MIN |
| ABKHF | 3L | 0.00E+00 | | N/A | | 3400 | | | | |
| ABPHC | 3L | N/A | | 0.00E+00 | | 1498 | | | ** | |
| ABPVC | 3L | N/A | | 0.00E+00 | | 768 | | | | |
| A906C | 3L | N/4 | | 0.00E+00 | | 1044 | | | | |
| ABGVC | 3L | N/A | | N/A | | 1044 | | | | |
| ABREC | 21 | N/A | | 0.00E+00 | | 642 | | | | |
| ABRVC | 31 | N/A | | 0.00E+00 | | 600 | | | | |
| ABKHE | 4 | 8.408-08 | 13 | N/A | | . 3400 | | 0.005-00 | 1.70E+00 | |
| ABPHC | 41 | N/A | | 8.408-08 | 13 | 1498 | | 3.74E+02 | 5. 60E+01 | |
| ABPVC | 41 | N/A | | 8.405-09 | 17 | 743 | | 1.975+02 | 1.40F+01 | |
| ARGSC | 41 | N/A | | 8.405-08 | 17 | 1044 | | 7.515+07 | 7.867+01 | |
| 490.00 | 41 | N/A | | N/4 | | . 1044 | | 7.615+07 | 2.00E+01 | |





RADERER

20205050

22222

I-132

File: OFSITREG.WK1 Page 3 Date 20-Aug-87

OFFSITE TRANSPORTATION - AIR LEG FOR C141 AIRCRAFT

Agent Available and Released

| SCENARIO | NO. | APG FRED. | RANGE Factor | LBAD FREQ. | RANGE Factor | AGENT AVAILABLE | LBS. SPILLED | LBS. DETONATED | LBS. Emitted | DURATION TIME |
|----------|-----|--------------|-----------------|---------------|-----------------|--------------------|-----------------|-------------------|-----------------|------------------|
| | | ******** | | | | | | ******** | | |
| ABRGC | 4L | N/A | | 8.40E-08 | 13 | 642 | | 1.60E+02 | 4.80E+01 | |
| ABRVC | 4L | N/A | | 8.40E-08 | 13 | 600 | | 1.50E+02 | 1.10E+01 | |
| ABKHF | 51 | 0.00E+00 | | N/A | | 3400 | | | | |
| ABPHC | 5L | N/A | | 1.80E-09 | 57 | 1498 | 5.80E+01 | 1.20E+01 | | 24 HR |
| ABPVC | 5L | N/A | | 1.80E-09 | 57 | 768 | 3.00E+01 | 6.00E+00 | | 24 HR |
| ABOGC | 5L | N/A | | 1.80E-09 | 57 | 1044 | 7.20E+01 | 1.40E+01 | | 24 HR |
| ABQVC | 5L | N/A | | N/A | | 1044 | 7.20E+01 | 1.40E+01 | | 24 HR |
| ABRGC | 5L | N/A | | 1.29E-08 | 57 | 642 | 5.99E+02 | 4.30E+01 | | 24 HR |
| ABRVC | 5L | N/A | | 1.29E-08 | 57 | 600 | 5.60E+02 | 4.00E+01 | | 24 HR |





File: OFSITRE6.WK1 Page 1 Date 20-Aug-87

ł

and a start was strong and strong a

OFFSITE TRANSPORTATION - AIR LEG FOR C5A AIRCRAFT

Agent Available and Released

| SCENARIO | NC. | APS FREQ. | RANGE Factor | LBAD FREQ. | RA nge Factor | AGENT AVAILABLE | LBS. SPILLED | LBS. DETONATED | LBS. EMITTED | DURATION TIME |
|----------|-----|--------------|-----------------|---------------|-------------------------|--------------------|-----------------|-------------------|-----------------|------------------|
| | | | | | ********* | | | | | |
| AAKHS | 17 | 1.00E-07 | 11 | N/A | | 13600 | 1.16E+04 | | | 24 HR |
| AAPHC | 17 | N/A | | 1.00E-07 | 11 | 5790 | 4.19E+03 | 8.98E+02 | | 24 HR |
| AAPVC | 1T | N/A | | 1.00E-07 | 11 | 3072 | 2.15E+03 | 4.61E+02 | | 24 HR |
| AABGC | 11 | N/A | | 1.00E-07 | 11 | 4176 | 2.92E+03 | 6.262+02 | | 24 HR |
| AAQVC | 11 | N/A | | N/A | | 4176 | 2.92E+03 | 6.26E+02 | | 24 HR |
| AAR6C | 11 | N/A | | 1.00E-07 | 11 | 2568 | 1.80E+03 | 3.85E+02 | | 24 HR |
| AARVC | 11 | N/A | | 1.00E-07 | 11 | 2400 | 1.68E+03 | 3.60E+02 | | 24 HR |
| AAKHF | 21 | 8.20E-08 | 11 | N/A | | 13600 | | | 6.8E+02 | 1 HR |
| AAPHC | 21 | N/A | | 8.20E-08 | 11 | 5790 | | 1.50E+03 | 2.2E+02 | 20 MIN |
| AAPVC | 2T | N/A | | 8.20E-08 | 11 | 3072 | | 7.58E+02 | 5.3E+01 | 20 HIN |
| AAQSC | 21 | N/A | | 8.20E-08 | 11 | 4176 | | 1.04E+03 | 3.1E+02 | 20 MIN |
| AAGVC | 21 | N/A | | N/A | | 4176 | | 1.042+03 | 7.8E+01 | 20 MIN |
| AARSC | 21 | N/A | ** | 8.202-03 | 11 | 2568 | | 6.42E+02 | 1.9E+02 | 20 MIN |
| AARVC | 21 | N/A | | 8.20E-08 | 11 | 2400 | | 6.00E+02 | 4.5E+01 | 20 MIN |
| AAKHE | 31 | 2.90E-08 | 12 | N/A | | 13600 | | | 6.8E+02 | 1 HR |
| AAPHC | 31 | N/A | | 2.908-08 | 12 | 5990 | | 1.50E+03 | 2.2E+02 | 20 MIN |
| AAPVC | 31 | N/A | | 2.90E-08 | 12 | 3072 | | 7.63E+02 | 5.8E+01 | 20 MIN |
| AAQGC | 31 | N/A | | 2.908-08 | 12 | 4176 | | 1.04E+03 | 3.1E+02 | 20 MIN |
| AABVC | 31 | N/A | | N/A | | 4176 | | 1.04E+03 | 7.8E+01 | 20 MIN |
| AARGE | 31 | N/A | | 2.90E-08 | 12 | 2568 | | 6.42E+02 | 1.9E+02 | 20 MIN |
| AARVC | 31 | N/A | | 2.90E-08 | 12 | 2400 | | 5.00E+02 | 4.5E+01 | 20 MIN |
| AAKHE | 4T | 1.00E-07 | 13 | N/A | | 13600 | | | 6.8E+02 | 1 KR |
| AAPHC | 41 | N/A | | 1.COE-07 | 13 | 5990 | | 1.50E+03 | 2.2E+02 | 20 MIN |
| AAPVE | 4T | N/A | | 1.00E-07 | 13 | 3072 | | 7.6SE+02 | 5.3E+01 | 20 NIN |
| AARAC | 4T | N/A | | 1.00E-07 | 13 | 4176 | | 1.04E+03 | 3.1E+02 | 20 MIN |
| AABVC | 41 | N/A | | N/A | | 4176 | | 1.04E+03 | 7.8E+01 | 20 MIN |
| AARSE | 41 | N/A | | 1.00E-97 | 13 | 2568 | | 6.42E+02 | 1.92+02 | 20 MIN |
| AARVC | 4T | N/A | | 1.00E-07 | 13 | 2400 | | 6.00E+02 | 4.5E+01 | 20 HIN |
| AAKHE | 51 | 0.00E+00 | | N/A | | - 3400 | | | | |
| AAPHC | 51 | N/A | | 2.10E-09 | 57 | 1499 | 5.80E+01 | 1.20E+01 | | 24 HR |
| AAPVE | 51 | N/A | | 2.10E-09 | 57 | 768 | 3.00E+01 | 6.00E+00 | | 24 HR |
| AADGC | 51 | N/A | | 2.10E-07 | 57 | 1044 | 7.20E+01 | 1.40E+01 | | 24 HR |
| AARVC | 51 | N/A | | N/4 | | 1044 | 7.20E+01 | 1.40E+01 | | 24 HR |
| AARSC | 57 | N/A | | 1.512-08 | 57 | 542 | 5.99E+02 | 4.30E+01 | | 24 HS |
| AARVC | 51 | NZA | | 1.51E-08 | 57 | 600 | 5.602+02 | 4.002+01 | | 24 HR |
| AAXHS | 15 | 4.00E-36 | 11 | N/A | | - 13600 | 1.16E+04 | (| | 24 HR |
| AAPSC | 1F | N/A | | 2.80E-06 | 15 | 3323 | 2.33E+03 | 4.99E+02 | | 24 HR |
| AAPHE | 1F | N/A | | 2.80E-06 | 1 | 5970 | 4.19E+03 | 8.792+02 | | 24 HR |
| AAFVC | 1F | N/A | | 2.80E-06 | 1 | 3072 | 2.15E+03 | 4.61E+02 | | 24 HR |
| AAOGE | 15 | N/A | | 2.802-06 | 1 | 4176 | 2.92E+03 | 5 6.25E+02 | | 24 HR |
| AAGVC | 1F | N/A | | N/A | | - 4176 | 2.92E+03 | £.26E+02 | | 24 HR |
| AARGC | 15 | N/A | | 2.50E-06 | 1 | 1 2569 | 1.802+0. | 3.85E+92 | ~~ | 24 HR |
| AAFVO | 15 | N/A | | 2.90E-06 | 1 | 2400 | 1.68E+03 | 3.60E+02 | | 24 HR |
| AALLE | 25 | 3.20E-04 | 11 | N/A | - | - 13600 | | | 5.3E+02 | 1 HR |
| AGEHE | 75 | N/A | | 2.305-06 | 1 | 1 5990 | | 1.502+03 | 2.2E+02 | 2 20 MIN |
| AAFVC | 75 | N/A | | 2.302-05 | . 1 | 1 3072 | | 7.68E+02 | 5.8E+0 | 20 MIN |
| 44050 | 25 | N/A | | 2.305-04 | • | 1 4176 | | 1,)4E+03 | 3.1E+02 | 20 MIN |
| AASLO | 25 | N/A | | N/A | - | - 4174 | | 1.04F+03 | 7.95+0 | 20 N.N |

DEFSITE TRANSPORTATION - AIR LEG FOR CSA AIRCRAFT

Agent Available and Released

| SCENARIO | NO. | APG FREQ. | RANGE Factor | LBAD FREQ. | Range Factor | AGENT AVAILABLE | LØS. SPILLED | LBS. DETONATED | LBS. EMITTED | DURATION TIME |
|----------|--------------|-----------------|-----------------|----------------------|-----------------|--------------------|-----------------|-------------------|--------------------|------------------|
| AARGC | 2F | N/A | | 2.30E-06 | 11 | 2568 | | 6.42E+02 | 1.9E+02 | 2C MIN |
| AARVC | 2F | N/A | | 2.30E-06 | 11 | 2400 | | 6.00E+02 | 4.5E+01 | 20 NIN |
| AAKHE | 3F | 0.00E+00 | | N/A | | 13600 | | | | |
| AAPHC | 3F | N/A | | 0.002+00 | 12 | 5990 | | | | |
| AAPVC | 3F | N/A | | 0.00E+00 | 12 | 3072 | | | | |
| 00060 | 3E | N/A | | 0.00F+00 | 12 | 4176 | | | | |
| AGOVC | 3F | N/A | | N/A | | 4176 | | | | |
| AARGC | 3E | N/A | | 0.00E+00 | 12 | 2568 | | | | |
| AARVC | 3F | N/A | | 0.00E+00 | 12 | 2400 | | | | |
| | 45 | 3.00E-07 | 13 | N/A | | 13600 | | · | A 8E+02 | 1 HR |
| | AF | N/A | | 2 20F-07 | 13 | 5990 | | 1.50E+03 | 2 25+02 | 20 NTN |
| | AF | N/A | | 7 20E-07 | 17 | 3072 | | 7 485+07 | 5 9F+01 | 20 MIN |
| 12000 | 45 | N/A | | 2 205-07 | 13 | 4174 | | 1 045+03 | 3.15+07 | 20 015 |
| | 46 | N/A | | N/A | | A174 | | 1 046+07 | 7 95+01 | 20 NIN 20 NIN |
| | 46 | N/A | | 2 205-07 | 17 | 11/0 | | 1.04E+0J | 1.00101 | 20 HIN 20 MTH |
| AAQUC | 45 | 97.15 N/A | | 2.200-07 | 10 | 2300 | | 6.42ET02 | 1.767VL 8 FELOI | 20 NEN 20 NEN |
| AAFUC | - TT F.C | 0 005400 | | 1.LUE-VI | 13 | 7400 | | 0.VVE+V2 | 4.JETV1 | 20 111 |
| ANDUC | JF | 0.002700 N/A | | 8/8 4 505-09 | | 1400 | 5 905101 | 1 205401 | | 14 UD |
| AADUC | Jr FC | N/A N/A | | 4.30E-07 | 57 | 1910 | 3.000-01 | 1.20CTV1 | | 24 65 |
| | 50 | 117 H 117 A | | 4 505-00 | 16 57 | 100 | 7 205101 | 1. 40E+00 | | 21 00 |
| AADUC | JF | 117 M 317 A | | 4.JUC-07 | JI | 1044 | 7.205101 | 1.400101 | | 24 AN 04 HD |
| HHEVL | Jr Fr | 11/H 11/A | | N/H 7 975 AG | | 1044 | F 005:07 | 1.496+91 | | 24 68 |
| AADUC | JF 55 | 517 H H 7 A | | 3.23C-V5 7 37E A0 | J/ 57 | 342 | 0.77E7VL | 4.000101 | | 24 05 |
| AAKUC | - 10 - 11 | N/A 5 (AC A7 | | 3.235-98 | 2/ | 600 | 3.602+02 | 4.002+01 | | 24 MK |
| AADUC | 10 | 3.602-07 | 11 | N/H E (05 07 | | 15600 | 1.16E*V4 | 0.000.00 | | 24 FR |
| AADUC | 11 | N/H | | J.6VE-97 | 11 | 5750 | 4.175+03 | 8.97E+U2 | | 24 88 |
| RAPVL | 1 L . | N/A | | 3.60E-07 | 11 | 3072 | 2.132+93 | 4.61E+02 | | 24 HR |
| ANGUO | 11 | N/A | | 3.60E-07 | 11 | 41/6 | 2.92E+03 | 5.26E+02 | | 24 HK |
| RREAC | 10 | N/A | | N/A | | 41/5 | 2.922+03 | 6.26E+02 | | 24 HR |
| AARGU | n. | N/A | | 5.60E-07 | 11 | 2568 | 1.802+03 | 3.85E+02 | | 24 HR |
| AARVC | 16 | N/A | | 5.60E-0/ | 11 | 2400 | 1.68E+03 | 3.60E+02 | | 24 HR |
| AAKHE | 26 | 4.60E-07 | 11 | N/A | | 13600 | | | 6.8E+02 | I HR |
| ARPHC | 2L | N/A | | 4.00E-07 | 11 | 5990 | | 1.502+03 | 2.2E+02 | 20 MIN |
| AAPVC | 2L | NZA | | 4.60E-07 | 11 | 3072 | •- | 7.68E+02 | 5.3E+01 | 20 MIN |
| AABEC | 2L | N/A | | 4.60E-07 | 11 | 4176 | | 1.04E+03 | 3.1E+02 | 20 MIN |
| AAGVC | 2L | N/A | | N/A | | 4176 | | 1.04E+03 | 7.SE+01 | 20 MIN |
| AAREC | 2L | NZA | | 4.60E-07 | 11 | 2568 | | 6.42E+02 | 1.9E+02 | 20 MIN |
| AARVC | 21 | N/A | | 4.60E-07 | 11 | 2400 | | 6.00E+02 | 4.5E+01 | 20 MIN |
| AAKHF | 3L | 0.00E+00 | | N/A | | 13600 | | | | |
| AAPHC | 3L | N/A | | 0.00E+00 | 12 | 5990 | | | | |
| AAPVC | ΓL | N/A | | 0.00E+00 | 12 | 3072 | | | | |
| JARGC | 3L | N/A | | 0.00E+09 | 12 | 4176 | | | | |
| AAQVC | 3c | N/A | | N/A | | 4175 | | | | |
| AAR6C | 3L | N/ A | | 0.00E+00 | 12 | 2563 | •- | | | |
| AARVC | 3L | N/A | | 0.00E+00 | 12 | 2400 | | | | |
| hay HF | 41 | 6.70E-07 | 13 | N/A | | 17600 | | | a.8E÷02 | 1 HR |
| Paphc | ŧ٤ | N - A | | 6.70E-07 | 13 | 5990 | | 1.50E+03 | 2.2E+02 | 20 MIN |
| AAPVC | 41 | N/A | | 6.70E-07 | : 7 | 3072 | | 7.682+02 | 5.3E+01 | 20 MIN |
| AAGEC | 4L | N/A | | 6.70E-07 | 17 | 417a | | 1.04E+03 | 0.1E+02 | 20 MIN |





 ~ 2

1212

1.1

NAMES AND ADDRESS PARAMAN

File: OFSITRE6.#K1 Page 3 Date 20-Aug-87

OFFSITE TRANSPORTATION - AIR LEG FOR CSA AIRCRAFT

Agent Available and Released

| SCENARIO | NC. | APG FREQ. | RANGE Factor | LBAD FREQ. | RANGE Factor | AGENT AVAILABLE | LBS. SFILLED | LBS. DETONATED | LBS. ENITTED | DURATION TIME |
|----------|-----|--------------|-----------------|---------------|-----------------|--------------------|-----------------|-------------------|-----------------|------------------|
| AADUD | | N/A | | N/G | | 4175 | | 1.04E+03 | 7.8E+01 | 20 MIN |
| AADCC | 41 | N/N | | 6.705-07 | 13 | 2568 | | 6.42E+02 | 1.9E+02 | 20 MIN |
| AAKEL | 9L | N/ M | | 6.70E -07 | 17 | 2400 | | 6.00E+07 | 4.5E+01 | 20 MIN |
| AARVC | 4L | N/ 8 | | 0./VE-V/ | 14 | 2100 | | | | |
| AAKHE | 5L | 0.00E+00 | | N/A | | | | | | |
| AAPHC | 51 | N/A | | 1.44E-08 | 57 | 1478 | 5.80E+01 | 1.20E+01 | | 24 HR |
| 455UC | 51 | N/C | | 1.445-08 | 57 | 768 | 3.00E+01 | 6.00E+00 | | 24 HR |
| MALADO | | NZA | | 1 445-08 | 57 | 1044 | 7.20E+01 | 1.40E+01 | | 24 HR |
| AAQSC | JL | 878 | | 1.445-00 | | 4044 | 7 205.01 | 1 405-01 | | 24 40 |
| AAQVC | 5L | N/A | | N/A | | - 1044 | 7.20E+01 | 1.400+01 | | 27 08 |
| AAPSC | 51 | N/A | | 1.03E-07 | 57 | 642 | 5.99E+02 | 4.30E+01 | | 24 HR |
| AAPVC | 51 | N/A | | 1.03E-07 | 57 | 600 | 5.60E+02 | 4.00E+01 | | 24 HR |





N. 1993 - Menderal Managar, Menderal Internation Internation International Internation Internation (International

APPENDIX J SUPPORTING INFORMATION FOR HANDLING ANALYSIS

CARE CARACTER AND CONTRACTOR



J.1. HANDLING LEAKING M55 ROCKETS

Sources for the information contained in this section are Refs. J-1 and J-2. Table J-1 summarizes handling operations for sending sites while Table J-2 presents operations for receiving sites. Leaking M55 rockets are detected during storage by the igloo monitors or by the hand-held sniffers used by handlers before transporting a pallet. Leakers are isolated in the storage igloo by a two-man team of handlers. Entering the igloo in Level A protective clothing (M-3 TAP suits), the handlers pinpoint the leaker and move its pallet to ground level. They unpack the pallet, removing only those rockets necessary to expose the leaker, and place the removed, nonleaking rockets in a holding fixture.

マンシンシンション

They spread a plastic sheet on the ground in the area in which the leaker is to be isolated, placing the tools necessary to complete the isolation and the isolation container itself (a PIG) on the sheet. The cover of the PIG is removed, and the leaker is hand-carried to the PIG and placed in it. The cover of the PIG is closed, and its exterior and the tools are decontaminated using sodium carbonate, which is collected in the plastic sheet.

The handlers pick the decontaminated PIG up by its handles, carry it outside and place it on the truck that will carry it to the designated leaker-storage igloo, where it remains until it is transported (two PIGs per pallet) to the demil site. For other PIG operations, the PIG's handlers need wear only Level D protective clothing.



• 557

| Step | Equipment per Person | Error |
|--|-------------------------|--|
| Pick up munitions and unload outside (igloo/ warehouse apron). | e. fk. | puncture-1 drop-2 collision-1 |
| Load munitions onto truck. | d. fk. | puncture-1 drop-2 collision-1 |
| Transport munitions to maintenance facility (ton containers, mines, car- tridges, leaking rockets). | truck | (NA) |
| Unload munitions from truck at maintenance facility. | d. fk. | drop-2 collision-1 puncture-1 |
| Replace plugs and valves on ton containers. | operator | valve improperly installed. drop-2 |
| Lift up ton layer of drums (containing 3 mines each) after their fuses have been removed and the drums have been placed back on pallet. | operator | drop |
| Remove propellants from cartridges. | operator | |
| Leaking rockets placed inside PIGS; PIG is put on pallet. | operator | |
| Pick up munitions and load onto truck. | d. fk. | drop-2 puncture-1 collision-1 |
| Return munitions to storage (igloo, warehouse). | truck | NA |

TABLE J-1HANDLING OPERATIONS TASK ANALYSIS (SENDINC SITES)

K. Y

(

ŝ

| Step | Equipment per Person | Error |
|--|-------------------------|-------------------------------------|
| Unload munitions from truck (igloo/warehouse apron). | d. fk. | drop-2 puncture-1 collision-1 |
| Place munitions back into storage (igloo, warehouse). | e. fk. | drop-2 puncture-1 collision-1 |
| Pick up munitions in storage and place out- side (igloo, warehouse). | e. fk. | drop-2 puncture-1 collision-1 |
| Load munitions and transport to vault. | d. fk. | drop-1 puncture-1 collision-1 |
| Load munitions into vault. | d. fk. | drop-1 |
| Load vault onto truck. | d. fk. | drop-2 collision-1 |
| Transport vault to packing area. | truck | NA |
| Pick up vault from truck and unload at packing area. | d. fk. | drop-1 collision-1 |
| Place vault inside CAMPACT at packing area. | d. fk. | drop-1 |
| Pick up vault at packing area and load onto truck to take to holding area. | d. fk. | drop-1 collision-1 |
| Transport to holding area. | truck | NA |
| Unload vault from truck at holding area. | d. fk. | drop-2 collision-1 |
| Pick up vault and load onto train car. | d. fk. | drop-2 collision-1 |

ard second booten reveal



TABLE J-1 (Continued)

AT + 17 + A *+ 4*

| Step | Equipment per Person | Error |
|---|-------------------------|-------------------------------------|
| Pick up vault from train and unload at holding area. | d. fk. | drop-2 collision-1 |
| Pick up vault at holding area and load onto truck. | d. fk. | drop-2 collision-1 |
| Transport vault to unpacking area. | truck | NA |
| Unload vault from truck at unpacking area. | d. fk. | drop-2 collision-1 |
| Remove vault from CAMPACT. | d. fk. | drop-2 collision-1 |
| Remove munitions from vault in the packing area. | d. fk. | drop-1 puncture-1 |
| Load munitions onto truck at the packing area. | d. fk. | drop-1 collision-1 |
| Transport munitions to storage area. | truck | NA |
| Unload munitions from truck outside storage igloo. | d. fk. | drop-2 collision-1 puncture-1 |
| Transfer munitions from outside to the inside of the storage igloo. | e. fk. | drop-2 collision-1 puncture-1 |
| Transfer munitions from inside to outside the storage igloo. | e. fk. | drop-2 puncture-1 collision-1 |
| Load munitions onto truck. | d. fk. | drop-2 puncture-1 |
| Transport munitions to MHI. | truck | COILISION-I NA |
| Unload munitions from truck outside MHI. | d. fk. | drop-2 collision-1 puncture-1 |





N. K. K.

| Step | Equipment per Person | Error |
|---|-------------------------|-------------------------------------|
| Pick up munitions outside MHI and place inside. | e. fk. | drop-2 collision-1 puncture-1 |
| Pick up munitions inside MHI and place outside. | e. fk. | drop-2 collision-1 puncture-1 |
| Pick up munitions outside MHI and unload onto conveyor/elevator of MDB. | d. fk. | drop-2 collision-1 |

TABLE J-1 (Continued)



J.2. IN-STORAGE DETECTION OF LEAKING M55 ROCKET

To ensure proper packaging for transportation, it is important that leaking M55 rockets be detected before they are placed in vaults so they can be packaged as leakers. Given the overall tendency of the stockpile to leak and the increased tendency of some production lots within the stockpile to leak more often than others, special surveillance measures have been instituted to detect leakers. The likelihood that a leaking rocket is packaged as a nonleaker is a function of the failure of those special measures. XXXXXXI DODODAL ROADOIL ROAVAN

NUCK STOCK

57777779 ufficteda 20000599 secondo hadaaaa

B7226024 Kateria Kateria Kateria Kateria

Any time a rocket-storage igloo is entered, first-entry monitoring is performed. This involves taking an air sample to be laboratoryanalyzed before entering. If the agent levels exceed acceptable limits, full protective clothing must be worn while entering the igloo and isolating the leaking-rocket. Procedures for handling a leaking M55 rocket are explained in detail in Chapter 8 of Ref. J-1.

Because of their history as leakers, GB rockets are monitored on a regular basis. All igloos holding GB rockets are monitored weekly, while those containing lots made up of known leakers ("leaker lots") are airsampled daily.

In addition, enhanced storage monitoring inspections (SMIs) are conducted quarterly for all rocket igloos. These SMIs include an overall inspection of igloo condition and a 100% visual inspection of all rocket shipping and firing tubes. For the visual inspection, pallets <u>must be moved and dismantled</u>, and individual tubes <u>must be rotated so</u> <u>handlers can observe the whole surface of each</u>. OR <u>For the visual</u> <u>inspection</u>, <u>handlers walk between rows of stacked pallets</u>, <u>observing</u> <u>each pallet and the ground around each pallet for signs of leakage</u>. During the visual inspection, hand-held "sniffers" are used to check for leaks that can't be seen by the handlers. Also, a statistical sample of each lot is selected for air sampling inside the shipping and firing tubes.

The likelihood that a leaker in the stockpile goes undetected is a function of human errors (Is the monitoring performed correctly and on time?), hardware failures (Does the sniffer function correctly?), and leaker location (Can it be detected where it is, given current monitoring practice?). There are three types of human errors possible: Not monitoring an igloo at all, using the sniffer incorrectly, and overlooking a pallet or row of pallets while monitoring the igloo. Not monitoring an igloo constitutes a failure of administrative control. The administrative controls should be designed to prevent such an omission, but the probability of their not being followed is approximately 10^{-2} (with an error factor of 5) for most cases. Here, because records of igloos' agent levels are retained and because the igloos are most likely monitored in order (and each is too big to miss altogether), the lower bound of 2 x 10^{-3} is used as a reasonable worst-case estimate for these conditions. Taking 2 x 10^{-3} as the computed upper bound, the new error probability is estimated as 4×10^{-4} , the likelihood that an igloo is overlooked during the SMI.

Using the sniffer incorrectly implies that the operator has not turned it on, is using an uncalibrated sniffer, is not bringing the sniffer into the range of all potentially leaking pallets, does not notice when a leaker is indicated, etc. Because this check is performed frequently (e.g., GB igloos are monitored weekly), because several dozen pallets are involved in each check, because the monitoring is carried out by a two-man team, and because the leaker-indication alarm is hard to miss, it is estimated that the likelihood of the operator's failing to detect a leaker because of his using the sniffer incorrectly is negligible. This includes his failure to monitor a pallet entirely since the protocol he follows specifies that he will check all pallets

(which he does by walking the pallet aisles in market fashion, checking both sides of each aisle).

The likelihood of a leaker's being detected depends more on equipment reliability than it does on human reliability. If the hand-held sniffers have failed but seem to be in good working order (for example, if they have been miscalibrated), they may not detect a leaker. Also, their sensitivity to leaker position is an important variable. If the leaker is located in the center of a pallet and the pallet is located at the bottom of a stack of pallets, a small leak may not be detected by the sniffer. With little or no airflow in the igloo, the agent vapor may never reach the sniffer (or the igloo sensor). Therefore, the probability of detection is mainly dependent on leaker location and sniffer sensitivity.

J.3. TON CONTAINER VALVE REPLACEMENT

J.3.1. INTRODUCTION

One scenario of interest involves the replacement of existing valves and plugs on ton containers (TCs). For any demilitarization option involving offsite transportation, the valves and the plugs must be replaced by plugs. This calls for handling, movement, and replacement activities on the TCs. There is some probability that the replacement will be made incorrectly, and that the plugs will leak after installation. This discussion addresses that probability.

Valve replacement is different for TCs holding different agents. For TCs of GB, no offsite transportation will be required, and their current valves and plugs (many of which are showing signs of corrosion and some of which have already had to be replaced) will have been replaced with new plugs prior to the start date for the demil operations. Since offsite transportation is not necessary, these new plugs will not have to be replaced again before demilitarization. Therefore, TCs of GB will not be addressed in this analysis.

Before TCs holding either HD or VX can be transported for disposal at a national or regional site, their values and plugs do have to be replaced. The specifics of this replacement have not been determined, so a generic description of representative activities involved in value replacement is provided for analysis. Value replacement does not change any of the assumptions made about TC integrity during transportation or demil.

J.3.2. VALVE-REPLACEMENT ACTIVITIES

TC valve replacement will be performed <u>in situ</u>, no onsite transportation will be necessary. The TCs will be moved by forklift to a clear area on one side of their storage site during replacements, which will be made on one TC at a time. The handlers will wear Level A protective clothing (an M-3 TAP suit) for the operation and will use a TC "cradle," a device designed to hold the TC during valve replacement. The bed of the cradle rotates to move the TC from the horizontal to the vertical and can be locked into position by using a cotter pin to secure a bolt inserted through aligned holes in the frame and the bed.

The TC will be lifted from its stored, horizontal position using a lifting beam attached to a forklift. The forklift will be used to place the TC into the cradle (which should be locked into the horizontal orientation) so that its aft end can be rotated up. The TC will be secured in the cradle by fastening two chains across its girth, from one side of the cradled bed to the other. The bed will be rotated and locked so that the aft end of the TC is vertical. Handlers will use wrenches and taps as necessary to remove existing plugs and replace them with new, steel plugs one at a time. The handlers will work to prespecified torque limits for plug tightening. It may be necessary to rethread the plug holes before the new plugs can be installed.

Following plug installation, the bed of the cradle will be returned to the horizontal, and the TC will be released, lifted using the forklift and lifting beam, and placed and secured in the cradle bed so that its forward end can be rotated up. The bed will be rotated and locked so that the forward end of the TC is vertical. Handlers will use wrenches and taps as necessary to remove existing plugs and replace them with new, steel plugs one at a time. The valves will be removed and will be replaced with new, steel plugs. Handlers will work to prespecified torque limits for plug tightening. It may be necessary to

rethread holes before the new plugs can be installed. The bed will be returned to the horizontal, and the TC will be released, lifted using the forklift and lifting beam, and returned to its storage position. 522222

J.3.3. ERROR IDENTIFICATION

Obviously, it is hoped that the exposure of agent to the air is minimized. This translates operationally into the handlers replacing any removed values or plugs as quickly as possible. There may be instances, however, when quick (within 10 min) replacement is not achieved. Since these operations involve only HD and VX containers, the consequences of leaving an open port on one end of the container for as long as 10 min are negligible. This is because of the stable nature of those agent under the described conditions and because the cross-section of the TC presents a small surface area of exposure. Even the handlers themselves are not likely to be subjected to significant exposure since their contact with the used values and plugs (sources of contact contamination) should be brief.

It is possible that, while performing the replacement, the handlers will drop foreign matter (e.g., tools) into the TC. This event has no immediate or future consequences that can be predicted at present, and probably represents a no-cost error. The foreign material will most likely be left in the TC through its demil.

Another potential error involves incorrect installation of the new plugs. The handlers could cross-thread the plug so that a seal is not achieved or they could apply too little or too much torque to the plug. If the plug is cross-threaded or it too little torque is applied, the plug may leak as a result. If the leak is serious, it will be detected when the TC is returned to the horizontal. If the leak is not serious, it could remain undetected for some time, especially if the replugged TCs are stored in the same area where the replacement operations are taking place. Still, it is likely to be small enough to be contained in the lop of the TC. If the surface of the TC is cleaned before or after the valve replacement, even a small leak is likely to be noticed before transport. The joint probability that an installation error is made, that it leads to a leak too small to be detected immediately, and that the leak is not noticed during the inspection before transport is negligible.

If too much torque is applied to a plug, it will probably not be detected. Neither is it likely to have any adverse effect. This is considered a no-cost error.





J.3.4. CONCLUSION

The factors that characterize TC valve replacement include the agent involved (HD and VX), the time to complete the operation for a single TC (probably less than 30 min), and the replacement parts used (all plugs). These conditions argue for there being no significant risk to the public associated with valve replacement for HD and VX TCs. The risk to the handler is another matter and is discussed in the following section.

KSS33333

Performant and the second

J.3.5. SPECIAL HANDLING FIXTURES

At times, it may be necessary to replace TC values and plugs outside. If an emergency such as a large or obvious leak exists, special equipment has been developed to minimize agent release during the operation. This equipment is a portable glove box designed to be used with the same cradle that is used in normal operations. While the protection afforded the general public in terms of release mitigation may be substantial, the increased risk to the handler may argue against full-time use of the portable glove box.

The glove box is a portable shroud that fits over the end of a TC. The shroud is about 18 in. high and its diameter is slightly larger than that of the TC. The bottom edge has a rubber gasket that forms a friction seal with the sides of the TC. The shroud is slipped over the end of the TC until its top is about 12 in. from the valve. The side of the shroud has two discharge vents from which hoses to M-6 filters are connected. The top is 3/8-in. thick sheet of Lexan through which four hand ports have been cut. The M-6 filter fan pulls air through the ports and out the discharge vents, minimizing or eliminating any agent release.

The handlers (who sometimes must stand on a step-platform attached to the cradle to reach the plug and valve assemblies) must access the plugs and valve through the glove box ports. The distance from the valve to the top of the shroud make manipulations difficult, and the limited-access area makes visual contact with the working surface where hands are inside difficult. The Lexan itself is prone to scratches, we the inside is often contaminated with splashes from the operations. These factors also limit visual access.

Working inside a glove-box apparatus while in Level A process clothing makes performing manipulative tasks extremely diffitime to complete each replacement will likely take three tops as as it would if the glove box were not used. The handless







will increase their per-operation exposure time. The number of TCs modified during any given crew's shift will, therefore, decrease significantly.

Another area of concern involves the level of protection afforded by the M-3 TAP suit. An examination of the accident and incident data base for chemical munitions operations reveals that, of the measurable worker exposures, the majority were either caused or contributed to by failures of their protective equipment. These failures were themselves the products of equipment failures, errors in administrative control, or undetected damage to the suit during operations. The handlers' lack of visual access to tasks involving the use of sharp tooling and their increased time at the agent release site imply that the use of the portable glove box for valve and plug replacement for the whole TC stockpile may likely be more hazardous for them than if it is not used.



J.4. TRANSPORTATION CONTAINER MONITORING

In estimating the likelihood that a handler will open a transportation container without monitoring its contents for leaks is estimated as 1E-3. This human-error probability is taken from Table 20-22, item 9 (Ref. J-3) and represents a case in which a checker will fail to check equipment status when that status affects the checker's own safety. Since the containers are loaded elsewhere (or at least by other people), the unloading handler should be cautious in dealing with them. Since he has no control over ensuring a "clean" container interior by performing the loading himself, he will probably want to protect himself by making sure he monitors the container before opening it. This error suggests that he will overlook one out of every thousand containers he should monitor.

J.5. REFERENCES

- J-1. "Probabilities of Selected Hazards in Disposition of M55 Rockets," M55-CS-02, USATHAMA, September 22, 1985.
- J-2. "Transportation of Chemical Agent and Munitions: A Concept Plan," (Preliminary Draft Report for review and comment), USATHAMA, April 3, 1987.
- J-3. "Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Applications," NUREG/CR-1278, U.S. Nuclear Regulatory Commission, May 1983.



A U.S. GOVERNMENT PRINTING OFFICE 1987-2 0 1 - 4 4 2 / 8 0 1 3 9

END DATE FILMED DTIC July 88